

# Current Developments in Biotechnology and Bioengineering

### Human and Animal Health Applications

*Edited by* Vanete Thomaz-Soccol, Ashok Pandey, Rodrigo R. Resende



AMSTERDAM • BOSTON • HEIDELBERG • LONDON • NEW YORK • OXFORD PARIS • SAN DIEGO • SAN FRANCISCO • SINGAPORE • SYDNEY • TOKYO Elsevier Radarweg 29, PO Box 211, 1000 AE Amsterdam, Netherlands The Boulevard, Langford Lane, Kidlington, Oxford OX5 1GB, United Kingdom 50 Hampshire Street, 5th Floor, Cambridge, MA 02139, United States

Copyright © 2017 Elsevier B.V. All rights reserved.

No part of this publication may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or any information storage and retrieval system, without permission in writing from the publisher. Details on how to seek permission, further information about the Publisher's permissions policies and our arrangements with organizations such as the Copyright Clearance Center and the Copyright Licensing Agency, can be found at our website: www.elsevier.com/permissions.

This book and the individual contributions contained in it are protected under copyright by the Publisher (other than as may be noted herein).

#### Notices

Knowledge and best practice in this field are constantly changing. As new research and experience broaden our understanding, changes in research methods, professional practices, or medical treatment may become necessary.

Practitioners and researchers must always rely on their own experience and knowledge in evaluating and using any information, methods, compounds, or experiments described herein. In using such information or methods they should be mindful of their own safety and the safety of others, including parties for whom they have a professional responsibility.

To the fullest extent of the law, neither the Publisher nor the authors, contributors, or editors, assume any liability for any injury and/or damage to persons or property as a matter of products liability, negligence or otherwise, or from any use or operation of any methods, products, instructions, or ideas contained in the material herein.

#### Library of Congress Cataloging-in-Publication Data

A catalog record for this book is available from the Library of Congress

#### British Library Cataloguing-in-Publication Data

A catalogue record for this book is available from the British Library

ISBN: 978-0-444-63660-7

For information on all Elsevier publications visit our website at https://www.elsevier.com/



Publisher: John Fedor Acquisition Editor: Kostas Marinakis Editorial Project Manager: Anneka Hess Production Project Manager: Vijayaraj Purushothaman Designer: Greg Harris

Typeset by TNQ Books and Journals



# List of Contributors

L.M. Alvarenga Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil

F. Antunes Centro de Química e Bioquímica, Faculdade de Ciências, Universidade de Lisboa, Portugal

A.M.M. Basso University of Brasília (UnB), Brasília-DF, Brazil; EMBRAPA – Genetic Resources and Biotechnology, Brasília-DF, Brazil

P. Billiald University of Paris-Sud, Châtenay-Malabry, France; Muséum National d'Histoire Naturelle - CNRS UMR 7245, Paris, France

T. Callese Wake Forest School of Medicine, Winston Salem, NC, United States

K.A.T. Carvalho Pelé Pequeno Príncipe Research Institute & Pequeno Principe Faculty, Curitiba, Brazil

F. Castiglione Institute for Applied Mathematics (IAC), National Research Council of Italy (CNR), Rome, Italy

L.R. Castilho Federal University of Rio de Janeiro, Rio de Janeiro, Brazil

J.C. Corrêa-Velloso Department of Biochemistry, Institute of Chemistry, University of São Paulo, São Paulo, SP, Brazil

O. Dalagostin Universidade Federal de Pelotas (UFPel), Pelotas, Rio Grande do Sul, Brazil

N. de Cássia Oliveira Paiva Federal University of Minas Gerais, Belo Horizonte, Brazil

W.M.S. Degrave Laboratory for Functional Genomics and Bioinformatics, Oswaldo Cruz Institute–Fiocruz, Rio de Janeiro, RJ, Brazil

O.A. Dellagostin Universidade Federal de Pelotas (UFPel), Pelotas, Rio Grande do Sul, Brazil R.V.B. de Medeiros Federal University of Minas Gerais, Belo Horizonte, Brazil

J. de Moura Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil

M.V. Dias-Souza Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil

L. Edgar Wake Forest School of Medicine, Winston Salem, NC, United States

L. Felicori Federal University of Minas Gerais, UFMG, Belo Horizonte, Brazil

S.R. Felix Universidade Federal de Pelotas (UFPel), Pelotas, Rio Grande do Sul, Brazil

R. Fendrich Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil

C. Gentile Wake Forest School of Medicine, Winston Salem, NC, United States

A.Q. Gomes Escola Superior de Tecnologia da Saúde de Lisboa, IPL, Lisboa, Portugal; Instituto de Medicina Molecular, Faculdade de Medicina, Universidade de Lisboa, Lisboa, Portugal

V.A.M. Goulart Federal University of Minas Gerais, Belo Horizonte, Brazil

M.F. Grossi-de-Sa Catholic University of Brasília (UCB), Brasília-DF, Brazil; EMBRAPA – Genetic Resources and Biotechnology, Brasília-DF, Brazil

D. Guedes Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil

D. Igel Wake Forest School of Medicine, Winston Salem, NC, United States

S. Jorge Universidade Federal de Pelotas (UFPel), Pelotas, Rio Grande do Sul, Brazil

R. Katari Wake Forest School of Medicine, Winston Salem, NC, United States

H. Lima McMaster University, Hamilton, Ontario, Canada

D. Marasco University of Naples Federico II, Naples, Italy

H.S. Marinho Centro de Química e Bioquímica, Faculdade de Ciências, Universidade de Lisboa, Lisboa, Portugal

K. McNamara Wake Forest School of Medicine, Winston Salem, NC, United States

T. Mendes Federal University of Minas Gerais, UFMG, Belo Horizonte, Brazil

L. Mendonça-Lima Laboratory for Functional Genomics and Bioinformatics, Oswaldo Cruz Institute–Fiocruz, Rio de Janeiro, RJ, Brazil

S. Mondal National Institute of Animal Nutrition and Physiology, Bangalore, India

A. Mor National Institute of Animal Nutrition and Physiology, Bangalore, India

R. Nakamura Instituto de Tecnologia do Paraná, Curitiba, Paraná, Brazil

R.M. Nardi Drummond Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil

I.C. Nascimento Department of Biochemistry, Institute of Chemistry, University of São Paulo, São Paulo, SP, Brazil

S. Ndoja McMaster University, Hamilton, Ontario, Canada

A.A. Nery Department of Biochemistry, Institute of Chemistry, University of São Paulo, São Paulo, SP, Brazil

S. Nolasco Escola Superior de Tecnologia da Saúde de Lisboa, IPL, Lisboa, Portugal; Instituto Gulbenkian de Ciência, Oeiras, Portugal; Instituto de Investigação Científica Tropical, CVZ, CIISA, Faculdade de Medicina Veterinária, Universidade de Lisboa, Lisboa, Portugal

G. Orlando Wake Forest School of Medicine, Winston Salem, NC, United States

A. Pandey Center of Innovative and Applied Bioprocessing, (a national institute under Dept of Biotechnology, Ministry of S&T, Govt of India), Mohali, Punjab, India

**R.C. Parreira** Federal University of Minas Gerais, Belo Horizonte, Brazil; InstitutoNanocell, Divinópolis, Brazil

P.B. Pelegrini EMBRAPA – Genetic Resources and Biotechnology, Brasília-DF, Brazil; Diagene Diagnósticos Moleculares Ltda, Brasília-DF, Brazil

C. Perottoni Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil; Instituto de Tecnologia do Paraná (TECPAR), Curitiba, Paraná, Brazil

M.C.X. Pinto Federal University of Minas Gerais, Belo Horizonte, Brazil; InstitutoNanocell, Divinópolis, Brazil; Federal University of Goiás, Goiânia, Brazil

**G.S. Prado** Catholic University of Brasília (UCB), Brasília-DF, Brazil; EMBRAPA – Genetic Resources and Biotechnology, Brasília-DF, Brazil

C. Real Centro de Química e Bioquímica, Faculdade de Ciências, Universidade de Lisboa, Portugal

I.J. Reddy National Institute of Animal Nutrition and Physiology, Bangalore, India

**R.R. Resende** Federal University of Minas Gerais, Belo Horizonte, Brazil; InstitutoNanocell, Divinópolis, Brazil

R.K. Sanson Instituto de Tecnologia do Paraná, Curitiba, Paraná, Brazil

V.L. Santos Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil

T. Scheuer Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil; Instituto de Tecnologia do Paraná (TECPAR), Curitiba, Paraná, Brazil

M.G. Severo Divisão de vacinas virais do instituto de tecnologia do Estado do Paraná - TECPAR, Curitiba, Paraná, Brazil

H. Soares Centro de Química e Bioquímica, Faculdade de Ciências, Universidade de Lisboa, Lisboa, Portugal; Escola Superior de Tecnologia da Saúde de Lisboa, IPL, Lisboa, Portugal; Instituto Gulbenkian de Ciência, Oeiras, Portugal

C.R. Soccol Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil

L.T. Souza Federal University of Minas Gerais, Belo Horizonte, Brazil

R. Tamburrini Wake Forest School of Medicine, Winston Salem, NC, United States

V. Thomaz-Soccol Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil

P. Tieri Institute for Applied Mathematics (IAC), National Research Council of Italy (CNR), Rome, Italy

F.C.P. Tonelli Federal University of Minas Gerais, Belo Horizonte, Brazil

F.M.P. Tonelli Federal University of Minas Gerais, Belo Horizonte, Brazil; InstitutoNanocell, Divinópolis, Brazil

H. Ulrich Department of Biochemistry, Institute of Chemistry, University of São Paulo, São Paulo, SP, Brazil

L.P.S. Vandenberghe Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil

F. Vitola Universidade Federal do Paraná (UFPR), Curitiba, Paraná, Brazil

J.P. Zambon Wake Forest School of Medicine, Winston Salem, NC, United States

A.S. Zeferino Instituto de Tecnologia do Paraná, Curitiba, Paraná, Brazil

#### xxi

# About the Editors

#### Vanete Thomaz-Soccol

Professor Thomaz-Soccol is a Level 1A research fellow at the Brazilian National Council for Scientific and Technological Development at the Federal University of Paraná (UFPR), Brazil. She graduated with a degree in veterinary medicine from UFPR in 1978, a master of science in biochemistry from UFPR in 1981, and a Ph.D. in parasitology at the Faculty of Medicine, Université Montpellier I, France, in 1993, and earned a postdoctorate in molecular parasitology at the Institute of Research and Development in France in 1998. Professor Thomaz-Soccol is a senior professor and heads the Centre for Molecular Biology for the PhD program of Bioprocess Engineering and Biotechnology at UFPR. She is experienced in the area of Parasitology and Biotechnology.

She has supervised more than 100 masters and doctoral students, and created the MSc and PhD programs in microbiology, parasitology, and pathology at UFPR, Curitiba, and in industrial biotechnology at Positive University, Curitiba. Her main focus is in the area of the development of biotechnological inputs for human and veterinary immunodiagnostic and vaccine composition, related to infectious parasitic diseases. She has 492 publications/communications, including 10 patents and design copyrights, 04 books, 31 book chapters, and 240 original and review papers, with a h index of 28 and more than 5422 citations (Google Scholar).

#### Ashok Pandey

Professor Ashok Pandey is Eminent Scientist at the Center of Innovative and Applied Bioprocessing, Mohali (a national institute under the Department of Biotechnology, Ministry of Science and Technology, Government of India), and former chief scientist and head of the Biotechnology Division at the CSIR's National Institute for Interdisciplinary Science and Technology at Trivandrum. He is an adjunct professor at Mar Athanasios College for Advanced Studies Thiruvalla, Kerala, and at Kalasalingam University, Krishnan Koil, Tamil Nadu. His major research interests are in the areas of microbial, enzyme, and bioprocess technology, which span various programs, including biomass to fuels and chemicals, probiotics and nutraceuticals, industrial







enzymes, solid-state fermentation, etc. He has more than 1100 publications and communications, which include 16 patents, 50+ books, 125 book chapters, and 425 original and review papers, with an h index of 75 and more than 23,500 citations (Google Scholar). He has transferred several technologies to industries and has been an industrial consultant for about a dozen projects for Indian and international industries.

Professor Pandey is the recipient of many national and international awards and fellowships, which include Elected Member of the European Academy of Sciences and Arts, Germany; Fellow of the International Society for Energy, Environment and Sustainability; Fellow of the National Academy of Science (India); Fellow of the Biotech Research Society, India; Fellow of the International Organization of Biotechnology and Bioengineering; Fellow of the Association of Microbiologists of India; honorary doctorate degree from the Université Blaise Pascal, France; Thomson Scientific India Citation Laureate Award, United States; Lupin Visiting Fellowship; Visiting Professor at the Université Blaise Pascal, France, the Federal University of Parana, Brazil, and the École Polytechnique Fédérale de Lausanne, Switzerland; Best Scientific Work Achievement Award, Government of Cuba; UNESCO Professor; Raman Research Fellowship Award, CSIR; GBF, Germany, and CNRS, France fellowships; Young Scientist Award; and others. He was chairman of the International Society of Food, Agriculture and Environment, Finland (Food & Health) during 2003–04. He is the Founder President of the Biotech Research Society, India (www.brsi.in); International Coordinator of the International Forum on Industrial Bioprocesses, France (www.ifibiop.org); chairman of the International Society for Energy, Environment & Sustainability (www.isees.org); and vice president of the All India Biotech Association (www.aibaonline.com). Professor Pandey is editor-in-chief of Bioresource Technology, Honorary Executive Advisor of the Journal of Water Sustainability and Journal of Energy and Environmental Sustainability, subject editor of the Proceedings of the National Academy of Sciences (India), and editorial board member of several international and Indian journals, and also a member of several national and international committees.

#### Rodrigo R. Resende

Professor Rodrigo R. Resende graduated with degrees in pharmacy and biochemistry in 2002 and biological sciences 2003 from the Federal University of Alfenas. He earned a Ph.D. in biochemistry from the University of São Paulo in 2007 and a postdoctorate in Biophysics in 2008 and in physics of nanomaterials in 2009. He is the head of the Cell Signaling and Nanobiotechnology Laboratory and an adjunct professor at the Department of Biochemistry and Immunology, Federal University of São João del-Rei (UFSJ), Minas Gerais. He was the founding president of the Brazilian Society for Cell Signaling and founding president of the Nanocell Institute, a nonprofit organization for research and development, and the promotion of



sciences and education in Brazil (www.institutonanocell.org.br). Also he was the founder and creator of the popular science journal *Nanocell News* (www.nanocellnews.com.br). He created the master of science in biotechnology degree at UFSJ, Minas Gerais. His research expertise is in the areas of biochemical technology, nanobiotechnology, cell signaling, stem cell differentiation, direct reprogramming of somatic cells, and tissue engineering. He has published over 75 papers, 6 books, and 23 book chapters, and has 9 patents. He has supervised 14 masters, doctoral, and postdoctoral students.



### Preface

This book is a part of the comprehensive series, *Current Developments in Biotechnology and Bioengineering* (Editor-in-Chief: Ashok Pandey), composed of nine volumes. Within this series, the current volume is dedicated to human and animal health applications in biotechnology. In the 1970s, researchers developed DNA recombinant technology, which facilitated the production of recombinant proteins. One of the first molecules produced by this technology was somatostatin. Through genetic engineering techniques, the gene coding for somatostatin was inserted into *Escherichia coli* cells, which expressed a recombinant protein with activity similar to the human hormone. Over the ensuing years, other heterologous systems were introduced for producing recombinant human molecules, including yeast, insect, and mammalian cells. The new class of biologically derived therapeutic molecules was called *biopharmaceuticals*.

Actually, 60% of the new drugs approved by the Food and Drug Administration are biopharmaceuticals that were developed for the treatment of cancer and infectious, cardiovascular, and inflammatory diseases. Biopharmaceuticals now comprise a diverse number of molecules, such as coagulation factors, hormones, enzymes, monoclonal antibodies, and vaccines. Owing to their complexity in three-dimensional structure and size, each process applied to produce a molecule will result in a biopharmaceutical with specific physicochemical properties and biological activity. The biotechnological industrial sector has made outstanding contributions to medical progress resulting in improvements to life.

This book is composed of nine parts. In the first part, Chapters 1 and 2 present the global market and perspectives on biotechnology health, in which different classes of biopharmaceutical products are described and the manufacturing technologies used to obtain these products are discussed. Part 2 is devoted to research and development in immunotechnology, such as interferon production by the biotechnology industry for viral infections and cancer treatments, and monoclonal antibodies that have revolutionized the fields of research and medicine. Chapter 5 introduces and explores new perspectives in recombinant antibody production, a unique source of biomolecules that are used as pharmacologic tools and which also should contribute to a revival of immunodiagnostic and biomedical applications as yet unimagined.

In the third part, combinatorial libraries are described, which include the impacts of the aptamer technology on diagnostics, biotechnology, and therapy; combinatorial preparation; and high-throughput screening of arrays of compounds that are the basis of numerous platforms of pharmaceutical drug discovery processes. Finally the concept, innovations, applications, and future of phage display technology are presented.

The fourth part is on gene and cell therapy. Chapter 9 discusses the power of mathematical networks and modeling for the identification of key components related to rheumatoid arthritis and how to predict the response of different individuals to infections. Chapter 10 reviews the biogenesis pathways and general functions of microRNAs, PIWI RNAs, and long noncoding RNAs. It presents current knowledge concerning the use of noncoding RNAs in diagnosis, prognosis, and therapeutics and discusses their role in the development of the immune system and the regulatory functions of  $H_2O_2$  during the course of metazoan evolution.

Part 5 deals with tissue engineering. Chapter 11 presents biomaterials, which are derived from naturally occurring molecules and those that recapitulate key motifs of biomolecules within biologically active synthetic materials. The most significant biological features of the extracellular matrix are discussed and several engineering methods currently being implemented to design and tune synthetic scaffolds to mimic these features are presented. Chapter 12 presents protocols to isolate adipose-derived stem cells, strategies to characterize these cells, an example of stem cell differentiation into bone cells (generating material to perform bone repair in vivo) and some techniques that can be applied to verify the structure of the newly generated bone tissue. In Chapter 13, the current state of bioengineering and regenerative medicine is discussed with respect to each abdominal organ. Chapter 14 offers new approaches to reprogramming mesenchymal stem cells for tissue engineering on a biofunctionalized scaffold for tissue repair and for restoring its function.

The next part is about biofilms and biosurfactants in health, which describes the most promising technology using microbial surfactants, which have attracted attention as potential substitutes for, additions to, currently employed antimicrobial compounds, mainly for biofilm infection control. Chapter 15 discusses the potential use of these biomolecules in formulating drug delivery nanocarrier systems, including liposomes and polymer-based carriers. Chapter 16 highlights the use of bacteriocins as a new strategy for planktonic cells and biofilm control.

The field of vaccinology has yielded several effective vaccines that have significantly reduced the impact of a number of important diseases. Biotechnology has made the development of new vaccines safer and less costly. Chapter 17 reviews technologies that are currently available for the development of recombinant veterinary vaccines. Chapter 18 describes new technologies related to the production of acellular pertussis vaccine against whooping cough; Chapter 19 presents a case description about the biotechnological development of a diphtheria subunit vaccine using diphtheria toxin subunit B as the immunogenic protein. Chapter 20 discusses serum-free rabies vaccine production and Chapter 21 discusses the lyophilization process.

Biotechnological strategies needed to develop better pharmaceuticals against leishmaniasis are presented in Chapter 22. Finally, Chapter 23 focuses on maternal recognition of pregnancy and genes controlling maternal recognition of pregnancy events that facilitate the development of optimal reproductive management strategies and paradigms to augment embryo survival.

We hope that this book will be of special interest to academicians, researchers, graduate students, and industry scientists working in the field. We would like to thank the authors and reviewers of the chapters for their cooperation and for their preparedness in revising the articles on a scheduled timeline. We thank Dr. Kostas Marinakis, book acquisition editor, Ms. Anneka Hess, and the entire production team at Elsevier for their help and support in bringing out this volume. Without their commitment, efficiency, and dedicated work, this volume could not have been accomplished.

Editors Vanete Thomaz-Soccol Ashok Pandey Rodrigo R. Resende

# 1

# Biopharmaceutical Products: An Introduction

L.R. Castilho FEDERAL UNIVERSITY OF RIO DE JANEIRO, RIO DE JANEIRO, BRAZIL

#### 1.1 Introduction

The term "biopharmaceuticals" was coined in the 1980s specifically to refer to human health products generated or produced by means of modern molecular biological methods using engineered organisms, and to distinguish these products from traditional biological products directly extracted from natural biological sources such as proteins obtained from plasma or plants [1]. Techniques usually considered to be the landmarks in "modern Biotechnology", recombinant DNA technology and hybridoma technology, both developed in the 1970s, were important for the development of this class of products. However, besides "biopharmaceuticals," several other terms, such as biotherapies, immunotherapies, biologics, and biological products, have been used to refer to these products [2,3]. In 2002, in an attempt to avoid the problem that the various terms have different meanings for different people, Walsh [4] discussed the issue of nomenclature and proposed the use of the following definition:

A biopharmaceutical is a protein or nucleic acid-based pharmaceutical substance used for therapeutic or in vivo diagnostic purposes, which is produced by means other than direct extraction from a native (non-engineered) biological source.

Currently, according to this definition by Walsh [4], "biopharmaceuticals" is the most widely adopted term, but it is often used interchangeably with "biologicals," "biologics," and "biological products" [3]. In terms of the biopharmaceutical market, recombinant therapeutic proteins largely dominate. Thus, it is also common to use the term "biopharmaceuticals" to refer to them specifically. However, if one refers to "biopharmaceutical products" or to the "biopharmaceutical sector" as an industrial segment, recombinant vaccines and nucleic acid—based therapies are usually included. Besides these, other biotechnological products for human health are cell therapies, in which the cells are the product used to regenerate or engineer tissues, and molecules used for in vitro or in vivo diagnostics.

#### 1.2 The Biopharmaceutical Sector: Market Aspects

The first biopharmaceutical product to be approved for human use in 1982 was human insulin produced in recombinant *Escherichia coli*. Four years later, there were other biopharmaceutical landmarks: the approvals of recombinant tissue plasminogen activator (t-PA) produced in animal cells (Chinese hamster ovary cells, CHO), of the monoclonal antibody OKT3 produced in hybridoma cells and of recombinant hepatitis B vaccine produced in *Saccharomyces cerevisiae* [5,6].

Since the commercialization of insulin almost 35 years ago as the first recombinant therapeutic protein, many other products followed, leading to a continuous increase in the sales of the biopharmaceutical sector. Until July 2014, 246 biopharmaceutical products had been licensed in the United States and in the European Union (EU), containing 166 different active ingredients. In the 20 years between 1995 and 2014, the number of new biopharmaceutical products approved in these regions was approximately constant, between 50 and 60 every 5 years. Among the 54 products licensed between 2010 and July 2014, about 40% were not truly new products (i.e., they were biosimilars, improvements of existing products, or products previously approved in other regions). Thus, only 32 of them, containing 30 different new active ingredients, can be considered genuinely new products [1].

Despite the decades that have passed since approval of recombinant insulin, the biopharmaceutical sector cannot yet be considered a mature industry, and many new developments regarding new products and new technologies are under way.

In the period 2010–14, there were several important landmarks:

- A first biosimilar monoclonal antibody (mAb) (i.e., a product comparable to the innovator biopharmaceutical product whose patent has expired) was approved in the EU: infliximab biosimilar products (Remsina and Inflectra) from two companies (Celltrion and Hospira) were licensed in 2013.
- In the so-called "highly regulated markets," for the first time, there was approval of a gene therapy product: Glybera (alipogene tiparvovec), produced by the company UniQure, licensed in the EU in 2012.
- For the first time, a biopharmaceutical produced in plant cells was approved: Elelyso (glucocerebrosidase/taliglucerase alfa), expressed in recombinant carrot cells propagated in bioreactors and licensed by Protalix Biotherapeutics/Pfizer in 2012 in the United States.
- Ruconest, produced by the company Pharming using transgenic rabbits, was the second biopharmaceutical product obtained from the milk of transgenic animals to be licensed. It has been approved in 2010 in the EU and in 2014 in the United States. The first product had been Atryn, a recombinant human antithrombin III produced from the milk of transgenic goats, approved in 2009 in the United States.

The pharmaceutical sector as a whole, and more specifically the biopharmaceutical sector, has been steadily growing over past decades. This is related, among other factors, to the continuous increase in the average life expectancy of the world population. This is

particularly true in the case of biopharmaceuticals, because many of them are indicated for disorders that are more prevalent at advanced ages, such as cancer and inflammatory disorders. Between 1980 and 2012, for example, life expectancy grew from 75 to 82 years in Australia, 65 to 75 in China, 73 to 81 in Germany, 58 to 73 in Turkey, 67 to 77 in Mexico, and 61 to 68 years considering the worldwide average [7]. In 2013, for example, the pharmaceutical sector grew 3% in the United States (data by IMS Health, cited in EFPIA [8]). In emerging economies, such as Brazil, growth was even more pronounced (14% in Brazil in 2013), because in these countries economic and social factors such as a reduction in the inequality of income distribution, can have a significant impact on access of the population to drugs.

Around the world, according to a report by the McKinsey company [9], the annual growth rate of the pharmaceutical industry was 4%; it reached 8% in the case of the biopharmaceutical sector alone. Investments in R&D of biopharmaceuticals usually give a better return than in the case of synthetic drugs, because the success rate of biopharmaceuticals is usually two times higher than that of small molecules (13% of biopharmaceuticals that enter phase I of clinical trials reach the commercialization phase). Maybe motivated by this, the number of submitted biopharmaceutical patents has been growing 25% per year since 1995, ensuring continuous growth in the number of new biopharmaceuticals being developed [9].

This continuous growth of the biopharmaceutical pipeline will certainly further increase the importance of the biopharmaceutical sector, which in 2014 accounted for approximately 20% of all pharmaceutical sales [9]. This forecast is further corroborated by the fact that among the approximately 10,000 therapeutic products under development and about 41,000 clinical trials under way in 2012, about 40% referred to biopharmaceuticals [10]. However, to develop a new biological entity for human health, as of 2012 an investment of approximately USD 1.5 billion and a period of 12–13 years were needed. Thus, the biotech industry is known to be one of the most R&D-intensive sectors, spending approximately 14.4% of net sales in R&D [8]. The biopharmaceutical sales, which in 2014 reached USD 163 billion, exceed the reported gross domestic product (GDP) of 75% of the economies (156 out of 214 countries) included in the World Bank GDP ranking database [1,9]. A large part of these sales is due to 37 biopharmaceutical products that are considered blockbusters, i.e., that have sales higher than USD 1 billion per year.

Data displayed in Table 1.1 [11] show the important role of biopharmaceuticals in the global pharmaceutical industry. All 10 top-selling pharmaceutical products, which together accounted for over USD 75 billion sales in 2013, individually reached over USD 5 billion sales. Furthermore, it is noteworthy that:

- the top-selling pharmaceutical product was the mAb Humira (adalimumab), indicated for arthritis treatment
- among the five top-selling pharmaceuticals, four were biopharmaceuticals, all of which were mAb-based products
- among the 10 top-selling pharmaceuticals, seven were biopharmaceuticals, six of which were mAb-based products

Product	Main Indication	Company	Sales in 2012 (10 <sup>6</sup> USD)	Sales in 2013 (10 <sup>6</sup> USD)	Growth Rate 2012–13 (%)
Humira	Arthritis	AbbVie	9265	10,659	15.0
Remicade	Arthritis	J&J/Merck & Co	8215	8944	8.9
Rituxan	Non-Hodgkin lymphoma	Biogen Idec/Roche	8266	8583	3.8
Enbrel	Arthritis	Pfizer/Amgen	7973	8325	4.4
Seretide/Advair	Asthma/chronic obstructive pulmonary disease	GlaxoSmithKline	7887	8243	4.5
Lantus	Diabetes	Sanofi	6586	7589	15.2
Avastin	Colorectal cancer	Roche	6217	6746	8.5
Herceptin	Breast cancer	Roche	6352	6557	3.2
Crestor	Dyslipidemia	AstraZeneca	6253	5622	-10.1
Abilify	Schizophrenia	Otsuka/BMS	5308	5158	-2.8

Table 1.1	Ten Worldwide	<b>Top-Selling</b>	Pharmaceutical	Products
-----------	---------------	--------------------	----------------	----------

Entries that are biopharmaceuticals are shown in italics.

Adapted from FirstWord Pharma. Pharma's 50 Biggest Selling Drugs: AbbVie's Humira Joins the \$10 Billion Club, 2014 http://www.firstwordpharma.com/node/1194000#axzz3Y2ixQ0qJ (access in June 2015).

#### 1.3 Types of Biopharmaceutical Products for Human Health

Among the different types of biopharmaceutical products, recombinant therapeutic proteins, vaccines, and products for in vitro and in vivo diagnostics are most mature, whereas nucleic acid—based therapies and cell therapies are less consolidated, with a still low number of approved products. In terms of clinical trials (phases I, II, and III), as of 2013, 245 were ongoing focusing on cell therapy, 99 focused on gene therapy and 127 focused on antisense therapies [12].

#### 1.3.1 Recombinant Therapeutic Proteins

During the 20th century, advances in knowledge about the molecular basis of diseases revealed that many disorders were related to deficiencies in specific proteins. Because in healthy organisms the physiologic levels of such proteins are usually low, commercial manufacture of these proteins by isolating them from material obtained from healthy donors was shown in almost all cases to be unfeasible. In the 1970s, the development of recombinant DNA (rDNA) technology made it virtually possible to produce any desired human protein by means of cloning and expressing the respective human gene in a host cell line then to be cultivated on a large scale under controlled conditions.

Walsh [13] proposed the classification of recombinant therapeutic proteins as firstand second-generation biopharmaceuticals. The former are proteins that have an amino acid sequence identical to the native protein found in healthy organisms, intended for simple replacement or an increase in circulating levels in deficient patients. On the other hand, second-generation biopharmaceuticals are engineered versions of human proteins which have been modified by protein engineering techniques to achieve improved therapeutic properties such as:

- generation of a product with an action that is either quicker or slower
- manipulation of the product half-life
- alteration of immunogenicity
- development of hybrid or fusion proteins

Protein engineering techniques used to develop second-generation biopharmaceuticals can involve different strategies:

- modification of the amino acid sequence, such as in commercial insulin and t-PA products
- deletion of domains, such as in commercial blood coagulation Factor VIII and also t-PA products
- generation of hybrid or fusion proteins, such as chimeric and humanized antibodies, bispecific antibodies, and proteins fused to immunoglobulin G (IgG) or albumin
- posttranslational engineering, such as the creation of additional glycosylation sites in erythropoietin or the pegylation of several proteins

In addition to the seven groups that have traditionally been used to classify therapeutic proteins [5], the approval of several fusion proteins in past years and the large number of such products in development have resulted in a further group. Thus, biopharmaceutical proteins can be divided into:

- **1.** recombinant cytokines
- 2. recombinant hematopoietic growth factors
- **3.** other recombinant growth factors
- 4. recombinant hormones
- 5. recombinant blood factors
- 6. recombinant enzymes
- 7. monoclonal antibodies and antibody-drug conjugates (ADCs)
- 8. fusion proteins

Among the wide range of products, some such as interleukins and erythropoietins seem to have attained saturation in terms of supply and demand, because no new product with such active ingredients was approved in 2010–14 [1]. On the other hand, therapeutic mAbs have gained increasing relevance in terms of new approvals, the total number of products under commercialization, production volume, and market value. Among the more recent products, almost all are humanized or fully human. Main therapeutic indications are different types of cancer and inflammatory/autoimmune diseases, such as rheumatoid arthritis and lupus. From 2010 until July 2014, mAbs represented 26.5% of the new biopharmaceutical products that received approval.

Sales of mAbs reached USD 63 billion in 2013, accounting for approximately 40% of biopharmaceutical sales. Because mAbs are usually administered in high doses to a relatively broad number of patients, their production volume is also significant compared with other products: It was estimated that in 2010 approximately 7 tons of purified mAb active ingredients were produced and the forecast for 2016 was 13.4 tons [1]. Among mAbs approved and under development, a small fraction consisted of ADCs [14]. The two most recent ADC products were Kadcyla (trastuzumab emtansine), sold by Roche and approved in the EU and the United States in 2013, and Adcetris (brentuximab vedotin), sold by the company Takeda and approved in the United States in 2011 and in the EU in 2012.

Fusion proteins are nonnative molecules obtained by the molecular combination of the gene sequences coding for two or more proteins, protein fragments, or peptides. The aim is to confer a longer half-life, manipulate cytotoxicity, or enable novel delivery and targeting routes [15]. Usually a linker (e.g., a peptide) is used in the construction of stable, bioactive fusion proteins [16,17]. The first fusion protein to be approved for human use, in 1998, was Enbrel (etanercept), which consists of two molecules of the 75-kDa fragment of tumor necrosis factor (TNF)- $\alpha$  receptor fused to the Fc portion of an IgG1, which is responsible for extending the half-life of the resulting fused product. Because etanercept binds reversibly to TNF- $\alpha$ , it is used to treat TNF-dependent inflammatory diseases such as arthritis. Right after its introduction to the market in 1998, Enbrel presented a great sales performance, which at first led to difficulties in production capacity to meet the unexpectedly high product demand. Enbrel continues to be the fourth most sold pharmaceutical product worldwide, with sales in 2013 at over USD 8.3 billion [11].

According to Belsey and Somers [16], of 43 projects involving fusion proteins, 19% were either already on the market or under analysis by regulatory agencies, 53% were in clinical phase I/II or II, and 28% were in clinical phase II/III or III. In terms of therapeutic indication, 63% of these projects focused on oncology or autoimmune diseases. Mellado and Castilho [5] compiled therapeutic recombinant proteins approved until that moment, indicating also information related to the production technology, such as the cell line/expression system and process details (bioreactor type, operation mode, etc.). A compilation up to July 2014 can be found in Walsh [1], but fewer process details have been included.

#### 1.3.2 Nucleic Acid–Based Therapies

Nucleic acid—based therapies are composed of gene therapies and antisense therapies. In both cases, the main targets are different types of cancer and monogenic disorders. The aim of these therapies is to modulate expression of a given gene that is associated with the disease. By means of gene therapy, the genetic material of patient cells is modified. A gene that is inappropriately activated can be blocked, a gene that has a defect can be corrected or replaced, or a heterologous gene can be introduced to enable the cell to produce some useful protein [18].

Gene therapy can be carried out ex vivo or in vivo. In the former case, cells are collected from the patient and genetically manipulated in the laboratory, and then expanded in vitro and reintroduced to the same patient. In the case of in vivo therapy, the genetic material is introduced to the patient and gene delivery can be either systemic or site-directed. In this case, the main challenges are related to directing the genetic material adequately to the target cells or tissues, as well as to achieving an effective expression of the desired gene. Viral vectors are commonly used to carry the gene, aiming at efficient and stable expression.

The first gene therapy product to be approved for human use was the product Gendicine, licensed in China in 2003. The product is manufactured by the company Shenzhen Sibiono GeneTech and is based on an adenoviral vector containing the p53 tumor supressor gene. In up to 50–70% of human tumors, the p53 gene is mutated and the p53 pathway is inactivated [19]. The expression of heterologous p53 protein helps control the tumor. Gendicine is produced using HEK 293 cells in fixed-bed bioreactors.

Long after the approval of Gendicine, the first gene therapy product was licensed in the so-called highly regulated markets: Glybera was approved in 2012 in the EU. It is based on an adeno-associated viral vector (AAV1) containing a gene that codes for lipoprotein lipase. This is a key enzyme in the metabolism of triglyceride-rich lipoproteins, and its deficiency causes hypertriglyceridemia and pancreatitis, and can be lethal. Considering that since 1989 approximately 2000 gene therapy clinical trials have been approved (64% of which were in the United States and 26% of which were in the EU [1]), the approval of Glybera in the EU can be considered a landmark. However, the cost of EUR 53,000 per vial or EUR 1.1 million per treatment (21 vials per patient) creates doubts as to whether the product will be successful and whether it will open the path to an increase in the number of licensed gene therapy products [20].

On the other hand, antisense therapies are based on a different mechanism for modulating the expression of a gene related to a disorder: Nucleic acid molecules are used to block the target mRNA molecules. Single-strand oligonucleotides, small interfering RNAs, micro RNAs, ribozymes, and other antisense compounds can be used to block expression of a given gene: for example, an oncogene in the case of cancer therapies [21].

In 2013 the US Food and Drug Administration (FDA) approved an antisense therapy, the first to have a systemic delivery and to target a chronic disorder, and thus to be administered lifelong. The active ingredient is an antisense oligonucleotide and the therapeutic indication is the rare disorder "homozygous familial hypercholesterolemia," which results in very high cholesterol levels and may cause heart attacks and even death in very young patients (e.g., at age 30 years). The product is named Kynamro and is commercialized by the companies Isis and Genzyme. It blocks the production of apolipoprotein B, an atherogenic protein that transports cholesterol in the bloodstream [22].

Although Kynamro is a single-strand oligonucleotide with just 20 nucleotides, and thus is produced by chemical synthesis and not rDNA, antisense therapies have historically been referred to as biopharmaceutical products [1]. Many other antisense

products are currently under clinical development and an increase in the number of approved products is expected to occur.

#### 1.3.3 Cell Therapies

Cell therapies consist of a medical intervention aiming at reestablishing the structure and function of a tissue by means of using cells; i.e., in this case the cells are the product. Cell therapy is applied to treat dysfunctions caused by trauma or disease, as well as by degeneration processes which can be premature or result from aging. Some cell therapies have been used successfully for a long time, such as in cardiac diseases, eliminating the need for transplantation; in serious burns, decreasing hospitalization time and mortality; in cerebrovascular accidents (CVA), reducing cases, reducing hospitalization time and sequelae; and in autologous bone marrow transplant.

Many advances related to cell therapies are intrinsically related to advances in stem cell research. A landmark for therapies based on adult stem cells was the first transplantation of bone marrow cells in the 1950s [23]. However, the derivation of mouse embryonic stem cells in 1981 [24] and of human embryonic stem cells in 1998 [25] attracted unprecedented and widespread scientific and clinical interest in the potential use of these cells for regenerative medicine and tissue engineering, because the pluripotency of these cells allows them to be differentiated into any cell type in the organism.

A further landmark consisted of the development of cell reprogramming techniques by Takahashi and Yamanaka [26,27]. These authors showed that adult, fully differentiated cells can be reprogrammed to the pluripotent state by means of the heterologous expression of genes associated with pluripotency. These reprogrammed cells, known as induced pluripotent stem (iPS) cells, are a major focus of research for the development of cell therapies and are also used as tools to screen for new drugs and in basic studies of diseases. A further advance of great scientific and clinical interest is the direct generation of functional differentiated cells by reprogramming, e.g., fibroblasts. Vierbuchen et al. [28] showed that mouse fibroblasts can be directly converted to functional neuronal cells through expression of three transcription factors (Ascl1, Myt1l, and Brn2). These neurons were termed induced neuronal cells. Until 2015, no therapies based on cells derived from pluripotent stem cells had been approved by the FDA [29]. The only clinical trial using cells derived from pluripotent stem cells, sponsored by the company Geron, was approved in 2009 by the FDA, but it was canceled by the sponsor in 2011 [30].

#### 1.3.4 Vaccines

In public health, there is no tool as effective (and also cost-effective) as vaccination. Not even antibiotics have had the effect of reducing population mortality as pronouncedly as have vaccines. Through vaccination, nine of the main human diseases have been controlled in several regions of the world: diphtheria, tetanus, yellow fever, pertussis, poliomyelitis, measles, mumps, rubeola, and smallpox. The latter was officially eradicated worldwide in 1979 as a consequence of effective vaccination actions leaded by the World Health Organization. In addition, great progress has been achieved through vaccination against influenza, hepatitis B, pneumococcus, rotavirus, and *Haemophilus influenza* type B.

The origin of vaccine development dates to the end of the 18th century, when Edward Jenner observed that milkmaids who were in contact with the cowpox virus, which does not cause the disease in humans, were protected against infection with the human smallpox virus [31]. Traditional vaccines that were developed since then, and especially during the 20th century, can target bacterial or viral infections, and generally have as active ingredient the corresponding pathogen in either an attenuated or inactivated form, eliciting an immune response and protective antibody formation in vaccinated individuals, but without causing the disease.

In past decades, advances have been made in the development of recombinant vaccines based on the expression of genes coding proteins of the pathogen. The first recombinant vaccine to be approved was the one against hepatitis B in 1986. It contains the hepatitis B surface antigen, which forms multimers containing about 85–155 molecules of the antigen, forming tridimensional virus-like particles (VLPs) that mimic the virus and are immunogenic [6]. Two recombinant vaccines (Cervarix and Gardasil) against human papilloma virus, produced by GSK and Merck & Co., respectively, have been approved. These contain the recombinant form of the main protein of the virus capsid (the L1 protein). This protein forms in vitro tridimensional VLPs that consist of 72 pentamers of L1. These VLPs are immunogenic when administered to humans.

Recombinant vaccines currently account for approximately 10% of approved biopharmaceutical products. Besides an increase in approved VLP-based vaccines, new vaccines based on plasmid DNA or on mRNA are expected to be licensed in the next years. Between 2010 and July 2014, recombinant vaccines that were approved include [1]:

- Bexsero (Novartis), a multicomponent subunit vaccine against meningitis B produced in *E. coli* and approved in the EU in 2013
- Flublok (Protein Sciences), a vaccine based on VLPs of recombinant hemagglutinin from three different influenza virus strains, produced in insect cells and approved in the United States in 2013
- Provenge (sipuleucel-T, Dendreon), a vaccine approved in 2010 in the United States and in 2013 in the EU, consisting of autologous peripheral blood cells combined with prostate acid phosphatase and recombinant G-CSF produced in insect cells

#### 1.3.5 In Vitro and In Vivo Diagnostic Products

Hybridoma technology, developed by Koehler and Milstein [32], made the generation of antibodies with predefined specificities possible. These antibodies were named monoclonal antibodies (mAbs). Besides their therapeutic use (previously discussed), these antibodies are widely employed for in vitro and in vivo diagnostics. In the case of in vitro diagnotics, antibodies are mainly used in laboratory immunoassays, such as enzyme-linked immunosorbent assays (ELISA), blots and others. mAbs of murine origin are frequently used. Owing to the small quantities used in the assays, the production scale of these antibodies is significantly lower than that of therapeutic mAbs. Moreover, the purity requirements, although high, are considerably lower than for injectable therapeutic mAbs.

In the case of in vivo diagnostic mAbs, labeled antibodies (e.g., radiolabeled) are injected into patients, aiming at the in situ detection of tumors and other conditions. In this case, because the mAb is injected into the patient, quality and manufacturing requirements resemble those applicable to therapeutic mAbs [33]. Because there is usually no repeated administration of the product, murine mAbs are acceptable.

The number of approved in vivo diagnosis mAb-based products is not large [34]. One example is the product ProstaScint, approved for commercialization in 1996 in the United States, which is a murine mAb labeled with In<sup>111</sup>, used as an in vivo diagnostic imaging agent in patients diagnosed with biopsy-proven prostate cancer who are at high risk for metastases. Another example is LeukoScan, approved in 1997 in the EU, which is a technetium-99m—labeled mAb fragment specific for NCA-90, a surface antigen found on activated granulocytes, and is indicated to localize infections and inflammations in bone in patients with suspected osteomyelitis.

#### 1.4 Production Processes for Recombinant Therapeutic Proteins

Recombinant protein therapeutics represent the most mature and economically most important class of biopharmaceutical products. Therefore, the focus of this section is on the technologies used to manufacture them. However, many similarities exist among the production processes for the different classes of biopharmaceutical products, so much of the discussion here also applies to the production of gene therapy vectors, viral vaccines, diagnostic mAbs, and cell therapies.

#### 1.4.1 Product Complexity Versus Expression System

Recombinant proteins used as therapeutic agents or as vaccines are frequently glycoproteins and are usually large and complex, with different posttranslational modifications such as glycosylation,  $\gamma$ -carboxylation, sulfation, phosphorylation, acetylation, methylation, and cleavage of peptides [35]. These posttranslational modifications are important because they affect the biological activity, stability, and immunogenicity of the recombinant products. Posttranslational modifications which either lack or result in structures that differ from corresponding human standards can result in recombinant proteins with low biological activity, decreased stability, and high immunogenicity, which make their commercial use unfeasible. Glycosylation can further affect protein folding, transport, target recognition, and binding. For these reasons, for the production of a recombinant protein, it is crucial to know its structure beforehand and carefully select an adequate expression host.

Among different hosts, the structure of glycosylated structures attached to the amino acid chain can vary a lot, because the potential diversity of chemical structures is enormous owing to:

- the sequence of monossacharides
- the position of glycosidic bounds
- the configuration (alpha or beta) of the glycosidic bond
- the number of ramification points
- the position of ramifications

Human glycoproteins usually hold glycans of the so-called complex type, with several ramifications that yield a "multiantennary" structure. Different monosaccharides are added to an initial core rich in mannose, and the antennas terminate with N-acetyl neuraminic acid molecules, also known as sialic acid. When a human protein is produced in its recombinant form for replacement therapies, the aim is to produce a copy that is as similar as possible to the native protein found in healthy organisms. Thus, the glycosylation pattern of the host cell should be as similar as possible to the human glycosylation pattern.

The glycan structure depends on the availability of precursor sugars, culture conditions (e.g., medium composition, temperature, pH, metabolites), and the pool of glycosidases and glycosyltransferases present in the host cell. Differently from transcription and translation, in glycosylation there is no template for biosynthesis.

Because *E. coli*, the workhorse among bacterial expression systems, does not have the cellular machinery for glycosylation processing [36], it has been used as the host for the expression of nonglycosylated proteins such as insulin, or of glycosylated proteins whose activity is not strongly affected by the lack of glycans, such as occurs with granulocyte colony stimulating factor (G-CSF).

Mammalian cells have a glycosylation pattern similar to that of human cells, and therefore represent the most widely used expression system for the production of biopharmaceutical products. Yeast, insect, and plant cells typically produce high-mannose glycostructures which can be immunogenic to humans.

One example of highly glycosylated protein is erythropoietin (EPO), in which 40% of the molecular mass refers to the glycidic portion and 60% to the amino acid chain. Changes in the glycosylation pattern strongly affect the isoelectric point, biological activity, and stability of EPO molecules. Thus, international pharmacopoeias establish which glycoforms of EPO can be present in the final product, according to their isoelectric point. Undesirable glycoforms must be minimized during the cell culture process and those eventually present need to be removed during the purification process.

In specific cases in which the presence of exposed mannose residues is exceptionally required for protein action, such as in native glucocerebrosidase, it is interesting to produce the recombinant glycoprotein in an expression system that yields high-mannose glycostructures. In the case of glucocerebrosidase, the product manufactured by Genzyme in CHO cells needs to have an additional cleavage step after purification, to expose mannose residues of the glycan core. Therefore, Protalix developed a recombinant glucocerebrosidase expressed in cultured carrot cells, so that the recombinant protein is directly synthesized bearing exposed mannose residues [37]. However, this is an unusual situation, because almost all human glycoproteins have a complex-type glycosylation pattern characterized by terminal sialic acid molecules.

The effects of glycosylation can also be explored to improve product properties: for example, using protein engineering techniques to generate hyperglycosylated molecules. Amgen developed a hyperglycosylated EPO which has two extra glycosylation sites in addition to the four sites present in native EPO. The resulting molecule has a threefold higher half-life, which decreases the frequency of required injections.

Other relevant aspects regarding the choice of the expression system are:

- if the host cell line is able to secrete the recombinant protein or if it accumulates the product intracellularly
- if the cell line has robust growth in large-scale stirred bioreactors
- if the cell line has characteristics that increase safety against contamination, e.g., of viral origin

Mammalian cells are the main expression system in the biopharmaceutical industry; as of 2014 they responded to the production of 52% of approved biopharmaceuticals. Bacteria represented 19%, yeast 16.5%, human cells 4%, and other hosts 8.5% [1]. Among the mammalian cells, the CHO cell line predominates, responding to 35.5% of approved biopharmaceuticals (or 68.2% of those produced in mammalian cells).

This is due to different factors in addition to the adequate glycosylation pattern conferred by CHO cells. These cells secrete recombinant proteins in the culture medium, which makes the purification processes easier and cheaper. Furthermore, they are robust, can be cultivated in suspension in large-scale bioreactors, and are a poor substrate for the replication of human viruses, which represents an additional safety factor in the event of viral contamination of the process.

#### 1.4.2 Manufacturing Technology: Upstream Processing

Despite the more sophisticated bioprocess requirements presented by animal cells, including the need for complex, expensive culture media composed typically on the order of 50 components, they account for approximately 60–65% of all approved biopharmaceuticals, if mammalian, human, and insect cells are taken together [1,38]. Owing to the growing demand for biopharmaceuticals, there are continuously great efforts to improve mammalian cell-based technologies, with the aim of increasing productivity and reducing costs [38,39]. Mammalian cells have been cultured in vitro in the laboratory for over 100 years, but it was just in the past 60–70 years that research aiming at technological aspects and scale-up has been intensified, at first motivated by advances in the area of viral vaccines produced by mammalian cell culture [40,41].



FIGURE 1.1 Overview of different bioreactor operation modes: batch, fed-batch, simple continuous mode (chemostat), and continuous mode with cell recycle (perfusion).

Different types of bioreactors can be used for animal cell culture, such as roller bottles and hollow-fiber bioreactors. However, most cell culture processes aiming at producing biopharmaceuticals both on a laboratory and industrial scale are based on the use of stirred-tank bioreactors for the cultivation of suspension-adapted cells [38,42].

Bioreactors can be run under different operation modes, which differ basically in the way in which medium is fed to them (Fig. 1.1). Most commercial products are still produced by batch and fed-batch processes, owing to their lower complexity. In fed-batch processes, the addition of a concentrated nutrient solution extends the culture time and allows higher titers to be obtained. The development, in the past 10–20 years, of fed-batch feed solutions with optimized composition, added at optimized feed rates to the bioreactor, enabled obtaining high cell concentrations (on the order of  $30 \times 10^6$  cells/mL) and high product concentrations (on the order of 10 g/L).

Simple continuous cultures, on the other hand, have traditionally been used primarily for physiologic studies [43] and do not represent a good alternative to production processes. Because of the low specific growth rate of animal cells, the highest dilution rate that can be used with no cell washout is relatively low, and thus the maximum concentration of cells that can be achieved is not high.

However, if a cell retention device is coupled with a continuous bioreactor, resulting in a continuous process with cell recycle, also known as a perfusion process, continuous replacement of culture medium can be carried out at increasing flow rates, supplying a progressively increasing cell concentration with nutrients [44]. Cell retention in the bioreactor allows very high cell densities to be achieved, resulting in the continuous harvest of large product amounts at high dilution rates, thus giving volumetric productivities that are usually 10 times higher than in fed-batch processes. When a desired high cell concentration is achieved, a controlled cell bleed can be carried out, avoiding oxygen transfer limitations and allowing a steady state to be achieved under which the bioreactor can be operated for weeks or months [45,46]. The residence time of product in the bioreactor is short, which contributes to its quality and consistency. The high cell density in the bioreactor also allows the culture medium to contain fewer nutritional supplements, because the cells secrete to the medium growth factors and other autocrine factors [47]. However, maintenance of a continuously operating system attached to a cell retention device (usually placed in an external recirculation loop) for a long period increases both the operational complexity and contamination risk [48].

In the past 2 decades, for stable proteins such as mAbs, fed-batch processes have become popular as a low-complexity process alternative that, when adequately developed, allows high concentrations of cells and product to be obtained. The use of perfusion processes on an industrial scale was mainly restricted to sensitive proteins such as enzymes and blood coagulation factors, which needed to be quickly removed from the bioreactor to maintain their functionality. However, a change in the technological paradigm has begun in the biopharmaceutical industry, and there is a growing trend to adopt continuous perfusion processes for both stable and unstable molecules, eventually integrated into continuous purification steps. This trend has arisen from the growing need for process intensification, cost reduction, and the development of more flexible plants.

#### 1.4.3 Manufacturing Technology: Downstream Processing

Purity requirements for injectable products for human use are very high. Strict regulatory guidelines establish limits for critical contaminants, such as DNA and host cell proteins. For these reasons, purification processes in the biopharmaceutical industry consist of many sequential steps, including several sophisticated chromatography steps, to reach the degree of purity required for the product. The need for sequential downstream processing with many steps, exploiting differences in properties between the protein of interest and other biomolecules (such as the charge, size, surface hydrophobicity, and biospecific affinity), results in losses of product along the processing steps. Thus, as product purity becomes higher along processing, product yield and recovery decrease.

For biopharmaceutical products produced in bacteria, generally found as insoluble aggregates (inclusion bodies) in the cytoplasm, there are additional steps of cell lysis, protein renaturation, and separation of cell debris, as well as increased difficulty owing to the large amount of intracellular contaminants released upon cell lysis. For these reasons, cumulative product loss is significant, leading to decreased global yields that can be as low as 10%. Even when mammalian cells that usually secrete the recombinant protein to the extracellular medium are used, thus with no extra load of intracellular contaminants, global yield is not high, generally in the range of 30–70%. Global recoveries achieved in the downstream process have a direct impact, given a desired production capacity, on the bioreactor scale and on the sizing of separation equipment, thus affecting both capital and operational costs.

Thus, considering the manufacturing process as a whole, downstream processing steps can represent up to 90% of production costs of therapeutic proteins [49,50]. The fraction of manufacturing costs owing to product purification are strongly related to the number of steps required, because the processing time increases and the product yield decreases with an increase in the number of purification steps.

A way to reduce the number of downstream processing steps is to replace low selectivity techniques with more selective ones, such as affinity chromatography, thus achieving the same result with a lower number of steps. Affinity adsorption relies on the selectivity of reversible interactions between the protein of interest and a ligand, frequently resulting in very high purification factors at relatively high recoveries. However, affinity ligands are usually expensive, and the number of cycles that can be used is often limited because they are usually of biological origin and can be prone to degradation. On the other hand, ion exchange and hydrophobic interaction chromatography steps are carried out using adsorbents that are more stable and less expensive. Thus, the use of these techniques can contribute to decreasing downstream processing costs. Separation of molecules from a mixture in these cases is based on the differences of charge and surface hydrophobicity, respectively. Despite being less selective than affinity chromatography, they can result in efficient purification processes which are satisfactory in terms of both cost and performance.

#### 1.5 Technological Challenges and Future Perspectives

Many changes are expected in the near future in the biopharmaceutical industry [9]. Global innovative companies will have to focus on product innovation to remain at the technological frontier, exploring new operational designs. Biosimilar companies will have to focus on costs, quality, and scale, because for them speed, innovation in manufacturing processes, and operational excellence will determine their survival. Contract manufacturing organizations (CMOs) will have to keep at the frontier of operational efficiency and innovation in manufacturing processes, warranting a high reputation in terms of services and performance.

Although the biopharmaceutical industry has traditionally been conservative and has been characterized mainly by incremental innovations [51], true evolution in the production technologies and operational capacity will be required in the future, because simple technological and operational gains will not be enough to increase productivity and quality and to reduce costs as much as needed. Companies will have to take different issues into consideration [9]:

- decrease costs related to the manufacturing process and to quality issues, improving technology from cell line development to downstream processing
- increase operational agility and equipment use at no expense of quality aspects, to eliminate bottlenecks, produce multiple products in fewer lines, and respond quickly to the demands of a volatile market
- increase production capacity and adopt new technologies enabling the design of flexible plants to suit specific product and market conditions, which can be based on stainless-steel equipment, disposable systems, or hybrid combinations of them
- be able to introduce new products and new technologies quickly to push a greater number of molecules toward commercial launches

- decide which activities should be carried out internally and which should be outsourced to CROs (contract research organizations) or to CMOs
- be able to keep talented human resources who should be able to handle these issues and future challenges that will emerge in this rapidly evolving sector

Indeed, a major technological shift currently seems to be under way [52]. According to Warikoo et al. [53], by the beginning of 2010 approximately 90% of the biopharmaceutical products were produced on an industrial scale through batch or fed-batch processes, but almost all innovative companies and suppliers were studying continuous bioprocessing [54]. The business context, with rising competition from off-patent biopharmaceuticals, broad pipelines of products, and rapidly evolving market demands, started challenging industry conservatism and pushing technological development toward continuous processes, which could provide higher productivity, improved product quality, lower costs, and enhanced product/scale flexibility [51,53].

Advances in disposable systems for upstream and downstream processing, as well as encouraging signs given by regulators, have further contributed to increasing industry interest in a new, continuous-flow production platform [52]. This new technology should be applicable to both mAbs and nonantibody proteins and should integrate upstream and downstream operations, allowing extra-low production cycles. A new paradigm toward integrated continuous biomanufacturing is currently being established in the biopharmaceutical industry, and the facilities of the future will be efficient, simple, compact, automated, mobile, and cost-effective, distributed around many countries and offering multiproduct and multiphase (clinical/commercial) manufacturing capability [53,55].

#### References

- [1] G. Walsh, Biopharmaceutical benchmarks 2014, Nature Biotechnology 32 (2014) 992-1000.
- [2] H. Walter, Biotherapies, immunotherapies, targeted therapies, biopharmaceuticals... which word should be used? Médecine/Sciences (Paris) 30 (2014) 567–575.
- [3] J. Neves, B. Sarmento, Mucosal Delivery of Biopharmaceuticals: Biology, Challenges and Strategies, Springer Science & Business Media, New York, 2014.
- [4] G. Walsh, Biopharmaceuticals and biotechnology medicines: an issue of nomenclature, European Journal of Pharmaceutical Sciences 15 (2002) 135–138.
- [5] M.C.M. Mellado, L.R. Castilho, Recombinant therapeutic proteins, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor and Francis, Andover, 2008, pp. 389–408 (Chapter 16).
- [6] A. Roldão, M.C.M. Mellado, L.R. Castilho, M.J.T. Carrondo, P.M. Alves, Virus-like particles in vaccine development, Expert Review of Vaccines 9 (2010) 1149–1176.
- [7] Kalorama, Global Market for Biomarkers A Kalorama Information Market Intelligence Report, 2013. Available to acquire from: http://www.kaloramainformation.com.
- [8] EFPIA, The Pharmaceutical Industry in Figures, 2014. http://www.efpia.eu/uploads/Figures\_2014\_ Final.pdf (access in June 2015).

- [9] McKinsey, Rapid Growth in Biopharma: Challenges and Opportunities, 2014. www.mckinsey.com/ insights/health\_systems\_and\_services/rapid\_growth\_in\_biopharma (access in June 2015).
- [10] BioPlan Associates, Ninth Annual Report and Survey of Biopharmaceutical Manufacturing Capacity and Production a Study of Biotherapeutic Developers and Contract Manufacturing Organizations, 2012. http://www.bioplanassociates.com/images/9th\_Annual\_2012\_ReportTable\_of\_Contents.pdf (access in June 2015).
- [11] FirstWord Pharma, Pharma's 50 Biggest Selling Drugs: AbbVie's Humira Joins the \$10 Billion Club, 2014. http://www.firstwordpharma.com/node/1194000#axzz3Y2ixQ0qJ (access in June 2015).
- [12] PhRMA, Innovation in the Biopharmaceutical Pipeline: A Multidimensional View, 2013. http://www.phrma.org/sites/default/files/pdf/2013innovationinthebiopharmaceuticalpipelineanalysisgroupfinal.pdf (access in June 2015).
- [13] G. Walsh, Second-generation biopharmaceuticals, European Journal of Pharmaceutics and Biopharmaceutics 58 (2004) 185–196.
- [14] Y. Feng, Z. Zhu, W. Chen, P. Prabakaran, K. Lin, D.S. Dimitrov, Conjugates of small molecule drugs with antibodies and other proteins, Biomedicines 2 (2014) 1–13.
- [15] S.R. Schmidt, Fusion proteins as biopharmaceuticals applications and challenges, Current Opinion in Drug Discovery & Development 12 (2009) 284–295.
- [16] M. Belsey, G. Somers, Analyzing and forecasting the fusion protein market and pipeline, in: S.R. Schmidt (Ed.), Fusion Protein Technologies for Biopharmaceuticals: Applications and Challenges, John Wiley and Sons, New York, 2013, pp. 25–38 (Chapter 2).
- [17] X. Chen, J.L. Zaro, W.C. Shen, Fusion protein linkers: property, design and functionality, Advanced Drug Delivery Reviews 65 (2013) 1357–1369.
- [18] K.M. Lima, S.A. Santos, C.L. Silva, J.M. Rodrigues Jr., Gene therapy, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor and Francis, Andover, 2008, pp. 489–504 (Chapter 21).
- [19] K. Suzuki, H. Matsubara, Recent advances in p53 research and cancer treatment, Journal of Biomedicine and Biotechnology (2011) 7, 978312.
- [20] S. Ylä-Herttuala, Glybera's second act: the curtain rises on the high cost of therapy, Molecular Therapy 23 (2015) 217–218.
- [21] A. Tekewe, Gene therapy and antisense therapy: new approaches for cancer therapy a review, International Journal of Drug Research and Technology 2 (2012) 144–162.
- [22] Isis Pharmaceuticals, Genzyme and Isis Announce Submission of U.S. NDA for Kynamro™ (Mipomersen Sodium) in Homozygous Familial Hypercholesterolemia, 2013. http://ir.isispharm.com/phoenix.zhtml?c=222170&p=irol-newsArticle\_Print&ID=1678219 (access in June 2015).
- [23] F.R. Appelbaum, Hematopoietic-cell transplantation at 50, New England Journal of Medicine 357 (2007) 1472–1475.
- [24] M.J. Evans, M.H. Kaufmann, Establishment in culture of pluripotential cells from mouse embryos, Nature 292 (1981) 154–156.
- [25] J.A. Thomson, J. Itskovitz-Eldor, S.S. Shapiro, M.A. Waknitz, J.J. Swiergiel, V.S. Marshall, J.M. Jones, Embryonic stem cell lines derived from human blastocysts, Science 282 (1998) 1145–1147.
- [26] K. Takahashi, S. Yamanaka, Induction of pluripotent stem cells from mouse embryonic and adult fibroblast cultures by defined factors, Cell 126 (2006) 663–676.
- [27] K. Takahashi, K. Tanabe, M. Ohnuki, M. Narita, T. Ichisaka, K. Tomoda, S. Yamanaka, Induction of pluripotent stem cells from adult human fibroblasts by defined factors, Cell 131 (2007) 861–872.
- [28] T. Vierbuchen, A. Ostermeier, Z.P. Pang, Y. Kokubu, T.C. Südhof, M. Wernig, Direct conversion of fibroblasts to functional neurons by defined factors, Nature 463 (2010) 1035–1041.

- [29] FDA, What Are Stem Cells? How Are They Regulated?, 2015. http://www.fda.gov/AboutFDA/ Transparency/Basics/ucm194655.htm (access in June 2015).
- [30] D. Lukovic, M. Stojkovic, V. Moreno-Manzano, S.S. Bhattacharya, S. Erceg, Perspectives and future directions of human pluripotent stem cell-based therapies: lessons from Geron's clinical trial for spinal cord injury, Stem Cells and Development 23 (2014) 1–4.
- [31] S. Riedel, Edward Jenner and the history of smallpox and vaccination, Proceedings (Baylor University. Medical Center) 18 (2005) 21–25.
- [32] G. Köhler, C. Milstein, Continuous cultures of fused cells secreting antibody of predefined specificity, Nature 256 (1975) 495–497.
- [33] WHO, Guidelines for Assuring the Quality of Monoclonal Antibodies for Use in Humans, World Health Organization Technical Report Series No. 822, 1992, http://www.who.int/biologicals/areas/ WHO\_TRS\_822\_A3\_mAbs.pdf (access in July 2015).
- [34] B.T. Smith, Introduction to Diagnostic and Therapeutic Monoclonal Antibodies, 2012. http://pharmacyce.unm.edu/nuclear\_program/freelessonfiles/Vol17Lesson1.pdf (access in July 2015).
- [35] M. Butler, Post-translational modifications of recombinant proteins, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor and Francis, Andover, 2008, pp. 129–146 (Chapter 6).
- [36] J.M. Carton, W.R. Strohl, Protein therapeutics (introduction to biopharmaceuticals), in: R. Ganellin, S.M. Roberts, R. Jefferies (Eds.), Introduction to Biological and Small Molecule Drug Research and Development – Theory and Case Studies, Elsevier, Amsterdam, 2013, pp. 127–159 (Chapter 4).
- [37] Y. Shaaltiel, D. Bartfeld, S. Hashmueli, G. Baum, E. Brill-Almon, G. Galili, O. Dym, S.A. Boldin-Adamsky, I. Silman, J.L. Sussman, A.H. Futerman, D. Aviezer, Production of glucocerebrosidase with terminal mannose glycans for enzyme replacement therapy of Gaucher's disease using a plant cell system, Plant Biotechnology Journal 5 (2007) 579–590.
- [38] M. Jesus, F.M. Wurm, Manufacturing recombinant proteins in kg-ton quantities using animal cells in bioreactors, European Journal of Pharmaceutics and Biopharmaceutics 78 (2011) 184–188.
- [39] S. Meeks, C. Josephson, Should hemophilia treaters switch to albumin-free recombinant factor VIII concentrates? Current Opinion in Hematology 13 (2006) 457–461.
- [40] O. Merten, Introduction to animal cell culture technology past, present and future, Cytotechnology 50 (2006) 1–7.
- [41] G. Kretzmer, Industrial processes with animal cells, Applied Microbiology and Biotechnology 59 (2002) 135–142.
- [42] K. Jayapal, K. Wlaschin, W.S. Hu, M. Yap, Recombinant protein therapeutics from CHO cells 20 years and counting, Chemical Engineering and Processing 103 (2007) 40–47.
- [43] J.B. Griffiths, Animal cell culture processes batch or continuous? Journal of Biotechnology 22 (1992) 21–30.
- [44] E. Chico, G. Rodriguez, A. Figueredo, Bioreactors for animal cells, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor and Francis, Andover, 2008, pp. 221–258 (Chapter 9).
- [45] F.M. Wurm, Production of recombinant protein therapeutics in cultivated mammalian cells, Nature Biotechnology 22 (2004) 1393–1398.
- [46] L.R. Castilho, R.A. Medronho, Animal cell separation, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor and Francis, Andover, 2008, pp. 273–294 (Chapter 11).
- [47] S. Woodside, B. Bowen, J. Piret, Mammalian cell retention devices for stirred perfusion bioreactors, Cytotechnology 28 (1998) 163–175.

- [48] A.A. Shukla, J. Thoemmes, Recent advances in large-scale production of monoclonal antibodies and related proteins, Trends in Biotechnology 28 (2010) 253–261.
- [49] A.M. Moraes, L.R. Castilho, S.M.A. Bueno, Product purification processes, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor and Francis, Andover, 2008, pp. 295–328 (Chapter 12).
- [50] M.T. Cunha, M.R. Aires-Barros, J.M.S. Cabral, Extraction for rapid protein isolation, in: R. Hatti-Kaul, B. Mattiasson (Eds.), Isolation and Purification of Proteins, Marcel Dekker, New York, 2003, pp. 321–372.
- [51] U. Gottschalk, Biomanufacturing: time for change? Pharmaceutical Bioprocessing 1 (2013) 7–9.
- [52] L.R. Castilho, Continuous animal cell perfusion processes: the first step toward integrated continuous biomanufacturing, in: G. Subramanian (Ed.), Continuous Processing in Pharmaceutical Manufacturing, Wiley-VCH, New York, 2014, pp. 306–326 (Chapter 6).
- [53] V. Warikoo, R. Godawat, K. Brower, S. Jain, D. Cummings, E. Simons, T. Johnson, J. Walther, M. Yu, B. Wright, J. McLarty, K.P. Karey, C. Hwang, W. Zhou, F. Riske, K. Konstantinov, Integrated continuous production of recombinant therapeutic proteins, Biotechnology and Bioengineering 109 (2012) 3018–3029.
- [54] A. Sinclair, Addressing the process & economic dimensions (workshop introductory talk), in: Proceedings of the Integrated Continuous Biomanufacturing Conference, October 2013, Barcelona, Spain, Engineering Conferences International, New York, 2013.
- [55] K.B. Konstantinov, The promise of continuous bioprocessing, in: Proceedings of the Integrated Continuous Biomanufacturing Conference, October 2013, Barcelona, Spain, Engineering Conferences International, New York, 2013.

### **Biopharmaceuticals and Biosimilars**

A.M.M. Basso<sup>1,3</sup>, G.S. Prado<sup>2,3</sup>, P.B. Pelegrini<sup>3,4</sup>, M.F. Grossi-de-Sa<sup>2,3,\*</sup>

<sup>1</sup>UNIVERSITY OF BRASÍLIA (UNB), BRASÍLIA-DF, BRAZIL; <sup>2</sup>CATHOLIC UNIVERSITY OF BRASÍLIA (UCB), BRASÍLIA-DF, BRAZIL; <sup>3</sup>EMBRAPA — GENETIC RESOURCES AND BIOTECHNOLOGY, BRASÍLIA-DF, BRAZIL; <sup>4</sup>DIAGENE DIAGNÓSTICOS MOLECULARES LTDA, BRASÍLIA-DF, BRAZIL

#### 2.1 Introduction

Until the 19th century, pharmaceuticals provided to the population were based on plantderived natural compounds and were applied in the form of infusions, teas, and creams. In 1885, Louis Pasteur attenuated the rabies virus in the laboratory and applied it to a boy who was bitten by a dog with rabies disease, thereby saving his life [1]. It was the first report of virus manipulation as medicine in the history of human health sciences.

Later, in 1928, Alexander Fleming developed the penicillin antibiotic. Immediately, pharmaceutical companies such as Pfizer, Squibb, and Merck tried to synthesize it for large-scale production without success. Hence, the North Agriculture Regional Search Laboratory in the United States developed a methodology to produce penicillin but did not have a specific application. The need for the large-scale production of penicillin occurred during the Second World War (1939–45), because the number of American soldiers dying from bacterial infections and gangrene, accounting for 12–15% of all men. The administration of penicillin to wounded soldiers reduced mortality to 3%, saving hundreds of lives [2,3].

The post–Second World War period was the beginning of the biopharmaceutical industry, when the production of large-scale antibiotics started, as did the development of novel drugs from animal sources [4]. During the 1970s, researchers developed DNA recombinant technology, which facilitated the production of recombinant proteins. One of the first molecules produced by this technology was somatostatin [5]. Through genetic engineering techniques, the gene coding for somatostatin was inserted into *Escherichia coli* cells, which expressed a recombinant protein with activity similar to the human hormone [5]. In 1982, the same procedure was applied for the production and commercialization of the first recombinant insulin [6].

Over the ensuing years, other heterologous systems were introduced for producing recombinant human molecules, including yeasts, insects, and mammalian cells. The new

<sup>\*</sup>Corresponding Author.

class of biologically derived therapeutic molecules was called biopharmaceuticals [7]. Hence, 60% of new drugs approved by the Food and Drug Administration (FDA) until 2013 were biopharmaceuticals that were developed for the treatment of cancer and infectious, cardiovascular, and inflammatory diseases [8]. Biopharmaceuticals have been on the market for the past 20 years and now comprise a diverse number of molecules, such as coagulation factors, hormones, enzymes, monoclonal antibodies (mAbs), and vaccines. Because of their complexity in three-dimensional structure and size, each process applied to produce a molecule will result in a biopharmaceutical with specific physicochemical properties and biological activity [9].

# 2.2 Biopharmaceuticals: Classes, Variations, and Applications

The composition of biopharmaceuticals can be peptides, proteins, or nucleic acids [10]. Cell-based systems can also be considered biopharmaceuticals for cell therapy applications [11]. Thus, we can group the broad nature of biopharmaceuticals into distinct classes (Figs. 2.1 and 2.2).

Among the protein-based biopharmaceuticals, mAbs comprise one of the most structurally and functionally complex classes. They are produced using in vitro display techniques [12] (i.e., phage, ribosome, bacterial, yeast, or mRNA display), hybridoma technology [13], or heterologous expression systems [14]. Most are used to prevent and treat cancer (e.g., Rituximab, Bevacizumab, Trastuzumab, Cetuximab) and autoimmune diseases (e.g., Adalimumab, Infliximab, Golimumab) [15,16].

Enzymes are another complex class and include molecules such as glucocerebrosidase,  $\alpha$ -galactosidase, hyaluronidase, and asparaginase. In the case of hormones, the structural complexity is even wider and ranges from single-chain polypeptides and large heterodimeric proteins such as human growth hormone (somatotropin) and follicle-stimulating hormone (FSH), respectively, to simpler proteins and small peptides such as insulin and glucagon [15,17].

The blood factors group is composed of blood-clotting proteins (coagulation factors) such as factor VIII and factor IX, which are applied in the treatment of hemophilia A and B, respectively. Similarly, thrombolytic proteins (tissue plasminogen activator), anticoagulants (hirudin), hematopoietic growth factors (erythropoietin (EPO)), and granulocyte colony-stimulating factor (G-CSF) are biopharmaceutical classes used to treat cardiovascular diseases [15,17,18].

Furthermore, growth factors for non-blood cell types are used as therapeutic molecules. They include the vascular endothelial growth factor and somatomedins (insulin-like growth factors). Other molecules from this group include osteogenic proteins (bone morphogenetic protein) such as eptotermin- $\alpha$  for bone tissue regeneration at a clinical level. Finally, cytokines comprising cell signaling modulator molecules in immune responses are composed of two main subclasses: interferons (IFNs), such as IFN-alfa, IFN-beta, and IFN-gamma, which are used to treat hepatitis B and C, multiple



FIGURE 2.1 Groups/classes of biopharmaceuticals: some examples of each: (A) mAb: immunoglobulin G2a (PDB ID 1IGT); (B) IFNs: alfa, beta, and gamma (*left*), and three-dimensional structure of IFN-beta (*right*) (PDB ID 1AU1); (C) enzyme: α-galactosidase (PDB ID 1R46); (D) hormone: somatotropin (PDB ID 1HGU); (E) cell growth factor: vascular endothelial growth factor (PDB ID 1WQ8); (F) IL-2 (PDB ID 1M47); (G) thrombolytic: tissue plasminogen activator (PDB ID 1A5H); (H) blood factor: coagulation factor IX (PDB ID 1PFX); (I) colony-stimulating factor: G-CSF (PDB ID 1RHG); (J) hematopoietic growth factor: EPO (PDB ID 1BUY); (K) anticoagulant: hirudin (PDB ID 1DWC); (L) vaccine: hepatitis B surface antigen; (M) aptamer: pegaptanib. *PDB ID*, Identification code in *Protein Data Bank*. *PDB* (http://www.rcsb.org/pdb/home/home.do); E. NG, D.T. Shima, P. Calias, E.T. Cunningham, D.R. Guyer, A.P. Adamis, Pegaptanib, a targeted anti-VEGF aptamer for ocular vascular disease. Nature Reviews Drug Discovery 5 (2006) 123–132; F.J. Van Hemert, H.L. Zaaijer, B. Berkhout, V.V. Lukashov, Occult hepatitis B infection: an evolutionary scenario. Virology Journal 5 (2008) 1–13.



FIGURE 2.2 Venn diagram grouping biopharmaceutical classes according to their nature. Among the proteins, there are coagulation factors, cell growth factors, colony-stimulating factors, monoclonal antibodies, hematopoietic growth factors, anticoagulants, thrombolytics, and enzymes. Among the nucleic acids, there are siRNAs and aptamers. Hormones can be divided into peptides (e.g., glucagon) or proteins (e.g., insulin), and vaccines can be divided into all groups, although DNA and peptide-based vaccines have reached only the clinical phases to date and are not currently available on the market for use in humans. A.M.M. Basso, P.B. Pelegrini, F. Mulinari, A.B. Viana, L.P. Silva, M.F. Grossi-de-Sa, Recombinant glucagon: a differential biological activity. AMB Express 5 (20) (2015) 1–9.

sclerosis, and some types of cancer; and interleukins (ILs), such as IL-1, IL-2, and IL-12, which are used to activate immune cells, especially T-cells, and to treat autoimmune diseases [15,17,19].

Therefore, there is great diversity in protein therapeutic classes. However, many of the molecules used to prevent several diseases fit into the vaccines group. Prophylactic vaccines are made of peptides, DNA, or recombinant proteins, and some of these recombinant proteins have been produced and licensed for the market [20]. The hepatitis B surface antigen, for instance, is a well-known protein-based vaccine used against hepatitis B and has already been commercialized, but no peptide-based vaccine has reached the market yet [21].

Moreover, there are three types of nucleic acid biopharmaceuticals: DNA vaccines, small interfering RNA (siRNA)-based formulations, and oligonucleotide-based formulations (also called aptamers). These classes comprise fewer biopharmaceuticals, although their production is gradually and increasingly used to prevent and treat several human and animal diseases [10,22].

However, DNA vaccines generally have decreased immunogenicity compared with other vaccine types, and there are still some ethical concerns about their effects and mechanism of action [23]. Nonetheless, aptamers are oligonucleotides composed of DNA or RNA and have binding affinity to proteins, which enables them to modulate protein activity through, e.g., pathogenic protein inhibition. This makes them attractive and flexible molecules for a wide range of therapeutic goals [24]. Pegaptanib sodium (Macugen) is a relatively recent RNA-based aptamer and the first licensed for clinical use in humans to treat neovascular age-related macular degeneration, with considerable and proven success [25].

Considering all of the biopharmaceutical classes and subclasses presented to date, they can be divided into numerous types according to their structure modifications. This increases their therapeutic applicability, coverage, safety, efficacy, and/or accuracy in mechanism of action. Hence, some incremental variations in the active site must be considered to produce higher-quality and advantageous pharmaceuticals, which are called second- and third-generation biopharmaceuticals [28].

In general, these variations result in immunogenicity reduction or elimination, pharmacokinetic profile changes, increases in half-life time, or even the generation of new proteins (hybrid proteins), such as fusion proteins, composed of different polypeptide chains fused to each other [28]. Etanercept is one example. It is a fusion protein composed of a cytokine receptor that contains an Fc antibody region fused to its C-terminus end [29].

In addition, many pharmaceutical proteins, such as mAbs, FSH, EPO, and G-CSF, are glycoproteins [30]. Their glycosylation pattern, that is, the set of sugar residues attached along the polypeptide chain at glycosylation sites, may positively or negatively influence the biological activity and immunogenicity, which also depend on the organism from which the molecule is obtained [31]. In most cases, variations concerning glycosylation or punctual amino acids lead to active ingredients that maintain the same function but exhibit a novel nomenclature and activity level. In this way, there are different forms of the glucocerebrosidase enzyme (i.e., imiglucerase, taliglucerase- $\alpha$ , and velaglucerase- $\alpha$ ) owing to these differences in chains and sources [32].

New technologies, including PEGylation [33], and nanostructuration [34], however, have been used to increase the stability and consequently the half-life of medicines, or to create an address through nanostructured systems-based delivery, respectively. Another approach is to attach protein (mAb-like) scaffolds on pharmaceutical structures to increase the specificity and improve biological or physicochemical properties [35,36].

Therefore, the development of various biopharmaceuticals has also improved the safety of biotechnological products and increased their application in preventing, controlling, or curing several diseases such as cancer, rheumatoid arthritis, Alzheimer's disease, heart attacks, diabetes, dermatitis, Crohn's disease, multiple sclerosis, cystic fibrosis, hemophilia, hepatitis, heart failure, thrombosis, leprosy, and lupus. For multiple sclerosis, for example, recombinant  $\beta$ -interferon has improved the quality of life of many patients, reducing the number of relapses in 30% of individuals [37]. For rheumatoid arthritis, the new biotherapies reversed joint degenerations only 48 h after the beginning
of treatment [37]. Moreover, new biopharmaceuticals can selectively attack cancer cells and have shown less toxic effects than have chemotherapy procedures [37].

Biopharmaceuticals are only some of the pharmaceutical industry products. Chemical and natural medicines differ from biopharmaceuticals owing to their low molecular weight and structure because they are simple molecules with well-defined physicochemical properties that are administered orally and are quickly absorbed by the blood through capillaries. Unlike them, biopharmaceuticals have complex molecular structures and high molecular weight (200- to 1000-fold higher than traditional medicines). Because of their size and sensitivity, most biopharmaceuticals require a parenteral administration route and are primarily absorbed by the lymphatic system [38,39].

Originally, biopharmaceuticals were obtained by purifying molecule-containing tissue extracts and were named active components of a biological source [37,40]. After the advent of modern biotechnology, genetic engineering techniques allowed the production of biopharmaceuticals from heterologous expression of genes by using genetically modified organisms and microorganisms, and became active components of a biotechnological source [37,41].

## 2.3 Biosimilars

After 20 years, a generation of patents expired, which led to the production of a second generation of biopharmaceuticals, the so-called biosimilars [42]. Although the definition of biosimilars differs among countries, depending on the regulations applied by national agencies, in general terms, biosimilars correspond to drugs that are identical to the original biopharmaceutical products [43]. Biosimilars are developed using production processes which show differences in cell lines or production hosts regarding the reference biopharmaceutical process, to produce higher concentrations of the final product or a molecule with increased biological activity [44,45]. However, it is difficult to demonstrate that the biosimilar product is truly similar to the biopharmaceutical of reference (including structural differences and manufacturing variations). In this way, the current demand proposes a specific evaluation and approach to safety development for each biosimilar produced compared with the biopharmaceutical of reference [42].

In 2012, 26 biosimilars were in advanced production stages, following current guidelines (Table 2.1). Furthermore, 47 other biosimilars were in clinical study phases in 2015 (Table 2.2).

Over the past several years, pharmaceutical companies have demonstrated increased interest in commercializing biosimilar products [47]. Therefore, in the near future, it is expected that each developed biosimilar will be evaluated individually to better confirm its quality, safety, and efficacy compared with the biopharmaceutical of reference.

Molecule (Brand Name)	Developing Company	Current Status
G-CSF (Neutroval)	Teva Pharmaceuticals	BLA filed
G-CSF	Mochida/Fuji Pharma	BLA filed
Somatropin	JCR Pharmaceuticals	BLA filed
Trastuzumab	Celltrion	BLA filed
Infliximab	Celltrion	BLA filed
Follitropin-alfa (Gonadopin)	Dong-A Pharmaceuticals	Market <sup>a</sup>
Long-acting G-CSF (Neugranin)	Teva Pharmaceuticals	Phase III
Follitropin-alfa	Teva Pharmaceuticals	Phase III
Follitropin-alfa	Finox Biotech	Phase III
Etanercept	Celltrion	Phase III
Etanercept	LG Life Sciences	Phase III
Etanercept	Hanwha Chemical; Merck	Phase III
Darbepoietin-alfa (Avent)	Avestagen	Phase III
Erythropoietin	Hospira	Phase III
Interferon-beta	Reliance Life Sciences	Phase III
Interferon-alfa-2b	Amoytop Biotech	Phase III
G-CSF	Dong-A Pharmaceuticals	Phase II
Rituximab	Teva Pharmaceuticals	Phase II
Etanercept	Cephalon	Phase II
Etanercept (TuNex)	Mycenax	Phase II
Trastuzumab	GreenCross	Phase I
G-CSF	Hospira	Phase I
Pegylated G-CSF	Hospira	Phase I
RecombinanthCG	Dong-A Pharmaceuticals	Phase I
Interferon-beta	Dong-A Pharmaceuticals	Phase I (Brazil)
Folitropin-alfa	JCR Pharmaceuticals	Phase I

 Table 2.1
 Biosimilar Products and Stages of Development. Adapted from [45]

*BLA*, Biologic license application (biological products approved for marketing under the provisions of the Public Health Service (PHS) Act); *G-CSF*, granulocyte colony-stimulating factor.

<sup>a</sup>Phase III for ovulation induction.

S. Ariyanchira, Biosimilar market posts steady gains. BioMarket Trends 32 (12) (2012). Avaiable: http://www.genengnews.com/gen-articles/biosimilar-market-posts-steady-gains/4137/ (accessed 12.04.15).

## 2.4 Heterologous Systems for Protein Expression

To produce biopharmaceuticals and biosimilars via biotechnological routes, various heterologous systems can be used that are based on several microorganisms and macroorganisms, each of which has intrinsic traits in its protein expression and processing machinery (Table 2.3) [48–50]. Thus, selecting an organism as an expression system is a crucial and definitive step for the production and activity of recombinant biopharmaceuticals, once the system predicts the molecule's biological properties.

Hence, bacterial systems and mammalian cells are the most commonly used vehicles for expressing recombinant biopharmaceuticals. Bacteria are selected preferentially for simpler molecules, which do not require complex structure changes to keep their

Market	Number of Approved Biosimilars
Australia	8
Canada	3
Europe	20
India <sup>a</sup>	3
Japan	6
Latin America	4
South Korea	2
United States <sup>b</sup>	1

Table 2.2Biosimilars Approved for Commercializationin 2015. Adapted from [46]

<sup>a</sup>Does not include biosimilar approved prior to implementation of regulatory guidelines.

<sup>b</sup>Omnitrope (somatropin/HGH) was launched by Sandoz in 2007 under special ruling.

## **Table 2.3**Comparison of Recombinant Biopharmaceuticals and BiosimilarsExpression Systems, Based on Different Parameters [49–51]

System Parameters	Bacteria	Fungi	Mammalian Cells	Transgenic Animals	Transgenic Plants
Global cost	Low	Medium	High	High	Very low
Production time	Short	Medium	Long	Very long	Long
Scale-up capacity	High	High	Very low	Low	Very high
Product quality	Low	Medium	Very high	Very high	High
Glycosylation	Absent	Incorrect	Correct	Correct	Correct
Risk of product contamination	High (toxins)	Low	High (pathogens)	High (pathogens)	Low
Production scale	Limited	Limited	Limited	Limited	Unlimited
Production cost	Medium	Medium	High	High	Low
Storage cost	Medium	Medium	High	High	Low
Scale-up cost	High	High	High	High	Low
Insert size	—	_	Limited	Limited	Somewhat limited
Multimeric protein assembly	No	Partial	Yes	Yes	Yes
Spread	Easy	Easy	Difficult	Possible	Easy
Molecule homogeneity	Low	Medium	Medium	High	High
Protein folding	Low	Medium	High	High	High
Molecule yield	Medium	High	Medium/High	High	High
Safety	Low	Medium	Medium	High	High

J.K. Ma, P.N.W. Drake, P. Christou, The production of recombinant biopharmaceutical proteins in plants, Nature Reviews Genetics 4 (2003) 794–805; D.A. Goldstein, J.A. Thomas, Biopharmaceuticals derived from genetically modified plants, The Quarterly Journal of Medicine 97 (2004) 705–716; A.A.B. Viana, P.B. Pelegrini, M.F. Grossi-de-Sá, Plant biofarming: novel insights for peptide expression in heterologous systems, Biopolymers 98 (2012) 416–427.

biological activity. In other cases, mammalian cells are chosen generally for more complex molecules with longer polypeptide chains that require accurate post-translational modifications. Genes can be expressed in different biological systems; defining which system is advantageous for expressing the recombinant molecule form is a first step. In this respect, the most suitable expression system is the one that enables the production of a biologically active and safe therapeutic biopharmaceutical at the lowest cost [14].

Different microorganisms can be used to produce biopharmaceuticals at high expression levels. Among bacteria, *E. coli* is the most notorious and is used to produce a wide set of molecule types [51]. Filamentous fungi (e.g., *Aspergillus* spp., *Fusarium* spp., *Trichoderma* spp.) [52] and yeasts (e.g., *Saccharomyces cerevisiae*, *Pichia pastoris*, *Hansenula polymorpha*) [53] are also common choices of fungal systems for the expression of therapeutic proteins. In addition, protozoans such as *Leishmania tarentolae* [54] are becoming useful heterologous expression systems, and insect cells (*Spodoptera frugiperda* and *Trichoplusia ni*) [55] are another choice when performing baculovirus-mediated protein expression. Among mammalian cells, Chinese hamster ovary cells, human embryonic kidney, NS0, and SP2/0-Ag14 cell lines are the most commonly employed, especially for the expression of mAbs [56–58].

Furthermore, transgenic animals and plants are the most widely adapted expression systems for the large-scale production of recombinant molecules, once the available biomass quantity is sufficient to ensure high amounts of the molecule of interest [19]. In animals, many heterologous proteins are expressed and secreted into the fluids or organic systems of fishes, birds, rabbits, pigs, cows, goats, and sheep. Blood, milk, urine, semen, eggs, and embryos can be the final destination from which molecules are purified [59].

However, among all of the expression systems, transgenic plants seem to be advantageous in almost all parameters related to satisfactory yield and functionality [60,61]. Therefore, plants tend to become the most promising and economically feasible alternatives of biofactories in the large-scale production of biopharmaceuticals, considering their numerous advantages [62–64]. However, some systems, including algae (*Chlamydomonas reinhardtii*) [65,66] and mosses (*Physcomitrella patens*) [67,68], are new alternatives and have been used to express growth factors and mAbs.

## 2.5 Financial Market

During the 1970s, most pharmaceutical industries were located in the United States. However, lately, the participation of European and Asiatic pharmaceutical companies has increased in the market because multinationals have opened factories in different countries and because of increasing interest in patenting pharmaceutical processes and molecules in several places at once [69]. Currently, the cost required to produce pharmaceuticals is around US \$1 billion, because it takes up to 10 years from the development of the process to produce a biomolecule until the release of the drug to the market [70].

Moreover, the current pharmaceutical global market is US \$300 billion per year, and this amount might reach US \$400 billion within the following 3 years. Ten major pharmaceutical companies control one-third of this market, with sales exceeding US \$10 billion per year and with a 30% profit margin. Six of the 10 richest pharmaceutical industries in the world are located in the United States and four are in European countries. In this way, Japan and European and American countries will be responsible for 85% of the world pharmaceutical market in the 21st century [69].

In the pharmaceutical industry in Europe, the largest companies are located in Germany, France, Italy, Spain, and the United Kingdom. However, in 2012, the leading manufacturers were located in Switzerland, Germany, Italy, the United Kingdom, France, and Ireland; together, they had a value of  $\in 30$  million, employing more than 700,000 people [71]. In 2014 European pharmaceutical companies invested  $\in 30,5$  million in research and development, employing 707,000 people, which generates three to four times more employments indirectly, thus demonstrating the importance and interest in investments in the sector worldwide [72].

Considering other countries, the biopharmaceutical sectors in China, India, and Israel are increasing considerably, because investments in internal strategies for developing their own products are enhanced annually and they are promoting advances at a national level. Similarly, Brazil is developing its own strategies to overcome expenditures of biopharmaceuticals by the Ministry of Health, because the country still depends on imports of these products owing to the lack of national production [38].

In 2008, the 30 most commercialized biopharmaceuticals yielded US \$42 billion to the companies, with the expectation of a 56.5% increase by 2014 [73]. mAbs occupy the first positions (39% of total sales) as biopharmaceuticals most in demand, followed by insulin and its analogues, EPO and IFNs, respectively [17,38]. Nevertheless, the high expenditures required for the development process to produce biopharmaceuticals and for preclinical and clinical testing are not the only reason why these drugs are commercialized at such an expensive value. Patenting biopharmaceuticals, which enables companies to have exclusivity and licensure for the commercialization of certain drugs, is a main cause for high prices [74]. In 2014, 81.4% of all patent applications deposited were from pharmaceutical companies, followed by universities (10.1%) and biotechnological industries (8.5%). Among the companies, Merck occupied the first position in patent submission, followed by Eli Lilly, Pfizer, Roche, and Upjohn [8] (Table 2.4).

Hence, the marketing of biopharmaceuticals was enhanced from only 10% of the total pharmaceutical market in 2002, increasing to 18% in 2010. For 2016, the share of biopharmaceuticals is expected to reach 21% of the total market [75]. Companies such as Amgen, Genentech, Genzyme, Johnson & Johnson, Novo Nordisk, Serono, Novartis, Sepracor, Chiron, Celgene, Bayer, Merck & Co., Abbott, Pfizer, Eli Lilly & Co., Roche, MedImmune, Gilead Sciences, Shire Pharmaceuticals, Cephalon, and Biogen have become established as the main biopharmaceutical companies worldwide and are the

Company	Patents
Merck	65 (4.7%)
Eli Lilly	41 (3.3%)
Pfizer	39 (2.8%)
Roche	38 (2.8%)
Upjohn	38 (2.8%)

**Table 2.4**Biopharmaceutical Companies andNumber of Patents Deposited. Adapted from [76]

Five companies with major patents in 1374 examined (M.S. Kinch, J. Raffo, Economic Research Working Paper N°.24. Economics and Statistics Series, WIPO (Word Intelectual Property Organization), 2015 pp. 1–14.

reason for the exponential growth of biopharmaceuticals in the market (Table 2.5) [39,77]. In Brazil, Oswaldo Cruz Foundation (Fiocruz) set up a laboratory to produce biopharmaceuticals as a strategy to overcome the expenditures incurred by importing biological medicines.

Drugs whose global revenues exceed billions of dollars per year are denominated as blockbusters, such as Remicade (US \$7 billion), Avastin, Enbrel, and Humira (US \$6 billion), which were developed during the 1990s [78]. Production of these biopharmaceuticals started in the 1990s. Thus, the patents for some began to expire in 2011, and others will expire by 2015 once the validation for each patent is reached, that is 20 years [79]. In 2011, biological products yielded US \$142 billion, which is equivalent to 19% of the total biopharmaceutical market [80] and is expected to reach 20% by 2016 [81].

Because several biopharmaceuticals were initially produced in the 1990s, many patents are now expiring [79,82]. This situation opens opportunities for the development of new processes for the production of biosimilars. Compared with the drug of reference, biosimilars have similar quality, safety, and efficacy [74].

Hence, over the past several years, both biopharmaceutical companies and chemical pharmaceutical industries have become interested in producing biosimilars. Whereas biopharmaceutical companies are investing in biosimilars to diversify their portfolio and create new revenue streams [83], chemical pharmaceutical industries aim for growth in manufacturing synthetic drugs at the lowest possible cost through economies of scale owing to patent litigation skills [84]. Thus, the biosimilars market offers chemical and pharmaceutical industries an opportunity to increase their manufacturing support value and to diversify their portfolios.

The first biosimilar that was developed and licensed for therapeutic use (Omnitrope), a human growth hormone medicine produced by Sandoz, reached the market in 2006 and was approved for commercialization in the European Union (EU) to treat growing disorders such as Turner's syndrome [85,86]. Since then, various biosimilars have been developed and licensed and were expected to represent approximately 2% of total biological medicine sales by the end of 2015 [38]. During this year, there were already 246 biosimilars in the EU and the United States alone [86].

				An (\$U	inual Sal IS Millioi	es ns)
Product	Active Ingredient	Company	Therapeutic Indication	2008	2009	2010
Remicade	Infliximab	Merck & Co.	Autoimmune diseases	5856	6631	7324
Enbrel	Etanercept	Pfizer	Autoimmune diseases	6191	5916	6808
Humira	Adalimumab	Abbott	Autoimmune diseases	4500	5500	6500
Avastin	Bevacizumab	Roche	Cancer (several types)	2908	5837	6061
Rituxan	Rituximab	Roche	Non-Hodgkin's lymphoma,	2852	5710	5962
			leukemia, rheumatoid arthritis			
Herceptin	Trastuzumab	Roche	Breast cancer; gastric cancer	1819	4940	5093
Lantus	Insulin glargine	Sanofi-Aventis	Diabetes	3259	4096	4668
Epogen/Eprex	Epoetin-a	Amgen/J & J	Anemia	4916	4814	4458
Novolog	Insulin aspart	Novo Nordisk	Diabetes	2503	3020	3666
Neulasta	Pegfilgrastim	Amgen	Febrile neutropenia	3318	3355	3558
Aranesp	Darbepoetin-α	Amgen	Anemia	3137	2652	2486
Novolin	Insulin	Novo Nordisk	Diabetes	2194	2103	2198
Humalog	Insulin Lispro	Eli Lilly & Co.	Diabetes	1736	1959	2054
Pegasys	Peginterferon-alfa-2a	Roche	Hepatitis C	1534	1553	1543
Novo Seven	Factor VIIa	Novo Nordisk	Hemophilia A and B	1189	1314	1493
Lucentis	Ranibizumab	Roche	Macular edema and	887	1124	1368
			degeneration			
Sandostatin	Octreotide acetate	Novartis	Several tumors	1100	1155	1291
Neupogen	Filgrastim	Amgen	Febrile neutropenia	1341	1288	1286
Neo	Epoetin-a	Roche	Anemia	1664	1463	1205
Recormon						
Humulin	Insulin	Eli Lilly & Co.	Diabetes	1063	1022	1089
Synagis	Palivizumab	Astra Zeneca	Prevention against RSV	1230	1082	1038
Gardasil	Quadrivalent HPV vaccine	Merck & Co.	Prevention against HPV	1403	1118	988
Norditropin	Somatotropin	Novo Nordisk	Growth failure	718	818	893
Genotropin	Somatotropin	Pfizer	Growth failure	989	887	885
Forteo	Teriparatide	Eli Lilly & Co.	Osteoporosis	779	816	830
	·	<b>,</b>	Total annual sales	59,086	70,174	74,746

Table 2.5	Top 25 Biotechnology Medicines Based on Reported Worldwide Sales
Adapted f	om [87]

HPV, human papillomavirus; J & J, Johnson & Johnson; RSV, respiratory syncytial virus.

Adapted from R.J.Y. Ho, Biotechnology and Biopharmaceuticals: Transforming Proteins and Genes into Drugs, second ed., Wiley Blackwell, New Jersey, 2013.

Biosimilars are a promising market for investment [88] because their final cost and prices on the market can be 50–80% less than those of the original biopharmaceuticals [89,90]. Thus, more than 250 companies are currently involved in the development and/or commercialization of biopharmaceuticals and biosimilars worldwide [86,91]. Estimates from the market indicate that biosimilar inclusion will become 4% of total biopharmaceutical sales over the next decade, amounting to a savings of US \$44.2 billion. This global market is estimated to reach US \$2 billion in 2018 and to have

a growth rate of more than 20% between 2013 and 2018. However, this market depends on decisions from the FDA [43,90].

In emerging countries such as Brazil, Russia, India, China, and South Korea, biopharmaceutical companies receive incentives and have advantages in producing biosimilar products, including a large internal demand for biopharmaceuticals, lower labor costs, and government support [91]. With the increase in small local biopharmaceutical companies, big pharma industries are facing a challenge to continue selling their products at higher prices while searching for and developing new products.

In fact, big pharma will confront the challenge of losing market exclusiveness once some companies develop biosimilar products from other original industries while searching for new biopharmaceuticals in their development pipeline [43].

However, there are commercial disputes among pharmaceutical companies that may cause an increase in the final price. Currently, the main concern in the US market is whether companies producing biosimilars conflict with any previously granted patents that belong to biopharmaceutical-producing companies whose main patents have already expired [92].

Hence, to control this process, the Innovation Act and Price Competition of Biosimilars (BPCIA) was created in 2010 (part of the Affordable Care Act and the Patient Protection). BPCIA's purposes are to help companies reach a consensus concerning patent disputes and to share important information. According to the BPCIA essential standards, biosimilar producers must deliver, within 20 days, documents to the FDA and to the company that developed the reference drug such that the sale of the new drug containing the biosimilar molecule can be legalized. Nevertheless, biosimilar producers usually deliver the documents merely to the FDA, claiming that sharing information with other companies will occur only under certain conditions, which is often not accepted by the company that developed the reference drug [92]. Consequently, the judicial processes from those disputes may return more expensive drugs to the market and retard patent submissions [92].

## 2.6 Legislation History and World Regulations

#### 2.6.1 History

The first legislations and regulations regarding the pharmaceutical industry were released in the United States at the end of the 19th century. During the Mexican–American war (1846–48), the US government issued the first law related to pharmaceuticals, called the "Import Drugs Act of 1848," to control the quality and purity of imported drugs used to treat several infectious diseases. Fifty years later, the first law regarding biological products was created. The Biologics Control Act of 1902 demanded previous licensing for the production and distribution of any biological product (vaccine, sera, toxins, and viruses) and established rules about product labeling, license number, and expiration dates [93].

The Food and Drugs Act of 1902 was a landmark in the creation of the current FDA, which occurred later in 1906. The Act stipulated the mandatory labeling of ingredients to all drugs and food that was commercialized in the United States. The law also controlled and prohibited the transportation and adulteration of food and drugs in the country and, for the first time, had an impact on the overall national system of drug and food safety [93].

As the pharmaceutical industry of the United States continued to grow, new rules were established. Hence in 1938, after the discovery of the deadly effects of sulfanilamide in several patients, the government released the Food, Drug, and Cosmetic Act, in which every drug had to be proven safe before being authorized for commercialization in the country. In 1962, even after being approved as safe by the FDA, thalidomide caused deformities in babies born to mothers who consumed the drug during pregnancy. Therefore, the Kefauver–Harris Amendments were released, in which both the safety and efficacy of a drug had to be confirmed before marketing [93].

Currently, the FDA is the main agency worldwide that is responsible for drug quality control. Along with the Current Good Manufacturing Practice (CGMP) regulations, agencies in the United States guarantee safety in the manufacturing, processing, packaging, and use of every new drug released to the market. Moreover, the federal american agency determines whether the pharmaceutical industry contains the necessary facilities, equipment, and technical team to produce the new drug that is in process for approval [93].

To standardize all of the steps to produce and commercialize new drugs, the FDA follows four federal regulations (Federal Food, Drug, and Cosmetic Act; 21 Code of Federal Regulations Part 201 and Part 211; Federal Register Notices for Proposed Changes; and Final Changes to CGMP), as well as Guidance, Manuals, and Compliances [94]. The Guidance and Guidelines are instructions for industries on how to process, manufacture, produce, test, and evaluate the drug that has been approved or is in the approval phase [94]. Nevertheless, industries should follow the Guideline on the Preparation of Investigational New Drug Products (issued November 1992, posted March 2, 1998), the Guidance for Industry: Investigating Out-of-Specification Test Results for Pharmaceutical Production (October 2006), and the Standing, Operating Procedures for New Drug Application/Abbreviated New Drug Application Field Alert Reports (issued October 30, 1998; posted November 2, 1998).

In this way, the Hatch–Waxman Act regularizes rules for the chemical pharmaceutical market (add in Section 505) of the Federal Food, Drug and Cosmetics Act). The Act stimulates independent pharmaceutical companies to create novel synthetic drugs at affordable and accessible prices for the population. Thus the Hatch–Waxman Act was a success because it maintained the incentive for drug discovery and development. However, this act does not apply to biomolecules [94].

Therefore, other regulations, such as the Life-Saving Medicine Act, were released to determine whether the infrastructure of a pharmaceutical company is appropriate for the production of biosimilar drugs [94,95]. Hence, the effort resulted in the creation

of the Biologics Price Competition and Innovation Act, which allowed the FDA to authorize the review and approval of biosimilars to the market by evaluating biosimilarity or interchangeability with the reference product in terms of animal studies and clinical trials in addition to an analytical evaluation [41].

Furthermore, the FDA uses the Biological License Application to evaluate and approve new biological products. In this process, industries must submit a report demonstrating all research and results developed during preclinical and clinical studies, biological activity evaluation of the drug, and physicochemical and manufacturing information [96]. The purpose is to prove that the product is "safe, pure, and potent." After approval, the pharmaceutical industry must follow CGMP regulations to standardize the facilities, production process, packaging, quality control procedures, stability and activity tests, expiration date analyses, and production records, among others (CFR Parts 210, 211, and 600) [97].

#### 2.6.2 World Regulations

Each country's administration department that handles these manners regulates all changes in importing and exporting biopharmaceuticals. These organizations and administrations are important political players in the pharmaceutical industry. Politics has a role in regulating the manufacture, distribution, and sale of pharmaceuticals for the safety of the consumer. Governments want to ensure that the medicines are safe and effective for all users. Each nation has its own system for ensuring the safety and distribution of drugs.

Hence, over the past decade, biopharmaceuticals have been responsible for an expansive increment in the biotechnology and pharmaceutical industry. Despite the high costs involved in research and in biopharmaceutical development and production, regulations were created to balance intellectual property and companies' investments to patients who depend on free access to medical drugs [94].

In this way, the European Medicines Agency (EMA) has established and documented a robust legal regulatory guideline for the approval of biosimilars. Since the legalization of the first guideline in 2005, 14 biosimilars have been approved for commercialization in European countries (Table 2.6). Nevertheless, not all biosimilars reach the market even after a long period of investment and research. For example, the drug Alpheon (BioPartners GmbH), which consists of the biosimilar interferon-alfa-2a, had purity incompatibilities and unexpected side effects compared with the drug of reference, and therefore was rejected by the EMA for release to the market [43,94].

The technological changes that occurred as a result of troublesome experiences with previous cases led to the release in 2013 of the Committee of Medical Products for Human Use. The EMA has a robust regulatory process and guidelines are frequently revised as a result of experiences with the approval of biosimilars over time [43,98].

Product	Common Name (International Nonproprietary Name)	Company	Reference Product	Year of Approval
Omnitrope	Somatropin	Sandoz International Limited	Genotropin	2006
Valtropin	Somatropin	Biopartners GmbH	Humatrope	2006
Binocrit	Epoetin-alfa	Sandoz International Limited	Eprex	2007
Epoetin Alfa Hexal	Epoetin-alfa	HEXAL AG	Eprex	2007
Abseamed	Epoetin-alfa	Medicine Arzneimittel Putter GmbH Co. KG	Eprex	2007
Retacrit	Epoetin-zeta	Hospira, Inc.	Eprex	2007
Silapo	Epoetin-zeta	SAASA Arneimittel AG	Eprex	2007
Biograstim	Filgrastim	CT Arzneimittel GmbH	Neupogen	2008
Filgrastim Ratiopham <sup>a</sup>	Filgrastim	Ratiopham GmbH	Neupogen	2008
Ratiograstim	Filgrastim	Ratiopham GmbH	Neupogen	2008
Tevagrastim	Filgrastim	Ratiopham GmbH	Neupogen	2008
Zarzio	Filgrastim	Sandoz International Limited	Neupogen	2009
FilgrastimHexal	Filgrastim	HEXAL AG	Neupogen	2009
Nivestim	Filgrastim	Hospira, Inc.	Neupogen	2010

Table 2.6 Approved Biosimilars in Europe up to 2010. Adapted from [94]

<sup>a</sup>Note that Filgrastim Ratiopharm was withdrawn on April 20, 2011, at the request of the sponsor. Up to December 2010.

In 2014, in the United States, the FDA drafted Guidance to support biopharmaceuticals and biosimilar industries and created a list of biologicals called the "Purple Book." Two biosimilar applications were filed in this book during 2014: filgrastim and infliximab [43].

Other countries such as Japan, Switzerland, Canada, and Australia, as well as the World Health Organization (WHO) adopted the same scientific principles from the EU for biosimilar approval. Although some countries are based on the same regulation guides, some items are distinct (Table 2.7). In addition, in 2010 the workshop entitled "WHO Guidelines on Evaluating Similar Biotherapeutic Products" took place to facilitate the "global harmonization" of biosimilar evaluation [94]. In this event, the discrepancies in regulations between the WHO and those of countries such as Canada, Korea, Japan, and the EU were highlighted and discussed.

Because regulations and guidelines for biosimilars are still under optimization, several scientific factors remain unresolved: (1) how similar molecules are considered a biosimilar; (2) the degree of similarity (individual or collective) between molecules; (3) statistical models showing total biosimilarity between molecules; (4) the criteria used to assess biosimilarity (average against variability; one criterion or flexible criteria); (5) interchangeability; and (6) standardization of the design of studies (potential uses for adaptive design in biosimilar studies). Therefore, with some factors unresolved, the development of new regulations and guidelines to "global harmonization" is necessary [94].

	World Health Organization	Canada	Korea	European Union	Japan	
Torm	Similar Biothorapoutic	Subsequent Entry	Piocimilars	Piocimilars	Follow on	
Tellin			DIOSITTIIIdTS	DIOSITIIIdTS	FUILOW-UIT	
Scope	Recomb	biologics SLBS		Mainly recombinant	Becombinant	
Scope	Neconic	finant protein drugs		nrotein drugs	nroteins drugs	
Efficacy	Double-blind or c	bserved-blind.	Fauivalence	Comparability mar	nins should be	
Entercy	equivalence or non	inferiority design	design	prespecific an	d iustified	
Reference product	Authorized in juri	isdiction with well-es	tablished	Authorized in	Authorized in	
nererence produce	requi	latory framework	abiblica	Furopean Union	lapan	
Stability		Accelerated degrad	ation studies	Laropean emeri	Not necessary	
	St	udies under various	stress condition	S	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
Purity of		Process- and p	product-related	impurities		
manufacture						
Physicochemical		Primary and	higher-order s	tructure		
		Posttranslational modifications				
Biological activity		Qualitative	e measure of fu	inction		
	Quar	Quantitative measure (e.g., enzyme assays or binding assays)				
Nonclinical studies	In vitro (e.g., receptor-binding, cell-based assays)					
	In vitro (pharmacodynamic PD activity, at least one repeat dose toxicity study, antibody					
	measurements, local tolerance)					
PK study design	Single dose, steady-state studies, or repeated determinations of pharmacokinetics PK					
and criteria		Cros	sover or paralle	4		
	Include absorption and elimination characteristics					
		Traditional 80–125% equivalence range is used				
PD	PD markers should be selected and comparative PK/PD studies may be appropriate					
Safety principles	Prelicensing safety data and risk management plan					
	Generic approach is not appropriate for Follow-on Biologic FOB					
	Follow-on biologics should be similar to the reference in terms of quality, safety, and efficacy				ty, and efficacy	
	Stepwise comparability approach: Similarity of the SBP to KBP Reference biotherapeutic produc			rapeutic product		
	in terms of quality is prerequisite for reduction of nonclinical and clinical data required for				a required for	
	C	asa bu casa approach	approvai	classes of products		
	Case-by-case approach for unreferit classes of products					

# **Table 2.7**Comparison of Biopharmaceutical Regulations Among DifferentCountries. Adapted from [94]

SBP or SBPs - similar biotherapeutic products. SEB - subsequent-entered biologics. PD - pharmacodynamics tests. PK - pharmacokinetics tests. FOB - follow-on biologics. RBP - Reference biotherapeutic product. S.C. Chow, Biosimilars: Design and Analysis Follow-on Biologics, first ed., CRC Press, Florida, 2014.

Currently, the guidelines of regulatory agencies from the EU, United States, and Japan, among others, regulate general principles of biosimilars, including fabrication and preclinical and clinical tests. Moreover, comparability tests to reference products in all development phases are made to ensure the quality, efficacy, and safety of biosimilars [44,49]. Furthermore, to avoid serious adverse effects, pharmaco-vigilance and immunogenicity tests are required [44,49].

#### 2.6.3 Regulations in Brazil

The biosimilar market is growing in emerging countries; in Brazil, the costs of importing biopharmaceuticals as estimated by the Ministry of Health represent 50% of all annual health care expenditures. Considering that the Brazilian government is responsible for the health care of 205 million people, the amount spent on biopharmaceuticals reaches billions of dollars every year [43,91].

The pharmaceutical industry in Brazil has been developed with the goal of the national independence of drug production, but local research and development have had less than 10 years of application. Nevertheless, the first policies created in the country date to 1973, when Law 5991 initiated a new step in drug safety regulation. Three years later, Law 6360 demanded new procedures for the registration of any pharmaceutical product commercialized in Brazil. At that time, there was nothing in place concerning the bioequivalence or interchange of drugs [99,100].

The new Constitution of Brazil, established in 1998, included Laws 8080/90 and 8142/90, which created the Unique Health System (Sistema Único de Saúde - SUS), whose purpose was to "provide health to all." With these laws, access to health treatments and several medicines could be obtained free or for a charge by the population. However, the pharmaceutical industry still depended on international companies and the import of medicines, and thus Ministry of Health expenses increasing yearly [101,102].

In 1993, the Brazilian government set up the Program for Specialized Medicines. In this context, drugs commercialized in the country were categorized into three groups based on their price and application. Moreover, it was possible to estimate the amount of a specific drug needed for Brazilian patients and to predict the value to be spent with pharmaceutical products by the government. Five years later, the National Drug Policy showed that more than 50% of the population actually had access to only 20% of all commercially available drugs [103].

Therefore, to decrease the high government expenditures with the acquisition of pharmaceutical products, enhance the access of drugs to the population, and increase the local production of medicines, the Law of Generics was released in 1999. With this law, the prices of drugs decreased, a wider variety of medicines went on the national market, and rules on interchangeability with new medicines and bioequivalence started to be part of Brazil's health priority [104].

Hence, the need for novel resolutions and laws as a result of the fast development of the pharmaceutical sector in Brazil led to the release of Collegiate Director Resolution (RDCs) 133/134 and 16/17 by the National Agency for Sanitary Vigilance (ANVISA), during 2003 and 2007, respectively. These resolutions included the requirements for pharmaceutical equivalence and pharmacokinetic studies, as well as for the registration of generic drugs. However, although the registration of generic products has increased approximately 3600% since then, the registration of biopharmaceuticals has increased four times less. Furthermore, the dependence of Brazil on importing specialized

medicines and biological products reached almost US \$3 billion dollars in 2012 alone, corresponding to 30% of all expenditures by the Ministry of Health. Although biological products corresponded to less than 10% of all drugs commercialized in the country, they represented 50% of the cost of all medicines in Brazil [41,105–108].

Thus, in 2004, the government introduced the Law for Technological Innovation (Law 10,973) and in 2007, the Politics for National Biotechnology. The purposes were to incentivize the development of innovative products and processes and stimulate the improvement of Brazilian pharmaceutical companies. By 2005, the ProPharma Programs, with the financial support of BNDES (National Development Bank), invested 44% of the amount into the increase of national companies, 40% into production, and 16% into research and development [41,109].

Therefore, considering that the commercial balance showed a deficit of more than 9 billion dollars on the national production, export, and import of biopharmaceuticals in Brazil, public institutions started looking for partnerships with international private companies to speed up the release of national biopharmaceutical products into the market. Until 2011, 64 partnerships were settled among 15 national laboratories and 40 private companies, working on producing drugs. In 2014, there were 14 new pharmaceutical companies in partnership with six Brazilian public laboratories investing in the production of 14 different biosimilar molecules [110].

In Brazil, first-generation biotechnological products (e.g., insulin, growth hormones) were under constant development and registration by national companies, especially Bio-Manguinhos. However, second-generation products are the new future for pharmaceutical industries worldwide. The first patents are expiring and others will expire in approximately 2020. This reflects a global race for the development of novel technologies to produce biosimilars at the same time that specific regulations are being released for these products, including in Brazil [111]. Moreover, Brazilian private companies such as the FarmaBrasil Group are investing in technology and training and prioritizing the development of biosimilar products, although the focus on producing new molecules has not been discarded [111].

Along with the politics involved in the improvement of the biopharmaceutical industry and following the demand for updated regulations, ANVISA released new RDCs (49, 50, 55) and Ordinances (506/2012 and 837/2012) to establish registration and postregistration requirements for biological products. It also requires stability studies for the registration and postregistration of biological products. Guidelines and criteria for development of a productive development partnerships and the construction of a Health Industrial Complex and its Managing Committee were included in the new ordinances [112–116].

The legal basis for the production and registration of biotechnological products in Brazil has been under consolidation since 2010, and its rules are in accordance with international standards. These regulations include guidelines from EMA and FDA, with modifications that fit the Brazilian market. Those guidelines are constantly being revised according to the technical complexity, economic and financial importance the regulations demmand for the registration and maintenance of biotechnological products into the market [117].

The Ministry of Health invested \$17 billion in the development of 100% national drugs [119]. There are already two biosimilar molecules on the market in Brazil because of the National Politics for Biotechnology Development: botulinum toxin and the vaccine H1N1, which benefit more than one million patients in the country. Last year, the biopharmaceutical Etanercept (Enbrel), which is used against rheumatoid arthritis and was developed by partnerships between Brazilian and international companies, obtained registration for commercialization by ANVISA. Moreover, at least nine biosimilar molecules and five vaccines produced in Brazil will be released into the market by 2017 [118].

In addition to boosting the production and development of public laboratories and national private companies, the partnerships have provided savings for the Ministry of Health of \$2 billion. Furthermore, they reduced Brazilian Social Policy problems and strengthened the country's Industrial Complex and Health Innovation, thus allowing SUS to transform its productive structure and permitting the Health System of Brazil to be compatible with new technological world standards [111]. The expectation for 2015 was that the new drugs would provide savings for the Ministry of Health of more than \$500 million, which could be spent in other areas of the country's health system.

## 2.7 Perspectives: Pharmacogenomics and Next-Generation Biopharmaceuticals

With advances in modern biotechnology, the possibility of developing biomarkers for the diagnosis and treatment of several diseases has become a reality. Moreover, improvement in technologies has allowed the use of pharmacogenomics to indicate personal treatment for a certain disease [119].

In parallel with the development and expansion of novel automated techniques, obtaining the personal genetic profile of a large number of individuals and correlating this profile with their pharmacodynamics profile has also become a reality. The results will lead to the ability to determine exactly which biopharmaceutical doses and formulations are more suitable for administration to each patient [37,120].

Along with pharmacogenomics studies, the development of next-generation biopharmaceuticals together with meaningful contributions of nanotechnology are even more promising because they will provide safer and more effective medicines against many diseases. As a result, biologics may soon become even more effective, which will also allow for the reduction of side effects, especially for serious diseases such as cancer, and will significantly increase quality and expectancy life. [34,37].

### References

- [1] C. Hanrahan, Making sense of bisimilars in the US market, Healthcare Interventional Journal 2 (2014) 1–7.
- [2] W.E. Sneader, Drug Discovery (The History), Van Nostrand's Scientific Encyclopedia, 2005. Available at: http://samples.sainsburysebooks.co.uk/9780470015520\_sample\_379861.pdf (accessed on 01.06.15).
- [3] P. Neushul, Science, government and the mass production of penicillin, The Journal of the History of Medicine and Allied Sciences 48 (1993) 371–395.
- [4] D. Walls, R. Mcgrath, S.T. Loughran, A digest of protein purification, Protein Chromatography: Methods and Protocols, Methods in Molecular Biology 681 (2011) 3–23.
- [5] J. Pless, From somatostatin to Sandostatin®: history and chemistry, Metabolism 4 (1992) 5–6.
- [6] M.W.J. Strachan, B.M. Frier, Insulin Therapy. A Pocket Guide, Springer London Heidelberg, New York Dordrecht, 2013.
- [7] S. Simoens, Biosimilar medicines and cost-effectiveness, Clinic Economics and Outcomes Research 201 (2011) 29–36.
- [8] M.S. Kinch, J. Raffo, Sources of innovation: an assessment of intellectual property, Drug Discovery Today 20 (2014) 500–504.
- [9] H. Conor, Making sense of biosimilars in the US market, Healthcare Interventions Journal 2 (2014) 2–8.
- [10] G. Walsh, Biopharmaceuticals: recent approvals and likely directions, Trends in Biotechnology 23 (2005) 553–558.
- [11] H. Schellekens, Follow-on biologics: challenges of the 'next generation', Nephrology Dialysis Transplantation 20 (2005) 31–36.
- [12] C.R. Geyer, J. McCafferty, S. Dubel, A.R.M. Bradbury, S.S. Sidhu, Recombinant antibodies and *in vitro* selection technologies, in: G. Proetzel, H. Ebersbach (Eds.), Antibody Methods and Protocols, vol. 901, Springer, New York, 2012.
- [13] G. Köhler, C. Milstein, Continuous cultures of fused cells secreting antibody of predefined specificity, Nature 256 (1975) 495–497.
- [14] A. Berlec, B. Strukelj, Current state and recent advances in biopharmaceutical production in *Escherichia coli*, yeasts and mammalian cells, Journal of Industrial Microbiology and Biotechnology 40 (2013) 257–274.
- [15] G. Walsh, Biopharmaceutical benchmarks 2010, Nature Biotechnology 28 (2010) 917–924.
- [16] A.L. Nelson, E. Dhimolea, J.M. Reichert, Development trends for human monoclonal antibody therapeutics, Nature Reviews Drug Discovery 9 (2010) 767–774.
- [17] S. Aggarwal, What's fueling the biotech engine -2012 to 2013, Nature Biotechnology 32 (2014) 32-39.
- [18] K. Swiech, V. Picanço-Castro, D.T. Convas, Human cells: new platform for recombinant therapeutic protein production, Protein Expression and Purification 84 (2012) 147–153.
- [19] J. Zhu, Mammalian cell protein expression for biopharmaceutical production, Biotechnology Advances 30 (2012) 1158–1170.
- [20] S. Rabindran, V. Yusibov, Novel expression systems for vaccine production, in: E. Wen, R. Ellis, N.S. Pujar (Eds.), Vaccine Development and Manufacturing, Wiley, New Jersey, 2014.
- [21] WHO World Health Organization, Hepatitis B vaccines, Weekly Epidemiological Record 84 (2009) 405–420.

- [22] S.D. Patil, D.G. Rhodes, D.J. Burgess, DNA-based therapeutics and DNA delivery systems: a comprehensive review, The AAPS Journal 7 (2005) 61–77.
- [23] M.A. Liu, DNA vaccines: an historical perspective and view to the future, Immunological Reviews 239 (2011) 62–84.
- [24] H. Ulrich, C.A. Trujillo, A.A. Nery, J.M. Alves, P. Majumder, R.R. Resende, A.H. Martins, DNA and RNA aptamers: from tools for basic research towards therapeutic applications, Combinatorial Chemistry & High Throughput Screening 9 (2006) 619–632.
- [25] E. NG, D.T. Shima, P. Calias, E.T. Cunningham, D.R. Guyer, A.P. Adamis, Pegaptanib, a targeted anti-VEGF aptamer for ocular vascular disease, Nature Reviews Drug Discovery 5 (2006) 123–132.
- [26] F.J. Van Hemert, H.L. Zaaijer, B. Berkhout, V.V. Lukashov, Occult hepatitis B infection: an evolutionary scenario, Virology Journal 5 (2008) 1–13.
- [27] A.M.M. Basso, P.B. Pelegrini, F. Mulinari, A.B. Viana, L.P. Silva, M.F. Grossi-de-Sa, Recombinant glucagon: a differential biological activity, AMB Express 5 (20) (2015) 1–9.
- [28] P.J. Carter, Introduction to current and future protein therapeutics: a protein engineering perspective, Experimental Cell Research 317 (2011) 1261–1269.
- [29] T. Kohno, L.T. Tam, S.R. Stevens, J.S. Louie, Binding characteristics of tumor necrosis factor receptor-Fc fusion proteins vs anti-tumor necrosis factor mAbs, Journal of Investigative Dermatology Symposium Proceedings 12 (2007) 5–8.
- [30] G. Walsh, R. Jefferis, Post-translational modifications in the context of therapeutic proteins, Nature Biotechnology 24 (2006) 1241–1252.
- [31] R.G. Werner, K. Kopp, M. Schlueter, Glycosylation of therapeutic proteins in different production systems, Acta Paediatrica 96 (2007) 17–22.
- [32] R.J. Solá, K. Griebenow, Glycosylation of therapeutic proteins: an effective strategy to optimize efficacy, BioDrugs 24 (2010) 9–21.
- [33] G. Pasut, F.M. Veronese, State of the art in PEGylation: the great versatility achieved after forty years of research, Journal of Controlled Release 161 (2012) 461–472.
- [34] S. Mitragotri, P.A. Burke, R. Langer, Overcoming the challenges in administering biopharmaceuticals: formulation and delivery strategies, Nature Reviews Drug Discovery 13 (2014) 655–672.
- [35] M. Gebauer, A. Skerra, Engineered protein scaffolds as next-generation antibody therapeutics, Current Opinion in Chemical Biology 13 (2009) 245–255.
- [36] A. Beck, T. Wurch, C. Bailly, N. Corvaia, Strategies and challenges for the next generation of therapeutic antibodies, Nature Reviews Immunology 10 (2010) 345–352.
- [37] A.G. Roche Holding, Medicamentos biológicos. Available on: http://www.roche.com.br/portal/ roche-brazil/medicamentos\_biologicos (accessed 25.11.14).
- [38] C. Reis, A. Landim, J.P. Pieroni, Lições da experiência internacional e propostas para incorporação da rota biotecnológica na indústria farmacêutica brasileira, BNDES Setorial 34 (2011) 5–44.
- [39] Amgen, Biologics and Biosimilars: An Overview, Amgen Inc., Thousand Oaks, 2012.
- [40] E.S. Ferro, Biotecnologia translacional: hemopressina e outros peptídeos intracelulares, Estudos Avançados 24 (2010) 109–1210.
- [41] A.M.M. Basso, M.F. Grossi-de-Sa, P.B. Pelegrini, Biopharmaceutical and biosimilar products in Brazil: from political to biotechnological overview, Journal of Bioequivalence and Bioavailabilibty 5 (2013) 60–66.
- [42] L. Fryklund, M. Ritzén, G. Bertilsson, M.H. Arnlind, Is the decision on the use of biosimilar growth hormone based on high quality scientific evidence? – a systematic review, European Journal of Clinical Pharmacology 70 (2014) 509–517.

- [43] L.R. Tsuruta, et al., Biosimilars advancements: moving on to the future, Biotechnology Progress (2015). http://dx.doi.org/10.1002/btpr.2066.
- [44] H. Mellstedt, D. Niederwieser, H. Ludwig, The Challenge of Biosimilars, Annals of Oncology 19 (2008) 411–419.
- [45] S. Ariyanchira, Biosimilar market posts steady gains, BioMarket Trends 32 (12) (2012). Avaiable: http://www.genengnews.com/gen-articles/biosimilar-market-posts-steady-gains/4137/ (accessed 12.04.15).
- [46] J.E. Sackman, M.J. Kuchenreuther, The Bullish Outlook for Biosimilars. Available: http://www. processdevelopmentforum.com/articles/the-bullish-outlook-for-biosimilars/ (accessed 12.04.15).
- [47] P. Wilson, C. Wood, Biosimilar Esas: a comparative review, Journal of Renal Care 41 (2015) 53-61.
- [48] M. Nagasaki, Y. Ando, Clinical development and trial design of biosimilar product: a Japanese perspective, Journal of Biopharmaceutical Statistics 24 (2014) 1165–1172.
- [49] J.K. Ma, P.N.W. Drake, P. Christou, The production of recombinant biopharmaceutical proteins in plants, Nature Reviews Genetics 4 (2003) 794–805.
- [50] D.A. Goldstein, J.A. Thomas, Biopharmaceuticals derived from genetically modified plants, The Quarterly Journal of Medicine 97 (2004) 705–716.
- [51] A.A.B. Viana, P.B. Pelegrini, M.F. Grossi-de-Sá, Plant biofarming: novel insights for peptide expression in heterologous systems, Biopolymers 98 (2012) 416–427.
- [52] N.H. Tolia, L. Joshua-Tor, Strategies for protein coexpression in *Escherichia coli*, Nature Methods 3 (2006) 55–64.
- [53] O.P. Ward, Production of recombinant proteins by filamentous fungi, Biotechnology Advances 30 (2012) 1119–1139.
- [54] E. Çelik, P. Çalik, Production of recombinant proteins by yeast cells, Biotechnology Advances 30 (2012) 1108–1118.
- [55] S. Klatt, S. Konthur, Secretory signal peptide modification for optimized antibody-fragment expression-secretion in *Leishmania tarentolae*, Microbial Cell Factories 97 (2012) 1–10.
- [56] M.M.J. Cox, Recombinant protein vaccines produced in insect cells, Vaccine 30 (2012) 1759–1766.
- [57] M. Butler, Animal cell cultures: recent achievements and perspectives in the production of biopharmaceuticals, Applied Microbiology and Biotechnology 68 (2005) 283–291.
- [58] V. Agrawal, M. Bal, Strategies for rapid production of therapeutic proteins in mammalian cells, BioProcess International 10 (2012) 32–48.
- [59] J.W. Larrick, D.W. Thomas, Producing proteins in transgenic plants and animals, Current Opinion in Biotechnology 12 (2001) 411–418.
- [60] L. Houdebine, Production of pharmaceutical proteins by transgenic animals, Comparative Immunology, Microbiology and Infectious Diseases 32 (2009) 107–121.
- [61] P.N. Desai, N. Shrivastava, H. Padh, Production of heterologous proteins in plants: strategies for optimal expression, Biotechnology Advances 28 (2010) 427–435.
- [62] P. Ahmad, M. Ashraf, M. Younis, X. Hu, A. Kumar, N.A. Akram, F. Al-Qurainy, Role of transgenic plants in agriculture and biopharming, Biotechnology Advances 30 (2012) 524–540.
- [63] G. Giddings, G. Allison, D. Brooks, A. Carter, Transgenic plants as factories for biopharmaceuticals, Nature Biotechnology 18 (2000) 1151–1155.
- [64] H. Daniell, N.D. Singh, H. Mason, S.J. Streatfield, Plant-made vaccine antigens and biopharmaceuticals, Trends in Plant Science 14 (2009) 669–679.
- [65] S. Tiwari, P.C. Verma, P.K. Singh, R. Tuli, Plants as bioreactors for the production of vaccine antigens, Biotechnology Advances 27 (2009) 449–467.

- [66] M. Tran, B. Zhou, P.L. Pettersson, M.J. Gonzalez, S.P. Mayfield, Synthesis and assembly of a fulllength human monoclonal antibody in algal chloroplasts, Biotechnology and Bioengineering 104 (2009) 663–673.
- [67] E. Specht, S. Miyake-Stoner, S. Mayfield, Micro-algae come of age as a platform for recombinant protein production, Biotechnology Letters 32 (2010) 1373–1383.
- [68] M. Schuster, W. Jost, G.C. Mudde, S. Wiederkum, C. Schwager, E. Janzek, F. Altmann, J. Stadlmann, C. Stemmer, G. Gorr, *In vivo* glyco-engineered antibody with improved lytic potential produced by an innovative non-mammalian expression system, Biotechnology Journal 2 (2007) 700–708.
- [69] E.L. Decker, R. Reski, Glycoprotein production in moss bioreactors, PlantCell Rep 31 (2012) 453-460.
- [70] B. Verbist, G. Klambauer, L. Vervoort, W. Talloen, , QSTAR Consortium, Z. Shkedy, T.O. Bender, A. Gohlmann, H.W.H.H. Sepp, Using transcriptomics to guide lead optimization in drug discovery projects: lessons learned from the QSTAR project, Drug Discovery Today 20 (2015) 505–513.
- [71] WHO World Health Organization, Pharmaceutical Industry, 2015. Available at: http://www.who. int/trade/glossary/story073/en/ (accessed 21.02.15).
- [72] EFPIA European Federation of Pharmaceutical Industries and Associations. The Pharmaceutical Industry in Figures, 2015. Available at: http://www.efpia.eu/uploads/Figures\_2015\_Key\_data.pdf (accessed 15.01.2016).
- [73] EFPIA, The Pharmaceutical Industry in Figures, European Federation of Pharmaceutical Industries and associations, Brussels, 2013.
- [74] G.S. Burrill, Co, Biotech 2010 Life Sciences: Adapting for Success, Editora Burrill & Company, São Francisco, 2010.
- [75] P. Dylst, A. Vulto, B. Godman, S. Simonens, Generic medicines: solutions for a sustainable drug market? Applied Health and Economic Health Policy 11 (2013) 437–443.
- [76] M.S. Kinch, J. Raffo, Economic Research Working Paper. N. 24. Economics and Statistics Series, WIPO (World Intelectual Property Organization) (2015) 1–14.
- [77] V. Pimentel, R. Gomes, A. Landim, J. Pieroni, P.P. Filho, Saúde como desenvolvimento: perspectivas para a atuação do BNDES no complexo industrial da saúde, BNDES 60 anos: perspectivas setoriais 2 (2012) 300–332.
- [78] B. Carlo-Fernandez, J.L. Martinez-Hurtado, Biosimilars: company strategies to capture value the biologics market, Pharmaceuticals 5 (2012) 51393–51408.
- [79] Evaluate Pharma, World Preview 2018 Embrancing the Patent Cliff, *In* "Technical report", Evaluate Pharma, London, 2012.
- [80] V. Pimentel, R. Gomes, A. Landim, J. Pieroni, P.P. Filho, Health to Development: Perspectives for the Performance of BNDES in Industrial Health Complex, 2012 (In Portuguese). Available at: http://www.bndes.gov.br/SiteBNDES/bndes/bndes\_pt/Institucional/Publicacoes/Consulta\_Expressa/ Tipo/BNDES\_Setorial/201210\_8.html (accessed 28.11.14).
- [81] E.C. Li, R. Abbas, I.A. Jacobs, D. Yin, Considerations in the early development of biosimilar products, Drug Discovery Today 20 (2) (2015) 1–9.
- [82] M. Kessel, The problem with today's biopharmaceutical business: an outsider's view, Nature Biotechnology 29 (2011) 27–33.
- [83] M. Chui, Despite global downturn generics makers stand to reap great rewards in China's evolving pharmaceutical market, Journal of Genetics and Medical Business 6 (2009) 230–236.
- [84] J. Woodcock, J. Griffin, R. Behramn, B. Cherney, T. Crescenzi, B. Frase, D. Hixon, C. Joneckis, S. Kozlowski, A. Rosenberg, L. Schrager, E. Shacter, R. Temple, K. Webber, H. Winkle, The FDA's assessment of follow-on protein products: a historical perspective, Nature Reviews Drug Discovery 6 (2007) 437–442.

- [85] G. Walsh, Biopharmaceutical benchmarks 2014, Nature Biotechnology 32 (2014) 992–1000.
- [86] R.A. Huml, Investment considerations for pharmaceutical product portfolios, in: Optimization of Pharmaceutical R&D Programs and Portifolios, Springer International Publishing, Switzerland, 2015, pp. 49–69.
- [87] R.J.Y. Ho, Biotechnology and Biopharmaceuticals: Transforming Proteins and Genes into Drugs, Second ed., Wiley Blackwell, New Jersey, 2013.
- [88] B. Godman, W. Shrank, B. Wettrmark, M. Andersen, I. Bishop, T. Burkhardt, K. Garuolienè, M. Kalaba, O. Laius, R. Joppi, C. Sermet, U. Schwabe, I. Teixeira, F.C. Tulunay, K. Wendykowska, C. Zara, L. Gustafsson, Use of generics – a critical cost containment measure for all healthcare professionals in Europe? Pharmaceuticals 3 (2010) 2470–2494.
- [89] K.V. Vooren, A. Curto, L. Garanttini, Biosimilar versus generic drugs: same but different? Applied Health and Economic Health Policy 13 (2015) 125–127.
- [90] S. Mukherjee, RAND Study: Biosimilars Could Save US \$44 Billion in 10 Years. BioPharma Dive. Available at: www.biopharmadive.com/news/rand-study-biosimilars-could-saveus-44-billion-in-10-years/328924/ (accessed 31.12.14).
- [91] C. Mintz, Biosimilars in Emerging Markets. BioProcess Online, July 2013. Available at: http://www. bioprocessonline.com/doc/biosimilarsin-emerging-markets-0001 (accessed 25.01.15).
- [92] C. Willyard, Biosimilar makers flout forced patent negotiations in US, Nature Medicine 21 (2015) 100–101.
- [93] H. Rahalkar, Historical overview of pharmaceutical industry and drug regulatory affairs, Pharmaceutical Regulatory Affairs (2012) S11:002.
- [94] S.C. Chow, Biosimilars: Design and Analysis Follow-on Biologics, first ed., CRC Press, Florida, 2014.
- [95] B.A. Liang, Regulating follow-on biologics, Harvard Journal Legislation 44 (2007) 363-373.
- [96] M. Mathieu, Biologics Development: A Regulatory Overview, Parexel International, S.A., Whaltman, 2004.
- [97] L. Tsang, N. Cortez, Biopharmaceuticals: definition and regulations, in: J.A. Cavagnaro (Ed.), Preclinical Safety Evaluation of Biopharmaceuticals: A Sicence-Based Approach to Facilitating Clinical Trials", John Wiley& Sons Inc., Nova Iorque, 2008, pp. 3–20 (Chapter 1).
- [98] EMA. EMA/CHMP/437/04, Rev 1. Guideline on Similar Biological Medicinal Products, 2015. Available at: http://www.ema.europa.eu/docs/en\_GB/document\_library/Scientific\_guideline/ 2014/10/WC500176 768.pdf (accessed 15.02.15).
- [99] Brasil. Ministério da Saúde, Lei n. 5991 de 17 de Dezembro de 1973. Dispõe sobre o Controle Sanitário do Comércio de Drogas, Medicamentos, Insumos Farmacêuticos e Correlatos, e dá outras Providências, 1973.
- [100] Brasil. Ministério da Saúde, Lei n. 6.360, de 23 de setembro de 1976. Dispõe sobre a Vigilância Sanitária a que ficam sujeitos os Medicamentos, as Drogas, os Insumos Farmacêuticos e Correlatos, Cosméticos, Saneantes e Outros Produtos, e dá outras Providências, 1976.
- [101] Brasil. Ministério da Saúde, Lei n. 8.080, de 19 de setembro de 1990. Dispõe sobre as condições para a promoção, proteção e recuperação da saúde, a organização e o funcionamento dos serviços correspondentes e dá outras providências, 1990.
- [102] Brasil. Ministério da Saúde, Lei n. 8.142, de 28 de dezembro de 1990. Dispõe sobre a participação da comunidade na gestão do Health Unic System (SUS) e sobre as transferências intergovernamentais de recursos financeiros na área da saúde e dá outras providências, 1990.
- [103] Ministério da Saúde, Portaria 2.577/GM, de 27 de outubro de 2006. Aprova medicamentos de Dispensação Excepcional, 2006.

- [104] Brasil, Lei n. 9.787, de 10 de fevereiro de 1999. Altera a Lei n. 6.360, de 23 de setembro de 1976, que dispõe sobre a vigilância sanitária, estabelece o medicamento genérico, dispõe sobre a utilização de nomes genéricos em produtos farmacêuticos e dá outras providências, 1999.
- [105] ANVISA. Agência Nacional de Vigilância Sanitária, Resolução n. 134, de 13 de julho de 2001. Determina a todos os estabelecimentos fabricantes de medicamentos, o cumprimento das diretrizes estabelecidas no Regulamento Técnico das Boas Práticas para a Fabricação de Medicamentos, conforme ao Anexo I da presente Resolução, 2001.
- [106] ANVISA. Agência Nacional de Vigilância Sanitária, Resolução RDC n. 133, de 29 de maio de 2003. Dispõe sobre o registro de medicamento similar e dá outras providências, 2003.
- [107] Ministério da Saúde. ANVISA Agência de Vigilância Sanitária, Resolução RDC n. 17, de 16 de abril de 2010. Dispõe sobre as Boas Práticas de Fabricação de Medicamentos, 2010.
- [108] Ministério da Saúde. ANVISA Agência de Vigilância Sanitária, Resolution of the Executive Board RDC n. 16, de 28 de março de 2013. Aprova o Regulamento Técnico de Boas Práticas de Fabricação de Produtos Médicos e Produtos para Diagnósticos de Uso *in Vitro* e dá outras providências, 2013.
- [109] Brasil, Lei n. 10.973, de 2 de dezembro de 2004. Dispõe sobre incentivos à inovação e à pesquisa científica e tecnológica no ambiente produtivo e dá outras providências, 2004.
- [110] Ministério da Saúde, Brasil vai investir 35 bilhões em medicamentos até 2016, in: Pfarma. com.br (Semerene, B.), 2013. Available at: http://pfarma.com.br/noticia-setor-farmaceutico/ mercado/1314-brasil-vai-investir-vai-investir-35-bilhoes-em-medicamentos-ate-2016.html (accessed 10.06.15).
- [111] ABIFINA, Associação Brasileira das Indústrias de Química Fina, Biotecnologia e suas Especialidades. Biofármacos: umhorizonte de oportunidades, FACTO 38 (2013) 40.
- [112] Ministério da Saúde. ANVISA Agência de Vigilância Sanitária, Resolução RDC n. 55, de 14 de novembro de 2012. Dispõe sobre os detergentes enzimáticos de uso restrito em estabelecimentos de assistência à saúde com indicação para limpeza de dispositivos médicos e dá outras providências, 2012.
- [113] Ministério da Saúde, Portaria n. 506, de 21 de março de 2012. Instituo o Programa para o Desenvolvimento do Complexo Industrial da Saúde (PROCIS) e seu Comitê Gestor, 2012.
- [114] Ministério da Saúde, Portaria n. 837, de 18 de abril de 2012. Define as diretrizes e os critérios para o estabelecimento das Parcerias para o Desenvolvimento Produtivo (PDP), 2012.
- [115] Ministério da Saúde. ANVISA Agência de Vigilância Sanitária, Resolução n. 49, de 31 de outubro de 2013. Dispõe sobre a regularização para o exercício de atividade de interesse sanitário do microempreendedor individual, do empreendimento familiar rural e do empreendimento econômico solidário e dá outras providências, 2013.
- [116] Ministério da Saúde. ANVISA Agência de Vigilância Sanitária, Resolução RDC n.50, de 6 de novembro de 2013. Altera o Art. 4° da Resolution of the Executive Board n. 25, de 16 de junho de 2011, que dispõe sobre os procedimentos gerais para utilização dos serviços de protocolo de documentos no âmbito da ANVISA, 2013.
- [117] C. Reis, A. Landim, J.P. Pieroni, Complexo Industrial da Saúde: Lições da Experiência internacional e propostas para incorporação da rota biotecnológica na indústria farmacêutica brasileira, BNDES Setorial 34 (2011) 5–44.
- [118] Datamark, Brazilian Biosimilars Market, 2012. Available at: http://www.biosimilarnews.com/ brazilian-biosimilars-market (accessed 30.04.15).
- [119] K.R. Crews, J.K. Hicks, C. Pui, M.V. Pelling, W.E. Evans, Pharmacogenomics and individualized medicine: translating science into practice, Clinical Pharmacology & Therapeutics 92 (2012) 467–475.
- [120] A. Squassina, M. Manchia, V.G. Manolopoulos, M. Artac, C. Lappa-Manakou, S. Karkabouna, K. Mitropoulos, M. Del Zompo, G.P. Patrinos, Realities and expectations of pharmacogenomics and personalized medicine: impact of translating genetic knowledge into clinical practice, Pharmacogenomics 11 (2010) 1149–1167.

# Interferons

#### W.M.S. Degrave\*, L. Mendonça-Lima

LABORATORY FOR FUNCTIONAL GENOMICS AND BIOINFORMATICS, OSWALDO CRUZ INSTITUTE-FIOCRUZ, RIO DE JANEIRO, RJ, BRAZIL

## 3.1 Interferons: Biological Functions and Clinical Uses

Although the phenomenon of interference in the process of viral infection caused by a previous (viral) infection had been known for several decades, in 1957, Isaacs and Lindenmann [1] decisively characterized the responsible factor that was released by chick chorioallantoic membrane after exposure to heat-inactivated influenza virus. That factor interfered with the growth of live virus in fresh pieces of membrane and was called "interferon" (IFN). Intensive research showed that, apparently, all viruses can induce IFN in most types of vertebrate animal cells and, in principle, IFN can inhibit all types of viruses. IFNs were shown to possess potent antiviral and antiproliferative activities, and also immune and inflammatory modulation roles in the human body through cell–cell interactions [2]. Such activities inspired high hopes for the development of medical applications, and the genes for human IFN-alfa, -beta, and -gamma were cloned and characterized in the early 1980s [3–10].

Their expression [11-17] and clinical investigation contributed to the development of the biotechnology industry with Genentech, Biogen, Hoffmann-La Roche, and others. Indeed, the expression, purification, and clinical trials set high hopes that human IFN would be a magic bullet against viral infections, and especially against cancer.

After the IFN-alfa, beta, and gamma encoding genes, many other interleukins (ILs) were cloned and characterized, starting with IL-2 [18], and initiated important biotechnological and biopharmaceutical portfolios. Many were also cloned from other vertebrate organisms; later, genome projects and bioinformatic analyses amplified the scientific knowledge and cytokine product portfolio.

IFN-alfa proved to be one of the most important biopharmaceutical products [19], especially because of its application in the treatment of viral hepatitis [20] and also because of its oncologic indications, with a market value of about US \$1 billion per year, whereas IFN-beta has progressed toward more widespread use for the treatment of multiple sclerosis. IFN-gamma, also called "immune IFN," was expected to be the most prominent of the IFNs, but has not found much clinical use thus far. It is applied in

<sup>\*</sup>Corresponding Author.

chronic granulomatosis and rheumatic diseases. A lot of research was invested in the discovery and further characterization of additional molecules with IFN-like activities, including the bioinformatic analyses of genomic sequences; a "consensus interferon" was marketed, and synthetic modified variants of IFN with better stability, systemic half-life, and modified activities are being engineered.

To date, three main types of IFN have been described: type I IFNs, composed of IFNalfa, -beta, -epsilon, -kappa, -tau, -omega, -zeta (the latter is also called limitin [21]), and -delta (only in mice); type II IFNs, with IFN-gamma as its sole member; and type III IFNlambda (also called IL-28 and -29). Their respective biological activities, physiologic roles in the host defense system, mechanism of action, and gene/protein structures have been studied in greater detail, and are briefly described subsequently.

IFNs can be described as a group of cell–cell signaling proteins that are produced and released by host cells in the presence of a variety of pathogens ranging from viruses to bacterial and fungal products, parasites, and tumor cells [22,23], besides natural and synthetic nucleic acids, several polymers, and mitogens. They are secreted and trigger the deployment of protective measures in neighboring cells. As such, IFNs can be considered cytokines [24] that activate innate and acquired immune responses such as natural killer cells and macrophage activities and increased expression of the major histocompatibility complex, which promotes antigen presentation and mediates cellular immunity. IFN proteins range from 19 to 45 kDa, with 143–172 amino acids. All IFNs are monomers, except for IFN-gamma, which is a homodimer. They are frequently modified by glycosylation, but these posttranslational modifications are often not critical for their biological function.

Human IFNs are classified according to the receptors to which they bind, provoking a signaling cascade in the target cell [24].

Type I IFNs [25] are found in all mammals, and orthologs have been found in birds, reptiles, amphibians, and fish species [26]. Type I IFNs are defined according to their binding to a specific multichain receptor complex located on the cell surface, and known as IFNAR (for IFN-a/b receptor), consisting of at least two different subunits, IFNAR1 and IFNAR2 [27]. They are encoded by at least 23 IFN genes with a length of 1-2 kb, which are clustered on the short arm of human chromosome 9, at band p21 [28,29], resulting in 13 subgroups: IFN-a1, -a2, -a4, -a5, -a6, -a7, - a8, -a10, -a13, -a14, -a16, -a17, and -a21. It is not clear whether all of these genes are expressed after cell stimulation, and the expression varies in different cell types [30]. Only two species of human alfa-IFNs are glycosylated: IFN-a2b is O-glycosylated at the threonine residue in position 106 and IFN-a14c is N-glycosylated at the asparagine residue at position 72 [31]. IFN-alfa is a major cytokine produced by fibroblasts and monocytes in response to a viral infection [32]. Furthermore, type I IFNs produced by malignant cells or tumorinfiltrating dendritic cells seem to have a role in anticancer immunity by controlling the autocrine or paracrine circuits that underlie cancer immune surveillance [33]. The only type II IFN in humans is IFN-gamma, which is secreted by type I T helper (Th) cells (Th1 response) and has a pivotal role in the immune system because it is activated by IL-12 and suppresses type II Th cells (Th2). IFN-gamma binds to the oligomeric IFN-gamma receptor (IFNGR) consisting of IFNGR1 and IFNGR2 chains [34].

Type III IFNs bind to a receptor complex composed of IL10R2 (or CRF2-4) and IFNLR1 (CRF2-12) chains and is important in some types of virus infections [35].

Binding of secreted IFNs to their specific cell surface receptors on neighboring cells leads to the activation of signal transducer and activator of transcription complexes (STATs), which constitute a family of specific transcription factors responsible for the regulation of specific genes related to the immune response [34]. In the case of the IFNs, STATs are activated via phosphorylation by receptor-associated Janus-activated kinases and associate with other factors to move to the nucleus of the cell, where the complex binds to specific nucleotide sequences called IFN-stimulated response elements in the promoter regions of IFN-stimulated genes, inducing their transcription and subsequent expression. Several additional signaling complexes can bind to IFN-activated site elements in gene promoters, some specific for type I or II IFNs [34], and the expression of a wide variety of genes is thus influenced [36,37].

Subsequent antiviral activity of the type I IFN response is mediated by several defense systems in the cell, such as protein kinase RNA-activated and 2',5'-oligoadenylate synthetase/RNase L system, the adenosine deaminase ADAR1, the Mx GTPases, and cellular microRNAs [38]. In infected cells, expression of IFN itself is activated by both viral RNA and DNA through sensing of double-stranded RNA by membrane-bound Toll-like receptor, triggering activation of transcription factors such as IRF3 and nuclear factor- $\kappa$ B, and retinoic acid-inducible gene 1, melanoma differentiation-associated protein 5, and DAI cytosolic viral nucleic acid sensors [38,39].

In addition, mitogens and several other cytokines can induce IFN production. In response to IFN, cells slow down protein synthesis, degrade mRNA (both viral and host), and end up promoting apoptosis of the infected cells [40-42].

#### 3.1.1 Interferon Protein Structures

Human IFN-alfas are synthesized as a 188- to 189-residue-long precursor protein. The first 23 residues correspond to a signal peptide, yielding a protein with a calculated molecular mass of 18.5 kDa. The experimentally determined molecular masses range from 19 to 26 kDa, owing to the presence of posttranslational modifications [23]. The gene sequences encoding the IFN-alfa subfamily show a high level of overall sequence similarity and the corresponding protein sequences share between 70% and 80% homology and about 35% identity with human IFN-beta [43]. Several highly conserved residues in IFN-alfa proteins were shown to be crucial for their function (lysine residues close to the N-terminus and a tyrosine residue near the C-terminus) [44]. Conformational stability is enhanced through four cysteine residues at positions 1, 29, 99, and 139 that are involved in the formation of two disulfide bonds between the Cys29 and Cys139 residues that are necessary for biological activity, and between Cys1 and Cys99, which is functionally nonessential [45,46].

After crystallization of IFN-alfa, its three-dimensional structure could be elucidated [47,48]. IFNs-alfa and -beta are structured in a single domain, which is mostly alpha-helical.

The globular structure is composed of five major alpha-helices connected by a long loop and three shorter loops. The amino acids essential for the structuring of the protein, as well as for interaction with the IFN receptor, are known [49,50]. Several intrinsically disordered regions seem to have a functional role in the protein [51].

IFN-gamma shows little homology with type I IFN protein sequences and is encoded on chromosome 12, map 12q24.1. Unlike the other IFNs [4,7], its gene has three intervening sequences [10], the first of which contains a repetitive element. The monomeric molecular weight (MW) of the protein is around 35 kDa. The precursor protein is 166 amino acids, of which 24 N-terminal residues form a signal peptide. It is homodimeric in solution, with an MW of 70 kDa. The IFN-gamma protein has a structure similar to the type I IFNs, because it is constituted primarily by alpha-helical structures, with six helices in each subunit, without beta sheets. The dimeric structure is stabilized by the intertwining of helices with multiple intersubunit interactions [52].

#### 3.1.2 Clinical Applications

Although several cases of viral infection resistance to IFN are known [53,54], as also demonstrated in the case of HIV [55] and in several cancers [56], many preparations of IFNs have been tested in clinical trials.

IFNs are potent biopharmaceuticals but they pose many hurdles for clinical applications. Because they are synthesized in the body in low quantities and show variable levels of species specificity, only human recombinant IFN can be used for medicinal applications. Moreover, it was soon evident that because IFNs act mainly on the level of cell–cell interactions, their systemic application, mostly by intramuscular or sometimes subcutaneous injection, is not straightforward, although fairly well tolerated.

IFN treatment has frequently been shown to be on the limit of the compromise between therapeutic effect and toxicity. Often, flulike symptoms are experienced within hours of IFN-alfa administration, which can be alleviated by paracetamol administration and tolerance later on. However, in a significant percentage of patients, treatment has to be interrupted owing to cardiovascular, central nervous system, or autoimmune reactions.

#### 3.1.3 Interferon-Alfa in Antiviral Treatment

The most successful and widespread clinical application of (alfa)-IFN has been in the treatment of chronic hepatitis infections, especially hepatitis C (HCV) (formerly non-A/ non-B). HCV infection is a global threat; an estimated 130–170 million people are chronically infected worldwide [57] and over 350–1000 deaths occur each year as a result of detrimental HCV-related liver diseases such as cirrhosis and cancer [58]. Given either as monotherapy or combined with ribavirin, IFN-alfa has been the treatment of choice for patients with chronic HCV infection [59]. New chemotherapeutic tools have become available and the use of IFN may be quickly coming to an end [60].

The IFN-alfa2 subtype has been the only IFN in use for the treatment of chronic hepatitis B and C, and leukemia [19,61]. Natural recombinant alfa-IFNs approved by the Food and Drug Administration (FDA) as therapeutics are mainly IFN-alfa2a (Roferon A1) and IFN-alfa2b (Intron A1) subtypes, and their PEGvlated forms (Pegasys1 and PEG-Intron1). Introduction of the PEGylated IFN in 2001 led to the improvement of its antiviral effectiveness, mostly owing to the longer half-life in the bloodstream, which required only one injection per week instead of the previous two to three and combination with ribavirin therapy [62]. However, variable success rates have been observed in different countries, especially with different HCV subtypes [63]. Thus, there has been an intensive search for new therapies. Often, additional antiviral drugs such as protease inhibitors Telaprevir or Boceprevir, or the nucleotide analog Sofosbuvir are given in combination treatments. Interestingly, human genetic polymorphism near the gene IL28B that encodes IFN lambda3 was shown to influence patients' response to treatment and also natural clearance of the HCV genotype 1 virus [64,65]. The high cost and frequency of relapse led to a renewed effort for the development of alternative drugs, which have since reached the market [20,66].

IFN-alfa was also shown to be successful in the treatment of several papilloma viral infections such as genital warts and laryngeal papillomatosis [67].

#### 3.1.4 Antitumor Treatment

The antiproliferative activities of crude preparations of IFN-alfa reported in the late 1970s and early 1980s elicited high hopes for anticancer treatments [68]. Encouraging results were obtained for breast cancer, lymphomas, and some myelomas and sarcomas. Clinical studies with purified recombinant IFN-alfa in the 1980s were less successful mostly owing to collateral effects, and the clinical use of IFN-alfa became more restricted. Schering Plough's Intron A (IFN-alfa2b produced in *Escherichia coli*) has been on the market since 1986. IFN-alfa2a and 2b remained successful in the treatment of hairy cell leukemia (a B-lymphocyte neoplasm) and also malignant melanoma, condylomata acuminata (genital warts), ovarian carcinoma, follicular lymphoma, metastatic renal-cell carcinoma, and AIDS-related Kaposi sarcoma [69]. IFN-alfa is also used as an adjuvant with other therapeutics to treat patients with bladder and renal cancers, and melanoma [70].

#### 3.1.5 Interferon-Beta

IFN-beta, which was first cloned, sequenced, and expressed in 1982 [6,7], did not immediately find a satisfactory medical application, but was later used to treat relapsing-remitting multiple sclerosis, and slowed disease progression [71,72]. IFN-beta is thought to inhibit IFN-gamma and TNF-alpha production, thus down-regulating the proin-flammatory response. The mechanism of action of IFN-beta is complex, involving effects at multiple levels of cellular function. IFN-beta appears to increase expression and concentration of antiinflammatory agents directly while down-regulating the expression

of proinflammatory cytokines. IFN-beta treatment may reduce the trafficking of inflammatory cells across the blood—brain barrier and increase nerve growth factor production, leading to a potential increase in neuronal survival and repair. Overall, therapy with IFN-beta leads to a reduction of neuron inflammation [73].

Recombinant IFN-beta can be expressed and purified with relative ease in *E. coli* or in Chinese hamster ovary (CHO) cells (Avonex and Rebif). Production in *E. coli*, such as Betaferon/Betaseron, yields a nonglycosylated product, which has not affected its biological activities negatively [74]. In Betaferon, the cysteine 17 was substituted by a serine residue.

Type I IFNs are also promising for their use as vaccine adjuvants, but the subtype of IFN for this purpose could have a significant influence on the adaptive immunity generated at specific immune compartments. For instance, IFN-beta could be used to enhance systemic T cell immune responses, whereas IFN-epsilon could be more promising as an adjuvant to enhance mucosal T cell immunity in the lung and the gut mucosae [75].

#### 3.1.6 Interferon-Gamma

IFN-gamma, a type II IFN, formerly called immune IFN, is a critical cytokine in innate and adaptive immunity against infection by a variety of viral pathogens, and some bacterial and protozoal infections. It activates other immune cells, such as macrophages and natural killer cells, and has an essential immune-stimulatory and immunemodulatory effect. However, it is also involved in several autoimmune diseases [76]. Because of its complex and pivotal role in the immune system, clinical applications have not been straightforward, and IFN-gamma is used on few occasions, such as for chronic granulomatous disease. It is normally produced in *E. coli* and is purified with standard chromatography techniques.

#### 3.1.7 Other Interferons

IFN-epsilon is a type I IFN that was identified more recently. It is highly and constitutively expressed in the brain, but its biochemical and biological characteristics are not fully understood. It has less antiviral, cytotoxic, and antiproliferative activity, but the differential regulation of genes related to the central nervous system by recombinant human IFN-epsilon has been reported, which suggests a role for this IFN in the structure and operation of the brain [77]. However, one of its main functions seems to be in the protection of the female reproductive tract against viral and bacterial infection [78]. Contrary to other type I IFNs, IFN-epsilon is not induced by known pattern recognition receptors that signal to induce effector cytokines in response to pathogens that normally trigger the innate immune system. However it was shown to be constitutively expressed by epithelial cells of the female reproductive tract and also in the male reproductive organs, and to be hormonally regulated, possibly with a crucial role in the defense against sexually transmitted pathogens [79]. IFN-kappa is also a member of the type I IFN family and is expressed in keratinocytes. The gene is found on chromosome 9, adjacent to the type I IFN cluster [80,81].

IFN-omega has only one functional form described to date (*IFNW1*), but has several pseudogenes: *IFNWP2*, *IFNWP4*, *IFNWP5*, *IFNWP9*, *IFNWP15*, *IFNWP18*, and *IFNWP19* in humans. Many nonprimate placental mammals express multiple IFN-omega subtypes [82].

IFN-zeta, also called limitin, displays antiviral, immunomodulatory, and antitumor effects, but it has much less lympho-myelosuppressive activities than does IFN-alfa, and a narrow range of biological activities [83].

Additional type I IFNs, such as alfa-omega, nu, and IFN-delta, are found in several mammal species [84], as well as IFN-tau [85], which is a multifunctional type I IFN secreted by trophectoderm, and acts as the pregnancy recognition signal in ruminants. It also displays antiviral, antiproliferative, and immunomodulatory bioactivities [86].

The most recently discovered type III IFN group consists of four IFN-lambda proteins called IFN-lambda 1, 2, 3, and 4, (also called IL-29, IL-28A, and IL-28B) [87,88], but their biological activity has been less well characterized.

## 3.2 Interferon Production and Formulation

Initial IFN production systems were established using mammalian cell lines in culture flasks; soon after, immortalized cancer cell lines were discovered to secrete larger amounts of IFN-alfa. For example, a human lymphoblastoid cell line called Namalwa, stimulated with Sendai virus, was used as a large-scale production system, yielding a mixture of at least eight different IFN-alfa subtypes [89]. This mixture was then purified from the culture supernatant using standard process chromatography systems. Much improvement was also made with the use of monoclonal antibodies for the sensitive detection of IFN in immunoassays. Wellferon was one of the first products produced in this way. It contained a mixture of nine IFN-alfa subtypes. Such "natural" IFN preparations were later shown to elicit less relapsed or refractory clinical responses, at least partly owing to the much lower frequency of the production of IFN-neutralizing antibodies [90]. However the search for purified single-IFN species, large-scale and cheaper production systems, and the newly developed recombinant DNA technologies that became available in the early 1980s quickly led to the cloning and sequencing of IFNalfa, -beta, and -gamma complementary DNAs (cDNAs) from humans, mice, and other mammals, also followed by the characterization of their genomic gene sequences. Many cDNAs were expressed in a variety of heterologous systems such as E. coli, yeast, fungi, and mammalian cell lines (CHO and monkey kidney cell lines). However E. coli remained the system of choice because most human IFN-alfa and -beta is not glycosylated, and although IFN-gamma is glycosylated, the form expressed in E. coli shows similar biological activity. Cloning gene sequences and expressing them in high volumes

and cell densities with relatively cheap culture media is easy, fast, and accessible, but has as a major drawback extensive and costly purification steps that are needed to remove unwanted contaminants from the product, and it is often difficult to achieve full biologically active protein conformation. Escherichia coli does not normally secrete heterologous proteins into the culture medium, which poses the challenge of removing all unwanted nucleic acids, contaminating proteins and pyrogenic activities from the protein solution to reach more than 99% purity. Also, many heterologous proteins are trapped in inclusion bodies in *E. coli* expression systems, and much work has been invested to avoid or minimize this phenomenon. In some cases in which biologically active conformations are not needed, inclusion bodies can be an advantage, providing a partial purification step, for example in the production of antigens with nonconformational epitopes. In other cases, however, different promoter systems were developed to minimize inclusion body formation through better transcriptional control, slower expression, or induction and expression at lower temperatures, for example at 25°C or 20°C. Synthetic gene construction and codon use adaptation of the heterologous sequences to the E. coli codon preference has been added to the arsenal of tools. However, careful research into optimization of downstream processing technologies is needed, such as inclusion body cleanup and subsequent solubilization with specialized buffers, followed by renaturation of the recombinant protein. The optimization of renaturation protocols must be achieved case by case, because many factors have a role in this process, most of them poorly understood, and variable degrees of loss of the active protein must be accounted for [91-93]. Purification protocols must be optimized, and usually include precipitation steps, and chromatographic techniques ranging from gel filtration and ion exchange, metal affinity, hydrophobic interaction, and also affinity chromatography using immobilized anti-IFN monoclonal antibodies can be employed, besides ultrafiltration, crystallization, and more specific techniques. Large-scale chromatographic systems are expensive; they use process scale columns, with medium pressure and often a final high-pressure high-performance liquid chromatography (HPLC) step. The purified protein can be crystallized before final formulation, which adds to purity and stability, and micro- and nanocrystals are increasingly used as part of the formulation technology (Fig. 3.1).

Many different approaches have been used to overexpress human IFN-alfa in *E. coli* using a variety of promoter systems and even more purification schemes [94–96]. New tools for optimization of codon use have been employed, with promising success [97,98], although many factors influencing protein translation and folding, including codon bias, are still only partially understood. Alternatively, other expression systems, such as *Bacillus subtilis* [99,100], *Pichia pastoris* [101,102], mammalian cells such as CHO [103] or Human Embryonic Kidney cells (HEK 293) [104], insect cells using baculoviral vectors [105], and cell-free protein synthesis systems [106], were employed and evaluated for the production of recombinant human IFNs.

Several IFN-alfa 2a formulations have been marketed, and even more for IFN-alfa 2b such as Intron A from Schering Plough, and many new formulations mostly from new or



FIGURE 3.1 Overview of recombinant protein expression and purification scheme suitable for production of IFN in a bacterial, yeast, or cell system.

regional companies, such as Realderon from Teva-Israel, and many formulations: Viferon, Alpharona, Reaferon, Infagel, Recolin, Altevir, Kipferon, Giaferon, Genferon, Grippferon, Opthalamoferon, and Gerpferon, from several Russian companies [107]. To increase serum half-life, high—molecular weight groups such as polyethylene glycol (PEGylation) can be added through reaction of free amino groups of the IFN protein with activated PEG such as methoxy-polyethylene glycol or other variants, resulting in more

effective drugs, which contain, in addition to mono-PEGylated molecules, some free and di-PEGylated forms [108,109]. Typically, some 40 kDa of linear or branched PEG chains are added. They are thought to provide a slower release into the bloodstream, but also a slower clearance resulting, on average, in a threefold longer half-life. Examples are Pegasys, Reiferon Retard, Peg-Intron, Pegetron, and ViraferonPeg, among others (Table 3.1).

Type of Interferon	Trade Name	Company	Observations
IFN-alpha-n1	Wellferon	GSK	A "natural" mixture of >20 IFN-alpha, purified from a Sendai-virus induced lymphoblastoid cell line, for treatment of hairy cell leukemia, chronic hepatitis B and genital warts.
IFN-alpha-n3	Alferon-N	Interferon Sciences Hemispherx Biopharma	A "natural" mixture of up to 14 IFN- alpha, purified from Sendai-induced pooled human leukocytes.
IFN-alpha-Le	Multiferon	Swedish Orphan Biovitrum	A mixture of 6 IFN-alpha
IFN-alpha 2a	Roferon A	Roche	Neoplasms of the lymphatic and hematopoietic system; chronic hepatitis B and C. Produced in <i>E. coli</i>
Interferon-alpha	Intron A	Schering/Merck	Hairy cell leukemia, melanoma, follicular
2b	Viraferon	Shering Plow	lymphoma, chronic hepatitis B and C,
	Reliferon Uniferon	Reliance life Sciences (India) Getz life Sciences (Pakistan)	genital warts. Produced in <i>E. coli</i>
INF-alpha 2c	Berofor	Boehringer Ingelheim	Hairy-cell leukemia, viral eye infections
PEGylated	Pegasys	Hoffmann-La Roche	Chronic hepatitis B and C
IFN-alpha 2a	Reiferon- Retard	Egypt	
PEGylated	Peg-Intron	Schering Plow	Chronic hepatitis B and C
IFN-alpha 2b	Viraferon-Peg Pegetron	Shering Plow Canada	Combination with ribavirin
IFN-beta 1a	Rebif	Merck Serono	Relapsing forms of multiple sclerosis.
	Avonex	Biogen	Produced in CHO cells, glycosylated
IFN-beta 1b	Betaseron	Chiron	
	Betaferon	Shering	
Interferon- gamma 1b	Actimmune	Horizon Pharma	Chronic granulomatous disease, Severe malignant osteopetrosis
	Imukin	Boehringer Ingelheim	
Consensus IFN-alpha	Infergen (IFN-alfacon-1, CIFN or IFN-Con1)	Amgen/Intermune	

#### Table 3.1 Types of Interferon and Commercial Pharmaceutical Forms

## 3.3 Modified or New (Synthetic) Interferons and Formulation

IFNs have many drawbacks as therapeutic molecules, as mentioned previously. Improving activity, eliminating toxicity and unwanted side effects, increasing serum halflife, stability, and shelf life are among the desired modifications, besides optimization of heterologous expression through codon use modifications. Indeed, growing tools for gene synthesis and high content screening make the design and selection of optimal variants accessible, but they are limited by the high costs of clinical trials. Synthetic variants of IFN-alfa are commercially available, and several other modified IFNs are under development. The first such IFN that reached the market will be discussed briefly as an illustration of the overall development and production process for such molecules.

The best known synthetic IFN-alfa, called consensus IFN-a (IFN alfacon-1; Infergen), was launched by Amgen in 1997 [110,111]; it claimed to improve the antiviral activity of IFN-alfa [112,113]. Consensus IFN is thus an artificial recombinant second-generation type I IFN. Its gene sequence was derived after a multiple alignment analysis of 14 human IFN-alfa subtype protein sequences that assigned essentially the most frequently occurring amino acid in each position. The resulting sequence differs from that of IFNa2 at 20 of 166 amino acids (approximately 88% homology), and has about 34% homology with human IFN-beta. The resulting protein sequence was then back-translated using codon optimization into a DNA sequence, cloned in suitable expression plasmids, and introduced in *E. coli*. Induction through the use of adequate promoter systems by either the lambda P<sub>L</sub> system or a T7 polymerase system and isopropyl β-D-1thiogalactopyranoside induction resulted in mostly inclusion bodies, with variable yields. These are then recovered through centrifugation and washing steps, solubilized, refolded, and IFN purified through chromatography. Evaluation of safety, biochemical, and pharmacologic activity was done in rodent and primate models, as well as in vitro on a number of human and animal cell lines. Preclinical tests included pharmacokinetic studies of Infergen in hamsters and rhesus monkeys, demonstrating similar absorption and elimination profiles using several administration routes (intramuscular, subcutaneous, or intravenous). Peak serum concentrations in hamsters and monkeys were measured at several time points. Systemic exposure was evaluated, as well as bioavailability and clearance. Pharmacologic and toxicologic profiles were compared with those of other IFN-alfas. Adverse effects at different doses were also monitored, as well as reproductive toxicology. Initial clinical trials were encouraging, and the consensus IFN was claimed to display higher biological activity (antiviral, antiproliferative, and cytokine-inducing activity) than the native type I IFNs, also in patients [114]. Infergen was approved by the FDA for use in treating chronic HCV infection in 1997, with Amgen as the producer, who passed the rights to InterMune in 2001.

The product is supplied in a sterile liquid dosage form as single-use vials containing  $9 \mu g$  IFN alfacon-1 in 0.3 mL 100 mM NaCl and 25 mM NaPO<sub>4</sub>. The production process

[115] follows a typical general design for recombinant protein production (Fig. 3.1). A Master Cell Bank of the plasmid-transformed recombinant *E. coli* strain is constructed and verified for contamination using microbial tests for contaminant strains of bacteria and phages. The presence of bacteriophages would invalidate the cell bank for several reasons.

Fermentation of *E. coli* is monitored for pH, temperature, dissolved oxygen, cell density, and eventually nutrient monitoring. After cell growth to the desired density, induction and expression take place for the optimum time. Cells are then collected, usually through (continuous) centrifugation or tangential flow filtration and the obtained cell paste is stored for quality assurance testing. This normally includes bioburden and product yield evaluation and testing for contamination, yiability, protein concentration, genotype verification, plasmid retention, and paste yield. At this point, validation of plasmid stability has been confirmed by continuous monitoring of peptide mapping and N-terminal sequence of end-of-production protein, and complete DNA sequencing of the target gene at the end of fermentation. This ensures that the strain, plasmid, IFN coding gene, and resulting protein continue as expected. IFN purification starts with cell lysis (usually through a large-scale pressure homogenizer), centrifugation (of inclusion bodies) or clearing steps, contaminant removal through inclusion body washing, solubilization, oxidation and refolding, precipitation, and clarification, followed by chromatography steps, formulation into final buffer and eventual addition of preservatives, and sterile filtration of purified bulk before filling. The purification process is controlled by monitoring homogenizer pressure, time and temperature limits, bioburden, chromatogram records, and process vield.

The process thus developed must be validated through production of several purified bulk lots derived from at least three cell paste lots, and verification of the absence of host cell and product contamination, purity [HPLC, sodium dodecyl sulfate-polyacrylamide gel electrophoresis (SDS-PAGE), and peptide mapping], identity, and specification of the product, including full biological activity of the IFN, and consistent process yields. Biological activity is measured in validated assays with reference material (natural or reference IFN). Full biochemical and biophysical characterization of the recombinant IFN must, of course, be performed, including complete amino acid sequence determination by sequencing of peptides and mass spectrometry; peptide mapping; amino-terminal isoform analysis by isoelectric focusing; anion exchange and reverse-phase chromatography (HPLC); size exclusion chromatography and laser light scattering; SDS-PAGE electrophoresis; circular dichroism; fluorescence spectroscopy; acid sensitivity; and UV absorption curves (calculation of extinction coefficient). Product stability must be demonstrated, both on bulk and final product, the latter for a shelf life of 24 months at 2–8°C. Therefore, stability assays must be performed for several biological and biochemical parameters in a comprehensive stability program at  $-70^{\circ}$ C,  $-20^{\circ}$ C,  $4^{\circ}$ C, and  $37^{\circ}$ C. Experimental stability tests must also simulate shipping conditions and freeze-thaw and temperature cycling.

Manufacturing facilities must fully comply and be certified for Good Manufacturing Practices, including environmental impact evaluation and control for the production facility and processes. Several other modified IFNs are under development, such as a second-generation IFN-gamma with high activity, through shortening of the C-terminus of the protein, and a third generation with higher stability through introduction of a disulfide bridge in the molecule, leading to a tighter packed protein [116]. Also, several groups have invested continued efforts in the optimization of IFN expression, especially in *E. coli*, through the use of new promoter or induction systems, and even through expression in cell-free systems, although this is currently not a viable large-scale production alternative. However, cell-free systems can be used to obtain sufficient biological material for functional assays in a fast approach, for example, for screening purposes.

#### 3.3.1 Future Perspectives

The main application of IFN-alfa, as an antiviral agent in the treatment of chronic HCV, seems to have come to an end with newer, more effective drugs reaching the market that have fewer side effects. IFN-beta is still promising in the slowing down of the progression of multiple sclerosis. IFNs and IFN-gamma are also used for clinical treatment of some specific tumors. Will this lead to a full winding down of IFN as a drug? Because of its intrinsic cell–cell interaction profile and inherent toxicity and often only partial success when used on a systemic basis, this would not be surprising. However, a number of questions remain unanswered related to the clinical use of IFN. Several options are still available for further development, including a more detailed characterization of the functional role and applications of less known IFNs, such as type III IFN-lambdas and type I IFN-epsilon, -omega, -kappa, and -tau. Indeed, it is possible that specific niche treatments can be discovered for these cytokines.

Equally important could be the construction of a number of variant IFNs based on (parts of) natural sequences, but combining different structural and functional elements into the molecules, and systematic screening for desired characteristics in adequate functional assays. Tools now available in synthetic gene construction and the availability of high-content screening systems should facilitate such an endeavor.

However, in our view, the most promising applications of IFNs have yet to be developed as elements in the construction of targeted cell therapy. One can envisage that many future medicines will target specific cell types: for example, immune cells, cancer cells, and neurons. More and more tools are being developed to target such cells physically using drug or effector conjugated antibodies. These can be single-chain antibodies, classic monovalent antibody fragments (Fab and scFv), and engineered variants such as diabodies, triabodies, minibodies, and single-domain antibodies, or aptamers. Coupling IFN molecules, engineered or not, to such constructs could be a strategy to obtain the unique benefits of these classes of cell signaling proteins, unharnessing their full strength in the complex pathways of immune and host defense systems.

## References

- [1] A. Isaacs, J. Lindenmann, Virus interference. I. The interferon, Proceedings of the Royal Society of London. Series B, Biological Sciences 147 (1957) 258–267.
- [2] S. Pestka, J.A. Langer, K.C. Zoon, C.E. Samuel, Interferons and their actions, Annual Review of Biochemistry 56 (1987) 727–777.
- [3] T. Taniguchi, Y. Fujii-Kuriyama, M. Muramatsu, Molecular cloning of human interferon cDNA, Proceedings of the National Academy of Sciences USA 77 (1980) 4003–4006.
- [4] S. Nagata, N. Mantei, C. Weissmann, The structure of one of the eight or more distinct chromosomal genes for human interferon-alpha, Nature 287 (1980) 401–408.
- [5] J. Weissenbach, Y. Chernajovsky, M. Zeevi, L. Shulman, H. Soreq, U. Nir, D. Wallach, M. Perricaudet, P. Tiollais, M. Revel, Two interferon mRNAs in human fibroblasts: in vitro translation and *Escherichia coli* cloning studies, Proceedings of the National Academy of Sciences USA 77 (1980) 7152–7156.
- [6] R. Derynck, J. Content, E. DeClercq, G. Volckaert, J. Tavernier, R. Devos, W. Fiers, Isolation and structure of a human fibroblast interferon gene, Nature 285 (1980) 542–547.
- [7] W. Degrave, R. Derynck, J. Tavernier, G. Haegeman, W. Fiers, Nucleotide sequence of the chromosomal gene for human fibroblast (beta 1) interferon and of the flanking regions, Gene 14 (1981) 137–143.
- [8] R. Devos, H. Cheroutre, Y. Taya, W. Degrave, H. Van Heuverswyn, W. Fiers, Molecular cloning of human immune interferon cDNA and its expression in eukaryotic cells, Nucleic Acids Research 10 (1982) 2487–2501.
- [9] P.W. Gray, D.V. Goeddel, Structure of the human immune interferon gene, Nature 298 (1982) 859-863.
- [10] Y. Taya, R. Devos, J. Tavernier, H. Cheroutre, G. Engler, W. Fiers, Cloning and structure of the human immune interferon-gamma chromosomal gene, EMBO Journal 1 (1982) 953–958.
- [11] R. Derynck, E. Remaut, E. Saman, P. Stanssens, E. De Clercq, J. Content, W. Fiers, Expression of human fibroblast interferon gene in *Escherichia coli*, Nature 287 (1980) 193–197.
- [12] S. Nagata, H. Taira, A. Hall, L. Johnsrud, M. Streuli, J. Ecsödi, W. Boll, K. Cantell, C. Weissmann, Synthesis in *E. coli* of a polypeptide with human leukocyte interferon activity, Nature 284 (1980) 316–320.
- [13] W. Fiers, E. Remaut, R. Devos, H. Cheroutre, R. Contreras, D. Gheysen, W. Degrave, P. Stanssens, J. Tavernier, Y. Taya, J. Content, The human fibroblast and human immune interferon genes and their expression in homologous and heterologous cells, Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences 299 (1982) 29–38.
- [14] E. Remaut, P. Stanssens, W. Fiers, Inducible high level synthesis of mature human fibroblast interferon in *Escherichia coli*, Nucleic Acids Research 11 (1983) 4677–4688.
- [15] S.J. Scahill, R. Devos, J. Van der Heyden, W. Fiers, Expression and characterization of the product of a human immune interferon cDNA gene in Chinese hamster ovary cells, Proceedings of the National Academy of Sciences USA 80 (1983) 4654–4658.
- [16] R. Devos, C. Opsomer, S.J. Scahill, J. Van der Heyden, W. Fiers, Purification of recombinant glycosylated human gamma interferon expressed in transformed Chinese hamster ovary cells, Journal of Interferon & Cytokine Research 4 (1984) 461–468.
- [17] G. Simons, E. Remaut, B. Allet, R. Devos, W. Fiers, High-level expression of human interferon gamma in *Escherichia coli* under control of the pL promoter of bacteriophage lambda, Gene 28 (1984) 55–64.

- [18] W. Degrave, J. Tavernier, F. Duerinck, G. Plaetinck, R. Devos, W. Fiers, Cloning and structure of the human interleukin 2 chromosomal gene, EMBO Journal 2 (1983) 2349–2353.
- [19] E.C. Borden, G.C. Sen, G. Uze, R.H. Silverman, R.M. Ransohoff, G.R. Foster, G.R. Stark, Interferons at age 50: past, current and future impact on biomedicine, Nature Reviews Drug Discovery 6 (2007) 975–990.
- [20] M.H. Heim, 25 years of interferon-based treatment of chronic hepatitis C: an epoch coming to an end, Nature Reviews Immunology 13 (2013) 535–542.
- [21] K. Oritani, Y. Tomiyama, Interferon-zeta/limitin: novel type I interferon that displays a narrow range of biological activity, International Journal of Hematology 80 (2004) 325–331.
- [22] S. Pestka, The interferons: 50 years after their discovery, there is much more to learn, The Journal of Biological Chemistry 282 (28) (2007) 20047–20051.
- [23] G.C. Sen, P. Lengyel, The interferon system. A bird's eye view of its biochemistry, The Journal of Biological Chemistry 267 (1992) 5017–5020.
- [24] S. Pestka, C.D. Krause, M.R. Walter, Interferons, interferon-like cytokines, and their receptors, Immunological Reviews 202 (2004) 8–32.
- [25] Y.J. Liu, IPC: professional type 1 interferon-producing cells and plasmacytoid dendritic cell precursors, Annual Review of Immunology 23 (2005) 275–306.
- [26] S.A. Samarajiwa, W. Wilson, P.J. Hertzog, Type I interferons: genetics and structure, in: A. Meager (Ed.), The Interferons: Characterization and Application, Wiley-VCH Verlag GmbH & Co, 2006, pp. 3–34.
- [27] N.A. De Weerd, S.A. Samarajiwa, P.J. Hertzog, Type I interferon receptors: biochemistry and biological functions, The Journal of Biological Chemistry 282 (2007) 20053–20057.
- [28] M.O. Díaz, The human type I interferon gene cluster, Seminars in Virology 6 (1995) 143–149.
- [29] M.P. Hardya, C.M. Owczarek, L.S. Jermiin, M. Ejdebäck, P.J. Hertzog, Characterization of the type I interferon locus and identification of novel genes, Genomics 84 (2004) 2331–2345.
- [30] P. Genin, A. Vaccaro, A. Civas, The role of differential expression of human interferon a genes in antiviral immunity, Cytokine and Growth Factor Reviews 20 (2009) 283–295.
- [31] T.A. Nyman, N. Kalkkinen, H. Tolo, J. Helin, Structural characterization of N-linked and O-linked oligosaccharides derived from interferon-alpha2b and interferon-alpha14c produced by Sendaivirus-induced human peripheral blood leukocytes, European Journal of Biochemistry 253 (1998) 485–493.
- [32] G.D. Kalliolias, L.B. Ivashkiv, Overview of the biology of type I interferons, Arthritis Research & Therapy 12 (Suppl. 1) (2010) S1.
- [33] L. Zitvogel, L. Galluzzi, O. Kepp, M.J. Smyth, G. Kroemer, Type I interferons in anticancer immunity, Nature Reviews Immunology 15 (2015) 405–414.
- [34] L.C. Platanias, Mechanisms of type-I- and type-II-interferon-mediated signaling, Nature Reviews Immunology 5 (2005) 375–386.
- [35] P. Hermant, T. Michiels, Interferon- $\lambda$  in the context of viral infections: production, response and therapeutic implications, Journal of Innate Immunity 6 (2014) 563–574.
- [36] S.A. Samarajiwa, S. Forster, K. Auchettl, P.J. Hertzog, Interferome: the database of interferon regulated genes, Nucleic Acids Research 37 (2009) D852–D857.
- [37] L.B. Ivashkiv, L.T. Donlin, Regulation of type I interferon responses, Nature Reviews Immunology 14 (2014) 36–49.
- [38] I.M. Pedersen, G. Cheng, S. Wieland, S. Volinia, C.M. Croce, F.V. Chisari, M. David, Interferon modulation of cellular microRNAs as an antiviral mechanism, Nature 449 (2007) 919–922.
- [39] A. Takaoka, Z. Wang, M.K. Choi, H. Yanai, H. Negishi, T. Ban, Y. Lu, M. Miyagishi, T. Kodama, K. Honda, Y. Ohba, T. Taniguchi, DAI (DLM-1/ZBP1) is a cytosolic DNA sensor and an activator of innate immune response, Nature 448 (2007) 501–505.
- [40] M.J. De Veer, M. Holko, M. Frevel, E. Walker, S. Der, J.M. Paranjape, R.H. Silverman, B.R. Williams, Functional classification of interferon-stimulated genes identified using microarrays, Journal of Leukocyte Biology 69 (2001) 912–920.
- [41] A. Takaoka, S. Hayakawa, H. Yanai, D. Stoiber, H. Negishi, H. Kikuchi, S. Sasaki, K. Imai, T. Shibue, K. Honda, T. Taniguchi, Integration of interferon-alpha/beta signaling to p53 responses in tumor suppression and antiviral defense, Nature 424 (2003) 516–523.
- [42] V. Fensterl, G.C. Sen, Interferons and viral infections, BioFactors 35 (2009) 14-20.
- [43] N.A. El-Baky, E.M. Redwan, Therapeutic alpha-interferons protein: structure, production, and biosimilar, Preparative Biochemistry and Biotechnology 45 (2015) 109–127.
- [44] L. Runkel, L. Pfeffer, M. Lewerenz, D. Monneron, C.H. Yang, A. Murti, S. Pellegrini, S. Goelz, G. Uzé, K. Mogensen, Differences in activity between alpha and beta type I interferons explored by mutational analysis, The Journal of Biological Chemistry 273 (1998) 8003–8008.
- [45] R.M. Roberts, L.M. Liu, A. Alexenko, New and atypical families of type I interferons in mammals: comparative functions, structures, and evolutionary relationships, Progress in Nucleic Acid Research and Molecular Biology 56 (1997) 287–325.
- [46] R. Wetzel, Assignment of the disulphide bonds of leukocyte interferon, Nature 289 (1981) 606-607.
- [47] D.L. Miller, H.F. Kung, S. Pestka, Crystallization of recombinant human leukocyte interferon-A, Science 215 (1982) 689–690.
- [48] S. Matsuda, T. Senda, S. Itoh, G. Kawano, H. Mizuno, Y. Mitsui, New crystal form of recombinant murine interferon-beta, The Journal of Biological Chemistry 264 (1989) 13381–13382.
- [49] M. Karpusas, A. Whitty, L. Runkel, P. Hochman, The structure of human interferon-beta: implications for activity, Cellular and Molecular Life Sciences 54 (1998) 1203–1216.
- [50] R. Radhakrishnan, L.J. Walter, A. Hruza, P. Reichert, P.P. Trotta, T.L. Nagabhushan, M.R. Walter, Zinc mediated dimer of human interferon-alpha(2b) revealed by X-ray crystallography, Structure 4 (1996) 1453–1463.
- [51] N.A. El-Baky, V.N. Uversky, E.M. Redwan, Human consensus interferons: Bridging the natural and artificial cytokines with intrinsic disorder, Cytokine and Growth Factor Reviews 6 (2015) S1359–S6101.
- [52] S.E. Ealick, W.J. Cook, S. Vijay-Kumar, M. Carson, T.L. Nagabhushan, P.P. Trotta, C.E. Bugg, Threedimensional structure of recombinant human interferon-gamma, Science 252 (1991) 698–702.
- [53] V. Navratil, B. de Chassey, L. Meyniel, F. Pradezynski, P. André, C. Rabourdin-Combe, V. Lotteau, Systems-level comparison of protein-protein interactions between viruses and the human type I interferon system network, Journal of Proteome Research 9 (2010) 3527–3536.
- [54] S.H. Seo, E. Hoffmann, R.G. Webster, Lethal H5N1 influenza viruses escape host anti-viral cytokine responses, Nature Medicine 8 (2002) 950–954.
- [55] D. Guha, V. Ayyavoo, Innate immune evasion strategies by human immunodeficiency virus type 1, ISRN AIDS 2013 (2013) 954806.
- [56] R.J. Critchley-Thorne, D.L. Simons, N. Yan, A.K. Miyahira, F.M. Dirbas, D.L. Johnson, S.M. Swetter, R.W. Carlson, G.A. Fisher, A. Koong, S. Holmes, P.P. Lee, Impaired interferon signaling is a common immune defect in human cancer, Proceedings of the National Academy of Sciences USA 106 (2009) 9010–9015.
- [57] J.P. Messina, I. Humphreys, A. Flaxman, A. Brown, G.S. Cooke, O.G. Pybus, E. Barnes, Global distribution and prevalence of hepatitis C virus genotypes, Hepatology 61 (2015) 77–87.

- [58] H.S. Te, D.M. Jensen, Epidemiology of hepatitis B and C viruses: a global overview, Clinical Liver Disease 14 (2010) 1–21.
- [59] S. Chevaliez, J.M. Pawlotsky, Interferon-based therapy of hepatitis C, Advanced Drug Delivery Reviews 59 (2007) 1222–1241.
- [60] L.H. Calabrese, P.P. Cacoub, For patients with rheumatic disease and hepatitis C infection: the end of interferon, RMD Open 1 (2015) e000008.
- [61] J.U. Gutterman, Cytokine therapeutics lessons from interferon-alpha, Proceedings of the National Academy of Sciences USA 91 (1994) 1198–1205.
- [62] M.W. Fried, M.L. Shiffman, K.R. Reddy, C. Smith, G. Marinos, F.L. Goncales Jr., D. Haussinger, M. Diago, G. Carosi, D. Dhumeaux, A. Craxi, A. Lin, J. Hoffman, J. Yu, Peginterferon alfa-2a plus ribavirin for chronic hepatitis C virus infection, The New England Journal of Medicine 347 (2002) 975–982.
- [63] I.S. Jamall, S. Yusuf, M. Azhar, S. Jamall, Is pegylated interferon superior to interferon, with ribavarin, in chronic hepatitis C genotypes 2/3? World Journal of Gastroenterology 14 (2008) 6627–6631.
- [64] D. Ge, J. Fellay, A.J. Thompson, J.S. Simon, K.V. Shianna, T.J. Urban, E.L. Heinzen, P. Qiu, A.H. Bertelsen, A.J. Muir, M. Sulkowski, J.G. McHutchison, D.B. Goldstein, Genetic variation in IL28B predicts hepatitis C treatment-induced viral clearance, Nature 461 (2009) 399–401.
- [65] D.L. Thomas, C.L. Thio, M.P. Martin, Y. Qi, D. Ge, C. O'Huigin, J. Kidd, K. Kidd, S.I. Khakoo, G. Alexander, J.J. Goedert, G.D. Kirk, S.M. Donfield, H.R. Rosen, L.H. Tobler, M.P. Busch, J.G. McHutchison, D.B. Goldstein, M. Carrington, Genetic variation in IL28B and spontaneous clearance of hepatitis C virus, Nature 461 (2009) 798–801.
- [66] B.P. Lam, T. Jeffers, Z. Younoszai, Y. Fazel, Z.M. Younossi, The changing landscape of hepatitis C virus therapy: focus on interferon-free treatment, Therapeutic Advances in Gastroenterology 8 (2015) 298–312.
- [67] V. Gerein, E. Rastorguev, J. Gerein, P. Jecker, H. Pfister, Use of interferon-alpha in recurrent respiratory papillomatosis: 20-year follow-up, Annals of Otology, Rhinology & Laryngology 114 (2005) 463–471.
- [68] W.E. Stewart II, The Interferon System, Springer Verlag, Vienna, 1979.
- [69] T. Floros, A.A. Tarhini, Anticancer cytokines: biology and clinical effects of interferon-α2, interleukin (IL)-2, IL-15, IL-21, and IL-12, Seminars in Oncology 42 (2015) 539–548.
- [70] P.M. Thalanayar, S.S. Agarwala, A.A. Tarhini, Melanoma adjuvant therapy, Chinese Clinical Oncology 3 (2014) 26.
- [71] D. Paolicelli, V. Direnzo, M. Trojano, Review of interferon beta-1b in the treatment of early and relapsing multiple sclerosis, Biologics: Targets and Therapy 3 (2009) 369–376.
- [72] B.C. Kieseier, The mechanism of action of interferon- $\beta$  in relapsing multiple sclerosis, CNS Drugs 25 (2011) 491–502.
- [73] W.B. Veldhuis, S. Floris, P.H. van der Meide, I.M. Vos, H.E. de Vries, C.D. Dijkstra, P.R. Bär, K. Nicolay, Interferon-beta prevents cytokine-induced neutrophil infiltration and attenuates blood-brain barrier disruption, Journal of Cerebral Blood Flow & Metabolism 23 (2003) 1060–1069.
- [74] L. Runkel, W. Meier, R.B. Pepinsky, M. Karpusas, A. Whitty, K. Kimball, M. Brickelmaier, C. Muldowney, W. Jones, S.E. Goelz, Structural and functional differences between glycosylated and non-glycosylated forms of human interferon-beta (IFN-beta), Pharmaceutical Research 15 (1998) 641–649.
- [75] D.K. Wijesundara, Y. Xi, C. Ranasinghe, Unraveling the convoluted biological roles of type I interferons in infection and immunity: a way forward for therapeutics and vaccine design, Frontiers in Immunology 5 (2014) 412.

- [76] J.R. Schoenborn, C.B. Wilson, Regulation of interferon-gamma during innate and adaptive immune responses, Advances in Immunology 96 (2007) 41–101.
- [77] F.W. Peng, Z.J. Duan, L.S. Zheng, Z.P. Xie, H.C. Gao, H. Zhang, W.P. Li, Y.D. Hou, Purification of recombinant human interferon-epsilon and oligonucleotide microarray analysis of interferonepsilon-regulated genes, Protein Expression and Purification 53 (2007) 356–362.
- [78] K.Y. Fung, N.E. Mangan, H. Cumming, J.C. Horvat, J.R. Mayall, S.A. Stifter, N. De Weerd, L.C. Roisman, J. Rossjohn, S.A. Robertson, J.E. Schjenken, B. Parker, C.E. Gargett, H.P. Nguyen, D.J. Carr, P.M. Hansbro, P.J. Hertzog, Interferon ε protects the female reproductive tract from viral and bacterial infection, Science 339 (2013) 1088–1092.
- [79] P. Hermant, C. Francius, F. Clotman, T. Michiels, IFN-ε is constitutively expressed by cells of the reproductive tract and is inefficiently secreted by fibroblasts and cell lines, PLoS One 8 (2013) e71320.
- [80] D.W. LaFleur, B. Nardelli, T. Tsareva, D. Mather, P. Feng, M. Semenuk, K. Taylor, M. Buergin, D. Chinchilla, V. Roshke, G. Chen, S.M. Ruben, P.M. Pitha, T.A. Coleman, P.A. Moore, Interferon-kappa, a novel type I interferon expressed in human keratinocytes, The Journal of Biological Chemistry 276 (2001) 39765–39771.
- [81] B. Nardelli, L. Zaritskaya, M. Semenuk, Y.H. Cho, D.W. LaFleur, D. Shah, S. Ullrich, G. Girolomoni, C. Albanesi, P.A. Moore, Regulatory effect of IFN-kappa, a novel type I IFN, on cytokine production by cells of the innate immune system, The Journal of Immunology 169 (2002) 4822–4830.
- [82] J. Bekisz, H. Schmeisser, J. Hernandez, N.D. Goldman, K.C. Zoon, Human interferons alpha, beta and omega, Growth Factors 22 (2005) 243–251.
- [83] K. Oritani, Y. Kanakura, IFN-zeta/limitin: a member of type I IFN with mild lymphomyelosuppression, Journal of Cellular and Molecular Medicine 9 (2005) 244–254.
- [84] M. Cochet, D. Vaiman, F. Lefèvre, Novel interferon delta genes in mammals: cloning of one gene from the sheep, two genes expressed by the horse conceptus and discovery of related sequences in several taxa by genomic database screening, Gene 433 (2009) 88–99.
- [85] J.L. Martal, N.M. Chene, L.P. Huynh, R.M. L'Haridon, P.B. Reinaud, M.W. Guillomot, M.A. Charlier, S.Y. Charpigny, IFN-tau: a novel subtype I IFN1. Structural characteristics, nonubiquitous expression, structure-function relationships, a pregnancy hormonal embryonic signal and crossspecies therapeutic potentialities, Biochimie 80 (1998) 755–777.
- [86] F.W. Bazer, W. Ying, X. Wang, K.A. Dunlap, B. Zhou, G.A. Johnson, G. Wu, The many faces of interferon tau, Amino Acids 47 (2015) 449–460.
- [87] J. Vilcek, Novel interferons, Nature Immunology 4 (2003) 8-9.
- [88] Y.F. Lu, D.B. Goldstein, T.J. Urban, S.S. Bradrick, Interferon-λ4 is a cell-autonomous type III interferon associated with pre-treatment hepatitis C virus burden, Virology 476 (2015) 334–340.
- [89] T.A. Nyman, H. Tolo, J. Parkkinen, N. Kalkkinen, Identification of nine interferon-alpha subtypes produced by Sendai virus-induced human peripheral blood leucocytes, Biochemical Journal 329 (1998) 295–302.
- [90] D.R. Strayer, W.A. Carter, Recombinant and natural human interferons: analysis of the incidence and clinical impact of neutralizing antibodies, Journal of Interferon & Cytokine Research 32 (2012) 95–102.
- [91] S.M. Singh, A.K. Panda, Solubilization and refolding of bacterial inclusion body proteins, Journal of Bioscience and Bioengineering 99 (2005) 303–310.
- [92] A.P.J. Middelberg, Preparative protein refolding, Trends in Biotechnology 20 (2002) 437-443.
- [93] R.R. Burgess, Refolding solubilized inclusion body proteins, Methods in Enzymology 463 (2009) 259–282.

- [94] K.R. Babu, S. Swaminathan, S. Marten, N. Khanna, U. Rinas, Production of interferon-alpha in high cell density cultures of recombinant *Escherichia coli* and its single step purification from refolded inclusion body proteins, Applied Microbiology and Biotechnology 53 (2000) 655–660.
- [95] H.K. Lim, K.H. Jung, D.H. Park, S.I. Chung, Production characteristics of interferon-alpha using an L-arabinose promoter system in a high-cell-density culture, Applied Microbiology and Biotechnology 53 (2000) 201–208.
- [96] P. Srivastava, P. Bhattacharaya, G. Pandey, K.J. Mukherjee, Overexpression and purification of recombinant human interferon alpha2b in *Escherichia coli*, Protein Expression and Purification 41 (2005) 313–322.
- [97] N. Acosta-Rivero, J.C. Sanchez, J. Morales, Improvement of human interferon HUIFNalpha2 and HCV core protein expression levels in *Escherichia coli* but not of HUIFNalpha8 by using the tRNA(AGA/AGG), Biochemical and Biophysical Research Communications 296 (2002) 1303–1309.
- [98] B.K. Chung, D.Y. Lee, Computational codon optimization of synthetic gene for protein expression, BMC Systems Biology 6 (2012) 134.
- [99] H. Kakeshita, Y. Kageyama, K. Endo, M. Tohata, K. Ara, K. Ozaki, K. Nakamura, Secretion of biologically-active human interferon-β by *Bacillus subtilis*, Biotechnology Letters 33 (2011) 1847–1852.
- [100] J.A. Rojas Contreras, M. Pedraza-Reyes, L.G. Ordoñez, N.U. Estrada, A.P. Barba de la Rosa, A. De León-Rodríguez, Replicative and integrative plasmids for production of human interferon gamma in *Bacillus subtilis*, Plasmid 64 (2010) 170–176.
- [101] M.A. Khan, N. Hassan, N. Ahmad, M.I. Khan, A.U. Zafar, F. Khan, T. Husnain, Studies to analyze the relationship between IFNα2b gene dosage and its expression, using a *Pichia pastoris*-based expression system, Yeast 31 (2014) 13–28.
- [102] D. Wang, H. Ren, J.W. Xu, P.D. Sun, X.D. Fang, Expression, purification and characterization of human interferon-γ in Pichia pastoris, Molecular Medicine Reports 9 (2014) 715–719.
- [103] B.K. Chung, F.N. Yusufi, Mariati, Y. Yang, D.Y. Lee, Enhanced expression of codon optimized interferon gamma in CHO cells, Journal of Biotechnology 167 (2013) 326–333.
- [104] M. Loignon, S. Perret, J. Kelly, D. Boulais, B. Cass, L. Bisson, F. Afkhamizarreh, Y. Durocher, Stable high volumetric production of glycosylated human recombinant IFNalpha2b in HEK293 cells, BMC Biotechnology 8 (2008) 65.
- [105] W.S. Chen, O.B. Villaflores, T.R. Jinn, M.T. Chan, Y.C. Chang, T.Y. Wu, Expression of recombinant human interferon-γ with antiviral activity in the bi-cistronic baculovirus-insect/larval system, Bioscience, Biotechnology, and Biochemistry 75 (2011) 1342–1348.
- [106] N.A. El-Baky, S.H. Omar, E.M. Redwan, The anti-cancer activity of human consensus interferonalpha synthesized in cell-free system, Protein Expression and Purification 80 (2011) 61–67.
- [107] I. Dmitrij, D.I. Bairamashvili, M.L. Rabinovich, Russia through the prism of the world biopharmaceutical market, Biotechnology Journal 2 (2007) 1–17.
- [108] B. Podobnik, B. Helk, V. Smilović, Š. Škrajnar, K. Fidler, S. Jevševar, A. Godwin, P. Williams, Conjugation of PolyPEG to interferon alpha extends serum half-life while maintaining low viscosity of the conjugate, Bioconjugate Chemistry 26 (2015) 452–459.
- [109] X. Hu, A. Seddighzadeh, S. Stecher, Y. Zhu, J. Goyal, M. Matson, T. Marbury, W. Smith, I. Nestorov, S. Hung, Pharmacokinetics, pharmacodynamics, and safety of peginterferon beta-1a in subjects with normal or impaired renal function, The Journal of Clinical Pharmacology 55 (2015) 179–188.
- [110] O.N. Ozes, Z. Reiter, S. Klein, L.M. Blatt, M.W. Taylor, A comparison of interferon-Con1 with natural recombinant interferons-alpha: antiviral, antiproliferative, and natural killer-inducing activities, Journal of Interferon & Cytokine Research 12 (1992) 55–59.

- [111] L.M. Blatt, J.M. Davis, S.B. Klein, M.W. Taylor, The biological activity and molecular characterization of a novel synthetic interferon-alpha species, consensus-interferon, Journal of Interferon & Cytokine Research 16 (1996) 489–499.
- [112] M.J. Tong, K.R. Reddy, W.M. Lee, P.J. Pockros, J.C. Hoefs, E.B. Keeffe, F.B. Hollinger, E.J. Hathcote, H. White, R.T. Foust, D.M. Jensen, E.L. Krawitt, H. Fromm, M. Black, L.M. Blatt, M. Klein, J. Lubina, Treatment of chronic hepatitis C with consensus interferon: a multicenter, randomized, controlled trial, Hepatology 26 (1997) 747–754.
- [113] E.N. Fish, S.A. Harrison, T. Hassanein, The role of consensus interferon in the current treatment of chronic hepatitis C viral infection, Gastroenterology and Hepatology 4 (2008) 1–12.
- [114] T.H. Witthöft, Review of consensus interferon in the treatment of chronic hepatitis C, Biologics 2 (2008) 635–643.
- [115] FDA (2015). http://www.fda.gov/downloads/Drugs/DevelopmentApprovalProcess/HowDrugsare DevelopedandApproved/ApprovalApplications/TherapeuticBiologicApplications/ucm093414.pdf.
- [116] Fraunhofer (2015). http://www.igb.fraunhofer.de/en/competences/molecular-biotechnology/ therapeutic-proteins/interferons\_pharmaproteins.html.



# Monoclonal Antibodies

#### S. Ndoja, H. Lima\* MCMASTER UNIVERSITY, HAMILTON, ONTARIO, CANADA

### 4.1 Introduction to the Immune System

#### 4.1.1 Origins of the Immune System

The evolution of multicellularity in organisms can explain the origin and function of the immune system. The interaction of different organisms or the cohesion of cells of unicellular beings resulted in multicellular organisms. The evolutionary process of cell recognition of the same origin and interdependence with others from a different origin has been happening since the first microorganisms got together to form the first multinuclear organism. In addition to larger size, multicellularity increased biological complexity through the formation of new biological structures [1].

It may seem as if the immune system is active only in a disease state. In reality, it is always active [2]. Therefore, the absence of disease does not mean a lack of interaction with microorganisms. In a homeostatic condition, the immune system constantly interacts with other cells such as resident microbiological flora and organisms from the external world.

Studies have revealed the unexpected importance of host and microorganism interactions for evolution. Resident microbes in the human body, which form the microbiota, can be viewed as an integrated system [3]. Resident microbes are continuously interacting with the immune system, and depending on their location, their compositional diversity, or the lack of it, they can affect the health state of the organism. This interaction between a host and microbes has prompted the possibility of considering the microbiome as another organ [4]. The discovery of the importance of the microbiome for human health was a crucial advancement in evolutionary biology in the first decade of the 21st century [5]. Advances in genomic mapping allowed insight into microbiome—host interactions, information that was previously unavailable to immunologists. With this new information, current understanding of the role of microbes is changing; they are not assumed not be pathogens or symbionts, but cohabitants of the same genetic milieu of the host. Each organism is the result of a combination of germs and host DNA integrated into a complex system [6]. During evolutionary history,

<sup>&</sup>lt;sup>\*</sup>Corresponding Author.

microbes and their hosts have cohabited, with natural selection acting upon the entire superorganism to drive evolution [7].

The current understanding of host-microbe interactions engenders the possibility that infection and immunity are an important factor in organismal evolution. This hypothesis favors the paradigm that there is a need to preserve the coexistence of organisms in the same biological structure for lateral gene exchange among microbes. Thus, infection is necessary for both microbe and host evolution. The immune system can thus be considered the metagenomic immune system (MIS).

The idea of the MIS is a new interpretation of immunity. It is a different perception of the basics of the immune system as part of a new evolutionary theory. The evolution of biological systems by integrating genetic material is an emergent theory that takes into account recent scientific facts. The data show that life results by assimilating one genomic system into other genomic arrangements. Therefore, genetic exchange among organisms is a priority instead of a competitive process. This is a satisfactory interpretation for understanding biological diversity by evolution and the immune system origin. Therefore, the main driving force of immune system development is the process of interaction among biological beings. This new paradigm includes the interpretations that the immune response of one organism does not react, but rather facilitates interrelation with other organisms.

#### 4.1.2 Innate and Adaptive Immunity

Evolutionary processes have led to the generation of two types of immune processes: innate and adaptive [8,9]. The difference between the innate and adaptive immunity is not because the cells have a different role in each system, but rather how they recognize and interact with microorganisms and other cells. Whereas the adaptive immunity recognizes an antigen by its specific epitope, the innate immunity relies on evolution-arily conserved targets such as the pathogen-associated molecular pattern (PAMP). They are recognized by the pattern recognition receptors. PAMPs are found in both pathogenic and nonpathogenic microorganisms. Therefore, the term "PAMPs" is instead more aptly characterized as microbe-associated molecular patterns. The innate immune system serves functions beyond recognizing and clearing pathogens. The innate immune system also serves as a clearing mechanism for damaged cells that are no longer viable, as well as transformed cells that do not operate within their normal parameters. Unique molecular patterns, can be recognized by the innate immune system. The innate immune system can recognize approximately 10<sup>3</sup> molecular patterns.

Although the development of the adaptive immune system has led to an increase in genomic complexity of vertebrates, this complexity appears to be irrelevant in terms of the species survival. Invertebrates did not develop adaptive immunity and are also able to survive the pathogen interactions successfully with only the innate immune system [10].

Whereas the targets that the innate immune system recognizes are common in any microorganism structure; such as double-stranded DNA, the adaptive immune system

"learns" from exposure to microorganisms and can form "memory" to mount a more adept immune response with subsequent exposures. The main cells in the adaptive immune system are the T-cells and B-cells. Both T-cell and B-cell receptors have the ability to recognize a particular target on a microorganism, typically a segment of an extracellular protein. The segment that is recognized by B-cells and T-cells is known as an epitope.

The primary activity of B-cells is the ability to secrete their receptor in a soluble form, producing antibodies that are specific to epitopes on antigens. Once tagged, antibodies are able to recruit other immune cells to carry out a subsequent function. Antibodies act as molecular tags, directing other immune cells to destroy the tagged substance. Antibodies are Y-shaped proteins with a variable and a constant region (Fig. 4.1). The variable region, as the name suggests, varies between antibodies, and it is where antibody—antigen interactions occur. In the variable region, the antibody specificity is determined.

Antibodies can be divided into five classes: immunoglobulin (Ig)M, IgD, IgG, IgA, and IgE, which are determined by their constant region. The IgG class of Igs contains four subtypes, named IgG1–4 [11]. The constant regions interact with fragment crystallizable region receptors (FcR), which have a high affinity for each particular class. FcRs are named in relation to the class of antibodies for which they have the highest affinity:  $Fc\alpha R$ ,  $Fc\gamma R$ ,  $Fc\delta R$ , and  $FcR\mu R$ , respectively. Different categories of Igs are thought to have various functions. IgM and IgG3 have a strong affinity for complement fixation. IgG1 can cross the placental barrier and has mild complement fixation ability. IgE demonstrates high affinity to mast cell and basophils [12]. The specificity of an antibody in tagging proteins has made it an invaluable tool in the field of biotechnology.



**FIGURE 4.1** An immunoglobulin protein is composed of two proteins called the light and heavy chains, which are joined by disulfide bonds. Each heavy and light chain has a constant and a variable region. The antigen recognition area is found in the variable region.

# 4.2 Monoclonal Antibody Development

The modern era of immunology began in 1890 with the discovery of antibodies as key components of protective immunity. However, it was not until the early 20th century when the use of antibodies was brought to fruition by Paul Ehrlich, one of the fathers of immunology, in his "magic bullet" theory. Ehrlich rationalized that because the interactions of antigen and antibody were already known, it stood to reason that a drug could be bound to a distinct carrier exhibiting pharmacologic activity only in the target tissue. Thus the undesirable effects of its action in other tissues would be greatly reduced. As an added benefit, the efficacy of the drugs could be increased, which would decrease the necessary dose to achieve therapeutic effects.

Although antibodies seem like an ideal candidate, the immune response to any antigen is polyclonal. The polyclonal nature of the immune response makes it difficult to purify antibodies with a desired specificity in mass quantities. It was not until 1975, with the discovery of cell hybridization techniques by J. F. Georges Kohler and Cesar Milstein, who created hybridomas, that the use of mAbs could be mass produced. The creation of hybridomas involves extracting splenocytes of mice immunized with a desired antigens and fusing them with myeloma cells. This technique makes use of an agent, such as polyethylene glycol, which facilitates the fusion of the adjacent plasma membranes. Myeloma cells are screened for two key features: They do not produce their own antibody and lack the hypoxanthine-guanine phosphoribosyl transferase (HGPRT) gene. HGPRT is used in nucleotide synthesis and is essential to cell division. The HPGRT gene has a redundancy because there is a de novo synthesis route through dihydrofolate reductase (DHFR). By using the molecule aminopterin, which inhibits DHFR, the cell is unable to divide without supplementation.

In the creation of hybridomas, three cell populations are present: myeloma cells, hybridomas, and splenocytes. When these cell cultures are grown in hypoxanthine, aminopterin, and thymidine (HAT) culture medium, the myeloma cells are unable to divide because there is no HPGRT, and they die. Splenocytes are unable to grow indefinitely by the average finite life, and eventually die off. This results in only hybridomas remaining. Hybrids can grow indefinitely and begin to multiply, with rapid formation of colonies (Fig. 4.2). The hybridoma cells are cloned and supernatants are tested for antibody production. Extensive tests are conducted to ensure the specificity of the antibodies and are seeded from a single cell colony so that cells in one medium are from one clone and thus produce only one type of antibody. They can be produced large amounts of antibody in vitro (in culture medium) or in vivo (in ascites in mice). This discovery earned Kohler and Milstein the Nobel prize for Medicine and Physiology in 1984.

With the invention of hybridomas, it became possible to mass produce large amounts of mAbs for clinical use. One could produce mAbs specific to a microorganism or tumor specific antigen, to interfere with immune system capacity. However, because these antibodies originated from a mouse source, the human immune system recognized



**FIGURE 4.2** Breakdown of myeloma creation. Step 1: a murine carrier is immunized and splenocytes are extracted. Step 2: The murine splenocytes are fused with the splenocytes. Step 3: The hybridoma cells are cultured in HAT medium and the hybridoma producing the desired antibody is isolated. Step 4: The isolated hybridoma cell colony is expanded.

these antibodies as foreign and produced human anti-mouse antibodies that neutralized the efficacy of mAbs. With the advent of genetic engineering, it became possible to replace the murine antibody constant region gene for the same human genomic sequence. These "chimeric" antibodies are more tolerable to the human immune system while providing the same specificity [13]. However, to minimize immune activation, the less murine the antibody has, the more efficacious it is. Further technologies have been developed to create mAbs that are more human, in which part of the variable region is human; that is, they are humanized, or the entire antibody is human: it is a fully human antibody [14] (Fig. 4.3). The ability to mass produce mAbs with a desired specificity has revolutionized both the clinical and research fields.

# 4.3 Monoclonal Antibody Nomenclature

The seemingly infinite possible different targets, different classes, and different sources of antibodies require a naming convention for mAbs. Conventions for naming antibodies and antibody fragments have been established in the United States. Four or more syllables in each name reveal much information about each product. mAb nomenclature includes the suffix "mab", for monoclonal antibody. One or two letters preceding the "mab" suffix provides much information about the source of the antibody: "u" for human source, "o" for mouse, "a" for rat, "i" for primate, "xi" for chimeric, and "zu" for humanized.



**FIGURE 4.3** Murine antibodies are completely foreign and unmodified (A). In a chimeric antibody, the constant region is modified to be human (B). Humanized antibodies have part of the variable region edited to be of human origin (C). A fully human antibody contains constant and variable regions of human origin (D).

The target of the antibody or the indication of disease is designated by an additional internal syllable in the jargon of the governing body of nomenclature, called "infix," which is the prefix. These include syllables: "ba" for bacterial, "li" for the immuno-modulator, "mel" for melanoma, "pr (a)" for the prostate tumor, "gov" for the tumor gonad (ovary), "ci (r)" for cardiovascular, "vi" for viral, "col" for cervix tumor, "Sea" for breast tumor, and "got" for gonad tumors (testicle); however, all of these have been combined under the infix "tu" for tumor. The prefix is preceded by a compatible syllable that designates the end of the product name. With the nomenclature convention, words such as biciromab, sevirumab, edobacumab, and nebacumab start to make sense.

If another molecule is added to the antibody, a separate word is added to the name. If a toxin is added, the "tox" should be part of the second word. For example, "aritox" refers to the toxin ricin, and "sudotox," the exotoxin of *Pseudomonas aeruginosa*. Pendetide or pentetate words are also used for holding all linker antibodies to a toxin or isotope.

A summary of the nomenclature would be thus: Endings with "mab" are mAbs; endings in "ximab" are monoclonal chimeric and can form neutralizing antibodies; final "zumab" is humanized mAbs and is less able to generate neutralizing antibodies; final "cept" involves fusion with the Fc portion of human IgG1 (Table 4.1).

Prefix	Т	arget infix	Sour	Suffix	
	- ba -	Bacteria	- a -	Rat	
	- ci -	Cardiovascular	- e -	Hamster	
Variable	- fu -	Fungal	- i -	Primate	- mab
	- li -	Immunomodulation	- 0 -	Mouse	
	- <i>OS</i> -	Bone	- u -	Human	
	- vi -	Viral	- xi -	Chimeric	
	- tu -	Tumor	- <i>zu</i> -	Humanized	

 Table 4.1
 Monoclonal Antibody Nomenclature

# 4.4 Monoclonal Antibodies in Research

With their ability to target a desired protein specifically, mAbs have revolutionized the research field by allowing the quantification of protein assays. Through the use of molecular biology techniques, it is possible to conjugate the constant region of antibodies with another product. Fluorescent and colorimetric labels are commonly tagged onto antibodies. Conjugated antibodies, coupled with spectrometers, allow researchers to quantify or confirm the presence of a protein target in a sample. Using mAbs, researchers have developed many techniques.

#### 4.4.1 Enzyme-Linked Immunosorbent Assay

One commonly used technique in both in the laboratory and clinic, is the enzymelinked immunosorbent assay (ELISA). As the name suggests, an enzyme, commonly horseradish peroxidase (HRP), is linked to a specific antibody [15]. The benefit of conjugating HRP is that it can undergo a chemical reaction with substrates such as 2,2'azino-*bis*(3-ethylbenzothiazoline-6-sulphonic acid) and 3,3',5,5'-tetramethylbenzidine to induce a color change. The change in color can be used visually to confirm the presence of a protein or, by using a reference assay and spectrophotometers, quantitatively to measure the concentration of a given protein. Typically the conjugated antibody is a secondary antibody specific to a class of antibodies. This allows for the use of the conjugated antibody in different tests.

A common form of ELISA is the sandwich ELISA. In this model, antibodies unique to the desired antigen, the substrate, are coated on a plate. The substrate is added to the plate and enough time is allotted for it to bind to the antibody. The sample is then washed to remove any unbound substances. A detection antibody is then added, which then binds to the substrate that is bound to antibodies that were coated on the plate (Fig. 4.4). After an allotted time, the plate is washed to remove any unbound antibodies. A secondary antibody specific to the constant region of the detection antibody is then added. The secondary antibody is conjugated with HRP. After any unbound secondary



**FIGURE 4.4** A plate is first coated with the capture antibody (A). The substrate to be detected is added to the plate (B). The detection antibody is added to form a sandwich (C). The conjugated secondary antibody is added (D). Substrate is added to induce a color change (E).

antibodies are washed off, a substrate that induces a color change in HRP is added and the color change is observed.

#### 4.4.2 Western Blot

Similar to ELISAs, western blot (WB) can be used to detect the presence of a protein through the use of antibodies [16]. However, WB protocol can also give insight into the molecular weight of a protein as well as the relative concentration. Cell or tissues are lysed to gain access to cellular proteins. The lysates are then separated by gel electrophoresis according to their molecular weight. After separation the proteins are transferred to a membrane where antibodies are applied to probe the protein of interest.

#### 4.4.3 Flow Cytometry

Antibodies tagged with a fluorescent molecule have uses beyond ELISA. The use of fluorescent-conjugated antibodies is a powerful tool in flow cytometry. Flow cytometry relies on the principle of fluorescence resonance energy transfer (FRET). FRET relies on two closely positioned fluorescent molecules. In this configuration, the proximity of the



FIGURE 4.5 FRET allows for energy transfer of the emission spectrum of the first molecule (green (gray in print versions)) to be absorbed by the second molecule (red (dark gray in print versions)).

two molecules allows the emission spectrum of the first molecule to be the absorption spectrum of the second molecule (Fig. 4.5). As energy is lost between absorption and emission, this results in the emanation gamut of the whole system to be different from the absorption molecule, allowing for easy differentiation between the two spectra.

In flow cytometry, antibodies are conjugated to a fluorophore, which takes advantage of the FRET paradigm. Cell samples are stained for the desired molecular markers and suspended in a medium, which prevents cell clumping. More than one molecular marker can be detected at the same time. However, a limitation exists in the differentiation of spectra owing to spectral overlap. The emission spectrum of the fluorophores has a range, and these spectra can overlap each other, making it difficult to determine which fluorophore is being activated. Once stained, the samples are run through the machine, which uses hydrodynamic focusing to move the cells in a single line (Fig. 4.6). A laser is then used to activate the appropriate fluorophores and a detector measures the emission spectra.

Once cells are in a single line, a laser is used to excite the fluorophores conjugated to the antibodies. The emitted light is then recorded for each cell, and by using multiparametric analysis, a detailed understanding of the cellular composition of a sample can be inferred. Beyond measuring the presence of an antigen, flow cytometry can be used to purify samples. By preparing a fluorescence-conjugated mAb tag to identify the desired cell population, a desired population of cells could be isolated from a heterogeneous sample. In this paradigm, the solution in which the cell samples are suspended is run through the detector. Immediately after the sample is measured it is split into droplets that can contain one more many cells, depending on the size. The droplets are given a charge, and using electrostatic deflection, the sample can be isolated into a desired tube.



FIGURE 4.6 Hydrodynamic focusing allows a sample of cells to be run in a single line for analysis.

#### 4.4.4 Mass Cytometry

Similarly to flow cytometry, mass cytometry allows insight into cell or tissue composition. The main difference between mass cytometry and flow cytometry is that in mass cytometry antibodies are conjugated to heavy metal ions. One advantage of using heavy metal ions is that the output peaks are more discreet and therefore less prone to spectral overlap. This allows more sensitivity in the measurements and allows for a greater number of simultaneous probes.

#### 4.4.5 Immunohistochemistry

Immunohistochemistry uses antibodies to visualize targets on a tissue level [17]. There are various techniques for staining tissues, but the principle is that antibodies specific for an extracellular protein or protein complex are used on a tissue sample. The antibodies contain either radiative or fluorescent conjugates. The stained tissue is then visualized to give a visual representation of the target in the tissue.

# 4.5 Clinical Uses of Monoclonal Antibodies

mAbs can be applied in various areas of medicine. mAbs are capable of recognizing specific molecules present on the cell surface or in body fluids, affecting their function without affecting the remaining cells of the organism. Therefore they can mark or destroy tumor cells, inactivate enzymes, stimulate or turn off receptors, activate and reduce physiologic functions, and stop pathologic processes. mAbs have significant application in oncology, the treatment of autoimmune diseases, and transplant rejection. They can act as markers for diagnostic tests and as scientific research tools in medicine. The following section is devoted to providing an overview of some Food and Drug Administration (FDA)-approved mAbs used to treat diseases in the fields of clinical immunology, dermatology, hematology, and rheumatology (Table 4.2).

#### 4.5.1 Alemtuzumab

Alemtuzumab is a humanized mAb specific for CD52. The CD52 glycoprotein is expressed on the surface of B- and T-lymphocytes and is not present on progenitors, monocytes, and granulocytes [18]. Two forms of Alemtuzumab are approved: murine and humanized. In vitro studies suggest that alemtuzumab mechanisms of action are mediated through antibody-dependent cell-mediated cytotoxicity, the complement pathway, as well as direct induction of apoptosis [19,20]. Whereas in vitro studies have suggested possible mechanisms, in vivo mechanisms for therapeutic effects remain to be elucidated. The specificity of CD52 to B-cells and T-cells as well as the commonality between the two allow CD52 to be an ideal target for decreasing B-cell and T-cell counts when they are too high, such as in malignancies. Alemtuzumab is currently used to treat hematologic malignancies, such as chronic lymphocytic leukemia, cutaneous T-cell

Nonproprietary	Trade			Indication First Approved			
Name	Name	Target	Format	or Reviewed	EU year	US year	
Adalimumab	Humira	TNF	Human IgGl	Rheumatoid arthritis	2003	2002	
Ado-	Kadcyla	HER2	Humanized	Breast cancer	2013	2013	
trastuzumabemtansine		6553	lgGl £	<b>NA</b> 101 1 1 1	2012	2014	
Alemtuzumab	Lemtrada	CD52	Humanized IgGl	Multiple sclerosis	2013	2014	
Alirocumab	Praluent	PCSK9	Human IgGl	High cholesterol	In review	In review	
Basiliximab	Simulect	IL-2R	Chimeric IgGl	Prevention of kidney transplant rejection	1998	1998	
Belimumab	Benlysta	BLyS	Human IgGl	Systemic lupus erythematosus	2011	2011	
Bevacizumab	Avastin	VEGF	Humanized IgGl	Colorectal cancer	2005	2004	
Blinatumomab	Blincyto	CD 19, CD3	Murine β	Acute lymphoblastic leukemia	In review	2014	
Brentuximab	Adcetris	CD30	Chimeric	Hodgkin lymphoma,	2012	2011	
vedotin			lgGl £	systemic anaplastic			
				large cell lymphoma			
Canakinumab	llaris	IL-1β	Human IgGl	Muckle–Wells syndrome	2009	2009	
Catumaxomab	Removab	EPCAM/CD3	Rat/mouse bs mAb	Malignant ascites	2009	NA	
Certolizumab pegol	Cimzia	TNF	Humanized Fab PEG	Crohn disease	2009	2008	
Cetuximab	Erbitux	EGFR	Chimeric IgGl	Colorectal cancer	2004	2004	
Denosumab	Prolia	RANK-L	Human lgG2	Bone loss	2010	2010	
Dinutuximab	Unituxin	GD2	Chimeric IgGl	Neuroblastoma	In review	2015	
Eculizumab	Soliris	C5	Humanized IgG2/4	Paroxysmal nocturnal hemoglobinuria	2007	2007	
Evolocumab	Repatha	PCSK9	Human lgG2	High cholesterol	In review	In review	
Golimumab	Simponi	TNF	Human IgGl	Rheumatoid and psoriatic arthritis, ankylosing spondylitis	2009	2009	
Ibritumomab tiuxetan	Zevalin	CD20	Murine IgGl	Non-Hodgkin lymphoma	2004	2002	
Idarucizumab	(Pending)	Dabigatran	Humanized Fab	Reversal of dabigatran- induced anticoagulation	In review	In review	
Infliximab	Remicade	TNF	Chimeric IgGl	Crohn disease	1999	1998	
Ipilimumab	Yervoy	CTLA-4	Human IgGl	Metastatic melanoma	2011	2011	
Mepolizumab	(Pending)	IL-5	Humanized IgGl	Severe eosinophilic asthma	In review	In review	
Natalizumab	Tysabri	a4 integrin	Humanized IgG4	Multiple sclerosis	2006	2004	
Necitumumab	(Pending)	EGFR	Human IgGI	Non-small cell lung cancer	In review	In review	
Nivolumab	Opdivo	PD1	Human IgG4	Melanoma	In review	2014	
Obinutuzumab	Gazyva	CD20	Humanized IgGl§	Chronic lymphocytic leukemia	2014	2013	

#### Table 4.2 Food and Drug Administration–Approved Monoclonal Antibodies

Nonproprietary	Trade			Indication First Approved		
Name	Name	Target	Format	or Reviewed	EU year	US year
Ofatumumab	Arzerra	CD20	Human IgGl	Chronic lymphocytic leukemia	2010	2009
Omalizumab	Xolair	lsE	Humanized IgGl	Asthma	2005	2003
Palivizumab	Synagis	RSV	Humanized IgGl	Prevention of respiratory syncytial virus infection	1999	1998
Panitumumab	Vectibix	EGFR	Human IgG2	Colorectal cancer	2007	2006
Pembrolizumab	Keytruda	PD1	Humanized IgG4	Melanoma	In review	2014
Pertuzumab	Peijeta	HER2	Humanized IgGl	Breast cancer	2013	2012
Ramucirumab	Cyramza	VEGFR2	Human IgGl	Gastric cancer	2014	2014
Ranibizumab	Lucent is	VEGF	Humanized IgGI	Fab macular degeneration	2007	2006
Raxibacumab	(Pending)	B. anthrasis PA	Human IgGl	Anthrax infection	NA	2012
Rituximab	MabThera, Rituxan	CD20	Chimeric IgGl	Non-Hodgkin lymphoma	1998	1997
Secukinumab	Cosentyx	IL-17a	Human IgGl	Psoriasis	2015	2015
Siltuximab	Sylvant	IL-6	Chimeric IgGl	Castleman disease	2014	2014
Tocilizumab	RoActemra, Actemra	IL-6R	Humanized IgGI	Rheumatoid arthritis	2009	2010
Trastuzumab	Herceptin	HER2	Humanized IgGl	Breast cancer	2000	1998
Ustekinumab	Stelara	IL-12/23	Human IgGl	Psoriasis	2009	2009
Vedolizumab	Entyvio	α4β7 integrin	Humanized IgGl	Ulcerative colitis, Crohn disease	2014	2014
Alemtuzumab	Campath- 1H	CD52	Humanized IgGl	Chronic myeloid leukemia	2001	2001
Daclizumab	Zenapax	IL-2R	Humanized IgGI	Prevention of kidney transplant rejection	1999	1997
Efalizumab	Raptiva	CD11a	Humanized IgGl	Psoriasis	2004	2003
Gemtuzumab ozogamicin	Mylotarg	CD33	Humanized IgG4	Acute myeloid leukemia	NA	2000
Murom on ab-CD 3	Orthoclone Okt3	CD3	Murine IgG2a	Reversal of kidney transplant rejection	1986	1986
Tositumomab-1131	Bexxar	CD20	Murine lgG2a	Non-Hodgkin lymphoma	NA	2003

#### Table 4.2 Food and Drug Administration–Approved Monoclonal Antibodies—cont'd

 $\beta$ , bispecific tandem scFv; *bs*, bispecific; *EGFR*, epidermal growth factor receptor; *EU year*, first European Union approval year; *Fab*, fragment antigen binding; *Ig*, immunoglobulin; *IL*, interleukin; *NA*, not approved; £, immunoconjugate; *PEG*, pegylated; *TNF*, tumor necrosis factor; *US year*, first US approval year; *VEGF*, vascular endothelial growth factor; *WD*, withdrawn from market; §, glycoengineered.

Modified from: J.M. Reichert, Ph.D., Reichert Biotechnology Consulting LLC; table updated March 19, 2015.

lymphoma (CTL), and T-cell lymphomas. The murine (rat) form of Alemtuzumab, marketed as Campath 1G, has been used extensively in allogeneic transplant bone marrow in minimizing rejection.

The main side effects are fever, hypotension, tremors, skin rash, and dyspnea, most common in the first weeks of drug infusion. Significant hematologic toxicity and immunosuppressive clinical features are observed. Alemtuzumab induces profound lymphopenia, with wide range of infections opportunistic observed in patients who are using this drug. There are reports of severe and prolonged myelosuppression and, in some studies, aplasia and hypoplasia bone marrow, even in the recommended dose. The incidence of these complications is greater with increasing doses. Serious autoimmune anemia and thrombocytopenia have also been reported.

#### 4.5.2 Anti-CD20: Rituximab and Ibritumomab Tiuxetan

Rituximab is a chimeric monoclonal IgG1 antibody with specificity for CD20. The variable region of a murine antibody specific to CD20 is fused with a constant human  $\kappa$  light chain and heavy chain. The antibody heavy chain has been selected to induce antibody-dependent cytotoxicity. CD20 is a transmembrane protein located on pre– B-lymphocytes and mature B-lymphocytes, but not on stem cells, pro-B cells, normal plasma cells, or other normal tissues. The CD20 antigen is present in more than 95% B-cell non-Hodgkin lymphomas (NHLs), which makes it an ideal target for identifying cancerous cells. The mechanisms of rituximab in fighting NHL are thought to be through direct signaling, cytotoxicity complement-dependent cellular cytotoxicity (CDCC), and antibody-dependent cytotoxicity (ADCC) [21,22].

Significant indirect results suggested that rituximab has direct immune system– independent cytotoxic effects. Rituximab binding has been theorized to induce apoptotic cellular signals through mechanisms such as lipid rafts, Src-family kinases, and caspases [23–25]. Rituximab immune-mediated cytotoxic effects are carried out through its Fc region. Two main effects are CDCC and ADCC. After antibodies have tagged an antigen, through their Fc region, they induce complement fixation that activates the complement pathway [26]. ADCC uses secondary immune cells to eliminate targets tabbed by antibodies [27]. Cells such as macrophages, natural killer (NK) cells, and eosinophils have Fc receptors that can recognize a tagged cell. Once recognized, they perform downstream functions that destroy the tagged cell.

Rituximab was the first unconjugated antibody approved by FDA in 1997 for the treatment of relapsed lymphomas. In 2002, it became the main anticancer drug in the world. It is indicated for cases of lymphoma and non-Hodgkin low-grade and follicular CD20. It is effective when combined with chemotherapy and has demonstrated promise in long-term response in treating aggressive lymphoma. The combination with cyclophosphamide– hydroxydaunorubicin–oncovin–prednisone (CHOP) therapy increased response rates and survival. Toxic events observed with the combination are not greater or more toxic than those present in therapy isolated with CHOP. The use of rituximab against autoimmune diseases such as rheumatoid arthritis (RA), immune thrombocytopenic purpura, autoimmune hemolytic anemia, systemic lupus erythematosus (SLE), pemphigus vulgaris, and multiple sclerosis (MS) is under investigation. The success of this mAb has prompted the development of conjugated antibodies. Ibritumomab tiuxetan (Zevalin) is a combination of ibritumomab, a CD20-specific mAb, and tiuxetan, a chelator with a radioisotope. The benefit of ibritumomab tiuxetan is that, in addition to direct signaling, CDCC, and ADCC, it is able to kill cells using the radioactive isotope, which improves its ability to eliminate target cells [28,29].

#### 4.5.3 Anti–Tumor Necrosis Factor: Infliximab, Adalimumab, Certolizumab Pegol (Cimzia), and Golimumab

Tumor necrosis factor  $(TNF)\alpha$  is a molecule part of the TNF family. As the name suggests, this class of molecule has the ability to induce cellular apoptosis. TNF $\alpha$ , the canonical TNF molecule, is highly active in host defense [30]. When TNF function is impaired, the host's ability to fight off intracellular bacteria is greatly impaired. TNF $\alpha$  is a potent proinflammatory cytokine with the capacity to recruit immune cells to the site of infection. For circulating immune cells to reach the site of infection, they must cross through the endothelial walls of the circulatory system, a process known as extravasation.  $TNF\alpha$ has been shown to inhibit Fas ligand expression, an important molecule in the inhibition of leukocyte extravasation [31]. The overproduction of  $TNF\alpha$  present in autoinflammatory diseases has been long observed [32] to be specifically associated with T-cell-mediated tissue damage [33]. Blocking TNF $\alpha$  biological activity carries out therapeutic effects of anti-TNFa mAbs. Two anti-TNF mAbs, infliximab and adalimumab, are currently in the market. Infliximab is a chimeric mAb ( $IgG_1$ ) with specificity to TNF $\alpha$ . The FDA approved it in 1998 for use in patients with Crohn disease with moderate to severe activity that responded inadequately to conventional therapy. In 1999, its approval was widened for use in the treatment of RA in combination with methotrexate therapy. In 2000, its use was expanded for other inflammatory diseases. Indicated for adults with moderate to severe active RA, its use was approved for reducing signs and symptoms of the disease and inhibition of structural progression damage of joints for patients who have not obtained effective response to disease-modifying antirheumatic drugs. After its use in patients with RA, there is a rapid fall in serum acute phase inflammatory markers: C-reactive protein, erythrocyte sedimentation rate, and cytokines [interleukin (IL)-6]. The success of infliximab has prompted the creation of other anti-TNFa mAbs.

Adalimumab is a fully human mAb IgG1 that specifically binds to TNF $\alpha$  and neutralizes the biological function of blocking TNF. Its interaction with TNF receptors on the p55 and p75 cell surface also modulates the biological responses induced or regulated by TNF, including changes in the levels of adhesion molecules responsible for leukocyte migration (ELAM-1, VCAM-1, and ICAM-1). An advantage of adalimumab is that its fully human nature makes it less immunogenic, unlike infliximab, which is a chimeric mAb [34].

Another fully human anti-TNF $\alpha$  mAb is golimumab (Simponi). Because it is an entirely human mAb, it shares similar characteristics with adalimumab. In vitro studies suggest that golimumab possess a higher affinity to soluble TNF $\alpha$  than does adalimumab; however, this has not been tested in vivo [35]. Studies suggest that adalimumab and golimumab possess similar efficacy after indication. However, golimumab demonstrated superior outcomes for sustained outcomes [36].

Certolizumab pegol (CZP) (Cimzia) is a pegylated mAb. Unlike a classical mAbs with a variable and constant region, CZP is composed of a variable region fragment known as fragment antigen binding, with polyethylene glycol conjugated to it. The addition of polyethylene glycol increases the half-life of the antibody fragment comparable to that of unmodified mAbs [37]. CZP lacks a constant region and is therefore unable to perform ADCC or complement binding. The similar therapeutic effects of CZP compared with infliximab, adalimumab, and golimumab suggest that Fc, mediated, and complement fixation are not large contributors to the therapeutic effects of mAb therapy [38] (Fig. 4.7).

#### 4.5.4 Belimumab

Belimumab is a human IgG1 mAb specific for the B-lymphocyte stimulator (BLyS), sometimes known as B-cell activating factor. It has been approved for use in SLE. SLE is a heterogeneous autoimmune disease with a wide array of clinical symptoms ranging from



**FIGURE 4.7** Three anti-TNF $\alpha$  biologics. Infliximab is a chimeric mAb with a murine variable region and a human constant region (A). Adalimumab is a fully human mAb (B). Certolizumab pegol is a humanized mAb fragment with two polyethylene glycol molecules attached to it (C).

fever to fatigue rash. Over time, SLE can affect major organs, resulting in renal, cardiac, pulmonary, and central nervous system toxicities. Although the pathogenesis of SLE is not well understood, there are indications that B-cell malfunction is involved. Autoantibodies, particularity anti-double-stranded DNA antibodies, are present in 70% of patients. Immune complexes are thought to contribute to the clinical manifestations seen in SLE, and higher BLyS levels are correlated with a worsening of clinical symptoms [39]. BLyS has been shown to inhibit B-cell apoptosis and promotes differentiation of B-cells into antibody-producing plasma cells. Moreover, patients with SLE have elevated BLyS levels, which makes it an ideal target when designing mAbs for patients with SLE [39,40]. By binding to the soluble form of BLyS, Belimumab prevents BLyS from binding to its receptors and inhibits its downstream functions.

#### 4.5.5 Ipilimumab

Ipilimumab is a human IgG1 mAb specific for cytotoxic T-lymphocyte antigen 4 (CTLA 4). CTLA 4, like CD 28, binds to the CD 80:86 (B7.1:B7.2) ligands [41]. However, unlike CD28, which sends a stimulatory signal, CTLA 4 acts as an immune regulator and inhibits T-cell activation. One of the defining characteristics of cancer is its ability to hide from the immune system [42]. Normally, when cells are transformed into cancerous cells, the immune system can eliminate them early on [43]. However, the ability of cancerous cells to not induce immune activation when they proliferate uncontrollably makes it difficult for the immune system to destroy abnormal cells. By blocking CTL 4, an immune suppressant, it is thought to make the immune system more sensitive and able to find and eradicate cancerous cells. Currently, Ipilimumab is accepted for treatment of metastatic melanoma [44].

#### 4.5.6 Mepolizumab

Mepolizumab is a humanized IgG1 mAb specific for IL5. It is approved for management of severe eosinophilic asthma. Severe eosinophilic asthma differs from day-to-day asthma symptoms in that it does not respond to typical inhalants [45]. Asthma exacerbations are associated with airway inflammation and morbidity. Mepolizumab, an IL5 mAb, had shown disappointing results in asthma treatment in three clinical trials [46–48]. However, interest emerged in the use of mepolizumab in a subset of patients with asthma who are unresponsive to prednisone treatments. These patients demonstrate a high concentration of eosinophils in their sputum [49]. Eosinophils have been thought to be implicated in severe asthma owing to their presence in postmortem examinations of those who died from an acute severe asthma attack [50].

#### 4.5.7 Natalizumab

Natalizumab is a humanized IgG4 mAb specific for the  $\alpha$ 4 integrin molecule. It is used in the treatment of MS. MS is a demyelination disease of the central nervous system. Axonal

loss and plaques in the central nervous system characterize MS. Lymphocyte passage across the blood-brain barrier is thought to be an important factor in disease etiology [51,52].  $\alpha 4\beta 1$  is an integrin glycoprotein expressed on leukocytes that interact with vascular-cell adhesion molecule 1 (VCAM-1).  $\alpha 4\beta 1$ -VCAM-1 interactions are essential to diapedesis. Natalizumab is a humanized mAb specific for the  $\alpha 4$  subunit of the integrin proteins (both  $\alpha 4\beta 1$  and  $\alpha 4\beta 7$ ). Natalizumab blocks the interaction of the integrin and the endothelial receptors, inhibiting diapedesis and in turn inflammation [53]. Randomized trials of natalizumab have demonstrated promise as a treatment for MS [54,55].

#### 4.5.8 Omalizumab

Omalizumab is a humanized monoclonal anti-IgE antibody with a high degree of human characteristics greater than 95% (IgG1-k). It recognizes the constant region of IgE antibodies, Fc $\epsilon$ , and prevents it from binding to the high-affinity Fc $\epsilon$ R receptors. FcR interactions can elicit various downstream effects. When these effects go awry, pathologies can occur. The  $\varepsilon$  region of IgE antibodies has been strongly correlated in asthma and allergic rhinitis. These two diseases are thought to have effects mediated through the interaction of IgE antibodies and their corresponding Fc receptors: in the case of IgE, the Fc receptor is Fc $\epsilon$ . When IgE constant region is recognized by the high -affinity Fc $\epsilon$ receptor, on cells such as mast cells, and neutrophils, there is a surge of activity that is implicated in this pathogenesis. The importance of IgE antibodies in disease pathology has prompted the creation of anti-IgE mAbs. By preventing IgE antibodies from being recognized by Fce receptors, omalizumab prevents the release of inflammatory mediators from mast cells and basophils such as histamine, leukotrienes, and cytokines. Omalizumab prevents asthma attacks after provocation test by inhalation allergens in both immediate and delayed response in patients with the mild allergic disease. Other applications have been suggested. Effective use in atopic dermatitis has been reported.

Omalizumab can reduce asthma exacerbations in patients with moderate and severe stages of the disease. It can reduce symptoms and improve parameters of pulmonary function, as well as reduce necessary doses of corticosteroids and beta-agonists, resulting in improvement in the quality of life for patients. In patients with eosinophilia, omalizumab has been shown to improve numbers to a clinical level. In patients with seasonal allergic rhinitis, omalizumab has been shown to decrease nasal and ocular symptoms as well as reduce the need for drugs such as antihistamines.

Administration of omalizumab can be done intravenously or subcutaneously. After administration, there is a rapid dose-dependent drop in serum free IgE (>99%). The blockage of IgE–Fc $\epsilon$ R interaction results in a down-regulation of high-affinity FC $\epsilon$ R1 expression and subsequent release of histamine receptor by antigen-stimulated basophils. The dosage of anti-IgE needed must be individualized according to the total IgE levels and the weight of each patient. The formation anti-IgE–IgE complexes results in the disappearance of circulating free IgE. Because these complexes cannot bind the IgE

receptors, they are devoid of IgE-specific biological activity. Owing to antibody humanization and the small=size anti-IgE–IgE complex, omalizumab does not trigger autoimmune diseases because these complexes are unable to activate the complement system.

#### 4.5.9 Programmed Death 1 Blockers: Nivolumab and Pembrolizumab

Like CTLA 4, programmed death 1 (PD1) inhibits immune activation. Its ligand, PD-L1, is a B7 family ligand. However, unlike B7, it cannot interact with CD28 and CTLA4 [56]. PD-L1 has been shown to be up-regulated in various cancer tissues [57], one system where the PD-L1 blockade has shown promise is in melanoma [58,59]. Nivolumab is a human IgG4 mAb specific for PD1 [60]. It is approved for management of melanoma. By blocking PD1, nivolumab blocks the PD-L1–PD1 interaction, and dampens immune inhibition by PD-L1 on melanoma cells. It is hypothesized that blocking PD1 melanoma cells are no longer able to avoid immune surveillance. Therefore, the immune system can destroy the malignant cells. Although monotherapy with nivolumab has shown promise, combination therapy of nivolumab with ipilimumab shows strong results displaying rapid tumor progression [61]. Pembrolizumab is a humanized IgG4 mAb with specificity for PD1 [62]. Pembrolizumab is an approved mAb under the FDA's fast-track program. It showed promise in patients for whom nivolumab therapy had failed.

#### 4.5.10 Secukinumab

Secukinumab is a human IgG1 mAb specific for IL17A. Psoriasis etiology is poorly understood. Beyond IL12 and IL23, IL17 has been implicated in psoriasis [63,64]. IL17A is a proinflammatory cytokine essential for microbial pathogen defense. It is released primarily by a subset of CD4 cells known as Th17 cells [65,66]. IL17 has been shown to interact directly with keratinocytes to stimulate the release of molecules that are elevated in psoriasis [67]. Secukinumab, an mAb specific for IL17A, is an approved mAb drug that has shown promise in the treatment of moderate to severe psoriasis [68–70]. By binding to IL17A, secukinumab can inhibit the interaction of IL17A with its receptors and reduce its immunologic functions.

#### 4.5.11 Tocilizumab

Tocilizumab is a humanized mAb specific for the IL6 receptor. It is indicated for treatment of RA. IL6 is a potent proinflammatory cytokine affecting leukocytes, neutrophils, and osteoclasts [71]. IL6-activated cells are found in high numbers in RA. Beyond affecting immune cells, IL6 is a main player in activating the acute phase reaction, a feature correlated with disease activity in RA [72]. The presence of IL6 effects in RA has made blocking the IL6 receptor an attractive candidate in the treatment of RA. Tocilizumab has shown promise as an effective treatment of RA in randomized control trials [73], and has even shown to be more efficient as a monotherapy compared with the standard methotrexate regiment [74]. A systematic review of the literature has shown support for the efficacy of tocilizumab as a therapeutic agent for RA [75].

#### 4.5.12 Ustekinumab

Ustekinumab is a human IgG1 mAb specific for the p40 subunit common to IL12 and IL23. It is used to treat psoriasis. Psoriasis is a common chronic skin autoimmune disease. It is characterized by a thickening of the epidermal layer caused by the proliferation of keratinocytes [76]. Skin plaques, a common manifestation, can be found in the elbows, knees, umbilical, and cosmetically sensitive areas such as the face, feet, hands, scalp, and genitalia [77]. Beyond the dermatologic manifestations, psoriasis is associated with comorbidities such as depression, cardiovascular disease, Crohn disease, obesity, and psoriatic arthritis [78]. Although the etiology of psoriasis is not understood, elevated cytokines levels have been highly correlated with disease severity [64]. Beyond cytokine levels, the T-helper subtype 1 (Th1) is commonly seen in psoriatic lesions [79,80]. One of the main cytokines driving a Th1 phenotype is IL12 [81]. IL12 is a heterodimer protein composed of IL12A and IL12B, released by antigen-presenting cells in response to antigen presentation [82]. IL12 is joined by two disulfide bonds that are glycosylated, termed p35 and p40. IL12 activates NK cells and has been shown to polarize CD4 response to a Th1 response. IL23 is another heterodimer protein that shares the same p40. IL23 has demonstrated activity on NK cells as well as T-cells [83]. Ustekinumab has shown significant promise in a double-blind randomized controlled trial at reducing signs and symptoms of psoriasis [84]. A long-term follow-up of 1500 patients treated for over 4 years demonstrated safety and tolerability [85].

#### 4.5.13 Vedolizumab

Vedolizumab is a humanized IgG1 mAb specific to  $\alpha 4\beta7$  integrin protein. It is used in the treatment of ulcerative colitis (UC), and Crohn disease.  $\alpha 4\beta 7$  is an integrin glycol protein expressed on T- and B-cells [86] that binds to mucosal addressin-cell adhesion molecule-1 (MAdCAM-1) [87], which is found on intestinal vasculature [88] and is highly expressed in an inflamed gut tissue. The interaction of integrin and MAdCAM-1 is essential to leukocyte migration into the gut [89]. UC is a chronic form of an inflammatory bowel disease characterized by abdominal cramps and bloody diarrhea [90]. One proposed theory for the pathogenesis of UC is thought to be an overactive T-cell population [91]. In theory, by blocking T-cell diapedesis into the gut, UC symptoms should subside, which makes MAdCAM-1-integrin interactions an attractive candidate. Targeted therapy for  $\alpha 4\beta 7$  has been shown to be effective as an acute treatment for UC [92] as well as for long-term maintenance [93]. Crohn disease is another form of an inflammatory disease of the intestine. It is characterized by thickening of the colon wall, with inflammation localized to patches anywhere in the digestive track, unlike UC, in which inflammation is largely continuous in the large intestine [94]. An overactive T-cell population is thought to underlie the etiology of both Crohn disease and UC [95]. Vedolizumab has shown efficacy in the induction and maintenance therapy for Crohn disease in randomized control trials [96].

## 4.6 Conclusion and Perspectives

The development of mAbs by genetic engineering has provided new treatment to numerous diseases. The ability of mAbs to bind to a predetermined epitope stimulated the development of these new drugs. The complexity of the molecule made of polypeptide chains with large—molecular weight domains associated with multifunctionality was the initial hurdle to their production and use. The mouse origin and the difficulty in selecting the appropriate target structures on the cell surface can be added to the list of initial difficulties in the development of mAbs. The replacement of most murine sequences with equivalent human sequences, a concomitant decrease in immunogenicity, and identifying cell surface components of mAbs have become valuable resources in the treatment of infectious, inflammatory, autoimmune, and neoplastic diseases. In addition, mAbs have significant application as markers for diagnostic tests and scientific research tools in medicine.

Continued success in the development of therapy-based antibodies will require extensive clinical research detailing how to use these compounds properly, which patients will benefit, and long-term side effects. Immunologic research also benefits because it will reveal important aspects of the mechanisms' basic action of antibodies and the fundamental immunologic mechanisms involved.

# References

- C. Simpson, The evolutionary history of division of labour, Proceedings of the Royal Society of London B: Biological Sciences 279 (1726) (2012) 21–116.
- [2] P. Quattroni, R.M. Exley, C.M. Tang, New insights into pathogen recognition, Expert Review of Anti-Infective Therapy 9 (8) (2011) 577–579.
- [3] G.M. Weinstock, Genomic approaches to studying the human microbiota, Nature 489 (7415) (2012) 250–256.
- [4] F. Bäckhed, R.E. Ley, J.L. Sonnenburg, D.A. Peterson, J.I. Gordon, Host-bacterial mutualism in the human intestine, Science 307 (5717) (2005) 1915–1920.
- [5] E.A. Grice, J.A. Segre, The human microbiome: our second genome, Annual Review of Genomics and Human Genetics 13 (2012) 151.
- [6] M. Gerstein, Genomics: ENCODE leads the way on big data, Nature 489 (7415) (2012) 208.
- [7] E. Salvucci, Selfishness, warfare, and economics; or integration, cooperation, and biology, Frontiers in Cellular and Infection Microbiology 2 (2012).
- [8] G.W. Litman, J.P. Rast, S.D. Fugmann, The origins of vertebrate adaptive immunity, Nature Reviews Immunology 10 (8) (2010) 543–553.
- [9] W.E. Paul, Bridging innate and adaptive immunity, Cell 147 (6) (2011) 1212-1215.
- [10] T. Kaisho, K. Takeda, The study of innate immunity in Japan: a historical perspective, International Immunology 21 (4) (2009) 313–316.

- [11] F. Shakib, D. Stanworth, Human IgG subclasses in health and disease (A review) Part I, Ricerca in clinica e in laboratorio 10 (3) (1980) 463–479.
- [12] H.L. Spiegelberg, Biological role of different antibody classes, International Archives of Allergy and Immunology 90 (Suppl. 1) (1989) 22–27.
- [13] G.P. Adams, L.M. Weiner, Monoclonal antibody therapy of cancer, Nature Biotechnology 23 (9) (2005) 1147–1157.
- [14] G.J. Weiner, Building better monoclonal antibody-based therapeutics, Nature Reviews Cancer 15 (6) (2015) 361–370.
- [15] T. Tahara, K. Usuki, H. Sato, et al., A sensitive sandwich ELISA for measuring thrombopoietin in human serum: serum thrombopoietin levels in healthy volunteers and in patients with haemopoietic disorders, British Journal of Haematology 93 (4) (1996) 783–788.
- [16] M.W. Bolt, P.A. Mahoney, High-efficiency blotting of proteins of diverse sizes following sodium dodecyl sulfate-polyacrylamide gel electrophoresis, Analytical Biochemistry 247 (2) (1997) 185–192.
- [17] L.A. Sternberger, P.H. Hardy, J.J. Cuculis, H.G. Meyer, The unlabeled antibody enzyme method of immunohistochemistry preparation and properties of soluble antigen-antibody complex (horseradish peroxidase-antihorseradish peroxidase) and its use in identification of spirochetes, Journal of Histochemistry & Cytochemistry 18 (5) (1970) 315–333.
- [18] Y. Hu, M.J. Turner, J. Shields, et al., Investigation of the mechanism of action of alemtuzumab in a human CD52 transgenic mouse model, Immunology 128 (2) (2009) 260–270.
- [19] C.S. Zent, C.R. Secreto, B.R. LaPlant, et al., Direct and complement dependent cytotoxicity in CLL cells from patients with high-risk early-intermediate stage chronic lymphocytic leukemia (CLL) treated with alemtuzumab and rituximab, Leukemia Research 32 (12) (2008) 1849–1856.
- [20] A.P. Mone, C. Cheney, A.L. Banks, et al., Alemtuzumab induces caspase-independent cell death in human chronic lymphocytic leukemia cells through a lipid raft-dependent mechanism, Leukemia 20 (2) (2006) 272–279.
- [21] G.J. Weiner, Rituximab: mechanism of action, Seminars in Hematology 47 (2) (2010) 115–123.
- [22] T. Van Meerten, R.S. Van Rijn, S. Hol, A. Hagenbeek, S.B. Ebeling, Complement-induced cell death by rituximab depends on CD20 expression level and acts complementary to antibody-dependent cellular cytotoxicity, Clinical Cancer Research 12 (13) (2006) 4027–4035.
- [23] C. Bezombes, J.-J. Fournié, G. Laurent, Direct effect of rituximab in B-cell–derived lymphoid neoplasias: mechanism, regulation, and perspectives, Molecular Cancer Research 9 (11) (2011) 1435–1442.
- [24] J.P. Deans, H. Li, M.J. Polyak, CD20-mediated apoptosis: signalling through lipid rafts, Immunology 107 (2) (2002) 176–182.
- [25] M.R. Smith, Rituximab (monoclonal anti-CD20 antibody): mechanisms of action and resistance, Oncogene 22 (47) (2003) 7359–7368.
- [26] X. Zhou, W. Hu, X. Qin, The role of complement in the mechanism of action of rituximab for B-cell lymphoma: implications for therapy, The Oncologist 13 (9) (2008) 954–966.
- [27] A. Capron, M. Capron, J.P. Dessaint, M. Fougereau, J. Dausset, ADCC as primary mechanisms of defence against metazoan parasites, in: Immunology 80. Progress in Immunology IV, Fourth International Congress of Immunology, Paris, July 1980. Theme 12., vol. 2, Academic Press, Inc., 1980, pp. 782–793.
- [28] G.A. Wiseman, L.I. Gordon, P.S. Multani, et al., Ibritumomab tiuxetan radioimmunotherapy for patients with relapsed or refractory non-Hodgkin lymphoma and mild thrombocytopenia: a phase II multicenter trial, Blood 99 (12) (2002) 4336–4342.
- [29] K. Hanaoka, M. Hosono, Y. Tatsumi, et al., Heterogeneity of intratumoral 111In-ibritumomab tiuxetan and 18F-FDG distribution in association with therapeutic response in radioimmunotherapy for B-cell non-Hodgkin's lymphoma, EJNMMI Research 5 (1) (2015).

- [30] K. Pfeffer, Biological functions of tumor necrosis factor cytokines and their receptors, Cytokine and Growth Factor Reviews 14 (3–4) (2003) 185–191.
- [31] M. Sata, K. Walsh, TNFα regulation of fas ligand expression on the vascular endothelium modulates leukocyte extravasation, Nature Medicine 4 (4) (1998) 415–420.
- [32] R.N. Maini, M.J. Elliott, F.M. Brennan, M. Feldmann, Beneficial effects of tumour necrosis factoralpha (TNF-alpha) blockade in rheumatoid arthritis (RA), Clinical and Experimental Immunology 101 (2) (1995) 207–212.
- [33] M. Ramos-Casals, P. Brito-Zerón, S. Muñoz, et al., Autoimmune diseases induced by TNF-targeted therapies: analysis of 233 cases, Medicine 86 (4) (2007).
- [34] P.J. Anderson, Tumor necrosis factor inhibitors: clinical implications of their different immunogenicity profiles, Seminars in Arthritis and Rheumatism 34 (5 Suppl1) (2005) 19–22.
- [35] D. Shealy, A. Cai, K. Staquet, et al., Characterization of golimumab, a human monoclonal antibody specific for human tumor necrosis factor α, mAbs 2 (4) (2010) 428–439.
- [36] K. Thorlund, E. Druyts, K. Toor, E.J. Mills, Comparative efficacy of golimumab, infliximab, and adalimumab for moderately to severely active ulcerative colitis: a network meta-analysis accounting for differences in trial designs, Expert Review of Gastroenterology & Hepatology 9 (5) (2015) 693–700.
- [37] A. Nesbitt, G. Fossati, M. Bergin, et al., Mechanism of action of certolizumab pegol (CDP870): *in vitro* comparison with other anti-tumor necrosis factor  $\alpha$  agents, Inflammatory Bowel Diseases 13 (11) (2007) 1323–1332.
- [38] S.C. Horton, S. Das, P. Emery, Certolizumab pegol in rheumatoid arthritis: a review of Phase III clinical trials and its role in real-life clinical practice, International Journal of Clinical Rheumatology 6 (5) (2011) 517–527.
- [39] A.K. Wiglesworth, K.M. Ennis, D.R. Kockler, Belimumab: a BLyS-specific inhibitor for systemic lupus, Annals of Pharmacotherpy 44 (12) (2010) 1955–1961.
- [40] W. Stohl, D.M. Hilbert, The discovery and development of belimumab: the anti-BLyS-lupus connection, Nature Biotechnology 30 (1) (2012) 69–77.
- [41] C.A. Chambers, M.S. Kuhns, J.G. Egen, J.P. Allison, CTLA-4-mediated inhibition in regulation of T cell responses: mechanisms and manipulation in tumor immunotherapy, Annual Review of Immunology 19 (1) (2001) 565–594.
- [42] D. Hanahan, R.A. Weinberg, Hallmarks of cancer: the next generation, Cell 144 (5) (2011) 646-674.
- [43] R. Kim, M. Emi, K. Tanabe, Cancer immunoediting from immune surveillance to immune escape, Immunology 121 (1) (2007) 1–14.
- [44] F.S. Hodi, S.J. O'Day, D.F. McDermott, et al., Improved survival with ipilimumab in patients with metastatic melanoma, New England Journal of Medicine 363 (8) (2010) 711–723.
- [45] J.M. FitzGerald, P.G. Gibson, Asthma exacerbations 4, Prevention. Thorax 61 (11) (2006) 992–999.
- [46] J.C. Kips, B.J. O'Connor, S.J. Langley, et al., Effect of SCH55700, a humanized anti-human interleukin-5 antibody, in severe persistent asthma: a pilot study, American Journal of Respiratory and Critical Care Medicine 167 (12) (2003) 1655–1659.
- [47] M.J. Leckie, A. ten Brinke, J. Khan, et al., Effects of an interleukin-5 blocking monoclonal antibody on eosinophils, airway hyper-responsiveness, and the late asthmatic response, The Lancet 356 (9248) (2000) 2144–2148.
- [48] P. Flood-Page, C. Swenson, I. Faiferman, et al., A study to evaluate safety and efficacy of mepolizumab in patients with moderate persistent asthma, American Journal of Respiratory and Critical Care Medicine 176 (11) (2007) 1062–1071.
- [49] P. Haldar, C.E. Brightling, B. Hargadon, et al., Mepolizumab and exacerbations of refractory eosinophilic asthma, The New England Journal of Medicine 360 (10) (2009) 973–984.

- [50] J.C. Houston, S. De Navasquez, J.R. Trounce, A clinical and pathological study of fatal cases of status asthmaticus, Thorax 8 (3) (1953) 207–213.
- [51] A. Prat, J. Antel, Pathogenesis of multiple sclerosis, Current Opinion in Neurology 18 (3) (2005) 225–230.
- [52] C. Ffrench-Constant, Pathogenesis of multiple sclerosis, Lancet 343 (8892) (1994) 271–275.
- [53] T.A. Yednock, C. Cannon, L.C. Fritz, F. Sanchez-Madrid, L. Steinman, N. Karin, Prevention of experimental autoimmune encephalomyelitis by antibodies against  $\alpha 4\beta$ l integrin, Nature 356 (6364) (1992) 63–66.
- [54] C.H. Polman, P.W. O'Connor, E. Havrdova, et al., A randomized, placebo-controlled trial of natalizumab for relapsing multiple sclerosis, New England Journal of Medicine 354 (9) (2006) 899–910.
- [55] D.H. Miller, O.A. Khan, W.A. Sheremata, et al., A controlled trial of natalizumab for relapsing multiple sclerosis, New England Journal of Medicine 348 (1) (2003) 15–23.
- [56] C. Blank, T. Gajewski, A. Mackensen, Interaction of PD-L1 on tumor cells with PD-1 on tumorspecific T cells as a mechanism of immune evasion: implications for tumor immunotherapy, Cancer Immunology, Immunotherapy 54 (4) (2005) 307–314.
- [57] D.M. Pardoll, The blockade of immune checkpoints in cancer immunotherapy, Nature Reviews Cancer 12 (4) (2012) 252–264.
- [58] J. Gadiot, A.I. Hooijkaas, A.D.M. Kaiser, H. van Tinteren, H. van Boven, C. Blank, Overall survival and PD-L1 expression in metastasized malignant melanoma, Cancer 117 (10) (2011) 2192–2201.
- [59] S. Spranger, R.M. Spaapen, Y. Zha, et al., Up-regulation of PD-L1, IDO, and Tregs in the melanoma tumor microenvironment is driven by CD8+ T cells, Science Translational Medicine 5 (200) (2013) 200ra116.
- [60] S.L. Topalian, M. Sznol, D.F. McDermott, et al., Survival, durable tumor remission, and long-term safety in patients with advanced melanoma receiving nivolumab, Journal of Clinical Oncology 32 (10) (2014) 1020–1030.
- [61] J.D. Wolchok, H. Kluger, M.K. Callahan, et al., Nivolumab plus ipilimumab in advanced melanoma, New England Journal of Medicine 369 (2) (2013) 122–133.
- [62] C. Robert, A. Ribas, J.D. Wolchok, et al., Anti-programmed-death-receptor-1 treatment with pembrolizumab in ipilimumab-refractory advanced melanoma: a randomised dose-comparison cohort of a phase 1 trial, Lancet 384 (9948) (2014) 1109–1117.
- [63] J.G. Krueger, S. Fretzin, M. Suárez-Fariñas, et al., IL-17A is essential for cell activation and inflammatory gene circuits in subjects with psoriasis, Journal of Allergy and Clinical Immunology 130 (1) (2012) 145–154.
- [64] O. Arican, M. Aral, S. Sasmaz, P. Ciragil, Serum levels of TNF-α, IFN-γ, IL-6, IL-8, IL-12, IL-17, and IL-18 in patients with active psoriasis and correlation with disease severity, Mediators of Inflammation 2005 (5) (2005) 273–279.
- [65] C. Gu, L. Wu, X. Li, IL-17 family: cytokines, receptors and signaling, Cytokine 64 (2) (2013) 477-485.
- [66] M. Kawaguchi, M. Adachi, N. Oda, F. Kokubu, S.K. Huang, IL-17 cytokine family, Journal of Allergy and Clinical Immunology 114 (6) (2004) 1265–1273.
- [67] E. Guttman-Yassky, M.A. Lowes, J. Fuentes-Duculan, et al., Low expression of the IL-23/Th17 pathway in atopic dermatitis compared to psoriasis, The Journal of Immunology 181 (10) (2008) 7420–7427.
- [68] R.G. Langley, B.E. Elewski, M. Lebwohl, et al., Secukinumab in plaque psoriasis results of two phase 3 trials, The New England Journal of Medicine (2014) 3–8.
- [69] K.A. Papp, R.G. Langley, B. Sigurgeirsson, et al., Efficacy and safety of secukinumab in the treatment of moderate-to-severe plaque psoriasis: a randomized, double-blind, placebo-controlled phase II dose-ranging study, British Journal of Dermatology 168 (2) (2013) 412–421.

- [70] I.B. McInnes, J. Sieper, J. Braun, et al., Efficacy and safety of secukinumab, a fully human antiinterleukin-17A monoclonal antibody, in patients with moderate-to-severe psoriatic arthritis: a 24-week, randomised, double-blind, placebo-controlled, phase II proof-of-concept trial, Annals of the Rheumatic Diseases 73 (2) (2014) 349–356.
- [71] T. Naka, N. Nishimoto, T. Kishimoto, The paradigm of IL-6: from basic science to medicine, Arthritis Research 4 (Suppl. 3) (2002) S233–S242.
- [72] D. Aletaha, V.P. Nell, T. Stamm, et al., Acute phase reactants add little to composite disease activity indices for rheumatoid arthritis: validation of a clinical activity score, Arthritis Research & Therapy 7 (4) (2005) R796–R806.
- [73] J.S. Smolen, A. Beaulieu, A. Rubbert-Roth, et al., Effect of interleukin-6 receptor inhibition with tocilizumab in patients with rheumatoid arthritis (OPTION study): a double-blind, placebocontrolled, randomised trial, The Lancet 371 (9617) (2008) 987–997.
- [74] G. Jones, A. Sebba, J. Gu, et al., Comparison of tocilizumab monotherapy versus methotrexate monotherapy in patients with moderate to severe rheumatoid arthritis: the AMBITION study, Annals of the Rheumatic Diseases 69 (01) (2010) 88–96.
- [75] M.M. Schoels, D. van der Heijde, F.C. Breedveld, et al., Blocking the effects of interleukin-6 in rheumatoid arthritis and other inflammatory rheumatic diseases: systematic literature review and meta-analysis informing a consensus statement, Annals of the Rheumatic Diseases 72 (4) (2013) 583–589.
- [76] F.O. Nestle, D.H. Kaplan, J. Barker, Psoriasis, New England Journal of Medicine 361 (5) (2009) 496-509.
- [77] E. Christophers, Psoriasis–epidemiology and clinical spectrum, Clinical and Experimental Dermatology 26 (4) (2001) 314–320.
- [78] U. Mrowietz, J.T. Elder, J. Barker, The importance of disease associations and concomitant therapy for the long-term management of psoriasis patients, Archives of Dermatological Research 298 (7) (2007) 309–319.
- [79] I. Kryczek, A.T. Bruce, J.E. Gudjonsson, et al., Induction of IL-17+ T cell trafficking and development by IFN-γ: mechanism and pathological relevance in psoriasis, The Journal of Immunology 181 (7) (2008) 4733–4741.
- [80] M.A. Lowes, T. Kikuchi, J. Fuentes-Duculan, et al., Psoriasis vulgaris lesions contain discrete populations of Th1 and Th17 T cells, Journal of Investigative Dermatology 128 (5) (2008) 1207–1211.
- [81] R. Manetti, P. Parronchi, M.G. Giudizi, et al., Natural killer cell stimulatory factor (interleukin 12 [IL-12]) induces T helper type 1 (Th1)-specific immune responses and inhibits the development of IL-4-producing Th cells, The Journal of Experimental Medicine 177 (4) (1993) 1199–1204.
- [82] G. Trinchieri, Interleukin-12 and the regulation of innate resistance and adaptive immunity, Nature Reviews Immunology 3 (2) (2003) 133–146.
- [83] S. Aggarwal, N. Ghilardi, M.-H. Xie, F.J. de Sauvage, A.L. Gurney, Interleukin-23 promotes a distinct CD4 T cell activation state characterized by the production of interleukin-17, Journal of Biological Chemistry 278 (3) (2003) 1910–1914.
- [84] A. Gottlieb, A. Menter, A. Mendelsohn, et al., Ustekinumab, a human interleukin 12/23 monoclonal antibody, for psoriatic arthritis: randomised, double-blind, placebo-controlled, crossover trial, Lancet 373 (9664) (2009) 633–640.
- [85] K.A. Papp, C.E.M. Griffiths, K. Gordon, et al., Long-term safety of ustekinumab in patients with moderate-to-severe psoriasis: final results from 5 years of follow-up, British Journal of Dermatology 168 (4) (2013) 844–854.
- [86] U.H. Von Andrian, B. Engelhardt,  $\alpha 4$  integrins as therapeutic targets in autoimmune disease, New England Journal of Medicine 348 (1) (2003) 68–72.

- [87] D.J. Erle, M.J. Briskin, E.C. Butcher, A. Garcia-Pardo, A.I. Lazarovits, M. Tidswell, Expression and function of the MAdCAM-1 receptor, integrin alpha 4 beta 7, on human leukocytes, The Journal of Immunology 153 (2) (1994) 517–528.
- [88] S. Arihiro, H. Ohtani, M. Suzuki, et al., Differential expression of mucosal addressin cell adhesion molecule-1 (MAdCAM-1) in ulcerative colitis and Crohn's disease, Pathology International 52 (5–6) (2002) 367–374.
- [89] T.A. Springer, Traffic signals for lymphocyte recirculation and leukocyte emigration: the multistep paradigm, Cell 76 (2) (1994) 301–314.
- [90] S. Danese, C. Fiocchi, Ulcerative colitis, New England Journal of Medicine 365 (18) (2011) 1713–1725.
- [91] R.B. Sartor, Mechanisms of disease: pathogenesis of Crohn's disease and ulcerative colitis, Nature Clinical Practice Gastroenterology & Hepatology 3 (7) (2006) 390–407.
- [92] B.G. Feagan, G.R. Greenberg, G. Wild, et al., Treatment of ulcerative colitis with a humanized antibody to the  $\alpha 4\beta 7$  integrin, New England Journal of Medicine 352 (24) (2005) 2499–2507.
- [93] B.G. Feagan, P. Rutgeerts, B.E. Sands, et al., Vedolizumab as induction and maintenance therapy for ulcerative colitis, New England Journal of Medicine 369 (8) (2013) 699–710.
- [94] J.E. Lennard-Jones, Classification of inflammatory bowel disease, Scandinavian Journal of Gastroenterology 24 (S170) (1989) 2–6.
- [95] C. Fiocchi, Inflammatory bowel disease: etiology and pathogenesis, Gastroenterology 115 (1) (1998) 182–205.
- [96] W.J. Sandborn, B.G. Feagan, P. Rutgeerts, et al., Vedolizumab as induction and maintenance therapy for Crohn's disease, New England Journal of Medicine 369 (8) (2013) 711–721.

# 5

# Recombinant Antibodies: Trends for Standardized Immunological Probes and Drugs

L.M. Alvarenga<sup>1, \*</sup>, J. de Moura<sup>1</sup>, P. Billiald<sup>2,3</sup>

<sup>1</sup>UNIVERSIDADE FEDERAL DO PARANÁ (UFPR), CURITIBA, PARANÁ, BRAZIL; <sup>2</sup>UNIVERSITY OF PARIS-SUD, CHÂTENAY-MALABRY, FRANCE; <sup>3</sup>MUSÉUM NATIONAL D'HISTOIRE NATURELLE - CNRS UMR 7245, PARIS, FRANCE

# 5.1 Introduction

In 1975, G. Kohler and C. Milstein (Nobel Prize in Medicine, 1984) described hybridoma technology, which consists of immortalizing antibody-secreting cells [1]. Fairly simple to achieve with mouse or rat plasma cells, this technology has led to the selection of a multitude of monoclonal antibodies directed against various antigens. Soon after that, a wide range of applications have emerged. Every month and every week has seen new monoclonal antibodies appearing with well-established properties including high antigen-specificity and that can be produced continuously in a stable form [2]. These antibodies have led to the identification of cell markers (CD). They also have contributed to the understanding of the structure—function relationship of complex proteins and to elucidating the mechanisms of interaction between biomolecules. The immunoassay sector has greatly taken great advantage of this discovery by developing robust immuno reagents (Fig. 5.1). On July 8, 2015, a simple search for the term "immunoassay" in the PubMed database revealed more than  $4.5 \times 10^5$  original articles. The market of "immunodiagnostic" represents today \$1.6 billion with over two million antibodies sold for research by more than 300 companies [3].

However, the use of these monoclonal antibodies of animal origin has been a dramatic failure for therapeutic applications in humans [4,5]. Only three antibodies have been approved by the Food and Drug Administration (FDA): an anti-CD3 IgG (Orthoclone OKT3; Muromomab), which is used in the prevention of acute allograft rejection, and also two radiolabeled anti-CD20 antibodies (Bexxar, <sup>131</sup>I-Tositumomab; and Zevalin, <sup>111</sup>In and <sup>90</sup>Y Ibritumomab) prescribed as second-line in the treatment of particular lymphomas. This setback has several origins [6]. Indeed, serious reactions including fatal allergic ones

\*Corresponding Author.

**1960:** First radioimmunoassay (RIA) of circulating hormones. The principle comprises precipitation of plasmatic hormone, using polyclonal antibodies in competition with a low concentration of radiolabeled hormone. These tests are sensitive but long (24 hours).

**1971:** First sandwich immunoassays in which the target antigen is captured by an antibody immobilised on a plastic surface. The immune complexes formed are detected by a secondary antibody chemically coupled to an enzyme (peroxidase, phosphatase alkaline  $\beta$ -galactosidase).

**1975:** The hybridoma technology will revolutionise the immunoassay sector. It becomes possible to immortalise rodent cells able to secrete a monoclonal antibody unique. Thousands of murine monoclonal antibodies are thus produced against antigens. Simultaneously, new materials are developed to immobilise antibodies (microtiter plates, beads). Assays are simplified, become faster and allow the simultaneous analysis of several biological samples.

**1980-1990:** The methods consisting in coupling antibodies to non-isotopic tracers generalise and become more effective. New luminescent or fluorescent tracers amplify signals and contribute to increase the sensitivity of the assays. They offer many advantages over isotopes. In the 1990s, these immunoassay techniques are automated and quickly spread

**Early 2000:** New formats mono- or multivalent antibodies, mono- or bispecific are created by molecular engineering. Directed evolution methods modulate the specificity of these antibodies and adapt to highly specific immunoassays. The technology of phage antibody enables the selection of specific antibodies *In vitro* against any antigen without recourse to prior immunisation of animals.

**2015:** More than 30 recombinant antibodies have been approved for pharmacological treatments. The rapid growth in demand for therapeutic antibody production has been well accommodated by the industrialization of antibody manufacturing. However, the areas of immunoassay and diagnostic have not yet experienced the same growth.

FIGURE 5.1 Antibody and immunoassay over time.

have occurred during or after administration of these molecules owing to their xenogenic nature. In addition, it is now well established that the effector function of the mouse antibody Fc domain is not optimal in humans. Finally, the mouse antibody half-life in human plasma may be too short for pharmacologic effects.

In this context, the past 30 years have seen continuous and progressive innovations in the design of new antibody formats and in vitro selection and maturation of high-affinity antibody fragments specific for any sort of antigens. Complementary DNA (cDNA) cloning and protein expression in recombinant host have enabled us to create new formats of antigen-binding molecules including single-chain antibody fragment (scFv), which are simply made of the light chain variable domain fused via a short peptide linker to the heavy chain variable domain of an antibody. scFvs can be used to design more sophisticated antibody formats often better adapted to specific uses than conventional antibodies [7,8]. In particular, they are templing tools for antibody drug designers because their small size enables the generation of fusion proteins without limiting tissue penetration and with pharmacologic properties optimized for specific applications. Owing to their small size, they have higher tissue permeability, do not elicit Fcassociated cytotoxicity, and still exhibit antigen-binding properties. In addition, combinatorial scFv-phage libraries have been created [9]. They allow in vitro selection of antibody fragments against any antigen. Finally, mice have been genetically engineered to produce fully human antibodies [10]. These transgenic mice are now used as platforms for the discovery of high-affinity antibodies with inherent qualities for successful drug development [11] (Fig. 5.2).



**FIGURE 5.2** Design of therapeutic antibodies: a long quest for more humanity. Therapeutic antibodies can be produced after reengineering murine monoclonal antibodies, whose domains are represented in red (dark grey in print versions) (A), leading to chimeric or humanized antibodies that are less immunogenic, while preserving antigen-binding properties (antibody domains from human origin are represented in white). Alternatively, in vitro panning of phage libraries displaying antibody fragments followed by fusion with human antibodies can also be performed by using transgenic mice (C) or single peripheral blood B-cells and cDNA cloning (D). The suffix used in the international nonproprietary name (INN) for murine, chimeric, humanized, and fully human antibodies are indicated as well as the year of first approval.

Thirty-six antibodies have been marketed in the United States and Europe for therapeutic applications (January 2015). The FDA approved seven in the past year and no slowdown is expected. In 2014, global sales of all approved therapeutic antibodies were US \$78 billion. Sixteen antibodies achieved blockbuster status, each generating a net annual turnover higher than US \$1 billion. In addition, five launches nearly reached sales of US \$1 billion in 2015 year [12].

In the immunodiagnostic sector, inputs of these recombinant technologies are still modest, probably because conventional poly- or monoclonal antibodies all meet current needs. Their immunogenicity is not a drawback for in vitro use and their production is cheap. However, recombinant DNA technologies may offer new resources and suggest the design of innovative reagents with even unsuspected scope.

# 5.2 Natural Antibodies: Structure and Applications

In response to the injection of a foreign antigen, vertebrates produce antibodies, which are proteins secreted by plasma cells, and that serve as effectors of humoral immunity. These antibodies are usually heterogeneous, with each B-cell clone producing a monoclonal antibody that recognizes a particular structural motif (epitope). All of these monoclonal antibodies are part of the polyclonal antibody serum.

The structure of an antibody such as immunoglobulin G (IgG) is well understood from early studies carried out by R. Porter and G. Edelman (Nobel Prize in Medicine, 1972) between 1950 and 1960 [13]. IgG is an Ig monomer consisting of four polypeptide chains: two identical polypeptide chains called heavy (H) of approximately 50 kDa, and two identical light chains (L) (25 kDa) (Fig. 5.3). Disulfide bonds, in a "Y" configuration, join the four chains in which the light chains bracket the heavy chains, starting at the mouth of the "Y" and continuing through the variable region. The nature of the heavy chain ( $\gamma$ ,  $\mu$ ,  $\alpha$ ,  $\delta$ , and  $\varepsilon$ ) determines the "class" of the antibody as IgG, IgM, IgA IgD, or IgE, respectively. Differences in structures and functions in the constant domains of the heavy chains distinguish some subclasses among them (e.g.,  $\gamma$ 1–4 for human IgGs). The light chains consist of two types (kappa or lambda).

Within an IgG, each heavy polypeptide chain (H) or light (L) is organized into domains: a variable domain of about 110 residues at the N-terminus of the light (VL) and heavy (VH) chains, combined with a constant domain (CL) for the light chain or three constant domains for the heavy chain (CH1, CH2, and CH3). In each heavy chain, peptides join the CH1 and CH2 domains. Disulfide bonds between the two peptides create a structure called the hinge region, which gives the molecule great flexibility and the ability to link: for example, two epitopes that are in opposite directions relative to the other [14].

The association of the variable domains VH and VL forms the binding site of an antibody to its target antigen. The sequence variability of the V domains is not equally distributed. There are four regions called framework regions or frameworks (FRs) in which well-conserved residues determine the domain structure (Fig. 5.3). Between the



**FIGURE 5.3** Immunoglobulin G and functional fragments. (A) Monomeric antibody (IgG) made of two light chains and two heavy chains each organized in domains. (B) Bivalent fragment F(ab')<sub>2</sub> obtained after limited enzymatic digestion with pepsin. (C) Monovalent Fabs and Fc fragments obtained after limited enzymatic digestion with papain. (D) Fv variable fragment consisting of the VH and VL domain. Each variable domain is organized in three FRs and three hypervariable loops or CDRs. (E) Three-dimensional structure of the Fv. Hypervariable loops L1, L2, and L3, and H1, H2, and H3 constitute a pocket (paratope) that interacts with the target antigen.

FRs of the same area, there are three "loops" held together in close proximity, also called hypervariable loops or "CDRs," owing to the variability of their particular sequence. The spatial approximation of the loops of both V-domains (six loops) is the paratope, which interacts with the epitope of the target antigen. Despite their high sequence variability, five of the six loops adopt just a small repertoire of main-chain conformations, called "canonical structures" [15]. These conformations are first determined by the length of the loops and second by the presence of key residues at certain positions in the loops and in the FRs that determine the conformation through their packing, hydrogen bonding, or the ability to assume unusual main-chain conformations.

The idea of using antibodies outside the natural humoral immune response was born even before their formal identification. First, von Behring, the first Nobel Prize in Medicine (1901), and Kitasato discovered the concept of serum therapy: that is to say, the treatment of diseases by the injection of blood serum from immune animals [16]. The first therapeutic serum was marketed in the early 1900s for the treatment of infectious diseases (diphtheria and tetanus), and soon after that for the specific treatment of envenomings. Expansion of antibodies as immunoassay tools or diagnostic in clinical biology began in the 1960s with the polyclonal antibodies obtained by immunization animals (rabbits, goats, and sheep) against antigens of human origin. Even if successful, the batch-to-batch reproducibility of polyclonal antibody preparation is a major drawback to their use.

The possibility of producing an infinite homogeneous population of antibodies (monoclonal antibodies) by hybridoma technology was then used in the 1980s to generate an original murine antibody with constant and well-defined specificity and affinity [1].

Today, hybridomas and monoclonal antibodies have become an industrial product, and new concepts and technologies have emerged in biological sciences, in both research and in vitro diagnostics. Thus, immunocytometry has become a key diagnostic tool for blood cell differential analysis combining morphologic characteristics (size and relative granularity) with the presence of immunophenotypic markers. The same antibodies are also exploited in immunohistochemistry to label a particular antigen in tissue. The enzyme immunoassays that use antibodies coupled to a colorimetric enzyme also experienced an extraordinary expansion that is the result of their high sensitivity, the need for simple apparatus, and the possibility of being robotic.

The scope of monoclonal antibodies is wide [17]. However, there are situations in which they are not suited. In the field of in vivo diagnosis, as in therapy, murine monoclonal antibodies will be quickly recognized as a foreign substance by the patient's immune system. The presence of an Fc fragment, which has no role in antigen recognition, and effector functions, which are not effective in humans, is a major drawback. The fragments derived from the proteolysis of antibodies [Fab and F(ab')<sub>2</sub>] as well as IgG (150 kDa) always include constant domains unnecessary in the recognition of the antigen. Their molecular weight remains high, limiting their ability to diffuse into tissues. Finally, the functional qualities of these antibodies are limited by the immune response of the animal that produced them and cannot be changed, altered, or enhanced [18].

# 5.3 Recombinant Antibodies: Wide Range of Alternative Antibody Fragments

Since the early 1990s, the development of molecular cloning techniques and the discovery of polymerase chain reaction have given new insights that led to the identification of antibody genes and their expression by genetically modified organisms. These technologies rapidly led to a wide variety of pieces that can be adapted to new applications (Fig. 5.4) [19].

#### 5.3.1 Simple Monovalent Antibody Fragments

The variable fragment (Fv) of an antibody simply consists of the V-domains of light and heavy chains. Fvs are the minimal antigen-binding structure derived from antibodies.


**FIGURE 5.4** Recombinant antibodies: unedited structures for new applications. (A) Molecular engineering allows designing chimeric antibodies in which domains of human origins substitute for murine constant domains. In humanized antibodies, only the hypervariable loops are of murine origin. (B) Various formats consisting in antibody variable domains can also be created. These include monovalent and monospecific fragments, multivalent and monospecific fragments, bivalent, and bispecific fragments, and finally, bifunctional fragments.

However, these Fvs have rarely been used. They are not obtainable after limited proteolysis of whole antibodies. In addition, they are not stable owing to the lack of a covalent bond between the two antibody domains. Therefore, new monovalent antibody structures were invented. The first single-chain Fv was produced with Cys residues introduced by site-directed mutagenesis into each of the variable domains to induce disulfide cross-domain without altering the specificity of antigen recognition. Today, far more common are scFvs in which the variable domains of an antibody are produced by genetically modified organisms in the form of a fusion protein in which they are joined together by a peptide bond [20]. These molecules were created to facilitate phage display, where it is highly convenient to express the antigen-binding domain as a single peptide. As an alternative, scFv can be created directly from subcloned heavy and light chains derived from a hybridoma. The sequence and length of the peptide linker have to be adapted to balance the stability and functionality of the structure. The linker is usually rich in glycine for flexibility, as well as serine or threonine for solubility, and can either connect the N-terminus of the VH with the C-terminus of the VL, or vice versa. Thus, the scFv represents the basic module from which it is possible to create monovalent or original multivalent antibody formats. It can be produced free or as a fusion protein suitable for a wide range of therapeutic or diagnostic applications.

Some single monomeric variable domains (VL or VH), also called single-domain antibodies (sdAb), have been reported with specific ability to bind to antigen [21]. In addition, new antibody classes consisting of heavy-chain homodimers with no associated light chains have been discovered in camelids and cartilaginous fishes. In both cases, the antigen binding is encapsulated within a single immunoglobulin domain 13-14 kDa in size, also called V<sub>H</sub>H (in camelids) or VNAR (in sharks) containing three or two hypervariable loops, respectively [22,23]. These molecules have been shown to be just as specific as a regular antibody, and in some cases they are more robust. They show great structural similarity to the human VH and are easy to transform into bacterial cells for bulk production, which makes them ideal for research purposes.

#### 5.3.2 Fragments of Monospecific Multivalent Antibody

In the scFv molecule, the length and flexibility of the peptide bridging both V-domains are important from a structural and functional point of view. Thus, substituting the pentadecapeptide  $(Gly_4Ser)_3$  for a pentapeptide  $(Gly_4Ser)$  promotes the formation of dimers in which the VH domain of each subunit joins the VL domain of the other, thus leading to structures called "diabody," monospecific but bivalent, and whose size is equivalent to that of a Fab (50 kDa) [24]. Diabodies have been shown to have dissociation constants up to 40-fold lower than corresponding scFvs, meaning that they have a much higher apparent affinity to their target.

Reducing the linker to a single residue or eliminating it leads to trimeric, trivalent ("triabody") or tetramers, tetravalent ("tetrabody"), with the main consequence being an even higher avidity for the target antigen. Functional analysis in preclinical studies also showed a better retention in target tissues, and lower systemic clearance.

#### 5.3.3 Bispecific Antibody Fragments

The idea to develop bispecific antibodies capable of binding different cells or molecules with potential applications, particularly in the field of cancer immunotherapy, dates back to 1980. The methods described at the time consisted mainly of developing hybrids hybridoma or chemically coupling two Fab derived from the proteolysis of different monoclonal antibodies, but they all experienced failure. However, molecular engineering has allowed the main difficulties to be bypassed [25]. It gives the ability to create bispecific antibody fragments consisting only of the variable domains of two different antibodies. Several arrangements are possible: heterodimeric diabody in which one of the subunits is composed of VH of a first antibody associated with the VL of the second antibody with a different antigen-binding specificity. The second subunit consists of the

VH of the second antibody associated with the VL of the first antibody. Such a structure is compact and rigid. Much more flexible are tandem-scFvs consisting in two scFv associated together with a short peptide bond. Other techniques involve molecular domains that have a propensity to self-assemble. Thus scFv can be produced by fusion with amphipathic  $\alpha$  helices that are coiled together like the strand of a rope or constant domains of antibodies that, by associating with their partner, will lead to bivalent molecules.

#### 5.3.4 Bifunctional Antibody Fragments or Fusion Proteins

Most immunoassays performed make use of antibody molecules previously chemically conjugated to a tracer (a protein or an enzyme). The chemical reactions employed involve nonspecific cross-linking agents, which lead to immunoconjugates whose structure is heterogeneous and tends to aggregate, leading to altered functional properties of one or both partners. For instance, luciferase has never been efficiently coupled to an antibody although its luminescence properties are commonly used in reporter gene techniques.

It is also a reason for the failure of immunotoxins for therapeutic purposes. Today, genetic fusion of an scFv to a protein tracer is possible and offers many advantages: the coupling is complete, equimolar, and homogeneous [26,27]. This homogeneous structure results in quick and simple purification processes. The technology is reproducible and the functional properties of the two entities may be subsequently improved by mutagenesis.

#### 5.3.5 Alternative Nonantibody Scaffolds for Molecular Recognition

Even if antibodies and their recombinant fragments are considered the paradigm for binding proteins, miniproteins are currently developed as alternatives for the generation of novel binding motifs. Usually, these molecules consist of rigid scaffolds that are stabilized by alpha-helices, beta-sheets, or disulfide-constrained secondary structural elements. They are tolerant to multiple substitutions or to the incorporation of a customized affinity domain. Phage display techniques allow the expression of libraries and the selection of specific binders [28,29].

These antibody mimetics belong to various scaffolds also designated as affibodies, DARPins, anticalins, avimers, versabodies, and duocalin.

## 5.4 Main Sources of Recombinant Antibodies

Beside polyclonal and monoclonal antibodies, there are several strategies to obtain recombinant antibodies, two of which are widely used (Fig. 5.5). The first consists of starting from a monoclonal antibody–secreting hybridoma as a source of V-domains. The cDNA is cloned and sequenced. Then, a gene encoding an antibody fragment (e.g., scFv) is synthesized and incorporated into an expression vector, which will be used to



FIGURE 5.5 Sources of antibodies and applications, antibodies specific for an antigen can be prepared after in vivo selection (conventional immunization) (A) or after in vitro selection (biopanning of immune or naïve libraries) (B). Polyclonal antibodies are purified from the serum of immunized animals. Monoclonal antibodies are secreted from immortalized B-cells (hybridoma). Both are widely used for in vitro immunoassays. Recombinant antibodies are usually engineered after reformatting recombinant scFvs. They are mainly two sources of scFvs. First, they derive from well-characterized hybridomas. Alternatively, they are selected after in vitro panning of phage-scFv libraries. In both cases, the strategy can lead to antibodies with therapeutic potential (reformatted chimeric, humanized, or fully human antibodies).

transform host cells before being expressed as a soluble recombinant protein. Although this approach is successful, it remains limited to the original murine antibody for which the secreting hybridoma has been isolated previously [30]. We published a detailed stepby-step protocol allowing cloning and expression of recombinant antigen-binding molecules from hybridoma within 50 working days [31]. Coupled with directed evolution technologies, this strategy permits modulating affinity and antigenic specificity and also "humanizes" the molecule for in vivo applications [32]. In addition, these antibody fragments can easily be reformatted into whole immunoglobulin backbones in a highthroughput independent manner. Most of antibodies approved for therapeutic purposes have been designed according to this strategy. Another technology consists of creating in vitro display methods such as a phage, yeast, and ribosome display [33,34]. The most widely spread technology consists of using combinatorial libraries of antibody fragments exposed on the phage surface and selecting phages expressing the antibody fragment specific to an antigen of interest (Fig. 5.6). This technique, originally developed for the presentation and selection of peptides, has been adapted to recombinant antibody fragments (scFv, Fab, and sdAb) [35]. It consists first of creating a library of antibody fragments, usually scFv, in which the variable domains are associated in random combinations. A VH can be married to any VL, and vice versa. It recreates the diversity of the natural repertoire.

Besides the fact that these libraries can consist of antibody fragments of both human and animal origin, there is interesting in offering highly complex libraries capable of reaching 10<sup>11</sup> combinations. The principle bottlenecks are the transfection efficiency of the host and loss of antibody functionality owing to frameshift mutations [36].

Depending on the origin of the variable domains that will constitute the antibody fragments, there are three types of libraries. "naive" libraries are constructed from cDNA, encoding the antibody variable domains of nonimmunized donors. Antigen-biased libraries are constructed using cDNA from donors previously immunized so that



**FIGURE 5.6** Affinity selection of antigen-specific scFvs from a phage display library. (A) DNA encoding millions of different scFv fragments is cloned into the genome of filamentous bacteriophage linked to one of the phage coat protein genes. (B) Each DNA variant is packed into a separate phage particle, and the antibody fragment displayed on the phage coat protein. (C) Phage displaying scFv that bind to the target antigen are selected using biopanning cycles of binding, washing, and elution. (D) Eluted phages are reinfected into suppressive *E. coli* cells and amplified for further rounds of affinity selection. (E) Clones from the enriched library are characterized for binding properties using appropriate technique. (F) Free scFvs are then produced after transformation of suppressive *E. coli* strain.

antigen-specific immunoglobulin genes are overrepresented. These libraries are mostly of animal origin; they may also come from individuals recovering from infectious diseases or producing autoantibodies. Finally, synthetic or semisynthetic libraries are constructed from synthetic genes in which the diversity of the framework regions is significantly reduced.

A naive repertoire avoids animal immunization and can produce recombinant antibodies in a shorter period than immune or synthetic approaches. However, affinities from these libraries are usually lower. In addition, naive libraries are often subjected to licensing fees with restricted access to protect intellectual property rights. Consequently, most of the time, recombinant antibodies with superior functional properties are produced from immunized animals.

The scFv repertoire subsequently created is cloned into a phage or a phagemid and the library is propagated in suppressive bacteria (Sup E and Sup F). The phage produced by bacteria contains the phagemid that encodes the antibody fragment to be exposed at the surface of a coat protein. Phages exposing at their surface the antibody fragments are then screened against the antigen of interest usually immobilized on a microtitration plate. Bacteriophages presenting an antibody fragment capable of interacting with the antigen on their surface are captured. They are eluted after the plate is washed and the clones selected are recovered by infecting new suppressive bacteria. This step, also known as "biopanning," is repeated four to five times and is used to enrich the population in bacteriophages carrying antibody fragments with affinity for the target antigen; it mimics the natural immune response. Once screening is completed, phagemids contained in the selected bacteriophages are cloned into nonsuppressive bacteria to generate the antibody fragment of interest as a soluble free molecule. It is the presence of an "Amber" codon at the end of the gene encoding the antibody fragment that allows this action.

Advantages of the phage-antibody technology are numerous. First, they offer faster production pipelines relative to conventional monoclonal antibodies. Their production avoids the need for animals, in accordance with legislation on animal protection. This strategy provides greater opportunity for postproduction improvements and alteration by in vitro genetic manipulation. Recombinant antibodies can be generated against extremely potent toxins that would not be possible in vivo [37,38]. Although phage display technology was established more than 20 years ago, and despite widespread use in early-stage antibody discovery, to date only a limited number of approved antibody products have been created via phage display.

## 5.5 Recombinant Antibodies as Emerging Molecules for Pharmacologic Treatments

Hybridoma technology gave rise to the hope for the rapid development of therapeutic applications. Indeed, it offers the opportunity to keep an unlimited amount of murine

antibody with unique specificity and reproducible affinity. Thus the first hybridomaderived murine monoclonal anti-CD3 was introduced into clinical practice in the mid-1980s to prevent organ rejection. However, this strategy faced early failure owing to side effects, mainly immunologic [39]. Thus, modulation of the immune response has been studied by two approaches to reduce the human anti-mouse antibody (HAMA) responses. The first one consists of producing chimeric antibodies that are less immunogenic, in which constant domains of a murine monoclonal antibody are replaced by constant domains of human origin. The second approach is to produce humanized or entire human antibody fragments even less immunogenic. It took a decade for the first chimeric monoclonal antibody, abciximab for hemostasis, to be approved by the FDA in 1994 [40]. The first humanized monoclonal antibody; daclizumab (Zenapax) for kidney transplant rejection, was approved for clinical use by the FDA in 1997 [41]. Humanization alleviated the HAMA response to various degrees, but many other drawbacks became evident. For example, the humanization process is demanding of technology and the process may result in reduced antigen-binding affinity and decreased efficacy. Two major approaches were developed to avoid the human immune response to murine-derived monoclonal antibodies and to overcome the technical challenges associated with humanizing murine monoclonal antibodies. The first is to express human antibody fragments on bacteriophage surfaces [42]. Adalimumab (Humira), the first fully human monoclonal antibody derived from a bacteriophage display antibody library, was approved by the FDA in 2002 for the treatment of rheumatoid arthritis [43]. The second approach was the use of transgenic mice to produce fully human antibodies [44]. Panitimumab (Vectibix), an antiepidermal growth factor receptor antibody approved for colorectal cancer therapy in 2006, was the first fully human therapeutic antibody derived from a transgenic mouse system [45].

Thus, to date, recombinant antibodies have attracted a great deal of attention by the pharmaceutical industry. Antibody therapeutics have reached higher approval success rates and similar development phase lengths compared with those of small-molecule drugs. They have become the centerpiece in the design of targeted therapeutic strategies and constitute a well-accepted class of therapeutics that now covers the fields of cancer, infectious diseases, transplantation, allergy, asthma, and some autoimmune diseases (Fig. 5.7) [46,47].

# 5.6 Recombinant Antibodies: New Tools for Research and Clinical Analysis

The areas of immunoassay and diagnostic use of recombinant antibodies have not experienced the same growth as the pharmaceutical market. The heterologous original murine monoclonal antibody is not a problem for in vitro applications and conventional antibodies currently meet expressed needs.

2014	Vedolizumab, Siltuximab, Ramucirumab, Pembrolizumab, Nivolumab, Blinatumomab, Secukinumab
2013	Objnutuzumah Itolizumah Trastuzumah emtansine Ustekinumah
2013	Ravihacumah Pertuzumah
2011	Brentuximab, Belimumab, Inilimumab, Belatacent
2010	Tocilizumah, Denosumah
2009	Ofatumumah, Canakinumah, Catumaxomah#, Golimumah, Ustekinumah
2008	Certolizumab Pegol. Rilonacent. Rominlostin
2007	Eculizumah
2006	Ranibizumab, Panitumumab, Natalizumab
2005	Metuximab★, Abatacept
2004	Omalizumab, Bevacizumab, Cetuximab, Fanolesomab
2003	Tositumomab <sup>*</sup> , Alefacept#
2002	Adalimumab, Ibritumomab, Efalizumab
2001	Alemtuzumab
2000	Gemtuzumab🛇
1999	
1998	Trastuzumab, Infliximab, Basiliximab, Palivizumab, Etanercept
1997	Rituximab, Daclizumab*, Sulesomab#
1996	Arcitumomab 🛇, Imciromab, Capromab, Nofetumomab
1995	
1994	Abciximab
1993	
1992	Satumomab
1991	
1990	
1989	
1988	
1987	
1986	Muromomab*

**FIGURE 5.7** Timeline of antibodies and immunoadhesins approved for therapeutic application or imagery in humans, Approval limited to the United States (\*), China ( $\star$ ), Europe (#), or withdrawn ( $\odot$ ). Therapeutic class: oncohematology in red, immunology and inflammation in blue, cardiology in brown, ophthalmology in orange, infectiology in green, orphan diseases in pink, transplantation in black, imagery in light blue. Murine (-momab), chimeric (-ximab), humanized (-zumab), fully human (-mumab).

The unique specificity of monoclonal antibodies, the possibility to obtain unlimited amounts of murine antibody with unique specificity and reproducible affinity, to attach on solid supports, or to conjugate with radiolabeled tracers, fluorescent, or enzymatic tracers, led to a wide range of tests that allow in vitro titration of a wide variety of substances.

These tests are rapid and mostly sensitive and can be performed in a reproducible manner. They are used routinely in clinical biology for the titration of hormones, the search for tumor markers, infectious agents, and toxic substances. They are also used for veterinary diagnosis, environmental quality control in the food industry, and many other sectors.

However, there is room for improvement. Phage-antibody technology makes it possible to select scFvs or Fabs specific for nearly any antigen without depending on the immune response of animals, which allows the fields of their applications to be expanded. Difficulties experienced when coupling antibodies to protein tracers by conventional chemical methods could be avoided by adopting gene fusion technology [48]. The gene of a functional antibody fragment can be fused in a reading frame with that of an enzyme, cloned into an expression vector, and introduced into a host cell that will produce the fusion protein. In this case, various host cells have been tested with success, including bacteria, yeast, insect, or mammalian cells.

The overexpression of the fusion protein in bacteria is fast, and certainly, from a practical and economic point of view, the most interesting. Secretion into the bacterial periplasm leads to soluble fusion protein, properly folded and bifunctional with a specific antigen recognition site and an active colorimetric enzyme. Several examples of antibody fragments, scFvs, or Fabs expressed in fusion with enzymes (alkaline phosphatase or acetylcholinesterase), peptides, streptavidin, or fluorescent protein have been designed for the detection and titration of clinically relevant antigens (toxins or hormones) [49]. Internal tags can even be designed [50]. The sensitivity of the immunoassays performed with these fusion proteins readily achieved a few picomoles. This sensitivity can certainly be increased further and reach that of radioimmunoassay. This may require improving the specificity and affinity of the antibody, and also the physical and catalytic properties of enzymes. By combining in vitro selection steps (directed or random mutagenesis) or in vivo, using bacterial strains causing mutations, the affinity of some antibodies has been increased by a factor of 1000. A high-affinity monoclonal antibody for progesterone, which also cross-reacts with steroid structural analogs, has been reformatted by these techniques. This led to an scFv strictly specific for progesterone without loss of affinity, giving it a strong potential to develop a rapid, sensitive, and specific titration kit [51].

Antibody engineering also offers the prospect of developing specific biosensors capable of recognizing an antigen in a heterogeneous medium such as biological fluids, and instantly transduces the event directly into a measurable signal. Thus, Bedouelle et al. established a method of processing recombinant antibody fragments in fluorescent biosensors [52]. They first identified residues of an scFv in the vicinity of the antigen-binding site that may be mutated to cysteine without affecting the antigen recognition function. Cysteine introduced into the mutants with these criteria was then coupled to a fluorophore and the mutants for which the antigen-binding causes a change in the structural environment of the fluorophore were selected. This structural change immediately induces an increase in fluorescence intensity proportional to the antigen concentration. The process has been patented and could be applied to the detection, dosing, and monitoring of many antigens, hormones, markers, or infectious agents in biological fluids using the biosensor solution free or immobilized on a protein array.

The antibodies used in in vivo diagnostics sector remain murine antibodies, even if they are few. Originally, it is their high specificity associated with the possibility of combining them with gamma emitters that stimulated their in vivo use in medical imaging to target specific antigens associated with tumors. A gamma camera is used to detect sites, which concentrate the radiolabeled probes. A multitude of antibodies have been developed for immunoscintigraphy but fewer than a dozen experienced commercial development with limited application in monitoring metastatic colorectal, ovarian, or prostate cancer, and evaluation of myocardiac necrosis consecutive to infarction. Here, the limiting factor is not the potential immunogenicity of these molecules but rather the low contrast generally observed between healthy and pathologic tissues. The size of the probes (whole IgG or their Fab) limits the diffusion and their ability to reach the target antigen embedded in a tissue or a tumor. Their plasma half-life is relatively high even for Fabs that can be cleared by the kidney, so the specific signal–background ratio is low [53].

Studies in animal models have shown the potential benefits of recombinant antibody fragments much more diverse in size and valences. Thus, bivalent diabodies are rapidly eliminated from the plasma compartment and disseminate better to target tissues. The signal—noise ratio is thereby increased and more contrasted images are observed. The sensitivity could be improved by using positron emission tomography or antibodies coupled to fluorescent reagents [54]. However, no molecule has yet been placed on the market, perhaps because of the costs of development and the habits that make immunoscintigraphy unsuitable to be used as the first-line approach, but only in the case of relapses and when results of conventional examinations (scanner or ultrasound) are doubtful.

Another area in which the recombinant antibodies nourish high hopes is that of the proteomic study of biological samples. These hopes are fed in part by the enormous diversity of antibodies that carry combinatorial libraries and the possibilities offered by "microarray" technologies that consist of attaching proteins or antibodies to silicone glass nanosurfaces and developing rapid sensitive and multiplex analysis [55]. Random screening of nonimmune libraries has already demonstrated the ability to clone antibody fragments directed against conserved epitopes or nonimmunogenic antigens that had never been achieved by conventional hybridoma technology. Antibodies directed against the protein of the endoplasmic reticulum B-cells or antibodies capable of distinguishing the posttranslationally activated forms of intracellular proteins as Rab6 or guanosine-5'-triphosphate tubulin are interesting examples [56]. The screening of antibody libraries against proteins extracted from a healthy or diseased tissue by twodimensional electrophoresis and characterized by mass spectrometry could lead to the selection of an antibody of interest (Fig. 5.8). These antibodies could be immobilized at specific positions on nanosurfaces capturing molecules of interest contained in a biological sample, and after washing, immune complexes of various disclosure systems could be implemented with the position of the signal on the chip and its intensity indicating the result for each of the parameters analyzed [57]. Several detection systems are being evaluated, including nanoelectrodes, sandwich reactions involving a second fluorescent antibody, surface plasmon resonance, or matrix-assisted laser desorption/ ionization mass spectrometry. Simultaneous analysis of the healthy and pathologic samples (differential analysis) would provide added value in early diagnosis, prognosis, or therapeutic monitoring of complex pathologies. These objectives could be achieved by using technological alternatives: for example, microfluidic approaches in which the



**FIGURE 5.8** Potential application of recombinant antibodies for global analysis of tissue proteome. (A) The proteomes derived from normal and malignant tissue are isolated. (B) scFv-phage libraries are panned against these proteomes. Highly specific scFv-phages are selected and free scFvs are produced in nonsuppressive bacteria and purified from periplasmic extracts. (C) The antigen-specific scFvs are spotted on the chips. (D) The samples to be analyzed are then allowed to react with the functionalized chip. (E) The readout is shown as a three-dimensional histogram.

natural molecules in solution are captured by antibodies coupled to colored beads and then run in microcapillaries where other specific antibodies have been immobilized. These immune complexes retain color to a preset position, thereby allowing their detection.

## 5.7 Producing Recombinant Antibodies

As consequence of growing interest in recombinant antibodies, many different production systems have been evaluated for their expression, ranging from bacterial hosts to transgenic animals. The choice of the most suitable expression system depends on several factors. The first is the antibody format (whole antibody or nonglycosylated fragment). Other factors that will influence this choice are the scale and total annual production, the application (therapeutic, diagnostic, and experimental tools), and finally the downstream process, whose cost may have a dramatic impact on the product.

#### 5.7.1 Prokaryotic Systems

Microbial expression systems are attractive for the expression of antibody fragments that do not require posttranslational modifications such as phosphorylation and glycosylation. The preferred host is Escherichia coli K12-derived strains, even if several grampositive bacteria have been used for expression such as Bacillus subtilis, Streptomyces *lividans*, and *Staphylococcus carnosus* [58,59]. The expression is typically achieved by fusion to N-terminal signal peptides that address the protein to the periplasmic space of E. coli, where chaperones assist the correct folding and disulfide bridges formation that stabilize the structure (Fig. 5.9). This expression system is attractive. Large biomass density and quick growth can be reached using unexpansive serum-free media. However, recovery of the properly folded and functional antibody fragment strongly depends on the amino acid sequence. Most of the time, large amounts of antibody fragments remain sequestered in the cytoplasm as inclusion bodies. When exported to the periplasm the protein can undergo undesired degradation by periplasmic proteases. In an academic research environment, the yield is rarely higher than a few milligrams of functional antibody fragment per liter of shaken flask cultures, but it is sufficient to allow purification and characterization of early-stage material.

Few therapeutic antibody fragments have been marketed and no more than 50 antibody fragments are currently developed and tested in clinical studies [60,61].



**FIGURE 5.9** Expression of recombinant antibody fragment in *E. coli.* (A) When produced in the cytoplasm, antibody fragments form inclusion bodies and refolding is a tedious step difficult to standardize. (B) Under the control of a signal sequence the unfolded premature antibody fragment is transported across the inner membrane into the periplasm. The protein folds and forms disulfide bridges in the oxidative environment of the periplasm. (C) Using engineered strain, it becomes possible to translocate the correctly folded native protein into the extracellular medium. This strategy greatly simplifies the downstream process.

However, these studies have contributed to develop new *E. coli* expression technologies that allow the secretion of recombinant proteins in their native form directly into the culture medium. The expression yield can reach a couple of grams per liter, which is achieved without cell disruption. These features contribute to simplify the downstream process and reduce the amount of endotoxins that are considered to be the biggest drawback compared with nonbacterial expression technologies [62].

#### 5.7.2 Yeast Systems

Yeast expression systems have been used for therapeutic protein expression since the early 1980s. They grow quickly in well-defined medium in the absence of animal-derived growth factors and can secrete large amounts of recombinant protein. In this way, they constitute a valuable alternative when antibody fragments are difficult to express in bacteria. Until 2009, several biopharmaceutical proteins were produced in Saccharomyces cerevisiae, but other expression systems were also successfully developed independently using the methylotrophic yeast *Pichia pastoris* [63,64]. Yeast are even more robust than E. coli in cell culture and can grow to high cell densities, which increases the yield per a given volume of cell suspension. Thus, protein production tends to be faster and cheaper than in mammalian systems. Typical expression levels after 120 h fermentation processes can reach 2-5 g/L for a wide range of proteins including Fab fragments. The main advantages of yeasts over bacterial expression systems are the absence of endotoxins and their ability to secrete recombinant proteins into the culture medium using an eukaryotic protein-synthesis pathway. In addition, protein folding and posttranslational modifications can occur in *Pichia* so that it could be used to express glycoproteins. However N-linked glycosylation patterns differ significantly from that of humans, which is a major drawback of yeasts compared with mammalian cells.

#### 5.7.3 Mammalian Cell Systems

Mammalian cell expression systems have emerged as the method of choice for the production of whole monoclonal antibodies because they support expression of larger proteins than can bacteria (Fig. 5.10). They allow assembling multidomain proteins and the addition of sugar motifs that profoundly affect the function of the Fc fragment. Patterns of glycosylation are highly species-specific and mammalian cells have glycosylation repertoires most similar to humans.

Human embryo kidney (HEK-293) has long been considered to be the host of choice for transient expression of recombinant proteins. Improvements in media composition and host cell engineering have led to generate higher-yielding transient transfection systems based on suspension-adapted Chinese hamster ovary (CHO) K1 cells [65]. These systems allow the production of up to 100 mg/L of antibody within 10 days after transient transfection. However, protein expression remains an empirical process and results are never guaranteed even with the most seemingly appropriate expression system.



FIGURE 5.10 Overview of the most popular systems for the production of recombinant antibodies and antibody fragments.

In the biopharmaceutical industry, CHO has become the workhorse for the production of whole IgGs in front of other cell lines including mouse myeloma (NSO), HEK-293, and human embryonic retinoblast (PER.C6) [66,67]. Companies have mainly focused on CHO and have been able to optimize industrial upstream processes to reach typical titers ranging from 1 to 5 g/L after several days. Concerning expression of antibody fragments (Fab and scFv), interest in mammalian cells remains questionable. Indeed, they are more difficult to culture than are microbial cells and protein production tends to be slower and more expensive in mammalians versus microbial systems. In addition to the complexity of the protein folding, other points have to be considered, including the downstream process, whose cost may have a dramatic impact on the product. In this way, the relative purity of the protein secreted in CHO cell culture medium may be advantageous compared with *E. coli* periplasmic extracts but viral clearance is absolutely required and cost-effective.

#### 5.7.4 Alternative Expression Systems

To date, approved biopharmaceuticals are produced in a limited number of expression systems, especially in *E. coli*, *P. pastoris*, and CHO cells. However, unconventional systems are catching up. An interesting one is the use of transgenic animals as bioreactors. They are capable of posttranslational modifications that closely match those of human proteins and offer most economical efficient technological models of production.

Recombinant antibodies were produced in transgenic animals for the first time at the end of the 1990s and many companies are actively working on developing this technology [68]. So far, antibodies have been produced mainly in transgenic goats at a relatively high concentration (0.5–10 mg/mL) without affecting the health and lactation of animals. These antibodies are stable and highly efficient, and one can estimate that herds of several dozen of transgenic goats whose milk contains a specific monoclonal antibody could fully meet the world demand for the treatment of most pathologies [69]. One of the main challenges in this strategy will be to develop an economically viable downstream process, because milk is a complex mixture that can still be infected with pathogenic agents.

#### 5.8 Conclusions and Perspectives

The possibilities offered by molecular engineering to adapt the format and valence o recombinant antibodies to specific applications, increase their affinity, and modulate their specificity will greatly expand the range of their applications. First, as a therapeutic agent: nearly 40 molecules received marketing authorization in areas as diverse as infectious diseases, oncology, hematology, and treatment of cardiovascular, inflammatory, or autoimmune diseases. Nearly 500 are currently in clinical trials. The number of antibody fragments remains much lower. Only three Fabs have yet been marketed: Abciximab (Reopro) for thrombosis; Ranibizumab (Lucentis) for wet age-related macular degeneration; and Certolizumab pegol (Cimzia) for Crohn disease.

In the immunoassay sector, although conventional monoclonal antibodies remain essential tools, the panel of their targets is limited to a few thousand proteins. This sector should quickly take advantage of technological progress and see if its scope can be broadened.

Altering the intrinsic properties of an antibody (size, affinity, and specificity) is possible even if the methods used are still difficult to generalize. The preparation of recombinant bifunctional immunoconjugates that are homogeneous in structure and perfectly adapted to the requirements of the immunoassay is a reality. Intensive research is carried out in this area and recombinant immunoenzymatic conjugates will soon appear on the diagnostic market. Through in vitro selection technologies from combinatorial libraries, probes specific to various antigens may be isolated without immunizing animals. Thus the repertoire of available antibodies increases significantly regardless of the target proteins, including poorly immunogenic proteins. The microfluidics and the ability to create recombinant antibody arrays pave the way to the development of rapid assays with simultaneous detection of several predetermined antigens in biological samples. Chips could then be made to determine the protein expression profiles of healthy and cancerous cells, discover new biomarkers and the molecular mechanisms of diseases, or identify new therapeutic targets.

### References

- G. Kohler, C. Milstein, Continuous cultures of fused cells secreting antibody of predefined specificity, Nature 256 (1975) 495–497.
- [2] A. Bradbury, A. Plückthun, Reproducibility: standardize antibodies used in research, Nature 518 (2015) 27–29.
- [3] M. Baker, Reproducibility crisis: blame it on the antibodies, Nature 521 (2015) 274–276.
- [4] O. Leavy, Therapeutic antibodies: past, present and future, Nature Reviews Immunology 10 (2010) 297.
- [5] P. Chames, M. Van Regenmortel, E. Weiss, D. Baty, Therapeutic antibodies: successes, limitations and hopes for the future, British Journal of Pharmacology 157 (2009) 220–233.
- [6] C. Sgro, Side-effects of a monoclonal antibody, muromonab CD3/orthoclone OKT3: bibliographic review, Toxicology 105 (1995) 23–29.
- [7] E. Laffl, R. Sodoyer, Monoclonal and recombinant antibodies, 30 years after..., Human Antibodies 14 (2005) 33-55.
- [8] P.J. Hudson, C. Souriau, Engineered antibodies, Nature Medicine 9 (2003) 129–134.
- [9] C. Marks, J.D. Marks, Phage libraries a new route to clinically useful antibodies, The New England Journal of Medicine 335 (1996) 730–733.
- [10] N. Lonberg, Human monoclonal antibodies from transgenic mice, Handbook of Experimental Pharmacology 181 (2008) 69–97.
- [11] L.L. Green, Transgenic mouse strains as platforms for the successful discovery and development of human therapeutic monoclonal antibodies, Current Drug Discovery Technology 1 (2014) 74-84.
- [12] J.M. Reichert, Antibodies to watch in 2015, MAbs 7 (2015) 1-8.
- [13] T.N.K. Raju, Structural studies of immunoglobulins, The Lancet 354 (1999) 1040.
- [14] R.A. Goldsby, T.J. Kindt, B.A. Osborne, Kuby Immunology, sixth ed., WH Freeman & Co., 2006.
- [15] C. Chothia, A.M. Lesk, Canonical structures for the hypervariable regions of immunoglobulins, Journal of Molecular Biology 196 (1987) 901–911.
- [16] E. Behring, S. Kitasato, Ueber das Zustandekommen der Diphterie-Immunitaet und der Tetanus-Immunitatet bei Thieren, Deutsch Med Wochenschr (1890) 1113–1114.
- [17] W.J. Payne Jr., D.L. Marshall, R.K. Shockley, W.J. Martin, Clinical laboratory applications of monoclonal antibodies, Clinical Microbiology Reviews 1 (1988) 313–329.
- [18] L. Riechmann, M. Clark, H. Waldmann, G. Winter, Reshaping human antibodies for therapy, Nature 332 (1988) 323–327.
- [19] P. Holliger, P.J. Hudson, Engineered antibody fragments and the rise of single domains, Nature Biotechnology 23 (2005) 1126–1136.
- [20] S. Farajnia, V. Ahmadzadeh, A. Tanomand, K. Veisi, S.A. Khosroshahi, L. Rahbarnia, Development trends for generation of single-chain antibody fragments, Immunopharmacology Immunotoxicology 36 (2014) 297–308.
- [21] W. Chen, Z. Zhu, X. Xiao, D.S. Dimitrov, Construction of a human antibody domain (VH) library, Methods in Molecular Biology 525 (2009) 81–99.
- [22] S.D. Nuttall, Overview and discovery of IgNARs and generation of VNARs, Methods in Molecular Biology 911 (2012) 27–36.
- [23] S. Muyldermans, Nanobodies: natural single-domain antibodies, Annual Review of Biochemistry 82 (2013) 775–797.

- [24] O. Dolezal, L.A. Pearce, L.J. Lawrence, A.J. McCoy, P.J. Hudson, A.A. Kortt, ScFv multimers of the anti-neuraminidase antibody NC10: shortening of the linker in single-chain Fv fragment assembled in V(L) to V(H) orientation drives the formation of dimers, trimers, tetramers and higher molecular mass multimers, Protein Engineering 13 (2000) 565–574.
- [25] C. Grandjenette, M. Dicato, M. Diederich, Bispecific antibodies: an innovative arsenal to hunt, grab and destroy cancer cells, Current Pharmaceutical Biotechnology 16 (2015) 670–683.
- [26] M. Mousli, I. Turki, H. Kharmachi, M. Saadi, K. Dellagi, Recombinant single-chain Fv antibody fragment-alkaline phosphatase conjugate: a novel in vitro tool to estimate rabies viral glycoprotein antigen in vaccine manufacture, Journal of Virological Methods 146 (2007) 246–256.
- [27] L.M. Alvarenga, J. Muzard, A. Ledreux, C. Bernard, P. Billiald, Colorimetric engineered immunoprobe for the detection and quantification of microcystins, Journal of Immunological Methods 406 (2014) 124–130.
- [28] A. Skerra, Alternative non-antibody scaffolds for molecular recognition, Current Opinion in Biotechnology 18 (2007) 295–304.
- [29] F. Zoller, U. Haberkorn, W. Mier, Miniproteins as phage display-scaffolds for clinical applications, Molecules 16 (2011) 2467–2485.
- [30] N.E. Weisser, J.C. Hall, Applications of single-chain variable fragment antibodies in therapeutics and diagnostics, Biotechnology Advances 27 (2009) 502–520.
- [31] C. Fields, D. O'Connell, S. Xiao, G. Lee, P. Billiald, J. Muzard, Creation of recombinant antigenbinding molecules derived from hybridomas secreting specific antibodies, Nature Protocols 8 (2013) 1125–1148.
- [32] Y. Safdari, S. Farajnia, M. Asgharzadeh, M. Khalili, Antibody humanization methods a review and update, Biotechnology & Genetic Engineering Reviews 29 (2013) 175–186.
- [33] U. Schmitz, A. Versmold, P. Kaufmann, H.G. Frank, Phage display: a molecular tool for the generation of antibodies–a review, Placenta 21 (2000) S106–S112.
- [34] M.J. Feldhaus, R.W. Siegel, Yeast display of antibody fragments: a discovery and characterization platform, Journal of Immunological Methods 290 (2004) 69–80.
- [35] A.R. Bradbury, J.D. Marks, Antibodies from phage antibody libraries, Journal of Immunological Methods 290 (2004) 29–49.
- [36] D. Ponsel, J. Neugebauer, K. Ladetzki-Baehs, K. Tissot, High affinity, developability and functional size: the holy grail of combinatorial antibody library generation, Molecules 16 (2011) 3675–3700.
- [37] H.R. Nejad, J.F. Mehrabadi, J. Nourozi, M. Mahdavi, Construction of scFv library of human antibodies against tetanus toxin, Human Antibodies 23 (2014) 21–26.
- [38] S. Miethe, C. Rasetti-Escargueil, Y. Liu, S. Chahboun, T. Pelat, A. Avril, A. Frenzel, T. Schirrmann, P. Thullier, D. Sesardic, M. Hust, Development of neutralizing scFv-Fc against botulinum neurotoxin A light chain from a macaque immune library, MAbs 6 (2014) 446–459.
- [39] S.L. Smith, Ten years of Orthoclone OKT3 (muromonab-CD3): a review, Journal of Transplant Coordination 6 (1996) 109–119.
- [40] D. Faulds, E.M. Sorkin, Abciximab (c7E3 Fab). A review of its pharmacology and therapeutic potential in ischaemic heart disease, Drugs 48 (1994) 583–598.
- [41] F. Vincenti, R. Kirkman, S. Light, G. Bumgardner, M. Pescovitz, P. Halloran, J. Neylan, A. Wilkinson, H. Ekberg, R. Gaston, L. Backman, J. Burdick, Interleukin-2-receptor blockade with daclizumab to prevent acute rejection in renal transplantation. Daclizumab Triple Therapy Study Group, The New England Journal of Medicine 338 (1998) 161–165.
- [42] T.J. Vaughan, A.J. Williams, K. Pritchard, J.K. Osbourn, A.R. Pope, J.C. Earnshaw, J. McCafferty, R.A. Hodits, J. Wilton, K.S. Johnson, Human antibodies with sub-nanomolar affinities isolated from a large non-immunized phage display library, Nature Biotechnology 14 (1996) 309–314.

- [43] M.E. Weinblatt, E.C. Keystone, D.E. Furst, L.W. Moreland, M.H. Weisman, C.A. Birbara, L.A. Teoh, S. A. Fischkoff, E.K. Chartash, Adalimumab, a fully human anti-tumor necrosis factor alpha monoclonal antibody, for the treatment of rheumatoid arthritis in patients taking concomitant methotrexate: the ARMADA trial, Arthritis and Rheumatism 48 (2003) 35–45.
- [44] N.D. Russell, J.R. Corvalan, M.L. Gallo, C.G. Davis, L. Pirofski, Production of protective human antipneumococcal antibodies by transgenic mice with human immunoglobulin loci, Infection and Immunity 68 (2000) 1820–1826.
- [45] L. Saltz, C. Easley, P. Kirkpatrick, Panitumumab, Nature Reviews Drug Discovery 5 (2006) 987–988.
- [46] J.M. Reichert, Metrics for antibody therapeutics development, MAbs 6 (2010) 695–700.
- [47] A.L. Nelson, J.M. Reichert, Development trends for therapeutic antibody fragments, Nature Biotechnology 27 (2009) 331–337.
- [48] C. Lindbladh, K. Mosbach, L. Bülow, Use of genetically prepared enzyme conjugates in enzyme immunoassay, Trends in Biochemical Sciences 18 (1993) 279–283.
- [49] N. Aubrey, C. Devaux, E. di Luccio, M. Goyffon, H. Rochat, P. Billiald, A recombinant scFv/ streptavidin-binding peptide fusion protein for the quantitative determination of the scorpion venom neurotoxin, Biological Chemistry 382 (2001) 1621–1628.
- [50] J. Muzard, S. Adi-Bessalem, M. Juste, F. Laraba-Djebari, N. Aubrey, P. Billiald, Grafting of protein L-binding activity onto recombinant antibody fragments, Analytical Biochemistry 388 (2009) 331–338.
- [51] O. Dubreuil, M. Bossus, M. Graille, M. Bilous, A. Savatier, M. Jolivet, A. Ménez, E. Stura, F. Ducancel, Fine tuning of the specificity of an anti-progesterone antibody by first and second sphere residue engineering, The Journal of Biological Chemistry 26 (2005) 24880–24887.
- [52] H. Bedouelle, M. Renard, L. Belkadi, P. England, Biosensors, Process for Obtaining Them and Their Applications, 2007. Patent n° EP1742055.
- [53] A.M. Wu, P.D. Senter, Arming antibodies: prospects and challenges for immunoconjugates, Nature Biotechnology 9 (2005) 1137–1146.
- [54] T. Olafsen, S.J. Sirk, S. Olma, C.K. Shen, A.M. Wu, ImmunoPET using engineered antibody fragments: fluorine-18 labeled diabodies for same-day imaging, Tumour Biology 3 (2012) 669–677.
- [55] C. Wingren, C.A. Borrebaeck, Antibody microarrays: current status and key technological advances, OMICS 10 (2006) 411–427.
- [56] A. Dimitrov, M. Quesnoit, S. Moutel, I. Cantaloube, C. Poüs, F. Perez, Detection of GTP-tubulin conformation in vivo reveals a role for GTP remnants in microtubule rescues, Science 322 (2008) 1353–1356.
- [57] C.A. Borrebaeck, Antibodies in diagnostics from immunoassays to protein chips, Immunology Today 21 (2000) 379–382.
- [58] O. Spadiut, S. Capone, F. Krainer, A. Glieder, C. Herwig, Microbials for the production of monoclonal antibodies and antibody fragments, Trends in Biotechnology 32 (2014) 54–60.
- [59] K. Graumann, A. Premstaller, Manufacturing of recombinant therapeutic proteins in microbial systems, Biotechnology Journal 2 (2006) 164–186.
- [60] A. Plückthun, Mono- and bivalent antibody fragments produced in *Escherichia coli*: engineering, folding and antigen binding, Immunological Reviews 130 (1992) 151–188.
- [61] C. Chen, B. Snedecor, J.C. Nishihara, J.C. Joly, N. McFarland, D.C. Andersen, J.E. Battersby, K.M. Champion, High-level accumulation of a recombinant antibody fragment in the periplasm of *Escherichia coli* requires a triple-mutant (degP, prc spr) host strain, Biotechnology and Bioengineering 85 (2004) 463–474.

- [62] G. Seidel, S. Di Cesare, Improving primary recovery. Simple and efficient platform approach reduces impurity load for downstream processing, Genetic Engineering & Biotechnology News 31 (2011) 1–3.
- [63] B. Gasser, M. Maurer, J. Gach, R. Kunert, D. Mattanovich, Engineering of *Pichia pastoris* for improved production of antibody fragments, Biotechnology and Bioengineering 94 (2006) 353–361.
- [64] A.M. Gonçalves, A.Q. Pedro, C. Maia, F. Sousa, J.A. Queiroz, L.A. Passarinha, *Pichia pastoris*: a recombinant microfactory for antibodies and human membrane proteins, Journal of Microbiology and Biotechnology 23 (2013) 587–601.
- [65] V. Agrawal, B. Yu, R. Pagila, B. Yang, C. Simonsen, O. Beske, A high-yielding, CHO-K1-based transient transfection system rapid production for therapeutic protein development, BioProcess International 11 (2013) 1–3.
- [66] M. Chartrain, L. Chu, Development and production of commercial therapeutic monoclonal antibodies in mammalian cell expression systems: an overview of the current upstream technologies, Current Pharmaceutical Biotechnology 9 (2008) 447–467.
- [67] T. Jostock, H.P. Knopf, Mammalian stable expression of biotherapeutics, Methods in Molecular Biology 899 (2012) 227–238.
- [68] L.M. Houdebine, Antibody manufacture in transgenic animals and comparisons with other systems, Current Opinion in Biotechnology 13 (2002) 625–629.
- [69] O.G. Maksimenko, A.V. Deykin, Y.M. Khodarovich, P.G. Georgiev, Use of transgenic animals in biotechnology: prospects and problems, Acta Naturae 5 (2013) 33–46.

## Impacts of Aptamer Technology on Diagnostics, Biotechnology, and Therapy

J.C. Corrêa-Velloso, I.C. Nascimento, A.A. Nery, H. Ulrich\* DEPARTMENT OF BIOCHEMISTRY, INSTITUTE OF CHEMISTRY, UNIVERSITY OF SÃO PAULO, SÃO PAULO, SP, BRAZIL

## 6.1 Systematic Evolution of Ligands by Exponential Enrichment Technique

The systematic evolution of ligands by exponential enrichment (SELEX) technique describes the reiterative screening of a combinatorial RNA or single-stranded (ss) DNA library for binding affinity and specificity against desired target molecules, followed by reverse transcription—polymerase chain reaction (RT-PCR) or PCR amplification of eluted target binders, renewing an already selected library for the next cycle. The SELEX process mimics natural selection, in which best-fitting species are amplified for the next generation. RNA or DNA molecules with high target-binding affinity and specificity obtained after completion of various *in vitro* selection cycles are denominated aptamers (from Latin *aptus*, which means "to fit").

The basic design of the initial library is similar to both RNA and DNA selection processes using a DNA template composed of a random inner region (20–70 random nucleotides) flanked by constant sequences containing the primer annealing sequences as well as RNA polymerase initiation sites (in case of RNA aptamer selection). For DNA aptamer selection, the double-stranded (ds) pool is denatured into its single strands, whereas for RNA aptamer selection, the dsDNA pool needs first to be *in vitro* transcribed into RNA.

For the formation of secondary and tertiary structures of ssDNA or RNA, pools are denatured and then presented to their targets, to which they bind in a sequence- and structure-specific manner.

Following the regular flow of the selection cycles and removal of unbound DNA or RNA molecules, target binders are eluted and amplified by RT-PCR or PCR procedures to restore the DNA library for the next cycle of *in vitro* selection. After consecutive SELEX

<sup>\*</sup>Corresponding Author.

cycles, the heterogeneous random library containing up to  $10^{15}$  different sequences is purified to a homogeneous population of aptamers binding with high affinity and specificity to their selection target. Individual aptamers are identified from this final pool by DNA sequencing, and the obtained aptamers are characterized for binding affinity and biological activity (Fig. 6.1).

Resistance against nuclease activity can be achieved during the SELEX process by enzymatic incorporation of modified nucleotides using PCR or in vitro transcription reaction [1,2]. Stability and specificity of already identified aptamers can be further optimized by post-SELEX modifications, a particular feature of the SELEX process that is feasible to introduce modifications after selection. These modifications include truncation of the sequences, incorporation of modified bases by chemical synthesis, and attachment of biotin moieties and fluorescence reporters.



**FIGURE 6.1** In vitro selection of high-affinity aptamers for *in vivo* applications. DNA and RNA aptamers are selected from a randomized oligonucleotide library by reiterative SELEX rounds. In case of the DNA aptamer, the double-stranded DNA pool needs to be denatured for purification of the single-stranded sense strand, which is then presented to its selection target. After removal of unbound and low-affinity bound DNA molecules, target-bound DNA molecules are eluted and amplified by PCR in the presence of a biotinylated primer. After denaturation, the unbiotinylated sense strand can be purified by polyacrylamide gel electrophoresis and used for the next SELEX cycle. In the case of development of an RNA aptamer, the double-stranded DNA pool is *in vitro* transcribed to the RNA pool, which is then used for SELEX. Eluted target binders are reversed-transcribed to cDNA, amplified by PCR, and again *in vitro* transcribed to give the RNA pool used in the next SELEX cycle. RNA and DNA molecules can be protected against nuclease attacks during the selection process by incorporation of modified nucleotides using enzymatic reactions, or they can be chemically modified after the SELEX process. *HPLC*, high-performance liquid chromatography; ss, single-stranded; ds, double-stranded.

The origin of SELEX studies goes back to the work of the research groups of Larry Gold and Jack Szostak [3,4], who first identified RNA molecules specifically interacting with T4 bacteriophage DNA polymerase or small molecules such as adenosine triphosphate (ATP). In fact, Ellington and Szostak showed that the selected RNA aptamers bound to ATP but not to related nucleotides such as deoxyadenosine triphosphate, indicating that this technology would generate more specific ligands than those present in nature.

These original works suggested that high-affinity RNA ligands could be developed for any target molecule. In the following years, subsequent studies revealed that aptamers possess target recognition features, which are similar to those of antibodies [5]. Because they are highly specific molecules, aptamers are able to distinguish between isoforms of the same protein, i.e., protein kinase C [6]. Moreover, they are able to distinguish by the matter of 10,000-fold affinity between theophylline and caffeine; the only difference between the two molecules is a methyl group at the nitrogen atom N-7 [7]. Aptamers are even able to distinguish different conformational forms of the same protein [8–10]. These capabilities position oligonucleotide molecules as potential candidates for replacing the antibody technology in several applications.

Aptamers are being developed into novel tools, such as therapeutics and/or diagnostic agents against a broad spectrum of bioactive peptides and growth factors, with tremendous impact on therapeutic and diagnostic approaches. As a result of these developments, starting in the early 1990s, the most prominent molecule was a therapeutic aptamer targeting the isoform 165 of vascular endothelial growth factor (VEGF), which was approved for the treatment of disease-related neovascularization in macular eye disease, commercially named MACUGEN [11,12]. A new aptamer aimed at ophthalmologic applications has been developed targeting platelet-derived growth factor (PDGF), commercially named FOVISTA [13].

SELEX applications were extended from targeting soluble proteins to complex targets, which are comprehensive structures such as transmembrane proteins needing the membrane environment to maintain activity, or entire membranes or even whole cells [14–18]. Remarkably, the use of complex targets for aptamer selection is strategically advantageous over targeting purified proteins, because there is no concern regarding conformational changes, which may occur in purified or recombinant-expressed cell surface proteins [19]. Cell SELEX, an improvement of the original technique, exploits differences in surface molecule expression between any two types of cells. Consequently, panels of aptamers can be obtained, which specifically recognize the molecular signature of the target cell type. A specific probe for a distinct cell type can be obtained with no knowledge of molecular differences between, for example, cancer cells or parasite-infected cells and normal cells.

For these applications, the combinatorial RNA or ssDNA library is incubated with target cells. Unbound oligonucleotides are removed by washing steps, and bound sequences are eluted from the cell surfaces and amplified by RT-PCR or PCR for subsequent SELEX cycles. To overcome limitations in aptamer selection owing to unspecific binding of RNA or DNA pools to sites on cell surfaces other than the desired target site, subtractive SELEX cycles have been established. The preselected RNA or DNA pool is exposed to control cells and only the unbound RNA or DNA fractions are used for the next SELEX cycle.

Therefore, all sequences bound to cell surface epitopes commonly expressed by both cell types have been discarded. After various cycles of target selection and subtraction rounds, the final SELEX pool consists of a homogeneous population of aptamers exclusively binding to cells used as targets for *in vitro* selection [19]. Identified aptamers can then be employed as capture agents for cell separation and identification of their target proteins on the cell surface [20].

The basis of aptamer-target recognition relies on the fact that SELEX carried out with RNA or DNA combinatorial libraries consists of up to  $10^{15}$  different sequences. As result, such molecules can fold into a vast number of different complex structures, with a high probability that from this library of possible structures an RNA molecule with an appropriate secondary structure can be identified for any given target protein.

Aptamers can also be developed from ssDNA libraries, although secondary structures formed by ssDNA are less stable than those of RNA molecules [21]. DNA aptamers have some advantages over RNA aptamers, because there is less need for chemical modifications to grant nuclease resistance. Furthermore, although they are synthesized by the PCR reaction, modifications can easily be introduced using modified primers.

Aptamers are known to interact with their selection targets with micro- to femtomolar binding affinities and identify the sequence and conformation of protein surface features (epitopes) such as antibodies do. Interactions between aptamers and a peptide target are disrupted by sequence substitutions [5].

Sequence-specific recognition result from the establishment of hydrogen (H-) bonds between bases of ribonucleotides or deoxyribonucleotides and amino acid side chains of the target peptide or protein. Structure-specific recognition results from the formation of H-bonds with oligonucleotide backbone sugars or phosphates and the protein chain. Furthermore aptamers are capable to bind their targets with different solution conformations [5], ensuring optimal molecular shape complementarities [22].

This structural plasticity enables small target molecules to be incorporated into the aptamer structure, whereas in the case of binding to larger proteins, aptamers integrate into binding pocket of the protein. This mechanism of induced fit also provides a mechanism for the observation that hidden target sites, which cannot be accessed by antibodies, are accessible for aptamers.

## 6.2 Therapeutics Features of Aptamers

Over the past three decades, aptamers have been generated against hundreds of molecular targets. DNA and RNA aptamers have been selected against proteins of therapeutic interest, such as cytokines, proteases, kinases, immunoglobulins, gene regulatory factors, cell-surface receptors, cell-adhesion molecules, and growth factors (Table 6.1). Moreover, aptamers were also developed against pathogenic bacteria [23],

Target	Dissociation Constant (K <sub>d</sub> ) (nM)	Possible Therapeutic Application	References
Growth Factor, Hormones			
Fibroblast growth factor 2, basic	0.35	Prevents angiogenesis	[31]
Substance P	40	Prevents tumor development	[32]
Epidermal growth factor receptor	2.4	Prevents tumor development	[33]
Gonadotropin-releasing membrane hormone 1	50	Prevents tumor development	[34]
Receptors			
Cocaine binding site of the nicotinic acetylcholine receptor	2 and 12	Prevents toxic effects of cocaine	[14]
Receptor tyrosine kinase RETC634Y	35	Prevents tumor development	[35]
Neurotensin receptor 1	1.5	Prevents tumor development/	[36]
		manages pain	
Nociceptin	110	Manages pain	[37]
Viral Proteins			
HIV-1 reverse transcriptase	1	Inhibits viral replication	[38]
HIV-1 rev	<1	Inhibits viral replication	[39]
HIV-1 integrase	10	Inhibits viral replication	[40]
HIV gp120	5	Inhibits viral infectivity	[41]
NS3 protease	10	For treatment of hepatitis	[42]
		C viral infection	
Coagulation Regulatory Proteins			
α-thrombin	25	Prevents thrombosis	[43]
Factor IXa	0.65	Prevents thrombosis	[44]
Cell Adhesion Molecules			
L-selectin	3	Modulates inflammation	[45]
Epithelial cell-adhesion molecule	38	Prevents tumor development	[46]
Pathogenic Proteins			
Acetylcholine-specific autoantibodies	60	For treatment of myasthenia gravis	[47]
Bovine prion protein	6.8	For treatment of prion and	[48]
		Alzheimer disease	
Whole Organism			
Trypanosoma cruzi	40-400	Diagnostics and treatment of	[49]
		Chagas disease	
Trypanosoma brucei	60	Diagnostic and treatment of	[50]
		"sleeping sickness"	

## Table 6.1 Dissociation Constants ( $K_d$ ) of Aptamers Binding to Targets of Therapeutic Interest

viral particles [24] and cancer cells [25], and are able to distinguish tumor tissue such as microvessels of rat glioblastoma from normal tissue [26]. The use of aptamers instead of antibodies is arising in almost every application, for which they are suitable, because they have many advantages over antibodies in engineering and medicine applications:

- Production and quality control: The *in vitro* method for nucleic acid aptamer selection is simple compared with the complex *in vivo* antibody production process [27]. No organisms are required for aptamer production, resulting in no batch-to-batch variation among aptamers for a specific target. In addition, aptamers are chemically produced in a readily scalable process, which renders products, which are not susceptible to viral or bacterial contamination [28].
- **2.** *Ligand specificity and affinity:* It has been shown that aptamers often exhibit high affinities toward the selected target compared with antibodies and often yield dissociation constants ( $K_d$ ) of ligand-target complexes in the nano- or sub-nanomolar range (Table 6.1).
- **3.** *Immunogenicity and toxicity:* Aptamers are reported to be low-immunogenic and low-toxic molecules. Whereas the efficacy of many monoclonal antibodies often depends on the antigen's ability to trigger an immune response, nucleic acid fragments are not typically recognized by the human immune system as nonself entities [29].
- **4.** *Variety of targets:* Molecules that do not elicit strong immune response, such as toxins, ions, or small molecules, are not eligible for antibody production. However, aptamers can be generated with high affinity and specificity against targets that are not recognized by antibodies [30].
- **5.** *Stability:* Oligonucleotides are more thermally stable and maintain their structures over repeated cycles of denaturation/renaturation whereas antibodies are easily denatured and lose their tertiary structure at high temperatures. Aptamers recover their native conformation and can bind to targets after reannealing, whereas antibodies easily undergo irreversible denaturation [27]. In addition, aptamers can be stored for extended periods (more than 1 year) as lyophilized powders [30]. These biochemical features of oligonucleotide aptamers allow for their use in a wide range of assay conditions.

Under biological conditions, unmodified RNA and DNA molecules are prone to nuclease-mediated degradation. Because oligoribonucleotides are particularly susceptible to the attack of nucleases present both, in the serum and inside cells, the development of chemical modifications to improve aptamer stability has been crucial for biological applications. Accordingly, modified nucleotides can be introduced into libraries before the initiation of SELEX by polymerases that accept modified nucleotide triphosphates as substrates. This modification step yields aptamers, whose nucleotide sequences are partially or completely substituted with one or more modifications. Among the most commonly used modifications are 2'-amino pyrimidines [51,52], 2'-fluoropyrimidines [53,44], and 2'-O-methyl ribose pyrimidines and purines [54,55].

Another modification is composed by internucleotide linkages often inserted at the 5'-ends of aptamers, such as high-molecular mass polyethylene glycol conjugation [56,57]. Such modifications result in an enhanced resistance to nuclease attacks, improving aptamers' half-lifes in biofluids from minutes to several hours [30]. In rat plasma, the half-life of the thrombin aptamer was extended from 24 min to 6 h with no effect on target-binding affinity [58].

To enhance stability, the sugar backbone of an oligonucleotide can also be substituted by enantiomers, built from nonnatural L-nucleotides, generating aptamers also known as Spiegelmers (from the German word for mirror, *spiegel*) [59–61]. Because SELEX cannot be directly performed with L-ribonucleotides, because wild-type RNA polymerases do not accept L-nucleotide triphosphate as substrates for PCR amplification and transcription reactions, the first step in the production of Spiegelmers is to synthesize an enantiomer of the target. After performing SELEX using oligonucleotides composed of natural D-nucleotides, once the aptamer is selected, the sequence is used to chemically synthesize the enantiomers of the nucleotide sequences with L-nucleotides. Because of this change in chirality of the nucleotides, Spiegelmers are more nuclease-resistant than D-nucleotide sequences. The synthetic production of the enantiomeric target is a limitation of this approach, because only relatively small proteins or peptides are eligible for Spiegelmer methodology [62].

## 6.3 From Basic Research to Therapeutics: Application of Aptamers

The feasibility of aptamer development for almost every type of molecular target makes this class of high-affinity class important not only for basic research but also for several therapeutic approaches, such as *in vivo* imaging diagnosis and therapy. Basic steps for aptamer development and possible applications in biotechnology and therapy are shown in Fig. 6.2.



FIGURE 6.2 Flowchart for the development of aptamers and their potential applications.

Some aptamers are found in nature as ligand-binding sites of special RNA structures called riboswitches. Commonly found in bacteria, riboswitches play roles in modulation of many fundamental biochemical pathways by sensing metabolites, such as guanine, adenine, flavin mononucleotide, lysine and glycine [63]. The typical riboswitch contains two distinct functional domains. An aptamer domain binds the effector molecule, and the second domain, the expression platform, is responsible for the interface between the binding event and the change in gene expression [64]. Riboswitches are commonly found in the 5'-untranslated region (UTR) of mRNAs, exerting their regulatory control over the transcript in a *cis*-fashion upon binding of small molecule ligands [64,65]. Because no protein is required for binding of metabolites to the aptamer domain of riboswitches, this regulation offers an immediate feedback response. Insertion of aptamers into the 5'-UTR of mRNAs provides a specific handle for controlling gene expression [66,67]. The binding of the ligand promotes a reversible conformational change in the aptamer domain that regulates the translation of the mRNA [66]. Another promising approach is the reengineering of natural riboswitches, aiming at conditional control of gene expression and cellular processes [68-71]. A vaccination strategy against Venezuelan equine encephalitis virus using this method was successful in controlling the immune response, RNA replication expression, and viral replication [71].

Owing to their ability of recognizing a large variety of targets, aptamers can also be used as biosensors in biosecurity applications. Ions, such as lead (Pb<sup>2+</sup>), potassium (K<sup>+</sup>), and mercury (Hg<sup>2+</sup>), can be detected by aptamers [72,73]. The detection of K<sup>+</sup> ions can be used as an indicator of several kidney diseases. Chang and colleagues detected this ion in urine by the formation of a G-guadruplex, a secondary RNA structure, in the ATP-binding aptamer, because the folding of this aptamer depends on the concentration of K<sup>+</sup> ions [72]. The same approach of ion-dependent aptamer folding was employed to detect both, Pb<sup>2+</sup> and Hg<sup>2+</sup>, in samples of ponds and soil [73]. Besides monitoring small molecules and drugs in biological samples, aptamer-based biosensors can be used to measure environmental contamination. For example, trinitrotoluene concentrations can be readily detected (within minutes) in water and soil samples [74]. The same aptamer technique is used for the screening of milk samples for the antibiotic neomycin [75].

Besides measurement and detection of specific targets, aptamers can be useful for imaging of biomolecules. Numerous analytical techniques, such as electrochemical, colorimetric, optical and mass-sensitive methods, can be employed to detect targets [76]. Aptamers specific for p68 in liver tumors and for small-cell lung cancer cells demonstrated potential as bioimaging probes [77–79].

Prostate-specific membrane antigen (PSMA), an important prostate cancer marker, is expressed in the vasculature of many solid tumors, especially in prostate. A dual-aptamer probe consisting of an A10 aptamer for PSMA(+) cells and a DUP-1 aptamer for PSMA(-), binds to prostate cancer cells (LNCaP and PC3), but not to normal (PNT2) or other cancer cells [80]. In addition, a drug-delivering system using the same A10 aptamer effectively introduced doxorubicin, an anticancer drug, into prostate cancer cells [81].

## 6.4 Aptamers in Clinical Trials

The search for the term "aptamer" in public clinical trials database http://clinicaltrials. gov lists 26 clinical trials as of August 2015, summarized in Table 6.2. These clinical trials demonstrate the clinical relevance of a range of aptamers and their targets in a variety of clinical conditions.

Trial Number	Aptamer	Target	Disease/condition	Phase
NCT00950638	ARC1905	Complement component 5 (C5)	Age-related macular degeneration	Phase I
NCT01089517	E10030	PDGF	Neovascular age-related macular degeneration	Phase II
NCT02397954	Zimura	Complement component 5 (C5)	Idiopathic polypoidal choroidal vasculopathy	Phase II
NCT00709527	ARC1905	Complement component 5 (C5)	Neovascular age-related macular degeneration	Phase I
NCT00569140	E10030	PDGF	Neovascular age-related macular degeneration	Phase I
NCT00021736	EYE001	VEGF	Neovascular age-related macular degeneration	Phase II/III
NCT00113997	REG1	Coagulation factor IXa	Healthy	Phase I
NCT02387957	Fovista	PDGF	Neovascular age-related macular degeneration	Phase II
NCT00056199	EYE001	VEGF	Von Hippel—Lindau disease	Phase I
NCT00215670	Pegaptanib sodium (Macugen)	VEGF	Neovascular age-related macular degeneration	Phase II/III
NCT01944839	Fovista	PDGF	Neovascular age-related macular degeneration	Phase III
NCT01940900	Fovista	PDGF	Neovascular age-related macular degeneration	Phase III
NCT00321997	EYE001	VEGF	Neovascular age-related macular degeneration	Phase II/III
NCT00312351	Pegaptanib sodium	VEGF	Neovascular age-related macular degeneration	Phase IV
NCT01940887	Fovista (E10030)	PDGF	Neovascular age-related macular degeneration	Phase III
NCT00040313	Pegaptanib sodium (Macugen)	VEGF	Diabetic macular edema	Phase II
NCT01034410	AS1411	Nucleolin	Acute myeloid leukemia	Phase II
NCT01194934	NOX-A12	CXCL12	Hematopoietic stem cell transplantation	Phase I
NCT00976729	NOX-E36	CCL2	Chronic inflammatory diseases/type 2 diabetes	Phase I
			Mellitus/systemic lupus erythematosus	
NCT00976378	NOX-A12	CXCL12	Autologous stem cell transplantation	Phase I
NCT00694785	ARC1779	Von Willebrand factor	Von Willebrand disease	Phase II
NCT00632242	ARC1779	Von Willebrand factor	Von Willebrand factor—related platelet function disorders	Phase II
NCT02079896	Lexaptepid Pegol (NOX-H94)	Hepcidin	Erythropoiesis-stimulating	Phase I/II
NCT01191372	ARC19499	TFPI	Hemophilia	Phase I
NCT01487044	Pegaptanib sodium	VEGF	Diabetic macular edema	
NCT01487070	Pegaptanib sodium (Macugen)	VEGF	Proliferative diabetic retinopathy	Phase I

 Table 6.2
 List of Aptamers in Clinical Trial According to clinicaltrial.gov database

PDGF, platelet-derived growth factor; TFPI, tissue factor pathway inhibitor; VEGF, vascular endothelial growth factor.

**ARC1905 (Zimura):** The target of this therapeutic aptamer is complement component, an important mediator of inflammation processes. A phase 1 study was completed with the objective of evaluating safety and tolerability in subjects with dry age-related macular degeneration (AMD) in both eyes. In other completed phase I study it was tested in combination with Lucentis, a monoclonal antibody fragment that binds and inactivates the VEGF. VEGF is a protein involved in angiogenesis. With the same strategy, a phase II study currently recruiting participants is testing the anti-C5 aptamer combined with anti-VEGF therapy in subjects with idiopathic polypoidal choroidal vasculopathy.

**E10030** (Fovista): An anti-PDGF aptamer was tested in a completed phase I study in subjects with neovascular AMD to evaluate its safety, tolerability, and pharmacokinetics. In another completed phase II study the aptamer was tested alone or combined with Lucentis in subjects with subfoveal choroidal neovascularization resulting from AMD. In a phase II study that is recruiting participants, the anti-PDGF aptamer is being tested in association with anti-VEGF therapy (Avastin, Lucentis, or Eylea) in subjects with neovascular AMD. Similarly, in two phase III studies, one recruiting and other active but not recruiting, patients with neovascular AMD are being evaluated for the safety and efficacy of combined Fovista and Lucentis therapy compared with Lucentis monotherapy. Another phase III study is recruiting subjects with the same disease to test therapy with Fovista associated with Avatin or Eylea compared with Avatin or Eylea monotherapy. Preliminary results of a phase 1 study that had tested the combination of E10030 and ranibizumab showed partial regression of choroidal neovascularization [82].

**Pegaptanib sodium (EYE001, Macugen):** This anti-VEGF aptamer was coupled to a 40-kD polyethylene glycol to increase *in vivo* half-life. Pegaptanib sodium was tested in a completed phase II/III study with the objective of assessing the effectiveness of stabilizing and/or improving vision in patients with exudative AMD. Two other studies, one a completed phase II/III study and the other a terminated phase IV study, aimed to compare the safety and efficacy of three different doses of pegaptanib sodium in patients with wet AMD. The safety, tolerability, and pharmacokinetics of intravitreous injections of 1 or 3 mg/eye pegaptanib were also tested in subjects with exudative AMD in a completed phase II/III study. In other two studies pegaptanib was tested in patients with diabetic retinal edema. In the first, a completed phase II study, the purpose was to test pegaptanib's safety and effectiveness in stabilization or improved vision compared with sham injection. In the second study, whose status was identified as unknown and no phase specified, the goal was establish the efficacy of pegaptanib in restoring VEGF levels. The safety of multiple injections of pegaptanib was shown in a phase II study [83]. The results of another phase II clinical trial revealed improvements in visual acuity in subjects treated with pegaptanib [84]. González and collaborators showed that intravitreal pegaptanib injection leads to regression of diabetic retinal neovascularization [85]. The safety and efficacy of intravitreal injections of Macugen in subjects with proliferative diabetic retinopathy were the aim of a completed phase I study. The ability of EYE001 to reduce retinal thickening and improve vision in patients

with Von Hippel–Lindau syndrome, who usually have angiomas in the retina and optic nerve, was tested in a phase I study.

**REG1:** REG1 is an aptamer-based anticoagulation system composed of two components. The first is an aptamer that binds and inhibits factor IXa (pegnivacogin), and the second is a complementary sequence that inactivates the anticoagulant effect of the first component (anivamersen). In a phase I completed study, the system was tested in healthy volunteers to check the safety and dosing of REG1. In phase I studies it was shown in healthy volunteers that pegnivacogin inhibits factor IX activity in a dosedependent way, quickly increases clotting, and activates partial thromboplastin time [86]. Subsequently it was shown that in patients with stable coronary artery disease, anivamersen quickly reversed the action of pegnivacogin [87]. Then in healthy subjects who had demonstrated previous results of safety and reproducibility, the researchers tested the use of multiple doses of the REG1 system [88]. Phase II studies evaluated the applicability of the REG1 system in elective percutaneous coronary interventions [89]. The REG1 system was able to suppress ischemic events and thrombotic complications [90].

**AS1411:** This aptamer targets nucleolin, one of the most abundant proteins. This protein is overexpressed in tumor cells and its presence in the cell surface is related to the growth and metabolic activity of cells. The AS1411 aptamer has been tested in combination with cytarabine in patients with primary refractory or relapsed acute myeloid leukemia in a phase II study.

**NOX-A12:** This aptamer binds and neutralizes CXCL12, a chemokine involved in metastasis, neovascularization, cell homing, and tissue regeneration. It was tested in two phase I studies in healthy volunteers to assess safety, tolerability, pharmacokinetics, and pharmacodynamics of an intravenous dose of NOX-A12 and to evaluate the effect on mobilization of hematopoietic stem cells. The intention is to use this aptamer in patients with lymphoma undergoing autologous stem cell transplantation.

**NOX-E36:** This aptamer targets the proinflammatory chemokine CCL2, which is involved in the recruitment of monocytes and T cells to sites of injury and inflammation. In a completed phase I study, safety and tolerability were shown in healthy subjects. Together with pharmacokinetic and pharmacodynamic data, this clinical trial aimed to establish the dose and route of administration of NOX-E36 in patients with chronic inflammatory diseases, type 2 diabetes mellitus, and systemic lupus erythematosus.

**ARC1779:** This aptamer targets the A1 domain of von Willebrand factor (vWF), inhibiting their binding to platelet glycoprotein Ib. The safety and tolerability of ARC1779 was tested in patients with vWF-related platelet disorders in a completed phase II study. Another phase II clinical trial using ARC1779 in patients with vWF-related platelet disorder was withdrawn before enrollment. Results of the phase I study showed safety and tolerability of ARC1779 in healthy volunteers. Moreover, the inhibition of vWF activity and platelet function occurred dose-dependently [91]. The results of a phase II study provided the proof of concept that ARC1779 inhibits vWF and prevents a

desmopressin-induced platelet drop in patients with type 2b vWF disease [92]. The aptamer was tested in patients with thrombotic thrombocytopenic purpura (TTP) for the safety, pharmacokinetics, and pharmacodynamics of ARC1779. Results showed that ARC1779 aptamer is well tolerated and inhibits vWF, leading to increased platelet count, which suggests that treatment with ARC1779 may be beneficial in patients with TTP [93,94].

**NOX-H94:** This aptamer binds hepcidin, a hormone produced in the liver that is involved in the control of plasma levels of iron. In addition, hepcidin has been identified as a possible mediator of anemia of chronic diseases. NOX-H94 is currently being tested in a clinical trial phase I/II, aiming to assess the safety and efficacy in increasing hemoglobin concentration in patients receiving dialysis who have functional iron deficiency and erythropoiesis-stimulating agent hyporesponsiveness.

**ARC19499:** This aptamer is an antitissue factor pathway inhibitor. It was used in a terminated study whose purpose was to test the safety and tolerability of ARC19499 in patients with hemophilia.

## 6.5 Conclusions

Aptamers have turned into potent tools in research with promising applications in diagnostics and therapy. These high-affinity binders developed by *in vitro* selection and chemical synthesis can be developed for almost every target molecule. Improved selection protocols have made possible the development of aptamers that recognize complex targets on cell surfaces or even entire cells or organisms. Aptamers are ideal for therapeutic and diagnostic applications in medicine and biotechnology, because they can easily be modified to enhance stability and nuclease resistance. The large number of aptamers in clinical trials suggests that they will soon compete with antibodies for therapeutic applications.

## Acknowledgments

Aptamer research in H.U.'s laboratory is financed by grants from São Paulo Research Foundation (FAPESP) and National Council of Technological and Scientific Development (CNPq), Brazil. J.C.V. is grateful for a postdoctoral fellowship from CNPq.

## References

- [1] H. Ulrich, C. Trujillo, A. Nery, J.M. Alves, P. Majumder, R.R. Resende, A.H. Martins, DNA and RNA aptamers: from tools for basic research towards therapeutic applications, Combinatorial Chemistry & High Throughput Screening 9 (2006) 619–632.
- [2] M. Faria, H. Ulrich, Sugar boost: when ribose modifications improve oligonucleotide performance, Current Opinion in Molecular Therapeutics 10 (2008) 168–175.
- [3] C. Tuerk, L. Gold, Systematic evolution of ligands by exponential enrichment: RNA ligands to bacteriophage T4 DNA polymerase, Science 249 (1990) 505–510.

- [4] A.D. Ellington, J.W. Szostak, In vitro selection of RNA molecules that bind specific ligands, Nature 346 (1990) 818–822.
- [5] W. Xu, A.D. Ellington, Anti-peptide aptamers recognize amino acid sequence and bind a protein epitope, Proceedings of the National Academy of Sciences of the United States of America 93 (1996) 7475–7480.
- [6] R. Conrad, A.D. Ellington, Detecting immobilized protein kinase C isozymes with RNA aptamers, Analytical Biochemistry 242 (1996) 261–265.
- [7] R.D. Jenison, S.C. Gill, A. Pardi, B. Polisky, High-resolution molecular discrimination by RNA, Science 263 (1994) 1425–1429.
- [8] G.P. Hess, H. Ulrich, H.G. Breitinger, L. Niu, M. Gameiro, C. Grewer, S. Srivastava, J.E. Ippolito, S.M. Lee, V. Jayaraman, S.E. Coombs, Mechanism-based discovery of ligands that counteract inhibition of the nicotinic acetylcholine receptor by cocaine and MK-801, Proceedings of the National Academy of Sciences of the United States of America 97 (2000) 13895–13900.
- [9] A. Rhie, L. Kirby, N. Sayer, R. Wellesley, P. Disterer, I. Sylvester, A. Gill, J. Hope, W. James, A. Tahiri-Alaoui, Characterization of 2'-fluoro-RNA aptamers that bind preferentially to disease-associated conformations of prion protein and inhibit conversion, Journal of Biological Chemistry 278 (2003) 39697–39705.
- [10] S. Sekiya, F. Nishikawa, K. Noda, P.K.R. Kumar, T. Yokoyama, S. Nishikawa, In vitro selection of RNA aptamers against cellular and abnormal isoform of mouse prion protein, Nucleic Acids Symposium Series (2004) 49 (2005) 361–362.
- [11] E.W.M. Ng, D.T. Shima, P. Calias, E.T. Cunningham, D.R. Guyer, A.P. Adamis, Pegaptanib, a targeted anti-VEGF aptamer for ocular vascular disease, Nature Reviews Drug Discovery 5 (2006) 123–132.
- [12] C.A. Trujillo, A.A. Nery, J.M. Alves, A.H. Martins, H. Ulrich, Development of the anti-VEGF aptamer to a therapeutic agent for clinical ophthalmology, Clinical Ophthalmology 1 (2007) 393–402.
- [13] M.J. Tolentino, A. Dennrick, E. John, M.S. Tolentino, Drugs in Phase II clinical trials for the treatment of age-related macular degeneration, Expert Opinion on Investigational Drugs 24 (2015) 183–199.
- [14] H. Ulrich, J.E. Ippolito, O.R. Pagán, V.A. Eterović, R.M. Hann, H. Shi, J.T. Lis, M.E. Eldefrawi, G.P. Hess, In vitro selection of RNA molecules that displace cocaine from the membranebound nicotinic acetylcholine receptor, Proceedings of the National Academy of Sciences of the United States of America 95 (1998) 14051–14056.
- [15] K.N. Morris, K.B. Jensen, C.M. Julin, M. Weil, L. Gold, High affinity ligands from in vitro selection: complex targets, Proceedings of the National Academy of Sciences of the United States of America 95 (1998) 2902–2907.
- [16] M.H. Magdesian, R. Giordano, H. Ulrich, M.A. Juliano, L. Juliano, R.I. Schumacher, W. Colli, M.J.M. Alves, Infection by *Trypanosoma cruzi*: identification of a parasite ligand and its host cell receptor, The Journal of Biological Chemistry 276 (2001) 19382–19389.
- [17] D.A. Daniels, H. Chen, B.J. Hicke, K.M. Swiderek, L. Gold, A tenascin-C aptamer identified by tumor cell SELEX: systematic evolution of ligands by exponential enrichment, Proceedings of the National Academy of Sciences of the United States of America 100 (2003) 15416–15421.
- [18] D. Shangguan, Y. Li, Z. Tang, Z.C. Cao, H.W. Chen, P. Mallikaratchy, K. Sefah, C.J. Yang, W. Tan, Aptamers evolved from live cells as effective molecular probes for cancer study, Proceedings of the National Academy of Sciences of the United States of America 103 (2006) 11838–11843.
- [19] J.A. Phillips, Y. Xu, Z. Xia, Z.H. Fan, W. Tan, Enrichment of cancer cells using aptamers immobilized on a microfluidic channel, Analytical Chemistry 81 (2009) 1033–1039.
- [20] A.A. Nery, C. Wrenger, H. Ulrich, Recognition of biomarkers and cell-specific molecular signatures: aptamers as capture agents, Journal of Separation Science 32 (2009) 1523–1530.

- [21] K. Harada, A.D. Frankel, Identification of two novel arginine binding DNAs, The EMBO Journal 14 (1995) 5798–5811.
- [22] T. Hermann, D.J. Patel, Adaptive recognition by nucleic acid aptamers, Science 287 (2000) 820-825.
- [23] Y. Huang, X. Chen, N. Duan, S. Wu, Z. Wang, X. Wei, Y. Wang, Selection and characterization of DNA aptamers against *Staphylococcus aureus* enterotoxin C1, Food Chemistry 166 (2015) 623–629.
- [24] A. Orabi, M. Bieringer, A. Geerlof, V. Bruss, An aptamer against the matrix binding domain on the hepatitis B virus capsid impairs virion formation, Journal of Virology (2015).
- [25] H. Shi, W. Cui, X. He, Q. Guo, K. Wang, X. Ye, J. Tang, Whole cell-SELEX aptamers for highly specific fluorescence molecular imaging of carcinomas in vivo, PLoS One 8 (2013) e70476.
- [26] M. Blank, T. Weinschenk, M. Priemer, H. Schluesener, Systematic evolution of a DNA aptamer binding to rat brain tumor microvessels: selective targeting of endothelial regulatory protein pigpen, The Journal of Biological Chemistry 276 (2001) 16464–16468.
- [27] S.M. Nimjee, C.P. Rusconi, B.A. Sullenger, Aptamers: an emerging class of therapeutics, The Annual Review of Medicine 56 (2005) 555–583.
- [28] D.H.J. Bunka, P.G. Stockley, Aptamers come of age at last, Nature Reviews Microbiology 4 (2006) 588–596.
- [29] F.C. Richardson, B.C. Tennant, D.J. Meyer, K.A. Richardson, P.C. Mann, G.R. McGinty, J.L. Wolf, P.M. Zack, R.A. Bendele, An evaluation of the toxicities of 2'-fluorouridine and 2'-fluorocytidine-HCl in F344 rats and woodchucks (*Marmota monax*), Toxicologic Pathology 27 (1999) 607–617.
- [30] P.S. Pendergrast, H.N. Marsh, D. Grate, J.M. Healy, M. Stanton, Nucleic acid aptamers for target validation and therapeutic applications, The Journal of Biomolecular Techniques 16 (2005) 224–234.
- [31] D. Jellinek, C.K. Lynott, D.B. Rifkin, N. Janjić, High-affinity RNA ligands to basic fibroblast growth factor inhibit receptor binding, Proceedings of the National Academy of Sciences of the United States of America 90 (1993) 11227–11231.
- [32] D. Eulberg, K. Buchner, C. Maasch, S. Klussmann, Development of an automated in vitro selection protocol to obtain RNA-based aptamers: identification of a biostable substance P antagonist, Nucleic Acids Research 33 (2005) 1–10.
- [33] N. Li, H.H. Nguyen, M. Byrom, A.D. Ellington, Inhibition of cell proliferation by an anti-egfr aptamer, PLoS One 6 (2011) e20299.
- [34] S. Leva, A. Lichte, J. Burmeister, P. Muhn, B. Jahnke, D. Fesser, J. Erfurth, P. Burgstaller, S. Klussmann, GnRH binding RNA and DNA Spiegelmers: a novel approach toward GnRH antagonism, Chemistry & Biology 9 (2002) 351–359.
- [35] L. Cerchia, F. Ducongé, C. Pestourie, J. Boulay, Y. Aissouni, K. Gombert, B. Tavitian, V. De Franciscis, D. Libri, Neutralizing aptamers from whole-cell SELEX inhibit the RET receptor tyrosine kinase, PLoS Biology 3 (2005) 697–704.
- [36] D.A. Daniels, A.K. Sohal, S. Rees, R. Grisshammer, Generation of RNA aptamers to the G-proteincoupled receptor for neurotensin, NTS-1, Analytical Biochemistry 305 (2002) 214–226.
- [37] D. Faulhammer, B. Eschgfäller, S. Stark, P. Burgstaller, W. Englberger, J. Erfurth, F. Kleinjung, J. Rupp, S.D. Vulcu, W. Schröder, S. Vonhoff, H. Nawrath, C. Gillen, S. Klussmann, Biostable aptamers with antagonistic properties to the neuropeptide nociceptin/orphanin FQ, RNA 10 (2004) 516–527.
- [38] C. Tuerk, S. MacDougal, L. Gold, RNA pseudoknots that inhibit human immunodeficiency virus type 1 reverse transcriptase, Proceedings of the National Academy of Sciences of the United States of America 89 (1992) 6988–6992.
- [39] L. Giver, D.P. Bartel, M.L. Zapp, M.R. Green, A.D. Ellington, Selection and design of high-affinity RNA ligands for HIV-1 rev, Gene 137 (1993) 25–31.

- [40] P. Allen, S. Worland, L. Gold, Isolation of high-affinity RNA ligands to HIV-1 integrase from a random pool, Virology 209 (1995) 327–336.
- [41] M. Khati, M. Schüman, J. Ibrahim, Q. Sattentau, S. Gordon, W. James, Neutralization of infectivity of diverse R5 clinical isolates of human immunodeficiency virus type 1 by gp120-binding 2'F-RNA aptamers, Journal of Virology 77 (2003) 12692–12698.
- [42] S. Sekiya, F. Nishikawa, K. Fukuda, S. Nishikawa, Structure/function analysis of an RNA aptamer for hepatitis C virus NS3 protease, Journal of Biochemistry 133 (2003) 351–359.
- [43] L.C. Bock, L.C. Griffin, J.A. Latham, E.H. Vermaas, J.J. Toole, Selection of single-stranded DNA molecules that bind and inhibit human thrombin, Nature 355 (1992) 564–566.
- [44] C.P. Rusconi, E. Scardino, J. Layzer, G.A. Pitoc, T.L. Ortel, D. Monroe, B.A. Sullenger, RNA aptamers as reversible antagonists of coagulation factor IXa, Nature 419 (2002) 90–94.
- [45] D. O'Connell, A. Koenig, S. Jennings, B. Hicke, H.L. Han, T. Fitzwater, Y.F. Chang, N. Varki, D. Parma, A. Varki, Calcium-dependent oligonucleotide antagonists specific for L-selectin, Proceedings of the National Academy of Sciences of the United States of America 93 (1996) 5883–5887.
- [46] Y. Song, Z. Zhu, Y. An, W. Zhang, H. Zhang, D. Liu, C. Yu, W. Duan, C.J. Yang, Selection of DNA aptamers against epithelial cell adhesion molecule for cancer cell imaging and circulating tumor cell capture, Analytical Chemistry 85 (2013) 4141–4149.
- [47] S.W. Lee, B.A. Sullenger, Isolation of a nuclease-resistant decoy RNA that can protect human acetylcholine receptors from myasthenic antibodies, Nature Biotechnology 15 (1997) 41–45.
- [48] K. Murakami, F. Nishikawa, K. Noda, T. Yokoyama, S. Nishikawa, Anti-bovine prion protein RNA aptamer containing tandem GGA repeat interacts both with recombinant bovine prion protein and its beta isoform with high affinity, Prion 2 (2008) 73–80.
- [49] H. Ulrich, M.H. Magdesian, M.J.M. Alves, W. Colli, In vitro selection of RNA aptamers that bind to cell adhesion receptors of *Trypanosoma cruzi* and inhibit cell invasion, The Journal of Biological Chemistry 277 (2002) 20756–20762.
- [50] M. Homann, H.U. Göringer, Combinatorial selection of high affinity RNA ligands to live African trypanosomes, Nucleic Acids Research 27 (1999) 2006–2014.
- [51] D. Jellinek, L.S. Green, C. Bell, C.K. Lynott, N. Gill, C. Vargeese, G. Kirschenheuter, D.P. McGee, P. Abesinghe, W.A. Pieken, Potent 2'-amino-2'-deoxypyrimidine RNA inhibitors of basic fibroblast growth factor, Biochemistry 34 (1995) 11363–11372.
- [52] Y. Lin, Q. Qiu, S.C. Gill, S.D. Jayasena, Modified RNA sequence pools for in vitro selection, Nucleic Acids Research 22 (1994) 5229–5234.
- [53] J. Ruckman, L.S. Green, J. Beeson, S. Waugh, W.L. Gillette, D.D. Henninger, L. Claesson-Welsh, N. Janjić, 2'-fluoropyrimidine RNA-based aptamers to the 165-amino acid form of vascular endothelial growth factor (VEGF165): inhibition of receptor binding and VEGF-induced vascular permeability through interactions requiring the exon 7-encoded domain, The Journal of Biological Chemistry 273 (1998) 20556–20567.
- [54] P.E. Burmeister, S.D. Lewis, R.F. Silva, J.R. Preiss, L.R. Horwitz, P.S. Pendergrast, T.G. McCauley, J.C. Kurz, D.M. Epstein, C. Wilson, A.D. Keefe, Direct in vitro selection of a 2'-O-methyl aptamer to VEGF, Chemistry & Biology 12 (2005) 25–33.
- [55] P.E. Burmeister, C. Wang, J.R. Killough, S.D. Lewis, L.R. Horwitz, A. Ferguson, K.M. Thompson, P.S. Pendergrast, T.G. McCauley, M. Kurz, J. Diener, S.T. Cload, C. Wilson, A.D. Keefe, 2'-Deoxy purine, 2'-O-methyl pyrimidine (dRmY) aptamers as candidate therapeutics, Oligonucleotides 16 (2006) 337–351.
- [56] Y. Takafuji, J.I. Jo, Y. Tabata, Simple PEG modification of DNA aptamer based on copper ion coordination for tumor targeting, Journal of Biomaterials Science, Polymer Edition 22 (2010) 1179–1195.

- [57] L. Tan, K.G. Neoh, E.T. Kang, W.S. Choe, X. Su, PEGylated anti-MUC1 aptamer-doxorubicin complex for targeted drug delivery to MCF7 breast cancer cells, Macromolecular Bioscience 11 (2011) 1331–1335.
- [58] L. Reyderman, S. Stavchansky, Pharmacokinetics and biodistribution of a nucleotide-based thrombin inhibitor in rats, Pharmaceutical Research 15 (1998) 904–910.
- [59] K.U. Bilik, E. Ergüven, S. Klussmann, F. Jarosch, P.Y. Wielinga, T.A. Lutz, T. Riediger, In-vitro and in-vivo antagonistic action of an anti-amylin Spiegelmer, Neuroreport 18 (2007) 1855–1859.
- [60] S. Helmling, C. Maasch, D. Eulberg, K. Buchner, W. Schröder, C. Lange, S. Vonhoff, B. Wlotzka, M.H. Tschöp, S. Rosewicz, S. Klussmann, Inhibition of ghrelin action in vitro and in vivo by an RNA-Spiegelmer, Proceedings of the National Academy of Sciences of the United States of America 101 (2004) 13174–13179.
- [61] W.G. Purschke, D. Eulberg, K. Buchner, S. Vonhoff, S. Klussmann, An L-RNA-based aquaretic agent that inhibits vasopressin in vivo, Proceedings of the National Academy of Sciences of the United States of America 103 (2006) 5173–5178.
- [62] A.D. Keefe, S. Pai, A. Ellington, Aptamers as therapeutics, Nature Reviews Drug Discovery 9 (2010) 537–550.
- [63] A. Ramesh, W.C. Winkler, Metabolite-binding ribozymes, Biochimica et Biophysica Acta Gene Regulatory Mechanisms 1839 (2014) 989–994.
- [64] C.A. Wakeman, W.C. Winkler, C.E. Dann, Structural features of metabolite-sensing riboswitches, Trends in Biochemical Sciences 32 (2007) 415–424.
- [65] W.C. Winkler, R.R. Breaker, Regulation of bacterial gene expression by riboswitches, The Annual Review of Microbiology 59 (2005) 487–517.
- [66] S. Hanson, K. Berthelot, B. Fink, J.E.G. McCarthy, B. Suess, Tetracycline-aptamer-mediated translational regulation in yeast, Molecular Microbiology 49 (2003) 1627–1637.
- [67] K. Beilstein, A. Wittmann, M. Grez, B. Suess, Conditional control of mammalian gene expression by tetracycline-dependent hammerhead ribozymes, ACS Synthetic Biology 4 (2015) 526–534.
- [68] Y. Nomura, Y. Yokobayashi, Reengineering a natural riboswitch by dual genetic selection, Journal of the American Chemical Society 129 (2007) 13814–13815.
- [69] N. Muranaka, K. Abe, Y. Yokobayashi, Mechanism-guided library design and dual genetic selection of synthetic OFF riboswitches, ChemBioChem: European Journal of Chemical Biology 10 (2009) 2375–2381.
- [70] C.A. Kellenberger, J. Sales-Lee, Y. Pan, M.M. Gassaway, A.E. Herr, M.C. Hammond, A minimalist biosensor: quantitation of cyclic Di-GMP using the conformational change of a riboswitch aptamer, RNA Biology (2015).
- [71] C.L. Bell, D. Yu, C.D. Smolke, A.J. Geall, C.W. Beard, P.W. Mason, Control of alphavirus-based gene expression using engineered riboswitches, Virology 483 (2015) 302–311.
- [72] C.C. Huang, H.T. Chang, Aptamer-based fluorescence sensor for rapid detection of potassium ions in urine, Chemical Communications 12 (2008) 1461–1463.
- [73] C.W. Liu, C.C. Huang, H.T. Chang, Highly selective DNA-based sensor for lead(II) and mercury(II) ions, Analytical Chemistry 81 (2009) 2383–2387.
- [74] E. Ehrentreich-Förster, D. Orgel, A. Krause-Griep, B. Cech, V.A. Erdmann, F. Bier, F.W. Scheller, M. Rimmele, Biosensor-based on-site explosives detection using aptamers as recognition elements, Analytical and Bioanalytical Chemistry 391 (2008) 1793–1800.
- [75] N. De-los-Santos-Álvarez, M.J. Lobo-Castañón, A.J. Miranda-Ordieres, P. Tuñón-Blanco, Modified-RNA aptamer-based sensor for competitive impedimetric assay of neomycin B, Journal of the American Chemical Society 129 (2007) 3808–3809.

- [76] K.M. Song, S. Lee, C. Ban, Aptamers and their biological applications, Sensors 12 (2012) 612–631.
- [77] V. Bagalkot, L. Zhang, E. Levy-Nissenbaum, S. Jon, P.W. Kantoff, R. Langery, O.C. Farokhzad, Quantum dot-aptamer conjugates for synchronous cancer imaging, therapy, and sensing of drug delivery based on bi-fluorescence resonance energy transfer, Nano Letters 7 (2007) 3065–3070.
- [78] T. Kunii, S. Ogura, M. Mie, E. Kobatake, Selection of DNA aptamers recognizing small cell lung cancer using living cell-SELEX, Analyst 136 (2011) 1310–1312.
- [79] J. Mi, Y. Liu, Z.N. Rabbani, Z. Yang, J.H. Urban, B.A. Sullenger, B.M. Clary, In vivo selection of tumor-targeting RNA motifs, Nature Chemical Biology 6 (2010) 22–24.
- [80] K. Min, K.-M. Song, M. Cho, Y.-S. Chun, Y.-B. Shim, J.K. Ku, C. Ban, Simultaneous electrochemical detection of both PSMA(+) and PSMA(-) prostate cancer cells using an RNA/peptide dual-aptamer probe, Chemical Communications 46 (2010) 5566–5568.
- [81] K. Min, H. Jo, K. Song, M. Cho, Y.S. Chun, S. Jon, W.J. Kim, C. Ban, Dual-aptamer-based delivery vehicle of doxorubicin to both PSMA(+) and PSMA(-) prostate cancers, Biomaterials 32 (2011) 2124–2132.
- [82] E.M. Lad, B.G. Hammill, L.G. Qualls, F. Wang, S.W. Cousins, L.H. Curtis, Anti-VEGF treatment patterns for neovascular age-related macular degeneration among medicare beneficiaries, American Journal of Ophthalmology 158 (2014) 537–543.
- [83] Eyetech Study Group, Anti-vascular endothelial growth factor therapy for subfoveal choroidal neovascularization secondary to age-related macular degeneration: phase II study results, Ophthalmology 110 (2003) 979–986.
- [84] E.T. Cunningham Jr., A.P. Adamis, M. Altaweel, L.P. Aiello, N.M. Bressler, D.J. D'Amico, M. Goldbaum, D.R. Guyer, B. Katz, M. Patel, S.D. Schwartz, Macugen Diabetic Retinopathy Study Group, A phase II randomized double-masked trial of pegaptanib, an anti-vascular endothelial growth factor aptamer, for diabetic macular edema, Ophthalmology 112 (2005) 1747–1757.
- [85] V.H. González, G.P. Giuliari, R.M. Banda, D.A. Guel, Intravitreal injection of pegaptanib sodium for proliferative diabetic retinopathy, British Journal of Ophthalmology 93 (2009) 1474–1478.
- [86] C.K. Dyke, S.R. Steinhubl, N.S. Kleiman, R.O. Cannon, L.G. Aberle, M. Lin, S.K. Myles, C. Melloni, R.A. Harrington, J.H. Alexander, R.C. Becker, C.P. Rusconi, First-in-human experience of an antidote-controlled anticoagulant using RNA aptamer technology: a phase 1a pharmacodynamic evaluation of a drug-antidote pair for the controlled regulation of factor IXa activity, Circulation 114 (2006) 2490–2497.
- [87] M.Y. Chan, M.G. Cohen, C.K. Dyke, S.K. Myles, L.G. Aberle, M. Lin, J. Walder, S.R. Steinhubl, I.C. Gilchrist, N.S. Kleiman, D.A. Vorchheimer, N. Chronos, C. Melloni, J.H. Alexander, R.A. Harrington, R.M. Tonkens, R.C. Becker, C.P. Rusconi, Phase 1b randomized study of antidote-controlled modulation of factor IXa activity in patients with stable coronary artery disease, Circulation 117 (2008) 2865–2874.
- [88] M.Y. Chan, C.P. Rusconi, J.H. Alexander, R.M. Tonkens, R.A. Harrington, R.C. Becker, A randomized, repeat-dose, pharmacodynamic and safety study of an antidote-controlled factor IXa inhibitor, The Journal of Thrombosis and Haemostasis 6 (2008) 789–796.
- [89] M.G. Cohen, D.A. Purdy, J.S. Rossi, L.R. Grinfeld, S.K. Myles, L.H. Aberle, A.B. Greenbaum, E. Fry, M.Y. Chan, R.M. Tonkens, S. Zelenkofske, J.H. Alexander, R.A. Harrington, C.P. Rusconi, R.C. Becker, First clinical application of an actively reversible direct factor IXa inhibitor as an anticoagulation strategy in patients undergoing percutaneous coronary intervention, Circulation 122 (2010) 614–622.
- [90] T.J. Povsic, J.P. Vavalle, J.H. Alexander, L.H. Aberle, S.L. Zelenkofske, R.C. Becker, C.E. Buller, M.G. Cohen, J.H. Cornel, J.D. Kasprzak, G. Montalescot, P.S. Fail, I.J. Sarembock, R. Mehran, RADAR Investigators, Use of the REG1 anticoagulation system in patients with acute coronary syndromes undergoing percutaneous coronary intervention: results from the phase II RADAR-PCI study, EuroIntervention 10 (2014) 431–438.
- [91] J.C. Gilbert, T. DeFeo-Fraulini, R.M. Hutabarat, C.J. Horvath, P.G. Merlino, H.N. Marsh, J.M. Healy, S. BouFakhreddine, T.V. Holohan, R.G. Schaub, First-in-human evaluation of anti-von Willebrand factor therapeutic aptamer ARC1779 in healthy volunteers, Circulation 116 (2007) 2678–2686.
- [92] B. Jilma, P. Paulinska, P. Jilma-Stohlawetz, J.C. Gilbert, R. Hutabarat, P. Knöbl, A randomised pilot trial of the anti-von Willebrand factor aptamer ARC1779 in patients with type 2b von Willebrand disease, Thrombosis and Haemostasis 104 (2010) 563–570.
- [93] P. Jilma-Stohlawetz, M.E. Gorczyca, B. Jilma, J. Siller-Matula, J.C. Gilbert, P. Knöbl, Inhibition of von Willebrand factor by ARC1779 in patients with acute thrombotic thrombocytopenic purpura, Thrombosis and Haemostasis 105 (2011) 545–552.
- [94] P. Jilma-Stohlawetz, P. Knöbl, J.C. Gilbert, B. Jilma, The anti-von Willebrand factor aptamer ARC1779 increases von Willebrand factor levels and platelet counts in patients with type 2B von Willebrand disease, Thrombosis and Haemostasis 108 (2012) 284–290.

# 7

# Synthetic Peptide Libraries: Chemical Diversity to Reach Lead Compounds

D. Marasco

# 7.1 Introduction

A combinatorial peptide library is a method in which a vast number of various peptides are synthesized; its use was first reported by Geysen et al. in 1984 using the pin method. Since then, many papers describing different methods of synthesizing and screening peptide libraries have been published [1]. In this approach, a huge number of diverse peptides are produced and successively screened in:

- 1. drug discovery and development,
- 2. epitope mapping,
- **3.** target validation, and
- 4. structure-activity studies.

This method revealed the ability to identify biologically active peptides successfully, including antibacterial peptides, ligands for cell surface receptors, opioid receptor antagonists, protein kinase inhibitors, and substrates. This combinatorial approach is mainly based on three methods by which peptides are selected in an unlabeled and soluble form or still linked to their support, either synthetic or biological, or to a labeling molecule.

These methods can be described as follows:

- **1.** Peptide libraries are synthesized and cleaved from the solid support to be screened as free compounds.
- 2. Synthetic combinatorial libraries of peptides are assayed on their solid support.
- **3.** A phage display is used, which enables the selection of clones of interest rather than screening, because large phage libraries can be panned against a target molecule by standard protocols.

In this chapter the first two synthetic methods will be examined.

# 7.2 Preparation of Synthetic Peptide Libraries

Synthetic combinatorial libraries, especially peptide libraries, are generally prepared by two different methods. The first is the "divide, couple, recombine" method, also referred to as "pool/split, portion/mix, and **mix-and-split**." It was first conceived of by Furka in 1991 [2] and was further developed by the groups of Hruby and Geysen [3] as a robust methodology to create large arrays of peptides to be used for the selection of target ligands. Resin beads are used as microreactors onto which steps of amino acid coupling and redistribution are carried out to increase the number of new sequences exponentially at each step. It consists of three basic steps: splitting, coupling, and mixing. First, resin beads are split into a number of aliquots that equals the number of building blocks to be used in the synthesis (splitting). To each resin aliquot, one building block is coupled and the reactions are driven to completion (coupling). Successively beads are randomly and thoroughly mixed (mixing) and then resplit into the same number of aliquots, achieving a set of homogeneous and equimolar collections of compounds. Repetitive execution of these basic steps for "n" times will produce a rapid increase in the newly generated molecules whereas the bead number remains constant. Because the resin beads encounter one reactant at a time, no more than one compound per bead can be generated [one bead one compound (OBOC)]; therefore the total number of final compounds is limited by the number of initial beads. Furthermore, as reactions are driven to completion, all peptides are generated in equal ratios in the mixtures. The basic principles of the mix-and-split method are outlined in Fig. 7.1A in a simplified version, in which three different amino acids are used as building blocks (A-B-C). The same steps can be performed regardless of the number or types of monomer units, which enormously increases the number of new sequences and their chemical diversity. With this procedure, couplings are carried out on spatially separated samples, adding single amino acids to each separate resin aliquot. As a result, both the number of peptides and their equimolar ratio are preserved in each aliquot. By choosing suitable sets of acid-resistant linkers and side chain protections, the compounds can be left on the beads but freed of the protecting groups. In this way, peptides can be tested still attached to the beads, exploiting the high local concentration of compounds on their surface. However, the use of suitable linkers allows the partial release of molecules, also permitting testing in solution. A major drawback of OBOC libraries is the limited set of peptides. Thus, the library's complexity is dictated by the initial number of beads.

The second method is the "amino acid mixture" method, **PRE-MIX**, in which the amino acids chosen for the library's construction as building blocks are premixed and then coupled to a single resin batch. This method is faster than mix-and-split and allows the preparation of large libraries, thus producing wide chemical diversity. Conversely, the distribution of any single peptide within the final mixtures depends on the relative kinetic rates of individual protected amino acids. To overcome the kinetic differences, the use of "smart" mixtures prepared in molar ratios adjusted according to different coupling rates of preactivated amino acids was proposed [4]. However, alternative



**FIGURE 7.1** Scheme of basic principles of the synthesis of peptide library built with three randomized positions and three different amino acids as building blocks (for a matter of simplicity). Three amino acids, after three randomization steps, generate 27 different sequences. A, Mix and split method, repetitive cycles of resin mixing, and repooling are performed until the requested length and complexity are reached. By repeating this procedure n times, a complexity of  $3^n$  is achieved. B, Premix method: the first n - 1 randomization steps are performed on a single resin batch by acylating with the premixed amino acids. Before the nth coupling, the resin is subdivided in three equal aliquots and isolated residues are coupled to each different aliquot.

methods can be applied, such as the use of large excesses of mixtures (50- to 100-fold) to compensate for the different reactivities. In Fig. 7.1B the basic principles of this approach are described in a simplified version in an iterative screening format.

Nevertheless, independently of the synthetic method, how are mixtures screened to arrive at one active compound? Various deconvolution methods have been employed, most commonly:

- **1. Iterative deconvolution**, which consists of an iterative process of screening and resynthesis of smaller sublibraries in the attempt to fractionate a mixture into its most active constituent(s), as described in Fig. 7.2A [3]; and
- **2. Positional scanning**, in which the mixtures are presynthesized in a number of sublibraries for each variable position. Each compound present in a given mixture has a common individual building block at a given position, whereas the remaining positions are composed of mixtures of all of the building blocks used to prepare the library [5]. Positional scanning synthetic peptide libraries (PS-SPCLs) are generated by making individual libraries with one amino acid held constant while the remaining amino acids are varied. The peptide sequence is scanned by creating additional unique combinatorial libraries, each holding a different amino acid constant. A tripeptide positional scanning library provides three distinct library subsets such that each holds one amino acid position constant, as described in Fig. 7.2B [6].



FIGURE 7.2 Scheme of the most common deconvolution methods of a library with three randomized positions and 20 building blocks. A, Iterative screening: Iterative process of screening and resynthesis of smaller sublibraries in the attempt to fractionate a mixture into its most active constituent(s). B, Positional scanning synthetic peptide combinatorial libraries: mixtures are presynthesized in a number of sublibraries for each variable position. Each compound present in a given mixture has a common individual building block at a given position, whereas the remaining positions are composed of mixtures of all of the building blocks used to prepare the library.

Besides deconvolution methods, libraries containing  $\alpha$ -amino acids are easy to sequence by Edman degradation using automated protein sequencing. However, Edman degradation requires a free N-terminus, so libraries with more complex peptide structures, such as cyclic or branched peptides or peptides containing  $\beta$ - or  $\gamma$ -amino acids, necessitate inclusion of a chemical tag in the bead structure. These chemical tags can be incorporated into the interior of the bead so as to not interfere with binding of the library peptides to targets of interest, and then subsequently sequenced using either Edman microsequencing or mass spectrometry [7].

A novel method for generating beads with interior tags, termed bilayer beads, was proposed, in which the library compound resides on the outer layer of the bead and the coding tag remains in the bead interior [8]. Such topologically segregated bifunctional beads can easily be prepared with a biphasic method. In this method, TentaGel beads are first thoroughly swollen in water. After excess water is drained, a limiting amount of amino-protecting reagent (e.g., 0.25 Eq.), such as N-(9-fluorenylmethyloxycarbonyloxy)succinimide (Fmoc-OSu) dissolved in an organic solvent mixture of dichloromethaneether (55:45, v/v), is added to the water-swollen beads. Under these conditions, only the outer layer of the TentaGel bead is derivatized. These bifunctional beads can be applied to OBOC peptide libraries with structures that cannot be sequenced with conventional methods [9]. In such an approach the surface density of the bead is crucial. As an example, it was applied to select multimeric ligands in which the surface density of the displayed peptide resulted in a high local concentration of the ligand (about 100 mM on a TentaGel bead), which can lead to the selection of low-affinity peptides. To avoid this problem, the selection can be performed in the presence of compounds or the peptide density can be reduced through a bilayer bead encoding approach. Alternatively, the beads can be spatially separated [10].

Similar to a phage display, synthetic libraries are relatively inexpensive and easy to generate; however, they are not constrained to natural amino acids such as those generated by phage technique, and can include both unnatural and D-amino acids, in addition to secondary structures constraints not tolerated by the phage. Indeed, completely unnatural peptoid libraries have been used to select cell-binding ligands [11]. Because unnatural and D-amino acids, synthetic libraries have the potential to identify stable peptide sequences rapidly, and posttranslational modifications such as glycosylation and phosphorylation can be incorporated into the library design as well. The design of synthetic libraries also makes them ideal for use in optimization of known ligands. Peptides previously isolated by phage display or structure-based methods can be used as lead compounds for synthetic peptide library construction, allowing for rapid generation of optimized peptides with higher affinity or specificity.

Synthetic libraries in soluble format offer a series of advantages such as the higher specificity of the interaction of bioactive selected peptides that can be synthesized as single library and aliquoted for use in multiple assays with multiple targets and lead sequences can be rapidly optimized. Furthermore, they are adaptable to almost any selection technique. They can be incubated with cells or receptors, typically in a high-throughput fashion such as in a 96-well plate or a microarray. For example, screens can be made for binding versus competitor fluorescently tagged natural ligands, looking for loss of fluorescence [12] or using biotinylated peptide and streptavidin—horseradish peroxidase as a detection reagent [13]. However, they are not commercially available and usually require additional synthesis for retesting. Because this approach depends on the idea that each amino acid of single lead sequence contributes individually to binding to the target of interest, this may make it difficult to determine ideal peptide sequences when multiple peptide motifs exist for the given target [14]. Thus after the ideal amino acids at each peptide position are determined, all possible combinations of peptides using these ideal amino acids should be generated and further tested for binding. Although PS-SPCLs are not used as frequently for the initial isolation of cell-targeting peptides, they are an excellent way to optimize lead peptides isolated from a phage displayed or bacterial peptide library [15,16].

On the other hand, OBOC libraries have unique features such as the selection of peptides on whole cells in vitro or ex vivo assays, but the presence of the linker provides potential steric hindrance between the peptide sequence and cellular receptor and is not suitable for in vivo screening in animals.

Although synthetic procedures generate millions of peptides, they are partially limited by fact that the sequence of the peptide on a particular bead must be determined by either sequencing or using additional encoding procedures as described in deconvolution methods (Fig. 7.2).

In an alternative method, reaction reagents and amino acid monomers can be spotted (SPOT technique) onto a support surface such as a cellulose membrane, creating spatially addressable peptide arrays containing about 100 sequences/cm<sup>2</sup> [17].

However, often the most common cellulose membranes used for SPOT are acid sensitive and lack mechanical stability in the synthesis process [18,19].

Further miniaturization of peptide arrays relies on the parallel synthesis of individually addressable peptide microchips that appeared efficient and versatile. This approach is based on conventional peptide chemistry with in-solution removal of acidlabile protecting groups using photogenerated reagents and digital photolithography [20,21]. These peptide microchips resulted in assays suitable for epitope binding to screen quickly and systematically for sequences specific to the human p53 antibody Pab240 [22,23].

Cell-binding ligands that can deliver biologically active cargo to a specific cell type or a diseased cell are highly sought [24,25]. Soluble target proteins can be screened for binding to OBOC libraries using several different approaches, all of which rely on the ability to distinguish protein-bound beads selectively. The protein of interest can be labeled with a tag (i.e., fluorescent or colorimetric dye, biotin, enzyme, radionuclide, epitope tag) that allows the protein-bound beads to be detected. Most commonly, an enzyme-linked colorimetric assay is employed [7]. The protein target can be directly labeled with alkaline phosphatase, or if a primary antibody against the protein is available, an alkaline phosphatase secondary antibody can be employed. The peptide content of isolated positive beads is then determined using Edman sequencing or mass spectrometry. The success of the screen depends on the stringency of the selection conditions and effective negative screens.

The greatest potential of combinatorial chemistry is represented by the number and variety of screenable compounds. Major efforts have focused on the development of methodologies to further increase molecular diversity: "dynamic mixtures" are based generally on reversible reactions in these; reactants and products are present in thermodynamic equilibrium. The most potent binding compound is selected among different combinations of mixture components, producing a shift of the equilibrium by subtracting products. However, although innovative and promising, these libraries have been limited so far to a small number of reversible reactions and libraries of moderate size [26,27]. Similarly, "libraries from libraries" represent an innovation respect to the traditional combinatorial chemistry. By this approach, combinatorial libraries of peptides once built on a solid phase are subsequently modified to maximize chemical diversity. Oxidations, reductions, alkylations, and acylations are performed, exponentially increasing the number of new compounds [28].

# 7.3 Peptides as Drugs

Peptide-based drug discovery processes have raised renewed interest because of the substantial failure, in most cases, of the screening of small molecules in discovering antagonists of protein—protein interactions (PPI) at the molecular basis of several human diseases [29,30].

Peptides are highly specific and show minimized risks of systemic toxicity, because their degradation products are amino acids and have a short half-life. The advantages of peptide-based therapeutics derive from unique chemicophysical properties including the modularity of amino acids and amide bonds; high target affinity, specificity, and potency; relatively low cost; ease of synthesis and storage; and reduced antigenicity [31]. Despite these favorable properties, they often have poor bioavailability and low metabolic stability. Thus major challenges for the development of peptide drugs appear to be optimization of their pharmacokinetics and setup of successful delivery strategies.

The most commonly used strategies to transform peptides into small peptidomimetics are outlined in Fig. 7.3, but process results are often difficult and intricate [32]. Once the primary structure of the biologically active peptide has been determined, the first step is to identify the minimum active sequence (MAS) required for the activity. This step involves testing truncated peptides from the C- and N-termini alternatively. Subsequently, the influence of individual amino acid on the biological activity is determined by systematically replacing each residue in the peptide with specific amino acids, typically alanine or D-amino acids.



FIGURE 7.3 Flowchart of strategies for the conversion of peptides into small peptidomimetics.

Synthetic peptide libraries of single compounds are often employed in this context: Ala-, Pro-, and D-Aaa scanning truncated and overlapping peptidomimetic libraries are often synthesized and screened to identify MAS [15].

After the structure-activity relationship (SAR) of each amino acid in the sequence has been assessed, the bioactive conformation is investigated by introducing constraints at various positions to reduce the conformational flexibility of the peptide; indeed, to fix crucial features for biological activity, peptides should be as small and rigid as possible. On the length, generally peptides with fewer than six residues are good starting points [33], whereas for peptides between 6 and 15 residues are challenging even if they often contain a well-defined key core and can include other dispensable positions. Conformational flexibility can be reduced by introducing local or global constraints at various positions in the peptide. These constraints can be achieved by incorporating modified amino acids, chemical modifications of the N-, the C-termini, and of the backbone, as well as by side chain-to-side chain cyclization including the formation of disulfide bridges. Many cyclic peptides, pseudopeptides, and peptidomimetics preserving their biological properties have been further modified to increase their resistance to degradation and elimination, bioavailability, and selectivity to become good drug candidates [34,35]. Usually the exact binding conformation of both peptide and protein in a complex is difficult to determine by experimental structural techniques (X-ray crystallography or nuclear magnetic resonance), but also the docking of peptides to binding partners in silico is one of the most complex modeling problems owing to the intrinsic flexibility of the ligand [36]. SAR studies of the constrained analogs can be used in an iterative process to provide information about the receptor-bound and biologically active conformation.

# 7.4 Recent Applications

Synthetic peptidomimetic libraries have found many applications in select bioactive compounds in many fields, such as antigens, inhibitors and activators of PPIs, kinases, proteases and tyrosinases, and ligands of receptors and catalysts [37].

Here interesting applications will be described.

To evaluate the protease activity of several enzymes, often synthetic peptide libraries have been employed. A study was focused on high-temperature requirement A2 (HtrA2). This protein belongs to the HtrA family of adenosine triphosphate—independent serine proteases. The primary function of HtrA2 includes maintaining the mitochondria homeostasis, cell death (by apoptosis, necrosis, or anoikis) and contribution to the cell signaling. HtrA2 protease substrate specificity was delineated via a combinatorial chemistry approach that led to the selection of novel intramolecularly quenched substrates. A combinatorial chemistry approach and the mix-and-split method with iterative scanning performed in solution were employed. Two separate substrate libraries to characterize the nonprime and prime binding pockets of HtrA2 were synthesized. The general formula of the synthesized peptide libraries were:

- **1.** the nonprime library,  $ABZ-X_4-X_3-X_2-X_1-ANB-NH_2$ , in which in position  $X_4 = X_3 = X_2$  the set of proteinogenic amino acid residues, excluding Cys, was present, whereas Ala, Abu, Val, Nva, Ile, Leu, Nle, Ser, or Thr was introduced in position  $X_1$ ; ABZ-2-aminobenzoic acid (fluorescence donor); ANB-NH2-amide of 5-amino-2-nitrobenzoic acid (quencher of fluorescence); and
- **2.** the prime library, ABZ-peptide- $X_{10}$ - $X_{20}$ - $X_{30}$ -Tyr(3-NO<sub>2</sub>)-NH<sub>2</sub>, in which in position  $X_{10} = X_{20} = X_{30}$  the set of proteinogenic amino acid residues, excluding Cys, was introduced.

For all synthesized compounds, the highest HtrA2-mediated hydrolysis efficiency and selectivity among tested HtrA family members was observed for ABZ-Ile-Met-Thr-Abu-Tyr-Met-Phe-Tyr(3-NO<sub>2</sub>)-NH<sub>2</sub>, which displayed a specificity constant kcat/KM value of 14,535  $M^{-1}$  s<sup>-1</sup> [38].

In past decades peptides were considered a class of anticancer agents that could specifically target cancer cells with lower toxicity to normal tissues, offering new opportunities for cancer prevention and treatment [39]. Indeed considerable progress has been made to identify peptides that specifically bind to tumor cells and tumor vasculature, defined as tumor-homing peptides [40]. The principle behind the peptide homing strategy is that they should recognize only molecules that are up-regulated in tumors, and therefore would not recognize normal cells from the corresponding organ [41]. Numerous peptide ligands have been isolated against various types of receptors or cells, such as RGD-containing peptides against integrin receptors in angiogenic tumor vasculature [42], or specific for platelet-derived growth factor receptor- $\beta$  receptor in pericytes and endothelial cells [43], KRK-containing peptides directed to angiogenic blood vessels and tumor cells [44], and a peptide recognizing thrombin receptor [45].

Tissue-specific homing peptides have also been reported for pancreatic  $\beta$ -cells, as well as specific peptides for tumor cells, especially lung tumor. There is also a peptide named GE11, which specifically recognizes the epidermal growth factor receptor [46]. The most widely used peptides in the targeted-delivery applications are integrin-targeting RGD-peptides, the first tumor-targeting peptides discovered. Integrins  $\alpha\nu\beta3$  and  $\alpha\nu\beta5$  are overexpressed in tumor endothelium and some epithelial cells during tumor growth, angiogenesis, invasion, and metastasis. Therefore, they represent an interesting molecular target for a tumor-homing approach [47]. RGD is a cell-adhesion motif present in many proteins of the extracellular matrix. This motif is recognized by  $\alpha\nu\beta3$  and  $\alpha\nu\beta5$  integrin receptors. The binding affinities of some of the RGD-containing derivatives for  $\alpha\nu\beta3$  range from 3.2 to 100 nM. The addition of specific amino acid residues to peptide sequence motifs, such as RGD, induces binding to cell-attachment proteins and strongly enhances the binding affinity of the peptide. The injection of the modified peptides induced antigen-specific serum antibodies [48].

In the employment of synthetic peptide libraries, few peptides have been selected against purified cell surface cancer biomarkers using OBOC libraries [49]. However, these synthetic libraries have been more widely exploited for selections on whole cells such as breast cancer MDA-MB 231 toward receptor  $\alpha 3\beta 1$  [50], leukemia Jukart cells toward  $\alpha 4\beta 1$  [51], ovarian cancer cells such as CAOV-3, SKOV-3, ES-2 [7,52], lung cancer cells such as A-549, H-1650 [53,54], and prostate cells such as DU 145 [55,56]. This may be partly because of the ease with which positive beads can be isolated by visual inspection. Biased selections can be performed in which the target receptor is overexpressed and the selection occurs within the context of whole cells. For example, the cvclic peptide cGRGDdvc was isolated from OBOC screening using  $\alpha v\beta 3$  integrintransfected K562 leukemia cells [57]. The parental K562 cell line was used for negative selections to remove ligands that bound to other receptors with respect to  $\alpha v\beta 3$ . However, most OBOC screens have been performed on cell lines without overexpression of a specific receptor. Nonetheless, all of the peptides isolated from OBOC libraries for which receptors have been identified bind integrin receptors. Much of this is likely due to library synthesis, because many of the libraries are designed with a bias toward sequences that are known to bind integrins. As such, it cannot be ruled out that the peptides bind other cellular targets as well. Unlike peptides isolated from phagedisplayed libraries, the ligands identified from OBOC libraries frequently bind multiple cancer cell lines. This is not surprising considering that integrins are often up-regulated in many cancers [58].

On the other hand, the PS-SPCLs are ideal for screenings for cellular effects induced by peptides. Kang and co-workers used integrin microarrays to select peptides that inhibit angiogenesis. From this screening a novel peptide ligand containing a PDZ-binding motif (Ser-Asp-Val), named P11, was selected. During the screening fluorescently labeled fibronectin was mixed with the peptide library before addition to the microarray. Library peptides that inhibited the receptor—fibronectin interaction were thus visible by a reduced fluorescent signal [59]. In a similar fashion, an  $\alpha 5\beta 1$  microarray and a PS-SPCL were used to screen for peptides that inhibit  $\alpha 5\beta 1$ -fibronectin interactions [60,61]. RGDs have been found by panning on purified integrins, on whole cells, and in vivo [62]. The specificity of the peptide for a particular integrin varies depending on the flanking sequence around the RGD [63,64]. High-affinity binding to integrins is achieved with only three amino acids; any library with a diversity of  $8.0 \times 10^3$  is statistically likely to contain the RGD binding motif. For this reason and the fact that integrins are often overexpressed in cancer cells and tumor vasculature, it is not surprising that RGD sequences are often isolated.

Similarly, numerous NGR-containing peptides have been found over the years [65]. This peptide motif binds to aminopeptidase N (CD13) but can also undergo a spontaneous deamination to form isoDGR, a ligand for  $\alpha\nu\beta3$  and  $\alpha5\beta1$  integrins [66].

In an applicative perspective, the possibility of incorporating unnatural amino acids in synthetic peptide libraries makes them useful for organo-catalysis. Metallopeptide catalysts and artificial metalloenzymes built from peptide scaffolds and catalytically active metal centers possess a number of exciting properties that could be exploited for selective catalysis [67]. Control over metal catalyst secondary coordination spheres, compatibility with library based methods for optimization and evolution, and biocompatibility stand out in this regard. A wide range of unnatural amino acids (UAAs) have been incorporated into peptide and protein scaffolds using several distinct methods, and the resulting UAA-containing scaffolds can be used to create novel hybrid metal– peptide catalysts. Promising levels of selectivity have been demonstrated for several hybrid catalysts, and these provide a strong impetus and important lessons for the design of and optimization of hybrid catalysts [68].

New strategies involving computer-customized combinatorial libraries offer enormous potential for the design of more "focused" and "smart" chemical libraries with maximal diversity. Libraries can be seen as a source of bioactive molecules that are selected on the basis of the biochemical properties predefined by appropriate assay settings. These settings represent the fishing-out method. The more diverse the library will be, the higher will be the probability of selecting good "hits." The concept of "diversity" is therefore of utmost importance in choosing a library for a particular screening, and good libraries must first fulfill the requirement of "highest diversity" instead of the "highest complexity." Molecules with overlapping structures will not or only poorly contribute to the overall probability to find a positive hit. The diversity of a library is generally associated with its complexity (the number of different components). However, this is not always true, and low-complexity libraries can display a higher diversity than can libraries with a huge number of components. In fact, the synthesis of random combinatorial libraries of peptides generates a large number of "quasiduplicates" deriving from the strong similarity between several side chains.

Using common amino acids, in L- or D-configuration, sequences where Glu is replaced by Asp, Leu by Ile or Val, Gln by Asn, and so on, can display similar properties. Such residues, although different in their propensity to adopt secondary structures, can be considered almost equivalent in terms of intrinsic physicochemical properties, as for example the capacity to establish external interactions or to fit in a crucial recognition site. On the contrary, in large libraries, the need to manage large arrays of tubes and codes can puzzle the way to identifying lead compounds and slow down the synthetic and deconvolution steps. In the perspective of simplifying the synthesis and deconvolution procedures without affecting the probability of finding active peptides, our approach focused on the general properties of L- and D-amino acids, reaching a compromise between the need to maintain the highest possible diversity and reducing the number of building blocks. We called these libraries "simplified libraries," intending with this any new ensemble of possible sequences achievable with a reduced and nonredundant set of amino acids. The distribution of residue molecular weights is also an important parameter in cases of deconvolution of the library by mass spectrometry approaches [39].

We have further evolved this approach by applying "focused simplified libraries" in which, through an Ala-scan approach, we defined hot spots of a natural peptide, the KIR domain of SOCS1, in the interaction with JAK2 and randomized few positions within it, to find a more active peptidomimetic named PS5 [15,16]. Indeed the chances of finding a hit are much higher in a focused combinatorial library than in a completely random library because the structural diversity is already positively biased for a given problem, e.g., binding to a specific target protein. Thus to identify new ligands it is often sufficient to use much smaller libraries with only a couple of hundred different members. The use of small but focused libraries, rather than completely random mixtures, has already been successfully applied in pharmaceutical and medicinal chemistry [33,69].

# 7.5 Conclusions and Perspectives

Major advantages of peptides rely on their typical high affinity and selectivity, their high potency, minimized drug–drug interactions, and low accumulation capacity and toxicity.

Peptide libraries can have a pivotal role in the rapid identification of high- or medium-affinity target ligands that can be subsequently optimized in terms of potency, selectivity, and stability. Indeed the construction of peptide libraries is commonly used to identify and characterize ligand receptor-specific interactions and search for novel ligands for protein purification. The easiness of preparation, characterization, and robustness of the available chemistry have pushed tens of laboratories around the world to use synthetic peptide libraries as a source of chemical diversity and a versatile way to select new active molecules.

Further improvement in chemical and biological resistance of affinity ligands encouraged the "intelligent" design and synthesis of chemical libraries of low—molecular weight bio-inspired mimic compounds. For instance, high-throughput synthesis and screening of compound collections through phenotypic or biochemical assays often yields useful compounds discovered, relative to the high cost in time and resources expended [70]. The large complexity achievable by using nonnatural amino acids, by introducing secondary structure diversity elements or changing the basic scaffolds to enhance molecular surfaces and target avidity, represents an indisputable advantage with respect to small-molecule libraries. These major points, coupled with the ease of preparation, characterization, and handling, will increasingly set forth synthetic peptide libraries to the attention of the drug discovery chemists community, making them an invaluable tool for drug discovery.

The use of novel automated synthesis and screening strategies, such as simplified focused peptide libraries, can increase the process throughput and speed the discovery procedure, making the discovery of new active peptides a fast and efficient process.

### References

- [1] G. Tribbick, S. Rodda, Combinatorial methods for discovery of peptide ligands which bind to antibody-like molecules, Journal of Molecular Recognition 15 (2002) 306–310.
- [2] A. Furka, F. Sebestyen, M. Asgedom, G. Dibo, General method for rapid synthesis of multicomponent peptide mixtures, International Journal of Peptide & Protein Research 37 (1991) 487–493.
- [3] R.A. Houghten, C. Pinilla, S.E. Blondelle, J.R. Appel, C.T. Dooley, J.H. Cuervo, Generation and use of synthetic peptide combinatorial libraries for basic research and drug discovery, Nature 354 (1991) 84–86.
- [4] J.M. Ostresh, J.H. Winkle, V.T. Hamashin, R.A. Houghten, Peptide libraries: determination of relative reaction rates of protected amino acids in competitive couplings, Biopolymers 34 (1994) 1681–1689.
- [5] C. Pinilla, J.R. Appel, P. Blanc, R.A. Houghten, Rapid identification of high affinity peptide ligands using positional scanning synthetic peptide combinatorial libraries, Biotechniques 13 (1992) 901–905.
- [6] B.P. Gray, K.C. Brown, Combinatorial peptide libraries: mining for cell-binding peptides, Chemical Reviews 114 (2014) 1020–1081.
- [7] O.H. Aina, R. Liu, J.L. Sutcliffe, J. Marik, C.X. Pan, K.S. Lam, From combinatorial chemistry to cancer-targeting peptides, Molecular Pharmaceutics 4 (2007) 631–651.
- [8] X. Wang, J. Zhang, A. Song, C.B. Lebrilla, K.S. Lam, Encoding method for OBOC small molecule libraries using a biphasic approach for ladder-synthesis of coding tags, Journal of the American Chemical Society 126 (2004) 5740–5749.
- [9] X. Wang, L. Peng, R. Liu, B. Xu, K.S. Lam, Applications of topologically segregated bilayer beads in 'one-bead one-compound' combinatorial libraries, Journal of Peptide Research 65 (2005) 130–138.
- [10] S. Aggarwal, J.L. Harden, S.R. Denmeade, Synthesis and screening of a random dimeric peptide library using the one-bead-one-dimer combinatorial approach, Bioconjugate Chemistry 17 (2006) 335–340.
- [11] J.H. Lee, A.M. Meyer, H.S. Lim, A simple strategy for the construction of combinatorial cyclic peptoid libraries, Chemical Communications (Camb) 46 (2010) 8615–8617.
- [12] J.G. Paez, P.A. Janne, J.C. Lee, S. Tracy, H. Greulich, S. Gabriel, P. Herman, F.J. Kaye, N. Lindeman, T. J. Boggon, K. Naoki, H. Sasaki, Y. Fujii, M.J. Eck, W.R. Sellers, B.E. Johnson, M. Meyerson, EGFR mutations in lung cancer: correlation with clinical response to gefitinib therapy, Science 304 (2004) 1497–1500.

- [13] C.L. Denholt, P.R. Hansen, N. Pedersen, H.S. Poulsen, N. Gillings, A. Kjaer, Identification of novel peptide ligands for the cancer-specific receptor mutation EFGRvIII using a mixture-based synthetic combinatorial library, Biopolymers 91 (2009) 201–206.
- [14] R. Liu, A.M. Enstrom, K.S. Lam, Combinatorial peptide library methods for immunobiology research, Experimental Hematology 31 (2003) 11–30.
- [15] N. Doti, P.L. Scognamiglio, S. Madonna, C. Scarponi, M. Ruvo, G. Perretta, C. Albanesi, D. Marasco, New mimetic peptides of the kinase-inhibitory region (KIR) of SOCS1 through focused peptide libraries, Biochemical Journal 443 (2012) 231–240.
- [16] S. Madonna, C. Scarponi, N. Doti, T. Carbone, A. Cavani, P.L. Scognamiglio, D. Marasco, C. Albanesi, Therapeutical potential of a peptide mimicking the SOCS1 kinase inhibitory region in skin immune responses, European Journal of Immunology 43 (2013) 1883–1895.
- [17] R. Frank, H. Overwin, SPOT synthesis. Epitope analysis with arrays of synthetic peptides prepared on cellulose membranes, Methods in Molecular Biology 66 (1996) 149–169.
- [18] A. Kramer, U. Reineke, L. Dong, B. Hoffmann, U. Hoffmuller, D. Winkler, R. Volkmer-Engert, J. Schneider-Mergener, Spot synthesis: observations and optimizations, Journal of Peptide Research 54 (1999) 319–327.
- [19] C.Y. Cho, E.J. Moran, S.R. Cherry, J.C. Stephans, S.P. Fodor, C.L. Adams, A. Sundaram, J.W. Jacobs, P.G. Schultz, An unnatural biopolymer, Science 261 (1993) 1303–1305.
- [20] E. LeProust, J.P. Pellois, P. Yu, H. Zhang, X. Gao, O. Srivannavit, E. Gulari, X. Zhou, Digital lightdirected synthesis. A microarray platform that permits rapid reaction optimization on a combinatorial basis, Journal of Combinatorial Chemistry 2 (2000) 349–354.
- [21] X. Gao, E. LeProust, H. Zhang, O. Srivannavit, E. Gulari, P. Yu, C. Nishiguchi, Q. Xiang, X. Zhou, A flexible light-directed DNA chip synthesis gated by deprotection using solution photogenerated acids, Nucleic Acids Research 29 (2001) 4744–4750.
- [22] C.W. Stephen, P. Helminen, D.P. Lane, Characterisation of epitopes on human p53 using phagedisplayed peptide libraries: insights into antibody-peptide interactions, Journal of Molecular Biology 248 (1995) 58–78.
- [23] J.P. Pellois, X. Zhou, O. Srivannavit, T. Zhou, E. Gulari, X. Gao, Individually addressable parallel peptide synthesis on microchips, Nature Biotechnology 20 (2002) 922–926.
- [24] J.W. Kehoe, B.K. Kay, Filamentous phage display in the new millennium, Chemical Reviews 105 (2005) 4056-4072.
- [25] H.R. Hill, P.G. Stockley, Phage presentation, Molecular Microbiology 20 (1996) 685–692.
- [26] S. Otto, R.L. Furlan, J.K. Sanders, Recent developments in dynamic combinatorial chemistry, Current Opinion in Chemical Biology 6 (2002) 321–327.
- [27] O. Ramstrom, T. Bunyapaiboonsri, S. Lohmann, J.M. Lehn, Chemical biology of dynamic combinatorial libraries, Biochimica et Biophysica Acta 1572 (2002) 178–186.
- [28] A. Nefzi, J.M. Ostresh, Y. Yu, R.A. Houghten, Combinatorial chemistry: libraries from libraries, the art of the diversity-oriented transformation of resin-bound peptides and chiral polyamides to low molecular weight acyclic and heterocyclic compounds, Journal of Organic Chemistry 69 (2004) 3603–3609.
- [29] A. Loregian, G. Palu, Disruption of protein-protein interactions: towards new targets for chemotherapy, Journal of Cellular Physiology 204 (2005) 750–762.
- [30] P. Braun, A.C. Gingras, History of protein-protein interactions: from egg-white to complex networks, Proteomics 12 (2012) 1478–1498.
- [31] J. Vagner, H. Qu, V.J. Hruby, Peptidomimetics, a synthetic tool of drug discovery, Current Opinion in Chemical Biology 12 (2008) 292–296.

- [32] P.L. Scognamiglio, C. Di Natale, G. Perretta, D. Marasco, From peptides to small molecules: an intriguing but intricated way to new drugs, Current Medicinal Chemistry 20 (2013) 3803–3817.
- [33] P.L. Scognamiglio, N. Doti, P. Grieco, C. Pedone, M. Ruvo, D. Marasco, Discovery of small peptide antagonists of PED/PEA15-D4alpha interaction from simplified combinatorial libraries, Chemical Biology & Drug Design 77 (2011) 319–327.
- [34] H. Pajouhesh, G.R. Lenz, Medicinal chemical properties of successful central nervous system drugs, NeuroRx 2 (2005) 541–553.
- [35] B.D. Welch, A.P. VanDemark, A. Heroux, C.P. Hill, M.S. Kay, Potent D-peptide inhibitors of HIV-1 entry, Proceedings of the National Academy of Sciences of the United States of America 104 (2007) 16828–16833.
- [36] A. Russo, P.L. Scognamiglio, R.P. Hong Enriquez, C. Santambrogio, R. Grandori, D. Marasco, A. Giordano, G. Scoles, S. Fortuna, In silico generation of peptides by replica exchange Monte Carlo: docking-based optimization of maltose-binding-protein ligands, PLoS One 10 (2015) e0133571.
- [37] D. Marasco, G. Perretta, M. Sabatella, M. Ruvo, Past and future perspectives of synthetic peptide libraries, Current Protein and Peptide Science 9 (2008) 447–467.
- [38] M. Wysocka, A. Wojtysiak, M. Okonska, N. Gruba, M. Jarzab, T. Wenta, B. Lipinska, R. Grzywa, M. Sienczyk, K. Rolka, A. Lesner, Design and synthesis of new substrates of HtrA2 protease, Analytical Biochemistry 475 (2015) 44–52.
- [39] D. Wu, Y. Gao, Y. Qi, L. Chen, Y. Ma, Y. Li, Peptide-based cancer therapy: opportunity and challenge, Cancer Letters 351 (2014) 13–22.
- [40] A. Gautam, P. Kapoor, K. Chaudhary, R. Kumar, G.P. Raghava, Tumor homing peptides as molecular probes for cancer therapeutics, diagnostics and theranostics, Current Medicinal Chemistry 21 (2014) 2367–2391.
- [41] P. Laakkonen, M.E. Akerman, H. Biliran, M. Yang, F. Ferrer, T. Karpanen, R.M. Hoffman, E. Ruoslahti, Antitumor activity of a homing peptide that targets tumor lymphatics and tumor cells, Proceedings of the National Academy of Sciences of the United States of America 101 (2004) 9381–9386.
- [42] F. Danhier, A. Le Breton, V. Preat, RGD-based strategies to target alpha(v) beta(3) integrin in cancer therapy and diagnosis, Molecular Pharmaceutics 9 (2012) 2961–2973.
- [43] Z. Wang, D. Kong, Y. Li, F.H. Sarkar, PDGF-D signaling: a novel target in cancer therapy, Current Drug Targets 10 (2009) 38–41.
- [44] L. Zhang, E. Giraudo, J.A. Hoffman, D. Hanahan, E. Ruoslahti, Lymphatic zip codes in premalignant lesions and tumors, Cancer Research 66 (2006) 5696–5706.
- [45] S.R. Coughlin, How the protease thrombin talks to cells, Proceedings of the National Academy of Sciences of the United States of America 96 (1999) 11023–11027.
- [46] Z. Li, R. Zhao, X. Wu, Y. Sun, M. Yao, J. Li, Y. Xu, J. Gu, Identification and characterization of a novel peptide ligand of epidermal growth factor receptor for targeted delivery of therapeutics, FASEB Journal 19 (2005) 1978–1985.
- [47] R.A. Bartolome, A. Pelaez-Garcia, I. Gomez, S. Torres, M.J. Fernandez-Acenero, B. Escudero-Paniagua, J.I. Imbaud, J.I. Casal, An RGD motif present in cadherin 17 induces integrin activation and tumor growth, Journal of Biological Chemistry 289 (2014) 34801–34814.
- [48] M. Toporkiewicz, J. Meissner, L. Matusewicz, A. Czogalla, A.F. Sikorski, Toward a magic or imaginary bullet? Ligands for drug targeting to cancer cells: principles, hopes, and challenges, International Journal of Nanomedicine 10 (2015) 1399–1414.
- [49] K.S. Lam, Q. Lou, Z.G. Zhao, J. Smith, M.L. Chen, E. Pleshko, S.E. Salmon, Idiotype specific peptides bind to the surface immunoglobulins of two murine B-cell lymphoma lines, inducing signal transduction, Biomedical Peptides, Proteins, and Nucleic Acids 1 (1995) 205–210.

- [50] N. Yao, W. Xiao, X. Wang, J. Marik, S.H. Park, Y. Takada, K.S. Lam, Discovery of targeting ligands for breast cancer cells using the one-bead one-compound combinatorial method, Journal of Medicinal Chemistry 52 (2009) 126–133.
- [51] J.R. Falsey, M. Renil, S. Park, S. Li, K.S. Lam, Peptide and small molecule microarray for high throughput cell adhesion and functional assays, Bioconjugate Chemistry 12 (2001) 346–353.
- [52] O.H. Aina, J. Marik, R. Liu, D.H. Lau, K.S. Lam, Identification of novel targeting peptides for human ovarian cancer cells using "one-bead one-compound" combinatorial libraries, Molecular Cancer Therapeutics 4 (2005) 806–813.
- [53] D. Lau, L. Guo, R. Liu, J. Marik, K. Lam, Peptide ligands targeting integrin alpha3beta1 in non-small cell lung cancer, Lung Cancer 52 (2006) 291–297.
- [54] M. Mikawa, H. Wang, L. Guo, R. Liu, J. Marik, Y. Takada, K. Lam, D. Lau, Novel peptide ligands for integrin alpha 4 beta 1 overexpressed in cancer cells, Molecular Cancer Therapeutics 3 (2004) 1329–1334.
- [55] M.E. Pennington, K.S. Lam, A.E. Cress, The use of a combinatorial library method to isolate human tumor cell adhesion peptides, Molecular Diversity 2 (1996) 19–28.
- [56] I.B. DeRoock, M.E. Pennington, T.C. Sroka, K.S. Lam, G.T. Bowden, E.L. Bair, A.E. Cress, Synthetic peptides inhibit adhesion of human tumor cells to extracellular matrix proteins, Cancer Research 61 (2001) 3308–3313.
- [57] W. Xiao, Y. Wang, E.Y. Lau, J. Luo, N. Yao, C. Shi, L. Meza, H. Tseng, Y. Maeda, P. Kumaresan, R. Liu, F.C. Lightstone, Y. Takada, K.S. Lam, The use of one-bead one-compound combinatorial library technology to discover high-affinity alphavbeta3 integrin and cancer targeting arginine-glycineaspartic acid ligands with a built-in handle, Molecular Cancer Therapeutics 9 (2010) 2714–2723.
- [58] L. Seguin, J.S. Desgrosellier, S.M. Weis, D.A. Cheresh, Integrins and cancer: regulators of cancer stemness, metastasis, and drug resistance, Trends in Cell Biology 25 (2015) 234–240.
- [59] J.Y. Bang, E.Y. Kim, D.K. Kang, S.I. Chang, M.H. Han, K.H. Baek, I.C. Kang, Pharmacoproteomic analysis of a novel cell-permeable peptide inhibitor of tumor-induced angiogenesis, Molecular & Cellular Proteomics 10 (2011). M110 005264.
- [60] E.Y. Kim, J.Y. Bang, S.I. Chang, I.C. Kang, A novel integrin alpha5beta1 antagonistic peptide, A5-1, screened by protein chip system as a potent angiogenesis inhibitor, Biochemical and Biophysical Research communication 377 (2008) 1288–1293.
- [61] R. Pasqualini, R.E. Millikan, D.R. Christianson, M. Cardo-Vila, W.H. Driessen, R.J. Giordano, A. Hajitou, A.G. Hoang, S. Wen, K.F. Barnhart, W.B. Baze, V.D. Marcott, D.H. Hawke, K.A. Do, N.M. Navone, E. Efstathiou, P. Troncoso, R.R. Lobb, C.J. Logothetis, W. Arap, Targeting the interleukin-11 receptor alpha in metastatic prostate cancer: a first-in-man study, Cancer (2015).
- [62] E. Ruoslahti, RGD and other recognition sequences for integrins, Annual Review of Cell and Developmental Biology 12 (1996) 697–715.
- [63] A.N. Elayadi, K.N. Samli, L. Prudkin, Y.H. Liu, A. Bian, X.J. Xie, I.I. Wistuba, J.A. Roth, M.J. McGuire, K.C. Brown, A peptide selected by biopanning identifies the integrin alphavbeta6 as a prognostic biomarker for nonsmall cell lung cancer, Cancer Research 67 (2007) 5889–5895.
- [64] W. Arap, R. Pasqualini, E. Ruoslahti, Cancer treatment by targeted drug delivery to tumor vasculature in a mouse model, Science 279 (1998) 377–380.
- [65] A. Corti, F. Curnis, W. Arap, R. Pasqualini, The neovasculature homing motif NGR: more than meets the eye, Blood 112 (2008) 2628–2635.
- [66] F. Curnis, A. Cattaneo, R. Longhi, A. Sacchi, A.M. Gasparri, F. Pastorino, P. Di Matteo, C. Traversari, A. Bachi, M. Ponzoni, G.P. Rizzardi, A. Corti, Critical role of flanking residues in NGR-to-isoDGR transition and CD13/integrin receptor switching, Journal of Biological Chemistry 285 (2010) 9114–9123.

- [67] A. Lombardi, D. Marasco, O. Maglio, L. Di Costanzo, F. Nastri, V. Pavone, Miniaturized metalloproteins: application to iron-sulfur proteins, Proceedings of the National Academy of Sciences of the United States of America 97 (2000) 11922–11927.
- [68] J.C. Lewis, Metallopeptide catalysts and artificial metalloenzymes containing unnatural amino acids, Current Opinion in Chemical Biology 25 (2015) 27–35.
- [69] G.E. Mulder, J.A. Kruijtzer, R.M. Liskamp, A combinatorial approach toward smart libraries of discontinuous epitopes of HIV gp120 on a TAC synthetic scaffold, Chemical Communications (Camb) 48 (2012) 10007–10009.
- [70] D.J. Payne, M.N. Gwynn, D.J. Holmes, D.L. Pompliano, Drugs for bad bugs: confronting the challenges of antibacterial discovery, Nature Reviews Drug Discovery 6 (2007) 29–40.

# 8

# Biotechnological Role of Phage-Displayed Peptides for the Diagnosis of Neglected Tropical Diseases

J. de Moura\*, L.M. Alvarenga, V. Thomaz-Soccol UNIVERSIDADE FEDERAL DO PARANÁ (UFPR), CURITIBA, PARANÁ, BRAZIL

# 8.1 Introduction

Thirty years ago at the University of Missouri, George P. Smith [1] reported, for the first time, an in vitro strategy for identifying peptide ligands for antibodies. Based on the specificity and affinity between molecules, filamentous phages displaying peptides (phage display) on the surface were selected from a combinatorial library of random peptides.

Although the phage display technology has evolved in the context of protein bioengineering, the potential of this new approach was noted years later, taking into consideration the increase in articles related to the technique.

Over the years, owing to its simplicity, cost effectiveness, reproducibility, and versatility for generating a wide range of biological ligands to almost any target, phage display has become a well-established platform for a variety of applications including screening of antibody repertoires [2], selection of binders in vivo [3], discovery of new therapies [4], inhibitors of pathologic antibodies [5], design of vaccines [6,7], studies about receptor–ligand interactions [8], epitope mapping [9], and identification of immunodominant epitopes for serologic diagnosis including neglected diseases [10–12].

This practicality in identifying binders with desired properties from any number of possibilities is feasible because this technology is based on linkage between the surfacedisplayed peptide or protein and the genetic material that encodes it, i.e., the physical association between the phage phenotype and its genotype is guaranteed. In fact, as an interconnection between molecular biology and bioengineering, the gene encoding the displayed molecule is fused to that of a phage coat protein. This strategy provides an efficient exposure of the candidate ligands to the target molecules, ensures the selective

<sup>\*</sup>Corresponding Author.

isolation of identical bacteriophages from a single host cell clone, and allows the primary structure of the phage display-derived binder to be easily determined by nucleotide sequencing.

Considering the efficiency of phage display as a selection method, regardless of research purpose, layout and strategy of screening, or which library is used (peptide, antibody, or protein), the selection of ligands requires some rounds of panning. This selective process is composed of an incubation that allows interaction with the target and posterior removal of nonspecific phages. The recovery of particles with potential affinity occurs after an efficient elution process and sequential enrichment of specific binding phages through the infection of host cells, usually *Escherichia coli*. Successful screening of phage-display libraries after each panning can be evaluated by titering the infective phages before each amplification [13]. In this phase, each infected host cell is individualized as a clone that can be recovered and stored. Thus, unamplified phages can be assayed individually just after each panning to assess their reactivity or affinity with the target.

Undoubtedly, the library quality in terms of complexity, diversity, and stability has a direct influence on efficient screening. Nevertheless, other technical elements, such as the use of high-quality affinity reagents and the process of elution, have to be considered before starting the selection of phage-display random peptide libraries.

The epitope mapping of polyclonal antibodies from patients is an alternative to obtain antigens and/or immunogens that are able to be used as binders in immunodiagnosis or to elicit specific and efficient immunologic response against an etiologic agent, especially if this agent is the cause of a neglected disease. Taking into account that usually these patients have concomitant diseases, the antibodies, as good targets, should be as specific as possible, i.e., should be able to be recovered after binding to the original antigen with high affinity.

In this same context, another important element is the elution of the bound phage. Typically, when the ligand is not known or there is insufficient quantity of antibody binders to be used in a specific competitive elution, phages are eluted with severe pH alteration, i.e., in a nonspecific manner. Employing two different models: a *bacillus* (*Mycobacterium leprae*) [11,12] and a metazoan parasite of tissues (*Taenia saginata*) [14], in this chapter, an overview about some important aspects relating to each of these procedures will be discussed.

# 8.2 Hansen Disease: Example of Rarity of Antigens and Patients' Antibodies

"I have examined a great number of them—small staff-like bodies, much resembling bacteria, lying within the cells." One hundred forty years ago, these were the words used by Gerhard Henrik Armauer Hansen (1841–1912), a Norwegian physician, to describe for the first time the *M. leprae*, the causative agent of leprosy [15]. As a tribute to him, this

chronic infectious illness is also known as Hansen disease and affects mainly the skin, mucosa, and peripheral nerves.

Although the skin and neurologic examination is fundamental, the diagnosis is confirmed by the presence of acid-fast bacilli in skin smears or biopsies. Noninvasive procedures are not available. For this reason serologic assays using the major antigenic glycolipid in the *bacillus*, phenolic glycolipid 1 (PGL-1) have been developed. Although PGL-1 is considered a good antigen [16], its application on a large scale is unfeasible because *M. leprae* is an uncultivated bacteria. Considering the epidemic character of Hansen disease in developing countries, more than ever the development of synthetic antigens [11,12,17] is desirable.

A possible alternative to the use of a native antigen in diagnosis would be the identification of peptide mimotopes (or mimic epitopes) from a phage-displayed peptide library [18] using patients' antibodies as the target. However, the successful screening of a phage-display library depends on the specificity and affinity between the ligand and its target. Thus obtaining specific antibodies after binding to the antigen is necessary.

Particularly in relation to M. leprae, two problems are evident:

- **1.** the difficult of obtaining the crude antigen, because it is a noncultivable bacteria cultured only on the footpad of nine-banded armadillos [19], and
- **2.** the low level of antibodies found in patients with type 1 T helper–dominant immune response against the bacilli [20].

The affinity purification of patients' antibodies using *M. leprae* antigens bound to polyvinylidene difluoride (PVDF) [21] membrane was the alternative to using minimal amounts of bacilli to obtain specific and rare antibodies from patients' sera. The next section discusses the most important steps of the protocol used by Alban and colleagues [11,12] to provide anti–*M. leprae* antibodies used as targets for the phage-display peptide library.

# 8.3 Antibody Purification Based on Affinity to Membrane-Blotted Antigens

# 8.3.1 Obtaining Crude Human Immunoglobulin by Ammonium Sulfate Precipitation

Precipitating human immunoglobulins by using ammonium sulfate [22] reduces contaminants and concentrates antibodies from a large starting volume of serum. Besides being easy to perform, it is a rapid and inexpensive method.

Protocol 1.

- **1.** The ammonium sulfate saturated solution (4.3 M) should be prepared at least 24 h before use.
- **2.** Centrifuge serum for 30 min at  $10,000 \times g$  at 4°C. Remove any remaining lipids that would compromise the salt precipitation.

- **3.** Determine the total sample volume and transfer it to a beaker containing a stir bar. Place the beaker into an ice bath and on a magnetic stirrer.
- **4.** Stir gently and add the ammonium sulfate—saturated solution drop by drop slowly and continuously to produce a 50% final saturation, i.e., add the saturated solution to the sample in the proportion of 1:2.

Adding the saturated solution very slowly guarantees the homogeneity of the salt concentration at the dripping site and consequently prevents the precipitation of contaminants.

- **5.** After dripping the entire volume of ammonium sulfate, keep the beaker in the stirring plate at 4°C overnight to ensure complete precipitation.
- **6.** Transfer the milky solution to conical tubes and centrifuge at  $10,000 \times g$  for 15 min at 4°C.
- **7.** Remove the supernatant and carefully invert the tube on an absorbent paper to drain the residual liquid.
- **8.** Initially, dissolve the pellet in 20% of the original volume in phosphate-buffered saline (PBS) (0.05 M, pH 7.4) by drawing the solution into and out of the pipette carefully.
- **9.** After total dissolution, complete the volume by adding PBS up to 40% of the original volume and dialyze it against PBS (100 vol) at 4°C for 3 h. New buffer changes should be done after 6 and 12 h of dialyzing (overnight) to remove any residual ammonium sulfate from the antibody solution.

*In the last dialysis, to prevent bacterial contamination, it is recommended to add 0.03% thimerosal.* 

**10.** Centrifuge to remove aggregates, verify the concentration, make aliquots to prevent repeated freeze—thaw cycles, and keep it at  $-70^{\circ}$ C or make working aliquots and keep them at 4°C for a few months.

#### 8.3.2 Affinity Purification of Human Antibodies Using Antigens Immobilized on Blot Membranes

An efficient method of antibody purification results in the selective enrichment of specific molecules based on their functionality. This process depends on the stability of the antigen and of the preservation of its antigenicity when immobilized on a matrix.

The purification of human antibodies from PVDF [21] or nitrocellulose [10] membranes, as illustrated in Fig. 8.1, containing immobilized antigens has been shown to be a useful and convenient method when both antibodies as antigens are scarce. Especially in relation to ethical aspects, sometimes the serum volume collected from patients is limited and it is necessary to work with minute quantities.

Although some renaturation of proteins occurs during the process [23], antibodies whose recognition depends on the antigen conformation might not be purified using this methodology.



FIGURE 8.1 Overview of some steps to purify immunoglobulins from human serum based on the affinity to membrane-blotted antigens.

#### Protocol 2.

- **1.** First, prepare a polyacrylamide gel (sodium dodecyl sulfate-polyacrylamide gel electrophoresis) in terms of acrylamide concentration, according to the specifics and characteristics of the sample.
- **2.** If the sample consists of crude antigen, run 1-10 mg of protein on one large lane in a 1.5-mm tick, 10-cm-wide gel size. On the other hand, if the antigen is a purified protein,  $20-100 \ \mu \text{g}$  may be used.
- **3.** Once the protein(s) has been resolved, transfer them to a PVDF membrane prewetted with methanol or ethanol according to the manufacturer's instructions.

Protein transfer to a membrane support is done to make the proteins accessible to ligands such as antibodies. Therefore this support could also be a nitrocellulose membrane. Although nitrocellulose is cheaper than the PVDF membrane, the latter is recommended for the affinity purification of antibodies because of its higher protein binding capacity (180  $\mu$ g/cm<sup>2</sup> against 100  $\mu$ g/cm<sup>2</sup>) and because of its higher resistance to mechanical damage and to a wide operating range of pH, which makes it resilient to the elution and reprobing process.

**4.** Visualize the proteins by staining the PVDF membrane within freshly prepared Ponceau solution [0.5% (w/v) in 1% (v/v) acetic acid] and rinse it with 50\% methanol to destain.

Nitrocellulose membrane must be rinsed with distilled water until the bands become clearly visible.

**5.** Trim away any excess membrane and block it with 0.3% (v/v) Tween-20 in PBS [24] for 1 h at room temperature.

In the case of using purified proteins, cut out the band of interest after staining and destaining and transfer the small membrane pieces to a 1.5-mL microtube.

- **6.** Wash the membrane with PBS containing 0.05% (v/v) Tween 20 (PBS-T) three times for 5 min.
- **7.** Place the membrane into a appropriately sized plastic bag containing 1 mg/mL crude immunoglobulin fractions in PBS-T and close it using a plastic bag sealer. Incubate at 4°C overnight with gentle agitation.

Using this method of antibody purification, it is reasonable to assume that small amounts of antibody will be recovered, considering that the serum concentration of an antibody against an immunodominant antigen is generally between 50 and 200  $\mu$ g/mL [25] and that recovery of a specific antibody is between 25% and 33% [21].

**8.** Remove the membrane from the plastic bag and wash it with PBS-T six times for 5 min.

More stringent washes by using PBS and 0.1% Tween 20 can be used to reduce nonspecific binding and to select high-affinity antibodies.

**9.** Elute the antibodies by incubating the membrane with 2 mL of 0.1 M glycine, 0.15 M NaCl, pH 2.8, at room temperature for 30 min, and agitation.

If the elution is done from small membrane pieces, add around 200  $\mu$ L of 0.1 M glycine, 0.15 M NaCl, pH 2.8, to the microtube or in enough of a volume to maintain them completely submerged in solution.

- **10.** Transfer the glycine solution to a tube containing a volume of 1 M Tris–HCl, pH 9.0, to neutralize the pH (7.0–7.5).
- 11. Repeat the purification process; the membrane can be reprobed several times.
- **12.** Pool eluates, dialyze it against PBS, and verify the antibody concentration spectrophotometrically by absorbance at 280 nm.

This dosage method is interesting because the sample is recoverable. At 280 nm, human antibody solution at concentration of 1 mg/mL gives an absorbance of 1.38.

**13.** Add bovine serum albumin (BSA) to 1 mg/mL to reduce loss from adsorption to tubes and to stabilize molecules. Make aliquots to prevent repeated freeze—thaw

cycles and keep at  $-70^{\circ}$ C for long-term storage or working aliquots and keep at  $4^{\circ}$ C for a few months.

### 8.4 Parasitic Infections: Elution Strategy of Phage-Displayed Peptides When Antigens Are Available

The World Health Organization estimates that more than a quarter of the total population in the world, approximately 1.4 billion people, are affected by helminth infections, especially in the developing countries [26]. These data, which are alarming in themselves, do not include protozoan infections. It is not rare that besides been caused by more than one parasitic worm, helminth infections coexist with other etiologic agents such as *Plasmodium* spp. [27], a phenomenon that increasingly requires the use of specific antigens for differential immunodiagnosis.

Considering the number of patients, the helminth dimension, and the efforts for developing techniques for in vitro cultivation of these parasites [28], limitations in obtaining a crude antigen would not be expected in some fundamental steps of phage-display library selection, as occurs with *M. leprae bacillus*. However, one of the most relevant problems relating to search of immunodominant proteins from parasitic helminths lies in the fact that most of them have complex cycles and different evolutionary stages.

The taeniosis/cysticercosis complex is one example. *T. saginata*, a tapeworm species classified within the Platyhelminthes phylum, is a parasite in the small intestine of humans (taeniosis), the only definitive host. The ingestion of *T. saginata* eggs scattered in pastures or water by cattle, the intermediate host, culminates with the development of the infective larval stage in the musculature (cysticercosis). The life cycle is perpetuated when a human eats raw or improperly cooked bovine meat containing a viable larval stage [29].

Habits of consuming undercooked steaks facilitates the continuity of the taeniosis/ cysticercosis complex and makes taeniosis an important social and public health problem in developing countries. Some efforts must be done to break the parasite cycle, such as to establish wider health education regarding the prevention of parasitosis is and high standards of veterinary control over slaughter practices, and to adopt modern and more sensitive diagnostic tools [30].

To propose an alternative to the current method of inspection by visual detection, the phage-display technique was employed as a biotechnological approach able to screen epitopes recognized by antibodies from bovines infected with *T. saginata* larvae to develop a new generation of antigens. To this end, specific antibodies purified against larvae antigens were used for immunopanning phage-displayed libraries [14]. The libraries used in this work were described by Bonnycastle and colleagues [18] and obtained from J. Scott (Simon Fraser University, Burnaby, BC, Canada).

In this topic, by using the bovine cysticercosis as model, the reactivity of ligands after two different elution protocols (acid and competitive) was verified. The results obtained with both strategies were compared after probing the original antibodies against synthetic sequences corresponding to each peptide phage using the SPOT method [31,32].

# 8.4.1 Screening Phage Displayed Peptide Library: Acid $\times$ Competitive Elution

Generally, considering that bacteriophages are able to tolerate extreme conditions and remain infective, the elution of high-affinity binders from phage libraries can be done by changing drastically the pH, i.e., by using 0.1 M glycine–HCl, pH 2.2 [9,10] or 0.1 M triethanolamine, pH 11 [33], solutions.

Although the acid—basic elution can result in the selection of binders to denatured antibodies or targets [34], this nonspecific elution is convenient if the original antigen or ligand of the antibody or target is not known, is not available, or is obtainable in low amounts, conditions that restrict the elution based on the binding characteristics of the target.

To demonstrate the utility and differences of both of these elution strategies, the following provides a step-by-step guide of how to perform the selection procedure for screening phage-displayed peptide libraries.

#### 8.4.2 Panning Phage Display Library

Protocol 3.

**1.** For the first panning round, coat an immunotube with  $5 \mu g/mL$  bovine immunoglobulin purified against *T. saginata* crude antigen diluted in 1.5 mL 0.1 M carbonate buffer, pH 8.6. Seal the tube with parafilm and incubate overnight at 4°C.

To select binders with high affinity and specificity to the antibodies, in the subsequent pannings it is recommended that the target concentration be gradually decreased. Thus in the following selections, the immunotube must be coated with 1.0 and 0.5  $\mu$ g/mL of antibodies, respectively.

- **2.** Using a tube rotator, wash the immunotube five times for 5 min with TBS/0.05% Tween 20, pH 7.5 (TBS-T) and fill the tube completely with TBS-T containing 3% BSA previously filtered on a 0.22-µm-pore-size membrane filter. Seal the tube and incubate for 2 h at 37°C to block nonspecific binding sites.
- **3.** After washing, add an appropriate titer, generally 100-fold the library complexity [18] of each library in a final volume of 1.5 mL TBS-T containing BSA 1 mg/mL. Seal the tube and incubate overnight at 4°C by using a tube rotator.

Because BSA is used as a blocking agent, it is recommended to use the protein in the binding buffer to avoid the recovery of BSA-specific binders.

**4.** Remove unbound and nonspecific phages by washing the immunotube 10 times with TBS-T 0.5%, five times for 5 min each with TBS-T 0.05%, and finally twice with TBS.

If a nonspecific elution, such as drastic changes in pH, is chosen as a strategy, an appropriate washing stringency must be done to result in a successful affinity selection. Practically, the washing conditions must eliminate target-unrelated binders so that the selecting process is not compromised.

- 5. Specific and nonspecific elution
  - **5.1** For nonspecific elution, elute the bound phage from the antibody by adding 1.5 mL of 0.1 M glycine–HCl, pH 2.2, containing 1.0 mg/mL BSA for 30 min at room temperature. Transfer the eluted phages to a fresh 2-mL microtube and neutralize the pH immediately by adding 70  $\mu$ L 2 M Tris–HCl, pH 9.0. Check the pH by using test strips.

The introduction of a sonication process has shown good results [35]. The adapted protocol consists of removing nonspecific phages by washing the immunotube with TBS-T as described and rinsing it four times with glycine–HCl (0.1 M, pH 2.2). Another 1.5 mL of glycine–HCl solution is added to the tube. After sealing, the tube is inserted into a sonicator water bath (50 kHz) for 10 min and the eluate is neutralized as described previously.

**5.2** For a specific elution, elute bound phage-displayed peptides by adding a concentration of antigen 100-fold higher than that of immunoglobulins immobilized in the immunotube. After incubation for 2 h at room temperature using a tube rotator, the eluted phages can be cloned by infecting host cells.

To remove lower-affinity binders, a previous elution can be done with the competitive antigen for 60 min at room temperature. After removing the eluate, a second elution is carried out under the same conditions for 24 h [36].

Typically, when the competitor antigen is a known protein, a recombinant protein, for example, specific elution is performed in the presence of 10- to 100-fold excess molar protein. In the competitive elution just described, the competitor agent was not known; thus the crude parasite antigen concentration used was 100-fold (w/w) higher than that of purified immunoglobulins.

**6.** Before amplification, save an aliquot of the eluted phages for titering (around  $50 \ \mu L$ ) to evaluate the output input ratio of phages after each round of selection.

#### 8.4.3 Amplification

After each round of panning, the amplification procedure is necessary to enrich bacteriophages carrying peptides with high affinity and specificity to the target. However, some promising clones may have difficulty infecting host cells; consequently, they may not be efficiently amplified. On the other hand, some peptides have characteristics that do not interfere with the replication process. Thus some resultant peptide sequences selected with more frequency are not necessarily those with higher affinity to the ligand, but those whose infection process may be easier if compared with others.

#### Protocol 4.

**1.** On the day before phage elution, pick a single colony of host cell from a freshly Luria Bertani (LB) agar plate and inoculate it into 10 mL LB medium without antibiotics in a 50-mL aerated tube. Grow the preculture overnight, shaking at 225 rpm at 37°C.

Each day a new culture of host cells is necessary, prepare a preculture from a newly prepared plate.

It is possible to prepare a preculture from a thawed glycerol stock of bacteria stored at  $-80^{\circ}$ C by using 100-fold dilution into autoclaved LB medium. Avoid rethawed glycerol stocks; this can reduce the bacterial viability.

**2.** During the elution process, place 5 mL from the preculture in a 500-mL Erlenmeyer flask containing prewarmed 100 mL LB to prepare an exponentially growing culture of host cells: in this case, *Escherichia coli* K91 cells. Incubate cells at 37°C, shaking at 225 rpm until the culture reaches 1.8 at 600 nm (approximately 3–4 h). After that, incubate the culture for 30 min, shaking at 50 rpm at 37°C to pili formation.

After approximately 2 h 30 min, start reading the bacterial suspension diluted onehalf in LB medium. When the  $A_{600}$  of the dilution reaches approximately 1.6, incubate the culture slow the rotation (50 rpm) for 30 min to pili formation

**3.** Incubate eluted phages with 5 mL of pilated bacteria for 10 min at 37°C without agitation, and add to prewarmed 95 mL LB medium containing a low concentration of the appropriate selective antibiotic: in this case, 0.2 μg/mL tetracycline. After new 30-min incubation at 37°C at 225 rpm, add tetracycline to yield a final concentration of 20 μg/mL and grow the culture overnight.

Tetracycline hydrochloride is not soluble in water. For a stock solution of 20 mg/mL, add 200 mg powder in a final volume of 10 mL 60% ethyl alcohol. Vortex until completely dissolved, filter, and stock in the dark at  $-20^{\circ}$ C. Wrap tetracycline containing plates in aluminum foil to prevent light exposure.

**4.** Centrifuge cell culture suspension  $(3000 \times g, 10 \text{ min})$  and transfer the supernatant to a 250-mL centrifuge flask and add 15% (v/v) of autoclaved polyethylene glycol (PEG) 8000–NaCl solution (20% PEG 8000, 2.5 M NaCl). Homogenize by shaking and keep the centrifuge flask at 4°C for 12–16 h.

During the autoclaving process, the PEG–NaCl solution may separate into two phases. Let it cool and then shake it; the phase separation is reversible.

- **5.** After centrifugation (8000  $\times$  g, 40 min, 4°C), collect the phage pellet, resuspending it in 3 mL TBS (50 mM Tris-HCl, 150 mM NaCl, pH 7.5) by shaking at 150 rpm, for 30 min and 37°C. Transfer it to 1.5-mL microtubes and centrifuge the phage solution  $10,000 \times g$  for 15 min.
- **6.** Proceed with titer determination and store the supernatant at 4°C in TBS containing 0.02% NaN<sub>3</sub> for subsequent rounds of panning.

#### Phage Titration 8.4.4

Titration is the procedure used to obtain the phage concentration, i.e., to determine the number of infectious particles per milliliter of stock solution. In fact, this information is necessary to know the volume of phage stock to be used in the next panning.

Also important is the evaluation of output-input phage ratio after each round of panning to determine the phage recovery efficiency. A gradual increase of output-input phage ratio between pannings is the first evidence of a successful screening.

Titering phages is also convenient because the individual colonies obtained from plates after each panning could be immediately assayed against the antibody or ligand.

#### Protocol 5.

- **1.** Pick a single colony from an agar plate and inoculate into 10 mL LB medium without antibiotics, as described earlier. Incubate at 37°C, at 225 rpm overnight.
- **2.** Add 5 mL from the preculture in a 500-mL Erlenmeyer flask containing prewarmed 100 mL of LB. Incubate the flask at 37°C, shaking at 225 rpm, and continue culturing until the culture reaches 1.8 at 600 nm. After that, incubate the culture for 30 min, shaking at 50 rpm at 37°C to pili formation.
- **3.** For titration of eluted (output) phages, prepare 100-fold serial dilution as following: **a.** Add 10  $\mu$ L eluted phage to 990  $\mu$ L LB in a 1.5-mL microtube (10<sup>-2</sup> dilution). **b.** Add 10  $\mu$ L of 10<sup>-2</sup> dilution to 990  $\mu$ L LB in a 1.5-mL microtube (10<sup>-4</sup> dilution). **c.** Add 10  $\mu$ L of 10<sup>-4</sup> dilution to 990  $\mu$ L LB in a 1.5-mL microtube (10<sup>-6</sup> dilution).
- **4.** For titration of amplified (input) phages, prepare a 100-fold serial dilution as described, including  $10^{-8}$ ,  $10^{-10}$ , and  $10^{-12}$  dilutions.
- 5. Transfer 200-µL K91 cells from an exponentially growing culture to autoclaved 1.5mL microtubes and add 10 µL of phage dilutions. To bacterial cells add an aliquot from  $10^{-4}$  and  $10^{-6}$  dilutions of output phages and  $10^{-8}$ ,  $10^{-10}$ , and  $10^{-12}$  dilutions of input phages, respectively (Fig. 8.2). Mix well by carefully pipetting up and down. Incubate at 37°C for 15 min without shaking and for 15 more min of shaking at 225 rpm at 37°C;
- **6.** Prewarm freshly prepared 100-mm LB agar plates containing 20  $\mu$ g/mL tetracycline at 37°C for 2 h. Label one plate per sample, including one for uninfected K91 cells used as a control. After overnight incubation, the control plate containing only K91 E. coli and dilution medium should have no plaques.
- 7. Gently add 200  $\mu$ L of each infected K91 cell onto the prewarmed plates and spread them until the plate is completely dry. Invert and place the plate overnight in a 37°C incubator to allow colonies to grow and be isolated.



**FIGURE 8.2** Procedure to determine the phage stock titer. A series of 100-fold dilutions is obtained by putting 10  $\mu$ L of the stock solution into 990  $\mu$ L of LB medium (10<sup>-2</sup>). Each subsequent dilution is made from the previous dilution, as illustrated. After that, 10  $\mu$ L of each dilution is incubated with host cells, which are then spread onto LB agar plates.

**8.** Observe plates and determine the titer in transducing units per mL (TU/mL) according the equation:

 $TU/mL = number \ of \ colonies \times fold \ dilution \times 100$ 

For example:

$$50 imes 10^8 imes 100 = 5.10^{11} \, \mathrm{TU/mL}$$

The multiplication factor 100 is used to obtain the phage concentration per milliliter once  $10 \,\mu$ L corresponds to the volume of phages used to infect K91 host cells.

Generally, it is recommended to select the plate with a number of colonies around 100 to determine the titer with respect to minimizing the statistical counting error [37]. If it is not possible, i.e., if all plates have fewer colonies than expected, an independent determination is recommended for each sample, and then it is best to take the average value from it to obtain the sample titer [38].

**9.** Based on the measurements published by Day and Wiserman [39], another possibility is to determine the phage concentration (phage particles per milliliter) by absorbance at 269 and 320 nm [40] according to the following formula:

Particles/mL =  $(A_{269} - A_{320}) \times (6.10^{16})/(DNA \text{ bases in the phage genome})$ 

### 8.4.5 Cloning

Individual colonies derived from titering plates of the final eluate, generally the third one, or output phages after each panning round should be cloned, propagated for further analysis, and/or saved in sterile 96-well culture plates in medium containing 20% glycerol at  $-80^{\circ}$ C.

Protocol 6.

- **1.** Fill the wells of two microplates with 200  $\mu$ L LB medium containing 20  $\mu$ g/mL tetracycline and inoculate each with a single colony from the agar plate using toothpicks or sterile pipette tips.
- **2.** Cover the microplates with a polystyrene lid and incubate at 37°C overnight using an orbital shaker at 225 rpm.
- **3.** The next day, spin one of the plates in a centrifuge equipped with a microplate adaptor at  $1500 \times g$  for 5 min and collect the supernatants to subject them to an enzyme-linked immunosorbent assay (ELISA) analysis.
- **4.** To save the clones or to perform reanalysis at a later time, add an appropriate amount of glycerol to the second microplate to get a final concentration of 20% and, for high viability, store at  $-80^{\circ}$ C.

# 8.5 Phage Enzyme-Linked Immunosorbent Assay

It is possible to use several ELISA configurations to select phages according their reactivity to antibody or target. Here, the interaction between the phage-displayed peptide and its target is evaluated by coating a microplate with the antibodies initially used as targets in the rounds of panning.

Protocol 7.

- **1.** Coat 96-well microtiter plate overnight at 4°C with 10  $\mu$ g/mL antibody in 100  $\mu$ L of 0.1 M carbonate buffer, pH 8.6.
- **2.** The next day, wash the plate with PBS-T three times.
- **3.** Add 150  $\mu L$  of 5% skim milk in PBS-T to each well and incubate the plate at 37°C for 1 h.
- **4.** After washing the plate three times with PBS-T, add 100  $\mu$ L of each phage supernatant isolated (Protocol 6) diluted 1:2 in PBS-T containing 2% skim milk and incubate at 37°C for 2 h.

As negative and positive controls use 100  $\mu$ L of wild-type phage supernatant and the native antigen, respectively, in concentrations previously established.

- **5.** Wash the plate three times with PBS-T.
- **6.** To detect phage binding, add to each well 100  $\mu$ L of anti M13/fd bacteriophage antibody—horseradish peroxidase diluted in PBS-T containing 2% skim milk according to the manufacturer's recommendations, and incubate the plate at 37°C for 1 h.

**7.** Wash the plate five times with PBS-T. Add to each well 100  $\mu$ L of a substrate chromogen solution containing 0.02% (v/v) hydrogen peroxide and 0.2 mg/mL *o*-phenylenediamine in 0.1 M citrate-phosphate buffer, pH 5.0, and incubate for 15 min at room temperature.

After incubation, if the microplate is read immediately, the color development should be measured spectrophotometrically at the wavelength of 450 nm. On the other hand if a stopping agent is add to each well, for example,  $20 \,\mu\text{L}$  of  $2 \,M H_2 \text{SO}_4$ , the orange color is measurable at 492 nm.

**8.** Save all binding clones by using 20% glycerol as described in Protocol 6. Amplify them to isolate single-stranded DNA (ssDNA) and use the primer 5'-TCG GCA AGC TCT TTT AGG-3' for sequencing.

To analyze phage-displayed peptides by sequencing, a high concentration of ssDNA may be necessary. Sometimes the procedure of amplification does not result in enough material. For this reason it is recommended to prepare ssDNA from PEG–NaCl-precipitated phages.

# 8.6 Characterization of Peptide Sequences

In dealing with tropical neglected diseases, one of the most important applications of the phage display is to identify mimotopes with effective antigen-like characteristics such that the original antigen can be replaced in diagnostic assays without losing sensitivity and specificity [41].

Besides antigenicity, if phage-displayed peptides are selected for binding to pathogen-specific antibodies capable of performing important immunologic roles such as neutralization [9], activation of the classical complement pathway [42], and antibody-dependent cellular cytotoxicity [43], and are able to induce response immune, they may prove to be good vaccine candidates.

Interest in identifying new diagnostic targets using the phage-display technique is based on the fact that most functional antibodies recognize conformational antigenic determinants [44]. In addition, the technical procedures of selecting immunodominant epitopes of the etiologic agent direct them to immunoglobulins to obtain a more specific and sensitive diagnostic test.

Use of the ELISA, as described earlier, is an initial trial to validate the interaction between phage-displayed peptides and antibodies used initially as targets. This immunochemical assay determines the relative affinity between the phage-presented peptide and the antibody without considering whether this interaction depends on the peptide conformation on the phage surface [18].

The chemical synthesis of the peptides (mimotopes) on the cellulose membrane (SPOTsynthesis) [31] may be an effective approach to confirm antigen—antibody reactivity. This technique allows straightforward localization and the precise identification of each key residue essential for recognition [32]. In addition to being a method based on the same procedure used in conventional chemical synthesis on resin, the cost per peptide sequence corresponds to less than 1% of a soluble synthetic peptide [45], not to mention the possibility of regenerating and reprobing the membrane several times with excellent reproducibility.

SPOT technology is a parallel chemical synthesis of peptides [32] on small circular regions on a cellulose membrane, in which protected and activated amino acids are spotted positionally addressed, forming a pattern of small spots.

On each spot a different peptide is synthesized. Although longer peptides may be prepared, a peptide length between 12 and 15 amino acids is commonly used.

Once a mimotope has been identified by phage display, assayed by ELISA and its sequence is known (Fig. 8.3A), it is possible to determine the contribution of individual amino acid residues to binding energy of the antigen—antibody interactions. It is necessary to prepare a series of alanine analogs of the epitope (alanine scanning) (Fig. 8.3B) and probe them with antibody [32,46]. In peptides with alanine in their original sequence, this residue can be substituted by serine.

Comparing the reaction of the antibody against the wild type and an alaninescanning mutant peptide, it is possible to indicate, according to the decrease, or sometimes the lack of reactivity, which amino acid residue is crucial to mimotope– antibody interactions (Fig. 8.3C).

In Fig. 8.4A, four phage display-derived-peptides corresponding to mimotopes of *T. saginata* obtained by competitive elution [14] were synthesized on a cellulose



**FIGURE 8.3** Alanine scanning. (A) Illustration of a filamentous phage that displays peptides at the N-terminus of the pVIII coat protein. (B) Representation of the reaction between a wild-type peptide and analogs with alanine replacement with a specific antibody revealed with an enzyme-labeled secondary antibody and enhanced chemiluminescence detection on an SPOT membrane. (C) Determination of which residues are important to the antigen—antibody interaction. Each bar represents the reactivity of a serine or alanine-mutated peptide. Reactivity of the wild-type peptide (*first bar*) was considered 100%.



**FIGURE 8.4** Comparison of anti *Taenia saginata* antibody reaction and eight cellulose-bound phage-displayed peptides, obtained after competitive (A) and acid elution (B). The wild-type peptide (WT) is synthesized in the first spot of the sequence following by its analogs with alanine replacement. The amino acid residues substituted by alanine (or serine) are indicated on the top of each spot.

membrane in which an alanine was used to replace each amino acid residue sequentially. The same procedure was performed with *T. saginata* mimotope sequences (Fig. 8.4B) eluted in a nonspecific manner, i.e., through a change of pH.

There was a significant difference between the antigenic reactivity of cellulose-bound phage-displayed peptides from specific and nonspecific elution when the membranes were probed with a positive bovine serum pool under the same conditions and at the same time. These data reinforce the concept that the competitive elution is a better strategy to affinity select specific ligands.

Although the specific elution is the best alternative, both strategies present similar results regarding the absence of reactivity in some synthetic peptides spotted on the

Clone	Amino Acid Sequence	Residues (n)	Elution
Pep 1	H F Y Q I T W L P N T F P A R	15	Competitive elution [14]
Pep 2	T C I W Q W P D W A C K	12	
Pep 3	V H T S I R P R C Q P R A I T P R	17	
Pep 4	M	17	
Pep 5	D C G T D V V Y F E C R	12	Acid elution
Pep 6	T V H T Q I L L C P P N F W P V T	17	
Pep 7	V K V T W Y Y Q C E T A N K T E P	17	
Pep 8	S C T H L V M S T P C T	12	

 Table 8.1
 Peptide (Pep) Sequences Selected by Specific and Nonspecific

 Elution
 Peptide (Pep) Sequences Selected by Specific and Nonspecific

membrane, such as peptides 2 and 8 (Fig. 8.4). Considering that both peptides present two cysteines and that no treatment on the membrane was done to obtain cysteinebridged peptides [32], one hypothesis is that the conformation of these peptides is probably essential for recognition by antibodies, and this characteristic is preserved only when they are exposed on the surface of the phage.

Curiously, no consensus sequences have been selected between the phage-displayed peptides (Table 8.1). Only Pep 1 (HFYQITWL**PNTF**PAR) [14], eluted by competition with *T. saginata* crude antigen, presented some homology with a highly conserved sequence from GK-1 (GYYYPSD**PNTF**YAPPYQ) [47], a peptide from KETc7, a protein found in the Taeniidae family [48] with immunogenic [49] and antigenic characteristics [50] against cysticercosis caused by *Taenia solium*.

Regarding neglected tropical diseases, phage display technology integrated with other technologies such as SPOT synthesis may eventually increase the possibility of obtaining high-affinity ligands so as to contribute to the identification of molecules suitable for diagnosis or to the rational design of mimotope-based vaccines.

# 8.7 Conclusion and Perspectives

Thirty years ago, George P. Smith described the phage display technique without realizing the impact that it would have on the generation of monoclonal antibodies, drug development, protein engineering, cancer therapy, design of vaccines, and the diagnosis of several diseases. Considered an inexpensive, simple, and resourceful technique, phage display, which is aligned to the chemical synthesis of mimotope/epitope sequences, has been consolidating as a technology of great importance in the development of alternative diagnostic methods mainly with regard to neglected tropical diseases.

# Acknowledgments

We thank Janaína Capelli-Peixoto for her skillful technical contributions. We also are grateful to Elisa Haddad de Oliveira for drawing the illustrations.

# References

- [1] G.P. Smith, Filamentous fusion phage: novel expression vectors that display antigens on the virion surface, Science 228 (1985) 1315–1317.
- [2] J. McCafferty, A.D. Griffiths, G. Winter, D.J. Chiswell, Phage antibodies: filamentous phage displaying antibody variable domains, Nature 348 (1990) 552–554.
- [3] R. Pasqualini, E. Ruoslahti, Organ targeting in vivo using phage display peptide libraries, Nature 380 (1996) 364–366.
- [4] B.A. Desimmie, M. Humbert, E. Lescrinier, J. Hendrix, S. Vets, R. Gijsbers, R.M. Ruprecht, U. Dietrich, Z. Debyser, F. Christ, Phage display-directed discovery of LEDGF/p75 binding cyclic peptide inhibitors of HIV replication, Molecular Therapy 20 (2012) 2064–2075.
- [5] S. Villard, S. Lacroix-Desmazes, T. Kieber-Emmons, D. Piquer, S. Grailly, A. Benhida, S.V. Kaveri, J. M. Saint-Remy, C. Granier, Peptide decoys selected by phage display block in vitro and in vivo activity of a human anti-FVIII inhibitor, Blood 102 (2003) 949–952.
- [6] J. Capelli-Peixoto, C. Chávez-Olórtegui, D. Chaves-Moreira, J.C. Minozzo, J. Gabardo, K.N. Teixeira, V. Thomaz-Soccol, L.M. Alvarenga, J. de Moura, Evaluation of the protective potential of a *Taenia solium* cysticercus mimotope on murine cysticercosis, Vaccine 29 (2011) 9473–9479.
- [7] C.M. Toledo-Machado, L.L. Bueno, D. Menezes-Souza, R.A. Machado-de-Avila, C. Nguyen, C. Granier, D.C. Bartholomeu, C. Chávez-Olórtegui, R.T. Fujiwara, Use of phage display technology in development of canine visceral leishmaniasis vaccine using synthetic peptide trapped in sphin-gomyelin/cholesterol liposomes, Parasites & Vectors 8 (2015) 133–140.
- [8] J. Mandelin, M. Cardó-Vila, W.H. Driessen, P. Mathew, N.M. Navone, S.H. Lin, C.J. Logothetis, A.C. Rietz, A.S. Dobroff, B. Proneth, R.L. Sidman, R. Pasqualini, W. Arap, Selection and identification of ligand peptides targeting a model of castrate-resistant osteogenic prostate cancer and their receptors, Proceedings of the National Academy of Sciences of the United States of America 112 (2015) 3776–3781.
- [9] J. de Moura, L. Felicori, V. Moreau, G. Guimarães, C. Dias-Lopes, L. Molina, L.M. Alvarenga, P. Fernandes, F. Frézard, R.R. Ribeiro, C. Fleury, C. Nguyen, F. Molina, C. Granier, C. Chávez-Olórtegui, Protection against the toxic effects of *Loxosceles intermedia* spider venom elicited by mimotope peptides, Vaccine 29 (2011) 7992–8001.
- [10] R.C. Hell, P. Amim, H.M. de Andrade, R.A. de Avila, L. Felicori, A.G. Oliveira, C.A. Oliveira, E. Nascimento, C.A. Tavares, C. Granier, C. Chávez-Olórtegui, Immunodiagnosis of human neuro-cysticercosis using a synthetic peptide selected by phage-display, Clinical Immunology 131 (2009) 129–138.
- [11] S.M. Alban, J. de Moura, J.C. Minozzo, M.T. Mira, V. Thomaz-Soccol, Identification of mimotopes of *Mycobacterium leprae* as potential diagnostic reagents, BMC Infectious Diseases 13 (2013) 42.
- [12] S.M. Alban, J. de Moura, V. Thomaz-Soccol, S. Bührer Sékula, L.M. Alvarenga, M.T. Mira, C.C. Olortegui, J.C. Minozzo, Phage display and synthetic peptides as promising biotechnological tools for the serological diagnosis of leprosy, PLoS One 9 (2014) e106222.
- [13] G.P. Smith, V.A. Petrenko, Phage display, Chemical Reviews 97 (1997) 391-410.
- [14] R.L. Fogaça, J. Capelli-Peixoto, I.B. Yamanaka, R.P. de Almeida, J.C. Muzzi, M. Borges, A.J. Costa, C. Chávez-Olortegui, V. Thomaz-Soccol, L.M. Alvarenga, J. de Moura, Phage-displayed peptides as capture antigens in an innovative assay for *Taenia saginata*-infected cattle, Applied Microbiology and Biotechnology 98 (2014) 8887–8894.
- [15] G.A. Hansen, Undersogelser angaende spedalskhedens arsager, Norsk Mag Lacgevidensk 4 (1874)
   1-88 [English translation: G.A. Hansen, Causes of leprosy, International Journal of Leprosy 23 (1955) 307-309.].
- [16] S.N. Cho, D.L. Yanagihara, S.W. Hunter, R.H. Gelber, P.J. Brennan, Serological specificity of phenolic glycolipid I from *Mycobacterium leprae* and use in serodiagnosis of leprosy, Infection and Immunity 41 (1983) 1077–1083.
- [17] T. Fujiwara, S.W. Hunter, S.N. Cho, G.O. Aspinall, P.J. Brennan, Chemical synthesis and serology of disaccharides and trisaccharides of phenolic glycolipid antigens from the leprosy bacillus and preparation of a disaccharide protein conjugate for serodiagnosis of leprosy, Infection and Immunity 43 (1984) 245–252.
- [18] L.L.C. Bonnycastle, J.S. Mehroke, M. Rashed, X. Gong, J.K. Scott, Probing the basis of antibody reactivity with a panel of constrained peptide libraries displayed by filamentous phage, Journal of Molecular Biology 258 (1996) 747–762.
- [19] R.W. Truman, J.L. Krahenbuhl, Viable *M. leprae* as a research reagent, International Journal of Leprosy and other Mycobacterial Diseases 69 (2001) 1–12.
- [20] R.L. Modlin, Th1-Th2 paradigm: insights from leprosy, Journal of Investigative Dermatology 102 (1994) 828–832.
- [21] K. Ritter, Affinity purification of antibodies from sera using polyvinylidenedifluoride (PVDF) membranes as coupling matrices for antigens presented by autoantibodies to triosephosphate isomerase, Journal of Immunological Methods 21 (1991) 209–215.
- [22] A.F. Habeeb, R.D. Francis, Preparation of human immunoglobulin by ammonium sulfate precipitation, Vox Sanguinis 31 (1976) 423–434.
- [23] H.P. Hauri, K. Bucher, Immunoblotting with monoclonal antibodies: importance of the blocking solution, Analytical Biochemistry 159 (1986) 386–389.
- [24] B. Batteiger, W.J. Newhall, R.B. Jones, The use of tween 20 as a blocking agent in the immunological detection of proteins transferred to nitrocellulose membranes, Journal of Immunological Methods 55 (1982) 297–307.
- [25] N.S. Lipman, L.R. Jackson, L.J. Trudel, F. Weis-Garcia, Monoclonal versus polyclonal antibodies: distinguishing characteristics, applications, and information resources, ILAR Journal 46 (2005) 258–268.
- [26] P.J. Hotez, J.R. Herricks, Helminth elimination in the pursuit of sustainable development goals: a "Worm Index" for human development, PLoS Neglected Tropical Diseases 9 (2015) e0003618.
- [27] S.J. Brooker, R.L. Pullan, C.W. Gitonga, R.A. Ashton, J.H. Kolaczinski, N.B. Kabatereine, R.W. Snow, Plasmodium-helminth coinfection and its sources of heterogeneity across East Africa, Journal of Infectious Diseases 205 (2012) 841–852.
- [28] J.D. Smyth, In Vitro Cultivation of Parasitic Helminths, CRC Press, Boca Raton, 1990, pp. 276.
- [29] E.P. Hoberg, *Taenia* tapeworms: their biology, evolution and socioeconomic significance, Microbes and Infection 4 (2002) 859–866.
- [30] OIE/WHO/FAO, in: K.D. Murrell (Ed.), Guidelines for the Surveillance, Prevention and Control of Taeniasis/cysticercosis, World Organization for Animal Health, Paris, 2005.
- [31] R. Frank, Spot-synthesis: an easy technique for the positionally addressable, parallel chemical synthesis on a membrane support, Tetrahedron 48 (1992) 9217–9232.
- [32] C. Granier, S. Villard, D. Laune, Mapping and characterization of protein epitopes using the SPOT method, in: J.E. Celis (Ed.), Cell Biology, Four-Volume Set, 3rd Edition: A Laboratory Handbook, vol. 1, Elsevier Academic Press, Burlington, 2005, pp. 519–525.
- [33] I. Van Rooy, S. Cakir-Tascioglu, P.O. Couraud, I.A. Romero, B. Weksler, G. Storm, W.E. Hennink, R. M. Schiffelers, E. Mastrobattista, Identification of peptide ligands for targeting to the blood-brain barrier, Pharmaceutical Research 27 (2010) 673–682.
- [34] K. Sakamoto, Y. Ito, T. Hatanaka, P.B. Soni, T. Mori, K. Sugimura, Discovery and characterization of a peptide motif that specifically recognizes a non-native conformation of human IgG induced by acidic pH conditions, Journal of Biological Chemistry 284 (2009) 9986–9993.

- [35] M. Lunder, T. Bratkovic, U. Urleb, S. Kreft, B. Strukelj, Ultrasound in phage display: a new approach to nonspecific elution, Biotechniques 44 (2008) 893–900.
- [36] R.M. Wright, H. Gram, A. Vattay, S. Byme, P. Lake, D. Dottavio, Binding epitope of somatostatin defined by phage-displayed peptide libraries, Biotechnology (N Y) 13 (1995) 165–169.
- [37] M.F. Renschler, W.J. Dower, R. Levy, Identification of peptide ligands for the antigen binding receptor expressed on human B-cell lymphomas. Combinatorial peptide libraries, in: S. Cabilly (Ed.), Methods in Molecular Biology, vol. 87, Human Press, Inc., Totowa, NJ, 1998, pp. 209–234.
- [38] M.J. McGuire, S. Li, K.C. Brown, Biopanning of phage displayed peptide libraries for the isolation of cell-specific ligands, Methods in Molecular Biology 504 (2009) 291–321.
- [39] L.A. Day, R.L. Wiseman, A comparison of DNA packaging in the virions of fd, Xf, and Pf1, in: D.T. Denhardt, D. Dressler, D.S. Ray (Eds.), The Single-Stranded DNA Phages, Cold Spring Harbor Laboratory, Cold Spring Harbor, NY, 1978, pp. 605–625.
- [40] G.P. Smith, J.K. Scott, Libraries of peptides and proteins displayed on filamentous phage, Methods in Enzymology 217 (1993) 228–257.
- [41] S. Deroo, C.P. Muller, Antigenic and immunogenic phage displayed mimotopes as substitute antigens: applications and limitations, Combinatorial Chemistry & High Throughput Screening 4 (2001) 75–110.
- [42] M. Li, Z. Yan, W. Han, Y. Zhang, Mimotope vaccination for epitope-specific induction of anti-CD20 antibodies, Cellular Immunology 239 (2006) 136–143.
- [43] L. Yang, R. Xing, C. Li, Y. Liu, L. Sun, X. Liu, Y. Wang, Active immunization with Tocilizumab mimotopes induces specific immune responses, BMC Biotechnology 15 (2015) 46.
- [44] D.C. Benjamin, J.A. Berzofsky, I.J. East, F.R. Gurd, C. Hannum, S.J. Leach, E. Margoliash, J.G. Michael, A. Miller, E.M. Prager, M. Reichlin, E.E. Sercarz, S.J. Smith, P.E. Todd, A.C. Wilson, The antigenic structure of proteins: a reappraisal, Annual Review of Immunology 2 (1984) 67–101.
- [45] K. Hilpert, D.F. Winkler, R.E. Hancock, Peptide arrays on cellulose support: SPOT synthesis, a time and cost efficient method for synthesis of large numbers of peptides in a parallel and addressable fashion, Nature Protocols 2 (2007) 1333–1349.
- [46] S. Piquer, L. Valera, V. Lampasona, B. Jardin-Watelet, S. Roche, C. Granier, F. Roquet, M.R. Christie, T. Giordano, M.L. Malosio, E. Bonifacio, D. Laune, Monoclonal antibody 76F distinguishes IA-2 from IA-2beta and overlaps an autoantibody epitope, Journal of Autoimmunity 26 (2006) 215–222.
- [47] D. Rassy, R.J. Bobes, G. Rosas, V.H. Anaya, K. Brehm, B. Hernandez, J. Cervantes, S. Pedraza, J. Morales, N. Villalobos, A.S. de Aluja, J.P. Laclette, C.M. Nunes, G.F. Biondi, G. Fragoso, M. Hernandez, E. Sciutto, Characterization of S3Pvac anti-cysticercosis vaccine components: implications for the development of an anti-cestodiasis vaccine, PLoS One 5 (2010) e11287.
- [48] K. Manoutcharian, G. Rosas, M. Hernandez, G. Fragoso, A. Aluja, N. Villalobos, L.F. Rodarte, E. Sciutto, Cysticercosis: identification and cloning of protective recombinant antigens, Journal of Parasitology 82 (1996) 250–254.
- [49] A. Toledo, C. Larralde, G. Fragoso, G. Gevorkian, K. Manoutcharian, M. Hernández, G. Acero, G. Rosas, F. López-Casillas, C.K. Garfias, R. Vázquez, I. Terrazas, E. Sciutto, Towards a *Taenia solium* cysticercosis vaccine: an epitope shared by *Taenia crassiceps* and *Taenia solium* protects mice against experimental cysticercosis, Infection and Immunity 67 (1999) 2522–2530.
- [50] G. Gevorkian, K. Manoutcharian, C. Larralde, M. Hernandez, J.C. Almagro, M. Viveros, J. Sotelo, E. Garcia, E. Sciutto, Immunodominant synthetic peptides of *Taenia crassiceps* in murine and human cysticercosis, Immunology Letters 49 (1996) 185–189.

## Systems and Synthetic Biology Applied to Health

T. Mendes<sup>1, \*</sup>, F. Castiglione<sup>2</sup>, P. Tieri<sup>2</sup>, L. Felicori<sup>1, \*</sup>

<sup>1</sup>FEDERAL UNIVERSITY OF MINAS GERAIS, UFMG, BELO HORIZONTE, BRAZIL; <sup>2</sup>INSTITUTE FOR APPLIED MATHEMATICS (IAC), NATIONAL RESEARCH COUNCIL OF ITALY (CNR), ROME, ITALY

# 9.1 Systems Biology: The Dawn of a Holistic Vision in Biomedicine

The concept of systems biology, which has its roots in the first studies of enzyme kinetics at the beginning of the 20th century, has been used extensively in the life sciences only starting from the beginning of the current century. Modern systems biology can be seen through a variety of aspects, (not comprehensively) encompassing:

- **1.** a field of study that focuses on the complex interactions among biological systems components;
- **2.** a paradigm antithetical to reductionism (integrating instead of reducing, and observing the whole instead of single parts);
- **3.** a research protocol, i.e., a recursive sequence of steps that includes hypothesis and modeling, experimental validation, quantitative description, improved hypothesis and modeling, and so on; and
- **4.** a socioscientific phenomenon that regards the strategy devoted to pursuing the integration of massive, heterogeneous data as coming from different experimental sources and different methodologies and instrumentation, as well as people from disparate scientific backgrounds [1].

Typically the main reason referred to as when taking into account a "system" is that "the whole is greater than the sum of the parts." The interaction of the system's parts enables peculiar properties or functions to emerge that would not been expected from the single elements each on their own. Such emergent properties arise out of "more elementary" entities, and yet are "novel" or "irreducible" with respect to them. This irreducible nature makes emergent properties not easily predictable; in other words, there is no linear or proportional input—output relation.

\*Corresponding Authors.

Through this perspective, systems biology is a holistic approach (from the Greek word  $\ddot{o}\lambda o_{\zeta}$  meaning "all," "entire," or "whole"), according to the concept that the parts of a given system are intimately interconnected, and that they are explicable only by reference to the whole. This view differs from reductionism, i.e., the description and analysis of a (complex) phenomenon in terms of its simplest or fundamental elements. It is to be stressed that reductionism has been and still is fundamental to an understanding of the nature of biological constituents. However, with modern high-throughput (e.g., next-generation sequencing) and computational technologies, there is the chance to move on, and to understand and reconstruct the single parts into the whole.

In modern science systems, biology can be considered an integrated approach that aims for:

- **1.** the comprehension of the *structure* of the system, that can be real (e.g., network of neurons interconnected through synapses) as well as more abstract (e.g., metabolic and cellular signaling networks, genetic regulation networks);
- **2.** the comprehension of the *dynamics* of the system by means of qualitative and quantitative analysis (kinetics) of fluxes and flows among the system's elements, and related mathematical modeling;
- **3.** the comprehension of *system control* and its regulation procedures, i.e., the principles that drive the dynamics; and
- **4.** the comprehension of the *original design* of the system and the principles of self-organization (the "instruction manual" that you need to put all of the parts together).

#### 9.2 Approaches to Systems Biology

Because it is a wide concept, cataloging all approaches related to systems biology comprehensively is not a trivial task. Virtually, any (recursive) workflow involving the extensive use of mathematical and computational methods for hypothesis drawing and predictions based on experimental data and able to be validated can be classified as a systems biology approach. This section is not the place for a complete review of such approaches, so it will briefly introduce some of the main techniques and processes. For a wider essay with a focus on drug development see Ref. [2].

Integrating and analyzing high-throughput data, from multiple experimental sources, gathered using different "omics" technology platforms using informatics (combined computational, statistical, and mathematical) tools and techniques is considered a main systems biology approach. Through this perspective, the states of genes, proteins, and posttranslational modifications, for example, can be assessed in a large-scale systemic fashion (working all together) to understand how they are regulated, expressed, or modified during a specific process (physiologic function, disease, etc.) toward the generation of a testable hypothesis about the components involved (signaling pathways or set of genes) or the discovery of a process or disease markers.

To give an idea of the complexity of biological systems, a group composed of member with different expertise and 15 different laboratories studied together how a nonpathogenic bacterium as *Bacillus subtilis* completely changed its metabolism when growing in different nutrient sources as glucose or malate. More than 30 different conditions were analyzed and different methods were developed to better understand the metabolic networks involved in the bacteria adaptation [3,4]. This kind of approach was extremely important in the sense that it creates several computational tools to lead with data complexity and at the same time opened up the perspective to apply the same approach for understanding pathogenic organisms and how they are involved in disease processes.

Computational and mathematical modeling of single or multiple biological scales (from the gene level to the whole organism), as well as of pathophysiologic processes and biological modules using existing information (e.g., databases, literature) aims to generate predictions for target selection (prioritization of genes or proteins) for clinical guidance or the design of clinical trials. Network biology and agent-based simulation, for example, can be attributed to this branch.

Finally, ex vivo human cell-based systems, i.e., combination of human cells, can be used to screen emergent and complex systemic behavior in response to pharmacologic treatment or other perturbations.

#### 9.3 The Network: A New, Powerful Paradigm in Biology

Networks describe how things connect and interact, graphically representing players (*nodes* and *vertices*) and their relationships (*links* and *edges*) in a mathematically tractable object called a *graph*. Lying among the main approaches that fall under the wider systems biology's umbrella, the network paradigm and graph theory is a specific branch of mathematics that is extensively used in many fields going well beyond the biological field, such as social, engineering and informational sciences [5]. Indeed, network science is a broad interdisciplinary academic field that has practical applications in the study of several types of complex networks, e.g., telecommunications networks, computer networks, biological networks, cognitive/semantic networks, and social networks, a phenomenon also testified to by the flourishing of dedicated institutes and conferences.

In biomedicine, the network abstraction enables complex biological systems to be considered as a whole, as well as to undergo mathematical analysis and quantification, aiming to discover major systemic properties and provide an accurate and analytic view at a glance of actors and their relational linkages and functions. Indeed, the cooperation of biological systems is possible through the organization and coordination of the many different elements and processes of the organism. In this view, the network approach highlights how the characteristics and behavior of single elements influence the network structure (i.e., the *topology*), as well as how the network structure impinges upon processes spreading over the system (i.e., the network dynamics), or how perturbations may affect network performance. In this regard, the network abstraction represents a useful functional view that can complement analyzes and approaches from molecular biology and from many different "omics" [6].

#### 9.4 Network Measures for Biological Systems

Several measures and parameters can be used to analyze network structure, component relevance, key nodes, general efficiency, etc. A few are briefly mentioned here referring to more complete works in the field [7,8].

The *degree* of a node is the number of edges linked to it. In the paradigmatic case of protein—protein interactions (PPIs), the degree is the number of *physical interactions* that a specific protein shows with other proteins. In other types of network, e.g., in social networks, in which the nodes are persons, the degree can be the number of her or his friends and acquaintances.

The *path length* is the number of nodes that need to be crossed when going from one node to another. Many different paths can be chosen to reach one node from another, but the *shortest path length* is particularly significant: when averaged on all node couples, it provides a rough measure of the network navigability.

The average number of neighbors of a node in the network (sometimes referred to as *connectivity*) shows how densely the network is populated with links and may also be represented by means of a normalized version of this parameter, i.e., the network density (a value between 0 and 1).

Of particular interest in biology, some *centrality* measures account for interesting biological characteristics and behaviors. For example, the *betweenness centrality* of a node attests to the grade of control this node exerts over the interactions of other nodes in the network, and its biological relevance has been demonstrated in cancer as well as other biological fields [9].

#### 9.5 Types of Biological Networks

Because of the power of such network representation, diverse biological networks have been figured out and analyzed with a certain degree of success. Networks can be scanned in a single picture and in a graph-theoretic framework for all known events, processes, and associations related to the system. A noncomprehensive list of the most important biological network types is provided next.

*PPI networks* are among the most well-known and used network representations in molecular biology. PPI networks are built by connecting two or more proteins in the presence of intentional physical contacts established between them. PPI networks, also referred to as *interactomes*, are a powerful tool to visualize, analyze, and understand the key nodes, structure, and dynamics of numerous physiologic and pathologic processes, and provide a way to integrate data of different nature.

*Intracellular signaling networks* represent molecular and soluble signals generally going from the cell surface to the cell nucleus (pathways) through a branched cascade of several different events such as PPIs, phosphorylation reactions, and metabolic processes.

Transcription factor regulatory networks are built by analyzing bound DNA elements within the gene proximal regulatory regions. The aim is to map the combinatorial

cross-regulation of several hundreds of sequence-specific transcription factors that eventually define regulatory networks controlling cellular identity, features, and function.

*Protein phosphorylation networks* represent phosphorylation events such as different phosphorylated forms, phosphorylation state—specific PPIs, and kinase—substrate relationships.

*Metabolic networks* represent sets of metabolic and physical processes that determine the physiologic and biochemical properties of a cell, such as interactions between enzymes and metabolites. Metabolic networks, now mostly available online by mean of dedicated databases such as KEGG, among others, are powerful tools for studying and modeling metabolism.

*Genetic regulatory networks* (GRNs) show interactions between usually large numbers of genes, which involve downstream-regulated genes as well as their regulators. As for other types of biological networks, GRNs are used to visualize and analyze the regulatory relationships.

*Co-expression networks* depict genetic maps in which genes are connected if their transcripts are co-regulated.

Finally, *disease networks* map pathologies and disease genes linked by known disorder–gene associations, indicating common origins of many diseases.

#### 9.6 Networks Studies Applied to Understand Key Molecules and Disease Processes

With the explosion of high-throughput technologies, the integration of data coming from the various "omics" (genomics, transcriptomics, glycomics, proteomics, and metabolomics, among others) represents a powerful strategy aimed to reconstruct and analyze complex multidimensional interactions, and subsequently to enable deeper understanding.

Nevertheless, data integration is never a trivial task, and issues coming from the lack of standardization or shared biological identities [10], just to mention a few, need to be faced to overcome major hurdles in integrative workflows. In this perspective, the PPI network framework represents an assessed integrative approach that has been used in computational biology to understand disease pathogenesis, prioritize cancer-associated genes, and predict functional linkages among genes, among many other examples [11–13]. Here are two examples of such integrative procedure used to frame in a systemic landscape a paradigmatic autoimmune disease [rheumatoid arthritis (RA)] and a major inflammatory signaling pathway [nuclear factor- $\kappa$ B (NF- $\kappa$ B)].

#### 9.7 Rheumatoid Arthritis

RA is a paradigmatic autoimmune and autoinflammatory disease, and its complex etiology makes the effective use of current therapies challenging, also in terms of

the difficulties in controlling side effects and adverse events in the numerous districts involved [14].

With the intentions of collecting available molecular knowledge into a single picture and of providing a wider framework to estimate the systemic impact of therapeutic targets on the numerous biochemical compartments implicated in RA, some of us compiled and analyzed a multi-omic data map that assembles genomic, epigenomic, transcriptomic, posttranscriptomic, proteomic, and (for the first time in a systemic manner) host-microbiome interface data [15] (Fig. 9.1).

Thirteen different sources (online databases and literature screening) supplied all data sets that enlisted molecules experimentally associated to RA (*core* data set, 377 proteins from manual curation of literature sources), and additional molecules and pathways strongly yet not explicitly associated to RA (*extended* data set, 4709 proteins). The core set constitutes a more specific RA map, whereas its extension offers a more systemic and practically usable map, notably in terms of the significance of the statistics run on the extended map. Such comprehensive systemic information was integrated at the functional level of PPIs to obtain the multi-omic landscape map of the disease, finally undergoing mathematical analysis and pharmacologic treatment simulation.



FIGURE 9.1 RA is a systemic, inflammatory disease with unclear etiology. A systems biology, multiple "omic" approach has been used to help disentangle its multifactorial causation. A large number of data from "omics" approaches, including genomics, epigenomics, transcriptomics, posttranscriptomics, proteomics, and host-microbiome interfaces to gut intestinal bacteria metagenomics (1) have been integrated using computational tools and manual curation (2) to obtain two main data sets, one specifically related to experimental evidences, and its extension, able to offer a more systemic view (3). From these sets, protein-protein interaction-based RA maps have been derived (4) and undergone network analysis and simulation (5). Hypotheses and predictions about candidate pharmacologic targets and the outcome of existing medication use from such a mathematical and computational approach are provided, ready to be tested, and validated in a laboratory environment (6).

The topological analysis evidenced, among other results, the extraordinary relevance of the mediator growth factor receptor-bound protein 2 (GRB2) and the notable network role of interleukin-1 receptor associated kinase-4 (IRAK4) (Fig. 9.1).

GRB2 is an effective target and mediator in several oncopathologies (mainly leukemia) and a crucial player in the host-microbiome interaction of *Helicobacter pylori* and *Prevotella intermedia*, besides being considered a marker of RA in synoviocytes. Interestingly, a member of the *Prevotella* genus (*Prevotella copri*) was associated to RA [16] as a specific marker of the gut intestinal microbiome dysbiosis that goes along the disease. From the network perspective, this seemingly dispersed knowledge converged in a connected map setting up a robust rationale for accounting GRB2 as a target for RA.

IRAK4 has a critical role in initiating response to foreign pathogens; because of this it was presented to the American College of Rheumatology as a potential treatment for RA, but according to the network perspective, this choice comes with a warning. Indeed, the multi-omic map evidenced that whereas a regression of some aspects of the disease was observed, the control on IRAK4 (by IRAK4 inhibitors) also exerts a significant influence on the response to pathogens, with an impact on immune-specific dendritic cells of patients with RA [17], an event that may severely limit the competent innate host response in case of bacterial infections.

In this work novel information was gathered by condensing knowledge from approximately 4000 selected molecules and about 15 public databases, with the aim of drawing hypotheses in support of basic research and further clinical practice. The work focusing on two major areas of application: (1) support in identifying novel drug targets, and (2) support in identifying potential contraindication to novel therapies.

#### 9.8 The Nuclear Factor-kB Pathway Interactome

It is progressively acknowledged that the number of elements that impinge upon phenotypic outcomes of signal transduction pathways may be higher than that taken into consideration from canonical representations [18], and that pathways may work as larger "mega-networks" rather than as linear cascades [19]. Such considerations pushed some of us to compile a larger reference map of NF- $\kappa$ B, a protein complex that promotes both induction and repression of more than 400 genes. In mammalian cells, there are five NF- $\kappa$ B family members, and different NF- $\kappa$ B complexes can be formed from their homo- and hetero-dimers. The combinatorial control and signaling cross-talk of the NF- $\kappa$ B pathway are intricate, and limited analyses to the canonical NF- $\kappa$ B pathway elements may turn out to be insufficient in uncovering key mechanisms or actors in the regulation of this system.

With this perspective, data from multiple existing sources were integrated and analyzed under a network approach to chart the NF- $\kappa$ B interactome map. This type of comprehensive approach can provide a deeper understanding about how specific stimuli trigger particular subsets of NF- $\kappa$ B target genes or about the regulatory hierarchy that rules the selection of target gene expression.

To compile the NF- $\kappa$ B map, multiple data sources, i.e., manually collected data from literature, binary PPI data, protein annotation data, and NF- $\kappa$ B downstream gene data, were mined and integrated into a single picture. Three different sets of proteins were compiled, showing evidence of involvement in the upstream regulation process of NF- $\kappa$ B, and one set of downstream genes and related proteins, whose expression appears to be regulated by NF- $\kappa$ B. Starting from these sets, three NF- $\kappa$ B interactomes were derived by using PPI data from multiple databases and then checked for experimental evidence of physical interactions. Interactomes then underwent mathematical analysis (centrality measures computation) and results were interpreted through a biological perspective. Confirmation of known key elements in feedback loops suggested biological insights about other high-ranking (high betweenness centrality) proteins that were not reported in the literature as crucial regulatory components.

This systems-oriented map may have helped improve the understanding of one of the most complex signaling networks and highlighted the impact of single elements, their feedback, and cross-talk regulations on cellular processes. Besides confirming previous studies and insights, this interactome mapping emphasized that the number of elements impinging upon the outcome of NF- $\kappa$ B pathways is much higher than that usually taken under consideration, pushing for a rethinking of canonical pathway representations.

#### 9.9 Introduction to Mathematical Modeling

As discussed previously, systems biology is an active area of research that exploits the availability of large biological experimental data sets and the power of modern computers. Mathematical modeling of biological phenomena constitutes a way to analyze a system by breaking it down into parts and studying their dynamics. It is a reductionist approach. The systems biology viewpoint is to reassemble the constituents and evaluate the mutual influence of the various components through a holistic view. This task is accomplished by means of both classical mathematical methodologies such as differential equations and computational tools developed after the advent of automatic calculus.

Mathematics is a language. It allows the expression of the relationships among entities entitled in a phenomenon. In addition, it does so in a logical and precise way. The description might be wrong but the way to progress from the assumptions to the conclusions is exact.

The most difficult part in writing a mathematical model is not to derive the conclusions but to describe the system and formulate the problem. So, if it cannot provide sure answers to the desired questions, how can it be useful? It depends on how well the problem is formulated. In other words, what are the assumptions that can be done to make the system conceptually manageable? There is no other way to describe real life than to simplify it; therefore a mathematical description has to be based on choices taken to include what is relevant to the purpose of understanding a certain aspect of the system and to leave out what is not. The usefulness of describing nature by mathematical equations lies in the process of studying the phenomenon: that is, in the fact of reasoning about something by using the rigid logic of mathematics. The formulation of the problem itself becomes a way to see the system through a different perspective, which forces researchers to note otherwise elusive things. Once formulated, the mathematical (and now computational) model is a formidable instrument to confirm or reject hypotheses about how things really work, and/or to find points of intervention to control the system under investigation.

Mathematics has been "the" language to describe physical phenomena from the very beginning of the scientific revolution of the 17th century, but only relatively a short time ago its attention was turned to biological phenomena. Throughout the centuries biology has shown a higher resilience to mathematical investigation compared with physical sciences because of the enormous complexity of living matter compared with nonliving things: a complexity stemming from the redundancy, multifactorial, and dynamism of the mechanisms governing living things [1].

This complexity originates mainly from the distributed/systemic characteristic of the phenomena and from the enormous variety of components involved, ranging from molecules and cells to organs and organisms.

One of the first biomathematical models, which dates back to the beginning of the 20th century, is the Lotka–Volterra system of nonlinear differential equations, which describes the dynamics of two species, in which one is the predator and the other the prey [20]. This model has inspired many other variants and generalizations that have proven useful in predicting the temporal evolution of two or more interacting species in generic ecosystems under disparate constraints (e.g., availability of resources or land-scape restrictions). The model formulation describing, for example, the relationship between foxes (F) that predate rabbits (R) looks like:

dR/dt = a R - b F RdF/dt = c F R - d F

The sense of these sentences in mathematical language is that there is an interdependence between the time variation of the prey and that of the foxes. The fate of the two species is joined because there cannot be foxes without rabbits and the prosperity of the rabbit species depends on the abundance of foxes. The model assumes unconstrained access to food resources for the rabbits. Those, two lines can be "spelled out" as follows. Rabbits grow at a rate and are killed by the foxes at a rate that is proportional to the fox population (bF). Foxes grow proportionally to the food availability (cR) and die by aging at a constant rate (d). Rabbits also die, but for them it is a growth rate greater than the death rate that is assumed.

Models of this type are constituents of much more complex descriptions of biochemical reactions such as the Belousov–Zhabotinsky (BZ) oscillating chemical reaction, often referred to as a paradigmatic example of nonequilibrium dynamics characterized by the propagation of waves in a reactive medium. There are various forms

of BZ reactions, each using a different chemical "recipe." One of the simplest is based on a mix of hypobromous acid, bromide, cerium-4, bromite, and a general organic species. In its simple form it may be comprehended in terms of the following three concurrent and competing processes [21]: (1) reduction of bromate to bromine; (2) introduction of hypobromous acid as a reducing agent for bromate; and (3) reduction of the catalyst formed from the previous processes. The chemical formulation would be:

$$A + Y \rightarrow X + P,$$
  

$$X + Y \rightarrow 2P,$$
  

$$A + X \rightarrow 2X + 2Z,$$
  

$$2X \rightarrow A + P,$$
  

$$B + Z \rightarrow hY + Q,$$

where the variables represent the concentrations of the specific reactants and h and Q are constants. In the classical mathematical formalism of ordinary differential equations, the system would look like:

$$dX/dt = AY - XY + AX - 2X2,$$
  
$$dZ/dt = AY - XY + hBZ,$$
  
$$dZ/dt = 2AX - BZ.$$

Oscillating patterns characterize the temporal evolution of this mathematical system, whereas traveling (in space) waves would emerge in the spatially extended version of the differential system, therefore written by means of partial differential equations.

These examples are meant to help trace a line between reality and mathematics whereby the obtained models become toy systems in the hands of the researcher who can then play to understand and discover. Those systems are much simpler than reality for the purpose of keeping the complexity manageable. Systems biology, however, points to a unified vision in which components are brought together to connect, interact, and compete. The latest anticipation in the field of mathematical modeling is to build on existing models by creating tools, standards, and theoretical frameworks to unwind the task of linking models together. This job often entails bridging different physical scales (i.e., time and length), leading to the complex domain of multiscale biomodeling [22].

The advent of digital computers has given the means of stretching the reach of classical mathematical models that otherwise would have been strongly limited by the availability of "analytical" solutions of the model systems. In simple language, once formulated in differential equation form expressing the dynamical evolution of quantities such as concentrations or cells counts, a system of differential equation needs to be

mathematically disentangled to obtain the "solution": that is, a formula explicitly stating the relationship between the quantity and the variable "time." This is not an easy task, and most mathematical models do not yet have an explicit solution. Computers detour this venture by digitally iterating rules that match the passage of infinitesimally small intervals of time, eventually uncovering the relationship between the variable and time. Actual desktop computers can handle thousands of differential equations in a matter of seconds. That is a powerful tool in the hands of a systems biologist.

#### 9.10 From Classical Modeling to Microscopic Simulations

Modeling by differential equations is not the only method. There is a "new kind of science" that is gaining consensus in the theoretical biology domain [23]. This "new science" has different names. It is discrete mathematics in the computer and adheres to a simple idea: to represent reality by enacting its simple constituents, which are particles, cells, molecules, and agents, whatever individual entity composes a system. A system of cooperating cells is therefore not described by means of lumped variables such as those expressing the foxes and the rabbits in the differential equation example given earlier, but by means of a number of unique representations of single foxes and rabbits. In the digital world of a computer, this designates an individual portion of memory in which to annotate the different characteristics of the agents; this is instrumental to represent the processes carefully, resulting in the overall dynamics at a population level. In the example, not all foxes are equal; some are bigger and stronger, eat more, and run faster, and thus predate more. Likewise rabbits are not all nothing-more-than runners; some may be more inclined to hide or are less risk-takers than others. This modeling paradigm that here we call microscopic simulation or microsimulation, gives a rationale to those details that makes life so diverse and complex. It is a computational paradigm rather than a mathematical one in the classical sense, and it exists because the rules that describe how the entities behave and interact among them and with the environment are effortlessly executed in the machines.

Stanislaw Ulam and John von Neumann originally conceived the microsimulation concept in the 1940s while they were contemporaries at Los Alamos National Laboratory and were interested in laying the theoretical foundation of self-replicating machines. Although some researchers studied microsimulation throughout the 1950s and 1960s, it was not until the 1970s and John H. Conway's Game of Life that interest in the subject expanded beyond academia. The Game of Life is the most famous of a larger class of models. It takes the name of cellular automata (CA) and consists of a checkerboard on which cells live and die. The dynamic evolution of the combined fate of the living "cells" is governed by simple logical (and local!) rules. Because the entities are stored in binary digits and the rules, no matter how complex, are coded as Boolean functions (i.e., transformations of bits), it is not surprising to realize that CAs are worthy children of the computer age.

#### 9.11 Microscopic Simulation of the Immune System

Microscopic simulations provide researchers with a tool to study complex biological occurrences. They represent the agents and dynamical rules and then set the systems free to evolve over time through agent interactions, with a minimum of a priori assumptions about the dynamic itself. Macroscopic system behavior is then observed as an emergent property. The richness of details one can take into account in microscopic CA-like simulations makes this methodology appealing for the description of biological systems whose behavior and heterogeneity of interacting components are not safely reducible to some stylized or simple mechanism.

Paradigmatic of such complex phenomena is the immune system, which has been studied for a long time by means of detailed simulations [24]. This model originates from a couple of ideas. The first is to use discrete entities to represent the population of lymphocytes, other immune cells, and antigens such as bacteria or viruses. The second is to use binary strings to represent molecular binding sites so that a digital match (computer scientists call it the Hamming distance) can identify the degree of molecular binding affinity. This microsimulation model includes both innate and adaptive immunity. The adaptive branch of the immune system is further specified in both cellular and humoral immunity. The whole adaptive immunity model is bolted on the clonal selection theory of the Nobel laureate F. M. Burnet [25], developed on the tracks first highlighted by P. Ehrlich at the end of the 19th century. To have a fully equipped clonal selection process as a response to an antigenic insult, the model has to be equipped with the concept of specificity. Here is where the mentioned binary string and affinity comes into play: More lymphocyte clones of different specificity are represented (it is a bit-string polyclonal model, as opposed to monoclonal models in which only a single population of genetically identical lymphocytes is represented). Major classes of cells of the lymphoid lineage (lymphocytes T-helper and cytotoxic, lymphocytes B- and deriving antibody-producing plasma cells, and natural killer cells) and some of the myeloid lineage (macrophages and dendritic cells) are represented in the model. Other working immunology-grounded assumptions are hematopoiesis; lymphocytes maturation, and thymic selection of T-cells; Hayflick limit in the number of duplications; aging and memory of past infections; hypermutation of antibodies; bystander effect; cell activation and anergy; cell interaction and cooperation; and antigen digestion and presentation. Each of these subjects identifies a component or a process of the immune system and is implemented by algorithmic rules whose complexity sets it apart from what could be reasonably be manageable with differential equations.

The microsimulation modeling paradigm allows a step further in the representation of the immunologic specificity. For example, it was implemented a novel approach in which the lymphocyte receptors as well as antigenic binding regions are not binary strings but the molecular primary structure [26]. This modification is a great step toward realism but it carries a number of complications and requires a number of strong assumptions that, it is hoped, in a not so distant future will be removed by adopting new more reliable techniques.

The whole method relies on techniques developed in immunoinformatics, a new experimental and theoretical discipline emerging from growing knowledge gathered for decades in experimental immunology and immunogenomics [27]. In immunology, what is of utmost importance is to "predict" which part of the antigen will constitute an immunogenic epitope; broadly speaking, there are two ways of doing it. The first is to simulate the chemical–physical interactions between peptides and major histocompatibility complex (MHC) molecules. The second is to resort to machine learning and statistical methods to extract and generalize information from available experimental data of, for instance, MHC–peptide sequences [28].

This simulation framework test hypotheses or elucidate causal relationships among the various subsystem components actually fulfilling the core requirements of the systems biology approach. For example, a basic immunological question is the contribution of the humoral and the cellular branch of immunity in viral infections by selectively tune processes pertaining to viral evolution [29]. Or to investigate the role of MHC haplotype heterozygosity and homozygosity with respect to the influenza virus to show that there is an advantage to heterozygosity (Fig. 9.2) [26]. Furthermore, one can compare vaccine formulations to elucidate the protective role of the immune memory



**FIGURE 9.2** Mathematical models transform consolidated knowledge coming, for instance from experiments, into new knowledge by enacting hypothetical conditions through numerical calculations. The advantage to human leukocyte antigen (HLA) haplotype heterozygosity was demonstrated (A\*0201, A\*0301, B\*5304, B\*5309, DRB3\*0302, and DRB5\*0202) with respect to homozygosity (A\*0201, B\*5304, and DRB3\*0302) when simulating infection by flu influenza A serotype H1N1 (genome ID: HU13275) [26]. DB, database and Ag, antigen.

elicited by priming with either influenza vaccines or influenza infection. In this specific example, reported in Castiglione [30], a number of computer experiments were performed to test the level of influenza-specific CD4<sup>+</sup> T-cell cross-reactivity resulting from various vaccination protocols consisting of priming with the split (i.e., an approximation of the vaccine consisting of the 11 proteins of the virus at equal concentrations) A-Brisbane-59–2007 H1N1 influenza A virus vaccine followed by various boosts such as the hemagglutinin molecule of virus H1N1 influenza A California-04-2009 alone or a set of molecules from the swine-origin influenza virus vaccine consisting of molecules of the A-California-04-2009 virus. Other protocols were included in this comparison, including a simulated version of a "live" attenuated virus (i.e., encompassing the major steps of the lifecycle of the virus) consisting of the total 11 molecules of the A-Brisbane-59–2007 H1N1 influenza A virus. The resulting dynamics has proven to be in qualitative and to a certain extent quantitative agreement with previous ex vivo studies. In particular, the level of immunoglobulin G simulated by the computer model was in agreement with previous experimental results [30].

Complete control over the setup of virtual experiments and deep analysis of the results of such a detailed model allowed the effect of different rules to be calculated for each coding for a biological hypothesis, idea, or immunologic mechanism. To mention a few, it is possible to simulate the consequences of varying the vaccine dose or vaccination schedule, its formulation (soluble antigen, complex and conjugate, virus-like particle, the effect of using adjuvant, etc.), and the difference between injecting a compound or the single components separately, among others.

#### 9.12 Basic Aspects of Synthetic Biology

The "omics" technology and system biology approaches have been used extensively to describe complex biological networks through a deep characterization of molecular components or biological parts that compose the organisms (i.e., genes, different RNA families, proteins, and metabolites) and their interactions, as discussed earlier using networks. The massive knowledge acquired may be used to design or redesign new genetic circuits and biological parts with several applications to biotechnology. These purpose-driven approaches leading to fabrication of biological components and systems that do not already exist in the natural world are named synthetic biology. Considering biological organisms to be programmable manufacturing systems, synthetic biologists use engineering concepts to supplied genetic instructions allowing execution of the desired function. These concepts are based on modularized propriety of biological parts that can be combined in different ways and will then work as intended.

Work involving synthetic biology is highly dependent on a set of well-characterized biological parts. These biological parts are DNA sequences that encode proteins as enzymes, transporters, or regulatory elements as promoters, noncoding RNAs, and different binding sites for several elements. Because of the acid nucleic nature of biological parts, the design and build of engineering biologically based devices and systems is closely linked to DNA editing involving classical molecular biology experiments as gene synthesis, polymerase chain reaction (PCR), digestion with restriction enzyme, and cloning and assembly of biological parts that will produce the desired function. Because a range of vectors and protocols can be used to isolate and stock these biological parts, combination of components is usually laborious, expensive, and slow work. A stimulating challenge is standardization of biological parts libraries allowing their use by many research groups to build different synthetic devices. In 2002, Knight [31] developed the concept of BioBrick standardization of biological genetic parts. Each BioBrick contacting the functional sequence between a prefix composed of restriction site to enzymes *Eco*RI, *Not*I, and *Xba*I and a suffix containing site to enzymes *SpeI*, *Not*I, and *Pst*I. Two genetic parts may be connected through a ligation link among compatible overhangs from the first BioBrick, previously digested by *SpeI* and a second BioBrick cut by *Xba*I creating an uncuttable junction (Fig. 9.3A). This protocol allows easy assembly with a controlled number and order of BioBricks.

BioBricks are suitable for several applications, but the system often fails in the construction of protein fusions owing to the production of a sequence that encodes an in-frame stop codon. Other approaches have been proposed to solve this issue [32,33]. BglBricks (Fig. 9.3B) is one option based on the presence of the *B*glII enzyme site as a prefix and *Bam*HI as a suffix that when linked produces a scar sequence encoding a dipeptide glycine and serine [32]. Therefore BblBricks are suitable to build synthetic devices with protein fusion. An alternative methodology regardless of the restriction endonucleases frequently used to join biological parts is the Gibson assembly approach (Fig. 9.3C). DNA sequences that will be connected using this approach should have an overlapping region that after digestion with a 5'-exonuclease will result in single-stranded regions. These single-stranded cohesive ends overlapping regions of different fragments can anneal. The DNA polymerase extends paired overlapping regions and the DNA ligase seals nicks in the assembled DNA. Gibson assembly was used to produce the first genome composed exclusively of chemically synthesized DNA [34].

After suitable biological parts are selected to produce the desired result, they generally are inserted inside a *chassis*, where regulatory sequences will exert their effects and proteincoding sequences will be transcribed and translated. A chassis is a cell that may harness its own biochemistry to produce the intended work, and thus organism models with wellestablished DNA manipulation protocols are often used [35]. Moreover, availability of a complete genome sequence is desirable to accelerate research using the selected organism. The most common chasses used in laboratories are bacteria as *Escherichia coli* and *Bacillus subtilis* [35], but these chasses may not be feasible for the production of compounds, proteins, or synthetic circuits depending on specific eukaryotic aspects. Among eukaryotic chassis, yeast, is often chosen owing to rapid growth and easy genetic manipulation [36,37]. Mammalian cells represented by lineages such as Chinese hamster ovary and HeLa derived from human cervical cancer cells are also used mainly for potential applications that require



FIGURE 9.3 Approaches to biological parts assembly. (A) BioBrick system. Each biological part is between Xbal (blue letters—dark gray in print version) and Spel (red letters—gray in print version) endonuclease sites. Parts can be fixedly connected by DNA ligase when the first part is digested by Spel enzyme and the second part is cleaved by Xbal enzyme, forming cohesive ends that after ligation form a scar sequence (underscore region) not accessible to the same enzymes, allowing large construction containing several parts. Analysis of three possible reading frames (first, second, and third labels) of the produced scar sequence shows a stop codon in the third frame. (B) BglBrick system. The Xbal and Spel sites were replaced by BglII (green letters—gray in print version) and BamHI (pink letters—light gray in print version), respectively, that produce a scar sequence without a stop codon to all possible reading frames. (C) Gibson assembly. Synthetic DNA containing overlapping regions are digested by 5' exonuclease, allowing annealing between cohesive single-strand ends from different fragments. After PCR extension using DNA polymerase, the fragments are joined using DNA ligase.

mammalian characteristics [38] as well as implants in humans or animal models [39]. Besides living chasses, cell-free devices based on cell extracts or in vitro transcription and translation are emerging as amenable and safety systems and are ideal for producing toxic or inhibitory products for living chasses [35,40–42].

Methods for genome editing of chasses have experienced significant breakthroughs during through the discovery and technical improvement of transcriptional activator-like effector (TALE) nuclease, zinc finger nuclease (ZFN), and clustered regularly interspaced short palindromic repeat (CRISPR)/CRISPR-associated protein-9 nuclease (Cas9) systems. TALE are DNA-binding proteins first identified in *Xanthomonas* plant pathogens that mimic eukaryotic transcription factors regulating plant genes during pathogenesis [43,44]. Chimeric proteins obtained by fusion of a TALE repeat array with a nonspecific FokI nuclease have the ability to cleave genomic DNA from living cells (Fig. 9.4A). Other chimeric fusion of protein containing zinc finger domain previously known by DNA binding ability, with FokI nuclease originated in ZFN systems was used for genome editing [45,46] and allowed specific double breaking of genomic DNA (Fig. 9.4B). CRISPR together with Cas9 was initially identified to participate in the adaptive immune system of several bacteria and archaea against bacteriophages and other foreign nucleic acids [47] and has been optimized to work in different cells such as bacteria, yeast, protozoa, plant and mammalian lineages [48-50]. The CRISPR/Cas9 system uses guide RNA that binds to target genome sequences, initiating a double-strand break after a protospacerassociated motif represented by the NGG trinucleotide sequence (Fig. 9.4C). All of



**FIGURE 9.4** Genome editing tools. Double-strand break of target region in genome sequence can be performed: (A) CRISPR/Cas9 system has a guide RNA (gRNA) that directs CRISPR-associated nuclease 9 (Cas9) to the target site and the cleavage occurs before a protospace adjacent motif (PAM) composed of NGG trinucleotides; (B) TALE system that directs a nonspecific *Fok*I enzyme to target site of each strand; (C) ZFN system uses zinc finger DNA-binding function that also directs *Fok*I to the target site of each genomic strand. Double-strand breaks can be repaired by nonhomologous end joining or homologous recombination that inserts DNA fragments from the donor sequence with overhanging ends with the cleaved regions.

these systems allow the degradation of specific DNA fragments or can be replaced by recombination to insert single and large nucleotide alterations [45,46,50–52].

Intermediate work in synthetic biology based on the design of new genetic circuits to control some system propriety involves the concept of logic gates well-established in electrical engineering [53–55]. Logic gates are implemented as a Boolean function that can assume only an ON or OFF stage in a determined moment, and the output of a system depends on the combination of its inputs (Fig. 9.5A and B). For instance, AND logic gates of a hypothetical system that receives two inputs will produce the output only when both inputs are present (Fig. 9.5B). On the other hand, OR or NOR logic gates will produce the output when at least one input or no one is present, respectively. Moreover, simple logic gates can be combined, resulting in highly complex control of the system [56]. From the biological point of view, logic gates may be built using transcriptional or posttranscriptional control stimulated or inhibited by different elements such as RNA, proteins, metabolites, light, pH, and heat that control a determined output such as a



FIGURE 9.5 Genetic circuits based on AND, OR, or NOR logic gates with two inputs. (A) Schematic representation. (B) Expected logic outputs based on the presence (1) or absence (0) of each input. (C) Synthetic biological circuit example. In synthetic biological AND circuit [63], green fluorescence protein (GFP) production (output) is controlled by a promoter that produces the protein after binding of a heterodimeric complex (RS). The R complex component is stimulated by the presence of arabinose (input A) and the S subunit is produced under isopropyl β-D-1-thiogalactopyranoside (IPTG) control (input B), resulting in output production only when both inputs are present. To the OR logic gate [64], YFP production (output) is under the control of two independent promoters stimulated by arabinose (input A) and anhydrotetracycline (input B), respectively. Only one input or both are sufficient to trigger YFP production. The NOR logic gate [64] is represented by a repressor component independently stimulated by arabinose (input A) or anhydrotetracycline (input A). The repressor inhibits production of YFP (output) and this protein is only produced in the absence of both inputs.

protein, metabolic reaction, or photon emission (Fig. 9.5C) [57]. A toggle switch composed of two repressors and two constitutive promoters arranged in a mutually inhibitory network and developed in *E. coli* showed interesting properties for biotechnological application such as bistability [58]. In addition to switches, synthetic biological oscillators have been developed [59] and several mechanisms for transcriptional or translate control of synthetic circuits such as riboswitches [60,61] have enabled significant advances in synthetic biology. Eventual problems in synthetic genetic circuits are associated with the production of transcripts even in the absence of an activating signal, and they often needs computational and experimental optimization [62].

Especially in human and animal health, synthetic biology is a useful tool for improving diagnosis, treatment, and preventive health care. The next sections provide examples of applications of synthetic biology approaches for the development of improved diagnostic devices and optimized drug delivery systems and biological chasses for drugs and other compound production (Fig. 9.6). Furthermore, the application of synthetic biology to prevent disease will be discussed based on vaccine system design and genetic correction by genome editing. Although the potential of synthetic biology application is widely discussed and recognized, the field is just emerging and the vast majority of published works have only presented proof of concept results.

#### 9.13 Synthetic Biology and Disease Diagnosis

An ideal biomarker for disease diagnosis should have high sensitivity and specificity to identify patients affected by the disease compared with healthy individuals or patients with other diseases. An advantage of using synthetic biological devices to diagnosis disease is the ability of these systems to identify different targets and integrate signals produced by each target in a more specific response. In addition, the synthetic device may be optimized to amplify the signal response, making it more sensitive even at low target levels. Disease diagnosis based on multiple targets uses logic gates concepts to detect different elements (compounds) in a sample; only when all target elements are present, will the output be detectable [65-67]. For instance, a switch that allows production of two stable states implemented in E. coli was used to evaluate environmental conditions of murine gut and correctly discriminated mice previously treated with anhydrotetracycline and untreated animals [68]. However, biosensors containing synthetic genetic circuits that work as logic gates can present a number of problems such as low sensibility and signal-to-noise ratio, and thus they should be optimized through analysis of thresholding, digitalization, and amplification of biologically relevant signals [69].

The versatility of synthetic biological devices to indicate the presence of liver metastasis commonly associated with more advanced stages of certain types of cancer including colorectal, breast, and pancreatic [70,71] was demonstrated using an engineered *E. coli* strain [72]. Gene encoding lacZ enzyme was inserted in *E. coli* Nissle 1917 strain and administrated orally to mice with liver metastatic tumors. Owing to the



FIGURE 9.6 Current health-based biological synthetic devices. (A) Disease diagnostic tools. (B) Synthetic systems applied to disease, treatment, and prevention.

natural and selective propensity of this bacteria strain to colonize tumor tissues, it proliferates within 24 h after reaching liver metastatic tumors, resulting in a high level of lacZ secretion that can be detected in urine samples by a colorimetric test. Furthermore, another synthetic genetic circuit was built to identify microRNAs of specific cancer cells and resulted in apoptosis as a reporter response by an RNA interference mechanism [73]. The synthetic circuit was transiently expressed within a HeLa cervical cancer cell line and normal cells leading to apoptosis only in cancer cells.

In addition to synthetic genetic circuits, synthetic biology may be applied to produce chimeric proteins to increase the efficiency of traditional diagnostic approaches. This approach is applied but not limited to immunodiagnostic methods based on the detection of specific antibodies. Protein engineering applied to the production of synthetic chimeras composed of different epitopes sequences may contribute to increase the specificity and sensibility of immunodiagnostic assays [74,75]. This approach was validated for Lyme disease caused by *Borrelia burgdorferi* using a chimeric protein that reached 98% sensitivity and 100% specificity for disease diagnosis [76].

Synthetic biology also provided significant advances for the detection of human bacterial pathogens using engineered bacteriophages [77,78]. Bacteriophages are viruses with the ability specifically to infect and replicate inside target bacteria by injecting their acid nucleic content that incorporates into the bacterial genome or remains a stable episome replicating with their host. Determined environmental and biological conditions lead to the production of viral proteins by bacterial transcription and translation machinery, resulting in a large quantity of new viral complete particles that are released after disruption of the host cell in a process called the lytic cycle [79,80]. Heterologous genes associated with the different report system were added to some bacteriophages that specifically infect human clinical bacterial pathogens such as Mycobacterium tuberculosis [81–83], Yersinia pestis [78,84], and Bacillus anthracis [85], allowing identification of these pathogens in samples such as blood, serum, processed sputum, nasal swab, and laboratory cultures. A report system used to detect the pathogens was based on the addition of a the gene encoding luciferase enzyme that uses endogenous and exogenous substrate and cofactors to produce light that can be detected by different devices [86-89]. Moreover, systems based on the detection of specific fluorescence produced by the exogenous gene encoding proteins such as green fluorescent protein and yellow fluorescent protein were also developed [90]. Strong bacterial promoters were inserted downstream of both reporter genes in the altered phage genomes, contributing to the expression of the high level of reporter genes in the early stage of infection. The current reference standard of diagnosis of these clinical pathogens requires slow cultivation that needs a specific laboratory structure. The synthetic device based on bacteriophages may be used and provides fast and accurate results owing to the short time of the bacteriophage lytic cycle and the specificity of different phages for each pathogen, respectively. Another important advantage is that the engineered bacteriophage may also be used quickly to detect food and waterborne pathogens in samples of water and food that will be consumed by the population [91].

Because the clinical use of cell-based synthetic devices has been hampered owing to feasibility and biosafety issues, another promising approach based on synthetic biology to diagnosis disease has been developed in cell-free systems [41,92]. Most early work aimed to develop cell-free systems used fresh or frozen cell extracts containing all necessary biological machinery and liposome technology to assemble artificial cells [40,93–96]. Pardee and collaborators developed a versatile and stable cell-free synthetic gene network system embedded onto paper that could easily be distributed and stored as well as activated by adding water [42]. This technology was used as a glucose and mRNA strain-specific Ebola virus sensor.

#### 9.14 Synthetic Biology and Disease Therapy

Several approaches have been described using synthetic biological devices to treat disease including metabolic engineering of different organisms to produce therapeutic compounds at a high yield [97-100], design smart drug delivery systems [39,101,102] and edit the genome of host cells, targeting pathogens [103,104]. Metabolic engineering has allowed the production of natural drugs that are not viable by chemical synthesis owing to their structural complexity and that are present in low levels at their original source. One of the most successful cases of metabolic engineering has been applied for the production of artemisinin, an antimalarial drug previously isolated from medicinal plant Artemisia annua [105]. The artemisinin biosynthetic pathway was extensively studied [106,107] and the heterologous expression of key enzymes for drug biosynthesis from mevalonate and deoxyxylulose 5-phosphate pathways was performed in E. coli [98] and veast [99] chasses, resulting in high-titer production of artemisinin. Moreover, the original plant source was optimized by genetic engineering to overexpress its own genes evolving in artemisinin biosynthesis as well as heterologous genes that contribute to the production of high levels of the drug [108,109]. The genes were modified or inserted using an Agrobacterium transformation system that uses the ability of bacteria from the genus Agrobacterium to infect plant cells and integrate regions from large tumorinducing and rhizogenic plasmid into the plant nuclear genome [110]. Another example of metabolic engineering of *E. coli* in drug production is the isoprenoid pathway optimization for the overproduction of taxol [100], a potent anticancer drug first isolated from the conifer plant Taxus brevifolia. Production of drugs using metabolic engineered microorganisms is described in Ref. [111].

Exciting advances in synthetic biology have allowed the designer of complex circuits to control the production of drugs and therapeutic products in the host cells when required owing to pathologic events. These systems are usually responsible for small molecules or others stimulator signals that trigger the production of therapeutic components. The microencapsulated immortal human HeLa cell line containing a synthetic genetic circuit that secrets urate oxidase was built to recognize increases in uric acid levels in the bloodstream [39]. This system was validated through peritoneal implantation in mouse models exhibiting pathologic urate levels and resulted in a decrease in uric acid crystal deposits that may compromise renal function with valuable potential for application in patients with tumor lysis syndrome and gout. Promising results have also been obtained with a synthetic multifunctional mammalian pH sensor that was able to identify increases in diabetic ketoacidosis levels in mice and produced insulin until homeostasis recuperation [101]. Unlike responses to endogenous metabolites, glycemic control caused by type II diabetes in a mouse model was also reached using a synthetic genetic device that produced glucagon-like peptide 1 when stimulated with an external blue light [102]. Long-time durability, the possibility of pathology control in the early stage of disease, and better fine-tuning of therapy to disease progression and reduction are advantages of a self-sufficient control system compared with classical therapy.

Genome editing technologies have also been used as potential therapeutics through modifying human cells to avoid the progression of disease. The most successful example is the alteration of the genome of target cells from patients infected with HIV, to produce resistant lineages that are refractory to virus entry [103,104,112]. In these works, key regions from different host receptors necessary for virus invasion, such as CCR5, CXCR4 from T-cells, or hematopoietic stem cells, were mutated by site-directed technology such as ZNF, TALE, or CRISPR/Cas9, and virus-resistant cells were transplanted back into the patients or humanized mouse models [112].

#### 9.15 Synthetic Biology and Disease Prevention

Synthetic biology has contributed to disease prevention mainly through designer or regulatory devices for DNA or RNA vaccines [113–115] and genetic correction by genome-editing methodologies [116,117]. The first approach is based on modulation of the immune system to early recognition and elimination of infecting pathogens or malignant cells in the early stages of disease establishment. The use of DNA as a vaccine was established in the early 1990s through the direct introduction of a plasmid or viral vectors containing the DNA sequence encoding antigens with the ability to trigger a protective immune response [118,119]. On the other hand, direct vaccination with mRNA has also been used to provide an antigen source for adaptive immunity, simultaneously stimulating innate immunity mediated by interaction with Toll-like receptor 7 [120]. DNA or RNA vaccines have several advantages compared with traditional approaches, including an increase in stimulation of both B- and T-cell response, vaccine stability, and low-cost, large-scale manufacture [121]. The rational modification of a synthetic Gag protein in DNA vaccine against HIV type 1 by an increase in CG content and optimization of codons that increase more than 10 times the antigen expression in target mammalian cells enhanced the immunogenicity and efficacy of the vaccine [122]. Synthetic biology has also contributed to an increase in the yield of antigens produced by heterologous expression through optimization of selected chasses as well as mRNA structures and gene sequences based on codon use of producing organisms [113]. Furthermore, protein engineering has contributed to the production of chimeric proteins generating multivalent vaccines that are safer and more efficient [114,123,124].

Synthetic biology has been applied to disease prevention using genome-editing tools to correct genetic alterations that could develop metabolic and genetic syndromes. For instance, the CRISP/Cas9 system has applied as proof of concept for functional genome repair of stem [116] and embryonic [48,117] mammalian cells, including human cells. Intestinal stem cells with mutated transmembrane conductor receptor recovered from patients with cystic fibrosis were rescued through homologous recombination directed by the CRISP/Cas9 system, resulting in fully functional alleles [116]. Mouse zygotes with the mutated Crygc gene associated with dominant cataract disorder were also submitted to genome correction, resulting in progeny with a repaired Crygc allele, and were disease-free [117]. The estimated efficiency in rescuing

the mutated  $\beta$ -globin gene that caused  $\beta$ -thalassemia in tripronuclear human zygotes was approximately 25%, and the author observed off-target cleavage along unexpected sites including exonic regions of C1QC and transthyretin genes [48]. Therefore potential problems observed in all works as the absence of complete understanding of the molecular mechanism, evaluation of stability, and off-target effects should be addressed before clinical application approval.

Fig. 9.6 shows the tools, molecules, devices, and systems produced by synthetic biology that are important in diagnosis and therapeutics.

#### 9.16 Biosafety and Ethical Aspects Surrounding Synthetic Biology

Although the initial results from applying synthetic biology to the development of new health solutions are extremely exciting, studies are in the early stages and many issues of biosafety and ethics should be addressed before clinical use. Regarding safety, it is difficult to assess the risks of extensively genetically modified organisms. To deal with these problems, a group from Harvard Medical School developed a bacteria strain that depends on a specific amino acid, which does not occur in nature, to grow up. In this way, this bacterium only grows up in a laboratory under specific conditions [125].

The efficacy of such synthetic systems needs to be thoroughly tested to ensure the beneficial effects and track possible unexpected problems including side effects. The stability of construction is another important aspect associated with new technologies that should be validated to guarantee the quality of a synthetic biological product. Furthermore, necessary equipment to stock and maintain the viability of synthetic devices should be at hand. Efficient discard protocols of live synthetic systems should be available to avoid environmental contamination and reduce the risks of unexpected effects during interaction with other biotic elements. All of these issues depend on an elaborate training program for synthetic biologists and others involved, and rigorous standardization of protocols to assess quality.

These issues may become more significant with the progress of synthetic biology [126].

#### 9.17 Conclusion and Future Perspectives

Systems biology and synthetic biology are emerging disciplines that have a high impact on several areas of bioscience, especially for the understanding, diagnosis, and treatment of diseases. A deep understanding of biological systems is far from being achieved and for a better understanding of a system as a whole, several challenges should be overcome: (1) the quality and standardization of "omics" data; (2) multisource data integration; (3) the development of methods to evaluate the dynamics of a system; (4) adaptation of computational tools to become more friendly; and (5) a decrease in the time to process data [127]. Once these challenges are overcome, systems biology approaches will hold the promise of substantially improving the current state of the art in medicine by providing the ability to distinguish among multiple disease states and to enable identification of the molecular causes underlying diseases [127]. This is particular important for multifactorial diseases such as cancer [128–130], obesity [131], RA [132], and cardiac disease [127], among several others.

In addition, with the advance of technology regarding DNA sequencing and synthesis and DNA manipulation such as Gibson assembly and TALEN, ZFN, and CRISPR/Cas9 tools, the understanding and treatment of diseases will completely change. DNA sequencing and synthesis is an easy, fast, and cheap method that allows rapid design, fabrication, and testing of synthetic systems. The Gibson assembly method allows a synthetic cell to be built [34] and has an impact in the construction of synthetic influenza vaccine viruses for rapid response to pandemics [133]. There is now a great potential for creating new vaccines using synthetic genomes. Nucleases associated with zinc finger proteins have affected the treatment and cure of diseases such as HIV [103,112]. With the advent of CRISPR/Cas9, one can imagine a series of applications to treat diseases such as Yao and Chen summarized regarding cancer therapy [134]. This deep understanding not only of clinical symptoms but also of the molecular components and how they interact to result in a specific disease is a prerequisite for the development of approaches to personalized medicine.

#### References

- [1] H. Kitano, Systems biology: a brief overview, Science 295 (2002) 1662-1664.
- [2] E.C. Butcher, E.L. Berg, E.J. Kunkel, Systems biology in drug discovery, Nature Biotechnology 22 (2004) 1253–1259.
- [3] R.J. Kleijn, J.M. Buescher, L. Le Chat, et al., Metabolic fluxes during strong carbon catabolite repression by malate in *Bacillus subtilis*, The Journal of Biological Chemistry 285 (2010) 1587–1596.
- [4] J.M. Buescher, W. Liebermeister, M. Jules, et al., Global network reorganization during dynamic adaptations of *Bacillus subtilis* metabolism, Science 335 (2012) 1099–1103.
- [5] A.L. Barabási, Network Science, Cambridge University Press, Cambridge, 2016.
- [6] A.-L. Barabási, Z.N. Oltvai, Network biology: understanding the cell's functional organization, Nature Reviews Genetics 5 (2004) 101–113.
- [7] S. Bornholdt, H.G. Schuster, Handbook of Graphs and Networks: From the Genome to the Internet, Wiley-VCH, 2003. Wolfgang.
- [8] M. Dehmer, F. Emmert-Streib, A. Graber, A. Salvador, Applied Statistics for Network Biology: Methods in Systems Biology, Wiley-Blackwell, Hoboken, 2011.
- [9] A. Platzer, P. Perco, A. Lukas, B. Mayer, Characterization of protein-interaction networks in tumors, BMC Bioinformatics 8 (2007) 224.
- [10] P. Tieri, C. Nardini, Signalling pathway database usability: lessons learned, Molecular BioSystems 9 (2013) 2401–2407.
- [11] W. Huang, P. Wang, Z. Liu, L. Zhang, Identifying disease associations via genome-wide association studies, BMC Bioinformatics 10 (2009) S68.

- [12] K. Wu, C.V. Rao, Computational methods in synthetic biology: towards computer-aided part design, Current Opinion in Chemical Biology 16 (2012) 318–322.
- [13] B. Lehner, I. Lee, Network-guided genetic screening: building, testing and using gene networks to predict gene function, Briefings in Functional Genomics & Proteomics 7 (2008) 217–227.
- [14] I.B. McInnes, G. Schett, The pathogenesis of rheumatoid arthritis, The New England Journal of Medicine 365 (2011) 2205–2219.
- [15] P. Tieri, X. Zhou, L. Zhu, C. Nardini, Multi-omic landscape of rheumatoid arthritis: re-evaluation of drug adverse effects, Frontiers in Cell and Developmental Biology 2 (2014) 59.
- [16] J.U. Scher, A. Sczesnak, R.S. Longman, et al., Expansion of intestinal *Prevotella copri* correlates with enhanced susceptibility to arthritis, eLife (2013) e01202.
- [17] E.Y. Chiang, X. Yu, J.L. Grogan, Immune complex-mediated cell activation from systemic lupus erythematosus and rheumatoid arthritis patients elaborate different requirements for IRAK1/4 kinase activity across human cell types, The Journal of Immunology 186 (2011) 1279–1288.
- [18] A. Friedman, N. Perrimon, Genetic screening for signal transduction in the era of network biology, Cell 128 (2007) 225–231.
- [19] I.D.C. Fraser, R.N. Germain, Navigating the network: signaling cross-talk in hematopoietic cells, Nature Immunology 10 (2009) 327–331.
- [20] V. Volterra, Fluctuations in the abundance of a species considered mathematically, Nature 118 (1926) 558–560.
- [21] J. Tyson, What everyone should know about the Belousov–Zhabotinsky reaction, in: S.A. Levin (Ed.), Frontiers in Mathematical Biology, vol. 100, Springer Berlin Heidelberg, Berlin, 1994, pp. 569–587.
- [22] A. Cappuccio, P. Tieri, F. Castiglione, Multiscale modelling in immunology: a review, Briefings in Bioinformatics (2015).
- [23] S. Wolfram, A New Kind of Science, first ed., Wolfram Media, Champaign, 2002.
- [24] F. Castiglione, F. Celada, Immune System Modeling and Simulation, first ed., CRC Press, Boca Raton, 2015.
- [25] F. Burnet, The Clonal Selection Theory of Acquired Immunity, first ed., Cambridge University Press, Cambridge, 1959.
- [26] N. Rapin, O. Lund, M. Bernaschi, F. Castiglione, Computational immunology meets bioinformatics: the use of prediction tools for molecular binding in the simulation of the immune system, PLoS One 5 (2010) e9862.
- [27] B. Korber, M. LaBute, K. Yusim, Immunoinformatics comes of age, Computational Biology 2 (2006) 484–492.
- [28] C. Lundegaard, O. Lund, C. Keşmir, et al., Modeling the adaptive immune system: predictions and simulations, Bioinformatics 23 (2007) 3265–3275.
- [29] B. Kohler, R. Puzone, P.E. Seiden, F. Celada, A systematic approach to vaccine complexity using an automaton model of the cellular and humoral immune system I. Viral characteristics and polarized responses, Vaccine 19 (2000) 862–876.
- [30] F. Castiglione, B. Ribba, O. Brass, Comparing in silico results to in vivo and ex vivo of influenzaspecific immune responses after vaccination or infection in humans, in: S. Baschieri (Ed.), Innovation in Vaccinology, Springer Netherlands, Dordrecht, 2012, pp. 17–42.
- [31] T. Knight, Idempotent vector design for standard assembly of BioBricks idempotent vector design for standard assembly of BioBricks, Structure (2003) 1–11.
- [32] J.C. Anderson, J.E. Dueber, M. Leguia, et al., BglBricks: a flexible standard for biological part assembly, Journal of Biological Engineering 4 (2010) 1–12.

- [33] I. Phillips, P. Silver, A new BioBrick assembly strategy designed for facile protein engineering, 2006, pp. 1–6. DSpace@MIT.
- [34] D.G. Gibson, J.I. Glass, C. Lartigue, et al., Creation of a bacterial cell controlled by a chemically synthesized genome, Science 329 (2010) 52–56.
- [35] R. Kelwick, J.T. MacDonald, A.J. Webb, P. Freemont, Developments in the tools and methodologies of synthetic biology, Frontiers in Bioengineering and Biotechnology 2 (2014) 1–23.
- [36] M.S. Siddiqui, K. Thodey, I. Trenchard, C.D. Smolke, Advancing secondary metabolite biosynthesis in yeast with synthetic biology tools, FEMS Yeast Research 12 (2012) 144–170.
- [37] H. Kim, S.J. Yoo, H.A. Kang, Yeast synthetic biology for the production of recombinant therapeutic proteins, FEMS Yeast Research 15 (2014) 1–16.
- [38] X. Duportet, L. Wroblewska, P. Guye, et al., A platform for rapid prototyping of synthetic gene networks in mammalian cells, Nucleic Acids Research 42 (2014) 13440–13451.
- [39] C. Kemmer, M. Gitzinger, M. Daoud-El Baba, et al., Self-sufficient control of urate homeostasis in mice by a synthetic circuit, Nature Biotechnology 28 (2010) 355–360.
- [40] O. Kovtun, S. Mureev, W. Jung, et al., *Leishmania* cell-free protein expression system, Methods 55 (2011) 58–64.
- [41] C.E. Hodgman, M.C. Jewett, Cell-free synthetic biology: thinking outside the cell, Metabolic Engineering 14 (2012) 261–269.
- [42] K. Pardee, A.A. Green, T. Ferrante, et al., Paper-based synthetic gene networks, Cell 159 (2015) 940–954.
- [43] S. Kay, S. Hahn, E. Marois, et al., A bacterial effector acts as a plant transcription factor and induces a cell size regulator, Science 318 (2007) 648–651.
- [44] P. Römer, S. Hahn, T. Jordan, et al., Plant pathogen recognition mediated by promoter activation of the pepper Bs3 resistance gene, Science 318 (2007) 645–648.
- [45] F.D. Urnov, E.J. Rebar, M.C. Holmes, et al., Genome editing with engineered zinc finger nucleases, Nature Reviews Genetics 11 (2010) 636–646.
- [46] D. Carroll, Genome engineering with zinc-finger nucleases, Genetics 188 (2011) 773-782.
- [47] R. Barrangou, C. Fremaux, H. Deveau, et al., CRISPR provides acquired resistance against viruses in prokaryotes, Science 315 (2007) 1709–1712.
- [48] P. Liang, Y. Xu, X. Zhang, et al., CRISPR/Cas9-mediated gene editing in human tripronuclear zygotes, Protein & Cell 6 (2015) 363–372.
- [49] L. Bortesi, R. Fischer, The CRISPR/Cas9 system for plant genome editing and beyond, Biotechnology Advances 33 (2015) 41–52.
- [50] J.E. Dicarlo, J.E. Norville, P. Mali, et al., Genome engineering in *Saccharomyces cerevisiae* using CRISPR-Cas systems, Nucleic Acids Research 41 (2013) 4336–4343.
- [51] D. Reyon, M.L. Maeder, C. Khayter, et al., Engineering customized TALE nucleases (TALENS) and TALE transcription factors by fast ligation-based automatable solid-phase high-throughput (FLASH) assembly, Current Protocols in Molecular Biology 103 (2013) 12.16.1–12.16.18.
- [52] A.A. Nemudryi, K.R. Valetdinova, S.P. Medvedev, S.M. Zakian, TALEN and CRISPR/Cas genome editing systems: tools of discovery, Acta Naturae 6 (2014) 19–40.
- [53] A.S. Khalil, J.J. Collins, Synthetic biology: applications come of age, Nature Reviews Genetics 11 (2010) 367–379.
- [54] T.S. Moon, C. Lou, A. Tamsir, et al., Genetic programs constructed from layered logic gates in single cells, Nature 491 (2012) 249–253.
- [55] D.J. Sayut, Y. Niu, L. Sun, Construction and enhancement of a minimal genetic AND logic gate, Applied and Environmental Microbiology 75 (2009) 637–642.

- [56] R. Baron, O. Lioubashevski, E. Katz, et al., Logic gates and elementary computing by enzymes, The Journal of Physical Chemistry A 110 (2006) 8548–8553.
- [57] T. Miyamoto, S. Razavi, R. Derose, T. Inoue, Synthesizing biomolecule-based boolean logic gates, ACS Synthetic Biology 2 (2013) 72–82.
- [58] T.S. Gardner, C.R. Cantor, J.J. Collins, Construction of a genetic toggle switch in *Escherichia coli*, Nature 403 (2000) 339–342.
- [59] B. Novák, J.J. Tyson, Design principles of biochemical oscillators, Nature Reviews Molecular Cell Biology 9 (12) (2008) 981–991.
- [60] C. Berens, F. Groher, B. Suess, RNA aptamers as genetic control devices: the potential of riboswitches as synthetic elements for regulating gene expression, Biotechnology Journal 10 (2015) 246–257.
- [61] A. Wittmann, B. Suess, Engineered riboswitches: expanding researchers' toolbox with synthetic RNA regulators, FEBS Letters 586 (2012) 2076–2083.
- [62] M.S. Dasika, C.D. Maranas, OptCircuit: an optimization based method for computational design of genetic circuits, BMC Systems Biology 2 (2008) 1–19.
- [63] B. Wang, R.I. Kitney, N. Joly, M. Buck, Engineering modular and orthogonal genetic logic gates for robust digital-like synthetic biology, Nature Communications 2 (2011) 1–9.
- [64] A. Tamsir, J.J. Tabor, C.A. Voigt, Robust multicellular computing using genetically encoded NOR gates and chemical "wires", Nature 469 (2011) 212–215.
- [65] D. Wang, Y. Fu, J. Yan, et al., Molecular logic gates on DNA origami nanostructures for microRNA diagnostics, Analytical Chemistry 86 (2014) 1932–1936.
- [66] J. Wang, E. Katz, Digital biosensors with built-in logic for biomedical applications, Israel Journal of Chemistry 51 (2011) 141–150.
- [67] E.M. Cornett, E.A. Campbell, G. Gulenay, et al., Molecular logic gates for DNA analysis: detection of rifampin resistance in *M. tuberculosis* DNA, Angewandte Chemie International Edition 51 (2012) 9075–9077.
- [68] J.W. Kotula, S.J. Kerns, L.A. Shaket, et al., Programmable bacteria detect and record an environmental signal in the mammalian gut, Proceedings of the National Academy of Sciences of the United States of America 111 (2014) 4838–4843.
- [69] A. Courbet, D. Endy, E. Renard, et al., Detection of pathological biomarkers in human clinical samples via amplifying genetic switches and logic gates, Science Translational Medicine 7 (2015), 289ra83.
- [70] R. Konopke, J. Roth, A. Volk, et al., Colorectal liver metastases: an update on palliative treatment options, The Journal of Gastrointestinal and Liver Diseases 21 (2012) 83–91.
- [71] M. Selzner, M.A. Morse, J.J. Vredenburgh, et al., Liver metastases from breast cancer: long-term survival after curative resection, Surgery 127 (2000) 383–389.
- [72] T. Danino, A. Prindle, G.A. Kwong, et al., Programmable probiotics for detection of cancer in urine, Science Translational Medicine 7 (2015), 289ra84.
- [73] Z. Xie, L. Wroblewska, L. Prochazka, et al., Multi-input RNAi-based logic circuit for identification of specific cancer cells, Science 333 (2011) 1307–1311.
- [74] M. Hernández, L. Pozo, I. Gómez, A. Melchor, Chimeric synthetic peptide as antigen for immunodiagnosis of HIV-1 infection, Biochemical and Biophysical Research Communications 272 (2000) 259–262.
- [75] R.M. Khaitov, S.M. Andreev, Prospects for the use of synthetic antigens in immunodiagnosis, Journal of Biomedical Science 1 (1990) 553–564.
- [76] P.D. Burbelo, A.T. Issa, K.H. Ching, et al., Rapid, simple, quantitative, and highly sensitive antibody detection for lyme disease, Clinical and Vaccine Immunology 17 (2010) 904–909.

- [77] A.E. Smartt, S. Ripp, Bacteriophage reporter technology for sensing and detecting microbial targets, Analytical and Bioanalytical Chemistry 400 (2011) 991–1007.
- [78] D. Schofield, N.J. Sharp, C. Westwater, Phage-based platforms for the clinical detection of human bacterial pathogens, Bacteriophage 2 (2012) 105–283.
- [79] M. Delbrück, The growth of bacteriophage and lysis of the host, The Journal of General Physiology 23 (1940) 643–660.
- [80] M.R.J. Clokie, A.D. Millard, A.V. Letarov, S. Heaphy, Phages in nature, Bacteriophage 1 (2011) 31-45.
- [81] R. McNerney, B.S. Kambashi, J. Kinkese, et al., Development of a bacteriophage phage replication assay for diagnosis of pulmonary tuberculosis, Journal of Clinical Microbiology 42 (2004) 2115–2120.
- [82] M. Piuri, W.R. Jacobs Jr., G.F. Hatfull, Fluoromycobacteriophages for rapid, specific, and sensitive antibiotic susceptibility testing of *Mycobacterium tuberculosis*, PLoS One 4 (2009) e4870.
- [83] A. Dusthackeer, V. Kumar, S. Subbian, et al., Construction and evaluation of luciferase reporter phages for the detection of active and non-replicating tubercle bacilli, Journal of Microbiological Methods 73 (2008) 18–25.
- [84] K.V. Sergueev, Y. He, R.H. Borschel, et al., Rapid and sensitive detection of *Yersinia pestis* using amplification of plague diagnostic bacteriophages monitored by real-time PCR, PLoS One 5 (2010) e11337.
- [85] Y. Fujinami, Y. Hirai, I. Sakai, et al., Sensitive detection of *Bacillus anthracis* using a binding protein originating from gamma-phage, Microbiology and Immunology 51 (2007) 163–169.
- [86] D.A. Schofield, C. Westwater, Phage-mediated bioluminescent detection of *Bacillus anthracis*, Journal of Applied Microbiology 107 (2009) 1468–1478.
- [87] D.A. Schofield, I.J. Molineux, C. Westwater, "Bioluminescent" reporter phage for the detection of category a bacterial pathogens, Journal of Visualized Experiments 53 (2011) e2740.
- [88] D.A. Schofield, I.J. Molineux, C. Westwater, Diagnostic bioluminescent phage for detection of *Yersinia pestis*, Journal of Clinical Microbiology 47 (2009) 3887–3894.
- [89] M.H. Hazbón, N. Guarín, B.E. Ferro, et al., Photographic and luminometric detection of luciferase reporter phages for drug susceptibility testing of clinical *Mycobacterium tuberculosis* isolates, Journal of Clinical Microbiology 41 (2003) 4865–4869.
- [90] P.F. Riska, Y. Su, S. Bardarov, et al., Rapid film-based determination of antibiotic susceptibilities of *Mycobacterium tuberculosis* strains by using a luciferase reporter phage and the Bronx box, Journal of Clinical Microbiology 37 (1999) 1144–1149.
- [91] A.E. Smartt, T. Xu, P. Jegier, et al., Pathogen detection using engineered bacteriophages, Analytical and Bioanalytical Chemistry 402 (2012) 3127–3146.
- [92] M.T. Smith, K.M. Wilding, J.M. Hunt, et al., The emerging age of cell-free synthetic biology, FEBS Letters 588 (2014) 2755–2761.
- [93] M.K. Takahashi, J. Chappell, C.A. Hayes, et al., Rapidly characterizing the fast dynamics of RNA genetic circuitry with cell-free transcription-translation (TX-TL) systems, ACS Synthetic Biology 4 (2014) 503–515.
- [94] Z.Z. Sun, E. Yeung, C.A. Hayes, et al., Linear DNA for rapid prototyping of synthetic biological circuits in an *Escherichia coli* based TX-TL cell-free system, ACS Synthetic Biology 3 (2014) 387–397.
- [95] E. Karzbrun, A.M. Tayar, V. Noireaux, R.H. Bar-Ziv, Synthetic biology. Programmable on-chip DNA compartments as artificial cells, Science 345 (2014) 829–832.
- [96] S.H. Hong, I. Ntai, A.D. Haimovich, et al., Cell-free protein synthesis from a release factor 1 deficient *Escherichia coli* activates efficient and multiple site-specific nonstandard amino acid incorporation, ACS Synthetic Biology 3 (2014) 398–409.

- [97] R.D. Woodyer, Z. Shao, P.M. Thomas, et al., Heterologous production of fosfomycin and identification of the minimal biosynthetic gene cluster, Chemistry & Biology 13 (2006) 1171–1182.
- [98] V.J.J. Martin, D.J. Pitera, S.T. Withers, et al., Engineering a mevalonate pathway in *Escherichia coli* for production of terpenoids, Nature Biotechnology 21 (2003) 796–802.
- [99] D.-K. Ro, E.M. Paradise, M. Ouellet, et al., Production of the antimalarial drug precursor artemisinic acid in engineered yeast, Nature 440 (2006) 940–943.
- [100] P.K. Ajikumar, W.-H. Xiao, K.E.J. Tyo, et al., Isoprenoid pathway optimization for Taxol precursor overproduction in *Escherichia coli*, Science 330 (2010) 70–74.
- [101] D. Ausländer, S. Ausländer, G. Charpin-El Hamri, et al., A synthetic multifunctional mammalian pH sensor and CO<sub>2</sub> transgene-control device, Molecular Cell 55 (2014) 397–408.
- [102] H. Ye, M. Daoud-El Baba, R.-W. Peng, M. Fussenegger, A synthetic optogenetic transcription device enhances blood-glucose homeostasis in mice, Science 332 (2011) 1565–1568.
- [103] P. Tebas, D. Stein, W.W. Tang, et al., Gene editing of CCR5 in autologous CD4 T cells of persons infected with HIV, The New England Journal of Medicine 370 (2014) 901–910.
- [104] J. Van Lunzen, T. Glaunsinger, I. Stahmer, et al., Transfer of autologous gene-modified T cells in HIV-infected patients with advanced immunodeficiency and drug-resistant virus, Molecular Therapy 15 (2007) 1024–1033.
- [105] D.L. Klayman, Qinghaosu (artemisinin): an antimalarial drug from China, Science 228 (1985) 1049–1055.
- [106] P.S. Covello, K.H. Teoh, D.R. Polichuk, et al., Functional genomics and the biosynthesis of artemisinin, Phytochemistry 68 (2007) 1864–1871.
- [107] N. Schramek, H. Wang, W. Römisch-Margl, et al., Artemisinin biosynthesis in growing plants of Artemisia annua. A <sup>13</sup>CO<sub>2</sub> study, Phytochemistry 71 (2010) 179–187.
- [108] B. Liu, H. Wang, Z. Du, et al., Metabolic engineering of artemisinin biosynthesis in Artemisia annua L. Plant Cell Reports 30 (2011) 689–694.
- [109] G.D. Brown, L.K. Sy, In vivo transformations of artemisinic acid in Artemisia annua plants, Tetrahedron 63 (2007) 9548–9566.
- [110] S.B. Gelvin, Agrobacterium-mediated plant transformation: the biology behind the "gene-jockeying" tool, Microbiology and Molecular Biology Reviews 67 (2003) 16–37.
- [111] S.Y. Lee, H.U. Kim, J.H. Park, et al., Metabolic engineering of microorganisms: general strategies and drug production, Drug Discovery Today 14 (2009) 78–88.
- [112] W. Gu, Genome editing-based HIV therapies, Trends in Biotechnology 33 (2015) 172–179.
- [113] K. Kindsmuller, R. Wagner, Synthetic biology: impact on the design of innovative vaccines, Human Vaccines 7 (2011) 658–662.
- [114] S.F. Nunes, C. Hamers, M. Ratinier, et al., A synthetic biology approach for a vaccine platform against known and newly emerging serotypes of bluetongue virus, Journal of Virology 88 (2014) 12222–12232.
- [115] E. De Gregorio, R. Rappuoli, From empiricism to rational design: a personal perspective of the evolution of vaccine development, Nature Reviews Immunology 14 (2014) 505–514.
- [116] G. Schwank, B.K. Koo, V. Sasselli, et al., Functional repair of CFTR by CRISPR/Cas9 in intestinal stem cell organoids of cystic fibrosis patients, Cell Stem Cell 13 (2013) 653–658.
- [117] Y. Wu, D. Liang, Y. Wang, et al., Correction of a genetic disease in mouse via use of CRISPR-Cas9, Cell Stem Cell 13 (2013) 659–662.
- [118] D.C. Tang, M. DeVit, S.A. Johnston, Genetic immunization is a simple method for eliciting an immune response, Nature 356 (1992) 152–154.

- [119] J.A. Wolff, R.W. Malone, P. Williams, et al., Direct gene transfer into mouse muscle in vivo, Science 247 (1990) 1465–1468.
- [120] M. Fotin-Mleczek, K.M. Duchardt, C. Lorenz, et al., Messenger RNA-based vaccines with dual activity induce balanced TLR-7 dependent adaptive immune responses and provide antitumor activity, Journal of Immunotherapy 34 (2011) 1–15.
- [121] M. Ingolotti, O. Kawalekar, D.J. Shedlock, et al., DNA vaccines for targeting bacterial infections, Expert Review of Vaccines 9 (2010) 747–763.
- [122] L. Deml, A. Bojak, S. Steck, et al., Multiple effects of codon usage optimization on expression and immunogenicity of DNA candidate vaccines encoding the human immunodeficiency virus type 1 Gag protein, Journal of Virology 75 (2001) 10991–11001.
- [123] E.J. Tarcha, V. Basrur, C.Y. Hung, et al., Multivalent recombinant protein vaccine against coccidioidomycosis, Infection and Immunity 74 (2006) 5802–5813.
- [124] G. Dakshinamoorthy, A.K. Samykutty, G. Munirathinam, et al., Multivalent fusion protein vaccine for lymphatic filariasis, Vaccine 31 (2013) 1616–1622.
- [125] E. Dolgin, Synthetic biology: safety boost for GM organisms, Nature 517 (2015) 423.
- [126] H. König, D. Frank, R. Heil, C. Coenen, Synthetic genomics and synthetic biology applications between hopes and concerns, Current Genomics 14 (2013) 11–24.
- [127] F. Molina, M. Dehmer, P. Perco, et al., Systems biology: opening new avenues in clinical research, Nephrology Dialysis Transplantation 25 (2010) 1015–1018.
- [128] M.M. Shen, Cancer: the complex seeds of metastasis, Nature 520 (2015) 298-299.
- [129] M.M. Gubin, X. Zhang, H. Schuster, et al., Checkpoint blockade cancer immunotherapy targets tumour-specific mutant antigens, Nature 515 (2014) 577–581.
- [130] A.V. Ullal, V. Peterson, S.S. Agasti, et al., Cancer cell profiling by barcoding allows multiplexed protein analysis in fine-needle aspirates, Science Translational Medicine 6 (2014), 219ra9.
- [131] A.E. Locke, B. Kahali, S.I. Berndt, et al., Genetic studies of body mass index yield new insights for obesity biology, Nature 518 (2015) 197–206.
- [132] Y. Okada, D. Wu, G. Trynka, et al., Genetics of rheumatoid arthritis contributes to biology and drug discovery, Nature 506 (2014) 376–381.
- [133] P.R. Dormitzer, P. Suphaphiphat, D.G. Gibson, et al., Synthetic generation of influenza vaccine viruses for rapid response to pandemics, Science Translational Medicine 5 (2013), 185ra68.
- [134] S. Yao, C. Chen, CRISPR/Cas9 mediated genome editing of epigenetic factors for cancer therapy, Human Gene Therapy 26 (2015) 463–471.

# 10

## Noncoding RNAs as Critical Players in Regulatory Accuracy, Redox Signaling, and Immune Cell Functions

A.Q. Gomes<sup>2,4</sup>, C. Real<sup>1</sup>, F. Antunes<sup>1</sup>, H.S. Marinho<sup>1</sup>, S. Nolasco<sup>2,3,5</sup>, H. Soares<sup>1,2,3,\*</sup>

<sup>1</sup>CENTRO DE QUÍMICA E BIOQUÍMICA, FACULDADE DE CIÊNCIAS, UNIVERSIDADE DE LISBOA, LISBOA, PORTUGAL; <sup>2</sup>ESCOLA SUPERIOR DE TECNOLOGIA DA SAÚDE DE LISBOA, IPL, LISBOA, PORTUGAL; <sup>3</sup>INSTITUTO GULBENKIAN DE CIÊNCIA, OEIRAS, PORTUGAL; <sup>4</sup>INSTITUTO DE MEDICINA MOLECULAR, FACULDADE DE MEDICINA, UNIVERSIDADE DE LISBOA, LISBOA, PORTUGAL; <sup>5</sup>INSTITUTO DE INVESTIGAÇÃO CIENTÍFICA TROPICAL, CVZ, CIISA, FACULDADE DE MEDICINA VETERINÁRIA, UNIVERSIDADE DE LISBOA, LISBOA, PORTUGAL

## 10.1 Noncoding RNA Molecules: From Evolution to Human Disease Therapy

The use of genome-wide approaches and the large output of genome sequencing technologies have revealed that the mammalian transcriptome is much more complex than initially thought because it includes a large number of small noncoding RNAs (sncRNAs) and long noncoding RNAs (lncRNAs) [1,2]. As the genome sequence analysis of diverse organisms was accomplished, no direct correlation was found between genome size or the number of protein-coding genes and the complexity of the organism, even when the generation of protein diversity by alternative splicing and post-translational regulation was taken into account [3]. On the other hand, it was established that 75% of the human genome is transcribed into RNA, whereas only 3% is transcribed into mRNAs [4-6], which clearly shows that the number of noncoding RNAs (ncRNAs) is probably much higher than that of genes coding for proteins. This fact, together with growing data regarding ncRNAs, suggests that organism complexity better correlates with the percentage of each genome that is transcribed into these molecules [7].

<sup>&</sup>lt;sup>\*</sup>Corresponding Author.

Supporting this view is the fact that most ncRNAs display complex patterns of expression during differentiation and development, as well as tissue-specific expression.

A great number of different classes of ncRNAs, based on their length, biogenesis pathways, features, and functions have been reported in the literature [8,9]. However, three classes of sncRNAs have been the most studied, specifically short interfering RNAs (siRNAs), microRNAs (miRNAs) and PIWI-interacting RNAs (piRNAs). The siRNAs and miRNAs are small RNAs about 20–30 nucleotides (nt) long that are produced from double-stranded RNA (dsRNA) precursors that can be exogenously introduced into, or formed in cells by, gene transcription of both sense and antisense DNA strands, transcription of pseudogenes, and inverted repeats. These ncRNAs are involved in mechanisms related to the silencing of specific genes by mRNA degradation and/or translational repression, and also to cell/genome defense against virus, mobile repetitive DNA sequences, retro-elements, and transposons [10].

The less characterized class of sncRNAs is the piRNAs, small RNAs of about 24–31 nt in length, that have been essentially detected in germline cells of mammals, fish, and *Drosophila melanogaster* [11,12]. These sncRNAs have important roles in germline development by suppressing transposon activity. In fact, disruption of the piRNA biogenesis pathway is associated with increased transposon expression, and in mouse and fish it causes germline-specific cell death and sterility [13]. In *Drosophila* heat shock protein 90 kDa (Hsp90), PIWI and Hop form a protein complex that seems to be required to prevent phenotypic variation despite genotypic variation and environmental influences (canalization) [14]. Hsp90 seems to be able to avoid phenotypic variation by preventing the mutagenic activity of transposons, probably because it is a capacitor [15,16]. Thus piRNA pathway–dependent mechanisms may have been critical during evolution in the establishment of developmental robustness.

The ability of piRNAs to protect genomes against parasitic nucleic acids and the importance of this protection for development seem to have been established early in evolution because protozoa ciliates have a mechanism that resembles that of piRNAs. In these organisms the polyploid macronucleus (the somatic nucleus) maintains the vegetative growth of the cell. This nucleus develops from the diploid micronucleus (the germline nucleus) by complex programmed genome reorganization after sexual conjugation. This process requires chromosome fragmentation and elimination of germline-specific sequences (internal eliminated sequences, transposons, and other repeated sequences). This occurs by a mechanism that compares the germline and somatic genomes followed by a genomic subtraction between meiosis-specific germline sncRNAs (small RNAs that resemble piRNAs) and longer noncoding transcripts from the maternal somatic genome [17]. Consequently, an epigenetic memory of rearrangement patterns is preserved through sexual generations. This evidence associated with the involvement of piRNAs in developmental robustness strongly supports the role of piRNAs in the ancestral development of complex organisms.

Besides sncRNAs, a large percentage of the transcriptomes consists of ncRNA transcripts that are longer than -200 nt in length and are designated as lncRNAs [2,18,19]. The estimated number of gene members integrating this class of ncRNAs ranges from 10,000 to >200,000 [21]. These molecules are transcribed from diverse origins such as intergenic regions and promoter regions but can also be interleaved, overlapping, or antisense to genes encoding proteins [22]. Transcribed pseudogenes can also be an origin for lncRNAs [23]. Similar to primary transcripts produced by transcription of genes encoding proteins, lncRNAs are processed acquiring a 5' cap and a poly (A)<sup>+</sup> tail and can undergo alternative splicing and RNA editing [24,25]. In past years a wide variety of regulatory functions have been ascribed to lncRNAs [26] such as epigenetic regulation and transcriptional and posttranscriptional control including translation regulation. These regulatory events take place during both developmental and differentiation processes, which make lncRNAs active players in dosage compensation (X chromosome inactivation), genomic imprinting, cell differentiation, and organogenesis. It is also clear that the number of lncRNA species increases in genomes of developmentally complex organisms, which strongly suggests that these RNA molecules were probably critical in the evolution of multicellular organisms [7]. The biological importance of lncRNAs is also supported by the fact that they display complex patterns of expression and different subcellular localizations.

Interestingly it was shown that some lncRNAs can be transcribed from mitochondrial genomes [27]. Taking into account the assumed endosymbiont origin of mitochondria, these results suggest that ncRNAs may also exist and have important regulatory roles in prokaryotes. In fact, the transcriptomes of distinct species of bacteria as different as, e.g., Helicobacter pylori, Bacillus subtilis, Escherichia coli, and Mycoplasma pneumoniae contain antisense RNA (asRNA) transcripts for a high percentage of genes. The sizes of asRNAs are diverse, ranging from 100 to 300 nt (e.g., SymR [28], GadY [29], and SyR7 [30], to 700–3500 nt) [31]. Interestingly, asRNAs share functional similarities with eukaryotic trans-acting regulatory RNAs: They are involved in the regulation of transcription termination, degradation of protein-coding RNAs, control of translation, transcriptional interference, and enhanced stability of their respective target transcripts [31]. On the other hand, bacterial sncRNAs (sRNAs) have been identified, e.g., in E. coli and Salmonella typhimurium. These sncRNAs are usually smaller than 300 nt and are able to pair with mRNAs, leading to the activation or inhibition of mRNA translation (RNAIII) [32], mRNA stabilization (GadY) [29], or mRNA degradation (RyhB) [33]. Although many of these mechanisms are distinct from those operating in eukaryotic cells, many also require cooperation between RNAs and protein partners [34].

Consequently, ncRNAs seem to have been evolutionarily important for the establishment of complex metazoa, but growing data concerning these molecules also suggest that they were already present and performing biological roles before the split of eukaryotes and prokaryotes. These facts refurbish discussion regarding the hypothesis of a primordial "RNA world" that proposes that the first organisms could rely only on RNA molecules, and only later on was the evolution of a more complex system based on proteins established [35,36]. Interestingly, data concerning bacterial riboswitches shows that similar to sRNAs, these RNA structures, located within the 5'-untranslated region (UTR) of mRNAs that regulate gene expression at the level of transcription, translation, or splicing [37,38], can also act in *trans* [39]. Moreover, specific ligands are able to change the riboswitch conformation, and in certain cases it was proposed that mRNA
decay mediated by the riboswitch could be a target to RNase E. If this is the mechanism and the regulator riboswitch can control gene expression precisely by modulating the RNase cleavage site, this would be additional support for the "RNA first" model [34].

On the other hand, several data point to the fact that, at least for some ncRNAs such as XIST, H-19, and the brain-specific Human Accelerated RNA [40], sequence conservation interspecies is not observed. Also a group of brain miRNAs found in humans and chimpanzees is not conserved away from primates, and some seem to be speciesspecific, which suggests a recent origin [41]. This evidence has been used to support the theory that miRNA relative nonconservation suggests that metazoan ncRNAs are a relatively recent evolutionary macro event and not vestiges of the old RNA world [42]. Alternatively, this can indicate that evolution of miRNAs is still occurring, and that along with ancient, highly conserved miRNAs, there are a number of emerging miRNAs.

Beyond this discussion remains the fact that new data strongly support the idea that RNAs evolve together with proteins and DNA, constituting a complex system of information in prokaryotic and eukaryotic genomes. In fact, the progressive characterization of these molecules has been having a considerable impact on our view of pathophysiologic states, thus providing new insights into the molecular basis of several diseases. It was shown that human  $\alpha$ -thalassemia may be caused by the up-regulation and translocation of an antisense long noncoding transcript to nearby the  $\alpha$ -globin gene. This asRNA is responsible for the dense methylation of a CpG island involved in the transcriptional regulation of the  $\alpha$ -globin gene that becomes silenced during development. subsequently causing  $\alpha$ -thalassemia [43]. Also, single-nucleotide polymorphism associated with an increased risk for autoimmune thyroid disease (AITD) simultaneous occurs in the 3'-UTR of the zinc finger gene ZFAT (a gene in the AITD susceptibility region) and in the promoter region of the small antisense transcript of ZFAT, with a critical role in B-cell function by inhibiting the expression of the antisense transcript, which in turn causes increased ZFAT mRNA stability [44]. These examples are merely illustrative, because the implication of ncRNAs in human diseases is continuously growing. For example, lncRNAs have been associated with distinct cancers [45,46], neurologic diseases [45,47], several heritable syndromes (e.g., hemolysis, elevated liver enzymes, and low platelets; syndrome congenital skeletal malformation brachydactyly type E) [48], facioscapulohumeral muscular dystrophy [49], and schizophrenia 2 [50]. Because in most of these cases lncRNAs have been associated with the diseases through abnormal expression patterns, they may become important for diagnosis as biomarkers.

Also, the involvement of miRNAs in regulating early embryonic development, neuronal, muscle and lymphocyte development, and their functional association with various stem cell populations, in the context of neural function and cardiovascular biology, in metabolism and aging, and during cancer, positioned them as critical molecules in maintaining homeostasis [51]. In fact, results coming from gain- and loss-of-function studies have revealed prominent roles for miRNAs: for example, in cardiovascular disorders including myocardial infarction, cardiac hypertrophy, heart failure, angiogenesis, vascular stenosis, and fibrosis [52–57]. In this context, ncRNAs are

emerging as "hopeful" molecules attracting attention not only for basic research but also for biomarker discovery and therapeutic applications that have been explored with the development of nanotechnologies [8,45,52,58]. These potential therapeutic applications of ncRNAs have been reinforced by the accomplishment of successful phase I and phase II clinical trials of Santaris Pharma's anti-miR against miR-122, miravirsen, for the therapy of patients infected with the hepatitis C virus (HCV) [52]. Also, the first cancertargeted miRNA drug, MRX34, a liposome-based miR-34 mimic, entered in phase I clinical trials in April 2013 for patients with advanced hepatocellular carcinoma [8,59]. On the other hand, attention is also focused on lncRNAs by designing inhibitors of the natural antisense transcript (NAT) class of lncRNAs, known as antagoNAT oligonucleotides, and exploring their potential therapeutic application. The success of the first clinical data showing the safety and efficacy of an anti-miR supports the pursuit of ncRNAs as a new class of drug targets [45].

In this chapter we will give an overview of the biogenesis pathways of miRNA, piRNA, and siRNA, as well as lncRNAs, focusing on the miRNA and lncRNAs gene expression regulatory roles specifically in redox signaling and immune cell functions [60,61]. We will also explore the interconnection between cellular redox homeostasis, immune system function, and the complex networks of ncRNA actions.

## 10.2 Biogenesis Pathways of Noncoding RNAs: An Overview

## 10.2.1 Small Noncoding RNAs

Small ncRNAs such as miRNAs, siRNAs, and piRNAs, are produced by different biogenesis pathways and regulatory mechanisms, but they share the common feature of using nucleotide sequence complementarity to identify target genes and regulate their expression.

#### 10.2.1.1 Short Interfering RNAs and MicroRNAs

Both siRNAs and miRNAs originate in the cleavage of dsRNA precursors. Regarding siRNAs, long dsRNAs can be produced from transposons, replicating virus, or endogenous genes that produce transcripts that are able to base pair, or when dsRNA is introduced experimentally in cells triggering gene silencing by a process designated by RNA interference (RNAi). These long dsRNA molecules are then processed by the Dicer ribonuclease, a member of the ribonuclease III (RNase III) family that originates small RNA duplexes with 2-nt overhangs at their 3' ends and phosphate groups at their 5' ends [62].

miRNAs are single-stranded RNAs conserved throughout the phylogenetic tree and their exact number is far from being estimated because identification of miRNAs occurs continuously. For example, the latest release (version June 21, 2014) of the miRBase database (http://www.mirbase.org), a primary miRNA sequence repository, has 28,465 entries that result in 35,828 mature miRNAs from 223 species including animals, plants, unicellular algae, and viruses [63]. Contrary to siRNAs, these sncRNAs are generated

from endogenous transcripts that are transcribed by RNA polymerase II, although some have been described as products of RNA polymerase III [64], in long dsRNA precursor transcripts (possessing several hundreds to thousands of nucleotides) designated by primary miRNA (pri-miRNA) that are capped and polyadenylated [9,65]. miRNAs were shown to be encoded within the cellular genome in three main ways: as unique genes, as intronic sequences within protein-coding genes, or as polycistronic miRNAs/single transcript encoding multiple miRNAs [66]. As a consequence of these encoding possibilities, their regulation can be subject to different mechanisms through alternative processing or miRNA processing.

The pri-miRNA precursors are processed by the ribonuclease (RNase) III Drosha-DGCR8/Pasha nuclear complex that originates smaller precursor miRNAs of 60–100 nt that possess a hairpin structure. These precursor miRNAs (pre-miRNAs) are transported to the cytoplasm through nuclear pores by the system of exportin5-RanGTP [67]. There, they are further cleaved by the Dicer enzyme into approximately 22-nt double-stranded mature miRNAs (miRNA–miRNA\* duplexes, where miRNA was assumed to be the antisense, or guide/mature strand and miRNA\* the sense, or passenger strand). Also, it has been generally presumed that the passenger strand was degraded. However, this view is now questioned by evidence showing that functional mature sequences can be derived from both arms of the hairpin [68,69]. Moreover, a shift between both arms of the precursor to produce functional mature microRNAs may occur between species or may be tissue- or developmental stage–specific [70].

As mentioned, miRNAs may derive from short intronic hairpins. These are termed "mirtrons" and are generated by an alternative, noncanonical biogenesis pathway that was identified both in vertebrates and invertebrates [71–73]. In this case, precursor miRNA (pre-miRNA)-like hairpins are produced by the action of the splicing machinery and the lariat-de-branching enzyme. This pathway converges with the canonical one in the export of mirtrons to the cytoplasm where both types of hairpins are going to be processed by Dicer into double-stranded miRNAs. Then, it is supposed that helicases separate both miRNAs strands and the mature strand is loaded into the RNA-induced silencing complex (RISC) a hetero-oligomeric complex integrating an Argonaute (AGO) protein, Dicer, and a dsRNA-binding protein (the TRBP in humans). Only one strand associates with the AGO protein, becoming the guide strand; the nonguide strand is cleaved during loading of mature miRNA strand into RISC [74,65]. The strand with the less stably paired 5' end is preferentially chosen to be loaded into the AGO proteins [75].

AGO proteins are the direct binding partners of small RNAs and are a highly conserved protein family found from humans to Archaea [74]. They have evolved into specialized clades (or subfamilies) [76]: the AGO subfamily or AGO clade, the first member was identified in *Arabidopsis thaliana* ARGONAUTE 1 (AGO1); the PIWI subfamily first identified in *D. melanogaster P*-element-induced wimpy testis (PIWI); and the WAGO subfamily (worm-specific AGO) of specific proteins of *Caenorhabditis elegans*. In humans, the AGO clade and PIWI clade are composed of four members each [77]. RISC-specific functions depend on each AGO protein, and probably on other proteins that are able

to recognize it and integrate the complex. Thus AGO1 and AGO2 (AGO2-RISCs are functionally distinct, silencing different types of target RNAs by different mechanisms [78,79]. The sorting of different ncRNAs into different RISC complexes implies the selection of individual AGO proteins, and this selection seems to rely on the intrinsic RNA duplex structure. For example, in the case of duplexes displaying unpaired central regions, as is generally observed in the case of miRNAs, the small RNA is routed to AGO1. When the complementarity between the miRNA bound to AGO1 and the target RNA is high, this causes miRNA tailing and 3'- to 5'-trimming. If the duplex displays perfect base pairing, as in the case of most siRNAs, it is driven to AGO2 [79,80]. The discrimination between AGO1 and AGO2 also seems to depend on the action of Hen1, an enzyme that adds the 2'-O-methyl group at the 3' ends of small RNAs bound to AGO2, but not those bound to AGO1 [81,82]. This methyl group is known to block tailing and trimming of the miRNA.

A second, non-Dicer-dependent pathway of miRNA biogenesis has been discovered [83,84]. This alternative pathway relies on AGO2 to mediate final processing of the miRNA. Indeed, AGO2-mediated cleavage of pre-miRNAs, followed by uridylation and trimming, generates functional miRNAs independent of Dicer. Also, here the secondary structure of the pre-miRNA seems to be important in determining the pathway of biogenesis, with distinct loop and hairpin structures determining whether the miRNA precursor is cleaved by Dicer or AGO2. Of the miRNAs processed by this pathway, miR-451 is the best characterized so far, and it contains a small, distinctive loop structure [83].

Interestingly, loading of the double-stranded siRNA or miRNA to the AGO protein requires interaction with a complex containing the molecular chaperones heat shock cognate protein 70 kDa and Hsp90, which maintains the AGO protein in a conformation that allows loading of the RNA duplex [85–87].

Finally, the guide-strand drives the RISC complex toward the target RNA that has a nucleotide sequence complementary to that of siRNA or miRNA (for miRNAs, this so-called "seed" sequence or short sequence at nt 2–8 on the 5′ end of the miRNA typically binds to the 3′-UTRs of their target mRNAs), and after pairing this is degraded or its translation is abolished [10,81,88]. It is estimated that approximately 85% of miRNA-mediated regulation in mammals occurs at the level of mRNA decay [89].

The target mRNA choice seems to require additional factors that interact with Dicer. In *Drosophila*, the Loqs-PB Dicer-partner cleaves pre-miR-307a, generating a longer miRNA isoform with a distinct seed sequence and target specificity [90]. In mammals, the mammalian TRBP homologue also interacts with Dicer to cleave pre-miR-132 generating a longer miRNA and consequently targets different mRNA molecules [91]. In *Schizosaccharomyces pombe*, AGO1 protein also seems to be associated with transcriptional silencing by inducing heterochromatin formation because it is a component of the RNA-induced transcriptional silencing complex [92]. Besides AGO1, this complex contains Chp1 (H3K9- binding protein) a poorly characterized protein Tas3, and siRNAs originated from centromeric repeat sequences [93], and is involved in recruitment of methyltransferases to specific genomic regions, where it methylates histone H3 at lysine

9 (H3K9). These data suggest the existence of cross-talk between siRNAs and chromatin remodeling [88] that does not seem to be a yeast specific characteristic because RNAi has also been linked to transcriptional silencing in humans [94].

### 10.2.2 The PIWI-Interacting RNAs

The piRNAs are the most recently described sncRNAs and the least characterized. piRNAs have been described to have important roles in transposon silencing, epigenetics, and posttranscriptional regulation of gene expression [81]. They differ from siRNAs and miRNAs because their biogenesis does not involve the transcription of long dsRNA and does not require Dicer activity and they interact with the specific clade of AGO proteins: the PIWI proteins [PIWI, Aubergine (Aub), and AGO3] [74]. Also, unlike siRNAs and miRNAs, which are expressed in most cells and tissue types, piRNAs are preferentially expressed in germline cells, as observed in mammals, zebrafish, and *D. melanogaster* [12,13].

piRNAs were originally described during studies aimed at clarifying how the tandemly repeated *Stellate* (*Ste*) gene was silenced in the *Drosophila* male germline. This is a critical event for spermatogenesis progression and its failing causes male infertility. The authors observed that *Ste* silencing depended on a Y-chromosomal locus (designated Suppressor of Stellate) that has been shown to contain tandem repeats with complementary sequences to the *Ste* gene [95–97]. The further cloning and characterization of abundant piRNAs in the *Drosophila* male and female germline revealed that they include a previously discovered class of small ncRNAs called repeat-associated siRNAs [11] and are derived largely from retrotransposons and other repetitive sequence elements that thus could not be assigned to specific chromosomal loci [98–100]. However some piRNAs can also map to specific genomic loci such as in *Drosophila* and mammals where they are found as clusters in large pericentromeric or subtelomeric domains, generally spanning from 50 to 100 kb and enriched in numerous transposable DNA elements and their fragments [100]. Other piRNAs are transcribed from 3'-UTRs of protein-coding genes and dispersed euchromatin copies of transposable elements [100,101].

The main function of the PIWI-piRNA pathway in the germline has been suggested to be the transposon suppression. This not only because piRNAs map largely to transposable elements but also because PIWI loss-of-function mutations or depletion lead to drastically increased levels of transposons that are accompanied by severe infertility phenotypes [95,102–107]. In addition, mutations in *Drosophila* piRNA biogenesis factors result in disruption of transposon silence, causing DNA breaks to accumulate during female germline development and leading to defects in the posterior and dorsoventral axis [100,108–110]. However further studies suggested that observed developmental defects are indeed a consequence of DNA damage resulting from transposon derepression [111]. These data support the idea that piRNAs are required for the integrity of germline cell genomes protecting them from damaging transposons. Strikingly, if new transposons are actively introduced into piRNA clusters, and if they are hereditable by

progeny, novel piRNAs (able to control the new transposons) will be produced showing that mechanisms that drive adaptation to transposon invasion may be mediated by the piRNAs pathway [100,112]. Other studies showed that a subset of piRNA genes is implicated in the telomere protection complex assembly [111].

The somatic functions of PIWI have long emerged since their discovery. PIWIs are expressed in diverse cell types ranging from naive pluripotent stem cells to differentiated somatic cells, with most somatic expression related to various totipotent and pluripotent stem cells [113]. The expression of PIWIs in mammalian somatic tissues implies the existence of somatic piRNAs. In fact, the analysis of high-throughput small RNA sequencing data in *Drosophila*, mouse, and rhesus macaque samples showed that piRNAs are abundant in other tissues such as in the germline [22].

The piRNA biogenesis pathway is still poorly understood, although it is well accepted that there are two different mechanisms. In the first case, a single-stranded RNA precursor is processed in pre-piRNAs (primary piRNA biogenesis) that are preferentially cleaved at U residues. Those pre-piRNAs are then loaded onto PIWI proteins and are subsequently 3' trimmed and methylated [100,114]. Therefore they commonly have a uridine at the 5' end and a 5'-monophosphate. At the 3' end, piRNAs have a 2'-O-methyl nucleotide modification, a feature that they share with the plant miRNAs but not with mammalian miRNAs [81,115]. This process finally originates mature piRNA-induced silencing complexes (piRISCs) [100,114,116,117]. In the second case, piRISCs that possess an active Slicer activity can trigger secondary piRNA biogenesis. In the presence of corresponding sense and antisense precursor RNAs, PIWI and/or Aub directs the slicing of sense strand transposon transcripts [100,118]. The sliced sense strands are then bound by AGO3, and this complex directs the slicing of antisense transposon transcripts [100,114,119], generating its 5' end (ping-pong amplification cycle) [100,118]. The two piRNAs engaged in ping-pong have opposite orientations and exhibit a characteristic 10nt 5' overlap (ping-pong signature) [100,118].

The mechanism that allows the production of piRNAs from a single-strand precursor is still a matter of debate [120]. In fact, until recently there were no clues about how precursor piRNA transcripts were selected from the global pool of transcripts to be driven to the subsequent steps of the biogenesis pathways. However, the discovery that the *Drosophila* protein Rhino, a homologue of the heterochromatin protein 1 (HP1), generally accepted to be involved in transcriptional silencing [121], binds to many piRNA clusters and is required for piRNA transcription [112], brought new insights into the transport of precursor piRNA transcripts. In *S. pombe* HP1 binds to H3K9me3 histones in heterochromatin and is also an RNA-binding protein that is able to accept newly synthesized transcripts from its bound loci and to transport them to sites where they will be degraded [122]. Therefore, it was proposed that this protein will be an excellent candidate to route precursor piRNA transcripts for subsequent processing steps occurring at specific cellular sites such as an electron-dense granular material found between the nucleus and the cytoplasm at the nuclear pores of germ cell nuclei (nuage) [123,124].

Supporting this idea is the fact that Rhino was found to colocalize with the DEAD box protein UAP56 at nuclear speckles bound to piRNA precursors.

The next step of piRNAs biogenesis is the 5' end formation. Although some controversy surrounds the protein Zucchini (Zuc) concerning its enzymatic activity, resolution of the crystal structure of mouse and *Drosophila* Zuc clearly supports the nuclease activity of the protein [125–127]. Moreover, Zuc is able to catalyze the hydrolysis of single-stranded DNA and RNA *in vitro* [126,127]. These facts, together with the observation that Zuc mutants induce the up-regulation of two telomere-specific transposable elements (Het-A and Tart) and the expression of Stellate protein in the *Drosophila* germline, events associated with the inability to produce piRNAs [128], strongly suggest that Zuc is a key protein for the cleavage of the single-stranded RNA precursor in the piRNA pathway. Consequently, Zuc is a strong candidate to be the 5' nuclease involved in primary piRNA biogenesis. However, proof of this involvement requires additional experimental support because *in vitro* Zuc does not seem to specifically generate fragments with a 5'-U base, which is the feature of primary piRNAs, and also mice depleted in Zuc still have low amounts of processed piRNAs [120].

The following step of piRNAs biogenesis requires the loading of piRNAs onto PIWI (PIWI, Aub, and AGO3) proteins just before the generation of piRNAs mature 3' end. Detailed sequence analysis of the piRNAs associated with the PIWI subfamily [100,118] in the *Drosophila* female germline showed that the most abundant piRNAs are mainly generated from the antisense strand of retrotransposons, and these preferentially associate with PIWI and Aub proteins [100,118]. On the other hand, those present in the single major somatic cluster are mainly originated from the sense strand and are associated with AGO3. Similar to what was observed for the loading of siRNAs into AGO proteins, which requires the chaperone system Hsp90-Hsp70, it was shown in Drosophila that loading of piRNAs onto PIWI proteins also involves the evolutionarily conserved chaperone Shutdown (Shu) protein [129]. The Shu chaperone belongs to the immunophilin class, harbors a peptidyl-prolyl-cis/trans-isomerase domain (PPIase) and a TPR domain [130], and collaborates with Hsp90 at the loading step of piRNAs onto PIWI proteins. Finally the 3' end of piRNAs should be trimmed but the responsible enzyme was not yet identified. piRNAs are heterogeneous in length and show greater variability at the 3' end than at the 5' end, which suggests the activity of a 3-5' exonuclease. As mentioned, 3' maturation requires methylation that seems to protect piRNAs against nontemplated uridylation and destabilization [120]. Hen1 is also the methylase that adds a methyl group to the piRNA at the 2'OH position of the 3' end base [82]. This methylation seems to be a common feature to sncRNAs that base pair to target RNA molecules, i.e., miRNAs and siRNAs [131,132].

The role of PIWI–piRNAs pathways has been extended to numerous somatic functions. Indeed this pathway is important not only for transposon silencing but also genome rearrangement and epigenetic programming during development, with biological roles in stem cell function in regeneration, memory, and possibly cancer [113,133]. In *Drosophila* development the role of piRNAs is clear during maternal-to-zygotic transition, because it was shown that the expression of a piRNA cluster was required [134] to promote degradation of maternal mRNAs such as the embryonic posterior morphogen Nanos (Nos). piRNAs and their associated proteins are involved in recruiting the CCR4 deadenylation complex to specific Nos mRNAs, thus promoting their decay. If Nos mRNA is not degraded, head development defects are observed in the *Drosophila* embryo. Because the expressed piRNA cluster is transcribed from transposable elements, this has led to the suggestion of a close relationship between transposable elements and the regulation of gene expression during development [134]. Another example of the participation of piRNAs in development was revealed by deep sequencing of extracted RNA from the mouse hippocampus. piRNAs are necessary for spine morphogenesis because antisense suppression of this piRNA in hippocampus causes a decrease in the dendrite spine area [135].

Growing evidence also indicates that PIWI and piRNAs have a crucial part in epigenetic regulation through alterations in methylation patterns. Interestingly, in Drosophila ovarian somatic cells piRNAs mediate the inhibition of transcription of hundreds of transposon copies by promoting methylation of H3K9 on transposons and their genomic surroundings [136]. Moreover, in *Drosophila* the somatic silencing of the transposable elements depends on maternally transmitted piRNAs from the *tirant* retrotransposon locus, which correlates with an increase in histone H3K9 trimethylation of the active tirant element [137]. Also, in the Drosophila early embryo it was observed that piRNAs were implicated in heterochromatin formation and that this was necessary for heterochromatin establishment in nongonadal somatic cells, with an impact in the phenotype of the adult [138]. In addition, in Aplysia sensory neurons, a PIWI-piRNA complex is involved, in a serotonin-dependent way, in repression of the transcriptional repressor of memory CREB2, because it is involved in the methylation of a conserved CpG island in the CREB2 promoter [139]. Therefore, the role of the PIWI-piRNA pathway is extended to modifications in the gene expression patterns of neurons involved in long-term memory storage. The complexity of the regulatory networks involving the PIWI-piRNA pathway is also evident because they also enhance transcriptional activation [140]. Participation of the PIWI-piRNA pathway in *de novo* methylation is not restricted to invertebrates, because it was also found in mammalian genomes, as in the case of the differentially methylated region of the imprinted mouse *Rasgrf1* locus [141].

PIWI-piRNA pathways were also shown to have an important role in stem cell biology. Planarians are flatworms able to regenerate the whole body from a small part because they possess a population of stem cells called neoblasts that have the ability to originate all somatic and germ cells [142,143]. Remarkably, in planaria depletion of smedwi-2 and smedwi-3 proteins, which are members of the PIWI/AGO family, cause a decrease in piRNA levels and impair regeneration [144,145]. These results indicate that smedwi-2 and -3 are required for proper function of neoblasts, e.g., in replacing aged tissue.

PIWI proteins have also been implicated in regeneration processes. It was shown that the colonial ascidian tunicate *Botrylloides leachi* (sea squirt urochordate) can regenerate functional adults from microscopic vasculature fragments. This regeneration seems to rely on cells that contour the internal vasculature epithelium of blood vessels after an insult and express *de novo* PIWI [146,147]. After PIWI expression these cells are activated, change morphology, and migrate to places of the vasculature lumen, where they proliferate and differentiate, regenerating a functional organism. If the PIWI synthesis is inhibited, these cells are unable to proliferate or differentiate and subsequent regeneration does not occur [146,147].

There is also growing evidence associating the expression of PIWI with cancer cells. For example, in *Drosophila* it was shown that mutations in PIWI delay tumor growth in a sensitized *lethal* [3] malignant brain tumor [l(3)mbt]-mutant background [148]. In addition, it was observed that other genes encoding members of the piRNA pathways are induced in l(3)mbt tumors. It was also suggested that PIWI is expressed in response to an increase in transposon activity in cancer cells, but additional experiments are required to clarify these issues. However, it is possible that, with better understanding of the role of the PIWI-piRNA pathway in cancer biology, new avenues may be open to cancer therapy.

The role of PIWIs and piRNAs in the development of evolution has been mentioned in this chapter. In fact, during evolution, piRNAs may have been important components of a mechanism that guaranteed correct genomic rearrangements during macronucleus development (the nuclei that is transcriptionally active and the somatic line in ciliated ancestral unicellular organisms), but they might also act as capacitors with a critical role in regulating phenotypic variation and morphologic evolution by controlling how and when genetic variation is expressed. Together they seem to be involved in maintaining balance between the generation of "genomic variability" and "genomic stability," which is also linked to integrity. Therefore, they may have been critical for the sustained evolution of genomes allowing the establishment of complexity integrating environmental information. In this view, it is tempting to speculate that these two features may also be related to their involvement in stem cell maintenance and self-renewal, and hypothetically in cancer regulation.

## 10.2.3 Long Noncoding RNAs

Through the extensive analysis of data obtained from complementary DNA libraries and RNA sequencing data by high-throughput transcriptome projects, it became clear that the mammalian genome produces a large transcriptome of lncRNA (defined as ncRNA > 200 nt in length). However lncRNA's biological importance is just emerging because only a small percentage of lncRNA has been functionally characterized [7,149]. Growing interest in its biological role has been strongly stimulated by evidence that connects them to several human diseases [45,46,48,49].

Although a comprehensive classification of these molecules is still missing and some effort has been made to establish comprehensive conceptual guidelines [150], lncRNA's

transcriptional origins are commonly used for this purpose. Therefore lncRNA has been designated as promoter-associated long RNAs [25], NATs or opposite-strand transcripts [151], long intergenic ncRNA (lincRNA) [152], and enhancer associated RNAs (eRNAs) [153,154]. This criterion sometimes creates noise because often one lncRNA can be classified into more than one class. For example, intergenic RNAs can also be found in introns contributing to a significant fraction of ncRNAs in mammalian cells, and therefore the term "lincRNA" is inadequate [155].

The international FANTOM Consortium demonstrated that at least 25–40% of mammalian protein-coding genes display overlapping transcription [156,157]. In fact, lncRNAs can be transcribed as RNA molecules that display a primary nucleotide sequence that is total or partially complementary, and then able to pair to RNA transcripts with known function; therefore, they are designated NATs [158].

NATs can be designated as *cis*-NATs in cases in which they are originated by transcription of the antisense strand of a given genomic locus. The *cis*-NAT sequence will be complementary to that of the sense transcript of the same locus. Alternatively, they can be transcribed from different genomic loci and may have only partially complementarity with the sense counterpart designated as *trans*-NATs [159]. Thus, in the same cell a pair of sense-antisense transcripts of a genomic locus can be found, and they may have different degrees of overlapping. There are examples in which these pairs of senseantisense RNAs overlap by their 5' regions (5' to 5'), by their 3' regions (3' to 3'), or do not overlap or fully overlap (in this last case, we have two overlap genes) [26].

An extensive analysis of human lncRNAs produced by the GENCODE consortium, composed of 9277 manually annotated genes producing 14,880 transcripts [19], revealed a set of features of lncRNAs: (1) they seem to be synthesized and processed by pathways resembling those of genes encoding proteins; (2) they tend to be transcripts that possess two exons; (3) they are expressed at levels in general lower than those of genes encoding proteins presenting patterns of expression that are more tissue-specific; (4) they preferentially localize at the chromatin and nucleus although there are cytoplasmic pools; and (5) some seem to be able to be processed into small RNAs. Sequence analysis of lncRNA suggests that they do not possess open-reading frames with better features than those found at random [19].

Moreover, it seems that compared with gene-encoding proteins, lncRNAs are subjected to a lower selective pressure but higher than neutrally evolving sequences (ancestral repeat sequences). The promoter region of lncRNAs is the region under a higher selective pressure with levels of conservation equivalent to the promoter region of gene-encoding proteins [19,152,153,160,161]. This may reflect the expression specificity that these molecules seem to possess; on the other hand, it may be the operative mechanism used by these lncRNAs. It is conceivable that their function may be conserved although their molecular mechanism of action does not require extensive sequence conservation.

The roles of lncRNAs have been ascribed to their ability to interact with proteins and protein complexes originating ribonucleoprotein complexes that participate in a variety of different steps of gene expression regulation [18,20], and to establish base pair interactions with DNA and RNA that can mediate the role of the referred ribonucleoprotein complexes. These abilities are shared with small ncRNAs, as for example miRNAs and small nucleolar RNAs and other ribonucleoprotein complexes. Nevertheless lncRNAs have striking distinctive features because they seem to be able to adopt complex secondary structures and other high-order structures that create a new level of information for interaction and recognition, and which cannot be found among sncRNAs [7,18,20]. These assumptions have led to the proposal of regulatory models in which, by the ability of changing their secondary structure, lncRNAs will be sensors of signals. The lncRNA's structure changes will affect its ability to be recognized and interact with a partner or to be processed. In both cases, structure changes will allow the fast and focused transduction of information [42]. Therefore, attention should be paid to how secondary structures are conserved throughout interspecies lncRNAs. Interestingly, in these models lncRNAs display mechanisms of action with similarities to those of bacteria riboswitches [37,38].

In this scenario it has been proposed that because of their ability to pair with nucleic acids, lncRNAs could trap specific sites of transcription factors, facilitate interactions between distinct proteins by acting as scaffolds, recruit and guide proteins (i.e., those involved in chromatin remodeling), etc. [18,20,42,162].

The documented diversity of roles of lncRNAs in gene expression regulation will be analyzed and discussed later this chapter.

## 10.3 MicroRNAs: Key Posttranscriptional Regulators of Gene Expression

Since their initial discovery as regulators of larval development in nematodes [163], miRNAs have been increasingly involved in regulating important cellular processes of multicellular eukaryotes, including cell proliferation, development, differentiation, apoptosis, and oncogenic transformation [164].

miRNAs mediate the repression of target gene expression, acting as sequencespecificity guides for the RNAi machinery. They are estimated to affect most mammalian coding genes [165]. The growing list is curated in the miRBase, which currently accounts for approximately 1881 miRNA species in the human genome and 1193 in the mouse genome [63] that regulate 60% of all protein-coding genes [81]. Because most mRNA targets contain multiple miRNA binding sites, each miRNA can regulate multiple genes. As such, deregulated miRNA levels can perturb the expression of many genes and may be responsible for the onset of several diseases, some of which are addressed in this chapter.

#### 10.3.1 MicroRNA-Mediated Regulation of Gene Expression

As mentioned, it has become increasingly clear that the major mechanism by which miRNAs function is at the mRNA transcriptional level rather than the translational level [89].

mRNA decay is the most frequent miRNA-mediated transcriptional inhibition mechanism used by miRNAs to inhibit target gene expression, but they can also promote direct mRNA cleavage and chromatin reorganization, resulting in decreased mRNA levels. miRNA-mediated mRNA decay can occur via deadenylation, decapping or 5' to 3' degradation of the mRNA [166]. mRNA cleavage is a rare miRNA-mediated repressive mechanism in animals but it is frequent in plants and normally occurs when there is full complementarity between miRNA and its mRNA target [167]. miRNAs also have the capacity to reorganize chromatin by increasing methylation of the targeted mRNA promoters, thereby inhibiting their expression [168]. Interestingly, the repressed mRNAs, AGO proteins, and miRNAs frequently accumulate in processing bodies, which are cytoplasmic structures enriched in the mRNA degradation machinery but in which the translational machinery is normally absent [169].

The second mechanism of miRNA-mediated activity includes repression of translation initiation and/or elongation, premature termination, and nascent polypeptide degradation. Endogenous let-7 micro-ribonucleoproteins (miRNPs) or the tethering of AGO proteins to reporter mRNAs in human cells inhibit m(7)G-cap-dependent translation initiation, which suggests that miRNPs interfere with the recognition of the cap [170]. In addition miRNA-repressed mRNAs contain 40S but not 60S components, which indicates that miRNAs repress translation initiation by preventing the 60S subunit from joining miRNA-targeted mRNAs [171]. The fact that repressed mRNA targets seem to be associated with polyribosomes points to the fact that miRNAs can also repress translation at the elongation step [172–174]. Silencing by miRNAs can also occur before completion of the nascent polypeptide chain, causing a decrease in translational read through at a stop codon, with ribosomes on repressed mRNAs dissociating more rapidly after a block of initiation of translation than those of control mRNAs [174]. These observations suggest a role for miRNAs in ribosome drop-off-mediated repression.

Curiously, some studies provide exceptional evidence for miRNA-mediated transcriptional [175] and translational [176] activation. For example, miRNA-373 was shown to induce expression of genes with complementary promoter sequences [175], whereas miRNA-10a can bind to the 5'-UTR of ribosomal protein mRNAs and enhance their translation [177]. Furthermore, several studies have demonstrated that miRNAs and their associated complexes (microRNPs) can elicit both stimulation of gene expression and the typical repressive roles [178,179].

#### 10.3.2 Impact of MicroRNAs in Immune Cell Development and Functions

The immune system is one of the best, if not the best, characterized developmental systems in mammals. Briefly, this system is generated from hematopoietic stem cells (HSCs) through a series of ordered events of lineage commitment, differentiation, proliferation, and cell migration. This includes developmental programs of ordered T-cell receptor (TCR) and immunoglobulin gene segments that are included on each lymphocyte to form a single antigen receptor (TCR in the case of T cells and BCR in the case of B cells). It also includes differentiation of the cells into distinct cellular subsets with distinct effector functions, and innate and adaptive responses to antigens. Cell interactions between lymphocytes of different subsets and between lymphocytes and other cells of the hematopoietic system, such as dendritic cells, have a critical role and may be accompanied by the generation of immunologic memory. The surface markers of immune cell subsets and of signaling pathways as well as the transcriptional programs underlying developmental progression, cellular selection, and cell migration are extremely well characterized.

Given the considerable knowledge about the functions of immune cells, it was surprising to find that there was an additional unrecognized layer involved in control of the development and physiology of this system that is mediated by miRNAs.

After an initial demonstration that overexpression of an miRNA in HSCs affected B lymphopoiesis *in vivo* upon HSC transplantation [180], it became apparent from experiments in which Dicer was conditionally inactivated in T or B lymphocytes of the mouse that miRNAs are critical for lymphocyte development and differentiation.

A deficiency in miRNAs was first suggested to affect B-cell development by the conditional deletion of AGO2 in hematopoietic cells, which resulted in partial deficiency in miRNAs and compromised development of B and erythroid cells [181]. When Dicer was deleted in the B-cell lineage from the earliest stage of B-cell development, an almost complete block at the proto pre-B transition resulted, which at least partly resulted from apoptosis of Dicer-deficient pre-B cells [182].

In the case of T-cell development, deletion of Dicer in immature thymocytes led to a 10-fold reduction in total thymocyte numbers, an effect that was not as drastic on the periphery [183]. Also, there were major differences in T-cell differentiation in Dicerdeficient mice, which showed a major failure to repress interferon-gamma (IFN-gamma) expression [184]. Ultimately, disruption of this balance between miRNA expression patterns and cytokine production leads to exacerbated inflammation and autoimmunity. In fact, in mice deficient for Drosha or Dicer specifically in the regulatory T-cell population, there is a loss of regulatory T cell–mediated suppressive activity, which results in the development of fatal autoimmunity [185,186]. Also, the absence of miRNAs throughout the T-cell compartment results in spontaneous inflammatory disease [187].

Whereas these examples clearly show the global importance of miRNAs in the immune system, unique spatial and temporal expression patterns in the hematopoietic lineages suggest multiple roles for miRNA in immune cell differentiation and functions. The importance of individual miRNAs became apparent in 2007, when it was observed that the genetic inactivation of miR-155 affected a large spectrum of immune reactions ranging from cytokine production by T and B cells to antigen presentation by dendritic cells and the germinal center B-cell response [188,189]. Since then, many other miRNAs were implicated in the regulation of immune cell function, including early hematopoietic cell development and myeloid and lymphoid

cell development and function. Individual miRNAs have been shown to repress the expression of HSC-relevant genes and affect the production of hematopoietic progenitor and lineage-positive cells. For example, miR-221 and miR-222 regulate KIT expression, which is thought to affect stem cell homeostasis [190]. miR-223 is a myeloid-specific miRNA that promotes monocyte differentiation by repressing nuclear factor I/A production [191]. It also represses C/EBP $\beta$  and myeloid ELF1-like factor 2C (MEF2C) expression to modulate neutrophil differentiation and proliferation [192]. Expression profiling showed that stimulation of monocytes with lipopolysaccharide (LPS) induced the expression of miR-132, miR-146, and miR-155 [193,194]. miR-146 targets Traf6 and Irak1, components of the Toll-like receptor (TLR) signaling pathway that is activated by LPS, which suggests a negative feedback loop [194].

In T cells, microRNAs were shown to be more relevant during the early thymic differentiation rather than at later differentiation stages and during peripheral homeostasis, which are both largely miRNA-independent. During thymic development, T-cell survival and selection are influenced by the miR-17-92 cluster [which targets mRNAs encoding BCL-2—interacting mediator of cell death (BIM)], and phosphatase and tensin homologue (PTEN), and miR-181a (which targets mRNAs encoding phosphatases such as dual-specificity protein phosphatase 5 (DUSP5), DUSP6, SH2-domain—containing protein tyrosine phosphatase 2, and protein tyrosine phosphatase, nonreceptor type 22 (PTPN22), controlling TCR signaling thresholds in T lymphocytes [195]. On the periphery, mature T-cell differentiation is modulated, among others, by miR-326, which promotes skewing toward TH17 cells by targeting the mRNA encoding ETS1 [196] and miR-155, which promotes skewing toward T-helper 1 (Th1) cells by repressing macrophage-activating factor or Treg survival by repressing the expression of suppressor of cytokine signaling 1 [197].

Taking these findings together, miRNA control has emerged as a general regulatory mechanism in immune development and immune responses. In the next section, we provide updated information on mechanisms employed by miRNAs to regulate immune functions.

## 10.3.3 Mechanisms by Which miRNAs Exert Their Function in the Immune System

Based on their characteristics, such as on the combined protein—miRNA gene expression program, miRNAs are able to confer unique properties to a gene regulatory pathway in immune cells [198,199]. miRNAs are particularly important for regulating cellular responses that are time dependent. In fact miRNAs are thought to act on a shorter time scale than protein transcriptional repressors. For example, in inflammation there is an induction of both miRNAs and transcriptional factors, but because miRNAs do not require translation or translocation back into the nucleus to begin repressing their targets, they become active more quickly [198]. This may indeed be important in innate

responses in which a short time frame is critical to begin the fight against pathogenic organisms. On the other hand, miRNAs may act more slowly than protein factors that could directly target and inactivate miRNA targets [199]. Because miRNAs act at the RNA level, a protein that has been already produced and is active will remain active until miRNA action down-regulates the amount of new protein being produced. For example, miR-21 causes delayed inhibition of programmed cell death 4, a proinflammatory protein, following induction by nuclear factor (NF)-κB [200].

Another important notion is that although the mechanism of action of miRNAs is the same in different cells, they exert distinctive functions in different cell types. This is mainly because each cell type has a specific transcriptional program that limits the number of relevant targets for a given miRNA.

Also relevant is the fact that miRNA inhibition of target mRNAs does not lead to a complete knockdown of the respective protein. Indeed, unlike transcription factors that act as "switches," miRNAs are "fine-tuners" of gene expression, which is of extreme importance for the immune cells that can respond quickly to small changes in gene expression. This "fine-tuning," however, can still be modulated. In fact, distinct cellular concentrations of a given miRNA can dictate a different protein output of its target genes, as demonstrated in a study in which the interaction of miR-150 and its target c-Myb was addressed quantitatively [201]. When miR-150 was ectopically expressed at graded concentrations in lymphocyte progenitors *in vivo*, dose-dependent down-regulation of c-Myb protein levels was observed, with a 35% reduction caused by the highest concentration of the transgenic miRNA. There was a concomitant dose-dependent developmental block at the pro- to pre-B transition, which was proven to be mainly caused by the reduction in c-Myb protein levels [201].

Finally, miRNAs can regulate immune cell responses by modulating negative regulators, controlling distinct nonsynergistic regulatory pathways, or even controlling simultaneously functionally interrelated pathways. For example, by down-regulating several negative regulators of TCR signaling (PTPN22, DUSP5, and DUSP6), miR-181 changes the outcome of signaling triggered by TCR engagement by peptide-major histocompatibility complexes (MHCs): double-positive thymocytes (expressing high levels of miR-181), but not mature T cells, respond to self-antigens and are positively or negatively selected [195]. miR-155 is an example of a miRNA regulating distinct, nonsynergistic pathways, involved in controlling the generation of class-switched B cells and c-myc-immunoglobulin H translocations. miR-155 down-regulates AID, an enzyme essential for class switch recombination, and the transcription factor Pu.1, which counteracts the differentiation of class-switched cells [202]. The net effect, possibly involving additional miR-155 targets, is the promotion of this process. As for c-mvcimmunoglobulin H translocations, miR-155 down-regulates AID and other target genes, which synergize with AID in mediating these events [203]. A final example is that of the miR-17-92 cluster, which regulates components of functionally interrelated pathways by targeting PTEN (an inhibitor of the PI3K pathway, which promotes cell cycle progression), BIM (a proapoptotic factor), p21 (an inhibitor of the G1/S cell-cycle transition),

and Rbl2 (a member of the Rb family of pocket proteins that negatively regulate cell cycle), and E2F1, E2F2, and E2F3. The final outcome is cell proliferation, which correlates with the fact that miR-17-92 is highly expressed in proliferating progenitor cells during lymphocyte development, decreasing two- to threefold upon maturation [204,205].

Although there is still a lot to uncover regarding miRNA-mediated regulation, these small noncoding molecules are unequivocal key players in immune cell regulation.

## 10.3.4 MicroRNA-Mediated Regulation of Cellular Redox State

Cellular redox state is another important process regulated by miRNAs and its misregulation is frequently associated with immune-mediated responses such as inflammation, as further explored next.

Cells form reactive oxygen species (ROS) such as superoxide radical  $(O_2^-)$  or hydrogen peroxide  $(H_2O_2)$  and reactive nitrogen species such as nitric oxide (NO) or peroxynitrite  $(ONOO^-)$ . There is increasing evidence that ROS such as  $H_2O_2$  have a key role in the redox regulation of protein activity and signaling pathways through the use of protein thiols and disulfides as functional switches. Therefore, cells keep a tight control of ROS steady-state levels through antioxidants to prevent molecular oxidative damages and disruption of redox signaling and control, i.e., oxidative stress.

Work from several laboratories has shown that miRNAs can modulate the intracellular and extracellular redox state. In particular, the expression of antioxidant enzymes such as catalase and superoxide dismutase (SOD) and also nitric oxide synthase, which are responsible for the synthesis of NO, is controlled by miRs (Fig. 10.1).

In human retinal pigment epithelial cells (ARPE-19), a sublethal  $H_2O_2$  concentration (200  $\mu$ M) up-regulates the expression of miR-30b, a member of the miR-30 family, and leads to a decrease in catalase expression at both the transcript and protein levels. miR-30b antisense (antagomiR) not only suppressed the miR-30b mimics-mediated inhibition of catalase gene expression but also dramatically increased the expression of catalase, confirming that miR-30b regulates catalase expression [206]. Suppression of miR-551b also increased catalase expression and decreased ROS levels in a dose-dependent manner in human lung cancer cell lines with apoptosis resistance (A549-AR and H460-AR cells) that express low levels of catalase compared with wild-type cells [207]. Because inhibition of miR-551b in A549-AR cells increased catalase 3'-UTR luciferase reporter activity and the reporter activity was higher in A549 WT cells that express lower levels of miR-551b, miR-551b probably binds directly to catalase 3'-UTR to suppress catalase expression.

There is also evidence that miRs regulate SOD. In fact, cells overexpressing miR-21 have unchanged SOD1 levels but reduced SOD2 and SOD3 levels [208,209]. The 3'-UTR of SOD3, the extracellular SOD, represents an authentic miR-21 binding site and SOD3 is a direct target of miR-21. However, SOD2 expression is not regulated directly by miR-21 but indirectly through tumor necrosis factor- $\alpha$  (TNF- $\alpha$ ). TNF- $\alpha$  is a direct target



**FIGURE 10.1** Regulation of cellular redox state by microRNAs (miRNAs). Control of the redox state of the cell through regulation of major antioxidant enzymes catalase, superoxide dismutase (SOD), and the nitric oxide synthase responsible for the synthesis of nitric oxide. The main cellular roles and physiopathologic states that are mediated by these miRNAs are also indicated. Data were collected from http://www.ncbi.nlm.nih.gov/gene/ID (miR21, ID: 406,991; miR30b, ID: 407,030; miR-155, ID:406,947; miR-221, ID:407,006; and miR-222, ID:407,007).

of miR-21 and its levels are decreased in cells overexpressing miR-21. A consequence of miR-21–induced changes in SOD2 and SOD3 expression levels is that overexpression of miR-21 in the human bronchial epithelial cell line NL20 leads to increased superoxide  $(O_2^-)$  levels and reduced  $H_2O_2$  levels. A study also showed that in primary cortical neurons overexpression of miR-424 decreases cellular injury induced by 200  $\mu$ M  $H_2O_2$  for 1 h through a mechanism involving increased SOD2 activity [210].

A role for miRNAs in regulating endothelial nitric oxide synthase (eNOS) has long been suspected since the knocking down of Dicer, increased eNOS expression [211]. In addition, miR-221 and miR-222 mimics partially reverse the increase in eNOS elicited by Dicer down-regulation, although neither of these miRNAs directly targets the eNOS mRNA. It was shown that miR-155 directly targets eNOS mRNA by binding to its 3'-UTR to decrease enzyme expression and NO production [212]. Proinflammatory cytokines such as TNF- $\alpha$ , interleukin (IL)-1, interferon-gamma, and bacterial LPS increase miR-155 expression in human endothelial cells, most probably via AP-1 and NF- $\kappa$ B [212]. Conversely, inhibition of miR-155 partially prevents the TNF- $\alpha$ -induced decrease in eNOS expression [212]. It was also shown that miRNA-24 overexpression in human umbilical vein endothelial cells regulates eNOS expression by leading to the inhibition of its expression [213].

Whereas miRNAs function as regulators of stress-related factors, the reverse is also true, i.e., stress conditions can regulate miRNA expression. Indeed, together with inflammatory signals, these are part of a growing list of factors that modulate miRNA expression, as detailed in the next section.

### 10.3.5 Modulation of MicroRNA Expression in Immune Cells and During Cellular Stress Responses

miRNAs are mostly regulated during their biogenesis at the transcription and processing levels, and there is also evidence that some of these processes are influenced by immune challenge, inflammation, and other forms of cellular stress.

The expression of miRNAs is initially controlled at the transcriptional level by transcription factors that act in specific cell types during development or in response to environmental challenges, as further detailed in Sections 10.3.5.1-10.3.5.3.

The next layer of regulation is at the posttranscriptional level, affecting processing and including various stages of regulation from the initial primary transcript to the delivery of mature, single-stranded miRNAs to their target mRNAs. Regulators of miRNA processing bind to the stem or loop regions of miRNA precursors and influence their processing by Drosha and/or Dicer. Adenosine deaminase acting on RNA is an example of an RNA-modifying enzyme that is up-regulated during inflammation and can introduce mutations in double-stranded miRNA precursors, thereby changing the targeting specificity of the miRNA [214]. After being transported to the cytoplasm, some pre-miRNAs are posttranscriptionally modified by terminal uridyl transferase 4 (TUT4), which mediates uridylation of pre-let-7 miRNA upon recruitment of Lin28 [215,216]. This modification prevents maturation of pre-let-7 miRNA, thus inhibiting its function.

In addition to transcriptional and posttranscriptional regulation of miRNAs, there is evidence suggesting that under some conditions, such as after the onset of cellular stress, miRNAs were associated with stress granules [217]. This suggests that miRNAs can also be regulated by subcellular localization. However, further investigation is required to understand the role of these organelles in miRNA biology.

Finally, epigenetic mechanisms also seem to act on miRNA expression. A genomic analysis of miRNA sequences has revealed that CpG islands are present both upstream and downstream of miRNAs coding sequences. A significant portion of these CpG islands (40%) was shown to overlap predicted transcription start sites [218]. In addition, DNA hypermethylation and histone modifications of miRNA genomic regions have been associated with altered miRNA activities with a potential oncogenic role [219], which opens an unexplored area regarding the epigenetic regulation of miRNA gene expression in normal tissues, tumor development, and progression.

#### 10.3.5.1 Immune Signals Influencing MicroRNA Expression

The transcription of certain miRNAs found in immune cells, called immuno-miRs [220], was shown to be modulated in response to inflammatory stimuli and proinflammatory cytokines. For example, miR-155 and miR-146a transcription is up-regulated in response to inflammatory stimuli such as TLR ligands or proinflammatory cytokines [193,194]. TGF- $\beta$  is a potent immunosuppressive cytokine implicated in the inhibition of T-cell activation and the down-regulation of proinflammatory cytokines. It was shown to specifically affect the expression of several miRNAs which themselves have immuno-modulatory functions [221–223]. TGF- $\beta$  promotes the induction of miR-155 expression levels in human intestinal CD4<sup>+</sup> lamina propria T cells, which in turn targets Itk (a key signaling molecule downstream of the TCR), leading to decreased expression of IL-2 and impaired T-cell activation [223]. In human CD8<sup>+</sup> T cells, TGF- $\beta$  was shown to induce the three members of miR-23a cluster (miR-23a, -27a, and -24), which exhibited immuno-modulatory functions [222]. miR-27a and miR-24 targeted the proinflammatory cytokine IFN-gamma, whereas miR-23a targeted CD107a, an essential component of the cytotoxic granule machinery [222].

In addition to these specific roles, TGF- $\beta$  exerts pleiotropic effects on the miRNome because its downstream signaling effector proteins, the Smad protein family, have been shown to have an impact on the generation of miRNAs at the pri- to pre-miRNA processing step performed by the Drosha microprocessor complex [224,225]. In fact, Smad proteins activated by TGF- $\beta$  signaling bind to a consensus sequence found within the stem region of numerous pri-miRNAs. Both Smads and p53 associate with Drosha via p68 and increase the processing of their specific target pri-miRs [224,225]. KH-type splicing regulatory protein binds sequences in the loop region of several miRNA precursors and recruits Drosha and Dicer to the pri- and pre-miRNA, respectively, enhancing their processing [226]. In contrast, Lin28 represses both Drosha and Dicer processing of most let-7 family members by binding conserved sequences in their precursor loops. This mechanism involves Lin28 recruitment of TUT4 to pre-let-7, causing it to be oligo-uridylated and degraded [227].

Cytokines and immune factors other than TGF- $\beta$  were shown to influence miRNA expression. Interleukin-2, primarily produced by antigen-activated T cells, was actually the first cytokine shown to have an effect on the miRNA repertoire. Together with TCR activation induced miR-182 expression, targeting the transcription factor FOXO-1, a negative regulator of T-cell proliferation [228].

These examples highlight an important layer of immune-mediated miRNA regulation and the transcriptional networks associated with it. In the next sections we will focus on stress-related factors that also affect the different stages of miRNA biogenesis.

#### 10.3.5.2 Modulation of MicroRNA Biogenesis by Hydrogen Peroxide

As described earlier (Section 10.3.4), miRNAs can regulate the cellular redox state, but the reverse is also true: stress conditions can affect miRNA levels. In this section we will explore how stress factors, namely hydrogen peroxide ( $H_2O_2$ ), regulate miRNA biogenesis.

Why focus our attention in  $H_2O_2$ ? The biological effects of ROS depend on their concentration, with high levels being responsible for toxic effects, whereas low to moderate levels mediate important molecular pathways. Among ROS,  $H_2O_2$  is one of the most abundant species whose biological effects have been characterized. The specificity of  $H_2O_2$  biological effects results from activation of sensor proteins with diverse reactivity toward  $H_2O_2$  and from localized production of  $H_2O_2$  that activates signaling pathways in localized cellular spots. Its targets include signaling enzymes such as phosphatases [229], kinases [230], and transcription factors [231] and, as reviewed here, ncRNAs.

Biogenesis of miRNA is down-regulated by  $H_2O_2$  by several potential mechanisms. As described in Section 10.2.1.1, pri-miRNAs are cleaved by Drosha to generate pre-miRNAs in the nucleus [232]. Phosphorylation of Drosha at Ser300 or Ser302 by glycogen synthase kinase-3 $\beta$  (GSK-3 $\beta$ ) is required for Drosha nuclear localization [233,234]. GSK-3 $\beta$  is inhibited by phosphorylation of Ser9 catalyzed by AKT kinase [235], and so by activating the PI3K–AKT pathway  $H_2O_2$  indirectly inhibits GSK-3 $\beta$  [236,237]. Thus, we may speculate that  $H_2O_2$  may inhibit the nuclear localization of Drosha (Fig. 10.2).

In addition, DiGeorge critical region-8 (DGCR8), which forms a complex with Drosha to cleave primary miRNA, may also indirectly be modulated by H<sub>2</sub>O<sub>2</sub>, because the binding of ferric heme enhances its activity [238]. H<sub>2</sub>O<sub>2</sub> may down-regulate the cellular level of ferric heme by (1) activating the gene expression of heme oxygenase 1 [239], the enzyme responsible for converting heme to biliverdin, carbon monoxide, and ferrous ions, and by (2) inhibiting the first enzyme of the heme biosynthetic pathway,  $\delta$ -aminolevulinate synthase-2 (Fig. 10.2), with an inhibition constant measured *in vitro* of  $K_i = 5 \,\mu$ M [240]. Heme oxygenase 1 is under the control of NRF2, a redox-sensitive transcription factor that is activated by H<sub>2</sub>O<sub>2</sub> [241]; such control may have a significant impact on miRNA abundance because heme oxygenase modulates miRNA by a multitude of mechanisms other than heme availability [242].

After their formation by the Drosha–DGCR8 complex, pre-miRNAs are subsequently exported into the cytoplasm where they are further processed by Dicer [232], which is down-regulated by H<sub>2</sub>O<sub>2</sub>. In primary cerebromicrovascular endothelial cells derived from young rats (3 months old), treatment with 0.1  $\mu$ M H<sub>2</sub>O<sub>2</sub> decreased Dicer mRNA expression by 40% after 24 h [243]. Interestingly there was a decrease in Dicer mRNA with age that was reversed by overexpressing catalase [243]. In 3T3-F442A preadipocytes, rapid decreases of Dicer mRNA (90 min) were observed for higher H<sub>2</sub>O<sub>2</sub> concentrations. However, delivery of H<sub>2</sub>O<sub>2</sub> as continuous production by glucose oxidase yielded biphasic effects, because increased Dicer mRNA levels were observed for lower H<sub>2</sub>O<sub>2</sub> production rates, whereas higher rates of H<sub>2</sub>O<sub>2</sub> production decreased Dicer mRNA levels [244]. Dicer protein levels were nearly abolished 24 h after treatment with 250  $\mu$ M H<sub>2</sub>O<sub>2</sub> for 15 min in human JAR trophoblast cells (Fig. 10.2) [245]. The mechanism underlying H<sub>2</sub>O<sub>2</sub>-mediated Dicer inhibition is unknown, but it has been suggested that is mediated by the let-7 family of microRNAs [246], which are known to mediate suppression of Dicer (Fig. 10.2) [247,248] and are up-regulated by H<sub>2</sub>O<sub>2</sub> [249].



**FIGURE 10.2** Modulation of microRNA (miRNA) biogenesis by  $H_2O_2$ . Biogenesis of miRNA is down-regulated by  $H_2O_2$  at critical steps of the pathway. Drosha nuclear translocation: phosphorylation (P) of Drosha by glycogen synthase kinase 3 $\beta$  (GSK3 $\beta$ ), which is downstream of the PI3K–AKT pathway is regulated by  $H_2O_2$ . Biosynthesis of DiGeorge critical region-8 (DGCR8): binding of ferric heme enhances its activity and  $H_2O_2$  by activating gene expression of heme oxygenase 1 may lead to increased degradation of heme, thus reducing its availability.  $H_2O_2$  inhibits  $\delta$ -aminolevulinate synthase-2, the first enzyme of the heme pathway, which consequently down-regulates heme synthesis. Finally, Dicer, a member of the ribonuclease III family, which creates small RNA duplexes with 2-nucleotide overhangs at their 3' ends and phosphate groups at their 5' ends, is down-regulated by  $H_2O_2$  (see text for references). *Pre-microRNA*, precursor miRNA; *Pri-microRNA*, primary miRNA; *RISC*, RNA-induced silencing complex.

#### 10.3.5.3 Regulation of Specific MicroRNAS by Hydrogen Peroxide

In addition to a role as a general inhibitor of miRNA biogenesis,  $H_2O_2$  is also responsible for regulating particular miRNAs that regulate specific cellular responses (Table 10.1). This suggests that  $H_2O_2$  mediates several steps of miRNA processing and that miRNAs might be important downstream effectors of the redox signaling events. The mechanism by which  $H_2O_2$  regulates the expression of specific miRNA is still largely unknown. Table 10.1 shows the  $H_2O_2$ -regulated miRNA for which the  $H_2O_2$  dose and the recipient cell type are known. These data suggest that cell type,  $H_2O_2$  concentration, and exposure time are key factors in regulating miRNA expression and might account for the differences observed in the  $H_2O_2$ -dependent regulation of particular miRNAs. Nevertheless, some miRNAs such as miRNA-1, miRNA-21, and miRNA-200c, were similarly regulated by  $H_2O_2$  despite the different cellular contexts and  $H_2O_2$  doses.

Concerning miRNA-1, it is involved in myocyte progenitor determination during development [264,265] and is up-regulated in patients with coronary artery disease [266].

			MicroRNA (miRNA)		
miRNA	Target	Hydrogen Peroxide	Effect	Cell Type	References
miRNA-1	Insulin-like growth factor-1 (untranslated region)	Up-regulated 200 μM (16 h)	Proapoptotic	Induced pluripotent cells (induced from normal human foreskin fibroblast, SCRC-1041)	250
		Up-regulated 100—400 μM (6 h)	Proapoptotic (by Akt phosphoinhibition)	Cardiomyoblasts	251
miRNA-145	Aortic expression of contractile proteins (e.g., myocardin)	Up-regulated 300 μM (12 h)		Aortic smooth muscle cells	252
	BCL2/adenovirus E1B 19-KD protein- interacting protein 3	Down-regulated 50 μM (0.5 —8 h)	Antiapoptotic	Cardiomyocytes (neonatal rat ventricle myocytes)	253
miR-466h-5p		Up-regulated 1—5 mM	Proapoptotic role	B/CMBA.Ov (mouse cells)	254
miRNA-30b	Catalase	Up-regulated 200 $\mu$ M	Retinal pigmented epithelial oxidative stress	ARPE-19	206
miRNA-34a	sirt1	Up-regulated 200—500 μM	Proapoptotic role	Rat cardiomyocytes	255
miRNA-302	ARID4a and CCL5	Down-regulation 200 $\mu$ M	Progression to s-phase (cell cycle)	AG01522 normal human fibroblasts	256
miRNA-21	Maspin and programmed cell death 4	Up-regulated depending on p47phox expression	Protumorigenic effect	Prostate cancer (PC-3M-MM2)	257
	?Unknown target	Up-regulation 800 $\mu$ M (1 h)		RAW 264.7 (macrophages)	258
miRNA-21, let-7 family (and others array)	Programmed cell death 4	Up-regulated 25–200 µM (6 h)		Rat aortic smooth muscle cells (primary cultures)	249

## Table 10.1 Hydrogen Peroxide Regulates microRNA Expression

Continued

miRNA	Target	Hydrogen Peroxide	MicroRNA (miRNA) Effect	Cell Type	References
miR-27a, miR27b, miR-29b, miR-24-2,	?, Unknown target	Down-regulation 800 $\mu$ M (1 h)	Proinflammatory	RAW 264.7 (macrophages)	258
miRNA-200c (and others s miR-200 family miRNA- 200a 200b, 141, 429)	ZEB1	Up-regulated 200 μM	Proapoptotic role	Myoblast c2c12 human umbilical vein endothelial cells (HUVEC)	259
miR-9, miR-200c, miR- 708, miR-377, miR-26b, miR-296, miR-369, miR- 32, miR-1956, miR- 1190, miR-135b and miR-201		Up-regulated 200 μM (6 h)		Primary hippocampal neuron cultures from SAMR1 mice	260
mir-291a, miR-190b, miR-297c, miR-713 and miR-470		Down-regulated 200 μM (6 h)		Primary hippocampal neuron cultures from SAMR1 mice	261
miR-126		Down-regulated 50–200 μM	Increase survival	HUVEC line CRL-1730	262
miR-214		Up-regulated 30—100 μM (6 h)	Proapoptotic role	Primary neonatal rat cardiac ventricular myocytes	263
miR-92a	SIRT1, KLF2, and KLF4	Up-regulated	Proinflammatory	Cultured endothelial cells	264

## Table 10.1 Hydrogen Peroxide Regulates microRNA Expression—cont'd

In the presence of  $H_2O_2$  (100–400  $\mu$ M), miRNA-1 is up-regulated in both iPS cells and cardiomyoblasts, promoting apoptosis and decreasing insulin-like growth factor-1 (IGF-1) signaling activation [250]. The promoter region of miR-1 contains two potential binding sites for Foxo3 that regulate the expression of this miR in myocytes (C2C12) and neonatal cardiomyocytes [267]. Foxo3 is a transcription factor that mediates the oxidative stress response and might be involved in  $H_2O_2$ -dependent up-regulation of miR-1. IGF-1 and the IGF-1 receptor, which are key regulators of myocyte function and survival, were identified as targets of miRNA-1, which indicates a potential mechanism for  $H_2O_2$ -induced apoptosis [267,251].

miR-21 is known as an oncomiR because its overexpression in multiple human cancers targets tumor-supressor genes expression [268]. miR-21 is up-regulated in several cell types by extracellular  $H_2O_2$  (25–800  $\mu$ M) or indirectly by NADPH oxidase (NOX) overexpression. Increased expression of miR-21 in prostate cancer cells depends on the high levels of ROS produced by NOX; the activation of Akt is important for this induction [257]. Besides cancer, miR-21 is also involved in cardiac disease because it increases the ERK–MAP kinase activation pathway in cardiac fibroblasts by targeting SPRY1, a potent inhibitor of the Ras–MEK–ERK pathway [55].

miR-200c is up-regulated by  $H_2O_2$  (200  $\mu$ M) in myoblasts (C2C12), endothelial cells (HUVEC), and primary hippocampal neurons [259]. The miR-200 family targets ZEB1 and ZEB2, which are transcriptional repressors of E-cadherin promoting the mesenchymal-to-epithelial transition [269]. In endothelial cells, the down-regulation of ZEB1 owing to oxidative stress also increases apoptosis and senescence [259].

 $H_2O_2$ , among other stress stimuli, was shown to up-regulate miR-92 expression via Sterol regulatory element-binding protein 2n (SREBP2) transactivation in endothelial cells [263]. As a consequence of this activation, the expression of antiinflammatory genes, including SIRT1, KLF2, and KLF4 decreases, resulting in increased inflammasome and impaired eNOS-NO bioavailability [263]. Thus, oxidative stress—induced miR-92a increases endothelial innate immunity and the SREBP2-miR-92a—inflammasome may be a crucial pathway linking oxidative stress, inflammation, and endothelial dysfunction.

ROS, and  $H_2O_2$  in particular, are produced by macrophages for microbial killing but also have an important effect in regulating macrophage activity in inflammation and antigen presentation [270]. The regulation of miR expression is an important mechanism by which  $H_2O_2$  fine-tunes macrophage function [237]. This is the case of miR27b, which is down-regulated in the presence of  $H_2O_2$ , leading to an increase in NF- $\kappa$ B signaling activation [258].

These last examples clearly show that miRNAs can promote cross-talk between oxidative stress and immune functions, which is further explored in the next section.

#### 10.3.6 MicroRNAs in the Crossroad Between Immune Functions and Oxidative Stress

Here we focus on the role of miRNAs as regulators of oxidative stress and immune responses and how misregulation of these mechanisms can cause disease. The correlation between inflammatory and immune cells (such as mast cells, neutrophils, leukocytes, macrophages, monocytes, eosinophils, dendritic cells, phagocytes, and natural killer cells) and oxidative stress is well known. The generation of ROS, including  $H_2O_2$ , occurs in persistently inflamed tissues and also in cardiovascular disease and aging [271]. In extreme cases these can cause DNA damage and contribute to carcinogenesis by activating oncogenes and/or inactivating tumor suppressor genes [272]. The loss-of-function (inactivation) of tumor suppressor genes and the gain-of-function (activation) of oncogenes often cooperate to induce inflammation-driven carcinogenesis [273]. Next we will address the action of specific miRNAs on molecules involved in oxidative stress and immune dysfunction.

#### 10.3.6.1 Toll-Like Receptor Pathways and Reactive Oxygen Species Activation in Inflammatory Responses

TLRs are an evolutionarily conserved receptor family that regulate antimicrobial host defense in plants, invertebrates, and mammals [271]. They recognize bacterial compounds and activate intracellular signaling pathways resulting in inflammatory responses. There is evidence that TLRs such as TLR-4 can mediate ROS activation of NF- $\kappa$ B. It is known that, on the one hand, H<sub>2</sub>O<sub>2</sub> can trigger nuclear translocation of NF- $\kappa$ B, and on the other hand, agents that stimulate NF- $\kappa$ B activation increase the generation of intracellular ROS [274]. In agreement with this, several antioxidants prevent NF- $\kappa$ B nuclear translocation and activation, which suggests that NF- $\kappa$ B activation is induced when intracellular ROS generation is increased (Fig. 10.3).

A study by Asehnoune et al. [275] addressed the mechanisms by which ROS modulate the transcriptional activity of NF-KB in response to TLR-4-dependent signaling. Antioxidant treatment inhibited LPS-stimulated production of inflammatory cytokines and activation of the kinases (IKK-a and IKK-b) and IRAKs in neutrophiles. These results indicated that TRL-4-mediated signaling is oxidant-dependent and that ROS can modulate NF- $\kappa$ B-dependent transcription through their involvement in early TLR-4mediated cellular responses [275]. The specific role of miRNAs in ROS production in the context of TLR activation is not known. However, several miRNAs are involved in the negative regulation of innate immune and inflammatory responses, which target genes involved in TLR pathways (Fig. 10.3) [276,190]. It has been reported that the TLR-4 ligand, LPS, increased the expression of miR-146a, miR-155, and miR-132 in monocytes [194]. Ligands for TLR2, TLR-4, and TLR-5 also induce miR-146a/b expression. Importantly, TRAF6 and IRAK-1, which are the key molecules in TLR-mediated NF- $\kappa$ B activation, are miR-146 targets [194], which suggests that miR-146a may function as a negative feedback regulator of TLR-mediated signaling in innate immune and inflammatory responses (Fig. 10.3).

miR-155 expression can also be stimulated by ligands for TLR3 and TRL4 and also cytokines such as IFN-beta and IFN-gamma via TNF- $\alpha$  autocrine signaling [277]. miR-155 targets FADD, IKK-, and receptor-interacting serine/threonine-protein kinase-1 mRNA [278], which suggests that miR-155 is a component of the innate immune response to inflammatory stimulators that enhances TNF- $\alpha$  translation.



FIGURE 10.3 Complex regulatory role of miRNAs in the network of innate immune and inflammatory responses mediated by tumor necrosis factor receptors (TNFR) and Toll-like receptors (TLR) involving the nuclear factor kB (NF-kB) activation pathways and reactive oxygen species (ROS) signaling. TLR and TNFR can mediate ROS activation of NF-KB. Activation of the TNFR or the TLR-4 receptors induces the assembly of signaling complexes that promote activation of the  $I\kappa\alpha$  and  $I\kappa\beta$  kinase (IKK) complex. These signaling complexes are composed of myeloid differentiation primary response gene 88 (MyD88), interleukin-1 receptor-associated kinase 1 and 4 (IRAK1/4) and TNFR-associated factor 6 (TRAF6) for TLR-4 receptor, and by TNFR type 1-associated DEATH domain protein (TRADD), receptor-interacting serine/threonine-protein kinase 1 (RIP1), and TNFR-associated factors 2 and 5 (TRAF 2/5) for TNFR. Assembly of these complexes results in the phosphorylation (black/white P) and subsequent degradation by the proteasome of  $I\kappa\beta$  and in the NF- $\kappa$ B translocation into the nucleus, where it activates the target genes. The recruitment and activation of the IKK complex depend on polyubiquitination of RIP1 and TRAF6, respectively. These signaling pathways are regulated by miRNAs. The TLR-4 ligand, lipopolysaccharide, increases the expression of miR-146a, miR-155, and miR-132. miR-155 expression can also be stimulated by cytokines such as interferon (IFN)- $\beta$  and IFN- $\gamma$  via TNF- $\alpha$  autocrine signaling. TRAF6 and IRAK-1 are miR-146 targets, which suggests that miR-146 may function as a negative feedback regulator of TLR-mediated signaling in innate immune and inflammatory responses. miR-125b inhibits TNF- $\alpha$  production, whereas NLRP3 is negatively regulated by miR-223. miR-let7 and miR-21 target TLR-4 mRNA at the posttranscriptional level. Many of these miRNAs are also involved in the cellular redox state, such as miR-21, which targets mRNAs of mitochondrial superoxide dismutase (SOD2) and extracellular superoxide dismutase 3 (SOD3), and in turn is up-regulated in several cell types by extracellular H<sub>2</sub>O<sub>2</sub> or indirectly by NADPH oxidase (NOX) overexpression. miR-155 that down-regulates endothelial nitric oxide synthase (NO Synthase). It has been suggested that Dicer inhibition by H<sub>2</sub>O<sub>2</sub> is mediated by the let-7 family of miRNAs that are known to suppress the role of Dicer and are up-regulated by H<sub>2</sub>O<sub>2</sub> (see text for references).

LPS, the TLR-4 ligand, was shown to repress miR-125b expression. Because this miRNA targets TNF- $\alpha$ , its suppression leads to increased production of TNF- $\alpha$  during macrophage inflammatory responses [278]. TLR-4 itself can be a target of another miRNA: miR-let-7. In fact, it was observed that overexpression of miR-let-7 significantly decreased TLR-4 expression, and stimulation with *Cryptosporidium parvum* decreased miR-let-7 in a MyD88/NF- $\kappa$ B -dependent manner, while increasing expression of TLR-4 [279].

Upon NF- $\kappa$ B activation by TLR-mediated ROS induction, several genes are upregulated, including the Nod-like receptor protein NLRP3 and pro-IL-1ß [280]. The protein NLRP3 has emerged as a central regulator in the inflammatory process and is implicated directly in hereditary cryopyrinopathies and indirectly in diseases such as gout, type 2 diabetes, and atherosclerosis. Upon activation, NLRP3 forms a so-called inflammasome complex with the adaptor molecule ASC, which controls the activation of caspase-1, the enzyme that processes the immature form of the proinflammatory cytokine IL-1 $\beta$  into the active protein. An alternative priming mechanism has been put forward that results in the posttranslational activation of NLRP3 by deubiquitination and requires TLR-induced ROS [281]. Importantly, NLRP3 regulation seems to depend on a particular miRNA, miR-223, which has been identified as a negative regulator of NLRP3 [282,283] (Fig. 10.3). Interestingly, miR-223-deficient mice exhibit phenotypes consistent with deregulated NLRP3, such as neutrophilia, spontaneous lung inflammation, and increased susceptibility to endotoxin challenge [284]. ROS can induce TLRs, which in turn activate downstream molecules such as NF- $\kappa$ B or NLRP3, involved in inflammatory processes that can themselves be subject to regulation by miRNAs.

#### 10.3.6.2 Reactive Oxygen Species, Aging, and Immune Senescence

The immune system is essential to combat pathogenic stress. Age-associated immune dysfunction is also known as "immune senescence" and results from cooperative as well as antagonistic changes in multiple pathways. The mechanisms that mostly contribute to senescence are thymic involution, oxidative stress, proteostasis, telomere attrition, DNA damage signaling, epigenetic alterations, increased inflammation, and transcriptional alterations.

Regarding oxidative stress, it remains unclear whether increased ROS levels, for example, are the cause or consequence of aging. However it is becoming increasingly clear that not all ROS are deleterious. For example, data show that hematopoietic stem cells and early progenitors contain lower levels of ROS than their more differentiated progeny and that these may be critical in maintaining their stem cell potential [285].

Regarding ROS action, it appears that at small levels of  $H_2O_2$ , the main ROS present *in vivo*, and also  $O_2^-$ , have a key role in cellular signaling whereas at higher ROS levels deleterious effects occur [231,286,287].

Specifically in T cells, redox unbalance creating oxidative stress have a significant role in the induction of lowered T-cell responsiveness. This ROS-mediated response is observed in both aging and chronic pathologic conditions [288]. Inhibition of T-cell proliferative response and activation has also been observed as a consequence of oxidative stress-mediated depletion of intracellular stores of glutathione [289,290]. In addition, redox conditions strongly influence CD4<sup>+</sup> T-helper cell polarization, skewing them toward a more Th2 type than a Th1 type, similar to that observed during aging [291].

In the context of immune senescence there are some reports of miRNA-mediated regulation of the generation of memory cells, inflammatory gene inductions, chromatin remodeling, and MHC expression implicating a role of miRNA in regulating immune decline in aging [292,293]. Also, the miR-17–92 cluster (further discussed subsequently) has been implicated in aging in various human cell types, including CD8<sup>+</sup> T-cell subtypes [294].

It still remains unclear which specific miRNAs regulate the various pathways, including those regulating ROS signaling and aging, to understand better how senescence is regulated at the immune system level.

#### 10.3.6.3 Reactive Oxygen Species, Inflammation, and Cancer

As referred to earlier, immune cells present in inflamed tissue can generate ROS, which ultimately may lead to DNA damage and can contribute to carcinogenesis. Indeed, loss-of-function mutations of tumor suppressor genes and the gain-of-function of oncogenes often cooperate to induce inflammation-driven carcinogenesis [273]. Under oxidative stress conditions, caused by accumulation of ROS during chronic inflammation, for example, DNA strand brakes, base modifications, or illicit cross-links can occur. These can lead to replication errors and the subsequent genomic instability might be associated with tumor initiation [295,296]. Also, chronic exposure to UV radiation can lead to inflammatory tissue damage and ROS generation [297], which might cause *Ras* oncogene activation and p53 tumor suppressor gene inactivation, thus contributing to skin carcinogenesis [298]. Thus, chronic inflammation can lead to the generation of cellular ROS; or oxidative stress resulting from an imbalanced cellular redox regulation or exposure to environmental stress can induce tissue inflammation locally, both of which contribute to carcinogenesis.

Although there is no direct evidence of miRNAs linking ROS production, inflammation, and cancer, several miRNAs were shown to have significant roles in inflammationassociated carcinogenesis [299]. For example, when the proinflammatory cytokine IL-6 is stably overexpressed it causes induction of miR-let-7a, causing the constitutive phosphorylation of STAT3 in malignant cholangiocytes [300]. By contrast, IL-6 overexpression in malignant cholangiocytes repressed the expression of miR-370, miR-148, and miR-152, leading to induction of DNMT-1 and subsequent inactivation of methylation-sensitive tumor suppressor genes such as Rassf1a and p16INK4a [300,301]. Thus, by activating STAT3, IL-6 serves as a bridge between inflammation and cancer. Interestingly, miR-155, which is overexpressed in many human cancers, is inversely correlated at the expression level with SOCS1, an endogenous inhibitor of the IL-6-STAT3 signaling pathway, in human breast cells [302]. In fact, when miR-155 is overexpressed in breast cancer cells, the JAK-STAT3 signaling pathway is activated, an effect similar to the incubation of these cells with IL-6 or LPS [302]. Furthermore, diffuse large B-cell lymphomas displayed increased levels of miR-155, which correlated with down-regulation of the TNF- $\alpha$ -induced expression of the inositol phosphatase SHIP1. Treatment with TNF- $\alpha$  antagonists such as infliximab diminished miR-155 expression, restoring the level of SHIP1 and reducing cell proliferation [303].

It appears that several miRNAs that are regulated by cytokines can function as a powerful link between inflammation and cancer. Given that some cytokines can modulate enzymes involved in the production of ROS (e.g., TNF- $\alpha$  can decrease eNOS expression) [212], it is possible that miRNAs controlling the production of these cytokines in a tumor context might alter the oxidative status of a tumor microenvironment.

Overall, these small ncRNAs, the miRNAs, have been increasingly implicated in several human disorders and are starting to be used as biomarkers in several immune and stress-related dysfunctions. Increasing knowledge of their regulation, mechanism of action, and targets will benefit their use as therapeutic agents in clinical settings, as further discussed in Section 10.5.

# 10.4 Long Noncoding RNAs: How Far Are Their Strategies to Regulate Gene Expression?

In past years we have seen the discovery of a variety of functions for lncRNAs covering almost all aspects of gene expression regulation ranging from epigenetic to translational, including transcriptional and posttranscriptional control. Although the illustrative examples are growing throughout the phylogenetic tree, the molecular mechanisms underlying these functions are still poorly understood. However, it is assumed that through their ability to pair with nucleic acids, and owing to their complex secondary structures, these molecules are able to fine-tune regulatory proteins to target gene promoter regions, GpC regulatory islands, mRNAs, and other proteins, creating scaffolds for interactions.

### 10.4.1 Chromatin Remodeling Complexes and Epigenetics

The lncRNA "*Xist*" was one of the first examples showing that in nuclei, lncRNAs are involved in chromatin modulation by recruiting histone modifiers and chromatin remodeling complexes [26]. *Xist* acts in *cis* and promotes chromosome X inactivation by forming transcriptionally inactive heterochromatin. This lncRNA recruits and binds the Polycomb repressive complex 2 (PRC2) together with H2K27 to the inactive X chromosome [304]. This regulation is complex because not only do independent PRC2 pathways seem to occur [7] but *Xist* is also positively and negatively regulated by other lncRNAs such as *Tsix*. *Tsix* inhibits *Xist* transcription by promoting epigenetic alterations in *Xist* promoter [305]. Other examples of lncRNAs, such as Air, Kcnq1ot1, Mira, and HOTAIR [306–310], which are able to induce chromatin imprinting by recruiting chromatin remodeling complexes, have been described. These new examples showed that lncRNAs

regulate gene expression by acting in *cis* and/or *trans*, which means that they are able to repress or induce gene expression that are near their transcription sites and/or at independent loci, respectively. For example, the histone H3K4 methyltransferase MLL1 complex is recruited by the lncRNA (Mistral-Mira) and the interaction of MLL1 with chromatin-associated Mira triggers dynamic changes in chromosome conformation that mediate activation of *Hoxa6* and *Hoxa7* transcription, which culminates in transcriptional activation of genes involved stem cell differentiation [308,309]. Another example is that of HOTAIR, a *trans*-acting lncRNA that serves as a scaffold for two histone modification complexes binding to PRC2 and to LSD1 (in complex with CoREST/REST). This lncRNA coordinates the targeting of PRC2 and LSD1 to chromatin for coupled histone H3 at lysine 27 methylation and lysine 4 demethylation, leading to subsequent gene silencing [310]. Transcription of the lncRNA ANRIL is induced by transcription factor E2F1 in an ataxia-telangiectasia mutated-dependent manner after DNA damage. The high levels of ANRIL prevent the expression of INK4a, INK4b, and ARF tumor suppressor genes at the end of DNA damage response [311]. Also, the chromobox 7 (CBX7) of PRC1 binds to ANRIL, and both CBX7 and ANRIL repress the INK4b/ARF/INK4a locus and control senescence by H3K27 methylation [312].

The involvement of lncRNAs as a chromatin modulator in organ development is illustrated by the role of two lncRNAs involved in mouse cardiac development designated by braveheart (*Bvht*; also known as *Gm20748*) [313]) and *Foxf1* adjacent noncoding developmental regulatory RNA (*Fendrr*) [314]. Both *Bvht and Fendrr* interact with critical factors required for epigenetic regulation, namely, PRC2 and Trithorax protein group–MLL1 complexes; this strongly supports the idea that they are involved in cardiac commitment and correct heart development, respectively [313,314].

The use of co-immunoprecipitation and RNAi assays revealed that other members of the class of long intergenic noncoding RNAs (lincRNAs), that range in size from about 300 nt to several thousands of nucleotides, associated with chromatin-modifying complexes to specific genomic loci regulating gene expression [315]. Moreover, the ncRNA promoter-associated RNA (pRNA) is complementary to the rDNA promoter and mediates *de novo* CpG methylation of rRNA genes (rDNA). In fact, pRNA interacts with the target site of the transcription factor TTF-I, originating a DNA–RNA triplex that is specifically recognized by the DNA methyltransferase DNMT3b [316].

## 10.4.2 Long Noncoding RNA Structural Labels and Partners of Transcription Factors

Besides their ability to bind and recruit complexes involved in chromatin remodeling, lncRNAs are able to regulate gene-encoding proteins by interacting with their promoter regions (pRNAs). This is based on their capacity to base pair with DNA, probably creating local recognizable structures, "labels," that are able to recruit, allow recognition, and promote interaction with transcription regulatory proteins. lncRNAs may also modulate transcription factors and/or establish bridges between them and co-regulators working

as scaffolds. For example, the lncRNA PANDA is transcribed from the CDKN1A gene promoter (encoding a cell-cycle inhibitor) and is induced by DNA damage in a p53dependent manner [317]. Interestingly, PANDA interacts with the transcription factor NF-YA, inhibiting its ability to up-regulate proapoptotic genes. Therefore, whereas CDKN1A mediates cell-cycle arrest, PANDA promotes cell survival by inhibiting the induction of apoptotic genes. There is also evidence that this link is exploited by tumor cells. The regulatory network of lncRNAs in response to DNA damage and involving p53 seems to be complex. In fact, Huarte et al. [318] described an lncRNA named lincRNAp21 that is transcribed from a locus 15 kb upstream of *CDKN1A* that is induced by p53 and mediates p53-dependent gene repression. Therefore, the upstream regulatory region of the CDKN1A gene regulates expression of multiple coding and noncoding transcripts involved in the response to DNA damage. Another example comes from the studies of Zhu et al. that showed that the lncRNA ANCR, required to maintain the undifferentiated cell state within the epidermis, is able to associate with the enhancer of zeste homolog 2. This association results in inhibition of the expression of the transcription factor Runx2, which is required for osteoblast differentiation [319].

Activation of gene transcription through cross-talks between lncRNAs and transcription factors is also mediated by the ncRNA Evf-2 that forms a stable complex in vivo with the homeobox transcription factor Dlx2, activating its activity and with an important role in forebrain development and neurogenesis [320]. This type of interaction is also relevant in the role of several lncRNAs associated with pluripotency or specific differentiation events [321]. In fact, some lncRNAs are induced upon the reprogramming of fibroblasts to induced pluripotent stem cells (iPSCs) [322]. Illustrating this is the case of the lincRNA -RoR (Regulator of Reprogramming), which is regularly enriched in human iPSCs regardless of their origins and regulates the reprogramming that leads to pluripotent stem cells. [322]. Many of these ncRNAs are coexpressed with transcription factors known to be involved in regulating pluripotency. Indeed, experimental data show that lincRNAs are direct regulatory targets of several pluripotency-associated transcription factors (including Oct4, Sox2, Nanog, cMyc, nMyc, Klf4, Zfx, Smad, and Tcf3) [323]. The depletion of these lncRNAs from the cells, by performing loss-of-function assays, has consequences for gene expression that are comparable to the knocking down of embryonic stem cell regulators [323]. These authors observed that the knockdown of many lincRNAs causes the exit from the pluripotent state or, alternatively, the upregulation of pathways of commitment programs [323].

Despite available experimental data, the full scenario of how lncRNAs regulate transcription at the promoter level is far from being complete because many pieces of information are still missing from the puzzle. For example, experimental data show that enhancers can also be transcribed originating enhancer-noncoding transcripts (eRNAs) that probably have regulatory roles [324,325]. Moreover, long (antisense) pRNAs have the potential to form double-stranded molecules that can be processed into endo-siRNAs. These endo-siRNAs are able to be pair to the promoters, inducing transcriptional gene silencing [326–328] or activation [175,176,329] in a way similar to short pRNAs [22].

The regulatory activity of lncRNAs is not limited to transcriptional factors and co-regulators but also extends to the general transcription factors and to polymerase II. For example, a lncRNA that is transcribed from an upstream minor promoter of the dihydrofolate reductase (DHFR) gene mediates the regulation of basal transcription by forming a stable RNA–DNA triplex within the major promoter of the DHFR gene, preventing the binding of TFIIB, a component of the transcription initiation complex [330].

## 10.4.3 Long Noncoding RNAs: Linking Chromatin Remodeling and Transcription Regulation

lncRNAs are also able to combine regulatory mechanisms coordinating the activity of transcription factors, co-regulators, and chromatin remodeling complexes. The RNAbinding protein TLS is a critical transcriptional regulatory sensor of DNA damage signals that is allosterically regulated by RNA binding. Ligation of the protein to RNA causes the shift from an inactive to an active conformation that in turn allows the binding and inhibition of the CBP/p300 histone acetyltransferase activities on the repressed cyclin D1 (CCND1) gene [171]. Recruitment of TLS to the promoter of cyclin D1 is directed by low-copy number NcRNA<sub>CCND1</sub> transcripts that therefore negatively regulate CCND1 gene transcription. In the central nervous system, where lncRNAs are considerable abundant, they have been associated with neurogenesis and the evolution of the complexity of brain. For example, a microarray analysis for lncRNAs in human embryonic cells (hESCs) differentiating into neurons showed that some hESC-specific lncRNAs were involved in pluripotency maintenance [331]. Knockdown studies showed that depletion of these lncRNAs inhibited neurogenesis, and immunoprecipitation studies revealed interaction with the transcription factor SOX2 and the SUZ12 component of the PRC2 in undifferentiated hESCs [331].

## 10.4.4 Long Noncoding RNAs and Alternative Splicing

Several examples also illustrate that even alternative splicing is a regulatory step under the control of lncRNAs. Indeed it was shown that long noncoding antisense RNAs may base pair with their sense RNAs, an event that will hide the splice sites and have an impact on which splice variant will be produced. In fact, the antisense transcript RevErbA $\alpha$  regulates the thyroid hormone receptor  $\alpha$  gene (TR $\alpha$ ) splicing controlling the balance between the TR $\alpha$ 1 and TR $\alpha$ 2 mRNA levels [332]. lncRNAs with snoRNA (Small nucleolar RNA) ends corresponding to positions of intronic snoRNA were also described [333]. In fact, during exonucleolytic trimming, the sequences between the snoRNAs are not degraded, leading to the accumulation of lncRNAs flanked by snoRNA sequences but lacking 5' caps and 3' poly(A) tails. These sno-lncRNAs associate strongly with Fox family splicing regulators and produce changes in splicing patterns. Furthermore, these sno-lncRNAs map to a genomic region that is deleted in patients with Prader–Willi syndrome, which strongly suggests an association of these sno-lncRNAs with the disease [333]. Supporting these observations are data comparing transcriptomes of human, rhesus, and mouse that demonstrating that

the expression of sno-lncRNAs is species-specific and that their processing is closely linked to alternative splicing of their parent genes [334].

lncRNA interact with multiple proteins including members of the importin- $\beta$  superfamily, and regulate the nuclear trafficking nuclear factor of activated T cells [335]. The abundant 4 kb ncRNA, named nuclear-enriched autosomal transcript 1 (NEAT1), localizes at "paraspeckles" [336] and contributes to the formation of these dynamic structures of the interchromatin space that are implicated in mRNA nuclear retaining [336]. Paraspeckles have been proposed to regulate transcription (for example, via the sequestration of key transcription factors) [337,338], control alternative splicing patterns [339], and modulate the DNA damage response [340].

## 10.4.5 Long Noncoding RNA Mediators of Messenger RNA Stability and Translation

By base pairing with their sense messenger RNAs (mRNAs), long noncoding antisense RNAs are also able to control the stability of their counterpart mRNA, creating a scaffold to recruit proteins involved in RNA degradation, such as Staufen 1 (STAU1), a protein that binds to double-stranded mRNA and induces its decay [341]. In the case of the 3.7 kb lncRNA, terminal differentiation-induced ncRNA (TINCR), it directly binds to the STAU1 protein creating the TINCR-STAU1 complex that mediates the stabilization of epidermis differentiation mRNAs, such as keratin 80. TINCR is required for high levels of key differentiation genes, many of which are mutated in human skin diseases, including FLG, LOR, ALOXE3, ALOX12B, ABCA12, CASP14, and ELOVL3 [342]. TINCR-mRNA interaction occurs through a 25-nt "TINCR box" motif that is strongly enriched in the interacting mRNAs [342]. Another example is that of the lncRNA named  $\beta$ -site APPcleaving enzyme 1-antisense (BACE1-AS) that base pairs with the mRNA (BACE1) encoding the  $\beta$ -secretase an enzyme responsible for  $\beta$ -amyloid production, stabilizing it [343]. These examples show that the role of lncRNAs in regulating mRNA stability is also complex because they have the ability to increase or decrease the stability of the target coding RNAs.

Formation of the sense–antisense duplex complexes may also be involved in regulating protein synthesis by regulating translation, as in the case of PU.1 mRNA that encodes the transcription factor PU.1, an important regulator of hematopoiesis. In this case the translation of PU.1 is negatively regulated by the noncoding antisense polyadenylated transcript that exists at low levels but with a longer half-life than that of the sense PU.1 transcript [344]. The suppressor tumor protein p53 pathway is also negatively regulated by the lncRNA co-repressor 1 (Trp53cor1; also known as lincRNA-p21) [345]. The lincRNA-p21 associates with the ubiquitous ribonucleioprotein HuR that regulates cell proliferation, survival, carcinogenesis, and the stress and immune responses. HuR performs these functions mainly by associating with subsets of mRNAs and by increasing their stability and/or modulating their translation [346,347]. In HeLa cells, low HuR levels cause lincRNA-p21 accumulation that increases its association with *JUNB*  and *CTNNB1* mRNAs and causes the inhibition of their translation. High HuR levels decreased lincRNA-p21 levels, which in turn derepressed JunB and  $\beta$ -catenin translation, increasing the levels of these proteins [345]. Similarly, the lncRNA urothelial carcinoma-associated 1 (UCA1) interacts with hnRNP I, forming a functional ribonucleoprotein complex that increases UCA1 stability. Interestingly, hnRNP I enhances the translation of p27 by interacting with the 5'-UTR of p27 mRNAs, an event that is competitively inhibited by the presence of UCA1. This observation correlates with the fact that UCA1 has an oncogenic role in breast cancer [348].

Another mechanism that most probably underlies lncRNA's roles regulating mRNA stability is based on its potential to form siRNAs that will be involved in mRNA decay. This mechanism involves the formation of a sense-antisense pair of transcripts able to pair up and then be processed into siRNAs. These pairs can be created directly from the same loci (cis-NATs) or from different loci (trans-NATs). Interestingly, in rice a small number of pseudogenes are transcribed originating antisense transcripts that after pairing with the coding gene or a paralogous pseudogene transcript are processed into siRNAs [349]. Similarly, in mammalians pseudogene transcripts can be processed into siRNAs with the capacity to suppress gene expression in mouse oocytes [350,141]. Consequently, NATs may regulate gene expression through a mechanism similar to that of miRNAs and siRNAs, a hypothesis supported by several examples. During spermatogenesis, a complex process that is accompanied by microtubule cytoskeleton remodeling, it was observed that the gene encoding TBCA, a protein that binds to β-tubulin and is involved in the folding and dimerization of new tubulin heterodimers (the building blocks of microtubules), is regulated by a *Tbca* pseudogene that is transcribed in both directions [351]. During testis maturation, *Tbca* pseudogene levels decrease, whereas those of the *Tbca* mRNA progressively increase, which suggests that this *Tbca* lncRNA is required to maintain the undifferentiated state of spermatids. Another example comes from the regulation of the gene codifying for nitric oxide synthase (NOS2A) that is regulated by its nonsense counterpart. Similarly to the pair anti-Tbca/Tbca, anti-NOS2A RNA and the NOS2A mRNA have reverse patterns of expression in undifferentiated hESCs and in hESCs differentiating into neurogenic precursors [158].

Interestingly, NATs may be interlinked with miRNAs and siRNAs through the regulatory molecular mechanisms involved in gene expression regulation but they may also be components of the regulatory networks that control miRNAs and siRNA, creating a multilayer regulatory system. For example, lncRNAs may compete with coding mRNAs for specific miRNAs, binding and sequestering them acting as "miRNA sponges" [352]. Thus these lncRNAs have been designated competing endogenous RNAs (ceRNAs). In myogenesis lncRNA, muscle differentiation 1 (linc-MD1) was one of the first lncRNAs to be identified [353] but more have already been described [7]. Indeed, linc-MD1 regulates the time of muscle differentiation by acting as a ceRNA in mouse and human myoblasts [353]. linc-MD1 seems to operate by acting as a "sponge" of miR-133, regulating the transcription factors MAML1 and MEF2C involved in activating muscle-specific gene expression. Interestingly, linc-MD1 levels are strongly reduced in muscle cells of patients with Duchenne muscular dystrophy [353]. In hESC self-renewal, lincRNA-RoR regulates a core of pluripotency transcription factors (Oct4, Nanog, and Sox2) by linking them to the regulatory network of miRNAs by behaving as an endogenous miRNA sponge [354].

These various examples clearly show that lncRNAs display a vast repertoire of strategies to regulate protein-encoding genes. A central role of these molecules is their involvement in regulating gene expression during development by interacting with ubiquitous regulatory proteins to form cell type—specific RNA—protein complexes that coordinate cell type—specific gene expression programs. The number of examples of lncRNAs involved in regulating development and differentiation is continuously growing, which indicates that lncRNAs has a broad role in development. Examples such as lincRNA-EPS, which is involved in erythroid terminal differentiation [355], and lncRNAs, which participates in regulating adipogenesis, support this idea [356]. Also, as detailed in the next section, lncRNAs have been shown to have increasingly important roles in immune cell differentiation and function.

Interestingly, lncRNAs seem to be active, in general, in the cell undifferentiated states repressing target genes whose expression is critical for cell fate. In these processes they work as integrative molecules concentrating and distributing and/or combining signals among chromatin-modifying complexes, transcriptional, splicing and translational regulatory molecules, and also factors regulating mRNA transport and stability, therefore decreasing transduction noise.

## 10.4.6 Role of Long Noncoding RNAs in the Immune System

Although miRNAs are extremely abundant and regulate many aspects of immune cell differentiation and function (Section 10.3.2), lncRNAs, which, as mentioned earlier, have mainly been studied in the context of genomic imprinting, pluripotency, and cell differentiation control and chromosome dynamics, are emerging as important regulators of immune cell differentiation and activation. Interestingly, the expression profiles and molecular and biological functions of lncRNAs involved in all of these processes (including regulation of immune cell function) have been compiled and integrated in the NONCODE database (http://www.noncode.orghttp://www.noncode.org) that includes the first integrated collection of expression and functional lncRNA data obtained from reannotated microarray studies [357]. Several genome-wide studies have revealed that immune cells express large numbers of lncRNAs. For example, CD8<sup>+</sup> T cells express many hundreds of lncRNAs, 96 of which are specific to lymphocytes and are dynamically regulated during differentiation or activation (29 of transcripts were specific for CD8<sup>+</sup> T cells, 21 of the ncRNAs were significantly modulated during memory T-cell differentiation, 81 were during effector cell activation, and 4 were significantly regulated during both transitions) [358]. Interestingly, some of these overlap either with immunologically important protein-coding genes or with miRNAs and siRNAs, which suggests that proteins and different functional RNA species might regulate lncRNA functions.

Unique signatures of lncRNAs expression were also observed in response to severe acute respiratory virus or influenza virus [359], many of which were regulated downstream

of type I IFN signaling. Rhabdomyosarcoma cells are also reported to express lncRNAs when they are infected with enterovirus 71 [360].  $\text{CD11c}^+$  dendritic cells also express a large number of lnRNAs when stimulated with LPS, an activator of TLR-4 signaling [152]. The lnc-IL7R that overlaps with the 3'-UTR region of the human IL-7 receptor  $\alpha$ -subunit gene (IL7R), is also involved in inflammatory response. This lncRNA was able to decrease the LPS-induced inflammatory response, as observed by the increased expression of LPS-induced E-selectin, VCAM-1, IL-6, and IL-8 in lnc-IL7R knockdown cells [361].

It was also shown that lncRNAs can be important regulators of LPS-induced innate immune response through TLR-4 signaling in bone marrow–derived macrophages [362]. As reviewed in Pagani et al., CD4<sup>+</sup> T cells are reported to express many lncRNAs [363].

The molecular functions of some of the lncRNAs expressed in these immunologic contexts are beginning to be explored. They were shown to be important to both adaptive and innate immunity. The next two examples refer to lncRNAs acting in acquired immunity. The lncRNA Theiler's Murine Encephalitis Virus Possible Gene1 (*Tmevpg1*) also named NEttoie Theiler's Pas Salmonella (*NeST*), initially proposed to control Theiler virus persistence in mice [364], is the first identified enhancer lncRNA of the immune system to regulate expression of a master cytokine such as IFN-gamma [364–366]. Located downstream from the IFNG gene and transcribed from the antisense strand relative to IFNG, NeST acts as an enhancer-like lncRNA by enhancing transcription of INF-gamma. Earlier studies demonstrated that transcription of *Tmevpg1/NeST* was Th1 selective and dependent on Stat4 and T-bet, transcription factors involved in Th1 cell differentiation [364]. NeST RNA binds WDR5, which alters histone 3 lysine 4 trimethylation at the IFN-gamma locus. Consequently, both IFN-gamma RNA and IFN-gamma protein levels are increased in activated CD8<sup>+</sup> T cells [365]. In Th2 cells, lincR-Ccr2-5'AS is transcribed in the direction opposite the chemokine Ccr2 and is located between Ccr2 and Ccr3 genes [367]. Silencing of lincR-Ccr2-5'AS leads to lower expression of the neighboring Ccr1, Ccr2, Ccr3, and Ccr5 genes. These chemokines are required for trafficking of Th2 cells to the lungs, and knockdown of lincR-Ccr2-5' AS also reduced migration of Th2 cells to the lung.

lncRNAs also have relevant roles in innate immunity [368]. TNF-α was shown to regulate many lncRNA in murine fibroblasts, including 54 pseudogene lnRNAs, several of which exhibited a selective expression in response to specific cytokines and microbial components in an NF- $\kappa$ B-dependent manner. One of them is Lethe, which interacts with the NF- $\kappa$ B subunit RelA to inhibit RelA DNA binding and target gene activation [369]. Another lncRNA exerting its functions in innate cells is KIR, an antisense lncRNA expressed in progenitor or pluripotent cell lines that when overexpressed in NK cells causes decreased expression of the KIR protein-coding gene [370]. KIR antisense lncRNA overlaps with KIR-coding exons 1 and 2, as well as a proximal promoter upstream of KIR. Transcription of KIR antisense lncRNA appears to be regulated by myeloid zinc finger 1, which leads to silencing of KIR through an unknown mechanism. Also it was observed that during innate activation of THP1 macrophages, a group of 159 lincRNAs was
differentially expressed. In this group, linc1992/THRIL was shown to regulate the transcription of TNF- $\alpha$  gene by binding to its promoter by forming a complex with a heterogeneous nuclear ribonucleoprotein L. Moreover, expression of linc1992/THRIL was correlated with the severity of symptoms in patients with Kawasaki disease, an acute inflammatory disease of childhood [371].

HOTAIRM1 is an lncRNA specifically expressed in myeloid cells and appears to have an important role during granulocyte maturation [372]. It is antisense to the HOXA genes and is transcribed in NB4 promyelocytic leukemia cells. When these are activated by all*trans* retinoic acid, which induces granulocyte maturation, HOTAIRM1 is required for the expression of HOXA1, HOXA4, CD11b, and CD18.

A study demonstrated that lincRNA-Cox2 is an lncRNA significantly induced upon TLR2 engagement by a synthetic bacterial lipoprotein (Pam3CSK4) [373]. Hundreds of immune genes were shown to be both positively and negatively regulated by lincRNA-Cox2 during an innate immune response. Many of the target genes are repressed (including chemokines, chemokine receptors, and IFN-stimulated genes) at the transcriptional level and are mediated by an RNP complex consisting of lincRNA-Cox2 and hnRNPA/B or hnRNPA2/B1.

Interestingly, NEAT has two isoforms, 3.7 kb NEAT1 and 23 kb NEAT2/MALAT-1, that do not share significant homology with each other but are conserved within the mammalian lineage, which suggests a functional role for these lncRNAs [374]. As mentioned, NEAT2 is a component of nuclear speckles [316]. Also, this variant was induced by a TLR3 ligand (polyI:C), infection with herpes simplex virus-1(HSV-1), and the measles virus (MV). Moreover, the paraspeckle protein, SFPQ, binds to the IL-8 promoter blocking its expression. In cells stimulated by PolyI:C or infected with HSV-1 and MV, this regulator SFPQ protein is displaced to the NEAT1 promoter, which allows formation of the paraspeckles and the activation of IL-8 [337].

This evidence strongly indicates that lncRNAs regulate several molecular functions in the immune system.

#### 10.4.7 Long Noncoding RNAs as Sensors of Environmental Conditions: Role in Stress Response

It is possible to envisage a system in which environmental and stress conditions can affect development throughout the establishment of a web of lncRNAs regulating gene expression in coordination with distinct regulatory proteins. In fact, data suggest that they also seem to be sensors for environmental conditions mediating cellular response, for example, to stress conditions. In the plant *Arabidopsis*, winter cold induces the synthesis of a sense transcript from flowering locus c (*FLC*) called *COLD-ASSISTED INTRONIC NON-CODING RNA* (*COLDAIR*), which has similarities with lncRNAs by directly interacting with the PRC2 complex and targets it to *FLC*, establishing stable repressive chromatin at *FLC* and an epigenetic memory [375]. In fact, in cold weather the H3 at *FLC* is methylated, which inhibits the expression of genes required to trigger

flowering. The locus FLC is also differentially regulated by an antisense transcript IncRNA, called COLD-INDUCED LONG ANTISENSE INTRAGENIC RNA (COOLAIR). In fact, changes in the abundance, splicing, and polyadenylation of this lncRNA are observed in response to different environmental conditions and are associated with altered chromatin regulation and differential *FLC* expression [376]. In mammals there is not vet much information relative to the role of lncRNAs, but this is probably because only a few studies have been performed. For example, the lncRNA Uchl1 is involved in the translation enhancement rate of the ubiquitin carboxy terminal hydrolase L1 (UCHL1) mRNA by promoting its association with polysomes [377]. The inhibition of mTOR activity by stress causes the repression of cap-dependent translation. Under these conditions, *Uchl1-as1* is transported from the nucleus to the cytoplasm, where it can base pair with the Uchl1 mRNA and stimulate its cap-independent translation. The UCHL1 protein is a specific neuronal protein involved in the rapamycin-neuroprotective function associated with neurodegenerative diseases. Also, Adapt33/gadd7 is an lncRNA associated with polysomes in hamster cells, which is involved in the stress response to many agents including the adaptive response to hydrogen peroxide [378-381].

The impact of lncRNAs in regulating gene expression is vast, but in the future, new roles of lncRNAs may arise because some contain open-reading frames with the potential to produce small peptides that may have roles *in vivo*. If this will be true we probably have only a faint image of their evolution, functions, and molecular mechanisms of action.

### 10.5 Final Considerations

## 10.5.1 Are Noncoding RNAs the Desired Miracle Molecules in Human Disease Therapeutics?

From what was analyzed in this chapter, there is no doubt that ncRNAs have vast widespread regulatory activity in different steps of gene expression throughout the phylogenetic tree, creating various regulatory layers with an impact on development, differentiation, the maintenance of genome integrity, and homeostasis. Moreover, they are able to establish an incredible dynamic meshwork with regulatory proteins to sense and signal from internal and external environments, and improve the accuracy of signaling pathways by decreasing noise and by increasing the speed of responses. These and some other ncRNA features render these molecules appealing for diagnosis (as biological markers) and as therapeutic targets, and strong efforts have already been made to that purpose. In fact, an excellent review described and discussed the potential role of ncRNAs as noninvasive biomarkers and already explored RNA therapeutic drugs under study for the treatment of several diseases such as neurodegenerative diseases, myopathies, and cancer, as well as difficulties and future challenges [8]. Regarding miRNAs, because they are stable molecules and easily identified in body fluids, they may

constitute excellent specific biomarkers; experimental data already support this view in the case of lung cancer, colorectal cancer, and diabetes [382]. There are also examples showing the existence of correlations between circulating miRNA levels and the response to anticancer agents. In this case, this type of assay may be used to follow the effectiveness or resistance of certain therapeutic molecules [383], which also opens doors for personalized medicine.

On the other hand, numerous strategies have been used to regulate the levels of critical miRNAs, taking advantage of their structures and biogenesis pathways. Three main approaches have been blockage/loss-of-function and down-regulating or up-regulating their expression [8,384]. The strategies are diverse; for example, misex-pressed miRNAs may be trapped using artificially designed target RNAs that compete with endogenous targets (miR sponges), sequester them, and abolish their functions. For example, small antisense oligonucleotides or locked nucleic acids can be designed to antagonize specific miRNAs [384]. miRNAs function can also be suppressed by decreasing their levels, by promoting degradation (siRNAs, ribozymes, and deoxyribozymes). In all of these approaches there will be an increase in the expression levels of the protein-encoding genes normally down-regulated by the miRNA. Alternatively, their function can be blocked by using small molecule inhibitors of specific miRNAs [385]. The interaction of small molecules and miRNAs was termed "small molecule compounds" (SMIR) by Melo and Calin et al. [386].

Similarly, antagomirs, which are cholesterol-conjugated synthetic RNAs with a 2'-O-methyl linkage and phosphorothioate modification that are complementary to the full sequence of the targeted miRNA, can pair with the miRNA, blocking its function.

Another strategy is to restore the correct levels and function of the ncRNAs, e.g., using small molecules (nonspecific induction of miRNA expression), employing double-stranded synthetic RNAs that mimic endogenous miRNAs (miRNA mimics) or using miRNA expression vectors [387]. Most of these distinct methodologies are already being used in preclinical studies and some are currently in phase I and IIa [8]. Indeed, consulting the ClinicalTrials.gov website (US National Institutes of Health; http://www. clinicaltrials.gov), we find 272 ongoing clinical studies (June 2015) focused on the role of miRNAs that clearly showed developing efforts to explore these molecules clinically. Most of these studies involve the role of miRNAs in cancer, but many other human diseases are under evaluation, such as neurodegenerative diseases (e.g., Alzheimer), amyotrophic lateral sclerosis, autism spectrum disorder syndrome, coronary heart disease, multiple sclerosis, diabetes, Duchenne muscular dystrophy and a few dealing with immunologic problems at the level of allergic diseases.

Regarding lncRNAs, many studies using conventional siRNA treatment to downregulate lncRNAs have oncogenic roles, because many of these ncRNAs have modified patterns of expression in different types of cancer cells. Another approach is to deplete lncRNAs using antisense oligonucleotides, as well as ribozymes or deoxyribozymes, which use different interaction mechanisms with the target lncRNAs. The use of synthetic RNA molecules that form hairpin structures that mimic lncRNAs is also an alternative to removing specific lncRNAs. Finally, by using single-stranded oligonucleotides that are able to inhibit the formation of sense—anti-sense RNA pairs through competition, leading in general to an increase in the mRNA counterpart levels, it is possible to increase the encoded protein levels directly [388]. Therefore this is one of the few ways to increase the synthesis of a specific protein directly, an output difficult to achieve in general with other conventional drug designs [389]. Experimental data illustrating this come from an assay in which the continuous infusion of an oligonucleotide with a phosphorothioate backbone targeting cytosolic Cu,Zn-superoxide dismutase (SOD1) was injected into the right lateral ventricle of rats [390]. In the brain and spinal cord of these rats, decreases in SOD1 mRNA and protein levels were observed. Interestingly, amyotrophic lateral sclerosis, a progressive neurodegenerative disease, is characterized by mutated forms of SOD1; currently there is no helpful treatment [47].

Combined strategies have also been used and are promising. Studies *in vitro* have shown that combined treatment with BCL-2 siRNA and miR-15a synergistically improved methotrexate-induced apoptosis of Raji cells [391,367].

Many of these strategies clearly have advantages and are extremely promising compared with traditional therapies, but as expected, some limitations can also be described. In most cases the use of oligonucleotides as therapeutic agents requires complex formulation and delivery. However, the development of nanotechnologies and strong investments in discovering and using new materials, nanoparticles, and alternative drug delivery strategies will probably allow most difficulties to be to overcome eventually. Moreover, each strategy has weak spots [392]. For example, one disadvantage that needs to be taken into consideration is that SMIRs can have off-target effects in both the tissue of interest and throughout the body and are more complex to design compared with oligonucleotides. In the case of sponges, they have considerably low binding affinities at the concentrations required to trap miRNAs effectively. Therefore, they should be used in excess, which can increase the risk of unwanted toxicities. Another obstacle is the difficulty of assessing the efficacy of a given anti-miR treatment because it was observed that in certain cases, anti-miRs do not always reduce the corresponding miRNA levels, but instead abolish their role by creating a highly stable heteroduplex, as in the case of miravirsen and miR-122 [224]. Because Northern blot and polymerase chain reaction are the preferred techniques to assess anti-miR effects, in this case the effect of the anti-miR is not correctly followed. Moreover, there are no data about the potential cellular toxicity of the continuous accumulation of these complexes in vivo. Attention should also be given to innate immune responses. For example, the evaluation of a high-profile study using siRNAs targeted to vascular endothelial growth factor (VEGF) showed that an off-target immune response was indeed responsible for the observed suppression of VEGF [393].

In addition, miRNAs have multiple mRNA targets, many of which are involved in a specific pathway or in redundant pathways, which can be an advantage. Although this can useful, it deserves attention because monitoring anti-miR effects may require a more massive analysis of targets by using, for example, expression array analyses, to have a

complete scenario of the impact of the treatment. Attention should also be given to the fact that many more systematic studies on biogenesis and the molecular mechanism of the roles of ncRNA are required.

In fact, despite remarkable multidisciplinary development in the miRNA field, it is still difficult to predict the phenotypic consequences of manipulating miRNAs in vivo. Many of these difficulties are still rooted in our lack of knowledge of ncRNA biology. For example, processing of an miRNA results in an intermediate duplex of two potentially mature products that derive from the two arms (5' and 3') of the precursor hairpin [70]. Moreover, data showed that both precursor arms may create two functionally mature microRNAs. Thus alternative mature products produced from the same precursor microRNAs will have different targeting properties and therefore different biological functions, which should be kept in mind in a therapeutic strategy. Furthermore, by using 20-O-methyl oligonucleotides complementary to conserved terminal loops of the corresponding pri-miRNAs, Michlewski et al. [394] showed that these oligomers were able to block their processing efficiently. They also demonstrated that mutations in the terminal loop of pri-miR-18a that do not affect the structural architecture of the stem abolish its efficient processing. Thus they proposed the existence of auxiliary factors that bind to conserved terminal loops. No clues are available concerning the behavior of these factors under therapeutic strategies. Also, although miRNAs frequently have hundreds of conserved targets, most miRNA knockouts lack obvious effects on viability, fertility, visible morphology, and behavior [395-397], and in certain cases there are substantial discrepancies between knockouts and inhibitor-based methods [398]. Data are also available showing that manipulation of the levels and functions of a given ncRNA may be adequate for a given signaling pathway but cause problems in another in which this ncRNA participates. For example, antagomirs to let-7 and miR-122 have potential therapeutic applications for metabolic disorders and/or HCV [8], but these sncRNAs also have tumor-suppressor functions. That many miRNAs have greater effects in certain environmental backgrounds [399] highlights the need for genetic and environmental interaction studies. Therefore, much remains to be understood about how miRNAs are participating in biological pathways. Of crucial future importance will be efforts to further understand basic cellular disease mechanisms based on ncRNA networks. In Drosophila there is growing evidence suggesting that miRNA gain-of-function may be a much more frequent cause of disease than miRNA loss-of-function [400,401], which should be kept in mind and explored. Finally, it is starting to be well-accepted that miRNAs are modulators of protein-coding gene targets by fine-tuning their expressions rather than by causing drastic inhibition. Although this might be extremely important for specific cellular processes, it could also create background noise in therapeutic strategies.

The challenge of establishing ncRNAs as new therapeutics requires better knowledge of their molecular mechanisms and functions *in vivo* and overcoming biological barriers (e.g., intravascular barriers, immune cells, receptor-mediated endocytosis and endosomes), to clearly determine the side effects and find approaches for controlled delivery and sustained release.

#### 10.5.2 Noncoding RNAs and the Evolution of Regulatory Molecular Mechanism Accuracy: The Development of Multicellularity and Plasticity

Our view of RNA molecules has rapidly changed in the past decade. Essentially they have transformed from intermediate molecules in the information flux between DNA and proteins to ubiquitous active players in regulating gene expression from prokarvotes to eukaryotes. In the past years it also became clear that they have roles by being true partners of proteins to control gene expression at different levels of regulation, not only cooperating with them, but also creating structural environments that recruit, position, and allow or create interactions among proteins. Therefore, their ability to pair up with RNA or DNA allows the construction of a new level of information still based on nucleotide sequence that is used to mediate more accurate recognitions and interactions. This idea could have been anticipated much earlier if we had paid more attention to the role of RNAs in ribonucleoprotein complexes such as the spliceosome, telomerase, or even ribosomes. Without ribosomal RNAs, the peptidyl transferase activity of the ribosomes will be compromised as well as their decoding functions and specific tRNA acceptance [402]. This probably means that complex networks between proteins and RNAs were established early in evolution and have contributed to several basic processes progressively acquired accuracy, culminating in protein synthesis. In most cases in which a cellular mechanism requires precise recognition and precise interaction, RNA molecules are involved as key players or assist in the process. The prominent role of all ncRNAs in development and differentiation programs may also reflect this feature. From an evolutionary point of view, it seems that the "invention" of proteins such as telomerases was a critical step in establishing accurate spatial and temporal regulatory processes that probably allowed the evolution of eukaryotic complexity and later, the appearance of multicellularity. For example RNA editing [403], a mechanism based on protein or protein-RNA complexes responsible for the RNA editing reaction, requires a "guide RNA" molecule which, through base-pairing with the target RNA molecule, determines the editing site. By this mechanism an mRNA sequence may be posttranscriptionally altered, and consequently a new protein will be synthesized and information variability is generated. In ciliates, the development of the somatic line from the germline, a process that requires complex genomic rearrangements and processes, is essentially based on a mechanism that requires RNA pairing involving small RNAs that resemble piRNAs [17] and guides the process of rearrangements. This mechanism keeps an epigenetic memory throughout different generations. In fact, piRNAs have been associated with developmental robustness and the maintenance of genomic integrity against transposons and other parasitic nucleic acids, two events that are probably related. It would be interesting to analyze the precise regulation of these processes under challenging environmental conditions. In fact, ncRNAs are also emerging as interfaces between environmental conditions and epigenetic programs and/ or the regulation of gene expression. It is expected that in these complex regulatory

networks several hubs exist. These may be composed of ncRNAs and proteins or, alternatively, only proteins or ncRNAs, which will also allow the rapid and better integration of different environmental signals. Experimental efforts to identify these critical molecules will be welcome.

This scenario is rapidly evolving in plants, where numerous studies have shown the importance of ncRNAs in response to environmental stress conditions such as temperature, heavy metals, or oxidative stress caused by  $H_2O_2$  [404]. A study by Zhang et al. [405], based on expressed sequence tags (EST) analyses in *Arabidopsis*, showed that 25.8% of ESTs containing miRNAs were found in stress-induced plant tissues. Rice seedlings growing under exposure to  $H_2O_2$  show complex patterns of expression of miRNAs, where some are up-regulated and others down-regulated. In these studies the validated targets of the  $H_2O_2$ -responsive miRNAs were demonstrated to have important roles in transcriptional regulation, nutrient transport, auxin homeostasis, cell proliferation, and programmed cell death [406]. This picture resembles that observed in animal cells, in which examples of the involvement of ncRNAs in stress response are also starting to emerge.

In the view of an "RNA world hypothesis," it is tempting to speculate that one of the first roles of RNA was to maintain the viability and integrity of "cell precursors" defending them from destructive invader molecules (stress agents), keeping at same time the fidelity of the first ongoing molecular mechanisms. Although these mechanisms could then be explored later and developed into gene regulation, the ancestral defense functions of ncRNAs would be present and operating in modern cells.

During evolution, some of the stress molecules, in particular  $H_2O_2$ , became signaling molecules requiring cells to evolve mechanisms to distinguish responses to toxic  $H_2O_2$ from those to signaling  $H_2O_2$ . In fact, it is now well accepted that  $H_2O_2$  is a key regulatory molecule involved in biological processes as diverse as immune cell activation, vascular remodeling, proliferation, survival, migration, adhesion, cell polarity, apoptosis, and senescence [286,407–410], by regulating different levels of gene expression from transcription factors to molecules involved in posttranscriptional regulation [231,286]. The attractive question to put forward is whether ncRNAs were involved in establishing  $H_2O_2$ as a prominent regulator in multicellular organisms. In these organisms distinct types of cells are exposed to a new environment and information is mainly received via receptors in the plasma membrane. Indeed, ncRNAs seem to be able to modulate the levels of  $H_2O_2$  by regulating the expression of catalase and SODs, which would support an affirmative answer. However this answer might be too simplistic because H<sub>2</sub>O<sub>2</sub> also regulates miRNA biogenesis. On the other hand, it is known that H<sub>2</sub>O<sub>2</sub> regulates several transcription factors through redox-sensitive cysteine residues at their DNA-binding sites, such as NF-KB, AP-1, HIF-1a, and p53 [231]. Conformational changes resulting from oxidation of amino acid residues have also been implicated in regulating proteins such as TRX, IB, RAS, and Akt [378]. H<sub>2</sub>O<sub>2</sub> can also modulate protein function by changing the oligomerization state of proteins or their ability to interact with protein partners, as in the case of the transcription factor NRF2 [231]. Consequently, it is

tempting to suggest that  $H_2O_2$  and certain ncRNAs could be partners directly collaborating in regulating certain signaling pathways exploring together protein conformation alterations and interactions. If this is true, this type of cross-talk between  $H_2O_2$  and ncRNAs would have contributed to establish  $H_2O_2$  as a regulator molecule.

Finally, it would also be desirable to investigate further how far different classes of ncRNAs (sncRNA and lncRNA) interact with each other, not only at the level of their biogenesis pathways but also at the functional level. Such valuable information would help to understand some puzzling results and could be directly translated to new therapeutic strategies. This information is missed to a certain extent because the different classes of ncRNAs have essentially been investigated separately and in most cases at uneven levels.

This is especially true for the studies of ncRNAs in the immune system, in which miRNAs have been thoroughly investigated but lncRNAs are just beginning to be functionally characterized. One possible explanation could be that, unlike miRNAs, many lncRNAs appear to be expressed at low levels in immune cells [373]. However, these low levels might be enough to produce strong phenotypes, as in the case of *NeST*, which can regulate the expression of INF-gamma and susceptibility to a viral and bacterial pathogen by CD8<sup>+</sup> T cells [366]. Also, unlike most miRNAs, many lncRNAs are not conserved between human and mouse [19], which makes cross-species studies more difficult. In fact, a future challenge would be to determine whether this species-specificity would be responsible for major differences between the mouse and human immune systems. Indeed, identification of these differences might be relevant to determine, in general, the relevance of ncRNAs to the complexity of across species, because these differences are not reflected in their genome sizes.

Importantly, many cells of the immune system, including T cells (CD4<sup>+</sup> T cells and  $\gamma\delta$  T cells [411]) are endowed with the feature of plasticity, which implies that a given effector phenotype can revert rapidly to another differentiation state. This allows the immune system to adapt quickly to the many challenges it faces. Understanding how the network of coding and ncRNAs acts to regulate this plasticity in a coordinated fashion is of major importance in immunology and could ultimately have an impact on the development on novel therapeutic targets for immune-mediated diseases.

We can envisage that the ability of ncRNAs to allow cells to cope with stress conditions, in the context of multicellularity, was explored and contributed to set  $H_2O_2$  as a regulatory molecule but also to allow specific cell types to respond to environmental signals by changing their differentiated state to a new one more adapted to deal with the initial challenge, i.e., by exhibiting plastic behavior. In this view we can assume the capacity of ncRNAs to protect cells against destructive molecules as one of their primitive features that was progressively redefined in the course of evolution, helping organisms to attain higher regulatory complexity levels.

In the field of ncRNAs a fantastic development has occurred in past years but, despite the large bulk of information that is available, we are still missing important data regarding the complexity and mechanisms underlying the establishment of regulatory networks between RNAs and proteins. Therefore, efforts should continue to be made because it is expected that this knowledge may soon be directly translated to medicine, improving human health and even other areas such as agriculture by producing stress-tolerant plants through manipulating ncRNAs [404]. Hence, these evolutionary-old molecules that were critical in the emergence and complexity of life will continue to surprise us with their biological roles, which will probably render them important biotechnological tools.

#### Acknowledgments

Supported by Fundação para a Ciência e a Tecnologia (FCT), Portugal; UID/MULTI/00612/2013.

#### References

- E.P. Consortium, E. Birney, J.A. Stamatoyannopoulos, A. Dutta, R. Guigo, T.R. Gingeras, et al., Identification and analysis of functional elements in 1% of the human genome by the ENCODE pilot project, Nature 447 (2007) 799–816.
- [2] J.S. Mattick, I.V. Makunin, Non-coding RNA, Human Molecular Genetics 15 Spec No 1 (2006) R17–R29.
- [3] R.J. Taft, M. Pheasant, J.S. Mattick, The relationship between non-protein-coding DNA and eukaryotic complexity, Bioessays 29 (2007) 288–299.
- [4] E. Pennisi, Genomics. ENCODE project writes eulogy for junk DNA, Science 337 (1159) (2012) 61.
- [5] E.P. Consortium, An integrated encyclopedia of DNA elements in the human genome, Nature 489 (2012) 57-74.
- [6] S. Djebali, C.A. Davis, A. Merkel, A. Dobin, T. Lassmann, A. Mortazavi, et al., Landscape of transcription in human cells, Nature 489 (2012) 101–108.
- [7] A. Fatica, I. Bozzoni, Long non-coding RNAs: new players in cell differentiation and development, Nature Reviews Genetics 15 (2014) 7–21.
- [8] H. Ling, M. Fabbri, G.A. Calin, MicroRNAs and other non-coding RNAs as targets for anticancer drug development, Nature Reviews Drug Discovery 12 (2013) 847–865.
- [9] A.P. Aalto, A.E. Pasquinelli, Small non-coding RNAs mount a silent revolution in gene expression, Current Opinion in Cell Biology 24 (2012) 333–340.
- [10] D. Moazed, Small RNAs in transcriptional gene silencing and genome defence, Nature 457 (2009) 413–420.
- [11] A.A. Aravin, M. Lagos-Quintana, A. Yalcin, M. Zavolan, D. Marks, B. Snyder, et al., The small RNA profile during *Drosophila melanogaster* development, Developmental Cell 5 (2003) 337–350.
- [12] N.C. Lau, A.G. Seto, J. Kim, S. Kuramochi-Miyagawa, T. Nakano, D.P. Bartel, et al., Characterization of the piRNA complex from rat testes, Science 313 (2006) 363–367.
- [13] A.A. Aravin, G.J. Hannon, J. Brennecke, The Piwi-piRNA pathway provides an adaptive defense in the transposon arms race, Science 318 (2007) 761–764.
- [14] V.K. Gangaraju, H. Yin, M.M. Weiner, J. Wang, X.A. Huang, H. Lin, Drosophila Piwi functions in Hsp90-mediated suppression of phenotypic variation, Nature Genetics 43 (2011) 153–158.
- [15] S.L. Rutherford, S. Lindquist, Hsp90 as a capacitor for morphological evolution, Nature 396 (1998) 336–342.

- [16] V. Specchia, L. Piacentini, P. Tritto, L. Fanti, R. D'Alessandro, G. Palumbo, et al., Hsp90 prevents phenotypic variation by suppressing the mutagenic activity of transposons, Nature 463 (2010) 662–665.
- [17] S. Duharcourt, G. Lepere, E. Meyer, Developmental genome rearrangements in ciliates: a natural genomic subtraction mediated by non-coding transcripts, Trends in Genetics 25 (2009) 344–350.
- [18] J.L. Rinn, H.Y. Chang, Genome regulation by long noncoding RNAs, Annual Review of Biochemistry 81 (2012) 145–166.
- [19] T. Derrien, R. Johnson, G. Bussotti, A. Tanzer, S. Djebali, H. Tilgner, et al., The GENCODE v7 catalog of human long noncoding RNAs: analysis of their gene structure, evolution, and expression, Genome Research 22 (2012) 1775–1789.
- [20] P.J. Batista, H.Y. Chang, Long noncoding RNAs: cellular address codes in development and disease, Cell 152 (2013) 1298–1307.
- [21] J.T. Lee, Epigenetic regulation by long noncoding RNAs, Science 338 (2012) 1435–1439.
- [22] B.X. Yan, J.X. Ma, Promoter-associated RNAs and promoter-targeted RNAs, Cellular and Molecular Life Sciences 69 (2012) 2833–2842.
- [23] R.C. Pink, K. Wicks, D.P. Caley, E.K. Punch, L. Jacobs, D.R. Carter, Pseudogenes: pseudo-functional or key regulators in health and disease? RNA 17 (2011) 792–798.
- [24] P. Carninci, T. Kasukawa, S. Katayama, J. Gough, M.C. Frith, N. Maeda, et al., The transcriptional landscape of the mammalian genome, Science 309 (2005) 1559–1563.
- [25] P. Kapranov, J. Cheng, S. Dike, D.A. Nix, R. Duttagupta, A.T. Willingham, et al., RNA maps reveal new RNA classes and a possible function for pervasive transcription, Science 316 (2007) 1484–1488.
- [26] A. Werner, M. Carlile, D. Swan, What do natural antisense transcripts regulate? RNA Biology 6 (2009) 43-48.
- [27] O. Rackham, A.M. Shearwood, T.R. Mercer, S.M. Davies, J.S. Mattick, A. Filipovska, Long noncoding RNAs are generated from the mitochondrial genome and regulated by nuclear-encoded proteins, RNA 17 (2011) 2085–2093.
- [28] M. Kawano, L. Aravind, G. Storz, An antisense RNA controls synthesis of an SOS-induced toxin evolved from an antitoxin, Molecular Microbiology 64 (2007) 738–754.
- [29] J.A. Opdyke, J.G. Kang, G. Storz, GadY, a small-RNA regulator of acid response genes in *Escherichia coli*, Journal of Bacteriology 186 (2004) 6698–6705.
- [30] J. Georg, B. Voss, I. Scholz, J. Mitschke, A. Wilde, W.R. Hess, Evidence for a major role of antisense RNAs in cyanobacterial gene regulation, Molecular Systems Biology 5 (2009) 305.
- [31] J. Georg, W.R. Hess, Cis-antisense RNA, another level of gene regulation in bacteria, Microbiology and Molecular Biology Reviews 75 (2011) 286–300.
- [32] A. Toledo-Arana, F. Repoila, P. Cossart, Small noncoding RNAs controlling pathogenesis, Current Opinion in micro Biology 10 (2007) 182–188.
- [33] E. Masse, F.E. Escorcia, S. Gottesman, Coupled degradation of a small regulatory RNA and its mRNA targets in *Escherichia coli*, Genes & Development 17 (2003) 2374–2383.
- [34] D. Lalaouna, M. Simoneau-Roy, D. Lafontaine, E. Masse, Regulatory RNAs and target mRNA decay in prokaryotes, Biochimica et Biophysica Acta 1829 (2013) 742–747.
- [35] J.P. Dworkin, A. Lazcano, S.L. Miller, The roads to and from the RNA world, Journal of Theoretical Biology 222 (2003) 127–134.
- [36] N. Lehman, RNA in evolution, Wiley Interdisciplinary Reviews RNA 1 (2010) 202–213.
- [37] L. Bastet, A. Dube, E. Masse, D.A. Lafontaine, New insights into riboswitch regulation mechanisms, Molecular Microbiology 80 (2011) 1148–1154.

- [38] R.R. Breaker, Prospects for riboswitch discovery and analysis, Molecular Cell 43 (2011) 867–879.
- [39] E. Loh, O. Dussurget, J. Gripenland, K. Vaitkevicius, T. Tiensuu, P. Mandin, et al., A trans-acting riboswitch controls expression of the virulence regulator PrfA in *Listeria monocytogenes*, Cell 139 (2009) 770–779.
- [40] K.S. Pollard, S.R. Salama, N. Lambert, M.A. Lambot, S. Coppens, J.S. Pedersen, et al., An RNA gene expressed during cortical development evolved rapidly in humans, Nature 443 (2006) 167–172.
- [41] E. Berezikov, F. Thuemmler, L.W. van Laake, I. Kondova, R. Bontrop, E. Cuppen, et al., Diversity of microRNAs in human and chimpanzee brain, Nature Genetics 38 (2006) 1375–1377.
- [42] G. St Laurent 3rd, C. Wahlestedt, Noncoding RNAs: couplers of analog and digital information in nervous system function? Trends in Neurosciences 30 (2007) 612–621.
- [43] C. Tufarelli, J.A. Stanley, D. Garrick, J.A. Sharpe, H. Ayyub, W.G. Wood, et al., Transcription of antisense RNA leading to gene silencing and methylation as a novel cause of human genetic disease, Nature Genetics 34 (2003) 157–165.
- [44] S. Shirasawa, H. Harada, K. Furugaki, T. Akamizu, N. Ishikawa, K. Ito, et al., SNPs in the promoter of a B cell-specific antisense transcript, SAS-ZFAT, determine susceptibility to autoimmune thyroid disease, Human Molecular Genetics 13 (2004) 2221–2231.
- [45] C. Wahlestedt, Targeting long non-coding RNA to therapeutically upregulate gene expression, Nature Reviews Drug Discovery 12 (2013) 433–446.
- [46] T. Gutschner, S. Diederichs, The hallmarks of cancer: a long non-coding RNA point of view, RNA Biology 9 (2012) 703–719.
- [47] C. Pastori, C. Wahlestedt, Involvement of long noncoding RNAs in diseases affecting the central nervous system, RNA Biology 9 (2012) 860–870.
- [48] A. Troy, N.E. Sharpless, Genetic "Inc"-age of noncoding RNAs to human disease, The Journal of Clinical Investigation 122 (2012) 3837–3840.
- [49] D.S. Cabianca, V. Casa, B. Bodega, A. Xynos, E. Ginelli, Y. Tanaka, et al., A long ncRNA links copy number variation to a polycomb/trithorax epigenetic switch in FSHD muscular dystrophy, Cell 149 (2012) 819–831.
- [50] J.K. Millar, J.C. Wilson-Annan, S. Anderson, S. Christie, M.S. Taylor, C.A. Semple, et al., Disruption of two novel genes by a translocation co-segregating with schizophrenia, Human Molecular Genetics 9 (2000) 1415–1423.
- [51] G. Stefani, F.J. Slack, Small non-coding RNAs in animal development, Nature Reviews Molecular Cell Biology 9 (2008) 219–230.
- [52] E. van Rooij, E.N. Olson, MicroRNA therapeutics for cardiovascular disease: opportunities and obstacles, Nature Reviews Drug Discovery 11 (2012) 860–872.
- [53] A. Bonauer, G. Carmona, M. Iwasaki, M. Mione, M. Koyanagi, A. Fischer, et al., MicroRNA-92a controls angiogenesis and functional recovery of ischemic tissues in mice, Science 324 (2009) 1710–1713.
- [54] M. Xin, E.M. Small, L.B. Sutherland, X. Qi, J. McAnally, C.F. Plato, et al., MicroRNAs miR-143 and miR-145 modulate cytoskeletal dynamics and responsiveness of smooth muscle cells to injury, Genes & Development 23 (2009) 2166–2178.
- [55] T. Thum, C. Gross, J. Fiedler, T. Fischer, S. Kissler, M. Bussen, et al., MicroRNA-21 contributes to myocardial disease by stimulating MAP kinase signalling in fibroblasts, Nature 456 (2008) 980–984.
- [56] E. van Rooij, L.B. Sutherland, J.E. Thatcher, J.M. DiMaio, R.H. Naseem, W.S. Marshall, et al., Dysregulation of microRNAs after myocardial infarction reveals a role of miR-29 in cardiac fibrosis, Proceedings of the National Academy of Sciences of the United States of America 105 (2008) 13027–13032.

- [57] E. van Rooij, L.B. Sutherland, N. Liu, A.H. Williams, J. McAnally, R.D. Gerard, et al., A signature pattern of stress-responsive microRNAs that can evoke cardiac hypertrophy and heart failure, Proceedings of the National Academy of Sciences of the United States of America 103 (2006) 18255–18260.
- [58] M.S. Weinberg, M.J. Wood, Short non-coding RNA biology and neurodegenerative disorders: novel disease targets and therapeutics, Human Molecular Genetics 18 (2009) R27–R39.
- [59] A.G. Bader, miR-34-a microRNA replacement therapy is headed to the clinic, Frontiers in Genetics 3 (2012) 120.
- [60] J. Contreras, D.S. Rao, microRNAs in inflammation and immune responses, Leukemia 26 (2012) 404–413.
- [61] R.M. O'Connell, D.S. Rao, D. Baltimore, microRNA regulation of inflammatory responses, Annual Review of Immunology 30 (2012) 295–312.
- [62] G. Meister, M. Landthaler, A. Patkaniowska, Y. Dorsett, G. Teng, T. Tuschl, Human Argonaute2 mediates RNA cleavage targeted by miRNAs and siRNAs, Molecular Cell 15 (2004) 185–197.
- [63] A. Kozomara, S. Griffiths-Jones, miRBase: annotating high confidence microRNAs using deep sequencing data, Nucleic Acids Research 42 (2014) D68–D73.
- [64] G.M. Borchert, W. Lanier, B.L. Davidson, RNA polymerase III transcribes human microRNAs, Nature Structural & Molecular Biology 13 (2006) 1097–1101.
- [65] M. Jinek, J.A. Doudna, A three-dimensional view of the molecular machinery of RNA interference, Nature 457 (2009) 405–412.
- [66] V.N. Kim, J. Han, M.C. Siomi, Biogenesis of small RNAs in animals, Nature Reviews Molecular Cell Biology 10 (2009) 126–139.
- [67] Y. Zeng, B.R. Cullen, Structural requirements for pre-microRNA binding and nuclear export by Exportin 5, Nucleic Acids Research 32 (2004) 4776–4785.
- [68] S. Griffiths-Jones, J.H. Hui, A. Marco, M. Ronshaugen, MicroRNA evolution by arm switching, EMBO Reports 12 (2011) 172–177.
- [69] J.S. Yang, M.D. Phillips, D. Betel, P. Mu, A. Ventura, A.C. Siepel, et al., Widespread regulatory activity of vertebrate microRNA\* species, RNA 17 (2011) 312–326.
- [70] A. Marco, J.I. Macpherson, M. Ronshaugen, S. Griffiths-Jones, MicroRNAs from the same precursor have different targeting properties, Silence 3 (2012) 8.
- [71] E. Berezikov, W.J. Chung, J. Willis, E. Cuppen, E.C. Lai, Mammalian mirtron genes, Molecular Cell 28 (2007) 328–336.
- [72] K. Okamura, J.W. Hagen, H. Duan, D.M. Tyler, E.C. Lai, The mirtron pathway generates microRNAclass regulatory RNAs in Drosophila, Cell 130 (2007) 89–100.
- [73] J.G. Ruby, C.H. Jan, D.P. Bartel, Intronic microRNA precursors that bypass Drosha processing, Nature 448 (2007) 83–86.
- [74] G. Meister, Argonaute proteins: functional insights and emerging roles, Nature Reviews Genetics 14 (2013) 447–459.
- [75] A. Khvorova, A. Reynolds, S.D. Jayasena, Functional siRNAs and miRNAs exhibit strand bias, Cell 115 (2003) 209–216.
- [76] C.R. Faehnle, L. Joshua-Tor, Argonautes confront new small RNAs, Current Opinion in Chemical Biology 11 (2007) 569–577.
- [77] T. Sasaki, A. Shiohama, S. Minoshima, N. Shimizu, Identification of eight members of the Argonaute family in the human genome, Genomics 82 (2003) 323–330.
- [78] T.A. Farazi, S.A. Juranek, T. Tuschl, The growing catalog of small RNAs and their association with distinct Argonaute/Piwi family members, Development 135 (2008) 1201–1214.

- [79] K. Forstemann, M.D. Horwich, L. Wee, Y. Tomari, P.D. Zamore, Drosophila microRNAs are sorted into functionally distinct argonaute complexes after production by dicer-1, Cell 130 (2007) 287–297.
- [80] Y. Tomari, T. Du, P.D. Zamore, Sorting of Drosophila small silencing RNAs, Cell 130 (2007) 299–308.
- [81] H. Siomi, M.C. Siomi, On the road to reading the RNA-interference code, Nature 457 (2009) 396-404.
- [82] K. Saito, Y. Sakaguchi, T. Suzuki, T. Suzuki, H. Siomi, M.C. Siomi, Pimet, the Drosophila homolog of HEN1, mediates 2'-O-methylation of Piwi- interacting RNAs at their 3' ends, Genes & Development 21 (2007) 1603–1608.
- [83] S. Cheloufi, C.O. Dos Santos, M.M. Chong, G.J. Hannon, A dicer-independent miRNA biogenesis pathway that requires Ago catalysis, Nature 465 (2010) 584–589.
- [84] D. Cifuentes, H. Xue, D.W. Taylor, H. Patnode, Y. Mishima, S. Cheloufi, et al., A novel miRNA processing pathway independent of Dicer requires Argonaute2 catalytic activity, Science 328 (2010) 1694–1698.
- [85] M. Johnston, M.C. Geoffroy, A. Sobala, R. Hay, G. Hutvagner, HSP90 protein stabilizes unloaded argonaute complexes and microscopic P-bodies in human cells, Molecular Biology of the Cell 21 (2010) 1462–1469.
- [86] T. Iki, M. Yoshikawa, M. Nishikiori, M.C. Jaudal, E. Matsumoto-Yokoyama, I. Mitsuhara, et al., In vitro assembly of plant RNA-induced silencing complexes facilitated by molecular chaperone HSP90, Molecular Cell 39 (2010) 282–291.
- [87] S. Iwasaki, M. Kobayashi, M. Yoda, Y. Sakaguchi, S. Katsuma, T. Suzuki, et al., Hsc70/Hsp90 chaperone machinery mediates ATP-dependent RISC loading of small RNA duplexes, Molecular Cell 39 (2010) 292–299.
- [88] K.M. Creamer, J.F. Partridge, RITS-connecting transcription, RNA interference, and heterochromatin assembly in fission yeast, Wiley Interdisciplinary Reviews RNA 2 (2011) 632–646.
- [89] H. Guo, N.T. Ingolia, J.S. Weissman, D.P. Bartel, Mammalian microRNAs predominantly act to decrease target mRNA levels, Nature 466 (2010) 835–840.
- [90] R. Fukunaga, B.W. Han, J.H. Hung, J. Xu, Z. Weng, P.D. Zamore, Dicer partner proteins tune the length of mature miRNAs in flies and mammals, Cell 151 (2012) 533–546.
- [91] H.Y. Lee, J.A. Doudna, TRBP alters human precursor microRNA processing in vitro, RNA 18 (2012) 2012–2019.
- [92] S.I. Grewal, S.C. Elgin, Transcription and RNA interference in the formation of heterochromatin, Nature 447 (2007) 399–406.
- [93] A. Verdel, S. Jia, S. Gerber, T. Sugiyama, S. Gygi, S.I. Grewal, et al., RNAi-mediated targeting of heterochromatin by the RITS complex, Science 303 (2004) 672–676.
- [94] D.H. Kim, L.M. Villeneuve, K.V. Morris, J.J. Rossi, Argonaute-1 directs siRNA-mediated transcriptional gene silencing in human cells, Nature Structural & Molecular Biology 13 (2006) 793–797.
- [95] K. Sato, M.C. Siomi, Piwi-interacting RNAs: biological functions and biogenesis, Essays in Biochemistry 54 (2013) 39–52.
- [96] V.V. Vagin, A. Sigova, C. Li, H. Seitz, V. Gvozdev, P.D. Zamore, A distinct small RNA pathway silences selfish genetic elements in the germline, Science 313 (2006) 320–324.
- [97] A.A. Aravin, N.M. Naumova, A.V. Tulin, V.V. Vagin, Y.M. Rozovsky, V.A. Gvozdev, Double-stranded RNA-mediated silencing of genomic tandem repeats and transposable elements in the *D. mela-nogaster* germline, Current Biology 11 (2001) 1017–1027.
- [98] K. Saito, K.M. Nishida, T. Mori, Y. Kawamura, K. Miyoshi, T. Nagami, et al., Specific association of Piwi with rasiRNAs derived from retrotransposon and heterochromatic regions in the Drosophila genome, Genes & Development 20 (2006) 2214–2222.

- [99] H. Yin, H. Lin, An epigenetic activation role of Piwi and a Piwi-associated piRNA in *Drosophila melanogaster*, Nature 450 (2007) 304–308.
- [100] J. Brennecke, A.A. Aravin, A. Stark, M. Dus, M. Kellis, R. Sachidanandam, et al., Discrete small RNAgenerating loci as master regulators of transposon activity in Drosophila, Cell 128 (2007) 1089–1103.
- [101] N. Robine, N.C. Lau, S. Balla, Z. Jin, K. Okamura, S. Kuramochi-Miyagawa, et al., A broadly conserved pathway generates 3'UTR-directed primary piRNAs, Current Biology 19 (2009) 2066–2076.
- [102] J.C. Peng, H. Lin, Beyond transposons: the epigenetic and somatic functions of the Piwi-piRNA mechanism, Current Opinion in Cell Biology 25 (2013) 190–194.
- [103] D.N. Cox, A. Chao, J. Baker, L. Chang, D. Qiao, H. Lin, A novel class of evolutionarily conserved genes defined by piwi are essential for stem cell self-renewal, Genes & Development 12 (1998) 3715–3727.
- [104] G. Wang, V. Reinke, A C. elegans Piwi, PRG-1, regulates 21U-RNAs during spermatogenesis, Current Biology 18 (2008) 861–867.
- [105] M.A. Carmell, A. Girard, H.J. van de Kant, D. Bourc'his, T.H. Bestor, D.G. de Rooij, et al., MIWI2 is essential for spermatogenesis and repression of transposons in the mouse male germline, Developmental Cell 12 (2007) 503–514.
- [106] S. Kuramochi-Miyagawa, T. Kimura, T.W. Ijiri, T. Isobe, N. Asada, Y. Fujita, et al., Mili, a mammalian member of piwi family gene, is essential for spermatogenesis, Development 131 (2004) 839–849.
- [107] W. Deng, H. Lin, miwi, a murine homolog of piwi, encodes a cytoplasmic protein essential for spermatogenesis, Developmental Cell 2 (2002) 819–830.
- [108] Y. Chen, A. Pane, T. Schupbach, Cutoff and aubergine mutations result in retrotransposon upregulation and checkpoint activation in Drosophila, Current Biology 17 (2007) 637–642.
- [109] S. Chambeyron, A. Popkova, G. Payen-Groschene, C. Brun, D. Laouini, A. Pelisson, et al., piRNAmediated nuclear accumulation of retrotransposon transcripts in the Drosophila female germline, Proceedings of the National Academy of Sciences of the United States of America 105 (2008) 14964–14969.
- [110] C. Klattenhoff, D.P. Bratu, N. McGinnis-Schultz, B.S. Koppetsch, H.A. Cook, W.E. Theurkauf, Drosophila rasiRNA pathway mutations disrupt embryonic axis specification through activation of an ATR/Chk2 DNA damage response, Developmental Cell 12 (2007) 45–55.
- [111] J.S. Khurana, W. Theurkauf, piRNAs, transposon silencing, and Drosophila germline development, The Journal of Cell Biology 191 (2010) 905–913.
- [112] C. Klattenhoff, H. Xi, C. Li, S. Lee, J. Xu, J.S. Khurana, et al., The Drosophila HP1 homolog Rhino is required for transposon silencing and piRNA production by dual-strand clusters, Cell 138 (2009) 1137–1149.
- [113] R.J. Ross, M.M. Weiner, H. Lin, PIWI proteins and PIWI-interacting RNAs in the soma, Nature 505 (2014) 353–359.
- [114] S. Houwing, L.M. Kamminga, E. Berezikov, D. Cronembold, A. Girard, H. van den Elst, et al., A role for Piwi and piRNAs in germ cell maintenance and transposon silencing in Zebrafish, Cell 129 (2007) 69–82.
- [115] T. Ohara, Y. Sakaguchi, T. Suzuki, H. Ueda, K. Miyauchi, T. Suzuki, The 3' termini of mouse Piwiinteracting RNAs are 2'-O-methylated, Nature Structural & Molecular Biology 14 (2007) 349–350.
- [116] S. Kawaoka, N. Izumi, S. Katsuma, Y. Tomari, 3' end formation of PIWI-interacting RNAs in vitro, Molecular Cell 43 (2011) 1015–1022.

- [117] K.A. Senti, J. Brennecke, The piRNA pathway: a fly's perspective on the guardian of the genome, Trends in Genetics 26 (2010) 499–509.
- [118] L.S. Gunawardane, K. Saito, K.M. Nishida, K. Miyoshi, Y. Kawamura, T. Nagami, et al., A slicermediated mechanism for repeat-associated siRNA 5' end formation in Drosophila, Science 315 (2007) 1587–1590.
- [119] H. Ishizu, H. Siomi, M.C. Siomi, Biology of PIWI-interacting RNAs: new insights into biogenesis and function inside and outside of germlines, Genes & Development 26 (2012) 2361–2373.
- [120] M.J. Luteijn, R.F. Ketting, PIWI-interacting RNAs: from generation to transgenerational epigenetics, Nature Reviews Genetics 14 (2013) 523–534.
- [121] S.I. Grewal, S. Jia, Heterochromatin revisited, Nature Reviews Genetics 8 (2007) 35-46.
- [122] C. Keller, R. Adaixo, R. Stunnenberg, K.J. Woolcock, S. Hiller, M. Buhler, HP1(Swi6) mediates the recognition and destruction of heterochromatic RNA transcripts, Molecular Cell 47 (2012) 215–227.
- [123] K.A. al-Mukhtar, A.C. Webb, An ultrastructural study of primordial germ cells, oogonia and early oocytes in *Xenopus laevis*, Journal of Embryology and Experimental Morphology 26 (1971) 195–217.
- [124] R.F. Ketting, The many faces of RNAi, Developmental Cell 20 (2011) 148-161.
- [125] F. Voigt, M. Reuter, A. Kasaruho, E.C. Schulz, R.S. Pillai, O. Barabas, Crystal structure of the primary piRNA biogenesis factor Zucchini reveals similarity to the bacterial PLD endonuclease Nuc, RNA 18 (2012) 2128–2134.
- [126] H. Nishimasu, H. Ishizu, K. Saito, S. Fukuhara, M.K. Kamatani, L. Bonnefond, et al., Structure and function of Zucchini endoribonuclease in piRNA biogenesis, Nature 491 (2012) 284–287.
- [127] J.J. Ipsaro, A.D. Haase, S.R. Knott, L. Joshua-Tor, G.J. Hannon, The structural biochemistry of Zucchini implicates it as a nuclease in piRNA biogenesis, Nature 491 (2012) 279–283.
- [128] A. Pane, K. Wehr, T. Schupbach, Zucchini and squash encode two putative nucleases required for rasiRNA production in the Drosophila germline, Developmental Cell 12 (2007) 851–862.
- [129] D. Olivieri, K.A. Senti, S. Subramanian, R. Sachidanandam, J. Brennecke, The cochaperone shutdown defines a group of biogenesis factors essential for all piRNA populations in Drosophila, Molecular Cell 47 (2012) 954–969.
- [130] K. Munn, R. Steward, The shut-down gene of *Drosophila melanogaster* encodes a novel FK506binding protein essential for the formation of germline cysts during oogenesis, Genetics 156 (2000) 245–256.
- [131] S.L. Ameres, M.D. Horwich, J.H. Hung, J. Xu, M. Ghildiyal, Z. Weng, et al., Target RNA-directed trimming and tailing of small silencing RNAs, Science 328 (2010) 1534–1539.
- [132] S.L. Ameres, J.H. Hung, J. Xu, Z. Weng, P.D. Zamore, Target RNA-directed tailing and trimming purifies the sorting of endo-siRNAs between the two Drosophila Argonaute proteins, RNA 17 (2011) 54–63.
- [133] S. Siddiqi, I. Matushansky, Piwis and piwi-interacting RNAs in the epigenetics of cancer, Journal of Cellular Biochemistry 113 (2012) 373–380.
- [134] C. Rouget, C. Papin, A. Boureux, A.C. Meunier, B. Franco, N. Robine, et al., Maternal mRNA deadenylation and decay by the piRNA pathway in the early Drosophila embryo, Nature 467 (2010) 1128–1132.
- [135] E.J. Lee, S. Banerjee, H. Zhou, A. Jammalamadaka, M. Arcila, B.S. Manjunath, et al., Identification of piRNAs in the central nervous system, RNA 17 (2011) 1090–1099.
- [136] G. Sienski, D. Donertas, J. Brennecke, Transcriptional silencing of transposons by Piwi and maelstrom and its impact on chromatin state and gene expression, Cell 151 (2012) 964–980.

- [137] A. Akkouche, T. Grentzinger, M. Fablet, C. Armenise, N. Burlet, V. Braman, et al., Maternally deposited germline piRNAs silence the tirant retrotransposon in somatic cells, EMBO Reports 14 (2013) 458–464.
- [138] T. Gu, S.C. Elgin, Maternal depletion of Piwi, a component of the RNAi system, impacts heterochromatin formation in Drosophila, PLoS Genetics 9 (2013) e1003780.
- [139] P. Rajasethupathy, I. Antonov, R. Sheridan, S. Frey, C. Sander, T. Tuschl, et al., A role for neuronal piRNAs in the epigenetic control of memory-related synaptic plasticity, Cell 149 (2012) 693–707.
- [140] X.A. Huang, H. Yin, S. Sweeney, D. Raha, M. Snyder, H. Lin, A major epigenetic programming mechanism guided by piRNAs, Developmental Cell 24 (2013) 502–516.
- [141] T. Watanabe, S. Tomizawa, K. Mitsuya, Y. Totoki, Y. Yamamoto, S. Kuramochi-Miyagawa, et al., Role for piRNAs and noncoding RNA in de novo DNA methylation of the imprinted mouse Rasgrf1 locus, Science 332 (2011) 848–852.
- [142] J. Baguna, E. Salo, R. Romero, Effects of activators and antagonists of the neuropeptides substance P and substance K on cell proliferation in planarians, The International Journal of Developmental Biology 33 (1989) 261–266.
- [143] Y. Wang, R.M. Zayas, T. Guo, P.A. Newmark, Nanos function is essential for development and regeneration of planarian germ cells, Proceedings of the National Academy of Sciences of the United States of America 104 (2007) 5901–5906.
- [144] D. Palakodeti, M. Smielewska, Y.C. Lu, G.W. Yeo, B.R. Graveley, The PIWI proteins SMEDWI-2 and SMEDWI-3 are required for stem cell function and piRNA expression in planarians, RNA 14 (2008) 1174–1186.
- [145] P.W. Reddien, N.J. Oviedo, J.R. Jennings, J.C. Jenkin, A. Sanchez Alvarado, SMEDWI-2 is a PIWI-like protein that regulates planarian stem cells, Science 310 (2005) 1327–1330.
- [146] Y. Rinkevich, A. Rosner, C. Rabinowitz, Z. Lapidot, E. Moiseeva, B. Rinkevich, Piwi positive cells that line the vasculature epithelium, underlie whole body regeneration in a basal chordate, Developmental Biology 345 (2010) 94–104.
- [147] Y. Rinkevich, A. Voskoboynik, A. Rosner, C. Rabinowitz, G. Paz, M. Oren, et al., Repeated, long-term cycling of putative stem cells between niches in a basal chordate, Developmental Cell 24 (2013) 76–88.
- [148] A. Janic, L. Mendizabal, S. Llamazares, D. Rossell, C. Gonzalez, Ectopic expression of germline genes drives malignant brain tumor growth in Drosophila, Science 330 (2010) 1824–1827.
- [149] P. Johnsson, L. Lipovich, D. Grander, K.V. Morris, Evolutionary conservation of long non-coding RNAs; sequence, structure, function, Biochimica et Biophysica Acta 1840 (2014) 1063–1071.
- [150] G. St Laurent, C. Wahlestedt, P. Kapranov, The landscape of long noncoding RNA classification, Trends in Genetics 31 (2015) 239–251.
- [151] M. Magistri, M.A. Faghihi, G. St Laurent 3rd, C. Wahlestedt, Regulation of chromatin structure by long noncoding RNAs: focus on natural antisense transcripts, Trends in Genetics 28 (2012) 389–396.
- [152] M. Guttman, I. Amit, M. Garber, C. French, M.F. Lin, D. Feldser, et al., Chromatin signature reveals over a thousand highly conserved large non-coding RNAs in mammals, Nature 458 (2009) 223–227.
- [153] U.A. Orom, T. Derrien, M. Beringer, K. Gumireddy, A. Gardini, G. Bussotti, et al., Long noncoding RNAs with enhancer-like function in human cells, Cell 143 (2010) 46–58.
- [154] T.K. Kim, M. Hemberg, J.M. Gray, A.M. Costa, D.M. Bear, J. Wu, et al., Widespread transcription at neuronal activity-regulated enhancers, Nature 465 (2010) 182–187.

- [155] G. St Laurent, D. Shtokalo, M.R. Tackett, Z. Yang, T. Eremina, C. Wahlestedt, et al., Intronic RNAs constitute the major fraction of the non-coding RNA in mammalian cells, BMC Genomics 13 (2012) 504.
- [156] S. Katayama, Y. Tomaru, T. Kasukawa, K. Waki, M. Nakanishi, M. Nakamura, et al., Antisense transcription in the mammalian transcriptome, Science 309 (2005) 1564–1566.
- [157] P.G. Engstrom, H. Suzuki, N. Ninomiya, A. Akalin, L. Sessa, G. Lavorgna, et al., Complex loci in human and mouse genomes, PLoS Genetics 2 (2006) e47.
- [158] S.A. Korneev, E.I. Korneeva, M.A. Lagarkova, S.L. Kiselev, G. Critchley, M. O'Shea, Novel noncoding antisense RNA transcribed from human anti-NOS2A locus is differentially regulated during neuronal differentiation of embryonic stem cells, RNA 14 (2008) 2030–2037.
- [159] G. Lavorgna, D. Dahary, B. Lehner, R. Sorek, C.M. Sanderson, G. Casari, In search of antisense, Trends in Biochemical Sciences 29 (2004) 88–94.
- [160] A.C. Marques, C.P. Ponting, Catalogues of mammalian long noncoding RNAs: modest conservation and incompleteness, Genome Biology 10 (2009) R124.
- [161] J. Ponjavic, C.P. Ponting, G. Lunter, Functionality or transcriptional noise? Evidence for selection within long noncoding RNAs, Genome Research 17 (2007) 556–565.
- [162] F. De Lucia, C. Dean, Long non-coding RNAs and chromatin regulation, Current Opinion in Plant Biology 14 (2011) 168–173.
- [163] R.C. Lee, R.L. Feinbaum, V. Ambros, The C. elegans heterochronic gene lin-4 encodes small RNAs with antisense complementarity to lin-14, Cell 75 (1993) 843–854.
- [164] D.P. Bartel, MicroRNAs: target recognition and regulatory functions, Cell 136 (2009) 215–233.
- [165] R.C. Friedman, K.K. Farh, C.B. Burge, D.P. Bartel, Most mammalian mRNAs are conserved targets of microRNAs, Genome Research 19 (2009) 92–105.
- [166] Q. Jing, S. Huang, S. Guth, T. Zarubin, A. Motoyama, J. Chen, et al., Involvement of microRNA in AU-rich element-mediated mRNA instability, Cell 120 (2005) 623–634.
- [167] C. Llave, Z. Xie, K.D. Kasschau, J.C. Carrington, Cleavage of Scarecrow-like mRNA targets directed by a class of Arabidopsis miRNA, Science 297 (2002) 2053–2056.
- [168] D.H. Kim, P. Saetrom, O. Snove Jr., J.J. Rossi, MicroRNA-directed transcriptional gene silencing in mammalian cells, Proceedings of the National Academy of Sciences of the United States of America 105 (2008) 16230–16235.
- [169] J. Liu, M.A. Valencia-Sanchez, G.J. Hannon, R. Parker, MicroRNA-dependent localization of targeted mRNAs to mammalian P-bodies, Nature Cell Biology 7 (2005) 719–723.
- [170] R.S. Pillai, S.N. Bhattacharyya, C.G. Artus, T. Zoller, N. Cougot, E. Basyuk, et al., Inhibition of translational initiation by Let-7 MicroRNA in human cells, Science 309 (2005) 1573–1576.
- [171] B. Wang, A. Yanez, C.D. Novina, MicroRNA-repressed mRNAs contain 40S but not 60S components, Proceedings of the National Academy of Sciences of the United States of America 105 (2008) 5343–5348.
- [172] P.A. Maroney, Y. Yu, J. Fisher, T.W. Nilsen, Evidence that microRNAs are associated with translating messenger RNAs in human cells, Nature Structural & Molecular Biology 13 (2006) 1102–1107.
- [173] C.P. Petersen, M.E. Bordeleau, J. Pelletier, P.A. Sharp, Short RNAs repress translation after initiation in mammalian cells, Molecular Cell 21 (2006) 533–542.
- [174] J.R. Lytle, T.A. Yario, J.A. Steitz, Target mRNAs are repressed as efficiently by microRNA-binding sites in the 5' UTR as in the 3' UTR, Proceedings of the National Academy of Sciences of the United States of America 104 (2007) 9667–9672.

- [175] R.F. Place, L.C. Li, D. Pookot, E.J. Noonan, R. Dahiya, MicroRNA-373 induces expression of genes with complementary promoter sequences, Proceedings of the National Academy of Sciences of the United States of America 105 (2008) 1608–1613.
- [176] L.C. Li, S.T. Okino, H. Zhao, D. Pookot, R.F. Place, S. Urakami, et al., Small dsRNAs induce transcriptional activation in human cells, Proceedings of the National Academy of Sciences of the United States of America 103 (2006) 17337–17342.
- [177] U.A. Orom, F.C. Nielsen, A.H. Lund, MicroRNA-10a binds the 5' UTR of ribosomal protein mRNAs and enhances their translation, Molecular Cell 30 (2008) 460–471.
- [178] S. Vasudevan, J.A. Steitz, AU-rich-element-mediated upregulation of translation by FXR1 and Argonaute 2, Cell 128 (2007) 1105–1118.
- [179] S. Lee, S. Vasudevan, Post-transcriptional stimulation of gene expression by microRNAs, Advances in Experimental Medicine and Biology 768 (2013) 97–126.
- [180] C.Z. Chen, L. Li, H.F. Lodish, D.P. Bartel, MicroRNAs modulate hematopoietic lineage differentiation, Science 303 (2004) 83–86.
- [181] D. O'Carroll, I. Mecklenbrauker, P.P. Das, A. Santana, U. Koenig, A.J. Enright, et al., A slicer-independent role for Argonaute 2 in hematopoiesis and the microRNA pathway, Genes & Development 21 (2007) 1999–2004.
- [182] M. Haneklaus, M. Gerlic, M. Kurowska-Stolarska, A.A. Rainey, D. Pich, I.B. McInnes, et al., Cutting edge: miR-223 and EBV miR-BART15 regulate the NLRP3 inflammasome and IL-1beta production, Journal of Immunology 189 (2012) 3795–3799.
- [183] S.B. Koralov, S.A. Muljo, G.R. Galler, A. Krek, T. Chakraborty, C. Kanellopoulou, et al., Dicer ablation affects antibody diversity and cell survival in the B lymphocyte lineage, Cell 132 (2008) 860–874.
- [184] B.S. Cobb, T.B. Nesterova, E. Thompson, A. Hertweck, E. O'Connor, J. Godwin, et al., T cell lineage choice and differentiation in the absence of the RNase III enzyme Dicer, The Journal of Experimental Medicine 201 (2005) 1367–1373.
- [185] S.A. Muljo, K.M. Ansel, C. Kanellopoulou, D.M. Livingston, A. Rao, K. Rajewsky, Aberrant T cell differentiation in the absence of Dicer, The Journal of Experimental Medicine 202 (2005) 261–269.
- [186] X. Zhou, L.T. Jeker, B.T. Fife, S. Zhu, M.S. Anderson, M.T. McManus, et al., Selective miRNA disruption in T reg cells leads to uncontrolled autoimmunity, The Journal of Experimental Medicine 205 (2008) 1983–1991.
- [187] A. Liston, L.F. Lu, D. O'Carroll, A. Tarakhovsky, A.Y. Rudensky, Dicer-dependent microRNA pathway safeguards regulatory T cell function, The Journal of Experimental Medicine 205 (2008) 1993–2004.
- [188] M.M. Chong, J.P. Rasmussen, A.Y. Rudensky, D.R. Littman, The RNAseIII enzyme Drosha is critical in T cells for preventing lethal inflammatory disease, The Journal of Experimental Medicine 205 (2008) 2005–2017.
- [189] A. Rodriguez, E. Vigorito, S. Clare, M.V. Warren, P. Couttet, D.R. Soond, et al., Requirement of bic/ microRNA-155 for normal immune function, Science 316 (2007) 608–611.
- [190] T.H. Thai, D.P. Calado, S. Casola, K.M. Ansel, C. Xiao, Y. Xue, et al., Regulation of the germinal center response by microRNA-155, Science 316 (2007) 604–608.
- [191] N. Felli, L. Fontana, E. Pelosi, R. Botta, D. Bonci, F. Facchiano, et al., MicroRNAs 221 and 222 inhibit normal erythropoiesis and erythroleukemic cell growth via kit receptor down-modulation, Proceedings of the National Academy of Sciences of the United States of America 102 (2005) 18081–18086.

- [192] F. Fazi, A. Rosa, A. Fatica, V. Gelmetti, M.L. De Marchis, C. Nervi, et al., A minicircuitry comprised of microRNA-223 and transcription factors NFI-A and C/EBPalpha regulates human granulopoiesis, Cell 123 (2005) 819–831.
- [193] T. Fukao, Y. Fukuda, K. Kiga, J. Sharif, K. Hino, Y. Enomoto, et al., An evolutionarily conserved mechanism for microRNA-223 expression revealed by microRNA gene profiling, Cell 129 (2007) 617–631.
- [194] R.M. O'Connell, K.D. Taganov, M.P. Boldin, G. Cheng, D. Baltimore, MicroRNA-155 is induced during the macrophage inflammatory response, Proceedings of the National Academy of Sciences of the United States of America 104 (2007) 1604–1609.
- [195] K.D. Taganov, M.P. Boldin, K.J. Chang, D. Baltimore, NF-kappaB-dependent induction of microRNA miR-146, an inhibitor targeted to signaling proteins of innate immune responses, Proceedings of the National Academy of Sciences of the United States of America 103 (2006) 12481–12486.
- [196] Q.J. Li, J. Chau, P.J. Ebert, G. Sylvester, H. Min, G. Liu, et al., miR-181a is an intrinsic modulator of T cell sensitivity and selection, Cell 129 (2007) 147–161.
- [197] C. Du, C. Liu, J. Kang, G. Zhao, Z. Ye, S. Huang, et al., MicroRNA miR-326 regulates TH-17 differentiation and is associated with the pathogenesis of multiple sclerosis, Nature Immunology 10 (2009) 1252–1259.
- [198] J.C. Dudda, B. Salaun, Y. Ji, D.C. Palmer, G.C. Monnot, E. Merck, et al., MicroRNA-155 is required for effector CD8<sup>+</sup> T cell responses to virus infection and cancer, Immunity 38 (2013) 742–753.
- [199] O. Hobert, Gene regulation by transcription factors and microRNAs, Science 319 (2008) 1785–1786.
- [200] L.A. O'Neill, F.J. Sheedy, C.E. McCoy, MicroRNAs: the fine-tuners of Toll-like receptor signalling, Nature Reviews Immunology 11 (2011) 163–175.
- [201] F.J. Sheedy, E. Palsson-McDermott, E.J. Hennessy, C. Martin, J.J. O'Leary, Q. Ruan, et al., Negative regulation of TLR4 via targeting of the proinflammatory tumor suppressor PDCD4 by the microRNA miR-21, Nature Immunology 11 (2010) 141–147.
- [202] C. Xiao, D.P. Calado, G. Galler, T.H. Thai, H.C. Patterson, J. Wang, et al., MiR-150 controls B cell differentiation by targeting the transcription factor c-Myb, Cell 131 (2007) 146–159.
- [203] E. Vigorito, K.L. Perks, C. Abreu-Goodger, S. Bunting, Z. Xiang, S. Kohlhaas, et al., microRNA-155 regulates the generation of immunoglobulin class-switched plasma cells, Immunity 27 (2007) 847–859.
- [204] Y. Dorsett, K.M. McBride, M. Jankovic, A. Gazumyan, T.H. Thai, D.F. Robbiani, et al., MicroRNA-155 suppresses activation-induced cytidine deaminase-mediated Myc-Igh translocation, Immunity 28 (2008) 630–638.
- [205] A. Ventura, A.G. Young, M.M. Winslow, L. Lintault, A. Meissner, S.J. Erkeland, et al., Targeted deletion reveals essential and overlapping functions of the miR-17 through 92 family of miRNA clusters, Cell 132 (2008) 875–886.
- [206] C. Xiao, L. Srinivasan, D.P. Calado, H.C. Patterson, B. Zhang, J. Wang, et al., Lymphoproliferative disease and autoimmunity in mice with increased miR-17-92 expression in lymphocytes, Nature Immunology 9 (2008) 405–414.
- [207] R. Haque, E. Chun, J.C. Howell, T. Sengupta, D. Chen, H. Kim, MicroRNA-30b-mediated regulation of catalase expression in human ARPE-19 cells, PLoS One 7 (2012) e42542.
- [208] X. Xu, A. Wells, M.T. Padilla, K. Kato, K.C. Kim, Y. Lin, A signaling pathway consisting of miR-551b, catalase and MUC1 contributes to acquired apoptosis resistance and chemoresistance, Carcinogenesis 35 (2014) 2457–2466.

- [209] X. Zhang, W.L. Ng, P. Wang, L. Tian, E. Werner, H. Wang, et al., MicroRNA-21 modulates the levels of reactive oxygen species by targeting SOD3 and TNFalpha, Cancer Research 72 (2012) 4707–4713.
- [210] F. Fleissner, V. Jazbutyte, J. Fiedler, S.K. Gupta, X. Yin, Q. Xu, et al., Short communication: asymmetric dimethylarginine impairs angiogenic progenitor cell function in patients with coronary artery disease through a microRNA-21-dependent mechanism, Circulation Research 107 (2010) 138–143.
- [211] P. Liu, H. Zhao, R. Wang, P. Wang, Z. Tao, L. Gao, et al., MicroRNA-424 protects against focal cerebral ischemia and reperfusion injury in mice by suppressing oxidative stress, Stroke; A Journal of Cerebral Circulation 46 (2015) 513–519.
- [212] Y. Suarez, C. Fernandez-Hernando, J.S. Pober, W.C. Sessa, Dicer dependent microRNAs regulate gene expression and functions in human endothelial cells, Circulation Research 100 (2007) 1164–1173.
- [213] H.X. Sun, D.Y. Zeng, R.T. Li, R.P. Pang, H. Yang, Y.L. Hu, et al., Essential role of microRNA-155 in regulating endothelium-dependent vasorelaxation by targeting endothelial nitric oxide synthase, Hypertension 60 (2012) 1407–1414.
- [214] W. Zhang, L. Yan, Y. Li, W. Chen, N. Hu, H. Wang, et al., Roles of miRNA-24 in regulating endothelial nitric oxide synthase expression and vascular endothelial cell proliferation, Molecular and Cellular Biochemistry 405 (2015) 281–289.
- [215] W. Yang, T.P. Chendrimada, Q. Wang, M. Higuchi, P.H. Seeburg, R. Shiekhattar, et al., Modulation of microRNA processing and expression through RNA editing by ADAR deaminases, Nature Structural & Molecular Biology 13 (2006) 13–21.
- [216] J.P. Hagan, E. Piskounova, R.I. Gregory, Lin28 recruits the TUTase Zcchc11 to inhibit let-7 maturation in mouse embryonic stem cells, Nature Structural & Molecular Biology 16 (2009) 1021–1025.
- [217] I. Heo, C. Joo, Y.K. Kim, M. Ha, M.J. Yoon, J. Cho, et al., TUT4 in concert with Lin28 suppresses microRNA biogenesis through pre-microRNA uridylation, Cell 138 (2009) 696–708.
- [218] A.K. Leung, J.M. Calabrese, P.A. Sharp, Quantitative analysis of Argonaute protein reveals microRNA-dependent localization to stress granules, Proceedings of the National Academy of Sciences of the United States of America 103 (2006) 18125–18130.
- [219] H.K. Saini, S. Griffiths-Jones, A.J. Enright, Genomic analysis of human microRNA transcripts, Proceedings of the National Academy of Sciences of the United States of America 104 (2007) 17719–17724.
- [220] C. Baer, R. Claus, C. Plass, Genome-wide epigenetic regulation of miRNAs in cancer, Cancer Research 73 (2013) 473–477.
- [221] B.J. Kroesen, N. Teteloshvili, K. Smigielska-Czepiel, E. Brouwer, A.M. Boots, A. van den Berg, et al., Immuno-miRs: critical regulators of T-cell development, function and ageing, Immunology 144 (2015) 1–10.
- [222] H. Takahashi, T. Kanno, S. Nakayamada, K. Hirahara, G. Sciume, S.A. Muljo, et al., TGF-beta and retinoic acid induce the microRNA miR-10a, which targets Bcl-6 and constrains the plasticity of helper T cells, Nature Immunology 13 (2012) 587–595.
- [223] P.A. Chandran, A. Keller, L. Weinmann, A.A. Seida, M. Braun, K. Andreev, et al., The TGF-betainducible miR-23a cluster attenuates IFN-gamma levels and antigen-specific cytotoxicity in human CD8(+) T cells, Journal of Leukocyte Biology 96 (2014) 633–645.
- [224] L.M. Das, M.D. Torres-Castillo, T. Gill, A.D. Levine, TGF-beta conditions intestinal T cells to express increased levels of miR-155, associated with down-regulation of IL-2 and itk mRNA, Mucosal Immunology 6 (2013) 167–176.

- [225] B.N. Davis, A.C. Hilyard, G. Lagna, A. Hata, SMAD proteins control DROSHA-mediated microRNA maturation, Nature 454 (2008) 56–61.
- [226] M. Chekulaeva, W. Filipowicz, Mechanisms of miRNA-mediated post-transcriptional regulation in animal cells, Current Opinion in Cell Biology 21 (2009) 452–460.
- [227] M. Trabucchi, P. Briata, M. Garcia-Mayoral, A.D. Haase, W. Filipowicz, A. Ramos, et al., The RNAbinding protein KSRP promotes the biogenesis of a subset of microRNAs, Nature 459 (2009) 1010–1014.
- [228] S.R. Viswanathan, J.T. Powers, W. Einhorn, Y. Hoshida, T.L. Ng, S. Toffanin, et al., Lin28 promotes transformation and is associated with advanced human malignancies, Nature Genetics 41 (2009) 843–848.
- [229] A.B. Stittrich, C. Haftmann, E. Sgouroudis, A.A. Kuhl, A.N. Hegazy, I. Panse, et al., The microRNA miR-182 is induced by IL-2 and promotes clonal expansion of activated helper T lymphocytes, Nature Immunology 11 (2010) 1057–1062.
- [230] J.J. Tanner, Z.D. Parsons, A.H. Cummings, H. Zhou, K.S. Gates, Redox regulation of protein tyrosine phosphatases: structural and chemical aspects, Antioxidants & Redox Signaling 15 (2011) 77–97.
- [231] T.H. Truong, K.S. Carroll, Redox regulation of protein kinases, Critical Reviews in Biochemistry and Molecular Biology 48 (2013) 332–356.
- [232] H.S. Marinho, C. Real, L. Cyrne, H. Soares, F. Antunes, Hydrogen peroxide sensing, signaling and regulation of transcription factors, Redox Biology 2 (2014) 535–562.
- [233] T.M. Rana, Illuminating the silence: understanding the structure and function of small RNAs, Nature Reviews Molecular Cell Biology 8 (2007) 23–36.
- [234] X. Tang, M. Li, L. Tucker, B. Ramratnam, Glycogen synthase kinase 3 beta (GSK3beta) phosphorylates the RNAase III enzyme Drosha at S300 and S302, PLoS One 6 (2011) e20391.
- [235] X. Tang, Y. Zhang, L. Tucker, B. Ramratnam, Phosphorylation of the RNase III enzyme Drosha at Serine300 or Serine302 is required for its nuclear localization, Nucleic Acids Research 38 (2010) 6610–6619.
- [236] D.A. Cross, D.R. Alessi, P. Cohen, M. Andjelkovich, B.A. Hemmings, Inhibition of glycogen synthase kinase-3 by insulin mediated by protein kinase B, Nature 378 (1995) 785–789.
- [237] M. Shaw, P. Cohen, D.R. Alessi, The activation of protein kinase B by H<sub>2</sub>O<sub>2</sub> or heat shock is mediated by phosphoinositide 3-kinase and not by mitogen-activated protein kinase-activated protein kinase-2, Biochemical Journal 336 (Pt 1) (1998) 241–246.
- [238] A.S. Blair, E. Hajduch, G.J. Litherland, H.S. Hundal, Regulation of glucose transport and glycogen synthesis in L6 muscle cells during oxidative stress. Evidence for cross-talk between the insulin and SAPK2/p38 mitogen-activated protein kinase signaling pathways, The Journal of Biological Chemistry 274 (1999) 36293–36299.
- [239] I. Barr, A.T. Smith, Y. Chen, R. Senturia, J.N. Burstyn, F. Guo, Ferric, not ferrous, heme activates RNA-binding protein DGCR8 for primary microRNA processing, Proceedings of the National Academy of Sciences of the United States of America 109 (2012) 1919–1924.
- [240] V. de Oliveira-Marques, L. Cyrne, H.S. Marinho, F. Antunes, A quantitative study of NF-kappaB activation by H<sub>2</sub>O<sub>2</sub>: relevance in inflammation and synergy with TNF-alpha, Journal of Immunology 178 (2007) 3893–3902.
- [241] L. De Franceschi, M. Bertoldi, L. De Falco, S. Santos Franco, L. Ronzoni, F. Turrini, et al., Oxidative stress modulates heme synthesis and induces peroxiredoxin-2 as a novel cytoprotective response in beta-thalassemic erythropoiesis, Haematologica 96 (2011) 1595–1604.
- [242] G. Covas, H.S. Marinho, L. Cyrne, F. Antunes, Activation of Nrf2 by H<sub>2</sub>O<sub>2</sub>: de novo synthesis versus nuclear translocation, Methods in Enzymology 528 (2013) 157–171.

- [243] M. Kozakowska, K. Szade, J. Dulak, A. Jozkowicz, Role of heme oxygenase-1 in postnatal differentiation of stem cells: a possible cross-talk with microRNAs, Antioxidants & Redox Signaling 20 (2014) 1827–1850.
- [244] Z. Ungvari, Z. Tucsek, D. Sosnowska, P. Toth, T. Gautam, A. Podlutsky, et al., Aging-induced dysregulation of dicer1-dependent microRNA expression impairs angiogenic capacity of rat cerebromicrovascular endothelial cells, Journals of Gerontology Series A: Biological Sciences and Medical Sciences 68 (2013) 877–891.
- [245] M.A. Mori, P. Raghavan, T. Thomou, J. Boucher, S. Robida-Stubbs, Y. Macotela, et al., Role of microRNA processing in adipose tissue in stress defense and longevity, Cell Metabolism 16 (2012) 336–347.
- [246] J.L. Wiesen, T.B. Tomasi, Dicer is regulated by cellular stresses and interferons, Molecular Immunology 46 (2009) 1222–1228.
- [247] X. Cheng, C.H. Ku, R.C. Siow, Regulation of the Nrf2 antioxidant pathway by microRNAs: new players in micromanaging redox homeostasis, Free Radical Biology & Medicine 64 (2013) 4–11.
- [248] S. Tokumaru, M. Suzuki, H. Yamada, M. Nagino, T. Takahashi, let-7 regulates Dicer expression and constitutes a negative feedback loop, Carcinogenesis 29 (2008) 2073–2077.
- [249] J.J. Forman, A. Legesse-Miller, H.A. Coller, A search for conserved sequences in coding regions reveals that the let-7 microRNA targets Dicer within its coding sequence, Proceedings of the National Academy of Sciences of the United States of America 105 (2008) 14879–14884.
- [250] Y. Lin, X. Liu, Y. Cheng, J. Yang, Y. Huo, C. Zhang, Involvement of MicroRNAs in hydrogen peroxide-mediated gene regulation and cellular injury response in vascular smooth muscle cells, The Journal of Biological Chemistry 284 (2009) 7903–7913.
- [251] Y. Li, H. Shelat, Y.J. Geng, IGF-1 prevents oxidative stress induced-apoptosis in induced pluripotent stem cells which is mediated by microRNA-1, Biochemical and Biophysical Research Communications 426 (2012) 615–619.
- [252] T. Chen, Z. Li, J. Yan, X. Yang, W. Salminen, MicroRNA expression profiles distinguish the carcinogenic effects of riddelliine in rat liver, Mutagenesis 27 (2012) 59–66.
- [253] S. Chettimada, H. Ata, D.K. Rawat, S. Gulati, A.G. Kahn, J.G. Edwards, et al., Contractile protein expression is upregulated by reactive oxygen species in aorta of Goto-Kakizaki rat, American Journal of Physiology - Heart and Circulatory Physiology 306 (2014) H214–H224.
- [254] R. Li, G. Yan, Q. Li, H. Sun, Y. Hu, J. Sun, et al., MicroRNA-145 protects cardiomyocytes against hydrogen peroxide (H(2)O(2))-induced apoptosis through targeting the mitochondria apoptotic pathway, PLoS One 7 (2012) e44907.
- [255] A. Druz, M. Betenbaugh, J. Shiloach, Glucose depletion activates mmu-miR-466h-5p expression through oxidative stress and inhibition of histone deacetylation, Nucleic Acids Research 40 (2012) 7291–7302.
- [256] K. Iekushi, F. Seeger, B. Assmus, A.M. Zeiher, S. Dimmeler, Regulation of cardiac microRNAs by bone marrow mononuclear cell therapy in myocardial infarction, Circulation 125 (2012) 1765–1773. S1-7.
- [257] M.G. Kumar, N.M. Patel, A.M. Nicholson, A.L. Kalen, E.H. Sarsour, P.C. Goswami, Reactive oxygen species mediate microRNA-302 regulation of AT-rich interacting domain 4a and C–C motif ligand 5 expression during transitions between quiescence and proliferation, Free Radical Biology & Medicine 53 (2012) 974–982.
- [258] S. Jajoo, D. Mukherjea, T. Kaur, K.E. Sheehan, S. Sheth, V. Borse, et al., Essential role of NADPH oxidase-dependent reactive oxygen species generation in regulating microRNA-21 expression and function in prostate cancer, Antioxidants & Redox Signaling 19 (2013) 1863–1876.

- [259] S. Thulasingam, C. Massilamany, A. Gangaplara, H. Dai, S. Yarbaeva, S. Subramaniam, et al., miR-27b\*, an oxidative stress-responsive microRNA modulates nuclear factor-kB pathway in RAW 264.7 cells, Molecular and Cellular Biochemistry 352 (2011) 181–188.
- [260] A. Magenta, C. Cencioni, P. Fasanaro, G. Zaccagnini, S. Greco, G. Sarra-Ferraris, et al., miR-200c is upregulated by oxidative stress and induces endothelial cell apoptosis and senescence via ZEB1 inhibition, Cell Death Differ 18 (2011) 1628–1639.
- [261] S. Xu, R. Zhang, J. Niu, D. Cui, B. Xie, B. Zhang, et al., Oxidative stress mediated-alterations of the microRNA expression profile in mouse hippocampal neurons, International Journal of Molecular Sciences 13 (2012) 16945–16960.
- [262] X.Q. Sui, Z.M. Xu, M.B. Xie, D.A. Pei, Resveratrol inhibits hydrogen peroxide-induced apoptosis in endothelial cells via the activation of PI3K/Akt by miR-126, Journal of Atherosclerosis and Thrombosis 21 (2014) 108–118.
- [263] G. Lv, S. Shao, H. Dong, X. Bian, X. Yang, S. Dong, MicroRNA-214 protects cardiac myocytes against H<sub>2</sub>O<sub>2</sub>-induced injury, Journal of Cellular Biochemistry 115 (2014) 93–101.
- [264] Z. Chen, L. Wen, M. Martin, C.Y. Hsu, L. Fang, F.M. Lin, et al., Oxidative stress activates endothelial innate immunity via sterol regulatory element binding protein 2 (SREBP2) transactivation of microRNA-92a, Circulation 131 (2015) 805–814.
- [265] Y. Zhao, E. Samal, D. Srivastava, Serum response factor regulates a muscle-specific microRNA that targets Hand2 during cardiogenesis, Nature 436 (2005) 214–220.
- [266] J.F. Chen, E.M. Mandel, J.M. Thomson, Q. Wu, T.E. Callis, S.M. Hammond, et al., The role of microRNA-1 and microRNA-133 in skeletal muscle proliferation and differentiation, Nature Genetics 38 (2006) 228–233.
- [267] B. Yang, H. Lin, J. Xiao, Y. Lu, X. Luo, B. Li, et al., The muscle-specific microRNA miR-1 regulates cardiac arrhythmogenic potential by targeting GJA1 and KCNJ2, Nature Medicine 13 (2007) 486–491.
- [268] L. Elia, R. Contu, M. Quintavalle, F. Varrone, C. Chimenti, M.A. Russo, et al., Reciprocal regulation of microRNA-1 and insulin-like growth factor-1 signal transduction cascade in cardiac and skeletal muscle in physiological and pathological conditions, Circulation 120 (2009) 2377–2385.
- [269] A.M. Krichevsky, G. Gabriely, miR-21: a small multi-faceted RNA, Journal of Cellular and Molecular Medicine 13 (2009) 39–53.
- [270] S.M. Park, A.B. Gaur, E. Lengyel, M.E. Peter, The miR-200 family determines the epithelial phenotype of cancer cells by targeting the E-cadherin repressors ZEB1 and ZEB2, Genes & Development 22 (2008) 894–907.
- [271] J.J. Haddad, C.S. Fahlman, Redox- and oxidant-mediated regulation of interleukin-10: an antiinflammatory, antioxidant cytokine? Biochemical and Biophysical Research Communications 297 (2002) 163–176.
- [272] T. Ha, L. Liu, J. Kelley, R. Kao, D. Williams, C. Li, Toll-like receptors: new players in myocardial ischemia/reperfusion injury, Antioxidants & Redox Signaling 15 (2011) 1875–1893.
- [273] J.K. Kundu, Y.J. Surh, Emerging avenues linking inflammation and cancer, Free Radical Biology & Medicine 52 (2012) 2013–2037.
- [274] K. Iwanaga, Y. Yang, M.G. Raso, L. Ma, A.E. Hanna, N. Thilaganathan, et al., Pten inactivation accelerates oncogenic K-ras-initiated tumorigenesis in a mouse model of lung cancer, Cancer Research 68 (2008) 1119–1127.
- [275] L. Flohe, R. Brigelius-Flohe, C. Saliou, M.G. Traber, L. Packer, Redox regulation of NF-kappa B activation, Free Radical Biology & Medicine 22 (1997) 1115–1126.

- [276] K. Asehnoune, D. Strassheim, S. Mitra, J.Y. Kim, E. Abraham, Involvement of reactive oxygen species in Toll-like receptor 4-dependent activation of NF-kappa B, Journal of Immunology 172 (2004) 2522–2529.
- [277] F.J. Sheedy, L.A. O'Neill, Adding fuel to fire: microRNAs as a new class of mediators of inflammation, Annals of the Rheumatic Diseases 67 (Suppl. 3) (2008) iii50-5.
- [278] M. Ojaniemi, V. Glumoff, K. Harju, M. Liljeroos, K. Vuori, M. Hallman, Phosphatidylinositol 3kinase is involved in Toll-like receptor 4-mediated cytokine expression in mouse macrophages, European Journal of Immunology 33 (2003) 597–605.
- [279] E. Tili, J.J. Michaille, A. Cimino, S. Costinean, C.D. Dumitru, B. Adair, et al., Modulation of miR-155 and miR-125b levels following lipopolysaccharide/TNF-alpha stimulation and their possible roles in regulating the response to endotoxin shock, Journal of Immunology 179 (2007) 5082–5089.
- [280] X.M. Chen, P.L. Splinter, S.P. O'Hara, N.F. LaRusso, A cellular micro-RNA, let-7i, regulates Toll-like receptor 4 expression and contributes to cholangiocyte immune responses against *Cryptosporidium parvum* infection, The Journal of Biological Chemistry 282 (2007) 28929–28938.
- [281] F.G. Bauernfeind, G. Horvath, A. Stutz, E.S. Alnemri, K. MacDonald, D. Speert, et al., Cutting edge: NF-kappaB activating pattern recognition and cytokine receptors license NLRP3 inflammasome activation by regulating NLRP3 expression, Journal of Immunology 183 (2009) 787–791.
- [282] C. Juliana, T. Fernandes-Alnemri, S. Kang, A. Farias, F. Qin, E.S. Alnemri, Non-transcriptional priming and deubiquitination regulate NLRP3 inflammasome activation, The Journal of Biological Chemistry 287 (2012) 36617–36622.
- [283] F. Bauernfeind, A. Rieger, F.A. Schildberg, P.A. Knolle, J.L. Schmid-Burgk, V. Hornung, NLRP3 inflammasome activity is negatively controlled by miR-223, Journal of Immunology 189 (2012) 4175–4181.
- [284] J.B. Johnnidis, M.H. Harris, R.T. Wheeler, S. Stehling-Sun, M.H. Lam, O. Kirak, et al., Regulation of progenitor cell proliferation and granulocyte function by microRNA-223, Nature 451 (2008) 1125–1129.
- [285] M. Diehn, R.W. Cho, N.A. Lobo, T. Kalisky, M.J. Dorie, A.N. Kulp, et al., Association of reactive oxygen species levels and radioresistance in cancer stem cells, Nature 458 (2009) 780–783.
- [286] H. Soares, H.S. Marinho, C. Real, F. Antunes, Cellular polarity in aging: role of redox regulation and nutrition, Genes & Nutrition 9 (2014) 371.
- [287] S. Ponnappan, U. Ponnappan, Aging and immune function: molecular mechanisms to interventions, Antioxidants & Redox Signaling 14 (2011) 1551–1585.
- [288] K.A. Gelderman, M. Hultqvist, J. Holmberg, P. Olofsson, R. Holmdahl, T cell surface redox levels determine T cell reactivity and arthritis susceptibility, Proceedings of the National Academy of Sciences of the United States of America 103 (2006) 12831–12836.
- [289] T. Hadzic, L. Li, N. Cheng, S.A. Walsh, D.R. Spitz, C.M. Knudson, The role of low molecular weight thiols in T lymphocyte proliferation and IL-2 secretion, Journal of Immunology 175 (2005) 7965–7972.
- [290] M.S. Williams, J. Kwon, T cell receptor stimulation, reactive oxygen species, and cell signaling, Free Radical Biology & Medicine 37 (2004) 1144–1151.
- [291] R. Paganelli, E. Scala, R. Rosso, A. Cossarizza, L. Bertollo, D. Barbieri, et al., A shift to Th0 cytokine production by CD4<sup>+</sup> cells in human longevity: studies on two healthy centenarians, European Journal of Immunology 26 (1996) 2030–2034.
- [292] K.P. Hoefig, V. Heissmeyer, MicroRNAs grow up in the immune system, Current Opinion in Immunology 20 (2008) 281–287.
- [293] E. Tsitsiou, M.A. Lindsay, microRNAs and the immune response, Current Opinion in Pharmacology 9 (2009) 514–520.

- [294] M. Hackl, S. Brunner, K. Fortschegger, C. Schreiner, L. Micutkova, C. Muck, et al., miR-17, miR-19b, miR-20a, and miR-106a are down-regulated in human aging, Aging Cell 9 (2010) 291–296.
- [295] M.S. Cooke, M.D. Evans, M. Dizdaroglu, J. Lunec, Oxidative DNA damage: mechanisms, mutation, and disease, FASEB Journal: official publication of the Federation of American Societies for Experimental Biology. 17 (2003) 1195–1214.
- [296] L.J. Marnett, Oxyradicals and DNA damage, Carcinogenesis 21 (2000) 361-370.
- [297] G.M. Halliday, Inflammation, gene mutation and photoimmunosuppression in response to UVRinduced oxidative damage contributes to photocarcinogenesis, Mutation Research 571 (2005) 107–120.
- [298] A. Sarasin, The molecular pathways of ultraviolet-induced carcinogenesis, Mutation Research 428 (1999) 5–10.
- [299] H. Osada, T. Takahashi, MicroRNAs in biological processes and carcinogenesis, Carcinogenesis 28 (2007) 2–12.
- [300] F. Meng, R. Henson, H. Wehbe-Janek, H. Smith, Y. Ueno, T. Patel, The MicroRNA let-7a modulates interleukin-6-dependent STAT-3 survival signaling in malignant human cholangiocytes, The Journal of Biological Chemistry 282 (2007) 8256–8264.
- [301] C. Braconi, N. Huang, T. Patel, MicroRNA-dependent regulation of DNA methyltransferase-1 and tumor suppressor gene expression by interleukin-6 in human malignant cholangiocytes, Hepatology 51 (2010) 881–890.
- [302] S. Jiang, H.W. Zhang, M.H. Lu, X.H. He, Y. Li, H. Gu, et al., MicroRNA-155 functions as an OncomiR in breast cancer by targeting the suppressor of cytokine signaling 1 gene, Cancer Research 70 (2010) 3119–3127.
- [303] I.M. Pedersen, D. Otero, E. Kao, A.V. Miletic, C. Hother, E. Ralfkiaer, et al., Onco-miR-155 targets SHIP1 to promote TNFalpha-dependent growth of B cell lymphomas, EMBO Molecular Medicine 1 (2009) 288–295.
- [304] J. Zhao, B.K. Sun, J.A. Erwin, J.J. Song, J.T. Lee, Polycomb proteins targeted by a short repeat RNA to the mouse X chromosome, Science 322 (2008) 750–756.
- [305] J.T. Lee, M.S. Bartolomei, X-inactivation, imprinting, and long noncoding RNAs in health and disease, Cell 152 (2013) 1308–1323.
- [306] T. Hung, H.Y. Chang, Long noncoding RNA in genome regulation: prospects and mechanisms, RNA Biology 7 (2010) 582–585.
- [307] F. Mohammad, T. Mondal, N. Guseva, G.K. Pandey, C. Kanduri, Kcnq1ot1 noncoding RNA mediates transcriptional gene silencing by interacting with Dnmt1, Development 137 (2010) 2493–2499.
- [308] K.C. Wang, Y.W. Yang, B. Liu, A. Sanyal, R. Corces-Zimmerman, Y. Chen, et al., A long noncoding RNA maintains active chromatin to coordinate homeotic gene expression, Nature 472 (2011) 120–124.
- [309] S. Bertani, S. Sauer, E. Bolotin, F. Sauer, The noncoding RNA Mistral activates Hoxa6 and Hoxa7 expression and stem cell differentiation by recruiting MLL1 to chromatin, Molecular Cell 43 (2011) 1040–1046.
- [310] M.C. Tsai, O. Manor, Y. Wan, N. Mosammaparast, J.K. Wang, F. Lan, et al., Long noncoding RNA as modular scaffold of histone modification complexes, Science 329 (2010) 689–693.
- [311] G. Wan, R. Mathur, X. Hu, Y. Liu, X. Zhang, G. Peng, et al., Long non-coding RNA ANRIL (CDKN2B-AS) is induced by the ATM-E2F1 signaling pathway, Cellular Signalling 25 (2013) 1086–1095.
- [312] K.L. Yap, S. Li, A.M. Munoz-Cabello, S. Raguz, L. Zeng, S. Mujtaba, et al., Molecular interplay of the noncoding RNA ANRIL and methylated histone H3 lysine 27 by polycomb CBX7 in transcriptional silencing of INK4a, Molecular Cell 38 (2010) 662–674.

- [313] C.A. Klattenhoff, J.C. Scheuermann, L.E. Surface, R.K. Bradley, P.A. Fields, M.L. Steinhauser, et al., Braveheart, a long noncoding RNA required for cardiovascular lineage commitment, Cell 152 (2013) 570–583.
- [314] P. Grote, L. Wittler, D. Hendrix, F. Koch, S. Wahrisch, A. Beisaw, et al., The tissue-specific lncRNA Fendrr is an essential regulator of heart and body wall development in the mouse, Developmental Cell 24 (2013) 206–214.
- [315] A.M. Khalil, M. Guttman, M. Huarte, M. Garber, A. Raj, D. Rivea Morales, et al., Many human large intergenic noncoding RNAs associate with chromatin-modifying complexes and affect gene expression, Proceedings of the National Academy of Sciences of the United States of America 106 (2009) 11667–11672.
- [316] K.M. Schmitz, C. Mayer, A. Postepska, I. Grummt, Interaction of noncoding RNA with the rDNA promoter mediates recruitment of DNMT3b and silencing of rRNA genes, Genes & Development 24 (2010) 2264–2269.
- [317] T. Hung, Y. Wang, M.F. Lin, A.K. Koegel, Y. Kotake, G.D. Grant, et al., Extensive and coordinated transcription of noncoding RNAs within cell-cycle promoters, Nature Genetics 43 (2011) 621–629.
- [318] M. Huarte, M. Guttman, D. Feldser, M. Garber, M.J. Koziol, D. Kenzelmann-Broz, et al., A large intergenic noncoding RNA induced by p53 mediates global gene repression in the p53 response, Cell 142 (2010) 409–419.
- [319] L. Zhu, P.C. Xu, Downregulated LncRNA-ANCR promotes osteoblast differentiation by targeting EZH2 and regulating Runx2 expression, Biochemical and Biophysical Research Communications 432 (2013) 612–617.
- [320] J. Feng, C. Bi, B.S. Clark, R. Mady, P. Shah, J.D. Kohtz, The Evf-2 noncoding RNA is transcribed from the Dlx-5/6 ultraconserved region and functions as a Dlx-2 transcriptional coactivator, Genes & Development 20 (2006) 1470–1484.
- [321] M.E. Dinger, P.P. Amaral, T.R. Mercer, K.C. Pang, S.J. Bruce, B.B. Gardiner, et al., Long noncoding RNAs in mouse embryonic stem cell pluripotency and differentiation, Genome Research 18 (2008) 1433–1445.
- [322] S. Loewer, M.N. Cabili, M. Guttman, Y.H. Loh, K. Thomas, I.H. Park, et al., Large intergenic noncoding RNA-RoR modulates reprogramming of human induced pluripotent stem cells, Nature Genetics 42 (2010) 1113–1117.
- [323] M. Guttman, J. Donaghey, B.W. Carey, M. Garber, J.K. Grenier, G. Munson, et al., lincRNAs act in the circuitry controlling pluripotency and differentiation, Nature 477 (2011) 295–300.
- [324] G. Natoli, J.C. Andrau, Noncoding transcription at enhancers: general principles and functional models, Annual Review of Genetics 46 (2012) 1–19.
- [325] U.A. Orom, R. Shiekhattar, Long non-coding RNAs and enhancers, Current Opinion in Genetics & Development 21 (2011) 194–198.
- [326] K.V. Morris, S.W. Chan, S.E. Jacobsen, D.J. Looney, Small interfering RNA-induced transcriptional gene silencing in human cells, Science 305 (2004) 1289–1292.
- [327] S. Napoli, C. Pastori, M. Magistri, G.M. Carbone, C.V. Catapano, Promoter-specific transcriptional interference and c-myc gene silencing by siRNAs in human cells, The EMBO Journal 28 (2009) 1708–1719.
- [328] P.G. Hawkins, S. Santoso, C. Adams, V. Anest, K.V. Morris, Promoter targeted small RNAs induce long-term transcriptional gene silencing in human cells, Nucleic Acids Research 37 (2009) 2984–2995.
- [329] B.A. Janowski, S.T. Younger, D.B. Hardy, R. Ram, K.E. Huffman, D.R. Corey, Activating gene expression in mammalian cells with promoter-targeted duplex RNAs, Nature Chemical Biology 3 (2007) 166–173.

- [330] I. Martianov, A. Ramadass, A. Serra Barros, N. Chow, A. Akoulitchev, Repression of the human dihydrofolate reductase gene by a non-coding interfering transcript, Nature 445 (2007) 666–670.
- [331] S.Y. Ng, R. Johnson, L.W. Stanton, Human long non-coding RNAs promote pluripotency and neuronal differentiation by association with chromatin modifiers and transcription factors, The EMBO Journal 31 (2012) 522–533.
- [332] M.L. Hastings, H.A. Ingle, M.A. Lazar, S.H. Munroe, Post-transcriptional regulation of thyroid hormone receptor expression by cis-acting sequences and a naturally occurring antisense RNA, The Journal of Biological Chemistry 275 (2000) 11507–11513.
- [333] Q.F. Yin, L. Yang, Y. Zhang, J.F. Xiang, Y.W. Wu, G.G. Carmichael, et al., Long noncoding RNAs with snoRNA ends, Molecular Cell 48 (2012) 219–230.
- [334] X.O. Zhang, Q.F. Yin, H.B. Wang, Y. Zhang, T. Chen, P. Zheng, et al., Species-specific alternative splicing leads to unique expression of sno-lncRNAs, BMC Genomics 15 (2014) 287.
- [335] A.T. Willingham, A.P. Orth, S. Batalov, E.C. Peters, B.G. Wen, P. Aza-Blanc, et al., A strategy for probing the function of noncoding RNAs finds a repressor of NFAT, Science 309 (2005) 1570–1573.
- [336] C.M. Clemson, J.N. Hutchinson, S.A. Sara, A.W. Ensminger, A.H. Fox, A. Chess, et al., An architectural role for a nuclear noncoding RNA: NEAT1 RNA is essential for the structure of paraspeckles, Molecular Cell 33 (2009) 717–726.
- [337] K. Imamura, N. Imamachi, G. Akizuki, M. Kumakura, A. Kawaguchi, K. Nagata, et al., Long noncoding RNA NEAT1-dependent SFPQ relocation from promoter region to paraspeckle mediates IL8 expression upon immune stimuli, Molecular Cell 53 (2014) 393–406.
- [338] T. Hirose, G. Virnicchi, A. Tanigawa, T. Naganuma, R. Li, H. Kimura, et al., NEAT1 long noncoding RNA regulates transcription via protein sequestration within subnuclear bodies, Molecular Biology of the Cell 25 (2014) 169–183.
- [339] D.R. Cooper, G. Carter, P. Li, R. Patel, J.E. Watson, N.A. Patel, Long non-coding RNA NEAT1 associates with SRp40 to temporally regulate PPARgamma2 splicing during adipogenesis in 3T3-L1 cells, Genes 5 (2014) 1050–1063.
- [340] X. Gao, L. Kong, X. Lu, G. Zhang, L. Chi, Y. Jiang, et al., Paraspeckle protein 1 (PSPC1) is involved in the cisplatin induced DNA damage response—role in G1/S checkpoint, PLoS One 9 (2014) e97174.
- [341] C. Gong, L.E. Maquat, IncRNAs transactivate STAU1-mediated mRNA decay by duplexing with 3' UTRs via Alu elements, Nature 470 (2011) 284–288.
- [342] M. Kretz, Z. Siprashvili, C. Chu, D.E. Webster, A. Zehnder, K. Qu, et al., Control of somatic tissue differentiation by the long non-coding RNA TINCR, Nature 493 (2013) 231–235.
- [343] M.A. Faghihi, F. Modarresi, A.M. Khalil, D.E. Wood, B.G. Sahagan, T.E. Morgan, et al., Expression of a noncoding RNA is elevated in Alzheimer's disease and drives rapid feed-forward regulation of beta-secretase, Nature Medicine 14 (2008) 723–730.
- [344] A.K. Ebralidze, F.C. Guibal, U. Steidl, P. Zhang, S. Lee, B. Bartholdy, et al., PU.1 expression is modulated by the balance of functional sense and antisense RNAs regulated by a shared cisregulatory element, Genes & Development 22 (2008) 2085–2092.
- [345] J.H. Yoon, K. Abdelmohsen, S. Srikantan, X. Yang, J.L. Martindale, S. De, et al., LincRNA-p21 suppresses target mRNA translation, Molecular Cell 47 (2012) 648–655.
- [346] M.N. Hinman, H. Lou, Diverse molecular functions of Hu proteins, Cellular and Molecular Life Sciences 65 (2008) 3168–3181.
- [347] K. Abdelmohsen, M. Gorospe, Posttranscriptional regulation of cancer traits by HuR, Wiley Interdisciplinary Reviews RNA 1 (2010) 214–229.
- [348] J. Huang, N. Zhou, K. Watabe, Z. Lu, F. Wu, M. Xu, et al., Long non-coding RNA UCA1 promotes breast tumor growth by suppression of p27 (Kip1), Cell Death & Disease 5 (2014) e1008.

- [349] X. Guo, Z. Zhang, M.B. Gerstein, D. Zheng, Small RNAs originated from pseudogenes: cis- or transacting? PLoS Computational Biology 5 (2009) e1000449.
- [350] O.H. Tam, A.A. Aravin, P. Stein, A. Girard, E.P. Murchison, S. Cheloufi, et al., Pseudogene-derived small interfering RNAs regulate gene expression in mouse oocytes, Nature 453 (2008) 534–538.
- [351] S. Nolasco, J. Bellido, J. Goncalves, A. Tavares, J.C. Zabala, H. Soares, The expression of tubulin cofactor A (TBCA) is regulated by a noncoding antisense Tbca RNA during testis maturation, PLoS One 7 (2012) e42536.
- [352] L. Salmena, L. Poliseno, Y. Tay, L. Kats, P.P. Pandolfi, A ceRNA hypothesis: the Rosetta Stone of a hidden RNA language? Cell 146 (2011) 353–358.
- [353] M. Cesana, D. Cacchiarelli, I. Legnini, T. Santini, O. Sthandier, M. Chinappi, et al., A long noncoding RNA controls muscle differentiation by functioning as a competing endogenous RNA, Cell 147 (2011) 358–369.
- [354] Y. Wang, Z. Xu, J. Jiang, C. Xu, J. Kang, L. Xiao, et al., Endogenous miRNA sponge lincRNA-RoR regulates Oct4, Nanog, and Sox2 in human embryonic stem cell self-renewal, Developmental Cell 25 (2013) 69–80.
- [355] W. Hu, B. Yuan, J. Flygare, H.F. Lodish, Long noncoding RNA-mediated anti-apoptotic activity in murine erythroid terminal differentiation, Genes & Development 25 (2011) 2573–2578.
- [356] L. Sun, L.A. Goff, C. Trapnell, R. Alexander, K.A. Lo, E. Hacisuleyman, et al., Long noncoding RNAs regulate adipogenesis, Proceedings of the National Academy of Sciences of the United States of America 110 (2013) 3387–3392.
- [357] D. Bu, K. Yu, S. Sun, C. Xie, G. Skogerbo, R. Miao, et al., NONCODE v3.0: integrative annotation of long noncoding RNAs, Nucleic Acids Research 40 (2012) D210–D215.
- [358] K.C. Pang, M.E. Dinger, T.R. Mercer, L. Malquori, S.M. Grimmond, W. Chen, et al., Genome-wide identification of long noncoding RNAs in CD8<sup>+</sup> T cells, Journal of Immunology 182 (2009) 7738–7748.
- [359] X. Peng, L. Gralinski, C.D. Armour, M.T. Ferris, M.J. Thomas, S. Proll, et al., Unique signatures of long noncoding RNA expression in response to virus infection and altered innate immune signaling, mBio 1 (2010).
- [360] Z. Yin, D. Guan, Q. Fan, J. Su, W. Zheng, W. Ma, et al., lncRNA expression signatures in response to enterovirus 71 infection, Biochemical and Biophysical Research Communications 430 (2013) 629–633.
- [361] H. Cui, N. Xie, Z. Tan, S. Banerjee, V.J. Thannickal, E. Abraham, et al., The human long noncoding RNA lnc-IL7R regulates the inflammatory response, European Journal of Immunology 44 (2014) 2085–2095.
- [362] A.P. Mao, J. Shen, Z. Zuo, Expression and regulation of long noncoding RNAs in TLR4 signaling in mouse macrophages, BMC Genomics 16 (2015) 45.
- [363] M. Pagani, G. Rossetti, I. Panzeri, P. de Candia, R.J. Bonnal, R.L. Rossi, et al., Role of microRNAs and long-non-coding RNAs in CD4(+) T-cell differentiation, Immunological Reviews 253 (2013) 82–96.
- [364] S. Vigneau, P.S. Rohrlich, M. Brahic, J.F. Bureau, Tmevpg1, a candidate gene for the control of Theiler's virus persistence, could be implicated in the regulation of gamma interferon, Journal of Virology 77 (2003) 5632–5638.
- [365] S.P. Collier, P.L. Collins, C.L. Williams, M.R. Boothby, T.M. Aune, Cutting edge: influence of Tmevpg1, a long intergenic noncoding RNA, on the expression of Ifng by Th1 cells, Journal of Immunology 189 (2012) 2084–2088.

- [366] J.A. Gomez, O.L. Wapinski, Y.W. Yang, J.F. Bureau, S. Gopinath, D.M. Monack, et al., The NeST long ncRNA controls microbial susceptibility and epigenetic activation of the interferon-gamma locus, Cell 152 (2013) 743–754.
- [367] G. Hu, Q. Tang, S. Sharma, F. Yu, T.M. Escobar, S.A. Muljo, et al., Expression and regulation of intergenic long noncoding RNAs during T cell development and differentiation, Nature Immunology 14 (2013) 1190–1198.
- [368] K. Imamura, N. Akimitsu, Long non-coding RNAs involved in immune responses, Frontiers in Immunology 5 (2014) 573.
- [369] N.A. Rapicavoli, K. Qu, J. Zhang, M. Mikhail, R.M. Laberge, H.Y. Chang, A mammalian pseudogene lncRNA at the interface of inflammation and anti-inflammatory therapeutics, eLife 2 (2013) e00762.
- [370] P.W. Wright, A. Huehn, F. Cichocki, H. Li, N. Sharma, H. Dang, et al., Identification of a KIR antisense lncRNA expressed by progenitor cells, Genes and Immunity 14 (2013) 427–433.
- [371] Z. Li, T.C. Chao, K.Y. Chang, N. Lin, V.S. Patil, C. Shimizu, et al., The long noncoding RNA THRIL regulates TNFalpha expression through its interaction with hnRNPL, Proceedings of the National Academy of Sciences of the United States of America 111 (2014) 1002–1007.
- [372] X. Zhang, Z. Lian, C. Padden, M.B. Gerstein, J. Rozowsky, M. Snyder, et al., A myelopoiesisassociated regulatory intergenic noncoding RNA transcript within the human HOXA cluster, Blood 113 (2009) 2526–2534.
- [373] S. Carpenter, D. Aiello, M.K. Atianand, E.P. Ricci, P. Gandhi, L.L. Hall, et al., A long noncoding RNA mediates both activation and repression of immune response genes, Science 341 (2013) 789–792.
- [374] J.N. Hutchinson, A.W. Ensminger, C.M. Clemson, C.R. Lynch, J.B. Lawrence, A. Chess, A screen for nuclear transcripts identifies two linked noncoding RNAs associated with SC35 splicing domains, BMC Genomics 8 (2007) 39.
- [375] J.B. Heo, S. Sung, Vernalization-mediated epigenetic silencing by a long intronic noncoding RNA, Science 331 (2011) 76–79.
- [376] R. Ietswaart, Z. Wu, C. Dean, Flowering time control: another window to the connection between antisense RNA and chromatin, Trends in Genetics 28 (2012) 445–453.
- [377] C. Carrieri, L. Cimatti, M. Biagioli, A. Beugnet, S. Zucchelli, S. Fedele, et al., Long non-coding antisense RNA controls Uchl1 translation through an embedded SINEB2 repeat, Nature 491 (2012) 454–457.
- [378] D.R. Crawford, G.P. Schools, K.J. Davies, Oxidant-inducible adapt 15 RNA is associated with growth arrest- and DNA damage-inducible gadd153 and gadd45, Archives of Biochemistry and Biophysics 329 (1996) 137–144.
- [379] D.R. Crawford, G.P. Schools, S.L. Salmon, K.J. Davies, Hydrogen peroxide induces the expression of adapt15, a novel RNA associated with polysomes in hamster HA-1 cells, Archives of Biochemistry and Biophysics 325 (1996) 256–264.
- [380] Y. Wang, D.R. Crawford, K.J. Davies, adapt33, a novel oxidant-inducible RNA from hamster HA-1 cells, Archives of Biochemistry and Biophysics 332 (1996) 255–260.
- [381] Y. Wang, K.J. Davies, J.A. Melendez, D.R. Crawford, Characterization of adapt33, a stress-inducible riboregulator, Gene Expression 11 (2003) 85–94.
- [382] X. Chen, Y. Ba, L. Ma, X. Cai, Y. Yin, K. Wang, et al., Characterization of microRNAs in serum: a novel class of biomarkers for diagnosis of cancer and other diseases, Cell Research 18 (2008) 997–1006.
- [383] H.L. Zhang, L.F. Yang, Y. Zhu, X.D. Yao, S.L. Zhang, B. Dai, et al., Serum miRNA-21: elevated levels in patients with metastatic hormone-refractory prostate cancer and potential predictive factor for the efficacy of docetaxel-based chemotherapy, The Prostate 71 (2011) 326–331.

- [384] H. Wang, Y. Jiang, H. Peng, Y. Chen, P. Zhu, Y. Huang, Recent progress in microRNA delivery for cancer therapy by non-viral synthetic vectors, Advanced Drug Delivery Reviews 81 (2015) 142–160.
- [385] C. Monroig Pdel, L. Chen, S. Zhang, G.A. Calin, Small molecule compounds targeting miRNAs for cancer therapy, Advanced Drug Delivery Reviews 81 (2015) 104–116.
- [386] S. Zhang, L. Chen, E.J. Jung, G.A. Calin, Targeting microRNAs with small molecules: from dream to reality, Clinical Pharmacology and Therapeutics 87 (2010) 754–758.
- [387] J.C. Henry, A.C. Azevedo-Pouly, T.D. Schmittgen, MicroRNA replacement therapy for cancer, Pharmaceutical Research 28 (2011) 3030–3042.
- [388] M.A. Faghihi, C. Wahlestedt, Regulatory roles of natural antisense transcripts, Nature Reviews Molecular Cell Biology 10 (2009) 637–643.
- [389] N. Rusk, AntagoNATs boost gene expression, Nature Methods 9 (2012) 437.
- [390] R.A. Smith, T.M. Miller, K. Yamanaka, B.P. Monia, T.P. Condon, G. Hung, et al., Antisense oligonucleotide therapy for neurodegenerative disease, The Journal of Clinical Investigation 116 (2006) 2290–2296.
- [391] L. Ding, X.M. Hu, H. Wu, G.X. Liu, Y.J. Gao, D.M. He, et al., Combined transfection of Bcl-2 siRNA and miR-15a oligonucleotides enhanced methotrexate-induced apoptosis in Raji cells, Cancer Biology & Medicine 10 (2013) 16–21.
- [392] C.V. Pecot, G.A. Calin, R.L. Coleman, G. Lopez-Berestein, A.K. Sood, RNA interference in the clinic: challenges and future directions, Nature Reviews Cancer 11 (2011) 59–67.
- [393] M.E. Kleinman, K. Yamada, A. Takeda, V. Chandrasekaran, M. Nozaki, J.Z. Baffi, et al., Sequenceand target-independent angiogenesis suppression by siRNA via TLR3, Nature 452 (2008) 591–597.
- [394] G. Michlewski, S. Guil, C.A. Semple, J.F. Caceres, Posttranscriptional regulation of miRNAs harboring conserved terminal loops, Molecular Cell 32 (2008) 383–393.
- [395] E.M. Small, E.N. Olson, Pervasive roles of microRNAs in cardiovascular biology, Nature 469 (2011) 336–342.
- [396] E.A. Miska, E. Alvarez-Saavedra, A.L. Abbott, N.C. Lau, A.B. Hellman, S.M. McGonagle, et al., Most *Caenorhabditis elegans* microRNAs are individually not essential for development or viability, PLoS Genetics 3 (2007) e215.
- [397] P. Smibert, E.C. Lai, Lessons from microRNA mutants in worms, flies and mice, Cell Cycle 7 (2008) 2500–2508.
- [398] K. Sun, E.C. Lai, Adult-specific functions of animal microRNAs, Nature Reviews Genetics 14 (2013) 535–548.
- [399] J.L. Brenner, K.L. Jasiewicz, A.F. Fahley, B.J. Kemp, A.L. Abbott, Loss of individual microRNAs causes mutant phenotypes in sensitized genetic backgrounds in *C. elegans*, Current Biology 20 (2010) 1321–1325.
- [400] F. Bejarano, D. Bortolamiol-Becet, Q. Dai, K. Sun, A. Saj, Y.T. Chou, et al., A genome-wide transgenic resource for conditional expression of Drosophila microRNAs, Development 139 (2012) 2821–2831.
- [401] S. Szuplewski, J.M. Kugler, S.F. Lim, P. Verma, Y.W. Chen, S.M. Cohen, MicroRNA transgene overexpression complements deficiency-based modifier screens in Drosophila, Genetics 190 (2012) 617–626.
- [402] A.Q. Gomes, S. Nolasco, H. Soares, Non-coding RNAs: multi-tasking molecules in the cell, International Journal of Molecular Sciences 14 (2013) 16010–16039.
- [403] A. Brennicke, A. Marchfelder, S. Binder, RNA editing, FEMS Microbiology Reviews 23 (1999) 297–316.

- [404] A. Macovei, S.S. Gill, N. Tuteja, microRNAs as promising tools for improving stress tolerance in rice, Plant Signaling & Behavior 7 (2012) 1296–1301.
- [405] B.H. Zhang, X.P. Pan, Q.L. Wang, G.P. Cobb, T.A. Anderson, Identification and characterization of new plant microRNAs using EST analysis, Cell Research 15 (2005) 336–360.
- [406] Y. Ding, Z. Chen, C. Zhu, Microarray-based analysis of cadmium-responsive microRNAs in rice (*Oryza sativa*), Journal of Experimental Botany 62 (2011) 3563–3573.
- [407] E.A. Veal, A.M. Day, B.A. Morgan, Hydrogen peroxide sensing and signaling, Molecular Cell 26 (2007) 1–14.
- [408] Y.S. Bae, H. Oh, S.G. Rhee, Y.D. Yoo, Regulation of reactive oxygen species generation in cell signaling, Molecules and Cells 32 (2011) 491–509.
- [409] M. Giorgio, M. Trinei, E. Migliaccio, P.G. Pelicci, Hydrogen peroxide: a metabolic by-product or a common mediator of ageing signals? Nature Reviews Molecular Cell Biology 8 (2007) 722–728.
- [410] T.R. Hurd, M. DeGennaro, R. Lehmann, Redox regulation of cell migration and adhesion, Trends in Cell Biology 22 (2012) 107–115.
- [411] N. Schmolka, K. Serre, A.R. Grosso, M. Rei, D.J. Pennington, A.Q. Gomes, et al., Epigenetic and transcriptional signatures of stable versus plastic differentiation of proinflammatory gammadelta T cell subsets, Nature Immunology 14 (2013) 1093–1100.

# 11

# Implications of Substrate Topographic Surface on Tissue Engineering

M.C.X. Pinto<sup>1,2,3</sup>, V.A.M. Goulart<sup>1</sup>, R.C. Parreira<sup>1,2</sup>, L.T. Souza<sup>1</sup>, N. de Cássia Oliveira Paiva<sup>1</sup>, R.R. Resende<sup>1,2,\*</sup>

<sup>1</sup>FEDERAL UNIVERSITY OF MINAS GERAIS, BELO HORIZONTE, BRAZIL; <sup>2</sup>INSTITUTONANOCELL, DIVINÓPOLIS, BRAZIL; <sup>3</sup>FEDERAL UNIVERSITY OF GOIÁS, GOIÂNIA, BRAZIL

## 11.1 Biomimetic Constructs for Tissue Engineering

"Tissue engineering is a multi-disciplinary field where biology, biochemistry, physiology, physics, and chemistry combine to produce material science modeling samples to be used in tissue engineering towards the development of biological substitutes that restore, maintain, or improve tissue function" [1,2]. The use of scaffolds on tissue engineering is based on temporary three-dimensional supports, which is accomplished with the extracellular matrix (ECM), allowing the proliferation and communication of host cells harvested from natural tissue in vitro and in vivo. Within this context scaffolds can have the following roles: maintaining cell viability and growth, permitting cell signaling and communication, and allowing cells to reach their target site (e.g., bone repair, damage nerves, wound healing, burned skin) as would cells from their original niche site. Also, scaffolds are supposed to supplement or replace missing, compromised, or injured *tissue* or organs to keep their normal function [3]. Actually the tissue engineer field has been progressed from this simple concept of scaffolds as a bioinvisible, inert, inoffensive, and long-lived biomaterial. Currently, the principle is that scaffolds should confer biofunctionality, along with biodegradability over time, allowing replacement of biomimetic material with the original tissue.

Scaffolds are expected to have more than functionally durable and stability. Dynamic cell-control signaling must be integrated into these biomimetic materials, getting into their temporary structure and improving their homing tissue. In order to acquire improved properties (e.g., durability, flexibility, physical functions, communication), scaffolds must incorporate biological functions and improve the cellular control

<sup>\*</sup>Corresponding Author.

conditions. [4]. Furthermore, for clinical applications of this new concept of scaffolds, it will be necessary for them to allow cell-seed accomplished with nanomaterials complexed with drugs or growth factors able to be made as a drug and biomolecules carriers [5-8]. Diverse stem cells origins from adult or embryonic tissues and induced pluripotent stem cells are being evaluated as technologies for cell therapy (allogeneic, xenogeneic, or autologous) [9]. However the production or acquisition of cells alone is only part of the answer. Scaffolds must ensure a three-dimensional (3D) structure that allows cells to grow, communicate, differentiate, migrate, and produce an ECM substituting the transplanted material with functional native tissue in solid implants [10]. It is mainly related to mammalian connective tissue that often undergoes repair with scarring rather than functional regeneration [10,11].

In other words, scaffolds need to be constructed based on cues that allow cell functions enabling tissue remodeling rather than just providing single signaling molecules (e.g., growth factors, cytokines, or pharmacologic agents) to induce tissue reconstruction [10,12]. Depending on the type of tissue that needs to be repaired or replaced, these molecules can be of different natures. Tissues can be classified as highly cellular or noncellular tissues (e.g., secretory, metabolic, barrier). For the first type, cell-to-cell contact dominates control: for example, cell signaling, communication (e.g., extracellular signals induce transcription factors activation and consequently gene expression) [9,13–18], cell adhesion, and cell perfusion and nutrient consumption (e.g., gases such as CO<sub>2</sub>, O<sub>2</sub>, and nutrients as glucose) are supported. In the case of noncellular or poorly cellular tissues such as connective and contractile tissues, properties to make scaffolds biomimetic materials are related to their 3D topographic and directional cues, which means their mechanical and substrate characteristics, along with external mechanical loading and neurovascular innervation.

Biomimesis is related to the criteria of spatial dimensions that a scaffold can form. That is the essence of 3D growth rather than a flat culture surface. Signals attached to substrate can act through all three spatial axes of the cell, which allows them to communicate within all directions to generate new information, resembling to that of original tissues. This new information can offer stochastic chemical reactions, spatially polarized or homogeneous signaling networks [19–21], with different cells communicating with each other in spatiotemporal domain, in contrast to 2D monolayer cultures in which cues have polarized directional, with just a two-coverage, one "dorsal" fluid–covered surface and a "ventral" plastic–attached surface. Obviously, non-epithelial cells such as fibroblasts or other stromal cells are not appropriate for this system.

Three-Dimensional scaffolds can be replaced by normal tissue depending on its properties like complaisance, perfusion (pore size) or degradation rate. Based on that, Scaffolds can be divided into (1) synthetic polymer based, especially polylactide/glycolic acids [2,22] and (2) bioartificial natural polymers such as collagen/collagen–glycosaminoglycan (e.g., Integra) sponges [23], collagen gels [24], and fibrin [25]; in addition, a wide range of possibilities can be classified.

This is a key division relating to the question of scaffold degradation. When soft fibrous tissues are replaced with 3D engineered tissue, the initial scaffold must be replaced with the production of new tissue from the host cells. Otherwise, nondegradable mineral-based scaffolds for bone repair are more appropriate. This level of control is almost impossible for conventional biodegradable polymers, because of the diversity of physicochemical conditions at any given implant, pathology, or injury site. This is because of the action of hydrolyzation/dissolution to be *independent* of cell activity. In scaffolds constructed with native proteins aggregated from the cellular matrix, they are susceptible to being cleaved, consumed, and replaced through the metabolism of ECM proteases (meaning biological turnover and remodeling) and tend to be replaced by cell action as new matrix is produced. However, for conventional polymers this is not possible because they disappear too rapidly, providing poor support; or more commonly they disappear too slowly, inhibiting or counteracting cellbased renewal [10]. However, using synthetic polymers or their blends, and synthetic and natural polymers blends with precisely controllable degradation rates (Table 11.1) can solve this problem [26].

Questions have been posed about how much or how little scaffold design must have similarity with the host tissue to be classified as biomimetic. Nowadays, tissue engineering success is achieved considering composition and structure, which is much simpler than using native tissue, and it is possible to manipulate cells and biomaterials in vitro [27]. An example is the use of bioprosthetic heart valves resembling bovine pericardium. Even lacking the trilayered structure of native leaflets, these artificial valves vield adequate function [27]. Considering the large range of tissues to mimic, we can consider extreme types of scaffold designs, such as those that are cell-rich and matrixrich [28]. Naturally, there are many intermediate types, which can be achieved by remodeling their own format into another over time. Thus it is possible to have 3D scaffolds structured for adherent cell masses using a cell-rich construct [29,30]. Those scaffolds allow soft tissue and organs (e.g., intestine, lung) and full stromal differentiation and matrix deposition resembling their embryonic tissue templates and with little connective tissue content. Otherwise, scaffolds that require significant mechanical functions as support and/or contraction force guides (such as bone tissue, nerves, and blood vessels) use matrix-rich constructs.

Considering how long grafts have been studied, more experience and sensibility were achieved that actually occurs in natural tissue development and wound repair. Although occasionally function must be restored immediately before new tissue formation for tissue repair to cover a skin wound or to repair a ruptured tendon, during tissue repair these scaffolds must create conditions to restore by tissue native function through the regenerative process (for example, the restoration of nerve function or integration of a bone graft). Thus an implantable device requires an initial design that mimics the structure and function of the native tissues forming an initial repair template while providing a reconstitution in situ by its environment for final structures and functions.

		Characteristics	Advantages	Disadvantages	Applications
Polymers	Polyglycolic acid (PGA)	Highly crystalline, hydrophilic, byproduct is glycolic acid	<ul> <li>Biocompatible and biodegradable</li> <li>Bioabsorption (2-4 weeks)</li> <li>Electrospinning yields diameters ~ 200 nm</li> <li>Good choice for high strength and elasticity and fast- degrading material</li> </ul>	<ul> <li>Fast degradation causes pH change</li> <li>Tissue may require buffering capacity</li> </ul>	<ul> <li>Bone tissue strengthening</li> <li>Cardiac grafts</li> <li>Collagen and cellular interaction</li> <li>Differentiation with mesen- chymal stem cells</li> </ul>
	Polylactic acid (PLA)	Hydrophobic, lower melting temperature, byproduct is lactic acid	<ul> <li>Biocompatible and biodegradable</li> <li>Bioabsorption (30 weeks)</li> <li>Good choice for drug delivery owing to predictable degradation</li> </ul>	<ul> <li>Larger-diameter fibers ~ microscale</li> </ul>	
	Polydioxanone (PDO)	Highly crystalline	<ul> <li>Biocompatible and biodegradable</li> <li>Degradation rate between PGA-PLA</li> <li>Shape memory</li> <li>Excellent flexibility</li> <li>Modulus comparable collagen and elastin</li> <li>Good source for future vascular grafts</li> </ul>	<ul> <li>Lack of knot retention</li> <li>Lack of adaptability to develop tissue</li> </ul>	
	Polycaprolactone (PCL)	Semicrystalline properties, easily copolymerized, byproduct is caproic acid	<ul> <li>Biocompatible and biodegradable</li> <li>Inexpensive</li> <li>Highly elastic</li> <li>Slow degradation (1–2 years)</li> <li>Good choice for human mesenchymal stem cell seeding to induce differentiation</li> </ul>	• No shape retention (highly elastic)	

 Table 11.1
 Synthetic Polymers Characteristics, Advantages and Disadvantages

Chapter 11
٠
Nanotopographic
Effects
on
Cell
Functions
291

Blends	Pga-pla (plga)	<ul> <li>75% PLA-25% PGA</li> <li>50% PLA-50% PGA</li> <li>Blended PLA and PGA together in HFP at ratios of 100:0, 75:25, 50:50, 25:75</li> </ul>	<ul> <li>Hydrophilicity proportional to composition of copolymer</li> <li>Degradation rate proportional to composition of copolymer</li> </ul>		<ul> <li>Cardiac tissue in mice for tissue regeneration</li> <li>Individual cardiomyocytes attachment at seeding</li> <li>Scaffold loaded with antibiotics for wound healing</li> <li>Mechanical properties such as tangential modulus, peak stress, and strain to failure, of copolymers and blends appear to be controlled by</li> </ul>
	PGA-PCL	Not wide variance in fiber diameters and is assumed that individual polymers become entangled in regular pattern dictated by their chemistry and concentration In addition, owing to long degradation time of PCL (typically 1–2 years), it would be reasonable to expect that blend would degrade slower than PGA	<ul> <li>PGA: high stress tolerance</li> <li>PCL: highly elastic</li> <li>Optimal combination PCL-PGA: 1/3</li> <li>Longer degradation time ~ 3 months (PCL: 2 years; PGA: 2-4 weeks)</li> </ul>		fiber/polymer composition Blend seems well-suited to tissue engineering applications in which high elasticity is driving factor
	PLA-PCL	PLA-CL (PLCL) can be produced at various weight percents from methylene chloride at 25°C. The 70:30 ratio was hard solid, 50:50 ratio was elastomer, and 30:70 ratio was gummy solid consisting of fused mesh. The two ratios that resulted in fibrous scaffolds, 70:30 and 50:50, were determined to have Young's modulus of 14.2 and 0.8 MPa, respectively [136]	<ul> <li>Greater elasticity than PGA + PCL</li> <li>Similar tensile strength to PLA</li> <li>~5% addition of PCL increased strain by eightfold</li> <li>Overall best synthetic extracel- lular matrix (ECM) for cardiac applications</li> </ul>	<ul> <li>Decreasing PLA + PCL ratios decreases strain capacity, optimized at 95:5</li> </ul>	An engineered vascular graft must be strong enough to accommodate large pressure increase while having enough elasticity to expand to accept large bolus of blood passively and contract to push blood downstream Furthermore, smooth muscle and endothelial cells were seeded on these scaffolds and cultured for 7 days, with favorable results
		Characteristics	Advantages	Disadvantages	Applications
---	--------------------	---	--	--	---
	PDO-PCL	Fibers ranged from 940 nm for pure 100 mg/mL PDO to 1.5 µm for 70:30 ratio of 100 mg/mL PDO:500 mg/mL PCL	<ul><li>PCL high elasticity</li><li>PDO shape memory</li></ul>	<ul> <li>Lower tensile capacity than PDO</li> <li>Lower elasticity than PDO</li> <li>Larger fiber diameter</li> </ul>	
Natural polymers: no controllable degradation rates	Elastin	The most linearly elastic biosolid known. It is a highly insoluble, hydrophobic protein Electrospun solubilized elastin (bovine neck ligament) at concentrations ranging from 200 to 333 mg/mL in HFP, resulting in the creation of scaffold of flat, ribbon-like fibers	<ul> <li>Linearly elastic biosolid</li> <li>Insoluble and hydrophobic</li> <li>Critical role in shape and energy recovery for organs</li> </ul>	<ul> <li>Less elastic than native elastin</li> <li>Needs to be combined with PDO to increase tensile strength</li> <li>Fiber ~300 nm (not as small as PDO ~ 180 nm)</li> <li>Varying diameter</li> </ul>	It constitutes large proportion of walls of arteries and veins to provide elastic recoil and prevent dissipation of transmitted pulsatile energy into heat. It is also a large component of ligaments and lung parenchyma owing to its elastic properties. The skin contains thin strands of elastin that help keep it taut and smooth
	Gelatin collagen	Gelatin (denatured collagen)	<ul><li>Biocompatible and biodegradable</li><li>Inexpensive</li></ul>	• Quick to dissolve	Has been used for many years as vascular prosthetic sealant, carrier for drug delivery, and dressing for wounds. Through electrospinning, has gained interest as tissue engineering scaffold
	Fibrillar collagen	The electrospun collagen solutions ranged in concentration from 0.03 to 0.10 g/mL of HFP resulting in mats/scaffolds composed of 100 nm to 5 µm diameter fibers Electrospun calfskin type I collagen revealed the 67 nm banding that is	<ul> <li>Fibril-forming (types I, II, III)</li> <li>Most abundant natural polymers in body</li> <li>Important role in ECM</li> <li>Type I: principal structure in ECM</li> <li>Type II: pore size and fiber diameter easily controlled</li> <li>Type III: still under investigation</li> </ul>	Typical procedures used to isolate and reprocess this natural scaffolding into engineered material may compromise many of its biological and structural properties	Fibril-forming collagens, types I, II, and III, are most abundant proteins (natural polymers) in the body, are found throughout the interstitial, and provide the overall structural integrity and strength to tissues. More important, the collagen structure (ECM) provides cells with appropriate

## Table 11.1 Synthetic Polymers Characteristics, Advantages and Disadvantages—cont'd

		characteristic of native collagen Hence, an electrospun collagen mat may be a truly biomimicking scaffold, since sub-micron diameter fibers possessing the natural collagen ultrastructure can be created			biological space for embryologic development, organogenesis, cell growth, and wound repair
Collagen blends	Fibrinogen	<ul> <li>Low concentration produced fibers within range of fibrinogen fibers in plasma clots (80, 310, 700 nm)</li> <li>Stress capacity comparable to collagen (80–100 MPa)</li> <li>Electrospun fibrinogen scaffolds can be significantly enhanced, with no loss of bioactivity, by blending the natural polymer with PDO</li> </ul>	<ul> <li>High surface area to volume ratio: Increases area available for clot formation</li> <li>Translates into relative surface area to volume ratio of 1300 cm<sup>2</sup>/cm<sup>3</sup> or relative surface area to weight ratio of 4.1 m<sup>2</sup>/g</li> </ul>		
Synthetic and natural polymer compilation		<ul> <li>PGA, PLA and PLGA most commonly used</li> <li>PDO most similar to elastin collagen blend (limited by shape memory)</li> <li>PCL most elastic and mixed frequently with other materials</li> <li>Provide nanoscale phys- ical features</li> <li>Collagen type I and III + PDO: best possible match for blood vessels</li> </ul>	(Elastin and PDO) dynamic compliance measurements produced values ranging from 1.2% to 5.6%/100 mmHg for set of three different mean arterial pressures, with 50:50 ratio closely mimicking compliance of native femoral artery Hydrated PDO-collagen samples were able to sustain mean peak stresses between 4.6 and 6.7 MPa, mean strain to failure values between 56.5% and 186.4%, and	Scaffolds were electrospun in following blends [PDO —collagen (by wt%)]: 100:0, 90:10, 80:20, and 70:30; when mixture of collagen types I and III was used, it was 50:50 blend of collagens. Scaffold fiber diameters for blends ranged from 210 to 340 nm, but increase in percentage of collagen did not have significant effect on mean fiber diameter	Different ratios of elastin and PDO (100:0-50:50 PDO:elastin by volume) create bioactive scaffolds that remain mechanically viable Mechanical properties of electrospun PDO-collagen blended vascular prosthetics fall within ranges of corresponding values for

Continued

,	,	, 5	3	
	Characteristics	Advantages	Disadvantages	Applications
	<ul> <li>Indication that blends of PDO and collagen may match mechanical and morphological requirements of a blood vessel's microenviron- ment (similar to PDO section)</li> </ul>	mean tangential moduli between 7.6 and 18.0 MPa Hydrated electrospun PDO– collagen structures achieved modulus between maximal modulus of elastin and minimal modulus of collagen. Peak stress exhibited by electrospun PDO–collagen is within range of elastin and near lower limit of collagen. Percent strain at break range of electrospun PDO– collagen spans both ranges of elastin and collagen and exceeds upper limit of collagen	(although addition of collagen to PDO decreased fiber diameter compared with PDO alone)	traditional vascular materials. Human dermal fibroblasts seeded onto PDO—collagen type III scaffolds displayed favorable cellular interactions; cells migrated into thickness of scaffolds containing collagen but merely migrated on seeded surface of 100% PDO scaffold

#### Table 11.1 Synthetic Polymers Characteristics, Advantages and Disadvantages—cont'd

*HFP*, Hexafluoroisopropanol (1,1,1,3,3,3-Hexafluoro-2-propanol)

From I.K. Kwon, S. Kidoaki, T. Matsuda, Electrospun nano-to microfiber fabrics made of biodegradable copolyesters: structural characteristics, mechanical properties and cell adhesion potential, Biomaterials 26 (2005) 3929–3939. Increasing discussion led to certain basic elements that biomimetic constructs must possess:

- 1. it should be 3D,
- 2. it should be rich in matrix that is able to undergo cellular remodeling, and
- **3.** it should have built-in physicochemical signals to direct cell behavior.

Essentially, such constructs must be susceptible to cell-mediated remodeling after implantation. During cellular remodeling the actual ECM environment rapidly becomes protein-based, mainly collagen. In this chapter we will focus on how it is important for 3D biomimetic collagen scaffolds, which are developed to be used for tissue engineering, to resemble ECM. Also, we will discuss carbon nanotube uses for tissue replacement and organizational levels of scaffold structure considering biomimetic elements of nanoscale, microscale, and macroscale.

#### 11.1.1 Collagen-Based Biomimetic Constructs

Collagen is encoded by a large gene family that is subject to strict patterns of temporal regulation. Collagen proteins form an intricate network with themselves and with other extracellular cell matrix proteins, providing a huge range of structures and functions and supporting the *body's structures* and binding or connecting all *tissues*. These complex fibrous protein networks have the main role of maintaining the structural and biology integrity of ECM for both soft and hard connective tissues. These interconnected networks also undergo constant and dynamic remodeling of tissue structure [31]. Furthermore, the ECM participates in cellular signaling and communication through adhesion receptors and its own cleaved proteins (matricellular signaling), which presents and regulates growth factors and cytokines to cells (matricrine signaling) [32]. ECM transduces hemodynamic forces and resists traction forces produced by cells (mechanical signaling) (Fig. 11.1) that occur during development, repair, or natural conditions.

A total of 28 different molecular structures of collagen family have been identified, each of which contains at least one triple-helical domain formed from three  $\alpha$ -chains [31]. The triple helix is composed of three polypeptide chains, each with the repeating triplet Gly-X-Y, in which X and Y are frequently proline (approximately 22% occurrence of each in type I collagen) and 4-hydroxyproline, respectively. Those triple helices can be constituted of homotrimers or heterotrimers with various  $\alpha$ -chain combinations. This huge variety in collagen type may be due to the splicing process, the addition of glycosyl and hydroxyl groups, the presence of other nonhelical domains, or how they assemble, determining their function [33]. Within the huge varieties of the collagen family, some types are more probable to occur as I, II, III, V, and XI, characterized as fibril-forming collagens assembled into highly organized supramolecular fibrils. The most abundant (about 70% of collagen, which constitutes



**FIGURE 11.1** (A) Scale of tissue elasticity ranging from softest (brain) to stiffest (bone). (B) Signals from growth factors bound to the ECM affect cell function by mediating gene expression through various kinases. (C) Cells cultured on gels that mimic a soft tissue environment anchor less strongly to the substrate than do cells cultured on gels that mimic a stiff tissue environment. (Shown in green (light gray in print versions)) It is possible to see focal adhesions. (Shown in red (gray in print versions)) The actin cytoskeleton is presented. *Modified from D.E. Discher, D.J. Mooney, P.W. Zandstra, Growth factors, matrices, and forces combine and control stem cells, Science 324 (2009) 1673–1677, AAAS.* 

mammal tissues) is fibrillar type I collagen, which has the key property of possessing distinct morphologies for different tissues performing different functions depending on its own interactions. Bone tissue is the association of inorganic crystals (hydroxyapatite) providing rigidly and shock-resistant bone tissue with a high Young's modulus [34]. Otherwise in tendons collagen acts like an elastomer, possessing low rigidity and high deformation, allowing rupture, whereas in cornea it has optical properties such as transparency [31]. Type IV collagen is the main nonfibrillar collagens; it is the major component of the basal lamina that forms a *sheet-like* structure, integrating other ECM fibrous (e.g., laminin). There are also more types of collagens including types IX, XII, XIV, XVI, XIX, XX, XXI, and XXII, localized on the surface of collagen fibrils depending on the tissue location. All of these collagens are fibrils that are associated with the fibrous collagens constituting interrupting triple helices. In various connective tissues, microfibrillar collagen type VI generates fine filaments producing a distinct network, whereas other types such as X and VIII, which possess a short chain, form hexagonal networks [35].

#### 11.1.2 Collagen Used as a Biomaterial

Use of collagen as a biomaterial dates back to ancient Egypt, when sutures were made from animal sinew. It is possible that collagen was used even earlier in the first century AD in both Greece and India. During this time collagen may have been used for plastic surgery to repair mutilations caused by battle and punishment [36]. In India a medical encyclopedia by Sushruta recorded a procedure to repair the cut-off noses of patients using a prosthetic nose. This prosthesis was made using skin graft from the patient's cheek [36]. This may be the first record of a prosthetic autograft, the ancient use of biomaterials for tissue engineering. Also, during the Second World War, the first group to report the use of collagen sutures and sheets was that of Professor F.O. Schmitt, who used it to cover denuded areas of skin [37]. A decade later, with great success, Rosenberg's group used modified bovine blood vessels as heterologous grafts for the first time. Many preclinical and clinical studies confirmed the efficiency of this graft [38]. Since these successful medical applications, many others also used biomaterials, e.g., in cardiovascular, plastic, or neurosurgery as injectable fillers for cosmetic surgery including as commercial products (e.g., Zyplast, Zyderm, Contigen), wound dressings [39], and tubes for nerve, skin repair, and blood vessels (e.g., Integra, Apligraf) [40]. Collagen has low antigenicity owing to its preserved amino acid sequence across species, which explains why it can be used in varied application; as a collagen it is an ECM protein and its breakdown products are not harmful, which allows the use of collagen from both porcine and bovine sources for clinical application. It is also possible to reduce the immunogenicity of implanted collagen by using a subproduct of collagen called atelocollagen. Its structure is the nonhelical telopeptide extremity of the collagen molecule [41]. Because collagen was used in medicine it was proof that was safe and biocompatible for tissue engineering applications.

The use of collagen as a construct in tissue engineering can be divided into two general strategies: (1) a *top-down* or decomposition approach, in which the system is essentially broken down, gaining insight into its compositional subsystems. In our concept, ECM from connective tissue is composed mainly with collagen. All cell content is removed in a reducing process; e.g., when collagen is subtract from cadavers or animals, allowing construction of an architecture that resembles that required by tissue; (2) a *bottom-up* approach usually meaning synthesis, when collagen molecular blocks are built, assembled into a scaffold. Therefore, whereas the *top-down* approach reduces the complexity of native tissues, breaking them down, the *bottom-up* approach tends to increases the complexity of the system, assembling the isolated collagen into building blocks for tissue engineering.

To reduce systems, a *top-down* approach commonly uses acidic and basic solutions that can be hypotonic and hypertonic, alcohols, detergents, and enzymes to promote decellularization of the tissue, thus obtaining new scaffolds. The origin material can be obtained from humans, rats, bovines, and porcines allowing the production of constructs for tissue regeneration. The detergent sodium dodecyl sulfate produced blastema

rings obtained from rabbit's pinna colonization [42], and another detergent, Triton X 100, made it possible to reconstruct heart valves [43].

Through the *bottom-up* approach, diverse biomimetic materials together with inorganic, metal, other ECM proteins (e.g., elastin), or nanomaterials (e.g., carbon nanotubes), to mention just a few examples, have been used to fabricate constructs that are used to regenerate tissue. It is commonly used for skin, muscle and bone regeneration and nerve repair, among others [4,34,44–48].

Advantages of using native ECM for grafts are that in vivo cells need integrated physical and chemical communication between themselves and the ECM. In fact, foreign materials do not interact directly with living cells, but upon contact with soluble matrix proteins such as fibronectin, vitronectin, and fibrinogen, they readily attach to the adsorbed layer of proteins in biological fluids [49]. All this joined with physiologic fluids in vivo or culture medium in vitro will affect cellular interaction [50], allowing an ECM 3D structure to be maintained [50,51]. Disadvantages of using "ready-made" ECM as grafts include its density and its relatively impervious structure. It can inhibit cells to repopulate tremendously in the scaffold, slowing process and becoming uncontrollable. Also, a foreign transplant can become immunogenic or promote infection because of residual antigens in animal tissues or even cryptic infectious materials. In addition, it is particularly difficult to remove such macromolecular and cellular elements throughout the dense 3D structure without disrupting it. This is why integrated cellular/ECM scaffolds that are simpler in terms of their composition and architecture represent the most promising emerging technologies for regenerative repair [10].

A huge diversity of work with porous and sponges collagen scaffolds (a pseudo-3D structure) has been conducted because these provide a useful substrate for the bulk delivery of cells [52,53]. Their major application is in tissues that do not need fibers to be aligned, such as in skin replacement. Otherwise, collagen hydrogen represents a 3D scaffold comparable to nanoscale fibers [24,25,54] and their primary use has been as a substrate for the growth and delivery of cells [55], because they are easily seeded interstitially within the fibril network. Other properties of collagen hydrogels include their ability to retain water; their capacity to maintain cells in a stretched position because of their tenacity; their condition to have nanoporosity and microporosity, allowing seeded cells to grow and arrange in 3D; their potential to be biodegradable, creating space for new cells; and their interconnectivity by pores that allowing nutrients and gases flow freely in and around the growing cell mass [56].

The development of the carbon nanotube–collagen (CNT-Col) gel fabrication process was critically influenced by the realization that the excess fluid present in untreated collagen gels is a result of casting. CNT-Col hydrogels rapidly produce dense fibrillar collagen sheets (100–200  $\mu$ m thick) with a tissue-like architecture, strong mechanical properties (greater than those of native tissue collagen), and biomimetic function (e.g., supporting high cell viability) [34,47]. This process is versatile in terms of the volume, density, and shapes that can be produced (e.g., sheets can be layered or rolled), which creates the possibility of fabricating a wide range of final scaffold structures. Indeed,

several studies have demonstrated the suitability of CNT-Col scaffolds in engineering biomimetic tissue constructs such as skin, bladder, muscle, blood vessels, bone, and cornea.

## 11.2 How Scaffold Is Structurally Organized

The regeneration and production of functional tissue are the main goals of tissue engineering. The human body develops from a single diploid cell called a zygote and contains an estimated 85 trillion cells by adulthood, of which more than 150 billion turn over every day. Those trillions of cells are divided into about 260 different phenotypes, many of which go on to proliferate and differentiate and can self-assemble into complex and integrated systems of tissues and organs over time and space [56]. Artificial biomaterial scaffolds are designed to support cell and tissue growth, aiming to recreate the complexity and, nanoscale, microscale, and mesoscale details observed in real organs at the level of the cell-matrix interaction [46,57–60]. All this complex structure is related to living organs, with specific functions resembling the level of organization of the target tissue, also providing the correct morphology of the cell-carrying template.

However, we can distinguish among nanoscale, microscale, and macroscale structural elements, which are part of the structure of constructs, improving cell capacity in regulating tissue/scaffold functions and their biofunctionality [12,47]. These characteristics have profound effects on cell physiology and communication, and also influence cell shape and gene expression, which are related to cell growth and the maintenance of original host cells [61]. In this section we present some major levels of organization of native tissue and give some examples of how to build biomimetic constructs incorporating such elements.

### 11.2.1 Nanoscale and Microscale Scaffold Structure of Biomimetic Elements

The ECM's fibers at nanoscale and microscale reveal details not found on artificial materials. Many different fibers with distinct roles participate in the ECM's structure, providing suitable conditions for cells to form complex and different tissues such as brain, heart, liver, and bone. A major property of eukaryotic cells is that their proteins from the cytoskeleton, such as actomyosin, are capable of modifying cell shape according to signaling molecules and the environment. There are many more cytoskeletal proteins from cells that interact with ECM proteins. All of these cell–ECM interacting proteins have a key role as a substrate for cellular adhesion and can drive cell signaling and differentiation. Molecules such as collagen, laminin, fibronectin, integrins, and various proteoglycans are part of the ECM [32]. Some of these proteins are part of cell secretion in response to stress on the cytoskeleton. Interaction between the cytoskeleton and the ECM is important in fibrosis, wound healing, tumor growth (metastasis), and connective tissue development [32,62,63]. ECM proteins present multiple motifs that

have as their target cell receptors. In mammals, over 24 different integrin heterodimers have been described and they recognize motifs such as Arg-Gly-Asp within proteins of the ECM such as fibronectin, osteopontin, and vitronectin [64,65]. When bound, integrins activate a cascade of intracellular signaling pathways, leading to changes in gene expression and affecting most aspects of cell behavior. An important property of integrins is their connection to form focal adhesions. In addition, integrin distribution, structure, and function in native tissue with 3D architecture are distinct from those in 2D culture [66,67].

In bone, for example, the 67-nm-wide periodic spacing observed in type I collagen is essential for the proper assembly of ions leading to the nucleation of hydroxyapatite crystals. At a higher scale, 50- to 500-nm-diameter collagen fiber of the ECM surrounds, interacts with, and signals osteoblasts, whereas osteocytes rely on microscopic 1- to  $5-\mu$ m-diameter canaliculi for their viability, communication, and proper function. Likewise, multiscale topographic characteristics of the ECM of all tissues have a key role in structure and signaling. Examples include the diverse porosity and topography of the epithelial basement membrane effects on cell polarity and migration during embryogenesis [68,69] and contact guidance provided by Schwann cells surrounded by basal lamina that guide axon growth in peripheral nerve regeneration [70].

Cells are sensitive not only to ECM adhesion but also to its rigidity and elasticity. It was observed that there is an inverse correlation between matrix density or rigidity and cell migration. Discher and collaborators demonstrated the importance of matrix elasticity on stem cell fate [71–73]. Depending on the elasticity of the surface on which the mesenchymal stem cells (MSCs) were grown, they could differentiate into lineages that corresponded to the stiffness of the native environment that was resembled (Fig. 11.1).

For example, MSCs cultured on soft gels (approximately 0.1 to 1 kPa) to mimic brain elasticity developed a neuronal morphology, with filopodia branching and spreading including expressing genes related to neuronal differentiation pattern. Furthermore, medium-stiffness gels (8–17 kPa), which mimic striated muscle elasticity, promoted differentiation to myogenic cells, and gels with the highest stiffness (25–40 kPa), to mimic bone elasticity, enhanced osteogenic differentiation.

Increasing matrix stiffness disrupts cell morphology and leads to increased proliferation. As shown in Fig. 11.1B, increasing ECM rigidity elevates the activity of the Ras homolog gene family member A, which subsequently induces cytoskeletal tension, decreases cell–cell contact, disrupts cell polarity, and increases the growth rate.

Furthermore, other architectural features of the ECM are important concerning cell behavior. For example, because cell metabolism occurs by consuming nutrients, it is necessary to remove their degradation residues, which corresponds to the diffusion of the system. Diffusion properties of extension and nature can be largely affected by the porosity and permeability of the ECM, which have a direct influence on cell processes. All of these characteristics demonstrate that physical cues are as important as chemical ones in many aspects of tissue engineering [10,11,74,75]. Unfortunately, the signaling mechanism and the way it works related to material-based signals are not yet clear.

This topology-dependent phenomenon is referred to as "ECM dimensionality sensing." Migrating cell morphology and directional persistence are markedly altered in 3D culture systems compared with standard 2D cultures [69,76,77].

A distinct effect between 2D and 3D cell culture ECM is that in the first, when the dorsal cell surface, which is in contact only with culture liquid, is substituted with another substrate (e.g., synthetic or even collagen gels), it is possible to modify the cell morphology and turn the cell physiology to migrating cells [78,79]. Also it causes profound effects on the cell cytoskeleton [80] and dynamics [81] resembling that of cells culture in 3D culture [82]. A 3D environment also influences solute *diffusion* and binds many effector *proteins, ensuring* cell viability and optimal function [56]. The scaffold architecture also affects cell attachment and spreading. Cell interaction with fiber-based scaffolds occurs within microscale architectures that flatten and spread, resembling that of the cell culture on flat surfaces, regarded as a 2D culture (Fig. 11.2). It is possible to create a *pseudo* third dimension in the system by causing modifications or disturbances during scaffold production, incorporating some pores, grooves, or channels with approximately the same cellular dimensions (Fig. 11.2). For example, a large pore size can create a pseudo-3D surface, which presents cells on flat surfaces, but distributed



FIGURE 11.2 Schematic demonstration of cell attachment on 2D, pseudo-3D, and 3D surfaces. (A) Cells binding to scaffolds with fiber-based microscale architectures flatten and spread as if cultured on flat 2D surfaces. These cells have a polarized relationship, with one side of the cell attached to the scaffold and the other exposed to physiologic media. (B) Scaffolds containing pores, grooves, and channels of cellular dimensions provide a pseudothird dimension in the system because cells are presented with a series of effectively flat surfaces, but in all three planes. (C) Scaffolds with nanoscale architectures provide a third dimension and have larger surface areas to adsorb proteins, with many more binding sites to cell membrane receptors, allowing cell attachment to the substrate through most of their membrane surface.

throughout an 3D plan. On the other hand, in 3D cultures, the truly biomimetic scaffold, all cell surface is able to be attached to the construct surface. This is what happens with constructs nanoscale structures that have large surface area to volume ratios absorbing proteins, and allow more sites with which ECM receptors can bind [12,83,84]. That why stem cells and more specialized cells (e.g., neural, osteoblast, smooth muscle) have a greater proliferation rate on nanofiber-based than non–fiber-based scaffolds [83]. However, the concept of 3D cell attachment is not as simple as that because each side of the cell in 3D scaffolds has different possibilities for attachment. Even when cells can attach on their 360 degree surface, the 3D cell attachment modifies cell behavior through migration and differentiation and their relationship with neighboring cells as contact and communication. In addition, asymmetries in the scaffold structure can promote local distinct attachment.

The diverse nature of tissue architecture requires that different microenvironments be provided for repair or regeneration. This principle can be demonstrated in the use of scaffolds with tissue-specific pore sizes of different lengths according to tissue rigidity and elasticity. This was demonstrated for in vitro bone tissue regeneration that required pore sizes ranging from 100 to 350  $\mu$ m, as well as 40–100  $\mu$ m for osteoid ingrowth, and 20 µm for the ingrowth of hepatocytes [85,86]. Adult mammalian skin could be cultivated within a construct with a pore size of  $20-125 \,\mu\text{m}$  and for liver regeneration with 45–150 um [87]. Research has described ideal pore sizes to improve cell physiology (proliferation, differentiation, migration, matrix production, and nutrient and gas diffusion) [2,22,23] using random, nondirectional materials [88,89]. It was demonstrated that fibroporous meshes made of small-diameter fibers are less susceptible to encapsulation than are other architectures [90-92]. Many studies also described how surface charge can modify vessel ingrowth into small-fiber, fibrous encapsulation, fiber porous, the number of cells within a scaffold, and material meshes. Meshes electrospun from negatively charged polyurethane with defined fiber diameters of about 5 µm and a mean fiber spacing of 65 µm may facilitate vessel ingrowth into fibroporous mesh biomaterials have been described [93,94]. Thus the use of small-fiber, fibroporous meshes combined with surface coatings of an appropriate surface charge might facilitate specific cell nuclei densities and vessel ingrowth responses without fibrous capsule formation [94].

Other research has shown that the pore size of a construct allows vascularization of macroporous biomimetic materials in vivo [95]. Blood vessels can grow ideally within pore sizes greater than 400  $\mu$ m. A pore size smaller than 400  $\mu$ m of diameter will restrain the blood vessels growth, which generates vessels with small diameter. [95]. Also many studies have been conducted to describe the relationships among distinct characteristics such as porosity, pore size, and shape, and how are they interconnected, all of which are a macroporous biomimetic material for tissue ingrowth [96–99]. Among those parameters, pore size is receiving the most attention. Generally (although it is an open question), the pore size is 50–100  $\mu$ m, and if they are interconnect with a larger pathway greater that 20  $\mu$ m it is possible to obtain blood vessels enabling cell invasion with a greater biological response [100–102]. The first studies demonstrated that small

pores induce faster tissue ingrowth and more tissue formation [103,104], although later studies demonstrated faster bone ingrowth with larger pores [98,105–108]. Someone proposed an explanation for this apparent controversy: "smaller pores expose a much larger surface to the invading tissue elements and thus favor deposition." If we consider each of these conditions on the porous structure of constructs, it is necessary to evaluate three properties; the number of pores on surfaces (porosity), how much these pores are interconnected and how great this pathway is (interconnection size), and how great the macropore is. All of this has a direct or indirect influence on tissue ingrowth. However there is no common point of view to explain the adequate macropore size for tissue ingrowth, probably because there is no efficiently controlled production of the porous surface on the constructs. Moreover, when someone varies one parameter, e.g., the pore size, other parameters are also varied, e.g., the porosity or the size of pore interconnections [95]. Also, the charge surface, elasticity and hardness can control cell differentiation and proliferation, which can take more variables to determine the properties of pore sizes. It will be enlightening if new research takes into account the charge surface, elasticity and hardness, and porosity and interconnections to evaluate tissue ingrowth including blood vessel formation.

In tissue the physiologic conditions, cell distribution, cell survival, migration, proliferation, and differentiation are ensured by ideal uniform distribution, high porosity, and pore interconnectivity [47]. However, such criteria can be successfully determined only at early culture stages when cells occupy the pores on scaffolds and fill them with ECM constituents that limit cell perfusion. This will promote a diverse randomly distribution of cells and ECM network, enabling more identical spatial and random distribution of the pores. In addition, this is not a good biomimetic template for spatial organization during normal development, and in healthy tissue, these structures resemble scar tissue that has disoriented fibers and a disoriented structure [10,11]. More than the scale of topography (nanoscale or microscale), the presence of ridges, grooves, pockets, and planes that design the pattern of the substrate, together with its symmetry independent of the underlying material's chemistry [109], contribute to modulate cell behavior [110].

For example, microposts have been used to study cell adhesion [111–113] and stimulate mesenchymal stem cell growth [114,115]. Microchannels have been used to guide cell morphology and migration of a large variety of cells [116] and microholes have been used to enhance osteogenesis [117,118] and cell differentiation [119], study cyto-skeletal morphology [120], or create platforms for the controlled growth of undifferentiated embryonic stem cells [121,122]. Other types of microgeometries have been used to optimize nonviral transfection systems [123] or investigate signal transduction in cell migration [124].

Controlled cellular alignment has a crucial role in the microarchitecture of many human tissues, dictating their biological and mechanical function. Furthermore, fiber alignment directly determines tissues' mechanical properties [125] and provides a dynamic topographic cue to resident cells [126,127].

#### 11.2.2 Biomimetic Elements of Macroscale Scaffold Structure

Many tissue-engineering approaches must supply a scaffold that immediately mimics the functional architecture of the damaged tissue. For example, heart valve replacements need to be functional from the start, as opposed to bone repair sites that could be supported and immobilized until functionally mature. Engineering homogeneous tissues that can mimic the gross architecture of an organ is easily acquired by scaffold construction using native (e.g., collagen, fibrin, hyaluronic acid, or other ECM matrix constituents) or synthetic (e.g., polylactic acid, caprolactone, or others polymeric gels) biomaterials that function as integral units providing a 3D support to cultured cells [27]. Another method that carries the potential for engineering composite tissue structures with macro/microscale heterogeneity (e.g., osteochondral, nerve fascia, and vascular wall structures) is that of sheet-based engineering such as layer-by-layer tissue reconstruction using either cell sheets [128] or biomaterial sheets [129].

Cell-sheet engineering approaches imply the use of temperature-responsive polymer culture surfaces in which cultured cells can be retrieved or harvested as intact sheets by nondestructive (protease-free) ways [29,128]. However, the use of cell-sheet approaches for tissue engineering is not suitable for the generation of matrix-rich, cell-sparse tissues such as skin, tendon, bone, or cartilage, unlike the scaffold-based approach, because cell sheets contain relatively little ECM [128]. In addition, it is also difficult to fabricate a 3D architecture similar to that found in cardiac tissue and liver. To overcome these difficulties, is possible to layer individual cell sheets producing millimeter-thick tissues and integrate multilayer cell sheets with other biomaterial sheet scaffolds that can provide structural support to the construct.

Undoubtedly, the optimal approach to successful large-construct fabrication must rely on engineering angiogenesis by promoting the new formation of a functional microvascular network that can be integrated with other scaffold components, supporting their viability. It is important, however, that scaffold structures be engineered around the establishment of angiogenesis.

As previously discussed, Nano- and Microtopography is important for cell-surface interactions. The 3D spatial organization of sheet-based scaffolds can assume more than one form depending on the tissue guidance systems, even if it be more flexible than fifiber-based guidance. In this case of guidance directed by micrometer diameter fibrils, the cells motility and migration are constrained to only one plane at a time. The organization of the scaffold fibers, as well as by the efficiency of cell guidance. In contrast, interface guidance, acquired with 3D scaffolds, is an example of single-plane constraint. In this case, the restriction of motion is in only one plane, with movement permissible in the other two planes between adjacent sheets (within the interface). However, free cell motion between the layers or interfaces and along the length of the construct is possible, with cells following the interface [87]. Whereas constraint is achieved entirely by the density of the substrate (in this case, collagen), migrant cells gradually degrade it. This

emphasizes the importance of temporary, early cues to guidance [87]. This form of guidance resembles that of connective tissues present in sliding surfaces between organs and conductive tissues, such as nerves and blood vessels [130–132]. This biomimetic approach, which can control tissue structure such as cell guidance, represents a potential method for improving scaffold biofunctionality by interacting between sheet/ layer 3D spatial organization and perfusion of resident cells within the participating layers [133].

## 11.3 Effects of Scaffold Physical Structure on Cell Behavior 11.3.1 Effects of Scaffold Topography on Cell Function

Cell function can be regulated by a nonbiological and noninvasive method termed engineering scaffold topography, which acts as an extracellular physical milieu without involving biomolecules. However, this method can provide a biomimetic cellstimulating signal, because cells in vivo have contacts at the mesoscale structure imitating natural tissue structures, unlike smooth interfaces. We can classify topography with regard to its scale and pattern. At the basement membrane level, composition includes pits, pores, protrusions, striations, particulates, and fibers. These are structures of nanoscale dimension (5-200 nm), which may be more biomimetic in affecting cell behavior than microtopography [134]. Regarding topography patterns, cell functions such as differentiation and proliferation were evaluated, and can be affected through cell alignment along anisotropic topographies represented by grooves and ridges, or isotropic topographies, in which protrusions or pits are randomly or evenly distributed [110]. Beyond the topography classification on scale and pattern, it is important to distinct whether the patterned substrate is fabricated to allow interstitial or superficial cell seeding: in other words, to have two or three dimensions, because it is imperative regarding stromal cells that react with surface topography in different ways compared with epithelial cells. Although 3D structures resemble tissue structures with much more reliability, the role of topography in cell function regulation has been studied mainly on 2D substrates [110]. Although 2D surfaces are valuable tools for studying basic cellular response to nanotopography, translation of these findings toward clinical application will require 3D structures.

## 11.4 Conclusions and Perspectives

In this chapter, the influence of different scaffolds on cell behavior and function has been described, as well as their organizational levels (nano, micro, and macro scaffold structure). These topics are closely related because the main goal when developing scaffolds for tissue engineering is to mimic, with greater similarity, the structure of the tissue's ECM. To do this, a biomimetic scaffold has been developed through different methods. A large number of reports has covered the cell's reaction on 2D and 3D substrates with different topographies. The cell–surface interaction and the cell's migration, proliferation, and differentiation in response to a specific nanotopography have also received attention. It is expected that this chapter will enlighten readers about the importance of scale and topography of scaffolds resembling the ECM structure in regenerative medicine, especially when it comes to their use in tissue engineering.

## Acknowledgments

The authors are grateful for CNPq (Conselho Nacional de Desenvolvimento Científico e Tecnológico), FAPEMIG (Fundação de Amparo à Pesquisa do Estado de Minas Gerais), and CAPES (Coordenação de Aperfeiçoamento de Pessoal Nível Superior) financial support.

## References

- K.I. Jeong, S.G. Kim, J.S. Oh, S.Y. Lee, Y.S. Cho, S.S. Yang, S.C. Park, J.S. You, S.C. Lim, M.A. Jeong, J. S. Kim, S.Y. Lee, Effect of platelet-rich plasma and platelet-rich fibrin on peri-implant bone defects in dogs, Journal of Biomedical Nanotechnology 9 (3) (2013) 535–537.
- [2] R. Langer, J.P. Vacanti, Tissue engineering, Science 260 (5110) (1991) 920-926.
- [3] L. Jin, Z.Q. Feng, M.L. Zhu, T. Wang, M.K. Leach, Q. Jiang, A novel fluffy conductive polypyrrole nano-layer coated PLLA fibrous scaffold for nerve tissue engineering, Journal of Biomedical Nanotechnology 8 (5) (2012) 779–785.
- [4] M.S. Peach, S.G. Kumbar, R. James, U.S. Toti, D. Balasubramaniam, M. Deng, B. Ulery, A.D. Mazzocca, M.B. McCarthy, N.L. Morozowich, H.R. Allcock, C.T. Laurencin, Design and optimization of polyphosphazene functionalized fiber matrices for soft tissue regeneration, Journal of Biomedical Nanotechnology 8 (1) (2012) 107–124.
- [5] B. Buyuktimkin, Q. Wang, P. Kiptoo, J.M. Stewart, C. Berkland, T.J. Siahaan, Vaccine-like controlled-release delivery of an immunomodulating peptide to treat experimental autoimmune encephalomyelitis, Molecular Pharmaceutics 9 (4) (2012) 979–985.
- [6] T.H.B. Eriksen, E. Skovsen, P. Fojan, Release of antimicrobial peptides from electrospun nanofibres as a drug delivery system, Journal of Biomedical Nanotechnology 9 (3) (2013) 492–498.
- [7] Q. Wang, N. Zhang, X.W. Hu, J.H. Yang, Y.M. Du, Alginate/polyethylene glycol blend fibers and their properties for drug controlled release, Journal of Biomedical Materials Research 82A (1) (2007) 122–128.
- [8] Q. Wang, N. Zhang, X.W. Hu, J.H. Yang, Y.M. Du, Chitosan polyethylene glycol blend fibers and their properties for drug controlled release, Journal of Biomedical Materials Research 85A (4) (2008) 881–887.
- [9] B.R. Sousa, R.C. Parreira, E.A. Fonseca, M.J. Amaya, F.M. Tonelli, S.M. Lacerda, P. Lalwani, A.K. Santos, K.N. Gomes, H. Ulrich, A.H. Kihara, R.R. Resende, Human adult stem cells from diverse origins: an overview from multiparametric immunophenotyping to clinical applications, Cytometry Part A 85 (1) (2014) 43–77.
- [10] R.A. Brown, J.B. Phillips, Cell responses to biomimetic protein scaffolds used in tissue repair and engineering, International Review of Cytology 262 (2007) 75–150.
- [11] J.J. Tomasek, G. Gabbiani, B. Hinz, C. Chaponnier, R.A. Brown, Myofibroblasts and mechanoregulation of connective tissue remodelling, Nature Reviews Molecular Cell Biology 3 (5) (2002) 349–363.

- [12] M.M. Stevens, J.H. George, Exploring and engineering the cell surface interface, Science 310 (5751) (2005) 1135–1138.
- [13] R.R. Resende, A. Adhikarid, J.L. da Costa, E. Lorençon, M.S. Ladeira, S. Guatimosim, A.H. Kiharag, L.O. Ladeira, Influence of spontaneous calcium events on cell-cycle progression in embryonal carcinoma and adult stem cells, Biochimica et Biophysica Acta (BBA) – Molecular Cell Research 1803 (2) (2010) 246–260.
- [14] R.R. Resende, J.L. da Costa, A.H. Kihara, A. Adhikari, E. Lorencon, Intracellular Ca<sup>2+</sup> regulation during neuronal differentiation of murine embryonal carcinoma and mesenchymal stem cells, Stem Cells and Development 19 (3) (2010) 379–394.
- [15] R.R. Resende, A. Adhikari, Cholinergic receptor pathways involved in apoptosis, cell proliferation and neuronal differentiation, Cell Communication and Signaling 7 (20) (2009).
- [16] M.C. Pinto, A.H. Kihara, V.A. Goulart, F.M. Tonelli, K.N. Gomes, H. Ulrich, R.R. Resende, Calcium signaling and cell proliferation, Cellular Signalling 27 (11) (2015) 2139–2149.
- [17] R.R. Resende, L.M. Andrade, A.G. Oliveira, E.S. Guimaraes, S. Guatimosim, M.F. Leite, Nucleoplasmic calcium signaling and cell proliferation: calcium signaling in the nucleus, Cell Communication Signaling 11 (2013) 14.
- [18] R.R. Resende, K.N. Gomes, A. Adhikari, L.R. Britto, H. Ulrich, Mechanism of acetylcholine-induced calcium signaling during neuronal differentiation of P19 embryonal carcinoma cells in vitro, Cell Calcium 43 (2) (2008) 107–121.
- [19] V. Paschon, G.S. Higa, R.R. Resende, L.R. Britto, A.H. Kihara, Blocking of connexin-mediated communication promotes neuroprotection during acute degeneration induced by mechanical trauma, PLoS One 7 (2012).
- [20] T. Glaser, R.R. Resende, H. Ulrich, Implications of purinergic receptor-mediated intracellular calcium transients in neural differentiation, Cell Communication and Signaling 11 (2013) 12.
- [21] A.H. Kihara, T.O. Santos, E.J. Osuna-Melo, V. Paschon, K.S.M. Vidal, P.S. Akamine, L.M. Castro, R.R. Resende, D.E. Hamassaki, L.R. Britto, Connexin-mediated communication controls cell proliferation and is essential in retinal histogenesis, International Journal of Developmental Neuroscience 28 (1) (2010) 39–52.
- [22] D.J. Mooney, C.L. Mazzoni, C. Breuer, K. McNamara, D. Hern, J.P. Vacanti, R. Langer, Stabilized polyglycolic acid fibre-based tubes for tissue engineering, Biomaterials 17 (2) (1960) 115–124.
- [23] I.V. Yannas, J.F. Burke, Design of an artificial skin. I. Basic design principles, Journal of Biomedical Materials Research 14 (1) (1980) 65–81.
- [24] E. Bell, B. Ivarsson, C. Merrill, Production of a tissue-like structure by contraction of collagen lattices by human fibroblasts of different proliferative potential in vitro, Proceedings of the national Academy of Sciences of the United States of America 76 (3) (1979) 1274–1278.
- [25] B.C. Isenberg, C. Williams, R.T. Tranquillo, Small-diameter artificial arteries engineered in vitro, Circulation Research 98 (2006) 25–35.
- [26] C.P. Barnes, S.A. Sell, E.D. Boland, D.G. Simpson, G.L. Bowlin, Nanofiber technology: designing the next generation of tissue engineering scaffolds, Advanced Drug Delivery Reviews 59 (14) (2007) 1413–1433.
- [27] L.E. Freed, F. Guilak, X.E. Guo, M.L. Gray, R. Tranquillo, J.W. Holmes, M. Radisic, M.V. Sefton, D. Kaplan, G. Vunjak-Novakovic, Advanced tools for tissue engineering: scaffolds, bioreactors, and signaling, Tissue Engineering 12 (12) (2006) 3285–3305.
- [28] P.R. Sreerekha, D. Menon, S.V. Nair, K.P. Chennazhi, Fabrication of fibrin based electrospun multiscale composite scaffold for tissue engineering applications, Journal of Biomedical Nanotechnology 9 (5) (2013) 790–800.

- [29] I. Elloumi-Hannachi, M. Yamato, T. Okano, Cell sheet engineering: a unique nanotechnology for scaffold-free tissue reconstruction with clinical applications in regenerative medicine, Journal of Internal Medicine 267 (1) (2010) 54–70.
- [30] A. Kushida, M. Yamato, C. Konno, A. Kikuchi, Y. Sakurai, T. Okano, Temperature-responsive culture dishes allow nonenzymatic harvest of differentiated Madin-Darby canine kidney (MDCK) cell sheets, Journal of Biomedical Materials Research 51 (2) (2000) 216–223.
- [31] L. Cen, W. Liu, L. Cui, W. Zhang, Y. Cao, Collagen tissue engineering: development of novel biomaterials and applications, Pediatric Research 63 (2008) 492–496.
- [32] A.K. Santos, F.M.P. Tonelli, D.A.F. Silva, K.N. Gomes, L.O. Ladeira, R.R. Resende, The role of cell adhesion, cell junctions, and extracellular matrix in development and carcinogenesis, in: Trends in Stem Cell Proliferation and Cancer Research, Springer Netherlands, 2013, pp. 13–26.
- [33] K. Gelse, E. Poschl, T. Aigner, Collagens-structure, function, and biosynthesis, Advanced Drug Delivery Reviews 55 (12) (2003) 1531–1546.
- [34] E.E. Silva, H.H.M.D. Colleta, A.S. Ferlauto, R.L. Moreira, R.R. Resende, S. Oliveira, G.T. Kitten, R.G. Lacerda, L.O. Ladeira, Nanostructured 3-D collagen/nanotube biocomposites for future bone regeneration scaffolds, Nano Research 2 (6) (2009) 462–473.
- [35] M.D. Shoulders, R.T. Raines, Collagen structure and stability, Annual Review of Biochemistry 78 (2009) 929–958.
- [36] L.M. Zimmerman, I. Veith, Great Ideas in the History of Surgery, Norman Surgery, San Francisco, 1993.
- [37] K.H. Stenzel, T. Miyata, A.L. Rubin, Collagen as a biomaterial, Annual Review of Biophysics and Bioengineering 3 (1974) 231–253.
- [38] N. Rosenberg, G.H. Lord, J. Henderson, J.W. Bothwell, E.R. Gaughran, Collagen arterial graft of bovine origin: seven-year observations in the dog, Surgery 6 (1970) 951–956.
- [39] R.S. Narins, P.H. Bowman, Injectable skin fillers, Clinics in Plastic Surgery 32 (2) (2005) 151-162.
- [40] D.M. Supp, S.T. Boyce, Engineered skin substitutes: practices and potentials, Clinics in Dermatology 23 (4) (2005) 403–412.
- [41] A.K. Lynn, I.V. Yannas, W. Bonfield, Antigenicity and immunogenicity of collagen, Journal of Biomedical Materials Research Part B: Applied Biomaterials 71B (2) (2004) 343–354.
- [42] S. Naderi, J. Khayat Zadeh, N. Mahdavi Shahri, K. Nejad Shahrokh Abady, M. Cheravi, J. Baharara, S.A. Banihashem Rad, A.R. Bahrami, Three-dimensional scaffold from decellularized human gingiva for cell cultures: glycoconjugates and cell behavior, Cell Journal 15 (2) (2013) 166–175.
- [43] E. Rieder, M.T. Kasimir, G. Silberhumer, G. Seebacher, E. Wolner, P. Simon, G. Weigel, Decellularization protocols of porcine heart valves differ importantly in efficiency of cell removal and susceptibility of the matrix to recellularization with human vascular cells, The Journal of Thoracic and Cardiovascular Surgery 127 (2) (2004) 399–405.
- [44] K.R. Remya, J. Joseph, S. Mani, A. John, H.K. Varma, P. Ramesh, Nanohydroxyapatite incorporated electrospun polycaprolactone/polycaprolactone-polyethyleneglycol-polycaprolactone blend scaffold for bone tissue engineering applications, Journal of Biomedical Nanotechnology 9 (9) (2013) 1483–2149.
- [45] R. Nirmala, H.S. Kang, H.M. Park, R. Navamathavan, I.S. Jeong, H. Kim, Silver-loaded biomimetic hydroxyapatite grafted poly(epsilon-caprolactone) composite nanofibers: a cytotoxicity study, Journal of Biomedical Nanotechnology 8 (1) (2012) 125–132.
- [46] F.M.P. Tonelli, A.K. Santos, K.N. Gomes, E. Lorencon, S. Guatimosim, L.O. Ladeira, R.R. Resende, Carbon nanotube interaction with extracellular matrix proteins producing scaffolds for tissue engineering, International Journal of Nanomedicine 7 (2012) 4511–4529.

- [47] R.R. Resende, E.A. Fonseca, F.M. Tonelli, B.R. Sousa, A.K. Santos, K.N. Gomes, S. Guatimosim, A.H. Kihara, L.O. Ladeira, Scale/topography of substrates surface resembling extracellular matrix for tissue engineering, Journal of Biomedical Nanotechnology 10 (7) (2014) 1157–1193.
- [48] F.M. Tonelli, V.A. Goulart, K.N. Gomes, M.S. Ladeira, A.K. Santos, E. Lorencon, L.O. Ladeira, R.R. Resende, Graphene-based nanomaterials: biological and medical applications and toxicity, Nanomedicine (2015) 2423–2450.
- [49] F. Grinnell, K. Toda, Activation of human keratinocyte fibronectin receptor function in relation to other ligand-receptor interactions, Journal of Investigative Dermatology 88 (1986) 382.
- [50] S.F. Badylak, D.O. Freytes, T.W. Gilbert, Extracellular matrix as a biological scaffold material: structure and function, Acta Biomaterialia 5 (1) (2009) 1–13.
- [51] S.F. Badylak, Regenerative medicine and developmental biology: the role of the extracellular matrix, The Anatomical Record Part B: The New Anatomist 287B (1) (2005) 36–41.
- [52] M. Radisic, M. Euloth, L. Yang, R. Langer, L.E. Freed, G. Vunjak-Novakovic, High-density seeding of myocyte cells for cardiac tissue engineering, Biotechnology and Bioengineering 82 (4) (2003) 403–414.
- [53] V. Kroehne, I. Heschel, F. Schugner, D. Lasrich, J.W. Bartsch, H. Jockusch, Use of a novel collagen matrix with oriented pore structure for muscle cell differentiation in cell culture and in grafts, Journal of Cellular and Molecular Medicine 12 (5A) (2008) 1640–1648.
- [54] D. Karamichos, R.A. Brown, V. Mudera, Complex dependence of substrate stiffness and serum concentration on cell-force generation, Journal of Biomedical Materials Research Part A 78A (2) (2006) 407–415.
- [55] D.G. Wallace, J. Rosenblatt, Collagen gel systems for sustained delivery and tissue engineering, Advanced Drug Delivery Reviews 55 (12) (2003) 1631–1649.
- [56] R.C. Dutta, A.K. Dutta, Cell-interactive 3D-scaffold; advances and applications, Biotechnology Advances 27 (4) (2009) 334–339.
- [57] E. Lorencon, A.S. Ferlauto, S. de Oliveira, D.R. Miquita, R.R. Resende, R.G. Lacerda, L.O. Ladeira, Direct production of carbon nanotubes/metal nanoparticles hybrids from a redox reaction between metal ions and reduced carbon nanotubes, ACS Applied Materials & Interfaces 1 (10) (2010) 2104–2106.
- [58] M.S. Ladeira, V.A. Andrade, E.R. Gomes, C.J. Aguiar, E.R. Moraes, J.S. Soares, E.E. Silva, R.G. Lacerda, L.O. Ladeira, A. Jorio, P. Lima, M.F. Leite, R.R. Resende, S. Guatimosim, Highly efficient siRNA delivery system into human and murine cells using single-wall carbon nanotubes, Nanotechnology 21 (38) (2010).
- [59] P.A. Martins-Junior, C.E. Alcantara, R.R. Resende, A.J. Ferreira, Carbon nanotubes: directions and perspectives in oral regenerative medicine, Journal of Dental Research 92 (2013) 575–583.
- [60] E. Lorencon, R.G. Lacerda, L.O. Ladeira, R.R. Resende, A.S. Ferlauto, U. Schuchardt, R.M. Lago, Thermal behavior of carbon nanotubes decorated with gold nanoparticles, Journal of Thermal Analysis and Calorimetry 105 (3) (2011) 953–959.
- [61] K.F. Leong, C.M. Cheah, C.K. Chua, Solid freeform fabrication of three-dimensional scaffolds for engineering replacement tissues and organs, Biomaterials 24 (13) (2003) 2363–2367.
- [62] R. Mentlein, K. Hattermann, J. Held-Feindt, Migration, metastasis, and more: the role of chemokines in the proliferation, spreading, and metastasis of tumors, in: Trends in Stem Cell Proliferation and Cancer, Springer Netherlands, 2013, pp. 339–358.
- [63] D.J. Behonick, Z. Werb, A bit of give and take: the relationship between the extracellular matrix and the developing chondrocyte, Mechanisms of Development 120 (11) (2003) 1327–1336.
- [64] M. Barczyk, S. Carracedo, D. Gullberg, Integrins, Cell and Tissue Research 339 (1) (2010) 269–280.

- [65] A. Pozzi, R. Zent, Extracellular matrix receptors in branched organs, Current Opinion in Cell Biology 23 (5) (2011) 547–553.
- [66] E. Cukierman, R. Pankov, D.R. Stevens, K.M. Yamada, Taking cell-matrix adhesions to the third dimension, Science 294 (5547) (2001) 1708–1712.
- [67] G. Zanatta, M. Rudisile, M. Camassola, J. Wendorff, N. Nardi, C. Gottfried, P. Pranke, C.A. Netto, Mesenchymal stem cell adherence on poly(D, L-lactide-co-glycolide) nanofibers scaffold is integrin-beta 1 receptor dependent, Journal of Biomedical Nanotechnology 8 (2) (2012) 211–218.
- [68] Y. Yang, K.W. Leong, Nanoscale surfacing for regenerative medicine, Wiley Interdisciplinary Reviews: Nanomedicine and Nanobiotechnology 2 (5) (2010) 478–495.
- [69] R.J. Petrie, A.D. Doyle, K.M. Yamada, Random versus directionally persistent cell migration, Nature Reviews Molecular Cell Biology 10 (8) (2009) 538–549.
- [70] A. Hurtado, J.M. Cregg, H.B. Wang, D.F. Wendell, M. Oudega, R.J. Gilbert, J.W. McDonald, Robust CNS regeneration after complete spinal cord transection using aligned poly-L-lactic acid microfibers, Biomaterials 32 (26) (2011) 6068–6079.
- [71] A.J. Engler, S. Sen, H.L. Sweeney, D.E. Discher, Matrix elasticity directs stem cell lineage specification, Cell 126 (4) (2006) 677–689.
- [72] K. Saha, A.J. Keung, E.F. Irwin, Y. Li, L. Little, D.V. Schaffer, K.E. Healy, Substrate modulus directs neural stem cell behavior, Biophysical Journal 95 (9) (2008) 4426–4438.
- [73] N.D. Leipzig, M.S. Shoichet, The effect of substrate stiffness on adult neural stem cell behavior, Biomaterials 30 (36) (2009) 6867–6878.
- [74] D.S. Gray, J. Tien, C.S. Chen, Repositioning of cells by mechanotaxis on surfaces with micropatterned Young's modulus, Journal of Biomedical Materials Research Part A 66A (3) (2013) 605–614.
- [75] G.D. Pins, M. Toner, J.R. Morgan, Microfabrication of an analog of the basal lamina: biocompatible membranes with complex topographies, The FASEB Journal 14 (3) (2000) 593–602.
- [76] J.T. Erler, V.M. Weaver, Three-dimensional context regulation of metastasis, Clinical & Experimental Metastasis 26 (1) (2009) 35–49.
- [77] R. Pankov, Y. Endo, S. Even-Ram, M. Araki, K. Clark, E. Cukierman, K. Matsumoto, K.M. Yamada, A Rac switch regulates random versus directionally persistent cell migration, Journal of Cell Biology (2005) 793–802.
- [78] K.A. Beningo, Y.L. Wang, Double-hydrogel substrate as a model system for three-dimensional cell culture, Methods in Molecular Biology 370 (2007) 203–212.
- [79] F. Rehfeldt, A.E. Brown, M. Raab, S. Cai, A.L. Zajac, A. Zemel, D.E. Discher, Hyaluronic acid matrices show matrix stiffness in 2D and 3D dictates cytoskeletal order and myosin-II phosphorylation within stem cells, Integrative Biology 4 (2012) 422–430.
- [80] R.M. Ezzell, M. Toner, K. Hendricks, J.C. Dunn, R.G. Tompkins, M.L. Yarmush, Effect of collagen gel configuration on the cytoskeleton in cultured rat hepatocytes, Effect of Collagen Gel Configuration on the Cytoskeleton in Cultured Rat Hepatocytes 208 (2) (1993) 442–452.
- [81] K.A. Myers, K.T. Applegate, G. Danuser, R.S. Fischer, C.M. Waterman, Distinct ECM mechanosensing pathways regulate microtubule dynamics to control endothelial cell branching morphogenesis, The Journal of Cell Biology 192 (2) (2011) 321–334.
- [82] J.S. Harunaga, K.M. Yamada, Cell-matrix adhesions in 3D, Matrix Biology 30 (7–8) (2011) 363–368.
- [83] L.A. Smith, X.H. Liu, P.X. Ma, Tissue engineering with nano-fibrous scaffolds, Soft Matter 4 (2008) 2144–2149.
- [84] J. Venugopal, S. Low, A.T. Choon, S. Ramakrishna, Interaction of cells and nanofiber scaffolds in tissue engineering, Journal of Biomedical Materials Research Part B: Applied Biomaterials 84B (1) (2008) 34–48.

- [85] K. Whang, K.E. Healy, D.R. Elenz, E.K. Nam, D.C. Tsai, C.H. Thomas, G.W. Nuber, F.H. Glorieux, R. Travers, S.M. Sprague, Engineering bone regeneration with bioabsorbable scaffolds with novel microarchitecture, Tissue Engineering 5 (1999) 35–51.
- [86] J.H. Brekke, J.M. Toth, Principles of tissue engineering applied to programmable osteogenesis, Journal of Biomedical Materials Research 93 (4) (1998) 380–398.
- [87] V. Mudera, M. Morgan, U. Cheema, S. Nazhat, R. Brown, Ultra-rapid engineered collagen constructs tested in an in vivo nursery site, Journal of Tissue Engineering and Regenerative Medicine 1 (3) (2007) 192–198.
- [88] Y.C. Lin, F.J. Tan, K.G. Marra, S.S. Jan, D.C. Liu, Synthesis and characterization of collagen/ hyaluronan/chitosan composite sponges for potential biomedical applications, Acta Biomaterialia 5 (7) (2009) 2591–2600.
- [89] C.D. Chin, K. Khanna, S.K. Sia, A microfabricated porous collagen-based scaffold as prototype for skin substitutes, Biomedical Microdevices 10 (3) (2008) 459–467.
- [90] D. Annis, A. Bornat, R.O. Edwards, A. Higham, B. Loveday, J. Wilson, An elastomeric vascular prosthesis, Transactions – American Society for Artificial Internal Organs 24 (1978) 209–214.
- [91] J.E. Sanders, S.E. Lamont, S.B. Mitchell, S.G. Malcolm, Small fiber diameter fibro-porous meshes: tissue response sensitivity to fiber spacing, Journal of Biomedical Materials Research Part A 72A (3) (2005) 335–342.
- [92] J.E. Sanders, S.D. Bale, T. Neumann, Tissue response to microfibers of different polymers: polyester, polyethylene, polylactic acid, and polyurethane, Journal of Biomedical Materials Research 62
   (2) (2002) 222–227.
- [93] J.E. Sanders, S.E. Lamont, A. Karchin, S.L. Golledge, B.D. Ratner, Fibro-porous meshes made from polyurethane micro-fibers: effects of surface charge on tissue response, Biomaterials 26 (2005) 813–818.
- [94] S. Hattori, J.D. Andrade, J.B. Hibbs, D.E. Gregonis, R.N. King, Fibroblast cell-proliferation on charged hydroxyethyl methacrylate copolymers, Journal of Colloid and Interface Science 104 (1) (1985) 72–78.
- [95] B. Feng, Z. Jinkang, W. Zhen, L. Jianxi, C. Jiang, L. Jian, M. Guolin, D. Xin, The effect of pore size on tissue ingrowth and neovascularization in porous bioceramics of controlled architecture in vivo, Biomedical Materials 6 (1) (2011).
- [96] M. Rumpler, A. Woesz, J.W. Dunlop, J.T. van Dongen, P. Fratzl, The effect of geometry on threedimensional tissue growth, Journal of the Royal Society Interface 5 (27) (2008) 1173–1180.
- [97] M. Navarro, S. del Valle, S. Martinez, S. Zeppetelli, L. Ambrosio, J.A. Planell, M.P. Ginebra, New macroporous calcium phosphate glass ceramic for guided bone regeneration, Biomaterials 25 (2004) 4233–4241.
- [98] O. Gauthier, J.M. Bouler, E. Aguado, P. Pilet, G. Daculsi, Macroporous biphasic calcium phosphate ceramics: influence of macropore diameter and macroporosity percentage on bone ingrowth, Biomaterials 19 (1–3) (1998) 133–139.
- [99] J.A. Henry, K. Burugapalli, P. Neuenschwander, A. Pandit, Structural variants of biodegradable polyesterurethane in vivo evoke a cellular and angiogenic response that is dictated by architecture, Acta Biomaterialia 5 (1) (2008) 29–42.
- [100] Y. Kuboki, Q. Jin, M. Kikuchi, J. Mamood, H. Takita, Geometry of artificial ECM: sizes of pores controlling phenotype expression in BMP-induced osteogenesis and chondrogenesis, Connective Tissue Research 43 (2–3) (2002) 529–534.
- [101] T.M. Chu, D.G. Orton, S.J. Hollister, S.E. Feinberg, J.W. Halloran, Mechanical and in vivo performance of hydroxyapatite implants with controlled architectures, Biomaterials 23 (5) (2002) 1283–1293.

- [102] F.R. Rose, L.A. Cyster, D.M. Grant, C.A. Scotchford, S.M. Howdle, K.M. Shakesheff, In vitro assessment of cell penetration into porous hydroxyapatite scaffolds with a central aligned channel, Biomaterials 25 (24) (2004) 5507–5514.
- [103] P.S. Eggli, W. Muller, R.K. Schenk, Porous hydroxyapatite and tricalcium phosphate cylinders with two different pore size ranges implanted in the cancellous bone of rabbits. A comparative histomorphometric and histologic study of bony ingrowth and implant substitution, Clinical Orthopaedics Related Research 232 (1988) 127–138.
- [104] M.D. Grynpas, R.M. Pilliar, R.A. Kandel, R. Renlund, M. Filiaggi, M. Dumitriu, Porous calcium polyphosphate scaffolds for bone substitute applications in vivo studies, Biomaterials 23 (2002) 2063–2070.
- [105] V. Karageorgiou, D. Kaplan, Porosity of 3D biomaterial scaffolds and osteogenesis, Biomaterials 26 (27) (2005) 5474-5491.
- [106] B.S. Chang, C.K. Lee, K.S. Hong, H.J. Youn, H.S. Ryu, S.S. Chung, K.W. Park, Osteoconduction at porous hydroxyapatite with various pore configurations, Biomaterials 21 (2005) 1291–1298.
- [107] J.H. Kuhne, R. Bartl, B. Frisch, C. Hammer, V. Jansson, M. Zimmer, Bone formation in coralline hydroxyapatite. Effects of pore size studied in rabbits, Acta Orthopaedics Scandinavica 65 (3) (1994) 246–252.
- [108] M.C. Wake, C.W. Patrick Jr., A.G. Mikos, Pore morphology effects on the fibrovascular tissue growth in porous polymer substrates, Cell Transplantation 3 (4) (1994) 339–343.
- [109] E.K. Yim, R.M. Reano, S.W. Pang, A.F. Yee, C.S. Chen, K.W. Leong, Nanopattern-induced changes in morphology and motility of smooth muscle cells, Biomaterials 26 (2005) 5405–5413.
- [110] J.Y. Lim, H.J. Donahue, Cell sensing and response to micro- and nanostructured surfaces produced by chemical and topographic patterning, Tissue Engineering 13 (8) (2007) 1879–1891.
- [111] J.L. Tan, J. Tien, D.M. Pirone, D.S. Gray, K. Bhadriraju, C.S. Chen, Cells lying on a bed of microneedles: an approach to isolate mechanical force, Proceedings of the National Academy of Sciences of the United States of America 100 (2003) 1484–1489.
- [112] R.D. Sochol, A.T. Higa, R.R.R. Janairo, S. Li, L.W. Lin, Unidirectional mechanical cellular stimuli via micropost array gradients, Soft Matter 7 (2011) 4606–4609.
- [113] S.N. Weng, J.P. Fu, Synergistic regulation of cell function by matrix rigidity and adhesive pattern, Biomaterials 32 (2011) 9584–9593.
- [114] E.J. Kim, C.A. Boehm, A. Mata, A.J. Fleischman, G.F. Muschler, S. Roy, Post microtextures accelerate cell proliferation and osteogenesis, Acta Biomaterialia 6 (2010) 160–169.
- [115] A. Mata, C. Boehm, A.J. Fleischman, G.F. Muschler, S. Roy, Connective tissue progenitor cell growth characteristics on textured substrates, International Journal of Nanomedicine 2 (2007) 389–406.
- [116] A. Mata, C. Boehm, A.J. Fleischman, G. Muschler, S. Roy, Analysis of connective tissue progenitor cell behavior on polydimethylsiloxane smooth and channel micro-textures, Biomedical Microdevices 4 (2002) 267–275.
- [117] A. Wilkinson, R.N. Hewitt, L.E. McNamara, D. McCloy, R.M.D. Meek, M.J. Dalby, Biomimetic microtopography to enhance osteogenesis in vitro, Acta Biomaterialia 7 (2011) 2919–2925.
- [118] H. Sahana, D.K. Khajuria, R. Razdan, D.R. Mahapatra, M.R. Bhat, S. Suresh, R.R. Rao, L. Mariappan, Improvement in bone properties by using risedronate adsorbed hydroxyapatite novel nanoparticle based formulation in a rat model of osteoporosis, Journal of Biomedical Nanotechnology 9 (2013) 193–201.
- [119] M.R. Steedman, S.L. Tao, H. Klassen, T.A. Desai, Enhanced differentiation of retinal progenitor cells using microfabricated topographical cues, Biomedical Microdevices 12 (2010) 363–369.

- [120] M. Nikkhah, J.S. Strobl, R. De Vita, M. Agah, The cytoskeletal organization of breast carcinoma and fibroblast cells inside three dimensional (3-D) isotropic silicon microstructures, Biomaterials 31 (2010) 4552–4561.
- [121] A. Khademhosseini, R. Langer, J. Borenstein, J.P. Vacanti, Microscale technologies for tissue engineering and biology, Proceedings of the National Academy of Sciences of the United States of America 103 (2006) 2480–2487.
- [122] H. Yukawa, H. Noguchi, S. Hayashi, Embryonic body formation using the tapered soft stencil for cluster culture device, Biomaterials 32 (2011) 3729–3738.
- [123] A.F. Adler, A.T. Speidel, N. Christoforou, K. Kolind, M. Foss, K.W. Leong, High-throughput screening of microscale pitted substrate topographies for enhanced nonviral transfection efficiency in primary human fibroblasts, Biomaterials 32 (2011) 3611–3619.
- [124] J.Y. Mai, C. Sun, S. Li, X. Zhang, A microfabricated platform probing cytoskeleton dynamics using multidirectional topographical cues, Biomedical Microdevices 9 (2007) 523–531.
- [125] B.M. Baker, R.L. Mauck, The effect of nanofiber alignment on the maturation of engineered meniscus constructs, Biomaterials 28 (2007) 1967–1977.
- [126] R. Murugan, S. Ramakrishna, Design strategies of tissue engineering scaffolds with controlled fiber orientation, Tissue Engineering 13 (2007) 1845–1866.
- [127] D.R. Nisbet, J.S. Forsythe, W. Shen, D.I. Finkelstein, M.K. Horne, Review paper: a review of the cellular response on electrospun nanofibers for tissue engineering, Journal of Biomaterials Application 24 (2009) 7–29.
- [128] J. Yang, M. Yamato, T. Shimizu, H. Sekine, K. Ohashi, M. Kanzaki, T. Ohki, K. Nishida, T. Okano, Reconstruction of functional tissues with cell sheet engineering, Biomaterials 28 (2007) 5033-5043.
- [129] R.A. Brown, M. Wiseman, C.B. Chuo, U. Cheema, S.N. Nazhat, Ultrarapid engineering of biomimetic materials and tissues: fabrication of nano- and microstructures by plastic compression, Advanced Functional Materials 15 (2005) 1762–1770.
- [130] G.A. Georgeu, E.T. Walbeehm, R. Tillett, A. Afoke, R.A. Brown, J.B. Phillips, Investigating the mechanical shear-plane between core and sheath elements of peripheral nerves, Cell and Tissue Research 320 (2005) 229–234.
- [131] J.B. Phillips, X. Smit, N. De Zoysa, A. Afoke, R.A. Brown, Peripheral nerves in the rat exhibit localized heterogeneity of tensile properties during limb movement, The Journal of Physiology 557 (2004) 879–887.
- [132] E.T. Walbeehm, A. Afoke, T. de Wit, F. Holman, S.E. Hovius, R.A. Brown, Mechanical functioning of peripheral nerves: linkage with the "mushrooming" effect, Cell and Tissue Research 316 (2004) 115–121.
- [133] T. Shimizu, H. Sekine, J. Yang, Y. Isoi, M. Yamato, A. Kikuchi, E. Kobayashi, T. Okano, Polysurgery of cell sheet grafts overcomes diffusion limits to produce thick, vascularized myocardial tissues, FASEB Journal 20 (2006) 708–710.
- [134] R.G. Flemming, C.J. Murphy, G.A. Abrams, S.L. Goodman, P.F. Nealey, Effects of synthetic microand nano-structured surfaces on cell behavior, Biomaterials 20 (1999) 573–588.
- [135] D.E. Discher, D.J. Mooney, P.W. Zandstra, Growth factors, matrices, and forces combine and control stem cells, Science 324 (2009) 1673–1677.
- [136] I.K. Kwon, S. Kidoaki, T. Matsuda, Electrospun nano-to microfiber fabrics made of biodegradable copolyesters: structural characteristics, mechanical properties and cell adhesion potential, Biomaterials 26 (2005), 3929e3939.

# 12

## Tissue Engineering: The Use of Stem Cells in Regenerative Medicine

F.M.P. Tonelli<sup>1,2,\*</sup>, N. de Cássia Oliveira Paiva<sup>1</sup>, R.V.B. de Medeiros<sup>1</sup>, M.C.X. Pinto<sup>1,2,3</sup>, F.C.P. Tonelli<sup>1</sup>, R.R. Resende<sup>1,2</sup>

<sup>1</sup>FEDERAL UNIVERSITY OF MINAS GERAIS, BELO HORIZONTE, BRAZIL; <sup>2</sup>INSTITUTONANOCELL, DIVINÓPOLIS, BRAZIL; <sup>3</sup>FEDERAL UNIVERSITY OF GOIÁS, GOIÂNIA, BRAZIL

## 12.1 Introduction

### 12.1.1 Regenerative Medicine and Tissue Engineering

Regenerative medicine is a multidisciplinary field that includes tissue engineering. Its main objective is to propose innovative ways to keep body tissues and organs working properly, promoting health, especially when some kind of impairment occurs in them [1]. An example is tissue loss after an accident, which requires assistance to achieve tissue regeneration to regain normal function (Fig. 12.1). Tissue engineering offers an alternative by creating structures that mimic normal tissue. Tissue engineering not only regenerates injured or diseased tissue but contributes to factors that can dictate cell fate and searches for new diagnotic technology [2].

SCs are important tools to provide substitute tissue. That is why modern tissue engineering history merges with the history of SCs. Although we can consider tissue engineering history to have begun in the 17th century (when grafts were used to repair bones in Amsterdam) [3] and to have evolved until hemopoietic cell transplantation became successful in the late 1950s [4], the foundation of modern tissue engineering was established only after 1998, when SC lines became available [5]. From that moment, evolution in the field of regenerative medicine became increasingly fast and effective. To develop a tissue engineering strategy, it is essential to pay attention to four important components [6,7]:

**1.** Cells: they must be able to reestablish the tissue's function. For instance, if bone tissue and SCs will be used, they must be able to differentiate into bone cells.

<sup>\*</sup>Corresponding Author.



FIGURE 12.1 Stem cells (SCs) and tissue engineering. Stem cells can be used to promote tissue regeneration. ASC, adult SCs; ESC, embryonic SCs; iPSC, induced pluripotent SCs.

- **2.** Culture medium: it should contain all necessary cytokines and growth factors to support cell differentiation into desirable cells, to proliferate and grow, and also to keep the desirable cells alive, performing their expected activities in that kind of tissue.
- **3.** The porous matrix: it is necessary for the porous matrix to mimic the tissue extracellular matrix, in which cells can infiltrate to regenerate the injured tissue. It must have desirable physical and chemical properties and be made of biocompatible, bioreabsorbable, and biodegradable materials, all of which support cell growth.
- **4.** The bioreactor: the other three components are placed into this one, which should offer ideal conditions for cell multiplication and correct displacement into the scaffold, by mimicking physiologic conditions.

In this chapter we will focus attention on briefly exploring SC types that can be used in tissue engineering to regenerate tissue.

### 12.1.2 Stem Cells

SCs are unspecialized cells that can differentiate into various cell types in addition to having the ability to proliferate to self-renew, maintaining the SCs population [8]. According to their source, SCs can be classified as embryonic SCs (ESCs) or adult SCs (ASCs). ESCs are not used as much owing to ethical concerns regarding their obtainment. An important technique was also developed to convert normal cells into SCs to generate induced pluripotent SCs (iPSCs).

#### 12.1.2.1 Embryonic Stem Cells

During the vertebrate's embryo development there is a stage called blastocyst that contains a region called inner cell mass (ICM). The ICM is the source of ESCs that are pluripotent [9]. They were first obtained in 1981 from mice [9], and only 17 years later from humans [10].

ESCs can differentiate into cells from ectoderm, endoderm, or mesoderm because they are pluripotent. Thus if the intention is to regenerate cardiac tissue, ESCs can be induced to differentiate into cardiomyocytes [11]. Nevertheless if the stimuli that trigger differentiation are not present, ESCs also have the ability to maintain themselves undifferentiated for years [12].

ESCs can originate from different cell types under specific stimulations. The same sample of human ESC can be stimulated to differentiate into neural tissue, cartilage, liver, and blood vessels. The condition that dictates the different fates of cells can be the different three-dimensional structure of scaffolds [13].

In some countries the use of ESCs in research is prohibited because there is an ethical impasse arising from religious concerns that using these cells in research means killing humans. This idea comes from the fact that to obtain ESCs it is necessary to fertilize a human oocyte with human sperm, cultivated it until it reaches the blastocyst stage, and then human development needs to be interrupted to obtain ESC from the ICM (Fig. 12.2).



**FIGURE 12.2** Embryonic stem call (ESC) obtainment. The ESC can be obtained from the inner cell mass (ICM); however, inducing fertilization to achieve the blastocyst stage and obtain ESCs means interrupting the natural development of the fertilized oocyte into a new human.

#### 12.1.2.2 Adult Stem Cells

ASCs are an alternative to using ESCs in tissue engineering because their isolation does not require the destruction of an embryo. ASCs can be found in differentiated tissues present in the adult body. Their main role is to maintain tissues' condition in the case of injury or damage caused by sickness, promoting tissue repair [14].

ASCs have important characteristics that make them suitable for regenerative medicine. They are easy to cultivate in vitro and possess a high rate of proliferation. However, ASCs and ESCs are not the same. For example, ASCs are multipotent and not pluripotent as are ESCs. That limits the application of ASCs because the differentiation of these cells is already committed to specific cell lines. For example, hematopoietic SC (the first ASC to be studied [15]) can be found in bone marrow, umbilical cord, and placenta blood, and originates in all blood cells but not neurons.

Another important ASC is multipotent SCs derived from adipose tissue [16], which can be isolated from lipoaspirate obtained from plastic surgery.

ASCs have been used successfully to treat leukemia [17] and hepatic cirrhosis [18], for example, and to repair cartilage defects [19,20]. Studies continue to be published in this field; to cite an example, the potential of resident adult renal stem/progenitor cells is under investigation with the aim to treat acute renal injury [21].

#### 12.1.2.3 Induced Pluripotent Stem Cells

iPSCs can also be obtained from adults, but this requires reprogramming of a common differentiated cell to an ESC-like state. These cells were first obtained from mouse somatic cells in the beginning of the 21st century [22]. iPSCs have a molecular signature and morphology that resembles ESCs, and as with ESCs, they can be directed to differentiate into a cell type of interest.

Reprogramming can be achieved, for example, by providing the sequences encoding the four protein factors to the cell: the products of genes *OCT4*, *SOX2*, *NANOG*, and *LIN28* [23]. The sequences can be delivered using adenovirus [24] and bacterial artificial chromosomes [25] or the recombinant proteins can be provided to the cells [26].

New protocols to generate iPSCs continue to be developed [27,28], always with the intention of developing an easy, inexpensive, and efficient methodology. Once iPSCs are obtained, it is possible, as with ESCs, to induce their differentiation into a desirable cell type. De Peppo and coworkers, for example, used iPSCs to achieve bone regeneration [29].

## 12.2 Stem Cell Isolation and Differentiation

#### 12.2.1 Isolation of Adipose-Derived Stem Cells

The isolation of adipose-derived SCs can be performed using as a start material adipose tissue removed during plastic surgery by liposuction aspiration. This protocol was used by Bunnell and coworkers [30] and it should be performed a maximum of 24 h after obtaining the adipose tissue.

First, wash the tissue sample extensively with phosphate-buffered saline (PBS) containing 5% penicillin–streptomycin to remove the debris. Place the sample on a sterile tissue culture plate with 0.075% collagenase type I in PBS containing 2% penicillin– streptomycin aiming for tissue digestion. Pipette the sample up and down several times to facilitate the process. Incubate the sample for 30 min at  $37^{\circ}$ C, 5% CO<sub>2</sub>.

Neutralize collagenase type I activity by adding 5 mL  $\alpha$ -minimal essential medium (MEM) containing 20% heat-inactivated fetal bovine serum (FBS) (Atlanta Biological, Atlanta, GA). Pipette the sample up and down to disintegrate aggregates of the adipose tissue. Transfer the sample to a 50-mL tube and centrifuge the sample at 2000 rpm for 5 min, avoiding solid aggregates.

Take the samples out of the centrifuge and shake it vigorously to disrupt the pellet thoroughly and mix the cells. Repeat the centrifugation step. Aspirate all of the collagenase solution above the pellet without disturbing the cells. Resuspend the pellet in 1 mL lysis buffer and incubate it for 10 min on ice. Wash with 20 mL PBS/2% penicillin– streptomycin and centrifuge it at 2000 rpm for 5 min. Remove the supernatant and resuspend the cell pellet in 3 mL stromal medium (alpha-MEM; Mediatech, Herndon, VA) supplemented with 20% FBS, 1% L-glutamine (Mediatech), and 1% penicillin– streptomycin (Mediatech). Filter the cell suspension through a 70- $\mu$ m cell strainer. Wash the cell strainer with an additional 2 mL stromal medium. Plate the sample containing the cells in a lysine-coated culture plate and incubate it at 37°C, 5% CO<sub>2</sub>.

## 12.2.2 Osteogenic Differentiation

The following protocol for in vitro osteogenic differentiation was developed by Pourebrahim and coworkers [31]. It is recommended that the adipose-derived SCs be in passage 3 culture for use.

Seed  $5 \times 10^6$  ASCs in  $3 \text{ mm} \times 3 \text{ mm} \times 3 \text{ mm}$  hydroxyapatite (HA)/ $\beta$ -tricalcium phosphate ( $\beta$ -TCP) scaffold (Ceraform, Teknimed, France): 60% HA and 40%  $\beta$ -TCP with a mean pore size of 200–800 mm. The medium should be an osteogenic one, containing 50 $\mu$ g/mL ascorbic acid 2-phosphate (Sigma, USA), 10 mM  $\beta$ -glycerophosphate (Sigma, USA), and 0.1 mM dexamethasone (Sigma, USA). Incubate the seeded scaffold in osteogenic medium at 37°C and 5% CO<sub>2</sub> for 21 days.

## 12.3 Cells Maintenance and Characterization

## 12.3.1 Adipose-Derived Stem Cells

Adipose-derived SCs isolated from adipose tissue can be identified by some characteristics such as adherence to cell culture plates after isolation, the ability of the cell to differentiate into various lineages, and the expression of specific markers [32]. The last is the least laborious one and it uses a flow cytometer.

Cells can be stained simultaneously with phycoerythrin conjugated monoclonal antibody to CD44 (ab58754; ABCAM Antibodies, Cambridge Science Park, UK), and

fluorescein isothiocyanate-conjugated monoclonal antibody to CD90 (ab22541; ABCAM Antibodies) before analyses through flow cytometry [31].

Adipose-derived SCs also express a high level of markers CD49d, CD105, CD13, and CD7 and can maintain this high expression for at least 3 months and seven passages of in vitro culture [33].

The capacity to differentiate into various lineages requires more time and effort. Adipose-derived SCs can be stimulated to differentiate, for example, into adipogenic cell types (culture medium supplemented with 0.5 mM isobutylmethylxanthine, 50  $\mu$ M indomethacin, and 0.5  $\mu$ M dexamethasone with a change of medium every 3 days until mature adipocytes are obtained), osteogenic cell types (culture medium supplemented with 1 nM dexamethasone, 2 mM  $\beta$ -glycerolphosphate, and 50  $\mu$ M ascorbate-2-phosphate, with a replacement of medium every 2–3 days), chondrogenic cell types [high-glucose Dulbecco's MEM supplemented with 500 ng/mL bone morphogenic protein-6, 10 ng/mL transforming growth factor- $\beta$ 3, 10<sup>-7</sup> M dexamethasone, 50  $\mu$ g/mL ascorbate 2-phosphate, 40  $\mu$ g/mL proline, 100  $\mu$ g/mL pyruvate, and 50 mg/mL insulin, human transferrin, and selenous acid and premix (Becton Dickinson: 6.25  $\mu$ g/mL insulin, 5.35 mg/mL linoleic acid, with a replacement of medium every 2–3 days)], and neural cell types [culturing in neurobasal medium supplemented with B27 (1:50), 20 ng/mL basic fibroblastic growth factor, and 20 ng/mL epidermal growth factor for 4–7 days] [30].

#### 12.3.2 Generated Bone Cells

Bone cells generated inside the scaffold can have characterization performed in vitro and function accessed in vivo. Among analyses that can be performed to characterize bone cells from in vitro or in vivo samples are the determination of osteopontin concentration, alkaline phosphatase activity, calcium content, gene expression profile, and histologic examination (with or without antibodies).

#### 12.3.2.1 Determination of Osteopontin Concentration, Calcium Content, and Alkaline Phosphatase Activity

Osteopontin is a protein with important roles in bone biomineralization [34] and remodeling [35]. Its concentration can be determined by using the culture medium present in in vitro cell culture through the human osteopontin enzyme-linked immunosorbent assay kit [36].

Cells in culture or from in vivo samples can be used to determine the calcium content; trichloroacetic acid in water [0.15 mL of 5% (vol/vol)] can be added to them for 30 min under repeated pipetting. Then the supernatants can be submitted to calcium (CPC) Liquicolor assay [36].

Alkaline phosphatase is an enzyme largely present and active in bone tissue, and its increased activity is observed during bone formation. The seeded scaffold can be exposed, on ice, to lysis buffer [PBS, 1% (vol/vol) Triton X-100, 0.5% (wt/vol) sodium

deoxycholate, 0.1% (wt/vol) sodium dodecyl sulfate, 0.1 mg/mL phenylmethylsulfonylfluoride, and 0.3% (vol/vol) aprotinin] until its disintegration. After centrifugation, the supernatant can be incubated at 37°C until the development of yellow color, in alkaline buffer and nitrophenyl-phosphate substrate solution. At this stage the reaction is stopped with 0.5 M NaOH. Enzyme activity can be accessed by absorbance reading at 405 nm and comparison with a standard curve may be obtained from *p*-nitrophenol solutions of known concentrations [36].

#### 12.3.2.2 Histologic Analyses (With or Without Antibodies)

Histologic analyses to evaluate bone regeneration can be performed using biopsies of bone obtained from animal submitted to in vivo seeded scaffold transplantation. The samples can be fixed in 10% buffered formalin and decalcified in formic acid and sodium citrate. After they are washed, submitted to dehydration, cleared, included in paraffin, and processed, they can be stained using Masson trichrome. Histomorphometric analysis can be performed with measurements of the newly formed bone through automated image analysis software (IHMMA, Version 1, Sbmu, Iran) [31].

Inclusion in paraffin is also useful for immunohistochemical analyses, and methylmethacrylate inclusion may be used for other histologic analyses.

Staining with hematoxylin—eosin and Masson trichrome procedures, it is possible to observe collagen generation (positive blue areas); staining the samples with Goldner, osteoid formation can be accessed (positive red areas) [36].

In histologic analyses using antibodies, it is possible to detect the presence of, for example, osteopontin, bone sialoprotein, and osteocalcin.

#### 12.3.2.3 Gene Expression Analyses

The analysis can be performed after RNA extraction and synthesis of complementary DNA strands. The hybrid DNA–RNA can be submitted to real-time polymerase chain reaction to quantify transcript levels of alkaline phosphatase, osteocalcin, collagen I, and osteopontin, for example, in relation to a housekeeping gene of preference such as glyceraldehyde 3-phosphate dehydrogenase [29,31].

RNA can also be used to generate a double-stranded labeled template (start material for microarray analyses) with the aim to access the expression rate of genes involved in osteogenesis [29].

## 12.4 Conclusions and Perspectives

SCs possess a large array of clinical applications owing to their large capacity to differentiate into different cell types according to the scientific challenge. Regenerative medicine has developed with SC technologies and made new strategies possible to surpass the barrier of immunologic rejection in transplants from a donor that is not the receptor. It is already possible to use SCs and it a common procedure for a person to transform this cell in vitro to a cell of interest inside a scaffold and to use the result to transplant it back to that person as a new tissue derived from his or her own cells [37]. Adipose-derived SCs, for example, are being used to repair defects in soft tissues [38,39] and to generate bone tissue as mentioned in the protocol section [31].

New scaffolds are being developed from biomaterials [40] and commercial options are available. It is also possible to use as scaffold decellularized extracellular matrix [41] originated from an intact organ that undergoes several washings, leaving only the extracellular matrix. Growing organs in vitro and ex vivo is now also a reality [42]. However, there are still some limitations to developing blood capillaries in the tissue graft (especially in the inside part of the engineered tissues), and the clinical applications of laboratory-made organs remain a distant prospect.

Another application for SCs lies in the diagnostic field, because some conditions result in an increase in SCs in bloodstream, which can be used as a biomarker. For example, after myocardial infarction, patients have circulating SCs in the peripheral blood; thus these cells may be an important marker for risk evaluation in patients with cardiac disease [43]. Patients affected by osteoporosis possess mesenchymal SCs circulating as a consequence of the impairment of osteoblast differentiation [44]. Despite all of the progress made in the field of SC, further research is required before they are safe enough to be applied largely to strategies in humans.

## References

- J.F. Stoltz, N. de Isla, Y.P. Li, D. Bensoussan, L. Zhang, C. Huselstein, Y. Chen, V. Decot, J. Magdalou, N. Li, L. Reppel, Y. He, Stem cells and regenerative medicine: Myth or reality of the 21th century, Stem Cells International 2015 (2015) 734731–734738.
- [2] R. Langer, J.P. Vacanti, Tissue engineering, Science 260 (1993) 920-926.
- [3] R.J. Kroeze, M.N. Helder, L.E. Govaert, T.H. Smit, Biodegradable polymers in bone tissue engineering, Materials 2 (2009) 833–856.
- [4] E.D. Thomas, H.L. Lochte, J.H. Cannon, O.D. Sahler, J.W. Ferrebee, Supralethal whole body irradiation and isologous marrow transplantation in man, Journal of Clinical Investigation 38 (1959) 1709–1716.
- [5] M. Amit, M.K. Carpenter, M.S. Inokuma, C.P. Chiu, C.P. Harris, M.A. Waknitz, J. Itskovitz-Eldor, J.A. Thomson, Clonally derived human embryonic stem cell lines maintain pluripotency and proliferative potential for prolonged periods of culture, Developmental Biology 227 (2) (2000) 271–278.
- [6] B.P. Chan, K.W. Leong, Scaffolding in tissue engineering: general approaches and tissue-specific considerations, European Spine Journal 17 (4) (2008) 467–479.
- [7] F.M.P. Tonelli, A.K. Santos, K.N. Gomes, E. Lorencon, S. Guatimosim, L.O. Ladeira, R.R. Resende, Carbon nanotube interaction with extracellular matrix proteins producing scaffolds for tissue engineering, International Journal of Nanomedicine 7 (2012) 4511–4529.
- [8] F.M.P. Tonelli, A.K. Santos, S.L. Silva, K.N. Gomes, L.O. Ladeira, R.R. Resende, Stem cells and calcium signaling, in: M.S. Islam (Ed.), Calcium Signaling (Advances in Experimental Medicine and Biology), vol. 740, Springer, Dordrecht, 2012.
- [9] M.J. Evans, M. Kaufman, Establishment in culture of pluripotential cells from mouse embryos, Nature 292 (1981) 154–156.

- [10] J.A. Thomson, J. Itskovitz-Eldor, S.S. Shapiro, M.A. Waknitz, J.J. Swiergiel, V.S. Marshall, J.M. Jones, Embryonic stem cell lines derived from human blastocysts, Science 282 (1998) 1145–1147.
- [11] J.Q. He, Y. Ma, Y. Lee, J.A. Thomson, T.J. Kamp, Human embryonic stem cells develop into multiple types of cardiac myocytes: action potential characterization, Circulation Research 93 (2003) 32–39.
- [12] L.M. Hoffman, M.K. Carpenter, Human embryonic stem cell stability, Stem Cell Reviews 1 (2005) 139–144.
- [13] S. Levenberg, N.F. Huang, E. Lavik, A.B. Rogers, J. Itskovitz-Eldor, R. Langer, Differentiation of human embryonic stem cells on three-dimensional polymer scaffolds, Proceedings of the National Academy of Sciences of the United States of America 100 (2003) 12741–12746.
- [14] R.S. Tuan, G. Boland, R. Tuli, Adult mesenchymal stem cells and cell-based tissue engineering, Arthritis Research & Therapy 5 (2003) 32–45.
- [15] J.E. Till, C.E. Mc, A direct measurement of the radiation sensitivity of normal mouse bone marrow cells, Radiation Research 14 (1961) 213–222.
- [16] J.M. Gimble, A.J. Katz, B.A. Bunnell, Adipose-derived stem cells for regenerative medicine, Circulation Research 100 (9) (2007) 1249–1260.
- [17] C. Karanes, G.O. Nelson, P. Chitphakdithai, E. Agura, K.K. Ballen, C.D. Bolan, D.L. Porter, J.P. Uberti, R.J. King, D.L. Confer, Twenty years of unrelated donor hematopoietic cell transplantation for adult recipients facilitated by the National Marrow Donor Program, Biology of Blood and Marrow Transplantation 14 (2008) 8–15.
- [18] S. Terai, T. Ishikawa, K. Omori, K. Aoyama, Y. Marumoto, Y. Urata, Y. Yokoyama, K. Uchida, T. Yamasaki, Y. Fujii, K. Okita, I. Sakaida, Improved liver function in patients with liver cirrhosis after autologous bone marrow cell infusion therapy, Stem Cells 24 (2006) 2292–2298.
- [19] S. Wakitani, M. Nawata, K. Tensho, T. Okabe, H. Machida, H. Ohgushi, Repair of articular cartilage defects in the patello-femoral joint with autologous bone marrow mesenchymal cell transplantation: three case reports involving nine defects in five knees, Journal of Tissue Engineering and Regenerative Medicine 1 (2007) 74–79.
- [20] P. Macchiarini, P. Jungebluth, T. Go, M.A. Asnaghi, L.E. Rees, T.A. Cogan, A. Dodson, J. Martorell, S. Bellini, P.P. Parnigotto, S.C. Dickinson, A.P. Hollander, S. Mantero, M.T. Conconi, M.A. Birchall, Clinical transplantation of a tissue-engineered airway, Lancet 372 (2008) 2023–2030.
- [21] F. Sallustio, G. Serino, F.P. Schena, Potential reparative role of resident adult renal stem/progenitor cells in acute kidney injury, BioResearch Open Access 4 (1) (2015) 326–333.
- [22] K. Takahashi, S. Yamanaka, Induction of pluripotent stem cells from mouse embryonic and adult fibroblast cultures by defined factors, Cell 126 (2006) 663–676.
- [23] J. Yu, M.A. Vodyanik, K. Smuga-Otto, J. Antosiewicz-Bourget, J.L. Frane, S. Tian, J. Nie, G.A. Jonsdottir, V. Ruotti, R. Stewart, I.I. Slukvin, J.A. Thomson, Induced pluripotent stem cell lines derived from human somatic cells, Science 318 (2007) 1917–1920.
- [24] M. Stadtfeld, N. Maherali, D.T. Breault, K. Hochedlinger, Defining molecular cornerstones during fibroblast to iPS cell reprogramming in mouse, Cell Stem Cell 2 (2008) 230–240.
- [25] K. Woltjen, I.P. Michael, P. Mohseni, R. Desai, M. Mileikovsky, R. Hamalainen, R. Cowling, W. Wang, P. Liu, M. Gertsenstein, K. Kaji, H.K. Sung, A. Nagy, *piggyBac* transposition reprograms fibroblasts to induced pluripotent stem cells, Nature 458 (2009) 766–770.
- [26] D. Kim, C.H. Kim, J.I. Moon, Y.G. Chung, M.Y. Chang, B.S. Han, S. Ko, E. Yang, K.Y. Cha, R. Lanza, K. S. Kim, Generation of human induced pluripotent stem cells by direct delivery of reprogramming proteins, Cell Stem Cell 4 (6) (2009) 472–476.
- [27] B.K. Chou, P. Mali, X. Huang, Z. Ye, S.N. Dowey, L.M.S. Resar, C. Zou, Y.A. Zhang, J. Tong, L. Cheng, Efficient human iPS cell derivation by a non-integrating plasmid from blood cells with unique epigenetic and gene expression signatures, Cell Research 21 (3) (2011) 518–529.

- [28] G.L. Boulting, E. Kiskinis, G.F. Croft, M.W. Amoroso, D.H. Oakley, B.J. Wainger, D.J. Williams, D.J. Kahler, M. Yamaki, L. Davidow, C.T. Rodolfa, J.T. Dimos, S. Mikkilineni, A.B. MacDermott, C.J. Woolf, C.E. Henderson, H. Wichterle, K. Eggan, A functionally characterized test set of human induced pluripotent stem cells, Nature Biotechnology 29 (2011) 279–286.
- [29] G.M. de Peppo, I. Marcos-Campos, D.J. Kahler, D. Alsalman, L. Shang, G. Vunjak-Novakovic, D. Marolt, Engineering bone tissue substitutes from human induced pluripotent stem cells, Proceedings of the National Academy of Sciences of the United States of America 110 (21) (2013) 8680–8685.
- [30] B.A. Bunnell, M. Flaat, C. Gagliardi, B. Patel, C. Ripoll, Adipose-derived stem cells: isolation, expansion and differentiation, Methods 45 (2) (2008) 115–120.
- [31] N. Pourebrahim, B. Hashemibeni, S. Shahnaseri, N. Torabinia, B. Mousavi, S. Adibi, F. Heidari, M.J. Alavi, A comparison of tissue-engineered bone from adipose-derived stem cell with autogenous bone repair in maxillary alveolar cleft model in dogs, International Journal of Oral and Maxillofacial Surgery 42 (2013) 562–568.
- [32] M. Dominici, K. Le Blanc, I. Mueller, I. Slaper-Cortenbach, F. Marini, D. Krause, Minimal criteria for defining multipotent mesenchymal stromal cells. The International Society for Cellular Therapy position statement, Cytotherapy 8 (2006) 315–317.
- [33] V. Folgiero, E. Migliano, M. Tedesco, S. Iacovelli, G. Bon, M.L. Torre, A. Sacchi, M. Marazzi, S. Bucher, R. Falcioni, Purification and characterization of adipose-derived stem cells from patients with lipoaspirate transplant, Cell Transplantation 19 (10) (2010) 1225–1235.
- [34] M.D. McKee, W.N. Addison, M.T. Kaartinen, Hierarchies of extracellular matrix and mineral organization in bone of the craniofacial complex and skeleton, Cells Tissues Organs 181 (3–4) (2005) 176–188.
- [35] S.T. Choi, J.H. Kim, E.J. Kang, S.W. Lee, M.C. Park, Y.B. Park, S.K. Lee, Osteopontin might be involved in bone remodelling rather than in inflammation in ankylosing spondylitis, Rheumatology (Oxford) 47 (12) (2008) 1775–1779.
- [36] D. Marolt, I.M. Campos, S. Bhumiratana, A. Koren, P. Petridis, G. Zhang, P.F. Spitalnik, W.L. Grayson, G. Vunjak-Novakovic, Engineering bone tissue from human embryonic stem cells, Proceedings of the National Academy of Sciences of the United States of America 109 (22) (2012) 8705–8709.
- [37] J.W. Cassidy, Nanotechnology in the regeneration of complex tissues, Bone and Tissue Regeneration Insights 5 (2014) 25–35.
- [38] E.H. Kim, C.Y. Heo, Current applications of adipose-derived stem cells and their future perspectives, World Journal of Stem Cells 6 (2014) 65–68.
- [39] Y.J. Kim, J.H. Jeong, Clinical application of adipose stem cells in plastic surgery, Journal of Korean Medical Science 29 (2014) 462–467.
- [40] J. Cutts, M. Nikkhah, D.A. Brafman, Biomaterial approaches for stem cell-based myocardial tissue engineering, Biomarker Insights 10 (2015) 77–90.
- [41] P.A. Santi, S.B. Johnson, Decellularized ear tissues as scaffolds for stem cell differentiation, Journal of the Association for Research in Otolaryngology 14 (2013) 3–15.
- [42] M.E. Scarritt, N.C. Pashos, B.A. Bunnell, A review of cellularization strategies for tissue engineering of whole organs, Frontiers in Bioengineering and Biotechnology 3 (2015) 43–51.
- [43] E. Paczkowska, M. Kucia, D. Koziarska, M. Halasa, K. Safranow, M. Masiuk, A. Karbicka, M. Nowik, P. Nowacki, M.Z. Ratajczak, B. Machalinski, Clinical evidence that very small embryonic-like stem cells are mobilized into peripheral blood in patients after stroke, Stroke 40 (4) (2009) 1237–1244.
- [44] L. Dalle Carbonare, M.T. Valenti, M. Zanatta, L. Donatelli, V. Lo Cascio, Circulating mesenchymal stem cells with abnormal osteogenic differentiation in patients with osteoporosis, Arthritis & Rheumatology 60 (11) (2009) 3356–3365.

# 13

## Tissue Engineering and Regenerative Medicine Solutions for the Abdominal Organs

R. Katari, K. McNamara, C. Gentile, L. Edgar, T. Callese, D. Igel, J.P. Zambon, R. Tamburrini, G. Orlando\*

WAKE FOREST SCHOOL OF MEDICINE, WINSTON SALEM, NC, UNITED STATES

## 13.1 Introduction

In recent decades, organ transplantation technology and techniques have progressed substantially, resulting in its emergence as the reference standard treatment for all diseases leading to irreversible organ failure [1]. Over 100,000 solid organ transplants are performed every year worldwide including approximately 70,000 kidneys and 20,000 livers [2]. Although success rates have improve dramatically over the years, the challenges of donor organ supply and complications related to immunosuppression remain chief obstacles in the way of further progress for the treatment of irreversible end organ damage. End-stage renal disease, for example, represents an extraordinary public health burden worldwide, and has contributed to the kidney's being the most commonly transplanted organ. In 2014, however, about 15,700 people received kidney transplants while over 100,000 candidates remained on the waiting list at the end of the year owing to the limited supply [3,4]. Those candidates who do receive organ transplantation can develop secondary neoplasms or other adverse sequelae associated with prolonged immunosuppression.

Developing methods in tissue engineering and regenerative medicine (TERM) have shown great promise for addressing these two challenges hampering contemporary transplantation. Investigators within TERM and its subdisciplines seek to restore function by facilitating the transition from a diseased state to a healthy one in tissues that might not perform this task spontaneously or sufficiently [5]. This is commonly accomplished in experimental and clinical settings via the therapeutic administration of cellular material, naturally derived or synthetic biomaterials, or a combination of the two [6]. One way in which biomaterials are often employed are as geospatially appropriate

<sup>&</sup>lt;sup>\*</sup>Corresponding Author.

"scaffolds" upon which to seed cells or to allow the migration and proliferation of cells in vivo as a way to replace damaged tissue.

Significant interest has been growing in the cell-scaffold method, particularly for its use in abdominal organs, many of which are solid, complex, and modular. Given the dramatic success of surgical transplantation for the treatment of diseases afflicting the abdominal organs, researchers have sought to employ TERM methodologies to bioengineer functional organs or organ-shaped scaffolds ex vivo for subsequent implantation into the patient. Indeed, it has been recognized that transplantation historically shares many of the same guiding principles with the more recently developed TERM disciplines [7]. For example, the idea of replacing diseased tissue by the direct administration of bioactive cellular material is certainly applicable to both. By seeding the appropriate cellular material onto a biocompatible organ-shaped scaffold, an investigator can bioengineer a functional organ for use as a replacement [8].

A new technology under investigation and experimental characterization employs whole-organ decellularization to generate a native three-dimensional extracellular matrix (ECM) scaffold that preserves native tissue architecture, including intricate vascular networks and structural components [9]. It has long been understood that cells require supporting structures and cues to grow, expand, and exert function [10]. In fact, biomaterial scaffolds are employed for this very purpose. Nevertheless, ECM, the defining feature of connective tissue, constitutes the supporting structure in vivo and is specialized to carry out this function, making it the optimal choice. Its interaction and cooperation with adjacent cells is absolutely necessary for physiologic functioning across the board [11]. Tissue decellularization, achieved via perfusion of detergent through an organ's vasculature, leaves behind ECM scaffolds that can be used as the ideal biomaterial structure destined for recellularization [12]. With regard to the state of the art of abdominal organ engineering, as will be discussed in subsequent sections, both experimental and clinical investigations have been extremely promising [13].

Whether by direct healing via administration of cellular material and biomaterials or the ex vivo fabrication of whole organs for subsequent transplantation, although it still very early in its development, TERM is not limited by an organ shortage because cell cultures can be expanded in the laboratory and biomaterials can be manufactured without donor morbidity or mortality. Furthermore, the use of autologous cells would overcome the need for brutal immunosuppressive regimes that carry significant risks for the patient [14]. The aim of this chapter is to review the most important developments in TERM as they pertain to each abdominal organ.

## 13.2 Cell Types in Organ Bioengineering

TERM approaches to organ bioengineering typically involve cellular material either alone or in conjunction with a supporting scaffold. Determining the proper cell type most suitable for regeneration and/or seeding onto a scaffold remains a chief objective for each abdominal organ system. Experimental and clinical investigations have commonly involved fully differentiated adult cells, organ-specific progenitor cells, or pluripotent stem cells [15].

Adult somatic cells hold the advantage of being well-defined and readily isolated. However, compared with progenitor and stem cells, their lifespan and regenerative potential are limited. Furthermore, they lose function and differentiability when removed from the native environment. Their use is confined to the specific organ from which they are obtained. For example, cultured hepatocytes or tubular cells can be used only for liver scaffolds or renal scaffolds, respectively, and they would likely be minimally proliferative. Stem cells, however, possess the notable capacity of self-renewal and differentiability. Their applicability has a broader range than do adult somatic cells. Within the category of stem cells, the two types most commonly used are adult stem cells and embryonic stem cells (ESCs) [16].

ESCs are pluripotent stem cells derived from the inner cell mass of the embryo during the blastocyst stage. As hinted at by their purpose in the developing organism, they hold the ability to differentiate into any cell type when induced via specific environmental stimuli. Furthermore, they can self-renew indefinitely [17]. Researchers have successfully induced their differentiation into cell types from all three germ layers in vitro including cardiac cells [18], endothelial cells [19], neurons [20], insulin-producing cells [21], and renal tubular cells [22]. In the context of abdominal bioengineering, the use of ESCs would find wide applicability for experimental and clinical purposes. Nevertheless their use is significantly limited by ethical issues [23] along with potential teratogenicity [24]. Induced pluripotent stem cells (iPSCs) are generated via special reprogramming of adult cells resulting in their dedifferentiation into pluripotent stem cells. This process does not involve manipulation or destruction of primordial embryos, thus avoiding the ethical obstacles posed by ESCs. They have been derived from several cell types in experimental investigations, including human keratinocytes [25] and umbilical cord blood mononuclear cells [26].

Adult stem cells, however, benefit from wide availability in the adult organism and the relative ease of acquisition. They are thought to perform a major role in cellular repopulation and the replenishment process over time [27]. Indeed, stem cells residing in anatomical "niches" are responsible for the continual replacement of damaged or dead tissue compartments in response to microenvironmental cues [28]. Wound healing and bone remodeling are familiar examples of tissue systems with high turnover. Tissue systems with recognized niche populations of stem cells include skin [29], fat [30], intestine [31], and kidney [32], among others. Unlike somatic cells, they can be induced to differentiate along lineages other than the organ system from which they are derived. For example, fat stem cells have been successfully differentiated into cartilage cells, hematopoietic cells, neurons, bone cells, and skeletal muscle cells [33]. Nevertheless their differentiation potential is generally recognized to be limited to the germ layer from which they are derived, unlike the pluripotency of the ESC.

Adult cells harvested for regenerative purposes can be autologous, i.e., acquired from the patient. Autologous cells are beneficial because they are shielded from the recipient patient's immune system, because they would not be recognized as nonself. Given their ability to circumvent rejection, autologous cells represent the most desirable cell type in TERM investigations. Allogeneic cells are another option; acquired from another individual, their use is complicated by potential immunogenicity [34]. Heterologous cells derived from nonhuman organisms (e.g., porcine) represent another option under investigation. Ideally, cells used for TERM purposes would be expanded ex vivo and either (1) introduced directly into the patient to repair the damaged organ system, or (2) seeded onto a supporting scaffold for eventual reimplantation into the patient.

## 13.3 Gastrointestinal Tract

The intestinal and bowel conduits are complex, hollow structures involved in nutrient passage and absorption, digestion, excretion, and the innate immune system. The human tract undergoes continuous turnover and renewal throughout life, a process driven by the activity of resident stem cells [35]. This preexisting process of regeneration and renewal underscores the unique applicability of methods in regenerative medicine for bowel engineering. Taking into consideration the microanatomical and functional characteristics of the organ, the endeavor would aim toward regeneration of smooth muscle, specialized neuronal tissue, and a mucosal-epithelial bilayer. Correspondingly, one of the chief obstacles is the functional regeneration of diverse motility patterns, because motility is essential to proper intestinal functioning. Nevertheless, preliminary investigations have been promising [36].

The stomach has a major role in the gastrointestinal tract as a storage reservoir and digesting organ. Methods in tissue engineering could find applicability in cases involving gastrectomy, peptic ulcer disease, caustic ingestion, or trauma. Simple approaches have involved the use of collagen scaffolds to patch stomach wall defects [37]. This approach produced an intact mucosal layer with proton pumping cells but failed to recapture motility functioning when challenged with acetylcholine. An affiliated group later attempted again to employ collagen scaffolds but soaked them in autologous blood and bone marrow aspirate [38]. These collagen constructs were sutured over 5-cm circular defects of the stomach wall in canine models. The regenerated stomach retained native tensile strength and histologically showed villous mucosa and underlying dense connective tissue. Nevertheless the authors concluded that their engineered construct, although of improved biocompatibility, was still short of a functional gastrointestinal segment.

Speer et al. isolated organoid units from mouse glandular stomach and seeded them onto biodegradable scaffolds composed of polyglycolic acid (PGA) coated with poly-Llactic acid (PLA) and type I collagen [39]. The seeded scaffolds were implanted into the omentum of adult mice and harvested at designated times for analysis. The constructs were found to grow and proliferate as expanding spheres with simple columnar epithelium organized into gastric glands and an adjacent muscularis. Mucous, enteroendocrine, chief, and parietal cells were all expressed in the regenerated epithelium. Maemura et al. employed biodegradable polymers seeded with stomach epithelial
organoid units obtained from neonatal rats in various investigations [40]. They replaced recipient rat stomachs with tissue-engineered stomachs and discovered no evidence of stenosis or obstruction at the sites of anastomoses. Furthermore, histologic evaluation demonstrated well-developed vascularized tissue and stratified smooth muscle layers [41]. They also used the seeded scaffolds to patch gastric wall defects in rat models, with considerable success [42].

The small intestine is the primary segment for nutrient absorption within the gastrointestinal tract. This process depends on the integrity of microvilli structures that line the intestinal epithelium. Early regenerative investigations employed autologous tissue patches to repair defects in the intestinal wall. Later studies underscored the potential use of absorbable biomaterials as a patch-scaffold to facilitate tissue ingrowth [43]. Commonly used biomaterials have been collagen scaffolds, PLA, and PGA, among others. Fibrin hydrogels demonstrated sufficient mechanical rigidity to allow self-organization of circular sphincteric and intestinal smooth muscles, even in humans [44,45].

The advent of organ decellularization allows for the development of natural scaffolding material with an intact ECM, preserving the structural and biochemical properties necessary for tissue growth while obviating the need for biocompatible polymers. Notably, the ECM preserves the villus-crypt architecture and vasculature that supports intestinal epithelial regeneration [46].

The Vacanti group employed intestinal organoid units acquired from the small intestine and remodeled them on biocompatible matrices [47]. These implants were able to rescue morbidity associated with massive bowel resection in rat models. Chen et al. implanted ECM-derived submucosal tubes in canine models with small bowel resections without seeding material, to ascertain the regenerative potential [48]. The investigation used four dogs and concluded that this method was not feasible, although patch repair showed the potential for wall defect regeneration.

Numerous studies reported the use of bone marrow–derived mesenchymal stem cells (MSCs) for the regeneration of intestinal tissue. Hori et al. were able to regenerate intestinal segments by seeding MSCs onto collagen scaffolds [49]. Their constructs showed a transient distribution of  $\alpha$ -smooth muscle actin-positive cells but failed to regenerate a muscle layer, which is essential for peristalsis. ESCs and iPSCs have also been used to regenerate functional or semifunctional intestinal tissue, under proper induction by growth factors [50,51]. Watson et al. reported on the generation of human intestinal organoids procured in vitro from human ESCs or pluripotent stem cells that were able to engraft and vascularize in mouse models [52]. In vivo transplantation resulted in the marked expansion and maturation of differentiated epithelium and mesenchyme and also demonstrated digestive functions.

Although clinical translation has not yet proven feasible, findings and ongoing studies are showing great promise. Given the limited results of employing cells or biodegradable scaffolds alone, adult or stem cell seeding onto ECM-based scaffolds will likely achieve optimal results in future investigations. Identifying the most suitable cell type remains a primary challenge for future studies, along with establishing a scaffold that recaptures the microenvironmental cues of native intestine.

#### 13.4 Bladder

The architecture simplicity of the hollow bladder along with its relatively straightforward singular functions (i.e., storage and passage) renders it particularly amenable to bioengineering solutions. Furthermore, the limitations of commonly employed detubularized bowel segments for bladder replacement underscore the advantages of a tissue engineering approach [53]. The application of regenerative medicine technologies for bladder repair has mostly involved two basic approaches: (1) a cell-free, scaffold-only approach; and (2) scaffolds seeded with cells. The scaffold-only strategy employs a biocompatible and geometrically appropriate matrix to create an environment permissive of cellular infiltration and proliferation, which constitute the regenerative process. The cell-scaffold approach uses uroepithelial cells, smooth muscle cells, and/or other cell types and seeds them onto biocompatible scaffolds to bioengineer a bladder construct ex vivo for the purpose of eventual implantation. In theory, the latter approach is advantageous owing to the preparation of a more fully differentiated and phenotypically mature construct before implantation. In this way, it more closely resembles contemporary methods in surgical transplantation.

Both naturally derived materials such as collagen and alginate and synthetic polymers such as PGA and PLA have been employed as bladder wall substitutes in both experimental and clinical investigations. Naturally derived materials such as collagen have the advantage of enhanced biocompatibility, whereas synthetic materials have the benefit of large-scale manufacture with closer control over microstructural and mechanical properties. Both have been confirmed as being compatible with uroepithelial and smooth muscle tissue in experimental settings [54]. Decellularized scaffolds such as intestinal submucosa and acellular bladder matrices have also been employed in experimental investigations [55,56]. Nevertheless, these approaches are limited by the absence of cellular material, which can result in scarring, reduced reservoir volume, and graft contraction [57]. Furthermore, studies suggest that this approach does not adequately regenerate the muscular compartment, resulting in suboptimal elasticity and contractility [58].

More recent investigations suggest that the use of autologous cells seeded on bladdershaped biomaterial scaffolds is the best strategy in bladder engineering [59]. Atala et al. reported the first human clinical study of cystoplasty using cell-seeded collagen scaffolds, with dramatic success [60]. Seven patients needing cystoplasty were implanted with bladders manufactured using autologous urothelial and muscle cells expanded in culture and seeded on bladder-shaped scaffolds. Follow-up examination (mean, 46 months) showed increased compliance, decreased end-filling pressures, increased capacities, and longer dry periods over time. Raghavan et al. obtained a full-thickness bladder biopsy from a 17-year-old with neurogenic bladder caused by an iatrogenic spinal injury [61]. Urothelial and smooth muscle cells were isolated from the biopsy, expanded in vitro, and seeded onto a biodegradable scaffold, which was allowed to mature. This "neobladder construct" was then implanted surgically to augment bladder capacity and wrapped with omentum to provide neurovascular support. No follow-up has yet been reported.

The most suitable cell type for reseeding in bladder engineering is still under investigation, even with the encouraging results using smooth muscle and urothelial cells. Of course, autologous cells are highly advantageous because of their lack of immunogenicity [62]. Urothelial progenitor cells concentrated in the bladder neck and trigone area represent a strong candidate for the source of autologous material [63]. Amniotic fluid—derived [64], bone marrow—derived [65], and adipose-derived stem cells [66] are other potential autologous options because they have all have been shown to differentiate into bladder tissue under proper guidance.

There has been rapid growth in bladder engineering technology over the past several years [67]. Indeed, clinical trials are finishing up or are ongoing [68]. With proof-of-concept more than established, investigators must now focus on optimizing protocols while clinical trials lend increasing credibility to the technology.

#### 13.5 Kidney

The kidney is a complex solid organ with important roles in endocrine, metabolic, and immunologic homeostasis. Specifically, the kidney functions to filter the blood and to concentrate urine with toxic metabolic waste products, which highlights the crucial role of the kidney. Currently, dialysis and transplantation represent the reference standard treatment modalities for chronic kidney disease. However, dialysis does not cure kidney damage; it assumes a portion of the kidney's functions, particularly filtration. As such, dialysis leaves much to be desired. With the growing worldwide burden of hypertension and diabetes, the prevalence of chronic kidney disease is reaching epidemic proportions [69,70] and the need for functional kidneys is rising correspondingly.

The translational success of TERM strategies for the replacement of renal function depends on a better understanding of the native repair and regenerative processes occurring in vivo at the cellular and molecular levels [71,72]. Because of the proportionally increased metabolic demand of the kidney and its waste and toxin filtration function, renal tubular cells are constantly under the threat of acute injury and oxidative stress. For this reason, these cells contain unique regenerative abilities when damaged. In fact, researchers have observed surviving renal tubular cells giving rise to a new population of the cells after physiologic kidney damage [73–75]. Nagaike et al. observed that unilateral nephrectomy induces mitogenesis and hypertrophy in the contralateral kidney [76]. Cochrane et al. induced cortical tubular cell atrophy, tubular dilation, and interstitial macrophage infiltration via ureteral obstruction in a murine model of renal

injury. They showed that reversal of the obstruction induced a rapid process of reconstruction and interstitial matrix expansion that ultimately restored the glomerular filtration rate [77]. However, continuous and supraphysiologic damage characteristic of chronic kidney disease overpowers the regenerative properties of these cells. Furthermore, the growth of new nephrons, i.e., frank nephrogenesis, has not been shown to occur [78].

Researchers have thus explored the potential of cell therapy to restore kidney function in the face of widespread damage. Cell-based approaches seek to achieve kidney repair and regeneration in situ upon therapeutic administration. They are based on the observation that exogenously supplied cells stimulate the repair and proliferative process [79]. Progenitor cells harvested from the proximal tubules, glomerulus, peritubules, and papillae have all demonstrated some level of therapeutic capacity [80,81]. Stem cells obtained from urine have also shown some potential to reverse kidney damage and aid in the repair process [82].

Recruitment of bone marrow—derived MSC, a type of adult stem cell, to renal tubular damage sites has enhanced regenerative outcomes [83]. It is thought that MSCs stimulate the release of more proliferative mediators from native cells to enhance their own regenerative process. MSCs can also secrete specific chemokines, cytokines, and growth factors to promote growth and cellular protection from further damage. Using the nephrotoxic drug cisplatin to induce acute kidney damage in mice models, researchers have looked at MSC-induced proliferation and regeneration of epithelial cells in damaged renal tubules [84]. Although the use of MSCs is promising for acute kidney injury, there is a paucity of current literature regarding their use in models of chronic kidney disease, which hosts a different set of challenges and conditions. A study using rat models found that chronic kidney disease leads to premature senescence of MSCs and inhibits their typical regenerative potential, potentially limiting their use [85].

Enhanced understanding of the regenerative properties of renal cells has led to another avenue for the treatment of kidney damage: the use of embryonic kidney tissue. These primordial cells have been shown to integrate within adult organ systems, richly vascularize, and form new, mature nephrons (i.e., frank nephrogenesis) [86,87]. Ureteric bud and metanephric mesenchyme cultures have shown, though the inherent ability of mesonephric duct tissue, the ability to form collecting ducts through tubulogenesis and epithelization. The in vitro tissue was implanted into mice models and survived for over 5 weeks, demonstrating glomerular vascularization in vivo and thereby pointing to the therapeutic potential of these primordial tissues [88]. Imberti et al. implanted renal primordia under the kidney capsule of male rats with kidney injury [89]. The grafts went on to develop glomeruli and tubuli that filtered blood and produced urine in cyst-like structures. Furthermore, they initiated a process of regeneration in host tissue segments indicated by increased cell proliferation and vessel growth.

Researchers have explored the potential to bioengineer kidneys de novo for eventual implantation into patients, harnessing the advantages of surgical transplantation. The most promising strategy under investigation has been the cell-scaffold approach. Native kidneys are stripped of cellular material using detergents while preserving the ECM upon which to seed new cellular material. ECM scaffolds produced from animal and human kidneys have been shown to retain their innate biomolecular and biophysical properties along with their external and internal three-dimensional architecture [90]. Upon seeding with cells, signaling cytokines and growth factors from the retained ECM can guide cell differentiation toward organ-specific phenotypes and promote vasculogenesis necessary for full integration into the host.

Bonandrini et al. used ESCs to recellularize rat whole-kidney scaffolds by perfusing them through the renal artery along with circulating cell medium [91]. The investigators observed the loss of pluripotency and the shift toward mesoendodermal lineage, supporting the idea that decellularized kidneys can undergo rapid recellularization of vascular structures and glomeruli and induce differentiation along appropriate pathways. Nakayama et al. decellularized both the kidneys and lungs of rhesus monkeys and reseeded them with undifferentiated ESCs [92]. Expression levels of tubule markers and other kidney genes were higher in cells cultured on kidney ECM compared with those on lung ECM, which lent evidence to the idea that ECM guides differentiation selectively.

Our group successfully decellularized porcine kidneys and implanted them into porcine hosts with evidence of preserved functionality [93]. Moving to more clinically relevant models, a source of human kidneys available for bioengineering investigations was identified: organs originally intended for transplant purposes but discarded because of functional or anatomical anomalies (approximately 2600 kidneys annually in the United States) [94]. Upon detergent-perfusion, the successful production of viable ECM scaffolds using discarded kidneys was confirmed [95]. In a later study, amniotic fluid– derived stem cells were seeded on discarded kidney scaffolds to assess the potential to recellularize [96]. The cells attached, proliferated, and furthermore synthesized and secreted various chemokines and growth factors involved in angiogenesis and matrix remodeling including vascular endothelial growth factor, transforming growth factor- $\alpha$ , interleukin-8, metalloproteinase-2, and tissue inhibitor of metalloproteinase-2. Furthermore the cells were found to express early kidney developmental markers after 2 weeks of culture on the ECM, suggesting regeneration.

Both cell therapy and cell-scaffold strategies demonstrated potential for the treatment of kidney disease, which is currently limited to dialysis and transplantation. Further studies are needed to determine the best cell type for renal regeneration: adult stem cells, progenitor cells, or frank stem cells.

# 13.6 Pancreas

The pancreas is functionally composed of two parts, the exocrine pancreas and endocrine pancreas. Despite deriving from the same embryologic cells, these components are structurally different and are important for different physiologic systems. Although it accounts for 90–95% of the pancreas mass, the exocrine pancreas is not essential for survival owing to exogenous pancreatic digestive enzyme replacement therapy [97]. In contrast, the endocrine pancreas accounts for only roughly 2% of the pancreas mass while serving a vital physiologic role. The endocrine pancreas is not localized to a specific region of the pancreas but is interspersed around islets of Langerhans (highly vascularized centers) over the entire tissue. There are five main types of endocrine pancreatic cells (alpha, beta, delta, gamma, and epsilon) but the beta cell has a particularly vital role in glucose homeostasis and energy metabolism, thus drawing the most attention for bioengineering research and treatments with regard to endocrine pancreatic deficiencies.

One of the most common pathologic endocrine deficiencies is type 1 diabetes mellitus, in which beta cells are destroyed by the host's own immune system. About 1 in 300 individuals in the United States is diagnosed with type 1 diabetes mellitus by age 18 years [98]. Therapies for type 1 diabetes mellitus focus on the supplementation of exogenous insulin that is normally produced by beta cells. However, this method requires strict patient compliance and adequate patient education. Innovations have removed the need for conscious dosing. Although exogenous insulin therapy is effective at preventing acute decompensation, fewer than 40% achieve and maintain therapeutic targets [99]. The most commonly accepted treatment regimens lower the incidence of diabetic emergency. However, they require lifelong pharmacologic intervention without the possibility of remission. Beta-cell replacement, whether by whole pancreas or individual islet cell transplantation, is currently the only method to restore long-term, stable euglycemia in type 1 patients with diabetes.

Pancreatic transplantation is an invasive option that provides the patient an opportunity to be free from strict glycemic control by way of exogenous insulin. However, it is typically offered to adults only in conjunction with kidney transplants because of the high risk of surgical complications and lifetime of immunosuppressant medications [100]. Nevertheless recipients exhibit greater improvements in the micro- and macrovascular complications of diabetes while enjoying a great quality of life. The 5-year organ survival rate for pancreatic transplantations is roughly 50% [101].

An alternative solution involves the ectopic implantation of purified islet cells and simultaneous depletion of T-cells to recapture beta cell function while suppressing the autoimmune component of the disease [102,103]. Human islet cell purification and implantation have the same 5-year insulin independence rate as pancreatic transplantation but it has been found have a lower risk of complication and represents a less invasive solution [104].

Contemporary investigations employ TERM solutions to treat the disease without the need for lifelong pharmacologic intervention. One of the most heavily researched areas involves the induction of beta cell regeneration within host islets. There is a current debate in the research community regarding the source material of endogenous beta cell regeneration: It is still unknown whether existing beta cells or resident stem cells are primarily responsible [105,106]. Insulin-producing cells derived from ESCs are physiologically closer to beta cells than any adult stem cell—based beta cell. The conversion of

human ESCs toward beta cell phenotypes has been achieved through the initial differentiation into endodermal cells that can be further differentiated into insulin-producing endoderm derivatives [107,108].

To avoid the controversy of ESC procurement, advancements in beta cell replacements have turned toward iPSCs. Through the use of retroviruses to manipulate gene expression, adult somatic cells can be reprogrammed and individualized to specific patients. In animal models, experimental therapy has already demonstrated long-term correction of hyperglycemia [109]. Although promising, the technology surrounding iPSCs still faces obstacles posed by premature senescence, subchromosomal abnormalities, and destruction of patient-derived cells caused by autoimmune phenomena.

The recellularization of ECM-based scaffolds has also been under investigation regarding the de novo fabrication of pancreas organs. It has been observed that islets cultured in vitro on ECM increase the longevity of insulin production in response to glucose [110], likely owing to natural growth factors within the ECM that direct cell lines toward beta cells. In addition, studies have hinted at the potential use of xenographic transplants. Sani et al. demonstrated that a porcine pancreas could be decellularized while maintaining ECM composition and patent vasculature, and can subsequently be seeded with human amniotic fluid—derived stem cells [111]. The seeded islets demonstrate the ability to support normal pancreatic function by showing an increased metabolic rate as well as insulin secretion compared with isolated islets. Goh et al. successfully decellularized mouse pancreata and recellularized them with acinar and beta cell lines [112]. They showed successful engraftment within the three-dimensional decellularized pancreas along with strong up-regulation of insulin gene expression. Their findings supported the utility of whole-pancreas ECM for enhancing pancreatic cell functionality. Other investigators have reported similar findings [110].

Whereas all of these technologies may restore the functionality of destroyed islets, the primary pathology of type 1 diabetes is autoimmune, which will remain a persistent obstacle if left unaddressed. Future strategies must incorporate adjuvant immunomodulation or autologous tissue into current technologies to terminate the chronic disease state and preserve endocrine pancreas function.

#### 13.7 Liver

The liver has an innate capability to regenerate beyond the potential of other abdominal organs. In the event of injury to less than 70% of its total, the liver can fully regenerate within 6 months of the trauma [113]. However, this occurs through cellular hyperplasia of the residual liver and is not considered true regeneration [114]. Instead there is an observed compensatory enlargement of the remnant liver to meet the functional demands of the body. These properties of the liver make it uniquely amenable to methods in abdominal organ bioengineering. Current investigations involve two distinct regenerative pathways. One, similar to approaches in other organ systems, involves the use of

synthetic scaffolds or decellularizing an existing liver. These scaffolds would then be reseeded with host cells and allowed to regrow either in vivo or in vitro. Another option is to manipulate the existing cellular physiology of liver hepatocytes to induce liver regeneration, eliminating the need for a scaffold because the host will be able to use existing liver tissue.

Research has focused on the use of decellularization and recellularization to manufacture livers with healthy, disease-free tissue [115,116]. For liver regeneration to become clinically relevant, more studies must be performed to demonstrate:

- **1.** that whole liver can be decellularized and that this can be applicable to human liver,
- **2.** that human hepatocytes can be seeded and maintain viability on a decellularized scaffold, and
- **3.** that the regenerated liver cells can match the prior functioning of the native liver in terms of protein synthesis and breakdown, detoxification, and bilirubin metabolism.

Studies have demonstrated a technique in which the murine liver can be successfully converted into an acellular scaffold by applying sodium dodecyl sulfate detergents that remove cellular material and debris while retaining the proper collagen structure, laminin basement membranes, and the vascular network [115]. In the same study, the scaffold had enough structural integrity to withstand cannulation and perfusion of the reseeding cells. A follow-up study illustrated the ability of the perfused cells to leave the vasculature and distribute among the matrix [117]. Using electron microscopy, rodent hepatocytes were able to migrate to decellularized sinusoidal spaces and display increased levels of urea synthesis and select drug metabolites compared with the sandwich hepatocyte cultures. This is particularly important because the liver requires a wide range of enzymes, cofactors, serum proteins, and acute phase reactants to be fully functional. This graft was able to withstand transplantation for 8 hours in a rat host before harvesting, which showed that the hepatocytes retained their morphology, position, and integrity.

Yagi et al. reported on the successful decellularization of porcine livers, thereby producing human-sized ECM scaffolds that were capable of supporting hepatocyte engraftment and reorganization in three dimensions [118]. Baptista et al. [13] also illustrated that it is possible to decellularize whole liver (in mice, rats, ferrets, rabbits, and adult pigs) and that they can be successfully seeded with human hepatocyte progenitors. This study used the inferior vena cava for decellularization and the portal vein as well as the vena cava for the perfusion of cells. Immunohistochemical staining suggested an architectural blueprint dictated by the acellular scaffold. Discrete populations of differing cells were found throughout the matrix, which suggested that the retention of glycosaminoglycans and collagen structure provides the necessary environmental signaling to regulate cell differentiation and dictates the fate of the progenitor hepatic and endothelial cells. Wang et al. [119] demonstrated that progenitor cells seeded onto a scaffold could differentiate into hepatocytes and cholangiocytes, illustrating that the

scaffold could dictate the differentiation toward distinct cell lineages. Porcine liver has been successfully decellularized and repopulated with human cells [120]. Analysis revealed that the cells were able to differentiate into mature hepatocytes while maintaining active metabolism, the ability to withstand physiologic shear stress from blood flow, and the ability to synthesize albumin. Implanted liver ECM has even been shown to induce hepatic regeneration via hepatocyte proliferation, which further illustrates the critical relationship between ECM and cells.

Observations on the natural regenerative capacity of injured livers led to the use of progenitor liver cells for bioengineering research. As mentioned, cellular hyperplasia replaces the lost volume after liver amputation. Current studies attempt to enhance the innate regenerative ability of the liver through terminal differentiation via resident bone marrow–derived or liver progenitor cells [121–123]. Infusing autologous bone marrow cells into patients with decompensated liver cirrhosis has already been demonstrated to be safe and in some cases clinically beneficial [124,125]. Another avenue is the use of cytokine granulocyte colony-stimulating factor (G-CSF) to activate hematopoietic stem cell differentiation and mobilization while increasing the ability of resident progenitor cells to respond to injury [126]. Stem cells isolated from the peripheral bloodstream of patients with liver disease infused with G-CSF were isolated and infused back into the patients. These patients showed significant improvements in liver enzyme levels as well as the serum bilirubin compared with the control group [127,128].

Despite these findings, researchers are unable to determine the specific progenitor cell type that actually aids the regeneration process. Current targets include the bone marrow–derived MSCs [129,130] and hematopoietic stem cells [131–133], whereas fetal liver progenitor cells also remain a possibility [134]. Fetal hepatoblasts have been shown to have the ability to differentiate mature hepatic phenotypes when seeded on decellularized scaffolds [135]; however these cells are not common in adult liver. More studies are required to determine the cell types needed to repopulate the scaffold to form a functional liver. Furthermore, liver viability and functionality need to be confirmed after implantation.

# 13.8 The Challenge of Immunosuppression

One of the pillars of modern bioengineering and transplant research is the desire to achieve durable immune tolerance. Current organ transplants require long-term treatment with brutal immunosuppressive protocols, which target an array of molecular sites to inhibit the communication between antigen presenting cells and T-cell receptors [136]. Long-term use of these drugs leads to comorbidities and complications that extend beyond the physiologic arena serviced by the transplanted organ. These effects include drug-specific toxicities, metabolic complications, and secondary malignancy, among others [137,138]. In fact, 40% of kidney transplant recipients have an increased risk of secondary malignancies compared with the age-matched nonimmunosuppressed population after 20 years of chronic immunosuppressant use. In the United States alone,

there are over 1000 individuals living with an intestinal transplant [139], approximately 14,000 with a pancreatic transplant [140], 11,000 with a lung transplant [141], over 27,000 with a heart transplant [142], over 59,000 with a liver transplant [143], and almost 200,000 with a kidney transplant [144]. This totals over 300,000 transplant recipients needing long-term immunosuppression and/or tolerance monitoring.

After anoxia, the leading cause of mortality in kidney transplant recipients is cardiovascular disease resulting from the cardiotoxicity of the long-term immunosuppressive treatments [145,146]. Thus there is a strong need to balance the safety risks of high immunosuppressive therapy with graft efficacy. TERM investigations may provide innovative treatment solutions that can eliminate the need for antirejection therapy.

To ensure the graft survival and functionality, adherence to long-term immunosuppression is vital. The costs of immunosuppressive agents and overall antirejection therapy costs create an immense financial burden, which makes it a leading cause of nonadherence to the therapy [147]. At risk, patients may be faced with the decision between affording their medications and maintaining an adequate livelihood. In these cases, patients are at risk for a lower quality of life owing to the cost of medication, increased morbidity as a result of the medications, graft failure because of insufficient adherence to the medications, and death as a result of any of these conditions. Pressure to engineer innovative tactics to minimize costs and improve long-term recipient outcomes stems from the current high costs associated with transplant maintenance, organ procurement, and transplantation. Immune tolerance afforded by TERM would serve to reduce the cost and amount of immunosuppression required by transplant recipients, thereby improving quality of life on multiple fronts.

Durable tolerance can be achieved by one of two avenues: increased genetic compatibility of the graft and recipient host or better immunologic tolerance of the recipient. These states are difficult to arrange and the latter usually requires immunosuppression. Liver transplantation has seen limited success as a result of fulfilling extensive criteria in both the donor and the recipient [148], although this has been less accessible in other organ systems. Notably, up to a third of monozygotic twin organ transplants require long-term immunosuppressant therapy as a result of slight genetic variance, which suggests that rejection extends beyond human leukocyte antigen matching [149]. To optimize TERM solutions and guide the use of exogenous cellular material, increased understanding of the intricate factors that dictate tolerance of foreign bodies by the host immune system will be required.

The field of bioengineering hollow organs such as the bladder and blood vessel from autologous cells has already successfully transplanted hollow organs without the need for immunosuppression [150,151]. Approximately 200 patients have undergone successful transplantation of bioengineered grafts (author's calculation), which surpasses the number of hollow organ transplant recipients successfully removed from of long-term immunosuppression. These results are an early indicator of future advancements in the biofabrication of complex organs that eliminate the morbidities associated with current antirejection therapy.

# 13.9 Conclusions and Perspectives

Large-scale clinical translation of abdominal organ bioengineering remains a distant possibility, because substantial work in the laboratory remains. Nevertheless the chief obstacles posed by current medical and surgical modalities (particularly transplantation) are concerning and have created a pocket of necessity within which TERM investigations are taking place. The use of autologous cellular material has the potential to obviate the need for lifelong antirejection therapies. Furthermore, de novo organ fabrication using cell-scaffold technology could theoretically provide a limitless supply of transplantable organs for patients on the waiting list, thus circumventing the challenge of organ shortage.

Further research is needed to identify the most suitable cell type for regenerative investigations. Although adult somatic cells specific to the organ system being treated have shown promise, stem cells have the advantage of being multipotent and renewable, which are particularly attractive qualities in the context of TERM. However, the abundance of subcategories and subniches of stem cells add more complexity to identifying the best candidate. Furthermore, the risks associated with their use need to be further characterized before they can be used safely in patients on a large scale.

The immunogenicity of decellularized scaffolds is another issue that needs to be assessed. Although their tolerability has been recognized both empirically and theoretically, suboptimal decellularization protocols can leave behind residual antigen activity, triggering an immune response in the host.

Investigators must also seek to better understand the interaction between ECM and cells both endogenous and exogenous. The improved knowledge base would likely result in improved cellular engraftment, migration, and differentiation within the acellular matrix. The source of this understanding could well lie in developmental biology and organogenesis, in which ECM-guided differentiation is paramount.

Although TERM solutions for abdominal organ engineering show great promise, current researchers must also keep an eye toward research funding limitations and the role of commercial entities in bringing their innovations from bench to bedside. With these last obstacles addressed, advances in abdominal organ bioengineering may well bring about a paradigm shift for the treatment of end-stage organ diseases.

TERM	Tissue engineering and regenerative medicine
ECM	Extracellular matrix
ESC	Embryonic stem cell
IPSC	Induced pluripotent stem cell
MSC	Mesenchymal stem cell

# Abbreviations

### References

- [1] World Health Organization, World Health Statistic, World Health Organization, Geneva, Switzerland, 2013.
- [2] World Health Organization, GKT1 Activity and Practices. http://www.who.int/transplantation/gkt/ statistics/en/.
- [3] National Kidney Foundation, Organ donation and transplant statistics. https://www.kidney.org/ news/newsroom/factsheets/Organ-Donation-and-Transplantation-Stats.
- [4] Organ Procurement and Transplantation Network. http://optn.transplant.hrsa.gov/converge/ latestData/rptData.asp.
- [5] R. Katari, A. Peloso, G. Orlando, Tissue engineering and regenerative medicine: semantic considerations for an evolving paradigm, Frontiers in Bioengineering and Biotechnology 2 (2014) 57.
- [6] R.S. Katari, A. Peloso, G. Orlando, Tissue engineering, Advances in Surgery 48 (2014) 137–154.
- [7] G. Orlando, S. Soker, R.J. Stratta, A. Atala, Will regenerative medicine replace transplantation? Cold Spring Harbor Perspectives in Medicine 3 (8) (2013).
- [8] A. Peloso, A. Dhal, J.P. Zambon, et al., Current achievements and future perspectives in wholeorgan bioengineering, Stem Cell Research & Therapy 6 (2015) 107.
- [9] J.P. Guyette, S.E. Gilpin, J.M. Charest, L.F. Tapias, X. Ren, H.C. Ott, Perfusion decellularization of whole organs, Nature Protocols 9 (6) (2014) 1451–1468.
- [10] M.J. Bissell, H.G. Hall, G. Parry, How does the extracellular matrix direct gene expression? Journal of Theoretical Biology 99 (1982) 31–68.
- [11] C. Frantz, K.M. Stewart, V.M. Weaver, The extracellular matrix at a glance, Journal of Cell Science 123 (Pt 24) (2010) 4195–4200.
- [12] P.M. Crapo, T.W. Gilbert, S.F. Badylak, An overview of tissue and whole organ decellularization processes, Biomaterials 32 (12) (2011) 3233–3243.
- [13] P.M. Baptista, M.M. Siddiqui, G. Lozier, S.R. Rodriguez, A. Atala, S. Soker, The use of whole organ decellularization for the generation of a vascularized liver organoid, Hepatology 53 (2) (2011) 604–617.
- [14] C. Mason, P. Dunnill, Assessing the value of autologous and allogeneic cells for regenerative medicine, Regenerative Medicine 4 (6) (2009) 835–853 [12].
- [15] D.A. Goldfarb, Tissue engineering stem cells, and cloning: opportunities for regenerative medicine, The Journal of Urology 173 (4) (2005) 1431.
- [16] N. Gjorevski, A. Ranga, M.P. Lutolf, Bioengineering approaches to guide stem cell-based organogenesis, Development 141 (9) (2014) 1794–17804.
- [17] A.H. Brivanlou, F.H. Gage, R. Jaenisch, T. Jessell, D. Melton, J. Rossant, Stem cells. Setting standards for human embryonic stem cells, Science 300 (5621) (2003) 913–916.
- [18] I. Kehat, D. Kenyagin-karsenti, M. Snir, et al., Human embryonic stem cells can differentiate into myocytes with structural and functional properties of cardiomyocytes, The Journal of Clinical Investigation 108 (3) (2001) 407–414.
- [19] S. Levenberg, J.S. Golub, M. Amit, J. Itskovitz-eldor, R. Langer, Endothelial cells derived from human embryonic stem cells, Proceedings of the National Academy of Sciences of the United States of America 99 (7) (2002) 4391–4396.
- [20] B.E. Reubinoff, P. Itsykson, T. Turetsky, et al., Neural progenitors from human embryonic stem cells, Nature Biotechnology 19 (12) (2001) 1134–1140.
- [21] S. Assady, G. Maor, M. Amit, J. Itskovitz-eldor, K.L. Skorecki, M. Tzukerman, Insulin production by human embryonic stem cells, Diabetes 50 (8) (2001) 1691–1697.

- [22] K. Narayanan, K.M. Schumacher, F. Tasnim, et al., Human embryonic stem cells differentiate into functional renal proximal tubular-like cells, Kidney International 83 (4) (2013) 593–603.
- [23] H.W. Denker, Potentiality of embryonic stem cells: an ethical problem even with stem cell source, Journal of Medical Ethics 32 (2006) 665–671.
- [24] A. Aldahmash, M. Atteya, M. Elsafadi, M. Al-Nbaheen, H.A. Al-Mubarak, R. Vishnubalaji, A. Al-Roalle, S. Al-Harbi, M. Manikandan, K.I. Matthaei, A. Mahmood, Teratoma formation in immunocompetent mice after syngeneic and allogeneic implantation of gerline capable mouse embryonic stem cells, Asian Pacific Journal of Cancer Prevention 14 (2013) 5705–5711.
- [25] T. Aasen, A. Raya, M.J. Barrero, et al., Efficient and rapid generation of induced pluripotent stem cells from human keratinocytes, Nature Biotechnology 26 (11) (2008) 1276–1284.
- [26] J. Wang, Q. Gu, J. Hao, et al., Generation of induced pluripotent stem cells with high efficiency from human umbilical cord blood mononuclear cells, Genomics Proteomics Bioinformatics 11 (5) (2013) 304–311.
- [27] A. Spradling, D. Drummond-barbosa, T. Kai, Stem cells find their niche, Nature 414 (6859) (2001) 98–104.
- [28] L. Li, T. Xie, Stem cell niche: structure and function, Annual Review of Cell and Developmental Biology 21 (2005) 605–631.
- [29] E. Fuchs, Skin stem cells: rising to the surface, Journal of Cell Biology 180 (2) (2008) 273-284.
- [30] B.R. Sousa, R.C. Parreira, E.A. Fonseca, M.J. Amaya, F.M. Tonelli, S.M. Lacerda, P. Lalwani, A.K. Santos, K.N. Gomes, H. Ulrich, A.H. Kihara, R.R. Resende, Human adult stem cells from diverse origins: an overview from multiparametric immunophenotyping to clinical applications, Cytometry. Part A 85 (2014) 43–77.
- [31] L.G. Van der flier, H. Clevers, Stem cells, self-renewal, and differentiation in the intestinal epithelium, Annual Review of Physiology 71 (2009) 241–260.
- [32] B. Bussolati, S. Bruno, C. Grange, et al., Isolation of renal progenitor cells from adult human kidney, The American Journal of Pathology 166 (2) (2005) 545–555.
- [33] B. Lindroos, R. Suuronen, S. Miettinen, The potential of adipose stem cells in regenerative medicine, Stem Cell Reviews 7 (2) (2011) 269–291.
- [34] T.B. Strom, L.J. Field, M. Ruediger, Allogeneic stem cells, clinical transplantation and the origins of regenerative medicine, Current Opinion in Immunology 14 (5) (2002) 601–605.
- [35] B.D. Simons, H. Clevers, Stem cell self-renewal in intestinal crypt, Experimental Cell Research 317 (19) (2011) 2719-2724.
- [36] K.N. Bitar, S. Raghavan, Intestinal tissue engineering: current concepts and future vision of regenerative medicine in the gut, Neurogastroenterology & Motility 24 (1) (2012) 7–19.
- [37] Y. Hori, T. Nakamura, K. Matsumoto, Y. Kurokawa, S. Satomi, Y. Shimizu, Experimental study on in situ tissue engineering of the stomach by an acellular collagen sponge scaffold graft, ASAIO Journal 47 (3) (2001) 206–210.
- [38] M. Araki, H. Tao, T. Sato, et al., Development of a new tissue-engineered sheet for reconstruction of the stomach, Artificial Organs 33 (10) (2009) 818–826.
- [39] A.L. Speer, F.G. Sala, J.A. Matthews, T.C. Grikscheit, Murine tissue-engineered stomach demonstrates epithelial differentiation, Journal of Surgical Research 171 (1) (2011) 6–14.
- [40] T. Maemura, M. Shin, M. Kinoshita, et al., A tissue-engineered stomach shows presence of proton pump and G-cells in a rat model, resulting in improved anemia following total gastrectomy, Artificial Organs 32 (3) (2008) 234–239.
- [41] T. Maemura, M. Shin, M. Sato, H. Mochizuki, J.P. Vacanti, A tissue-engineered stomach as a replacement of the native stomach, Transplantation 76 (1) (2003) 61–65.

- [42] T. Maemura, M. Kinoshita, M. Shin, et al., Assessment of a tissue-engineered gastric wall patch in a rat model, Artificial Organs 36 (4) (2012) 409–417.
- [43] M.K. Chen, E.A. Beierle, Animal models for intestinal tissue engineering, Biomaterials 25 (9) (2004) 1675–1681.
- [44] L. Hecker, K. Baar, R.G. Dennis, K.N. Bitar, Development of a three-dimensional physiological model of the internal anal sphincter bioengineered in vitro from isolated smooth muscle cells, American Journal of Physiology-Gastrointestinal and Liver Physiology 289 (2) (2005) G188–G196.
- [45] S. Somara, R.R. Gilmont, R.G. Dennis, K.N. Bitar, Bioengineered internal anal sphincter derived from isolated human internal anal sphincter smooth muscle cells, Gastroenterology 137 (1) (2009) 53-61.
- [46] R.R. Resende, E.A. Fonseca, F.M.P. Tonelli, B.R. Sousa, A.K. Santos, K.N. Gomes, S. Guatimosim, A.H. Kihara, L.O. Ladeira, Scale/Topography of substrates surface resembling extracellular matrix for tissue engineering, Journal of Biomedical Nanotechnology 10 (2014) 1157–1193.
- [47] T.C. Grikscheit, A. Siddique, E.R. Ochoa, et al., Tissue-engineered small intestine improves recovery after massive small bowel resection, Annals of Surgery 240 (5) (2004) 748–754.
- [48] M.K. Chen, S.F. Badylak, Small bowel tissue engineering using small intestinal submucosa as a scaffold, Journal of Surgical Research 99 (2) (2001) 352–358.
- [49] Y. Hori, T. Nakamura, D. Kimura, et al., Experimental study on tissue engineering of the small intestine by mesenchymal stem cell seeding, Journal of Surgical Research 102 (2) (2002) 156–160.
- [50] S.R. Finkbeiner, J.R. Spence, A gutsy task: generating intestinal tissue from human pluripotent stem cells, Digestive Diseases and Sciences 58 (5) (2013) 1176–1184.
- [51] J.R. Spence, C.N. Mayhew, S.A. Rankin, et al., Directed differentiation of human pluripotent stem cells into intestinal tissue in vitro, Nature 470 (7332) (2011) 105–109.
- [52] C.L. Watson, M.M. Mahe, J. Múnera, et al., An in vivo model of human small intestine using pluripotent stem cells, Nature Medicine 20 (11) (2014) 1310–1314.
- [53] R.E. Hautmann, Urinary diversion: ileal conduit to neobladder, The Journal of Urology 169 (3) (2003) 834-842.
- [54] J.L. Pariente, B.S. Kim, A. Atala, In vitro biocompatibility assessment of naturally derived and synthetic biomaterials using normal human urothelial cells, Journal of Biomedical Materials Research 55 (1) (2001) 33–39.
- [55] B.P. Kropp, M.K. Rippy, S.F. Badylak, et al., Regenerative urinary bladder augmentation using small intestinal submucosa: urodynamic and histopathologic assessment in long-term canine bladder augmentations, The Journal of Urology 155 (6) (1996) 2098–2104.
- [56] J. Wefer, K.D. Sievert, N. Schlote, et al., Time dependent smooth muscle regeneration and maturation in a bladder acellular matrix graft: histological studies and in vivo functional evaluation, The Journal of Urology 165 (5) (2001) 1755–1759.
- [57] A. Atala, Tissue engineering in urologic surgery, Urologic Clinics of North America 25 (1) (1998) 39–50.
- [58] H. Orabi, S. Bouhout, A. Morissette, A. Rousseau, S. Chabaud, S. Bolduc, Tissue engineering of urinary bladder and urethra: advances from bench to patients, Scientific World Journal 2013 (2013) 154564.
- [59] A. Atala, Tissue engineering of human bladder, British Medical Bulletin 97 (2011) 81–104.
- [60] A. Atala, S.B. Bauer, S. Soker, J.J. Yoo, A.B. Retik, Tissue-engineered autologous bladders for patients needing cystoplasty, Lancet 367 (9518) (2006) 1241–1246.
- [61] A.M. Raghavan, P.J. Shenot, Bladder augmentation using an autologous neo-bladder construct, Kidney International 76 (2) (2009) 236.

- [62] B.G. Cilento, M.R. Freeman, F.X. Schneck, A.B. Retik, A. Atala, Phenotypic and cytogenetic characterization of human bladder urothelia expanded in vitro, The Journal of Urology 152 (2 Pt 2) (1994) 665–670.
- [63] E.A. Kurzrock, D.K. Lieu, L.A. Degraffenried, C.W. Chan, R.R. Isseroff, Label-retaining cells of the bladder: candidate urothelial stem cells, American Journal of Physiology – Renal Physiology 294 (6) (2008) F1415–F1421.
- [64] P. De coppi, G. Bartsch, M.M. Siddiqui, et al., Isolation of amniotic stem cell lines with potential for therapy, Nature Biotechnology 25 (1) (2007) 100–106.
- [65] G. Anumanthan, J.H. Makari, L. Honea, et al., Directed differentiation of bone marrow derived mesenchymal stem cells into bladder urothelium, The Journal of Urology 180 (4 Suppl.) (2008) 1778–1783.
- [66] S. Salemi, M. Tremp, J.A. Plock, et al., Differentiated adipose-derived stem cells for bladder bioengineering, Scandinavian Journal of Urology (2015) 1–8.
- [67] A. Atala, Regenerative bladder augmentation using autologous tissue-when will we get there? The Journal of Urology 191 (5) (2014) 1204–1205.
- [68] D.B. Joseph, J.G. Borer, R.E. De Filippo, S.J. Hodges, G.A. Mclorie, Autologous cell seeded biodegradable scaffold for augmentation cystoplasty: phase II study in children and adolescents with spina bifida, The Journal of Urology 191 (5) (2014) 1389–1395.
- [69] K.U. Eckardt, J. Coresh, O. Devuyst, et al., Evolving importance of kidney disease: from subspecialty to global health burden, Lancet 382 (9887) (2013) 158–169.
- [70] T.J. Hoerger, S.A. Simpson, B.O. Yarnoff, M.E. Pavkov, N. Ríos Burrows, S.H. Saydah, D.E. Williams, X. Zhuo, The future burden of CKD in the United States: a simulation model for the CDC CKD initiative, American Journal of Kidney Diseases 65 (2015) 403–411.
- [71] J.P. Zambon, R.S. Magalhaes, I. Ko, et al., Kidney regeneration: where we are and future perspectives, World Journal of Nephrology 3 (3) (2014) 24–30.
- [72] M. Salvatori, A. Peloso, R. Katari, G. Orlando, Regeneration and bioengineering of the kidney: current status and future challenges, Current Urology Reports 15 (1) (2014) 379.
- [73] G. Orlando, S. Soker, R.J. Stratta, Organ bioengineering and regeneration as the new Holy Grail for organ transplantation, Annals of Surgery 258 (2013) 221–232.
- [74] A.J. Matas, J.M. Smith, M.A. Skeans, et al., OPTN/SRTR 2011 annual Data Report: kidney, Journal of Transplantation 13 (Suppl. 1) (2013) 11–46.
- [75] M.B. Herrera, B. Bussolati, S. Bruno, V. Fonsato, G.M. Romanazzi, G. Camussi, Mesenchymal stem cells contribute to the renal repair of acute tubular epithelial injury, International Journal of Molecular Medicine 14 (2004) 1035–1041.
- [76] M. Nagaike, S. Hirao, H. Tajima, et al., Renotropic functions of hepatocyte growth factor in renal regeneration after unilateral nephrectomy, Journal of Biological Chemistry 266 (34) (1991) 22781–22784.
- [77] A.L. Cochrane, M.M. Kett, C.S. Samuel, et al., Renal structural and functional repair in a mouse model of reversal of ureteral obstruction, Journal of the American Society of Nephrology 16 (12) (2005) 3623–3630.
- [78] A.J. Davidson, Uncharted waters: nephrogenesis and renal regeneration in fish and mammals, Pediatric Nephrology 26 (9) (2011) 1435–1443.
- [79] M.H. Little, Regrow or repair: potential regenerative therapies for the kidney, Journal of the American Society of Nephrology 17 (9) (2006) 2390–2401.
- [80] O. Pleniceanu, O. Harari-Steinberg, B. Dekel, Concise review: kidney stem/progenitor cells: differentiate, sort out, or reprogram? Stem Cells 28 (9) (2010) 1649–1660.

- [81] P. Romagnani, L. Lasagni, G. Remuzzi, Renal progenitors: an evolutionary conserved strategy for kidney regeneration, Nature Reviews. Nephrology 9 (3) (2013) 137–146.
- [82] Y. Zhang, E. McNeill, H. Tian, S. Soker, K.-E. Andersson, J.J. Yoo, A. Atala, Urine derived cells are a potential source for urological tissue reconstruction, The Journal of Urology 180 (2008) 2226–2233.
- [83] M. Morigi, B. Imberti, C. Zoja, et al., Mesenchymal stem cells are renotropic, helping to repair the kidney and improve function in acute renal failure, Journal of the American Society of Nephrology 15 (7) (2004) 1794–1804.
- [84] M. Morigi, M. Introna, B. Imberti, D. Corna, M. Abbate, C. Rota, D. Rottoli, A. Benigni, N. Perico, C. Zoja, et al., Human bone marrow mesenchymal stem cells accelerate recovery of acute renal injury and prolong survival in mice, Stem Cells 26 (2008) 2075–2082.
- [85] B.M. Klinkhammer, R. Kramann, M. Mallau, et al., Mesenchymal stem cells from rats with chronic kidney disease exhibit premature senescence and loss of regenerative potential, PLoS One 9 (3) (2014) e92115.
- [86] D.R. Abrahamson, P.L. St john, D.J. Pillion, D.C. Tucker, Glomerular development in intraocular and intrarenal grafts of fetal kidneys, Laboratory Investigation 64 (5) (1991) 629–639.
- [87] A.S. Woolf, S.J. Palmer, M.L. Snow, L.G. Fine, Creation of a functioning chimeric mammalian kidney, Kidney International 38 (5) (1990) 991–997.
- [88] E. Rosines, K. Johkura, X. Zhang, H.J. Schmidt, M. De Cambre, K.T. Bush, S.K. Nigam, Constructing kidney-like tissues from cells based on programs for organ development: toward a method of in vitro tissue engineering of the kidney, Tissue Engineering. Part A 16 (2010) 2441–2455.
- [89] B. Imberti, D. Corna, P. Rizzo, et al., Renal primordia activate kidney regenerative events in a rat model of progressive renal disease, PLoS One 10 (3) (2015) e0120235.
- [90] D.C. Sullivan, S.H. Mirmalek-sani, D.B. Deegan, et al., Decellularization methods of porcine kidneys for whole organ engineering using a high-throughput system, Biomaterials 33 (31) (2012) 7756–7764.
- [91] B. Bonandrini, M. Figliuzzi, E. Papadimou, M. Morigi, N. Perico, F. Casiraghi, C. Dipl, F. Sangalli, S. Conti, A. Benigni, et al., Recellularization of well-preserved acellular kidney scaffold using embryonic stem cells, Tissue Engineering. Part A 20 (2014) 1486–1498.
- [92] K.H. Nakayama, C.C. Lee, C.A. Batchelder, A.F. Tarantal, Tissue specificity of decellularized rhesus monkey kidney and lung scaffolds, PLoS One 8 (5) (2013) e64134.
- [93] G. Orlando, A.C. Farney, S.S. Iskandar, S.H. Mirmalek-Sani, D.C. Sullivan, E. Moran, T. AbouShwareb, D.C. Paolo, K.J. Wood, R.J. Stratta, et al., Production and implantation of renal extracellular matrix scaffolds from porcine kidneys as a platform for renal bioengineering investigations, Annals of Surgery 256 (2012) 363–370.
- [94] G. Orlando, C. Booth, Z. Wang, et al., Discarded human kidneys as a source of ECM scaffold for kidney regeneration technologies, Biomaterials 34 (2013) 5915–5925.
- [95] A. Peloso, A. Petrosyan, S. Da Sacco, et al., Renal extracellular matrix scaffolds from discarded kidneys maintain glomerular morphometry and vascular resilience and retains critical growth factors, Transplantation (September 2015) 1807–1816.
- [96] A. Petrosyan, G. Orlando, A. Peloso, Z. Wang, A.C. Farney, G. Rogers, R. Katari, S. Da Sacco, S. Sedrakyan, R.E. De Filippo, R.J. Stratta, S. Soker, L. Perin, Understanding the bioactivity of stem cells seeded on extracellular matrix scaffolds produced from discarded human kidneys: a critical step towards a new generation bio-artificial kidney, CellR4 3 (1) (2015) e1401.
- [97] J.G. D'Haese, G.O. Ceyhan, I.E. Demir, P. Layer, W. Uhl, M. Löhr, R. Rychlik, K. Pirilis, Y. Zöllner, B. Gradl, et al., Pancreatic enzyme replacement therapy in patients with exocrine pancreatic insufficiency due to chronic pancreatitis: a 1-year disease management study on symptom control and quality of life, Pancreas 43 (2014) 834–841.

- [98] Z. Tao, A. Shi, J. Zhao, Epidemiological perspectives of diabetes, Cell Biochemistry and Biophysics (2015) 1–5.
- [99] G. Orlando, R.J. Stratta, J. Light, Pancreas transplantation for type 2 diabetes mellitus, Current Opinion in Organ Transplantation 16 (1) (2011) 110–115.
- [100] R. Bottino, M. Trucco, Clinical implementation of islet transplantation: a current assessment, Pediatric Diabetes (September 2015) 393–401.
- [101] R. Kandaswamy, P.G. Stock, M.A. Skeans, et al., OPTN/SRTR 2011 annual Data Report: pancreas, American Journal of Transplantation 13 (Suppl. 1) (2013) 47–72.
- [102] M.D. Bellin, R. Kandaswamy, J. Parkey, H.-J. Zhang, B. Liu, S.H. Ihm, J.D. Ansite, J. Witson, P. Bansal-Pakala, A.N. Balamurugan, et al., Prolonged insulin independence after islet allotransplants in recipients with type 1 diabetes, American Journal of Transplantation 8 (2008) 2463–2470.
- [103] C. Ricordi, P.E. Lacy, E.H. Finke, B.J. Olack, D.W. Scharp, Automated method for isolation of human pancreatic islets, Diabetes 37 (1988) 413–420.
- [104] M.D. Bellin, F.B. Barton, A. Heitman, J.V. Harmon, R. Kandaswamy, A.N. Balamurugan, D.E.R. Sutherland, R. Alejandro, B.J. Hering, Potent induction immunotherapy promotes long-term insulin independence after islet transplantation in type 1 diabetes, American Journal of Transplantation 12 (2012) 1576–1583.
- [105] Y. Dor, J. Brown, O.I. Martinez, D.A. Melton, Adult pancreatic beta-cells are formed by selfduplication rather than stem-cell differentiation, Nature 429 (2004) 41–46.
- [106] X. Xu, J. D'Hoker, G. Stangé, S. Bonné, N. De Leu, X. Xiao, M. Van De Casteele, G. Mellitzer, Z. Ling, D. Pipeleers, et al., β cells can Be generated from endogenous progenitors in injured adult mouse pancreas, Cell 132 (2008) 197–207.
- [107] K.A. D'Amour, A.D. Agulnick, S. Eliazer, O.G. Kelly, E. Kroon, E.E. Baetge, Efficient differentiation of human embryonic stem cells to definitive endoderm, Nature Biotechnology 23 (2005) 1534–1541.
- [108] K.A. D'Amour, A.G. Bang, S. Eliazer, O.G. Kelly, A.D. Agulnick, N.G. Smart, M.A. Moorman, E. Kroon, M.K. Carpenter, E.E. Baetge, Production of pancreatic hormone-expressing endocrine cells from human embryonic stem cells, Nature Biotechnology 24 (2006) 1392–1401.
- [109] Z. Alipio, W. Liao, E.J. Roemer, et al., Reversal of hyperglycemia in diabetic mouse models using induced-pluripotent stem (iPS)-derived pancreatic beta-like cells, Proceedings of the National Academy of Sciences of the United States of America 107 (30) (2010) 13426–13431.
- [110] E. De Carlo, S. Baiguera, M.T. Conconi, S. Vigolo, C. Grandi, S. Lora, C. Martini, P. Maffei, G. Tamagno, R. Vettor, et al., Pancreatic acellular matrix supports islet survival and function in a synthetic tubular device: in vitro and in vivo studies, International Journal of Molecular Medicine 25 (2) (2010) 195–202.
- [111] S.H. Mirmalek-Sani, G. Orlando, J. McQuilling, R. Pareta, D. Mack, M. Salvatori, A.C. Farney, R.J. Stratta, A. Atala, E.C. Opara, S. Soker, Porcine pancreas extracellular matrix as a platform for endocrine pancreas bioengineering, Biomaterials 34 (22) (2013) 5488–5495.
- [112] S.K. Goh, S. Bertera, P. Olsen, J. Candiello, W. Halfter, G. Uechi, M. Balasubramani, S. Johnson, B. Sicari, E. Kollar, S.T. Badylak, I. Banerjee, Perfusion-decellularized pancreas as a natural 3D scaffold for pancreatic tissue and whole organ engineering, Biomaterials 34 (28) (2013) 6760–6772.
- [113] G.Y. Minuk, Hepatic regeneration: if it ain't broke, don't fix it, Canadian Journal of Gastroenterology 17 (2003) 418–424.
- [114] M.A. Kay, N. Fausto, Liver regeneration: prospects for therapy based on new technologies, Molecular Medicine Today 3 (1997) 108–115.
- [115] T. Shupe, M. Williams, A. Brown, B. Willenberg, B.E. Petersen, Method for the decellularization of intact rat liver, Organogenesis 6 (2010) 134–136.

- [116] A. Soto-Gutierrez, L. Zhang, C. Medberry, K. Fukumitsu, D. Faulk, H. Jiang, J. Reing, R. Gramignoli, J. Komori, M. Ross, et al., A whole-organ regenerative medicine approach for liver replacement, Tissue Engineering. Part C: Methods 17 (2011) 677–686.
- [117] B.E. Uygun, A. Soto-Gutierrez, H. Yagi, M.L. Izamis, M.A. Guzzardi, C. Shulman, J. Milwid, N. Kobayashi, A. Tilles, F. Berthiaume, M. Hertl, Y. Nahmias, M.L. Yarmush, K. Uygun, Organ reengineering through development of a transplantable recellularized liver graft using decellularized liver matrix, Nature Medicine 16 (2010) 814–820.
- [118] H. Yagi, K. Fukumitsu, K. Fukuda, et al., Human-scale whole-organ bioengineering for liver transplantation: a regenerative medicine approach, Cell Transplantation 22 (2) (2013) 231–242.
- [119] Y. Wang, C.B. Cui, M. Yamauchi, P. Miguez, M. Roach, R. Malavarca, M.J. Costello, V. Cardinale, E. Wauthier, C. Barbier, D.A. Gerber, D. Alvaro, L.M. Reid, Lineage restriction of human hepatic stem cells to mature fates is made efficient by tissue-specific biomatrix scaffolds, Hepatology 53 (2011) 293–305.
- [120] K. Linke, J. Schanz, J. Hansmann, T. Walles, H. Brunner, H. Mertsching, Engineered liver-like tissue on a capillarized matrix for applied research, Tissue Engineering 13 (2007) 2699–2707.
- [121] E. Schmelzer, E. Wauthier, L.M. Reid, The phenotypes of pluripotent human hepatic progenitors, Stem Cells 24 (2006) 1852–1858.
- [122] S. Sugimoto, K. Harada, T. Shiotani, S. Ikeda, N. Katsura, I. Ikai, T. Mizuguchi, K. Hirata, Y. Yamaoka, T. Mitaka, Hepatic organoid formation in collagen sponge of cells isolated from human liver tissues, Tissue Engineering 11 (2005) 626–633.
- [123] L. Zhang, N. Theise, M. Chua, L.M. Reid, The stem cell niche of human livers: symmetry between development and regeneration, Hepatology 48 (2008) 1598–1607.
- [124] S. Terai, T. Takami, N. Yamamoto, et al., Status and prospects of liver cirrhosis treatment by using bone marrow-derived cells and mesenchymal cells, Tissue Engineering Part B Reviews 20 (3) (2014) 206–210.
- [125] X.N. Pan, L.Q. Zheng, X.H. Lai, Bone marrow-derived mesenchymal stem cell therapy for decompensated liver cirrhosis: a meta-analysis, World Journal of Gastroenterology 20 (38) (2014) 14051–14057.
- [126] F. Liu, X. Pan, G. Chen, et al., Hematopoietic stem cells mobilized by granulocyte colonystimulating factor partly contribute to liver graft regeneration after partial orthotopic liver transplantation, Liver Transplantation 12 (7) (2006) 1129–1137.
- [127] X.Z. Duan, F.F. Liu, J.J. Tong, et al., Granulocyte-colony stimulating factor therapy improves survival in patients with hepatitis B virus-associated acute-on-chronic liver failure, World Journal of Gastroenterology 19 (7) (2013) 1104–1110.
- [128] E. Tsolaki, E. Athanasiou, E. Gounari, et al., Hematopoietic stem cells and liver regeneration: differentially acting hematopoietic stem cell mobilization agents reverse induced chronic liver injury, Blood Cells, Molecules, and Diseases 53 (3) (2014) 124–132.
- [129] I. Aurich, L.P. Mueller, H. Aurich, J. Luetzkendorf, K. Tisljar, M.M. Dollinger, W. Schormann, J. Walldorf, J.G. Hengstler, W.E. Fleig, B. Christ, Functional integration of hepatocytes derived from human mesenchymal stem cells into mouse livers, Gut 56 (2007) 405–415.
- [130] J. Chamberlain, T. Yamagami, E. Colletti, N.D. Theise, J. Desai, A. Frias, J. Pixley, E.D. Zanjani, C.D. Porada, G. Almeida-Porada, Efficient generation of human hepatocytes by the intrahepatic delivery of clonal human mesenchymal stem cells in fetal sheep, Hepatology 46 (2007) 1935–1945.
- [131] G. Almeida-Porada, C.D. Porada, J. Chamberlain, A. Torabi, E.D. Zanjani, Formation of human hepatocytes by human hematopoietic stem cells in sheep, Blood 104 (2004) 2582–2590.
- [132] E. Lagasse, H. Connors, M. Al-Dhalimy, M. Reitsma, M. Dohse, L. Osborne, X. Wang, M. Finegold, I.L. Weissman, M. Grompe, Purified hematopoietic stem cells can differentiate into hepatocytes in vivo, Nature Medicine 6 (2000) 1229–1234.

- [133] D.S. Krause, N.D. Theise, M.I. Collector, O. Henegariu, S. Hwang, R. Gardner, S. Neutzel, S.J. Sharkis, Multi-organ, multi-lineage engraftment by a single bone marrow-derived stem cell, Cell 105 (2001) 369–377.
- [134] S. Sakamoto, A. Yachi, T. Anzai, T. Wada, AFP-producing cells in hepatitis and in liver cirrhosis, Annals of the New York Academy of Sciences 259 (1975) 253–258.
- [135] P. Zhou, N. Lessa, D.C. Estrada, et al., Decellularized liver matrix as a carrier for the transplantation of human fetal and primary hepatocytes in mice, Liver Transplantation 17 (4) (2011) 418–427.
- [136] M. Moini, M.L. Schilsky, E.M. Tichy, Review on immunosuppression in liver transplantation, World Journal of Hepatology 7 (2015) 1355–1368.
- [137] T.M. Manzia, R. Angelico, L. Baiocchi, L. Toti, P. Ciano, G. Palmieri, M. Angelico, G. Orlando, G. Tisone, The Tor Vergata weaning of immunosuppression protocols in stable hepatitis C virus liver transplant patients: the 10-year follow-up, Transplant International 26 (2013) 259–266.
- [138] G. Orlando, S. Soker, K. Wood, Operational tolerance after liver transplantation, Journal of Hepatology 50 (2009) 1247–1257.
- [139] J.M. Smith, M.A. Skeans, S.P. Horslen, et al., OPTN/SRTR 2013 annual Data Report: intestine, American Journal of Transplantation 15 (Suppl. 2) (2015) 1–16.
- [140] R. Kandaswamy, M.A. Skeans, S.K. Gustafson, et al., OPTN/SRTR 2013 annual Data Report: pancreas, American Journal of Transplantation 15 (Suppl. 2) (2015) 1–20.
- [141] M. Valapour, M.A. Skeans, B.M. Heubner, J.M. Smith, M.I. Hertz, L.B. Edwards, W.S. Cherikh, E.R. Callahan, J.J. Snyder, A.K. Israni, et al., OPTN/SRTR 2013 annual Data Report: lung, American Journal of Transplantation 15 (2015) 1–28.
- [142] M. Colvin-Adams, J.M. Smith, B.M. Heubner, M.A. Skeans, L.B. Edwards, C.D. Waller, E.R. Callahan, J.J. Snyder, A.K. Israni, B.L. Kasiske, OPTN/SRTR 2013 annual Data Report: heart, American Journal of Transplantation 15 (2015) 1–28.
- [143] W.R. Kim, J.R. Lake, J.M. Smith, M.A. Skeans, D.P. Schladt, E.B. Edwards, A.M. Harper, J.L. Wainright, J.J. Snyder, A.K. Israni, et al., OPTN/SRTR 2013 annual Data Report: liver, American Journal of Transplantation 15 (2015) 1–28.
- [144] A.J. Matas, J.M. Smith, M.A. Skeans, et al., OPTN/SRTR 2013 annual Data Report: kidney, American Journal of Transplantation 15 (Suppl. 2) (2015) 1–34.
- [145] D.J.M.M. Boots, M.H.L. Christiaans, J.P. van Hooff, Effect of immunosuppressive agents on longterm survival of renal transplant recipients, Drugs 64 (2012) 2047–2073.
- [146] M. Gorgan, B. Bockorny, M. Lawlor, J. Volpe, M. Fiel-Gan, Pulmonary hemorrhage with capillaritis secondary to mycophenolate mofetil in a heart-transplant patient, Archives of Pathology & Laboratory Medicine 137 (2013) 1684–1687.
- [147] R.W. Evans, W.H. Applegate, D.M. Briscoe, et al., Cost-related immunosuppressive medication nonadherence among kidney transplant recipients, Clinical Journal of the American Society of Nephrology 5 (12) (2010) 2323–2328.
- [148] F. Fändrich, Tolerance in clinical transplantation: progress, challenge or just a dream? Langenbeck's Archives of Surgery 396 (2011) 475–487.
- [149] E. Day, P.K. Kearns, C.J. Taylor, J.A. Bradley, Transplantation between monozygotic twins: how identical are they? Transplantation 98 (2014) 485–489.
- [150] C.L. Ross, C. Booth, B. Sanders, et al., Regeneration and bioengineering of transplantable abdominal organs: current status and future challenges, Expert Opinion on Biological Therapy 13 (1) (2013) 103–113.
- [151] G. Niu, E. Sapoznik, S. Soker, Bioengineered blood vessels, Expert Opinion on Biological Therapy 14 (4) (2014) 403–410.

# 14

# Mesenchymal Stem Cells Seeded on Biofunctionalized Scaffold for Tissue Engineering

K.A.T. Carvalho

PELÉ PEQUENO PRÍNCIPE RESEARCH INSTITUTE & PEQUENO PRINCIPE FACULTY, CURITIBA, BRAZIL

# 14.1 Introduction

Regenerative medicine has been looking for new approaches and among these, biomaterials studies have emerged in recent years and have been developing and manipulating molecules, cells, tissues, and organs grown in laboratories to replace human body parts. Many such materials have been used for this purpose, be they synthetic or biological, to provide new medical devices.

In general, engineering technology involves the selection of an appropriate substrate to sustain and promote the growth of a particular injured tissue. Tissue engineering technology offers enhanced recovery of injured tissues and organs.

Previous works have demonstrated that pluripotent stem cells can be generated from somatic cells as mesenchymal stem cells (MSCs) from diverse origins. In addition, they can be induced from somatic cells known as induced pluripotent stem cells (iPSCs), via overexpression of key transcription factors, creating a new cell reprogramming method [1-3]. In addition, the substrate is key for the cells' proliferation and plays an important role in this way.

Three-dimensional (3D) substrates can be prepared based on the tissue of interest. Several types of scaffolds are known: nondegradable synthetic polymers; degradable synthetic polymers; nonhuman collagen gel, without pores; nonhuman collagen gel, with human collagen tissue pores; and decellularized tissue, for example, amniotic or cadaverous tissue [4,5].

The cell microenvironment is mainly composed of soluble growth factors, cellmatrix interactions, and cell-cell interactions. In natural tissues, the extracellular matrix composition, cell density, and physiological properties are often nonhomogeneous. The great challenge of tissue engineering is the distribution of cells throughout tissueengineered scaffolds as a biomimetic material that is as close as possible to the natural tissue, and it should be analogous to the extracellular matrix (ECM), providing the same function as the tissue [6].

Cell deposition is mainly determined by local wall shear stress, which, in turn, is strongly influenced by the architecture of the scaffold pore network. The deposition rate of suspended cells on a surface depends on many parameters such as advective transport and diffusivity, colloidal interactions, concentration of ligands and receptor binding strength, bond-forming kinetics, and available surface area [7–11].

## 14.2 Pluripotency of Stem Cells

Extensive clinical experiments with stem cells, particularly in the treatment of oncohematological diseases, opened up the possibility of trying stem cells to treat nonhematopoietic diseases. MSCs subsequently emerged as a promising source for the regeneration and repair of various tissues in the treatment of a range of diseases, because of their presence in all derived mesenchymal tissues in adult solid organs as well as in mesoderm from embryonic tissue. This, along with their pluripotency and the fact that they were easily obtainable, meant that MSCs represented an important source for studies in regenerative medicine [12].

Pluripotency is defined as the ability of a cell to differentiate into any of three germ layers. Pluripotency is a natural property of MSCs from both adult and embryonic tissues and of others that have been shown to be capable of differentiating into various types of mesoderm-derived cells as well as ectoderm cells, such as skeletal muscle cells, neurons, cardiomyocytes, osteocytes, chondrocytes, and endoderm cells. Nevertheless, the differentiation could be (1) spontaneous, (2) in the niche where the cells were transplanted into the tissue, or (3) induced by the medium conditions, the substrate, or gene transfection, as known for iPSCs. As MSCs are the majority of isolated cells and their expansion will be carried out under good manufacturing practice (GMP) conditions for the development of cellular therapy, this chapter emphasizes their characteristics, potential, and details, for example, the neurospheres (precursor of neurons) obtained by substrate interactions on a biofunctionalized scaffold.

#### 14.2.1 Mesenchymal Stem Cells

Originating from the embryonic mesoderm, MSCs persist in adult organisms, promoting the maintenance and replacement of various cell types that regenerate, particularly where the tissue was injured [13-20].

These cells were isolated ex vivo first by Friedenstein et al. [21] from the bone marrow of *Rattus norvegicus*. The authors reported having obtained a very abundant cell fraction that was not phagocytic and not hematopoietic, with a fibroblastoid aspect and the property of adhesion to plastic (the substrate) [21,22]. The last property is a characteristic that was described for the first time and that usually permits their isolation.

In subsequent years, the same group also showed that this cell fraction showed in vitro clonogenic and differentiation ability to form structures compatible with the bone,

cartilage, fat, muscle, connective, and fibrous aspects of tissue [23,24]. Only in the early 1990s was the term "mesenchymal stem cell" used to designate this cell fraction [25].

In the following 2 decades, MSCs became significantly more frequent in publications on stem cells. Showing the possibility of their isolation from a diverse number of tissues other than the bone marrow, such as placenta, umbilical cord blood, Wharton's jelly, amniotic fluid, amnion, fallopian tubes, peripheral venous blood, adipose tissue, liver, thymus, spleen, parathyroid, pancreas, lung, dental pulp, periosteum, dermis, and skeletal muscle [20,26-35].

The diversity of potential sources for the isolation of MSCs showed significant differences between these cells. Of the reported differences, we highlight those involving immunophenotypical patterns, proliferative capacity, and/or in vitro differentiation potential and the efficiency in therapeutic applications [34,36–44].

The need for standardization has been established and motivated the International Society for Cellular Therapy to set forth minimum criteria for the characterization of MSCs. These are:

- 1. adherence to plastic under conventional cultivation conditions;
- 2. the cell surface proteins expressed by MSCs include cluster of differentiation (CD): CD105, CD73, CD44, CD90, CD71, CD106, CD166, CD29 and Stro-1 (≥95%) combined with no expression of CD45, CD14 or CD11. They also do not express co-stimulatory molecules CD80, CD86 or CD40 or adhesion molecules CD31, CD18 or CD56 or major histocompatibility class II HLA-DR (≤2% positive).
- **3.** demonstration of capacity to differentiate into osteoblasts, chondroblasts, and adipocytes in vitro [45,46].

Despite the many sources of MSCs already described, however, two tissues are used in the majority of MSC reports: bone marrow and adipose tissue. The volume of publications on bone marrow-derived MSCs (BMMSCs) is justified both because this tissue is the one from which MSCs were isolated for the first time and because of the improvement in hematological disease treatments [19–21,47]. Regarding adipose tissue, it was described as a source of MSCs [48,49]. Reports on adipose-derived MSCs (AMSCs) have become as common as the current BMMSC publications. Some authors have pointed them out as an ideal source for the isolation of MSCs. As an easy way to obtain a greater volume of tissue compared to other adult tissues, the procedure is less invasive and can be done under local anesthesia and, therefore, it results in less discomfort to the donor, as for example, liposuction in aesthetic procedures that can then be donated for the research [20,43,50–53].

#### 14.2.2 Differences Between AMSCs and BMMSCs

Several studies comparing AMSCs and BMMSCs have been published, containing a great diversity of results, not infrequently conflicting. There are reports that AMSCs can be maintained for longer growing periods, with lower senescence rate, higher proliferation rate [39,41], increased differentiation potential, and greater stability than the cytogenetic BMMSCs [39].

Concerning the potential for differentiation, there are reports with differences for both of them. De Toni et al. [54], for example, reported higher rates of hematopoietic differentiation in vitro and in vivo from AMSCs than from BMMSCs. AMSCs also have been characterized as having the greatest potential for adipogenic differentiation and lower potential for osteogenic and chondrogenic differentiation than BMMSCs. However, the adipogenic, osteogenic, and chondrogenic potentials showed no difference in several reports. On the other hand, Zhu et al. [58] and Lindroos et al. [59] reported the occurrence of an influence of the donor's gender and age on the AMSC potential, but not on that of BMMSCs. Samples from male donors showed greater potential for osteogenic differentiation than those from female donors, which also showed a significant reduction in differentiation potential with increasing age of the donor, which was not observed in male donor cells [41,43,53–59].

Another distinction between AMSCs and BMMSCs is the immune phenotypic profile. In this context, there are also conflicting results and the markers CD34, CD49d, CD49f, CD54, CD106, and Stro-1 were the most frequently identified as differential between these MSCs [20].

The proteomic and transcriptomic profiles of AMSCs and BMMSCs present less controversial results, but need more study. There are some reports on the occurrence of different expression of approximately 10% of the genes. However, only 3.4% can be considered AMSC or BMMSC specific [39,43,60]. Such divergences are probably due to the differences between immunomodulatory properties [61].

MSC intrinsic properties have been described as molecular, immunocytochemical, and those having to do with isolation, expansion, differentiation, and cryopreservation. The fact that MSCs do not express the antigens of histocompatibility means that they can be used in allogeneic transplantation. Methods have been developed to isolate sub-populations of MSC fractions together by magnetic separation with the perspective that this subpopulation could obtain better results than the total population in replacement therapy for correction of a determined specific function. Magnetic sorting using magnetic beads has become a routine methodology for the separation of key cell populations from biological suspensions. Owing to the inherent ability of magnets to provide forces at a distance, magnetic cell manipulation is now a standardized process step in numerous processes in tissue engineering [62].

#### 14.2.3 Mesenchymal Stem Cells in Cellular Therapy

The therapeutic effect and safe use of MSCs in the treatment of various diseases have also been extensively investigated, also showing a great diversity of organisms, models, and correlations between the target disease and its morbidity and/or mortality [20,63-65].

Similarly, the variety of diseases is as broad as that of BMMSC studies, including clinical conditions with relatively low morbidity, such as allergic rhinitis [66] and tendon injuries [67], to conditions with high morbidity and/or mortality, as in stroke [42] and spinal cord injury [68].

As of this writing, BMMSC and AMSC studies are in the clinical phase. The possibility of delivering, in certain diseases, cells that are committed to a specific type of cell in precursor form and maintaining their proliferation rate could provide transplanted cells with better integration into the host tissue, for example, neuron precursors.

#### 14.2.4 Mesenchymal Stem Cell Expansion

Cell culture is a prerequisite for almost all of the potential MSC therapeutic applications. This is because even in tissue with high yield, for example, in the case of adipose tissue, the amount of AMSCs that can be isolated is usually insufficient to promote the desired therapeutic effect. In addition, there are factors that can provoke even smaller yield, such as the particular physiological condition of the donor organism, the presence of other disorders, or age [52,64,69–75].

The aim of MSC cultivation to increase their number is justified by the variability of cells used to start cultivation. For this purpose, the cells to be cultivated are arranged in a medium that attempts to mimic physiological conditions of the tissue and/or organism of origin. In this context, the incomplete knowledge of both the physiological conditions of origin, such as the difficulty in mimicking known specific conditions of, for example, 3D tissue architecture, and the cell–substrate interactions is a potentially interfering factor in cultivation [76].

The MSC cultivation step is required in studies aimed at the characterization of their potential for self-renewal and/or differentiation in vitro. Theoretically, because these potentials are appropriately known, it is possible to obtain both a high amount of MSCs and their differentiation into any tissue. In this context, where different sources of MSC are described and different methods of isolation and culture, the results are without being standardized [73–84].

Given this diversity, it will be necessary to apply principles of GMP in all stages of the study, from obtaining the sample to the evaluation of transplants performed. This makes it mandatory to keep detailed documentation of the parameters used in cell cultures, especially in cases in which changes are identified [46,85].

There are some reports on genetic modifications of MSCs in culture. These range from genetic alterations, such as amplification or deletion loci and changes in methylation pattern, structure, and/or number of chromosomes, to phenotypic changes, such as the expression of enzymes and/or CDs in the proliferative profile, differentiation potential, and/or even the morphology. Identifying these changes in MSCs, especially those intended for therapeutic applications, reveals the existence of a risk not yet properly quantified as well as a risk of transplanting cells with tumor potential. This risk is established by several changes reported in MSCs that were originally characterized as typical tumor cell markers [73–83].

#### 14.2.5 induced Pluripotent Stem Cells

The iPSC was described from mouse adult fibroblasts that could be reprogrammed with all embryonic stem cell (ESC) characteristics by the introduction of four transcription

factors known to be expressed in ESCs (*Oct4, Sox2, Klf4,* and *c-Myc*) and selecting cells that expressed endogenous *Nanog* or *Oct4* [86].

Pluripotency is generated naturally during mammalian development through the formation of the epiblast, the founder tissue of the embryo. In the embryo, *Nanog* specifically demarcates the nascent epiblast, coincident with the domain of X-chromosome reprogramming. Without *Nanog*, pluripotency does not develop, and the inner cell mass is trapped in a prepluripotent, indeterminate state that is ultimately nonviable. The acquisition of *Nanog* is mandatory to gain full reprogramming of pluripotent cells. Otherwise, the somatic cells could be transfected with embryonic genes and then regain their pluripotency [87].

#### 14.3 Substrate

The successful growth and maintenance of cells in vitro requires culture conditions that mimic in vivo conditions with respect to temperature, oxygen, carbon dioxide concentration, pH, osmolality, and nutrition. The tissue culture medium, the basal medium, provides these conditions as well as the growth factors and induced factors to differentiate the cells. Some cells require supplementation and serum to be added to the basal medium.

Serum is an effective growth-promoting supplement for practically all types of cells. These can be divided into specific polypeptides that stimulate cell growth (growth factors), carrier proteins, antioxidant agents, cell attachment factors, and nutrients as trace elements. Regarding the relation of cell to substrate, attachment factors guarantee anchorage to the substrate. With the development of cellular therapy, which needs to attempt GMP conditions, the use of a medium without serum is mandatory and justified as a way to avoid contamination by viruses and prions and the need to purify the final product of contaminating proteins as well as the great variability in serum components causing nonreproducible results [88].

In experimental cell biology, it is important to be aware of the inherent variability of serum, which renders it very difficult to study the specific effects of molecules such as growth factors, hormones, cytokines, adhesion molecules, or matrix components, all of which are present in serum at undefined and variable levels.

In this way, to mimic the natural conditions in ECM tissue, the substrate has an important role in culture for replacement of serum-containing medium. For these reasons, the matrix should be investigated for adhesion and growth properties of its integrated molecules as well as the functionality that promotes 2D or 3D architectures as in natural tissue. There is increased interest in developing scaffolds in 2D or 3D, while the monolayer in flask culture is losing its use.

Diverse substrate sources have now been developed in intention on promoting cell adhesion, proliferation, and induction to a differentiated state in undifferentiated MSCs for cardiomyospheres, myospheres, and neurospheres to obtain the precursos of cardiomyocytes, muscle cells, and neurons, respectively (Fig. 14.1). Each substrate could



FIGURE 14.1 Myospheres from primary cultivation (rat). Inverted light microscopy, original magnification 100×.

induce different cell types. For example, collagen could be directed to epithelial cells as used to develop vessels and heart devices [12,13,89–98].

The substrate's functional properties depend on its nature and on its capacity to interact with diverse molecules or macromolecules of varying polarity. The cell–substrate interactions present different adhesive behaviors and could be selective regarding the induction medium toward certain types of cells. Cells are polyanionic surfaces that interact with their opposite, polycationic surfaces. The common polystyrene used in flasks that is treated with some molecules, for example, lysine, a polycationic layer of adsorbed polylysine, or one with altered electrostatic charge, has been shown to be able to modify the adhesion proprieties [99,100].

Cells of many kinds adhere firmly to glass or plastic surfaces that have been pretreated with polylysine. The attachment takes place as soon as the cells make contact with the surface, and the flattening of the cells against the surface is quite fast. Cells that do not normally adhere to solid surfaces, such as some suspension mammalian cells in culture, could be attached to such surfaces. The attachment of cells to polylysine-treated surfaces can be exploited for a variety of experimental manipulations. In the preparation of samples for scanning or transmission electron microscopy, the living material may first be attached to a polylysine-coated plate and subjected to some experimental treatment (fertilization of an egg, for example), then transferred rapidly to a fixative, and further passed through processing for observation. Each step involves only the transfer of the plate from one container to the next. The cells are not detached [101].

Particle surface modification through poloxamer adsorption can significantly alter the charge, adhesion, and consequently handling properties of a material. The charge reduction on polystyrene spheres achieved by this modification technique is dependent on the concentration, molecular weight, and conformation of poloxamer at the particle surface. Adsorption isotherms of poloxamers on polystyrene particles follow a Langmuir profile and there is an apparent correlation between the extent of adsorption and the poloxamer ability to reduce electrostatic charge [101].

In this way, the property of the attached fibroblasts could be used in replanting methods to obtain purity in a primary culture derived from skeletal muscles. The primary culture derived from skeletal muscles consists mainly of myoblasts and fibroblasts and the purification of myoblast cells from such a mixed population can be achieved. During the initial week or two of culture, selective growth and passaging conditions allow the myoblasts to become the dominant cell type, and after 3 weeks in culture, nearly 100% of the cells stain positive for the myoblast-specific proteins desmin or fast myosin [88,91].

#### 14.4 Biomaterials

Biomaterials from plants and microorganisms such as bacteria and fungi could be a good choice for developings scaffolds, and the biopolymers were tried at first time and demonstrated successful results; these are explained by their versatility. Investigators have been looking for biomaterials apart from animal origin. Otherwise, they might incur in the same risks of medium supplementation with serum, with the virus and prion contaminations. These biomaterials needs to be xeno—free (without animal—derived components).

#### 14.4.1 Biopolymers

The structures and functionalities of polysaccharides, proteins, and lipids allow their utilization with this aim in biomimetic and nanotechnology systems. Cellulose is one of the most studied biopolymers, and several sources of this polymer have been evaluated.

#### 14.4.1.1 Cellulose

Cellulose is a linear polysaccharide consisting of monomer units of  $\beta(1 \rightarrow 4)$ -D-glucopyranose; it presents long-chain variable molecular weight, with the empirical formula  $(C_6H_{10}O_5)_n$ , with *n* having a minimum value of 200. Total hydrolysis of cellulose produces glucose molecules, whereas partial hydrolysis produces cellobiose molecules. Some algae, fungi, and bacteria also produce this biopolymer, which comprises the

structural basis of the plant cell wall. The cellulose present in plants has the same chemical structure as the cellulose produced by bacteria, but there are morphological distinctions and they exhibit different physical and mechanical properties [102,103].

The synthesis of cellulose produced by *Gluconacetobacter xylinus* is a complex process and involves a glucose polymerization unit of a linear chain of  $\beta(1 \rightarrow 4)$  glucans, the extracellular secretion of the chain, and the assembly and crystallization of these chains as knitted, resulting in a 3D gelatinous structure formed on the surface of a liquid medium. The physical and mechanical characteristics of the bacterial cellulose membrane arise from a single structure that differs from the plant pulp. The bacterial cellulose nanofibrils create a large surface area that allows the absorption of a large amount of water. The hydrogen bonds between the fibrillar units stabilize the entire structure, acquiring high mechanical strength, which results in a hard network structure. This linearity yields multiple elementary fibrils that aggregate into larger bundles, which can contain crystalline and amorphous regions. The degrees of crystallinity and crystal dimensions are dependent on the origin of the cellulose or on the modification to which it was submitted [104–106].

Specifically, cellulose from bacterial sources exhibits higher crystallinity and has distinct advantages over cellulose from other sources. The advantages include high purity, without the need to extract lignin or hemicelluloses, as well as high surface area [104].

The cellulose membrane has several characteristics that are desired in an ideal dressing for wounds, among which we can mention:

- **1.** the presence of nanopores, which do not allow the entrance of microorganisms to the wound bed, but allow high throughput of water vapors;
- 2. ease of the sterilization process;
- 3. low production cost;
- 4. ability to shape the surfaces of the wound and elasticity [107].

Several studies are under development to improve bacterial cellulose membranes. Despite these membranes creating a barrier over the wound against microorganisms, this polymer shows no antimicrobial activity. The insertion of benzalkonium chloride, silver and copper nanoparticles, zinc oxide, and other bactericidal substances reduces the risk of infection [108–110]. The incorporation into these membranes of drugs such as propranolol, ibuprofen, and lidocaine allows the composition of a transdermal drug delivery, controlled by micropores formed by the mesh and nanostructured bacterial cellulose [111].

The fabrication of nanostructured membranes allows control of the thickness of the fibers that form the mesh and the properties at the molecular level of the original material. The large surface of this material due to volume increases its potential for interaction, creating functional surfaces. Sublimation (chemical vapor deposition), the sol-gel spin, self-assembly (layer by layer), the Langmuir–Blodgett technique, and electrospinning are some techniques for manufacturing nanostructured film [112,113].

The aforementioned characteristics have increased interest in producing polymer nanofibers, which began in the 1990s, bringing the possibility of producing 3D structures formed by nanonetworks and pores of varying sizes, serving as supports for tissue engineering. This approach is based on cell—substrate interactions. The identification of an optimal scaffold for cell implantation is a further requirement [114].

# 14.5 Scaffold Interaction With Mesenchymal Stem Cells

Curiously, the biofunctionalized scaffolds that have been seeded with MSCs interact with the cells in ways related not only to their charge but also to some unknown mechanisms. This may be due to the complexity of those cells that have pluripotency properties allied with their particular surface proteins, in particular adhesion molecules, for example, the endoglin known as CD105. On the other hand, scaffolds based on biopolymers have molecules that mimic the integrins (adhesion molecules). The MSC–substrate interaction plays a role in gene expression and gene suppression of their transcription factors. These facts could be inferred by the MSC differentiation state reported between different biopolymers. AMSCs seeded on cellulose have been maintained in the undifferentiated state, whereas AMSCs seeded on other biopolymers extracted from a Brazilian plant (protected for patent development) have been induced to neuron precursors, which express nestin, a biomarker for neurons, as well as CD271 (not published). The consequences of the differences in MSC–substrate interactions determine the purpose and potential.

#### 14.5.1 Scaffold to Cellular Therapy

With this aim, Souza et al. [115] developed studies on both tissue engineering and delivery of cells for cellular therapy based on biofunctionalized scaffolds. The nanocomposites used were obtained after defibrillation using never-dried bacterial cellulose to form a pulp and to biofunctionalize the substrate, whose films were regenerated in the presence of hydrocolloid (10% w/w) with gellan gum, tamarind xyloglucan, and lysozyme. Afterward, this membrane was seeded with AMSCs and used as a curative for burned skin. The surface properties of this developed scaffold provided conditions for the adhesion and proliferation of AMSCs. It was demonstrated that the presence of stem cells may have accelerated the healing process and reduced the inflammatory infiltrate in the tissue. The nanostructured membranes seeded with MSCs were demonstrated to be efficient living bioactive dressings, effective for tissue regeneration (Figs. 14.2 and 14.3) [115,106].

#### 14.5.2 Scaffold to Reprogram Mesenchymal Stem Cells to Neurons

In addition, a substrate as a functionalized scaffold biopolymer could reprogram MSCs based on its cell–surface interaction; neurospheres are an example. Neurospheres are agglomerates of precursors of neurons arranged as shaped spheres marked with the



**FIGURE 14.2** Cellulose in the presence of hydrocolloid (10% w/w) with gellan gum, tamarind xyloglucan, and lysozyme.



FIGURE 14.3 Nanostructured membrane reconstituted with cellulose, gellan gum, and lysozyme, seeded with mesenchymal stem cells. Electron microscopy, original magnification  $1000 \times$ .

precursor neuron protein nestin and developed from undifferentiated MSCs seeded on the scaffold. The precursors of neurons could continue to differentiate in vitro or in vivo. In vivo, the neuron precursors may replace the neurons and repair the tissue to restore function. More preclinical studies are still needed to evaluate their effectiveness in cellular therapy. The mechanisms of this reprogramming are unknown, including its gene expression and metabolomics. These mechanisms should be understood. The

FIGURE 14.4 Neurospheres from human adipose mesenchymal stem cells seeded on biopolymer. Inverted light microscopy, original magnification  $100 \times$ .



manufacture of cellular products obtained through the interaction of MSCs and a biofunctionalized scaffold requires further studies aimed at standardization of GMP conditions to be safe [95,96] (Fig. 14.4).

# 14.6 Conclusions

The versatility and variety of substrates as biofunctionalized scaffolds should be explored for the development of new approaches in tissue engineering based on their relations with stem cells and particularly MSCs.

# Acknowledgments

I would like to thank Ana Carolina Irioda, MsC; Vanessa Ferreira de Mello dos Santos, postgraduate student; Carolina Maria Costa de Oliveira Souza, PhD; and Professor Maria Rita Sierakowski, PhD, for permitting use of the figures, and particularly Ana Carolina Irioda for assistance with the text.

# References

- [1] C.B. Rippol, M. Flaat, J. Klopf-Einermann, J.M. Fisher-Perkins, C.B. Trygg, B.A. Scruggs, M.L. McCants, H.P. Leonard, A.F. Lin, S. Zhang, M.E. Eagle, X. Alvarez, Y.T. Li, S.C. Li, J.M. Gimble, B.A. Bunnell, Mesenchymal lineage stem cells have pronounced anti-inflammatory effects in the twitcher mouse model of Krabbe's disease, Stem Cells 29 (2011) 67–77.
- [2] I. Linero, O. Chaparro, Paracrine effect of mesenchymal stem cells derived from human adipose tissue in bone regeneration, PLoS One 9 (2014) e107001.
- [3] J. Dittmer, B. Leyh, Paracrine effects of stem cells in wound healing and cancer progression, International Journal of Oncology 44 (2014) 1789–1798.

- [4] J.C. Francisco, R.C. Cunha, R.B. Simeoni, L.C. Guarita-Souza, R.J. Ferreira, A.C. Irioda, C.O.S. Souza, G.V.N. Srikanth, S. Nityanand, J.C. Chachques, K.T.C. Carvalho, Amniotic membrane as a potent source of stem cells and a matrix for engineering heart tissue, Journal Biomedical Science and Engineering 6 (2013) 1178–1185.
- [5] A. Lindenmair, S. Wolbank, G. Stadler, A. Meinl, A. Peterbauer-Scherb, J. Eibl, H. Polin, C. Gabriel, M. Griensven, H. Redl, Osteogenic differentiation of intact human amniotic membrane, Biomaterials 31 (2010) 8659–8665.
- [6] F.P.W. Melchels, B. Tonnarelli, A.L. Olivares, I. Martin, D. Lacroix, J. Feijen, J.D. Wendt, D.W. Grijpma, The influence of the scaffold design on the distribution of adhering cells after perfusion cell seeding, Biomaterials 32 (2011) 2878–2884.
- [7] Z. Adamczyk, T.G.M. Vandeven, Deposition of particles under external forces in laminar-flow through parallel-plate and cylindrical channels, Journal of Colloid and Interface Science 80 (1981) 340–356.
- [8] J. Sjollema, H.J. Busscher, Deposition of polystyrene latex-particles towardpolymethylmethacrylate in a parallel plate flow cell, Journal Colloid Interface Science 132 (1989) 382–394.
- [9] C. Cozensroberts, J.A. Quinn, D.A. Lauffenburger, Receptor-mediated adhesion phenomena model studies with the radial-flow detachment assay, Biophysical Journal 58 (1990) 107–125.
- [10] D.A. Hammer, D.A. Lauffenburger, A dynamic-model for receptor-mediated cell-adhesion to surfaces, Biophysical Journal 52 (1987) 475–487.
- [11] J.M. Meinders, H.J. Busscher, Adsorption and desorption of colloidal particles on glass in a parallel-plate flow chamber – influence of ionic-strength and shear rate, Colloid and Polymer Science 272 (1994) 478–486.
- [12] K. Takahashi, K. Tanabe, M. Ohnuki, M. Narita, T. Ichisaka, K. Tomoda, S. Yamanaka, Induction of pluripotent stem cells from adult human fibroblasts by defined factors, Cell 131 (2007) 861–872.
- [13] J.S. Odorico, D.S. Kaufman, J.A. Thomson, Multilineage differentiation from human embryonic stem cells lines, Stem Cells 9 (2001) 193–204.
- [14] D.M. Braile, M.F. Godoy, Stem cell thepary. A new perspective for the treatment of ischemic heart failure, Arquivos Brasileiros de Cardiologia 84 (2005) 357–359.
- [15] L. Da Silva Meirelles, P.C. Chagastelles, N.B. Nardi, Mesenchymal stem cells reside in virtually all post-natal organs and tissues, Journal of Cell Science 119 (2006) 2204–2213.
- [16] M. Mimeault, S.K. Batra, Concise review: recent advances on the significance of stem cells in tissue regeneration and cancer therapies, Stem Cell 24 (2006) 2319–2345.
- [17] M. Crisan, S. Yap, L. Casteilla, C.W. Chen, M. Corselli, T.S. Park, G. Andriolo, B. Sun, B. Zheng, L. Zhang, C. Norotte, P.N. Teng, J. Traas, R. Schugar, B.M. Deasy, S. Badylak, H.J. Buhring, J.P. Giacobino, L. Lazzari, J. Huard, B. Péault, A perivascular origin for mesenchymal stem cells in multiple human organs, Cell Stem Cell 3 (2008) 301–313.
- [18] L. Da Silva Meirelles, A.I. Caplan, N.B. Nardi, In search of the in vivo identity of mesenchymal stem cells, Stem Cells 26 (2008) 2287–2299.
- [19] H.K. Salem, C. Thiemermann, Mesenchymal stromal cells: current understanding and clinical status, Stem Cells 28 (2010) 585–596.
- [20] M. Strioga, S. Viswanathan, A. Darinskas, O. Slaby, J. Michalek, Same or not the same? Comparison of adipose tissue-derived versus bone marrow-derived mesenchymal stem and stromal cells, Stem Cells 21 (2012) 2724–2752.
- [21] A.J. Friedenstein, K.V. Petrakova, A.I. Kurolesova, G.P. Frolova, Heterotopic of bone marrow. Analysis of precursor cells for osteogenic and hematopoietic tissues, Transplantation 6 (1968) 230–247.

- [22] M. Karagianni, I. Brinkmann, S. Kinzebach, M. Grass, C. Weiss, P. Bugert, K. Bieback, A comparative analysis of the adipogenic potential in human mesenchymal stromal cells from cord blood and other sources, Cytotherapy 15 (2013) 76–88.
- [23] A.J. Friedenstein, R.K. Chailakhjan, K.S. Lalykina, The development of fibroblast colonies in monolayer cultures of guinea-pig bone marrow and spleen cells, Cell and Tissue Kinetics 3 (1970) 393–403.
- [24] A.J. Friedenstein, U.F. Deriglasova, N.N. Kulagina, A.F. Panasuk, S.F. Rudakowa, E.A. Luriá, I.A. Ruadkow, Precursors for fibroblasts in different populations of hematopoietic cells as detected by the in vitro colony assay method, Experimental Hematology 2 (1974) 83–92.
- [25] A.I. Caplan, Mesenchymal stem cell, Journal of Orthopaedic Research 9 (1991) 641-650.
- [26] C. Bari, F. Dell'Accio, P. Tylzanowski, F.P. Luyten, Multipotent mesenchymal stem cells from adult human synovial membrane, Arthritis & Rheumatis 44 (2001) 1928–1943.
- [27] S. Gronthos, D.M. Franklin, H.A. Leddy, P.G. Robey, R.W. Storms, J.M. Gimble, Surface protein characterization of human adipose tissue-derived stromal cells, Journal of Cellular Physiology 189 (2001) 54–63.
- [28] J.J. Minguell, A. Erices, P. Conget, Mesenchymal stem cells, Experimental Biology and Medicine 226 (2001) 507-520.
- [29] H.E. Young, T.A. Steele, R.A. Bray, J. Hudson, J.A. Floyd, K. Hawkins, K. Thomas, T. Austin, C. Edwards, J. Cuzzourt, M. Duenzl, P.A. Lucas, A.C. Black Jr., Human reserve pluripotent mesenchymal stem cells are present in the connective tissues of skeletal muscle and dermis derived from fetal, adult, and geriatric donors, The Anatomical Record 264 (2001) 51–62.
- [30] Y. Hu, L. Liao, Q. Wang, L. Ma, G. Ma, X. Jiang, R.C. Zhao, Isolation and identification of mesenchymal stem cells from human fetal pancreas, The Journal of Laboratory and Clinical Medicine 141 (2003) 342–349.
- [31] M. Krampera, S. Marconi, A. Pasini, M. Galiè, G. Rigotti, F. Mosna, M. Tinelli, L. Lovato, E. Anghileri, A. Andreini, G. Pizzolo, A. Sbarbati, B. Bonetti, Induction of neural-like differentiation in human mesenchymal stem cells derived from bone marrow, fat, spleen and thymus, Boné 40 (2007) 382–390.
- [32] T. Jazedje, P.M. Perin, C.E. Czeresnia, M. Maluf, S. Halpern, M. Secco, D.F. Bueno, N.M. Vieira, E. Zucconi, M. Zatz, Isolation and identification of mesenchymal stem cells from human fetal pancreas, The Journal of Laboratory and Clinical Medicine 28 (2009) 9–17.
- [33] P.C. Tsai, T.W. Fu, Y.M. Chen, T.L. Ko, T.H. Chen, Y.H. Shih, S.C. Hung, Y.S. Fu, The therapeutic potential of human umbilical mesenchymal stem cells from Wharton's jelly in the treatment of rat liver fibrosis, Liver Transplation 15 (2009) 484–495.
- [34] F. Mosna, L. Sensebé, M. Krampera, Human bone marrow and adipose tissue mesenchymal stem cells: a user's guide, Stem cells and Development 19 (2009) 1449–1470.
- [35] N.F. Lizier, A. Kerkis, C.M. Gomes, J. Hebling, C.F. Oliveira, A.I. Caplan, I. Kerkis, Scaling-up of dental pulp stem cells isolated from multiple niches, PLoS One [electronic resource, e39885.] 7 (2012) 1–12.
- [36] S. Gronthos, J. Brahim, W. Li, L.W. Fisher, N. Cherman, A. Boyde, P. DenBesten, P.G. Robey, S. Shi, Stem cell properties of human dental pulp stem cells, Journal of Dental Research 81 (2002) 531–535.
- [37] R.A. Panepucci, J.L. Siufi, W.A. Silva Jr., R. Proto-Siquiera, L. Neder, M. Orellana, V. Rocha, D.T. Covas, M.A. Zago, Comparison of gene expression of umbilical cord vein and bone marrow-derived mesenchymal stem cells, Stem Cells 22 (2004) 1263–1278.
- [38] F. Festy, L. Hoareau, S. Bes-Houtmann, A.M. Péquin, M.P. Gonthier, A. Munstun, J.J. Hoarau, M. Césari, R. Roche, Surface protein expression between human adipose tissue-derived stromal cells and mature adipocytes, Histochemistry and Cell Biology 124 (2005) 113–121.

- [39] R. Izadpanah, C. Trygg, B. Patel, C. Kriedt, J. Dufour, J.M. Gimble, B.A. Bunnell, Biologic properties of mesenchymal stem cells derived from bone marrow and adipose tissue, Journal of Cellular Biochemistry 99 (2006) 1285–1297.
- [40] R. Izadpanah, D. Kaushal, C. Kriedt, F. Tsien, B. Patel, J. Dufour, B.A. Bunnell, Long-term in vitro expansion alters the biology of adult mesenchymal stem cells, Cancer Research 68 (2008) 4229–4238.
- [41] S. Kern, H. Eichler, J. Stoeve, H. Klüter, K. Bieback, Comparative analysis of mesenchymal stem cells from bone marrow, umbilical cord blood, or adipose tissue, Stem Cells 24 (2006) 1294–1301.
- [42] Y. Kim, H. Kim, H. Cho, Y. Bae, K. Suh, J. Jung, Direct comparison of human mesenchymal stem cells derived from adipose tissues and bone marrow in mediating neovascularization in response to vascular ischemia, Cellular Physiology and Biochemistry 20 (2007) 867–876.
- [43] D. Noël, D. Caton, S. Roche, C. Bony, S. Lehmann, L. Casteilla, C. Jorgensen, B. Cousin, Cell specific differences between human adipose-derived and mesenchymal-stromal cells despite similar differentiation potentials, Experimental Cell Research 314 (2008) 1575–1584.
- [44] R. Gaebel, D. Furlani, H. Sorg, B. Polchow, J. Frank, K. Bieback, W. Wang, C. Klopsch, L.L. Ong, W. Li, N. Ma, G. Steinhoff, Cell origin of human mesenchymal stem cells determines a different healing performance in cardiac regeneration, PLoS One [electronic resource, e15652] 6 (2011) 1–14.
- [45] M. Dominici, K. Le Blanc, I. Mueller, I. Slaper-Cortenbach, F.C. Marini, D.S. Krause, R.J. Deans, A. Keating, D.J. Prockop, E.M. Horwitz, Minimal criteria for defining multipotent mesenchymal stromal cells. The International Society for Cellular Therapy position statement, Cytotherapy 8 (2006) 315–317.
- [46] L. Barkholt, E. Flory, V. Jekerle, S. Lucas-Samuel, P. Ahnert, L. Bisset, D. Büscher, W. Fibbe, A. Foussat, M. Kwa, O. Lantz, R. Mačiulaitis, T. Palomäki, C.K. Schneider, L. Sensebé, G. Tachdjian, K. Tarte, L. Tosca, P. Salmikangas, Risk of tumorigenicity in mesenchymal stromal cell-based therapies-bridging scientific observations and regulatory viewpoints, Cytotherapy 15 (2013) 753–759.
- [47] J. Tolar, K. Le Blanc, A. Keating, B.R. Blazar, Concise review: hitting the right spot with mesenchymal stromal cells, Stem Cells 28 (2010) 1446–1455.
- [48] P.A. Zuk, M. Zhu, H. Mizuno, J. Huang, J.W. Futrell, A.J. Katz, P. Benhaim, H.P. Lorenz, M.H. Hedrick, Multilineage cells from human adipose tissue: implications for cell-based therapies, Tissue Engineering 7 (2001) 211–228.
- [49] P.A. Zuk, M. Zhu, P. Peter Ashjian, D.A. De Ugarte, J.I. Huang, H. Mizuno, Z.C. Alfonso, J.K. Fraser, P. Benhaim, M.H. Hedrick, Human adipose tissue is a source of multipotent stem cells, Molecular Biology of the Cell 13 (2002) 4279–4295.
- [50] A.M. Rodriguez, C. Elabd, E.Z. Amri, G. Ailhaud, C. Dani, The human adipose tissue is a source of multipotent stem cells, Biochimie 87 (2005) 125–128.
- [51] T.E. Meyerrose, D.A. De Ugarte, A.A. Hofling, P.E. Herrbrich, T.D. Cordonnier, L.D. Shultz, J.C. Eagon, L. Wirthlin, M.S. Sands, M.A. Hedrick, J.A. Nolta, In vivo distribution of human adipose-derived mesenchymal stem cells in novel xenotransplantation models, Stem Cells 25 (2007) 220–227.
- [52] B.R. Grimes, C.M. Steiner, S. Merfeld-Clauss, D.O. Trauktuev, D. Smith, A. Reese, A.M. Breman, V. C. Thurston, G.H. Vance, B.H. Johnstone, R.B. Slee, K.L. March, Interphase FISH demonstrates that human adipose stromal cells maintain a high level of genomic stability in long-term culture, Stem Cells and Development 18 (2009) 717–724.
- [53] H. Mizuno, Adipose-derived stem cells for tissue repair and regeneration: ten years of research and a literature review, Journal of Nippon Medical School 76 (2009) 56–66.
- [54] F. De Toni, S. Poglio, A.B. Youcef, B. Cousin, F. Pflumio, P. Bourin, L. Casteilla, P. Laharrague, Human adipose-derived stromal cells efficiently support hematopoiesis in vitro and in vivo: a key step for therapeutic studies, Stem Cells and Development 20 (2011) 2127–2138.

- [55] Y. Sakaguchi, I. Sekiya, K. Yagishita, T. Muneta, Comparison of human stem cells derived from various mesenchymal tissues: superiority of synovium as a cell source, Arthritis and Rheumatism 52 (2005) 2521–2529.
- [56] G. Pachón-Peña, G. Yu, A. Tucker, X. Wu, J. Vendrell, B.A. Bunnell, J.M. Gimble, Stromal stem cells from adipose tissue and bone marrow of age-matched female donors display distinct immunophenotypic profiles, Journal of Cellular Physiology 226 (2011) 843–851.
- [57] I. Bochev, G. Elmadjian, D. Kyurkchiev, L. Tzvetanov, I. Altankova, P. Tivchev, S. Kyurkchiev, Mesenchymal stem cells from human bone marrow or adipose tissue differently modulate mitogen-stimulated B-cell immunoglobulin production in vitro, Cell Biology International 32 (2008) 384–393.
- [58] H. Zhu, X. Song, L.J. Jin, P. Jin, R. Guan, X. Liu, X.Q. Li, Comparison of intra-coronary cell transplantation after myocardial infarction: autologous skeletal myoblasts versus bone marrow mesenchymal stem cells, The Journal of International Medical Research 37 (2009) 298–307.
- [59] B. Lindroos, R. Suuronen, S. Miettinen, The potential of adipose stem cells in regenerative medicine, Stem Cell Reviews and Reports 7 (2011) 269–291.
- [60] H.W. Park, J.S. Shin, C.W. Kim, Proteome of mesenchymal stem cells, Proteomics 7 (2007) 2881–2894.
- [61] E. Ivanova-Todorova, M. Mourdjeva, D. Kyurkchiev, I. Bochev, E. Stoyanova, R. Dimitrov, T. Timeva, M. Yunakova, D. Bukarev, A. Shterev, P. Tivchev, S. Kyurkchiev, HLA-G expression is upregulated by progesterone in mesenchymal stem cells, American Journal of Reproductive Immunology 62 (2009) 25–33.
- [62] B.D. Plouffe, S.K. Murthy, L.H. Lewis, Fundamentals and application of magnetic particles in cell isolation and enrichment: a review, Reports on Progress in Physics 78 (2015) 016601.
- [63] N.P. Bochkov, V.A. Nikitina, O.A. Buyanovskaya, E.S. Voronina, D.V. Goldstein, N.P. Kuleshov, A.A. Rzhaninova, I.N. Chaushev, Aneuploidy of stem cells isolated from human adipose tissue, Bulletin of Experimental Biology and Medicine 146 (2008) 344–347.
- [64] O.A. Buyanovskaya, N.P. Kuleshov, V.A. Nikitina, E.S. Voronina, L.D. Katosova, N.P. Bochkov, Spontaneous aneuploidy and clone formation in adipose tissue stem cells during different periods of culturing, Bulletin of Experimental Biology and Medicine 148 (2009) 109–112.
- [65] J.M. Gimble, F. Guilak, B.A. Bunnell, Clinical and preclinical translation of cell-based therapies using adipose tissue-derived cells, Stem Cell Research & Therapy [electronic resource] 1 (2010) 19.
- [66] K.S. Cho, H.K. Park, H.Y. Park, J.S. Jung, S.G. Jeon, Y.K. Kim, H.J. Roh, IFATS collection: immunomodulatory effects of adipose tissue-derived stem cells in an allergic rhinitis mouse model, Stem Cells 27 (2009) 259–265.
- [67] A.C. Uysal, H. Mizuno, Differentiation of adipose-derived stem cells for tendon repair, Methods in Molecular Biology 702 (2011) 443–451.
- [68] H.H. Ryu, J.H. Lim, Y.E. Byeon, J.R. Park, M.S. Seo, Y.W. Lee, W.H. Kim, K.S. Kang, O.K. Kweon, Functional recovery and neural differentiation after transplantation of allogenic adipose-derived stem cells in a canine model of acute spinal cord injury, Journal of Veterinary Science 10 (2009) 273–284.
- [69] B. Puissant, C. Barreau, P. Bourin, C. Clavel, J. Corre, C. Bousquet, C. Taureau, B. Cousin, M. Abbal, P. Laharrague, L. Penicaud, L. Casteilla, A. Blancher, Immunomodulatory effect of human adipose tissue-derived adult stem cells: comparison with bone marrow mesenchymal stem cells, British Journal of Haematology 129 (2005) 118–129.
- [70] D. Garcia-Olmo, D. Herreros, M. Pascual, I. Pascual, P. De-La-Quintana, J. Trebol, M. Garcia-Arranz, Treatment of enterocutaneous fistula in Crohn's Disease with adipose-derived stem cells: a comparison of protocols with and without cell expansion, International Journal of Colorectal Disease 24 (2009) 27–30.

- [71] A. Spiropoulos, M. Theodosaki, K. Stefanaki, G. Paterakis, M. Tzetis, K. Giannikou, E. Petrakou, M. N. Dimopoulou, I. Papassotiriou, E.S. Roma, E. Kanavakis, S. Graphakos, E. Goussetis, Rapid clinical-scale propagation of mesenchymal stem cells using cultures initiated with immunose-lected bone marrow CD105<sup>+</sup> cells, Journal of Cellular and Molecular Medicine 15 (2011) 1983–1988.
- [72] T.I. van der Spoel, S.J. Jansen of Lorkeers, P. Agostoni, E. van Belle, M. Gyöngyösi, J.P. Sluijter, M.J. Cramer, P.A. Doevendans, S.A. Chamuleau, Human relevance of pre-clinical studies in stem cell therapy: systematic review and meta-analysis of large animal models of ischaemic heart disease, Cardiovascular Research 91 (2011) 649–658.
- [73] M.T. Duailibi, L.D. Kulikowski, S.E. Duailibi, M.V. Lipay, M.I. Melaragno, L.M. Ferreira, J.P. Vacanti, P.C. Yelick, Cytogenetic instability of dental pulp stem cell lines, Journal of Molecular Histology 43 (2012) 89–94.
- [74] A. Bentivegna, M. Miloso, G. Riva, D. Foudah, V. Butta, L. Dalprà, G. Tredici, Dna methylation changes during in vitro propagation of human mesenchymal stem cells: implications for their genomic stability? Stem Cells International [electronic resource] 2013 (2013) 192425.
- [75] M. Roemeling-van Rhijn, M. Khairoun, S.S. Korevaar, L. Lievers, D.G. Leuning, J.N.M. IJzermans, M.G.H. Betjes, P.G. Genever, C. van Kooten, H.J.W. de Fijter, T.J. Rabelink, C.C. Baan, W. Weimar, H. Roelofs, M.J. Hoogduijn, M.E. Reinders, Human bone marrow- and adipose tissue-derived mesenchymal stromal cells are immunosuppressive in vitro and in a humanized allograft rejection model, Journal of Stem Cell & Theraphy 6 (2013) 20780.
- [76] M.L. Alves da Silva, A. Crawford, J.M. Mundy, V.M. Correlo, P. Sol, M. Bhattacharya, P.V. Hatton, R. L. Reis, N.M. Neves, Chitosan/polyester-based scaffolds for cartilage tissue engineering: assessment of extracellular matrix formation, Acta Biomaterialia 6 (2010) 1149–1157.
- [77] A.C. Irioda, L. Zocche, C.M.C.O. Souza, R.J. Ferreira, E. Aliprandini, R.C. Cunha, J.C. Francisco, L.C. Guarita-Souza, M. Malvezzi, M.P. Beltrame, L.A.F. Mesquita, D. Kuczera, J.C. Chachques, K.A.T. Carvalho, Pap test as the first step in sreening genetic stability in cell-based therapy, Journal of Stem Cell Research and Therapy 1 (2011) 1000106.
- [78] J.C. Ra, I.S. Shin, S.H. Kim, S.K. Kang, B.C. Kang, H.Y. Lee, Y.J. Kim, J.Y. Jo, E.J. Yoon, H.J. Choi, E. Kwon, Safety of intravenous infusion of human adipose tissue-derived mesenchymal stem cells in animals and humans, Stem Cells Development 20 (2011) 1297–1308.
- [79] U. Ben-David, Y. Mayshar, N. Benvenisty, Significant acquisition of chromosomal aberrations in human adult mesenchymal stromal cells: response to Sensebé et al, Cells Stem Cell 10 (2012) 11.
- [80] N.J. Harrison, Genetic instability in neural stem cells: an inconvenient truth? The Journal of Clinical Investigation 122 (2012) 484–486.
- [81] H. Ueyama, T. Horibe, S. Hinotsu, T. Tanaka, T. Inoue, H. Urushihara, A. Kitagawa, K. Kawakami, Chromosomal variability of human mesenchymal stem cells cultured under hypoxic conditions, Journal of Cellular and Molecular Medicine 16 (2012) 72–82.
- [82] A. Borghesi, M.A. Avanzini, F. Novara, M. Mantelli, E. Lenta, V. Achille, R.M. Cerbo, C. Tzialla, S. Longo, A. De Silvestri, L.J. Zimmermann, P. Manzoni, M. Zecca, A. Spinillo, R. Maccario, O. Zuffardi, M. Stronati, Genomic alterations in human umbilical cord-derived mesenchymal stromal cells call for stringent quality control before any possible therapeutic approach, Cytotherapy 15 (2013) 1362–1373.
- [83] K. Mandel, Y. Yang, A. Schambach, S. Glage, A. Otte, R. Hass, Mesenchymal stem cells directly interact with breast cancer cells and promote tumor cell growth in vitro and in vivo, Stem Cells and Development 22 (2013) 3114–3127.
- [84] B.R. Sousa, R.C. Parreira, E.A. Fonseca, M.J. Amaya, F.M. Tonelli, S.M. Lacerda, P. Lalwani, A.K. Santos, K.N. Gomes, H. Ulrich, A.H. Kihara, R.R. Resende, Human adult stem cells from diverse origins: an overview from multiparametric immunophenotyping to clinical applications, Cytometry A 85 (2014) 43–77.
- [85] S.R.B.R. Sella, J.C. Francisco, T.M. Madalozzo, A.C. Irioda, B.F. Morghabel, R.J. Ferreira, L.C. Guarita–Souza, K.A.T. Carvalho, General requirement for quality and safety in stem cell and cell–based products for translational therapy, Cell Biology: Research & Therapy 3 (2014) 1–4.
- [86] K. Tanabe, K. Takahashi, S. Yamanaka, Induction of pluripotency by defined factors, Proceedings of the Japan Academy. Series B, Physical and Biological Sciences 90 (2014) 83–96.
- [87] J. Silva, J. Nichols, T.W. Theunissen, G. Guo, A.L. van Oosten, O. Barrandon, J. Wray, S. Yamanaka, I. Chambers, A. Smith, Nanog is the gateway to the pluripotent ground state, Cell 138 (2009) 722–737.
- [88] P. Zammit, J. Beauchamp, The skeletal muscle satellite cell: stem cell or son of stem cell? Differentiation 68 (2001) 193–204.
- [89] Y. Wei, Y. Li, C. Chen, Human skeletal muscle-derived stem cells retain stem cell properties after expansion in myosphere culture, Experimental Cell Research 317 (2011) 1016–1027.
- [90] R. Sarig, Z. Baruchi, O. Fuchs, U. Nudel, D. Yaffe, Regeneration and transdifferentiation potential of muscle-derived stem cells propagated as myospheres, Stem Cells 24 (2006) 1769–1778.
- [91] K.A. Westerman, A. Penvose, Z. Yang, P.D. Allen, C.A. Vacanti, Adult muscle "stem" cells can be sustained in culture as free-floating myospheres, Experimental Cell Research 316 (2010) 1966–1976.
- [92] A. Penvose, K.A. Westerman, Sca-1 is involved in the adhesion of myosphere cells to V 3 integrin, Biology Open 1 (2012) 839–847.
- [93] C. Poulet, E. Wettwer, T. Christ, U. Ravens, Skeletal muscle stem cells propagated as myospheres display electrophysiological properties modulated by culture conditions, Journal of Molecular and Cellular Cardiology 50 (2011) 357–366.
- [94] T. Nomura, T. Ueyama, E. Ashihara, K. Tateishi, S. Asada, N. Nakajima, K. Isodono, T. Takahashi, H. Matsubara, H. Oh, Skeletal muscle-derived progenitors capable of differentiating into cardiomyocytes proliferate through myostatin-independent TGF-beta family signaling, Biochemical and Biophysical Research Communications 365 (2008) 863–869.
- [95] G. Su, Y. Zhao, J. Wei, Z. Xiao, B. Chen, J. Han, L. Chen, J. Guan, R. Wang, Q. Dong, J. Dai, Direct conversion of fibroblasts into neural progenitorlike cells by forced growth into 3D spheres on lowattachment surfaces, Biomaterials 34 (2013) 5897–5906.
- [96] G. Su, Y. Zhao, J. Wei, J. Han, L. Chen, Z. Xiao, B. Chen, J. Dai, The effect of forced growth of cells into 3D spheres using low attachment surfaces on the acquisition of stemness properties, Biomaterials 34 (2013) 3215–3222.
- [97] M. Ieda, J.D. Fu, P. Delgado-Olguin, V. Vedantham, Y. Hayashi, B.G. Bruneau, D. Srivastava, Direct reprogramming of fibroblasts into functional cardiomyocytes by defined factors, Cell 142 (2010) 375–386.
- [98] R.R. Resende, E.A. Fonseca, F.M.P. Tonelli, B.R. Sousa, A.K. Santos, K.N. Gomes, S. Guatimosim, A. H. Kihara, L.O. Ladeira, Scale/Topography of substrates surface resembling extracellular matrix for tissue engineering, Journal of Biomedical Nanotechnology 10 (2014) 1157–1193.
- [99] K.S. Park, J. Ahn, J.Y. Kim, H. Park, H.O. Kim, S.H. Lee, Poly-L-lysine increases the ex vivo expansion and erythroid differentiation of human hematopoietic stem cells, as well as erythroid enucleation efficacy, Tissue Engineering. Part A 20 (2014) 1072–1080.
- [100] O.E. Cassidy, G. Rowley, I.W. Fletcher, S.F. Davies, D. Briggs, Surface modification and electrostatic charge of polystyrene particles, International Journal of Pharmaceutics 182 (1999) 199–211.
- [101] D. Mazia, G. Schatten, W. Sale, Adhesion of cells to surfaces coated with polylysine. Applications to electron microscopy, Journal Cell Biology 66 (1975) 198–200.
- [102] R. Chen, A paradigm shift in biomass technology from complete to partial cellulose hydrolysis: lessons learned from nature, Bioengineered 6 (2015) 69–72.

- [103] P. Ross, R. Mayer, M. Benziman, Cellulose biosynthesis and function in bacteria, Microbiological Reviews 55 (1991) 35–58.
- [104] P.C. Tischer, M.R. Sierakowski, H. Westfahl Jr., C.A. Tischer, Nanostructural reorganization of bacterial cellulose by ultrasonic treatment, Biomacromolecules 11 (2010) 1217–1224.
- [105] S. Wong, S. Kasapis, Y. Tan, Bacterial and plant cellulose modification using ultrasound irradiation, Carbohydrate Polymers 77 (2009) 280–287.
- [106] C.F. Souza, N. Lucyszyn, M.A. Woehl, I.C. Riegel-Vidotti, R. Borsali, M.R. Sierakowski, Property evaluations of dry-cast reconstituted bacterial cellulose/tamarind xyloglucanbiocomposites, Carbohydrate Polymer 93 (2013) 144–153.
- [107] W.K. Czaja, D.J. Young, M. Kawecki, R.M. Brown Jr., The future prospects of microbial cellulose in biomedical applications, Biomacromolecules 8 (2007) 1–12.
- [108] Y. Zheng, Z. Pan, R. Zhang, Overview of biomass pretreatment for cellulosic ethanol production, International Journal of Agriculture & Biological Engineering 2 (2009) 51–68.
- [109] B. Wei, G. Yang, F. Hong, Preparation and evaluation of a kind of bacterial cellulose dry films with antibacterial properties, Carbohydrate Polymers 84 (2011) 533–538.
- [110] C. Wu, G. Murtaza, M.A. Yameen, M.N. Aamir, M. Akhtar, Y. Zhao, Permeation study through bacterial cellulose membrane, Acta Poloniae Pharmaceutica 71 (2014) 297–300.
- [111] E. Trovatti, C.S.R. Freire, P.C. Pinto, I.F. Almeida, P. Costa, A.J.D. Silvestre, C. Pascoal Neto, C. Rosado, Bacterial cellulose membranes applied in topical and transdermal delivery of lidocaine hydrochloride and ibuprofen: In vitro diffusion studies, International Journal of Pharmaceutics 435 (2012) 83–87.
- [112] N. Rehman, M.I.G. Miranda, S.M.L. Rosa, D.M. Pimentel, S.M.B. Nachtigall, C.I.D. Bica, Cellulose and nanocellulose from maize straw: an insight on the crystal properties, Journal of Polymers and the Environment 22 (2014) 252–259.
- [113] M. Zuwei, M. Kotaki, S. Ramakrishna, Electrospun cellulose nanofiber as affinity membrane, Journal of Membrane Science 265 (2005) 115–123.
- [114] J. Doshi, D. Reneker, Electrospinning process and applications of electrospun fibers, Journal of Electrostatics 35 (1995) 151–160.
- [115] C.M. Souza, L.A. Mesquita, D. Souza, A.C. Irioda, J.C. Francisco, C.F. Souza, L.C. Guarita-Souza, M. R. Sierakowski, K.A. Carvalho, Regeneration of skin tissue promoted by mesenchymal stem cells seeded in nanostructured membrane, Transplantations Proceedings 46 (2014) 1882–1886.

## 15

# Biosurfactants as Antimicrobial and Antibiofilm Agents

V.L. Santos\*, R.M. Nardi Drummond, M.V. Dias-Souza UNIVERSIDADE FEDERAL DE MINAS GERAIS, BELO HORIZONTE, MINAS GERAIS, BRAZIL

## 15.1 Introduction

Microorganisms are ubiquitous in the biosphere, existing on a tour of terrestrial and aquatic habitats that currently exist on earth. Diverse habitats present many selective pressure issues in addition to carbon and energy sources that have confronted microorganisms. Thus, survival under these conditions should require adaptation and the development of unique cellular biochemistry and metabolism by these microbes. To colonize these challenging environments, microorganisms developed strategies that enable them to survive as free-living organisms, associated with surfaces as biofilms or associated with other organisms, establishing various forms of relationships such as symbiotic and pathogenic. The resulting adaptive biochemical mechanisms involve production of varied metabolites with diverse activities, such as hydrolytic enzymes, surface-active compounds, and bioactive molecules, including ribosomally and nonribosomally synthesized peptides.

Biosurfactants are amphiphilic molecules, having hydrophilic and hydrophobic domains that allow them to exist at polar and nonpolar medium edges. Hence, they tend to accumulate at interfaces (air–water and oil–water) as well as air–solid and liquid–solid surfaces, resulting in the reduction of repulsive forces between dissimilar phases, allowing the phases to mix and interact easily. Specifically, biosurfactants can reduce surface (liquid–air) and interfacial (liquid–liquid) tension [1]. When surfactants are prepared in solution, a decrease in surface or interfacial tension can be observed until the molecule reaches the critical micellar concentration (CMC). At this concentration, biosurfactant monomers are likely to self-assemble in complex arrangements like micelles, vesicles, or bilayers mediated by hydrogen bonding or hydrophobic or van der Waals interactions [2].

The interaction of biosurfactants with a substratum surface can alter the surface hydrophobicity, interfering in the processes of microbial adhesion and desorption to biotic and abiotic surfaces [3]. Biosurfactants have also been reported to have biological activity,

<sup>\*</sup>Corresponding Author.

including antibacterial, antifungal, antiviral, and anticancer. As a result, several of these biomolecules have merited renewed interest in both clinical and hygienic sectors due to their potential to kill microorganisms and to disperse microbial biofilms [4-7].

Although biofilms play an important positive role in a variety of ecosystems, they also have many negative effects, including infections in clinical settings. Resistance to antimicrobial and host defense mechanisms is a property of the biofilm lifestyle, leading to treatment failure and infection recurrence. The possibility of developing pathogen resistance to or unwanted side effects of bioactive microbial metabolites is less likely than with chemical antimicrobials for several reasons, including new inhibitory mechanisms and the fact that some of them, as antimicrobial peptides, are a natural part of the human antimicrobial defense system. Also, these novel antimicrobials can be used in combinatory approaches with synergic effects to combat infectious pathogens and their antimicrobial-resistance derivatives in their sessile and planktonic forms [8-10].

In past years, nanotechnology has become increasingly important for the clinical field; once promising applications are already in development in the areas of nanoparticles, drug-delivery nanocarriers, or phage therapy. Owing to their self-assembling and biochemical properties, biosurfactants have been attracting increased attention in the field of nanobiotechnology and can be used in various ways: as components of (micro and nano) colloidal systems, as functionalizing nanoparticles, as components of vesicles due to self-assembly properties (as MEL-A), entrapped in phospholipid-based liposomes, and in gene transfection [11-13].

Therefore, this chapter brings to light the state of the art of biofilms as well as technologies based in microbial surface-active agents already in use or in development to prevent or decrease the risk of biofilm-associated infections. Also their use immobilized in liposomes or associated with nanoparticles will be discussed.

## 15.2 Biofilms

Biofilms are complex microbial communities adhered on the surfaces of solid–liquid, liquid–air, liquid–liquid, or solid–air interfaces, embedded in their own microbial-originated matrix of protective and adhesive extracellular polymeric substances (EPSs), which are mainly polysaccharides, lipids, and proteins [14]. This matrix along with carbohydrate-binding proteins, pili, flagella, other adhesive fibers, and extracellular DNA acts as a stabilizing scaffold for the three-dimensional biofilm structure. This way of life in which microorganisms adopt a multicellular behavior facilitates their survival in diverse environmental niches, including those natural, such as living tissue of plants and animals, or artificial, such as clinical devices, food-contact surfaces in the food industry, or pipeline distribution systems and water storage or natural water systems [14,15]. In these environments, biofilm cells show greater resistance to environmental challenges, including desiccation, pH fluctuations, biocides, and antimicrobial therapy, than their planktonic counterparts.

The switch from planktonic to biofilm bacterial growth happens in response to environmental changes and comprises varied regulatory mechanisms and signaling pathways that can result in alterations in gene expression and spatiotemporal reorganization of bacterial cells [16]. This alters the expression of surface adhesion molecules and virulence factors and nutrient utilization and equips bacteria with varied properties that enable their existence in unfavorable conditions. In hospital settings, for instance, biofilm formation on vents and medical equipment allows pathogens to persist as reservoirs that can promptly spread to patients. Inside the host, pathogens can subvert immune defenses when associated with biofilms, being, therefore, associated with long-term disease.

Biofilm formation occurs as a result of sequential events: microbial surface attachment, cell proliferation, matrix production, and detachment. The conditioning film, formed by a layer of electrolytes and macromolecules such as DNA, proteins, biosurfactants, and humic acids adhered to the surface, is considered a precursor for the initial attachment of planktonic cells [17]. Once the microorganisms reach critical proximity to the surface, the determination of adhesion depends on the net sum of attractive or repulsive forces generated between the two surfaces and influenced by surrounding hydrodynamic forces [15,18]. Bacteria generally reach an irreversible attachment that may overcome shear forces and keep a persistent grip on the surface. This contact activates responses that result in changes in gene expression, up-regulating factors favoring sessility, such as those implicated in extracellular matrix formation [18,19]. This stage is controlled by *quorum sensing* (QS), an intracellular communication system designated for cell-density and population-based gene regulation [20].

A biofilm is considered mature when adherent bacteria produce EPSs, which contribute to the biofilm's vitality in several environmental conditions, protecting the biomass from desiccation, predation, oxidizing molecules, radiation, and damaging agents [14,21]. The matrix also acts as a virulence factor and confers to bacteria enhanced antimicrobial resistance and protection from host immune responses [22]. The EPSs act as an adsorbent or reactant, reducing the amount of antimicrobials available to interact with the biofilm; as well, their structure physically reduces the diffusion of antimicrobial agents by walling off access to regions of the biofilm. In this stage, it is possible to observe cavities that work as water transporter channels and planktonic cells throughout the biofilm community and also provide a unique environment for optimum nutrient absorption and waste disposal [16].

The final stage is the release of microorganisms back into their surroundings, where they return to their free-living state. Biofilm dispersal can be due to several reasons, including modifications in nutrient availability, oxygen oscillations and increase in toxic compounds, or varied stress-inducing conditions [23–25].

## 15.3 Current Approaches to Preventing or Removing Biofilms

As discussed previously, planktonic cells and young biofilms are often more susceptible to antimicrobial agents than mature biofilms. The large amounts of EPSs in the mature

biofilms can act as a diffusion barrier to antimicrobial agents. The high cell density in the mature biofilms can induce cell-to-cell communication (QS) systems, which up-regulate the expression of genes contributing to antimicrobial resistance [26,27] and release of protecting DNA in addition to increased transmission of resistance markers within the biofilm community [28]. Also, competition for nutrients can lead to differentiation of subpopulations and spatial physiological heterogeneity in mature biofilms, which may cause further antimicrobial resistance [21]. This pattern has been observed in several biofilm-forming bacteria on equipment or on indwelling medical devices and implants such catheters, mechanical heart valves, pacemakers, prosthetic joints, and contact lenses. This poses a critical problem because it enables pathogens as *Enterococcus* faecalis, Staphylococcus aureus, Staphylococcus epidermidis, Streptococcus viridans, Escherichia coli, Klebsiella pneumoniae, Proteus mirabilis, and Pseudomonas to persist as reservoirs that can readily spread to patients. Thus, occurrence of antimicrobial-resistant isolates of these species is global and is a serious problem, because of the limited efficacy of antimicrobial drugs and hospital hygiene and the resistance of biofilm-associated clinical strains. Notable biofilm-associated infections include chronic cystitis, endocarditis, otitis media, periodontitis, prostatitis, wound infections, and catheter- and stent-associated infections [29].

For S. aureus, the polymerizable mucopolysaccharide secreted by methicillinresistant S. aureus (MRSA) at the biofilm surface limits penetration of antibiotics, increases the amount of time required for vancomycin to penetrate the biofilm, and decreases the permeability ratio [30]. The long-term exposure of biofilms to drugs may promote adaptation to stress, thus further increasing bacterial resistance to antimicrobials [31,32]. Cells of this bacterium on a stationary phase, like those existing in biofilms. exhibited thickened cell walls and lower antimicrobial activity of vancomycin in vitro, thought to be attributable to the absence of actively replicating cells [33]. In another study, 40 MRSA isolates recovered from bloodstream infections displayed a four- and an eightfold increase in the minimal inhibitory concentration,  $MIC_{50}$  and  $MIC_{90}$ , respectively, when in biofilm. In time-kill studies, 15 mg/L vancomycin showed bactericidal activity against low-biofilm-producing strains with only a 1.8 log(10) CFU/mL difference in bacterial kill, compared to high-biofilm-producing strains (p < .001) (phenotype of 60% of the isolates). In studies based on biofilm formation during peritoneal dialysis, all the antimicrobials tested were effective in laboratory MIC tests, but lost their efficacy against S. aureus biofilms, except for gentamicin [34].

As exposed, the development of novel effective biofilm control methods is essential. The control of biofilm formation might be achieved by killing bacteria or targeting developmental stages of the biofilms like microbial attachment to surfaces, altering biofilm structure development and differentiation, killing of biofilm cells, and induction of biofilm dispersion. A large number of new strategies and approaches have been developed in the past few years, including antimicrobial locks, in the case of biofilm control in catheters [35]; the use of QS inhibitors [36]; bioactive microbial metabolites, including biosurfactants, exopolysaccharides, and bacteriocins [37,38]; enzymes that

dissolve biofilms by depolymerizing either polysaccharides or extracellular DNA [9,39]; and electrical fields [40], ultrasound [41], or photodynamic therapy [42] for enhancing antimicrobial penetration of biocides. Also, coadministration of enzymes or biosurfactants and antimicrobial [8,43] and lytic phages [44] has been evaluated. Drugdelivery nanocarrier systems such as liposomes and nanoparticles also have come up as attractive methods with a great potential in the treatment of biofilm infections, because of varied factors such as good biocompatibility and the wide range and extent of drugs they can be loaded with [11,45]. Another important point is the protection provided by the entrapment of the drug in the biological milieu, reducing toxicity and allowing the drug to reach the specific target.

### 15.4 Surface Active-compounds as Potential Antibiofilm Agents

Biosurfactants are amphipathic molecules produced by a variety of bacteria, yeasts, and filamentous fungi. Because they present both hydrophilic and hydrophobic moieties, these molecules are able to interact with surfaces, lowering surface and interfacial tensions, and to form micelles, emulsifying immiscible substances. These biomolecules can be excreted to the extracellular medium or remain attached to the cell surface, being denominated particulate biosurfactants [1].

The surface-active properties of biosurfactants provide the same activities of detergency, emulsification, deemulsification, foaming, and dispersion phases, making them potentially applicable in varied industrial sectors, including agriculture, food, cosmetics, pharmaceuticals, and petrochemicals. Other advantages of these biomolecules that have attracted the attention of researchers and industry are specificity, minimum toxicity, elevated biodegradability, extensive applicability, and efficiency at pH and temperature extremes [1,46-49]. In the biomedical field, biosurfactants are getting more attention for their antibiofilm and antimicrobial activity because of lower toxicity for plants and animals, high biodegradability, low irritancy, and compatibility with human skin [50]. The only factor that works against the widespread use of biosurfactants is the economics of their production; however, much effort has been given to facilitating the optimization process at both the engineering and the biological levels. Our group evaluated the production of these molecules from biodiesel refinery coproducts such as glycerol, diatomaceous earth impregnated with esters from filters used in biodiesel purification, and macaúba (Acrocomia acu*leata*) [47].

Biosurfactants are classified according to their chemical composition and microbial origin. The major classes include varied structures such as glycolipids, lipopeptides, polysaccharides, or complexes of proteins, phospholipids, fatty acids, and neutral lipids [1]. Thus, it is logical to expect varied properties and biological activities for different biosurfactants. Conversely, these molecules can be modified to suit different applications

by genetic engineering, biologically or biochemically altering the growth substrate or growth conditions. The physiological roles of these biomolecules are often unclear, but many are essential for the survival of microorganisms in the environment, including increased surface area and bioavailability of hydrophobic substrates, antagonistic activity, binding to heavy metals, bacterial pathogenesis, QS, biofilm formation, and desorption surfaces [51].

These tensoactive molecules can spontaneously adsorb to surfaces, altering properties such as wettability and charge [3,52] or even making changes in bacterial cell surface structures. Loss of lipopolysaccharides (LPSs) was observed in Pseudomonas aeruginosa strains treated with rhamnolipids (RLs) and this resulted in increased changes in cell surface hydrophobicity and decreases in the outermembrane proteins of bacteria [53,54]. Successive studies confirm that RL biosurfactants play an important role during the first steps of biofilm formation during cell attachment and when microcolonies emerge [5,55]. These biomolecules are responsible for maintaining the transport channels and directly influence biofilm structure, namely the creation of mushroom-like filaments [4,5]. RLs may be used to exclude other invasive species from the biofilm structure. Such a strategy allows the microbial community to keep its uniform structure and exclusiveness over a given niche [4]. A mannose-rich polysaccharide produced by E. coli impairs S. aureus adherence and colonization of the mature *E. coli* biofilm [56]. A uropathogenic strain of E. coli was shown to induce surface physicochemical alterations by releasing into the environment a group II capsular soluble polysaccharide, preventing biofilm formation by a wide range of gram-positive and gram-negative bacteria [57]. Polymers assembling on surfaces can increases their hydrophobicity and modulate microbial interaction with interfaces by limiting initial bacterial adhesion and by interfering with subsequent cell-cell contact within the biofilm. Also, a 546-kDa exopolysaccharide (A101) isolated from marine Vibrio sp. not only inhibited biofilm formation of many gram-negative and gram-positive bacteria but also disrupted established biofilms of P. aeruginosa, but not S. aureus. The polymer also affected cell-cell interactions of both *P. aeruginosa* and *S. aureus*, whereas the cell-surface interaction inhibition occurred only in S. aureus [58].

Other roles for biosurfactants include antibacterial, antifungal, and antiviral activities [38,50]. Because of their tensoactive properties, many surfactants can cause membrane disruption and cellular lysis by increasing membrane permeability, causing metabolite leakage by altering the physical membrane structure or by disrupting proteins; thus, interfering with important membrane functions such as energy generation and transport [59]. This has also been reported to lead to toxicity, lysis, pyrogenicity, mitogenicity, and immunogenicity, among other effects. Also, human-derived biosurfactants have received increased attention because of their role in immunity and defense [60]. In this context, the use of biosurfactants as alternatives to antibiofilm antimicrobial agents has extensively been explored [52,61].

#### 15.4.1 Lipopeptides

Lipopeptides (LPs) are low-molecular-mass biosurfactants produced by a wide spectrum of microorganisms including species of *Bacillus, Streptomyces, Lactobacillus, Streptococcus, Pseudomonas*, and *Arthrobacter* [7,62,63]. Structurally, LPs are composed of a hydrophilic peptide attached to a hydrophobic lipid or fatty acid. Among the bacterial LP producers, *Pseudomonas* and *Bacillus* have received the most attention. The advances in genome sequencing have highlighted the genus *Bacillus* as a potential source of antibiotic-like compounds, including aminoglycosides, polyketides, and several small proteinaceous and peptidal structures such as bacteriocins, oligopeptides, and LPs and other unusual antibiotics. The nonribosomal peptide and polyketide groups include toxins, siderophores, pigments, antibiotics, cytostatics, and immunosuppressants [7].

LPs from *Bacillus* are classified into three families (surfactin, iturin, and fengycin) depending on their amino acid sequence, the type of peptide cyclization, and the nature of, length of, and substitutive groups that compose the fatty acid chain. Thus, each family has variants within the same peptide length, but changing the type and position of amino acid residue. Also, each variant can have homologs of varied length and isomeric state of the fatty acid chain [64]. The surfactin family, for instance, includes the heptapeptide variants of the esperin, lichenysin, pumilacidin, and surfactin groups. The peptide chain is linked to a  $\beta$ -hydroxyl fatty acid (C12–C16) with linear, iso, or anteiso, branches. The iturin family has the variants iturin A and C; bacillomycin D, F, and L; and mycosubtilin, in which the heptapeptide is linked to a  $\beta$ -amino fatty acid chain of variable length (C14–C17). The third family includes fengycins A and B, also known as plipastatins if Tyr9 is D-configured. These decapeptides are linked to a  $\beta$ -hydroxyl fatty acid chain (C14-C18) that can also be in linear, iso, or anteiso forms, saturated or not. Furthermore, the described LP families of *Bacillus* have specific mechanisms of cyclization. In addition, various other LPs have been identified for *Bacillus* species, including kurstakin heptapeptide from *Bacillus thuringiensis* [65], the 12-amino-acid-containing maltacines from *Bacillus subtilis* [66], polymyxins from *Bacillus polymyxa* with high content in diaminobutyric acid [67], and the surfactin-like bamylocin A from *Bacillus amyloliquefaciens* [62].

LPs belonging to the surfactin, iturin, and fengycin families, are formed by nonribosomal peptide synthetases and/or polyketide synthases [37]. The nonribosomal peptide synthetases lead to a remarkable heterogeneity among the LPs as described below.

*Surfactin*: Surfactins are one of the most effective biosurfactants first isolated from *B. subtilis*; they can decrease water surface tension from 72 to 27 mN/m, at low concentrations like 0.005%. These biomolecules have several pharmaceutical applications such as inhibition of clot formation, formation of ion channels in membranes, and antibacterial, antifungal, antimycoplasma, antiviral, and antitumor activities [64,68,69]. Unfortunately, surfactins can also be indiscriminately cytotoxic with hemolytic activity [70].

Mireles et al. [71] demonstrated that surfactin from *B. subtilis* is able to inhibit Salmonella enterica biofilm adhesion on precoated urinary catheters and polyvinyl chloride microtiter wells. They also reported a similar effect for *E. coli*, whereas this pretreatment was completely ineffective against *P. aeruginosa*. Fengycin-like and surfactin-like molecules produced by B. subtilis and Bacillus licheniformis have also been described to cause dispersion of up to 90% of gram-positive S. aureus ATCC 29213 biofilms and up to 97% of E. coli CFT073 biofilm [72]. It was shown that surfactin induces nanoripples in lipid bilayers, which may explain the biosurfactant action on biofilm permeability or integrity, probably through the formation of some kind of channels within the biofilm increasing penetrability [73]. Surfactin has an anionic nature: the antiadhesive effect can be due to the electrostatic repulsion between bacteria and the molecules of surfactin adsorbed onto the polystyrene surface [74]. Three surfactin-producing B. subtilis strains, C4, M1, and G2III, which were obtained from honey samples and intestines of the bee Apis mellifera L., have been described to exhibit antagonistic potential against six different bacteriocin-resistant Listeria monocytogenes strains. Surfactin produced by the B. subtilis subsp. subtilis C4, G2III, and M1 strains inhibited these pathogens at concentrations of 0.125, 0.250, and 1 mg/mL, respectively [75].

Furthermore, the N3 LP, which is produced by *B. amyloliquefaciens* M1 strain, shows broad-spectrum antibacterial potential, including activity against multidrug-resistant *Vibrio* species, particularly *Vibrio* anguillarum, and *Shewanella* aquimarina, which was isolated from sick marine animals. This LP contains surfactin isoforms with a regular amino acid sequence (GLLVDLL) and hydroxy fatty acids that are 12–15 carbons in extent. The treatment with this molecule also resulted in injury to the cell membrane of *V. anguillarum* and rupture of the bacterial cell [76].

Fassi et al. [77] described the use of surfactin to increase the effect of enrofloxacin against planktonic *Mycoplasma pulmonis*. Rivardo et al. [72] reported a synergistic increase in the efficacy of various antimicrobials (ampicillin, cefazolin, ceftriaxone, ciprofloxacin, piperacillin, tobramycin, and trimethoprim/sulfamethoxazole) in the eradication of uropathogenic *E. coli* CFT073 biofilm when associated with V9T14 LP biosurfactant from *B. licheniformis*. The authors hypothesize that the action of V9T14 biosurfactant lies in its interaction with the bacterial membrane, which results in pore formation and altered membrane integrity, leading to an increase in entrance and efficacy of the antimicrobial drug.

*Iturin*: This cyclic LP produced by *B. subtilis* presents antimicrobial activity. The antifungal activity involves pore forming that forms small vesicles in the plasma membrane of yeast resulting in metabolite leakage. Iturin A is able to penetrate the cell wall without causing damage, but the membranes of cytoplasmic organelles are affected. Furthermore, it also significantly enhances the electrical conductance of biomolecular lipid membranes. Iturin A has been investigated as an effective antifungal molecule for deep mycosis. Other variants of the iturin group, including bacillomycin D and bacillomycin Lc, also present antimicrobial activity against *Aspergillus flavus*, but the varied lipid chain length apparently interfered with the activity of the LP against other fungi [78].

An LP from *B*. *subtilis*, showed antifungal activity, modifying the morphology and structure of the cell membrane of a yeast [59].

*Fengycin*: Antimicrobial and antibiofilm activities have been ascribed to fengycin LPs. Like most the aminopeptides, fengycin seems to act by raising the plasma membrane permeability of the target cell. It was observed that fengycin purified from *B. thuringiensis* strain SM1 was subject to self-assembly, presenting a globular micellar structure. The interfacial modification enhanced the spectrum of the microbial action confined to fungi to include some activity against both gram-positive *S. epidermidis* and gram-negative *E. coli* (MIC 500 μg/mL) [79]. The fungitoxic activity can be attributed to inhibition of the enzymes phospholipase A2 and aromatase [80]. Culture supernatant of *B. amyloliquefaciens* ANT1 containing several LP antibiotics, including surfactin, iturin A, and fengycin B, showed antimicrobial and antibiofilm activities against *Bacillus cereus* ATCC 17778, *Pseudomonas* ATCC 15442, *Aeromonas hydrophila* ATCC 7966, and *Aspergillus niger* ATCC 9642 [81]. More recently, it was shown that a mixture of iturin and fengycin produced by *B. amyloliquefaciens* strain AR2 inhibited *Candida* biofilm formation as well as dislodging preformed biofilm; the dosage of 6 mg/mL of biosurfactant had similar fungicidal activity for both planktonic cell and biofilm [63].

**Polymixins**: These compounds constitute a class of nonribosomally synthesized cyclic LPs, generally produced as secondary metabolites of *Bacillus* sp. There are two polymyxins available for clinical use, colistin (i.e., polymyxin E) and polymyxin B. They were discovered in the 1940s and were never subjected to contemporary drug-development procedures. They have a narrow antibacterial spectrum, mainly against gram-negatives. Clinical use of colistin and polymyxin B waned in the 1970s because of the early experience of nephrotoxicity and neurotoxicity after intravenous administration, related to apoptosis of kidney tubular cells, due to oxidative stress. However, the rapid increase in resistance to all other antimicrobials has demanded their resurgence in clinical contexts [82]. Parenteral colistin is much more commonly used internationally, although injectable polymyxin B is available in a number of countries, such as Brazil, Singapore, and the United States; in these three countries, both antimicrobials are available [83].

The molecular structure of polymyxin comprises a cyclic peptide chain and a hydrophobic tail. Each member of the polymyxins differs in the structure of fatty acids and variations in the amino acid residues. The LPs of the polymyxin B and E groups share a similar primary sequence with the only difference being at position 6, which is occupied by D-Phe in B and D-Leu in E. Another important difference between polymyxin B and colistin is that the former is administered parenterally as the sulfate salt, whereas the latter is administered as the sodium salt of colistin methanesulfonate, an inactive prodrug that undergoes hydrolysis in vivo and in vitro to form the active entity colistin [83].

The bactericidal action of the polymyxin antibiotics is thought to be initiated by their interaction with LPS of gram-negative bacteria, which causes LPS aggregation raising the surface charge of the LPS. The antibiotic may be internalized and then attached to the bacterial phosphatidylglycerol-rich membrane leaflets, which in sequence causes

leakage of cellular contents at higher peptide concentrations [84]. The mechanism of action of polymyxins involves an initial stage of interaction with lipid A of the LPS, leading to self-promoted uptake of polymyxins across the membrane, followed by cell death.

Polymyxins have been shown to inhibit alternative nicotinamide adenine dinucleotide dehydrogenase and malate:quinone oxidoreductase in *Mycobacterium smegmatis*; no such enzymatic study has been reported in gram-negatives [85]. It was also suggested that rapid killing of *Acinetobacter baumannii* by polymyxins is mediated by a hydroxyl radical death pathway [86]. In addition, the mechanisms of resistance to polymyxins involve alterations in the chemical composition or loss of LPS, thereby decreasing the interaction with polymyxins.

A complex of antibiofilm compounds produced by *Paenibacillus polymyxa* containing fusaricidin B and polymyxin D1 and surfactins reduced the biofilm biomass of *B. subtilis, Micrococcus luteus, P. aeruginosa, S. aureus,* and *Streptococcus bovis.* This complex of biosurfactants also inhibited the formation of multispecies biofilms such as self-assembling marine biofilm in coincubation assays by 99.3% and disrupted previously established biofilms by 72.4% [87]. Polymyxin E (colistin) is indicated as an early destructive therapy to set back the onset of chronic *P. aeruginosa* infections, which normally form biofilms, or irregular colonization in cystic fibrosis patients, in a combination of oral ciprofloxacin with colistin inhalation [88].

Tarquinio et al. (2014) investigated the activity of tobramycin and polymyxin E against biofilms of *P. aeruginosa* in polyvinyl chloride endotracheal tubes, simulating a critical condition, which is biofilm formation in medical devices [89]. These drugs were not effective, neither isolated nor combined, against biofilms formed overnight. However, bactericidal and bacteriostatic effects were observed against planktonic cells, and when coating strategies were employed, tobramycin used in a concentration of more than  $2 \times$  the MIC, alone or combined with polymyxin E, was more bactericidal than polymyxin E alone after overnight biofilm exposure.

The cyclic LPs produced by *Pseudomonas* were initially classified into four major groups, viscosin, amphisin, tolaasin, and syringomycin, considering their structural characteristics [90]. But, some structurally new LPs were identified, including arthrofactin of *Pseudomonas* (formerly *Arthrobacter*) sp. strain MIS38 [91], putisolvins I and II of *Pseudomonas putida* [92,93], pseudofactin of *Pseudomonas fluorescens* BD5 [94], and pseudodesmins A and B of a *Pseudomonas* strain obtained from salamander skin [95]. Furthermore, several linear LPs were also described, including syringofactins of *Pseudomonas syringae* pv. tomato strain DC3000 [96] and peptin31, a linear derivative of syringopeptin produced by *P. syringae* strain 31R1 displaying activities against *Rhodotorula pilimanae*, *Rhizoctonia solani*, and *Trichoderma harzianum* and also hemolytic and antibacterial activities [97].

**Putisolvin**: *P. putida* PCL 1445 produces two surfactant-active compounds referred to as putisolvins I and II, which influence the development of bacterial biofilms on polyvinyl chloride, but can be effective dispersal agents when added before or after the

development of biofilms of other *Pseudomonas* sp. strains [92]. The only difference between the compounds is that Val11 in putisolvin I is replaced by Ile or Leu in putisolvin II, which accounts for the mass difference of 14 Da. A possible mode of action of putisolvins I and II during the inhibition of biofilm formation could be by binding to the cell surface or to components of the cell surface, thereby influencing the outer membrane hydrophobicity and consequently the adhesion.

**Pseudofactin**: This is a cyclic lipodepsipeptide produced by *P. fluorescens* with a structure based on that of a palmitic acid bound to the terminal amino group of an eight-amino-acid peptide chain. The C-terminal carboxylic moiety of the last amino acid generates a lactone with the hydroxyl of the third amino acid, which is a threonine. Pseudofactin II produced by *P. fluorescens* BD5 lowered the adhesion to three types of surfaces (glass, polystyrene, and silicone) of bacterial strains of five species, *E. coli, E. faecalis, Enterococcus hirae, S. epidermidis,* and *P. mirabilis,* and two *Candida albicans* strains [94]. Pretreatment of a polystyrene surface with 0.5 mg/mL pseudofactin II inhibited bacterial adhesion by 36–90% and that of *C. albicans* by 92–99%. The same concentration of pseudofactin II dislodged 26–70% of preexisting biofilms grown on previously untreated surfaces. Pseudofactin II also inhibited the initial adhesion of *E. faecalis, E. coli, E. hirae,* and *C. albicans* strains to silicone urethral catheters. Total growth inhibition of *S. epidermidis* was observed at the highest concentration tested (0.5 mg/mL), which caused a partial (18–37%) inhibition of other bacteria and 8–9% inhibition of *C. albicans* growth.

#### 15.4.2 Glycolipids

Glycolipids are formed by various types of mono-, di-, tri-, and tetrasaccharide carbohydrates combined to hydrophobic fractions of one or more long-chain aliphatic fatty acids. The most investigated glycolipids are rhamnolipids produced by *Pseudomonas* sp. [98]; mannosylerythritol lipids produced by *Pseudozyma antarctica* (previously *Candida*) [99]; trehalose lipids produced by *Rhodococcus* sp., *Nocardia* sp., and *Mycobacterium* sp. [100]; and sophorolipids synthesized by yeasts [101].

**Rhamnolipids**: These comprise one or two molecules of rhamnose bound to one or two molecules of  $\beta$ -hydroxydecanoic acid. Although originally isolated from *P. aeruginosa*, analogues of this biomolecule have been isolated from strains of *Burkholderia* [102], *Renibacterium salmoninarum*, *Cellulomonas cellulans*, *Nocardioides*, and *Tetragenococcus koreensis* [103]. RLs are involved in *Pseudomonas* sp. biofilm formation by way of promoting motility, the inhibition of attachment, and the destruction of the matrix, preserving channels throughout the biofilm for transport of water and oxygen [104]. RLs have shown antimicrobial activity against a wide variety of fungi and pathogens in concentrations from 0.4 to 35 µg/mL. A mixture of RLs produced by *P. aeruginosa* 47T2 showed excellent antimicrobial properties, and low MIC values were found for *Serratia marcescens* (4 µg/mL), *Enterobacter aerogenes* (8 µg/mL), *K. pneumoniae* (0.5 µg/mL), *S. aureus* and *S. epidermidis* (32 µg/mL), *B. subtilis* (16 µg/

phytopathogenic fungal species: Chaetonium globosum mL), and (64 µg/ mL), Penicillium funiculosum (16 µg/mL), Gliocadium virens (32 µg/mL), and Fusarium solani (75  $\mu$ g/mL) [105]. The biosurfactants produced by *Rhodococcus erythropolis* exhibited high inhibitory activity against E. coli, P. aeruginosa, and the fungi A. niger and A. *flavus*, among the tested organisms [106]. The molecules affected the morphology of cells, showing enlarged, elongated, empty hosts or fragmented ones, consistent with extremely low viability. Direct action on the surface structure of the bacterial cell has also been related. In P. aeruginosa at low concentrations of RL, a decrease in the proportion of LPSs was observed, increasing the hydrophobicity of the cellular surface and causing changes in the membrane proteins and therefore on the surface morphology of the cell [54].

RLs produced by P. aeruginosa generate 41-71% inhibition of bacterial adherence on polystyrene surfaces, as tested on L. monocytogenes [107]. They are also capable of eliminating and interrupting the formation of biofilms by Bordetella bronchiseptica, a pathogen of the respiratory tract in many mammals [108]. Moreover, the potent antibiofilm activity of RLs was evaluated against several microbial species associated with biofilm formation on voice prostheses and silicone rubber in the presence and absence of adsorbed RLs [109]. The antiadhesive activity of RLs at various concentrations was significant against all the strains and depended on the microorganism tested, with a maximal initial reduction of adhesion rate (66%) reported for strains of Streptococcus salivarius and Candida tropicalis. Furthermore, RL conditioning of silicone rubber caused a reduction of 48% in cell adherence of tested S. epidermidis, S. salivarius, S. aureus, and C. tropicalis strains. In another study, RLs reduced the initial attachment of the gram-negative E. coli, P. putida, and P. aeruginosa and the gram-positive B. subtilis on both hydrophilic glass and hydrophobic octadecyltrichlorosilane (OTS)-modified surfaces [110], whereas for S. epidermidis, these molecules reduced the attachment only on OTS-modified glass. RLs could inhibit the growth of B. subtilis, S. epidermidis, and P. aeruginosa PAO1 but not the growth of E. coli, P. putida, and P. aeruginosa E0340. Also, RLs were effective in changing the cell surface hydrophobicity of the tested strains, although no clear effect was observed on *B. subtilis*. Despite the observed trends on cell detachment, the responsible mechanism remains to be elucidated.

Studies have also demonstrated the activity of RLs against fungal biofilms. Dusane et al. [111] tested the potential of RLs to prevent biofilm formation and disrupt preestablished biofilms of the yeast *Yarrowia lipolytica*. In their study, precoating of microtiter plate wells with this glycolipid effectively reduced *Y. lipolytica* biofilm formation by 50% compared to cetyltrimethyl ammonium bromide (CTAB), which inhibited by 29%, and sodium dodecyl sulfate (SDS), which decreased biofilms by <10% at their respective MIC values. Moreover, RLs displayed 55% dispersion of *Y. lipolytica* biofilms (formed for 3 days in microtiter plate wells and treated for 1 h), whereas 35% and 40% disruption was observed with CTAB and SDS, respectively, at their respective MIC values. This biomolecule was able also to disrupt fungal biofilms of *C. albicans* formed on polystyrene surfaces [32]. The authors showed that antiadhesive activity of RLs on *Candida* was concentration dependent. RLs from *Pseudomonas* spp. enhance the disinfection effect of NaOCl and peracetic acid/hydrogen peroxide on stainless steel surfaces contaminated with *L. monocytogenes* [112].

*Lunasan and rufisan*: *Candida sphaerica* UCP 0995 produces a biosurfactant known as lunasan, when using residual refined soy oil as a carbon source, and it has shown growth inhibition (5 mg/mL) as well as antiadhesive properties (10 mg/mL) on pathogens such as *S. aureus* (90% and 100%, respectively), *C. albicans* (64% and 100%), and *Streptococcus agalactiae* (92% and 100%) [113]. Antiadhesive and antimicrobial activities of rufisan, a biosurfactant produced by the yeast *Candida lipolytica* UCP0988, was described [114]. Crude biosurfactant presented antiadhesive potential at concentrations higher than 0.75 mg/L against most of the bacteria tested: *S. aureus, S. agalactiae*, and *Streptococcus mutans*. The antiadhesive properties were proportional to the concentration of the biosurfactant, whereas the antimicrobial activities were observed at higher biosurfactant concentrations.

Mannosylerythritol lipid: This biosurfactant contains 4-*O*-β-D-mannopyranosylmesoerythritol as the hydrophilic group and a fatty acid and/or an acetyl group as the hydrophobic moiety. The fatty acid chains can be short (2–8 carbon atoms) or long (10–18 carbon atoms). Mannosylerythritol lipid (MEL) is reported to be secreted by Ustilago sp. as a minor component along with cellobiose [115]. Regarding Schizonella *melanogramma* (shizonellin) [116], *Geotrichum candidum* [117], and *P*. (previously *Candida*) *antarctica*, it is secreted as a major component [118]. The type of MEL varies with the strain and the species of *Pseudozyma*. Hence its production can be used as an important taxonomic index to identify *Pseudozyma* yeasts. The MELs are classified as MEL-A, -B, -C, and -D considering the degree of acetylation at the C4 and C6 positions and their order of appearance on thin-layer chromatography. MEL-A is the diacetylated compound, MEL-B and MEL-C are monoacetylated at C6 and C4, respectively, and the completely deacetylated structure is known as MEL-D [119].

Mixtures of these four MELs, but predominantly MEL-A and MEL-B, isolated from *C. antarctica* T34 showed surface-active and antimicrobial activities, including activity against *C. albicans* [118]. Two MELs, schizonellin A and B, from the fungus *S. melanogramma* were effective against fungi, gram-positive bacteria, and some gramnegative bacteria [116].

*Cellobiose lipids*: Flocculosin, a cellobiose lipid from *Pseudozyma flocculosa*, possesses a C8-hydroxy acid and two acetyl groups in the cellobiose residue. It presents in vitro antifungal potential against various pathogenic yeasts that cause human mycoses, like *Candida lusitaniae*, *Cryptococcus neoformans*, *Trichosporon asahii*, and *C. albicans* [120]. It was also particularly effective against *Staphylococcus* species, including MRSA, and its antimicrobial potential was not modified by the presence of resistance mechanisms against methicillin and vancomycin [121]. This molecule was able to kill *C. albicans* cells quickly, in a dose-dependent manner. The K<sup>+</sup> efflux detected in flocculosin-treated cells suggested that the glycolipid acts by disrupting the cell membrane.

The cellobiose lipid secreted by *Cryptococcus humicola* presents 16-(tetra-*O*-acetyl- $\beta$ -cellobiosyloxy)-2-hydroxyhexadecanoic acid as the major product and 16-(tetra-*O*-acetyl- $\beta$ -cellobiosyloxy)-2,15-dihydrohexadecanoic acid as a minor product, whereas *Pseudozyma fusiformata* secreted mainly 16-[6-*O*-acetyl-2'-*O*-(3-hydroxyhexanoyl)- $\beta$ -cellobiosyloxy]-2,15-dihydroxyhexadecanoic acid. These compounds exhibit similar fungicidal activities against different yeasts including pathogenic *Cryptococcus* and *Candida* species. *Filobasidiella neoformans* cells died after 30 min incubation with 0.02 mg/mL cellobiose lipids. The same effect on ascomycetous yeast, including pathogenic *Candida* species, is achieved at 0.1–0.3 mg/mL cellobiose depending on the test culture used [122].

**Sophorolipids**: This group is generally recognized as disaccharide sophoroses (2-*O*- $\beta$ -D-glucopyranosyl-D-glucopyranose)  $\beta$ -glycosidically linked to the hydroxyl group at the last but one carbon of fatty acids. Sophorolipids (SLs) are produced by nonpathogenic yeasts, like *Candida bombicola, Candida apicola,* and *Candida bogoriensis*. Similar to other biosurfactants, SLs present antimicrobial potential. The mechanism of action is not merely restricted toward bacteria; they also present antifungal, antialgal, antimycoplasmal, and antiviral activities [123]. The biological activity of this group involves destabilization and alteration of cellular membrane permeability [124]. SLs inhibited the growth of several gram-positive bacteria in concentrations from 50 to 29,000 µg/mL, such as *B. subtilis, S. epidermidis, S. aureus, S. faecium, Propionibacterium acnes*, and *Corynebacterium xerosis* [125,126].

Biosurfactants could be useful to increase the efficacy of known antimicrobials and biocides. Coadministration of SL produced by C. *bombicola* ATCC 22214 enhanced the action of antimicrobials in representative gram-positive (*S. aureus*) and gram-negative bacteria (*E. coli*). Taking into consideration the self-assembly capacity of SL, it has been postulated that self-assembled SLs can span the bacterial cell membrane and thus make possible the entrance of antimicrobial molecules [127].

*Trehalose lipids and succinyl trehalose lipids*: Trehalose lipids (TL-1 and TL-2) do not show inhibition properties against gram-negative bacteria and yeast, but TL-1 presents antifungal activity by inhibiting the germination of conidia from *Glomerella cingulata* at 300 mg/L [128]. Succinyl trehalose lipids (STL-1 and STL-2) show antifungal and antiviral properties, as they are capable of inhibiting herpes simplex and influenza viruses at concentrations from 11 to 33 mg/L [1,129].

Trehalose lipids (TLs) synthesized by *Tsukamurella* sp. strain DSM 44370 jointly with trisaccharide and tetrasaccharide lipids presented partial activity against gram-positive bacteria, with the exception of *S. aureus*, whereas gram-negatives were poorly or not inhibited at all [130]. Studies performed to clarify the molecular interactions between this molecule and the lipid component of the cell membrane revealed that TLs improved the fluidity of phosphatidylethanolamine and phosphatidylserine membranes and formed domains in the fluid state, but did not change the macroscopic bilayer organization [131].

*Xylolipids*: Xylolipids produced by probiotic bacteria such as *Lactococcus lactis* using paraffin as a carbon source showed antibacterial activity against the pathogens *E. coli* and *S. aureus* resistant to multiple drugs, such as methicillin. These glycolipids have a great potential as an alternative to the use of pharmaceuticals for oral and dermal administration [132].

**Oligosaccharide lipids**: Oligosaccharide lipids (GL) produced by *Tsukamurella* sp. show certain inhibition properties against some gram-positive and gram-negative bacteria. GL-1, GL-2, and GL-3 inhibit the growth of *Bacillus megaterium* at concentrations of 50, 100, and 150  $\mu$ g/mL, respectively [131]. *Tsukamurella* sp. DSM 44370 produces a mix of GLs when cultured in sunflower oil, but when using marigold oil as the carbon source, the production is increased 60% and the proportion is modified on GL-3 mainly. When these glycolipids are modified through microbial lipases, molecules that are biologically active against the activation of the Epstein–Barr virus are formed [133].

**Other glycolipids**: Complexes of glycolipids from *Brevibacterium casei* MSA19 have been reported to disrupt and significantly inhibit single and mixed biofilms of pathogenic and nonpathogenic bacteria of human and fish at concentrations of  $30 \ \mu g/mL$  [134]. A glycolipid biosurfactant from *Lysinibacillus fusiformis* S9 restricted the biofilm formation of *E. coli* and *S. mutans* completely at a concentration of  $40 \ \mu g/mL$  but did not show any bactericidal activity. The biosurfactant inhibited bacterial attachment and biofilm formation equally on hydrophilic and hydrophobic surfaces like glass and catheter tubing [135]. *Serratia marcescens* is capable of producing a glycolipid composed of glucose and palmitic acid that prevented the adhesion of *C. albicans, P. aeruginosa,* and *Bacillus pumilus,* as well as interrupting the formation of biofilms of these cultures in microtitration plates [136].

#### 15.4.3 Biosurfactants From Lactobacillus

Biosurfactants isolated from lactobacilli have been shown to reduce the adhesion of pathogenic microorganisms to glass [137], silicone rubber [138], surgical implants [139], and voice prostheses [140]. Biosurfactant produced by *Lactobacillus paracasei* subsp. *paracasei* A20 was effective against gram-positive and gram-negative bacteria, yeasts, and filamentous fungi [141]. The biomolecule also showed antiadhesive activity against all the bacteria tested. But the antiadhesive activities were highest for *S. aureus, S. epidermidis*, and *S. agalactiae* compared to the values observed for *E. coli, C. albicans,* and *P. aeruginosa,* in contrast with the antimicrobial activity exhibited against these microorganisms at the same biosurfactant concentrations. Biosurfactants produced by *Streptococcus thermophilus* A and *L. lactis* 53 presented significant antimicrobial activity against different bacteria and yeasts obtained from explanted voice prostheses [142,143]. The total growth inhibition pattern of these biomolecules against *C. albicans, S. aureus,* and *S. epidermidis* was similar to those observed for the biosurfactants isolated from *L. paracasei* subsp. *paracasei* A20, with concentrations between 25 and 100 mg/mL [143]. Another probiotic strain, *Lactobacillus acidophilus*, was able to reduce the biofilm

formation of *S. mutans*, a primary dental cariogen, by inhibiting its attachment [144]. Velraeds et al. [137] also described the inhibition of pathogenic enteric bacteria adhesion by a biosurfactant produced by a *Lactobacillus* strain and then reported that the biosurfactant triggered a dose-related inhibition of the initial deposition speed of *E. coli* and other bacteria adherent on both hydrophobic and hydrophilic substrates. More recently, antimicrobial activities against clinical multidrug-resistant strains of *A. baumannii*, *E. coli*, and *S. aureus* (MRSA) by biosurfactant produced by *Lactobacillus jensenii* and *Lactobacillus rhamnosus* at 25–50 mg/mL was demonstrated. Antiadhesive and antibiofilm potentials were also proposed for the aforementioned pathogens ranging from 25 to 50 mg/mL. Biosurfactant caused membrane damage to *A. baumannii* and marked cell wall injury in *S. aureus* [145]. Our group showed that the combination of subinhibitory concentrations of biosurfactant produced by *L. jensenii* P6A and benzazoles presented a concentration-dependent synergism against *E. coli* and *C. albicans*.

#### 15.4.4 High-Molecular-Weight Biosurfactants

Polymeric biosurfactants are synthesized by several bacteria and are made of lipoproteins, proteins, polysaccharides, LPSs, or complexes containing these molecules [146]. These high-molecular-weight biosurfactants generally possess effective emulsifying activity rather than an interfacial tension reducer, and are called bioemulsifiers. The beststudied biopolymer is emulsan, an LPS isolated from *Acinetobacter calcoaceticus* RAG-1 ATCC 31012, and its molecular weight is on average 1000 kDa [147]. Its surface activity is attributable to the presence of fatty acids in its hydrophobic portion, encompassing 15% of its dry weight, which are bound to the polysaccharide backbone through ester and amide bonds [148]. An additional high-molecular-weight biosurfactant is alasan, a complex of protein and an anionic polysaccharide, with a molecular weight of approximately 1000 kDa, isolated from *Acinetobacter radioresistens* [149].

Yeasts also can produce emulsifiers, which are particularly interesting because of the food-grade status of several species, which allows their use in pharmaceutical and food-related industries. Rufino et al. [114] reported that a polymeric biosurfactant produced by *C. lipolytica* UCP 0988 showed antiadhesive properties of 88% for *S. aureus* and 91% for *Lactobacillus casei*, at a concentration of 0.75 mg/mL, as well as antimicrobial activities (64.6%) at a concentration of 6 mg/mL for *S. mutans* HG985, and *E. coli* inhibition was 5% at 6 mg/mL.

Biosurfactants with antiadhesive properties produced by yeasts have been described by our group. A glycolipid-type biosurfactant produced by the yeast *Trichosporon montevideense* CLOA72 diminished biofilm formation by *C. albicans* CC obtained from the apical tooth canal by up to 85% in polystyrene microplates [52]. It was suggested that this inhibition was the result of interactions of the biosurfactant with the cell or the polystyrene plate surface and resulting modifications of net surface charge. We described also that the incubation of *C. albicans* CC cells in the presence of the glycolipid inhibited germ tube or hypha formation during biofilm formation on the polystyrene plate surface, and the cells showed lower values of cell-surface hydrophobibity (evaluated by microbial adhesion to hydrocarbons test, MATH) compared to cells incubated without the biosurfactant. The inhibition of adhesion can result from both free surfactant molecules and micelles, which may adsorb on the cell surface of microorganisms and solid surfaces, blocking or facilitating microbial adherence to hydrophobic and hydrophilic interfaces [3]. Our group showed also that adhesion of a clinical isolate of *C. albicans* cells to epithelial buccal cells was reduced up to 87.4% with the use of biosurfactant at 16 mg/mL concentration. Measurements of the  $\zeta$  potential of the cell surface showing an alteration of net surface charge ranged from -20.1 to -8.3 mV during the titration with biosurfactant.

## 15.5 Potential Use of Biosurfactants in Nanotechnology

#### 15.5.1 Functionalized Nanoparticles

Nanoparticles are made of various inorganic or organic materials and range from 1 to 100 nm in size. Compared to bulk materials, nanoparticles present improved characteristics that might be of interest for several practical fields, such as optical, magnetic, and electric behavior under specific conditions, beyond their use as vectors for delivering substances to a specific target, such as drugs to their receptors [150]. Functionalization is the attachment of chemical (functional) groups to a polymer, material, or any other possible target, using routes of chemical synthesis. The mechanism by which nano-particles reach functionalization is dependent on the functional group attached to them, and on the aimed at structure—target interaction. Despite several controversies on the biological context of nanoparticles, functionalization can improve the recognition (by any biological or chemical entity), vectorization, and distribution of nanoparticles, and potential toxic effects for humans, animals, and the environment might be reduced [151,152].

One might ask, why use biosurfactants to functionalize nanoparticles? The answer relies first on the efficiency of stabilization and dispersion compared to other chemical synthesis strategies and on applied biocompatibility and biodegradability concepts: the use of biosurfactants can effectively improve the desired characteristics of nanoparticles, and this complex can exist in contact with biological tissues or systems, and the environment, without causing significant injury, harm, or damage to them. Biosurfactants also reduce the formation of other aggregates than vesicles owing to electrostatic forces of attraction and provide standardized morphology of nanoparticles. Thus, it is of interest to replace chemicals that often offer compatibility risks and environmental impacts with these compounds [51,153].

The activity of  $AgNO_3$  combined with the LP biosurfactant V9T14 from *B. licheniformis* was evaluated against a preformed *E. coli* biofilm on the Calgary Biofilm Device [8]. The combined use resulted in a reduction in the quantity of silver used to achieve greater antimicrobial impact. The concentrations of silver in the silver—biosurfactant solutions were 129- to 258-fold less than the concentrations needed when silver was used alone.

Silver nanoparticles (AgNPs) functionalized with Kocuran, an exopolysaccharide produced by *Kocuria rosea* strain BS-1 with repeating monosaccharide residues of D-glucose, D-mannose, D-galactose, and D-glucuronic acid (51.2 kDa), showed antimicrobial activity against *S. aureus* and *E. coli* [154]. The cell death was mainly due to hydroxyl radical induction and depletion of NADH. The improved antimicrobial activity of Kocuran-capped AgNPs may be credited to the surface functionalization with the EPS sugar residues, which makes possible the distribution and diffusion of glyconanoparticles into the cell wall, and the structure–activity relationship. The particles also inhibited the biofilm development by these bacteria in Kocuran-capped AgNP-coated silicone urethral catheters.

Reddy et al. [155] functionalized gold nanoparticles with surfactin and investigated the effects of proton concentrations and temperature on the morphology of these nanoparticles. Nanoparticles synthesized at pH 7 and 9 remained stable for 2 months, whereas aggregates were stable for only 24 h, at pH 5. At pH 7, nanoparticles presented a homogeneous distribution of shape and size. At pH 5, they were polydispersed, and at pH 9 they were anisotropic. The nanoparticles produced at room temperature were monodispersed and were more uniform compared to those formed at 4°C [155].

Spherical nickel oxide nanoparticles were synthesized by Palanisamy and Raichur [156], using RLs in a microemulsion prepared with *n*-heptane and water. The synthesized nickel particles were crystalline and spherical and presented piled lamellar sheets, as suggested by scanning and transmission electron microscopy. The particle size decreased as pH increased: at pH 11.6, particles were in the range of  $86 \pm 8$  nm, and at pH 12.5 the particles were in the range of  $47 \pm 5$  nm [156].

Purified RLs from *P. aeruginosa* were used to synthesize AgNPs by Kumar et al. [157]. These nanoparticles exhibited antimicrobial activity against both gram-positive and gram-negative pathogens and *C. albicans*, suggesting a broad-spectrum antimicrobial activity. The nanoparticles were spherical in shape and monodispersed, with average particle size of 15.1 nm, and had an adsorption peak at 410 nm [157].

Narayanan et al. [158] used RLs to functionalize ZnS nanoparticles of about 4.5 nm. The particles were stabilized in an aqueous environment and ranged up to 10 nm. The functional groups -OH, -CH2, -CH3, -C=O (carbonyl), -COO- (carboxyl), and -C-O-C- were successfully added to the nanoparticles, which moved from a narrow absorption (at 340 nm) to a broad emission spectrum (450 nm). Although the reason for the broadening of the emission spectra was not clear, this was used to confirm the capping efficiency [158].

Iron oxide nanoparticles were functionalized by Sangeetha et al. [159] with surfactin, RL, polyethylene glycol (PEG), and dextran. Cell viability assays using the L929 mouse fibroblast cell line indicated that pure surfactin, RL, and dextran showed cytotoxicity as the concentration increased; however, pure PEG exhibited biocompatibility at different concentrations. Fourier transform infrared (FTIR) and thermogravimetric analyses confirmed the presence of adsorbed moieties on nanoparticles [159].

#### 15.5.2 Liposome Overview

Liposomes are closed lipid systems that are widely studied in various scientific fields, the pharmaceutical, cosmetics, and food sciences being the most advanced in their exploration. These systems were originally designed for studying the behavior of biological membranes; their biotechnological advantages due to controlled release and entrapment protection were first explored only in the 1970s, by G. Gregoriadis [160]. Controlled release provides a sustainable delivery of the substance of interest, and liposome entrapment has several advantages: bioactive compounds can support the presence of reactive species, exposure to extreme pH and temperature, high ion concentrations, and, often, light exposure. Hydrophilic and hydrophobic molecules can be put in the same vesicle: the first are localized in the aqueous core and the second in the membranes. Different delivery mechanisms, like rupture, permeability, or fusion to other membranes, are dependent on the surface that the liposome will have contact with; nevertheless, this increases even more the versatility of liposomes, considering that the rate of delivery might be different in each delivery mechanism [160,161].

The main components of liposomes comprise phospholipids, sterols, and antioxidants, and they may contain a single lipid membrane (unilamellar) or multiple membranes (multilamellar), depending on how the formulation is designed. Liposome size may range from nanometers to micrometers, and when the size is up to 100 nm, they are called nanoliposomes. The sizing can be achieved by techniques such as sonication and extrusion in polymeric membranes (made of materials such as polycarbonate). Several methods of liposome preparation are well described, such as lipid film hydration, reverse-phase evaporation, and injection into organic solvent systems [160].

Liposome nanotechnology relies largely on the biophysical properties of phospholipids and their interactions with the other components of liposome formulations: when dissolved in water or other polar solvents, phospholipids undergo spontaneous micellization, and then assemble into vesicles, mimicking cellular phospholipid bilayers. Proper combinations of liposome components determine characteristics such as membrane fluidity, allowing conformational flexibility. Loaded molecules become protected from degradation by external factors like free radical species and degrading enzymes [162,163]. This is especially important when liposomes are designed for crossing biological barriers such as tissues and bacterial biofilms.

The encapsulation of natural or synthetic antimicrobial molecules is of great potential to overcome microbial resistance. Synthetic drugs have been successfully encapsulated by several research groups, and strong increases in drug potency are expected and experimentally observed in most of the studies. Nevertheless, as the mechanisms of action of drugs are very specific, not only do antimicrobial drugs show a narrow spectrum of action, but the risk of developing resistance is only delayed and not avoided by the entrapment [6]. Hence, the investigation of antimicrobial alternatives that do not induce resistance is of paramount importance.

As discussed in previous sections, biosurfactants have been explored as new natural antimicrobial molecules with a broad spectrum of action. The entrapment of these molecules in liposomes can enhance their biological activity and also provide even more value to them, such that they might be used in formulations for clinical and nonclinical use against microorganisms. The subject, however, as of this writing, remains poorly investigated.

#### 15.5.2.1 Interactions With Liposome Lipids

Hydrophobic moieties of biosurfactants and other natural antimicrobial compounds produced by bacteria usually comprise unsaturated or saturated hydrocarbon chains or fatty acids. In contrast, the main hydrophilic moieties are positively or negatively charged peptides (cations or anions), generating LPs, or mono-, di-, or polysaccharides, generating glycolipids. Thus, a range of structurally and chemically diverse molecules can be produced by different microorganisms, with different physicochemical properties [6,11].

Some of these molecules tend to aggregate at the interfaces where they are applied, using hydrogen bonds and hydrophobic and van der Waals interactions, forming an interfacial film that affects both the surface energy and the wettability of a surface. They also reduce the interfacial tension on the boundary existing between immiscible liquids, generating emulsions. In general, changes in the rheological behavior of any liquid or emulsified (semisolid) system are expected when biosurfactants are used, given that both micelles and liposomes can be formed. Glycolipids may form spherical micelles or disk-like or rod-like structures. MELs, in particular MEL-A, because of their molecular orientation and balance between hydrophilic and hydrophobic groups, can form vesicles larger than 10  $\mu$ m, regarded as of giant size. On the other hand, RLs show a pH-sensitive alteration of structures: micelles can be formed at pH 6.8–7, lipid particles at pH 6.6–6.2, lamellar structures at pH 6.5–6.0, and 50- to 100-nm vesicles can be formed at pH 5.8–4.3. This conversion can be observed because of the presence of a carboxyl group on the side chain of the molecule [164,165].

Various issues can be raised regarding liposome entrapment of antimicrobial molecules: because the exposed biological features can interfere with the stability of liposome formulations, it is important to understand what events might be involved within phospholipid interactions and how these can interfere in entrapment efficiency. In this context, Deleu et al. [166] used biophysical techniques like cryogenic transmission electron microscopy and differential scanning calorimetry to study the mechanism of membrane fengycin, using monolayers of 1,2-dipalmitoyl-sn-glycero-3perturbation by phosphocholine (DPPC). The mechanism seems to be based on physicochemical properties that provide a concentration-dependent state transition: in low concentrations, fengycin is monomeric and nonperturbing. Its insertion does not affect the phospholipid interfacial assembly, being dispersed as a monomer on the hydrophobic core. At medium concentrations, self-association of fengycin with phospholipids can be observed, affecting the lipid bilayer. At high concentrations, the lipid bilayer is completely disrupted into micelles, and membrane leakage and bioactivity are seen [166].

Using isothermal titration calorimetry, Andrushchenko et al. [167] investigated the thermodynamics of the interaction of the cathelicidin peptides tritrpticin and indolicidin, and five tritrpticin analogues, with large unilamellar vesicles composed of 1-palmitoyl-2-oleoyl-*sn*-glycero-3-phosphoethanolamine (POPE), 1-palmitoyl-2-oleoyl-*sn*-glycero-3-phosphoglycerol (POPG), and 1-palmitoyl-2-oleoyl-*sn*-glycero-3-phosphocholine (POPC). A natural *E. coli* polar lipid extract was also used. All peptides exhibited poor binding to POPC, but presented stronger binding to anionic POPE/POPG and *E. coli* membrane systems. The electrostatic interactions between the highly positively charged peptides and the negatively charged membrane surface of these lipids help to explain this effect [167].

Exploring a similar approach to this question, Deleu et al. [168] investigated the effects of surfactin on membrane structure using fluid-disordered and gel-phase models of membranes, using 1,2-dioleoyl-*sn*-glycero-3-phosphocholine and DPPC lipids, respectively. It was demonstrated that the surfactin concentration is critical for its behavior on the lipid membrane. Surfactin interacts with both the membrane polar heads and the acyl chain region, and the presence of rigid domains plays an essential role in surfactin entrapment. At concentrations below the CMC, surfactin inserted at the boundary between gel and fluid lipid domains, inhibited phase separation, and stiffened the bilayer without inducing any morphological changes in liposomes. At concentrations close to the CMC, phospholipids were solubilized. At higher concentrations, both fluid and rigid bilayers were dissolved into micelles and other structures of a wide size distribution.

Abbasi et al. [169] investigated the interactions of a monorhamnolipid produced by a *Pseudomonas* strain with dielaidoylphosphatidylethanolamine (DEPE) membranes. The authors observed that the amalgamation of monorhamnolipids into DEPE reduced the onset temperature of phase transition. The coexistence of lamellar and hexagonal phases in this complex was observed at 60°C with better miscibility of monorhamnolipids into DEPE is exclusively lamellar. The loss of stability was also demonstrated through shifts in the frequency of DEPE bands in FTIR analysis to higher wave numbers.

#### 15.5.3 Gene Transfection With Liposomes

Ladokhin et al. [170] assessed the interactions of indolicidin with large unilamellar vesicles formed from POPC and POPG. It was observed that indolicidin binds strongly to both neutral POPC and POPG vesicles, but the process was reversible. Fluorescence assays showed that indolicidin was inserted in the bilayer interface and induced membrane permeabilization. Leakage induction was more effective with POPG (anionic rather than neutral) vesicles than with POPC vesicles [170].

Inoh et al. [163] developed different liposome formulations for gene transfection with MEL-A, MEL-B, and MEL-C. Only MEL-A enhanced the efficiency of gene transfection. Maitani et al. [12] prepared liposomes using cholesterol, L-dioleoylphosphatidyl-ethanolamine (DOPE), and two biosurfactants,  $\beta$ -sitosterol  $\beta$ -D-glucoside (Sit-G) and

MEL-A, for transfecting a thymidine kinase gene. Vesicles of about 300 nm were generated, and using a luciferase marker gene and the measurement of the activity of the enzyme, it was demonstrated that Sit-G and MEL liposomes presented higher transfection efficiency compared to conventional liposomes.

MEL-A liposomes were also investigated by Igarashi et al. [171] in combination with DC-cholesterol and DOPE, aiming to increase the transfection efficiency of a plasmid in human cervix carcinoma HeLa cells. Vesicles of 169 nm were generated, and compared to conventional liposomes, MEL-A induced a significantly higher level of gene expression. Confocal microscopy indicated that the MEL-A liposome was widely distributed in the cytoplasm, and flow cytometry assays showed that DNA was strongly detected in the cytoplasm and around the nucleus. Also, Nakanishi et al. [13] showed that MEL-A liposomes remarkably promoted the efficiency of gene transfection into mammalian cultured cells with 1 h or shorter incubation time.

Inoh et al. [172] determined the efficiency of cationic liposomes containing MEL-A for small interfering RNA (siRNA) delivery, to interrupt gene transcription. Protein expression was strongly suppressed at 24 h after target cells were incubated for 30 min with MEL-A liposome-entrapped siRNA. Rapid siRNA delivery into the cytosol caused the suppression of protein expression, as demonstrated by confocal microscopic analysis. This transfection system was superior to commercially available transfection kits. Later, the group of Nakanishi et al. [173] investigated the effect of MEL-A in liposomes for transfection of plasmid DNA. Despite the entrapment efficiency being equal to that of conventional liposomes, the transfection efficiency of MEL-A liposomes was more than 10 times higher than that of conventional vesicles.

## 15.6 Conclusion and Perspectives

Biosurfactants comprise a set of molecules that are of great interest for various areas and represent a potential alternative to synthetic antimicrobial products. There is evidence that events of acquisition of resistance to antimicrobial biosurfactants due to natural selection are unlikely. When entrapped in liposomes, biofilms can be reached and disrupted more efficiently. The controlled release provided by liposomes preserves the chemical stability and antimicrobial activity for prolonged periods compared to free molecules. However, remaining challenges are to overcome the (often) unstable behavior of phospholipid—biosurfactant interactions, and to scale up the manufacturing process. Studies on liposome-entrapped biosurfactants will provide promising alternatives for safe and efficient control of microbial biofilm, such that drug resistance can be better managed than it is nowadays.

## References

 J.D. Desai, I.M. Banat, Microbial production of surfactants and their commercial potential, Microbiology and Molecular Biology Review 61 (1997) 47–64.

- [2] D. Kitamoto, T. Morita, T. Fukuoka, M. Konishi, T. Imura, Self-assembling properties of glycolipid biosurfactants and their potential applications, Current Opinion in Colloid and Interface Sciences 14 (2009) 315.
- [3] T.R. Neu, Significance of bacterial surface-active compounds in interaction of bacteria with interfaces, Microbiology Reviews 60 (1996) 151–166.
- [4] M.E. Davey, N.C. Caiazza, G.A. O'Toole, Rhamnolipid surfactant production affects biofilm architecture in *Pseudomonas* PAO1, Journal of Bacteriology 185 (2003) 1027–1036.
- [5] S.J. Pamp, T. Tolker-Nielsen, Multiple roles of biosurfactants in structural biofilm development by *Pseudomonas aeruginosa*, Journal of Bacteriology 189 (2007) 2531–2539.
- [6] V.L. Santos, M.V. Dias-Souza, Strategies Based in Microbial Enzymes and Surface Actives Compounds Entrapped in Liposomes for Bacterial Biofilms Control, in: NanoBioMaterials in Antimicrobial Therapy (Applications of NanoBioMaterials), vol. VI, 2016, pp. 385–418.
- [7] G. Aleti, A. Sessitsch, G. Brader, Genome mining: prediction of lipopeptides and polyketides from *Bacillus* and related Firmicutes, Computational and Structural Biotechnology Journal 13 (2015) 192–203.
- [8] F. Rivardo, M.G. Martinotti, R.J. Turner, H. Ceri, Synergistic effect of lipopeptide biosurfactant with antibiotics against *Escherichia coli* CFT073 biofilm, International Journal of Antimicrobial Agents 37 (2011) 324–333.
- [9] G. Donelli, I. Francolini, D. Romoli, E. Guaglianone, A. Piozzi, C. Ragunath, J.B. Kaplan, Synergistic activity of dispersin B and cefamandole nafate in inhibiti on of staphylococcal biofilm growth on polyurethanes, Antimicrobial Agents and Chemotherapy 51 (2007) 2733–2740.
- [10] L. Hongxia, P. Houbao, H. Zhinan, F. Guangli, L. Dapeng, The antimicrobial effects and synergistic antibacterial mechanism of the combination of  $\varepsilon$ -Polylysine and nisin against *Bacillus subtilis*, Food Control 47 (2015) 444–450.
- [11] L.R. Rodrigues, Microbial surfactants: fundamentals and applicability in the formulation of nano-sized drug delivery vectors, Journal of Colloid and Interface Science 449 (2015) 304–316.
- [12] Y. Maitani, S. Yano, Y. Hattori, M. Furuhata, K. Hayashi, Liposome vector containing biosurfactantcomplexed DNA as herpes simplex virus thymidine kinase gene delivery system, Journal of Liposome Research 16 (2006) 359–372.
- [13] M. Nakanishi, Y. Inoh, D. Kitamoto, T. Furuo, Nano vectors with a biorurfactant for gene transfection and drug delivery, Journal of Drug Delivery Science and Technology 19 (2009) 165–169.
- [14] H.C. Flemming, J. Wingender, The biofilm matrix, Nature Reviews Microbiolology 8 (2010) 623–633.
- [15] R.M. Donlan, Biofilms: microbial life on surfaces, Emergent Infectious Diseases 8 (2002) 881–890.
- [16] R.D. Monds, G.A. O'Toole, The developmental model of microbial biofilms: ten years of a paradigm up for review, Trends in Microbiology 17 (2009) 73–87.
- [17] L.D. Renner, D.B. Weibel, Physicochemical regulation of biofilm formation, MRS Bulletin 36 (2011) 347–355.
- [18] C. Beloin, A. Roux, J.M. Ghigo, *Escherichia coli* biofilms, Current Topics in Microbiology and Immunology 322 (2008) 249–289.
- [19] P. Bhomkar, W. Materi, V. Semenchenko, D.S. Wishart, Transcriptional response of *E. coli* upon FimH-mediated fimbrial adhesion, Gene Regulation and Systems Biology 4 (2010) 1–17.
- [20] S. Hooshangi, W.E. Bentley, From unicellular properties to multicellular behavior: bacteria quorum sensing circuitry and applications, Current Opinion in Biotechnology 19 (2008) 550–555.

- [21] M.C. Walters, F. Roe, A. Bugnicourt, M.J. Franklin, P.S. Stewart, Contributions of antibiotic penetration, oxygen limitation, and low metabolic activity to tolerance of *Pseudomonas* biofilms to ciprofloxacin and tobramycin, Antimicrobial Agents and Chemotherapy 47 (2003) 317–323.
- [22] D. Davies, Understanding biofilm resistance to antibacterial agents, Nature Reviews Drug Discovery 2 (2003) 114–122.
- [23] K. Sauer, M.C. Cullen, A.H. Rickard, L.A. Zeef, D.G. Davies, P. Gilbert, Characterization of nutrientinduced dispersion in *Pseudomonas* PAO1 biofilm, Journal of Bacteriology 186 (2004) 7312–7326.
- [24] S.H. Hong, J. Lee, T.K. Wood, Engineering global regulator Hha of *Escherichia coli* to control biofilm dispersal, Microbial Biotechnology 3 (2010) 717–728.
- [25] M.C. Rowe, H.L. Withers, S. Swift, Uropathogenic *Escherichia coli* forms biofilm aggregates under iron restriction that disperse upon the supply of iron, FEMS Microbiology Letters 307 (2010) 102–109.
- [26] T.R. De Kievit, R. Gillis, S. Marx, C. Brown, B.H. Iglewski, Quorum-sensing genes in *Pseudomonas* biofilms: their role and expression patterns, Applied Environmental Microbiology 67 (2001) 1865–1873.
- [27] T. Bjarnsholt, P.O. Jensen, M. Burmolle, *Pseudomonas* tolerance to tobramycin, hydrogen peroxide and polymorphonuclear leukocytes is quorum sensing dependent, Microbiology 151 (2005) 373–383.
- [28] M. Allesen-Holm, K.B. Barken, L. Yang, M. Klausen, J.S. Webb, S. Kjelleberg, S. Molin, M. Givskov, T. Tolker-Nielsen, A characterization of DNA release in *Pseudomonas* cultures and biofilms, Molecular Microbiology 59 (2006) 1114–1128.
- [29] M.R. Parsek, P.K. Singh, Bacterial biofilms: an emerging link to disease pathogenesis, Annual Reviews Microbiology 57 (2003) 677–701.
- [30] H. Yasuda, T. Koga, T. Fukuoka, In vitro and in vivo models of bacterial biofilms, Methods in Enzymology 310 (1999) 577–595.
- [31] K.K. Jefferson, D.A. Goldmann, G.B. Pier, Use of confocal microscopy to analyze the rate of vancomycin penetration through *Staphylococcus aureus* biofilms, Antimicrobial Agents and Chemotherapy 49 (2005) 2467–2473.
- [32] R. Singh, P. Ray, A. Das, M. Sharma, Penetration of antibiotics through *Staphylococcus aureus* and *Staphylococcus epidermidis* biofilms, Journal of Antimicrobial Chemotherapy 65 (2010) 1955–1958.
- [33] I. Williams, F. Paul, D. Lloyd, Flow cytometry and other techniques show that *Staphylococcus aureus* undergoes significant physiological changes in the early stages of surface-attached culture, Microbiology 145 (1999) 1325–1333.
- [34] L.P. Girard, H. Ceri, A.P. Gibb, M. Olson, F. Sepandj, MIC versus MBEC to determine the antibiotic sensitivity of *Staphylococcus aureus* in peritoneal dialysis peritonitis, Peritonial Dialysis International 30 (2010) 652–656.
- [35] P.B. Bookstaver, J.C. Williamson, B.K. Tucker, I.I. Raad, R.J. Sherertz, Activity of novel antibiotic lock solutions in a model against isolates of catheter-related bloodstream infections, Annals of Pharmacotherapy 43 (2009) 210–219.
- [36] J. Lönn-Stensrud, M.A. Landin, T. Benneche, F.C. Petersen, A.A. Scheie, Furanones, potential agents for preventing *Staphylococcus* epidermidis biofilm infections? Journal of Antimicrobial Chemotherapy 63 (2009) 309–316.
- [37] C.D. Sumi, W.Y. Byung, I.C. Yeo, Y.T. Hahm, Antimicrobial peptides of the genus *Bacillus*: a new era for antibiotics, Canadian Journal of Microbiology 61 (2014) 93–103.
- [38] F.F. Campos, S. Johann, V.L. Santos, Natural products as a source of potential drugs for the treatment of fungal infections, in: first ed., in: L.V. Berhardt (Ed.), Advances in Medicine and Biology, 39, Nova Science Publishers, New York, 2012, pp. 49–83 (Org.).

- [39] Y. Itoh, X. Wang, B.J. Hinnebusch, J.F. Preston, T. Romeo, Depolymerization of  $\beta$ -1,6-N-acetyl-dglucosamine disrupts the integrity of diverse bacterial biofilms, Journal of Bacteriology 187 (2005) 382–387.
- [40] J.L. Del Pozo, M.S. Rouse, J.N. Mandrekar, M.F. Sampedro, J.M. Steckelberg, R. Patel, Effect of electrical current on the activities of antimicrobial agents against *Pseudomonas* aeruginosa, *Staphylococcus aureus*, and *Staphylococcus epidermidis* biofilms, Antimicrobial Agents and Chemotherapy 53 (2009) 35–40.
- [41] Z. Hazan, J. Zumeris, H. Jacob, H. Raskin, G. Kratysh, M. Vishnia, N. Dror, T. Barliya, M. Mandel, G. Lavie, Effective prevention of microbial biofilm formation on medical devices by low-energy surface acoustic waves, Antimicrobial Agents and Chemotherapy 50 (2006) 4144–4152.
- [42] S. Perni, P. Prokopovich, C. Piccirillo, J. Pratten, I.P. Parkin, M. Wilson, Toluidine blue-containing polymers exhibit potent bactericidal activity when irradiated with red laser light, Journal of Materials Chemistry 19 (2009) 2715–2723.
- [43] M. Alipour, Z.E. Suntres, A. Omri, Importance of DNase and alginate lyase for enhancing free and liposome encapsulated aminoglycoside activity against *Pseudomonas aeruginosa*, Journal of Antimicrobial Chemotherapy 64 (2009) 317–325.
- [44] L. Carson, S.P. Gorman, B.F. Gilmore, The use of lytic bacteriophages in the prevention and eradication of biofilms of *Proteus mirabilis* and *Escherichia coli*, FEMS Immunology and Medical Microbiology 59 (2010) 447–455.
- [45] M. Rai, A. Yadav, A. Gade, Silver nanoparticles as a new generation of antimicrobials, Biotechnology Advancement 27 (2009) 76–83.
- [46] A.S. Monteiro, M.R.Q. Bonfim, V.S. Domingues, A. Corrêa Jr., E.P. Siqueira, C.L. Zani, V.L. Santos, Identification and characterization of bioemulsifier-producing yeasts isolated from effluents of a dairy industry, Bioresource Technology 101 (2010) 5186–5193.
- [47] A.S. Monteiro, V.S. Domingues, M.V.D. Souza, I. Lula, D. Gonçalves, E.P. Siqueira, V.L. Santos, Bioconversion of biodiesel refinery waste in the bioemulsifier by *Trichosporon mycotoxinivorans* CLA2, Biotechnology for Biofuels 5 (2012) 29.
- [48] V.L. Santos, A.S. Monteiro, V.S. Domingues, A.D.L. Júlio, Microbial Bioemulsifiers and environmental applications potential, in: Kylie M. Tannen (Ed.), Emulsification: Processes, New Technology and Current Applications, third ed., Nova Science Publishers, 2013, p. 162 (Org.).
- [49] J.O.P.A. Coutinho, M.P.S. Silva, P.M. Moraes, A.S. Monteiro, J.C.C. Barcelos, E.P. Siqueira, V.L. Santos, Demulsifying properties of extracellular products and cells of *Pseudomonas aeruginosa* MSJ isolated from petroleum-contaminated soil, Bioresource Technology 128 (2013) 646–654.
- [50] S. Cameotra, R. Makkar, Recent applications of biosurfactants as biological and immunological molecules, Current Opinion in Microbiology 7 (2004) 262–266.
- [51] P. Singh, S. Cameotra, Potential applications of microbial surfactants in biomedical sciences, Trends in Biotechnology 22 (2004) 142–146.
- [52] A.S. Monteiro, T.T. Miranda, I. Lula, A.M.L. Denadai, R.D. Sinisterra, M.M. Santoro, V.L. Santos, Inhibition of *Candida albicans* CC biofilms formation in polystyrene plate surfaces by biosurfactant produced by *Trichosporon montevideense* CLOA72, Colloids and Surfaces. B, Biointerfaces 84 (2011) 467–476.
- [53] R.A. Al-Tahhan, T.R. Sandrin, A.A. Bodour, R.M. Maier, Rhamnolipid-induced removal of lipopolysaccharide from *Pseudomonas aeruginosa*: effect on cell surface properties and interaction with hydrophobic substrates, Applied Environmental Microbiology 66 (2000) 3262–3268.
- [54] A. Sotirova, D. Spasova, E. Vasileva-Tonkova, D. Galabova, Effects of rhamnolipid-biosurfactant on cell surface of *Pseudomonas* aeruginosa, Microbiology Research 164 (2009) 297–303.
- [55] Y. Lequette, E.P. Greenberg, Timing and localization of rhamnolipid synthesis gene expression in *Pseudomonas* biofilms, Journal of Bacteriology 187 (2005) 37–44.

- [56] O. Rendueles, L. Travier, P. Latour-Lambert, T. Fontaine, J. Magnus, E. Denamur, J.M. Ghigo, Screening of *Escherichia coli* species biodiversity reveals new biofilm-associated anti-adhesion polysaccharides, Molecular Biology 2 (2011) e00043–e000411.
- [57] J. Valle, S. Da Re, N. Henry, T. Fontaine, D. Balestrino, P. Latour-Lambert, J.M. Ghigo, Broadspectrum biofilm inhibition by a secreted bacterial polysaccharide, Proceedings of the National Academy of Sciences 103 (2006) 12558–12563.
- [58] P. Jiang, J. Li, F. Han, G. Duan, X. Lu, Y. Gu, W. Yu, Anti-biofilm activity of an exopolysaccharide from marine bacterium *Vibrio* sp. QY101, PLoS One 6 (2011) e18514.
- [59] L. Thimon, F. Peypoux, J. Wallach, G. Michel, Effect of the lipopeptide antibiotic iturin A, on morphology and membrane ultrastructure of yeast cells, FEMS Microbiology Letters 128 (1995) 101–106.
- [60] L. Gakhar, J.A. Bartlett, J. Penterman, D. Mizrachi, P.K. Singh, R.K. Mallampalli, S. Ramaswamy, P. B. McCray, PLUNC is a novel airway surfactant protein with anti-biofilm activity, PLoS One 5 (2010) e9098.
- [61] M.A. Díaz De Rienzo, I.M. Banat, B. Dolman, J. Winterburn, P.J. Martin, Sophorolipid biosurfactants: possible uses as antibacterial and anti-biofilm agent, New Biotechnology 32 (2015) 720–726.
- [62] S.C. Lee, S.H. Kim, I.H. Park, S.Y. Chung, Y.L. Choi, Isolation and structural analysis of bamylocin A, novel lipopeptide from *Bacillus amyloliquefaciens* LP03 having antagonistic and crude oilemulsifying activity, Archives of Microbiology 188 (2007) 307–312.
- [63] R. Rautela, A.K. Singh, A. Shukla, S.S. Cameotra, Lipopeptides from *Bacillus* strain AR2 inhibits biofilm formation by *Candida albicans*, Antonie van Leeuwenhoek 105 (2014) 809–821.
- [64] M. Ongena, P. Jacques, *Bacillus* lipopeptides: versatile weapons for plant disease biocontrol, Trends in Microbiology 16 (2008) 115–125.
- [65] Y. Hathout, Y.P. Ho, V. Ryzhov, P. Demirev, C. Fenselau, Kurstakins: a new class of lipopeptides isolated from *Bacillus thuringiensis*, Journal of Natural Products 63 (2000) 1492–1496.
- [66] G. Hagelin, B. Indrevoll, T. Hoeg-Jensen, Use of synthetic analogues in confirmation of structure of the peptide antibiotics maltacines, International Journal of Mass Spectrometry 268 (2007) 254–264.
- [67] P.J. Bergen, C.B. Landersdorfer, J. Zhang, M. Zhao, H.J. Lee, R.L. Nation, J. Li, Pharmacokinetics and pharmacodynamics of 'old' polymyxins: what is new? Diagnostic Microbiology and Infectious Diseases 74 (2012) 213–223.
- [68] K. Arima, A. Kahinuma, G. Tamura, Surfactin, a crystalline peptide lipid surfactnt produced by *Bacillus subtilis*: isolation, characterization and its inhibition of fibrin clot formation, Biochemical and Biophysical Research Communications 31 (1968) 488–494.
- [69] F. Peypoux, J.M. Bonmatin, J. Wallach, Recent trends in the biochemistry of surfactin, Applied Microbiology and Biotechnology 51 (1999) 554–563.
- [70] L. D'Auria, M. Deleu, S. Dufour, M.P. Mingeot-Leclercq, D. Tyteca, Surfactins modulate the lateral organization of fluorescent membrane polar lipids: a new tool to study drug: membrane interaction and assessment of the role of cholesterol and drug acyl chain length, Biochimica et Biophysica Acta 1828 (2013) 2064–2073.
- [71] J.R. Mireles, A. Toguchi, R.M. Harshey, *Salmonella enterica* serovar Typhimurium swarming mutants with altered biofilm-forming abilities: surfactin inhibits biofilm formation, Journal of Bacteriology 183 (2001) 5848–5854.
- [72] F. Rivardo, R.J. Turner, G. Allegrone, H. Ceri, M.G. Martinotti, Anti-adhesion activity of two biosurfactants produced by *Bacillus* spp. prevents biofilm formation of human bacterial pathogens, Applied Microbiology and Biotechnology 83 (2009) 541–553.

- [73] R. Brasseur, N. Braun, K. El Kirat, M. Deleu, M.P. Mingeot-Leclercq, Y.F. Dufrene, The biologically important surfactin lipopeptide induces nanoripples in supported lipid bilayers, Langmuir 23 (2007) 9769–9772.
- [74] A.E. Zeraik, M. Nitschke, Biosurfactants as agents to reduce adhesion of pathogenic bacteria to polystyrene surfaces: effect of temperature and hydrophobicity, Current Microbiology 61 (2010) 554–559.
- [75] D.C. Sabaté, M.C. Audisio, Inhibitory activity of surfactin, produced by different *Bacillus subtilis* subsp. subtilis strains, against *Listeria monocytogenes* sensitive and bacteriocin-resistant strains, Microbiology Research 168 (2013) 125–129.
- [76] M.H. Xu, Y.J. Rong, M.X. Zhao, B. Song, Z.M. Chi, Antibacterial activity of the lipopetides produced by *Bacillus amyloliquefaciens* M1 against multidrug-resistant *Vibrio* spp. isolated from diseased marine animals, Applied Microbiology and Biotechnology 98 (2014) 127–136.
- [77] F.L. Fassi, H. Wroblewski, A. Blanchard, Activities of antimicrobial peptides and synergy with enrofloxacin against *Mycoplasma pulmonis*, Antimicrobial Agents and Chemotherapy 51 (2007) 468–474.
- [78] A.L. Moyne, R. Shelby, T.E. Cleveland, Bacillomycin D: an iturin with antifungal activity against *Aspergillus flavus*, Journal of Applied Microbiology 90 (2001) 622–629.
- [79] A. Roy, D. Mahata, D. Paul, S. Korpole, O.L. Franco, S.M. Mandal, Purification, biochemical characterization and self-assembled structure of a fengycin-like antifungal peptide from *Bacillus thuringiensis* strain SM1, Frontiers in Microbiology 4 (2013) 332.
- [80] S. Steller, J. Vater, Purification of the fengycin synthetase multienzyme system from *Bacillus subtilis* b213, Journal of Chromatography B Biomedical Sciences Applied 737 (2000) 267–275.
- [81] R.A. Nastro, A. Arguelles-Arias, M. Ongena, A. Di Costanzo, M. Trifuoggi, Antimicrobial activity of *Bacillus amyloliquefaciens* ANT1 toward pathogenic bacteria and mold. Effects on biofilm formation, Probiotics and Antimicrobial Proteins 5 (2013) 252–258.
- [82] J. Li, R.L. Nation, J.D. Turnidge, Colistin: the re-emerging antibiotic for multidrug-resistant Gramnegative bacterial infections, Lancet Infectious Diseases 6 (2006) 589–601.
- [83] P.J. Bergen, J. Li, C.R. Rayner, Colistin methanesulfonate is an inactive prodrug of colistin against *Pseudomonas aeruginosa*, Antimicrobial Agents and Chemotherapy 50 (2006) 1953–1958.
- [84] R.E.W. Hancock, Peptide antibiotics, Lancet 349 (1997) 418-422.
- [85] T. Mogi, Y. Murase, M. Mori, Polymyxin B identified as an inhibitor of alternative NADH dehydrogenase and malate: quinone oxidoreductase from the Gram-positive bacterium *Mycobacterium smegmatis*, Journal of Biochemistry 146 (2009) 491–499.
- [86] T.R. Sampson, X. Liu, M.R. Schroeder, C.S. Kraft, E.M. Burd, D.S. Weiss, Rapid killing of *Acinetobacter baumannii* by polymyxins is mediated by a hydroxyl radical death pathway, Antimicrobial Agents and Chemotherapy 56 (2012) 5642–5649.
- [87] G.A. Quinn, A.P. Maloy, S. McClean, B. Carney, J.W. Slater, Lipopeptide biosurfactants from *Paenibacillus polymyxa* inhibit single and mixed species biofilms, Biofouling 28 (2012) 1151–1166.
- [88] G. Doring, S.P. Conway, H.G.M. Heijerman, M.E. Hodson, N. Hoiby, A. Smyth, D.J. Touw, Antibiotic therapy against *Pseudomonas* in cystic fibrosis: a European consensus, European Respiratory Journal 16 (2000) 749–767.
- [89] K. Tarquinio, K. Confreda, J. Shurko, K. LaPlante, Activities of Tobramycin and Polymyxin E against Pseudomonas aeruginosa Biofilm-Coated Medical Grade Endotracheal Tubes. Antimicrobial Agents Chemotherapy 58 (2014) 1723–1729.
- [90] O. Nybroe, J. Sorensen, Production of cyclic lipopeptides by fluorescent pseudomonads., in: J-L Ramos (Ed.), Pseudomonas, biosynthesis of macromolecules and molecular metabolism., Kluwer Academic/ Plenum Publishers, New York, U.S.A, 2004, pp. 147–172.

- [91] N. Roongsawang, K. Hase, M. Haruki, T. Imanaka, M. Morikawa, S. Kanaya, Cloning and characterization of the gene cluster encoding arthrofactin synthetase from *Pseudomonas* sp. MIS38, Chemical Biology 10 (2003) 869–880.
- [92] I. Kuiper, E.L. Lagendijk, R. Pickford, J.P. Derrick, G.E.M. Lamers, J.E. Thomas-Oates, B.J.J. Lugtenberg, G.V. Bloemberg, Characterization of two *Pseudomonas putida* lipopeptide bio-surfactants, putisolvin I and II, which inhibit biofilm formation and break down existing biofilms, Molecular Microbiology 51 (2004) 97–113.
- [93] M. Kruijt, H. Tran, J.M. Raaijmakers, Functional, genetic and chemical characterization of biosurfactants produced by plant growth-promoting *Pseudomonas putida* 267, Journal of Applied Microbiology 107 (2009) 546–556.
- [94] T. Janek, M. Łukaszewicz, A. Krasowska, Antiadhesive activity of the biosurfactant pseudofactin II secreted by the Arctic bacterium *Pseudomonas fluorescens* BD5, BMC Microbiology 12 (2012) 24.
- [95] D. Sinnaeve, C. Michaux, J. Van Hemel, Structure and X-ray conformation of pseudodesmins A and B, two new cyclic lipodepsipeptides from *Pseudomonas* bacteria, Tetrahedron 65 (2009) 4173–4181.
- [96] A.D. Berti, N.J. Greve, Q.H. Christensen, M.G. Thomas, Identification of a biosynthetic gene cluster and the six associated lipopeptides involved in swarming motility of *Pseudomonas syringae* pv. tomato DC3000, Journal of Bacteriology 189 (2007) 6312–6323.
- [97] A. Fiore, L. Mannina, A.P. Sobolev, A.M. Salzano, A. Scaloni, I. Grgurina, M.R. Fullone, M. Gallo, C. Swasey, V. Fogliano, J.Y. Takemoto, Bioactive lipopeptides of ice-nucleating snow bacterium *Pseudomonas syringae* strain 31R1, FEMS Microbiology Letters 286 (2008) 158–165.
- [98] F.G. Jarvis, M.J. Johnson, A glyco-lipid produced by *Pseudomonas aeruginosa*, American Chemical Society 71 (1949) 4124–4126.
- [99] D. Kitamoto, K. Haneishi, T. Nakahara, T. Tabuchi, Production of mannosylerythritol lipids by *Candida antarctica* from vegetable oils, Agricultural Biological Chemistry 54 (1990) 37–40.
- [100] S. Lang, F. Wagner, Structure and properties of biosurfactants, in: N. Kosaric, W.L. Cairns, N.C.C. Gray (Eds.), Biosurfactants and Biotechnology, vol. 25, Dekker, New York, 1987, pp. 21–47. Surfactant science series.
- [101] D.G. Cooper, D.A. Paddock, Production of a biosurfactant from *Torulopsis bombicola*, Applied Environmental Microbiology 47 (1984) 173–176.
- [102] S.G. Costa, E. Deziel, F. Lepine, Characterization of rhamnolipid production by *Burkholderia glumae*, Letters in Applied Microbiology 53 (2011) 620–627.
- [103] A.M. Abdel-Mawgoud, F. Lepine, E. Deziel, Rhamnolipids: diversity of structures, microbial origins and roles, Applied Microbiology Biotechnology 86 (2010) 1323–1336.
- [104] B.R. Boles, M. Thoendel, P.K. Singh, Rhamnolipids mediate detachment of *Pseudomonas* from biofilms, Molecular Microbiology 57 (2005) 1210–1223.
- [105] E. Haba, A. Pinazo, O. Jauregui, Physicochemical characterization and antimicrobial properties of rhamnolipids produced by *Pseudomonas* 47T2 NCBIM 40044, Biotechnology and Bioengineering 81 (2003) 316–332.
- [106] A. Abdel-Megeed, A.N. Al-Rahmah, A.A. Mostafa, Biochemical characterization of anti-microbial activity of glycolipids produced by *Rhodococcus erythropolis*, Pakistan Journal of Botany 43 (2011) 1323–1334.
- [107] L.V. Araujo, F. Abreu, U. Lins, L.M. Melo, M. Santa Anna, D.M. Nitschke, Rhamnolipid and surfactin inhibit *Listeria monocytogenes* adhesion, Food Research International 44 (2011) 481–488.
- [108] Y. Irie, G.A. O'Toole, M.H. Yuk, Pseudomonas rhamnolipids disperse Bordetella bronchiseptica biofilms, FEMS Microbiology Letters 250 (2005) 237–243.

- [109] L.R. Rodrigues, I.M. Banat, H.C. van der Mei, J.A. Teixeira, R. Oliveira, Interference in adhesion of bacteria and yeasts isolated from explanted voice prostheses to silicone rubber by rhamnolipid biosurfactants, Journal of Applied Microbiology 100 (2006a) 470–480.
- [110] M. Sodagari, H. Wang, B.M. Newby, L.K. Ju, Effect of rhamnolipids on initial attachment of bacteria on glass and octadecyltrichlorosilane-modified glass, Colloids and Surfaces B – Biointerfaces 1 (2013) 121–128.
- [111] D.H. Dusane, S. Dam, Y.V. Nancharaiah, A.R. Kumar, V.P. Venugopalan, S.S. Zinjarde, Disruption of *Yarrowia lipolytica* biofilms by rhamnolipid biosurfactant, Aquatic Biosystem 8 (2012) 17.
- [112] T. Meylheuc, M. Renault, M.N. Bellon-Fontaine, Adsorption of a biosurfactant on surfaces to enhance the disinfection of surfaces contaminated with Listeria monocytogenes, International Journal of Food Microbiology 109 (2006) 71–78.
- [113] J.M. Luna, R.D. Rufino, L.A. Sarubbo, L.R.M. Rodrigues, J.A.C. Teixeira, G.M.C. Takaki, Evaluation antimicrobial and antiadhesive properties of the biosurfactant lunasan produced by *Candida sphaerica* UCP 0995, Current Microbiology 62 (2011) 1527–1534.
- [114] R.D. Rufino, J.M. Luna, L.A. Sarubbo, L.R. Rodrigues, J.A. Teixeira, G.M. Campos-Takaki, Antimicrobial and anti-adhesive potential of a biosurfactant Rufisan produced by *Candida lipolytica* UCP 0988, Colloids and Surfaces B – Biointerfaces 84 (2011) 1–5.
- [115] R. Haskins, J.A. Thorn, B. Boothroyd, Biochemistry of ustilaginales. XI. Metabolic products of *Ustilago zeae* in submerged culture, Canadian Journal of Microbiology 1 (1955) 749–756.
- [116] G. Deml, T. Anke, F. Oberwinkler, W. Max, Steglich Schizonellin A and B, new glycolipids from *Schizonella melanogramma*, Phytochemistry 191 (1980) 83–87.
- [117] M. Kurz, C. Eder, D. Isert, Z. Li, E.F. Paulus, M. Schiell, L. Toti, L. Vértesy, J. Wink, G. Seibert, Ustilipids, acylated β- D-mannopyranosyl D-erythritols from Ustilago maydis and Geotrichum candidum, Journal of Antibiotics 56 (2003) 91–101.
- [118] D. Kitamoto, H. Yanagishita, T. Shinbo, T. Nakane, C. Kamisawa, T. Nakahara, Surface active properties and antimicrobial activities of manosylerythritol lipids as biosurfactants produced by *Candida antarctica*, Journal of Biotechnology 29 (1993) 91–96.
- [119] U. Rau, L.A. Nguyen, S. Schulz, V. Wray, M. Nimtz, H. Roeper, H. Koch, S. Lang, Formation and analysis of mannosylerythritol lipids secreted by *Pseudozyma aphidis*, Applied Microbiology and Biotechnology 66 (2005) 551–593.
- [120] B. Mimee, C. Labbé, R. Pelletier, R.R. Bélanger, Antifungal activity of flocculosin, a novel glycolipid isolated from *Pseudozyma flocculosa*, Antimicrobial Agents and Chemotherapy 49 (2005) 1597–1599.
- [121] B. Mimee, R. Pelletier, R.R. Bélanger, In vitro antibacterial activity and antifungal mode of action of flocculosin, a membrane-active cellobiose lipid, Journal of Applied Microbiology 107 (2009) 989–996.
- [122] T. Kulakovskaya, A. Shashkov, E. Kulakovskaya, W. Golubev, A. Zinin, Y. Tsvetkov, A. Grachev, N. Nifantiev, Extracellular cellobiose lipid from yeast and their analogues: structures and fungicidal activities, Journal of Oleoginous Science 58 (2009) 133–140.
- [123] I.N.A. van Bogaert, C.D. Saerens, D. Muynck, W. Develter, E. Soetaert, E.J. Vandamme, Microbial production and application of sophorolipids, Applied Microbiology Biotechnology 76 (2007) 23–34.
- [124] B. Seung-Hak, X.X. Sun, Y.J. Lee, S.Y. Wang, K.N. Han, J.K. Choi, Mitigation of harmful algae blooms by sophorolipid, Journal of Microbiology Biotechnology 13 (2003) 651.
- [125] K. Hommel, L. Weber, A. Weiss, U. Himmelreich, O. Rilke, H.P. Kleber, Production of sophorose lipid by *Candida (Torulopsis) apicola* grown on glucose, Journal of Biotechnology 33 (1994) 147–155.
- [126] J.N. Sleiman, S.A. Kohlhoff, P.M. Roblin, S. Wallner, R. Gross, M.R. Hammerschlag, M.E. Zenilman, M.H. Bluth, Sophorolipids as antibacterial agents, Annals of Clinical & Laboratory Science 39 (2009) 60–63.

- [127] K. Joshi-Navare, A. Prabhune, A biosurfactant sophorolipid acts in synergy with antibiotics to enhance their efficiency, BMC Research International 1 (2013) 8.
- [128] D.H. Kitamoto, T. Isoda, Functions and potential applications of glycolipids biosurfactants from energy – saving materials to gene delivery carriers, Journal of Bioscience and Bioengineering 94 (2002) 187–201.
- [129] S. Lang, J.C. Philp, Surface-active lipids in *Rhodococci*, Antonie van Leeuwenhoek International Journal of General and Molecular Microbiology 74 (1998) 59–70.
- [130] E. Vollbrecht, U. Rau, S. Lang, Microbial conversion of vegetable oils into surface-active di-, tri-, and tetrasaccharide lipids (biosurfactants) by the bacterial strain *Tsukamurella* spec, Lipids 101 (1999) 389–394.
- [131] A. Ortiz, J.A. Teruel, M.J. Espuny, A. Marques, Interactions of a bacterial biosurfactant trehalose lipid with phosphatidylserine membranes, Chemistry and Physics of Lipids 158 (2009) 46–53.
- [132] P. Saravanakumari, K. Mani, Structural characterization of a novel xylolipid biosurfactant from *Lactococcus lactis* and analysis of antibacterial activity against multi-drug resistant pathogens, Bioresource Technology 101 (2010) 8851–8854.
- [133] L. Olaf, P. Olof, W. Victor, T. Harukuni, L. Siegmund, Production and modification of bioactive biosurfactants, Process Biochemistry 41 (2006) 2138–2145.
- [134] G.S. Kiran, B. Sabarathnam, J. Selvin, Biofilm disruption potential of a glycolipid biosurfactant from marine *Brevibacterium casei*, FEMS Immunology and Medical Microbiology 59 (2010) 432–438.
- [135] A.K. Pradhan, N. Pradhan, L.B. Sukla, P.K. Panda, B.K. Mishra, Inhibition of pathogenic bacterial biofilm by biosurfactant produced by *Lysinibacillus fusiformis* S9, Bioprocess Biosystems Engineering 37 (2014) 139–149.
- [136] H.D. Devendra, Y.V. Vinay, V.P. Nancharaiah, R.K. Venugopala, S.Z. Ameeta, Anti-biofilm potential of a glycolipid surfactant produced by a tropical marine strain of *Serratia marcescens*, Biofouling 27 (2011) 645–654.
- [137] M.M.C. Velraeds, H.C. Van Der Mei, G. Reid, H.J. Busscher, Inhibition of initial adhesion of uropathogenic *Enterococcus faecalis* by biosurfactants from *Lactobacillus* isolates, Applied Environmental Microbiology 62 (1996) 1958–1963.
- [138] H.J. Busscher, C. Van hoogmoed, G.I. Geertsema-Doornbusch, M. Van der Kuijl- Booij, H.C. van der Mei, *Streptococcus thermophilus* and its biosurfactants inhibit adhesion by *Candida* spp. on silicone rubber, Applied Environmental Microbiology 63 (1997) 3810–3817.
- [139] B.S. Gan, J. Kim, G. Reid, P. Cadieux, J.C. Howard, *Lactobacillus fermentum* RC-14 inhibits *Staphylococcus aureus* infection of surgical implants in rats, Journal of Infectious Diseases 185 (2002) 1369–1372.
- [140] L. Rodrigues, H.C. van der Mei, J. Teixeira, R. Oliveira, Influence of biosurfactants from probiotic bacteria on formation of biofilms on voice prosthesis, Applied Environmental Microbiology 70 (2004) 4408–4410.
- [141] E.J. Gudiña, J.A. Teixeira, L.R. Rodrigues, Isolation and functional characterization of a biosurfactant produced by *Lactobacillus paracasei*, Colloids and Surfaces B – Biointerfaces 76 (2010) 298–304.
- [142] L. Rodrigues, J.A. Teixeira, H.C. Van der Mei, R. Oliveira, Isolation and partial characterization of a biosurfactant produced by *Streptococcus thermophilus* A, Colloids and Surfaces B – Biointerfaces 53 (2006b) 105–112.
- [143] L. Rodrigues, H. van der Mei, J.A. Teixeira, R. Oliveira, Biosurfactant from *Lactococcus lactis* 53 inhibits microbial adhesion on silicone rubber, Applied Environmental Microbiology 66 (2004b) 306–311.

- [144] A. Tahmourespour, R. Salehi, R.K. Kermanshahi, *Lactobacillus acidophilus* derived biosurfactant effect on GTFB and GTFC expression level in *Streptococcus mutans* biofilm cells, Brazilian Journal of Microbiology 42 (2010) 330–339.
- [145] K. Sambanthamoorthy, X. Feng, R. Patel, S. Patel, C. Paranavitana, Antimicrobial and anti-biofilm potential of biosurfactants isolated from lactobacilli against multi-drug-resistant pathogens, BMC Microbiology 14 (2014) 197.
- [146] E.Z. Ron, E. Rosenberg, Natural roles of biosurfactants, Environmental Microbiology 3 (2001) 229–236.
- [147] E. Rosenberg, A. Zuckerberg, C. Rubinovitz, D.L. Gutnick, Emulsifier of *Arthrobacter* RAG-1: isolation and emulsifying properties, Applied Environmental Microbiology 37 (1979) 402–440.
- [148] I. Belsky, D.L. Gutnick, E. Rosenberg, Emulsifier of Arthrobacter RAG-1: determination of emulsifier-bound fatty acids, FEBS Letters 101 (1979) 175–178.
- [149] S. Navon-Venezia, E. Banin, E.Z. Ron, E. Rosenberg, The bioemulsifier alasan: role of protein in maintaining structure and activity, Applied Microbiology and Biotechnology 49 (1998) 382–438.
- [150] A.S. Reddy, C.Y. Chen, C.C. Chen, J.S. Jean, H.R. Chen, M.J. Tseng, C.W. Fan, J.C. Wang, Biological synthesis of gold and silver nanoparticles mediated by the bacteria *Bacillus subtilis*, Journal of Nanoscience Nanotechnology 10 (2010) 6567–6574.
- [151] A. Singh, J.D. van Hamme, O.P. Ward, Surfactants in microbiology and biotechnology: part 2. Application aspects, Biotechnology Advances 25 (2007) 99–121.
- [152] B.R. Singh, S. Dwivedi, A.A. Al-Khedhairy, J. Musarrat, Synthesis of stable cadmium sulfide nanoparticles using surfactin produced by *Bacillus amyloliquifaciens* strain KSU-109, Colloids and Surfaces B – Biointerfaces 85 (2011) 207–213.
- [153] S. Singh, V. D'Britto, A.A. Prabhune, C.V. Ramana, A. Dhawan, B.L.V. Prasad, A cytotoxic and genotoxic assessment of glycolipid-reduced and capped gold and silver nanoparticles, New Journal of Chemistry 34 (2010) 294–301.
- [154] C.G. Kumar, P. Sujitha, Kocuran, an exopolysaccharide isolated from *Kocuria rosea* strain BS-1 and evaluation of its in vitro immunosuppression activities, Enzyme Microbial Technology 5 (2014) 113–120.
- [155] A.S. Reddy, C.Y. Chen, C.C. Chen, J.S. Jean, C.W. Fan, H.R. Chen, J.C. Wang, V.R. Nimje, Synthesis of gold nanoparticles via an environmentally benign route using a biosurfactant, Journal of Nanoscience and Nanotechnology 9 (2009) 6693–6699.
- [156] P. Palanisamy, A.M. Raichur, Synthesis of spherical NiO nanoparticles through a novel biosurfactant mediated emulsion technique, Materials Science and Engineering 29 (2009) 199–204.
- [157] C.G. Kumar, S.K. Mamidyala, B. Das, B. Sridhar, G.S. Devi, M.S. Karuna, Synthesis of biosurfactantbased silver nanoparticles with purified rhamnolipids isolated from *Pseudomonas aeruginosa* BS-161R, Journal of Microbiology and Biotechnology 20 (2010) 1061–1068.
- [158] J. Narayanan, R. Ramji, H. Sahu, P. Gautam, Synthesis, stabilisation and characterisation of rhamnolipid-capped ZnS nanoparticles in aqueous medium, IET Nanobiotechnology 4 (2010) 29–34.
- [159] J. Sangeetha, S. Thomas, J. Arutchelvi, M. Doble, J. Philip, Functionalization of iron oxide nanoparticles with biosurfactants and biocompatibility studies, Journal of Biomedical Nanotechnology 9 (2013) 751–764.
- [160] G. Gregoriadis, Liposome Technology, third ed., CRC Press, Boca Raton, vol. I (2007) 346.
- [161] A. Koçer, Functional liposomal membranes for triggered release, Methods in Molecular Biology 605 (2010) 243–255.
- [162] M. Liu, M. Li, G. Wang, X. Liu, D. Liu, H. Peng, Q. Wang, Heart-targeted nanoscale drug delivery systems, Journal of Biomedical Nanotechnology 10 (2014) 2038–2062.

- [163] X. Wang, Y. Song, Y. Su, Q. Tian, B. Li, J. Quan, Y. Deng, Are PEGylated liposomes better than conventional liposomes? A special case for vincristine, Drug Delivery 29 (2015) 1–9.
- [164] Y. Inoh, D. Kitamoto, N. Hirashima, M. Nakanishi, Biosurfactant MEL-A dramatically increases gene transfection via membrane fusion, Journal of Controlled Release 94 (2004) 423–431.
- [165] Y. Ueno, Y. Inoh, T. Furuno, N. Hirashima, D. Kitamoto, M. Nakanishi, NBD-conjugated biosurfactant (MEL-A) shows a new pathway for transfection, Journal of Controlled Release 123 (2007) 247–253.
- [166] M. Deleu, M. Paquot, T. Nylander, Effect of fengycin, a lipopeptide produced by *Bacillus subtilis*, on model biomembranes, Biophysics Journal 94 (2008) 2667–2679.
- [167] V.V. Andrushchenko, M.H. Aarabi, L.T. Nguyen, E.J. Prenner, H.J. Vogel, Thermodynamics of the interactions of tryptophan-rich cathelicidin antimicrobial peptides with model and natural membranes, Biochimica et Biophysica Acta 1778 (2008) 1004–1014.
- [168] M. Deleu, J. Lorent, L. Lins, R. Brasseur, N. Braun, K. El Kirat, T. Nylander, Y.F. Dufrêne, M.P. Mingeot-Leclercq, Effects of surfactin on membrane models displaying lipid phase separation, Biochimica et Biophysica Acta 1828 (2013) 801–815.
- [169] H. Abbasi, F.J. Aranda, K.A. Noghabi, A. Ortiz, A bacterial monorhamnolipid alters the biophysical properties of phosphatidylethanolamine model membranes, Biochimica et Biophysica Acta 1828 (2013) 2083–2090.
- [170] A.S. Ladokhin, M.E. Selsted, S.H. White, Bilayer interactions of indolicidin, a small antimicrobial peptide rich in tryptophan, proline, and basic amino acids, Biophysical Journal 72 (2 Pt 1) (1997) 794–805.
- [171] S. Igarashi, Y. Hattori, Y. Maitani, Biosurfactant MEL-A enhances cellular association and gene transfection by cationic liposome, Journal of Controlled Release 112 (2006) 362–368.
- [172] Y. Inoh, T. Furuno, N. Hirashima, D. Kitamoto, M. Nakanishi, Rapid delivery of small interfering RNA by biosurfactant MEL-A-containing liposomes, Biochemical and Biophysical Research Communications 414 (2011) 635–640.
- [173] M. Nakanishi, Y. Inoh, T. Furuno, New transfection agents based on liposomes containing biosurfactant MEL-A, Pharmaceutics 5 (2013) 411–420.

## 16

# Bacteriocins as Antimicrobial and Antibiofilm Agents

V.L. Santos\*, R.M. Nardi Drummond, M.V. Dias-Souza UNIVERSIDADE FEDERAL DE MINAS GERAIS, BELO HORIZONTE, MINAS GERAIS, BRAZIL

## 16.1 Introduction

Biofilms are the predominant mode of growth for bacteria in most natural, industrial, and clinical environments, consisting of densely packed multispecies populations of cells encased in a self-synthesized polymeric matrix and attached to a tissue or surface [1]. Biofilms have been implicated in chronic disease (tooth decay, periodontitis, otitis media, necrotizing fasciitis, biliary tract infection, bacterial prostatitis, osteomyelitis, native valve endocarditis, cystic fibrosis pneumonia, and melioidosis), health careassociated infections (pneumonia, sutures, exit sites, urinary catheter cystitis), dissemination of community-acquired diseases, and effective hygienic processing and increased failure rate of antiinfective therapy, in addition to food-borne diseases. This surface-adhered complex bacterial community is highly resistant to antimicrobials compared to their planktonic analogues. Thus, the treatment of biofilms with antibiotics or other biocides is usually ineffective at eradicating them. As a result, novel therapeutic solutions other than the conventional antimicrobial therapies are an urgent need. Bacteriocins have attracted attention as potential substitutes for, or as additions to, currently used antimicrobial compounds, as such molecules are stable and potent against multidrug-resistant strains [2].

Bacteriocins are proteins or peptides ribosomally synthesized that are produced by species of the Bacteria and Archaea domains, and are usually active against strains of bacteria that are closely related to the producer strains or are unrelated (narrow and broad spectrum, respectively) [3–5]. Almost all bacteria can produce at least one bacteriocin and most of them remain unidentified [6]. Although the exact role of bacteriocins in nature is not well understood, these biomolecules may have an important role in colonizing or facilitating the introduction and/or dominance of a producer onto an already occupied niche [7]. Alternatively, bacteriocins may act as antimicrobial peptides, directly inhibiting competing strains or pathogens, or may function as signaling peptides, either signaling

<sup>\*</sup>Corresponding Author.

other bacteria through quorum sensing and bacterial cross-talk within microbial communities or signaling cells of the host immune system [8-10].

The exploitation of bacteriocins for microbial biofilm control is a relatively new research field. Nevertheless, most studies provide evidence of the promising potential of bacteriocins, given their potency, efficacy, resistance to heat and pH variation, possibility of chemical modification, low toxicity, availability of both broad- and narrow—spectrum activities, and the chance of in situ production by probiotic microorganisms [4,5]. Another appreciated characteristic of bacteriocins is their narrow inhibitory spectrum that provides an efficient means to direct activity toward certain pathogens without disturbing the commensal bacterial microbiota, which can contribute health-giving benefits to the mammalian host [11].

Bacteriocin molecules are also reported to inhibit relevant pathogens like Shiga toxin-producing *Escherichia coli* (STEC), enterotoxigenic *E. coli*, methicillin-resistant *Staphylococcus aureus* (MRSA), vancomycin-resistant enterococci (VRE), *Agrobacterium*, and *Brenneria* spp. [5,12]. The potency of bacteriocins against clinically important pathogens varies both across and within the various peptide subclasses.

In past years, the application of encapsulation technology in liposomes for bacteriocins has been evaluated [13–16]. Liposomes are colloidal carriers that form spontaneously when certain phospholipids are hydrated in aqueous media [17]. They are composed of relatively biocompatible and biodegradable material and consist of an aqueous space surrounded by one or more layers of natural and/or synthetic lipids. As liposomes contain both lipid and aqueous phases, they can be utilized for the entrapment, delivery, and release of hydrophilic, hydrophobic, and amphiphilic compounds [18]. This technology has arisen with great potential in the treatment of biofilm infections, owing to good biocompatibility, stability, ample range and extent of drugs that they can carry, as well as the protection provided by the encapsulation of the drug in the biological milieu, decreasing toxicity and allowing the drug to reach the specific site.

Therefore, in this chapter, we describe the various classes of bacteriocins and the contribution of these metabolites as well as the drug-delivery nanocarrier technologies for planktonic cells and biofilm control.

### 16.2 Classification

Historically, bacteriocins are classified according to the producer microorganism, chemical structure, heat stability, molecular mass, enzymatic sensitivity, presence of modified amino acids, amino acid sequence homologies, gene cluster organization, and mechanism of action. Because of the large diversity of bacteriocins, several approaches have been taken to classify these biomolecules [3-5,19,20].

One way used to classify the bacteriocins of gram-negative organisms divides these biomolecules into small peptides, such as microcins (Mccs), and larger proteins, such as colicins (Cols). Mccs have previously been divided on the basis of the presence (class I) or absence (class II) of posttranslational modifications [21] (Table 16.1). Class I Mccs includes MccB17, MccC7-C51, MccD93, and MccJ25 of low molecular mass (<5 kDa)
Class	Characteristics	Microcin	Receptor	Translocator	Molecular Mass (Da)	Producing Strain	References
1	Low-molecular-mass peptides (<5 kDa),	MccB17	OmpF	OmpF, SbmA	3094	Escherichia coli	[23]
	posttranslationally	MccC7-C51	OmpF	OmpF, YejABEF	1177	E. coli	[22]
	modified	MccD93	_	_	<1000	E. coli	[24]
		MccJ25	FhuA	Fhua, TonB, ExbBD, SbmA	2107	E. coli	[25]
ll: Larger (5—10 kDa) peptides, with or without	Class IIa: Require more than one gene for	MccL	Cir	Cir, TonB, ExbBD, SdaC	8884	E. coli	[26,27]
posttranslational modifications	synthesis and assemblage of functional peptides	MccV	Cir	Cir, TonB, ExbBD	8741	E. coli	[28]
		MccN	OmpF, OmpC, Phoe	TolAQR	7274	E. coli	[29]
	Class IIb: Linear peptides carrying a C-terminal	MccE492	FepA, Cir, Fiu	FepA, Cir, Fiu, TonB, ExbBD	7886	Klebsiella pneumoniae	[30,31]
	siderophore as posttranslational	MccM	FepA, Cir, Fiu	FepA, Cir, Fiu, TonB, ExbBD	7284	E. coli	[32]
	modification or not at C-terminus	MccH47	FepA, Cir, Fiu	FepA, Cir, Fiu, TonB, ExbBD	4865	E. coli	[32]
		147 <sup>a</sup>	FepA, Cir, Fiu	FepA, Cir, Fiu, TonB, ExbBD	_	E. coli	[33]
		G492 <sup>a</sup>	FepA, Cir, Fiu	FepA, Cir, Fiu, TonB, ExbBD	_	<i>K. pneumoniae</i> RYC49	[32]

#### Table 16.1 Classification Scheme for Gram-Negative Microcins

<sup>a</sup>Predicted by genome sequence in silico analysis.

and the class II Mccs are of higher molecular mass (5–10 kDa). Class II Mccs can be divided into subclasses IIa (such as MccL, MccV, and MccN24), which require three different genes to synthesize and assemble functional peptides, and IIb (MccE492, MccM, and MccH47), linear peptides with or without posttranslational modifications at the C-terminal region [22].

Cols are categorized into two groups (A and B) according to translocation across the outer membrane of sensitive bacteria (Table 16.2). Group A Cols use the Tol protein system, for instance: ColE1–E9, ColA, ColK, ColN, ColS4, ColU, Col28b, ColDF13, and group B uses the Ton system, such as Col5, Col10, ColB, ColD, ColM, MccV, ColIa, and ColIb [34].

As Cols reach and enter the target cell, they can be divided into three major classes based on the mechanisms of action:

- 1 pore formation, including ColA, ColB, ColE1, ColIa, ColIb, ColK, and ColN;
- **2** nuclease activity, with DNase, 16S rRNase, and tRNase activities, also with ColE2 to E9; and
- **3** peptidoglycanase activity: these proteins can digest the peptidoglycan precursor, leading to a failure to produce peptidoglycan and, thus, bacterial death, e.g., ColM [24].

The classification of bacteriocins from gram-positive bacteria used today has been modified in several steps based on the initial classification scheme proposed by Klaenhammer [3] (Table 16.3). The bacteriocins were grouped in four major classes (I–IV) based on their probable structure and mode of action as predicted at that time. In this classification, class I included lantibiotics (peptides <5 kDa posttranslationally modified resulting in lanthionine,  $\beta$ -methyllanthionine, dehydroalanine and dehydrobutyrine), class II comprised small (<10 kDa) unmodified heat-stable peptides, class III comprised large (>30 kDa) heat-labile molecules, and class IV contained complex molecules bound to lipids or carbohydrates. Both class I and class II bacteriocins are secreted by ATP-binding cassette (ABC) transporter proteins. Three subgroups of class II bacteriocins were also included: pediocin-like or Listeria-active with a YGNGVXC motif near the N-terminus and a GG leader peptide (IIa), two-component peptides with a GG leader peptide (IIb), and thiol-activated peptides that require reduced cysteine for activity (IIc). Nes et al. [76] regrouped the class II bacteriocins, retaining the pediocin-like bacteriocin in class IIa and the two-peptide bacteriocins in class IIb, but changing class IIc to include bacteriocins that contain a typical signal peptide and that are secreted by the translocase general secretion (Sec) pathway. These authors defined the circular bacteriocins as class IId and excluded class IV bacteriocins. In contrast, van Belkum and Stiles [77] subdivided the class II bacteriocins considering the number of cysteine residues and resultant disulfide bridges as a criterion for classification. In their classification, the bacteriocins that are secreted by the signal peptide pathway are not classified as a separate group because different types of bacteriocins are produced by the Sec pathway with a signal peptide as part of the prepeptide. In 2003, Kemperman et al. [78]

					Molecular Mass		
	Colicin	Antibacterial Activity	Receptor	System Translocation	(Da)	Producing Strain	References
Group A	А	Pore-forming	BtuB	OmpF, TolABQR	62,989	Citrobacter freundii,	[35—37]
						Escherichia coli	
	E1	Pore-forming	BtuB	ToIC, ToIABQR	57,279	E. coli	[38]
	К	Pore-forming	Tsx, OmpFA	OmpA, OmpF, TolAQR	59,611	E. coli	[39,40]
	Ν	Pore-forming	OmpF	OmpF, TolABQR	41,696	E. coli	[41]
	S4	Pore-forming	OmpW	OmpF, TolABQR	54,085	E. coli	[42]
	U	Pore-forming	OmpA	OmpF, LPS, TolABQR	66,289	Shigella boydii	[43]
	Alveicin A	Pore-forming	_	TolABQR	43,027	Hafnia alvei	[44]
	Alveicin B	Pore-forming	-	TolABQR	38,183	H. alvei	[44]
	Marcescin	Pore-forming	Omp4, OmpA, F, LPS	OmpF, TolABQR	47,462	Serratia	[45,46]
	28b					marcescens	
	E3	16S rRNase	BtuB	OmpF, TolABQR	57,960	E. coli	[47,48]
	E4	16S rRNase	BtuB	OmpF, TolABQR		E. coli	[37]
	E6	16S rRNase	BtuB	OmpF, TolABQR	58,011	E. coli	[37,49]
	Cloacin DF13	16S rRNase	LutA, OmpF	TolAQR	59,293	Enterobacter	[49—51]
						cloacae	
	E2	DNase	BtuB	OmpF, TolABQR	61,561	E. coli, Shigella	[35,47,52]
						sonnei	
	E7	DNase	BtuB	OmpF, TolABQR	61,349	E. coli	[53]
	E8	DNase	BtuB	OmpF, TolABQR	70,000	E. coli	[54,55]
	E9	DNase	BtuB	OmpF, TolABQR	61,587	E. coli	[55,56]
	E5	tRNase	BtuB	OmpF, TolABQR	58,254	E. coli, S. sonnei	[37]

 Table 16.2
 Classification Scheme for Gram-Negative Colicins

Continued

					Molecular Mass		
	Colicin	Antibacterial Activity	Receptor	System Translocation	(Da)	<b>Producing Strain</b>	References
Group B	Pyocin AP41	DNase	orf1—tolQRA, tolB,	OmpF, TolAQR	84,000 and 10,000	Pseudomonas	[57,58]
			oprL—orf2			aeruginosa	
	Pyocin S1	DNase	Ferripyoverdine	TonB, ExbBD	65,600 and 10,000	P. aeruginosa	[59,60]
	Pyocin S2	DNase	Type I ferripyoverdine (FpvAl)	TonB, ExbBD	74,000 and 10,000	P. aeruginosa	[59—62]
	Pyocin S3	DNase	Type II ferripyoverdine (FpvAII)	TonB, ExbBD	81,385 and 17,047	P. aeruginosa	[63,64]
	Pyocin S4	tRNase	Type I ferripyoverdine (FpvAl)	OmpW, OmpF, Tol		P. aeruginosa	[62,64]
	D	tRNase	FepA	TonB-ExbBD	74,683	E. coli	[35,65]
	G	Membrane lysis	Fiu	TonB, ExbBD	5500	E. coli	[35,66]
	Н	Membrane lysis	Fiu	TonB, ExbBD	100	E. coli	[66]
	Pesticin I	Muramidase	FhuA	FhuA, TonB-ExbBD	40,043	Yersinia pestis	[67]
	М	Inhibition of synthesis of murein and lipopolysaccharide	FhuA	FhuA, TonB-ExbBD	29,453	E. coli	[68]
	В	Pore-forming	FepA	TonB-ExbBD	54,742	E. coli	[35,69]
	la	Pore-forming	Cir	TonB-ExbBD	69,429	E. coli	[35,70]
	lb	Pore-forming	Cir	TonB-ExbBD	69,923	S. sonnei	[35,70]
	5	Pore-forming	Tsx	TolC, TonB-ExbBD	53,137	E. coli	[39,71]
	10	Pore-forming	Tsx	TolC, TonB-ExbBD	53,342	E. coli	[72]
	Pyocin S5	Pore-forming	FptA ferripyochelin	_	_	P. aeruginosa	[64,73]
	Fy	Pore-forming	YiuR	TonB	54,000	Yersinia frederiksenii	[74]
	Q	Pore-forming	Cir	TonB, ExbBD	ND	E. coli	[75]

 Table 16.2
 Classification Scheme for Gram-Negative Colicins—cont'd

				van Belkum and				
Class	Subclass	Klaenhammer [3]	Nes et al. [76]	Stiles [77]	Cotter et al. [4,5]	Heng et	al. [79]	Examples
Mantibiotic	A	Lantibiotic	Lantibiotics	Lantibiotics	Lantibiotics	Type A—linear	A1—nisin-like	Nisin A, U, and Z; subtilin, epidermin, gallidermin, Pep5, streptin
							A2—all SA- F22-like	SA-FF22, lactin 481, salivaricin A, nukacin ISK-1, mutacin II
	В					Type B—globular		Mersacidin, cinnamycin, ancovenin, duramycin B and C
	С					Type C—multicom	ponent	Lacticin 3247, lichenicidin, mutacin I
II/nonlantibiotic	lla	Pediocine-like or <i>Listeria</i> -active with YGNGVXC motif near N-terminus and GG leader peptide	Pediocine-like or Listeria-active with YGNGVXCXXXVVV motif near N- terminus and GG leader peptide	Cystibiotics_ with two disulfide bridges with YGNGVXC motif near N- terminus	Pediocine-like or <i>Listeria</i> -active with YGNGVXC motif near N-terminus and GG leader peptide	Type IIa—pediocin	-like	Pediocin PA1, enterocin CRL35, carnobacteriocin BM1
	llb	Two-peptide bacteriocin, requires both peptides for activity	Two-peptide bacteriocin, requires both peptides for activity	Cystibiotics with one disulfide bridge with YGNGVXC motif near N-terminus	Two-peptide bacteriocin, requires both peptides for activity	Type IIb—multicor	nponent	Lactococcin G, lactacin F, ABP118
	llc	Thiol-activated peptides that require reduced cysteine for activity	Bacteriocins secreted by signal peptide pathway	Cystibiotics with one disulfide bridge but without YGNGVXC motif near N- terminus	Cyclic peptide, N- and C-termini are covalently linked	Type IIc—miscellar	neous	Reuterin 6, anacyclamides
	lld	_	_	Thiolbiotics with one or no cysteine residues	Single non-pediocin- like peptides			Divergicin, MccV, MccS, epidermicin NI01, lactococcin A
	lle	_	_	Two-component peptides with GG leader peptide	Contain a serine-rich carboxy-terminal region			MccE492, M
	llf	_	-	Atypical bacteriocins	_			
III		Large, heat-labile proteins	Large, heat-labile proteins	Large, heat-labile proteins	Bacteriolysins: large, heat-labile proteins, often murein hydrolases	Large (>10 kDa) p Ila—lysins, type IIIb	oroteins; type o—nonlytic	Lysostaphin, enterolysin A, caseicin 80, helveticin J
IV		Protein complexes	_	_	_	Cyclic peptides		Enterocin AS-48

#### Table 16.3 Comparison of Classification Systems for Bacteriocins

recommended recognition of a new group (class V) comprising ribosomally synthesized, unmodified head-to-tail ligated cyclic antibacterial peptides.

Later, Cotter et al. [4] reviewed this classification system for lactic acid bacteria bacteriocins and proposed a two-class division: I, lanthionine-containing bacteriocins/ lantibiotics, and II, non-lanthionine-containing bacteriocins. Moreover, they also suggested that class III (large heat-labile murein hydrolases) should not be classified as bacteriocins but instead be named bacteriolysins, and class IV (the lipid- or carbohydrate-containing bacteriocins) be withdrawn. It was further foreshadowed that the Klaenhammer class II subgroups IIa and IIb be retained and that the class V (cyclic peptides) proposed by Kemperrmam et al. [78] be reassigned as class IIc. Subclass IId was proposed to include all of the remaining linear non-lanthionine-containing bacteriocins as certain Mccs, such as MccV and MccS, whereas the subclass IIe includes MccE492-like bacteriocins (formerly known as the class IIb Mccs) [5].

Heng and Tagg [79] disproved many statements made by Cotter et al. [4] and proposed a new classification designed to fit all bacteriocins produced by gram-positive bacteria. Class III (large bacteriocins) was retained and subdivided into IIIa (bacteriolysins) and IIIb (nonlytic proteins); the cyclic bacteriocins (class IIc) were reassigned as class IV; and as a consequence of upgrading the cyclic bacteriocins to their own class, subgroup IIc became the repository for all unmodified class II inhibitors other than the *Listeria*-active (IIa) and multicomponent bacteriocins (IIb).

Arnison et al. [20] recommended a nomenclature system for the biosynthesis of ribosomally synthesized and posttranslationally modified peptides (RiPPs), including only molecules <10 kDa and excluding posttranslationally modified proteins, for which more than 20 distinct compound classes have been described. Among the RiPPs, bacteriocins are detected in many classes, such as the lanthipeptides (lantibiotic nisin), linear azol(in)e-containing peptides (MccB17), lasso peptides (MccJ25), thiopeptides (thiostrepton, philipimycin, MccJ25), Mccs (MccB17, MccC7-51), linaridins (cypemycin), proteusins (polytheonamide A), sactipeptides (sactibiotics: subtilosin A, thuricin CD), bottromyces (bottromyces A2), cyanobactins (patellamide, anacyclamide A10), and glycocins (sublancin 168).

### 16.3 Mechanisms of Action

The bactericidal mechanisms of bacteriocins can be divided into those that work primarily at the cell envelope and those that work mainly within the cell, affecting gene expression and protein synthesis. Nisin, some lantibiotics, and class II bacteriocins inhibit cell wall biosynthesis by interacting with lipid II, which is the carrier of peptidoglycan monomers across the cytoplasmic membrane and is the target of vancomycin [80,81]. Because the binding sites of these bioactive molecules are in different locations, bacteriocins retain activity against vancomycin-resistant gram-positive pathogens [82].

Most bacteriocins kill target cells by inducing permeabilization of the cell membrane, and the activity is generally specific, given that they use specific receptors on the target cell surfaces. Nisin and various lantibiotics use lipid II as a docking molecule to ease pore formation in the cell membrane, leading to loss of membrane potential and cell death [83,84]. In addition, it has been shown that different bacteriocins produced by both grampositive and gram-negative species can use the mannose phosphotransferase system (Man-PTS) membrane components on sensitive cells as receptor molecules, inducing pore formation. These bacteriocins comprise pediocin-like bacteriocins, lactococcins A and B (lactococcal bacteriocins), and MccE492 from *Klebsiella pneumoniae*, which can target the Man-PTS in *E. coli* membranes [85–87]. Other receptors related to the activity of bacteriocins are increasingly being reported including the maltose ABC transporter for class IIc bacteriocin, garvicin ML [88]; a Zn-dependent metallopeptidase for class IId bacteriocin LsbB [89]; and an undecaprenyl pyrophosphate phosphatase for the class IIb bacteriocin lactococcin G [90].

The class IIe peptide MccE492 uses the FepA, CirA, or Fiu receptors, all of which are iron siderophore receptors for its translocation to the periplasm. To exert its bactericidal activity via pores in the inner membrane, MccE492 needs TonB at the level of the interaction with the cytoplasmic membrane of the target cells [87,91]. However, the interaction of MccE492 with the TonB system is independent of the presence of the receptors. The RiPP proteusin polytheonamide B acts through a mechanism that involves pore formation, but precisely how this occurs and whether a receptor is involved still have yet to be elucidated [92]. The peptide depolarizes the bacterial cytoplasmic membrane, simultaneously decreasing the membrane potential and the intracellular K<sup>+</sup> contents, which is consistent with the formation of transmembrane ion channels [21,22,92,93]. Large-conductance mechanosensitive channels (MscL) are crucial for the activity of sublancin 168, a lantibiotic produced by *Bacillus subtilis* strain 168. MscL may serve either as a direct target for this lantibiotic or as a gate of entry to the cytoplasm [94].

The polycyclic peptides duramycin, duramycin B, duramycin C, and cinnamycin inhibit the phospholipase A enzyme indirectly by specifically sequestering the substrate phosphatidylethanolamine [95,96].

Other bacteriocins can kill their target cells by interfering with DNA, RNA, and protein metabolism. MccB17 and MccJ25 pass through the outer membrane via the OmpF and FhuA TonB-dependent porin pathways, respectively. The inner-membrane protein SbmA transports the Mccs through the inner membrane to the cytoplasmic face. Inside the cell, MccB17 targets DNA gyrase [97], whereas MccJ25 inhibits transcription by blocking the secondary channel of RNA polymerase [98]. The translocation of MccC7-C51 through the inner layer of the *E. coli* cell wall occurs via the YejABEF transporter. Then, the Mcc is processed by one of the many broad-specificity cytoplasmic aminopeptidases of the bacterium, producing a modified aspartyladenylate that inhibits aspartyl-tRNA synthetase, blocking mRNA synthesis [20].

Thiostrepton, micrococcin P1, thiazomycin, and several other thiopeptides interfere with bacterial protein synthesis by a direct interaction with the 50S ribosomal subunit. These molecules bind near the GTPase-associated center that engages a number of translation factors during initiation and elongation. The ribosomethiopeptide complex prevents conformational alterations communicated from the translation factors to the ribosome, ultimately halting translocation along the mRNA template. Bottromycins block aminoacyl-tRNA binding to the 50S ribosome. Another subset of thiopeptides, including the thiomuracins and GE2270A, inhibits protein synthesis by binding to elongation factor Tu (EF-Tu) [20,99]. This ligation occludes a portion of the recognition site for the aminoacyl-tRNA substrate between EF-Tu and aminoacyl-tRNA.

An additional activity of lantibiotics including nisin, subtilin, and sublancin is the ability to prevent spore outgrowth by *Bacillus* and *Clostridium* species [100,101]. It was demonstrated that in *Bacillus anthracis* the initiation of germination is essential for the action of nisin. This lantibiotic rapidly and irreversibly inhibits spores growth by preventing the establishment of oxidative metabolism and the membrane potential in germinating spores and the shedding of the external spore structures [102].

Garvicin A and lactococcin 972 have been shown to inhibit cell wall biosynthesis, most probably by inhibiting septum formation [103,104]. Garvicin A is specifically active against other *Lactococcus garvieae* strains, whereas lactococcin 972 inhibits only closely related *Lactococcus* spp. The processes of cell elongation and septum formation are linked in gram-positive cocci, taking place at an annular band located in the equatorial zone of the cell. *Lactococcus lactis* cells treated with these bacteriocins presented only a primordial septum and suffered an elongation that preceded the arrest of macromolecular synthesis and the death of the cell [104]. This suggests that both bacteriocins inhibit septum invagination rather than septum initiation.

In past years, diverse studies have suggested that the continuous exposure of bacteria to bacteriocins may select cells resistant to them. Bacteriocin resistance may be either innate or acquired, and seems to be a complex phenomenon arising at different frequencies and by different mechanisms, even among strains of the same bacterial species [4]. This observation has encouraged studies focused on understanding the molecular mechanisms involved in resistance, on definition of the frequency at which susceptible organisms can develop resistance to a given bacteriocin, and on the definition of strategies that can be used to overcome such resistance. Resistance mechanisms generally involve changes in the bacterial cell envelope, which result in reduction or loss of bacteriocin binding or insertion, bacteriocin sequestering, bacteriocin efflux pumping, and bacteriocin degradation. Other general forms of resistance include the formation of spores or biofilms, which are a common mechanisms such as production of the nisin lytic protein nisinase and the phenomenon of immune mimicry [105].

## 16.4 Antimicrobial and Antibiofilm Activities of Bacteriocins

Table 16.4 shows a summarized overview of the antibiofilm activity, biomedical applications, and potential therapeutic uses of bacteriocins. The lantibiotics nisin, planosporicin, Pep5, epidermin, gallidermin, mersacidin, mutacin B-Ny266, lacticin 3147, and actagardine have strong antimicrobial activity against emergent pathogens such as *Streptococcus pneumoniae*, MRSA, VRE, various mycobacteria species, *Propionibacterium acnes*, and *Clostridium difficile* in vitro [127]. Even though lantibiotics are generally believed to be poorly active against gram-negative bacteria, purified lantibiotics such as nisin and epidermin have been found to kill various species such as *Neisseria* and *Helicobacter, Campylobacter, Haemophilus, E. coli*, and *Salmonella* species, in which the outer membrane is damaged [99,127,154].

Nisin A is mostly studied as a food additive, and it has been granted GRAS (generally recognized as safe) status. This bacteriocin is used as a food preservative particularly in dairy products, canned foods, plants, protein foods, and cured meat. Structurally, nisin A is a lantibiotic with 34 amino acid residues posttranslationally modified, with one lanthionine, four methyllanthionine rings, and unusual residues such as dehydroalanine and dehydrobutyrine [155]. The molecule is highly active against gram-positive bacteria such as *Listeria monocytogenes*, *S. aureus, Bacillus cereus, Lactobacillus plantarum, Lactobacillus acidophilus, Micrococcus luteus, Micrococcus flavus, Streptococcus mutans, Streptococcus sanguinis*, and *Enterococcus faecalis*, also promoting spore growth inhibition of *B. anthracis* and *Clostridium botulinum* and inhibiting food spoilage by *Clostridium butyricum, Clostridium tyrobutyricum,* and *Clostridium sporogenes* [156,157].

Nisin presents several variants, like F [158], Q [153], U [159], and Z [160], but the activity spectra are similar. Except for nisin U, produced by *Streptococcus uberis*, the variants are produced by *L. lactis* subsp. *lactis*. Nisins A and Z are structurally identical, except for a substitution of asparagine for histidine as amino acid residue 27 in nisin Z. The substitution has no influence on its antimicrobial activity, but provides nisin Z higher solubility and diffusion characteristics compared to nisin A [160]. Nisin U is active against mastitis-associated bacteria such as *Streptococcus pyogenes*, *S. uberis*, *Streptococcus agalactiae*, *Streptococcus dysgalactiae*, *Staphylococcus simulans*, *Staphylococcus cohnii*, *L. lactis*, *L. acidophilus*, and *Streptococcus mitis* [159]. Nisin F, described by De Kwaadsteniet et al. [158], inhibited the growth of *S. aureus* in the respiratory tract of rats when administered intranasally. The antimicrobial activity of low concentrations of this peptide (80–320 AU/mL) was slightly stimulated by lysozyme and lactoferrin. As of this writing, among the lantibiotics, only nisins A and Z have commercial applications.

Various studies have provided evidence that nisin can be active against *L. monocytogenes* and *Enterococcus hirae* growing in biofilms at surfaces such as stainless steel, rubber, and polyethyleneterephthalate on food-processing equipment [106,109].

Class I (Posttranslational Modifications)	Bacteriocin	Producer Microorganism	Antimicrobial Spectrum	Biomedical Application or Potential Therapeutic Use	References
(Lantibiotic/A)	Nisin A	Lactococcus lactis subsp. lactis	Listeria monocytogenes, Staphylococcus aureus, MRSA, Bacillus cereus, Bacillus anthracis, Lactobacillus plantarum, Lactobacillus acidophilus, Micrococcus luteus, Micrococcus flavus, Streptococcus mutans, Streptococcus sanguinis, Enterococcus faecalis, Actinomyces sp., Clostridium botulinum, Corynebacterium sp., Gardnerella sp., Lactococcus sp., Mycobacterium, Propionibacterium acnes	Food preservative: inhibition of pathogenic and toxigenic bacteria, inactivation of endospore formers. Antibiofilm activity: <i>L. monocytogenes</i> and <i>Enterococcus hirae</i> in stainless steel chips; <i>S. aureus</i> , <i>Staphylococcus epidermidis</i> , and <i>E. faecalis</i> in catheters; oral biofilm-associated diseases. Treatments: peptic ulcer in humans, bovine mastitis, experimental gingivitis in dogs. Commercially available.	[106–110]
Lantibiotic/Al	Epidermin	S. epidermidis	S. aureus, S. epidermidis, P. acnes	Topical treatment of acne, eczema, folliculitis, impetigo. Use in personal care products	[111,112]
Lantibiotic/AI	Gallidermin	Staphylococcus gallinarum	S. aureus, S. epidermidis, P. acnes	Inhibition of biofilm formation of <i>S. aureus</i> and <i>S. epidermidis</i> . Topical treatment of acne, eczema, folliculitis, impetigo. Use in personal care products	[112–114]
Lantibiotic/Al	Pep5	S. epidermidis	S. epidermidis, P. acnes	Inhibition of <i>S. epidermidis</i> adhesion to silicone catheters. Topical treatment of acne	[115]
Lantibiotic/AI	Subtilin	Bacillus subtilis	P. acnes, Staphylococcus spp., Streptococcus spp., Clostridium spp., Bacillus	Food preservative: inhibition of pathogenic and toxigenic bacteria, inactivation of endospore formers	[116]
Lantibiotic/AI	Mutacin B-Ny266	S. mutans	MRSA, VRE	Treatment of multidrug- resistant bacteria	[117,118]
Lantibiotic/A Lantibiotic/A	Nukacin ISK-1 Amylolysin	Staphylococcus warneri Bacillus amyloliquefaciens GA1	MRSA L. monocytogenes, S. aureus, S. epidermidis, B. cereus	MRSA biofilm Antimicrobial activity	[2] [81]
Lantibiotic/B	Mersacidin	Bacillus sp.	MRSA, VRE, <i>P. acnes</i>	Topical treatment of acne and staphylococcal and enterococcal infections. Active against MRSA in murine model	[119–121]

#### Table 16.4 Bacteriocins Reported in the Literature With Antibiofilm and Antimicrobial Activities

Lantibiotic/B	Actagardine	Actinoplanes liguriensis	MRSA, VRE, Clostridium difficile	Bovine mastitis, oral hygiene product, treatment of acne and CDAD	[122]
Lantibiotic/B	Duramycin	Streptomyces cinnamoneus	Bacillus subtilis	Treatment of cystic fibrosis increasing chloride secretion in lung epithelium and dry eye syndrome	[123]
Lantibiotic/C	Lacticin 3147	L. lactis subsp. lactis	L. monocytogenes, B. cereus, C. botulinum Clostridium tyrobutyricum, Streptococcus agalactiae, Streptococcus dysgalactiae, Streptococcus uberis, S. mutans, MRSA, VRE, PRSP	Prevent food spoilage and disease. Treatment of bovine mastitis, staphylococcal and enterococcal infection	[124–127]
Lantibiotic/C	Mutacin 1140	S. mutans	Streptococcus sobrinus, S. sanguinis, Streptococcus oralis, Streptococcus gordonii, Streptococcus mitis, Streptococcus parasanguinis	Oral biofilm-associated diseases.	[128,129]
Sactibiotics	Subtilosin A	B. subtilis, Bacillus atrophaeus, and B. amyloliquefaciens	E. faecalis, Streptococcus pyogenes, L. monocytogenes, Gardnerella vaginalis	Treatment of bacterial vaginosis (vaginal biofilm)	[130,131]
Sactibiotics	Thuricin CD	Bacillus thuringiensis	C. difficile	Inhibition of <i>C. difficile</i> in an ex vivo model of the colon. Treatment of CDAD	[11,132]
Bottromycin Glycocin	Bottromycin A2 Sublancin 168	Streptomyces bottropensis B. subtilis 168	MRSA, VRE, <i>Mycoplasma</i> S. aureus, B. cereus, S. pyogenes inhibition of spore growth of Bacillus and Clostridium	Nosocomial infections Food preservative in milk and dairy products, and bioactive against <i>S. aureus</i> in infected mice	[133] [94,100,134—136]
Thiopeptide Thiopeptide	LFF571 Nocathiacin	Catellatospora sp. Nocardia sp. and Amycolatopsis sp.	C. difficile Clinically important multidrug-resistant	Treatment of CDAD Control of <i>S. aureus</i> systemic	[137] [69—71,138—140]
MccC7-C51 type bacteriocins	MccC7	E. coli H22	pathogens (MRSA, MREF, VRE, and PRSP) Enterobacter, Escherichia, Klebsiella, Morganella, Salmonella, Shigella, and Varcinia	infection (mouse model) Inhibition of <i>Shigella flexneri</i> in intestinal tract of anotobiotic mouse model	[141]
Lasso	MccJ25	E. coli	E. coli, Salmonella enterica, Shigella	Antimicrobial potential in mouse model of <i>Salmonella</i> infection	[142]
Linear azole- containing peptide	MccB17	E. coli	Escherichia, Citrobacter, Klebsiella, Salmonella, Shigella, Pseudomonas spp.	Antimicrobial activity	[143,144]
Class II (Unmodified or Cyclic)	Bacteriocin Example	Producer Microorganism	Antimicrobial Spectrum	Biomedical Application	References
lla peptides	Pediocin PA-1	Pediococcus acidilactici, Pediococcus pentosaceus, Pediococcus parvulus, L. plantarum, Bacillus coagulans, Enterococcus faecium	L. monocytogenes, E. faecium, Leuconostoc mesenteroides, Aeromonas sp., Yersinia sp., S. aureus, Serratia liquefaciens, Pseudomonas fluorescens	Food preservative against food spoilage and food-related health hazards. Commercially available.	[145]

Continued

Class I (Posttranslational				Biomedical Application or Potential Therapeutic	
Modifications)	Bacteriocin	Producer Microorganism	Antimicrobial Spectrum	Use	References
lla peptides	Enterocin CRL35	Enterococcus mundtii	L. monocytogenes, herpes simplex virus I and II	Food preservative	[146]
lla peptides	Carnobacteriocin A (CbnA)	Carnobacterium piscicola, Carnobacterium maltaromaticum	Listeria sp., Enterococcus sp., Carnobacterium sp.	Food preservative	[147]
llb	Abp118	Lactobacillus salivarius subsp. salivarius	Listeria, Salmonella, Campylobacter	Antilisterial activity in mouse model. Probiotic microorganisms for pigs and poultry—control of pathogenic strains in the microbiota.	[148,149]
llc	Enterocin AS-48	E. faecium	L. monocytogenes, B. cereus, S. aureus, Staphylococcus carnosus, E. coli, S. enterica, Alicyclobacillus acidoterrestris, Bacillus spp., Paenibacillus spp., Geobacillus stearothermophilus, Brochothrix thermosphacta Lactobacillus sakei	Food preservation	[151]
lld	Epidermicin NI01	S. epidermidis	Staphylococcus hominis, S. warneri, E. faecalis, MRSA	Promising candidate for clinical use against multiresistant gram-positive pathogens	[152]
lld	Lactococcin A	L. lactis	L. lactis	Food preservative, narrow	[153]
lle	MccE492	Klebsiella pneumoniae	E. coli, K. pneumoniae, S. enterica, Enterobacter cloacae	Remains to be elucidated	[32]
lle	MccM	E. coli	E. coli, S. enterica, E. cloacae	Remains to be elucidated	[32]

<b>Table 10.7</b> Decicitocitis reported in the Electricity with Antibiothin and Antimicrobial Activities cont	Table 16.4	Bacteriocins Re	eported in the	Literature With	Antibiofilm and	Antimicrobial	Activities-cont	ťd
--	------------	-----------------	----------------	-----------------	-----------------	---------------	-----------------	----

CDAD, clostridium difficile-associated diarrhea; MREF, multidrug-resistant Enterococcus faecium; MRSA, methicillin-resistant Staphylococcus aureus; PRSP, penicillin-resistant Streptococcus pneumoniae; VRE, vancomycin-resistant enterococci.

Nisin A from *L. lactis* UQ2, a bacteriocinogenic strain isolated from Mexican-style cheese, reduced >5 log cycles of *L. monocytogenes* Scott A in planktonic and sessile cells on stainless steel chips, in biofilms per chip, in competitive-exclusion tests at  $37^{\circ}$ C [106].

Nisin A can be also used in the control of oral biofilm-associated diseases, as demonstrated in vivo on the treatment of gingivitis in beagle dogs [107], and in tooth decay [108]. This bacteriocin was effective in eradicating *E. faecalis* and *Streptococcus gordonii* cells in pure culture, and was similar to  $Ca(OH)_2$  regarding the removal of these species from the root canal system [108]. It also inhibited the growth of oral bacteria in planktonic form at low concentrations, and slowed the development of polymicrobial biofilm by altering its development and reducing biofilm biomass and thickness in a dose-dependent manner. The molecule is a cationic bacteriocin with biological activity on membranes, and the possible mechanism of action is associated with disruption of the coaggregation process, which is essential for stable biofilms [110].

It was also shown that nisin A and lacticin Q have strong bactericidal activities against MRSA, both in planktonic and in biofilm cells (nisin showed the highest bactericidal activity), whereas vancomycin showed activity against planktonic cells only [2]. It was suggested that pore formation leading to ATP efflux is important for the bactericidal activity against biofilm cells and the pore size needs to be large enough for ATP flow, probably higher than 1.5 nm in diameter. The pore size and stability time vary among bacteriocins. The pore diameter of lacticin 3147 is around 0.6 mm, which induces ion leakage, but not ATP [161]. SA-FF22 produced by *S. pyogenes* FF22 disrupts the membrane potential by forming small short-lived pores (from 0.5 to 0.6 nm in diameter and with life spans of a few milliseconds) [162]. Nisin A forms large stable pores on membranes containing lipid II, with diameters around 2.5 mm and with life spans of a few seconds [163]. Lacticin Q forms large toroidal pores on membranes without lipid II, with diameters ranging from 4.6 to 6.6 nm that induce ATP leakage [164]. Another possibility is that lipid II, a docking molecule for nisin A, contributes to specific binding to targeted cells, although it is not required for lacticin Q activity.

It is important to mention that the effectiveness of individual bacteriocins might be enhanced when combined with other antimicrobials or membrane-active substances. Nisin, for instance, presented synergistic activity with polymyxin E and clarithromycin when tested against *Pseudomonas aeruginosa* [165], and with Ramoplanin and other drugs against MRSA and VRE strains [166]. This molecule can be used combined with other compounds such as chelating agents [167]. It was reported that  $\varepsilon$ -polylysine ( $\varepsilon$ -PL) and nisin present synergic behavior against *E. coli*, *B. subtilis*, and *S. aureus*, and it was proposed that nisin promoted the uptake of  $\varepsilon$ -PL into cells, forcing their interaction with intracellular DNA [168].

*Staphylococcus* spp. strains can produce lantibiotics such as gallidermin, epidermin, Pep5, and nukacin [2,115,169]. Gallidermin and epidermin are natural variants produced by *Staphylococcus gallinarum* and *Staphylococcus epidermidis*, respectively, which differ from epidermin by a leucine rather than an isoleucine at position 6 [169]. Gallidermin is

effective against *S. aureus* and *S. epidermidis*, which are involved in biofilm formation in several implanted medical devices such as stents, catheters, and others. Among the hypotheses to explain its mechanism of action is the repression of biofilm-related genes such as *atl* (autolysin) and the operon *ica* (intercellular adhesin), associated with *N*-acetylglucosaminyltransferase synthesis and biofilm formation [113,114].

As of this writing, there are several lantibiotics in clinical studies, including duramycin, produced by *Streptomyces cinnamoneus*, for the treatment of cystic fibrosis, by increasing chloride secretion in lung epithelium [123], a disease in which lungs are infected with bacterial biofilms, and a derivative of actagardine for the treatment of *C. difficile* infections [170,171]. Additionally, mutacin 1140 from the oral pathogen *S. mutans*, strain JH1140, has been developed to control supragingival biofilm formation (SMaRT Replacement Therapy, developed by Oragenics), excluding other *S. mutans* strains. It is worth mentioning that some people, after a period of 14 years, retained the SMaRT strain after being inoculated just once [128,129].

*Bacillus* strains are able to produce varied lantibiotics. Amylolysin from *Bacillus amyloliquefaciens* GA1 showed an antibacterial spectrum directed toward gram-positive bacteria including *L. monocytogenes* (0.5  $\mu$ M), *S. aureus* (0.4  $\mu$ M), and *S. epidermidis* (2.8  $\mu$ M), or *B. cereus* (0.2  $\mu$ M) [81]. The peptide did not inhibit the growth of either basidiomycetous or ascomycetous yeasts (i.e., *Cryptococcus neoformans* and *Saccharomyces cerevisiae*, respectively) or gram-negative bacteria (i.e., *E. coli* and *P. aeruginosa*). However, it is effective against *Enterococcus faecium*, with a MIC value of 0.1  $\mu$ M, and against *E. faecalis* to a lesser extent.

Subtilin, a pentacyclic lantibiotic of 32 amino acid residues, is a natural analogue of nisin [172]. Mersacidin produced by *Bacillus* sp. strain HIL Y-85, 54728 [173], inhibits the growth of MRSA strains in vivo in mice [120]. The peptide inhibits cell wall synthesis of MRSA strains with efficiency equal to that reported for vancomycin [174]. Mersacidin is also active against *P. acnes* and may thus be used in the treatment of acne [119]. However, its mechanism of action differs from that of vancomycin, which provides the option of using the two substances in combination [175]. Another strain, *B. subtilis* subsp. *spizizenii* DSM15029, has been described as an unsuccinylated entianin AMP producer. This peptide differs from subtilin at three amino acid positions and is effective against *S. aureus, E. faecalis*, and other gram-positive bacteria [176].

More recently, Chopra et al. [177] identified sonorensin, a novel peptide produced by *Bacillus sonorensis* MT93. This peptide is part of the heterocycloanthracin subfamily of bacteriocins, a set of presumed peptides containing heterocyclic moieties of oxazole and/or thiazole. Sonorensin has a 6.27-kDa molecular mass and presents antimicrobial activity against food-borne pathogens like *L. monocytogenes* and *S. aureus*. The authors also demonstrated its effectiveness against biofilms of *S. aureus*.

Thiopeptides are mostly active against gram-positive bacteria with potent in vitro activity, but their poor solubility made their large-scale production undesirable. Nocathiacin I and derivatives, the main components of nocathiacins, thiazolyl peptides isolated from *Nocardia* sp. and *Amycolatopsis* sp., exhibit activity in vitro with a MIC in

the ng/mL range against many clinically important multidrug-resistant pathogens including MRSA, multidrug-resistant *E. faecium*, VRE, and fully penicillin-resistant *S. pneumoniae* [138,139]. In addition, nocathiacin I showed excellent efficacy in vivo in a systemic *S. aureus* infection mouse model [140]. Philipimycin from *Actinoplanes philippinensis* MA7347 showed strong antibacterial activities against gram-positive bacteria, including MRSA, and exhibited MIC values ranging from 0.015 to 1.0  $\mu$ g/mL. It was effective in vivo in a mouse model of *S. aureus* infection exhibiting an ED<sub>50</sub> value of 8.4 mg/kg [178].

Thiazomycin from *Amycolatopsis fastidiosa* was highly potent in vitro against grampositive bacteria and was highly efficacious in vivo against *S. aureus* in mice. It showed activity against bacteria resistant to  $\beta$ -lactams, vancomycin, macrolides, quinolones, and other antimicrobial-resistant bacterial phenotypes, such as protein synthesis inhibitors (linezolid, macrolide, gentamicin, chloramphenicol, and tetracycline), indicating no cross-resistance [179].

Thiomuracins produced by a rare actinomycete bacterium typed as a *Nonomuraea* species inhibited the growth of *S. aureus* and *E. faecalis*, but had no measurable activity against *E. coli* and *P. aeruginosa*. Possibly, this poor antimicrobial activity can be explained by the low penetration of the molecules in gram-negative bacterial outer membranes [180]. GE2270 A and its derivative LFF571 (Novartis) exhibit activity against *C. difficile* and various gram-positive bacteria, excluding *Bifidobacteria* and lactobacilli [137].

Among the other posttranslationally modified bacteriocins, the sactibiotic thuricin CD produced by *Bacillus thuringiensis* exhibits antimicrobial activity comparable to the activities of vancomycin and metronidazole, both used to treat *C. difficile*-associated diarrhea (CDAD) in an experimental model of the human distal colon [181]. Notably, nevertheless, thuricin CD did not modify significantly the commensal microbiota composition, whereas vancomycin and metronidazole caused a remarkable increase in Proteobacteria abundance compared to organisms of other phyla. Thuricin CD consists of two different peptides, Trn- $\alpha$  and Trn- $\beta$ , that work synergistically to kill *C. difficile* isolates, including CDAD-associated strains [132]. Bottromycin A2 exhibits potent activity against MRSA and VRE by targeting translation mechanisms [133].

Subtilosin A, a sactibiotic, is also an uncommon example of a modified bacteriocin produced by gram-positive bacteria that is active against some gram-negative bacterial strains [130]. Subtilosin A displays a narrow spectrum of activity against *E. faecalis, S. pyogenes,* and *L. monocytogenes* [130], as well as *Gardnerella vaginalis* [131]. The pathway of bacterial killing by this peptide remains to be described. Like nisin, the antimicrobial activity of subtilosin may be due to interactions with membrane-associated receptors [84].

The molecular mechanism of the action of subtilosin against *L. monocytogenes* remains unclear. However, the mechanism against the vaginal pathogen *G. vaginalis* involves depleting the transmembrane pH gradient portion of the protonmotive force, causing an efflux of intracellular ATP and eventually cell death [131,182]. Subtilosin A inhibits *G. vaginalis* in vitro and has strong potential for inclusion in alternative bacterial vaginosis (BV) therapies because it has spermicidal activity, being atoxic to epithelial cells and lactobacilli at the human vaginal mucosa [183,184]. BV is a polymicrobial biofilm infection, dominated by dense clusters of *G. vaginalis* and, to a lesser extent, *Atopobium* species that probably have a role in potentiating failures of the antimicrobial treatment recommended by the Centers for Disease Control, metronidazole and clindamycin orally or intravaginally [185]. Noll et al. [186] showed that subtilosin A acts synergistically when combined with glycerol monolaurate, lauric arginate, and  $\varepsilon$ -PL against pathogens associated with BV, but not with human lactobacilli. The cell-free supernatants of overnight cultures were active against *L. monocytogenes*. The first report of antiviral activity for this peptide was for subtilosin from *B. amyloliquefaciens* KATMIRA 1933 [187].

Various unmodified class II bacteriocins have been described with antimicrobial activity against gram-positive bacteria, including lactic acid bacteria class IIa bacteriocins, pediocin PA1, leucocin A, and others, which are effective against *L. monocytogenes* [188] and other gram-positive pathogens [189]. A synergic effect was observed for the combination of the class IIa enterocin CRL35 with several antibiotics against *L. monocytogenes* [190]. Pediocin PA1 has been used commercially for food preservation. The Irish laboratory Quest explored a fermented product of *Pediococcus acidilactici*, a pediocin PA1-producing strain, and generated ALTA 2431. Patents from the United States and Europe cover its use for this purpose [191].

Class IIb bacteriocins are composed of a dimer of peptides like lactacin F [150,192] and ABP-118 [193]. These molecules are involved in the formation of an active poration complex, the mechanism of action of this class. ABP-118 presented antilisterial activity in a mouse model [148].

The class IV peptide enterocin AS-48, an antimicrobial enterococcal bacteriocin, has been studied for use as a food preservative, as it is active against food-borne pathogenic strains like *E. coli, Salmonella* spp., *L. monocytogenes, S. aureus*, and *B. cereus* and the food-spoilage bacteria *Bacillus* spp. and *Paenibacillus* spp. [194–197].

Licheniocin 50.2 from *Bacillus licheniformis* VPS50.2 is effective against *L. monocytogenes*, MRSA, and hemolytic streptococci [198]. Regarding class IId bacteriocins, epidermicin NI0 produced by *S. epidermidis* strain 224 has also exhibited potent antimicrobial activity in the nanomolar range against pathogenic gram-positive bacteria including MRSA, enterococci, and biofilm-forming *S. epidermidis* strains. The peptide is highly stable against protease and keeps its antimicrobial potential in a wide range of pH, from 2 to 10. Bacterial exposure to this peptide has no effect on the development of resistance mechanisms [152]. There are few cases of class IIa bacteriocins produced by gram-positive bacteria that are effective against gram-negative pathogens [199].

Bacteriocins of gram-negative bacteria usually have a narrow spectrum of action restricted to closely related species, different from those of gram-positive organisms that are active against many strains of gram-positive species, but generally inactive against gram-negative bacteria. Jordi et al. [200] found that 20 strains of *E. coli* could express Col,

which inhibited five STEC strains of serotypes O26, O111, O128, O145, and O157:H7. These *E. coli* can cause diarrhea and hemolytic uremic syndrome in humans. Under conditions simulating the cattle rumen environment (anaerobiosis and 30% rumen fluid), the Cols ColE1, ColE4, ColE8-J, ColK, and ColS4 can inhibit the growth of STEC. Purified ColE1 and ColN were active against enterotoxigenic *E. coli* F4 (K88) and F18 in vitro, which caused postweaning diarrhea and edema disease in piglets [201]. The same group showed also that the inclusion of purified ColE1 in the diet of young pigs reduced the incidence of postweaning diarrhea by F18-positive *E. coli* [202].

The antimicrobial spectrum of modified Mccs has been investigated. MccC7-C51 was described to be active against some strains of *Escherichia*, *Enterobacter*, *Klebsiella*, *Salmonella*, *Shigella*, *Proteus*, and *Yersinia* spp. [203]. Lasso-type bacteriocins were active against *Escherichia* and *Salmonella* spp. strains [204], and MccB17, a linear peptide with an azole moiety, was effective against several gram-negative species, including *Escherichia*, *Citrobacter*, *Klebsiella*, *Salmonella*, *Shigella*, and *Pseudomonas* spp. [143,144]. Several Mccs (subclasses IId, MccV, MccL, IIe, MccE492, MccM3, and MccH47) were effective against some gram-negative strains [26,32,205,206]. Bacteriocin-producing *E. coli* strains could inhibit biofilm formation by non-bacteriocin-producing *E. coli* strains on catheter-related materials [207]. Other studies explored the influence of bacteriocins on biofilm formation by competing organisms [208,209]. Bacteriocin-producing strains have a selective advantage within multispecies biofilms, and it is possible that bacteriocins evolved to function largely in biofilms rather than as planktonic cells.

Purified ColA-43864 bacteriocin produced from *Citrobacter freundii* exhibited antimicrobial activity against a wide range of gram-negative bacteria and killed cells within biofilms in a dose-dependent manner [210]. The purified ColA-43864 eradicated a large number of cells within a biofilm, supporting the model that diffusion of ColA-43864 into the biofilm is not a limiting factor. But because of the biofilm complexity, it is possible that there are microenvironments within the biofilm where Col would not function because of suboptimal pH or other conditions [210].

## 16.5 Antimicrobial and Antibiofilm Investigations With Liposome-Entrapped Bacteriocins

Marciset et al. [211] encapsulated thermophilin 13A in liposomes. This is a broad-spectrum antimicrobial bacteriocin obtained from *Streptococcus thermophilus*. Thermophilin 13 is composed of the antibacterial peptide ThmA and the enhancing factor ThmB, which increases the activity of ThmA around 40 times. Both free and liposome-entrapped forms presented antilisterial activity; however, the bacteriocin exerted a pore-forming activity in both liposome lipids and targeted bacterial membranes.

Benech et al. [13] compared the antimicrobial activity of liposome-entrapped nisin Z, using proliposome H (Pro-lipo H), and free nisin Z, produced in situ by a strain of

*L. lactis*, on the inhibition of *Listeria innocua* in cheddar cheese. The nisin Z-producing strain and nisin Z-containing liposomes did not affect cheese production or chemical composition. After 6 months of monitoring, cheeses produced with the *L. lactis* strain presented  $10^4$  CFU/g of *L. innocua* and 12% of initial activity of the bacteriocin was preserved. Cheeses made with encapsulated nisin Z, on the other hand, contained less than 10 CFU/g of *L. innocua*, and 90% of the initial nisin activity was preserved.

Similarly, using different proliposome formulations, Laridi et al. [212] encapsulated nisin Z in various proliposomes (Pro-lipo H, Pro-lipo S, Pro-lipo C, and Pro-lipo DUO) and assessed its stability in milk under fermentation. Liposome encapsulation efficiency ranged from 9.5% to 47%, the last being for Pro-lipo H. Long-term stability of liposome-entrapped nisin Z was demonstrated for 27 days at 4°C in various media, including milk with different fat levels.

Park et al. [16] developed a pH-sensitive liposome-entrapped bacteriocin to control kimchi fermentation. The liposomes were prepared by reverse-phase evaporation using dipalmitoylphosphatidylcholine, dipalmitoylphosphatidylcholamine, dio-leoylphosphatidylcholine, and cholesterol. The bacteriocin liposome was stable at pH 6 and 7, but ruptured at pH 4. The pH of samples treated with bacteriocin liposomes was preserved at 4 or higher, whereas untreated samples showed pH 3.58 or lower. The pH of kimchi samples treated with the bacteriocin–liposomes was 3.9 during 90 days storage, whereas that of the untreated samples was 3.32. The growth of lactic acid bacteria was controlled in treated samples, but kept increasing in untreated samples.

Sosunov et al. [14] assessed the antimicrobial potential of five bacteriocins obtained from bacterial culture supernatants in vitro against a *Mycobacterium tuberculosis* strain, using mouse macrophages, and in vivo in high-dose infection of inbred mice. In this model, four of the five tested bacteriocins were more potent antimycobacterials than rifampicin in equal concentrations. Whereas the molecules presented no toxicity to mouse macrophages, they also did not inhibit mycobacterial growth in these cells at 0.01-0.1 mg/L. However, when used entrapped in liposomes, bacteriocin 5 isolated from *E. faecalis* was able to inhibit mycobacterial growth inside macrophages without damaging the cellular membrane.

Also exploring bacterial growth control in food, Malheiros et al. [213] developed liposomes of soybean phosphatidylcholine and encapsulated an antimicrobial peptide, P34, produced by a strain of *Bacillus* sp. The liposomes' diameter was around 150 nm. The antimicrobial assays were performed by agar diffusion using *L. monocytogenes* ATCC 7644. Interestingly, no significant difference was seen in the biological activity of free and encapsulated P34. The mechanism of action of the liposomes was adherence to the bacterial cell wall, but without fusion. Free bacteriocin inhibited the pathogen more quickly than the encapsulated P34; however, liposome-entrapped P34 was more stable throughout the study.

Later, Malheiros et al. [214] encapsulated the same peptide P34 in nanovesicles prepared with soy lecithin, and tested them against *L. monocytogenes* strains by direct addition to milk samples. After 4 days of storage at low temperature, the antimicrobial

activity of both free and encapsulated BLS P34 decreased around 50%, and this decrease was associated with interactions with milk proteins. However, no significant loss of activity was detected until 21 days of storage. The viable counts of *L. monocytogenes* exposed to free or encapsulated P34 were permanently inferior to those observed in controls without the bacteriocin. Like the previous study of this group, the encapsulation of P34 in nanovesicles did not influence its antimicrobial activity; however, the encapsulation in liposomes preserved the same inhibitory effect against *L. monocytogenes* for a prolonged time compared to free P34.

## 16.6 Conclusion and Perspectives

Bacteriocins represent a potential alternative to synthetic antimicrobial products. In clinical contexts, the emergence of drug-resistant pathogens makes the search for novel bacteriocins even more relevant. This prospect can be based on the isolation of new bacteria and assays for antimicrobial activity followed by characterization of the activity spectrum and of the biomolecule. However, this approach presents limitations related to the susceptibility of the indicator bacterial species to the various groups of Cols and the growth conditions used in the screening. This is mostly because bacteriocin production can be under the control of a quorum-sensing system, and the expression of various bacteriocins can require defined physicochemical conditions. In this context, genomic, transcriptomic, and proteomic advances can also contribute to the identification of novel bacteriocins and understanding of regulatory mechanisms, which are important for determining how to switch on production when desired or how to enhance production or activity. The generation of bacteriocin-overproducer recombinant strains or manipulation of these biomolecules to improve their overall efficacy and stability is also important to make their technological use feasible. An important point is the concern that continuous exposure of bacteria to bacteriocins may select for resistant cells, as observed for synthetic antimicrobials. An approach to avoid the emergence of bacteriocin-resistant bacteria may be the combined use of bacteriocins with different mechanisms of action, which may additionally allow the use of lower bacteriocin doses. Another possibility consists in the combined use of synthetic antimicrobials and bacteriocins. Whereas most synthetic drugs act as enzyme inhibitors, most bacteriocins target the membranes of sensitive cells, leading to cell death. Therefore, the development of cross-resistance to antimicrobials with different mechanisms of action is expected to occur less frequently. Bacteriocins are capable of inhibition of the main pathways used by synthetic drugs as well as novel pathways. In this context, knowledge of the mechanisms involved in resistance to bacteriocins and the target pathogen resistance profiles is essential for the development of effective bacteriocin-based microbial control strategies.

The strategy of entrapment of bacteriocins in liposomes can contribute to the effective use of these biomolecules. When entrapped in liposomes not only can

intracellular delivery be achieved, but also biofilms can be reached and disrupted more efficiently. The controlled release provided by liposomes can preserve the chemical stability and antimicrobial activity for prolonged periods compared to free molecules. Studies on liposome-entrapped molecules are highly necessary to accelerate the use of these safe and efficient molecules mainly against highly resistant microorganisms in clinical and industrial environments. As the technologies become more accessible, research shall evolve at a quick pace, and the association of these compounds with nanotechnologies, in the near future, will allow their use as effective options for therapeutic use against biofilm-related infectious diseases and in other contexts, such as environmental and industrial.

## References

- [1] H.C. Flemming, J. Wingender, The biofilm matrix, Nature Reviews Microbiology 8 (2010) 623-633.
- [2] K. Okuda, T. Zendo, S. Sugimoto, T. Iwase, A. Tajima, S. Yamada, K. Sonomoto, Y. Mizunoe, Effects of bacteriocins on methicillin-resistant *Staphylococcus aureus* biofilm, Antimicrobial Agents and Chemotherapy 57 (2013) 5572–5579.
- [3] T.R. Klaenhammer, Genetics of bacteriocins produced by lactic acid bacteria, FEMS Microbiology Review 12 (1993) 39-85.
- [4] P.D. Cotter, C. Hill, R.P. Ross, Bacteriocins: developing innate immunity for food, Nature Reviews Microbiology 3 (2005) 777–788.
- [5] P.D. Cotter, R.P. Ross, C. Hill, Bacteriocins a viable alternative to antibiotics? Nature Reviews Microbiology 11 (2013) 95–105.
- [6] M.A. Riley, J.E. Wertz, Bacteriocins: evolution, ecology, and application, Annual Review of Microbiology 56 (2002) 117–137.
- [7] M.A. Riley, J.E. Wertz, Bacteriocin diversity: ecological and evolutionary perspectives, Biochimie 84 (2002) 357–364.
- [8] H. Majeed, O. Gillor, B. Kerr, M.A. Riley, Competitive interactions in *Escherichia coli* populations: the role of bacteriocins, The ISME Journal 5 (2011) 71–81.
- [9] M. Gobbetti, M. De Angelis, R. Di Cagno, F. Minervini, A. Limitone, Cell-cell communication in food related bacteria, International Journal of Food Microbiology 120 (2007) 34–45.
- [10] M. Meijerink, S. van Hemert, N. Taverne, M. Wels, P. de Vos, P.A. Bron, H.F. Savelkoul, J. van Bilsen, M. Kleerebezem, J. Wells, Identification of genetic loci in *Lactobacillus plantarum* that modulate the immune response of dendritic cells using comparative genome hybridization, PLoS One 5 (2010) e10632.
- [11] M.C. Rea, A. Dobson, O. O'Sullivan, F. Crispie, F. Fouhy, P.D. Cotter, F. Shanahan, B. Kiely, C. Hill, R.P. Ross, Effect of broad- and narrow-spectrum antimicrobials on *Clostridium difficile* and microbial diversity in a model of the distal colon, Proceedings of the National Academy of Sciences of the United States of America 108 (Suppl. 1) (2011) 4639–4644.
- [12] R. Grinter, J. Milner, D. Walker, Bacteriocins active against plant pathogenic bacteria, Biochemical Society Transaction 40 (2012) 1498–1502.
- [13] R.O. Benech, E.E. Kheadr, R. Laridi, C. Lacroix, I. Fliss, Inhibition of *Listeria innocua* in cheddar cheese by addition of Nisin Z in liposomes or by in situ production in mixed culture, Applied and Environmental Microbiology 68 (2002) 3683–3690.

- [14] V. Sosunov, V. Mischenko, B. Eruslanov, E. Svetoch, Y. Shakina, N. Stern, K. Majorov, G. Sorokoumova, A. Selishcheva, A. Apt, Antimycobacterial activity of bacteriocins and their complexes with liposomes, The Journal of Antimicrobial Chemotherapy 59 (2007) 919–925.
- [15] P.S. Malheiros, Y.M.F. Micheletto, N.P. Silveira, A. Brandelli, Development and characterization of phosphatidylcholine nanovesicles containing the antimicrobial peptide nisin, Food Research International 43 (2010) 1198–1203.
- [16] S. Park, M. Kim, K.S. Han, S. Oh, Use of bacteriocin produced by *Lactococcus* sp. CU216 with pH sensitive liposome entrapment, Korean Journal for Food Science of Animal Resources 24 (2004) 97–102.
- [17] A.D. Bangham, R.W. Horne, Negative staining of phospholipids and their structural modification by surface-active agents as observed in the electron microscope, Journal of Molecular Biology 8 (1964) 660–668.
- [18] M.R. Mozafari, C. Johnson, S. Hatziantoniou, C. Demetzos, Nanoliposomes and their applications in food nanotechnology, Journal of Liposome Research 18 (2008) 309–327.
- [19] S. Dimov, P. Ivanova, N. Harizanova, I. Ivanova, Bioactive peptides used by bacteria in the concurrence for the ecological niche: general classification and mode of action (overview), Journal of Molecular Biology 3 (2005) 3–22.
- [20] P.G. Arnison, M.J. Bibb, G. Bierbaum, A.A. Bowers, T.S. Bugni, Ribosomally synthesized and posttranslationally modified peptide natural products: overview and recommendations for a universal nomenclature, Natural Product Report 30 (2012) 108–160.
- [21] S. Duquesne, D. Destoumieux-Garzon, J. Peduzzi, S. Rebuffat, Microcins, gene-encoded antibacterial peptides from enterobacteria, Natural Product Report 24 (2007) 708–734.
- [22] K. Severinov, E. Semenova, A. Kazakov, T. Kazakov, M.S. Gelfand, Low molecular weight post translationally modified microcins, Molecular Microbiology 65 (2007) 1380–1394.
- [23] R.E. Collin Thompson, K.A. Jolliffe, R.J. Payne, A. Maxwell, Fragments of the bacterial toxin microcin B17 as gyrase poisons, PLoS One 8 (2013) e61459.
- [24] J.L. Martinez, J.C. Perez-Diaz, Isolation, characterization, and mode of action on *Escherichia coli* strains of microcin D93, Antimicrobial Agents and Chemotherapy 29 (1986) 456–460.
- [25] K.A. Wilson, M. Kalkum, J. Ottesen, J. Yuzenkova, B.T. Chait, R. Landick, Structure of microcin J25, a peptide inhibitor of bacterial RNA polymerase, is a lassoed tail, Journal of the American Chemical Society 125 (2003) 12475–12483.
- [26] A.M. Pons, F. Delalande, M. Duarte, S. Benoit, I. Lanneluc, S. Sable, Genetic analysis and complete primary structure of microcin L, Antimicrobial Agents and Chemotherapy 48 (2004) 505–513.
- [27] P. Hauduroy, J. Papavassiliou, Identification of a new type of coline (colicine L), Nature 195 (1962) 730–732.
- [28] M.J. Fath, L.H. Zhang, J. Rush, R. Kolter, Purification and characterization of colicin V from *Escherichia coli* culture supernatants, Biochemistry 33 (1994) 6911–6917.
- [29] G. Corsini, E. Karahanian, M. Tello, K. Fernandez, D. Rivero, J.M. Saavedra, A. Ferrer, Purification and characterization of the antimicrobial peptide microcin N, FEMS Microbiology Letters 312 (2010) 119–125.
- [30] A.M. Pons, N. Zorn, D. Vignon, F. Delalande, A. Van Dorsselaer, G. Cottenceau, Microcin E492 is an unmodified peptide related in structure to colicin V, Antimicrobial Agents and Chemotherapy 46 (2002) 229–230.
- [31] V. de Lorenzo, A.P. Pugsley, Microcin E492, a low-molecular-weight peptide antibiotic which causes depolarization of the *Escherichia coli* cytoplasmic membrane, Antimicrobial Agents and Chemotherapy 27 (1985) 666–669.

- [32] G. Vassiliadis, D. Destoumieux-Garzon, C. Lombard, S. Rebuffat, J. Peduzzi, Isolation and characterization of two members of the siderophore-microcin family, microcins M and H47, Antimicrobial Agents and Chemotherapy 54 (2010) 288–297.
- [33] M.E. Poey, M.F. Azpiroz, M. Laviña, Comparative analysis of chromosome-encoded microcins, Antimicrobial Agents and Chemotherapy 50 (2006) 1411–1418.
- [34] C. Kleanthous, Swimming against the progress and challenges in our understanding of colicin translocation, Nature Reviews Microbiology 8 (2010) 843–848.
- [35] P. Fredericq, Actions antibiotiques réciproques chez les Enterobacteriaceae, Revue belge de pathologie et de médecine expérimentale 19 (1948) 1–107.
- [36] J. Morlon, R. Lloublés, S. Varenne, M. Chartier, C. Lazdunuski, Complex nucleotide sequence of the structural gene for the colicin A, a gene translated at a non-uniform rate, Journal of Molecular Biology 170 (1983) 271–285.
- [37] B.M. Males, B.A. Stocker, Colicins E4, E5, E6 and A and properties of btuB+ colicinogenic transconjugants, Journal of General Microbiology 128 (1982) 95–106.
- [38] M. Yamada, Y. Ebina, T. Miyata, T. Nakazawa, A. Nakazawa, Nucleotide sequence of the structural gene for colicin E1 and predicted structure of the protein, Proceedings of the National Academy of Sciences of the United States of America 79 (1982) 2827–2831.
- [39] H. Pilsl, V. Braun, Strong function-related homology between the pore-forming colicins K and 5, Journal of Bacteriology 177 (1995a) 6973–6977.
- [40] J. Smarda, Colicin K decreases the density of intramembrane particles (IMP) in the cell membrane of *Escherichia coli*, Journal of General Microbiology 134 (1988) 1283–1288.
- [41] A.P. Pugsley, Nucleotide sequencing of the structural gene for colicin N reveals homology between the catalytic C-terminal domains of colicin A and colicin N, Molecular Microbiology 1 (1987) 317–325.
- [42] H. Pilsl, D. Smajs, V. Braun, Characterization of colicin S4 and its receptor, OmpW, a minor protein of the *Escherichia coli* outer membrane, Journal of Bacteriology 181 (1999) 3578–3581.
- [43] D. Šmajs, H. Pilsl, V. Braun, Colicin U a novel colicin produced by *Shigella boydii*, Journal of Bacteriology 179 (1997) 4919–4928.
- [44] J.E. Wertz, M.A. Riley, Chimeric nature of two plasmids of *Hafnia alvei* encoding the bacteriocins alveicins A and B, Journal of Bacteriology 186 (2004) 1598–1605.
- [45] M.B. Viejo, D. Gargallo, S. Ferrer, J. Enfedaque, M. Regué, Cloning and DNA sequence analysis of a bacteriocin gene from *Serratia marcescens*, Journal of General Microbiology 138 (1992) 1737–1743.
- [46] J.F. Guasch, J. Enfedaque, S. Ferrer, D. Gargallo, M. Regue, Bacteriocin 28b, a chromosomally encoded bacteriocin produced by most *Serratia marcescens* biotypes, Research in Microbiology 146 (1995) 477–483.
- [47] H.R. Herschman, D.R. Helinski, Purification and characterization of colicin E2 and colicin E3, Journal of Biological Chemistry 242 (1967) 5360–5368.
- [48] K.S. James, N. Zinder, Highly purified colicin E3 contains immunity protein, Proceedings of the National Academy of Sciences of the United States of America 71 (1974) 3380–3384.
- [49] A. Akutsu, H. Masaki, T. Ohta, Molecular structure and immunity specificity of colicin E6, an evolutionary intermediate between E-group colicins and cloacin DF13, Journal of Bacteriology 171 (1989) 6430–6436.
- [50] P.J. van den Elzen, H.H. Walters, E. Veltkamp, H.J.J. Nijkamp, Molecular structure and function of the bacteriocin gene and bacteriocin protein of plasmid Clo DF13, Nucleic Acids Research 11 (1983) 2465–2477.

- [51] J.A. Thomas, M.A. Valvano, Role of tol genes in cloacin DF13 susceptibility of *Escherichia coli* K-12 stains expressing the cloacin DF13-aerobactin receptor IutA, Journal of Bacteriology 175 (1993) 548–552.
- [52] S.T. Cole, B. Saint-Joanis, A.P. Pugsley, Molecular characterization of colicin E2 operon and identification of its proteins, Molecular Genetics and Genomics 198 (1985) 465–472.
- [53] K.F. Chak, W.S. Kuo, F.M. Lu, R. James, Cloning and characterization of the Col E7 plasmid, Journal of General Microbiology 137 (1991) 91–100.
- [54] M. Toba, H. Masaki, T. Ohta, Colicin E8, a DNase which indicates an evolutionary relationship between colicins E2 and E3, Journal of Bacteriology 170 (1988) 3237–3242.
- [55] P.C. Cooper, R. James, Two new E-colicins, E8 and E9 produced by a strain of *Escherichia coli*, Journal of General Microbiology 130 (1984) 209–215.
- [56] C.J. Macdonald, K. Tozawa, E.S. Collins, C.N. Penfold, R. James, C. Leanthous, Characterization of a mobile protein-binding epitope in the translocation domain of colicin E9, Journal of Biomolecular NMR 30 (2004) 81–96.
- [57] E.R. Lafontaine, P.A. Sokol, Effects of iron and temperature on expression of the *Pseudomonas aeruginosa tolQRA* genes: role of the ferric uptake regulator, Journal of Bacteriology 180 (1998) 2836–2841.
- [58] B.W. Holloway, H. Rossiter, D. Burgess, J. Dodge, Aeruginocin tolerant mutants of *Pseudomonas aeruginosa*, Genetics Research 22 (1973) 239–253.
- [59] Y. Sano, H. Matsui, M. Kobayashi, M. Kageyama, Molecular structures and functions of pyocins S1 and S2 in *Pseudomonas aeruginosa*, Journal of Bacteriology 175 (1993) 2907–2916.
- [60] S. Ito, M. Kageyama, F. Egami, Isolation and characterization of pyocins from several strains of *Pseudomonas aeruginosa*, The Journal of General and Applied Microbiology 16 (1970) 205–214.
- [61] M. Ohkawa, F. Kageyama, F. Egami, Purification and properties of pyocin S2, Journal of Biochemistry 73 (1973) 281–289.
- [62] A. Elfarash, Q. Wei, P. Cornelis, The soluble pyocins S2 and S4 from *Pseudomonas aeruginosa* bind to the same FpvAI receptor, Microbiology 1 (2012) 268–275.
- [63] C. Duport, C. Baysse, Y. Michel-Briand, Molecular characterization of pyocin S3, a novel S-type pyocin from *Pseudomonas aeruginosa*, Journal of Biological Chemistry 270 (1995) 8920–8927.
- [64] R. Parret, R. de Mot, Novel bacteriocins with predicted tRNase and pore-forming activities in *Pseudomonas aeruginosa* PAO1, Molecular Microbiology 35 (2000) 472–473.
- [65] U. Roos, R.E. Harkness, V. Braun, Assembly of colicin genes from a few DNA fragments. Nucleotide sequence of colicin D, Molecular Microbiology 3 (1989) 891–902.
- [66] D.E. Bradley, Colicins G and H and their host strains, Canadian Journal of Microbiology 37 (1991) 751–757.
- [67] S.I. Patzer, R. Albrecht, V. Braun, K. Zeth, Structural and mechanistic studies of pesticin, a bacterial homolog of phage lysozymes, Journal of Biological Chemistry 287 (2012) 23381–23396.
- [68] J. Kock, T. Olschlager, R.M. Kamp, V. Braun, Primary structure of colicin M, an inhibitor of murein biosynthesis, Journal of Bacteriology 169 (1987) 3358–3361.
- [69] E. Schramm, J. Mende, V. Braun, R.M. Kamp, Nucleotide sequence of the colicin B activity gene cba: consensus pentapeptide among TonB-dependent colicins and receptors, Journal of Bacteriology 169 (1987) 3350–3357.
- [70] J. Konisky, F.M. Richards, Characterization of colicin Ia and colicin Ib purification and some physical properties, Journal of Biological Chemistry 245 (1970) 2972–2978.
- [71] H. Pilsl, V. Braun, Evidence that the immunity protein inactivates colicin 5 immediately prior to the formation of the transmembrane channel, Journal of Bacteriology 177 (1995a) 6966–6972.

- [72] H. Pilsl, V. Braun, Novel colicin 10: assignment of four domains to TonB-and TolC dependent uptake via the Tsx receptor and to pore formation, Molecular Microbiology 16 (1995b) 57–67.
- [73] A. Elfarash, J. Dingemans, L. Ye, A.A. Hassan, M. Craggs, C. Reimmann, M.S. Thomas, P. Cornelis, Pore-forming pyocin S5 utilizes the FptA ferripyochelin receptor to kill *Pseudomonas aeruginosa*, Microbiology 160 (2014) 261–269.
- [74] J. Bosák, P. Laiblova, J. Šmarda, D. Dědičová, D. Šmajs, Novel colicin F<sub>Y</sub> of *Yersinia frederiksenii* inhibits pathogenic *Yersinia* strains via YiuR-mediated reception, TonB import, and cell membrane pore formation, Journal of Bacteriology 194 (2012) 1950–1959.
- [75] J. Śmarda, V. Obdrzálek, Q. Colicin, Zentralblatt fur Bakteriologie und Hygiene 1, Abt Originale 200 (1966) 111–118.
- [76] I.F. Nes, D.B. Diep, L.S. Håvarstein, M.B. Brurberg, V. Eijsink, H. Holo, Biosynthesis of bacteriocins in lactic acid bacteria, Antonie Van Leeuwenhoek 70 (1996) 113–128.
- [77] M.J. van Belkum, M.E. Stiles, Nonlantibiotic antimicrobial peptides from lactic acid bacteria, Natural Products Reports 17 (2000) 323–365.
- [78] R. Kemperman, A. Kuipers, H. Karsens, A. Nauta, O. Kuipers, J. Kok, Identification and characterization of two novel clostridial bacteriocins, circularin A and closticin 574, Applied Environmental Microbiology 69 (2003) 1589–1597.
- [79] N.C.K. Heng, J.R. Tagg, Correspondence: What's in a name? Class distinction for bacteriocins, Nature Reviews Microbiology 4 (2006) 117–129.
- [80] G. Bierbaum, H.G. Sahl, Lantibiotics: mode of action, biosynthesis and bioengineering, Current Pharmaceutical Biotechnology 10 (2009) 2–18.
- [81] A.A. Arias, M. Ongena, B. Devreese, M. Terrak, M. Joris, Characterization of amylolysin, a novel lantibiotic from *Bacillus amyloliquefaciens* GA1, PLoS One 9 (2013) e83037.
- [82] C. Piper, P.D. Cotter, R.P. Ross, C. Hill, Discovery of medically significant lantibiotics, Current Drug Discovery Technologies 6 (2009) 1–18.
- [83] N.I. Martin, E. Breukink, Expanding role of lipid II as a target for lantibiotics, Future Microbiology 2 (2007) 513–525.
- [84] I. Widemann, E. Breukink, C. van Kraaji, O.P. Kuipers, G. Bierbaum, B. de Kruijff, H.G. Sahl, Specific binding of nisin to the peptidoglycan precursor lipid II combines pore formation and inhibition of cell wall biosynthesis for potent antibiotic activity, The Journal of Biological Chemistry 276 (2001) 1172–1179.
- [85] S. Bieler, F. Silva, C. Soto, D. Belin, Bactericidal activity of both secreted and nonsecreted microcin E492 requires the mannose permease, Journal of Bacteriology 188 (2006) 7049–7061.
- [86] D.B. Diep, M. Skaugen, Z. Salehian, H. Holo, I.F. Nes, Common mechanisms of target cell recognition and immunity for class II bacteriocins, Proceedings of the National Academy of Sciences of the United States of America 104 (2007) 2384–2389.
- [87] R. Lagos, M. Tello, G. Mercado, V. Garcia, O. Monasterio, Antibacterial and antitumorigenic properties of microcin E492, a pore-forming bacteriocin, Current Pharmaceutical Biotechnology 10 (2009) 74–85.
- [88] P.D. Cotter, An 'Upp'-turn in bacteriocin receptor identification, Molecular Microbiology 92 (2014) 1159–1163.
- [89] G. Uzelac, M. Kojic, J. Lozo, T. Aleksandrzak-Piekarczyk, C. Gabrielsen, T. Kristensen, I.F. Nes, D.B. Diep, A Zn-dependent metallopeptidase is responsible for sensitivity to LsbB, a class II leaderless bacteriocin of *Lactococcus lactis* subsp. *lactis* BGMN1-5, Journal of Bacteriology 195 (2013) 614–5621.
- [90] M. Kjos, C. Oppegård, D.B. Diep, I.F. Nes, J.W. Veening, J. Nissen-Meyer, T. Kristensen, Sensitivity to the two-peptide bacteriocin lactococcin G is dependent on UppP, an enzyme involved in cellwall synthesis, Molecular Microbiology 92 (2014) 1177–1187.

- [91] E. Strahsburger, M. Baeza, O. Monasterio, R. Lagos, Cooperative uptake of microcin E492 by receptors FepA, Fiu, and Cir and inhibition by the siderophore Enterochelin and its dimeric and trimeric hydrolysis products, Antimicrobial Agents and Chemotherapy 49 (2005) 3083–3086.
- [92] S. Matsuoka, N. Shinohara, T. Takahashi, M. Iida, M. Inoue, Functional analysis of synthetic substructures of polytheonamide B: a transmembrane channel-forming peptide, Angewandte Chemie 123 (2011) 4981–4985.
- [93] M. Iwamoto, H. Shimizu, I. Muramatsu, S. Oiki, A cytotoxic peptide from a marine sponge exhibits ion channel activity through vectorial-insertion into the membrane, FEBS Letters 584 (2010) 3995–3999.
- [94] T.R.H.M. Kouwen, E.N. Trip, E.L. Denham, M.J.J.B. Sibbald, J.Y.F. Dubois, J.M. van Dijl, The large mechanosensitive channel MscL determines bacterial susceptibility to the bacteriocin sublancin 168, Antimicrobial Agents and Chemotherapy 53 (2009) 4702–4711.
- [95] D.A. Phoenix, F. Harris, M. Mura, S.R. Dennison, The increasing role of phosphatidylethanolamine as a lipid receptor in the action of host defence peptides, Progress in Lipid Research 59 (2015) 26–37.
- [96] F.E. Märki, E. Hänni, A. Fredenhagen, J. van Oostrum, Mode of action of the lanthioninecontaining peptide antibiotics duramycin, duramycin B and C, and cinnamycin as indirect inhibitors of phospholipase A2, Biochemical Pharmacology 42 (1991) 2027–2035.
- [97] I. Mathavan, S. Séverine Zirah, S. Shahid Mehmood, H.G. Choudhury, C. Goulard, Y. Li, C.V. Robinson, S. Rebuffat, K. Konstantinos Beis, Structural basis for hijacking siderophore receptors by antimicrobial lasso peptides, Nature Chemical Biology 10 (2014) 340–342.
- [98] M.C. Bagley, J.W. Dale, E.A. Merritt, X. Xiong, Thiopeptide antibiotics, Chemical Reviews 105 (2005) 685–714.
- [99] K. Kuwano, N. Noriko Tanaka, T. Shimizu, K. Nagatoshi, S. Nou, K. Sonomoto, Dual antibacterial mechanisms of nisin Z against Gram-positive and Gram-negative bacteria, International Journal of Antimicrobial Agents 26 (2005) 396–402.
- [100] S.H. Paik, A. Chakicherla, J.N. Hansen, Identification and characterization of the structural and transporter genes for, and the chemical and biological properties of, sublancin 168, a novel lantibiotic produced by *Bacillus subtilis* 168, Journal of Biological Chemistry 273 (1998) 23134–23142.
- [101] A.S. Mazzotta, A.D. Crandall, T.J. Montville, Nisin resistance in *Clostridium botulinum* spores and vegetative cells, Applied and Environmental Microbiology 63 (1997) 2654–2659.
- [102] I.M. Gut, A.M. Prouty, J.D. Ballard, W.A. van der Donk, S.R. Blanke, Inhibition of *Bacillus anthracis* spore outgrowth by nisin, Antimicrobial Agents and Chemotherapy 52 (2008) 4281–4288.
- [103] A. Maldonado-Barragán, N. Cárdenas, B. Martínez, J.L. Ruiz-Barba, J.F. Fernández-Garayzábal, J. M. Rodríguez, Garvicin A, a novel class IId bacteriocin from *Lactococcus garvieae* that inhibits septum formation in *L. garvieae* strains, Applied and Environmental Microbiology 79 (2013) 4336–4346.
- [104] B. Martínez, A. Rodríguez, J.E. Suárez, Lactococcin 972, a bacteriocin that inhibit sseptum formation in lactococci, Microbiology 146 (2000) 949–955.
- [105] L.A. Draper, P.D. Cotter, C. Hill, R.P. Ross, Lantibiotic resistance, Microbiology and Molecular Biology Reviews 79 (2015) 171–191.
- [106] B.E. García-Almendárez, I.K.O. Cann, S.E. Martin, I. Guerrero-Legarreta, C. Regalado, Effect of *Lactococcus lactis* UQ2 and its bacteriocin on *Listeria monocytogenes* biofilms, Food Control 19 (2008) 670–680.
- [107] T.H. Howell, J.P. Fiorellini, P. Blackburn, S.J. Projan, J. De la Harpe, R.C. Williams, The effect of a mouthrinse based on nisin, a bacteriocin, on developing plaque and gingivitis in beagle dogs, Journal of Clinical Periodontology 20 (1993) 335–339.

- [108] S.R. Turner, R.M. Love, K.M. Lyons, An *in vitro* investigation of the antibacterial effect of nisin in root canals and canal wall radicular dentin, International Endodontic Journal 37 (2004) 664–671.
- [109] N.P. Guerra, A.B. Araujo, A.M. Barrera, A.T. Agrasar, C.L. Macias, J. Carballo, L. Pastrana, Antimicrobial activity of nisin adsorbed to surfaces commonly used in the food industry, Journal of Food Proteins 68 (2005) 1012–1101.
- [110] J.M. Shin, I. Ateia, J.R. Paulus, H. Liu, J.C. Fenno, A.H. Rickard, Y.L. Kapila, Antimicrobial nisin acts against saliva derived multi-species biofilms without cytotoxicity to human oral cells, Frontiers in Microbiology 18 (2015) 6–617.
- [111] N. Schnell, K.D. Entian, U. Schneider, F. Götz, H. Zahner, Prepeptide sequence of epidermin, a ribosomally synthesized antibiotic with four sulphide-rings, Nature 333 (1988) 276–278.
- [112] R.R. Bonelli, T. Schneider, H.G. Sahl, I. Wiedemann, Insights into *in vivo* activities of lantibiotics from gallidermin and epidermin mode-of-action studies, Antimicrobial Agents and Chemotherapy 50 (2006) 1449–1457.
- [113] S. Zoll, M. Schlag, A.V. Shkumatov, M. Rautenberg, D.I. Svergun, F. Götz, T. Stehle, Ligand-binding properties and conformational dynamics of autolysin repeat domains in staphylococcal cell wall recognition, Journal of Bacteriology 194 (2012) 3789–3802.
- [114] C. Gerke, A. Kraft, R. Süssmuth, O. Schweitzer, F. Götz, Characterization of the N-acetylglucosaminyltransferase activity involved in the biosynthesis of the *Staphylococcus epidermidis* polysaccharide intercellular adhesin, Journal of Biological Chemistry 273 (1998) 18586–18593.
- [115] M.B. Fontana, M.C.F. Bastos, A. Brandelli, Bacteriocins Pep5 and epidermin inhibit Staphylococcus epidermidis adhesion to catheters, Currents Microbiology 52 (2006) 350–353.
- [116] J. Parisot, S. Carey, E. Breukink, W.C. Chan, A. Narbad, B. Bonev, Molecular mechanism of target recognition by subtilin, a class I lanthionine antibiotic, Antimicrobial Agents and Chemotherapy 52 (2008) 612–618.
- [117] M. Mota-Meira, G. LaPointe, C. Lacroix, M.C. Lavoie, MICs of mutacin B-Ny266, nisin A, vancomycin, and oxacillin against bacterial pathogens, Antimicrobial Agents and Chemotherapy 44 (2000) 24–29.
- [118] M. Mota-Meira, H. Morency, M.C. Lavoie, *In vivo* activity of mutacin B-Ny266, Journal of Antimicrobial Chemotherapy 56 (2005) 869–871.
- [119] W.W. Niu, H.C. Neu, Activity of mersacidin, a novel peptide, compared with that of vancomycin, teicoplanin, and daptomycin, Antimicrobial Agents and Chemotherapy 35 (1991) 998–1000.
- [120] D. Kruszewska, H.G. Sahl, G. Bierbaum, U. Pag, S.O. Hynes, A. Ljungh, Mersacidin eradicates methicillin-resistant *Staphylococcus aureus* (MRSA) in a mouse rhinitis model, Journal of Antimicrobial Chemotherapy 54 (2004) 648–653.
- [121] A.N. Appleyard, S. Choi, D.M. Read, A. Lightfoot, S. Boakes, A. Hoffmann, Dissecting structural and functional diversity of the lantibiotic mersacidin, Chemical Biology 16 (2009) 490–498.
- [122] J.K. Kettenring, A. Malabarba, K. Vekey, B. Cavalleri, Sequence determination of actagardine, a novel lantibiotic, by homonuclear 2D NMR spectroscopy, Journal of Antibiotics 43 (1990) 1082–1088.
- [123] H. Grasemann, F. Stehling, H. Brunar, R. Widmann, T.W. Laliberte, L. Molina, G. Döring, F. Ratjen, Inhalation of moli1901 in patients with cystic fibrosis, Chest 131 (2007) 1461–1466.
- [124] M. Galvin, C. Hill, R.P. Ross, Lacticin 3147 displays activity in buffer against gram-positive bacterial pathogens which appear insensitive in standard plate assays, Letters in Applied Microbiology 28 (1999) 355–358.
- [125] M.P. Ryan, M.C. Rea, C. Hill, R.P. Ross, An application in cheddar cheese manufacture for a strain of *Lactococcus lactis* producing a novel broad-spectrum bacteriocin, lacticin 3147, Applied and Environmental Microbiology 62 (1996) 612–619.

- [126] M.P. Ryan, W.J. Meaney, R.P. Ross, C. Hill, Evaluation of lacticin 3147 and a Teat seal containing this bacteriocin for inhibition of mastitis pathogens, Applied and Environmental Microbiology 64 (1998) 2287–2290.
- [127] C. Piper, L.A. Draper, P.D. Cotter, R.P. Ross, C.J. Hill, A comparison of the activities of lacticin 3147 and nisin against drug-resistant *Staphylococcus aureus* and *Enterococcus* species, Antimicrobial Chemotherapy 64 (2009) 546–551.
- [128] J.D. Hillman, Genetically modified *Streptococcus mutans* for the prevention of dental caries, Antonie Van Leeuwenhoek 82 (2002) 361–366.
- [129] J.D. Hillman, J. Mo, E. McDonell, D. Cvitkovitch, C.H. Hillman, Modification of an effector strain for replacement therapy of dental caries to enable clinical safety trials, Journal of Applied Microbiology 102 (2007) 1209–1219.
- [130] C.E. Shelburne, F.Y. An, V. Dholpe, A. Ramamoorthy, D.E. Lopatin, M.S. Lantz, The spectrum of antimicrobial activity of the bacteriocin subtilosin A, Journal of Antimicrobial Chemotherapy 59 (2007) 297–300.
- [131] K.S. Noll, P.J. Sinko, M.L. Chikindas, Elucidation of the molecular mechanisms of action of the natural antimicrobial peptide subtilosin against the bacterial vaginosis-associated pathogen *Gardnerella vaginalis*, Probiotics and Antimicrobial Proteins 3 (2011) 41–47.
- [132] M.C. Rea, C.S. Sit, E. Clayton, P.M. O'Connor, R.M. Whittal, J. Zheng, J.C. Vederas, R.P. Ross, C. Hill, Thuricin CD, a posttranslationally modified bacteriocin with a narrow spectrum of activity against *Clostridium difficile*, Proceedings of the National Academy Science of United States of America 107 (2010) 9352–9357.
- [133] Y. Kobayashi, M. Ichioka, T. Hirose, K. Nagai, A. Matsumoto, H. Matsui, H. Hanaki, R. Masuma, Y. Takahashi, S. Omura, T. Sunazuka, Bottromycin derivatives: efficient chemical modifications of the ester moiety and evaluation of anti-MRSA and anti-VRE activities, Bioorganic & Medicinal Chemistry Letters 20 (2010) 6116–6120.
- [134] T.J. Oman, J.M. Boettcher, H. Wang, X.N. Okalibe, W.A. van der Donk, Sublancin is not a lantibiotic but an S-linked glycopeptide, Nature Chemical Biology 7 (2011) 78–80.
- [135] Y.S. Hsieh, B.L. Wilkinson, M.R. O'Connell, J.P. Mackay, J.M. Matthews, R.J. Payne, Synthesis of the bacteriocin glycopeptide sublancin 168 and S-glycosylated variants, Organic Letters 14 (2012) 1910–1913.
- [136] Q. Wang, X. Zeng, S. Wang, C. Hou, F. Yang, X. Ma, P. Thacker, S. Qiao, The bacteriocin sublancin attenuates intestinal injury in young mice infected with *Staphylococcus aureus*, The Anatomical Record 297 (2014) 1454–1461.
- [137] J.A. Leeds, M. Sachdeva, S. Mullin, J. Dzink-Fox, M.J. Lamarche, Mechanism of action of, and mechanism of reduced susceptibility to the novel anti-*Clostridium difficile* compound LFF571, Antimicrobioal Agents and Chemotherapy 56 (2012) 4463–4465.
- [138] C. Zhang, K. Herath, H. Jayasuriya, J.G. Ondeyka, D.L. Zink, J. Occi, G. Birdsall, J. Venugopal, M. Ushio, B. Burgess, P. Masurekar, J.F. Barrett, S.B. Singh, Thiazomycins, thiazolyl peptide antibiotics from *Amycolatopsis fastidiosa*, Journal of Natural Products 72 (2009) 841–847.
- [139] R. Chauhan, J. Abraham, Antimicrobial activity of *Amycolatopsis orientalis* JAR10 from agricultural field, International Journal of Pharmaceutical Science and Research 6 (2015) 755–757.
- [140] W. Li, J.E. Leet, H.A. Ax, D.R. Gustavson, D.M. Brown, L. Tureer, K. Brown, J. Clark, H. Yang, Nocathiacins, new thiazolyl peptide antibiotics from *Nocardia* sp. I. Taxonomy, fermentation and biological activities, The Journal of Antibiotics 56 (2003) 226–231.
- [141] L. Cursino, D. Smajs, J. Smarda, R.M. Nardi, J.R. Nicoli, E. Chartone-Souza, A.M. Nascimento, Exoproducts of the *Escherichia coli* strain H22 inhibiting some enteric pathogens both *in vitro* and in vivo, Journal of Applied Microbiology 100 (2006) 821–829.

- [142] F.E. Lopez, P.A. Vincent, A.M. Zenoff, R.A. Salomón, R.,N. Farías, Efficacy of microcin J25 in biomatrices and in a mouse model of *Salmonella* infection, Journal of Antimicrobial Chemotherapy 59 (2007) 676–680.
- [143] C. Asensio, J.C. Perez-Diaz, A new family of low molecular weight antibiotics from enterobacteria, Biochemical and Biophysical Research Communications 69 (1976) 7–14.
- [144] F. Baquero, F. Moreno, The microcins, FEMS Microbiology Letters 23 (1984) 117-124.
- [145] S.M. Devi, P.M. Halami, Detection and characterization of pediocin Pa-1/AcH like bacteriocin producing lactic acid bacteria, Current Microbiology 63 (2011) 181–185.
- [146] L. Acuña, G. Picariello, F. Sesma, R.D. Morero, A. Bellomio, A new hybrid bacteriocin, Ent35–MccV, displays antimicrobial activity against pathogenic Gram-positive and Gram-negative bacteria, FEBS Open Biology 2 (2012) 12–19.
- [147] J. Jasniewski, C. Cailliez-Grimal, I. Chevalot, J.B. Millière, A.M. Revol-Junelles, Interactions between two carnobacteriocins Cbn BM1 and Cbn B2 from *Carnobacterium maltaromaticum* CP5 on target bacteria and Caco-2 cells, Food and Chemical Toxicology 47 (2009) 893–897.
- [148] S. Corr, Y. Li, C.U. Riedel, P.W. O'Tooles, C. Hill, C.G. Gahan, Bacteriocin production as a mechanism for the antiinfective activity of *Lactobacillus salivarius* UCC118, Proceedings of the National Academy Science of the United States of America 104 (2007) 7617–7621.
- [149] E.F. O'Shea, P.M. O'Connor, E.J. Raftis, P.W. O'Toole, C. Stanton, P.D. Cotter, R.P. Ross, C. Hill, Production of multiple bacteriocins from a single locus by gastrointestinal strains of *Lactobacillus salivarius*, Journal of Bacteriology 193 (2011) 6973–6982.
- [150] R. Nandakumar, K. Talapatra, Quantitative profiling of bacteriocins present in dairy-free probiotic preparations of *Lactobacillus acidophilus* by nanoliquid chromatography-tandem mass spectrometry, Journal of Dairy Science 4 (2014) 1999–2008.
- [151] M.J. Grande-Burgos, R.P. Pulido, L.A.M. Del Carmen, A. Gálvez, R. Lucas, The cyclic antibacterial peptide enterocin AS-48: isolation, mode of action, and possible food applications, International Journal of Molecular Sciences 12 (2014) 22706–22727.
- [152] S. Sandiford, M. Upton, Identification, characterization, and recombinant expression of epidermicin NI01, a novel unmodified bacteriocin produced by *Staphylococcus epidermidis* that displays potent activity against Staphylococci, Antimicrobial Agents and Chemotherapy 56 (2012) 1539–1547.
- [153] T. Zendo, S. Koga, Y. Shigeri, J. Nakayama, K. Sonomoto, Lactococcin Q, a novel two-peptide bacteriocin produced by *Lactococcus lactis* QU 4, Applied and Environmental Microbiolology 72 (2006) 3383–3389.
- [154] H. Morency, M. Mota-Meira, G. LaPointe, C. Lacroix, M.C. Lavoie, Comparison of the activity spectra against pathogens of bacterial strains producing a mutacin or a lantibiotic, Canadian Journal of Microbiology 47 (2001) 322–331.
- [155] C.I. Cheigh, Y.R. Pyun, Nisin biosynthesis and its properties, Biotechnology Letters 27 (2005) 1641–1648.
- [156] Z. Tong, Y. Zhang, J. Ling, J. Ma, L. Huang, L. Zhang, In vitro study on the effects of nisin on the antibacterial activities of 18 antibiotics against *Enterococcus faecalis*, PLoS One 9 (2) (2014a).
- [157] Z. Tong, L. Ni, J. Ling, Antibacterial peptide nisin: a potential role in the inhibition of oral pathogenic bacteria, Peptides 60 (2014b) 32–40.
- [158] M. De Kwaadsteniet, K.T. Doeschate, L.M.T. Dicks, Nisin F in the treatment of respiratory tract infections caused by *Staphylococcus aureus*, Letters in Applied Microbiology 48 (2009) 65–70.
- [159] R.E. Wirawan, N.A. Klesse, R.W. Jack, J.R. Tagg, Molecular and genetic characterization of a novel nisin variant produced by *Streptococcus uberis*, Applied and Environmental Microbiology 72 (2006) 1148–1156.

- [160] J.W. Mulders, I.J. Boerrgter, H.S. Rollema, R.J. Siezen, W.M. de Vos, Identification and characterization of the lantibiotic nisin Z, a natural nisin variant, European Journal of Biochemistry 201 (1991) 581–584.
- [161] O. McAuliffe, M. Ryan, R.P. Ross, C. Hill, P. Breeuwer, T. Abee, Lacticin 3147, a broad-spectrum bacteriocin which selectively dissipates the membrane potential, Applied and Environmental Microbiology 64 (1998) 439–445.
- [162] A. Dufour, T. Hindre, D. Haras, J.P. Le Pennec, The biology of lantibiotics from the lacticin 481 group is coming of age, FEMS Microbiology Reviews 31 (2007) 134–167.
- [163] I. Wiedemann, R. Benz, H.G. Sahl, Lipid II-mediated pore formation by the peptide antibiotic nisin: a black lipid membrane study, Journal of Bacteriology 186 (2004) 3259–3261.
- [164] F. Yoneyama, Y. Imura, K. Ohno, T. Zendo, J. Nakayama, K. Matsuzaki, K. Sonomoto, Peptide-lipid huge toroidal pore, a new antimicrobial mechanism mediated by a lactococcal bacteriocin, lacticin Q, Antimicrobial Agents and Chemotherapy 53 (2009) 3211–3217.
- [165] A. Giacometti, O. Cirioni, F. Barchiesi, G. Scalise, In vitro activity and killing effect of polycationic peptides on methicillin-resistant *Staphylococcus aureus* and interactions with clinically used antibiotics, Diagnostic Microbiology and Infectious Disease 38 (2000) 115–118.
- [166] W. Brumfitt, M.R. Salton, J.M. Hamilton-Miller, Nisin, alone and combined with peptidoglycanmodulating antibiotics: activity against methicillin-resistant *Staphylococcus aureus* and vancomycin-resistant enterococci, Journal of Antimicrobial Chemotherapy 50 (2002) 731–734.
- [167] K. Steven, B. Sheldon, N. Klapes, T. Klaenhammer, Nisin treatment for inactivation of *Salmonella* species and other Gram-negative bacteria, Applied and Environmental Microbiology 57 (1991) 3613–3615.
- [168] L. Hongxia, P. Houbao, H. Zhinan, F. Guangli, L. Dapeng, The antimicrobial effects and synergistic antibacterial mechanism of the combination of ε-Polylysine and nisin against *Bacillus subtilis*, Food Control 47 (2015) 444–450.
- [169] R. Kellner, G. Jung, T. Horner, H. Zahner, N. Schnell, K.D. Entian, F. Götz, Gallidermin: a new lanthionine-containing polypeptide antibiotic, European Journal of Biochemistry 177 (1988) 53–59.
- [170] T. Sedgwick, M.J. Dawson, Novacta targets *Clostridium difficile* the natural way, MedNous 6 (2009) 8-9.
- [171] S. Donadio, S. Maffioli, P. Monciardini, M. Sosio, D. Jabes, Antibiotic discovery in the twenty-first century: current trends and future perspectives, Current Pharmaceutical Biotechnology 63 (2010) 423–430.
- [172] T. Stein, *Bacillus subtilis* antibiotics: structures, syntheses and specific functions, Molecular Microbiology 56 (2005) 845–857.
- [173] P. Sass, A. Jansen, C. Szekat, V. Sass, H.-G. Sahl, G. Bierbaum, The lantibiotic mersacidin is a strong inducer of the cell wall stress response of *Staphylococcus aureus*, BMC Microbiology 8 (2008) 186.
- [174] S. Chatterjee, D.K. Chatterjee, R.H. Jani, J. Blumbach, B.N. Ganguli, N. Klesel, M. Limbert, G. Seibert, Ersacidin, a new antibiotic from *Bacillus*. In vitro and in vivo antibacterial activity, The Journal of Antibiotics 45 (1992) 839–845.
- [175] H. Brötz, G. Bierbaum, A. Markus, E. Molitor, H.G. Sahl, Mode of action of the lantibiotic mersacidin: inhibition of peptidoglycan biosynthesis via a novel mechanism? Antimicrobial Agents and Chemotherapy 39 (1995) 714–719.
- [176] S.W. Fuchs, T.W. Jaskolla, S. Bochmann, P. Kötter, T. Wichelhaus, M. Karas, Entianin, a novel subtilin-like lantibiotic from *Bacillus subtilis* subsp. *spizizenii* DSM 15029T with high antimicrobial activity, Applied and Environmental Microbiology 77 (2011) 1698–1707.

- [177] L. Chopra, G. Singh, V. Choudhary, D.K. Sahoo, Sonorensin: an antimicrobial peptide, belonging to the Heterocycloanthracin subfamily of bacteriocins, from a new marine isolate, *Bacillus sonorensis* MT93, Applied and Environmental Microbiology 80 (2014) 2981–2990.
- [178] C. Zhang, J. Occi, P. Masurekar, J.F. Barrett, D.L. Zink, S. Smith, R. Onishi, S. Ha, O. Salazar, O. Genilloud, A. Basilio, F. Vicente, C. Gill, E.J. Hickey, K. Dorso, M. Motyl, B. Sheo, Isolation, structure, and antibacterial activity of philipimycin, a thiazolyl peptide discovered from *Actinoplanes philippinensis* MA7347, Journal of American Chemical Society 130 (2008) 12102–12110.
- [179] S.B. Singh, J. Occi, H. Jayasuriya, K. Herath, M. Motyl, K. Dorso, C. Gill, E. Hickey, M. Karen, K.M. Overbye, J.F. Barrett, P. Masurekar, Antibacterial evaluations of thiazomycin a potent thiazolyl peptide antibiotic from *Amycolatopsis fastidiosa*, The Journal of Antibiotics 60 (2007) 565–571.
- [180] R.P. Morris, J.A. Leeds, H.U. Naegeli, L. Oberer, K. Memmert, E. Weber, M.J. LaMarche, C.N. Parker, N. Burrer, S. Esterow, A.E. Hein, E.K. Schmitt, P.K. Morris, Ribosomally synthesized thiopeptide antibiotics targeting elongation factor Tu, Journal of American Chemical Society 131 (2009) 5946–5955.
- [181] A. Bartoloni, A. Mantella, B.P. Goldstein, R. Dei, M. Benedetti, S. Sbaragli, F. Paradisi, In vitro activity of nisin against clinical isolates of *Clostridium difficile*, Journal of Chemotherapy 16 (2004) 119–121.
- [182] S. van Kuijk, K.S. Noll, M.L. Chikindas, The species-specific mode of action of the antimicrobial peptide subtilosin against *Listeria monocytogenes* Scott A, Letters in Applied Microbiology 54 (2011) 52–58.
- [183] Y. Turovskiy, T. Cheryian, A. Algburi, R.E. Wirawan, P. Takhistov, P.J. Sinko, M.L. Chikindas, Susceptibility of *Gardnerella vaginalis* biofilms to natural antimicrobials subtilosin, ε-poly-l-lysine, and lauramide arginine ethyl ester, Infectious Diseases in Obstetrics and Gynecology 2012 (2012) 284762.
- [184] K.E. Sutyak, R.E. Wirawan, A.A. Aroutcheva, M.L. Chikindas, Isolation of the *Bacillus subtilis* antimicrobial peptide subtilosin from the dairy product-derived *Bacillus amyloliquefaciens*, Journal of Applied Microbiology 104 (2008) 1067–1074.
- [185] A. Swidsinski, W. Mendling, V. Loening-Baucke, S. Swidsinski, Y. Dörffel, J. Scholze, H. Lochs, H. Verstraelen, An adherent *Gardnerella vaginalis* biofilm persists on the vaginal epithelium after standard therapy with oral metronidazole, American Journal of Obstetrics & Gynecology 198 (2008) e1–81.
- [186] K.S. Noll, M.N. Prichard, A. Khaykin, P.J. Sinko, M.L. Chikindas, The natural antimicrobial peptide subtilosin acts synergistically with glycerol monolaurate, lauric arginate and ε-poly-L-lysine against bacterial vaginosis-associated pathogens but not human lactobacilli, Antimicrobial Agents and Chemotherapy 56 (2012) 1756–1761.
- [187] N.I. Torres, K.S. Noll, S. Xu, J. Li, Q. Huang, P.J. Sinko, Safety, formulation and in vitro antiviral activity of the antimicrobial peptide subtilosin against herpes simplex virus type 1, Probiotics and Antimicrobial Proteins 5 (2013) 26–35.
- [188] V.G. Eijsink, M. Skeie, P.H. Middelhoven, M.B. Brurberg, I.F. Nes, Comparative studies of class IIa bacteriocins of lactic acid bacteria, Applied and Environmental Microbiology 64 (1998) 3275–3281.
- [189] D. Drider, G. Fimland, Y. Hechard, L.M. McMullen, H. Prevost, The continuing story of class IIa bacteriocins, Microbiology and Molecular Biology Reviews 70 (2006) 564–582.
- [190] E. Salvucci, E.M. Hebert, F. Sesma, L. Saavedra, Combined effect of synthetic enterocin CRL35 with cell wall, membrane-acting antibiotics and muranolytic enzymes against Listeria cells, Letters in Applied Microbiology 51 (2010) 191–195.
- [191] K.O. Honikel, Principles of curing, in: F. Toldrá (Ed.), Handbook of Fermented Meat and Poultry, Blackwell Publishing Ltd, Oxford, UK, 2007, pp. 1–545.

- [192] G.E. Allison, C. Fremaux, T.R. Klaenhammer, Expansion of bacteriocin activity and host range upon complementation of two peptides encoded within the lactacin F operon, Journal of Bacteriology 176 (1994) 2235–2241.
- [193] S. Flynn, D. van Sinderen, G.M. Thornton, H. Holo, I.F. Nes, J.K. Collins, Characterization of the genetic locus responsible for the production of ABP-118, a novel bacteriocin produced by theprobiotic bacterium *Lactobacillus salivarius* subsp. *salivarius* UCC118, Microbiology 148 (2002) 973–984.
- [194] M. Sanchez-Hidalgo, M. Montalban-Lopez, R. Cebrian, E. Valdivia, M. Martinez-Bueno, M. Maqueda, AS 48 bacteriocin: close to perfection, Cellular and Molecular Life Science 68 (2011) 2845–2857.
- [195] H. Abriouel, R. Lucas, N. Ben Omar, E. Valdivia, A. Gálvez, Potential applications of the cyclic peptide enterocin AS-48 in the preservation of vegetable foods beverages, Probiotics and Antimicrobial Proteins 2 (2010) 77–89.
- [196] H. Khan, S. Flint, Y. Pak-Lam, Enterocins in food preservation, International Journal of Food Microbiology 141 (2010) 1–10.
- [197] M. Maqueda, A. Gálvez, M. Martínez-Bueno, M.J. Sánchez-Barrena, C. González, A. Albert, M. Rico, E. Valdivia, Peptide AS-48, prototype of a new class of cyclic bacteriocins, Current Protein and Peptide Science 5 (2004) 399–416.
- [198] T. Berić, S. Stanković, V. Draganić, M. Kojić, J. Lozo, D. Fira, Novel antilisterial bacteriocin licheniocin 50.2 from *Bacillus licheniformis* VPS50.2 isolated from soil sample, Journal of Applied Microbiology 116 (2014) 502–510.
- [199] E.A. Svetoch, V.P. Levchuk, V.D. Pokhilenko, B.V. Eruslanov, E.V. Mitsevich, I.P. Mitsevich, V.V. Perelygin, Y.G. Stepanshin, Inactivating methicillin-resistant *Staphylococcus aureus* and other pathogens by use of bacteriocins OR 7 and E 50–52, Journal of Clinical Microbiology 46 (2008) 3863–3865.
- [200] J.A.M. Jordi-Bart, B. Khalil, M.E. Caroline, F. van Heeswijk, L. van Knapen, J.A. Lipman, Sensitivity of Shiga toxin-producing *Escherichia coli* (STEC) strains for colicins under different experimental conditions, FEMS Microbiology Letters 204 (2001) 329–334.
- [201] C.H. Stahl, T.R. Callaway, L.M. Lincoln, S.M. Lonergan, K.J. Genovese, Inhibitory activities of colicins against *Escherichia coli* strains responsible for postweaning diarrhea and edema disease in swine, Antimicrobial Agents and Chemotherapy 48 (2004) 3119–3121.
- [202] S.A. Cutler, S.M. Lonergan, N. Cornick, A.K. Johnson, C.H. Stahl, Dietary inclusion of colicin e1 is effective in preventing postweaning diarrhea caused by F18-positive *Escherichia coli* in pigs, Antimicrobial Agents and Chemotherapy 51 (2007) 3830–3835.
- [203] J.F. Garcia-Bustos, N. Pezzi, E. Mendez, Structure and mode of action of microcin 7, an antibacterial peptide produced by *Escherichia coli*, Antimicrobial Agents and Chemotherapy 27 (1985) 791–797.
- [204] R. Soudy, L. Wang, K. Kaur, Synthetic peptides derived from the sequence of a lasso peptide microcin J25 show antibacterial activity, Bioorganic and Medicinal Chemistry 20 (2012) 1794–1800.
- [205] L.S. Havarstein, H. Holo, I.F. Nes, The leader peptide of colicin V shares consensus sequences with leader peptides that are common among peptide bacteriocins produced by Gram-positive bacteria, Microbiology 140 (1994) 2383–2389.
- [206] X. Thomas, D. Destoumieux-Garzon, J. Peduzzi, C. Alain, A. Blond, N. Birlirakis, C. Goulard, L. Dubost, R. Thai, J.C. Tabet, S. Rebuffat, Siderophore peptide, a new type of post-translationally modified antibacterial peptide with potent activity, The Journal of Biological Chemistry 279 (2004) 28233–28242.

- [207] V. Hancock, M. Dahl, P. Klemm, Probiotic *Escherichia coli* strain Nissle 1917 outcompetes intestinal pathogens during biofilm formation, Journal of Medical Microbiology 59 (2010) 392–399.
- [208] J. Kreth, J. Merritt, W. Shi, F. Qi, Competition and coexistence between *Streptococcus mutans* and *Streptococcus sanguinis* in the dental biofilm, Journal of Bacteriology 187 (2005) 7193–7203.
- [209] K. Tait, I.W. Sutherland, Antagonistic interactions amongst bacteriocin-producing enteric bacteria in dual species biofilms, Journal of Applied Microbiology 93 (2002) 345–352.
- [210] R.M.Q. Shanks, A. Dashiff, J.S. Alster, D.E. Kadouri, Isolation and identification of a bacteriocin with antibacterial and antibiofilm activity from *Citrobacter freundii*, Archives of Microbiology 194 (2012) 575–587.
- [211] O. Marciset, M.C. Jeronimus-Stratingh, B. Mollet, B. Poolman, Thermophilin 13, a nontypical antilisterial poration complex bacteriocin that functions without a receptor, The Journal of Biological Chemistry 272 (1997) 1477–1484.
- [212] R. Laridi, E.E. Kheadr, R.O. Benech, J.C. Vuillemard, C. Lacroix, I. Fliss, Liposome encapsulated nisin Z: optimization, stability and release during milk fermentation, International Dairy Journal 13 (2003) 325–336.
- [213] P.S. Malheiros, D.J. Daroit, N.P. Silveira, A. Brandelli, Effect of nanovesicle-encapsulated nisin on growth of Listeria monocytogenes in milk, Food Microbiology 27 (2010) 175–178.
- [214] P.S. Malheiros, V. Sant'Anna, M. Utpott, A. Brandelli, Antilisterial activity and stability of nanovesicle-encapsulated antimicrobial peptide P34 in milk, Food Control 23 (2012) 42–47.

# **Recombinant Veterinary Vaccines**

O.A. Dellagostin\*, S.R. Felix, S. Jorge\* UNIVERSIDADE FEDERAL DE PELOTAS (UFPEL), PELOTAS, RIO GRANDE DO SUL, BRAZIL

# 17.1 Introduction

Animal vaccines are almost as old as vaccination itself and the involvement of animals and veterinary medicine in vaccine development dates from the same period. When Edward Jenner first vaccinated farmhands against smallpox in the early 19th century, the antigen he used was obtained directly from cowpox pustules of infected cattle. In fact, the word vaccine is derived from the Latin *vacca*, literally meaning "cow." It would be a few years until the first formal veterinary vaccines were introduced, but the cowpox procedure was known to work in animals from the start. Soon after the dawn of vaccinology, safety issues began to be taken into account, especially regarding the live attenuated vaccines developed after the studies of Louis Pasteur and Robert Koch. However, these safety issues were less of a concern for animal vaccines, a trend that still persists, which allows for faster development of animal vaccines compared to human vaccines.

Vaccination is an effective method of preventing a wide range of diseases in diverse animal species. The field of vaccinology has yielded several effective vaccines that have significantly reduced the impact of a number of important diseases in both companion animals and livestock. In fact, today, the vast majority of licensed vaccines are live attenuated or killed/inactivated microorganisms [26,34,55]. Live attenuated vaccines that contain targeted deletions of genes known to be responsible for virulence can be, nevertheless, very effective because they induce both cellular and humoral immune responses [9,42]. However, mechanisms of attenuation are often poorly defined and the risk associated with the potential for reversion to a virulent phenotype is a major concern [40,54]. Killed/inactivated vaccines are typically safer, although they may be less effective than attenuated vaccines, because, although they are able to induce strong antibody responses, cellular immune responses are usually poor. However, given that there are still several diseases that lack an efficient vaccine, a need remains for better and safer vaccines to prevent, control, or eradicate these diseases [13,26,41]. To this end, efforts to develop more effective vaccines against a large number of diseases using recombinant DNA technology are in progress around the world.

17

<sup>\*</sup>Corresponding Authors.

The technique of using genomic information to study vaccine development in silico, without needing to cultivate the pathogen, is called reverse vaccinology [40,50]. This genome-based approach started with the complete genome sequence of the human pathogen *Haemophilus influenzae* in 1995 [17]. Since then, a large amount of whole genome sequencing of human and/or animal pathogens has been performed [39,53,55,56]. The advent of genetic engineering has given scientists the ability to rationally design both individual microbial components and whole organisms [6]. Furthermore, the application of functional genomics and structural biology has allowed the identification of many promising new antigens [49,50]. Molecular approaches, together with a clearer understanding of the genes responsible for virulence and identification of the determinants of protective immune responses, have provided new methods for the development of novel vaccines against infectious or metabolic diseases.

Reverse vaccinology starts from the pathogen genome sequence on which a computational analysis is carried out to predict in silico which genes are most likely to be potential vaccine candidates. The selected genes can be cloned and expressed in heterologous systems, followed by immunization in an animal model and evaluation of the immune response elicited by the antigens regarding their potential as vaccine candidates. Different antigens can be combined simultaneously, allowing for the development of vaccines against multiple strains of a pathogen. This aspect is especially attractive for animal vaccines, for which vaccine cocktails are a useful vaccination option. One drawback of reverse vaccinology is that it cannot be used to predict polysaccharides or lipids, which are often included in vaccines as active compounds.

## 17.2 Types of Recombinant Vaccines

Recombinant veterinary vaccines can be broadly grouped into three kinds: DNA (or RNA), subunit recombinant, and vectored vaccines. The recombinant protein (for subunit vaccines) is made in a heterologous system in the laboratory and injected into the vaccine recipient, whereas the immunogenic protein associated with a DNA vaccine is generated by the cells of the host. Recombinant DNA vaccines are based on the expression, by the host, of biological constructs encoding proteins from specific pathogens. Recombinant vector vaccines are experimental vaccines that use an attenuated virus or bacterium to either express and multiply the antigen in the host organism or introduce DNA into host cells. A fundamental step in recombinant vaccine development is the cloning of a DNA fragment insert into a plasmid vector. Many plasmids are commercially available for such uses. The gene to be replicated is inserted into copies of a plasmid containing genes that make cells resistant to particular antibiotics. The gene is also inserted into a multiple cloning site, which is a short region containing several commonly used restriction sites allowing the easy insertion of DNA fragments. Fig. 17.1 shows typical features of prokaryotic (used for subunit and vectored vaccines) and eukarvotic (used for DNA vaccines) plasmid vectors.



**FIGURE 17.1** The transcriptional unit in plasmid vectors consists of the coding sequence of the gene of interest placed under the control of a strong promoter. A bacterial origin of replication (Ori) and an antibiotic resistance gene allow replication and selection of the plasmid in bacteria. (A) Diagram showing the typical features of an eukaryotic expression vector used to construct DNA vaccines. (B) Diagram showing a prokaryotic plasmid vector used for cloning and expression of recombinant proteins used as subunit vaccines. (C) Recombinant attenuated bacterium used as a vaccine vector, and virus vector vaccine expressing a heterologous antigen.

#### 17.2.1 DNA Vaccines

DNA vaccines offer the potential for further advancements in the production of safe and effective veterinary vaccines. A DNA (or RNA) vaccine can be defined as a plasmid containing a viral, bacterial, or parasite gene that can be expressed in mammalian cells (in the case of infectious diseases) or a gene encoding a mammalian protein (in the case of noninfectious diseases, such as cancer, autoimmunity, and allergies) [6,11,34]. The gene of interest is inserted into a plasmid, along with appropriate genetic elements such as strong eukaryotic promoters for transcriptional control, a polyadenylation signal sequence for stable and effective translation, and a bacterial origin of replication. The plasmid is transfected into host cells and transcribed into mRNA, which is then translated, resulting in the production of an antigenic protein by the host cellular machinery. The expressed proteins should have authentic posttranslational modifications (e.g., glycosylation) and tertiary structure, and consequently, conformational epitopes of the antigen should be effectively represented. The expressed proteins are recognized as foreign by the host immune system and can induce cellular and humoral immune responses (Fig. 17.2). These vaccines can be administered via a number of routes and techniques, which can alter the immune response and distribution of the protein [12,41,45]. Currently, some DNA vaccines have already been approved for commercial use in animals (see Table 17.2).

#### 17.2.2 Vectored Vaccines

Vector vaccine technology uses a vector to deliver a protective protein(s) to the immune system of the vaccinated host. The vector will usually be a virus or bacterium.



**FIGURE 17.2** DNA vaccination induces a strong humoral and cellular immune response against the plasmidencoded antigen. The delivery of antigens occurs by the intracellular route via induced MHCI-restricted CD8<sup>+</sup> responses, whereas antigens presented through the extracellular route activate MHCII-restricted CD4<sup>+</sup> responses. *APC*, antigen-presenting cell.

No extensive purification would be required and several live delivery systems have been shown to elicit strong long-lasting immunity without the need for adjuvants [12,44,54]. Both bacteria and viruses have been investigated for delivery of foreign antigens. The nucleotide sequence of these vectors has been modified using genetic engineering by inserting one or several exotic antigens and adequate upstream (promoter) and downstream (terminator) regulatory sequences. For example, the hemagglutinin of the influenza virus is responsible for attachment of the virus to cell receptors and for fusion of the viral and cellular membranes. In horses, the bestknown vector vaccine against influenza comprises two canarypox recombinant viruses expressing hemagglutinin from the H3N8 lineage. This nonreplicative canarypox-based vaccine induces humoral immunity, primes cell-mediated immunity, and demonstrates a long duration of immunity. Another approach in vectored vaccines is that involving attenuated bacteria, which can also be engineered to express exotic antigens or overexpress antigens of interest. Cattle vaccinated with a BCG strain overexpressing the Ag85B antigen (protein found on the bacterial surface) showed improved protective immune responses against Mycobacterium bovis under experimental conditions [42].
#### 17.2.3 Subunit Vaccines

This approach consists of transferring a gene encoding an antigen to a nonpathogenic host, thereby making the production of the antigen safer and generally more efficient. The recombinant antigen will then be isolated, and the subunit vaccine will be delivered as a purified recombinant protein [26]. The first of these vaccines was the highly pure surface antigen of the human hepatitis B virus expressed in yeast [11]. These novel approaches were based on studying the microorganism in terms of pathogenicity and immunogenicity to identify factors involved in virulence suitable as vaccine candidates, and have been the basis of development of some currently available vaccines. Additional benefits of this type of vaccine are that they incorporate proteins in their most native form, facilitating correct protein folding and reconstitution of conformational epitopes [8,15,18,19]. By incorporating more than one protein into a subunit vaccine, it is possible to derive immunity to more than one strain or serotype of a bacterial pathogen [12].

Cloning of the gene coding for the antigen is often necessary to better characterize and produce the identified antigen, and thus the new protein can enter vaccine development. Escherichia coli has been used extensively as a host for heterologous protein expression, but potential limitations include the yield, folding, and posttranslational modifications of the recombinant protein [24,49]. An alternative host to E. coli is the methylotrophic yeast, Pichia pastoris. This yeast strain has emerged as a powerful and inexpensive expression system for the heterologous production of recombinant proteins that allows techniques for genetic modifications, secretion of expressed proteins, posttranslational modifications, and a high yield [20,21]. The continued use of alternative expression systems has many potential advantages; foremost among these are other yeast species and plant-based systems, which may provide access to oral vaccines as well as enhanced stability of antigens. Expression of antigens in heterologous systems provides both manufacturer and user safety by eliminating the need to use a virulent or partially virulent microbe to provide immunity. Potential drawbacks of subunit vaccines are their moderate immunogenicity and requirement for adjuvants to generate robust immune responses. Types and some of the characteristics of recombinant vaccines are shown in Table 17.1.

# 17.3 Immune Responses Induced by Vaccination

The severity of disease depends on host susceptibility factors, the infecting species or strain, and the infective dose. Host susceptibility factors, such as immune competence, age, sex, and animal species and breed, have been identified as risk factors. Specific acquired immunity operates through clonal selection, whereby the B lymphocytes that produce antibodies display them on the cell surface, and when the T helper and cytotoxic lymphocytes that recognize an antigen are activated, these cells multiply and carry out immune defensive actions. In addition, immune memory is established because part of the lymphocytes that responded to the primary infection are preserved and may later

Type of Recombinant Vaccine	Characteristics
Subunit (proteins)	Well-defined composition
	No risk of pathogenicity
	Possibility of cost-efficient production and purification
	Primarily humoral immune response
	Need for adjuvant
DNA-based	Humoral and cellular immune responses (antigen presentation by both MHC class I and class II molecules)
	Long-term persistence of immunogen
	Risk of integration into host genome not completely excluded
	Unstable and quite expensive production (for RNA vaccines)
Vector-based	Cellular and humoral immune responses In vivo amplification systems available Inefficient transfection

 Table 17.1
 Types and Characteristics of Recombinant Vaccines

be reactivated in response to a second invasion by the same type of microorganism. Cellular memory depends on the persistence of antigen-sensitized T cells that are able to rapidly respond to a secondary infection. In this context, vaccination acts in a way similar to primary infection. A protective immune response to vaccination may be due to humoral immunity, the action of sensitized T lymphocytes (cellular immunity), or a combination of these factors. Recombinant subunit vaccines induce protective immunity without the risk of side effects or immune reactions caused by other parts of pathogenic bacteria, and are known to generally induce predominantly T-helper 2 (Th2) responses with weak or absent Th1 responses. Differences in the immunity induced by recombinant antigens have been observed, which can be influenced by differences in vaccine constructs, the route of immunization, the correct folding, and/or other post-translational modifications that may contribute to the antigen's capacity to generate antibodies [10,14,21].

Vectored vaccines can generate antigens for long periods after immunization, in contrast to a subunit vaccination. This increased presence and level of vaccine antigen contributes to and helps sustain a durable immune response, even augmenting the selection of higher affinity antibody-secreting cells. In subunit vaccines, the use of multiple antigens in the same immunization may represent a promising alternative [19]. Prime—boost strategies, using DNA or protein immunization in conjunction with a vectored vaccine, have also been explored as a means to avoided preexisting immunity [22]. The enhanced memory response is in part linked to the intrinsic augmentation of immunity induced by the vector.

Mucosal immunology is gaining increasing attention as an area of great potential for the development of animal vaccines. Indeed, mucosal surfaces are the major point of entry for infectious agents into the body. The mucosa contain several defined lymphoid tissues that respond specifically to invading antigens, and this immune response can be either cellular or humoral [immunoglobulin A (IgA)] [10,30,31]. The characterization of a protective immune response requires the definition of which aspects of the immune reaction are responsible for the protection afforded (systemic, humoral, mucosal, or cellular). All this information is necessary to ensure that a given in vitro potency test is also relevant for assessing in vivo efficacy and for ensuring that the relevant immune response is being measured against the appropriate antigens.

The tools of molecular biology and understanding of the mechanisms of the immune response have helped to expand the development of vaccines. The understanding of the molecular mechanisms of innate immune responses has dramatically increased, leading to the discovery of new classes of receptors such as Toll-like receptors, Nod-like receptors, and Rig-like receptors. These molecules have evolved to sense microbial infection and trigger an immune response adapted to the invading pathogens. Importantly, the innate immune reactions triggered by these receptors are also required to enhance and modulate the antigen-specific immunity [25,52,57]. The reverse vaccinology approach and the immune response elicited are represented in Fig. 17.3.

# 17.4 Veterinary Vaccine Production

Recombinant vaccines are still controversial with the general public, especially with respect to the aspects of the technology that are poorly understood by nonscientists. Thus, a concern with mutation, for instance, hinders the acceptance of naked DNA vaccines for human use, although most are known to be safe and several DNA vaccines are in human clinical trial stages. The same can be said of vectored vaccines, using either viruses or bacteria. In veterinary medicine, however, public opinion is less of a concern, and products that are "taboo" for human use are readily accepted, with concerns being productivity (in the case of livestock) and economic viability. Following this trend, the first DNA vaccine for veterinary use was licensed in 2005, against the equine West Nile virus (WNV), whereas as of 2015 there are still no DNA vaccines for human use.

Vaccine development for the veterinary market has several advantages, especially concerning licensing, and some disadvantages compared to the development of human vaccines. Considering livestock vaccines, the guiding force behind acceptance will be economic viability and food safety. Therefore, while tissue reactions against foot and mouth disease (FMD) vaccines are a concern, the loss in productivity and the discarded meat around the local inflammation site is the greater problem, with animal welfare usually considered a secondary issue. This focus on productivity is especially strong in underdeveloped and developing countries, where animal welfare is only now being discussed. The development of livestock vaccines is further facilitated by the relatively short life expectancy of these animals, with beef-producing cattle rarely reaching 3 years of age, swine reaching no more than 200 days, and poultry being slaughtered at no older than 40 days; thus there is no need for long-term safety concerns. Likewise, the generation of relatively short-term immunity will not be a problem in several animal species.



FIGURE 17.3 Schematic of different approaches to recombinant vaccine development and the cellular and humoral immune responses elicited by vaccination. APC, antigen-presenting cell; T<sub>H</sub>, T-helper cell.

Of course, there are long-lived animals that must be vaccinated, such as dairy cattle, pets, and wildlife. The disadvantages brought on by the diversity of the veterinary market will be considered further on.

Another advantage in the development of veterinary vaccines is that researchers can work directly with the target species. Like vaccines for humans, most initial research occurs in laboratory animals such as rats and mice, as these are cheaper, and a vast array of biological resources are available. Nonetheless, the advancement from these lab animals to the target species is usually immediate, because the final preclinical assays can be conducted in the latter, and usually overlap with the clinical trials themselves. These advantages, compared to human vaccines, have resulted in the commercialization of several recombinant veterinary vaccines, a selection of which is listed in Table 17.2.

Vaccine	Animal Species	Pathogen	Vaccine Type	Manufacturer
EURIFEL FeLV	Cats	Feline leukemia virus	Vectored	Merial
PUREVAX Feline Rabies	Cats	Rabies virus	Vectored	Merial
Gavac	Cattle	Ripcephalus (Boophilus) microplus	Subunit	Heber Biotec
TickGUARD	Cattle	R. (Boophilus) microplus	Subunit	Fort Dodge
RECOMBITEK rDistemper	Dogs	Canine distemper virus	Vectored	Merial
PUREVAX Ferret	Ferrets	Canine distemper virus	Vectored	Merial
Distemper				
Apex-IHN	Fish	Infectious hematopoietic necrosis virus	DNA	Novartis
ProteqFlu TE	Horses	Influenza virus and tetanus toxin	Vectored	Merial
PROTEQ-FLU/RECOMBITEK	Horses	Influenza virus	Vectored	Merial
RECOMBITEK Equine WNV	Horses	West Nile virus	Vectored	Merial
West Nile-Innovator DNA	Horses	West Nile virus	DNA	Fort Dodge
Innovax ILT	Poultry	Infectious laryngotracheitis virus	Vectored	Merck
Trovac-AI-H5	Poultry	Avian influenza virus	Vectored	Merial
Vaxxitek HVT + IBD	Poultry	Marek disease virus	Vectored	Merial
Vectormune FP-ND	Poultry	Newcastle disease virus	Vectored	Biomune
Raboral	Raccoons and coyotes	Rabies virus	Vectored	Merial
HIDATIL EG95	Sheep and goats	Echinococcus granulosus	Subunit	Tecnovax
Bayovac CSF E2	Swine	Classical swine fever virus	Vectored	Bayer
P. Circumvent	Swine	Porcine circovirus	Subunit	InterVet
PleuroStar APP	Swine	Actinobacillus pleuropneumoniae	Subunit	Novartis
Porcilis APP	Swine	A. pleuropneumoniae	Subunit	Intervet
Porcilis Pesti	Swine	Classical swine fever virus	Vectored	Intervet
PRO-VAC Circomaster	Swine	Porcine circovirus	Subunit	Vencofarma
Suvaxyn PCV2	Swine	Porcine circovirus	Vectored	Fort Dodge

Table 17.2Selection of Recombinant Veterinary Vaccines CommerciallyAvailable as of 2015

Arguably, the most significant advantage in recombinant veterinary vaccine development, compared to human vaccines, is the looser restrictions on adjuvants. Subunit vaccines may be the most straightforward and safest approach to developing recombinant vaccines; however, these are the least immunogenic. This is further hindered by the fact that the only adjuvants approved for human use are aluminum crystals, which have very limited immune boosting capabilities, compared to other available adjuvants. The combination of protective antigens and adjuvants or delivery systems may result in much-needed effective recombinant vaccines. By definition, an adjuvant is an agent that stimulates the immune system and increases the host response to an antigen without itself conferring a specific antigenic effect [3,26]. Some adjuvants act by sequestering antigens in physically restricted areas, termed depots, to provide an extended time period of antigenic stimulation. This depot effect is essential for the efficacy of the majority of human and veterinary vaccine adjuvants, particularly with vaccines consisting of pathogen subunits or recombinant subunits [24]. The immunomodulatory effects are dependent upon the particular adjuvant used in conjunction with specific antigens. Thus, several veterinary vaccines are emulsions in oil. This relatively old-fashioned technology is, nonetheless, a powerful approach that allows for a strong inflammatory response and slow antigen liberation, exactly what recombinant subunit vaccines lack. Likewise, naked DNA vaccines may require (though several studies indicate otherwise) in vivo electroporation, or the addition of substances that enhance membrane permeability, for optimum function. These methods are readily available for animal use, though none are yet considered safe for human use. The same can be said of vectored vaccines, of which rabies or WNV vaccines, vectored in canarypox and vaccinia viruses, for example, are already available for the animal market, whereas no vectored vaccine is yet in use in humans (in the United States or Europe as of 2014).

Although the advantages and the relative ease in the development of recombinant veterinary vaccines mean that there are more of these currently commercially available, there are a few disadvantages that cut back research funding. The most important disadvantage concerning the veterinary vaccine development is market size, compared to human vaccines. These differences in market size are reflected in funding volume. For instance, the US National Institutes of Health allocated over \$25 billion in support of human medical research in 2012. In the same period, the National Institute for Food and Agriculture's budget for veterinary medicine was almost 100 times less. Privately financed research follows the same trend; a successful veterinary vaccine will return a fraction of what a human vaccine returns, and that will depend on market acceptance, which brings us to another setback in vaccine development.

The second most important difficulty in veterinary vaccine development is the large number of species, both of hosts and of disease agents, which can suffer from or cause veterinary ailments, respectively. Thus, economically important species, such as livestock and mainstream pets, will be favored in vaccine development over wildlife, alternative meats, and exotic pets. Likewise, diseases that cause greater economic losses will, naturally, receive greater priority; those that cause more severe health problems may not necessarily be prioritized. This uneven distribution of research funds and effort is understood and expected; however, this means that several diseases and animals are currently unprotected.

# 17.5 Recombinant Vaccines Against Bacterial and Fungal Diseases

#### 17.5.1 Bacterial Vaccines

Novel approaches in recombinant DNA technology, genomics, and structural biology have revolutionized the way vaccine candidates are developed and will make a significant impact on the generation of safer and more effective animal vaccines [34,40,50]. Reverse vaccinology has primarily been applied to bacterial pathogens to identify proteins that can be formulated into subunit vaccines (one or more protein antigens) [12,19,26,31]. Currently, most of the bacterial vaccines licensed for animal use include live attenuated and inactivated or killed microorganisms, with varying degrees of efficacy and side effects. The development of effective vaccines against bacterial diseases with a wide range of protection and few side effects is a basic requirement for any new human vaccine. Indeed, such characteristics are just as important in animal vaccine development [12]. Research into recombinant vaccines has identified antigens that can protect distinct animal species against several infectious bacterial diseases.

An example under development as of this writing is a recombinant vaccine against leptospirosis. It is a zoonotic disease and a major public health concern in tropical and subtropical regions. There are numerous descriptions of problems associated with heat-killed, whole-cell (bacterin-type) vaccine preparations, including severe side effects (pain, nausea, fever), short-term immunity, and serovar-restricted protection [1,7,12,28,46]. Consequently, research has focused on leptospiral recombinant antigens capable of eliciting protective immunity, especially identifying virulence determinants such as the surface-exposed antigens and those present in multiple serovars/serogroups of bacteria. Some these antigens have been identified as being protective in experimental trials [8,12,21,47].

Several studies have evaluated recombinant proteins in various forms of administration and formulations, seeking to develop more effective bacterial vaccines. For example, some recombinant antigens were evaluated individually, and others were associated with attenuated bacterial or viral vectors and evaluated as cocktail or fused antigens. However, researchers have presented divergent results on vaccine efficacy for the same antigen, particularly when using a subunit formulation. Structural differences in the recombinant proteins are probably the simplest explanation for these observations. This is a recognized problem in recombinant protein expression; however, it should be possible to overcome this by using alternative expression systems and accurate mapping of the protective epitopes.

# 17.5.2 Recombinant Vaccines Against Fungal Infections

Although fungi, namely yeast, are major players in recombinant vaccine development as cell factories, vaccines to control fungal diseases are not a major concern. Few studies toward the development of recombinant vaccines against fungi of veterinary importance have been undertaken. Genera such as *Sporothrix, Aspergillus*, and *Cryptococcus*, as well as a myriad of fish microsporidia, are known to cause important diseases in pet and production animals; however, little effort has been undertaken toward developing any vaccines against these diseases, let alone recombinant vaccines [5]. *Candida* spp. are the most studied fungi regarding recombinant vaccines; however, most of these studies have been focused on human candidiasis [32]. Nevertheless, a human preparation will probably work in animals, though there will be limited commercial interest to push it forward.

# 17.6 Recombinant Vaccines Against Parasitic Diseases

Parasites affecting health in veterinary medicine can be divided into three major groups for the sake of control and vaccine development strategies, i.e., protozoan parasites, helminths, and ectoparasites. Generally, parasites pose a greater challenge for recombinant vaccine development, regardless of the major group. Their eukaryotic genomes are larger and more complex than those of viruses and bacteria, and even those of most fungi of veterinary relevance. A sequenced genome is the first step for a reverse vaccinology approach. However, although genomes of viruses and bacteria have been sequenced with ease, with several isolates being sequenced at once, this is not always possible with eukaryotic parasites. Although advances in technology are reducing costs, the production of immunogenic subunit parasitic antigens continues to pose challenges, such as posttranslational modifications, glycosylation, phosphorylation, and lipidation, among others.

Immune protection strategies similar to other, simpler disease agents can also be used against protozoan parasites, because they infect tissues or circulate in the bloodstream. Gut parasites and ectoparasites, on the other hand, will pose a greater challenge, because most of the parasite is "external" to the host and antigens are mostly hidden from the immune system. Strategies to control helminths may target those stages of the life cycle that occur in the host's tissues, or even involve IgA stimulation against antigens of the parasite's exterior. However, the multicellular, mostly inaccessible, ectoparasites pose the greatest challenge to researchers, who must work either with mouth antigens or with hidden antigens, because of the lack of any other exposure of the parasite to the host's immune system.

## 17.6.1 Protozoan Parasites

Protozoan parasites cause a myriad of diseases of veterinary importance. In most cases, the parasites require more than one host, usually an intermediate arthropod host, for the completion of the cycle. This means that protozoan agents are more prevalent in tropical and subtropical climates, where the arthropod hosts are more abundant. Cattle ticks do not occur beyond parallels 33, for example; therefore, babesiosis and anaplasmosis are also not a concern beyond these latitudes. It is a similar case for protozoan diseases that require bloodsucking flies and/or mosquitoes for their transmission. Consequently, these diseases are more common in underdeveloped and developing countries of the tropics, and vaccine production should be focused, not only on the control of these diseases, but also to allow the importation of naïve animals from outside these regions. Nevertheless, there are currently few commercial vaccines against protozoan agents.

Leishmaniasis is one of the most important protozoan zoonoses in the world, causing disease mostly in dogs and humans. Leish-Tec (Hertape Calier Saúde Animal S.A., Brazil) is a leishmaniasis vaccine licensed in Brazil. Although it was released without greater field studies, the recombinant A2 antigen used has been shown to be protective [16]. This antigen is a virulence factor of the amastigote stage of the parasite, and was foremost

among others studied as potential vaccine candidates. Leish-Tec is an example of how strong adjuvants can make simple subunit vaccines viable. It uses saponin, a proinflammatory adjuvant, in its preparation, which has been shown to be a potent adjuvant for Th1 cytokines [51].

Although the only commercially available recombinant vaccine against protozoan diseases up to 2013 was Leish-Tec [33], other antigens are expected to soon follow in its footsteps. Successful, or partially successful, experimental vaccines against the livestock and pet parasites *Toxoplasma*, *Babesia*, *Neospora*, and *Giardia*, as well as avian coccidiosis, indicate that this is the most promising market for future veterinary recombinant vaccines, at least as far as parasitic diseases are concerned.

#### 17.6.2 Helminths

Medically relevant helminths can be divided into three greater families, namely, nematodes (roundworms), trematodes (flatworms), and cestodes (tapeworms). These large, multicellular parasites have been eluding vaccine development for years. Although a few traditional (nonrecombinant) antihelminth vaccines have been commercialized, namely against the cattle lungworm *Dictyocaulus viviparus*, there are, as of this writing, trials under way for a vaccine against ovine haemonchosis in Australia. Though theoretically more feasible than ectoparasite vaccines because of the "closer" relation with the host, success has been limited, possibly due to a lack of promising protein antigens, and experts have recommended more focus on glycans as vaccine targets for a long time now [23]. Furthermore, a possible approach to helminth control would be the stimulation of mucosal immunity through IgA; however, this approach is still hardly ever used.

Nematodes (roundworms) are the most important helminths in veterinary practice, causing the greatest productivity and economic losses in livestock (and other production species such as poultry and fish), and major health risks in the pet market. Efforts to control roundworms through vaccination have thus far been frustrated, but there has been a concentration of resources to develop anti-*Haemonchus* spp. vaccines, because of the economic importance of this parasite in sheep and goats, and its ability to rapidly develop resistance to drugs. These efforts have produced some promising antigens, especially gut glycoproteins [4]. However, recombinant approaches have not yet granted protective immunity, though minor amounts of purified native protein are protective and have been commercialized in Australia (Barbervax). To circumvent the lack of protection, production of the recombinant proteins in a helminthic expression system, *Caenorhabditis elegans*, has been attempted, but no better results have been obtained thus far [43].

Cestodes (tapeworms) are generally not a burden on production animals and cause minor health risks in pets. Nonetheless, these parasites are economically important because several of them cause food-borne or otherwise zoonotic diseases. There are major concerns regarding *Taenia* spp., infecting pigs and cattle, and the canine/ovine parasite *Echinococcus granulosus*. A recombinant vaccine targeting the larval stages of

*E. granulosus* in ruminants has been commercialized in Argentina, although success has been limited. The anti-rEG95 vaccine (HIDATIL EG95) claims to hinder the capability of oncospheres to cross the gut wall and cause infection [27]. Likewise, vaccine attempts against other tapeworms have been more successful against larval-stage antigens [29] than the adult worm. Nevertheless, from an epidemiological point of view, the control of the parasite in the definitive host may be more effective. Although the definitive hosts of *Taenia* spp. are humans, few studies have been conducted on vaccines at this level. Vaccines against canine echinococcosis, on the other hand, are a possibility, and several research efforts are being made in the development of recombinant vaccines for this host [38].

Trematodes (flatworms) of importance in veterinary medicine are the liver flukes of the *Fasciola* genus, specifically *Fasciola hepatica* and, to a lesser extent, *Fasciola gigantica*. Although they are zoonotic and do cause some economic losses, especially in sheep, these parasites are generally contained in endemic areas and environmental control is feasible. Consequently, the least effort, among cestodes, has been put in developing vaccines against flatworms. Nonetheless, the "internal" colonization of tissues (as opposed to "external" colonization of the gut) facilitates immune access, and promising recombinant antigens have been identified, though no commercial vaccine is available yet [35].

#### 17.6.3 Ectoparasites

As stated, ectoparasites offer the greatest challenge for vaccine development, not only because of their size and complexity, but also because most of the parasite never enters the host, and usually only the mouthpiece is exposed during feeding. Unlike for human medicine, ectoparasites are a major concern in veterinary medicine, both as causative agents of disease and as vectors for other, mostly protozoal, agents. Ticks cause great losses in cattle populations, and fleas and mites are a prominent concern in pet markets worldwide. Though the need, and indeed the economic opportunity, for the development of vaccines against fleas, ticks, and mites has been pointed out [37], major efforts have been concentrated on the development of recombinant vaccines against the cattle tick *Ripcephalus* (*Boophilus*) *microplus*.

Considering the complexity and difficulty expected in the development of ectoparasite vaccines, it is surprising that one of the first recombinant vaccines available for veterinary use was against *R. microplus*. First introduced (and developed) in Australia in 1994, TickGUARD (Fort Dodge, Australia) is a recombinant subunit vaccine produced in and purified from *E. coli*. The same antigen, midgut protein BM86, was later produced in yeast by Cuban researchers, and marketed in Cuba (and some other countries, including Brazil) as Gavac (Heber Biotec SA, Cuba). The BM86 antigen is a midgut protein responsible for nutrient absorption by the tick. These commercial vaccines have seen limited success outside their countries of origin, with protection rates dropping when the vaccines were used in Brazil, for instance [2]. Explanations for this difference include an alleged variation in the protein between locally circulating strains. However, studies have shown that BM86 is highly conserved, even among different tick species, and that efficacy does not seem to be associated with variations in the gene [58]. Furthermore, BM86 is a concealed or hidden antigen and is, therefore, not naturally exposed to the immune system. Antibody levels do not rise with each reinfestation, and the vaccine's effectiveness is only partial, intended to be used concomitant with conventional drugs.

There are no commercial vaccines available to control other ectoparasites of veterinary importance as of this writing. Nonetheless, research toward their development is ongoing. Because there is no natural immune response against flea antigens, all potential antigens studied have been "hidden," usually from the midgut of the parasite [48], following the, possibly misleading, experience of BM86. Promising results have been obtained [36]; however, progress on these vaccines is slow, and the associated flea bite allergy antigens have received more attention.

Likewise, the development of vaccines against mites and myiasis-causing flies (or fly bots) has yet to produce promising results. The analysis of expressed sequence tags seems to be an interesting approach to finding functional antigens in these complex parasites [36]. However, other than some homologs found between house dust mites and the sheep mite *Psoroptes ovis* that seem promising, other agents have still to produce convincing antigens worthy of greater research investments.

# 17.7 Recombinant Vaccines Against Viral Diseases

Licensed recombinant viral vaccines in veterinary medicine include those protecting against pseudorabies, rabies, canine distemper, Newcastle disease, and a strain of avian influenza. Vaccines are useful tools to control avian influenza especially when biosecurity and stamping out strategies alone are not successfully implemented. Inactivated vaccines are the most widely used avian influenza vaccines although viral-vectored vaccines based on fowlpox or Newcastle disease virus are also used in some countries [41,54]. Oral rabies vaccines were first proven to be successful in the late 1960s and have been used for prevention and control of wild species from rabies over the world. Oral immunization in free-roaming dogs is one of the most practical approaches to protect humans from rabies in developing countries. The development of recombinant DNA technology has initiated a new era in rabies control. Various recombinant constructs (e.g., animal poxviruses or human or canine adenovirus as the vector) expressing the rabies glycoprotein were tested in various target and nontarget wildlife species by the oral route. The only recombinant vaccine that is now produced and used in large quantities is a vaccinia-based recombinant product expressing the rabies glycoprotein (VRG). VRG shares many basic properties with the parental vaccinia virus (Copenhagen strain) but differs in other ways, which make the vector virus safer. The deletion of the thymidine kinase gene dramatically decreases the pathogenicity of the vaccine in mice when it is injected intracerebrally and intraperitoneally. In addition, no viral spread from currently known sites of viral replication has been observed, and oral vaccination of dozens of animal species, including wild animals, has not revealed any residual pathogenicity [41].

# 17.8 Conclusions and Perspectives

Traditional vaccines are generally protective, and the same advances expected in human recombinant vaccines are expected in veterinary vaccines. Biologically engineered vaccines provide targeted immunity and eliminate the need to work with dangerous agents; they reduce the time required for the onset of immunity, with excellent safety characteristics. Nevertheless, some issues are particular to animal vaccines, and recombinant DNA technology may be the best alternative to tackle these. The most relevant issue to be tackled in the near future is the differentiation among vaccinated and exposed animals. Most countries use serological tests in their disease control strategies; thus, vaccination interferes in the control. For instance, BCG vaccines are mostly vectored for cattle and pigs, because the antibodies would jeopardize control through tuberculinization. Likewise, FMD vaccines in countries that are disease-free are used only in outbreak situations, because herd-wide vaccination may result in international sanctions, mainly regarding the exportation of live animals, but also concerning meat and meat products. Recombinant vaccines, capable of protecting animals with a strong immune response against one or few antigens, are a safe bet to circumvent these restrictions, because the diagnosis of a positive animal could be made while ruling out vaccine immunity.

Recombinant vaccines are undoubtedly the future of veterinary disease prevention. In this light, the drivers for recombinant vaccine development must be more potent, safer, better characterized vaccines, in addition to developing vaccines that provide broader protection against multiple antigens.

# References

- B. Adler, A. de la Peña Moctezuma, *Leptospira* and leptospirosis, Veterinary Microbiology 140 (3–4) (2010) 287–296.
- [2] R. Andreotti, Performance of two Bm86 antigen vaccine formulation against tick using crossbreed bovines in stall test, Brazilian Journal of Veterinary Parasitology 15 (3) (2006) 97–100.
- [3] K.L. Bacelo, D.D. Hartwig, F.K. Seixas, R. Schuch, A. Moreira, S. da, M. Amaral, T. Collares, C.T. Vendrusculo, A.J.A. Mcbride, O.A. Dellagostin, Xanthan gum as an adjuvant in a subunit vaccine preparation against leptospirosis, BioMed Research International 2014 (2014) 1–10.
- [4] C.C. Bassetto, A.F.T. Amarante, Vaccination of sheep and cattle against haemonchosis, Journal of Helminthology 89 (5) (2015) 5171–5525.
- [5] J.L. Blanco, M.E. Garcia, Immune response to fungal infections, Veterinary Immunology and Immunopathology 125 (1–2) (2008) 47–70.
- [6] L. Bruno, M. Cortese, R. Rappuoli, M. Merola, Lessons from reverse vaccinology for viral vaccine design, Current Opinion in Virology 11 (2015) 89–97.

- [7] G.M. Cerqueira, M. Picardeau, A century of *Leptospira* strain typing, Infection, Genetics and Evolution 9 (5) (2009) 760–768.
- [8] M.L. Coutinho, H.A. Choy, M.M. Kelley, J. Matsunaga, J.T. Babbitt, M.S. Lewis, J.A.G. Aleixo, D.A. Haake, A LigA three-domain region protects hamsters from lethal infection by *Leptospira interrogans*, PLoS Neglected Tropical Diseases 5 (12) (2011) 1–10.
- [9] C. Da Costa, B. Walker, A. Bonavia, Tuberculosis Vaccines state of the art, and novel approaches to vaccine development, International Journal of Infectious Diseases 32 (2015) 5–12.
- [10] A. Dawson, R.E. Harvey, S.J. Thevasagayam, J. Sherington, A.R. Peters, Studies of the field efficacy and safety of a single-dose *Mycoplasma hyopneumoniae* vaccine for pigs, The Veterinary record 151 (18) (2002) 535–538.
- [11] I. Delany, R. Rappuoli, E. De Gregorio, Vaccines for the 21st century, EMBO Molecular Medicine 6 (6) (2014) 708-720.
- [12] O.A. Dellagostin, A.A. Grassmann, D.D. Hartwig, S.R. Félix, É.F. da Silva, A.J.A. Mcbride, Recombinant vaccines against leptospirosis, Human Vaccines 7 (11) (2011) 1215–1224.
- [13] S.P. Dunham, The application of nucleic acid vaccines in veterinary medicine, Research in Veterinary Science 73 (1) (2002) 9–16.
- [14] J.A. Ellis, How well do vaccines for *Bordetella bronchiseptica* work in dogs? A critical review of the literature 1977–2014, The Veterinary Journal 204 (1) (2015) 5–16.
- [15] A. Eshghi, P.A. Cullen, L. Cowen, R.L. Zuerner, C.E. Cameron, Global proteome analysis of *Leptospira interrogans*, Journal of Proteome Research 8 (10) (2009) 4564–4578.
- [16] E.A. Ferraz Coelho, C.A. Pereira Tavares, F.A. Amorim Carvalho, K.F. Chaves, K.N. Teixeira, R.C. Rodrigues, H. Charest, G. Matlashewski, R.T. Gazzinelli, A.P. Fernandes, Immune responses induced by the *Leishmania (Leishmania) donovani* A2 antigen, but not by the LACK antigen, are protective against experimental *Leishmania (Leishmania) amazonensis* infection, Infection and Immunity 71 (7) (2003) 3988–3994.
- [17] R.D. Fleischmann, M.D. Adams, O. White, R.A. Clayton, E.F. Kirkness, A.R. Kerlavage, C.J. Bult, J.F. Tomb, B.A. Dougherty, J.M. Merrick, Whole-genome random sequencing and assembly of *Haemophilus influenzae* Rd, Science 269 (5223) (1995) 496–512.
- [18] V. Galli, S. Simionatto, S.B. Marchioro, A. Fisch, C.K. Gomes, F.R. Conceição, O.A. Dellagostin, Immunisation of mice with *Mycoplasma hyopneumoniae* antigens P37, P42, P46 and P95 delivered as recombinant subunit or DNA vaccines, Vaccine 31 (1) (2012) 135–140.
- [19] V. Galli, S. Simionatto, S.B. Marchioro, G.H.F. Klabunde, F.R. Conceição, O.A. Dellagostin, Recombinant secreted antigens from *Mycoplasma hyopneumoniae* delivered as a cocktail vaccine enhance the immune response of mice, Clinical and Vaccine Immunology 20 (9) (2013) 1370–1376.
- [20] S. Ghosh, G. Nagar, Problem of ticks and tick-borne diseases in India with special emphasis on progress in tick control research: a review, Journal of Vector Borne Diseases 51 (4) (2014) 259–270.
- [21] D.D. Hartwig, T.L. Oliveira, F.K. Seixas, K.M. Forster, C. Rizzi, C.P. Hartleben, A.J.A. Mcbride, O.A. Dellagostin, High yield expression of leptospirosis vaccine candidates LigA and LipL32 in the methylotrophic yeast *Pichia pastoris*, Microbial Cell Factories 9 (1) (2010) 98.
- [22] D.D. Hartwig, K.M. Forster, T.L. Oliveira, M. Amaral, A.J.A. mcbride, O.A. Dellagostina, A primeboost strategy using the novel vaccine candidate, LemA, protects hamsters against leptospirosis, Clinical and Vaccine Immunology 20 (5) (2013) 747–752.
- [23] W.R. Hein, G.B.L. Harrison, Vaccines against veterinary helminths, Veterinary Parasitology 132 (3–4) (2005) 217–222.
- [24] A.I. Heinson, C.H. Woelk, M. Newell, The promise of reverse vaccinology, International Health 7 (2) (2015) 85–89.

- [25] K. Hoebe, E. Janssen, B. Beutler, The interface between innate and adaptive immunity, Nature Immunology 5 (10) (2004) 971–974.
- [26] S. Jorge, N.R. de Oliveira, S.B. Marchioro, A. Fisch, C.K. Gomes, C.P. Hartleben, F.R. Conceição, O.A. Dellagostin, The *Mycoplasma hyopneumoniae* recombinant heat shock protein P42 induces an immune response in pigs under field conditions, Comparative Immunology, Microbiology and Infectious Diseases 37 (4) (2014) 229–236.
- [27] E. Larrieu, E. Herrero, G. Mujica, J.L. Labanchi, D. Araya, C. Grizmado, A. Calabro, G. Talmon, G. Ruesta, A. Perez, A. Gatti, G. Santillán, M. Cabrera, M. Arezzo, M. Seleiman, L. Cavagión, M.G. Cachau, C.A. Alvarez Rojas, L. Gino, et al., Pilot field trial of the EG95 vaccine against ovine cystic echinococcosis in Rio Negro, Argentina: early impact and preliminary data, Acta Tropica 127 (2) (2013) 143–151.
- [28] P.N. Levett, Leptospirosis, Clinical Microbiology 14 (2) (2001) 296-326.
- [29] M.W. Lightowlers, Control of *Taenia solium* taeniasis/cysticercosis: past practices and new possibilities, Parasitology 140 (13) (2013) 1566–1577.
- [30] S.B. Marchioro, D. Maes, B. Flahou, F. Pasmans, R. Del Pozo Sacristán, K. Vranckx, V. Melkebeek, E. Cox, N. Wuyts, F. Haesebrouck, Local and systemic immune responses in pigs intramuscularly injected with an inactivated *Mycoplasma hyopneumoniae* vaccine, Vaccine 31 (9) (2013) 1305–1311.
- [31] S.B. Marchioro, R.D.P. Sácristan, A. Michiels, F. Haesebrouck, F.R. Conceição, O.A. Dellagostin, D. Maes, Immune responses of a chimaeric protein vaccine containing *Mycoplasma hyopneumoniae* antigens and LTB against experimental *M. hyopneumoniae* infection in pigs, Vaccine 32 (36) (2014) 4689–4694.
- [32] J. Mašek, E. Bartheldyová, P. Turánek-Knotigová, M. Škrabalová, Z. Korvasová, J. Plocková, Š. Koudelka, P. Škodová, P. Kulich, M. Křupka, K. Zachová, L. Czerneková, M. Horynová, I. Kratochvílová, A.D. Miller, D. Zýka, J. Michálek, J. Vrbková, M. Šebela, et al., Metallochelating liposomes with associated lipophilised norAbuMDP as biocompatible platform for construction of vaccines with recombinant His-tagged antigens: preparation, structural study and immune response towards rHsp90, Journal of Controlled Release 151 (2) (2011) 193–201.
- [33] M.M. Mcallister, Successful vaccines for naturally occurring protozoal diseases of animals should guide human vaccine research. A review of protozoal vaccines and their designs, Parasitology 141 (5) (2014) 624–640.
- [34] S. Mcvey, J. Shi, Vaccines in veterinary medicine: a brief review of history and technology, Veterinary Clinics of North America Small Animal Practice 40 (3) (2010) 381–392.
- [35] V. Molina-Hernández, G. Mulcahy, J. Pérez, Á. Martínez-Moreno, S. Donnelly, S.M. O'neill, J.P. Dalton, K. Cwiklinski, *Fasciola hepatica* vaccine: we may not be there yet but we're on the right road, Veterinary Parasitology 208 (1–2) (2015) 101–111.
- [36] A.J. Nisbet, J.F. Huntley, Progress and opportunities in the development of vaccines against mites, fleas and myiasis-causing flies of veterinary importance, Parasite Immunology 28 (4) (2006) 165–172.
- [37] A.J. Nisbet, A. Mackellar, H.W. Wright, G.P. Brennan, K.Y. Chua, N. Cheong, J.E. Thomas, J.F. Huntley, Molecular characterization, expression and localization of tropomyosin and paramyosin immunodominant allergens from sheep scab mites (*Psoroptes ovis*), Parasitology 133 (4) (2006) 515–523.
- [38] A.-F. Petavy, C. Hormaeche, S. Lahmar, H. Ouhelli, A. Chabalgoity, T. Marchal, S. Azzouz, F. Schreiber, G. Alvite, M.-E. Sarciron, D. Maskell, A. Esteves, G. Bosquet, An oral recombinant vaccine in dogs against *Echinococcus granulosus*, the causative agent of human hydatid disease: a pilot study, PLoS Neglected Tropical Diseases 2 (1) (2008) E125.

- [39] M. Pizza, V. Scarlato, V. Masignani, M.M. Giuliani, B. Aricò, M. Comanducci, G.T. Jennings, L. Baldi, E. Bartolini, B. Capecchi, C.L. Galeotti, E. Luzzi, R. Manetti, E. Marchetti, M. Mora, S. Nuti, G. Ratti, L. Santini, S. Savino, et al., Identification of vaccine candidates against serogroup B meningococcus by whole-genome sequencing, Science 287 (5459) (2000) 1816–1820.
- [40] R. Rappuoli, Reverse vaccinology, a genome-based approach to vaccine development, Vaccine 19 (17–19) (2001) 2688–2691.
- [41] L. Redding, D.B. Weiner, DNA vaccines in veterinary use, Expert Review of Vaccines 8 (9) (2009) 1251–1276.
- [42] C. Rizzi, M.V. Bianco, F.C. Blanco, M. Soria, M.J. Gravisaco, V. Montenegro, L. Vagnoni, B. Buddle, S. Garbaccio, F. Delgado, K.S. Leal, A.A. Cataldi, O.A. Dellagostin, F. Bigi, Vaccination with a BCG strain overexpressing Ag85B protects cattle against *Mycobacterium bovis* challenge, PloS One 7 (12) (2012) E51396.
- [43] B. Roberts, A. Antonopoulos, S.M. Haslam, A.J. Dicker, T.N. Mcneilly, S.L. Johnston, A. Dell, D.P. Knox, C. Britton, Novel expression of *Haemonchus contortus* vaccine candidate aminopeptidase H11 using the free-living nematode *Caenorhabditis elegans*, Veterinary Research 44 (1) (2013) 1–15.
- [44] F.K. Seixas, É.F. da Silva, D.D. Hartwig, G.M. Cerqueira, M. Amaral, M.Q. Fagundes, R.G. Dossa, O. A. Dellagostin, Recombinant *Mycobacterium bovis* BCG expressing the LipL32 antigen of *Leptospira interrogans* protects hamsters from challenge, Vaccine 26 (1) (2007) 88–95.
- [45] M. Sibila, M. Pieters, T. Molitor, D. Maes, Current perspectives on the diagnosis and epidemiology of *Mycoplasma hyopneumoniae* infection, The Veterinary Journal 181 (3) (2009) 221–231.
- [46] É.F. Silva, C.S. Brod, G.M. Cerqueira, D. Bourscheidt, N. Seyffert, A. Queiroz, C.S. Santos, A.I. Ko, O. A. Dellagostin, Isolation of *Leptospira noguchii* from sheep, Veterinary Microbiology 121 (1–2) (2007) 144–149.
- [47] É.F. Silva, M.A. Medeiros, A.J.A. Mcbride, J. Matsunaga, G.S. Esteves, J.G.R. Ramos, C.S. Santos, J. Croda, A. Homma, O.A. Dellagostin, D.A. Haake, M.G. Reis, A.I. Ko, The terminal portion of leptospiral immunoglobulin-like protein LigA confers protective immunity against lethal infection in the hamster model of leptospirosis, Vaccine 25 (33) (2007) 6277–6286.
- [48] G.M. Silver, P.J. Gaines, S.W. Hunter, J.D. Maddux, R.E. Thomas, N. Wisnewski, Identification, characterization, and cloning of an immunoglobulin degrading enzyme in the cat flea, *Ctenocephalides felis*, Archives of Insect Biochemistry and Physiology 51 (3) (2002) 136–150.
- [49] S. Simionatto, S.B. Marchioro, V. Galli, D.D. Hartwig, R.M. Carlessi, F.M. Munari, J.P. Laurino, F.R. Conceicao, O.A. Dellagostin, Cloning and purification of recombinant proteins of *Mycoplasma hyopneumoniae* expressed in *Escherichia coli*, Protein Expression and Purification 69 (2) (2010) 132–136.
- [50] S. Simionatto, S. Beutinger, D. Maes, *Mycoplasma hyopneumoniae*: from disease to vaccine development, Veterinary Microbiology 165 (3–4) (2013) 234–242.
- [51] M. Singh, D.T. O'Hagan, Recent advances in veterinary vaccine adjuvants, International Journal for Parasitology 33 (5–6) (2003) 469–478.
- [52] K. Takeda, S. Akira, Roles of Toll-like receptors in innate immune responses, Genes to Cells: Devoted to Molecular & Cellular Mechanisms 6 (9) (2001) 733–742.
- [53] H. Tettelin, N.J. Saunders, J. Heidelberg, A.C. Jeffries, K.E. Nelson, J.A. Eisen, K.A. Ketchum, D.W. Hood, J.F. Peden, R.J. Dodson, W.C. Nelson, M.L. Gwinn, R. Deboy, J.D. Peterson, E.K. Hickey, D.H. Haft, S.L. Salzberg, O. White, R.D. Fleischmann, et al., Complete genome sequence of *Neisseria meningitidis* serogroup B strain MC58, Science 287 (5459) (2000) 1809–1815.
- [54] R.A. Tripp, S.M. Tompkins, Virus-vectored influenza virus vaccines, Viruses (2014) 3055-3079.
- [55] M. Unnikrishnan, R. Rappuoli, D. Serruto, Recombinant bacterial vaccines, Current Opinion in Immunology 24 (3) (2012) 337–342.

- [56] A.T.R. Vasconcelos, H.B. Ferreira, C.V. Bizarro, S.L. Bonatto, M.O. Carvalho, P.M. Pinto, D.F. Almeida, L.G.P. Almeida, R. Almeida, L. Alves-filho, E.N. Assunc, R. Bogo, M.M. Brigido, M. Brocchi, H.A. Burity, A.A. Camargo, S.S. Camargo, M.S. Carepo, D.M. Carraro, et al., Swine and poultry pathogens: the complete genome sequences of two strains of *Mycoplasma hyopneumoniae* and a strain of *Mycoplasma synoviae*, Journal of Bacteriology 187 (16) (2005) 5568–5577.
- [57] H. Weighardt, G. Jusek, J. Mages, R. Lang, K. Hoebe, B. Beutler, B. Holzmann, Identification of a TLR4- and TRIF-dependent activation program of dendritic cells, European Journal of Immunology 34 (2) (2004) 558–564.
- [58] P. Willadsen, Anti-tick vaccines, Parasitology 129 (Suppl.) (2004) S367-S387.

# 18

# Development of Process to Produce Recombinant Component for Acellular Pertussis Vaccine

T. Scheuer<sup>1,2</sup>, V. Thomaz-Soccol<sup>1,\*</sup>, C. Perottoni<sup>1,2</sup>, O.A. Dellagostin<sup>3</sup>, L.P.S. Vandenberghe<sup>1</sup>, C.R. Soccol<sup>1</sup>

<sup>1</sup>UNIVERSIDADE FEDERAL DO PARANÁ (UFPR), CURITIBA, PARANÁ, BRAZIL; <sup>2</sup>INSTITUTO DE TECNOLOGIA DO PARANÁ (TECPAR), CURITIBA, PARANÁ, BRAZIL; <sup>3</sup>UNIVERSIDADE FEDERAL DE PELOTAS (UFPEL), PELOTAS, RIO GRANDE DO SUL, BRAZIL

# 18.1 Introduction

#### 18.1.1 Pertussis Disease

Pertussis, also known as whooping cough, is a disease of the respiratory tract that affects humans of all ages and has worldwide distribution. The disease cause great morbidity and mortality in young children, especially in babies less than age 1 year. This disease occurs as the result of infection caused by *Bordetella pertussis*, which releases a number of virulence factors, including pertussis toxin (PT), filamentous hemagglutinin (FHA), pertactin, and fimbriae [1,2].

Pertussis is primarily a toxin-mediated disease. The bacteria attaches to the cilia of the respiratory epithelial cells, producing toxins that paralyze the cilia and causing inflammation of the respiratory tract, which interferes with clearing of pulmonary secretions [3].

Main symptoms of this disease are violent coughing, which often provokes difficulties in breathing [2]. Severe and potentially fatal complications include convulsions, bronchopneumonia, and encephalopathy [4].

Typical pertussis, as defined by the World Health Organization [5], consists of three phases: catarrhal, paroxysmal and convalescent. The catarrhal phase, during which the disease is most contagious, manifests as a cold-like illness with coryza and a mild cough. The onset of these symptoms is usually 7–10 days after exposure. Proliferation of *Bo. pertussis* and its subsequent abundance in nasopharyngeal secretions provide a means of aerosol transmission via droplet spread or direct contact [6].

<sup>&</sup>lt;sup>\*</sup>Corresponding Author.

Progression from a mild asymptomatic cough to convulsive or violent attacks (paroxysmal cough) indicates onset of the paroxysmal phase. Excessive mucous production, vomiting, and lymphocytosis are additional manifestations that are commonly observed during this period of infection. A feature of the paroxysmal cough is the characteristic "whoop" sound that occurs as a result of inhalation through a narrow glottis. Coughing attacks can occur frequently and may be so severe that exhaustion and apnea can occur. This stage can last for up to 4 weeks, after which most who experience it enter a convalescent phase characterized by a gradual recovery. However, the severe cough can reemerge in the presence of an opportunistic infection [4].

#### 18.1.2 Historical Aspects

Whooping cough was recognized as a human acute respiratory disease in the 16th century. The etiologic agent, *Bo. pertussis*, was isolated and identified by Bordet and Gengou in 1906 [7].

In the 16th century, pertussis was one of the most common childhood diseases and a major cause of childhood mortality in the United States. Before the availability of pertussis vaccine in the 1940s, more than 200,000 cases of pertussis were reported annually. During the 6 years from 1940 through 1945, more than one million cases of pertussis were reported [8]. Whole-cell pertussis vaccines were first licensed in the United States in 1914 and became available combined with diphtheria and tetanus toxoids (as DTP) in 1948 [8].

After the introduction of whole-cell pertussis vaccine in the 1940s, the incidence of pertussis gradually declined, reaching 15,000 reported cases in 1960 (approximately 8 per 100,000 people). By 1970, the annual incidence was fewer than 5000 cases per year, and during 1980–1990 an average of 2900 cases per year was reported (approximately 1 per 100,000 people). Since widespread use of the vaccine began, the incidence has decreased more than 80% compared with the prevaccine era [8].

#### 18.1.3 The Pathogen and the Toxin

*Bo. pertussis* is a small gram-negative aerobic coccobacillus that appears singly or in pairs [9]. It is fastidious and requires special media for isolation [8]. This bacterium produces an array of virulence factors, some of which are considered to have a key role in the establishment of *Bo. pertussis* infection. These include FHA, pertactin (P69), PT, and adenylate cyclase toxin. Other factors such as fimbriae, dermonecrotoxin, tracheal cytotoxin, tracheal colonization factor, serum resistance factor, and lipopolysaccharide have important roles leading to clinical manifestations of pertussis [10,11].

Regarded as the central virulence factor during infection, PT elicits a number of deleterious consequences including leukocytosis, splenomegaly, histamine sensitization, hypoglycemia, and hypoproteinemia [12].

PT is one of the most complex soluble bacterial proteins [8] (Fig. 18.1). It is a multimeric 105-kDa protein composed of five different subunits: S1, S2, S3, S4, and S5 [13]. The S1 subunit catalyzes the adenosine diphosphate (ADP) ribosylation of G proteins in target



FIGURE 18.1 Schematic of pertussis toxin. The holotoxin is viewed perpendicular to the fivefold axis of the B-oligomer. Reprinted from P.E. Stein, A.B. Boodhoo, G.D. Armstrong, S.A. Cockle, M.H. Klein, R.J. Read, The crystal structure of pertussis toxin, Structure, 2 45–57, copyright 1994, with permission from Elsevier.

cells of mammals, whereas subunits S2 to S5 (oligomer B) are responsible for the binding of S1 to these cells. ADP ribosyl transferase enzymatic activity is associated with the domain located in the 180 N-terminal amino acids of S1 (termed C180), whereas the rest of this subunit is involved in G protein binding and interaction with the oligomer B [14].

Besides enzymatic activity, mainly owing to Arg9, Arg13, and Glu129 residues [15], the subunit S1 is also recognized as the immunogenic portion of PT once it promotes the maturation of dendritic cells. These cells are in charge of activation of antigen-specific T-and B-cells, which mediate adaptive immunity against *Bo. pertussis* [16].

This subunit expressed in *Mycobacterium bovis* Bacillus Calmette—Guérin was able to induce protective immunity against the toxin itself and against *Bo. pertussis* in a model of intracerebral challenge in mice [17]. In addition, a DNA vaccine comprising a plasmid encoding this subunit was able to induce immunity [18].

## 18.1.4 Epidemiology

The disease is endemic and has epidemic cycles of 3–5 years. It ranks fifth among vaccine-preventable diseases in children under age 5 years. Since 2010, a global trend of increasing pertussis cases has been observed, with reports in California and elsewhere on the American continent [19].

There were 16 million cases of pertussis worldwide in 2008; 195,000 children died of the disease, most of them in developing countries. However, pertussis is not confined in the developing world. There has been a resurgence of the disease in many developed countries, including the United States, Australia, England, Wales, and Ireland. In Australia, the incidence of pertussis has increased from 349 reported cases in 1991 to 38,722 in 2011 [20].

Even with high vaccine coverage (vaccination was responsible for avoiding 37 million cases of illness and 587,000 deaths in 2002), the fall of immunity 5–10 years after the last vaccine dose left adolescents and young adults susceptible to the disease, mainly owing to decreased protective antibody titer [21,22]. This reduction in immunity among adults may be related to the reduced circulation of *Bo. pertussis* in the vaccinated populations as well, which causes less frequent exposure to antigens, and thus less strengthening of natural immunity [23]. This can be considered a serious problem, because an increased risk of complications exists with increasing age, including hospitalizations for pneumonia and sinusitis [24]. Moreover, adolescents and adults may serve as reservoirs for the pathogen and could be an infection source for children, beginning the cycle of transmission to newborns and infants not yet immunized, and the disease remains endemic [25].

#### 18.1.5 Pertussis Vaccines

Because of the emergence of new production technologies and regulatory requirements, vaccine production has had constant and deep transformations. These new products have high value and are causing technological dependence in developing countries [26] that has led to the requirement for short-term replacement of some vaccines, including DTP, used in public health systems.

The DTP vaccine is composed of tetanus toxoid, diphtheria toxoid, and a detoxified bacterial suspension of *Bo. pertussis* [27]. This vaccine has been effective in immunization against pertussis. However, several adverse effects have been reported, such as local effects (redness and swelling), systemic effects (fever, vomiting, and diarrhea), and even seizures [28]. Furthermore, these effects are exacerbated with age, and therefore the last dose of vaccine is given to people age 5 or 6 years.

Many groups have started to study the possibility of producing a acellular pertussis vaccine (DTaP) vaccine that contains only purified antigens of *Bo. pertussis*. A literature review [29] showed that these acellular vaccines have been well tolerated and highly immunogenic in adolescents and adults. These vaccines, which have an antigenic content more than 50% lower than formulations for children and reduced adverse effects, demonstrated an estimated efficacy of 92% in a study among American teenagers [25].

Initially, the approach to production of this vaccine was a bacterial culture and subsequent purification of some proteins recognized as being immunogenic. The major disadvantage of this procedure is that purification of components causes a low yield, because only that fraction of the cell is exploited for vaccine production. Only one dose of acellular vaccine can be produced from the same fermentation broth volume that produces 10–20 doses of whole-cell vaccine [30]. In addition, the vaccine generally has low immunogenicity owing to the chemical detoxification process by which it must pass [15].

Actually, several vaccines available on the market allow the vaccination of adults and young babies (Table 18.1). However these vaccines are produced by multinationals and are too expensive to be acquired by public health programs.

Component	KINRIX (GSK)	Pentacel (Sanofi Pasteur)	TETRAXIM (Sanofi Pasteur)	Adacel (Sanofi Pasteur)
Diphtheria toxoid	25 Lf*	15 Lf*	30 I. U.**	2 Lf*
Tetanus toxoid	10 Lf*	5 Lf*	20 I. U.**	5 Lf*
Pertussis toxoid	25 µg	10 µg	25 µg	2.5 μg
Filamentous Hemagglutinin	25 µg	5 μg	25 µg	5 μg
Pertactin (69 kDa)	8 µg	3 µg	-	3 µg
Fimbriae	_	5 µg	_	5 µg
Age indication	4—6 years	6 weeks-4 years	2—13 years	4—64 years

 Table 18.1
 Composition of Main Commercial Acellular Pertussis Vaccines

\*Limit of flocculation;

\*\*International Units.

## 18.1.6 Recombinant Vaccines

The production of DTaP by recombinant technology can be accomplished in two main ways. The first, already analyzed in the review of Loosmore and colleagues [31], consists of the mutation of *Bo. pertussis* to change certain amino acids in the S1 subunit of the toxin, which is responsible for its toxicity. The second most studied technology is currently the overexpression of components in microbial cells.

Burnette and colleagues [32] performed cloning of all subunits of PT in *Escherichia coli* to identify key functional sites and map epitopes to obtain candidates for an acellular vaccine. Furthermore, cloning attempts aimed at producing not only PT but also other possible components of the DTaP vaccine were performed in microorganisms such as *Bacillus subtilis* [33], *Pichia pastoris, Saccharomyces cerevisiae* [34], and *Bacillus brevis* [35], besides insect cells [36].

The advantages of using these organisms are the low cost of the process, mainly because of the high yield of protein [35], and the greater ease of purification. Furthermore, recombinant production has the advantage of eliminating the step of detoxifying the protein [37].

Table 18.2 shows that since 1987, many studies in developing pertussis vaccines that were performed were related to recombinant technology. These studies involved many different host cells, including *Bo. pertussis* itself, besides some bacteria such as *E. coli* and *Ba. subtilis* and some yeasts such as *P. pastoris* and *S. cerevisiae*. This research used subunit S1 from PT and other molecules as potential vaccines, with good results in producing antibody response.

# 18.2 Technological Development: Case Description

To develop recombinant vaccine, three steps are necessary. The first involves all of the procedures to get the recombinant plasmid. The second step has the objective of

Cell	Objective	Author	Year
<i>Escherichia coli</i> K12 ΔH1 Δtrp	Produce each of the five subunits of pertussis toxin as fusion proteins and verify their immunologic potential against	Nicosia et al. [38]	1987
Bacillus subtilis	intracerebral challenge with <i>Bo. pertussis</i> in mice Expression and secretion of pertussis toxin subunit S1 with an	Runeberg-Nyman	1987
Escherichia coli	Expression of each of the five subunits of pertussis toxin at nonfusion forms to use as vaccine components lacking toxic effects of traditional pertussis vaccines	Bumette et al. [40]	1988
E. coli	Expression of S1 subunit with site-specific mutagenesis in critical region. A mutant containing a single amino acid substitution (Arg9-*Lys) with reduced enzymatic activity (approximately 0.02% of control) while retaining protective epitope	Bumette et al. [32]	1988
E. coli	Expression of nonfusion form of S-1 subunit (rS-1) and catalytic peptide of S1 subunit that is composed of the first 180 amino-terminal residues (C180 peptide)	Barbieri et al. [41]	1989
Bordetella pertussis Bordetella parapertussis and Bordetella bronchisontica	Construction of a number of genetically engineered alleles of pertussis toxin genes, replacing either one or two key amino acids within enzymatically active S1 subunit, and introduction into the chromosome of strains of <i>Bo. pertussis, Bo. parapertussis</i> , and <i>Bo. Bronchiseptica</i> , aiming to produce partures toxin molecules that are postavis and immunoceptic	Pizza et al. [15]	1989
Ba. subtilis	Expression and secretion of S1 subunit and test of its immunogenic potential	Õlander et al. [33]	1991
Saccharomyces cerevisiae S150- 2B e Pichia pastoris GS115	Expression of membrane protein pertactin (P69) from <i>Bo. pertussis</i> to high levels in multicopy transformants	Romanos et al. [34]	1991
Bacillus brevis	Expression and secretion of S2 subunit of pertussis toxin	Kozuka et al. [35]	1996
Streptococcus gordonii	Purification and immunogenicity of recombinant <i>Bo. pertussis</i> S1S3FHA fusion protein expressed by <i>S. gordonii</i> to produce mucosal vaccine	Lee et al. [1]	2002
	Production of DNA vaccine with plasmid DNA expressing pertussis toxin S1 subunit (pcDNA/S1) of <i>Bo. pertussis</i> and verification of its immunogenicity, for its ability to induce protection against PT challenge or <i>Bo. pertussis</i> infection in mice	Kamachi et al. [18]	2003
	Development of pertussis DNA vaccine expressing nontoxic C180 polypeptide of pertussis toxin S1 subunit	Kamachi et al. [42]	2007
Bo. pertussis	Construction of <i>Bo. pertussis</i> strains with enhanced production of genetically inactivated pertussis toxin and pertactin	Buasri et al. [43]	2012
<i>Bo. pertussis</i> BpCNIC0311	Synthesis of genetically detoxified toxoid (PTg). Filamentous hemagglutinin and type 2 pertactin aiming for suitable yields after purification and high immunogenic response	Quintana- Vázquez et al. [44]	2015

 Table 18.2
 Research Development of New Generation of Pertussis Vaccines



FIGURE 18.2 Example of flowchart of steps to obtain the recombinant plasmid (A), to obtain the recombinant cells (B), and to obtain recombinant S1 subunit of pertussis toxin (C).

cultivating recombinant cells, and the third is finally the production and purification of the recombinant protein (Fig. 18.2A–C).

An example is presented of a heterologous vaccine production using DTaP obtained from a recombinant antigen. It is produced by the cloning and expression of an immunogenic protein of *Bo. pertussis*. Fig. 18.2 presents the steps of the process for heterologous vaccine production.

# 18.2.1 Cultivation and Inactivation of Bordetella pertussis

Before the cloning step, the pathogenic microorganism has to be cultivated. For this, the freeze-dried content, of an ampoule of *Bo. pertussis* strain 137 (BP/T/01/99), is homogenized with sterile saline solution (NaCl 0.9%), inoculated in an Erlenmeyer flask containing 50 mL of Stainer–Scholte medium [45]. Then it is incubated in a shaker flask at 37°C at 150 rpm for 48 h. Gram staining is performed to confirm the identity and purity of the culture. Furthermore, the strain is transferred to plates containing solid medium charcoal agar base with niacin and incubated at 37°C for 5 days, and stored at 4°C.

Liquid cultures are distributed in 50-mL centrifuge tubes, to be inactivated in a water bath at 60°C for 30 min, and then centrifuged at  $5000 \times g$  for 20 min at 4°C for cell recovery; the pellets are stored at 4°C [46]. All of these procedures must be performed in a biosecurity area.

## 18.2.2 Extraction and Quantification of DNA of Bordetella pertussis

The pellet of one of the centrifuge tubes can be used for DNA extraction, which can be developed combining freeze-thaw and phenol-chloroform-isoamyl alcohol

techniques. To see whether the technique was successful and to quantify DNA, absorbance is read at 260 nm. Then an aliquot is applied to a 0.8% agarose gel and run at 80 V for 1 h. Next, it is colored with ethidium bromide to verify the quality of DNA.

In this example, the average concentration of the DNA samples determined by spectrophotometry at 260 nm is  $3000 \,\mu$ g/mL. The concentration of DNA obtained is determined by the equation:

 $[DNA](\mu g/mL) = Abs_{260} \times 50 \times FD$ 

where [DNA] is the DNA concentration (in  $\mu$ g/mL); Abs<sub>260</sub> is the absorbance measured at 260 nm, and FD is the dilution factor of the sample.

#### 18.2.3 Primers Design and Vector Choice

Some preliminary analyses are needed for proper amplification of the desired region of DNA. These analyses can be performed using the software Vector NTI (Invitrogen). This software is also useful for construction of the resulting plasmid in silico. The ptxA gene sequence encoding the S1 subunit of PT was obtained from GenBank (GeneID: 2,665,068, locus tag BP3783). Aiming the directional cloning of amplicons into the expression plasmid, restriction sites for the enzymes *Bam*HI and *Hind*III were introduced in forward and reverse primers, respectively. The region selected for amplification contains 684 bp and excludes the first 44 amino acids from the protein, eliminating the potential pathogenicity of the molecule. Including the nucleotides inserted by the primers, the fragment size expected after the polymerase chain reaction (PCR) is 698 base pairs (bp).

The vector used in that case report is the noncommercial vector pAE. The vector has several advantageous features, such as a replication origin in *E. coli*, multiple cloning sites, ampicillin resistance, being a strong promoter of phage T7, the possibility of isopropyl  $\beta$ -D-1-thiogalactopyranoside (IPTG) induction, and the inclusion of a polyhistidine tail to the cloned protein.

# 18.2.4 Amplification of Gene of Interest by Polymerase Chain Reaction

PCR is carried out using commercial DNA polymerase following the manufacturer's protocol, using  $1 \ \mu$ L of  $10 \ ng/\mu$ L DNA template for each  $50 \ \mu$ L of reaction mix. The conditions of PCR in this example are 95°C for 5 min (initial denaturation), 30 cycles of 95°C for 30 s (denaturation), 55°C for 30 s (annealing), 72°C for 1 min (extension), and 72°C for 5 min (final extension). The efficacy of amplification is determined by running samples on a 1.0% agarose gel at 80 V for 1 h.

The result of the gel is analyzed; if it shows only one band with the fragment of the desired size, it confirms the correct amplification of the gene. Furthermore, in negative control (a sample that passes through PCR procedure without primers) must be observed no bands (Fig. 18.3).



FIGURE 18.3 Agarose gel (0.8%) of polymerase chain reaction products. 1 Kb, molecular weight marker; NC, negative control of PCR; P, amplified sample.

## 18.2.5 Preparation of Vector and Gene for Ligation

Both the gene obtained by PCR cloning (after purification) and the vector, in this case pAE, are subsequently digested with restriction enzymes suitable for this gene, according to the used primers. In this case, *Bam*HI and *Hind*III are used, and the digestion is carried out according to the manufacturer's protocol. Their concentrations are determined by spectrophotometry at 260 nm and the efficacy of digestion is determined in a 1.0% agarose gel.

The efficacy of digestion is proved if various bands can be observed at the circular vector sample and only one band with the expected size (2822 bp) can be observed on the digested vector.

Normally, on an uncut plasmid sample, four different bands can be seen, corresponding to variations in DNA folding, once not only the size but also the shape affects migration in electrophoresis. When plasmid is cut, it migrates as linear DNA, so the presence of only one band is expected [47].

On the other samples any difference is seen once digestion eliminates only a few nucleotides, and then there is no significant difference in molecule size.

After digestion, the ligation reaction with T4 DNA ligase is executed overnight at 4°C following the manufacturer's protocol.

## 18.2.6 Transformation of Escherichia coli DH5α for Plasmid Replication

The plasmids obtained are used to transform *E. coli* DH5 $\alpha$  by heat shock according to the protocol for replication and verification of the success of ligation. In addition, a negative control is performed only with the vector.

The protocol for competent cells production is described. Ten isolated colonies of *E. coli* are grown in 20-mL Luria–Bertani (LB) medium at 37°C until  $A_{600} = 0.5$  (about 3–4 h), and then centrifuged at 3000 × g at 4°C for 10 min in ice-cold sterile centrifugation tubes. The cell sediment is suspended in 20 mL ice-cold 0.1 M MgCl<sub>2</sub> solution and then centrifuged at 3000 × g at 4°C for 10 min. After that, the cell sediment is suspended

in 20 mL ice-cold 0.1 M CaCl<sub>2</sub> solution and incubated for 30 min at 4°C. After the incubation, cells are centrifuged at  $3000 \times g$  at 4°C for 10 min; the cell sediment is resuspended in 1 mL ice-cold 0.1 M CaCl<sub>2</sub> solution; 100 µL glycerol (sterilized by autoclaving) is added and 100 µL of the cell suspension is frozen at  $-80^{\circ}$ C.

In the protocol of competent cell transformation, the suspension with the competent *E. coli* cells on ice (about 15 min) is thawed; 20  $\mu$ L competent of *E. coli* cell solution is added to 5 ng (2  $\mu$ L) plasmid DNA. The suspension is incubated for 30 min on ice and incubated for 90 s at 42°C and then for 2 min on ice. Then 200  $\mu$ L LB medium is added and incubated at 37°C for 2 h while agitating.

The transformations are spread on Petri dishes containing LB medium supplemented with 100 mg/L ampicillin (once the vector has the gene of resistance to this antibiotic) and incubated overnight at 37°C.

In this case, the growth of six colonies are observed on the plate, which is seeded with a culture transformed with plasmid pAE-S1. These colonies are subjected to screening with phenol—chloroform to confirm the presence of the plasmid. After screening by electrophoresis (Fig. 18.4), four colonies (P1, P2, P3, and P4) have the same running pattern as the negative control samples. P5 has a plasmid without an insert compared with the pAE sample. P6 shows a different pattern, although the plasmid seems to be smaller than the pAE. Even so, this running difference can occur as a result of variations in plasmid folding, as described in Section 18.2.5. Thus P6 was previously chosen for plasmid extraction. Nevertheless plasmids from P4 and P5 were also extracted for comparison after double digestion, and thus confirm the presence of an insert.

The selected colonies are inoculated in 5 mL LB medium with ampicillin, incubated overnight on a rotary shaker at 180 rpm and 37°C, and used for the extraction of plasmids.

To confirm the presence of inserts in the extracted plasmids, a double digestion is performed with the same enzymes and verified on a 1.0% agarose gel.

Double digestion of the extracted confirms the expected results only if the digestion of the plasmid has a 698-bp band. In this case, it occurred only with the P6 colony. Thus



**FIGURE 18.4** Screening of colonies after transformation. *1 Kb plus*, molecular weight marker; *E. coli*, DNA extracted from one *E. coli* DH5 $\alpha$  colony; *NC 1, NC 2, NC 3*, DNA extracted from three colonies of negative control; *pAE*, uncut vector; *P1–6*, DNA extracted from the six colonies that had grown in plates with selective LB medium.



**FIGURE 18.5** Verification of recombinant protein expression. *NC ni*, negative control before induction; *NC I*, negative control after induction; *P ni*, transformed cells before induction; *P i*, transformed cells after induction; *MW*, molecular weight marker.

this plasmid was chosen to transform *E coli* Bl21 pLysS cells and produce the recombinant protein.

# 18.2.7 Transformation of *Escherichia coli* BL21 pLysS and Induction of Recombinant Protein Expression

The selected plasmid is used to transforming *E. coli* Bl21 pLysS expression cells. Cells transformed by heat shock (as described previously) are inoculated in LB medium containing ampicillin and 25 mg/L chloramphenicol (to maintain the plasmid and pLysS) and incubated on rotary shaker at 180 rpm overnight at 37°C. Cells without transformation also incubated in nonselective LB broth are used as a negative control.

One milliliter of this preinoculum is inoculated in 10 mL LB medium with the same antibiotics and incubated on a rotary shaker at 37°C and 180 rpm until the culture reaches an absorbance [optical density (OD)] of 0.6 at 600 nm. Then, induction of expression is performed with 0.3 mM IPTG for 3 h. Confirmation of the recombinant protein expression is performed by electrophoresis sodium dodecyl sulfate–polyacrylamide gel electrophoresis (SDS-PAGE) on a 10% gel.

In SDS-PAGE a band can be observed with approximately 28 kDa overexpressed in the inducted sample (Fig. 18.5).

This confirms the expression of the recombinant protein and the efficiency of IPTG in the induction of protein expression, although a band can be also seen with less intensity corresponding to the recombinant protein in the noninducted ample. This shows that sometimes the system of expression control, regulated by the pLysS plasmid, is not efficiently completed.

# 18.2.8 Solubility Test to Verify the Presence of the Protein in the Supernatant or in Inclusion Bodies

One milliliter of the preinoculum is used to inoculate 9 mL of LB broth with antibiotics and is incubated overnight at 37°C and 150 rpm. Then the culture is used to inoculate

100 mL of the same medium, which is kept in a shaker under the same conditions until an OD of approximately 0.8 is reached. IPTG is added to a final concentration of 0.3 mM and induction occurs for 3 h. The final culture is centrifuged at 7000 × g and 4°C for 15 min; the pellet obtained is washed with PBS (137 mM sodium chloride, 2.7 mM potassium chloride, 10 mM disodium hydrogen phosphate, 1.8 mM potassium dihydrogen phosphate, pH 7.4), centrifuged again under the same conditions, and resuspended in 10 mL resuspension buffer (20 mM NaH<sub>2</sub>PO<sub>4</sub>, pH 8.0). Four pulses of sonication of 20 s each at the same time interval are performed. Then centrifugation is performed at 15,000 × g and 4°C for 10 min to precipitate inclusion bodies. The supernatant is stored at 4°C and is called the soluble fraction. The pellet is resuspended in 10 mL solubilization buffer (20 mM NaH<sub>2</sub>PO<sub>4</sub>, 6M urea, pH 8.0) and a new cycle of sonication and centrifugation is carried out. The supernatant obtained is called the insoluble fraction.

The solubility test demonstrates whether the protein is mainly in the insoluble or soluble fraction. In this case, the recombinant protein is mainly in the insoluble fraction, meaning in the inclusion bodies, whereas a big part of other proteins is extracted in the soluble fraction.

This is expected for recombinant proteins expressed in *E. coli*; this result was also obtained by Burnette [32], whose results showed an expressed recombinant S1 subunit of PT at high phenotypical levels (7–22% of total cell protein content) in inclusion bodies. Choi and colleagues [48] demonstrated that overexpressed proteins are often produced in the form of inclusion bodies, and showed several advantages, such as much higher levels of protein accumulation (usually greater than 25% of total proteins) than soluble proteins. Also, inclusion bodies can be initially isolated in a highly purified and concentrated state by simple centrifugation, leading to a significant decrease in downstream processing requirements for removing contaminating host proteins; they also are resistant to proteolysis by *E. coli* proteases, allowing high-yield protein production [48].

Nevertheless this result can lead to some difficulties. The protein needs to be refolded to be soluble in a common saline buffer, which is not simple for some proteins.

#### 18.2.9 Cell Growth and Protein Production Kinetics

Cell growth kinetics and its relation to the yield of the process must be observed. This is essential for process optimization and scale-up.

Thus during the fermentation process, 2-mL aliquots are collected every hour before verification of the influence of the induction time in cell growth and protein production. These fractions are centrifuged for 5 min at 11,000 × g. Pellets are resuspended in 200  $\mu$ L solubilization buffer, pass through a cycle of four sonication pulses of 5 s with the same time interval, and are again centrifuged under the same conditions. The supernatant is then stored for further analysis in SDS-PAGE. To determine the growth rate ( $\mu$ ) at different phases of the culture, the following equation is used:  $\mu = (\ln X_t - \ln X_0)/(t - t_0)$ , where  $X_t$  and  $X_0$  are the cell concentrations at times t and  $t_0$ , respectively. Cell

concentrations are determined by multiplying OD value by 0.35. Lima [49] described that for recombinant *E. coli* cultures, 1.0 U absorbance at 600 nm corresponds to a cell concentration of 0.35 g/L, on a dry basis.

In the kinetic analysis of cell growth of cells of *E. coli* BL21 pLysS, the lag phase was not observed for not-transformed cells. However, for cells transformed with the recombinant plasmid (P), a lag phase was observed until the third hour of incubation. After 5 h of cultivation, absorbance at 600 nm reached an average value of 0.7, and then induction with IPTG was performed. The exponential phase was observed from the third hour to half of the fifth hour of growth. After addition of the inducing agent, a decrease in the growth rate was observed, which was expected because overexpression and over-production of heterologous protein impose metabolic burden stress and subsequently decreases in cell growth [50].

The graphic of the natural logarithm of the cell concentration as a function of time represents the variation in growth rate ( $\mu$ ). The calculated  $\mu$  for each phase of the growth of transformed cells (P) is: 0.358 h<sup>-1</sup> (lag phase), 0.626 h<sup>-1</sup> (exponential phase), and 0.215 h<sup>-1</sup> (after induction).

Cell concentration reached 0.42 g/L at the end of fermentation for both the negative control and transformed cells of the fermentation broth.

Analysis of protein expression in each fermentation time indicates that the maximum expression was achieved at the end of fermentation, 3 h after induction, and that there was little change in other proteins present in the sample during this time.

# 18.2.10 Optimization of Expression

The most important factors when dealing with the expression of recombinant proteins are the concentration of inducer (IPTG) and the time and temperature of induction. To study these factors, a fractional factorial  $3^{(3-1)}$  experimental design can be used with three replicates at the center point. This type of design allows determination of the statistical significance of each factor studied. In this example, the levels studied of induction time (2, 3, and 4 h) were determined using the study previously described. The levels of other variables (temperature between 25°C and 37°C and IPTG between 0.1 and 0.9 mM) were set as are commonly used for recombinant protein expression in *E. coli* [51].

Using this kind of optimization methodology, in this example the best conditions achieved for the insoluble fraction of the culture were 37°C, 0.1 mM of IPTG, and 3 h of induction. In the soluble fraction the recombinant protein band was not observed for any condition. This study also showed that an increase of IPTG concentration has a negative effect on protein production, i.e., this variable must be used at a lower value to achieve maximal expression values. This is important when thinking about industrial processes, because IPTG is an expensive additive. If lower concentrations give a better response, costs are reduced.

Results obtained for the influence of IPTG concentration agree with those obtained by Cao and colleagues [52], who tested six different concentrations of this compound to

induce 1,3-propanediol oxidoreductase expression in *E. coli* and obtained the optimal value of 0.1 mM (the lower concentration tested).

#### 18.2.11 Protein Purification

Purification is performed by affinity chromatography using a prepacked column HisTrap FF 5 mL from GE Healthcare. These columns are filled with nickel resin, which has affinity for polyhistidine tails present in the recombinant protein. Twenty-five milliliters of the clarified fraction is injected in the column after balance with binding buffer. Elution is performed using imidazole, a compound with greater affinity for nickel than polyhistidine, and thus which is able to disconnect histidine from the resin. All buffers contain urea to ensure that inclusion bodies are solubilized. The protocols were adapted from those recommended by the manufacturer (GE Healthcare).

After each chromatographic step,  $150 \ \mu L$  of each fraction is aliquoted and stored at  $-80^{\circ}C$  for subsequent dosage of total protein content [53] and protein profile electrophoresis analysis.

SDS-PAGE of the samples acquired at purification of the clarified fraction (obtained as described earlier) demonstrates that prior purification could be done but it still has some proteins that were eluted together with the recombinant protein (Fig. 18.6).

This may be because of the low concentration of imidazole (5.0 mM) in the binding and washing buffer. The column protocol recommends concentrations between 20 and 40 mM of this compound to decrease nonspecific binding. However previous tests (data not shown) indicated that higher concentrations led to the elution of the protein in binding and washing steps.

Other strategies can be tested to improve purification, including decreasing NaCl concentration in all buffers, facilitating the protein binding to the resin, and then increasing imidazole concentration to avoid nonspecific binding.



**FIGURE 18.6** Samples of purification process in HisTrap FF column. *BP*, sample before injection in column; *E1*, *E2*, *E3*, *E4*, *E5*, and *E6*, eluted fractions; *NB*, not-bounded proteins; *W1*, *W2*, *W3*, and *W4*, fractions of column washing; *MW*, molecular weight marker.

Moreover, once proteins with similar molecular weight are excluded from the sample by affinity chromatography, two steps of ultrafiltration with membranes of 20 and 35 kDa, or gel filtration chromatography, can eliminate other proteins and concentrate the sample. It is also useful to change the buffer (to remove imidazole and urea and decrease the salt concentration) and promote protein refolding, so it can be used for biological tests. Finally, the protein is eluted in four fractions (E2, E3, E4, and E5); and their concentrations, determined by Bradford method are, respectively: 133.73; 328.902; 272.176, and 91.364  $\mu$ g/mL. Therefore, the major fractions of the recombinant protein are E3 and E4. Analysis of the gel image by software gave the proportion of the band of recombinant protein in function of total bands of each sample as 54.28%, 71.53%, 75.21%, and 95.79%, respectively. Therefore, in these four fractions, it can be estimated that the total yield of protein obtained was 16.52 mg/L of the total culture; considering the purity, the yield is estimated to be 12 mg/L.

This result is more than three times higher than that obtained by Buasri and coworkers [43], who constructed *Bo. pertussis* mutant strains to produce genetically inactivated PT. They achieved protein expression of  $3.77 \pm 0.53$  mg/L. Moreover, cultivation of *Bo. pertussis* requires a rich medium and thus is much more expensive than *E. coli* fermentation.

Another attempt to produce a fusion protein of subunits S1 and S3 of PT and FHA in *Streptococcus gordonii* achieved only 12  $\mu$ g of protein per liter of culture [1]. Yet the expression of an S2 subunit of PT in *Ba. brevis* yielded 70 mg protein per liter of culture [35].

# 18.2.12 Protein Identification by DNA Sequencing

Both forward and reverse strains of the insert were sequenced in the ABI-PRISM 3100 Genetic Analyzer (Applied Biosystems). Results obtained were compared with sequences of the BLAST (NCBI) database to confirm that the correct fragment had been amplified.

Comparison of the nucleotide sequence obtained from the cloned gene with the BLAST database resulted in 99 matches with 99% identity and an E-value of 0.0. The lower the E-value, or the closer it is to zero, the more "significant" the match is. These matches were mainly of *Bo. pertussis* PT S1 subunit (ptxA) gene, as expected. There were results from *Bordetella parapertussis* and *Bordetella bronchiseptica*, as well.

# 18.2.13 Identity Confirmation by Mass Spectrometry

A protein band obtained by SDS-PAGE is digested with trypsin (Promega) at 37°C for 24 h. The sample is hydrolyzed in a mixed-solution matrix ( $\alpha$ -cyano-4-hydroxycinnamic acid II). The mass spectrum is obtained using the spectrophotometer matrix-assisted laser desorption and ionization—time-of-flight/time-of-flight Autoflex II (Bruker Daltonics, Germany). Some signals present in the mass spectrum MS were fragmented in mode mass spectrometry/mass spectrometry. Peaks are analyzed with the computer program FlexAnalysis 3.0 (Bruker Daltonics).

In this case, this analysis returned some hits according to PT S1 subunit (*Bo. pertussis*) with a score of higher than 37, which indicates identity or extensive homology (data not shown).

# 18.3 Conclusions

In the 21st century pertussis, also known as whooping cough, remains a disease that kills children and adults worldwide. Although its incidence has decreased since the first vaccination in the 1940s, it has not yet been eradicated.

For 30 years, many groups around the world have studied new technologies for pertussis vaccination that could be used not only for children but also for teenagers and adults. Even so, there is no cheap and widely distributed new vaccine for all countries, developed or not.

This chapter intended to show that there many new technologies can be studied with the aim of producing this important vaccine. A case report showed that it is possible to amplify a specific DNA fragment from *Bo. pertussis*, clone it into other kind of bacteria, and produce a heterologous protein able to immunize against whooping cough with high specificity and efficiency. This report also explained each steps required to develop such a subunit vaccine.

# References

- [1] S. Lee, S.A. Halperin, J.B. Knight, A. Tait, Purification and immunogenicity of a recombinant *Bordetella pertussis* S1S3FHA fusion protein expressed by *Streptococcus gordonii*, Applied and Environmental Microbiology 68 (2002) 4253–4258.
- [2] CDC Centers for Disease Control and Prevention, Pertussis (Whooping Cough): Causes & Transmission. Available in: http://www.cdc.gov/pertussis/about/causes-transmission.html, (August 12, 2013).
- [3] CDC Centers for Disease Control and Prevention. Pertussis (Whooping Cough): Disease Specifics. Available in: http://www.cdc.gov/pertussis/clinical/disease-specifics.html, (June 25, 2015).
- [4] L.C. Olson, Pertussis, Medicine 54 (1975) 427-469.
- [5] Anonymous, Pertussis vaccines. WHO position paper, Weekly epidemiological Records 2005 (2005) 31–38. http://www.who.int/wer.
- [6] J.R. Kerr, R.C. Matthews, *Bordetella pertussis* infection: pathogenesis, diagnosis, management and the role of protective immunity, European Journal of Clinical Microbiology 19 (2000) 77–88.
- [7] J.D. Cherry, Epidemiological, clinical and laboratory aspects of pertussis in adults, Clinical Infectious Diseases 28 (1999) 122–126.
- [8] CDC Centers for Disease Control and Prevention. Pertussis (Whooping Cough). Available in: http://www.cdc.gov/vaccines/pubs/pinkbook/downloads/pert.pdf, (June 25, 2015).
- [9] K. Todar, *Bordetella Pertussis* and Whooping Cough, Todar's Online Textbook of Bacteriology, 2005. Available in: http://textbookofbacteriology.net/pertussis.html (June 25, 2015).
- [10] K.H.G. Mills, Immunity to Bordetella pertussis, Microbes and Infection 3 (2001) 655-677.

- [11] A.A. Weiss, E.L. Hewlett, Virulence factors of *Bordetella pertussis*, Annual Review of Microbiology 40 (1986) 661–686.
- [12] E.L. Hewlett, A commentary on the pathogenesis of pertussis, Clinical Infectious Diseases 28 (1999) 94–98.
- [13] C. Locht, L. Coutte, N. Mielcarek, The ins and outs of pertussis toxin, The FEBS Journal 278 (2011) 4668–4682.
- [14] K. Kamachi, Y. Arakawa, Expression of a C terminally truncated form of pertussis toxin S1 subunit effectively induces protection against pertussis toxin following DNA-based immunization, Infection and Immunity 72 (2004) 4293–4296.
- [15] M. Pizza, A. Covacci, A. Bartoloni, M. Perugini, L. Nencioni, M.T. Magistris, L. Villa, D. Nucci, R. Manetri, M. Bugnoli, F. Giovannoni, R. Olivieri, J.T. Barbieri, H. Sato, R. Rappuoli, Mutants of pertussis toxin suitable for vaccine development, Science 246 (1989) 497–500.
- [16] W. Hou, Y. Wu, S. Sun, M. Shi, Y. Sun, C. Yang, G. Pei, Y. Gu, C. Zhong, B. Sun, Pertussis toxin enhances Th1 responses by stimulation of dendritic cells, Journal of Immunology 170 (2003) 1728–1736.
- [17] I.P. Nascimento, W.O. Dias, R.P. Mazzantini, E.N. Miyaji, M. Gamberini, W. Quintilio, V.C. Gebara, D.F. Cardoso, P.L. Ho, I. Raw, N. Winter, B. Gicquel, R. Rappuoli, L.C. Leite, Recombinant *Mycobacterium bovis* BCG expressing pertussis toxin subunit S1 induces protection against na intracerebral challenge with live *Bordetella pertussis* in mice, Infection and Immunity 68 (2000) 4877–4883.
- [18] K. Kamachi, T. Konda, Y. Arakawa, DNA vaccine encoding pertussis toxin S1 subunit induces protection against *Bordetella pertussis* in mice, Vaccine 21 (2003) 4609–4615.
- [19] WHO World Health Organization, Pertussis, 2010. Available in: http://www.who.int/ immunization/topics/pertussis/en/print.html (June 23, 2015).
- [20] R.E. Black, S. Cousens, H.L. Johnson, J.E. Lawn, I. Rudan, D.G. Bassani, P. Jha, H. Campbell, C.F. Walker, R. Cibulskis, T. Eisele, L. Liu, C. Mathers, Global, regional, and national causes of child mortality in 2008: a systematic analysis, Lancet 375 (2010) 1969–1987.
- [21] WHO World Health Organization, Recommendations for Whole-Cell Pertussis Vaccine, Technical Report Series No 941, 2007. Annex 6. Available in: http://www.who.int/biologicals/publications/trs/ areas/vaccines/whole\_cell\_pertussis/Annex%206%20whole%20cell%20pertussis.pdf (January 31, 2012).
- [22] L.F. Bricks, Vacinas acelulares pertussis para adolescentes e adultos: revisão, Pediatria 29 (2007) 208-215.
- [23] M. Cagney, P.B. McIntyre, L. Heron, A. Giammanco, C.R. McIntyre, The relationship between pertussis symptomatology, incidence and serology in adolescents, Vaccine 26 (2008) 5547–5553.
- [24] M. Iskedjian, J.H. Walker, M.E.H. Hemels, Economic evaluation of an extended acellular pertussis vaccine programme for adolescents in Ontario, Canada, Vaccine 22 (2004) 4215–4227.
- [25] G.M. Lee, S. Lett, S. Schauer, C. Lebaron, T.V. Murphy, T.A. Lieu, Societal costs and morbidity of pertussis in adolescents and adults, Clinical Infectious Diseases 11 (2004) 1572–1580.
- [26] A. Homma, R.M. Martins, E. Jessouroum, O. Oliva, Desenvolvimento tecnológico: elo deficiente na inovação tecnológica de vacinas no Brasil. História, Ciências, Saúde, Manguinhos 10 (2003) 671–696.
- [27] A.F.R. Prestes, Avaliação de adjuvantes em novas formulações de vacina tríplice bacteriana, Programa de Pós-graduação Interunidades em Biotecnologia, USP/Instituto Butantan/IPT, São Paulo, 2008 (Master's dissertation).
- [28] Y. Sato, M. Kimura, H. Fukumi, Development of a pertussis component vaccine in Japan, The Lancet 21 (1984) 122–126.

- [29] W. Keitel, Cellular and acellular pertussis vaccines in adults, Clinical Infectious Diseases 28 (1999) 118–123.
- [30] H.G. Higashi, E. Luna, A.R. Precioso, M. Vilela, F.S. Kubrusly, W.O. Dias, I. Raw, Acellular and "low" pertussis vaccines: adverse events and the role of mutations, Revista do Instituto de Medicina Tropical de São Paulo 51 (2009) 131–134.
- [31] S. Loosmore, S. Cockle, G. Zealey, H. Boux, K. Phillips, R. Fahim, M. Klein, Detoxification of pertussis toxin by site-directed mutagenesis: a review connaught strategy to develop a recombinant pertussis vaccine, Molecular Immunology 28 (1991) 235–238.
- [32] W.N. Burnette, W. Cieplak, V.L. Mar, K.T. Kaljot, H. Sato, J. Keith, Pertussis toxin S1 mutant with reduced enzyme activity and a conserved protective epitope, Science 242 (1988) 72–74.
- [33] R. Ölander, A. Muotiala, J. Himanen, M. Karvonen, U. Airaksinen, K. Runeberg-Nyman, Imunogenicity and protective efficacy of pertussis toxin subunit S1 produced by *Bacillus subtilis*, Microbial Pathogenesis 10 (1991) 159–164.
- [34] M.A. Romanos, J.J. Clare, K.M. Beesley, F.B. Rayment, S.P. Ballantine, A.J. Makoff, G. Dougan, N.F. Fairweather, I.G. Charles, Recombinant *Bordetella pertussis* pertactin (P69) from the yeast *Pichia pastoris*: high-level production and immunological properties, Vaccine 9 (1991) 901–906.
- [35] S. Kozuka, Y. Yasuda, K. Tochikubo, Expression and secretion of the S2 subunit of pertussis toxin in *Bacillus brevis*, Vaccine 14 (1996) 1707–1711.
- [36] Y. Xu, S. Zhang, B. Bolgiano, Y. Tan, C. Asokanathan, H. Zhang, D. Xing, J. Wang, Comparison of recombinant and native pertactin of *Bordetella pertussis*, Vaccine 29 (2011) 1974–1980.
- [37] R. Rappuoli, M. Pizza, A. Covacci, A. Bartoloni, L. Nencioni, A. Podda, M.T. Magistris, Recombinant acellular pertussis vaccine – from the laboratory to the clinic: improving the quality of the immune response, FEMS Microbiology Immunology 105 (1992) 161–170.
- [38] A. Nicosia, M. Bartoloni, R. Rappuoli, Expression and immunological properties of the five subunits of pertussis toxin, Infection and Immunity 55 (1987) 963–967.
- [39] K. Runeberg-Nyman, O. Engstrom, S. Lofdahl, S. Ylostalo, M. Sarvas, Expression and secretion of pertussis toxin subunit S1 in *Bacillus subtilis*, Microbial Pathogenesis 3 (1987) 461–468.
- [40] W.N. Burnette, V.L. Mar, W. Cieplak, C.F. Morris, K.T. Kaljot, K.S. Marchitto, R.K. Sachdev, C. Locht, J.M. Keith, Direct expression of *Bordetella pertussis* toxin subunits to high levels in *Escherichia coli*, Biotechnology 6 (1988) 699–706.
- [41] J.T. Barbieri, B.K. Moloney, L.M. Mende-Mueller, Expression and secretion of the S-1 subunit and C180 peptide of pertussis toxin in *Escherichia coli*, Journal of Bacteriology 171 (1989) 4362–4369.
- [42] K. Kamachi, Y. Arakawa, Development of safer pertussis DNA vaccine expressing non-toxic C180 polypeptide of pertussis toxin S1 subunit, Vaccine 25 (2007) 1000–1006.
- [43] W. Buasri, A. Impoolsup, C. Boonchird, A. Luengchaichawange, P. Prompiboon, J. Petre, W. Panbangred, Construction of *Bordetella pertussis* strains with enhanced production of genetically inactivated Pertussis toxin and pertactin by unmarked allelic exchange, BMC Microbiology 12 (2012) 16.
- [44] D. Quintana-Vázquez, M. Delgado, G. Lemos, E. Coizeau, Y. Ramos, Y. Támbara, L.J. Gonzales, G.E. Guillén, A. Álvarez, Assessment of the *Bordetella pertussis* BpCNIC0311 strain as a producing strain of genetically detoxified toxoid (PTg), filamentous hemagglutinin (FHA) and type 2 pertactin (Prn2), Journal of Infectious Diseases and Therapeutics 3 (2015) 8–20.
- [45] D.W. Stainer, M.J. Scholte, A simply chemically defined medium for the production of phase I Bordetella pertussis, Journal of General Microbiology 63 (1971) 211–220.
- [46] R.K. Gupta, S.N. Saxena, S.B. Sharma, S. Ahuja, Studies on the optimal conditions for inactivation of *Bordetella pertussis* organisms with glutaraldehyde for preparation of a safe and potent pertussis vaccine, Vaccine 6 (1988) 491–496.

- [47] J. Sambrook, D.W. Russel, Molecular Cloning: A Laboratory Manual, third ed., Cold Spring Harbor Laboratory Press, New York, 2001.
- [48] H.J. Choi, K.C. Keum, S.Y. Lee, Review production of recombinant proteins by high cell density culture of *Escherichia coli*, Chemical Engineering Science 61 (2006) 876–885.
- [49] W.J.N. Lima, Produção de proteínas recombinantes utilizando *Escherichia coli* em cultivos em alta densidade celular, Universidade Estadual de Campinas, Campinas/SP, 2004 (Ph.D. thesis).
- [50] C.P. Papaneophytou, G.A. Kontopidis, Optimization of TNF- $\alpha$  overexpression in *Escherichia coli* using response surface methodology: purification of the protein and oligomerization studies, Protein Expression and Purification 86 (2012) 35–44.
- [51] S.Y. Kim, J.H. Seo, H.J. Cha, Enhancement of heterologous proteins expression in *Escherichia coli* by co-expression of nonspecific DNA-binding stress protein, Dps, Enzyme and Microbial Technology 33 (2003) 460–465.
- [52] Y. Cao, Q. Xia, B. Fang, Optimization of expression of *dha*T gene encoding 1,3-propanediol oxidoreductase from *Klebsiella pneumonia* in *Escherichia coli* using the methods of uniform design and regression analysis, Journal of Chemical Technology and Biotechnology 81 (2006) 109–112.
- [53] M.M. Bradford, A rapid and sensitive method for the quantitation of microgram quantities of protein utilizing the principle of protein-dye binding, Analytical Biochemistry 72 (1976) 248–254.

# 19

# Cloning and Expression of a Heterologous Protein With Imunological Potential Against *Corynebacterium diphtheriae*

C. Perottoni<sup>1,2</sup>, T. Scheuer<sup>1,2</sup>, C.R. Soccol<sup>1</sup>, O. Dalagostin<sup>3</sup>, L.P.S. Vandenberghe<sup>1</sup>, V. Thomaz-Soccol<sup>1,\*</sup>

<sup>1</sup>UNIVERSIDADE FEDERAL DO PARANÁ (UFPR), CURITIBA, PARANÁ, BRAZIL; <sup>2</sup>INSTITUTO DE TECNOLOGIA DO PARANÁ (TECPAR), CURITIBA, PARANÁ, BRAZIL; <sup>3</sup>UNIVERSIDADE FEDERAL DE PELOTAS (UFPEL), PELOTAS, RIO GRANDE DO SUL, BRAZIL

# 19.1 Introduction

Diphtheria mostly developed as an infection of the upper respiratory tract, causing fever, sore throat, and malaise. A thick, gray fibrin membrane, called a pseudomembrane, often forms over the site of infection as a result of bacterial growth, toxin production, necrosis of underlying tissue, and the host immune response [1]. It is a disease caused by *Corynebacterium diphtheriae* and can be a serious problem and even fatal, especially in children, owing to airway obstruction and tissue damage. The toxin that is produced by the pathogen can spread through the bloodstream and cause muscle paralysis, heart and kidney failure, and even death. About 5–10% of people who develop diphtheria can die; most experience permanent damage [2]. In 2000, 30,000 cases were observed and 3000 deaths were reported to be caused by diphtheria worldwide [3]. Even with vaccination campaigns, the disease is still a public health concern [1,3,4].

*C. diphtheriae* produces a toxin that can harm and destroy human tissues and organs. It often develops in the throat and sometimes the tonsils, but in the tropics it also causes ulcers on the skin. It can affect people of all ages, but it is most common in non-immunized children [1].

Because children are highly susceptible, artificial immunization at an early age is universally used. Toxoid is given in two or three doses (1 month apart) for primary immunization at age 3-4 months. A booster injection should be given about a year later,

<sup>\*</sup>Corresponding Author.
and it is advisable to administer several booster injections during childhood. Infants are usually immunized with a trivalent vaccine containing the diphtheria toxoid, pertussis vaccine, and tetanus toxoid (DPT or DTaP vaccine) worldwide [3].

As a preventable disease, it should be controlled as soon as the population is vaccinated. However, actual vaccines do not provide long-lasting immunity. The titer against diphtheria often disappears after some years and boosters are generally not provided to the population [3].

There have being some efforts in the development of a new class of vaccines, subunit vaccines capable of long-lasting immunity and with reduced side effects. The diphtheria vaccine has not yet achieved at this point, but it should be interesting for disease control, because the booster should be taken at any age, without restrictions [3].

# 19.2 Historical Aspects

Throughout history, diphtheria was a leading cause of death among children. Diphtheria and scarlet fever were sometimes difficult to distinguish. The illnesses were known by a variety of names, such as throat disease, throat distemper, throat ail, malignant croup, angina suffocative, gangrenous sore throat, canker ail, malignant quinsies, purid, malignant or pestilential sore throat; cynanche trachealis, malignant angina, "strangling angel of children," and "*Garotillos*," among others [5].

In the fourth century BC, Hippocrates described the symptoms of asphyxia and clinical evolution of what could be diphtheria. In Spain in 1613, an epidemic of diphtheria caused the year to be named "*El Año de los Garotillos*" ("Year of Strangulations") because of the many deaths caused by diphtheria suffocation [6].

In 1735 there was an epidemic in New England. In 1826, Pierre Philippe Bretoneau described the disease as a tough membrane that is formed in the throat, for the first time giving diphtheria disease its name. In England it was known as Boulogne sore throat, because it had spread in France in 1746 [6].

In 1883, Klebs demonstrated that *C. diphtheriae* was the agent of diphtheria. Loeffler, in 1884, had found that the organism could be cultured only from the naso-pharyngeal cavity, and postulated that the damage to internal organs resulted from a soluble toxin [1].

Roux and Yersin discovered the diphtheria toxin in 1888, and showed that animals injected with sterile filtrates of *C. diphtheriae* developed organ pathology indistinguishable from that of human diphtheria; this demonstrated that a potent exotoxin was the major virulence factor [1,7].

In the 1890s the antitoxin against diphtheria was developed by Shibasaburo Kitasato and Emil von Behring. The first vaccine was developed in the 1920s. In 1923, Gaston Ramon, a veterinarian of the Pasteur Institute in France, developed the diphtheria toxoid [1].

With the development and administration of the diphtheria vaccine, the incidence of diphtheria decreased significantly, but it still is a public health problem [3].

# 19.3 The Pathogen and the Toxin

*C. diphtheriae* is gram-positive, nonsporulating, unencapsulated, nonmotile *Bacillus*. Corynebacteria are classified as Actinobacteria and are related phylogenetically to Mycobacteria and Actinomycetes. They belong to the kingdom Bacteria, phylum: Actinobacteria, order: Actinomycetales, suborder: Corynebacterineae, family: Corynebacteriaceae, genus: *Corynebacterium*, species: *C. diphtheriae*. They undergo snapping movements just after cell division, which brings them into characteristic forms resembling Chinese letters or palisades [8].

Not all the strains of *C. diphtheriae* are pathogenic. The pathogenicity comes from the toxin that is produced only if the *Bacillus* itself is infected (lysogenized) with a bacteriophage, which carries the genetic information. This is a high conserved polypeptide encoded by the bacterial chromosome, the tox-gene [9-11]. The structural gene of the diphtheria toxin has 535 residues, a 58-kDa exo toxin, and two subunits linked by a disulfide bond [1,10,12].

Outside the host cell, the toxin is inactive. The nontoxic fragment B (Mr 37,195) of the toxin attaches to the heparin-binding epidermal growth factor (HB-EGF), a member of the superfamily of growth factors. Just after the attachment in the cell surface, the highly toxic fragment A is detached from the N terminus of the polypeptide chain by cleavage with trypsin [13]. At this point, the fragment A (Mr 21,167), is able to catalyze the adenosine diphosphate (ADP)-ribosylation of elongation factor 2 in protein translation in practically all eukaryotic cells [9,14,15].

The toxin can produce local tissue necrosis, and when absorbed into the bloodstream, it causes toxemia and systemic complications including paralysis owing to demyelinating peripheral neuritis and cardiac failure due to myocarditis [10,16].

It has four biotypes according to colony morphology (mitis, intermedius, gravis and belfanti), as well as into lysotypes based on corynebacteriophage sensitivity [1]. Most infections have been caused by gravis or mitis biotypes, but the clinical and public health management is identical for all toxigenic strains [10].

The diphtheria toxin (DT) is a molecule synthesized as a single chain that can be cleaved by trypsin into two fragments, A and B. The A fragment is located at the N terminus of the polypeptide chain, and catalyzes the ADP-ribosylation of elongation factor 2 in practically all eukaryotic cells. This mechanism blocks the protein transcription, making it impossible for the cells to continue normal metabolism, and resulting in cell death [17].

# 19.4 The Disease

There are two types of clinical manifestation of diphtheria: nasopharyngeal and cutaneous. The symptoms of pharyngeal diphtheria may appear as mild pharyngitis, but can evolve to hypoxia because of airway obstruction by the pseudomembrane. The cutaneous type is characterized by skin lesions covered by a gray-brown pseudomembrane as well. The lesions start as vesicles and quickly form small, clearly demarcated, and sometimes multiple ulcers [1,10,18]. Usually they appear on exposed parts, especially the legs.

Life-threatening complications, principally loss of motor function (e.g., difficulty in swallowing) and congestive heart failure, may develop as a result of the action of DT on peripheral motor neurons and the myocardium [1].

Classical nasopharyngeal diphtheria is characterized besides the membranous pharyngitis with fever, enlargement of the cervical lymph nodes, and edema, which results in a "bull neck" appearance. Although not always present, the pseudomembranes are typically gray, thick, fibrous, and firmly adherent. Dislocation of the membrane releases a great amount of toxin. There is gradually increasing hoarseness and stridor when it reaches the pharynges. It can cause nasal discharge of bloody secretions [1,10,16].

Diphtheria is no longer easily diagnosed on clinical grounds. Mild cases of the disease resemble streptococcal pharyngitis and the classical pseudomembrane may not develop, particularly in people who have been vaccinated. Because the disease is rare, principally in developing countries and regions with better hygienic and sanitary conditions, and many clinicians may never encounter a case, the clinical diagnosis is missed [1,19]. Not all laboratories routinely culture throat swabs for *C. diphtheriae*, which further increases the potential for missed or delayed diagnosis [10,19].

# 19.5 Epidemiology

Diphtheria is found worldwide, although it is not common in industrialized countries because of longstanding routine use of diphtheria-tetanus-pertussis (DTP) vaccine. However, large epidemics occurred in several Eastern European countries in the 1990s, and many cases have been reported in developed countries [20,21]. As an example, outbreaks emerged after the breakup of the former Soviet Union in the late 1980s. Therefore, in the constituent countries, vaccination rates fell so low that there was a surge in diphtheria cases. In 1991 there were 2000 cases of diphtheria in the USSR. By 1998, according to Red Cross estimates, there were as many as 200,000 cases in the Commonwealth of Independent States, with 5000 deaths [8].

The case-fatality rate for diphtheria has changed little over the past 50 years. The overall case-fatality rate for diphtheria is 5-10%, with higher death rates (up to 20%) among people younger than age 5 and older than age 40 years. Before that, the disease was fatal in half of cases [8].

Worldwide there are still cases of diphtheria, with 4887 cases reported by the World Health Organization (WHO) in 2011. In the same year, in south France, one case was registered, in a 40-year-old man [22]. Whereas diphtheria primarily affected younger children in the prevaccination era, an increasing proportion of cases today occur in unvaccinated or inadequately immunized adolescents and adults [20].

Much research involves diphtheria cases. They describe insufficient vaccination coverage; discontinuation of the vaccination schedule; insufficient boosters after

primary doses in childhood, and so on, trying to discover why diphtheria is still circulating. Because diphtheria is present only among humans, eradication of the disease is possible [17].

According to Myers, even with a booster, the protective rate for diphtheria was 94%, and people above age 19 years were more responsive to the vaccine [23].

Research developed 25–30 years ago [24] involved 403 vaccinated people screened for diphtheria antitoxin titers. The result was that 19% of the group was unprotected. The research reaffirms the continuous decline of diphtheria antitoxin titer after vaccination, even in people who were revaccinated at age 12. As a conclusion, the authors suggested that people who had their primary vaccination before age 2 years or revaccination more than 10 years ago should be revaccinated [24].

Adults were revaccinated and protective serum diphtheria antitoxin levels were attained by all patients receiving diphtheria toxoid [25]. Mild local reactions were reported by 22% of the patients, and the authors concluded that the revaccination is advisable.

Kadirova et al. [26] believed that resurgent cases are reported as a result of the lack of vaccination with the dissolving of the Soviet Union. In the same way, Galazka et al. [27] suggested high immunization coverage of target groups, prompt diagnosis, management of diphtheria cases, and rapid identification of close contacts with effective management to prevent secondary cases. These actions should be taken to prevent the incidence of new diphtheria cases [26,27].

Hardy [28] suggested that current vaccination recommendations also need to be reviewed to ensure that population immunity are adequate to prevent a resurgence of diphtheria in Europe and North America [28].

A vaccination campaign with DT vaccine targeting the adult population was analyzed by Christenson et al. [29] and a major improvement in immunity to diphtheria with only minor side effects was observed.

Nandi et al. [30] described the number of cases in India; 70% of affected patients were not previously immunized against diphtheria. The mortality rate was about 16% in early diagnoses and treatment with the antidiphtheritic serum and antibiotics were very important for the cure. It was also stated that inadequate immunization was responsible for continued outbreaks and deaths caused by diphtheria.

Research on the loss of immunity 10 years after vaccination revealed that antidiphtheria titers are usually below the protective amount [23,31,32]. Authors also showed that vaccination coverage is still at about 83%; according to WHO and UNICEF, there is a clear gap in the immunization campaign. Although the vaccine is effective in preventing diphtheria-related death, its overall effectiveness against symptomatic disease is only estimated to be 70–90%. Diphtheria outbreaks were reported among highly vaccinated populations in closed communities [32–37].

Karakus et al. [38] evaluated antibodies against the intact DT and DT subunits A and B in 1319 individuals. Those with a protective antibody level were reactive to subunit B; the authors suggested its use as a detection agent for the diphtheria titer.

Unfortunately, the situation is not different on the American continent: Many regions have outbreaks revealing that the vaccination coverage and immunity given are not efficient. There are endemic regions that often have outbreaks despite the vaccination program. Haiti, the Dominican Republic, and Brazil are countries that have noticed a great number of cases in the past 10 years. There are still sporadic cases in several countries, excluding those that have not reported to the authorities, and mis-diagnoses [39].

# 19.6 Diphtheria Vaccines

Diphtheria vaccines are based on diphtheria toxoid, an inactivated bacterial toxin that induces protective antitoxin. Diphtheria toxoid combined with tetanus and pertussis vaccines (DTwP) has been part of the WHO Expanded Program on Immunization since its inception in 1974. In 1980–2000, the total number of reported diphtheria cases was reduced to less than 90% [16,18]. After the introduction of routine immunization with vaccines containing diphtheria toxoid in the 1940s and 1950s, the incidence of diphtheria declined dramatically [27].

Actual vaccines containing the diphtheria component are still the crude protein toxoid. The 58-kDa toxin is the antigen and is converted to a toxoid with formaldehyde and is crudely purified. The cells are separated by centrifugation or membrane filtration, and the supernatant is treated with formaldehyde at 0.75% and stored for 4–6 weeks at 37°C to allow complete detoxification [7]. Then, the components are extracted from the native's bacteria; the toxins from tetanus and diphtheria and the cell lysate for pertussis are the components of the DTwP vaccine [40].

There are four combination vaccines used to prevent diphtheria, tetanus, and pertussis. The first, DTP vaccine, has components that include diphtheria (D), tetanus (T), and pertussis whole inactivated cells (wP) [41].

The second combination, DTaP, was developed because of the high incidence of collateral effects. Efforts in the sense of producing a safer vaccine with fewer unpleasant reactions were done. In this way, antigens were selected to induce immunity with fewer antigens than the whole cell vaccine. This model of vaccine would be safer but more expensive. The rate of redness, fever, and pain is about 90% lower. This vaccine, which was developed in Japan in 1981, is indicated for children aged 6 years and younger to protect them from these three diseases. Older children, adolescents, and adults should not take this vaccine [41].

The third combination, the Tdap, is indicated for adolescents and adults. It was licensed in the United States in 2005. The difference from the previous vaccine is the reduced diphtheria and pertussis toxoids concentration to prevent side effects. The pertussis component is acellular like the DTaP, but with a lower concentration of antigen. Existing vaccines with this technology are Adacel (Sanofi) and Boostrix (GlaxoSmithKline) [41,42].

The fourth combination is the DT vaccine with diphtheria and tetanus components, just like the DTP vaccine, but without the pertussis component The Td vaccine has a lower antigen concentration and is indicated for adolescents and adults [41].

#### 19.6.1 Subunit Vaccines

Since the 1920s, the diphtheria vaccine has been produced the same way: as a batch cultivation of the pathogen. The toxin is secreted from the cell, and only the supernatant is used to produce the vaccine. In this supernatant there is not only the necessary antigen for immunization but a large number of other components (from the media, cell residues, and cell metabolism products) are also present in this protein pool, which can contribute against the vaccine's side effects.

Producers conduct different downstream processes, with a low purity (around 70–80%) antigen end product [13]. A huge amount of research is being conducted in the sense of developing new technologies capable of improving the diphtheria vaccine with long-lasting immunity. One is the subunit vaccines [43].

The world already uses subunit vaccines, with safer results and rare side effects. The hepatitis B and the pertussis vaccine (aP) are already used and approved by the WHO, with success. The pertussis acellular vaccine contains inactivated pertussis toxin (protein) and may contain one or more other bacterial components. The pertussis toxin is detoxified either by treatment with a chemical or by using molecular genetic techniques, transforming the molecule of the toxin into a nontoxic epitope [44].

There are three forms of subunit vaccines: protein, polysaccharide, or conjugated protein-polysaccharide, using these molecules as antigen. The technology is based on the presence of an antigen to the immune system without viral or bacterial particles, using a specific, isolated protein of the pathogen. It can contain one or several antigens from the pathogen. A weakness of this technique is that isolated proteins, if denatured, may bind to antibodies different from the protein of the pathogen [45].

There is a subunit vaccine against diphtheria, the Td and the Tdpa, with the diphtheria toxin purified for chromatographic method and with lower antigen concentration. However, the cost of the downstream process is still an impediment to producing and delivering the vaccine throughout developing countries [5,41].

These kinds of vaccines also can provide a stronger immune response, with fewer side effects. This epitope for subunit vaccines can be obtained by purification processes, which can considerably increase the vaccine price. They can also be produced with the use of recombinant techniques. These techniques are leading the search for the ideal vaccine, because they are safer and cheaper, heat-stable, and easy to administer, preferably single-dose, and capable of inducing a broad immune response with lifelong memory in both adults and infants. The perspective is that the same may be developed for diphtheria as an improvement in immunization and the prevention of undesirable side effects [45].

As stated, subunit B is capable of inducing immunity against DT and much research in ongoing in this sense [38,46–49].

Table 19.1 lists some reports on the use of subunit B as a potential vaccine, with good results in producing antibody response and the ability to neutralize the whole DT when challenged.

Cell	Objective	Author	Year
Escherichia coli	Synthesis of diphtheria tox-gene products	Murphy J. et al. [50]	1974
E. coli K12	Cloned diphtheria toxin fragment A	Leong D. et al. [51]	1983
	T cell—specific antibody identifies regions of B-chain active in cell entry	Colombatti M. et al. [52]	1986
E. coli	Construction and expression of diphtheria toxin-encoding gene derivatives in <i>E. coli</i>	Zdanovsky et al. [17]	1992
	Corynebacterium diphtheriae dtx R homologue from Streptomyces lividans and Streptomyces pilosus	Gtünter-Seeboth et al. [53]	1995
E. coli	Subunit B conjugated to meningococcal as vaccine to diphtheria toxin and meningococcal proteins	Johnson N. et al. [54]	1997
E. coli	Two recombinant fragments of diphtheria toxin Diphtheria toxin CRM	Lobeck K et al. [55] Randall K. H. [57]	1998 2000
Mycobacterium bovis BCG	Diphtheria toxin CRM197 mutant	Miyaji E. N. et al. [58]	2001
Pichia pastoris	Targeted introduction of diphtheria toxin resistant mutation into chromosomal EF-2 locus of <i>P. pastoris</i> and expression of immunotoxin in EF-2 mutants	Liu Y. Y. et al. [59]	2003
E. coli	Recombinant diphtheria fusion toxin DT388IL3	Urieto J. O. et al. [60]	2004
Streptococcus gordonii	Recombinant diphtheria toxin fragment A	Lee C. W. et al. [56]	2004
E. coli BLR(DE3) pLysS	Murine C3d sequences to diphtheria toxin fragment B	Gor D. O. et al. [61]	2006
E. coli	Expression and purification of trivalent pertussis—diphtheria —tetanus toxin fusion protein	Aminian M. et al. [62]	2007
	Recombinant subunits A and B and new method of their estimation	Kaberniuk A. et al. [63]	2009
E. coli	Diphtheria toxin CRM197 mutant	Stefan A. et al. [64]	2010
	Biochemical and biological characteristics of cross-reacting material 197	Bröker M. et al. [46]	2011
Insect cells	Recombinant protein vaccines produced in insect cells	Cox M. M. J. et al. [47]	2012
	Subunit B using as drug delivery across blood—brain barrier	Chen Y. et al. [48]	2012
E. coli	A and B fragments of C. <i>diphtheriae</i> toxin toward recombinant diphtheria vaccine	Abulmagd S. et al. [49]	2013

**Table 19.1** Research Developed Containing Diphtheria Toxin or Its Subunits as the Main Objective or a Collaborative Molecule

# 19.7 Technological Development of a Diphtheria Vaccine: Case Description

Improvement in technology is necessary to reduce side effects and the vaccination of the world population with a better-quality product, as well as vaccination program uptake effort has been made to develop a new, safer product for immunization.

This case description presents a new diphtheria vaccine using recombinant technology. The steps of the development are shown in Fig. 19.1.

#### 19.7.1 Diphtheria Culture and DNA Extraction

The strain of *C. diphtheriae* was received courtesy of the Technology Institute of Paraná-TECPAR (Park-Williams 8). A freeze-dried sample of the strain, which was reported as a producer strain of the diphtheria vaccine [65] was employed. The cultivation was carried out in a modified culture medium for *C. diphtheriae* [66] with 100 mg/mL casein



FIGURE 19.1 Flowchart of subunit vaccine development.

hydrolysate and incubated at 37°C ( $\pm$ 1°C) for 72 hours [67]. The toxin was inactivated in a water bath at 100°C for 30 minutes.

The DNA extraction was performed by employing the defined protocol for grampositive bacteria DNA extraction; 60% purity was obtained with a concentration of DNA obtained at 270 mg/mL.

For DNA amplification by polymerase chain reaction (PCR), two primers were designed, corresponding to DT subunit B [193–534 base pairs (bp), the immunologic portion of the molecule], with the assistance of the software Vector NTI (Invitrogen). The sense primer was designed with *Bam*HI, and the antisense with *Hin*dIII to enable the insertion in the right direction in the vector.

The sequence of the gene that contains 1026 bp totalized 1046 bp with the nucleotide containing the restriction sites for the enzymes mentioned. The initial 579 bp of the toxin sequence was excluded from the toxin activity portion [68].

Gene amplification was prepared with GoTaq Flexi DNA Polymerase (Promega), following the manufacturer's protocol. The PCR product was purified using the GFX PCR Kit DNA and Gel Band Purification (GE Healthcare), following the manufacturer's protocol. The PCR purification results were analyzed in 1.0% agarose gel electrophoresis, 80 V, for 1 hour to check their quality and concentration. Only one band appeared in the gel, between molecular weight markers 1636 and 1018, indicating the interest sequence (1046 bp).

#### 19.7.2 Plasmid and Diphtheria Toxin Subunit B DNA Ligation

The pAE vector was chosen because it has several advantages: it is not commercial, it was replicated in *E. coli*, it has multiple cloning sites, it has an ampicillin-resistant gene, it is a strong promoter of T7 phage, and it involves isopropyl  $\beta$ -D-1-thiogalactopyranoside (IPTG) induction and the inclusion of a polyhistidine tail in the cloned protein. The binding reaction between the insert (DT subunit B) and plasmid pAE was performed using a T4 DNA ligase (Promega) overnight at 4°C following the manufacturer's protocol.

### 19.7.3 Plasmid Linking to Diphtheria Toxin Subunit B Insert Cloning

The plasmid containing the subunit B insert was transfected into *E. coli* DH5alfa with the thermic chock protocols (transformation of *E. coli*), to plasmid replication.

#### 19.7.4 Insert Sequencing

Both DNA strains (forward and reverse) of the insert (DNA sequence of DT subunit B) were sequenced in capillary electrophoresis (ABI PRISM3100 Genetic Analyzer; Applied Biosystems). The results were compared with BLAST sequencing database (NCBI) and resulted in 99 hits with 99% identity and an E-value of 0.0 for "*C. diphtheriae* strain DIFT020 DT (tox) pseudogene, partial sequence" deposit. V01536.1. The result matched with sequence B of DT.

#### 19.7.5 Protein Expression

The extracted plasmid was transfected into *E. coli* BL21 pLysS, using the same transfection protocol, to express the recombinant protein. To confirm the presence of plasmid with the insert, a screening was made and analyzed by electrophoresis (0.8% agarose gel). A cell blank was produced and stored at  $-80^{\circ}$ C until use.

Cells cultivation was then scaled up to 100 mL (on a rotary shaker at  $37^{\circ}C \pm 1^{\circ}C$  at 180 rpm) and carried out until the culture reached an optical density of 0.6–0.8 at 600 nm. At this point, 0.3 mM IPTG was added to the culture and incubated under the same conditions for 3 hours. Protein expression was verified by 10% sodium dodecyl sulfate–polyacrylamide gel electrophoresis (SDS-PAGE).

An overexpression band of about 40 kDa was observed after induction. The band indicated DT subunit B, because its weight was about 37 kDa. A recombinant protein band with lower intensity was detected in the noninduced fraction, which suggesting that the protein had been expressed before induction.

Protein solubility was verified by sonicating the pellet with phosphate buffer (20 mM  $Na_2HPO_4$ , pH 8.0) and dissolution in buffer (20 mM  $Na_2HPO_4$ , 500 mM NaCl, 6 M urea, pH 8.0) for further steps. Recombinant proteins can be secreted in supernatant or stored in inclusion corpuscles. Both fractions were verified by 10% SDS-PAGE. An advantage of protein expression in inclusion bodies is low toxicity to cells; these corpuscles are easily precipitated by centrifugation because they are substantially free of contaminant aggregates [69–71].

Choin et al. [72] developed a high-yield recombinant protein production system. The overexpressed proteins produced in inclusion bodies have several advantages. It is possible to accumulate high levels of recombinant protein (greater than 25% of total protein) compared with soluble proteins. The inclusion bodies may initially be isolated in a highly purified state and concentrate by simple centrifugation, decreasing downstream steps for removing contaminating proteins of the host cell. For these reasons, the insoluble fraction was chosen to carry out the following purification steps.

To verify the best cultivation conditions and the effects of inducer concentration (IPTG), temperature, and time on protein expression, an experimental design of fractional factorial 3 (3-1) with three repetitions at the central point was carried out. The best culture conditions were observed at 31°C, 0.5 mM IPTG, at 3 hours.

#### 19.7.6 Protein Identity: Western Blot and Protein Sequencing

The protein band obtained by SDS-PAGE was analyzed by the method adapted by Shevchenko et al. [73]. The mass spectrum (MS) was obtained using the matrix-assisted laser desorption/ionization-time-of-flight/time-of-flight-II Autoflex (Bruker Daltonics, Germany) spectrometer. Peaks were analyzed by Flex Analysis 3.0 software (Bruker Daltonics). The MS/MS spectra were combined and subjected to searches using Mascot

software. This analysis identified the insert sequence as "chain A, with refined structure of DT at 2.3 Å resolution," database NCBInr (NCBI Access No. gi576189). The protein expressed is positive to the sequence of DT subunit B.

The protein fractions (soluble and insoluble) were also analyzed by western blot whit anti-his6 antibodies that were positive, indicating the success of polyhistidine tail inclusion in the recombinant protein. Protein expression was verified in this step; it showed only a band without nonspecific reactions. For the presence of the polyhistidine tail, it is important to perform the purification step through nickel column affinity chromatography, the chosen purification mechanism.

#### 19.7.7 Cell Growing and Protein Production Kinetics

During protein expression, 2-mL aliquots were collected every hour to evaluate the influence of induction time in cell growth. The cell concentration was determined by a spectrophotometer at 600 nm. Two kinetic parameters were used to evaluate bacterial cell growth: specific growth rate ( $\mu$ ) and growth velocity (dx/dt). Cell concentrations were determined by multiplying the absorbance values by 0.35. According to Lima [74], in *E. coli* recombinant cultures, 1.0 absorbance unit at 600 nm is equivalent to a cell concentration of 0.35 g/L dry weight.

After 5 h of cultivation, induction was performed. The addition of IPTG to the culture caused the metabolism to deviate to overexpression of the recombinant protein, generating cellular stress and reducing cell growth [72]. The culture was stopped and the cell concentration at the end of the cultivation time was approximately 0.360 g/L.

The highest expression of recombinant protein was achieved within 3 h after induction with IPTG, and there was no difference in the expression of other proteins present in the culture.

#### 19.7.8 Purification

To perform chromatography, the cell pellet from 2 L cultivation was sonicated (four cycles of 20 s with the same time in an ice bath) with buffers. The obtained supernatant was analyzed with SDS-PAGE. The total protein content [75] and protein profile were determined by SDS-PAGE. The protein concentration of the samples obtained after the sonication cycles biomass was 1.300 g/mL and 1.100 mg/mL, respectively.

A prepacked HisTrap column FF 5 mL (GE Healthcare) with polyhistidine tail affinity was used. Purification was performed on an ÄKTA Purifier 100 automated system (GE Healthcare). Protocols were adapted according to the manufacturer's recommendations, with a flow rate of 5 mL/min. To avoid the presence of contaminants, elution gradient was performed with 5 column volumes of binding buffer (20 mM Na<sub>2</sub>HPO<sub>4</sub>, 500 mM NaCl, 6 M urea, 5 mM imidazole, dithiothreitol or DTT 2 mM, pH 7.4) and elution buffer (20 mM Na<sub>2</sub>HPO<sub>4</sub>, 500 mM NaCl, 6 M uremia, 500 mM imidazol, DTT 2 mM, pH 8.0).



**FIGURE 19.2** Purification process in HIS trap FF electrophoresis gel. E1-8, elution of recombinant protein; L1-3, wash; NL, unbound; PM, molecular weight.

Purification by liquid chromatography of the recombinant protein in a nickel affinity column is shown in Fig. 19.2. Electrophoresis on SDS-PAGE gel shows fractions obtained in the purification step.

The fractions collected from purification in elution showed that protein concentrations (g/mL) varied from 80.33 to 595.58 mg/mL. A pool of the fractions was made and called "purified protein" with a final concentration of 418  $\mu$ g/mL. Endotoxins were removed using the resin X-114, 1% v/v. The final protein concentration was then 348 mg/L. Protein loss in the lipopolysaccharide removal process was about 9%. The sample, which was free of pyrogens, had a recovery of 33% and could be used for subsequent inoculation in animals for verification of immunogenicity. Protein production and recovery are in accordance with others' reports [76].

# 19.8 Conclusions and Perspectives

Much research has been conducted to develop new technologies capable of improving not only a vaccine against diphtheria but also other vaccines, with the aim of reducing adverse effects and ensuring long-lasting immunity. One new technology is the subunit vaccine [43].

Diphtheria has a subunit vaccine, the Td and Tdpa, using chromatographic methods for purification and lower concentrations of antigen. However the cost of the downstream process is still an obstacle for the production and distribution of this type of vaccine in developing countries [5,41]. These kinds of vaccines are also capable of granting a stronger immune response, with fewer incidents of side effects. Nevertheless epitopes obtained by purification processes can significantly increase the price of the vaccine [5].

A cheaper alternative is to develop recombinant vaccines. These techniques are leading the development of an ideal vaccine that is safer and cheaper, promotes higher thermal stability, in a single dose, and is capable of inducing an immune response with lifetime memory, which can be used in adults and children [38,46–49].

For diphtheria, research developed in this direction is a promising alternative for a safer and more effective vaccine [38,46–49].

Nascimento et al. demonstrated the DT subunit B in *E. coli* to obtain high levels of protein. However, at the end of the process, low quantities of the protein were recovered. In this case, the expressed protein was in an insoluble form, probably in inclusion corpuscles. The refolding process is also a concern when working with inclusion corpuscles, mainly when the right conformation is needed for its activity, such as in vaccines for conferring the right immunity [69,70,76].

The DT is able to bind to cell surface in HB-EGF receptor. The receptor is able to perform the internalization of molecules attached into it in human cells. This mechanism is being used as a vehicle for drugs, vaccines, antitumor medicine, and molecules of interest.

The diphtheria toxoid can be used as an immunogenic molecule, as an adjuvant, or attached to other drugs. Because of such properties there are possibilities of developing vaccines with recombinant DT or its subunits. These studies indicate promising immunization against the disease [46-52,54-57,61-63,77].

Cytotoxins blocking HB-EGF, diagnostic system tests, and serum therapy with recombinant DT DNA sequence are also being developed [59,77]. Research into therapeutic drugs, such as bacterial flagellin administered in combination with DT, or thymocytes that co-stimulate interleukin-17 producers, are successfully tested against microbes [78].

There are still further advances in the use of recombinant derivatives for the treatment and prophylaxis of oncology. One is the fusion of genetically modified toxins selected for acute myeloid leukemia blasts, using the first 388 amino acid residues of DT fused to human interleukin-3, which has achieved good results in the treatment of cancer [60].

The design of autoimmune drugs, controlling the immune system reaction to infectious diseases and other diseases, is also a promising target for a new approach to DT as an adjuvant agent for developing new drugs [43,79].

The use of recombinant technology with DT as a potential and improved vaccine is a promising solution for disease control in the near future. In the case description of this chapter, steps including immunogenic verification and the scale-up of protein production are still needed. However, research into new methods and technologies for vaccination improvement are certainly the best method to find an effective solution for disease control as well as a reduction of its treatment costs.

# References

- J.R. Murphy, Corynebacterium diphtheria, in: S. Baron (Ed.), Medical Microbiology, fourth ed., Galveston (TX) University of Texas Medical Branch at Galveston, 1996.
- [2] CDC, Updated recommendations for use of tetanus toxoid, reduced diphtheria toxoid, and acellular pertussis (Tdap) vaccine from the Advisory Committee on Immunization Practices, MMWR. Morbidity and Mortality Weekly Report 60 (1) (2011) 13–15. http://www.cdc.gov/mmwr/preview/ mmwrhtml/mm6001a4.htm.
- [3] WHO, Subunit Vaccines, 2013. http://www.vaccine-safety-training.org/subunit-vaccines.html.

- [4] A.L. Mattos-Garibaldi, Diphtheria remains a threat to health in the developing world: an overview, Memorias do Instituto Oswaldo Cruz 98 (8) (2003) 987–993.
- [5] R. Rappuoli, S. Black, P.H. Lambert, New decade of vaccines 2. Vaccine discovery and translation of new vaccine technology, Lancet 378 (2011) 360–368.
- [6] E. Laval, El garrotilho (Difteria) en España (Siglos XVI y XVII), Revista Chilena de Infectología 23 (2006) 78-80.
- [7] A.M. Pappenheimer Jr., A.M. Pappenheimer, The diphtheria bacillus and its toxin: a model system, Journal of Hygiene Camb 93 (1984) 397–404.
- [8] Todar, Online Textbook of Bacteriology. The Good, the Bad, and the Deadly, 2010. http://textbookofbacteriology.net/kt\_toc.html.
- [9] N.B. Groman, Evidence for the active role of bacteriophage in the conversion of nontoxigenic *Corynebacterium diphtheriae* to toxin production, Journal of Bacteriology 69 (1) (1955) 9. http://jb. asm.org/content/69/1/9.full.pdf.
- [10] J.M. Bonnet, N.T. Begg, Control of diphtheria: guidance for consultants in communicable disease control, Communicable Disease and Public Health (1999) 242–249. http://www.hpa.org.uk/webc/ HPAwebFile/HPAweb\_C/1194947407702.
- [11] S. Buzzi, D. Rubboli, G. Buzzi, A.M. Buzzi, C. Morisi, F. Pironi, CRM 197 (nontoxic diphtheria toxin): effects on advanced cancer patients, Cancer Immunology, Immunotherapy 53 (2006) 1041–1048. http://dx.doi.org/10.1007/s00262-004-0546-4, 2004.
- [12] O. Gómez-Duarte, J. Galen, S.N. Chatfield, R. Rappuoli, L. Eields, M.M. Levine, Expression of fragment C of tetanus toxin fused to a carboxyl-terminal fragment of diphtheria toxin in *Slamonella typhi* CVD 908 vaccine strain, Vaccine 13 (1995) 1596–1602.
- [13] A.M. Pappenheimer Jr., Diphtheria toxin, Annual Review of Biochemistry 46 (1977) 69.
- [14] W. Cieplaks, M.H. Gaudin, L. Eidels, Diphtheria toxin receptor. Identification of specific diphtheria toxin-binding proteins on the surface of vero and BS-C-1 cells, The Journal of Biological Chemistry 262 (1987) 13246–13253.
- [15] W. Abdel-Fattah, V. Scheidt, S. Uthman, M.J.R. Stark, R. Schaffrath, Insights into diphthamide, key diphtheria toxin effector, Toxins 5 (5) (2013) 958–968.
- [16] WHO, Weekly Epidemiological Record, 2006. http://www.who.int/wer/2006/wer8103.pdf.
- [17] A.G. Zdanovsky, O.I. Kulaeva, N.K. Yankovsky, Construction and expression of diphtheria toxinencoding gene derivatives in *Escherichia coli*, Gene 116 (1996) 81–86.
- [18] CDC, Centers for Disease Control and Prevention. Epidemiology and prevention of vaccine preventable diseases, in: W. Atkinson, J. Hamborsky, S. Wolfe (Eds.), Chapter 6 Diphtheria, twelfth ed., Public Health Foundation, Washington, DC, 2012.
- [19] R.W. Bryad, Diphtheria 'The strangling angel' of children, Journal of Forensic and Legal Medicine 20 (2) (2013) 65–68.
- [20] CDC, Centers for disease control and prevention. Summary of notifiable diseases: United States, MMWR. Morbidity and Mortality Weekly Report 51 (53) (2004) 1–84.
- [21] WHO, The Immunological Basis for Immunization Series Module 2: Diphtheria, World Health Organization, Geneva, 1993. WHO/EPI/GEN/93.12, http://www.health.govt.nz/system/files/ documents/publications/4-diphtheria-v2.pdf.
- [22] C. Rousseau, E. Belchior, B. Broche, E. Badell, N. Guiso, I. Laharie, D. Levy-Bruhl, Diphtheria in the south of France, March 2011. Euro surveillance: bulletin europeen sur les maladies transmissibles, European Communicable Disease Bulletin 16 (19) (2011) 5–7.
- [23] M.G. Myers, C.W. Beckman, R.A. Vosdingh, W.A. Hankins, Primary immunization with tetanus and diphtheria toxoids: reaction rates and immunogenicity in older children and adults, JAMA 248 (19) (1982) 2478–2480.

- [24] K. Kjeldsen, O. Simonsen, I. Heron, Immunity against diphtheria 25–30 years after primary vaccination in childhood, Lancet 20 (1985) 900–902.
- [25] O. Simonsen, K. Kjeldsen, H.-A. Vendborg, I. Heron, Revaccination of adults against diphtheria I: responses and reactions to different doses of diphtheria toxoid in 30–70-year-old persons with low serum antitoxin levels, Acta Pathologica Microbiologica Scandinavica Series – Immunology 94C (1986) 1–6.
- [26] R. Kadirova, H.U. Kartoglu, P.M. Strebel, Clinical characteristics and management of 676 hospitalized diphtheria cases, Kyrgyz Republic, 1995, Journal Infectious Disease 181 (2000) S110–S115.
- [27] A.M. Galazka, S.E. Robertson, Immunization against diphtheria with special emphasis on immunization of adults, Vaccine 14 (1996) 845–857.
- [28] I.R.B. Hardy, S. Dittmann, R.W. Sutter, Resurgence of diphtheria in the new independent states of the former Sovietic union: current situation and control strategies, Lancet 347 (9017) (1996) 1739–1744.
- [29] B. Christenson, U. Hellstrom, S.P. Sylvan, Impact of a vaccination campaign on adult immunity to diphtheria, Vaccine 19 (9–10) (2000) 1133–1140.
- [30] R. Nandi, P. Purkayastha, A.K. Bhattacharjee, Diphtheria the patch remains, Journal of Laryngology and Otology 117 (2003) 807–810.
- [31] N. Khetsuriani, K. Zakikhany, S. Jabirov, N. Saparova, P. Ursu, K. Wannemuehler, S. Wassilak, A. Efstratiou, R. Martin, Seroepidemiology of diphtheria and tetanus among children and young adults in Tajikistan: nationwide population-based survey 2010, Vaccine 31 (42) (2013) 4917–4922.
- [32] H. Fredlund, T. Norén, T. Lepp, E. Morfeldt, B. Henriques-Normark, A case of diphtheria in Sweden, October 2011, Euro Surveillance 16 (50) (2011) 20038.
- [33] A. Krumina, I. Logina, M. Donaghy, B. Rozentale, I. Kravale, A. Griskevica, L. Viksna, Diphtheria with polyneuropathy in a closed community despite receiving recent booster vaccination, Journal Neurology, Neurosurgery & Psychiatry 76 (2005) 1555–1557.
- [34] C. Ohuabunwo, J. Perevoscikovs, A. Griskevica, P. Gargiullo, A. Brilla, L. Viksna, S. Glismann, M. Wharton, C. Vitek, Respiratory diphtheria among highly vaccinated military trainees in Latvia: improved protection from DT compared with Td booster vaccination, Scandinavian Journal of Infectious Disease 37 (2005) 813–820.
- [35] S. Perkins, R. Cordery, G. Nixon, A. Abrahams, J. Andrews, J. White, A. Efstratiou, S. Anaraki, Investigations and control measures following a non-travel-associated case of toxigenic *Cornyebacterium diphtheriae*, Euro Surveillance 15 (16) (2010) 19544.
- [36] E. Lindhusen-Lindhé, L. Dotevall, M. Berglund, Imported laryngeal and cutaneous diphtheria in tourists returning from western Africa to Sweden, March 2012, Euro Surveillance 17 (23) (2012) 20189.
- [37] S. Handysides, Woman from England acquired diphtheria on a Baltic cruise, Euro Surveillance 1 (7) (2013) 1094.
- [38] R. Karakus, K. Caglar, C. Aybay, Evaluation of human antibody responses to diphtheria toxin subunits A and B in various age groups, Clinical Microbiology and Infection 13 (2007) 1065–1071.
- [39] PAHO, Pan American Health Organization Vaccines: preventing disease and protecting health, PAHO, Washington, D.C., 2003 (Scientific and Technical Publication No. 596) 242, 306.
- [40] J.D. Bonzino, B. Bizzini, Tissue engineering and artificial organs, in: The Biomedical Engineering Handbook, third ed., Taylor & Francis Corp. CRC Press, Boca Raton, FL, 2006.
- [41] MMWR, Updated Recommendations for Use of Tetanus Toxoid, Reduced Diphtheria Toxoid and Acellular Pertussis (Tdap) Vaccine from the Advisory Committee on Immunization Practices, 2010. http://www.cdc.gov/mmwr/preview/mmwrhtml/mm6001a4.htm.

- [42] CDC, Composition of Selected Vaccines with Tetanus Toxoid, Diphtheria Toxoid, and Acellular Pertussis Components Licensed in the United States, 2006. http://www.cdc.gov/mmwr/preview/ mmwrhtml/rr55e223a2.htm.
- [43] S.I. Romaniuk, D.B. Kolibo, S.V. Komisarenko, Perspectives of application of recombinant diphtheria toxin derivatives, Bioorganicheskaia Khimiia 38 (6) (2012) 639–652.
- [44] M. Mort, Vaccine Safety Basics Subunit Vaccine (Purified Antigen), 2013. http://www.vaccinesafety-training.org/subunit-vaccines.html.
- [45] S. Liljeqvist, S. Stahl, Production of recombinant subunit vaccines: protein immunogens, live delivery systems and nucleic acid vaccines, Journal of Biotechnology 73 (1) (1999) 1–33.
- [46] M. Bröker, P. Costantino, L. Detora, E.D. Mcintosh, R. Rappuoli, Biochemical and biological characteristics of cross-reacting material 197 (CRM197), a non-toxic mutant of diphtheria toxin: use as a conjugation protein in vaccines and other potential clinical applications, Biologicals 39 (4) (2011) 195–204.
- [47] M.M. Cox, Recombinant protein vaccines produced in insect cells, Vaccine 30 (10) (2012) 1759–1766.
- [48] Y. Chen, L. Liu, Modern methods for delivery of drugs across the blood-brain barrier, Advanced Drug Delivery Reviews 64 (7) (2012) 640–665.
- [49] S. Abulmagd, M. Emara, S. Aziz, R. El-Domany, Evaluation and characterization of A and B fragments of *Corynebacterium diphtheriae* toxin towards recombinant diphtheria vaccine, Indian Journal of Medical Microbiology 31 (2013) 3–9.
- [50] J.R. Murphy, A.M. Pappenheimer, S.T.D. Borms, Synthesis of diphtheria *tox*-gene products in *Escherichia coli* extracts, Proceedings of the National Academy of Sciences of the United States of America 71 (1) (1974) 11–15.
- [51] D. Leong, K.D. Coleman, J.R. Murphy, Cloned diphtheria toxin fragment A expressed from the tox promoter and exported to the periplasm by the SecA apparatus of *Escherichia coli* K12, The Journal of Biological Chemistry 258 (1983) 15016–15020.
- [52] M. Colombatti, L. Greenfield, R.J. Youle, Cloned fragment of diphtheria toxin linked to T cellspecific antibody identifies regions of B chain active in cell entry, Journal of Biological Chemistry 261 (7) (1986) 3030–3035.
- [53] K. Gtünter-Seeboth, T. Schupp, Cloning and sequence analysis of the *Corynebacterium diphtheriae* dtxR homologue from *Streptomyces lividans* and *S. pilosus* encoding a putative iron repressor, Gene 166 (1995) 117–119.
- [54] N. Johnson, M.A. Pickett, P.J. Watt, I.N. Clarke, J.E. Heckels, Construction of an epitope vector utilizing the diphtheria toxin B-subunit, FEMS Microbiology Letters 146 (1) (1997) 91–96.
- [55] K. Lobeck, P. Drevet, M. LéOnetti, C. Fromen-Romano, F. Ducancel, E. Lajeunesse, C. Lemaire, A. Ménez, Towards a recombinant vaccine against diphtheria, Toxin Infection and Immunity 66 (1998) 418–423.
- [56] C.W. Lee, S.F. Lee, S.A. Halperin, Expression and immunogenicity of a recombinant diphtheria toxin fragment a in *Streptococcus gordonii*, Applied and Environmental Microbiology 70 (8) (2004) 4569–4574.
- [57] R.K. Randall, Biology and molecular epidemiology of diphtheria toxin and the tox gene, The Journal of Infectious Diseases 181 (1) (2000) S156–S167.
- [58] E.N. Miyaji, R.P. Mazzantini, W.O. Dias, A.L. Nascimento, R. Marcovistz, D.S. Matos, I. Raw, N. Winter, B. Gicquel, R. Rappuoli, L.C. Leite, Induction of neutralizing antibodies against diphtheria toxin by priming with recombinant *Mycobacterium bovis* BCG expressing CRM (197), a mutant diphtheria toxin, Infections and Immunity 69 (2001) 869–874.
- [59] Y.Y. Liu, J.H. Woo, D.M. Neville Jr., Targeted introduction of a diphtheria toxin resistant mutation into the chromosomal EF-2 locus of *Pichia pastoris* and expression of immunotoxin in the EF-2 mutants, Protein Expression and Purification 30 (2003) 262–274.

- [60] J.O. Urieto, T.F. Liu, J.H. Black, K.A. Cohen, P.D. Hall, M.C. Willingham, L.K. Pennell, D.E. Hogge, R.J. Kreitman, A.E. Frankel, Expression and purification of the recombinant diphtheria fusion toxin DT<sub>388</sub>IL3 for phase I clinical trials, Protein Expression and Purification 33 (2004) 123–133.
- [61] D.O. Gor, X. Ding, Q. Li, N.S. Greenspan, Genetic fusion of three tandem copies of murine C3d sequences to diphtheria toxin fragment B elicits a decreased fragment B-specific antibody response, Immunology Letters 102 (2006) 38–49.
- [62] M. Aminian, S. Sivam, C.W. Lee, S.A. Halperin, Expression and purification of a trivalent pertussis toxin-diphtheria toxin-tetanus toxin fusion protein in *Escherichia coli*, Protein Expression and Purification 51 (2) (2007) 170–178.
- [63] A.A. Kaberniuk, O.S. Oliĭnyk, D.V. Kolybo, S.V. Komisarenko, Toxin-neutralizing properties of antibodies to diphtheria toxin recombinant subunits A and B and a new method of their estimation, Ukrainskii Biokhimicheskii Zhurnal 81 (3) (2009) 92–101.
- [64] A. Stefan, M. Conti, D. Rubboli, L. Ravagli, E. Presta, Overexpression and purification of the recombinant diphtheria toxin variant CRM197 in *Escherichia coli*, Journal of Biotechnology 156 (4) (2010) 245–252.
- [65] T. Lampidis, L. Barksdale, Park-Williams number 8 strain of *Corynebacterium diphtheriae*, Journal of Bacteriology 105 (1) (1971) 77–85.
- [66] J.H. Mueller, P.A. Miller, The Journal of Immunology 40 (1941) 21-32.
- [67] P.G. Heinemann, Note on the concentration of diphtheria toxin, Journal of Biology and Chemistry 5 (1908) 27–30.
- [68] L. Greenfield, M.J. Bjorn, G. Horn, D. Fong, G.A. Buck, R.J. Collier, D.A. Kaplan, Nucleotide sequence of structural gen for diphtheria toxin carried by corynebacteriophage beta, Proceedings of the National Academy of Sciences of the United States of America 80 (1983) 6853.
- [69] J. Sambrook, D.W. Russel, Molecular Cloning, third ed., vol. 3, Cold Spring Harbor Laboratory Press, New York, 2001.
- [70] K. Terpe, Overview of bacterial expression systems for heterologous protein production: from molecular and biochemical fundamentals to commercial systems, Applied Microbiology and Biotechnology 72 (2006) 211–222.
- [71] A. Gholipour, A. Moosavian, H. Galehdari, M. Makvandi, H.R. Memari, A. Alvandi, Cloning and periplasmic expression of peptidoglycan-associated lipoprotein (PAL) protein of *Legionella pneumophila* in *Escherichia coli*, Jundishapur Journal of Microbiology 3 (2010) 1–9.
- [72] H.J. Choi, K.C. Keum, S.Y. Lee, Review production of recombinant proteins by high cell density culture of *Escherichia coli*, Chemical Engineering Science 61 (2006) 876–885.
- [73] A. Shevchenko, O.N. Jensen, A.V. Podtelejnikov, F. Sagliocco, M. Wilm, O. Vorm, P. Mortensen, A. Shevchenko, H. Boucherie, M. Mann, Linking genome and proteome by mass spectrometry: large-scale identification of yeast proteins from two-dimensional gels, Proceedings of the National Academy of Sciences of the United States of America 93 (25) (1996) 14440–14445.
- [74] W.J.N. Lima, Produção de proteínas recombinantes utilizando *Escherichia coli* em cultivos em alta densidade celular (PhD thesis), Universidade Estadual de Campinas, Campinas/SP, 2004.
- [75] M.M. Bradford, A rapid and sensitive method for the quantitation of microgram quantities of protein utilizing the principle of protein-dye binding, Analytical Biochemistry 72 (1976) 248–254.
- [76] D.V. Nascimento, E.M.B. Lemes, J.L.S. Queiroz, J.G. Silva Jr., H.J. Nascimento, E.D. Silva, R. Hirata Jr., A.A.S.O. Dias, C.S. Santos, G.M.B. Pereira, A.L. Mattos-Guaraldi, G.R.G. Armoa, Expression and purification of the immunogenically active fragment B of the Park Williams 8 *Corynebacterium diphtheriae* strain toxin, Brazilian Journal of Medical and Biological Research 43 (5) (2010) 460–466.

- [77] D. Yoon, D.S.H. Chu, C.W. Ng, E.A. Pham, A.B. Mason, D.M. Hudson, V.C. Smith, R.T. Macgillivray, D.T. Kamei, Genetically engineering transferrin to improve its in vitro ability to deliver cytotoxins, Journal of Controlled Release 133 (3) (2009) 178–184.
- [78] A. Weber, C. Zimmermann, G.M. Zu Hörste, B.C. Kieseier, H.P. Hartung, H.H. Hofstetter, Bacterial flagellin and diphtheria toxin co-stimulate IL-17-producing thymocytes, Cytokine 64 (2013) 221–226.
- [79] J.S. Peraino, M. Schenk, G. Li, H. Zhang, E.A. Farkash, D.H. Sachs, C.A. Huang, R. Duran-Struuck, Z. Wang, Development of a diphtheria toxin-based recombinant porcine IL-2 fusion toxin for depleting porcine CD25+ cells, Journal of Immunological Methods 15 (2013) 398–399.

# 20

# Veterinary Rabies Vaccine

R.K. Sanson<sup>1</sup>, A.S. Zeferino<sup>1</sup>, R. Nakamura<sup>1</sup>, V. Thomaz-Soccol<sup>2</sup>, C.R. Soccol<sup>2</sup>, \*

<sup>1</sup>INSTITUTO DE TECNOLOGIA DO PARANÁ, CURITIBA, PARANÁ, BRAZIL; <sup>2</sup>UNIVERSIDADE FEDERAL DO PARANÁ (UFPR), CURITIBA, PARANÁ, BRAZIL

# 20.1 Introduction

Rabies is acute encephalitis or meningoencephalitis resulting from to a lyssavirus infection [1]. It is known to be present on all continents except Antarctica. All warmblooded animals are susceptible to rabies. It enters the body through wounds or by direct contact with mucosal surfaces. Rabies virus replicates in the bitten muscle and gains access to motor end plates and motor axons to reach the central nervous system [2,3]. Most deaths occur in the absence of postexposure prophylaxis.

In more than 99% of all human rabies cases, the virus is transmitted from dogs. Half of the global human population lives in canine rabies-endemic areas [4]. The most cost-effective strategy for preventing rabies in people is to eliminate it in dogs.

Canine rabies can be eliminated, as it has been demonstrated in North America, Western Europe, Japan, and many areas in South America. However, canine rabies is still widespread; it occurs in over 80 countries and territories, which are predominantly in the developing world. Mass parenteral vaccination programs remain the mainstay of canine rabies control. Coverage of 70% of dog populations has been sufficient to control canine rabies in several settings [4].

Pasteur developed the first rabies vaccine in 1885. It was composed of crude suspensions of desiccated, infected brain and spinal cord of rabbit, sheep, or goat brain. Occasionally Pasteur's vaccine led to vaccine-induced rabies [5]. Viral inactivation methods using phenol or betapropiolactone had been developed, increasing safety in vaccination; however, nerve-tissue vaccines enhance allergic neuroparalytic accidents caused by encephalitogen associated with myelin in adult mammalian brain. These reactions range from transient to severe paralysis, in some cases resulting in death [5].

A significant reduction in neuroallergic side effects was obtained with vaccine prepared in suckling mouse brain and inactivated with ultraviolet light or betapropiolactone

<sup>&</sup>lt;sup>\*</sup>Corresponding Author.

[6]. It is imperative in the case of mouse suckling that they be under 9 days of age at the time of harvest [5].

After the 1950s, establishment of cell culture technology allowed significant progress in the propagation of virus in vitro and consequent development of viral vaccines [7]. Rabies vaccines developed and produced by cell culture technology have been demonstrated to be safer and more immunogenic than nerve-tissue vaccines [8].

Since 1984, World Health Organization has recommended discontinuation of the production and use of nerve-tissue vaccines and their replacement for cell culture vaccines [1]. In Brazil, nerve-tissue vaccine was replaced by cell culture vaccine in 2002 [1] for human public vaccination, and in 2010 for dog and cat mass vaccination.

Industrial production of viral vaccines preferably requires the use of continuous or immortalized cell lines as the basis for viral multiplication [7]. For human rabies vaccine production, the Vero cell line is the most widely used. For veterinary vaccine production, the cell lines used are baby hamster kidney cell line (BHK-21) and embryo hamster cell line [9–12].

The BHK-21 cell line was isolated from five 1-day-old hamsters [11] and consists of adherent fibroblasts that can also be adapted to suspension culture [13]. Suspension culture is widely used for veterinary vaccine production, owing to the lower costs of production and large cell quantities attained without the need of increasing superficial areas [13].

Effective in vitro maintenance and growth of animal cells requires culture conditions similar to those found in vivo with respect to temperature, oxygen concentration, pH, osmolality, and nutrients [14].

Culture medium composition is one of the most important factors in the culture of animal cells. Its functions are to provide appropriate pH and osmolality for cell survival and multiplication, as well as to supply all chemical substances required by the cells that they are unable to synthesize themselves [14].

Some of these substances can be provided by a culture medium consisting of low– molecular weight compounds, known as basal media. However, most basal media fail to promote successful cell growth by themselves and require supplementation with more complex and chemically undefined additives such as blood serum [14].

Fetal bovine serum (FBS) is the most frequently employed serum in animal cell culture. However calves, horses, and even human sera are also used [14]. FBS is the favorite because it has low immunoglobulin and high growth factor concentrations. However it is one of the most expensive components in culture medium [14]. Use of serum in cell culture is important because of nutritional factors, culture-stimulating factors (growth factors, hormones, and proteins), and protecting agents, both for biological protection (antitoxin, antioxidant, and antiprotease) and to prevent mechanical damage [14].

However using animal serum is potential risky for health, owing to the possible presence of adventitious agents such as virus and prions. In addition, serum can contain contaminants such as bacteria, fungi, and mycoplasmas, which can negatively affect cell culture. Another strong limitation to the use of sera is variability among different lots and suppliers, which hinders standardization of the culture medium and the reproducibility of culture performance [14].

Furthermore, studies have demonstrated that bovine serum is the major reason for allergic reactions in dogs [15]. A research was designed to investigate the relationship between immunoglobulin E (IgE) reactivity to the vaccines and immediate-type allergic reactions after vaccination in dogs. Sera from 10 dogs that developed immediate-type allergic reactions such as circulatory collapse, cyanosis, dyspnea, facial edema, and vomiting within 1 h after vaccination with rabies vaccines and sera from 50 dogs that did not develop allergic reactions after vaccination were collected. Then, IgE reactivity to FBS and stabilizer proteins included in the vaccines was measured in sera that had high levels of IgE to the vaccines. Of the eight dogs that developed immediate-type allergic reactions and had high levels of serum-specific IgE to the vaccines, seven had specific IgE directed to FBS [15].

Later, researchers investigated IgE-reactive components of FBS using sera from 16 dogs that exhibited allergic reactions after vaccination. Immunoblot analysis revealed that several FBS proteins strongly reacted with IgE in the sera from dogs that showed postvaccination allergic reactions. The 66-kDa band was detected in the sera from 14 of the 16 dog serum samples analyzed in the immunoblot analysis for FBS, and it was speculated to be albumin based on its molecular weight; however, serum IgE reactivity to bovine serum albumin could be detected in only four of the 14 dog samples. These findings demonstrated that a variety of FBS components including albumin could function as allergens in dogs that developed allergic reactions after vaccination [16].

Many attempts have been made to develop culture media that do not need serum supplementation: that is, serum-free media (SFM) formulations. One of the greatest difficulties is to design a culture medium that ensures replicate cultures that show stable genetic, metabolic, and kinetic behavior [14].

Despite all of the developments seen, currently no standard culture medium can be used universally because each cell line has a specific metabolic profile and requires an appropriate medium composition [14].

For instance, Cellvento BHK-200 medium is an SFM formulated with no animalderived component and optimized for the culture of suspension BHK-21 cells at high density, viability, and efficient propagation of viruses. The medium formulation was developed for the growth and maintenance of BHK-21 suspension cell lines used for viral vaccine production, and qualified for the production of foot and mouth disease virus (FMDV) [17].

Ex Cell 302 is an SFM specifically developed for the long-term growth of transformed Chinese hamster ovary (CHO) cells in suspension for the expression of antibodies or protein products. It was developed using only recombinant human proteins that have molecular weights less than 10 kDa [18].

VP-SFM is an SFM expressly designed for virus production, manufactured with no components of animal or human origin. VP-SFM is an ultra-low protein medium designed specifically for the culture of Vero cells. It is also appropriate for the growth of

COS-7 [African green monkey kidney cells (CV1 cells transformed with a defective mutant of SV40)], Madin–Darby canine kidney cell line, and BHK-21 in suspension. It is particularly suitable for growing viruses [19].

When using a new medium, the need for cell adaptation should be considered. The assessment of cellular performance in different culture media is usually challenging and time-consuming, because it requires the establishment of the concentration of total and viable cells over multiple subcultures. In addition, it may involve the quantification of residual nutrients, such as glucose and glutamine, of metabolic by-products such as lactate, ammonium, and alanine, and the concentration of the target product [14].

Cell adaptation is often carried out in static cultures employing tissue culture flasks. However problems seen at the time of adaptation can be minimized in suspension cultures, because this allows subculture during the log growth phase. Cultures in suspension also allow better medium oxygenation [14] and homogeneity.

Mammalian cells can be adapted to serum-free conditions by direct adaptation from serum-containing media or gradual weaning. Both procedures require healthy, viable cultures in the midlogarithmic growth phase. During the adaptation phase, growth rates will usually be somewhat slower than growth in media supplemented with serum.

Perrin et al. [20] adapted BHK-21 cells to grow in SFM developed by Merten et al. [21], for the production of an experimental rabies vaccine. Kallel and colleagues [22] also adapted the BHK-21 cell line to grow in different commercial SFM and evaluated the rabies vaccine produced. In both cases, adaptation was carried out with cells growing in monolayer.

This chapter presents a case study of a process of rabies virus production to a veterinary rabies vaccine, using the BHK-21 cell line in suspension adapted to commercial SFM. To achieve this technological development, the following strategies were used:

- Adapt the BHK-21 cell line to grow in different commercial SFM.
- Analyze the growth kinetics of adapted cells.
- Infect cells with rabies virus and analyze the behavior of infected cells.
- Verify the virus production capacity of SFM-adapted cells.
- Compare the results of adapted cells.

# 20.2 Case Study

# 20.2.1 Adaptation of Baby Hamster Kidney Cell Line-21 to Serum-Free Media

#### 20.2.1.1 Cell Line

The BHK-21C13 LVI cell line, adapted to suspension, in passage 118 was used. This cell bank was coming from Panaftosa (the Pan American Centre for Foot-and-Mouth Disease) with 84 passages. Cells were adapted to grow at Dulbecco Modified Eagle's medium (DMEM)/Ham's F12 1:1 supplemented with 3% of FBS to generate the master and working cell bank. Cells were stored at  $-196^{\circ}$ C.

#### 20.2.1.2 Culture Media

Standard culture medium, composed of DMEM/Ham's F12 1:1 supplemented with 3% FBS, was used for control cell culture.

The following serum free media were used: VP-SFM (Gibco–Invitrogen cell culture), Ex Cell 302 (SAFC Biosciences), and Cellvento BHK-200 (Merck Millipore). Media were prepared following manufacturers' instructions. No antibiotic was added.

Cryopreservation culture media were prepared using 7% dimethyl sulfoxide, 46.5% SFM, and 46.5% media supernatant of centrifuged cells.

#### 20.2.1.3 Strategies for Cell Adaptation to Serum-Free Media

Three different cell adaptation protocols were used.

#### 20.2.1.3.1 DIRECT MEDIUM CHANGE

For direct medium change (D), the cell line growing healthily in standard culture media, with a cell density higher than  $7.5 \times 10^5$  cells/mL, was centrifuged and resuspended in SFM at a concentration of  $2.0 \times 10^5$  cells/mL. Subcultures were made until cell adaptation.

#### 20.2.1.3.2 GRADUAL MEDIUM CHANGE

For gradual medium change (G), the cell line growing healthily in standard culture media, with a cell density higher than  $7.5 \times 10^5$  cells/mL, had SFM added in culture gradually during subculture. An increased quantity of SFM was added only when cells were growing healthily in a previous adaptation phase. In the last adaptation step, cells were centrifuged to remove all serum from cultivation and resuspended in SFM.

#### 20.2.1.3.3 GRADUAL SERUM REDUCTION (S)

For gradual serum reduction (S), the cell line growing healthily in standard culture media, with a cell density higher than  $7.5 \times 10^5$  cells/mL, was centrifuged and resuspended in SFM, at a concentration of  $2.0 \times 10^5$  cells/mL. SFM was supplemented with 3% FBS. Serum was decreased from culture gradually during subcultures. Serum was reduced only when cells had normal growth at a previous adaptation phase. In the last adaptation step, cells were centrifuged to remove all serum from cultivation and resuspended in SFM (Fig. 20.1).

During adaptation, a control culture in DMEM/Ham's F12 1:1 supplemented with 3% FBS (standard culture media) was held to compare growth behavior.

A direct media change protocol was used for adaptation of three serum-free culture media used in this development. The protocol of gradual media change was used to adapt cells to VP-SFM and Ex Cell 302 SFM. The gradual serum reduction protocol was used to adapt cells to Cellvento BHK-200 SFM. The choice of protocol followed the manufacturer's recommendations. Table 20.1 presents strategies used for cell adaptation.

#### 20.2.1.4 Cell Cultivation

Cell cultivation was performed in spinner flasks (Techne, United Kingdom) of 500 mL working volume, with 400 mL culture medium, incubated at 37°C and agitated at 55 rpm.



FIGURE 20.1 Protocols used for cell adaptation to serum free media. SCM, standard culture medium composed of DMEM/Ham's F12 1:1 supplemented with 3% of fetal bovine serum.

	Serum-Free Media				
Adaptation Strategy	VP-Serum-Free Media	Ex Cell 302	Cellvento Baby Hamster Kidney Cell Line-200		
Direct media change (D)	X	X	X		
Gradual media change (G) Gradual serum reduction (S)	Х	Х	х		

Table 20.1Strategies Used for Baby Hamster Kidney Cell Line-21 Cell LineAdaptation to Serum-Free Media

Inoculum was made with a cell density of  $2.0 \times 10^5$  cells/mL. Subcultures (cells dilution) were made when growth reached  $7.5 \times 10^5$  cell/mL density or higher.

Sometimes, during adaptation to SFM, even when cells did not reach  $7.5 \times 10^5$  cells/mL, a subculture was done, usually to try to enhance growth conditions. In addition, if cells could not reach a cell density higher than  $7.5 \times 10^5$  cells/mL, inoculum for subculture was increased to  $5 \times 10^5$  cells/mL, to try to make cells reach a higher density.

Samples were collected daily for cell count, viability determination, morphology verification, metabolite analysis, and pH measurement.

#### 20.2.1.5 Cell Count

Cell samples were diluted and stained with trypan blue (0.5%), then counted in a hemocytometer. Viable cells have an intact membrane and are not colored by trypan blue. Nonviable cells permit trypan blue to traverse the membrane and are colored by dye. Total cells, viable cells, and cell viability were determined. Morphology, light refringence, and size were evaluated.

#### 20.2.1.6 Metabolism Analysis

Glucose consumption and lactate production were determinate to verify growth limitation by inhibitory levels of lactate or by glucose scarcity.

A dual-channel biochemical analyzer YSI 2700 (Yellow Springs Instruments, EE.UU.) was used for glucose and lactate determination. Each channel contains immobilized enzyme on a polycarbonate membrane. One membrane has enzyme glucose oxidase immobilized and catalyzes the following reaction:

$$Glucose + O_2 + H_2O \rightarrow gluconic acid + H_2O_2$$
 [i]

In other membrane, the enzyme lactate oxidase catalyzes reaction [ii]:

Lactate 
$$+ O_2 \rightarrow piruvate + H_2O_2$$
 [ii]

Hydrogen peroxide generated independently in each membrane is oxidized in a platinum anode. Electron flow is linearly proportional to the concentration of hydrogen peroxide formed, and this, to the concentration of the substance in the sample. A 2% variation was considered acceptable between successive calibrations.

#### 20.2.1.7 Sedimentation Tests

Cell lines adapted to SFM were submitted to sedimentation tests. They consisted of taking cell cultures, with a cell density over than  $7.5 \times 10^5$  cells/mL, in a cold chamber (2–8°C) for 2 or 3 h until complete cell sedimentation. Culture media was carefully removed from the spinner (about 200 mL) and fresh media were added; then cells were incubated at 37°C and agitated at 55 rpm. This procedure was repeated for 3 consecutive days. Every day, samples were collected for cell count, viability, morphology, and metabolic analysis. To be approved in this test, cells had to keep their growth behavior and not grow in clumps.

A sedimentation test was performed to simulate industrial rabies virus production in which cells had to be settled for virus infection. Therefore, adapted cells had be robust enough for this operation.

#### 20.2.1.8 Cryopreservation

Seed banks were made with cell lines approved in sedimentation tests. Cryotubes with  $1 \times 10^{\circ}7$  cells and 1 mL cryopreservation media were frozen in a  $-80^{\circ}C$  freezer. A gradual temperature reduction rate of  $-1^{\circ}C/min$  was achieved using Mr. Frosty (Nalgene) and isopropyl alcohol. After they reached  $-80^{\circ}C$ , cells were stored in liquid nitrogen ( $-196^{\circ}C$ ).

#### 20.2.1.9 Growth Kinetics

Growth kinetics was carried out with all serum-free adapted cell lines and control culture, growing in standard culture media. Cell cultivation was performed with 400 mL culture medium at 37°C, in spinner flasks of 500 mL working volume (Techne, United Kingdom), agitated at 55 rpm. Inoculum was made at a cell density of  $2.5 \times 10^5$ . No subculture was done. Daily samples were collected for cell count, viability determination, morphology verification, metabolite analysis, and pH measurement. Cultivation was carried out until cell death.

Specific growth rate,  $\mu$  (cells/h), was estimated by Eq. [20.1]:

$$\mu = (LnX_2 - LnX_1)/(t_2 - t_1)$$
[20.1]

where *X* represents the viable cell density (mL), *t* represents the time points of sampling (h); and subscripts 1 and 2 stand for two succeeding sampling points.

#### 20.2.1.10 Adaptation to VP-Serum Free Media From Standard Culture Media 20.2.1.10.1 CONTROL CULTURE

Control cultures had standard behavior during subcultures. Cell morphology, size, and refringence were normal. Cell density reached  $7.5 \times 10^5$ /mL or higher after 2 days of cultivation; cell viability was higher than 80%, and usually cells doubled or tripled every day. Glucose consumption, lactate production, and pH values were similar during subcultures.

The growth kinetics of BHK-21 cells in standard culture media reached a maximal viable cell density of  $8.1 \times 10^5$  cells/mL on the fourth day of cultivation. The glucose concentration was not limited to growth nor did lactate concentration reach inhibitory levels, according to what Cruz et al. [23] described. Cultivation lasted 6 days, in which there was no lag phase, 2 days of exponential phase, 3 days of stationary phase, and 1 day of death phase. Maximal specific growth rate obtained was 0.036 h<sup>-1</sup>. Because glucose and lactate were not limited to growth, cells probably stopped growing because of ammonia production or the absence of another nutrient, such as glutamine.

#### 20.2.1.10.2 DIRECT MEDIUM CHANGE

In adaptation using direct media change from standard culture media to VP-SFM (VPD), over 18 days and 10 subcultures, cells had abnormal morphology, size, and growth. In some subcultures, cells grew in clumps. There was much cell debris in the medium and viability was very low, reaching 18% on the fourth adaption day. In this period, between some subcultures, cells were centrifuged and half of the culture media was replaced to remove inhibitory metabolites, adjust pH, and supply nutrients. In the 11th subculture, cells started to have behavior similar to that of control cells, but in next subculture (12th), cells formed aggregates, so three more subcultures were carried out to confirm cell behavior. Because the cells maintained their behavior and clumps no longer formed, they were submitted to a sedimentation test. During the test, no clump formed and the culture reached a density of  $1.8 \times 10^6$  cells/mL. Cells were considered adapted to VP-SFM culture media and were robust enough for virus production. A seed bank was made.

Adaptation to VP-SFM took 26 days and 15 passages.

The growth kinetics of cells by VPD had 1 day of lag phase, 1 day of log phase, 1 day of stationary phase, and 2 days of death phase. The maximal cell density,  $8.2 \times 10^5$  viable cells/mL, was obtained on the third day of cultivation. Cultivation lasted 6 days, just like

	Maximal Cell Density (10 <sup>6</sup> cell/mL)	Maximal Specific Growth Rate (h <sup>-1</sup> )	Time of Cultivation (day)
Control	0.81	0.036	6
VPD	0.82	0.052	6
EXD	2.09	0.113	13
EXG	0.95	0.030	17
CVTDvp	1.03	0.058	8
CVTGvp	1.09	0.041	8

**Table 20.2**Growth of Baby Hamster Kidney Cell Line-21 Cells in Spinner inDifferent Serum-Free Media and Different Adaptation Strategies

*CVTDvp*, Adaptation to Cellvento BHK 200 media by direct media change using cells previously adapted to VP-serumfree media (SFM); *CVTGvp*, adaptation to Cellvento BHK 200 media by gradual media change using cells previously adapted to VP-SFM; *EXD*, adaptation to Ex Cell 302 media by direct media change; *EXG*, adaptation to Ex Cell 302 media by gradual media change; *VPD*, adaptation to VP-SFM media by direct media change.

the control culture. Growth was not limited by the glucose or lactate concentration [23]. Because the log phase was more acute, maximal specific growth rate obtained at  $0.052 \text{ h}^{-1}$  was higher than the control culture. Cells lasted less time in the stationary phase and cell death was more gradual than in the control culture. The maximal cell density was lowest among serum-free adapted cells (Table 20.2).

#### 20.2.1.10.3 GRADUAL MEDIUM CHANGE

In adaptation using gradual media change from standard culture media to VP-SFM, the first step of media change was 75% standard culture medium and 25% VP-SFM. No alteration in cell behavior or metabolites was observed. After three subcultures, a new step of media substitution was done, with 50% of standard culture medium and 50% of VP-SFM. No alteration in cell behavior or metabolites was observed. The next step was 25% of standard culture medium and 75% of VP-SFM. In this step, the cell viability and cell growth decreased during the third and fourth subcultures, but in the next passage, cell growth behavior became normal and cell viability improved in the sixth subculture. Therefore the next media change step was done: 0% of standard culture medium and 100% of VP-SFM. In this adaptation stage, cell morphology was abnormal, clumps formed, and much cell debris was observed in the media. In some subcultures, cells were centrifuged and half of the culture media was replaced to remove inhibitory metabolites, adjust the pH, and supply nutrients. However, no improvement in cell morphology was observed, and glucose consumption and lactate production were higher than during previous adaptation steps and control cells. Because adaptation by direct media change was reached, this adaptation procedure was stopped and cells were not considered to be adapted.

Adaptation procedure took 49 days and was unsuccessful.

#### 20.2.1.11 Adaptation to Ex Cell 302 From Standard Culture Media 20.2.1.11.1 CONTROL CULTURE

Control culture had standard behavior during subcultures. Cell morphology, size, and refringence were normal. Cell density reached  $7.5 \times 10^5$  or higher after 2 days of

cultivation, cell viability was higher than 80%, and cells doubled or tripled every day. Glucose consumption, lactate production, and pH values were similar during subcultures.

#### 20.2.1.11.2 DIRECT MEDIUM CHANGE

In adaptation using direct media change from standard culture medium to Ex Cell 302 EXD, over 26 days and 10 subcultures, cells had an abnormal morphology, size, and growth. There was much cell debris in the medium but cell viability did not decrease as much as in adaptation to VP-SFM by direct media change. The lowest viability value was 72%. In this period, between some subcultures, cells were centrifuged and half of the culture media was replaced. In the 11th subculture, cells started to have growth and metabolite behavior similar to the control culture. Three more subcultures were done to confirm cell behavior. Because cells maintained their behavior, they were submitted to a sedimentation test. During the test, no clump formed and the culture reached a density of  $1.2 \times 10^6$  cells/mL. Cells were considered to be adapted to the Ex Cell 302 culture media and were robust enough for virus production. A seed bank was made.

Adaptation to Ex Cell 302 by direct media change took 36 days and 14 passages.

The growth kinetics of cells adapted to Ex Cell 302 SFM by direct media change EXD had atypical behavior. On the first day of cultivation, the cell density was  $1.95 \times 10^6$  viable cells/mL, about 10 times higher than the inoculum. The maximal specific growth rate was  $0.113 \text{ h}^{-1}$ , twice as high as that obtained for other kinetics (Table 20.2). The maximal cell density,  $2.09 \times 10^6$ , was obtained on the third cultivation day. This cultivation was the one with the best energetic efficiency, because with a similar glucose concentration available, a higher cell density and lower lactate production were obtained compared with other growth kinetics. Glucose scarcity was obtained on the fifth day, but cultivation lasted for 13 days, which suggests that another energy source, probably glutamine, was used during the stationary phase. The death phase lasted only 1 day. Kallel et al. [22] also adapted BHK-21 cells to Ex Cell 302 SFM, but the maximal cell density obtained was  $7 \times 10^5$  cells/mL and the maximal specific growth rate was  $0.016 \text{ h}^{-1}$ .

#### 20.2.1.11.3 GRADUAL MEDIUM CHANGE

In an adaptation using gradual media change from a standard culture medium to Ex Cell 302 EXG, the first step of media change was 75% of the standard culture medium and 25% of Ex Cell 302. No alteration in cell growth behavior or metabolism was observed, but the cell morphology and size were abnormal and the media had much cell debris. However, after four subcultures, a new step of media substitution was performed: 50% of standard culture medium and 50% of Ex Cell 302. No alteration in cell behavior or metabolites was observed. The cell morphology and size became normal and cell debris disappeared. The next step was 25% of standard culture medium and 75% of Ex Cell 302. In this step, four subcultures were done. Cell growth behavior was normal and glucose consumption was higher, which could have resulted from the higher cell density. Therefore the next media change step was done: 0% of standard culture medium and

100% of Ex Cell 302. In this adaptation stage, no problems were observed. Morphology, growth, and metabolites had the same behavior as for control cells. Thus the cells were submitted to a sedimentation test. No clumps formed and the culture reached a density of  $1.3 \times 10^6$  cells/mL. The cells were considered to be adapted to the Ex Cell 302 culture media and robust enough for virus production. A seed bank was made.

The growth kinetics of cells adapted to Ex Cell 302 SFM by gradual media change EXg reached maximal cell density,  $9.5 \times 10^5$  viable cells/mL, on the 13th day. Cell growth was the slowest, with maximal specific growth of  $0.030 \text{ h}^{-1}$ . However it had the longest cultivation time, lasting 17 days. This cultivation had 1 day of lag phase, 2 days of exponential phase, 12 days of maintenance phase, and 2 days of senescence phase. Cultivation reached the lowest pH value at the death phase, 6.01.

Adaptation to Ex Cell 302 by gradual media change took 36 days and 18 passages. Adaptation to Ex Cell 302 by gradual media change took the same time as adaptation to Ex Cell 302 by direct media change, but more subcultures were done. This was because in adaptation to Ex Cell 302 by direct media change, cells took longer to reach a cell density higher than  $7.5 \times 10^5$ .

#### 20.2.1.12 Adaptation to BHK From Standard Culture Media

#### 20.2.1.12.1 CONTROL CULTURE

The control culture had standard behavior during subcultures. The cell morphology, size, and refringence were normal. The cell density reached  $7.5 \times 10^5$  or higher after 2 days of cultivation. The cell viability was higher than 80% and cells doubled or tripled every day. In some subcultures, cells quadrupled in 24 h. Glucose consumption, lactate production, and pH values were similar during subcultures.

#### 20.2.1.12.2 DIRECT MEDIUM CHANGE

In adaptation using direct media change from standard culture medium to Cellvento BHK-200, over 50 days and 15 subcultures, cells had an abnormal morphology, size, and growth. In some subcultures, cells grew in clumps. There was much cell debris in the medium. In this period, between some subcultures, cells were centrifuged and half of the culture media was replaced to remove inhibitory metabolites, adjust the pH, and supply nutrients. However in no subculture did cells reached a density higher than  $5.8 \times 10^5$  cells/mL. No enhanced cell morphology or growth was observed with time, so this adaptation work was stopped without cell adaptation.

#### 20.2.1.12.3 GRADUAL SERUM REDUCTION

In an adaptation using a gradual serum reduction from standard culture medium to Cellvento BHK-200, the first step was to centrifuge cells from standard culture medium and resuspend cells in Cellvento BHK-200 with 3% FBS. In this step of adaptation, cells had normal morphology, growth, and metabolism. Four subcultures were performed. The first step of serum reduction was done (reduction to 1.5%). Cells also had a normal morphology, growth, and metabolism. Therefore after three subcultures, a new adaptation step was done: Cellvento BHK-200 supplemented with 0.75% FBS. This step

presented no problem with cell morphology, cell growth, or metabolism. The next step of serum reduction was carried out (Cellvento BHK-200 with 0.5% FBS) and presented no cultivation problems. Cell growth, morphology, and metabolites were similar to those of the control.

In next step, Cellvento BHK-200 supplemented with 0.25% FBS, cells started to show abnormal growth. In some subcultures, even with media replacement, they could not reach a density of  $7.5 \times 10^5$  cell/mL. The cell inoculum was increased to  $5.0 \times 10^5$  cells/mL but cells could not double. The cell morphology became irregular; the media had much cell debris and different-sized cells in culture.

After 41 passages and 116 days trying to adapt cells, work was stopped without cell adaptation.

Because Cellvento BHK-200 is a culture medium developed for use in veterinary vaccine production, its price is about 10 times cheaper than other SFM used in this technological development. Adaptation of BHK-21 with this medium has an important economic perspective. Therefore, failure in adaptation using previous strategies led to attempts to try a new strategy. In the new strategy, cells already adapted to SFM, VP-SFM, were used to adapt to Cellvento BHK-200. Two adaptation protocols were used: direct media change and gradual media change.

#### 20.2.1.13 Adaptation to BHK From VP-Serum-Free Media

#### 20.2.1.13.1 CONTROL CULTURE

The control culture demonstrated standard behavior during the subcultures. The cell morphology, size, and refringence were normal. The cell density reached  $7.5 \times 10^5$  or higher after 2 days of cultivation, the cell viability was higher than 80%, and cells doubled or tripled every day. In some subcultures, cells quadrupled. Glucose consumption, lactate production, and pH values were similar during subcultures.

#### 20.2.1.13.2 DIRECT MEDIUM CHANGE

In the adaptation using direct media change from VP-SFM to Cellvento BHK-200 (CVTDvp), the first two passages had a lot of cell debris, but the cell morphology, size, and refringence remained normal. During subcultures, cells could not reach a density higher than  $7.5 \times 10^5$  cells/mL, so in some subcultures inoculums with  $5.0 \times 10^5$  cells/mL were used to make cells double and reach  $1.0 \times 10^6$  cells/mL. In the 13th subculture, the cell density was  $1.0 \times 10^6$  cells/mL, from an inoculum of  $2.0 \times 10^5$  cells/mL. For the next two subcultures, the cells maintained their behavior and so were submitted to a sedimentation test. During the test, no clumps formed and the culture reached a density of  $2.1 \times 10^6$  cells/mL.

After 42 days and 15 passages, cells were adapted to Cellvento BHK-200 culture medium and were robust enough for the virus production process.

The growth kinetics of cells adapted to Cellvento BHK-200 media by direct media change (CVTDvp) took 1 day of lag phase, 3 days of log phase, no days in stationary

phase, and 4 days of death phase. The maximal cell density,  $1.03 \times 10^6$  viable cells/mL, was obtained on the fourth day of cultivation. Cultivation lasted 8 days, and growth was limited neither by glucose nor by lactate concentration [23]. The maximal specific growth rate obtained was 0.058 h<sup>-1</sup>.

#### 20.2.1.13.3 GRADUAL MEDIUM CHANGE

In the adaptation using gradual media change from VP-SFM to Cellvento BHK-200 (CVTGvp), the first step of media change was 75% VP-SFM and 25% Cellvento BHK-200. No alteration in cell growth behavior or metabolites was observed, but in the third passage cells grew in clumps. However a new step of media substitution was done: 50% VP-SFM and 50% Cellvento BHK-200. No alterations in cell behavior or metabolites were observed. The next step was 25% VP-SFM and 75% Cellvento BHK-200. In this step, cells showed standard behavior. Therefore the next media change step was done, but because adaptation to Cellvento BHK-200 failed in prior adaptation work, a more gradual media change was made in this adaptation process: 15% VP-SFM and 85% Cellvento BHK-200. In this adaptation stage, no problem was observed. A new adaptation step was performed in which 5% VP-SFM and 95% Cellvento BHK-200 were used. In this step, the subculture had a density of  $1.1 \times 10^6$  cells/mL and a viability of 99%, so the last step was done with 0% VP-SFM and 100% Cellvento BHK-200. Three subcultures were carried out and no abnormal situation was detected. Cells were submitted to the sedimentation test. During the test, no clumps formed and the culture reached a density of  $1.9 \times 10^6$  cells/mL. Cells were considered to be adapted to Cellvento BHK-200 culture media and robust enough for virus production. A seed bank was made.

The growth kinetics of cells by CVTGvp presented behavior similar to that of cells adapted to Cellvento BHK-200 by direct media change: 1 day of lag phase, 2 days of log phase, no day in stationary phase, and 5 days of death phase. Maximal cell density,  $1.09 \times 10^6$  viable cells/mL, was obtained on the third day of cultivation. Cultivation lasted 8 days and growth was limited neither by glucose nor by lactate concentration. The maximal specific growth rate obtained was 0.041 h<sup>-1</sup>.

Adaptation to Cellvento BHK-200 by gradual media change from cells previously adapted to VP-SFM took 42 days and 20 passages. Adaptation to Cellvento BHK-200 by gradual media change took the same time as adaptation to Cellvento BHK-200 by direct media change, but more subcultures were done. This was because in adaptation to Cellvento BHK-200 by direct media change, cells took longer to reach cell density higher than  $7.5 \times 10^5$  cells/mL.

It was expected that in direct media change cells would adapt to new media without the need for subcultures. This result was not found in any culture media used. At least 14 passages for Ex Cell 302 and 15 passages for VP-SFM were needed for cells to adapt.

H. Kallel et al. [22] adapted the BHK-21 cell line to three commercial SFM, HyQ PF CHO, HyQ PF CHO MPS, and Ex Cell 302, in a single step by direct switch from serumcontaining media, and adapted BHK-21 to Rencyte by gradual media change. In gradual adaptation, 26 days were necessary for adaptation. In this study, cells grew in a monolayer. After adaptation, cells no longer adhered but grew in suspension-forming clumps.

Merten et al. [21] adapted BHK-21 cells to SFM MD SS2 using same protocol as Kallel and coworkers [22]; cells also grew as clump cultures. In both studies, clump cultures were able to produce rabies virus. However in an aggregated culture, it is not possible to have homogeneity. Cells in the middle of aggregates have difficulty accessing nutrients and it is more difficult to have a virus infection. Besides, cells in clumps do not grow as healthy as do cells free in suspension. Thus cells in suspension adapted to SFM without forming clumps are a great possibility for industrial use.

A summary of results of cell adaptation to SFM is shown in Table 20.3. Considering that 26 days and 15 passages were needed to adapt cells to VP-SFM, the actual value of passages to adapt BHK-21 cells to Cellvento BHK-200 culture media was 30 in direct media change and 35 in gradual media change, and it took 68 days.

Fig. 20.2 shows a comparison among kinetics studies. The cultivation of cells adapted to Ex Cell 302 by direct media change EXd obtained a twice-higher cell density and cultivation lasted twice as long as the control cultivation. The cultivation of cells adapted to Ex Cell 302 by gradual media change EXG lasted three times longer than the control culture with a similar maximal cell density. Although cells adapted to Cellvento BHK-200 (CVTDvp and CVTGvp) did not have a stationary phase, the cell density was higher than that of the controls on the third through sixth cultivation days. Cells adapted to VP-SFM (VPD) had maximal cell density similar to control culture, but cell death started earlier than for controls.

In this technological development, it is possible to affirm that BHK-21 cells are better adapted to Ex Cell 302 serum-free culture media, and direct media change adaptation led to better cell adaptation.

	Media Exchange					
Adaptation strategy	$SCM \to VP\text{-}SFM$	SCM $\rightarrow$ Ex Cell 302	SCM $\rightarrow$ Cellvento BHK-200	VP-SFM $\rightarrow$ Cellvento BHK-200		
Direct media change (D)	26 days 15 passages	36 days 14 passages	Not adapted	42 days 15 passages		
Gradual media change (G)	Not adapted	36 days 18 passages	_	42 days 20 passages		
Gradual serum reduction (S)	_	_	Not adapted	_		

Table 20.3 Summary of Cell Adaptation to Serum-Free Media (SFM)

SCM, Standard culture medium composed of Dulbecco Modified Eagle's medium/Ham's F12 1:1 supplemented with 3% fetal bovine serum.



#### 20.2.2 Rabies Virus Production Using Baby Hamster Kidney Cell Line-21 Adapted to Serum-Free Media

All adapted cell lines were submitted to infection kinetics to verify the ability of adapted cells to produce rabies virus. A control culture was carried out using standard culture media for infection.

#### 20.2.2.1 Culture Media

Standard culture medium for infection, composed of DMEM/Ham's F12 1:1 supplemented with 1% FBS, was used during virus infection of the control cell culture.

VP-SFM (Gibco–Invitrogen cell culture), Ex Cell 302 (SAFC Biosciences), and Cellvento BHK-200 (Merck Millipore) were prepared following the manufacturers' instructions. No antibiotic was added.

#### 20.2.2.2 Virus Strain

Rabies Pasteur virus, supplied from the Federal University of Minas Gerais, was adapted to grow in BHK-21C13 in suspension in standard culture media for infection. The adapted virus generated TECPAR (Technology Institute of Paraná) master and working virus bank was stored at  $-80^{\circ}$ C. The virus strain provided by the TECPAR working virus bank was used for this study.

#### 20.2.2.3 Cell Infection and Virus Production

Cell infection was carried out on cultures with a cell density of about  $1.0 \times 10^6$  cells/mL. The pH was adjusted to 7.4 with sodium bicarbonate 7.5%. Cells were settled in a cold chamber and 300 mL media were removed. Viruses in a multiplicity of infection of 0.015, and diethylaminoethanol (1% final volume) were added to the spinner flask and cells

were incubated at  $33^{\circ}$ C, over 35 rpm intermittent agitation (5 min of agitation and 20 min stopped) for 90 min. Media were added and spinners were again incubated at  $33^{\circ}$ C over 55 rpm.

Samples were collected daily for cell count, metabolite analysis, pH measurement, virus titer determination, and direct immunofluorescence assay. The pH was adjusted to 7.4, cells were settled in a cold chamber, and half the volume of media was harvested. Fresh media were added and spinners were again incubated. This procedure was continued for 6 days.

#### 20.2.2.4 Direct Immunofluorescence Assays

Infected cell samples ( $50 \ \mu$ L) were settled in microscope slides, fixed with cold 80% acetone, and stained with Evans Blue dye (1:40,000) and fluorescein-labeled antirabies nucleocapsid immunoglobulin (Bio-Rad code 357-2114). Slides were analyzed under a fluorescence microscope to determine the level of cell infection. It was evaluated in terms of fluorescent focus inside the cells (Table 20.4). This procedure was adapted from Batista and collaborators [24].

Immunofluorescence assays were conducted to verify the evolution of cell infection.

#### 20.2.2.5 Rabies Virus Titration

To determine the rabies virus titer, monolayers of BHK-21 cells on 96-well microtiter plates were infected with sample dilutions and incubated at  $37^{\circ}$ C with 5% CO<sub>2</sub> for 22 h. Cells were then fixed with cold 80% acetone, washed with phosphate-buffered saline, and incubated with fluorescein-labeled antirabies nucleocapsid immunoglobulin (Bio-Rad code 357-2114) for 1 h at  $37^{\circ}$ C [25–27]. The microtiter plates were then observed under a fluorescence microscope and the titer was expressed as the 50% fluorescent focus-forming dose (FFD<sub>50</sub>), as calculated by the Spearman-Kärber method [28].

#### 20.2.2.6 Infection Kinetics of Control Culture

During all infection kinetics studies, the maximal cell densities of cultivations were obtained after infection, owing to the media harvest that provided nutrients and removed metabolites. In the control cultivation, cell viability decreased right after cell infection. Glucose and lactate metabolism after infection was similar to the metabolism of

Infection Class	Fluorescence Focus
0	Total absence of fluorescent focuses
I	1–25% of cells had fluorescence focuses
11	26–50% of cells had fluorescence focuses
III	51–75% of cells had fluorescence focuses
IV	76–100% of cells had fluorescence focuses

Table 20.4	<b>Relation Between Infection Class</b>
and Fluores	cence Focuses

	Rabies Virus Titer log <sub>10</sub> FFD <sub>50</sub> /mL						
Serum-Free Adapted Cells	1dpi	2dpi	3dpi	4dpi	5dpi	6dpi	Average
Control	3.23	4.93	5.47	5.04	5.04	4.63	4.72
VPD	4.90	5.25	5.54	5.54	5.62	5.11	5.33
EXD	2.96	4.92	5.54	5.04	5.4	5.25	4.85
EXG	2.83	4.91	4.9	5.32	4.9	5.18	4.67
CVTDvp	3.53	4.63	4.9	5.32	5.32	4.64	4.72
CVTGvp	3.23	4.63	4.9	5.05	4.77	4.9	4.58

**Table 20.5**Virus Titer During Infection Kinetics of Adapted Cellsto Serum-Free Media

*CVTDvp*, Adaptation to Cellvento BHK 200 media by direct media change using cells previously adapted to VP-serum-free media (SFM); *CVTGvp*, adaptation to Cellvento BHK 200 media by gradual media change using cells previously adapted to VP-SFM; *dpi*, days postinfection; *EXD*, adaptation to Ex Cell 302 media by direct media change; *EXG*, adaptation to Ex Cell 302 media by gradual media change.

noninfected cells. Since the second day postinfection (dpi), more than 75% of cells were infected. Titer peak (bold in Table 20.5) coincided with maximal cell density on dpi 3.

Taking into account that in the industrial production of rabies vaccine in TECPAR only a harvest with  $\log_{10}$  FFD<sub>50</sub> higher than 3.8 is used for veterinary vaccine production, the first harvest was the only one that did not meet this requirement (Table 20.5).

#### 20.2.2.7 Infection Kinetics of Cells Adapted to VP-Serum-Free Medium by Direct Media Change

In the infection kinetics of cells adapted to VP-SFM, the maximal cell density was  $2.3 \times 10^6$  viable cells/mL. Cell viability decreased only after dpi 3. Infection class IV cells were obtained on dpi 3. After infection, glucose consumption increased when cells reached class IV of infection, all available glucose had been consumed during harvest, media were replaced, and glucose was supplied, even though all glucose was consumed. This may be because rabies virus is a glycoprotein, and glucose was used to produce it. Since dpi 1, the virus titer met the vaccine production requirement. In this study, the infection of cells adapted to VP-SFM reached the highest virus titer and the highest average of virus titer (Table 20.5). This cultivation was the only one in which the maximal cell density did not coincide with the peak virus titer.

20.2.2.8 Infection Kinetics of Cells Adapted to Ex Cell 302 by Direct Media Change The infection of cells adapted to Ex Cell 302 by direct media change had the maximal cell density at dpi 3. A slight decrease in cell viability was observed right after cell infection. The virus titer peak (bold in Table 20.5) coincided with the maximal cell density and when more than 75% of cells were infected. Glucose consumption and lactate production increased after infection, but glucose scarcity was not observed. Only the first harvest did not reached the minimal virus titer for vaccine production.

#### 20.2.2.9 Infection Kinetics of Cells Adapted to Ex Cell 302 by Gradual Media Change

The infection of cells adapted to Ex Cell 302 by gradual media change reached the maximal cell density on dpi 4. A slight decrease in cell viability was observed right after cell infection. Infection class IV was obtained on the second day after infection. The virus titer peak (Table 20.5) coincided with the maximal cell density. Glucose consumption and lactate production increased after infection, but glucose scarcity was observed only on dpi 6. Only the first harvest did not reach the minimal virus titer for vaccine production.

#### 20.2.2.10 Infection Kinetics of Cells Adapted to BHK by Direct Media Change

The infection of cells adapted to Cellvento BHK-200 by direct media change reached maximal cell density at dpi 5. A high decrease in cell viability was observed right after cell infection. The virus titer peak (bold in Table 20.5) coincided with the maximal total cell density on dpi 4, and with the maximal viable cell density on dpi 5. One day after infection, less than 25% of cells were infected. On dpi 3 more than 75% of cells were infected. Glucose consumption and lactate production increased after infection. Glucose scarcity was observed only at the sixth harvest. The first harvest was the only that did not reach the minimal virus titer for vaccine production.

#### 20.2.2.11 Growth Kinetics of Cells Adapted to BHK by Gradual Media Change

Infection of cells adapted to Cellvento BHK-200 by gradual media change reached the maximal cell density at dpi 4, coinciding with the virus titer peak. A high decrease in cell viability was observed right after cell infection. One day after infection, less than 25% of cells were infected. More than 75% of cell infection was observed on dpi 3. Glucose consumption and lactate production increased after infection, but glucose scarcity was observed only at the sixth harvest. Only the first harvest did not reach the minimal virus titer for vaccine production.

#### 20.2.2.12 Comparison of Virus Titer in Serum-Free Adapted Cell

Fig. 20.3 shows the virus titer during infection kinetics studies. With the exception of cells adapted to VP-SFM that had all harvests with a virus titer meeting requirements for vaccine production, others had enough virus titer since dpi 2.

No relation was observed between the infection class and virus titer because even when less than 50% of cells were infected, the virus titer was higher than  $10^{3.8}$ . All adapted cells had the capacity to produce virus for vaccine production.

Cell adapted to VP-SFM culture media had the worst growth kinetics but the best virus production, demonstrating the importance of the virus infection study. VP-SFM is a culture medium developed for virus production, and probably has a formulation more


**FIGURE 20.3** Comparison of virus production of all adapted cell lines. The *red line* (gray in print versions) is the cutoff line. Only the harvest of virus that results in the virus titer above the red line is used for vaccine production.

appropriate for this indication. Studies should be carried out to investigate whether increasing the glucose supply during infection enhances virus production.

The results presented here indicate that cells adapted to SFM can produce rabies virus for vaccine production. However, virus harvests have to be formulated and challenged in animals to confirm vaccine potency.

Perrin et al. [20] carried out cell infection in a bioreactor using a perfusion system. The best virus titer was achieved on the third to fifth days after infection. The results of this technological development were shown in a high virus titer from the second to the sixth day postinfection.

Kallel et al. [22] infected BHK-21 adapted to SFM with rabies virus. The study was carried out in spinners. The virus titer started to increase after dpi 3 and 5, depending on the media used. Virus peaks were obtained only on dpi 9.

Although cells adapted to VP-SFM culture media showed the best virus titer results, attention should be paid to Cellvento BHK-200 culture media. Because this culture medium was developed for a veterinary vaccine, mainly mouth-and-foot disease, which requires large-scale production and low prices compared with human vaccines, the Cellvento BHK-200 price is about 10 times cheaper than other SFM developed for human viral vaccine production (VP-SFM) or for recombinant proteins for human use (Ex Cell 302).

Table 20.6 compares culture media costs using SFM produce 10 million of doses of rabies vaccine. Therefore Cellvento BHK-200 culture medium currently seems to be the most adequate for veterinary rabies vaccine production.

Culture Medium	Volume for 10 Million of Doses (L)	Medium Costs (EUR)
VP-serum-free media	10,000	400,000.00
Ex Cell 302	11,000	440,000.00
Cellvento Baby hamster kidney cell line-200	11,000	40,000.00

**Table 20.6**Cost Estimation of Serum-Free Culture Media for 10 Million of Dosesof Vaccine Production

# 20.3 Conclusion and Perspectives

Dog rabies mass vaccination is an extremely important strategy to control human rabies infection. In Brazil, about 30 million of doses are used annually for dog and cat mass vaccination. Veterinary rabies vaccines available on the market are produced using serum for cell cultivation. Knowledge of side effects caused in dogs by serum in rabies vaccine leads to development of a vaccine produced with SFM. This case study showed that it is possible to substitute serum in the production process using commercial SFM. BHK-21C13 cell in suspension is able to adapt in the three serum-free culture media used in this technological development (VP-SFM, Ex Cell 302, and Cellvento BHK-200). Kinetics studies of cell growth and rabies virus production demonstrated that the best results in growth kinetics do not mean the best results in virus production. All adapted cells are able to produce rabies virus for vaccine production. Different adaptation protocols of cell adaptation to SFM do not interfere with the results of virus production. The best virus productivity was reached using VP-SFM culture media; however, taking into account economic aspects, the best result was achieved using Cellvento BHK-200.

Vaccine potency must be determined to prove virus quality. Because vaccine potency is good, scale-up studies using SFM should be carried out to permit serum substitution in industrial veterinary rabies vaccine production.

In addition, BHK-21 adapted to SFM can be used to produce other veterinary viral vaccines, such as foot-and-mouth disease. In this case, replacement of serum in cell cultivation leads to an important improvement in vaccine, because bovine serum may have adventitious agents, such as virus and prions, which can infect vaccinated calves.

ВНК	Baby hamster kidney cell line
СНО	Chinese hamster ovary cell line
COS	African green monkey kidney cells (CV1 cells transformed with a defective mutant of SV40)
CVTD	Adaptation to Cellvento BHK 200 media by direct media change
CVTDvp	Adaptation to Cellvento BHK 200 media by direct media change using cells previously adapted
	to VP-SFM
CVTGvp	Adaptation to Cellvento BHK 200 media by gradual media change using cells previously adapted
	to VP-SFM
CVTS	Adaptation to Cellvento BHK 200 media by gradual serum reduction
DEAE	Diethylaminoethanol

### List of Abbreviations

DMEM	Dulbecco Modified Eagle's medium
DMSO	Dimethyl sulfoxide
dpi	Day postinfection
EXD	Adaptation to Ex cell 302 media by direct media change
EXG	Adaptation to Ex cell 302 media by gradual media change
FBS	Fetal bovine serum
FBS	Fetal bovine serum
FFD <sub>50</sub>	Fluorescent focus doses 50
FMDV	Foot and mouth disease virus
HCV	Hepatitis C vírus
HIV	Human immunodeficiency virus
lgE	Immunoglobulin E
MDCK	Madin-Darby canine kidney cell line
MDSS2	Serum free medium developed by Merten et al. [21].
MOI	Multiplicity of infection
NIL2	Embryo hamster cell line
Panaftosa	Pan American Centre for Foot-and-Mouth disease
PBS	Phosphate buffered saline
PV	Rabies Pasteur Virus
SCM	Standard culture medium, comprising DMEM/Ham's F12 1:1 supplemented with 3% of FBS
SCMI	Standard culture medium for infection, comprising DMEM/Ham's F12 1:1 supplemented with 1% of FBS.
SFM	Serum free medium
TECPAR	Technology Institute of Paraná
UFMG	Federal University of Minas Gerais
UFPR	Federal University of Paraná
VPD	Adaptation to VP-SFM media by direct media change
VPG	Adaptation to VP-SFM media by gradual media change
VP-SFM	Commercial serum free medium developed for human viral vaccine production
WHO	World Health Organization

# Acknowledgments

We are thankful to TECPAR (Instituto de Tecnologia do Paraná) for the Line BHK cell and financial support.

# References

- [1] WHO-World Health Organization, Expert Consultation on Rabies: Second Report, 2013. WHO technical report series 982.
- [2] T. Hemachudha, J. Laothamatas, C.E. Rupprecht, Human rabies: a disease of complex neuropathogenetic mechanisms and diagnostic challenges, Lancet Neurology 1 (2002) 101–109.
- [3] G. Ugolini, Rabies virus as a transneuronal tracer of neuronal connections, Advances in Virus Research 79 (2011) 165–202.
- [4] WHO, Expert Consultation on Rabies: First Report, 2004. WHO technical report series 931.

- [5] A.A. King, G.S. Turner, Rabies a review, Journal of Comparative Pathology 108 (1993) 1–39.
- [6] E. Fuenzalida, R. Palacios, Un método mejorado para la preparación de la vacuna antirabica, Bulletin of the Institute of Bacteriology, Chile 8 (1955) 3.
- [7] I.M.V.G.C. Mello, M.M. Conceição, S.A.C. Jorge, P.E. Cruz, P.M.M. Alves, M.J.T. Carrondo, C.A. Pereira, Viral vaccines: concepts, principles, and bioprocesses, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor & Francis Group, New York, 2008, pp. 435–458.
- [8] WHO, Rabies vaccines: WHO position paper, Weekly Epidemiological Record 32 (2010).
- [9] L. Diamond, Two spontaneously transformed cell lines derived from the same hamster embryo culture, Journal of Cancer 2 (1967) 143.
- [10] P. Precausta, Vaccin antirabique inactivé à usage vétérinaire preparé à partir de culture cellulaire, Symposia Series in Immunobiological Standardization 21 (1974) 162.
- [11] I. Macpherson, M. Stoker, Polyoma transformation of hamster cell clones an investigation of genetic factors affecting cell competence, Virology 16 (1962) 147–151.
- [12] T. Pay, A. Boge, F. Menard, P. Radlett, Production of rabies vaccine by an industrial scale BHK-21 suspension cell culture system, Developments in Biological Standardization 60 (1985) 171.
- [13] P. Léo, A.L.L. Galesi, C.A.T. Suazo, A.M. Moraes, Animal cells: basic concepts, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor & Francis Group, New York, 2008, pp. 13–38.
- [14] A.M. Moraes, R.Z. Mendonça, C.A.T. Suazo, Culture media for animal cells, in: L.R. Castilho, A.M. Moraes, E.F.P. Augusto, M. Butler (Eds.), Animal Cell Technology: From Biopharmaceuticals to Gene Therapy, Taylor & Francis Group, New York, 2008, pp. 111–128.
- [15] K. Ohmori, K. Masuda, S. Maeda, Y. Kaburagi, K. Kurata, K. Ohno, D.J. DeBoer, H. Tsujimoto, M. Sakaguchi, IgE reactivity to vaccine components in dogs that developed immediate-type allergic reactions after vaccination, Veterinary Immunology and Immunopathology 104 (2005) 249–256.
- [16] K. Ohmori, K. Masuda, S. Maeda, K. Ohno, D.J. DeBoer, M. Sakaguchi, H. Tsujimoto, Immunoblot analysis for IgE-reactive components of fetal calf serum in dogs that developed allergic reactions after non-rabies vaccination, Veterinary Immunology and Immunopathology 115 (2007) 166–171.
- [17] Merck Millipore, Product Information: 1.02376 Cellvento<sup>™</sup> BHK-200 Cell Culture Medium. Available at: <a href="http://www.merckmillipore.com/showBrochure?id=201307.058">http://www.merckmillipore.com/showBrochure?id=201307.058</a>>, (accessed on: September, 2012).
- [18] SAFC Bioscience, Product Information Ex Cell<sup>®</sup> 302 Serum-Free Medium for CHO Cells. Available at: <a href="http://www.sigmaaldrich.com/etc/medialib/docs/Sigma/Product\_Information\_Sheet/p14326">http://www.sigmaaldrich.com/etc/medialib/docs/Sigma/Product\_Information\_Sheet/p14326</a>. Par.0001.File.tmp/p14326.pdf
- [19] Gibco, VP-SFM<sup>™</sup>. Available at: <http://tools.lifetechnologies.com/content/sfs/manuals/3918% 20VP%20SFM.pdf>, 2008 (accessed on: September, 2012).
- [20] P. Perrin, S. Madhusudana, C. Gontier-Jallet, S. Petres, N. Tordo, O. Merten, An experimental rabies vaccine produced with a new BHK-21 suspension cell culture process: use of serum-free medium and perfusion-reactor system, Vaccine 13 (1995) 1244–1250.
- [21] O.-W. Merten, J. Kierulff, N. Castignolles, P. Perrin, Evaluation of the new serum-free medium (MDSSP) for the production of different biologicals: use of various cell lines, Cytotechnology 14 (1994) 47.
- [22] H. Kallel, A. Jouini, S. Majoul, S. Rourou, Evaluation of various serum and animal protein free media for the production of a veterinary rabies vaccine in BHK-21 cells, Journal of Biotechnology 95 (2002) 195–204.
- [23] H.J. Cruz, C.M. Freitas, P.M. Alves, J.L. Moreira, M.J.T. Carrondo, Effects of ammonia and lactate on growth, metabolism and productivity of BHK cells, Enzyme and Microbial Technology 27 (2000) 43–52.

- [24] A.M. Batista, P.S. Cruz, E. Almeida, A.E.B. Costa, K.C. Scheffer, L.B. Chaves, A.C.R. Silva, G.M.M. Caporale, Infection of BHK-21 cells cultivated in stationary monolayers by PV and CVS strains, Boletin Epidemiologico Paulista (BEPA) 6 (2009) 71.
- [25] J.S. Smith, P.A. Yager, G.M. Baer, A rapid fluorescent focus inhibition test (RFFIT) for determine rabies virus-neutralizing antibody, in: F.X. Meslin, M.M. Kaplan, H. Koproeski (Eds.), Laboratory Techniques in Rabies, fourth ed., World Health Organization, Geneva, 1996, pp. 181–192.
- [26] W.C. Moura, N.M.F.G. Gallina, R.M.M. Fuches, P.C. Romijn, J.P.G. Leite, Validation of a virus neutralization potency test in BHK-21 cells for rabies immunoglobulins in a two-center study, Journal of Virological Methods 154 (2008) 7–13.
- [27] R.M.M. Fuches, Validação de bioensaios para o estudo da imunogenicidade da vacina contra raiva, Tese (Doutorado na área de Biotecnologia) – Universidade de São Paulo/Instituto Butantan/ Instituto de Pesquisas Tecnológicas, 2010.
- [28] R.J. Lorenz, K. Bogël, Methods of calculation, in: F.X. Meslin, M.M. Kaplan, H. Koproeski (Eds.), Laboratory Techniques in Rabies, fourth ed., World Health Organization, Geneva, 1996, pp. 321–335.

# 21

# Development of a Rabies Vaccine in Cell Culture for Veterinary Use in the Lyophilized Form

### M.G. Severo<sup>1, \*</sup>, A.S. Zeferino<sup>2</sup>, C.R. Soccol<sup>3</sup>

<sup>1</sup>DIVISÃO DE VACINAS VIRAIS DO INSTITUTO DE TECNOLOGIA DO ESTADO DO PARANÁ -TECPAR, CURITIBA, PARANÁ, BRAZIL; <sup>2</sup>INSTITUTO DE TECNOLOGIA DO PARANÁ, CURITIBA, PARANÁ, BRAZIL; <sup>3</sup>UNIVERSIDADE FEDERAL DO PARANÁ (UFPR), CURITIBA, PARANÁ, BRAZIL

# 21.1 Rabies Virus and Rabies Vaccines

#### 21.1.1 Rabies: Disease, Epidemiology, and Burden

Rabies is a viral zoonosis that occurs in more than 100 countries and territories, as shown in Fig. 21.1 [1]. There were 61,000 estimated cases of death from rabies in the world in 2010, according to the informal group Partners for Rabies Prevention. The estimated annual cost of rabies is US \$6 billion, covering pre- and postexposure treatments and loss of productivity owing to the premature deaths of workers [2].

The infection usually occurs when an infected animal causes a transdermal bite or scratch in the victim. Transmission may also occur when infectious material, usually saliva, comes into direct contact with the victim's mucosa or with fresh skin wounds. The incubation period is typically 1–3 months, but it may vary from less than a week to more than a year [3].

Inoculated virus is transported to the central nervous system (CNS). Upon arrival in the brain, it replicates and disseminates rapidly in many different tissues including the salivary glands [3]. The acute disease includes pain or paresthesia close to the bite site and is often associated with fever, fatigue, and weakness in associated limbs. Nonspecific neurologic symptoms including headache and anxiety are often experienced before the development of encephalitis, and it is usually only at this point that patients seek medical help and admission to the hospital [4].

As the virus spreads through the CNS, progressive fatal encephalomyelitis develops, characterized by hyperactivity and fluctuating consciousness and, in cases of furious

<sup>\*</sup>Corresponding Author.



Distribution of risk levels for humans contacting rabies, worldwide, 2013

FIGURE 21.1 Rabies worldwide distribution and risk exposition [1].

rabies, hydrophobia, aerophobia, or both. Death occurs by cardiorespiratory arrest within a few days [3].

#### 21.1.2 Rabies: Classification, Proteins, and Cell Infection

The rabies virus (RABV) belongs to the order Mononegavirales and family Rhabdoviridae. This family has three genera: *Vesiculovirus* (whose prototype is the vesicular stomatitis virus), *Ephemerovirus* (whose prototype is the bovine ephemeral fever virus), and *Lyssavirus* (whose prototype is the RABV) [5]. RABV and variants known as "rabies-related" virus belong to one of the 14 *Lyssavirus* genotypes, all of which are capable of producing a rabies-like encephalomyelitis [1].

The *Lyssavirus* stable infective particles (virions) have a bullet-shaped form with an average length of 180 nm and average diameter of 75 nm [6]. Lyssaviruses, like other Rhabdoviruses, consist mainly of RNA (2-3%), protein (67-74%), lipid (20-26%), and carbohydrate (3%) as integral components (percent total mass) of their structure [7].

The RABV genome is a nonsegmented, single-stranded RNA that has a negative-sense polarity. This implies that the minus-strand genome RNA, when free of protein, is not infectious for the victim's cells [8]. The genome analysis reveals the presence of five proteins: a

nucleoprotein (N), a phosphoprotein (P), a matrix protein (M), an RNA polymerase RNAdependent (L), and a glycoprotein (G), which carries the main antigenic sites [3,9].

The cycle of infection and replication of RABV can be summarized in the stages: (1) adsorption of virions to host cell, (2) penetration of the virion into the cell, (3) removal of protein—RNA complex proteins, (4) transcription of viral RNA, (5) translation of mRNAs, (6) processing of proteins, (7) viral RNA replication, (8) assembly of virions, and (9) budding [10].

In the virion morphogenesis, the proteins N, P, and L newly synthesized in the cytoplasm adhere to nascent viral RNA to form the nucleocapsid. The nucleocapsid is then stabilized by the M protein, which is responsible for giving the nucleocapsid its bullet-shaped morphology. The protein G, which requires glycosylation, is processed in rough endoplasmic reticulum and then goes to the host cell membrane. Finally, the nucleocapsid—protein M interacts with cytoplasmic membrane fragments that contain protein G to finish virion assembling and proceeds to budding of these new virion to spread the virus further [8,11].

# 21.1.3 Rabies Vaccines: Definition, Generations, and Use for Humans and Animals

A vaccine is any preparation intended to produce immunity to a disease by stimulating the production of antibodies. The most common method of administering vaccines is by injection. However some vaccines are given by mouth or nasal spray [12]. The first rabies vaccine was developed by Louis Pasteur and Emile Roux in 1885 [13].

Speaking specifically about rabies, there were three generations of vaccine production: the first generation used brains of rabbits, sheep, cows, horses, and other animals as substrate for virus cultivation. In 1955, Fuenzalida and Palacios described a vaccine using suckling mouse brain as substrate [14], and Peck and collaborators described a vaccine using duck embryonated eggs as substrate [15]. The second generation used cell cultures to produce rabies vaccine. In 1960, Fenje announced the first vaccine to be produced on hamster kidney tissue [16]. In the following years there were many reunions and congresses that debated research all over the world to standard the work, which enabled the development and then rapid commercialization of vaccines for veterinary use. This development benefitted from four favorable factors: (1) cell lines such as baby hamster kidney (BHK) were accepted for the production of vaccines for use in animals; (2) viral multiplication is excellent in these cells; (3) these systems permitted industrial scaling-up of the cultures; and (4) the addition of adjuvant was authorized for these veterinary vaccines [17]. The third generation of rabies vaccines is available. The vaccinia-rabies glycoprotein recombinant vaccine is currently applied on a large scale in some European countries for the immunization of wildlife. The canarypox recombinant vaccine has been considered and successfully tested for human immunization [18]. However it is first necessary to understand the differences between vaccines for human and veterinary use.

#### 21.1.3.1 Rabies Vaccines for Human Use

Vaccines for human use have existed for more than 40 years and can be obtained as (1) nerve tissue vaccines and (2) concentrated and purified cell culture or embryonated eggbased vaccines (CCEEVs) [2].

Nerve tissue vaccines have a series of disadvantages compared with CCEEVs, such as the induction of more severe adverse reactions and less immunogenicity. Because of these, the World Health Organization (WHO) recommended discontinuation of nerve tissue vaccines for human use in 1984 and their replacement by CCEEVs. Many developing countries followed this recommendation, including Brazil, Bangladesh, China, Chile, India, Indonesia, Mexico, Paraguay, and Thailand. Other countries such as Ecuador, Peru, Myanmar, and Pakistan are investigating affordable ways to substitute nerve tissue. Nevertheless in a few countries, mainly in Asia and Latin America, populations at high risk for rabies still depend on nerve tissue for postexposure prophylaxis [2].

CCEEVs have proved to be safe and effective against RABV and are used for pre- and postexposure prophylaxis; they have been administered to millions of people across the globe [19]. CCEEVs contain RABV that has been propagated in cell substrates such as: (1) human diploid cells; (2) Vero cells; (3) primary chick embryo cells; and (4) embryonated duck eggs. Chick embryo and Vero cells are comparatively safe and effective; however the others are less expensive. In the process, after growth in cell culture (or embryonic egg), the viral harvest is concentrated, purified, inactivated, and lyophilized. To stabilize the vaccine, human albumin or processed gelatin can be used. These types of vaccines are made for individual dose by intramuscular injection and do not contain preservatives such as thimerosal. The result is a vaccine with more than 3 years of shelf life that must be stored at  $2-8^{\circ}$ C and protected from sunlight [2,19]. The WHO recommendations for characterization, production, and control apply only for inactivated rabies vaccines produced in cell culture or embryonated eggs [2].

The minimal acceptable potency of CCEEVs is 2.5 international units (IU) per intramuscular dose, as determined in the mouse protection potency test [20]. That way, quick CCEEVs administration after exposure allied with proper wound management and simultaneous administration of rabies immunoglobulins is almost invariably effective in preventing rabies, even after high-risk exposure [19].

In general, CCEEVs are safe and well-tolerated. However some adverse effects may occur, depending on a series of factors. The most common adverse effects observed in vaccinated people are (1) minor, transient erythema, pain, or swelling occurring at the site of injection in 35-45% of patients; and (2) transient fever, headache, dizziness, and gastrointestinal symptoms in 5-15% of vaccinated people. Serious adverse events are rare and include Guillain–Barré syndrome [21].

CCEEVs provide immunologic memory that presumably persists for the life of the individual even after titers of neutralizing antibodies decline [2]. Some studies have indicated that periodic booster doses of vaccine are not required after primary rabies vaccination, except as an additional precaution for people whose occupation puts them

at continual or frequent risk of exposure [22,23]. However, the WHO recommends that even vaccinated individuals who are subsequently exposed to rabies should receive a short course of postexposure prophylaxis [2].

For special cases of exposure to RABV, the rabies vaccine must be complemented with rabies immunoglobulin [24]: (1) people with incomplete vaccination; (2) people with class III exposure (single or multiple transdermal bites or scratches, licks on broken skin, or contamination of mucous membrane with saliva from licks and exposure to bats); (3) immunocompromised individuals, such as patients with AIDS, with class II exposure (nibbling of uncovered skin, minor scratches, or abrasions without bleeding). For class I exposure (touching or feeding animals, licks on intact skin, or contact of intact skin with secretions or excretions of a rabid animal or human) there is no requirement for postexposure prophylaxis [2,22].

Three classes of immunoglobulins are available for passive immunization: human rabies immunoglobulin, equine rabies immunoglobulin, and highly purified F(ab')2 fragments produced from equine immunoglobulin [25]. Rabies immunoglobulin should be given along with the first dose of vaccine in and around the wound site. Human immunoglobulin should be given at 20 IU/kg body weight, whereas equine immunoglobulin requires 40 IU/kg body weight because it has a shorter half-life in humans. Equine immunoglobulin is considerably less expensive than the human product, and most new equine preparations are potent, highly purified, and safe, with few adverse events [2].

#### 21.1.3.2 Rabies Vaccines for Veterinary Use

Veterinary vaccines have been developed for use against rabies in domestic mammals and wildlife. Vaccines should be administered by or under the supervision of a competent person such as a veterinarian, according to the producer's recommendations. Three types of vaccines are used for domestic animals: (1) live replication-competent vaccines for oral use; (2) injectable modified-live virus vaccines; and (3) injectable inactivated vaccines [2].

Live vaccines for oral use may be appropriate under specific conditions, although the parenteral route is preferred for dog vaccination. The oral route should be used as a complement to parenteral mass vaccination campaigns, to improve vaccination coverage of the dog population by targeting individuals inaccessible for injectable vaccines, such as stray dogs and isolated dogs. Because oral rabies vaccines are costly and safe distribution tends to be time-consuming, the cost—benefit ratio of administering these vaccines to dogs should be carefully assessed. The most common use of oral vaccines is to immunize wild animals. In this case, the liquid vaccine usually contained in a sachet or blister pack should be incorporated in bait, preferably adapted to the target species with regard to taste, size, and texture [2].

Modified-live vaccines are produced from a modified egg-adapted strain of virus serially passaged in embryonated chicken eggs and from strains adapted to cell culture. Examples of this vaccine are those made with virus adapted to cell culture, such as the SAD and ERA strains. These vaccines can cause rabies in vaccinated animals; therefore their production and vaccination should be discontinued owing a lack of safety [2].

Injectable inactivated (killed) vaccines are the ones most used in domestic animal species because they are safe and less expensive than other kinds. Inactivated vaccines are produced in cell culture with either primary cells or continuous cell lines infected with an adapted strain of RABV. Inactivated rabies vaccines can be used in combination with bacterins such as *Leptospira* and other viral agents such as canine distemper virus and canine parvovirus. Combined vaccines currently available for cats include various antigens such as feline panleukopenia virus, feline calicivirus, and feline herpesvirus [2].

Speaking specifically about veterinary vaccine production in cell culture, after cell growth and virus production, it is necessary to inactivate the virus.  $\beta$ -Propiolactone (BPL), UV light, and acetylethyleneimine are used as inactivating agents, and BPL is the one most used [26]. Once inactivated, adjuvants are added to increase the immune response to the antigen. The most used are aluminum hydroxide, aluminum phosphate, and saponin; less commonly used are oil adjuvants [27,28]. Inactivated rabies vaccines are available in liquid or lyophilized form [2].

A case of study presented in Section 21.3 of this chapter shows an example of the freeze-drying process using different stabilizers.

#### 21.1.4 Perfusion and Multiplicity of Infection

Animal cells grow substantially slower than most microorganisms and lack a protective cell wall; thus they are also more fragile. Therefore growth is limited to gentle culture conditions and relatively low cell concentrations. One way to increase cell concentration yet maintain gentle culture conditions is the perfusion method [29].

A perfusion culture is one in which waste medium is continuously removed from the culture and the displaced medium is replenished with fresh medium. The constant addition of fresh medium and elimination of waste products provides cells with the environment they require to achieve high cell concentrations and with higher productivity [29].

In the context of virus multiplication in cells, the concept of multiplicity of infection (MOI) must be explained. MOI is a term frequently used in virology that refers to the number of virions that are added per cell during infection. If one million virions are added to one million cells, the MOI is 1. If 10 million virions are added, the MOI is 10. Add 100,000 virions, and the MOI is 0.1 [30].

A high MOI is used when it is expected that every cell in the culture is infected. By contrast, low MOI is used when multiple cycles of infection are required. However, it is necessary to determine the virus titer to calculate the MOI [30].

MOI calculations establish some conditions. The most frequent is that all cells in a culture are identical concerning the ability to be infected. Deviations in size and surface properties can be disregarded for some cell cultures. However when it comes to multicellular animals there are substantial differences in cell types that will affect

susceptibility to infection. Under these conditions, it is experimentally difficult to determine how many virions infect different cells [30]. Perrin and collaborators tried to develop a rabies vaccine using BHK-21 cell suspension culture in a bioreactor with perfusion, with satisfactory results [31]. They used different MOIs for each stage of adaptation of the cells from monolayer to suspension: 0.003, 0.01, 0.1, and 0.3 cells<sup>-1</sup>. Kallel and coworkers tested various serum and protein-free media to produce rabies vaccines. In spinners, the MOI selected was 0.1 cells<sup>-1</sup>. In bioreactors the MOI was 0.1 or 0.3 [32]. Work with the same MOIs was conducted in Vero cells by Trabelsi and collaborators [33,34]. Guidolin and collaborators also tested two different MOIs (0.07 and 0.046 cells<sup>-1</sup>) to produce rabies vaccines in BHK cells, and obtained satisfactory potency results [35].

## 21.2 Freeze-Drying: Process and Use in Vaccines

#### 21.2.1 Lyophilization: Basic Process

Lyophilization, or freeze-drying, is defined as a stabilizing process in which the substance is first frozen and the quantity of the solvent is reduced by sublimation (primary drying). After, a secondary drying (desorption), the reduction achieves values that no longer support biological growth or chemical reactions, resulting in powder or a cake. The lyophilization process is composed of five phases: (1) formulation, (2) freezing, (3) primary drying submitted to vacuum, (4) secondary drying, and (5) closure of recipients [36].

The main principle involved in freeze-drying is sublimation, in which water passes directly from a solid state, in this case ice, to a vapor state without passing through the liquid state. The material to be dried is first frozen; after that it is subjected to heat under a high vacuum, using conduction and/or radiation, so the frozen liquid sublimes, leaving only dried components of the original liquid. The driving force for the removal of water during freeze-drying is the concentration gradient of water vapor between the drying front and the condenser [37,38].

Lyophilization is a multistage operation in which each step is critical. The major important characters of this scenario are all well-known and should be under strict control to achieve a successful operation: (1) the product, the active principle, which must keep its prime properties; (2) the surrounding medium with its bulking agents, stabilizers, antioxidants, cryoprotectors, lyoprotectors, and other possible constituents, which provide stabilization of the product; (3) the equipment, which must be flexible, reliable, and adapted to the desired use of freeze-drying (mass production of sterile/nonsterile drugs or ingredients, experimental research, and technical development); (4) the process, which has to be adapted to individual cases according to specific requirements and low-temperature behavior of the different products under treatment; (5) the final conditioning and storage parameters of the finished product, which will vary not only from one substance to another but also in relationship to its "expected therapeutic life" and marketing conditions. In other words, a freeze-dryer is not a conventional balance; it does not perform in the same way with different products. There is no universal recipe for a successful freeze-drying operation, and "this material cannot be freeze-dried" can be said only after challenging the system with all and each successive step of the process [39].

An important part of freeze-drying is the energy involved in the process. Energy in the form of heat must be applied to the product to encourage the removal of water in the form of vapor from the frozen product. The heat must be carefully controlled, because applying more heat than the evaporative cooling in the system can warm the product above its eutectic or collapse temperature. Heat can be applied in the system in two ways: directly applied through the thermal conductor shelf, such as is used in drying; or using ambient heat, as in manifold drying [40].

Heat enters the products by one of several mechanisms: (1) direct contact between the container base and the shelf; (2) conduction across the container base and then through the frozen mass to the drying front (sublimation interface); (3) gaseous convection between the product and residual gas molecules in the chamber; and (4) radiation, which is low owing to the low temperature encountered in freeze-drying. Of all four mechanisms, convection is certainly the most important [41].

Before the freeze-drying process, there is a stage of sample preparation. Because the product that will undergo lyophilization may be a solid, liquid, paste, or emulsion, great care must be taken not to impede its fundamental properties [39]. The sample preparation may include product concentration, formulation revision by the addition of components to increase stability and/or improve processing, or a decrease in high–vapor pressure solvent or increase in the surface area. To achieve those goals, several methods can be used: freeze concentration, solution phase concentration, formulation to preserve product appearance, formulation to stabilize reactive products, formulation to increase the surface area, and a decrease in high–vapor pressure solvents [37].

The basic freeze-drying process is divided into three main stages: freezing, primary drying, and secondary drying. There is also an optional stage, the annealing stage.

#### 21.2.1.1 Freezing Stage

The first step in lyophilization is to freeze samples that are aliquotted into containers. The containers are placed on a freeze-dryer shelf and the samples are either cooled from ambient temperature or cooled rapidly on a prechilled shelf [41].

Freezing is a critical period because the microstructure established by the freezing process usually represents the microstructure of the dried product and the product must be frozen to a low enough temperature to solidify completely. Two types of constituents in a solution will be cooled: the major part of products that are subjected to freeze-drying consist primarily of the solvent, usually water, and the solute, the materials dissolved or suspended in the solvent [37].

During this critical stage, all fluids in the sample become solid bodies, crystalline, amorphous, or glass. Usually water results in a complex ice network, but it might also be

imbedded in glassy structures or remain more or less firmly bound within the interstitial structures. This process is critical because two processes can occur at the same time: the solutes concentrate and might crystallize; the sample undergo volumetric expansion that might induce mechanical stresses that combine with the osmotic shock owing to the increasing concentration of interstitial fluids [39].

The freezing point of a sample can be determined using four methods: (1) the theoretical thermodynamic value, (2) cryomicroscopy, (3) differential scanning calorimetry (DSC), and (4) measurement of temperature and resistance. The most common method is DSC [42].

The freezing method and the final temperature of the frozen product can affect the posterior freeze-drying of the sample. Rapid cooling results in small ice crystals, which is useful in preserving structures to be examined microscopically; however the product becomes more difficult to freeze. Slower cooling results in large ice crystals a lower restrictive channel in the matrix during the drying process [37,39].

Products freeze in two ways, depending on their constitution. First there are eutectic mixtures, which are mixtures of substances that freeze at a temperature lower than the surrounding water. The lowest possible melting temperature overall of the mixing ratios for the involved component species is called the eutectic temperature; in other words, a product will be frozen only when all of the solute suspension is properly frozen, which is achieved only when the eutectic temperature is reached. It is important to freeze the product to below the eutectic temperature before beginning the freeze-drying process [37,42].

The second type of frozen product is a suspension that undergoes glass formation during the freezing stage, which happens to polymers rather than eutectic mixtures. In this case there is a concept called the glass transition temperature (Tg). For each polymer or polymer mixture, there is a different Tg, and when the sample is cooled below this temperature, it becomes hard and brittle, like glass. Above the TG the product retains its amorphous form. When freeze-drying occurs below the Tg the entire suspension becomes increasingly viscous as the temperature is lowered. Finally the products freeze at Tg, forming a vitreous solid, which makes the freeze-drying process extremely difficult [43,44].

#### 21.2.1.2 Annealing Stage

Once the product has been frozen, the drying process can be facilitated by an additional annealing process. Annealing is a holding step at the Tg. Annealing can be carried out as a way point during the initial cooling, but more commonly it is a postfreezing warming and holding step, followed by recooling. The freezing step and any postfreezing temperature deviations above Tg will determine the texture or morphology of the product [45].

Annealing is an optional step occasionally used to crystallize the formulation component. If the solute separates out in crystalline form, it is known as the eutectic temperature. In contrast, if an amorphous form is formed, the temperature is referred to as the Tg. Determination of this critical temperature is important for the development of an optimized lyophilization cycle [37,41].

The annealing process is most beneficial in formulations with amorphous components with relatively low Tg values and that require long drying times, or when crystalline components are not completely crystallized. The annealing process in not good for the freeze-drying process in all cases and its use must be evaluated for effectiveness because when excipients crystallize during increases in temperature, they can eliminate their stabilizing effects. That is the reason for the optional choice of this stage [41].

#### 21.2.1.3 Primary Drying Stage

After the product is frozen, the pressure within the freeze-dryer is reduced using a vacuum pump to provide conditions under which ice can be removed from the frozen samples via sublimation, resulting in a dry, structurally intact product. To achieve good primary drying, it is necessary to optimize the balance between the temperature and the pressure inside the chamber [42].

In pharmaceuticals, for example, the chamber pressures range from 30 to 300 mTorr, depending on the desired product temperature and characteristics of the freeze-dryer. The chamber pressure needs to be lower than the vapor pressure of ice to ease sublimation of ice and the transport of water vapor to the condenser where it is deposited as ice, a phenomenon called cold trap. If the chamber pressure exceeds the vapor pressure at the sublimation interface, mass transfer stops. However, if the chamber pressure is too low, the heat transfer to the product is limited, resulting in slower sublimation rates [37,42].

During this stage, frozen samples are maintained below the product collapse temperature and the eutectic melting temperature. The product collapse temperature is highly influenced by the Tg and is unique for each formulation. Below the Tg, the amorphous phase exists as glasslike material, inhibiting large-scale molecular movement. Therefore, sublimation efficiency during the primary drying stage is maximized by operating as close to Tg as possible yet below that temperature. If the Tg is exceeded during the primary drying phase, it can result in cake collapses, which in turn can lower the rate of water sublimation/desorption, cause physicochemical changes in the product and resulting in an inferior product. Drying just below the Tg (at least  $2-5^{\circ}$ C) not only increases the efficiency of the primary drying phase, but also prevents cake collapses and results in more uniform distribution of bound water at the completion of the lyophilization process [41,46].

The end of the primary drying stage is indicated by a visible temperature increase that occurs upon completion of sublimation of water from ice. Although a large percentage of the water is sublimed during the primary drying phase, approximately 20–50% of more tightly associated solvent water is still present in the dried cake [41,47].

Because primary drying cycles can require more than a week to complete, the choice of temperature and sublimation rate needed to avoid cake collapses must be balanced with practical and economic considerations [41].

#### 21.2.1.4 Secondary Drying Stage

After primary freeze-drying is complete and all ice has sublimed, in the area where ice has been removed, desorption of water from the cake occurs. This process is called secondary drying and has started in the primary drying phase. The product appears to be dry but the residual moisture content may be as high as 7–8%. Continued drying is necessary at a warmer temperature to reduce the residual moisture content to optimal values. This process is called isothermal desorption because the bound water is desorbed from the product. It is not an easy task because overdrying might be as bad as underdrying. For each product appropriate residual moisture has to be reached under given temperatures and pressures [37,39].

Secondary drying is normally continued at a product temperature higher than ambient but compatible with the sensitivity of the product, typically maintaining the product at a temperature between 20°C and 40°C for several hours. In contrast to processing conditions for primary drying, which use low shelf temperature and a moderate vacuum, desorption drying is facilitated by raising the shelf temperature and reducing the chamber pressure to a minimum if it is not already low enough. At this stage, it is possible to increase the shelf temperature because the product has an increased thermal stability and a higher Tg owing to previous dehydration. However, the temperature must not be increased rapidly, so the ramp rate to the secondary drying temperature needs to be moderate  $(0.1-0.3^{\circ}C/min)$  for amorphous substances to avoid surpassing the Tg of the lyophilized cake and avoid cake shrinkage [37,41,42].

Also, water remaining during secondary drying is more strongly bound, and thus requires more energy for its removal. Decreasing the chamber pressure to the maximum attainable vacuum has traditionally been thought to favor desorption of water. The rate of desorption and the obtainable moisture level are controlled by diffusion within the solute phase and desorption from the surface, and therefore mostly depend on product temperature [37].

The duration of the secondary drying phase is dictated by the desired residual moisture level in the freeze-dried product. This residual moisture is a small portion of strongly bound water molecules that stabilize the product by hydrogen bonding interactions. Usually higher residual moisture levels translate to an increase in the product degradation rate because sorbed water can promote molecular mobility and increase product deterioration, both chemically and physically. Therefore, staying within a specified range of residual moisture levels can be vital to maintaining a desirable level of product integrity [41,48].

#### 21.2.2 Components of a Freeze-Dryer and Methods of Freeze-Drying

The essential components of a freeze-dryer are the chamber, shelves and shelf fluid system, condenser, refrigeration system, vacuum system, and control system.

The chamber is a vacuum-tight box that contains a shelf or shelves for processing the product and that can also be fit with a stoppering system. It is typically made of stainless

steel and is usually highly polished on the inside and is insulated and clad on the outside. The door-locking arrangement can be made with a hydraulic or electric motor [49].

The shelves have an elaborate design because of the several functions it performs. Shelves act as heat exchangers, removing energy from the product during freezing and supplying energy to the product during the primary and secondary drying segments of the freeze-drying cycle. Depending on the type of freeze-dryer, it can have one or more shelves and can be manufactured in sizes up to  $4 \text{ m}^2$  in area [50].

Energy exchange demanded by the freeze-drying process during the drying stages of the cycle is traditionally done by circulating a fluid through the shelves at a desired temperature. Temperature is set in an external heat exchange system consisting of cooling heat exchangers and an electrical heater. Fluid circulated is normally silicone oil and the shelves are connected to the silicone oil system through fixed or flexible hoses. This is circulated at a low pressure in a sealed circuit using a pump [51].

The condenser serves as a cold trap. It is designed to trap the solvent, usually water, during the drying process. The process condenser consists of coils or sometimes plates that are refrigerated to the desired temperature. These refrigerated parts may be in a vessel separate from the chamber (external condenser) or they may be located within the same chamber as the shelves (internal condenser). For an internal condenser the refrigerated coils or plates are placed beneath the shelves on smaller machines and behind the shelves on larger machines, although the position of the condenser does not affect the trapping performance [52].

Concerning the refrigerating system, the product to be freeze-dried is either frozen before it is put into the dryer (prefreezing) or frozen while on the shelves; either way it requires a considerable amount of energy. Compressors (or sometimes liquid nitrogen) supply cooling energy. Because the compressor may perform two tasks, cooling both the shelves and the condenser, usually multiple compressors are needed [37].

A vacuum must be applied to remove the solvent in a reasonable time during the drying process. To achieve a low vacuum, a two-stage rotary vacuum pump is used. Typical mechanical vacuum pumps used in freeze-dryers are oil lubricated; however oil-free mechanical pumping systems are available. For large chambers, multiple pumps may be used. The vacuum pumping system, in conjunction with the condenser system, provides the necessary pressure to conduct the primary and secondary drying processes [36,37].

Control may be entirely or usually fully automatic for production machines. The control elements required are shelf temperature, pressure, and time. A control program sets these values as required by the designed process. The time may vary from a few hours to several days. Other data such as product temperatures and process condenser temperatures can also be recorded and logged [37].

Three methods of freeze drying are most commonly used: manifold, batch, and bulk [53].

In the manifold method, flasks ampoules or vials are individually attached to the ports of a drying chamber. The prefrozen product is quickly attached to the drying chamber or manifold to prevent warming. The vacuum must be quickly created in the product container and the operator relies on evaporative cooling to maintain the low temperature of the product. This procedure can be used only for relatively small volumes and product with high eutectic and collapse temperatures. Heat input can be affected simply by exposing the vessels to room temperature or via a circulating bath [42].

In batch drying, large numbers of similar-sized vessels containing the same product are placed together in a tray dryer. The product is usually prefrozen on the shelf of the tray dryer. Precise control of the product temperature and the amount of heat applied to the product during drying can be maintained. Vials located in the front portion of the shelf may vary in terms of heat exchange from the vials in the center of the shelf. These slight variations can result in small difference in residual moisture. Batch drying allows closure of all vials at the same time under the same atmospheric condition, ensuring a uniform environment in each vial and uniform product stability during storage. The vials can be stoppered in a vacuum or after backfilling with inert gas. Stoppering the vials under a slightly reduced pressure of dry nitrogen gas seems to be the favorite option [39,42].

Bulk drying is generally carried out in a tray dryer as in batch drying. However the product is poured into a bulk pan and dried as a single unit. Although the product is spread throughout the entire surface area of the shelf and may be the same thickness as the product in the vials, the lack of empty spaces within the product mass changes the rate of heat input [42].

Manifold drying has some advantages over batch tray drying. Because the vessels are attached to the manifold individually, each vial or flask has a direct path to the collector. This removes some of the competition for molecular space created in a batch system and is most ideally realized in a cylindrical drying chamber, where the distance from the collector to each product vessel is the same. However, for some products in which precise temperature control is required, manifold drying may not be suitable. In addition, batch drying is used to prepare large numbers of ampoules or vials of one product and is commonly used in the pharmaceutical industry [42].

#### 21.2.3 Freeze-Drying Applications

The most extensive use of the lyophilization process is involved in the health-care industry. This includes the lyophilization of pharmaceuticals such as chemical compounds, parenteral formulations, vaccines, and in vivo and in vitro diagnostic products. Included in this category are biotechnology products that are mainly protein based [36].

Another common use of lyophilization involves veterinary products. These products range from vaccines for individual household pets to large-scale applications such as the inoculation of herds of cattle and sheep or large flocks of poultry. These lyophilized products serve not only to protect animals and the investment of the producer but also to improve the quality of the product. These products also protect the consumer from diseases that may be transmitted by the consumption of animal products [37].

Freeze-drying is commonly applied to foods as well. The most widely known freezedried food product is coffee. However, freeze-dried coffee is not a lyophilized product in the true sense. Other freeze-dried food products may be used as additives, such as berries in breakfast cereals, or are complete products that eliminate the need for refrigeration, such as ice cream that can be eaten without reconstitution. The process has been popularized in the form of freeze-dried ice cream, an example of astronaut food, and is also applicable for products in which weight can be a factor, such as in backpacking. These freeze-dried meals, such as freeze-dried beef, require reconstitution by the addition of water and enhancement of flavor by heating the product. Except in special cases such as freeze-dried coffee, freeze-dried foods cannot compete economically with frozen or canned food products [36].

In chemical synthesis, products are often freeze-dried to make them more stable or easier to dissolve in water for subsequent use. In bioseparation, freeze-drying can also be used as a late-stage purification procedure, to remove solvents. Furthermore, it is capable of concentrating substances with low molecular weight that are too small to be removed by a filtration membrane [37].

Other applications of lyophilization include the freeze-drying of floral products, taxidermy, and the recuperation of wet books and manuscripts. There has been only limited success in the freeze-drying of floral products: only the flower portion of the plant appears to respond to the drying process, and there is often a slight change in color. Freeze-drying has been applied to the taxidermy of small animals. Many small mammals such as squirrels and raccoons have been successfully freeze-dried to provide specimens with a lifelike appearance. Despite success with this technique, it is currently used on a limited basis. Freeze-drying has been instrumental in the rescue of precious manuscripts, books, and documents that incurred water damage as a result of a fire or a flood. By quickly freezing all of the documents and then freeze-drying the documents in batches, many documents and valuable manuscripts have been restored to nearly their original condition [36].

#### 21.2.4 Freeze-Drying Use for Vaccines

Freeze-drying is an established process to improve the stability of labile drugs, especially proteins and complex vehicles such as viral vaccines, viral vectors, liposomes, and lipid–DNA complexes. However, during freezing and subsequent drying, the drug is exposed to diverse stress factors that can cause a significant loss of activity. During freezing, drug stability can be influenced by exposure to ice–water interfaces, salt and drug concentration effects, pH shifts owing to selective crystallization of buffer species, and mechanical damage by growing ice crystals. During drying, removal of stabilizing hydration shells can influence the stability of the drug, to mention only the most common known degradation causes [54,55].

Currently, most vaccines must be stored at refrigerated temperatures or stored as frozen formulations because of progressive loss of antigen thermal stability. To sustain vaccine activity, a cold chain ranging from  $-22^{\circ}$ C to 8°C is usually required to maintain vaccine temperature below that of ambient from the location of manufacture to the site of vaccine administration, using refrigerated trucks to distribute the vaccines. Unfortunately, in many remote areas of the world it is not technologically possible to maintain the cold chain temperature because it is error-prone, logistically difficult, and expensive. In those areas the lack of refrigerators, power failures, and other problems can expose vaccines to elevated temperatures for several weeks, which can ultimately make the vaccine lose its therapeutic properties [41].

Lyophilization has permitted the development of vaccines that would otherwise be too unstable for effective use, and approximately one-third of vaccines in use today are prepared using lyophilization. Although it has been effective in certain cases, this technique has several drawbacks:

- 1. the process does not guarantee increased stability;
- **2.** many lyophilized vaccines still require storage at 4–8°C; the rabies vaccine is one of them;
- **3.** many adjuvants included to enhance immunogenicity cannot be frozen, such as aluminum salts;
- **4.** lyophilized formulations must be reconstituted and are therefore more complicated to administer;
- 5. clinicians prefer a liquid product for safety and ease of administration;
- 6. the freeze-drying process is expensive and laborious; and
- **7.** lyophilization requires complex processing, which can lead to antigen instability [41].

Problems concerning vaccine stability are far from solved. The main reason for these problems is a lack of understanding regarding the mechanisms involved in the thermal degradation of vaccines, both in solution and in the dried state. Dried or not, the major barrier to the development of more thermally stable vaccines is a lack of detailed information regarding the mechanisms by which vaccine potency is lost. In most cases there is no information about the molecular basis of the observed instabilities. In contrast, considerably more information is available regarding the mechanisms of inactivation for pharmaceutical peptides, proteins, and DNA than for common viral- and bacterial based vaccines. A better understanding of actual mechanisms of inactivation of specific vaccines should permit a much more rational approach to developing thermostable vaccines by systematic formulation or through alterations in the intrinsic nature of the vaccine entity itself [41].

Ideally, successfully lyophilized vaccines have a uniform appearance with no obvious signs of cake collapse and less than 2% residual moisture, that can be safely stored at ambient temperatures for 1-2 years and are quickly and easily reconstituted to an active solution. A few vaccines consist of relatively "simple" homogeneous components. Others, however, more typically contain combinations of proteins, polysaccharides, nucleic acids, and/or lipids, arranged in a viral or bacterial macromolecular complex.

In large part because of their natural ability to generate a robust immune response that is optimally protective, viral particles or bacteria make up approximately half of available vaccines. The current vaccine thermal instability problem, however, most often lies within these macromolecular complex-based vaccines. For this reason, vaccine component stability will first be addressed in terms of these individual macromolecular components [41].

Therefore to ensure drug stability in the freeze-drying step and subsequent storage, stabilizing excipients have to be employed. Cryoprotectants such as salts, polyols, and sugars are used to stabilize the solution therapeutic in the freeze-thawing process. Lyoprotectants, especially the disaccharides sucrose and trehalose, can be used to stabilize the drug during drying. In addition, bulking agents such as mannitol can be added to enable faster drying times and maintain lyophilisates with an attractive appearance [54,55]. Hubálek described a list of different cryoprotectors used in microorganism conservation [56]: sulfoxides; alcohols and derivatives; saccharides and polysaccharides; amides and imides; heterocyclic compounds; amino acids, proteins, peptides, and polypeptides; complex compounds; and surfactants.

Disaccharides such as sucrose and trehalose are inert and have been used to stabilize liposome, protein, and viral formulations. Excipients such as mannitol, sucrose, glycine, glycerol, and sodium chloride are good tonicity adjusters. Amino acids such as glycine can lower the Tg if it is maintained in the amorphous phase [37].

#### 21.2.5 Advances and Alternatives in Freeze-Drying

#### 21.2.5.1 Advances

Freeze-drying is a process that has been used for decades. Yet there still plenty of room for improvement in many areas. The main objectives are related to equipment improvement, process improvement, and standards adoption.

The equipment has an important role in the lyophilization process; therefore equipment manufacturers should be held accountable for the quality and performance of their equipment [57]. Instead of worrying about the software, enterprises must think about improvements in freeze-dryer design and instrumentation. Regarding the drying chamber, the freeze-dryer can be enhanced in chamber design/configuration, construction materials, finishing materials, insulation, leak prevention, design and construction of doors and handles (to prevent the gorilla effect), shelves construction and spacing, heat transfer and the fluid chosen for this purpose, gas conductance, and shelf temperature. With regard to cleaning, choices are manual, cleaning-in-place, and flooding. Related to the condenser, there is room for improvement in plate and coil configuration, capacity, temperatures, isolation valves, and use of external and internal condensers. Regarding the vacuum system, the foreline (connection between the vacuum pump and the chamber) design and prevention of backstreaming of oil vapors derived from the vacuum pump are important [57].

When process improvement is mentioned, the main objective is to exit the trial-anderror process design to enter rational vaccine design. More effort must be made to include the lyophilization aspects of a formulation as a part of the acceptance criteria [57]. The stability of several vaccines has greatly benefitted from the development of lyophilized formulations; however a significant number of vaccines have yet been successfully stabilized by this approach [41]. Ideally, vaccines should follow some stages so as to be well developed: preformulation (characterization) of the active ingredient, formulation design and accelerated or normal stability testing, and formulation process development. A variety of techniques exist to characterize vaccine formulations physically, but some are more informative and easily accessible than others. On the other hand, a variety of hydrodynamic, spectroscopic, and calorimetric techniques are readily available for routine characterization of vaccine moieties that can provide a multidimensional picture of a vaccine that is potentially sensitive to stability changes when combined [41]. The use of each technique depends on the desired final product and the inherent need of each active ingredient.

In addition, it is time for all parties involved in the lyophilization process to adopt statistical standards and standard methods that will be applicable to the lyophilization process. Such standards, if effective, will lead to improved quality of lyophilized products, a reduction in costs, and a general increase in productivity. They will also increase the sale of freeze-drying equipment because of greater confidence in the process and its associated equipment. The ultimate goal is to develop standards of confidence in assessing the performance and quality of a product that will, in time, eliminate the need for a regulatory agency. The public is better served by an industry that can regulate itself rather than one that has to be closely regulated by a government agency [57].

Inside the lyophilization world there is even room for unconventional improvements and changes. All lyophilization operations have been carried out with aqueous solutions although other solvents can sublimate from the frozen state, including mineral solvents such as ammonia and carbon dioxide, and organic solvents such as diethyl amine, chloroform, cyclohexane, dimethylsulfoxide, and benzene [39]. However the use of other solvents must be studied before their application as vaccines because of the toxicity some of them can bring to the products.

Other efforts are used to change the batch system to a continuous system. For more than 65 years, freeze-drying has been almost exclusively done within the pharmaceutical and biological industries using batch operation. In contrast, the food industry, which had to face large-scale production, progressively shifted from batch to semicontinuous and even purely continuous operations. The results of adopting a continuous method are a substantial increase in productivity and a more homogeneous product [39]. In addition, the dimensions and weight of a continuous freeze-dryer are smaller compared with freeze-dryers used for batch drying, and the losses in case of malfunctioning are smaller in continuous freeze-dryers [57].

Finally, Worral and collaborators developed a method called Xerovac, in which the liquid vaccine formulation is placed on preheated shelves (at 40°C) of a conventional

freeze-dryer, reducing the pressure to 600 Torr [58]. When evaporation begins, the product temperature decreases and the pressure inside the chamber is carefully controlled to maintain the product temperature above the freezing point during the entire evaporation process. The method is used for live attenuated Rinderpest and Peste des Petits ruminant vaccines. Under these conditions viruses are capable of resisting 45°C for 14 days with minimal loss of potency. The degree of thermotolerance was similar to current thermostable freeze-dried vaccines but provided a shorter, cheaper, and simpler process in the absence of the potentially harmful freezing step [41,58].

#### 21.2.5.2 Alternatives

There are alternatives to producing dried products rather than freeze-drying; mainly spray-drying and spray—freeze drying. Both methods require the liquid formulation to be sprayed through a nozzle to produce droplets within a desired size range. For spray-dried formulations, these liquid droplets are sprayed into a high-temperature environment to facilitate evaporation. One major advantage of spray-drying with respect to vaccines is the potential to create dried formulations of aluminum adjuvant-containing vaccines. It is also possible that the production times and product uniformity for dry powder delivery formulations could be significantly more desirable than those for standard lyophilized formulations [41].

An improvement in this method is nano spray-drying, which has been used to manufacture a dry powder for inhalation containing ketoprofen lysinate, a nonsteroidal antiinflammatory drug able to control inflammation in patients with cystic fibrosis. This method was able to obtain a free-flowing powder with satisfying aerosol performance. Different from a standard spray-drying apparatus characterized by a pneumatic nozzle, in this case liquid feed droplets are generated by a piezoelectric system, vibrating a thin stainless-steel membrane [59].

Spray-freeze dried products require the droplets to be sprayed into a low-temperature bath (using liquid nitrogen) to freeze the droplets. The frozen particles are then dried in a standard freeze-dryer [41]. Uses of spray-freeze drying include pulmonary, colony, nasal, and ophthalmic drug deliveries, needle-free intradermal injection, and apparent solubility enhancement [60].

In addition to the challenges mentioned previously for standard lyophilization and dehydration techniques, these spraying technologies include the introduction of new air—liquid interfaces that could lead to macromolecular structural perturbations. Other challenges include the extreme high and low temperatures necessary to dry or freeze the liquid droplets effectively, which have the potential to be damaging; and degradation of macromolecules owing to increased surface areas. The high surface areas of the spray-dried products may not consistently lend themselves to the production of stable dried vaccines when antigens are particularly labile, and there is a possible need to dry-fill vials as a result of manufacturing such formulations, which could be a costly and wasteful process [41].

Also, there are other drying techniques with a potential not yet realized, such as airdrying, foam vacuum-drying, microwave vacuum-drying, supercritical fluid conebulization, and supercritical fluid co-precipitation, which could find applications in the preparation of more stable vaccines [41].

Air-drying is the most common method but it has the disadvantage of need longer drying periods and higher drying temperatures [61].

Supercritical-assisted nebulization has been used successfully to develop a stable dried powder formulation from aqueous protein solutions [63].

In microwave vacuum-drying, the heat is not transferred to a material. Instead, the material is induced to heat itself by directly transforming the electromagnetic energy into kinetic molecular energy [61]. Advantages of this method are a reduction in drying time, lower energy demand, and a high-quality product. Major uses of microwave vacuum-drying are pharmaceutical and food applications such as the drying of carrots [63] and potatoes [64]. This technique also can be used for fine chemicals in the chemical industry and semiconductor manufacturing [61].

# 21.3 Case Study

In the first two sections of this chapter, a review was presented concerning the RABV, the mechanisms behind cell infection with such a virus, and how this can be used to produce rabies vaccines. In addition, the reviewed explained how the lyophilization process occurs and how it can be used to produce vaccines in the freeze-dried form, thus enhancing their stability and shelf life.

To exemplify the matters explained in the first part of the chapter, an experiment was conducted to test different inocula of RABV in cell culture, to learn a bit more about MOI and the importance of an equilibrated amount of virus in the moment of infection to ensure good virus production, therefore producing a good vaccine, in this case for veterinary use.

After evaluating the results of the different MOIs tested, the harvests collected from the vessel with the best result were inactivated and used to formulate 12 different combinations of excipients that were freeze-dried using the same protocol, to study the effects of different excipient classes in the final product. The freeze-dried vessels were also evaluated.

#### 21.3.1 Technological Aspects

To perform the experiment, it was necessary to follow a series of stages to achieve the objectives. Fig. 21.2 shows a scheme of all of the stages during the experiment.

#### 21.3.1.1 Virus Infection Study

The medium used to grow BHK-21 was composed of a combination of Dulbecco's modified Eagle medium and Ham's F12 nutrient mixture medium, pH 7.8. The medium



FIGURE 21.2 Scheme for case of study. The top part concerns the virus infection study and the bottom part concerns the freeze-dry study.

used before cell infection was called the standard culture medium (SCM) and was supplemented with 3% fetal bovine serum (FBS). After infection with virus particles, the medium used was called the SCM for infection (SCMI) and was supplemented with 1% FBS.

The cell line used was BHK-21C13 LVI [Pan American Centre for Foot-and-Mouth Disease (Panaftosa)] with 84 passages. During the experiments, the cell line had 118 passages.

The RABV strain was Pasteur virus adapted for the infection of BHK-21C13 cells; it had one passage. The strain came from Minas Gerais' Federal University. During the experiments, the virus from the work bank was used directly to infect the spinners.

The cells were counted in a hemocytometer. Readings were performed using an Olympus optical microscope with  $\times 100$  magnification. The cells were stained with 0.5% (w/v) Trypan blue at a dilution in which at least 20 cells could be counted per quadrant [65].

A Cryotube with BHK-21 cells from the work bank was thawed at 37°C and used as the cell inoculum in a 500-mL spinner (named 1A) with 80 mL SCM. The spinner was incubated at 37°C at 55 rpm agitation and cell counting was made every day. When it achieved at least 800,000 cells/mL, more medium was added to dilute the cells to 200,000 cells/mL and the spinner was incubated again. This proceeded until the spinner had 500 mL SCM.

After that, 250 mL of the first spinner was transferred to a 1.5-L spinner. SCM was added to this second spinner to complete 1.25 L, which left it with 200,000 cells/mL again; this second spinner was incubated at 37°C at 55 rpm. When the concentration reached 1 million cells/mL, the sample was divided into four spinners (1.1A, 1.2A, 1.3A, and 1.4A), each completed to 1.5 L with SCM. The four spinners were incubated under the same conditions. When the concentration of the four spinners reached at least 1 million cells/mL each, three of the spinners were infected with the virus strain.

The virus titer for the experiment was 10<sup>6.18</sup> fluorescent focus-forming dose (FFD) per milliliter, titrated in a microplate. The MOI calculations were made using the formula shown in Fig. 21.3.

To study the effects of different initial virus concentrations in the spinners, three different MOI were selected for infection: MOI 0.01 and 0.03 FFD/cell and 1% of the total flask volume. The inoculum used corresponding to 1% of the total flask volume (for example, 15 mL in a 1.5-L spinner) was the positive control because of its recurrent use for rabies vaccine production on a bioreactor. Using the formula, it was calculated that the positive control had MOI 0.015 FFD/cell.

When 1 million cells/mL was reached in the spinners, the infection protocol was started. On day 0 of infection, the pH was adjusted to 7.4 with sodium bicarbonate 7.5% (w/v) and the spinner was placed in a cold room  $(2-8^{\circ}C)$  for 2 h to sediment the cells. The supernatant was carefully removed from the spinner, leaving a final volume of

$$Vol_{inoculum} = \frac{MOI \times Cel_{Tot}}{Anti \log_{10} Tit}$$

**FIGURE 21.3** Formula for calculating the correct volume for the infection of spinners. *Antilog Tit*, antilogarithm of virus titer (FFD/mL); *Cel*<sub>tot</sub>, total number of cells, multiplying cell concentration for volume of the flask; *MOI*, multiplicity of infection selected (FFD/cell); *Vol*<sub>inoculum</sub>, volume of inoculum required for the infection (in mL).

300 mL. Then diethylaminoethanol-dextran 0.5% (w/v) was added at 1% of the remaining volume to optimize adsorption of the viral particles in the cells. The spinner was incubated at  $33^{\circ}$ C in intermittent agitation: 20 min stopped and 5 min with 30-35 rpm agitation for 90 min.

The four spinners were categorized: spinner 1.1A became the positive control and received 15 mL virus; spinner 1.2A became the negative control and was not inoculated with virus; spinner 1.3A became MOI 0.01 and received 10 mL virus; and spinner 1.4A became MOI 0.03 and received 30 mL virus.

After inoculation, the volume was completed to 1.5 L with SCMI and incubated at  $33^{\circ}$ C at 55 rpm. The assay lasted 10 days after infection. The next days were called days postinfection (dpi). On each day, samples were taken for cell counting, infection rate, pH evaluation, and glucose and lactate determination, and two samples were frozen for subsequent virus titration. The pH was adjusted to 7.4 with 7.5% sodium bicarbonate and the spinner was placed in a cold room for cell sedimentation for 2 h. The maximum volume of medium was discarded without losing cells, and new SCMI was added to achieve 1.5 L again; then the spinner was incubated in 33°C at 55 rpm. From 3 to 8 dpi, the medium was harvest instead of discarded, placed in 2-L roller bottles, labeled, and stored at  $-80^{\circ}$ C.

After 10 days of experiment, the growth curves for the four spinners were measured in terms of viable cells (Fig. 21.4).



**FIGURE 21.4** Cell growth curves obtained for cultivation of four spinners in development experiments. After 10 days of infection, the negative control of assay achieved maximal cell concentration of 1.78 million cells/mL (viable cells) at dpi 6. The positive control achieved maximal cell concentration of 1.33 million cells/mL (viable cells) at dpi 6. The spinner MOI 0.01 reached a maximum of 1.33 million cells/mL (viable cells) at dpi 5. The spinner MOI 0.03 achieved a maximum with 1.22 million cells/mL (viable cells) at dpi 4.

The growth peaks for the different infected cultures were different: 6 dpi for positive control, 5 dpi for MOI 0.01, and 4 dpi for MOI 0.03. The peak for MOI 0.03 was achieved more quickly, probably because the virus concentration was the highest and started to have an effect on the culture earlier than the others. Nevertheless, the infected cultures achieved similar peaks of concentration and the growth oscillations were practically constant.

Results for the four spinners were similar to those obtained by Kallel and collaborators, although the four spinners used a higher amount of medium volume. In addition, cultures with the same volume in the bioreactor had better cell production than the spinner did [32]. This happened because the bioreactors possess better control for temperature, dissolved oxygen, pH, and agitation. Thus using bioreactors with the same volume as that of the experiment (1.5 L) would increase cell production. In addition, using a perfusion system coupled with a bioreactor would have even better results.

After counting viable and total cells, it was possible to calculate the viability as a percentage. Viability was obtained as a percentage, dividing viable cells by total cells and multiplying by 100. From the results, the highest, lowest, and mean values for viability were obtained (Table 21.1). The viability of the negative control was highest because of the absence of the virus. Therefore cell concentration was limited by other factors such as the absence of gas, pH, temperature control, agitation, and accumulation of some metabolites and debris that could interfere in cell growth. The virus does not kill the cell because new virions are released outside by budding. Uninfected cells replicate normally, although one that is infected cell no longer replicates, just produces more virus until its death. Therefore, with a lower quantity of virus as inoculum, the tendency is to achieve better viability because the cells take a longer time to become infected and die. The spinner with MOI 0.01 had the best viability among infected spinners because of the lower quantity of virus inoculated.

Chapman and collaborators studied the viability of BHK-21 cells infected with RABV and kept at two temperatures, 33°C and 36°C [66]. MOI 0.05 was used in this experiment, corresponding to 1% of the final volume of the vessel, just as for the positive control. The viability of cells after infection at 36°C decreased from 3 dpi to almost 0% [66]. However, the viability of infected cells at 33°C remained above 90%, just as was found in this experiment.

Spinner	Highest Viability	Lowest Viability	Mean Viability
Negative control	98.3% (6)	88.2% (0)	95.4%
Positive control	96.2% (3)	66% (8)	87.7%
Multiplicity of infection 0.01	95.4% (7)	85.6% (8)	91.1%
Multiplicity of infection 0.03	97.5% (3)	75.7% (8)	90.8%

Table 21.1 Rate of Cell Culture Viabilities for Four Spinners

Numbers in parentheses are days postinfection when viability was achieved.

$$\mu = \frac{\ln (X_n) - \ln (X_{n-1})}{(t_n - t_{n-1})}$$

**FIGURE 21.5** Formula for calculating specific growth rate of cell cultures. n and  $n^{-1}$ , two succeeded sampling points;  $\mu m$ , specific growth rate (h<sup>-1</sup>); t, time (h); X, cell concentration (cells/mL).

The specific growth rate ( $\mu$ ) can be calculated with a formula adapted for cell concentration values obtained in the assay (Fig. 21.5). The specific growth rate measures the rate of cell multiplication and is calculated every day after infection.

A summary of the specific growth rate for the spinners is shown in Table 21.2. A characteristic of animal cells such as BHK-21 is that with optimized growth conditions there is a low specific growth rate, between 0.01 and 0.05 h<sup>-1</sup>.  $\mu$  is significantly lower than specific growth rates shown by microorganisms, which makes the industrial process costlier [67]. The four spinners achieved a  $\mu_{max}$  compatible with the optimized cultures, although the spinners had poor process control, which means that the cultures grew satisfactorily.

Every day a sample of each spinner was taken in a microtube and centrifuged at 800 rpm for 5 min. Then the supernatant was injected in a YSI 2700 Select Dual-Channel Biochemistry Analyzer (Yellow Springs Instruments, Yellow Springs, OH) to determine the residual glucose and lactate production concentration. The results for metabolite consumption are shown in Table 21.3.

Spinner	$\mu_{max}$ (h <sup>-1</sup> )	Days Postinfection of $\mu_{max}$
Negative control	0.010	1
Positive control	0.013	2
Multiplicity of infection 0.01	0.014	5
Multiplicity of infection 0.03	0.020	1

 Table 21.2
 Maximum Specific Growth Rate for Four Spinners

**Table 21.3** Minimum Glucose Concentration and Maximum LactateProduction for Culture Cells in Four Spinners During 10 Days of Infection

Spinner	Minimum Glucose Concentration	Maximum Lactate Concentration
Negative control	1.20 g/L (8)	2.33 g/L (8)
Positive control	2.63 g/L (6)	1.42 g/L (7)
Multiplicity of infection 0.01	2.04 g/L (5)	1.81 g/L (5)
Multiplicity of infection 0.03	2.78 g/L (8)	1.14 g/L (5)

Numbers in parentheses are days postinfection when viability was achieved.

Animal cells need glucose and glutamine as carbon sources. In glucose metabolism practically all glucose is converted into lactose. Thus, the relation between glucose consumption and lactate production represents an important parameter for process monitoring and provides information about the metabolic flow inside cells [67,68]. Cruz and collaborators studied the effects of lactate inhibition on BHK cells [69]. When lactate concentration achieved 28 mM (which corresponds to 2.52 g/L), cell growth was reduced [61]. Therefore, in none of the spinners did the lactate reach a limiting concentration. In addition, in none of the spinners was the glucose totally consumed.

The direct immunofluorescence test was adapted from Batista and collaborators [70]. The lamina was observed in an Olympus immunofluorescence microscope with  $\times 100$  magnification. Samples were evaluated in terms of fluorescent focus inside the cells. The results are listed in Table 21.4.

Table 21.5 summarizes results for the infection rate found in the three infected spinners. Fig. 21.6 shows examples of the classification of the infection rate. However the infection rate did not reflect the production of viruses inside the cells and the quality of viruses produced. Direct immunofluorescence is a tool for process control and classifies

Class	Requirements
0	Total absence of fluorescent focuses
1	Form rare fluorescent focuses to 25% fluorescent focuses
2	Form 25–50% fluorescent focuses
3	Form 50–75% fluorescent focuses
4	>75% fluorescent focuses

Table 21.4Classes and Requirements forEvaluate Direct Immunofluorescence

 Table 21.5
 Infection Rates of Virus Inside Cells

Days Postinfection	Positive Control	Multiplicity of Infection 0.01	Multiplicity of Infection 0.03
1	Class 1	Class 1	Class 1
2	Class 3	Class 2	Class 4
3	Class 4	Class 4	Class 4
4	Class 4	Class 4	Class 4
5	Class 4	Class 4	Class 4
6	Class 4	Class 4	Class 4
7	Class 4	Class 4	Class 3
8	Class 3	Class 4	Class 3
9	Class 3	Class 4	Class 3
10	Class 3	Class 4	Class 4



FIGURE 21.6 Examples of classes in direct immunofluorescence assay. Cells are stained in red; virus is green. Magnification of images is 50 times; evaluation is performed 100 times. (A) Class 1: from rare focus to 25% of fluorescent focuses (Pos. Control at dpi 1); (B) Class 2: 25–50% of fluorescent focuses (MOI 0.01 at dpi 2); (C) Class 3: 50–75% of fluorescent focuses (Pos. Control at dpi 2); (D) Class 4: more than 75% fluorescent focuses (MOI 0.03 at dpi 4).

only the infection rate inside the cells. This technique can be used to predict on which dpi the harvests can be initiated. Normally when the culture reaches Class 4 it means that harvests can be made. This happened for all spinners at dpi 3, and on that day the harvests were initiated.

The RABV titration of samples was made following the Fuches protocol, which was previously validated by Moura and colleagues [71,72]. The virus titration is an important tool for evaluating viral production and quality at several stages of vaccine production [63]. It is a relatively fast in vitro test (compared with other in vivo tests) to evaluate the infection of virus produced in other cells and to compare the results with a standard virus. Samples for different dpis were centrifuged at 800 rpm for 5 min before distribution on a microplate and only the supernatants were used in the titration. Three microplates with 96 wells were made, each with 10 samples (one for each dpi) of one infection, in duplicate.

The virus samples from the three infected spinners collected during the 10 days of infection were titrated. The titration for the standard virus was  $10^{5.37}$  FFD/mL. The maximal titration value reached by positive control was  $10^{5.98}$  FFD/mL at dpi 4. The

maximum for MOI 0.01 was  $10^{5.37}$  FFD/mL at dpi 3, and for MOI 0.03 was  $10^{5.77}$  FFD/mL at dpi 4. Harvests of virus from infected spinners were made from dpi 3–8 and kept in an ultrafreezer at  $-80^{\circ}$ C.

The harvests are used for vaccine production, usually from a bioreactor. After several attempts to produce a vaccine with great potency using different titers, it was decided to use a cutoff at  $10^{3.8}$  FFD/mL and that harvests with a virus titer lower than the cutoff should not be used to produce a rabies vaccine. Fig. 21.7 shows the titration curve for the three infected spinners along with the cutoff.

On the days when harvests occurred, all of the infected spinners had six harvests above the cutoff. However, the positive control and MOI 0.03 obtained two more harvests above the cutoff that were not collected. Between the harvests above cutoff, it can be seen that the positive control had the largest graphic area above the cutoff.

This experiment was able to tell the influence in the different MOIs in the spinners and compare the results with an uninfected spinner. However it was important to obtain growth curves of the four spinners to analyze and compare the infected cultures with the negative control in terms of kinetics, specific growth rate, and metabolite consumption and production. Based on all of the information shown and calculated, it can be concluded that the positive control is the best inoculum for spinner infection. This viral inoculum is made by adding a viral volume corresponding to 1% of the volume of the vessel used in the experiment, which is the best way to infected a BHK-21 culture with RABV.



**FIGURE 21.7** Virus titration curves for the three infected spinners. Only harvests above cutoff are used for rabies vaccine production. The maximum titration value reached by positive control was 105.98 FFD/mL at dpi 4. The maximum for MOI 0.01 was 105.37 FFD/mL at dpi 3, and for MOI 0.03 was 105.77 FFD/mL at dpi 4. The best graphic area above cutoff means the best inoculum. In this case, the best inoculum corresponds to the positive control.

#### 21.3.1.2 Freeze-Dry Study

Harvests from the positive control were used for lyophilization experiments. However it is necessary to inactivate the harvest and concentrate it before its formulation and use in lyophilization.

The harvests that had titration above the cutoff were merged to form a pool of RABV. The mixture was inactivated using BPL and became an inactivated viral suspension (IVS).

Harvests from the spinner usually have some cellular debris and other undesirable metabolites. Thus it was necessary to clarify this IVS, washing it with NT (NaCl and Tris) buffer solution. The NT buffer used in the exchange is composed by NaCl 130 mmol/L and Tris-HCl 50 mmol/L, with pH adjusted to 7.6. Before the buffer exchange, it was necessary to concentrate the IVS to improve the wash results, and to use a minor amount of buffer solution.

The buffer exchange step was performed in a Pellicon tangential filtration system with 100 kDa of porosity (Merck-Millipore, Billerica, MA). The membrane was coupled with a peristaltic pump set up for 55 rpm. First, the IVS were concentrated from 2 L to 400 mL, discarding permeate. Next, buffer was continued added to the IVS to wash, totalizing five times volume, also discarding permeate. Finally, the volume was reestablished with the same buffer and thimerosal was added to preserve the IVS. The IVS were stored at cold room.

For lyophilization experiments, 11 excipients were selected. A summary of the selected excipients, their respective classes, and their estimated prices is given in Table 21.6.

Excipients were diluted in purified water (Milli-Q) for lyophilization experiments and prepared concentrated two times to prepare stock solutions of each excipient, which were used during all experiments. For sugars and polyalcohols, the stock solution was

Excipients	Туре	Estimated Cost (US\$) (per kg)
Sucrose	Sugar	\$86
Trehalose	Sugar	\$4211
Lactose	Sugar	\$193
Sorbitol	Polyalcohol	\$55
Mannitol	Polyalcohol	\$232
Proline	Amino acid	\$1263
Serine	Amino acid	\$1280
Glutamic acid	Amino acid	\$162
Dextran 40	Polymer	\$4522
Polyethylene glycol 3350	Polymer	\$254
Gelatin	Polymer	\$142

Table 21.6 Type and Cost of Selected Excipients

10% (w/v) concentrated, and for amino acids and polymers it was 2% (w/v) concentrated.

Each vaccine is a combination of IVS with a different excipient solution. The vaccine formulation was done by mixing together the IVS and the excipients at 1:1 proportion, homogenizing it and making 1-mL aliquots in 7.5-mL glass vials. The vials were partially closed with a rubber stopper to permit water vapor from sublimation to escape. Five vials were prepared for each excipient solution. An additional of five 1-mL vials were prepared with only the IVS diluted in purified water. This unprotected formulation was used to compare results.

The lyophilization experiments were developed in a freeze-dryer (Edwards RC-300; Edwards Vacuum, Crawley, West Sussex, UK). After preparation of the partially closed glass vials with vaccine formulations, the vials were disposed in the trays of the equipment. The trays were placed inside the freeze-dryer and the protocol was started.

The correct way to conduct this experiment would be to test each formulation separately, with more control parameters to optimize the protocol for each formulation. However, this experiment was made to determine the effects of different excipients in the freeze-dried cakes from rabies vaccines, so it was necessary to freeze-dry all formulations together using the same IVS and protocol.

The protocol (Table 21.7) was established based on other experiments related to lyophilization. It is useful to see the different times for freezing, primary drying, and secondary drying. The vaccines were not frozen before lyophilization, and parameters such as product temperature and DSC were not made. This protocol was used to freezedry the different vaccine formulations.

After freeze-drying occurred, the lyophilized cakes for the 12 formulations were obtained (Figs. 21.8 and 21.9).

The lyophilized vials were visually evaluated using three parameters: color, integrity, and uniformity. The color of the IVS with no excipient was white, so the excipient must not make modifications in the cake color to pass the color test. The cakes must have uniform mass distribution to pass the uniformity test. The cakes must not present cracks to pass the integrity test. Table 21.8 summarizes the visual aspects results for color, uniformity, and integrity.

Step	Temperature (°C)	Duration (h)
Freezing	-42	3
Primary drying	-40 to 5	24—96
Secondary drying	+25	3

 Table 21.7
 Description of Lyophilization Protocol



**FIGURE 21.8** Lyophilized cakes, part 1. (A) IVS without excipient; (B) lactose 5%; (C) mannitol 5%; (D) sucrose 5%; (E) sorbitol 5%; (F) trehalose 5%. All sugars resulted in good cakes as well as mannitol, the IVS without excipient formed a heterogeneous cake and the sorbitol cake collapsed.



**FIGURE 21.9** Lyophilized cakes, part 2. *G*, glutamic acid 1%; *H*, serine 1%; *I*, proline 1%; *J*, dextran 40 1%; *K*, gelatin 1%; *L*, PEG 3350 1%. All amino acids formed collapsed cakes, as well as PEG 3350. Dextran 40 formed an excellent cake and gelatin presented a heterogeneous cake.

Lyophilization of amino acids as excipients did not produce good cakes. The samples presented collapsed during lyophilization (a process also known as melt back). The same phenomenon happened with sorbitol. PEG 3350 cakes also melted back but they appeared to be drier than the sorbitol and amino acids cakes. IVS and gelatin did not pass the uniformity test because bubbles appeared. All excipients did not aggregate color in the cake and passed the color test. The best excipients in this step were sucrose, lactose, trehalose, and dextran.

Excipient	Color	Uniformity	Integrity
Pure inactivated viral suspension	Х		Х
Sucrose	Х	Х	Х
Trehalose	Х	Х	Х
Lactose	Х	Х	Х
Sorbitol	Х		
Mannitol	Х	Х	Х
Proline	Х		
Serine	Х		
Glutamic acid	Х		
Dextran 40	Х	Х	Х
Polyethylene glycol 3350	Х		
Gelatin	Х		Х

 Table 21.8
 Visual Aspects Evaluation of Lyophilized Cakes

X, cakes that passed the tests.

Excipient	Time of Reconstitution	Final pH
Pure inactivated viral suspension	X	7.58
Sucrose	Х	7.52
Trehalose	Х	7.56
Lactose	Х	7.54
Sorbitol	Х	7.55
Mannitol	Х	7.58
Proline	Х	7.56
Serine	Х	7.32
Glutamic acid	Х	3.47
Dextran 40	Х	7.56
Polyethylene glycol 3350	Х	7.57
Gelatin		7.4

Table 21.9 Time and pH Evaluation of Reconstituted Cakes

X, formulations that achieved the specified time of reconstitution.

Then the cakes were reconstituted with purified water. Two other parameters were evaluated under this condition: time of reconstitution and pH. The time of total reconstitution of cakes must not exceed 1 min. Results are shown in Table 21.9.

The only excipient that did not pass the time of the reconstitution test was gelatin, which did not solubilize in purified water in less than 1 min. Although the other excipients passed, the time of reconstitution was better in pure IVS, sucrose, mannitol, and dextran. Lactose and trehalose took awhile longer to become homogenized. The excipients with worse cakes, like all amino acids and PEG, took more time for reconstitution but achieved total homogenization in less than 1 min.
Exciniont	DM1	DMO	DM2	Moon Pacidual Maistura
		RIVIZ		
Pure inactivated viral suspension	4.16	3.99	2.73	$3.63\% \pm 0.78\%$
Sucrose	2.30	2.98	2.32	$2.53\% \pm 0.39\%$
Trehalose	2.51	3.33	3.18	$3.01\% \pm 0.44\%$
Lactose	2.34	2.86	3.28	$2.83\% \pm 0.47\%$
Sorbitol	NC	NC	NC	NC
Mannitol	1.93	2.78	2.72	$2.48\% \pm 0.47\%$
Proline	NC	NC	NC	NC
Serine	NC	NC	NC	NC
Glutamic acid	NC	NC	NC	NC
Dextran 40	3.15	2.95	2.74	$2.95\% \pm 0.21\%$
Polyethylene glycol 3350	NC	NC	NC	NC
Gelatin	3.42	3.10	2.85	$3.12\% \pm 0.29\%$

 Table 21.10
 Residual Moisture Results for Lyophilized Cakes

NC, sample was not calculated.

The Brazilian regulatory agency (MAPA) demands that the reconstituted vaccine have a pH between 6.8 and 8.5. The only excipient that did not pass the pH test was glutamic acid, which turned the vaccine more acid than was recommended [73].

The residual moisture test was done by volumetric Karl Fischer titration using Titroline KF equipment (Schott SI Analytics, Mainz, Rhineland-Palatinate, Germany). The three best lyophilized vials of each formulation were selected for the test. The samples of each vial were weighted and titrated using Karl Fischer equipment. Table 21.10 shows the results for residual moisture.

Cakes obtained for the amino acids and PEG 3350 could not be removed from lyophilized vials for injection in the titration equipment because the cakes were melted back and difficult to collect. Therefore, the residual moisture of these excipients could not be calculated. MAPA recommends that residual moisture not pass 3% [73]. The IVS without excipient addition and gelatin did not pass the test. The trehalose was slightly above that recommended but because the test is influenced by many factors (temperature, operating time, and air humidity), it can be said that trehalose passed the test; furthermore, the lyophilized trehalose cake will be reevaluated. The other excipients analyzed had residual moisture under regulatory claims.

The IVS with no excipient formed a cake that passed in the visual aspects tests and almost passed the residual moisture test. This happened, because there could still be traces of sugar, fetal bovine serum, amino acids and other metabolites that probably stabilized the IVS during lyophilization. Lang and Winter demonstrated that sodium chloride interferes in lyophilization in concentrations over 60 mmol/L by aggregating molecules [55]. The sodium chloride concentration in NT buffer is 50 mmol/L, so the presence of this salt only improves the tonicity of the vaccine.

The amino acids cakes melted back. It probably occurred because of its ineffectiveness in protecting the IVS when used as the only excipient, even though the excipients were present in the pure IVS. It may have better results when associated with excipients of a different kind. Probably with the addition of amino acids, the Tg would be lower.

The book *Laboratory Techniques in Rabies* describes two protocols using gelatin as an excipient [26]. In this experiment, gelatin provided good cakes but after reconstitution it did not homogenize within the desired time.

Trehalose was described as a good cryoprotector and lyoprotector by Gupta and collaborators and Sarkar and collaborators for a respiratory syncytial virus and peste des petits ruminants vaccine, respectively [74,75]. Caricati also tested different excipients for rabies vaccine lyophilization and obtained interesting results for trehalose, sucrose, and PEG 3350 [76]. This experiment also had good results for trehalose and sucrose in terms of visual aspects and residual moisture. However PEG 3350, which is used as a stabilizing agent in lyophilization, had poor results. This could be because PEG is usually coupled with other excipients to serve as a stabilizing agent; in this experiment, it was used alone.

Lang and Winter also tested mannitol as an excipient. When used alone, mannitol had good results for appearance and residual moisture. When coupled with trehalose, the resultant cake had a robust crystalline matrix, leading to a more elegant final product and enabling shorter drying times [55].

Sucrose at concentrations of 1–68% has frequently been used for the cryopreservation of microorganisms [56]. Law and Hull tested the stabilizing effect of sucrose for respiratory syncytial virus and its infectivity, and obtained good results [77]. Sehgal and Das studied the effect of freezing on the bean mosaic virus using sucrose as a stabilizer, and also obtained good results [78].

Dextran, sorbitol, and mannitol are inert excipients. They provide elegance to lyophilized cakes because they easily crystallize under low temperatures and are employed in biopharmaceutical vaccines that use lower doses per vial and consequently have lower mass. In these cases, the inert excipients provide more volume and can be coupled with amorphous excipients generating a two-phase solid: one amorphous with the active principle and one crystalline with the inert excipient.

#### 21.3.2 Conclusions

The growth curves were determined for four BHK-21 cells cultivated in spinners under different viral inocula. Parameters were calculated and discussed to comprehend cell growth before and after the infection and how the initial quantity of virus influenced cell growth. The virus titration showed that the positive control (1% of spinner volume) was the best viral inoculum.

Furthermore, the vaccine was formulated and freeze-dried with different excipients. The IVS with no excipient alongside trehalose, sucrose, lactose, mannitol, and dextran 40 presented lyophilized cakes with the best appearance. This experiment shows the trialand-error method, a common practice when freeze-drying a vaccine. However, the correct approach is the rational design along with better parameter controls to truly understand the necessities of the antigen to remain active. A rational design would help choose the best excipient before freeze-drying as well as design an adapted lyophilization protocol, which would result in a better product to be tested in vivo.

# 21.4 Perspectives

RABV is a well-known virus that has been studied for more than a century. The infection mechanisms are well-described and a large number of vaccines are available for human and veterinary use for pre- and postexposure to the virus. If the potency of a vaccine could be measured by some mathematical model that matches results with biological tests, it would enhance production and eliminate in vivo tests. Novel methods have been proposed to substitute potency tests as well as mathematical models to predict the effectiveness of vaccines; however in most cases in vitro and in vivo results do not match. Therefore these tests and models need to be improved to substitute biological tests in the future.

Freeze-drying is currently the most commonly used drying method to improve the stability of biopharmaceuticals. Since its beginning, most efforts to improve the process focus on increase drying rates to reduce residual moisture, decrease the cycle duration to optimize time, and enhance process efficiency to improve production. One important part that can no longer be overlooked is the characterization and formulation of vaccines to improve the rational design of products, which in turn improves the stability of vaccines, the main objective when freeze-drying. Also, effort must be spent on improving equipment and in some cases finding alternatives to freeze-drying to obtain pharmaceutical products that are safer, more effective, and cheaper.

# References

- [1] WHO World Health Organization, Rabies: Countries or Areas at Risk, 2013. Available in: http://www.who.int/rabies/Global\_distribution\_risk\_humans\_contracting\_rabies\_2013.png?ua=1.
- WHO World Health Organization, WHO Expert Consultation on Rabies (Second Report), 2013. WHO Technical Report Series 982. Available in: http://apps.who.int/iris/handle/10665/ 85346.
- WHO World Health Organization, Rabies Vaccines: WHO Position Paper, Weekly Epidemiological Record 32 (2010). Available in: http://www.who.int/wer/2010/wer8532.pdf.
- [4] N. Johnson, A.F. Cunningham, A.R. Fook, The immune response to rabies virus infection and vaccination, Vaccine 28 (2010) 3896–3901.
- [5] F.A. Murphy, C.M. Fauquet, D.H.L. Bishop, S.A. Ghabrial, A.W. Jarvis, G.P. Martelli, M.A. Mayo, M.D. Summers, Virus taxonomy: classification and nomenclature of viruses, in: F.A. Murphy (Ed.), Sixth Report of the International Committee on Taxonomy of Viruses, Springer-Verlag, New York, 1995.
- [6] M.C. Davies, M.E. Englert, G.R. Sharpless, V.J. Cabasso, The electron microscopy of rabies virus in cultures of chicken embryo tissues, Virology 21 (1963) 642–651.

- [7] W.H. Wunner, The chemical composition and molecular structure of rabies viruses, in: G.M. Baer (Ed.), The Natural History of Rabies, CRC Press, Boca Raton, 1991, pp. 31–67.
- [8] W.H. Wunner, Rabies virus, in: A.C. Jackson, W.H. Wunner (Eds.), Rabies, second ed., Academic Press, New York, 2007, pp. 23–69.
- [9] H.B.C.R. Batista, A.C. Franco, P.M. Roehe, Raiva: uma breve revisão (In Portuguese), Acta Scientiae Veterinariae 35 (2007) 125–144.
- [10] S.A. Nadin-Davis, Rabies: Virus and Disease, 2010. Available in: http://www.els.net/WileyCDA/ ElsArticle/refId-a0002244.html.
- [11] J.A.M. Hirose, A.E.M. Villegas, Estructura antigenica y mecanismos de infeccion del virus de la rabia (In Spanish), Ciencia Veterinaria 7 (1996) 67–102.
- [12] WHO World Health Organization, Vaccine Definition, 2015. Available in: http://who.int/topics/ vaccines/en/.
- [13] T.O. Bunn, Canine and feline rabies vaccines, past and present, in: G.M. Baer (Ed.), The Natural History of Rabies, CRC Press, Boca Raton, 1991, pp. 415–425.
- [14] E. Fuenzalida, R. Palacios, Un metodo mejorado para la preparacion de la vacuna antirrábica (In Spanish), Boletín del Instituto de Bacteriología de Chile 8 (1955) 3–11.
- [15] F.B. Peck, H.M. Powell, C.G. Culbertson, A new antirabies vaccine for human use, clinical and laboratory results using rabies vaccine made from embryonated duck eggs, Journal of Laboratory and Clinical Medicine 45 (1955) 679–683.
- [16] P. Fenje, A rabies vaccine from hamster kidney tissue cultures: preparation and evaluation in animals, Canadian Journal of Microbiology 6 (1960) 605–609.
- [17] M. Roumiantzeff, N. Ajjan, B. Montagnon, J.C. Vincent-Falquet, Rabies vaccines produced in cell culture, Annales de l'Institut Pasteur/Virologie 136 (1985) 413–424.
- [18] P. Sureau, Contribution to rabies prevention, Vaccine 10 (1992) 896-899.
- [19] WHO World Health Organization, WHO position paper on rabies vaccines, Weekly Epidemiological Record 85 (2010) 309–320.
- [20] L.A. Wilber, M.F.A. Aubert, The NIH test for potency, in: F.-X. Meslin, M.M. Kaplan, H. Koprowski (Eds.), Laboratory Techniques in Rabies, fourth ed., WHO Library, Geneva, 1996, pp. 360–368.
- [21] WHO World Health Organization, Grading of Scientific Evidence, Table III. Safety of Cell-culturebased Rabies Vaccines, 2010. Available in: http://www.who.int/entity/immunization/rabies\_grad\_ safety.pdf.
- [22] A. Strady, J. Lang, M. Lienard, C. Blondeau, R. Jaussaud, S.A. Plotkin, Antibody persistence following preexposure regimens of cell-culture rabies vaccines: 10-year follow-up and proposal for a new booster policy, Journal of Infectious Diseases 177 (1998) 1290–1295.
- [23] WHO World Health Organization, The Immunological Basis for Immunization Series, Module 17: Rabies, WHO Library, Geneva, 2011.
- [24] WHO World Health Organization, Guide for Post-exposure Prophylaxis, 2012. Available in: http:// www.who.int/rabies/human/postexp/en/.
- [25] J. Lang, P. Attanath, B. Quiambao, V. Singhasivanon, P. Chanthavanich, C. Montalban, C. Lutsch, S. Pepin-Covatta, V. Le Mener, M. Miranda, A. Sabchareon, Evaluation of the safety, immunogenicity, and pharmacokinetic profile of a new, highly purified, heat-treated equine rabies immunoglobulin, administered either alone or in association with a purified, Vero-cell rabies vaccine, Acta Tropica 70 (3) (1998) 317–333.
- [26] P. Reculard, Cell-culture vaccines for veterinary use, in: F.-X. Meslin, M.M. Kaplan, H. Koprowski (Eds.), Laboratory Techniques in Rabies, fourth ed., WHO Library, Geneva, 1996.

- [27] P. Precausta, J.-P. Soulebot, Vaccines for domestic animals, in: G.M. Baer (Ed.), The Natural History of Rabies, CRC Press, Boca Raton, 1991, pp. 445–459.
- [28] D.J. Briggs, D.W. Dressen, W.H. Wunner, Vaccines, in: A.C. Jackson, W.H. Wunner (Eds.), Rabies, first ed., Academic Press, New York, 2002, p. 379.
- [29] J. Furey, Continuous cell culture using the ATF system, Genetic Engineering News 20 (2000) 52–53.
- [30] V. Racaniello, Multiplicity of Infection, 2011. Available in: http://www.virology.ws/2011/01/13/ multiplicity-of-infection/.
- [31] P. Perrin, S. Madhusudana, C. Gontier-Jallet, S. Petres, N. Tordo, O.-W. Merten, An experimental rabies vaccine produced with a new BHK-21 suspension cell culture process: use of serum-free medium and perfusion-reactor system, Vaccine 13 (1995) 1244–1250.
- [32] H. Kallel, A. Jouini, S. Majoul, S. Rourou, Evaluation of various serum and animal protein free media for the production of a veterinary rabies vaccine in BHK-21 cells, Journal of Biotechnology 95 (2002) 195–204.
- [33] K. Trabelsi, S. Rourou, H. Loukil, S. Majoul, H. Kallel, Comparison of various culture modes for the production of rabies virus by Vero cells grown on microcarriers in a 2-L bioreactor, Enzyme and Microbial Technology 36 (2005) 514–519.
- [34] K. Trabelsi, S. Rourou, H. Loukil, S. Majoul, H. Kallel, Optimization of virus yield as a strategy to improve rabies vaccine production by Vero cells in a bioreactor, Journal of Biotechnology 121 (2006) 261–271.
- [35] R. Guidolin, M.C. Baltazar, F. Zelante, Produção de vacina antirrábica veterinária em suspensão de células BHK (In Portuguese), Revista de Microbiologia 14 (1983) 27–35.
- [36] T.A. Jennings, Introduction, in: Lyophilization: Introduction and Basic Principles, Informa Healthcare, New York, 2008, pp. 1–13.
- [37] V. Lavakumar, G.R. Nireesha, L. Divya, C. Sowmya, N. Venkateshan, M.N. Babu, Lyophilization/ Freeze drying – a review, International Journal of Novel Trends in Pharmaceutical Sciences 3 (4) (2013) 87–98.
- [38] H.A. Lieberman, L. Lachman, B.J. Schwartz, Pharmaceutical Dosage Form: Tablets, vol. I, Marcel Dekker, New York, 1989, p. 1.
- [39] L. Rey, Glimpses into the realm of freeze-drying: classical issues and new ventures, in: L. Rey, J.C. May (Eds.), Freeze Drying/Lyophilization of Pharmaceutical and Biological Products, third ed., Informa Healthcare, New York, 2010, pp. 1–28.
- [40] M. Saclier, R. Peczalski, J. Andrieu, Effect of ultrasonically induced nucleation on ice crystals' size and shape during freezing in vials, Chemical Engineering Science 65 (10) (2010) 3064–3071.
- [41] J. Rexroad, C.M. Wiethoff, L.S. Jones, C.R. Middaugh, Lyophilization and the thermostability of vaccines, Cell Preservation Technology 1 (2) (2002) 91–104.
- [42] S. Khairnar, R. Kini, M. Harwalkar, K. Salunkhe, S.R. Chaudha, A review on freeze drying process of pharmaceuticals, International Journal of Research in Pharmacy and Science 4 (1) (2013) 76–94.
- [43] S. Rambhatla, M.J. Pikal, Heat and mass transfer scale-up issues during freeze drying, I: atypical radiation and the edge vial effect, AAPS Pharmaceutical Science Technology 4 (2) (2003) 114–127.
- [44] T. Lam, R.G. Strickly, G.C. Visor, An unexpected pH effect on stability of moexipril lyophilized powder, Pharmaceutical Research 6 (2004) 971–975.
- [45] J.A. Searles, Freezing and annealing phenomena in lyophilization, in: L. Rey, J.C. May (Eds.), Freeze Drying/Lyophilization of Pharmaceutical and Biological Products, third ed., Informa Healthcare, New York, 2010, p. 55.

- [46] P. Sheehan, A.I. Liapis, Modeling of the primary and secondary drying stages of the freeze drying of pharmaceutical products in vials: numerical results obtained from the solution of a dynamic and spatially multi-dimensional lyophilization model for different operational policies, Biotechnology Bioengineering 60 (1998) 712–728.
- [47] M.J. Pikal, Freeze-drying of proteins I: process design, BioPharm 9 (1990) 18-27.
- [48] R.H.M. Hatley, F. Franks, Applications of DSC in the development of improved freeze-drying processes for labile biologicals, Journal of Thermal Analysis and Calorimetry 37 (1991) 1905–1914.
- [49] M.J. Hawe, P. Fries, The impact of the freezing stage in lyophilization: effects of the ice nucleation temperature on process design and product quality, American Pharmaceutical Review 5 (2002) 48–53.
- [50] A.A. Smith, Y.N. Krishna, S.G.K. Parimalakrishnan, J. Srikanth, Formulation and evaluation of tigecycline injection by lyophilization, Der Pharmacia Lettre 4 (5) (2012) 1426–1430.
- [51] T.A. Jennings, The freezing process, in: Lyophilization: Introduction and Basic Principles, Informa Healthcare, New York, 2008, pp. 261–281.
- [52] N. Tsinotides, D.S. Baker, The importance of freezing on lyophilization cycle development, Asian Journal of Biomedical and Pharmaceutical Sciences 19 (2002) 16–21.
- [53] T. Pikal, A.J. Cannor, E.H. Trappier, Optimization of lyo cyles and production of mathematical models, European Journal of Pharmaceutical Sciences 54 (2004) 132–154.
- [54] W. Wang, Lyophilization and development of solid protein pharmaceuticals, International Journal of Pharmaceuticals 203 (2000) 1–60.
- [55] R. Lang, G. Winter, Rational design of a stable, freeze-dried virus-like particle-based vaccine formulation, Drug Development and Industrial Pharmacy 35 (2009) 83–97.
- [56] Z. Hubálek, Protectants used in the cryopreservation of microorganisms, Cryobiology 46 (2003) 205–229.
- [57] T.A. Jennings, Effect of vacuum freeze-dryers: present and future, in: Lyophilization: Introduction and Basic Principles, Informa Healthcare, New York, 2008, pp. 549–619.
- [58] E.E. Worrall, J.K. Litamoi, B.M. Seck, G. Ayelet, Xerovac: an ultra-rapid method for the dehydration and preservation of live attenuated Rinderpest and Peste des Petit ruminants vaccines, Vaccine 19 (2000) 834–839.
- [59] R.P. Aquino, M. Stigliani, P. Del Gaudio, T. Mencherini, F. Sansone, P. Russo, Nanospray drying as a novel technique for the manufacturing of inhalable NSAID powders, The Scientific World Journal 2014 (2014) 1–7.
- [60] S. Wanninga, R. Süverkrüpa, A. Lamprechta, Pharmaceutical spray freeze drying, International Journal of Pharmaceutics 488 (1) (2015) 136–153.
- [61] M. Gupta, S. Prasad, V. Meda, Microwave Vacuum Drying of Fruits and Vegetables, 2015. Available in: http://www.on-line-foods.com/tech\_paper/mohitgupta.pdf.
- [62] S.P. Sellers, G.S. Clark, R.E. Sievers, J.F. Carpenter, Dry powders of stable protein formulations from aqueous solutions prepared using supercritical CO<sub>2</sub>-assisted aerosolization, Journal of Pharmaceutical Sciences 90 (6) (2001) 785–797.
- [63] T.M. Lina, T.D. Duranceb, C.H. Scamanb, Characterization of vacuum microwave, air and freeze dried carrot slices, Food Research International 31 (2) (1998) 111–117.
- [64] J. Bondaruku, M. Markowskia, W. Błaszczak, Effect of drying conditions on the quality of vacuummicrowave dried potato cubes, Journal of Food Engineering 81 (2) (2007) 306–312.
- [65] H.J. Phillips, Dye exclusion tests for cell viability, in: P.F. Kruse Jr., M.K. Patterson (Eds.), Tissue Culture: Methods and Applications, Academic Press, New York, 1973, pp. 406–408.

- [66] W.G. Chapman, I.A. Ramshaw, J. Crick, Inactivated rabies vaccine produced from the Flury LEP strain of virus grown in BHK-21 suspension cells, Applied Microbiology 26 (1973) 858–862.
- [67] A. Tonso, Monitoramento e operação de cultivos de células animais em sistemas de perfusão (In Portuguese) (Doctoral's thesis), Universidade de São Paulo, São Paulo, 2000.
- [68] A.P. Zeng, W.S. Hu, W.D. Deckwer, Variation of stoichiometric ratios and their correlation for monitoring and control of animal cell cultures, Biotechnology Progress 14 (1998) 434–441.
- [69] H.J. Cruz, C.M. Freitas, P.M. Alves, J.L. Moreira, M.J.T. Carrondo, Effects of ammonia and lactate on growth, metabolism, and productivity of BHK cells, Enzyme and Microbial Technology 27 (2000) 43–52.
- [70] A.M. Batista, P.S. Cruz, E. Almeida, A.E.B. Costa, K.C. Scheffer, L.B. Chaves, A.C.R. Silva, G.M.M. Caporale, Infection of BHK-21 cells cultivated in stationary monolayers by PV and CVS strains (In Portuguese), Boletim Epidemiológico Paulista (BEPA) 6 (2009) 1–7.
- [71] R.M.M. Fuches, Validação de bioensaios para o estudo da imunogenicidade da vacina contra raiva (In Portuguese) (Doctoral's thesis), Universidade de São Paulo, São Paulo, 2010.
- [72] W.C. Moura, N.M.F. Gallina, R.M.M. Fuches, P.C. Romijn, J.P.G. Leite, Validation of a virus neutralization potency test in BHK-21 cells for rabies immunoglobulins in a two-center study, Journal of Virological Methods 154 (2008) 7–13.
- [73] Brasil Ministério da Agricultura, Pecuária e Abastecimento, Portaria N° 228/1988 (In Portuguese), 1988. Available in: http://sistemasweb.agricultura.gov.br/sislegis/action/detalhaAto.do?method= visualizarAtoPortalMapa&chave=58090151.
- [74] C.K. Gupta, J. Leszczynski, R.K. Gupta, G.R. Siber, Stabilization of respiratory syncytial virus (RSV) against thermal inactivation and freeze-thaw cycles for development and control of RSV vaccines and immune globulin, Vaccine 14 (1996) 1417–1420.
- [75] J. Sarkar, B.P. Sreenivasa, R.P. Singh, P. Dhar, S.K. Bandyopadhyay, Comparative efficacy of various chemical stabilizers on the thermostability of a live-attenuated peste de petits ruminants (PPR) vaccine, Vaccine 21 (2003) 4728–4735.
- [76] A.T.P. Caricati, Estratégias do processo para a conservação da potência da vacina rábica de uso veterinário por liofilização (In Portuguese), Master's Dissertation, Universidade de São Paulo, São Paulo, 2012.
- [77] T.J. Law, R.N. Hull, Stabilizing effect of sucrose upon respiratory syncytial virus infectivity, Experimental Biological Medicine 128 (1968) 515–518.
- [78] O.P. Sehgal, P.D. Das, Effect of freezing on conformation and stability of the virions of southern bean mosaic virus, Virology 64 (1975) 180–186.

# 22

# Antileishmanial Biocompound Screening

#### F. Vitola<sup>1, a</sup>, R. Fendrich<sup>1, a</sup>, C.R. Soccol<sup>1</sup>, D. Guedes<sup>1</sup>, A. Pandey<sup>2</sup>, V. Thomaz-Soccol<sup>1, \*</sup>

<sup>1</sup>UNIVERSIDADE FEDERAL DO PARANÁ (UFPR), CURITIBA, PARANÁ, BRAZIL; <sup>2</sup>CENTER OF INNOVATIVE AND APPLIED BIOPROCESSING, (A NATIONAL INSTITUTE UNDER DEPT OF BIOTECHNOLOGY, MINISTRY OF S&T, GOVT OF INDIA), MOHALI, PUNJAB, INDIA

# 22.1 Introduction

Leishmaniosis is a disease that can manifest as a visceral or cutaneous form. The prevalence is estimated to be 12 million people in 98 countries, with about 1.5 million new cases annually. The infection is caused by protozoa of the genus *Leishmania* Ross, 1903 and is transmitted by the bite of female sandflies. Infections in humans are caused by more than 20 species of *Leishmania* [1]. The visceral form is characterized by affected internal organs and the cutaneous form can demonstrate sores on skin and mucosa (cutaneous or mucocutaneous forms). The severity of some manifestations impairs psychological and physical health and has social and economic impacts. Leishmaniosis is considered by the World Health Organization [1] to be the second most important protozoan disease with regard to public health. Available treatments for leishmaniosis have serious toxic side effects. Moreover, the drugs are expensive and there is the possibility of developing resistance.

Derivatives of pentavalent antimony (Pentostam) or meglumine antimoniate (Glucantime) are the reference standard treatment. They require a hospital stay of several weeks and have significant side effects. In addition, many cases of resistance are reported in India. Thus there is increasing use of amphotericin B or miltefosine as a lipid (despite the significant cost), since shorter hospital stay and increased tolerance are often observable.

Many molecules are tested each year to find new treatments for less severe cases. Few of them show in vivo activity and go beyond the preclinical stage of development. However the disease continues to evolve and cases of resistance are worrying, so it is essential to continue research.

<sup>&</sup>lt;sup>a</sup>These authors contributed equally to this work.

<sup>&</sup>lt;sup>\*</sup>Corresponding Author.

# 22.2 Primary Chemotherapy With Antimonials

Organic antimonials were first used a century ago as leishmanicidal compounds by Gaspar Vianna in 1913 for the treatment of cutaneous leishmaniasis (CL). The first commercial drug developed was urea stibamine, in India in 1922, but it had severe toxic effects. Other chemotherapeutics, also antimonials {based on pentavalent antimonials [Sb (+5)]} have been introduced as drugs of choice for treatment and have been used successfully for 60 years [2].

Pentavalent antimony compounds remain the first-line pharmaceuticals of choice for the treatment of leishmaniosis [3]. The compounds (Fig. 22.1A and B) that have this active principle have the trademarks names Glucantime (*N*-methyl meglumine antimoniate; manufactured by Aventis Pasteur) and Pentostam (sodium stibogluconate; Wellcome, manufactured by GlaxoSmithKline). Glucantime is widely prescribed as the initial treatment for CL and visceral leishmaniosis (VL). It is usually administered by intramuscular injections.

To be active against *Leishmania* spp., Sb (+5) has to enter the infected cell and cross the phagolysosomal membrane to attack intracellular amastigotes. To combat the infection, Sb (+5) must be reduced to the active form of the drug against the parasite, the trivalent antimonial [Sb (+3)] [4,5]. The whole molecular mechanism of antimonials, however, is not fully understood. Antimony compromises the thiol redox potential of the cell by inducing the efflux of intracellular thiols and by inhibiting trypanothione reductase [6].

However, pentavalent antimonials have toxic side effects and limited efficacy to control parasite proliferation, and drug resistance is frequently encountered [9,47]. Although its effectiveness still saves lives [10], it is an aggressive substance as a chemotherapeutic. This drug causes serious side effects such as cardiac, hematologic, liver, pancreatic, and renal toxicities that can result in death [11]. Because of the aggressiveness of the side effects, treatment often stops before the cure occurs, thereby contributing indirectly to the worldwide increasing frequency of resistance to antimonials [12,13]. Chemotherapeutic regimens for treating parasitic diseases have been derived empirically and are not particularly selective for the metabolic machinery of the parasite. Unsurprisingly, therefore, antiparasitic drugs are noted for causing moderate to severe toxicity in the human host [14].

Since the 1940s, a serious public health problem has been described: resistance to antimonials by some strains of *Leishmania* [15]. Although pentavalent antimonials still have their basis in treatment, resistance to this class of drug is becoming so high in some parts of the world that it is quickly becoming obsolete [13,16]. For example, the treatment of kala-azar by conventional dosages has failed particularly in northeast India since the 1970s, with 60% unresponsiveness reported [17].

Adenosine triphosphate—binding cassette (ABC) transporters mediate Glucantime sequestration in cellular organelles [18]. In leishmaniosis, ABC transporters are responsible for much of the decreased sensitivity to antimonials. Active movement of



**FIGURE 22.1** (A) Meglumine antimoniate (Glucantime). Systematic name: hydroxy-dioxostiborane; (2R,3R,4R,5S)-6-methylamino -hexane-1,2,3,4,5-pentol (B) Sodium stibogluconate (Pentostam). Systematic name: 2,4:2',4'-O-(oxydistibylidyne) bis [D-gluconic acid] Sb,Sb'dioxide trisodium salt nonahydrate (C) Amphotericin B. Systematic name: (1R,3S,5R,6R,9R,11R,15S,16R,17R,18S,19E,21E,23E,25E,27E,29E,31E,33R,35S,36R,37S)-33-[(3-amino-3,6dideoxy-β-D-mannopyranosyl)oxy]-1,3,5,6,9,11,17,37-octahydroxy-15,16,18-trimethyl-13-oxo14,39-dioxabicy-clo [33.3.1] nonatriaconta-19,21,23,25,27,29,31-heptaene-36-carboxylic acid. Adapted from: IUPAC (International Union of Pure and Applied Chemistry), Organic and Biochemical Nomenclature Recommendations (2010). Available at: www.iupac.org, and Disanejercito, Direccion de Sanidad del Ejercito Salud Operacional, Leishmaniosis (2008). Available on: http://www.disanejercito.mil.co/Salud\_Operacional\_Leishmaniosis.htm.).

drugs across membranes through these transporters is a cause of therapy failure. Previous reports evidenced that the augmented expression of ABC transporters enhances drug extrusion and disposition [4] and constitutes one of the main impediments to successful chemotherapy against parasites of the *Leishmania* genus [19].

Administration of therapeutic drugs combined with ABC transporter blockers is one way to actively avoid the increasing frequency of drug resistance. Glibenclamide (specific blocker) enhances the Glucantime effect to decrease the infection rate of macrophages [18]. Furthermore, the selective inhibition increases the cytoplasmic concentration of Glucantime and efficacy. However further investigations into some arsenite-resistant strains revealed that increased activity and expression of topoisomerase II (necessary for kinetoplast DNA) induces a multidrug-resistant phenotype [20,21] (Fig. 22.1A and B)

# 22.3 Second Generation of Antileishmanial Drugs

Pentavalent antimony compounds have been the mainstay of leishmanicidal therapy for half a century [3], but over the past decade, alternative drugs or new formulations of other standard drugs have become available and registered for use in some countries, whereas other drugs are under clinical trial for both forms of the disease [22–24]. Some representative second-line drugs are presented in the following subsections.

Second-line drugs have not experienced widespread use, mainly owing to the easy development of clinical resistance, toxicity, cost, and possible contribution to the increased incidence of leishmaniosis—HIV co-infections [22]. Similar to antimonial derivatives (Glucantime), drugs such as *bis*-amidines (pentamidine) and polyene macro-lides (amphotericin B) exhibit high liver, kidney, and heart toxicities. Still others, including paromomycin, allopurinol, and sitamaquine have been reported to have variable cure rates. Although miltefosine is effective in some cases, parasites can develop resistance against it [25]. Fluconazole, taken orally, was shown to be effective against CL [13].

Therapies combining second-line antiparasitic substances are currently being considered, especially to avoid the development of parasite resistance. Some drugs that were already tried include atovaquone, roxithromycin, and edelfosine [9]. A tested combination of paromomycin and other leishmanicidal agents with sodium stibogluc-onate in a standard dose gave no optimistic results. Immunotherapy combined with chemotherapeutic agents using stimulating cytokines or vaccines is still is a future promise [17].

#### 22.3.1 Amphotericin B

Amphotericin B (Fig. 22.1C) is a polyene amphoteric antibiotic that binds to cell membranes and alters their permeability, creating pores and leakage of ions [30]. It has been proven to be highly effective for the treatment of resistant strains of *Leishmania* to antimonials, in the form of amphotericin B deoxycholate (Fungizone) [2,26].

Amphotericin B makes complexes with substituted sterols, thus causing the pores that alter ion balance and result in death of the parasite [16].

It was first developed as an antifungal compound, valuable for use in the treatment of systemic fungal infections [27] by interacting with fungi cell membrane sterols and preferentially with ergosterol. Like fungi, *Leishmania* also has ergostane-based sterols as its major membrane sterol [28].

However, amphotericin B is an unpleasant drug because of its toxicity and the need for slow-infusion parenteral administration over 4 h [24]. It was also found to be highly nephrotoxic. Often a serious acute reaction is noted after the infusion, consisting of high fever, shaking chills, hypotension, anorexia, nausea, vomiting, headache, drowsiness, generalized weakness, dyspnea, and tachypnea.

The use of lipid formulations containing amphotericin B has been one of the greatest advances in the treatment of leishmaniosis, minimizing side effects of its use. Various safer colloidal and lipid formulations have been prepared [12]. Lipid formulations of amphotericin B, although expensive and administered parenterally, represent a major advance for treating VL, e.g., liposomal amphotericin B (AmBisome) [29].

The standard dosage of amphotericin B for treating New-World CL is 0.5–1.0 mg/kg per day through 20 days. An urgent technological need for this kind of treatment is an understanding of the immune response to leishmaniosis and field-applicable and affordable methods for diagnosis and treatment. Although introduced more recently, amphotericin B is not widely used because of cost of the treatment with this drug [12].

#### 22.3.2 Pentamidine

Pentamidine, a cationic diamidine (Fig. 22.2A) is a drug used against resistant parasites, but close clinical control of the patients is required because of side effects [30]. Pentamidine accumulates in the mitochondria; indeed, it was shown that this could greatly enhance the efficacy of mitochondrial respiratory chain complex inhibitors [31].

This drug does not accumulate in the mitochondria of in vitro pentamidine-resistant mutants, and the cytosolic fraction of the drug is extruded outside the cell [32]. Similarly to Glucantime, a possible candidate for the required efflux pump is the ABC transporter that was isolated by Coelho et al. [33], while selecting for pentamidine resistance. Acquired resistance has been reported against pentamidine (about 25%), although it was initially used with prolonged dosage as a second-line drug [17].

#### 22.3.3 Miltefosine

Miltefosine (Fig. 22.2B), which belongs to the alkylphosphocholine group, was approved for the treatment of human leishmaniosis [13]. It has been developed for treatment of both VL and CL by Zentaris, Impavido, and the World Health Organization [34]. Miltefosine is the first oral leishmanicidal drug; it was put in use in 2001, although it was originally developed as an anticancer agent [35]. An advantage is that it does not require refrigeration. There is a governmental program in India to face pentavalent antimonial



FIGURE 22.2 (A) Pentamidine. Systematic name: 4,4'-[pentane-1,5-diylbis(oxy)]dibenzenecarboximidamide (B) Miltefosine. Systematic name: 2-(hexadecoxy-oxido-phosphoryl)oxyethyl-trimethyl-azanium (C) Paromomycin. Systematic name: (2R,3S,4R,5R,6S)-5-amino-6-[(1R,2S,3S,4R,6S)-4,6-diamino-2-[(2S,3,4,5R)-4-[(2,3,4,5R,6S)-3-amino-6-(aminomethyl)-4,5-dihydroxy-oxan-2-yl]oxy-3-hydroxy-5-hydroxymethyl)oxolan-2-yl]oxy-3-hydroxy-cyclohexyl]oxy-2-(hydroxymethyl)oxane –3,4-diol. Adapted from: IUPAC (International Union of Pure and Applied Chemistry), Organic and Biochemical Nomenclature Recommendations (2010). Available at: www.iupac.org, and Disanejercito, Direccion de Sanidad del Ejercito Salud Operacional, Leishmaniosis (2008). Available on: http://www.disanejercito. mil.co/Salud\_Operacional\_Leishmaniosis.htm.).

resistance with the administration of Miltefosine, with the hope of eradicating the visceral leishmaniosis form [36].

Miltefosine was effective in experimental and clinical trials. Resistance to this drug is possible. Single point mutations in the *Leishmania* putative miltefosine transporter can lead to resistance [25]. This drug has a potent effect with a longer half-life when orally administered; thus it has high potential to induce parasite resistance [17]. Furthermore, Basu et al. [37] reported a gene isolated from *Leishmania infantum* that encodes a putative polypeptide (without similarities to known proteins) that confers resistance not only against antimonials but also against miltefosine. Other drawbacks are that this drug has teratogenicity, severe gastrointestinal side effects, and a narrow therapeutic window that makes its application restrictive owing to the high cost [38].

#### 22.3.4 Paromomycin

Paromomycin (Fig. 22.2C) is an aminoglycoside antibiotic used for various clinical infections. The spectrum of activity of paromomycin encompasses most gram-negative and many gram-positive bacteria, some protozoa, and many cestodes [36]. Paromomycin has been used to treat both VL in a parenteral formulation and CL in both topical and parenteral formulations [12].

This compound showed efficacy and safety in phase IV clinical trials against leishmaniosis [39]. Paromomycin inhibited the in vitro growth of *Leishmania donovani* promastigotes in a dose-dependent manner. Intracellular macrophage-amastigotes were more sensitive. As with Miltefosine, resistance to paromomycin could be induced experimentally in promastigotes in vitro [40]. The developed resistant strain was threefold less sensitive to paromomycin than the wild type.

Paromomycin has undergone extensive clinical trials in Indian kala-azar patients. Because it is an aminoglycoside, acquired resistance is likely to occur when it is used as a monotherapy. To overcome the problem of treatment failure in kala-azar and reduce the length of therapy, the combination of at least two effective leishmanicidal agents is a desirable option [17,30].

# 22.4 Antileishmanial Drug Development Review

The incidence of millions of new cases of *leishmaniosis* per year worldwide and deficiencies in current treatments point to the urgent need for new drugs to combat these parasitic diseases [41,42]. The massive displacement of populations, weather changes, growing resistance to available drugs, a lack of therapeutic alternatives and long treatments are creating a significant rise in the number of infected population people [43–45,91] with resulting increases in public health problems, mainly regarding poor individuals. Leishmaniosis is an almost neglected disease owing to its prevalence in developing countries. Big pharmaceutical companies are not motivated to search and develop antiparasite compounds with modest economic perspectives.

Nevertheless research for novel therapeutics is desirable because:

- 1. Parasites develop drug resistance;
- **2.** Patients have bad side effects owing to the high toxicity of current drugs during and after treatment;
- **3.** Treatment is expensive and the market is not supplied market owing to economic factors; and
- **4.** There is undesirable responsiveness in disease recovery and variable efficacy [46,47].

It is unlikely that a single formulation of one or more drugs effective against all forms of leishmaniosis will be developed. Several factors affect this development. Sites of infection impose different pharmacokinetic requirements and there is intrinsic variation in drug sensitivity for different *Leishmania* species.

There are other emerging problems to overcome:

- **1.** the need for drug activity in regions where resistance to pentavalent antimonials develops; and
- 2. efficacy in immunosuppressed patients.

The effects of neoplastic diseases and immunologic disorders are often worsened by immunosuppressive drugs such as corticosteroids and cytotoxics [48]. Treatment may be chosen according to disease responsiveness.

Antiparasitic compound availability is usually low in natural products, typically causing difficulties with research in this field. According to Bhuwan [49], practical aspects to consider in active natural substances screening are the low yield and sample representativeness issues, such as only one sample obtained from each source, which makes statistical analyses poor or impossible. In addition, the relative structural complexity of natural products is high and includes the occurrence of multiple stereo-isomers. Furthermore, isolated active principles rarely exhibit potent activity themselves, but require follow-up improvement to be effective [49].

# 22.5 Biotechnological Development of New Antileishmanial Drugs

Considering the drawbacks of these drugs, more attention should be given to the extracts and biologically active compounds isolated from natural sources, such as plant species [50,60] and fungi [51,52] by biotechnological methods. Several promising antileishmanial compounds have been reported over the past few years in various test systems. To identify the most promising compounds, candidate molecules should be compared with one or more standard antileishmanial drugs. If necessary, animal models should be used to assess parasitemia in mice or hamster models.

Many studies have reported the isolation of compounds from plants or by chemical synthesis. One advantage of using microorganisms to obtain active compounds is the possibility of rapidly increasing production by process optimization and genetic manipulation. Many fold increases have been recorded in the production of small metabolites, active principles of new pharmaceutical products. High productivity also facilitates the purification of these substances.

The cell biology of *Leishmania* and mammalian cells differs considerably and this distinctness extends to the biochemical level. This provides the promise that many of the parasite's proteins should be sufficiently different from those in the host to be successfully exploited as drug targets. Consequently, discovering new antileishmanial drugs should be considerably easier than discovering other new drugs, from a scientific perspective.

There is a need to develop new antiprotozoal drugs with novel structures and modes of action [53]. Indeed, over many years, several interesting drugs have been proposed. Unfortunately, few of these compounds have been validated rigorously.

Research into new active principles has been mainly based on these trends: screening of natural products, targeting of biochemical pathways, chemical derivatization of synthetic compounds, molecular modeling, and combination therapies.

A wide variety of products from different sources with leishmanicidal activity are attracting the interest of researchers across the world. Some examples from the past decade are presented.

A series of novel aryloxy benzocycloalkyl azoles prepared from cyclohexanone were highly active in vitro against *L. donovani* and exhibited significant in vivo activity in a hamster model [54]. The evaluation was performed in vitro against transgenic *L. donovani* promastigotes and intracellular amastigotes at various concentrations, taking sodium stibogluconate and pentamidine as a control. Cytotoxicity response was assessed using a mouse macrophage cell lineage. All of the compounds killed promastigotes and amastigotes in a concentration-dependent manner and inhibited parasites.

Hydromethanolic fractions of the aerial parts of the plant *Mitracarpus frigidus* showed potential antioxidant, antimicrobial, and leishmanicidal activity [55]. Phytoconstituents were investigated and the total amount of phenolic compounds and flavonoids was quantified to characterize the extracts.

Crypthophilic acids (from the plant family Scrophulariaceae) and other constituents of *Scrophularia cryptophila* were examined for their in vitro antiprotozoal and antimycobacterial potential [56]. Resin glycosides were studied, which are secondary metabolites exclusive to convolvulaceous plants. With the exception of two compounds, all glycosides tested were active against *L. donovani*; the best was Harpagide.

Derivatives, intermediates, and analogs of several drugs (alkyl-amino-cyclohexanols, cyclohexanediamine, cyclohexanediol, and cyclohexane-fused aziridine) were evaluated in vitro against promastigotes of *Leishmania amazonensis*, *Leishmania Braziliensis*, and *L. donovani*. All types of compounds assayed resulted in highly potent antileishmanials except one unable to form hydrogen bonds [57]. Pentamidine and amphotericin B were used as reference drugs.

An aminoglycoside, neomycin, a structural analog of paromomycin, at  $600-\mu M$  concentration, resulted in 30% inhibition of *L. donovani* growth for both the wild and resistant strains [40]. Earlier studies also reported a weak effect of neomycin on *L. donovani* growth [58].

Alkaloidal extracts of Brazilian plants (*Aspidosperma ramiflorum*) were active against the promastigote form of *L. amazonensis*. Acid and basic fractions obtained from ethanolic crude extract and pure compounds from the stem bark of this plant were tested. Mainly the basic alkaloidal fractions had good activity and also showing activity against gram-positive bacteria. The basic fraction was further fractionated on silica gel column chromatography in a bioassay-guided protocol, resulting in individual purified compounds for characterization [59].

Diverse naphtoquinones, which are present on different species of vegetables, have shown some level of activity against *Leishmania*. Benzoquinones, naphtoquinones, and mansonones demonstrated activity against Trypanosomatidae [53]. Diospirine, which showed activity against promastigote forms of *L. donovani*, was chemically modified to reduce its toxic side effects [67].

Nicotinamide (NAm), a form of vitamin  $B_3$ , significantly inhibited the intracellular growth of *Leishmania* amastigotes [60]. NAm may thus represent a new leishmanicidal agent that could potentially be used in combination with other drugs during therapy.

The alkaloid renieramycin A from sponges inhibited *L. amazonensis* [61]. Euplotin C, a sesquiterpene obtained from a marine ciliate, was active against *Leishmania major* and *L. infantum* [62]. A peptide derived from defensin of marine mussel inhibited *L. major* [63].

The specialized sterols of *Leishmania* and their close similarities to sterol composition and biosynthesis in yeast led to the assay of activity of a number of antifungal sterol biosynthesis inhibitors against *Leishmania* [64].

Pentamidine activity can be associated with alteration of L-arginine transport, because it inhibits this mechanism in protozoa. Production of nitric oxide (NO•) has a role as a cell signaling agent and its function as an antileishmanial effector molecule. The action of pentamidine isethionate (reference drug) and amidine derivatives (test compound) were studied on the NO• pathway of promastigotes and axenic amastigotes of *L. amazonensis* by measuring nitrite, a by-product of nitric oxide released into culture supernatants. NO• production by infective promastigotes was inhibited by the test compound in about 23% and by pentamidine in only 4%. In axenic amastigotes, the inhibition was 53% for the test compound. This fact is important because L-arginine is the substrate of the NO• pathway [65].

Some lichen galactomannans (polysaccharides) showed leishmanicidal activity in vitro by direct cytotoxicity and immunomodulating stimulation activity of macrophages. This activity was even improved by complexation with vanadium ions [66]. Monomeric and dimeric naphtoquinones were found to have in vitro leishmanicidal activity [67]. Another focus was given to finding oral drugs to minimize injectionassociated complications. Many substances, including antifungal ketoconazole, were reported to be effective [68].

Investigations of Atlantic Forest fungi were published, with the report of several active substances: antifungal, cytotoxic, trypanocidal, immunosuppressive, and leishmanicidal compounds [69] from *Agrocybe perfecta* (*Rick*) *Sing.*, *Oudemansiella canarii* (*Jungh.*) *Hohn*, and *Lentinus strigosus* (*Schwein.*). Other studies for the discovery of potential new drugs from natural sources were reported, including the screening of *Basidiomycota* species, using in vitro bioassays to assess the obtained compounds against human cancer cells lines, human peripheral blood mononuclear cells, recombinant *Trypanosoma cruzi*, and amastigote-like forms of *L. amazonensis* [56].

Vitola [51] reported leishmanicidal activities of several extracts obtained from *Perenniporia martiusii, Plectania* sp., and *Ganoderma stipitatum*—submerged mycelial cultivation by 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyl tetrazolium bromide (MTT) and tritiated thymidine methodologies. Some assayed extracts showed even more activity than Glucantime, used as positive control.

Fendrich [52] further fractionated *Perenniporia martiusii* extracts by using chromatographic techniques, and detected significant antileishmanial activities by using various in vitro models. More than one fraction exhibited activity. The most active fraction (fraction D) showed more activity than Glucantime in even lower concentrations. The cytotoxicities of the most active extracts were evaluated by assaying Vero cells with MTT. In fact, some active fractions stimulated Vero cell proliferation. This D fraction stimulated macrophages NO• production in vitro, which suggested that this substance had both direct and indirect activity mechanisms.

According to Chang and Miles [70], "only about 20% of all mushroom derived products are based on extracts from cultivated mycelia, and about 2% are from broth used for growing mycelia. Mycelia formed by growing pure cultures in submerged conditions are of constant composition, and this is the best technique for obtaining consistent and safe mushroom products. In addition, increased quality control of the products and year-round production under controlled conditions make mycelial-based products, including exopolymers from culture broth, the wave of the future."

# 22.6 Methods for Anti-Leishmania Drug Screening

Several methodologies (in vitro, in silico, and in vivo) are available to assess antiparasitic activity. Each has its advantages and drawbacks. In vitro tests, for example, are limited for previewing complex interactions that happen naturally in living environments. On the other hand, although in vivo tests on animal models give results more approximated to human reactions, they cannot simply be extrapolated. As better computational models of biochemical structures and even of whole cells or organisms are developed, new in silico assays are becoming more precise and approximate to real biological structures. It is still impossible to substitute in vitro or in vivo tests entirely by in silico assays. However it is a powerful approach for rapidly screening a great number of candidate chemical structures and for predicting the effects of molecular modifications.

Nevertheless it is necessary to reduce, replace, and refine experiments to minimize the suffering of animals. In silico, in vitro, and in vivo models should be used wisely in combination, to obtain the most active and safe new drugs in the fastest, cheapest, and most ethical way possible.

The screening process can be performed in several steps. For each stage, it is necessary to define the experimental model type (in silico, in vitro, or in vivo). Whenever possible, it is a better choice to start with in silico studies, to evaluate candidate substances without the need to spend materials or waste animals' lives. However, it is unusual to have detailed molecular information about new natural samples. In this case, it may be cheaper and more practical to perform some in vitro tests and even in vivo assays before deciding to deeply analyze the molecular composition of a given unknown sample.

For in vitro methods, it is necessary to choose the species of *Leishmania* and the medium in which to cultivate the parasites. It is possible to work with promastigotes or amastigotes (intracellular form). It is ethical to assay candidate substances in vitro both for their activity and their toxicity before in vivo tests. After a candidate substance passed all preliminary in vitro steps successfully, it is necessary to test it in animal models before prescribing it to humans. Tests with rodent and nonrodent species are required by authorities before allowing clinical studies of new drug candidates with human patients. Various clinical experiment phases are needed before the production, commercialization, and prescription of any given new pharmaceutical substance is authorized. In this process, efficacy and safety must be proven experimentally for the greatest number of cases possible.

In silico methods can also be used to evaluate the effect of chemical modifications of active substances to improve their efficacy further or to reduce side effects.

Fig. 22.3 presents suggested steps for testing metabolites or new drugs.



FIGURE 22.3 Flowchart showing an antileishmanial drug screening strategy.

#### 22.6.1 Methods for Working With Leishmania

#### 22.6.1.1 Leishmania Culturing

Parasite cultivation techniques are necessary for screening substances as drug candidates against leishmaniasis. The first step is to choose among *Leishmania* species. Several media [coeur-cerveau-sang de mouton (CCS), Roswell Park Memorial Institute medium (RPMI), and modified liver infusion tryptose (LIT)] can be used for *Leishmania* cultivation. Media compositions are described in Tables 22.1–22.3.

Other media such as M-199 medium {supplemented with 40 mM [4-(2-hydroxyethyl)-1-piperazineethanesulfonic acid],  $100 \mu$ M adenosine, 0.5 mg/L hemin, 10% heat-inactivated fetal bovine serum, and  $50 \mu$ g/mL gentamycin} or RPMI (Sigma) medium, added to 10% fetal calf serum, can be used for *Leishmania* culturing. RPMI and LIT cultures generally are conduced inside Roux bottles, CCS cultures inside 250-mL Erlenmeyer and Novy–MacNeal–Nicolle medium (NNN) in tubes (Table 22.2). The cultures are maintained inside an incubator at  $23 \pm 2^{\circ}$ C.

An inverted microscope is used to monitor Roux bottles culture characteristics, including concentration, motility, and contaminations. For Erlenmeyer cultures, samples are taken for observation on a common optical microscope. Whenever necessary, samples are also taken from Roux bottles for better observation on a common optical microscope. Contaminated and old cultures should be adequately sterilized and discarded. Weekly transferences to fresh media are necessary. Only cultures within a concentration range between  $10^7$  and  $10^9$  parasites per milliliter are used. An adequate inoculum rate is about 20%.

As described by GOMES et al. [71], "... the different *Leishmania* species are not equally easy to culture. Contamination is a constant problem, and variations in efficacy among different growth medium formulations or even batches may be encountered." For some assays, NNN medium was found to give better results [72].

Component	Concentration (g/L)
NaCl	5.0
Na <sub>2</sub> HPO <sub>4</sub>	7.5
Tryptose	5.0
Yeast extract	3.0
Brain heart infusion	3.0
KCI	0.4
Adjust pH to 7.4 with 1 M Na <sub>2</sub> HPO <sub>4</sub>	
Autoclave at 120°C for 30 min	
Add 2 mL hemine (20 mg/L) for each liter of LIT medium (previously filtered micro-pore filters)	ed through 0.2 $\mu$ M
Add 5 mL gentamicin (10 mg/mL) for each liter of LIT	
Add 10% previously decomplemented fetal calf serum	

 Table 22.1
 Modified Liver Infusion Tryptose (LIT) Medium Composition

Component	Concentration
NaCl	3.0
Bacto-agar or past-agar	5.0
Boil for 2—3 min	
Distribute in tubes (4 mL/tube)	
Autoclave at 120°C for 30 min	
Add 0.3 mL citrated rabbit blood per tube	
Let medium solidify in inclined position inside tubes	

Table 22.2 Novy–MacNeal–Nicolle Medium Composition

#### Table 22.3 CCS Medium Composition

Component	Concentration (g/L)
Brain heart infusion	37.0
Bacto-agar	15.0
Boil for 2—3 min	
Distribute in 125-mL Erlenmeyer flask (10 mL/flask)	
Autoclave at 120°C for 30 min	
Add 1 mL citrated rabbit blood per flask	
Let medium solidify inside flasks	
Add 4 mL physiologic serum	

*L. infantum* grows properly in RPMI medium and *Leishmania enriettii* is well-adapted to modified LIT medium. On the other hand, *Leishmania braziliensis* does not grow well on blood cell-free media (RPMI and modified LIT), but CCS medium contains rabbit blood cells, which interfere with the MTT assay. To assay this species with MTT, a blood cell disruption procedure and medium transference are necessary before applying the bioactivity evaluation methodology. A cell disruption protocol is described next.

#### 22.6.1.2 Blood Cell Disruption

- Filter about 20 mL of the liquid fraction of a CCS medium previously cultured with *Leishmania* ( $\sim 10^7$  cells/mL) through a sterilized cotton cloth and transfer it to a sterilized Falcon tube under aseptic conditions.
- Wash the CCS medium solid matrix and the Erlenmeyer flask originally containing it with 10 mL LIT modified medium; filter it and add it to the remaining suspension in the Falcon tube.
- Add 10 mL of physiologic serum and centrifuge the resulting suspension for 15 min at 5000g, at 23  $\pm$  2°C.
- Pipet and discard the supernatant.
- Add 30 mL of physiologic serum to the remaining pellet.
- Centrifuge the resulting suspension for 15 min at 5000g,  $23 \pm 2^{\circ}$ C.

- Pipet and discard the supernatant.
- Resuspend the remaining pellet in 25 mL of modified LIT medium (10% fetal calf serum).

#### 22.6.1.3 Leishmania Counting in Thoma Chamber

This technique is used to quantify parasite concentration in a culture's liquid. The Thoma chamber consists of a microscopy slide with a thin well of a precise volume and an imprinted microscopic grid. With the aid of an optical microscope, microorganisms contained in a liquid sample can be counted. The method consists of covering the well with a small glass slide and transferring about 10  $\mu$ L of the properly stained protozoan cultured liquid to be quantified inside the slide well by capillarity. With an optical microscope, individual parasites inside the grid squares (quadrants) are counted. Because the chamber volume is known (0.004 mm<sup>3</sup>), parasite concentration on the sample can be determined.

*Leishmania* culture samples need to be adequately diluted before counting (one-half, one-fifth, or one-tenth). A fixation/staining solution can be used for the dilution besides its main functions (fixation and staining of the parasites).

A microscope coupled with a video camera linked to a monitor can facilitate counting. A manual counter is also a good tool for high parasite concentrations.

Microorganisms within five quadrants are usually counted and the mean number is divided by the volume over one quadrant. The values are adequately corrected for previously applied dilutions.

#### 22.6.2 Experimental Models for Screening Leishmanicidal Activity

#### 22.6.2.1 In Vitro Antipromastigote Assay

Different species of *Leishmania* can be used for these experiments. In the new world, *L. amazonensis* is generally the model of choice for tests because the parasite grows quickly. In the old world, the most commonly used model is *L. major. L. enriettii* is an elegant model because it is not pathogenic for humans, which avoids laboratory accidents. Furthermore, replication of *L. enrietti* promastigotes is fast.

To screen candidate biotechnological metabolites against *Leishmania*, we can use 24- or 96-well plates. Typical protocols suggest properly diluting and distributing parasite suspensions containing  $10^6$  parasites/mL on multiple microwells and adding the chosen cultivation medium and the drug to be tested. The experiments should be performed with parasites in their logarithmic phase of growth.

After inoculations of the drugs to be tested, the parasites should be incubated at  $23 \pm 2^{\circ}$ C for 5 days without medium replacement. Every 24 h after treatment, aliquots are harvested and the effect on promastigotes growth is evaluated using a chamber for counting, and compared with untreated parasite culture. Cultures should be performed in triplicate. Glucantime can be used as a positive control.

In vitro tests are based on protozoa cultivation and evaluation of their survival or of alterations on their metabolism upon the addition of test substances. One simple method for assaying protozoa survival is to observe whether their motility is significantly reduced or stopped, using an optical microscope.

Other methods involve the assimilation of radioactive isotopes by protozoa as they grow. Measuring the concentration of incorporated radioactive compounds after an incubation period allows the protozoa growth rate to be inferred. Tritiated thymidine is a current methodology of this kind.

Alternatively, reagents that change color as they are metabolized can be used to measure metabolism rhythm alterations. Examples of these reagents are MTT, XTT {sodium 2,3,-bis(2-methoxy-4-nitro-5-sulfophenyl)-5-[(phenylamino)-carbonyl]-2H-tetrazolium} and WST-1 {4-[3-(4-Iodophenyl)-2-(4-nitro-phenyl)-2H-5-tetrazolio]-1,3-benzene disulfonate}. All of them are tetrazolium salts that change color as they are metabolized. Color intensities can be measured with a spectrophotometer and are proportional to the parasite metabolism rate [69]. Many bioactive compounds screening studies have been performed with this principle. These methods are easy to perform and fast to read. Furthermore, those reagents have the advantage of being nonradioactive. Drawbacks are that sensibility and accuracy are not as high as with DNA-based methods (radiolabeled thymidine or bromodeoxyuridine).

#### 22.6.2.2 In Vitro Antipromastigote Colorimetric Assay

MTT {(3-(4,5-Dimethylthiazol-2-yl)-2,5-Diphenyltetrazolium Bromide} is a yellow tetrazolium salt that turns purple when reduced (cleaved) to formazan compounds. It is useful for many bioassays because cellular metabolism can reduce MTT. Cellular metabolism rate can be inferred by the color of the medium after incubation with this reagent [73].

Because formazan salts are water-insoluble, they must be properly solubilized before reading (10% sodium dodecyl sulfate, 50% isopropanol can be used). Color intensity measurement is accomplished with a spectrophotometer. For microassays, a 96-well plate reader is better suited [73,74]. These methodologies are currently used to measure factor-induced cytotoxicity or cell necrosis, but cell viability can also be assayed. Also, because proliferating cells are metabolically more active than nonproliferating cells, these tetrazolium salt—based assays are frequently used to measure cell activation and proliferation [73–75]. Current applications include:

- · evaluating growth-inhibitory or cytotoxic effects of compounds,
- analyzing cytotoxic and cytostatic effects of potential drugs,
- analyzing cytopathic effects of viruses and screening compounds with antiviral activity,
- · screening antibodies for growth-inhibiting potential, and
- studying the effects of substances over immune system cells (proliferation or suppression).

#### 22.6.2.3 In Vitro Antipromastigote Radiolabeled Thymidine Assay

Cell proliferation can be measured with radiolabeled thymidine methodologies, based on the fact that proliferating cells incorporate free nucleotides such as thymidine when they are present in the culture medium. Thymidine molecules synthesized with radioactive isotopes (<sup>3</sup>H or <sup>14</sup>C) are added to culture medium. These are called radiolabeled thymidine molecules. As cells multiply, their DNA is replicated. Radiolabeled thymidine molecules are incorporated through this process.

After some incubation time, cells are collected with a cell harvester and transferred to a filter paper; nonincorporated radiolabeled thymidine molecules are washed. Pieces of the filter paper containing cells are transferred to a tube and immersed in a scintillation solution. The radioactivity emission intensity of each tube is measured with a scintillator and compared with the radioactivity emitted by the total concentration of thymidine initially added.

Many research groups have used radiolabeled thymidine methodologies to assess substance bioactivities against tumor cells, immune system cells, and even parasite cells [75]. These methodologies are elegant and have very high sensitivity. The main drawback is that they involve the manipulation of radioactivity.

An alternative methodology to DNA synthesis quantification involves the use of bromodeoxyuridine (BrdU), which is a modified but nonradioactive nucleotide that incorporates to newly synthesized DNA molecules. For BrdU, immunostaining techniques allow the method to be colorimetric or measured by fluorescence [76-78].

Assaying the activity of substances against promastigotes is a practical tool for screening new biotechnological drugs. Promastigotes are also useful as cytotoxicity indicators in the bioassay-guided fractionation of plant products. However there are significant differences between promastigotes and amastigotes in biochemistry and sensitivity to standard and experimental drugs [30].

#### 22.6.2.4 Antiamastigote Assays Using Macrophages Infected With Axenically Grown Amastigotes

Adhered peritoneal macrophages can be infected with *Leishmania* sp. promastigotes (stationary growth phase) at a parasite/macrophage ratio of 10:1 and incubated for 3 h at  $37^{\circ}$ C and 5% CO<sub>2</sub>. In the final of experiment phase the cells are washed with saline solution, fixed in methanol, and stained with Giemsa. The number of parasites is determined by examining three coverslips for each treatment. At least 200 infected macrophages are counted and results are expressed as the infectivity index. The positive control is Glucantime (50 mg/mL), and triplicates should be used for each concentration.

The review by Croft and coauthors [30] showed that different protocols were described for *Leishmania* mexicana [76,77], *L. braziliensis* [78], *L. donovani* [79,80], and *Leishmania infantum* [81]. However the amastigotes used must have confirmed biochemical and immunologic markers, and interpretation of data related to the high concentration of serum is required in some systems. Differences in drug sensitivity between axenic *L. donovani* amastigotes and intracellular amastigotes have been observed [30].

Generally, host cells used for amastigote infection are murine peritoneal macrophages or human-monocyte transformed macrophages. These cells can have variation in drug sensitivity [82]. For these differentiated nondividing macrophages, the rate of amastigote division in host cells and drug activity can be clearly determined. For species of the *donovani* and *infantum* complex, the slow rate of division of amastigotes in this model is a limitation [30]. Assays that use dividing host cells must ensure that the confounding effects of drug activity on both parasite and host cell number are considered. Mouse (J774) [83] and human (THP-1, U937, and HL-60) monocytic cell lines have been used in drug assays [84]. THP-1 cells can form a nondividing monolayer and make an attractive alternative to primary isolated macrophages, and have been used in antibacterial assays [85].

The sensitivity and drug activity are assessed by determining the percentage of infected macrophages and the parasite survival index (PSI), the mean number of intracellular parasites in an infected macrophage.

PSI is calculated as 100 - (number of amastigotes/100 infected macrophages in treated wells)/(number of amastigotes/100 infected macrophages in untreated wells). The cells are counted with an optical microscope [86–88].

If 100% of efficacy is measured for a substance, it is possible to confirm the result further by assaying the infected cells and positive control in NNN medium [66].

#### 22.6.2.5 Antiamastigote Assays Using Automated Screening

Major limitations of the amastigote—macrophage model are the absence of automation and dependence on microscopic evaluation. Promastigote assays using reazzurin (Alamar blue) and transfected parasites have been successful [89–92] but not within the clinically relevant amastigote—macrophage model.

The firefly luciferase gene has been successfully integrated into the genome of an *L. amazonensis* strain and has been evaluated in vitro. It was used in a drugscreening assay in 96-well plates and results correlated with those of microscopic evaluation. Transfected reference strains of *Leishmania*, such as *L. donovani* HU3, would be of further use to the drug screening community; work is under way to achieve this.

#### 22.6.2.5.1 PROPIDIUM IODIDE DYE

Propidium iodide (PI) is a fluorescent dye that is unable to penetrate intact cell membranes, and which is used to detect apoptosis cells on viability assays. PI binds the DNA of cells that lost membrane integrity intercalating between base pairs with low or absence of sequence preference. When PI is bound to nucleic acids, it emits an enhanced red fluorescence with a maximum at 617 nm wavelength. PI can be used in fluorescence microscopy, confocal laser scanning microscopy, flow cytometry, and fluorometry. PI is preferentially used in flow cytometry because it permits a rapid and reliable method. PI has been commonly used in assays to measure the effect of apoptosis by drugs on promastigote plasma membrane. Aliança and coworkers used this methodology with promastigotes of *L. amazonesis*. After staining with PI dye, cultures were incubated with candidate drugs and labeled. The intensity of the stain was measured in a flow cytometer and then analyzed on software [93]. Researchers also reported the use of PI cell viability assay to measure the apoptotic effect of drugs against amastigote forms [94]. Antileishmanial drug activity was tested against *L. amazonesis* axenic amastigotes. After an incubation period with the drug, the amastigotes were stained with PI at room temperature for 5 min. Then the stain intensity was measured by flow cytometry and analyzed on software [95]. The PI stain method through flow cytometry has the advantages to be accurate, which permits simultaneous evaluation of different drugs and automated reading of results. Other fluorochromes can also be used, such as (2',7'-bis-(2-carboxyethyl)-5-(and-6)-carboxyfluorescein, acetoxymethyl ester),SYTO-17, and a fluorescein conjugated with monoclonal antibodies [95].

#### 22.6.2.5.2 HIGH-THROUGHPUT SCREENING AND HIGH-CONTENT SCREENING

High-throughput screening (HTS) is a technology used to select active compounds in a range of small molecules widely used in the pharmaceutical industry to discovery new drugs. Some research groups have reported the use of this technology in in vitro assays to find new drugs against *Leishmania* [96,97]. The efficiency of HTS can be combined with assay information to provide measures of the molecule effect in the assay system, resulting in high-content screening (HCS) [97].

An HCS protocol was developed for in vitro drug screening for the intracellular amastigote form of *Leishmania* sp. to infect human macrophages [97]. This protocol consists of two principal steps: (1) a 384-well plate containing the host cells (THP-1), the parasites, the testing, drugs and the controls, is imaged on an automated microscope; (2) the images are analyzed using an image analysis algorithm.

The results are stored in a database and can be accessed with software that provides assay features using statistical analysis. The features include, among others, the number of cells, the number of infected cells, the infection ratio, and the total number of parasites per cell [97].

The HCS methodology provides a promising and relevant system for the discovery of new antileishmanial drugs, because it compares and quantifies drug activity against the intracellular form of the parasite.

#### 22.6.2.6 In Vivo Assays

To confirm the results of in vitro experiments, in vivo models should be used. In in vivo models, it is possible to study absorption (route of administration), distribution (different sites of infection), metabolism (pro-drugs or immunomodulators), and excretion of candidate drugs, and even to give an early indication of toxicity. In the literature, the most commonly used animal model is the mouse, because small amounts of test compounds are required, as well as a relatively small number of animals (five animals per group). The BALB/c mouse–*L. major* model has been used for others species such as *L. mexicana* and *L. amazonensis*. Lesions on the back of mice are also used to test topical formulations. To determine drug efficacy, changes in lesion size should be measured during and after the course of treatment. Poor animal models for

*L. tropica, L. aethiopica,* and *L. braziliensis* are a limitation of studies. Work on the *L. braziliensis* hamster model has shown the potential for drug testing [98]. For visceral leishmaniosis, dogs are frequently the model of choice [99].

In animal models, the determination of drug activity upon necropsy or biopsy depends on culture in NNN medium or microscopy to determine the level of infection. Molecular techniques such as real-time polymerase chain reaction are currently used [30,100-104]; however these are better used as preliminary evaluation assays. It is preferable to reisolate the parasite, to avoid false negatives or positives.

Before starting in vivo experiments, it is necessary obtain approval for the clinical trials from the Ethical Committee for Animal Care, following international rules (for details, see the 3R rules [105]). Some examples of animal models, inoculation procedures, and inoculation rates are given in Table 22.4.

#### 22.6.2.7 In Silico Assays

Theoretical knowledge can aid the screening of natural compounds for active molecules. With the proper analytic techniques, it should be possible to predict the potential of a novel natural sample before any in vitro or in vivo experiments.

The interaction of molecules can be predicted by physicochemical models. Computers allow the calculation of the interaction of highly complex molecules, such as those present in biological systems. Evolving knowledge of Leishmania genes, proteins, RNA molecules, and other metabolites, combined with knowledge about humans and vectors molecular profiles, provides a field for virtual experiments that can result in good insight for the development of better new antileishmanial drugs.

It has been 10 years since the first *Leishmania* genome was fully sequenced [121]. Since then many *Leishmania* species and strains have been fully sequenced. Genomic, transcriptomic and proteomic research can greatly contribute to the development of new antileishmanial drugs. New vaccines and chemotherapeutics, for example, could be projected based on molecular differences between humans and parasites. Proteins present exclusively on parasites are good candidates as chemotherapeutic targets or vaccine immunogens.

Moreno et al. [122] identified that tyrosine aminotransferase from *L. infantum* should be a good drug target candidate because it is significantly different from human proteins and is overexpressed during infection.

Khademvatan et al. [123] compared the activity of biurets and Glucantime against *Leishmania* previewed by an in silico model with experimental in vitro results. The most active substance pointed out by the in silico model was confirmed by in vitro results. Roldos et al. [124]. compared in silico, in vitro, and in vivo results of a new leishmanicidal drug (14-hydroxylunularin), with a positive correlation.

Furthermore, resistance development mechanisms may be profoundly comprehended by genomic and proteomic perspectives, thus allowing a rational approach toward developing new pharmaceutical molecules capable of effectively overcoming the parasites' defenses in the long term.

Species	Animal Model Choice	Objective of Study	Size of Inoculum/Inoculation Route	References
Leishmania (L.) infantum	BALB/c mice	Assess leishmanicidal effect of series of synthesized compounds containing nitrobenzene and sulfonamido moieties	$1 \times 10^7$ promastigotes per animal were inoculated by intracardiac route	[106]
L. infantum Leishmania (V.) braziliensis	Mice peritoneal macrophages from Balb/c, C57BL/6 and knock-out (TLR2 <sup>-/-</sup> , TLR4 <sup>-/-</sup> )	Measure production of nitric oxide, cytokines, mitogen-activated protein kinases, and nuclear factor-kB to evaluate macrophage modulation	Mice macrophages were primed with interferon-gamma and stimulated with purified LPG from <i>L. infantum</i> and <i>L. braziliensis</i>	[107]
Leishmania (L.) donovani	BALB/c mice (6 weeks old)	Evaluation of protective efficacy of tuzin as DNA vaccine obtained by reverse vaccinology approach using novel high-throughput immunologic screening assay	$1 \times 10^8$ stationary phase promastigotes suspended in 100 µL phosphate-buffered saline (PBS) (freshly transformed from spleen isolated amastigotes) were intravenously injected using 22-gauge needle	[108]
L.(V.) braziliensis Leishmania (V.) peruviana	Golden hamsters ( <i>Mesocricetus auratus</i> ) (4 weeks old)	Create model able to demonstrate reproducibly differences in infectivity among representative strains from <i>Leishmania</i> of subgenus <i>Viannia</i>	$1 \times 10^{6}$ parasites resuspended in 100 µL saline solution were inoculated at right footpad: Group I stationary phase promastigotes and Group II metacyclic promastigotes purified by Ficoll protocol.	[109]
L. (V.) braziliensis	Golden hamsters ( <i>Mesocricetus auratus</i> )	Evaluation of reproducibility of <i>L. braziliensis</i> infection as appropriate model for immunopathogenesis studies of cutaneous leishmaniosis, supporting its use in clinical, vaccine, and chemotherapy experimental protocols	$1 \times 10^{6}$ stationary growth phase promastigotes (third in vitro passages in supplemented Schneider's <i>Drosophila</i> medium) were inoculated intradermally in dorsal hind paw of hamsters.	[110]
L. (V.) braziliensis	Golden hamster ( <i>Mesocricetus auratus</i> )	Assess antileishmanial activity of metabolites, mainly flavanones, from <i>Picramnia gracilis</i> Tul., motivated by traditional use of this species in skin problems	$5 \times 10^8$ promastigotes ( <i>EpiR-GFP</i> ) in 100 µL PBS were inoculated in anesthetized hamsters at dorsal skin.	[111]

#### Table 22.4 In Vivo Antileishmanial Assays

Continued

Species	Animal Model Choice	Objective of Study	Size of Inoculum/Inoculation Route	References
L. (V.) braziliensis Leishmania (V.) guyanensis	C57BL/6 murine model for LRV dsRNA, that exacerbates disease in dose-dependent manner	Propose refinements of methods for successful <i>Leishmania</i> RNA virus detection for use as diagnostic tool relating to cutaneous leishmaniosis complications	$1 \times 10^{6}$ stationary phase promastigotes (with and without <i>Leishmania</i> RNA virus) were injected subcutaneously into base of hind footpad.	[112]
Leishmania (L.) major	BALB/c mice	Evaluation of in vitro and in vivo antileishmanial activity of <i>Pistacia khinjuk</i> stock alcoholic	$2 \times 10^{6}$ stationary growth phase promastigotes were subcutaneously injected into base of tail.	[113]
L. (L.) major	BALB/c mice	Evaluation of value of oleylphosphocholine for treatment of cutaneous leishmaniosis by in vivo tests	$1 \times 10^{7}$ stationary growth phase promastigotes were injected into base of tail.	[114]
Leishmania (L.) amazonensis	BALB/c (4—5 weeks old)	Test combination therapy of tamoxifen and amphotericin B in vivo	$1 \times 10^{6}$ stationary-phase transgenic promastigotes of <i>L. amazonensis</i> expressing luciferase were inoculated at base of tail.	[115]
L.(L.) amazonensis	Experimental model of cutaneous leishmaniosis in BALB/c mice	Test antileishmanial activities of six 2-nitrovinylfurans	$5 \times 10^6$ stationary phase were injected through intradermal route in footpads.	[116]
L. (L.) amazonensis	C57BL/6 mice	Characterize changes in pattern of immune response during subcutaneous vaccination with Leishvacin	$1 \times 10^5$ promastigotes from stationary phase cultures (4 days) were inoculated in hind footpad.	[117]
Leishmania (L.) tropica	BALB/c mice	Evaluate effects of ethanolic extract of Berberis vulgaris fruits and chloroform extract of Nigella sativa (folk medicine)	$1 \times 10^{6}$ stationary promastigotes were inoculated in male BALB/c mice macrophages.	[118]
Leishmania (L.) enriettii	Guinea pigs ( <i>Cavia</i> <i>porcellus</i> ) C57BL/6 mice and knockout mice (TLR2 <sup>-/-</sup> , TLR4 <sup>-/-</sup> )	Characterization of LPGs and glycoinositolphospholipids, two important macromolecules of disease immunopathology model	$1 \times 10^5$ promastigotes in 100 µL PBS were inoculated by intradermal inoculation.	[119]
L.(L.) major	Mouse models	Review immunologic aspects of disease	_	[120]

#### Table 22.4 In Vivo Antileishmanial Assays—cont'd

#### 22.6.3 Cytotoxicity Assays

Whenever a candidate drug shows activity in in silico or in vitro experimental models, the next logical step is to perform assays with animal models and then clinical tests with human volunteers. However before taking these steps, it is ethical to perform other important experiments to detect substances that would be toxic against human cells, besides their leishmanicidal effect. As with activity assays, there are many possible experimental systems for assaying a substance's cytotoxicity. It is possible to use in silico methods to compare the candidate drug molecule with molecular data banks, seeking for similarities between the sample and known toxic compounds. It is also possible to use computers to solve models predicting the interaction of test substances with human molecular structures.

There are many different in vitro experimental models to assay the cytotoxicity of a given substance. Cells from multiple organ sources (kidney, liver, heart, lung, and nervous system), cell types (epithelial, endothelial, myoskeletal, neural, germ line, leukocyte, etc.) and different species (zebrafish, rodent, sheep, human, etc.) can be cultivated and used for toxicity assays [125]. Most standard methodologies involve simply measuring the proliferation of cells in cultivation with and without the addition of candidate substances. Several methodologies are also available to identify cell death mechanisms, such as apoptosis or necrosis.

Cell proliferation can be measured with colorimetric methods (based on MTT, XTT, and WST-1, for example), or DNA synthesis detection methods (including methods based on tritiated thymidine and BrdU), as described in previous sections. Other assay mechanisms include the absorption or exclusion of dyes by living or dead cells [125].

Cellular components can also be used to assay drug toxicities. For example, DNA and RNA microarrays, immobilized proteins, and antibodies developed with the objective of interacting with known toxic compounds can aid in identifying toxic substances in new or complex samples.

The advent of high-throughput systems to perform in vitro assays allowed the screening of thousands of substances, pointing out an elevated number of molecular candidates for in vivo and clinical experiments. This screening power resulted in problems for choosing among candidates for further experimental phases.

Although in vitro cytotoxicity assay methods are applied extensively in the pharmaceutical industry, at least 10% of all drugs that reach clinical testing stages have toxicity problems and 3% are discontinued because of them [122]. It proves that in vitro or even in vivo assays are limited in predicting a substance's toxicity against whole human organisms; nevertheless these methodologies are fundamental to drug development processes.

# 22.7 Conclusions and Perspectives

Although leishmaniosis has afflicted humanity for a long time, the pursuit of better antileishmanial drugs continues to be a considerable challenge. Current medications have serious side effects and parasites continue to evolve, developing resistance and spreading to new areas, and infecting a rising number of individuals.

This situation constitutes a matter of public health concern. Research on new antileishmanial drugs is needed. Many groups within the scientific community are already involved but more work is required to bring about a satisfactory solutions. Governmental contributions are necessary in this process, because the big pharmaceutical companies do not see great economic perspectives in this area.

Public universities currently constitute the major responsibility for developing new therapeutic strategies against leishmaniosis. However, most public university laboratories face restrictive budgets and lack of automation and researchers frequently work independently in a noncollaborative manner. This scene contributes to slow and disorganized technological development.

Countless biomolecules are available for antiparasitic substances screening. The methodologies described here are ready for an evaluation of the activities of natural and synthetic products against *Leishmania* species.

Already discovered natural active extracts must be further fractioned to determine the active principles responsible for the antileishmanial effects observed. This suggests molecular groups for further antileishmanial biocompound screening. Characterization of these molecules and their mechanisms of action can also offer insights useful for antileishmanial drug design.

Biotechnological tools will be necessary to improve the productivity of some natural products. Even molecules from vegetable and animal sources can be expressed in microbial vectors aided by molecular biology tools. As active principles are identified, biotechnological processes parameters, such as process duration, aeration, agitation, temperature, pH, culture medium composition, and extractive procedures, can be optimized to maximize active biomolecule production and recovery.

The effect of the bioactive compounds can be improved by chemical manipulation of the molecules, such as complexation with vanadium ions. Combinations of active substances must be considered as well.

Edible active organisms such as mushrooms and vegetables should be analyzed through other methodologies to evaluate their potential as antiparasitic nutraceuticals. The edibility indicates that their compounds will not have significant toxicity against human cells. This is important because specificity is probably the main goal of drug design, besides effectiveness.

Besides screening new active substances, it is also important to create new methodologies to further select the best candidates for in vivo and clinical studies. Nevertheless molecular data banks have plenty of useful and unexplored information about *Leishmania* and antileishmanial drugs. With the proper in silico methods, this knowledge can result in novel breakthroughs that can lead to intelligent solutions.

Leishmanicidal compounds are not the only alternative to control the dissemination of leishmaniosis. Both vaccines and strategies to control vector insects are effective approaches and should be considered in parallel. This chapter may inspire researchers and supply experimental tools to contribute to the discovery and invention of new antileishmanial drugs.

### References

- [1] WHO (World Health Organization), Leishmaniosis and *Leishmania*/HIV Co-Infection, 2010. WHO report on global surveillance of epidemic-prone infectious diseases. Available at: http://www.who. int/csr/resources/publications/surveillance/Leishmaniosis.pdf.
- [2] W.L. Roberts, W.J. McMurray, P.M. Rainey, Characterization of the antimonial antileishmanial agent meglumine antimonate (glucantime), Antimicrobial Agents and Chemotherapy 42 (1998) 1076–1082.
- [3] P. Mitropoulos, D.O. Konidas, M. Durkin-Konidas, New world cutaneous leishmaniosis: updated review of current and future diagnosis and treatment, Journal of the American Academy of Dermatology 63 (2) (2010) 309–322.
- [4] P. Shaked-Mishan, N. Ulrich, M. Ephros, D. Zilberstein, Intracellular SbV reducing activity correlates with antimony susceptibility in *Leishmania donovani*, Journal of Biological Chemistry 276 (2001) 3971–3976.
- [5] F. Frezard, C. Demicheli, C.S. Ferreira, M.A. Costa, Glutathione-induced conversion of pentavalent antimony to trivalent antimony in maglumine antimoniate, Antimicrobial Agents Chemotherapy 45 (2001) 913–916.
- [6] S. Wyllie, M.L. Cunningham, A.H. Fairlamb, Dual action of antimonial drugs on thiol redox metabolism in the human pathogen *Leishmania donovani*, Journal Biology and Chemistry 279 (38) (2004) 39925–39932.
- [7] IUPAC International Union of Pure and Applied Chemistry, Organic and Biochemical Nomenclature Recommendations, 2010. Available at: www.iupac.org.
- [8] Disanejercito, Direccion de Sanidad del Ejercito Salud Operacional, Leishmaniosis (2008). Available on: http://www.disanejercito.mil.co/Salud\_Operacional\_Leishmaniosis.htm.
- [9] S. Singh, R. Sivakumar, Challenges and new discoveries in the treatment of leishmaniosis, Journal Infection Chemotherapy 10 (2004) 307–315.
- [10] K. Hajlaoui, F. Zeglaoui, N. Ezzine, S. Kastalli, N. Elfekih, B. Fazaa, M.H. Loueslati, M.R. Kamoun, Réaction à l'antimoniate de méglumine (Glucantime®) d'évolution fatale lors du traitement d'une leishmaniose cutanée, Annales de Dermatologie et de Vénéréologie 135 (4) (2008) 317–318.
- [11] S. Antinori, A. Cascio, C. Parravicini, R. Bianchi, M. Corbellino, Leishmaniosis among organ transplant recipients, The Lancet Infectious Diseases 8 (3) (2008) 191–199.
- [12] A.J. Davis, H.W. Murray, E. Handman, Drugs against leishmaniosis: a synergy of technology and partnerships, Trends in Parasitology 20 (2004) 73–76.
- [13] M. Ouellette, J. Drummelsmith, B. Papadopoulou, Leishmaniosis: drugs in the clinic, resistance and new developments, Drug Resistance Updates 7 (2004) 257–266.
- B. Ullman, Multidrug resistance and P-glycoproteins in parasitic protozoa, Journal of Bioenergetics and Biomembranes 27 (1995) 43–52.
- [15] I.L.K. Wong, L.M.C. Chow, The role of *Leishmania enriettii* multidrug resistance protein 1 (LeMDR1) in mediating drug resistance is iron-dependent, Molecular & Biochemical Parasitology 150 (2006) 278–287.
- [16] S. Sundar, L.B. Gupta, V. Rastogi, G. Agrawal, H.W. Murray, Short-course, cost-effective treatment with amphotericin B-fat emulsion cures visceral leishmaniosis, Transactions of the Royal Society of Tropical Medicine and Hygiene 94 (2000) 200–204.

- [17] T.K. Jha, Drug unresponsiveness & combination therapy for kala-azar, Indian Journal Medical Research 123 (2006) 389–398.
- [18] M. Padrón-Nieves, E. Díaz, C. Machuca, A. Romero, A.P. Sucre, Glibenclamide modulates Glucantime activity and disposition in *Leishmania major*, Experimental Parasitology 121 (2009) 331–337.
- [19] K. El Fadili, N. Messier, P. Leprohon, G. Roy, C. Guimond, N. Trudel, N.G. Saravia, B. Papadopoulou, D. Légaré, M. Ouellette, Role of the ABC transporter MRPA (PGPA) in antimony resistance in *Leishmania infantum* axenic and intracellular amastigotes, Antimicrobial Agents and Chemotherapy 49 (2005) 1988–1993.
- [20] G. Singh, C.S. Dey, Induction of apoptosis-like cell death by pentamidine and doxorubicin through differential inhibition of topoisomerase II in arsenite-resistant *L. donovani*, Acta Tropica 103 (2007) 172–185.
- [21] G. Singh, M. Takur, K. Pradip, C.S. Dey, Evidence for the presence of R250G mutation at the ATPase domain of topoisomerase II in an arsenite-resistant *Leishmania donovani* exhibiting a differential drug inhibition profile, International Journal of Antimicrobial Agents 33 (2009) 80–85.
- [22] BRAZIL- SVS/MS Secretaria de Vigilância em Saúde, Ministério Da Saúde Brasil, Atlas de Leishmaniose Tegumentar Americana – Diagnóstico Clínico e Diferencial, Brasília, 1a ed., 2006.
- [23] S.L. Croft, K. Seifert, V. Yardley, Current scenario of drug development for leishmaniosis, Indian Journal of Medical Research 123 (2006) 399–410.
- [24] B.L. Herwaldt, Leishmaniosis seminar, The Lancet 354 (October 2, 1999).
- [25] F.J.S. Perez-Victoria, F. Castanys, *Leishmania donovani* resistance to miltefosine involves a defective inward translocation of the drug, Antimicrobial Agents Chemotherapy 47 (2003) 2397–2403.
- [26] C.P. Thakur, R.K. Singh, S.M. Hassan, R. Kumar, S. Narain, A. Kumar, Amphotericin B deoxycholate treatment of visceral leishmaniosis with newer modes of administration and precautions: a study of 938 cases, Transactions of the Royal Society of Tropical Medicine and Hygiene 93 (1999) 319–323.
- [27] H. Weenen, M.H.H. Nkunya, D.H. Bray, L.B. Mwasumbi, L.S. Kinabo, V.A.E.B. Kilimali, Antimalarial compounds containing an alpha, beta-unsaturated carbonyl moiety from Tanzanian medicinal plants, Plant Medical 56 (1990) 371–373.
- [28] L.J. Goad, G.G. Holz, D.H. Beach, Sterols of *Leishmania* species. Implications for biosynthesis, Molecular Biochemistry and Parasitology 10 (1984) 161–170.
- [29] M. Larabi, V. Yardley, P.M. Loiseau, M. Appel, P. Legrand, A. Gulik, C. Bories, S.L. Croft, G. Barratt, Toxicity and antileishmanial activity of a new stable lipid suspension of amphotericin B, Antimicrobial Agents Chemotherapy 47 (12) (2003) 3774–3779.
- [30] K.L. Goa, D.M. Campoli-Richards, Pentamidine isethionate: a review of its antiprotozoal activity, phamacokinetic properties and therapeutic use in *Pneumocystis carinii* pneumonia, Drugs 33 (1987) 242–258.
- [31] A. Mehta, C. Shaha, Apoptotic death in *Leishmania donovani* promastigotes in response to respiratory chain inhibition: complex II inhibition results in increased pentamidine cytotoxicity, Journal of Biological Chemistry 279 (2004) 11798–11813.
- [32] M. Basselin, H. Denise, G.H. Coombs, M.P. Barrett, Resistance to pentamidine in *Leishmania mexicana* involves exclusion of the drug from the mitochondrion, Antimicrobial Agents Chemotherapy 46 (2002) 3731–3738.
- [33] A.C. Coelho, S.M. Beverley, P.C. Cotrim, Functional genetic identification of PRP1, an ABC transporter superfamily member conferring pentamidine resistance in *Leishmania major*, Molecular Biochemistry and Parasitology 130 (2003) 83–90.

- [34] S.L. Croft, M.P. Barrett, J.A. Urbina, Chemotherapy of trypanosomiases and leishmaniosis, Trends in Parasitology 21 (11) (2005) 508–512.
- [35] P. Escobar, S. Matu, C. Marques, S.L. Croft, Sensitivities of *Leishmania* species to hexadecylphosphocholine (miltefosine), ET-18-0CH3 (edelfosine) and amphotericin B, Acta Tropica 81 (2002) 151–157.
- [36] WHO (World Health Organization), Research on Leishmaniosis, October 2010. Accessed on: http:// www.who.int/leishmaniosis/.
- [37] M. Basu, J. Mookerjee, A. Banerjee, R. Saha, M. Singh, S. Naskar, K. Tripathy, G. Sinha, P.K. Pandey, K. Sundar, S. Bimal, P.K. Das, S.K. Choudhuri, S. Roy, Inhibition of ABC transporters abolishes antimony resistance in *Leishmania* infection, Antimicrobial Agents and Chemotherapy 52 (2008) 1080–1093.
- [38] H. Sindermann, S.L. Croft, K.R. Engel, W. Bommer, H.J. Eibl, C. Unger, J. Engel, Miltefosine (Impavido): the first oral treatment against leishmaniosis, Medical Microbiology and Immunology 193 (2004) 173–180.
- [39] P.K. Sinha, T.K. Jha, C.P. Thakur, D. Nath, S. Mukherjee, A.K. Aditya, S. Sundar, Clinical study: phase 4 pharmacovigilance trial of paromomycin injection for the treatment of visceral leishmaniosis in India, Journal of Tropical Medicine 2011 (2011) 1–7.
- [40] A. Jhingran, B. Chawla, S. Saxena, M.P. Barrett, R. Madhubala, Paromomycin: uptake and resistance in *Leishmania donovani*, Molecular & Biochemical Parasitology 164 (2) (2008) 111–117.
- [41] L.F. Oliveira, A.O. Schubach, M.M. Martins, S.L. Passos, R.V. Oliveira, M.C. Marzochi, Systematic review of the adverse effects of cutaneous *Leishmaniosis* treatment in the new world, Acta Tropica 118 (2011) 87–96.
- [42] A.M.G. Brito, D. dos Santos, S.A. Rodrigues, R.G. Brito, L. Xavier-Filho, Plants with anti-*Leishmania* activity: integrative review from 2000 to 2011, Pharmacognosy Reviews 7 (13) (2013) 34–41.
- [43] S.R. Wilkinson, J.M. Kelly, Trypanocidal drugs: mechanisms, resistance and new targets, Experimental Reviews in Molecular Medicine 11 (2009) 1–24.
- [44] M. Yasinzai, M. Khan, A. Nadhman, G. Shahnaz, Drug resistance in leishmaniosis: current drugdelivery systems and future perspectives, Future Medicinal Chemistry 5 (2013) 1877–1888.
- [45] S. Alsford, J.M. Kelly, N. Baker, D. Horn, Genetic dissection of drug resistance in trypanosomes, Parasitology 140 (2013) 1478–1491.
- [46] S.L. Croft, V. Yardley, Chemotherapy of leishmaniosis, Current Pharmacology Design 8 (2002) 319–342.
- [47] J.P. Dedet, Protozoan Diseases: Leishmaniosis, Université Montpellier, Montpellier, France, 2008. http://dx.doi.org/10.1016/B978-012373960-5.00488-3 (accessed in February, 2010).
- [48] P. Desjeux, J. Alvar, *Leishmania*/HIV co-infection: epidemiology in Europe, Annals of Tropical Medicine and Parasitology 97 (2002) S3–S15.
- [49] B.B. Mishra, R.R. Kale, R.K. Singh, V.K. Tiwari, Alkaloids: future prospective to combat leishmaniosis, Fitoterapia 80 (2) (2009) 81–90.
- [50] L.H. Rosa, K.M.G. Machado, A.L.T. Rabello, E.M.S. Fagundes, R.C. Oliveira, A. Rosa, L. Zani, Cytotoxic, immunosuppressive, trypanocidal and antileishmanial activities of *Basidiomycota* fungi present in Atlantic Rainforest in Brazil, Antonie van Leeuwenhoek (Gedrukt) 95 (2009) 227–237.
- [51] F.M.D. Vítola, Antileishmanial Biocompounds Screening on Submerged Mycelial Culture Broths of Twelve Macromycete Species, Master BIODEV dissertation. Master in Biotechnological Processes, Federal University of Parana, 2008.
- [52] R.C. Fendrich, Production and Testing of Antileishmanial Compounds Obtained by Submerged, Master BIODEV dissertation. Master in Biotechnological Processes, Federal University of Parana, 2010.

- [53] W.S. Martin, Structures and evaluation of biologically active constituents of *Cussonia zimmermannii* Harms, Tesis for Erlangung der Würde eines Doktors der Philosophie (2006) 17–29.
- [54] K. Bhandari, N. Srinivas, V. Marrapu, A. Verma, S. Srivastava, S. Gupta, Synthesis of substituted aryloxy alkyl and aryloxy aryl alkyl imidazoles as antileishmanial agents, Bioorganic & Medicinal Chemistry Letters 20 (2010) 291–293.
- [55] R.L. Fabri, M.S. Nogueira, F.G. Braga, E.S. Coimbra, E. Scio, *Mitracarpus frigidus* aerial parts exhibited potent antimicrobial, antileishmanial, and antioxidant effects, Bioresource Technology 100 (2009) 428–433.
- [56] D. Tasdemira, R. Brunb, S.G. Franzblauc, Y. Sezgind, I. Çalis, Evaluation of antiprotozoal and antimycobacterial activities of the resin glycosides and the other metabolites of *Scrophularia cryptophila*, Phytomedicine 15 (2008) 209–215.
- [57] O. Rebollo, E. Olmo, G. Ruiz, J.L. López-Pérez, A. Giménez, A.S. Felicianoa, Leishmanicidal and trypanocidal activities of 2-aminocyclohexanol and 1,2-cyclohexanediamine derivatives, Bioorganic & Medicinal Chemistry Letters 18 (2008) 184–187.
- [58] M. Maarouf, F. Lawrence, S. Brown, Biochemical alterations in paromomycin-treated *Leishmania donovani* promastigotes, Parasitology Research 83 (1997) 198–202.
- [59] J.C.A. Tanaka, C.C. Silva, I.C.P. Ferreira, G.M.C. Machado, L.L. Leon, A.J.B. Oliveira, Antileishmanial activity of indole alkaloids from *Aspidosperma ramiflorum*, Phytomedicine 14 (2007) 377–380.
- [60] D. Sereno, A. Monte Alegre, A. Silvestre, B. Vergnes, A. Ouaissi, *In vitro* antileishmanial activity of nicotinamide, Antimicrobial Agents Chemotherapy 49 (2005) 808–812.
- [61] Y. Nakao, T. Shiroiwa, S. Murayama, S. Matsunaga, Y. Goto, Y. Matsumoto, N. Fusetani, Identification of Renieramycin A as an Antileishmanial substance in a marine sponge *Neopetrosia* sp. Marine Drugs 2 (2004) 55–62.
- [62] D. Savoia, C. Avanzini, T. Allice, E. Callone, G. Guella, F. Dini, Antimicrobial activity of euplotin C, the sesquiterpene taxonomic marker from the marine ciliate *Euplotes crassus*, Antimicrobial Agents Chemotherapy 48 (2004) 3828–3833.
- [63] P. Roch, A. Beschin, E. Bernard, Antiprotozoan and antiviral activities of non-cytotoxic truncated and variant analogues of mussel defensin, Evidence-Based Complementary Alternative Medicine 1 (2004) 167–174.
- [64] C.W. Roberts, R. McLeod, D.W. Rice, M. Ginger, M.L. Chance, L.J. Goad, Fatty acid and sterol metabolism: potential antimicrobial targets in apicomplexan and trypanosomatid parasitic protozoa, Molecular Biochemistry and Parasitology 126 (2003) 129–142.
- [65] M. Genestra, A. Echevarria, F. Cysne, L. Vignolio-Alves, L.L. Leon, Effect of amidine derivatives on nitric oxide production by *Leishmania amazonensis* promastigotes and axenic amastigotes, Nitric Oxide 8 (2003) 1–6.
- [66] G.R. Noleto, A.L.R. Mercê, M. Iacomini, P.A.J. Gorin, V. Thomaz-Soccol, M.B.M. Oliveira, Effects of a lichen galactomannan and its vanadyl(IV) complex on peritoneal macrophages and leishmanicidal activity, Molecular and Cellular Biochemistry 233 (2002) 73–83.
- [67] O. Kayser, A.F. Kiderlen, H. Laatsch, S.L. Croft, *In vitro* leishmanicidal activity of monomeric and dimeric naphtoquinones, Acta Tropica 76 (2000) 131–138.
- [68] R.E. Saenz, H. Paz, J.D. Berman, Efficacy of ketoconazole against *Leishmania braziliensis* panamensis cutaneous leishmaniosis, The American Journal of Medicine 89 (2) (1990) 147–155.
- [69] C. Williams, O.A. Espinosa, H. Montenegro, L. Cubilla, T.L. Capson, E. Ortega-Barría, L.I. Romero, Hydrosoluble formazan XTT: its application to natural products drug discovery for *Leishmania*, Journal of Microbiological Methods 55 (2003) 813–816.

- [70] S.T. Chang, P.G. Miles, Mushrooms: Cultivation, Nutritional Value, Medicinal Effect, and Environmental Impact, CRC Press LLC, NW, 2000, pp. 1–477.
- [71] A.H.S. Gomes, I.M. Armelin, S.Z. Menon, V.L. Pereira-Chioccola, *Leishmania* (V.) *braziliensis*: detection by PCR in biopsies from patients with cutaneous leishmaniosis, Experimental Parasitology 119 (2008) 319–324.
- [72] E. Chouihi, F. Amri, N. Bouslimi, E. Siala, K. Selmi, N. Zallagua, R.B. Abdallah, A. Bouratbine, K. Aoun, Les cultures sur milieu NNN dans le diagnostic biologique des leishmaniosis, Pathologie Biologie (2008) 32–33.
- [73] Roche Applied Science, Cytotoxicity Assay Methods. Methods for Studying Cytotoxicity. Apoptosis, Cell Death, and Cell Proliferation Manual, 2008, p. 69. Acessed on: https://www.roche-appliedscience.com/PROD\_INF/MANUALS/CELL\_MAN/apoptosis\_068\_069.pdf.
- [74] Roche Applied Science, Cell Proliferation/viability Assay Methods. Methods for Studying Cell Proliferation and Viability in Cell Populations. Apoptosis, Cell Death, and Cell Proliferation Manual, 2008, p. 85. Acessed on: https://www.roche-appliedscience.com/PROD\_INF/MANUALS/ CELL\_MAN/apoptosis\_085.pdf.
- [75] Roche Applied Science, Cell Proliferation/viability Assay. Methods for Studying Cell Proliferation and Viability in Individual Cells. Apoptosis, Cell Death, and Cell Proliferation Manual, 2008, p. 101. Accessed on: https://www.roche-appliedscience.com/PROD\_INF/MANUALS/CELL\_MAN/ apoptosis\_101.pdf.
- [76] R.A. Neal, *Leishmania major*: culture media, mouse strains, and promastigote virulence and infectivity, Experimental Parasitology 57 (1984) 269–273.
- [77] P. Agnew, P. Holzmuller, Y. Michalakis, D. Sereno, J.L. Lemesre, F. Renaud, *In vitro* growth of *Leishmania amazonensis* promastigotes resistant to pentamidine is dependent on interactions among strains, Antimicrobial Agents Chemotherapy 45 (2001) 1928–1929.
- [78] P.A. Bates, Complete developmental cycle of *Leishmania mexicana* in axenic culture, Parasitology 108 (1994) 1–9.
- [79] H.L. Callahan, A.C. Portal, R. Devereaux, M. Grogl, An axenic amastigote system for drug screening, Antimicrobial Agents Chemotherapy 41 (1997) 818–822.
- [80] J.M. Balanco, E.M. Pral, S. da Silva, A.T. Bijovsky, R.A. Mortara, S.C. Alfieri, Axenic cultivation and partial characterization of *Leishmania braziliensis* amastigote-like stages, Parasitology 116 (1998) 103–113.
- [81] D. Sereno, J.L. Lemesre, Axenically cultured amastigote forms as an *in vitro* model for investigation of antileishmanial agents, Antimicrobial Agents Chemotherapy 41 (1997) 972–976.
- [82] M. Ephros, A. Bitnun, P. Shaked, E. Waldman, D. Zilberstein, Stage-specific activity of pentavalent antimony against *Leishmania donovani* axenic amastigotes, Antimicrobial Agents Chemotherapy 43 (1999) 278–282.
- [83] D. Sereno, P. Holzmuller, J.L. Lemesre, Efficacy of second line drugs on antimonyl-resistant amastigotes of *Leishmania infantum*, Acta Tropica 74 (2000) 25–31.
- [84] V. Yardley, S.L. Croft, S. De Doncker, J.C. Dujardin, S. Koirala, S. Rijal, The sensitivity of clinical isolates of *Leishmania* from Peru and Nepal to miltefosine, American Journal of Tropical Medicine and Hygiene 73 (2005) 272–275.
- [85] A.G. Tempone, D. Perez, S. Rath, A.L. Vilarinho, R.A. Mortara, H.F. de Andrade, Targeting *Leishmania* (*L.*) *chagasi* amastigotes through macrophage scavenger receptors: the use of drugs entrapped in liposomes containing phosphatidylserine, Journal of Antimicrobial Chemotherapy 54 (2004) 60–68.
- [86] A. Gebre-Hiwot, G. Tadesse, S.L. Croft, D. Frommel, An *in vitro* model for screening antileishmanial drugs: the human leukaemia monocyte cell line, THP-1, Acta Tropica 51 (1992) 237–245.
- [87] S. Carryn, F. Van Bambeke, M.P. Mingeot-Leclercq, P.M. Tulkens, Activity of beta-lactams (ampicillin, meropenem), gentamicin, azithromycin and moxifloxacin against intracellular *Listeria monocytogenes* in a 24 h THP-1 human macrophage model, Journal of Antimicrobial Chemotherapy 51 (2003) 1051–1052.
- [88] M.E. Ibrahim, M. Hag Ali, A.M. El-Hassan, T.G. Theander, A. Kharazmi, *Leishmania* resistant to sodium stibogluconate: drug associated macrophage-dependent killing, Parasitology Research 80 (1994) 569–574.
- [89] D. Sereno, M. Cavaleyra, K. Zemzoumi, S. Maquaire, A. Ouaissi, J.L. Lemesre, Axenically grown amastigotes of *Leishmania infantum* used as an *in vitro* model to investigate the pentavalent antimony mode of action, Antimicrobial Agents and Chemotherapy 42 (1998) 3097–3102.
- [90] A.P.D. Rodrigues, L.H.S. Farias, A.S.C. Carvalho, A.S. Santos, J.L.M. do Nascimento, et al., A novel function for Kojic acid, a secondary metabolite from *Aspergillus* fungi, as antileishmanial agent, PLoS One 9 (3) (2014) e9125.
- [91] T. Okuno, Y. Goto, Y. Matsumoto, H. Otsuka, Y. Matsumoto, Applications of recombinant *Leishmania amazonensis* expressing *egfp* or the b-galactosidase gene for drug. Screening and histopathological analysis, Experimental Animal 52 (2003) 109–118.
- [92] N. Singh, A. Dube, Short report: fluorescent *Leishmania*: application to anti-leishmanial drug testing, American Journal of Tropical Medicine and Hygiene 71 (2004) 400–402.
- [93] A.S.S. Aliança, The *in vitro* biological activity of the brazilian brown seaweed dictyota mertensii against *Leishmania amazonensis*, Molecules 19 (2014) 14052–14065.
- [94] T.S. Tiuman, Cell death in amastigote forms of *Leishmania amazonensis* induced by parthenolide, BMC Microbiology 14 (2014) 1–12.
- [95] S.M. Abdulah, et al., Comparison of different staining procedures for the flow cytometric analysis of U-937 cells infected with different *Leishmania*-species, Journal of Microbiology Method 37 (1999) 123–138.
- [96] J.L. Siqueira-Neto, et al., Antileishmanial high-throughput drug screening reveals drug candidates with new scaffolds, PLoS Neglected Tropical Diseases 4 (2010) 1–8.
- [97] J.L. Siqueira-Neto, et al., An image-based high-content secreening assay for compounds targeting intracellular *Leishmania donovani* amastigotes in human macrophages, PLoS Neglected Tropical Diseases 6 (2012) 1–9.
- [98] G.S. Goncalves, A.P. Fernandes, R.C. Souza, J.E. Cardoso, F. de Olivira-Silva, F.C. Maciel, et al., Activity of a paromomycin hydrophilic formulation for topical treatment of infections by *Leishmania (Leishmania) amazonensis* and *Leishmania (Vianna) braziliensis*, Acta Tropica 93 (2005) 161–167.
- [99] J. Moreno, J. Alvar, Canine leishmaniosis: epidemiological risk and the experimental model, Trends Parasitology 18 (9) (2002) 399–405.
- [100] V. Yardley, S.L. Croft, Animal models of cutaneous leishmaniosis, in: O. Zak (Ed.), Handbook of Animal Models of Infection, Academic Press, New York, London, 1999, pp. 775–781.
- [101] J. Samuelson, E. Lerner, R. Tesh, R. Titus, A mouse model of *Leishmania braziliensis braziliensis* infection produced by co-injection with sand fly saliva, Journal Experimental Medicine 173 (1991) 49–54.
- [102] S.G. Andrade, A. Carneiro Filho, A.J. de Souza, E.S. de Lima, Z.A. Andrade, Influence of treatment with immunosuppressive drugs in mice chronically infected with *Trypanosoma cruzi*, International Journal of Experimental Pathology 78 (1997) 391–399.
- [103] S. Bretagne, R. Durand, M. Olivi, J.F. Garin, A. Sulahian, D. Rivollet, Real-time PCR as a new tool for quantifying *Leishmania infantum* in liver in infected mice, Clinical and Diagnostic Laboratory Immunology Journal 8 (2001) 828–831.

- [104] L. Nicolas, E. Prina, T. Lang, G. Milon, Real-time PCR for detection and quantitation of *Leishmania* in mouse tissues, Journal of Clinical of Microbiology 40 (2002) 1666–1669.
- [105] W.M.S. Russell, R.L. Burch, The Principles of Humane Experimental Technique, Methuen, London, 1959, ISBN 0900767782.
- [106] M.A. Dea-Ayuela, E. Castillo, M. Gonzalez-Alvarez, C. Veja, M. Rolón, F. Bolás-Fernández, et al., *In vivo* and *in vitro* anti-leishmanial activities of 4-nitro-N-pyrimidinand N-pyrazin-2ylbenzenesulfonamides, and N2-(4-nitrophenyl)-N1-propylgly cinamide, Bioorganic & Medicinal Chemistry 17 (2009) 7449–7456.
- [107] I.C. Ibraim, R.R. de Assis, N.L. Pessoa, M.A. Campos, M.N. Melo, S.J. Turco, R.P. Soares, Two biochemically distinct lipophosphoglycans from *Leishmania braziliensis* and *Leishmania infantum* trigger different innate immune responses in murine macrophages, Parasites & Vectors 6 (2013) 54.
- [108] B.S. Lakshmi, R. Wang, R. Madhubala, *Leishmania* genome analysis and high-throughput immunological screening identifies tuzin as a novel vaccine candidate against visceral leishmaniasis, Vaccine 32 (2014) 3816–3822.
- [109] D. Gamboa, K. Torres, S. De Doncker, M. Zimic, J. Arevalo, J.C. Dujardin, Evaluation of an *in vitro* and *in vivo* model for experimental infection with *Leishmania (Viannia) braziliensis* and *L. (V.) peruviana*, Parasitology 135 (2008) 319–326.
- [110] A. Gomes-Silva, J.G. Valverde, R.P. Ribeiro-Romão, R.M. Plácido-Pereira, A.M. Da-Cruz, Golden hamster (*Mesocricetus auratus*) as an experimental model for *Leishmania (Viannia) braziliensis* infection, Parasitology 140 (2013) 771–779.
- [111] S.M. Robledo, W. Cardona, K. Ligardo, J. Henao, N. Arbeláez, A. Montoya, F. Alzate, J.M. Pérez, V. Arango, I.D. Vélez, J. Sáez, Antileishmanial effect of 5,3'-hydroxy-7,4'-dimethoxyflavanone of *picramnia gracilis Tul.* (Picramniaceae) fruit: *in vitro* and *in vivo* studies, Advances in Pharmacological Science 2015 (2015) 1–8.
- [112] H. Zangger, C. Ronet, C. Desponds, F.M. Kuhlmann, J. Robinson, M.A. Hartley, F. Prevel, et al., Detection of *Leishmania* RNA virus in *Leishmania* parasites, PLoS Neglected Tropical Diseases 7 (2013) 1–11.
- [113] B. Ezatpour, E. Saedi Dezaki, H. Mahmoudvand, M. Azadpour, F. Ezzatkhah, *In vitro* and *in vivo* antileishmanial effects of *Pistacia khinjuk* against *Leishmania tropica* and *Leishmania major*, Evidence-Based Complementary and Alternative Medicine 2015 (2015) 1–6.
- [114] A. Fortin, D.P. Caridha, S. Leed, F. Ngundam, J. Sena, T. Bosschaerts, et al., Direct comparison of the efficacy and safety of oral treatments with oleylphosphocholine (OIPC) and miltefosine in a mouse model of *L. major* cutaneous leishmaniasis, PLoS Neglected Tropical Disease 8 (2014) 1–7.
- [115] C.T. Trinconi, J.Q. Reimao, J.K.U. Yokoyama-Yasunaka, D.C. Miguel, S.R.B. Uliana, Combination therapy with tamoxifen and amphotericin B in experimental cutaneous leishmaniasis, Antimicrobial Agents Chemotherapy 58 (2014) 2608–2613.
- [116] S. Sifontes-Rodriguez, L. Monzote-Fidalgo, N. Castañedo-Cancio, A.M. Montalvo-Álvarez, Y. López-Hernández, N.M. Diogo, et al., The efficacy of 2-nitrovinylfuran derivatives against *Leishmania in vitro* and in vivo, Memorias do Instituto Oswaldo Cruz 110 (2015) 166–173.
- [117] M.B.H. Carneiro, L.M. de Andrade e Sousa, L.G. Vaz, L.M. Dos Santos, L. Vilela, C.C. de Souza, et al., Short-term protection conferred by Leishvacin® against experimental *Leishmania amazonensis* infection in C57BL/6 mice, Parasitology International 63 (2014) 826–834.
- [118] H. Mahmoudvand, F. Sharififar, M.S. Rahmat, R. Tavakoli, Evaluation of antileishmanial activity and cytotoxicity of the extracts of *Berberis vulgaris* and *Nigella sativa* against *Leishmania tropica*, Journal of Vectors Borne Diseases 51 (2014) 294–299.

- [119] L. Paranaíba, R.R. de Assis, P.M. Nogueira, A.C. Torrecilhas, J.H. Campos, A.C. Silveira, Leishmania enriettii: biochemical characterisation of lipophosphoglycans (LPGs) and glycoinositolphospholipids (GIPLs) and infectivity to *Cavia porcellus*, Parasites and Vectors 8 (2015) 1–14.
- [120] K.P. Chang, S.G. Reed, B.S. McGwire, L. Soong, *Leishmania* model for microbial virulence: the relevance of parasite multiplication and pathoantigenicity, Acta Tropica 85 (2003) 375–390.
- [121] C. Cantacessi, F. Dantas-Torres, M.J. Nolan, D. Otranto, The past, present, and future of *Leishmania* genomics and transcriptomics, Trends in Parasitology 31 (3) (2015) 100–108.
- [122] M.A. Moreno, A. Alonso, P.J. Alcolea, A. Abramov, M.G. Lacoba, J. Abendroth, S. Zhang, T. Edwards, D. Lorimer, P.J. Myler, V. Larraga, Tyrosine aminotransferase from *Leishmania infantum*: a new drug target candidate, International Journal for Parasitology: Drugs and Drug Resistance 4 (2014) 347–354.
- [123] S. Khademvatan, N. Adibpour, A. Eskandari, S. Rezaee, M. Hashemitabar, F. Rahim, *In silico* and *in vitro* comparative activity of novel experimental derivatives against *Leishmania major* and *Leishmania infantum* promastigotes, Experimental Parasitology 135 (2013) 208–216.
- [124] V. Roldos, H. Nakayama, M. Rolon, A. Montero-Torres, F. Trucco, S. Torres, C. Vega, Y. Marrero-Ponce, V. Heguaburu, G. Yaluff, A. Gomez-Barrio, L. Sanabria, M.E. Ferreira, A. Rojas de Arias, E. Pandolfi, Activity of a hydroxybibenzyl bryophyte constituent against *Leishmania* spp. and *Trypanosoma cruzi: in silico, in vitro* and *in vivo* activity studies, European Journal of Medicinal Chemistry 43 (2008) 1797–1807.
- [125] A. Astashkina, B. Mann, D.W. Grainger, A critical evaluation of in vitro cell culture models for high-throughput drug screening and toxicity, Pharmacology and Therapeutics 134 (2012) 82–106.

# 23

# Factors/Genes in Maternal Recognition of Pregnancy: An Overview

#### S. Mondal\*, A. Mor, I.J. Reddy

NATIONAL INSTITUTE OF ANIMAL NUTRITION AND PHYSIOLOGY, BANGALORE, INDIA

### 23.1 Introduction

Reproduction is a highly complex, luxurious biological process requiring a dialogue between the developing conceptus and maternal uterus that must be established during the periimplantation period for pregnancy recognition signaling and the regulation of gene expression by uterine epithelial and stromal cells. The greatest limitation to reproductive efficiency across mammalian species is early embryonic mortality. Early embryonic mortality denotes the death of fertilized ova and embryos up to the end of implantation [1]. It is a major source of embryonic and economic loss, with a mortality rate of up to 40% in animal production through repeat breeding and the increased cost of artificial insemination [2-4]. Most losses are characterized by asynchrony between conceptus (embryo/fetus and associated membranes) signals and uterine receptivity, resulting in defective implantation and/or placentation. Suboptimal communication, resulting from impairment of conceptus (embryo and associated extraembryonic membranes) development and/or from abnormal uterine receptivity [5], contributes to a high incidence of embryonic mortality. Understanding and unraveling the pregnancy recognition, secrets of implantation, embryo development, and reciprocal signaling networks between the embryo and uterus will lead to alleviation of the problems of infertility. Early embryonic development, including implantation of the fetus and maintenance of pregnancy, is critically influenced by embryo-maternal cross-talk [6]. The survival of the embryo during early embryonic life mostly depends on the efficiency with which the maternal recognition of pregnancy (MRP) is established [7]. Pregnancy recognition signaling, as related to sustaining the functional lifespan of the corpora lutea, is required to sustain the functional lifespan of the corpora lutea for the production of progesterone  $(P_4)$ , which is essential for uterine functions supportive of

<sup>&</sup>lt;sup>\*</sup>Corresponding Author.

implantation and placentation required for successful outcomes of pregnancy. In ruminants, the main signal for MRP is identified as interferon- $\iota$  (IFN $\tau$ ) secreted by the trophoectoderm of blastocysts for a limited period during early pregnancy. Oxytocin (OT) triggers the pulsatile release of prostaglandin (PG) $F_{2\alpha}$  by the endometrium at the end of an infertile estrous cycle to induce luteolysis. Interferon-i abrogates the uterine luteolytic mechanism by suppressing the normal pattern of the pulsatile release of  $PGF_{2\alpha}$ in late estrous cycle by a mechanism that involves the down-regulation of estrogen receptor in the uterine epithelium which in turn prevents a rise in OT receptor (OTR) [8]. IFN $\tau$  acts in concert with P<sub>4</sub> to induce the expression of genes for transport and/or secretion of histotrophs that includes nutrients such as glucose and arginine that activate the mechanistic target of rapamycin nutrient-sensing cell signaling pathway to stimulate proliferation, migration, differentiation, and translation of messenger RNAs (mRNAs) essential for growth and development of the conceptus. Several factors such as genetic abnormalities, nutrition, environment, hormonal imbalance, thermal stress, impaired maternal-fetal interaction, and infections are responsible for impaired communication between the embryo and the dam. There is an urgent need to unravel the complexities of maternal recognition of pregnancy and implantation to address two contrasting global issues: to improve fertility and develop novel contraceptives. The challenge is to understand the complex molecular dialogue between the conceptus and maternal uterine microenvironments that are characteristic of successful implantation and to use that knowledge to develop optimal reproductive management strategies and paradigms to improve embryo survival. Advances in genomic technologies will certainly enhance our understanding of the underlying biological processes involved in fertilization and the establishment of pregnancy and to explore gene profiles and the potential to identify genes responsible for improved embryo survival.

# 23.2 Maternal Recognition of Pregnancy

Higher rates of early embryonic mortality are a major cause of reproductive failure during the preimplantation period in livestock. These preimplantation losses are associated with inadequate luteal function during early pregnancy in ruminants [9]. The survival of embryo during early embryonic life mostly depends on the efficiency with which the MRP is established [7]. The term "MRP" was coined by Roger Short and can be defined as a delicate and sensitive process that involves the interaction between a conceptus and the maternal endometrium. It is defined as the uterine recognition of conceptus(es)' signal and extension of the ovarian corpus luteum (CL) lifespan, which occurs in a coordinated synchronous manner if pregnancy is established. It is critical so that the endometrium does not secrete PGF<sub>2α</sub>, lysing the CL and destroying the source of  $P_4$ , the hormone necessary to sustain a pregnancy. There are two categories of mechanism of MRP: luteotrophic and antiluteolytic. Luteotrophic directly promotes luteal function, supporting the CL, and antiluteolytic prevents the uterine release of luteolytic PGF, which causes the lysis of the CL. Humans, nonhuman primates, and rodents

maintain luteal function through luteotrophic mechanisms. In primates the maternal recognition signal is chorionic gonadotropin (CG). The syncytiotrophoblast cells secrete CG from days 8 to 10 postovulation for pregnancy recognition. CG stimulates the CL to produce  $P_4$  until there is a shift between luteal and placental  $P_4$  secretion. Once the shift occurs, CG secretion decreases. Rodents also maintain luteal function through luteotrophic mechanisms; prolactin is the MRP signal. Prolactin is the initial maternal recognition signal until day 12 of pregnancy; after day 12 the conceptus and uterine decidua take over, stimulating  $P_4$  secretion by secreting lactogenic hormones (prolactins). These hormones act on the luteal cells through the prolactin receptors to maintain luteal cell function and the secretion of  $P_4$  (Figs. 23.1 and 23.2).

Sheep, cattle, and goats have estrous cycles of about 17, 21, and 20 days, respectively. An ovulatory surge of luteinizing hormone (LH) occurs near the time of onset of estrus that leads to ovulation about 30 h later. With the development of CL, the concentration of P<sub>4</sub> in peripheral circulation peaks around the mid to late diestrus phase and luteolysis is induced by the pulsatile release of  $PGF_{2\alpha}$  from endometrial epithelium on day 15 and/or 16. The luteolytic mechanism that develops in the endometrial luminal epithelium (LE) and surface GE (sGE) requires sequential effects of  $P_4$ , estrogen, and oxytocin, acting through their respective receptors [10,11]. At estrus (day 0), estrogens from graafian follicles stimulate increased uterine estrogen receptor alpha (ER $\alpha$ ), P<sub>4</sub> receptor (PR), and OTR expression [12,13]. During early diestrus, P<sub>4</sub> from the newly formed CL stimulates accumulation of phospholipids in LE and sGE that can liberate arachidonic acid for the synthesis and secretion of  $PGF_{2\alpha}$ . During diestrus P<sub>4</sub> levels increase and act via PR to "block" expression of ER $\alpha$  and OTR in the endometrial LE and sGE [14]. Therefore  $ER\alpha$  and OTR expression is not detected between days 5 and 11 of the cycle, i.e., during most of diestrus. In sheep and cows endometrial receptors for OT were present between estrus and about day 4 of the cycle; they were low or undetectable



**FIGURE 23.1** Events of maternal recognition of pregnancy. *ER*, Estrogen receptor; *IFN*- $\tau$ -Interferon tau; *IFNAR*, type 1 IFN receptor; *IRF2*- Interferon regulatory factor two; *OTR*, Oxytocin receptor; *PR*, Progesterone receptor.



**FIGURE 23.2** IFNτ signaling in uterine glandular epithelium. *GAF*, gamma interferon activation factor; *GAS*, gamma interferon activation sequence; *IFNT*, IFNτ; *IFNAR*, type I IFN receptor; *IRF1*, IFN regulatory factor; *ISGF3G-IFN*, stimulated transcription factor-3, gamma 48-k Da; *ISG15*, IFNτ-stimulated gene-15; *ISGF3*, IFN-stimulated transcription factor-3; *JAK1*, janus kinase-1; *STAT*, signal transducers and activators of transcription.

between days 5 and 13 and increased rapidly between days 14 and 16 in sheep [15] or days 17 and 20 in cows [16]. Endometrial ER and PR were highest during the first 10-12 days after onset of estrus in cows, and then both declined to their lowest levels on about day 13. ER then increased between days 14 and 15, with OTR increasing rapidly between days 17 and 21. The cytosolic PR concentrations were highest during the second stage of estrous cycle (days 18-20 of the estrous cycle) and lowest during the second stage of the estrous cycle (days 5-10 of the cycle) in buffalo endometrium [17]. High P<sub>4</sub> levels during the second phase of the estrous cycle antagonize the ability of estrogen to promote the synthesis and/or replenishment of uterine steroid receptors whereas cytosolic estradiol receptor concentrations were higher in follicular phase compared with luteal phases [18]. Rising concentrations of estradiol in peripheral plasma shortly before estrus may be the physiologic cause of increased concentration of estradiol receptors in the follicular phase.

MRP in ruminants (sheep, cattle, and goats) requires the conceptus elongate to be from a spherical to a tubular and then a filamentous form to produce IFN $\tau$ , which is the pregnancy recognition signal that prevents the development of the endometrial luteolytic mechanism [10,19]. During MRP, the mononuclear cells of the conceptus trophectoderm synthesize and secrete IFN $\tau$  between days 10 and 21–25 with maximal production on days 14–16 [19]. IFN $\tau$  appears to be the sole factor produced by the conceptus that prevents development of the endometrial luteolytic mechanism. It is a novel member of the type I IFN family. It acts differentially on the endometrial LE, glandular epithelium (GE), and stroma to regulate expression of a number of IFNstimulated genes (ISGs) that have roles in endometrial differentiation and conceptus implantation. The IFN $\tau$  gene has a 595–base pair (bp) open reading frame that encodes a 195–amino acid preprotein containing a 23–amino acid signal sequence and is cleaved to yield the mature protein. The IFN $\tau$  genes expressed by ruminant conceptuses share approximately 70% homology with IFNw. It shows remarkable homology with the complementary DNA (cDNA) nucleotide sequence across ruminant species. Bovine, ovine, and caprine IFN $\tau$  are more similar to each other in sequence. In the coding region, the nucleotide sequences exhibit approximately 90% identity, and their inferred amino acid sequences about 80% identity. IFN $\tau$  is thought to act antiluteolytic by:

- 1. stabilization or up-regulation of P<sub>4</sub> receptors in the endometrium;
- 2. direct inhibition of endometrial estrogen receptors;
- 3. direct inhibition of endometrial oxytocin receptors;
- **4.** initiation of postreceptor mechanisms, which prevent oxytocin-induced release of PGF; and/or
- **5.** induction of the endometrium to synthesize an inhibitor of enzymes necessary for the synthesis of  $PGF_{2\alpha}$  [20,21].

In sheep, it is secreted between days 10 and 21 by the mononuclear trophoblast cells. In vitro culture (IVC) of sheep conceptus homogenates and analysis for radiolabeled proteins released into the culture medium indicate that ovine trophoblastic protein-1 (oTP-1) is secreted by mononuclear cells of ovine trophoectoderm. oTP-1 is secreted in two phases: one between days 10 and 21 of pregnancy and the other by chorion between days 25 and 45 of pregnancy. oTP-1 is thought to exert a paracrine antiluteolytic effect on endometrium because there is no evidence that it is transported from the uterus to affect CL directly. Secretion of oTP-1 (ng/uterine flushing) begins on about day 10 and increases with morphologic changes: from spherical (312 ng) to tubular (1380 ng) to filamentous (4455 ng) forms on days 12-13. Goat conceptus secretes caprine IFN $\tau$ between days 16 and 21, which is assumed to be an antiluteolytic signal. Intrauterine injections of recombinant ovine IFN $\tau$  in cyclic goats extend the CL lifespan in goats. Pulsatile release of oxytocin and  $PGF_{2\alpha}$  is suppressed in pregnant compared with cvclic goats between days 10 and 12 and estrus or day 20 of pregnancy, which suggests that the antiluteolytic mechanism mediated by IFN $\tau$  is similar for sheep, cows, and goats. The endocrine-exocrine theory of MRP in pigs was studied. In pigs, MRP and antiluteolysis are not mediated by trophoblastic IFNs, but by trophoblast-secreted estrogen from days 10.5 to 11 [20]. A second period of estrogen production occurs between days 15 and 25-30 of pregnancy. Injection of exogenous estrogen on days 11-15 of the estrous cycle results in CL maintenance for a period equivalent to or slightly longer than pregnancy (about 120 days later). This condition is referred to as pseudopregnancy [22], whereas a single injection of estradiol on day 9.5, 11, 12.5, 14, 15.5, or 14–16 results in interestrous

intervals of about 30 days. Estradiol must be administered to gilts on day 11 and days 14–16 or daily from days 11 to 15 to obtain interestrous intervals of greater than 60 days. This suggests that two phases of estradiol, similar to those produced by conceptuses on days 11–13 and days 15 to 25–30, are necessary for prolonged secretion of  $PGF_{2\alpha}$  into the uterine lumen. PGF from the uterus is taken up by the mesometrium and transferred to the ovary in arterial blood by a countercurrent system that exists in the broad ligament of the uterus [23]. Estrogen also changes the ratio of  $PGE_2$  to  $PGF_{2\alpha}$  [24,25] and maintains LH receptor levels in both CL and the uterus [26].  $PGE_2$  may protect the CL from the luteolytic effects of  $PGF_{2\alpha}$ . Estradiol may induce receptors for maternal hormones, e.g., prolactin, or conceptus secretory proteins, which influence the exocrine secretion of  $PGF_{2\alpha}$  [27]. The first estrogen signal may induce those receptors and the second estrogen signal may be required to replenish those receptors. Available evidence indicates that estrogens of blastocyst origin are essential for MRP in pigs and that secretory proteins from the conceptus, including IFNs, have other roles during early pregnancy in pigs, including cellular changes that lead to increased polarity of endometrial epithelial cells for exocrine secretion of components of histotrophs, including  $PGF_{2\alpha}$  [28].

The IFN<sup>T</sup> novel member of the type I IFN family acts differentially on the endometrial LE, GE, and stroma to regulate expression of a number of ISGs that are hypothesized to have roles in endometrial differentiation and conceptus implantation [29]. All type I IFNs bind a common receptor composed of two subunits, IFNAR1 and IFNAR2, to induce cell signaling via the Janus-activated kinase (JAK) and tyrosine kinase 2 (TYK2) pathways, respectively.

IFNτ binds to a common type I IFN receptor, IFNAR1 and IFNAR2, containing tyrosine kinases such as JAK1 and TYK2 to activate the JAK/STAT cell signaling pathway. IFNτ-mediated association of IFNAR subunits facilitates cross-phosphorylation and activation of JAK1 and Tyk2, which in turn phosphorylates the receptor and creates a docking site for STAT2. STAT2 is then phosphorylated, thus creating a docking site for STAT1, which is then phosphorylated. Phosphorylated STAT1 binds phosphorylated STAT2 to form a heterodimer and translocates to the nucleus after forming a hetero-trimeric transcriptional complex by binding with ISGF3G, collectively termed ISGF3, which binds to IFN-stimulated response element (ISRE) in promoter regions of ISGs and transactivates genes containing an ISRE(s), such as STAT1, STAT2, IRF9, B2M, ISG15, MIC, and OAS. In addition to STAT1/2 heterodimerization, type I IFN induces the formation of phosphorylated STAT1 homodimers termed GAF, which enter the nucleus and transactivate genes containing a GAS element(s) such as IRF1. IRF1 can also bind and transactivate ISRE-containing genes as well as IRFE-containing genes.

# 23.3 Gene Regulating Maternal Recognition of Pregnancy Events

Successful embryo development and survival include the formation of the blastocyst, implantation into the uterus, the formation of the placenta, the development of the

heart, and vascularization of both the embryo and fetus to assist nutrient deliverance. Among these developmental events, implantation of the embryo is a crucial step and its success mostly depends on the efficiency with which the MRP is established [7]. The MRP includes a series of events that are synchronized by the endocrine interaction between the mother and the embryo. The genes regulating MRP are discussed here.

#### 23.3.1 Prostaglandin E Synthase

PGs have, important roles in ovulation, luteolysis, implantation, decidualization and parturition. The production of PGs is primarily catalyzed by two rate-limiting enzymes, e.g., cyclooxygenase-1 (COX-1) and -2 (COX-2), also referred to as PG endoperoxidase H synthase-1 (PGHS-1) and -2 (PGHS-2), respectively. PGHS-1 and PGHS-2 are responsible for the conversion of arachidonic acid into PGG<sub>2</sub> and PGG<sub>2</sub> to PGH<sub>2</sub>, respectively. The downstream enzymes, PG E synthase (PGES) and PG F synthase (PGFS), catalyze the conversion of PGH<sub>2</sub> to PGE<sub>2</sub> and PGF<sub>2a</sub>, respectively. PGE<sub>2</sub> protects the CL from spontaneous regression and helps maintain pregnancy [30]. It enhances endometrial vascular permeability at the sites of blastocyst apposition and decidualization, blastocyst hatching, and implantation. In contrast,  $PGF_{2\alpha}$  acts as the luteolytic agent in ruminants, which regresses the CL and causes termination of the estrous cycle or pregnancy. PGES is a member of the membrane-associated protein in eicosanoid and glutathione metabolism superfamily, which consists of six proteins with divergent functions [31]. The activity of PGES has been detected both in cytosolic and microsomal fractions of various cells, and in most cells it requires glutathione (GSH) as a coenzyme for optimal activity. Cytosolic PGES (cPGES) is constitutively expressed in a wide variety of cells and tissues and is predominantly linked with COX-1 to promote immediate response during which relatively high concentrations of arachidonic acid are released in a short period [32], whereas microsomal PGES (mPGES) is a membrane-associated, inducible perinuclear enzyme with GSH-dependent activity and is expressed in variety of tissues including prostate, testis, and small intestine [31]. There are three isoforms of PGES in bovines: mPGES-1, mPGES-2, and cPGES [33]. Unlike mPGES-1, mPGES-2 does not require GSH for catalytic activity and is activated in the presence of a broad range of reducing agents. Although three forms of PGES have been characterized, mPGES-1 is the main PGES responsible for PGE<sub>2</sub> production in endometrial cells. The PGES gene structure has been evaluated in human [34]. The gene size is 14.8 kb and has been mapped to chromosome 9. The genomic sequences of human PGES (Acc. No. AC 007,936) indicate three exons (136, 83, and 1526 nt) separated by two introns (4.2 and 8.8 kb). The complete cDNA sequences of PGES have been characterized in rats (Acc. No. NM 0,211,583; AF 280,967), cattle (Acc. No. AY 032,727), horses (Acc. No. AY 057,096), and pigs (Acc. No. AY 857,634). Filion et al. [35] reported the complete cDNA sequences of 462 bp of bovine PGES (Acc. No. AY 032,727). Partial PGES gene sequences are available (Acc. No. AY260142) in cattle. The nucleotide sequence of buffalo PGES cDNA encoding the entire mature peptide except the first three amino acids is also available (Acc. No. DQ167808).

Expression of mRNA encoding for the PGES has been evaluated in a variety of tissues such as uterus, CL, placenta, and ovary. Expression of mPGES-1 mRNA was high between days 13 and 21 of the cycle in bovines, which indicated that the bovine endometrium becomes responsive to produce PGE<sub>2</sub> after day 13 of the cycle [36]. However no variation in PGES mRNA levels was observed during late diestrus and early pregnancy in horses [37]. Even infusion of PGE<sub>2</sub> did not extend the lifespan of CL and PGE<sub>2</sub> did not induce a luteotropic effect in cultured luteal cells [38]. Moreover, equine conceptus produced huge amounts of  $PGE_2$  during days 10–16 of the cycle.  $PGE_2$  produced by the endometrium supplemented  $PGE_2$  produced by the conceptus rather than playing a luteotropic role. Waclawik et al. [39] also reported that there was no significant variation in mPGES-1 expression throughout the estrous cycle in pigs. In contrast, abundant levels of mPGES-1 protein were expressed in endometrium during the proliferative phase, whereas a low expression was observed during the late-secretory phase of the menstrual cycle in humans owing to a decrease in  $PGE_2$  production by the endometrium [40]. The expression of mPGES-1 mRNA could not be detected in the cyclic endometrium of buffalo during the first (days 3-5) and second (days 6-15) stages but it was observed in the endometrium during the third stage (days 16-21) of the estrous cycle [41]. Limited reports are available on PGES regulation in uterus in vitro studies using cultured endometrial cells. Tanioka et al. [32] observed that GSH-dependent PGES activity increased in vivo threefold after lipopolysaccharide (LPS) challenge in cytosol of rat brain, but not of other tissues. PGES mRNA expression is also increased in the presence of LPS, tumor necrosis factor- $\alpha$  (TNF- $\alpha$ ), and IFN $\tau$  in epithelial cells [42]. They also reported that phorbol-12 myristate-13 acetate increased PGES mRNA, COX-2 mRNA, and PGE<sub>2</sub> production, which suggested that the expression of PGES was correlated with that of COX-2 for the production of PGE<sub>2</sub>. Increasing production modulated the PGE<sub>2</sub>–PGF<sub>2</sub> $\alpha$ ratio, leading to the establishment of a successful pregnancy.

#### 23.3.2 Prostaglandin F Synthase

PGFS is a monomeric protein with a molecular weight of about 36,000, which converts  $PGH_2$  to  $PGF_{2\alpha}$ . PGFS is a member of the aldo-keto reductase (AKR) superfamily based on substrate specificity, molecular weight, and amino acid sequence. The AKR superfamily consists of monomeric NAD(P)H-dependent oxido-reductase with a broad specificity of substrates: endogenous substrates such as steroid hormones, PGs, and xenobiotics such as drugs and carcinogens. The PGFS gene consists of nine exons and eight introns. The initiation codon is located at exon 1 and the stop codon at exon 9 and encodes a protein of 323 amino acid residues. The complete cDNA sequence of PGFS has been characterized in cattle (Acc. No. J03570), buffalo (Acc. No. DQ884879), sheep (Acc. No. AY135401), horses (Acc. No. AY304536), and pigs (Acc. No. AY863054). PGFS is known to exist in six isoforms: lung type PGFS (PGFS1), lung type PGFS found in liver (PGFS2), liver type PGFS (DDBX), and three other PGFS isolated from human, sheep, and *Trypanosoma brucei*. All of these isoforms belong to the AKR1C subfamily.

except *Trypanosoma*, which belongs to the AKR5A subfamily. In bovine, six isoforms of PGFS [PGFS1, PGFS2, DDBX, PGFS like 1 (PGFSL-1), PGFSL-2, and  $20\alpha$ -HSD] have been identified [43]. A new isoform, AKR1B5, possessing aldoreductase activity has been implicated as the most probable enzyme responsible for production of PGF<sub>2α</sub> in bovines [43]. Kuchinke et al. [44] reported the complete cDNA sequence of 972 bps of bovine lung PGFS (Acc. No. M86544). The cDNA sequences of bovine PGFSL-1 protein (Acc. No AY135400) and PGFSL-2 protein (Acc. No. AY135401) are of 972 bp. The nucleotide sequence of buffalo PGFSL-2 cDNA exhibited 98%, 87.6%, 87.3%, 77.2%, 68.9%, 68.7%, 65%, and 77.7% identity with bovine PGFSL-2, bovine lung PGFS, bovine liver PGFS, horses, pigs, monkeys, dogs, and humans, respectively [41].

The expression of PGFS mRNA has been studied in domestic ruminants including cattle, buffalo, sheep, horses, pigs, monkeys, and rats. Madore et al. [43] reported that no known PGFS enzymes belonging to AKR1C family (PGFS1, PGFS2, PGFSL-1, PGFSL-2, and DDBX) were expressed in the bovine endometrium during the estrous cycle. Interestingly, it was observed that AKR1B5 expression was up-regulated in the endometrium during the second half of the estrous cycle. PGFS expression was found to up-regulate around luteolysis in pigs [39]. However expression of PGFS mRNA was not up-regulated during luteolysis and pregnancy in horses, which suggests that PGFS does not have a crucial role in endometrial  $PGF_{2\alpha}$  synthesis. They also reported that expression of COX-2 increased around luteolysis (day 15) and this induction was observed in pregnant mares, which indicated that horse conceptus blocks endometrial PGF<sub>2 $\alpha$ </sub> synthesis by down-regulating the expression of COX-2. Palliser et al. [45] demonstrated that the expression of PGFS increased in placentome after dexamethasone induced and spontaneous labor onset in sheep. Mondal et al. [41] reported that the expression of PGFSL-2 mRNA could not be detected in buffalo uterine endometrium at any stage of the estrous cycle. In vitro studies have shown that  $IFN\tau$  down-regulated PGFS mRNA expression in bovine endometrium, which suggested that conceptus interferes with  $PGF_{2\alpha}$  synthesis by targeting PGFS gene expression [46].

#### 23.3.3 Cyclooxygenase-2

PGs are generated via the COX pathway, and COX is the rate-limiting enzyme for the conversion of arachidonic acid into PGH2. COX exists in to two isoforms that are encoded by two separate genes, COX-1 and COX-2, which are also known as PGHS-1 and PGHS-2. To date, three isoforms of COXs have been identified. COX-1 is a glycoprotein of 71 kDa, which is constitutively expressed in different tissues. COX-1 is encoded by a gene on chromosome 9 and has a role in tissue homeostasis by modulating several cellular processes ranging from cell proliferation to angiogenesis or platelet aggregation owing to thromboxane production. Although COX-1—deficient female mice are fertile, they have specific defects in parturition, whereas COX-2—deficient female mice are infertile with abnormalities in ovulation, fertilization, implantation, and decidualization. COX-2 is the inducible isoform, which is regulated by growth factors and different cytokines such as

interleukin (IL)1 $\alpha$ , IL6, or TNF- $\alpha$  and overexpressed during inflammation. The COX-2 gene is located on chromosome 1 and its promoter displays a nuclear factor- $\kappa$ B response element as well as other cytokine-dependent (i.e., IL6) response elements. Finally, COX-3 has been identified as a splice variant of COX-1 and it is present mainly in the brain and spinal cord. The role of COX-3 is not known. The requirement of COX-2 for normal blastocyst implantation and decidualization in mice is due to the role of COX-2—derived PGs in the regulation of vascular endothelial growth factor and angiopoietin signaling that influence uterine vascular permeability and angiogenesis. Parent et al. [42] correlated the expression of PGE<sub>2</sub> in bovine endometrium. Increasing this production will modulate the PGE2–PGF<sub>2 $\alpha$ </sub> ratio and contribute to the establishment of pregnancy.

#### 23.3.4 Interferon-τ

IFNs are cytokines with antiviral, antiproliferative, and immunomodulatory biological effects critical to immune responses that protect the body against viral infections and malignant cells. In many species of ruminants, IFN $\tau$  is the pregnancy recognition signal expressed only by mononuclear trophectoderm cells of ruminant conceptuses (the embryo and its extraembryonic membranes) into the uterine lumen on days 13-21 of pregnancy. It is a subclass of the 172-amino acid type I omega (w) IFNs that compete with IFN $\alpha$  and - $\beta$  for binding of a common cell surface receptor. IFN $\iota$  of cattle and sheep arise from multiple mRNAs approximately 1 kb in length, which arise from multiple genes. Like other type I IFN genes, IFN<sup>T</sup> genes are intronless and consists of 585 bp ORF coding for a 195-amino acid preprotein containing a 23-signal sequence which after cleavage yields a mature protein of 172 amino acids. In the coding region, the nucleotide sequences of bovine, ovine, and caprine IFN $\tau$  exhibit approximately 90% identity and their inferred amino acid sequence about 80% identity [19]. The predicted amino acid identity between IFN $\tau$  and IFN $_{\alpha 1}$  and IFNw is 50% and 72%, respectively. IFN $\tau$  acts differently on the endometrial LE, GE, and stroma to regulate expression of various ISGs that have crucial roles in endometrial differentiation and implantation of conceptus. The mechanism of action of IFN $\tau$  includes inhibition of estradiol receptors, consequent reduction in oxytocin receptors, and activation of a COX inhibitor that abrogates the oxytocin-dependent release of luteolytic pulses of  $PGF_{2\alpha}$  by uterine epithelium; therefore the CL continues to produce P<sub>4</sub> required for pregnancy.

#### 23.3.5 Osteopontin

Osteopontin (OPN) is a highly phosphorylated acidic glycoprotein that stimulates cell–cell adhesion, increases cell–extracellular matrix communication, and promotes cell migration. It is a negatively-charged acidic hydrophilic protein and is secreted into all body fluids. It has an arginine–glycine–aspartic acid (RGD) cell binding sequence, a calcium binding site, and two heparin binding domains. In general, OPN is a monomer ranging in length from 264 to 301 amino acids that undergoes extensive posttranslational

modification, including phosphorylation, glycosylation, and cleavage, resulting in molecular mass variants ranging from 25 to 75 kDa. It contains a hydrophobic leader sequence characteristic of a secreted protein, a potential calcium phosphate aspartite binding region of consecutive asparagine residues, a cell attachment GRGDS sequence, a thrombin cleavage site, and two glutamines that are recognized substrates for transglutaminase-supported multimer formation. OPN is hypothesized to serve as a bifunctional bridging ligand that mediates the adhesion between LE and trophoblast essential for implantation and placentation [47]. It binds to integrin heterodimers ( $\alpha v \beta 1$ ,  $\alpha\nu\beta3$ ,  $\alpha\nu\beta5$ ,  $\alpha\nu\beta6$ ,  $\alpha\nu\beta8$ ,  $\alpha4\beta1$ ,  $\alpha5\beta1$ , and  $\alpha8\beta1$ ) via its RGD sequence, and to  $\alpha4\beta1$  and  $\alpha 9\beta 1$  by other sequences to promote cell adhesion, spreading, and migration. During the perimplantation period of pregnancy in sheep, OPN mRNA is expressed only by the endometrial glands, is first detected in some glands of some ewes by day 13, and is present in all glands by day 19 [48]. Multiple integrin receptors for OPN are present on trophoblasts and LE of humans and domestic animals, some of which increase during the periimplantation period [47]. Evidence suggests that secreted OPN binds integrin receptors expressed on conceptus trophoblast and endometrial LE, where it can stimulate changes in proliferation, migration, survival, adhesion, and remodeling of the conceptus as it elongates, opposes, and adheres to the LE. OPN protein is expressed at high levels in the uterine glands and on the apical surface of LE during the midsecretory phase, in decidualizing stroma, and on invading cytotrophoblast in humans [49,50].

#### 23.3.6 Galectin

Galectins belongs to a family of calcium-independent  $\alpha$ -galactoside-binding proteins, the lectin superfamily, and is involved in immunomodulation, cell adhesion, and defense against invading microorganisms. In sheep, the size of the LGALS-15 gene transcript is 243 bp, which encodes a protein of 81aa. Ovine and caprine LGALS-15 were highly homologous at the mRNA (95%) and protein (91%) levels, and all contained a conserved carbohydrate recognition domain and RGD recognition sequence for integrin binding. Several forms of galectins [galectin-1 (LGALS-1), galectin-3 (LGALS-3), galectin-5 (LGALS-5), galectin-9 (LGALS-9), and galectin-15 (LGALS-15)] were discovered. Galectin-1 possesses adhesive and antiadhesive roles and is an immunomodulator in maternofetal tolerance as well as embryo implantation. Galectin-3 (LGALS-3) is a galactose-specific lectin, the expression of which increases significantly during the secretory phase of the menstrual cycle [51] endometrium, with expression in glandular and luminal epithelial but not stromal or immune cells [52]. A new galectin family member, galectin-15, was discovered in the endometrium of sheep and has a prospective role in trophectoderm attachment [53]. In sheep, the galectin-15 (LGALS-15) gene is expressed specifically in the endometrial LE. Galectin-9 has been identified in mid- and late-secretory and decidual phases in human and sGE of the uterus in concert with blastocyst elongation during the periimplantation period; it is also expressed in the uterus of cattle, goats, and pigs. Endometrial LGALS-15 mRNA levels increased after day 11 of both the estrous cycle and pregnancy, and were considerably increased after day 15 of pregnancy in goats. In situ hybridization detected abundant LGALS-15 mRNA in endometrial LE and sGE of early pregnant goats [54]. The proposed extracellular role of galectin-15 in the uterine lumen is functionally to bind and cross-link  $\beta$ -galactosides on glycoproteins, such as mucins, integrins, fibronectin, laminin, and other glycoproteins and glycolipids, thereby allowing it to function as a heterophilic cell adhesion molecule bridging the blastocyst and the endometrial LE.

#### 23.3.7 Integrin

Integrins are another family of glycoconjugate-containing members expressed on the uterine luminal epithelium, and have spatial and temporal relationships with blastocyst development and implantation. They also have a major role in cellular differentiation, motility, and adhesion. They comprise a large family of transmembrane heterodimeric glycoprotein receptors that are intimately associated with the cytoskeleton and signaling proteins [55]. Integrin subunits  $\alpha$  (v, 4, and 5) and  $\beta$  [1,3,5] are constitutively expressed on the apical surfaces of both conceptus trophoblast and endometrial LE during the periimplantation period of pregnancy [48]. Integrin receptors, composed of a and â subunits, participate in bidirectional signaling involving both "outside-in" (i.e., ligand activation at the cell surface to modify cytoskeletal organization, intracellular signaling, and gene expression) as well as "inside-out" (i.e., cytoplasmic domain transduction of intracellular signals to regulate ligand binding affinity) pathways [56]. Luminal epithelial expression of integrin subunits  $\dot{a}_4$ ,  $\dot{a}_5$ , and  $\hat{a}_1$  increases during the period of MRP, and  $\hat{a}_4$ ,  $\hat{a}_5$ ,  $\hat{a}_{v}$ ,  $\hat{a}_1$ , and  $\hat{a}_3$  are localized to implantation sites [57]. These subunits potentially result in the integrin receptors  $\dot{a}_v \hat{a}_1$ ,  $\dot{a}_v \hat{a}_3$ ,  $\dot{a}_v \hat{a}_5$ ,  $\dot{a}_4 \hat{a}_1$ , and  $\dot{a}_5 \hat{a}_1$  at the maternofetal interface during pregnancy and may function as part of an adhesion cascade that serves to generate both stable adhesion between apposing epithelial surfaces and activation of outside-in signal transduction [58]. Altered expression of integrins is correlated with several causes of infertility, null mutations of several integrins leads to periimplantation lethality, and functional blockade of selected integrins reduces the number of implantation sites [59]. In the sheep, receptivity to implantation does not appear to involve changes in either temporal or spatial patterns of integrin expression, but may depend on expression of other glycoproteins and extracellular matrix (ECM) proteins, such as galectin-15, OPN, and fibronectin, which are ligands for heterodimers of these integrins [48]. In species such as pigs, mice, and humans, interactions between specific integrins and ECM proteins frame the putative window of implantation [60].

#### 23.3.8 Serpin

Uterine serpins (SERPINA14) belong to a super family of serine protease inhibitors secreted from the uterine endometrium under the influence of  $P_4$  in various livestock: bovine [61] ovine [62], and porcine [63]. Serpins are a group of proteins with similar

structures that were first identified as a set of proteins able to inhibit proteases. The acronym "serpin" was originally coined because many serpins inhibit chymotrypsinlike serine proteases (serine protease inhibitors) [64]. The SERPINA14 gene consists of five exons and four introns that encode a protein of 446 amino acids. This gene is present in chromosome 21 and 18 in bovines and ovines, respectively. In case of pigs this is known as serpin D (heparin cofactor) and is present on chromosome 14. The size of this is 2098 bp and the protein size is 495 aa. The SERPINA14 protein performs diverse biological functions including direct nutrition to the conceptus, growth control, inhibition of proteolytic activities, and suppression of local invasion of the mouse uterus by distinct trophoblast subtypes. The ability of SERPINA14 to bind with members of the pregnancy-associated glycoproteins [61] and aspartic proteinases [65] and to the growth factor activin suggests its role as a carrier protein during gestation. SERPINA14 has been further implicated as an important GE during days 13–15 of the estrus cycle as well as during days 15-50 of pregnancy in sheep [66] and on day 25 of pregnancy in goats [62]. In bovines, SERPINA14 is reported to be expressed predominantly in the uterine endometrium during pregnancy [67]. UTMP and SPP1 are excellent markers for differentiation and overall secretory capacity of uterine GE during pregnancy in ewes [68].

#### 23.3.9 Fibroblast Growth Factor-2

Fibroblast growth factor-2 (FGF-2) belongs to the member of the fibroblast growth factor (FGF) gene family, FGF-2 is a wide-spectrum mitogenic, angiogenic, and neurotrophic factor expressed at low levels in many tissues and cell types and reaches high concentrations in brain and pituitary. The FGF-2 gene consists of three exons and two introns and encodes a protein of 155 amino acid residues. This gene is located in chromosome 17 in bovines. FGF-2 has been implicated in a multitude of physiologic and pathologic processes, including limb development, angiogenesis, wound healing, and tumor growth [69]. In ruminants, conceptus development beyond the blastocyst state requires input from uterine-derived factors. FGF-2 is expressed by the bovine endometrium throughout the estrus cycle and early pregnancy and stimulates trophectoderm expression of IFN- $\tau$ , the MRP factor in ruminants. During MRP, the mononuclear cells of the conceptus trophectoderm synthesize and secrete IFN $\tau$  between days 10 and 21–25 with maximal production on days 14-16 [19]. This antiluteolytic effect of IFN<sub> $\tau$ </sub> results in the maintenance of a functional CL and hence secretion of P<sub>4</sub> that is essential to maintain a uterine environment that supports events critical to successful development of the conceptus to term. Okumu et al. [70] reported that the expression of FGFR1 and FGFR4 was higher during early embryo development (days 7-13) but decreased on day 16 whereas FGFR2 expression was similar from days 7 through 13, with a significant increase by day 16 that was maintained until day 19 in cows. However, no pregnancy-dependent changes in endometrial FGF-2 mRNA abundance were detected until placental attachment was well under way in sheep [71].

#### 23.3.10 Insulin-Like Growth Factor-1

Insulin-like growth factor-1 (IGF-1) has been referred to as a "sulfation factor" and its effects were termed "nonsuppressible insulin-like activity" in the 1970s. In bovines, IGF-1 genes are present on chromosome 5 and it has five exons. The IGF-1 gene is conserved in humans, chimpanzees, Rhesus monkeys, dogs, mice, rats, chickens, zebrafish, and frogs. IGF-1 is produced primarily by the liver as an endocrine hormone as well as in target tissues in a paracrine/autocrine fashion. Production is stimulated by growth hormone (GH) and can be retarded by undernutrition, GH insensitivity, lack of GH receptors, or failure of the downstream signaling pathway after GH receptor including SHP2 and STAT5B. IGF-1 is expressed by uterine glands of cyclic and pregnant pigs and IGF-1 receptors (IGF-1R) are expressed by cells of the endometrium and conceptuses, which suggest paracrine and autocrine actions of IGF-1. Its primary action is mediated by binding to its specific receptor, IGF-1R, which is present on many cell types in many tissues. Binding to IGF-1R, a receptor tyrosine kinase, initiates intracellular signaling; IGF-1 is one of the most potent natural activators of the AKT signaling pathway, a stimulator of cell growth and proliferation, and a potent inhibitor of programmed cell death. It is likely that estrogens produced by the conceptus are important in inducing pregnant-specific uterine proteins and in facilitating uterine contractions. Although uterine proteins are important for embryonic and placental development, their possible role in MRP is not known [72]. Both the embryo and the endometrium produce significant quantities of IGF-1, which is independent of estrogen secretion [73] and has a role in autocrine-paracrine maternal-conceptus signaling.

#### 23.3.11 Mx

Mx belongs to the class of dynamin-like large guanosine triphosphatases known to be involved in intracellular vesicle trafficking and organelle homeostasis. Mx is a proteincoding gene, and proteins encoded by genes are key components of the antiviral state induced by type I IFN in many species [74]. Mx expression in the endometrial epithelium is apparently regulated during the estrous cycle and the conceptus induces Mx expression in the uterine epithelium, stroma, and myometrium. Mx expression may be influenced by  $P_4$  in the cyclic endometrium and may have a role in preparing the uterus to support the developing conceptus. Expression of Mx mRNA is up-regulated during pregnancy in cows and gilts whose conceptuses secrete IFNs during early pregnancy [75]. In cyclic ewes, mRNA abundance in endometrium increased from day 1 to peak levels on day 13 and then declined to day 15. In pregnant ewes, steady-state levels of Mx mRNA were first detected above the level in cyclic ewes on day 13 after mating, were greater than 10-fold higher on day 15, and remained elevated on day 19 [76]. Johnson et al. [77] reported that Mx mRNA continued to be strongly expressed in LE and superficial GE through day 17 of pregnancy in sheep, whereas ISG17 remained low to undetectable in these cells. Both ISG17 and Mx mRNA increased in the stratum compactum stroma (ST) between days 11 and 13, and expression extended into the deep GE and stratum spongiosum ST on days 15-17 in pregnant ewes.

# 23.4 Causes of Early Embryo Loss

#### 23.4.1 Genetic Abnormalities

Most embryonic loss occurs during the first 18 days of gestation owing to genetic abnormalities affecting cleavage, MRP, attachment, or formation of the placenta [78]. Genetic causes of embryo death include chromosomal defects, individual genes, and genetic interactions [79]. Embryos produced with chromosomal defects are unable to continue development and are progressively lost during early developmental stages. Abnormal segregation during gametogenesis and cleavage of embryos and polyspermy are main reasons for the occurrence of an euploidy. Sometimes a lethal gene is expressed that causes death of the embryo within the first 5 days of pregnancy. King [80] reported that abnormal chromosome numbers had resulted in abnormal growth of the embryo and usually death within the first trimester of gestation. Cytogenetic studies revealed that gross chromosomal defects account for 3-5% of embryo losses in bovines [81]. Chromosome abnormalities can cause embryonic death owing to the presence of tetraploid cells in one of the eight blastocysts at 12–16 days of age. Deficiency of uridine monophosphate synthase [82], a homozygous recessive condition, had been found to cause fetal death at days 40-50 of gestation [83]. Yadav et al. found about 4.2% and 10.7% of causes of aneuploid and diploids, respectively, in cattle [84].

#### 23.4.2 Nutrition

Nutrition has an important role in the maintenance of pregnancy. Cows will have less embryo mortality if they are gaining condition, whereas those losing condition will tend to have higher losses [85]. Dunne et al. [86] showed that sudden reductions in dry matter intake (DMI) around the time of artifical insemination (AI) adversely affected embryo survival in the heifer dam. When energy intake was reduced from twice maintenance to 0.8 times maintenance for 2 weeks immediately after AI, the embryo survival rate in heifers was consistently less than 40%. When heifers were provided with either a constant level of feed intake or changed from a lower to a higher level feed intake, the embryo survival was 65–71%. Intake of high-protein diets by pregnant cows has been shown to alter pH and the concentration of other ions in uterine secretions only during the luteal phase and not at estrus. The intake of high dietary protein can result in elevated blood concentration of ammonia, urea, or both, depending on the balance of protein fractions present in the rumen and the availability of fermentable carbohydrates. Higher plasma urea concentrations had been observed to interfere with normal inductive actions of P<sub>4</sub> on the microenvironment of uterus and thereby caused suboptimal conditions for the support of embryo development [87].

After parturition, the nutrient demands of the dairy cow had been found to increase dramatically as peak lactation yield was approached and typically exceeded dietary intake, resulting in a state of negative energy balance (NEB). High-producing cows of high genetic merit experience greater NEB in early lactation under grazing conditions than cows of lower genetic merit, notwithstanding the somewhat higher DMI capacity of the former [88]. Meta-analysis of the combined data from Kennedy et al. [88] and Horan et al. [89] showed that there were no interaction effects on conception rate from service x study x supplementation rate, service x study or study x supplementation rate. The rate of supplementation had no effect on first-service conception rate (56% v. 53%), but cows on the low level of supplementation had a lower second-service conception rate (39% v. 58%) than cows on the high level of supplementation. Morris et al. [90] and Wathes et al. [91] studied gene expression in the uterine tissue and spleen of cows with severe postpartum NEB and reported that these cows had increased expression of many key genes known to be involved in inflammatory responses, which is consistent with a remodeling of the postpartum uterus and the clearance of microbial infections.

#### 23.4.3 Hormonal Imbalance

 $P_4$  is essential for maintaining a suitable endometrial environment for early embryo development, implantation, and the maintenance of pregnancy. During preimplantation embryo development,  $P_4$  increases embryo growth and induces the secretion of IFN $\tau$  by acting on the endometrium to stimulate the production of a variety of embryotrophic factors [92]. In addition, P<sub>4</sub> is necessary to maintain uterine quiescence that allows for proper implantation of the embryo within the endometrium and for fetal development [93]. In cows it is believed that intraluteal  $P_4$  is implicated in a survival pathway in the CL by stimulating PGs, OT, and its own production in a stage-specific style and inhibiting apoptosis. The role of P<sub>4</sub> in embryonic loss has been extensively explored, and it is now widely accepted to have a pivotal role in the establishment of pregnancy, embryo survival, and development. Stronge et al. [94] and McNeill et al. [95] found a positive relationship between the probability of embryo survival and milk P<sub>4</sub> concentrations on days 5–7 and 4–6, respectively. Similarly, Franco et al. [96] observed that cows undergoing pregnancy loss had lower milk  $P_4$  concentration. Starbuck et al. [97] found that pregnancy loss before day 45 was greater in cows with the lowest plasma concentration of P<sub>4</sub> on days 28–37 of gestation. However, optimal or threshold blood P<sub>4</sub> concentrations below which embryo mortality may occur are not known. It is possible that because P<sub>4</sub> is transferred locally from the ovary to the uterus, peripheral concentrations of  $P_4$  may be less relevant for embryo survival than local (uterine tissue and lumen) concentrations [98]. There are many potential mechanisms by which low P<sub>4</sub> concentrations may compromise embryo survival because it is essential to regulate the most critical steps of pregnancy establishment and embryo development. Low  $P_4$  may lead to asynchronous uterine environment, lower uterine secretions [99], failure to inhibit uterine immune response against the conceptus [100], excessive concentrations of other hormones that may affect either oocyte quality or embryonic development [101], and lower antiluteolytic signal [102]. Inskeep (2004) pointed out four major periods during embryo development in cows when are susceptible to lower concentrations of P<sub>4</sub>, which result in embryonic mortality: the early postovulatory period (when lower P4 exists during the

preceding luteal phase); days four through 9 after mating; the period of maternal recognition of pregnancy (days 14–17 with a slightly elevated  $E_2$  concentration) and the late embryonic period (days 28–42). In addition, it is well-documented that a delay in the postovulatory  $P_4$  rise is most important for embryo development rather than the concentrations achieved at later stages in cows and ewes [103]. Even though studies reported lower concentrations of  $P_4$  in plasma or milk during the luteal phase or the pregnancy of lactating dairy cows, it is unclear whether this is due to reduced production by CL or increased metabolism of secreted  $P_4$ . Supplementation with P4 during early pregnancy has given unequivocal results; most studies reported increased PR in lactating cows but no or reduced effects in heifers [104].

#### 23.4.4 Thermal Stress

Heat stress is an environmental factor that affects the survival of embryos and accounts for approximately 10% of embryonic loss. Whether an embryo can survive the early stages of pregnancy depends, at least in part, on the degree to which the embryo can adjust its own physiology [105]. Embryonic loss occurs when there is disruption in the physiologic regulation of oviductal and uterine function owing to environmental stresses. Conception rates had been found to decline from averages of 40-60% in cooler months to only 10-20% during times of high temperature and humidity [106]. Conception rates in beef heifers exposed for 72 h to an elevated temperature  $(32^{\circ}C)$  or ideal temperature (21°C) immediately after breeding were 0% and 48%, respectively [107]. Interestingly, we observed that higher  $P_4$  levels during hotter months (June and July) in buffalo may be responsible for the suppression of reproductive efficiency associated with high environmental temperatures [108]. Environmental stress has a lethal effect on embryos both in vivo and in vitro. The preimplantation embryo is most susceptible to certain types of stresses very early in its development because its genome is largely repressed. However the embryo possesses cellular adaptive mechanisms to limit adverse effects of stress, as occurs in the case of heat stress. Heat shock induces the disruption of embryonic development by influencing several key events leading to a successful pregnancy. First, follicular dynamics are altered during heat stress: Persistent preovulatory follicles [109], lower steroidogenic capacity [110], and a decline in oocyte competence [111] have been found in cows exposed to a high environmental temperature. Exposure to heat  $(39.5-41.0^{\circ}C)$  during the first 48 h of IVC of bovine zygotes significantly reduced the rate of development to the eight-cell stage at 72 h of IVC and the rate of development to the morula or blastocyst stage at 144 h of IVC [112]. Furthermore, the reduction in the percentage of embryos that developed into blastocysts was especially notable in the heat-treated embryos (41°C for 12 h) at the two-cell [113] and one-cell stages [114]. Under heat stress conditions proteins synthesis is reduced [115] and free radical metabolism is increased [116]. However, as development proceeds, the embryo acquires mechanisms to become resistant to elevated temperatures, which involves heat-shock protein-70 (HSP-70) expression [115], antioxidant systems

activation [116], and the ability to undergo apoptosis [117]. Failure of any of these mechanisms to develop may lead to embryo death. Roth et al. [118] observed that exposure of cows during the early stage of follicular development hampers the oocyte quality and subsequent embryo development. The effect of in vitro heat stress on synthesis and secretion of protein and PG by oocyte conceptus and endometrium has been reported by Putney et al. [119]. The elevation of temperature of tissue incubation from  $39^{\circ}$ C to  $43^{\circ}$ C reduces the synthesis and secretion of protein by the conceptus and stimulates the release of PGF<sub>2 $\alpha$ </sub> by pregnant endometrium in cattle. Whenever the embryo is exposed to heat shock, the cell fails to synthesize HSP-70, which ultimately protects the embryo from heat shock. Interestingly, it has been observed that GSH, which removes free radicals from cells, is reduced by heat shock [116].

#### 23.4.5 Infectious Agents

Infectious diseases cause reproductive problems for livestock producers worldwide. With the development of cost-effective vaccination and other methods to control infection and diseases, infectious agents are becoming less of a factor for embryo survival. It is therefore essential to understand the importance of maintaining these agents to a minimum by using vaccination and medication. Bovine viral diarrhea (BVD) is the most prevalent cause of abortion in cattle, with an incidence of infection often in excess of 70% [120]. There are two forms of BVD: cytopathogenic (CP) and noncytopathogenic (NCP). BVD-CP induces apoptotic cell death whereas NCP replicates in cultured cells without inducing cell death and has the ability to cross the placenta. Because NCP can cross the placenta, any vaccination must be able to protect the fetus. Infection with BVD-NCP in the first 40 days of gestation causes early embryonic death, infection between 40 and 125 days of gestation results in persistently infected offspring or later abortion, and infection after 125 days of gestation is nonlethal, because the fetus is able to mount an immune response [121]. Embryos infected with BVD-NCP have a reduced rate of in vitro blastocyst formation when recovered at day 8 postbreeding [122]. Infectious bovine rhinotracheitis causes sporadic abortions throughout gestation. Abortion is most common after the fourth month of gestation; however Miller et al. [123] inoculated heifers on day 14 postbreeding and within 10 days two heifers showed a decline in  $P_4$  to concentrations similar to those seen at estrus, whereas an additional two heifers lost embryos on days 40 and 42 after breeding. Inoculation at day 14 postbreeding resulted in early embryonic death, but heifers inoculated on day 21 or 28 postbreeding calved normally [124]. The presence of *Actinomyces* pyogenes in the reproductive tract is often considered to result from other infections such as endometritis [125]. Visual embryonic decomposition and pus were found after abortion from cows infected between days 27 and 41 of pregnancy compared with cows given PG to terminate the pregnancy. Cornebacterium pyogenes, the most common bacteria in cattle uterus, is responsible for abortion and endometritis [126]. Vibrio is also responsible for infertility and causes early embryonic mortality in cows [127]. Brucella abortus has more of an effect during mid to

late gestation, and causes abortion. Bovine herpes virus-1 causes early embryonic death, cystic CL, and pathologic changes in the uterus [128].

#### 23.4.6 Impairment of Maternal–Fetal Interactions

The uterine environment provides the medium in which the embryo develops. In cows, the embryo lies free in the medium for 30 days or more, during which time extensive embryonic development takes place before the conceptus becomes firmly attached to the embryonic wall. During this period the composition of the uterine fluid constantly changes and strict synchrony exists between the uterine environment and the developing conceptus. In the early stages of gestation, embryo development has to synchronize with changes in the uterine environment to prevent abnormalities and failure to implant. Natural asynchrony between the embryo and the mother can arise from late onset of the first meiotic division, which leads to a delay in oocyte maturation [129]. Also, delayed fertilization may result in asynchrony, but undoubtedly the main cause of asynchrony is an altered secretory pattern of  $P_4$  [99]. The major cause of preimplantation embryo mortality is thought to be difficulties in signaling between the conceptus and the mother [130]. Failure of the conceptus to communicate its presence at the appropriate time by secreting adequate concentrations of IFN $\tau$  leads to pregnancy loss. Inadequate IFN $\hat{o}$ production may result from intrinsic defects of the embryo (e.g., delayed or retarded trophoblast development) or adverse alterations in the maternal environment (e.g., low  $P_4$  concentration in the uterus) [105]. Thatcher et al. [131] hypothesized that some embryonic losses in cows could be caused by the inability of the embryo to suppress the luteolytic cascade, likely because of a failure of the conceptus to produce luteotropic signals, or perhaps failure of the CL to respond to luteotropins. Failure of the embryo to produce steroids may compromise its own survival. Steroids of conceptus origin could have a local effect on the endometrium to stimulate production of histotrophe or factor(s) necessary for the apposition and attachment phases of implantation [3].

Defects in endometrial adenogenesis or function during the periimplantation period may cause embryo mortality [132]. The unequivocal role of secretions of endometrial glands as primary regulators of conceptus survival and development, onset of pregnancy recognition signals, and implantation is well-established [132]. Uterine secretions are thought to be particularly important for conceptus survival and development in ruminants because of the prolonged period of preimplantation conceptus development that precedes superficial attachment and placentation [133]. Failure of endometrial gland development or the servomechanism supposed to regulate endometrial function in early stages of pregnancy can lead to embryo mortality. Uterine pH is also thought to have an important role in determining whether the embryo might survive and develop. There is abundance of evidence that low pH in the uterine luminal fluid has detrimental effects on embryo survival and reproduction in general [134]. It has been suggested that the decrease in uterine pH caused by an excess of protein supplementation may influence endometrial secretory activity, creating a hostile or suboptimal environment for embryo development and thus compromising the establishment of pregnancy.

#### 23.4.7 Quality of Oocytes and Sperm

An embryo's fate is also dictated by events before fertilization: Factors affecting the health of the gametes ultimately may determine the ability of an embryo to survive and grow. Indeed, embryos formed from incompetent oocytes have a lower probability of successful development than do embryos developing from normal oocytes. Oocyte competence can be compromised by alterations in follicular dynamics. Oocytes belonging to persistent dominant follicles exhibit various morphologic abnormalities [135]; after fertilization they give rise to zygotes whose development is retarded and early embryonic death usually occurs before the 16-cell stage [136]. Cows with persistent preovulatory follicles show low fertility and have altered protein secretory patterns in the oviduct [137]. These events may be because of the larger size and the longer lifespan of a persistent follicle. Oocytes belonging to persistent follicles are at a too-advanced stage of maturation at the time of fertilization, which increases the probability of having poorquality zygotes [135]. Persistent follicles also produce greater quantities of estrogens than do normal follicles, with detrimental effects on embryo survival [138]. This abnormal follicular dynamic could result from the failure of normal luteal function that affects P<sub>4</sub> production and its regulation of LH secretion. The sequential relationship of low  $P_4$ , increased frequency of LH pulses, a persistent dominant follicle, and increased secretion of estradiol-17 $\beta$  is widely accepted as one a cause of early embryonic death [101]. Inadequate follicular development also reduced the ability of fertilized oocytes to develop to the blastocyst stage. Perry et al. [139] observed low pregnancy rates and high embryo losses in cows induced to ovulate prematurely, likely because the  $P_4$  concentration rose at a slower rate than in cows induced to ovulate later. This could have led to poorly developed embryos although, like the oocyte, the male gamete influences fertility by imparting characteristics to the embryo that influence its ability to proceed through development [105]. The main male-derived causes of embryo loss are chromosomal abnormalities, such as nondisjunction in spermatogenesis [140], 1/29 Robertsonian translocation [141], and a defective chromatin structure [142].

# 23.5 Strategies for Reducing Early Embryonic Wastage

The successful establishment of pregnancy depends on a delicate balance between luteolytic mechanisms inherent in the endometrium at the end of diestrus and antiluteolytic mechanisms orchestrated by the conceptus to change endometrial function and ultimately, block luteolysis. Several strategies, such as pharmacologic, nutritional, and management manipulations of the process of MRP, have endeavored to increase the probability of successful gestation in livestock.

#### 23.5.1 Development of Timed Insemination Programs

#### 23.5.1.1 Ovsynch

One program that has been extremely successful for inseminating cows at a fixed time for first service without the need to detect estrus is the Ovsynch program, in which injections of gonadotropin-releasing hormone (GnRH) are given 7 days before and 48 h after an injection of PGF<sub>2a</sub> and cows are inseminated 12-16 h after the second injection of GnRH. This system synchronizes follicle maturation without regression of the CL before GnRH-induced ovulation and timed insemination. In 1995, Pursley et al. [143] developed the hormonal program Ovsynch to synchronize ovulation in lactating dairy cows. Cows treated with Ovsynch yielded overall conception rates similar to those obtained after breeding to detected estrus (37% versus 39%, respectively) [144]. However a major limitation to Ovsynch is the wide variability in synchronization rates (defined as having a regressed CL and ovulation after the final GnRH) in Ovsynch-treated cows. Variations among dairy cows in synchronization rates to Ovsynch were primarily attributed to the stage of estrous cycle in which Ovsynch was initiated [145]. Cows started on Ovsynch at midcycle (days 5-9 of the cycle) had a greater probability of synchronizing to Ovsynch and therefore had a greater chance of conception than did cows at any other stage of the estrous cycle [146]. The key physiologic reasons for increased synchronization rate in midcycle are: (1) the presence of a functional dominant follicle capable of ovulating to first GnRH of Ovsynch; and (2) the presence of a CL that remained functional during the 7 days between first GnRH and  $PGF_{2\alpha}$  of Ovsynch. In contrast, during the early (days 1-4) or late (days >10) estrous cycle, the physiologic follicular and luteal scenarios were not conducive to synchronization to Ovsynch. When cows were started on Ovsynch early in the estrous cycle (days 1-4), ovulation to first GnRH of Ovsynch was impaired by the presence of an emerging follicular wave [145,146]. Moreover, when cows were started on Ovsynch in the late estrous cycle, spontaneous luteolysis was likely to occur before  $PGF_{2\alpha}$  of Ovsynch. In cows with spontaneous luteolysis before  $PGF_{2\alpha}$ , the dominant follicle was likely to trigger spontaneous ovulation before the final GnRH of Ovsynch. Successful synchronization of ovulation is less likely when cows are started on Ovsynch during the early or late estrous cycle compared with midcycle. Because insemination occurs at a fixed time after the final GnRH of Ovsynch, chances of conception to timed AI are lower in nonsynchronized cows. In buffalo [147,148], this protocol was successful in synchronizing ovulation in 70–90% of buffalo, with conception rates ranging from 33% to 60%.

#### 23.5.1.2 Presynch–Ovsynch

This program consists of two injections of  $PGF_{2\alpha}$  administered at an interval of 14 days with the second  $PGF_{2\alpha}$  administered 12–14 days before an OvSynch program. This program increased pregnancy rates 18% units (i.e., 25–43%) in lactating cyclic cows [146]. This stimulation in pregnancy rates was attributed to manipulation of the estrous cycle such that the OvSynch timed insemination program was initiated at the most favorable stages of the estrous cycle. Several presynchronization strategies were developed based on the idea that controlling ovarian function before the initiation of Ovsynch could improve synchronization rates to Ovsynch [149,150]. The most popular presynchronization strategies were based on the use of  $PGF_{2\alpha}$  at specific times before the initiation of Ovsynch. A presynchronization protocol before implementation of the Ovsynch protocol was developed by giving two injections of  $PGF_{2\alpha}$  14 days apart, with the second injection given 12 days before the first GnRH of the Ovsynch protocol. El-Zarkouny et al. [149] demonstrated that presynchronization before the GnRH-based protocol with a controlled internal drug release insert enhanced pregnancy rate compared with that achieved without presynchronization (46.8% versus 37.5%; P < .01), but this effect was not evident in anestrous cows. The success of the OvSynch program depends on whether lactating dairy cows are anestrus or cycling. Pregnancy rates were lower in cows that were not cycling at the time the OvSynch program was initiated (22% versus 42%). If anestrus cows ovulated to the first and second GnRH injections of OvSynch program, pregnancy rates appeared to be normal (39%).

#### 23.5.1.3 Heat Synch

Heat synch is an estrus synchronization protocol that substitutes an injection of estradiol cypionate (ECP) at 24 after the  $PGF_{2\alpha}$  for the last GnRH in the OvSynch protocol. Based on the synchronization of ovulation and pregnancy rates, ECP is used as an alternative to inducing ovulation in place of GnRH for timed insemination. Because lactating dairy cows have reduced concentrations of plasma estradiol in the preovulatory period and reduced intensity of estrus, the elevation of estradiol after ECP injection is a supplement for lactational-induced deficiency. If cows are anovulatory (they are anestrus or have not developed positive estradiol feedback), the Heatsynch program may not be as effective as the GnRH-based OvSynch program. This is because GnRH causes the direct secretion of LH.

#### 23.5.2 Nutritional Strategies

Energy status at service affects embryo survival. Sudden reductions in dry matter intake around the time of insemination adversely affect embryo survival in heifers. When energy intake was reduced from a high level twice their maintenance requirement to 0.8 times maintenance for 2 weeks immediately after AI, embryo survival rate in heifers was consistently less than 40%. When heifers were provided with either a constant level of feed intake or were changed from a low to a higher-level feed intake, embryo survival was consistently high at 65–71%. High dietary protein, resulting in high concentrations of urea in plasma and milk, has been associated with decreased fertility in dairy cattle [151]. By-products of nitrogen metabolism, such as ammonium ions, can have detrimental effects on the uterine environment, reducing sperm, ova, or embryo survival. Jordan and Swanson [152] reported that the increased supplementation of crude protein from 12.7% to 19.3% at 4 days postpartum and continuing for 91 days improved the number of service per conception from 1.47 to 2.47. Supplementation of 0.8 kg fishmeal per day substituted for 0.8 kg standard concentrate improved conception rate to all services (0.64 versus 0.44) and reduced the number the number of services per conception (1.62 versus 2.31) [153]. Dietary supplementation of omega-3 fatty acids [eicosapentaenoic acid (EPA), docosahexaenoic acid (DHA), and linoleic acid] increases the fertility rate and

reduces embryonic mortality by inhibiting uterine secretion of  $PGF_{2\alpha}$  [154]. These fatty acids are capable of decreasing the secretion of  $PGF_{2\alpha}$  and complement the antiluteolytic action of IFN $\tau$ . EPA and DHA are known to have distinct antiinflammatory and immunosuppressive effects that complement the normal immunosuppressive and antiinflammatory effects of P<sub>4</sub> and IFN $\tau$  in early pregnancy.

#### 23.5.3 Supplementation of Luteotrophic Agents

One approach to improving embryonic survival in cattle has been direct supplementation with  $P_4$ . However, direct  $P_4$  supplementation during the early or midluteal phases of the estrous cycle has not consistently increased pregnancy rates. P<sub>4</sub> treatment seemed to increase fertility in herds with low fertility. However, Macmillan et al. [155] reported that controlled internal drug release-containing P<sub>4</sub> was inserted for 6 or 12 days starting on days 5–7 and a significantly higher pregnancy rate (79%) was achieved than in untreated cattle, even though the fertility of the untreated cattle was high (66%). Treatment with human chorionic gonadotropin (hCG) may slightly improve fertility in some low-fertility herds, such as in heat-stressed cows in which conceptuses are slightly delayed in development [156], and also in embryo transfer recipients [157]. Treatment with GnRH at the time of insemination had a more consistent effect in increasing pregnancy rates in repeat-breeders than in first-service cattle [158]. An alternative strategy to increasing endogenous  $P_4$  concentration is to induce an accessory CL by ovulating the dominant follicle on day 5 of the estrous cycle with hCG [159]. In addition to increasing  $P_4$  concentrations by day 9, hCG injection resulted in all treated heifers having three follicular waves (the dominant follicle of the third wave did not reach 9-10 mm in size until approximately day 20, versus day 14 in the control heifers). Thus hCG treatment would reduce the estrogenic environment during the period of maternal recognition of pregnancy. Similarly, injecting hCG on day 5 after artificial insemination induced the formation of an accessory CL in 86.2% of cows (compared with 23.2% of control cows that had multiple ovulations), increased  $P_4$  concentrations and pregnancy rates [160].

Species	Dose and Route of IFN Used	Pregnancy Rate (%)	
		Treated	Control
Sheep	<b>1.</b> 2 mg of r bolFN- $\alpha$ i/m ly twice a day on days 12–14 after mating	78.3	74.8
	<b>2.</b> 2 mg of r bolFN- $\alpha$ i/m ly twice a day on days 12–15	81.6	76.3
	<ol> <li>2 mg of r bolFN-αi/m ly twice a day on day 12–16 Martinod et al. [161]</li> </ol>	80.5	73.2
	<b>1.</b> 2 mg of r bolFN- $\alpha_1$ 1 i/m ly twice a day on days 17–18	79.0	58.0
	<b>2.</b> 2 mg of r bolFN- $\alpha_1$ 1 i/m ly twice a day on days 11–18 SchalueFrancis et al. [162]	89.0	78.0
Cattle	<b>1.</b> 20 mg of bolFN- $\alpha$ i/m ly daily on days 14–17 after estrus	48.0	60.0
	<b>2.</b> A single injection of 40 mg of bolFN- $\alpha$ on day 13 after estrus	38.0	48.0
	<b>3.</b> An increasing dose of 0.01–10 mg, i/m ly, twice a day on days 11–19 Barros et al. [163]	44.0	56.0

#### 23.5.4 Supplementation With Interferon (IFN)

*r bolFN-a* ( $\alpha$ ), recombinant bovine interferon-( $\alpha$ )

#### 23.5.5 Use of Bovine Somatotropin

Administration of bovine somatotropin (bST) (500 mg Posilac, USA) at the time of the first GnRH injection or during the insemination in cows in a Presynch-OvSynch program increased pregnancy rates (57% compared with 42.6%; 148). Because bST was effective at insemination, it is likely that bST stimulated embryonic development and survival after insemination in lactating dairy cows. bST treatment increased the proportion of transferable embryos, the number of blastocysts present per flushing, and the pregnancy rates of recipients treated with bST or receiving embryos from donors treated with bST. In vitro administration of recombinant bST increased fertilization rates, hastened embryo development, and increased embryo quality [164], which indicated that both maternal and conceptus components were affected by bST treatment.

# 23.6 Conclusion

MRP is an incredibly useful biological system, a better understanding of which will advance our knowledge in several basic physiologic processes. Unraveling the secrets of MRP will assist biologists in developing a paradigm to reduce early embryonic loss. MRP is initiated at various times in the periimplantation period by the embryologic agent, which is not well-characterized. Until we understand the complex interaction between the conceptus and its maternal environment, it will not be possible to achieve a major reduction in embryonic growth retardation and mortality. Current biotechnological approaches will be of tremendous importance in understanding the complex mechanism of low reproductive efficiency in various livestock species to evolve safer and potent gene-based strategies to enhance production. Promising technologies include the use of adenoviruses, antisense oligodeoxynucleotides, morpholinos, and small inhibitory RNAs that could be used in vivo and ex vivo to perform gain-offunction and loss-of function studies of specific gene(s) in the endometrial epithelia and trophoblast. The sequencing of the genomes of domestic animals is expected to generate knowledge and reagents useful for understanding the basis of enhanced fertility as well as infertility. Understanding the signals that regulate uterine receptivity and implantation can be used to diagnose and identify the cause(s) of recurrent pregnancy loss and improve pregnancy rates in domestic animals and humans. Use of genomic technologies in animal agriculture will definitely bring about exciting changes in livestock production and the tailoring of animals to produce products needed by humanity. The challenge is to understand the complexity of key mechanisms that are characteristic of successful pregnancy in animals and to use that knowledge to enhance fertility and reproductive health of ruminant species in the livestock industry.

# Acknowledgments

We are thankful to the National Agricultural Science Fund (formerly the National Fund for Basic, Strategic, and Frontier Application Research in Agriculture), Ministry of Agriculture, ICAR, New Delhi for providing financial support to carry out the project work on Impact of Heat Stress and Nutritional Stress on Maternal Recognition of Pregnancy. The authors express their gratitude to Dr. A. Bandyopadhyay, Ex. National Coordinator and Dr. P. K. Agarwal, ADG (NASF) for their help and suggestions. We thank the Director, NIANP, for providing the necessary facilities for conducting the research work. The help rendered by A. Jagannath is duly acknowledged.

# References

- [1] M.R. Jainudeen, E.S.E. Hafez, Reproduction in Farm Animals, seventh ed., Lippinott Williams and Wilkins, USA, 2000.
- [2] J.M. Sreenan, M.G. Diskin, The extent and timing of embryonic mortality in cattle, in: J.M. Sreenan, M.G. Diskin (Eds.), Embryonic Mortality in Farm Animals, CEC, Martinus Nijhoff, 1986, pp. 142–158.
- [3] M.T. Zavy, Embryonic mortality in cattle, in: M.T. Zavy, R.D. Geisert (Eds.), Embryonic Mortality in Domestic Species, CRC Press, Boca Raton, 1994, pp. 99–140.
- [4] P. Humblot, Use of pregnancy specific proteins and progesterone assays to monitor pregnancy and determine the timing, frequencies and sources of embryonic mortality in ruminants, Theriogenology 56 (2001) 1417–1433.
- [5] T.E. Spencer, O. Sandra, E. Wolf, Genes involved in conceptus-endometrial interactions in ruminants: insights from reductionism and thoughts on holistic approaches, Reproduction 135 (2008) 165–179.
- [6] R.M. Roberts, Y. Chen, T. Ezashi, A.M. Walker, Interferons and the maternal-conceptus dialog in mammals, Seminars in Cell and Developmental Biology 19 (2008) 170–177.
- [7] R.M. Roberts, T. Schalue-Francis, Maternal recognition of pregnancy and embryonic loss, Theriogenology 33 (1990) 175–183.
- [8] F.W. Bazer, Mediators of maternal recognition of pregnancy in mammals, Proceedings of the Society for Experimental Biology and Medicine 199 (1992) 373–384.
- [9] S. Mondal, B.S. Prakash, Comparison of luteal function between cows and buffaloes during estrous cycle, Indian Journal of Dairy Science 55 (2002) 142–144.
- [10] T.E. Spencer, F.W. Bazer, Biology of progesterone action during pregnancy recognition and maintenance of pregnancy, Frontiers in Bioscience 7 (2002) 1879–1898.
- [11] T.E. Spencer, T.L. Ott, F.W. Bazer, Tau-Interferon: pregnancy recognition signal in ruminants, Proceedings of the Society for Experimental Biology and Medicine 213 (1996) 215–229.
- [12] D.C. Wathes, M. Hamon, Localization of oestradiol, progesterone and oxytocin receptors in the uterus during the oestrous cycle and early pregnancy of the ewe, Journal of Endocrinology 138 (1993) 479–492.
- [13] T.E. Spencer, F.W. Bazer, Temporal and spatial alterations in uterine estrogen receptor and progesterone receptor gene expression during the estrous cycle and early pregnancy in the ewe, Biology of Reproduction 53 (1995) 1527–1543.
- [14] J. McCracken, W. Schramm, W.C. Okulicz, Hormone receptor control of pulsatile secretion of PGF-2Alpha from the ovine uterus during luteolysis and its abrogation in early pregnancy, Animal Reproduction Science 7 (1984) 31–55.

- [15] A.P.F. Flint, E.L. Sheldrick, Ovarian oxytocin and maternal recognition of pregnancy, Journal of Reproduction and Fertility 76 (1986) 831–839.
- [16] H.H.D. Meyer, T. Mittermeier, D. Schams, Dynamics of oxytocin, estrogen and progesterone receptors in the bovine endometrium during estrous cycle, Acta Endocrinologica 118 (1986) 96–104.
- [17] S. Mondal, V. Kumar, I.J. Reddy, K. Singh, Endometrial cytosolic progesterone receptor concentrations in relation to phases of estrous cycle, in: Proceedings of 89th Session of Indian Science Congress Association Held at Lucknow, 3–7 January, 2002, pp. 67–68.
- [18] I.J. Reddy, V. Kumar, C.G. David, J. Ghosh, S. Mondal, Endometrial estradiol 17β receptor concentration in different stages of estrous cycle in buffaloes, in: Proc. of National Symposium on Recent Advances in Cryopreservation of Livestock Germplasm Held at Indira Gandhi Agricultural University from 28–29 January, 2005, pp. 162–163.
- [19] R.M. Roberts, A.D. Ealy, A.P. Alexenko, C.S. Han, T. Ezashi, Trophoblast interferons, Placenta 20 (1999) 259–264.
- [20] F.W. Bazer, Establishment of pregnancy in sheep and pigs, Reproduction Fertility and Development 1 (1989) 237–242.
- [21] M. Binelli, A. Guzeloglu, L. Badinga, D.R. Arnold, J. Sirois, T.R. Hansen, W.W. Thatcher, Interferon-T modulates phorbol ester-induced production of prostaglandin and expression of cyclooxygenase-2 and phospholipase-A2 from bovine endometrial cells, Biology of Reproduction 63 (2000) 417–424.
- [22] R.D. Geisert, M.T. Zavy, R.J. Moffatt, R.M. Blair, T. Yellin, Embryonic steroids and the establishment of pregnancy in pigs, Journal of Reproduction and Fertility Supplement 40 (1990) 293–305.
- [23] T. Krzymowski, J. Kotwica, S. Stefanczyk-Krzymowska, Uterine and ovarian countercurrent pathways in the control of ovarian function in the pig, Journal of Reproduction and Fertility Supplement 40 (1990) 179–191.
- [24] B.A. Akinlosotu, J.R. Diehl, T. Gimenez, Prostaglandin E2 counteracts the effects of PGF2 alpha in indomethacin treated cycling gilts, Prostaglandins 35 (1988) 81–93.
- [25] L.K. Christenson, D.B. Farley, L.H. Anderson, S.P. Ford, Luteal maintenance during early pregnancy in the pig: role for prostaglandin E2, Prostaglandins 47 (1994) 61–75.
- [26] A.J. Ziecik, M. Jedlinska, S.J. Rzucidlo, Effect of estradiol and progesterone on myometrial LH/hCG receptors in pigs, Acta Endocrinologica (Copenh) 127 (1992) 185–188.
- [27] K.H. Young, R.R. Kraeling, F.W. Bazer, Effect of pregnancy and exogenous ovarian steroids on endometrial prolactin receptor ontogeny and uterine secretory response in pigs, Biology of Reproduction 43 (1990) 592–599.
- [28] A. Cencic, M. Guillomot, S. Koren, C. La Bonnardiere, Trophoblastic interferons: do they modulate uterine cellular markers at the time of conceptus attachment in the pig? Placenta 24 (2003) 862–869.
- [29] S. Kim, Y. Choi, F.W. Bazer, T.E. Spencer, Identification of genes in the ovine endometrium regulated by interferon tau independent of signal transducer and activator of transcription 1, Endocrinology 144 (2003) 5203–5214.
- [30] R.R. Magness, J.M. Huie, G.L. Hoyer, T.P. Huecksteadt, L.P. Reynolds, G.J. Seperich, G. Whysong, C. W. Weems, Effect of chronic ipsilateral or contralateral intrauterine infusion of prostaglandin E<sub>2</sub> (PGE<sub>2</sub>) on luteal function of unilaterally overiectomized ewes, Prostagladin Mediators 6 (1981) 389–401.
- [31] P. Jakobsson, S. Thoren, R. Morgenstern, B. Samuelsson, Identification of human prostaglandin E synthase: a microsomal glutathione dependent, inducible enzyme constituting a potential novel drug target, The Proceedings of the National Academy of Sciences, USA 96 (1999) 7220–7225.

- [32] T. Tanioka, Y. Nakatani, N. Semmyo, M. Murakami, I. Kudo, Molecular identification of cytosolic prostaglandin E synthase that is functionally coupled with cyclooxygenase–1 in immediate prostaglandin E biosynthesis, Journal of Biological Chemistry 275 (2000) 32775–32782.
- [33] J. Parent, M.A. Fortier, Expression and contribution of three different isoforms of prostaglandin E synthase in the bovine endometrium, Biology of Reproduction 73 (2005) 36–44.
- [34] L. Forsberg, L. Leeb, S. Thoren, R. Morgenstern, P. Jacobsson, Human glutathione dependent prostaglandin E synthase: gene structure and regulation, FEBS Letter 471 (2000) 78–82.
- [35] F. Filion, N. Bouchard, A.K. Goff, J.G. Lussier, J. Sirois, Molecular cloning and induction of bovine prostaglandin E synthase by gonadotropins in ovarian follicles prior to ovulation in vivo, Journal of Biological Chemistry 276 (2001) 34323–34330.
- [36] J.A. Arosh, J. Parent, P. Chapdelaine, J. Sirois, M.A. Fortier, Expression of cyclooxygenases 1 and 2 and prostaglandin E synthase in bovine endometrial tissue during the estrous cycle, Biology of Reproduction 67 (2002) 161–169.
- [37] D. Boerboom, K.A. Brown, D. Vaillancourt, P. Poitras, A.K. Goff, K. Watanabe, M. Dore, J. Sirois, Expression of key prostaglandin synthases in equine endometrium during late diestrus and early pregnancy, Biology of Reproduction 70 (2004) 391–399.
- [38] D.K. Vanderwall, G.L. Woods, J.A. Weber, A.B. Lichtenwalner, Corpus luteum function in nonpregnant mares following intrauterine administration of prostaglandin E<sub>2</sub> or estradiol 17B, Theriogenology 42 (1994) 1069–1083.
- [39] A. Waclawik, A. Rivero-Muller, A. Blitek, M.M. Kaczmarek, L.J.S. Brokken, K. Watanabe, N.A. Rahman, A. Ziecik, Molecular cloning and spatiotemporal expression of prostaglandin F synthase and microsomal prostaglandin E synthase-1 in porcine endometrium, Endocrinology 147 (2006) 210–221.
- [40] S.A. Milne, G.B. Perchick, S.C. Boddy, H.N. Jabbour, Expression, localization and signaling of PGE2 and EP2/EP4 receptors in human nonpregnant endometrium across the menstrual cycle, Journal of Clinical Endocrinology and Metabolism 86 (2001) 4453–4459.
- [41] S. Mondal, V.P. Varshney, A. Mitra, Prostaglandin E and F synthases: genomic insight, in: Proceedings of National Symposium on Animal Biotechnology Held at NIANP, Bangalore, on 16th November, 2007, 2007, pp. 34–40.
- [42] J. Parent, P. Chapdelaine, J. Sirois, M.A. Fortier, Expression of microsomal prostaglandin E synthase in bovine endometrium: coexpression with cyclooxygenase type 2 and regulation by interferon-tau, Endocrinology 143 (2002) 2936–2943.
- [43] E. Madore, N. Harvey, J. Parent, P. Chapdelaine, J.A. Arosh, M.A. Fortier, An aldose reductase with  $20 \propto$  hydroxysteroid dehydrogenase activity is most likely the enzyme responsible for the production of prostaglandin  $F_{2\alpha}$  in the bovine endometrium, Journal of Biological Chemistry 278 (2003) 11205–11212.
- [44] W. Kuchinke, O. Barski, K. Watanabe, O. Hayaishi, A lung type prostaglandin F synthase is expressed in bovine liver: cDNA sequence and expression in *E.coli*, Biochemical and Biophysical Research Communications 183 (1992) 1238–1246.
- [45] H.K. Palliser, G.T. Ooi, J.J. Hirst, G. Rice, N.L. Dellions, R.M. Escalona, J.R. Young, Changes in the expression of prostaglandin E and F synthases at induced and spontaneous labour onset in the sheep, Journal of Endocrinology 180 (2004) 469–477.
- [46] C.W. Xiao, J.M. Liu, J. Sirois, A.K. Goff, Regulation of cyclooxygenase-2 and prostaglandin F synthase gene expression by steroid hormone and interferon tau in bovine endometrial cells, Endocrinology 139 (1998) 2293–2299.
- [47] G.A. Johnson, T.E. Spencer, R.C. Burghardt, F.W. Bazer, Ovine osteopontin: I. Cloning and expression of mRNA in the uterus during the peri-implantation period, Biology of Reproduction 61 (1999) 884–891.

- [48] G.A. Johnson, F.W. Bazer, L.A. Jaeger, H. Ka, J.E. Garlow, C. Pfarrer, T.E. Spencer, R.C. Burghardt, Muc-1, integrin, and osteopontin expression during the implantation cascade in sheep, Biology of Reproduction 65 (2001) 820–828.
- [49] M.F. Young, J.M. Kerr, J.D. Termine, U.M. sssWeever, M.G. Wang, O.W. McBride, L.W. Fisher, cDNA cloning, mRNA distribution and heterogeneity, chromosomal location, and RFLP analysis of human osteopontin (OPN), Genomics 7 (1990) 491–502.
- [50] K.B.C. Apparao, A.F. Chambers, B.A. Lessey, Progesterone receptor isoform-B modulates the expression of osteopontin in human endometrial epithelium during the window of implantation, in: Program of the Annual Meeting of the Society for Gynecologic Investigation; Toronto, Canada, 2001. Abstract 43.
- [51] M. Von Wolff, X. Wang, H.J. Gabius, T. Strowitzki, Galectin fingerprinting in human endometrium and decidua during the menstrual cycle and in early gestation, Molecular Human Reproduction 11 (2005) 189–194.
- [52] D.M. Smalley, K. Ley, L-selectin: mechanisms and physiological significance of ectodomain cleavage, Journal of Cellular and Molecular Medicine 9 (2005) 255–266.
- [53] J.L. Farmer, R.C. Burghardt, F.D. Jousan, P.J. Hansen, F.W. Bazer, T.E. Spencer, Galectin 15 (LGALS15) functions in trophectoderm migration and attachment, FASEB Journal 22 (2008) 548.
- [54] S.K. Lewis, J.L. Farmer, R.C. Burghardt, G.R. Newton, G.A. Johnson, D.L. Adelson, F.W. Bazer, T.E. Spencer, Galectin 15 (LGALS15): a gene uniquely expressed in the uteri of sheep and goats that functions in trophoblast attachment, Biology of Reproduction 77 (2007) 1027–1036.
- [55] F.G. Giancotti, E. Ruoslahti, Integrin signaling, Science 285 (1999) 1028-1032.
- [56] R.O. Hynes, Integrins: a family of cell surface receptors, Cell 48 (1987) 549-554.
- [57] G.J. King, B.A. Atkinson, H.A. Robertson, Implantation and early placentation in domestic ungulates, Journal of Reproduction and Fertility 82 (1982) 87–95.
- [58] R.C. Burghardt, J.A. Bowen, G.R. Newton, F.W. Bazer, Extracellular matrix and the implantation cascade in pigs, Journal of Reproduction and Fertility 52 (1997) 151–164.
- [59] M.J. Illera, E. Cullinan, Y. Gui, L. Yuan, S.A. Beyler, B.A. Lessey, Blockade of the alpha (v) beta (3) integrin adversely affects implantation in the mouse, Biology of Reproduction 62 (2000) 1285–1290.
- [60] D.D. Carson, I. Bagchi, S.K. Dey, A.C. Enders, A.T. Fazleabas, B.A. Lessey, K. Yoshinaga, Embryo implantation, Developmental Biology 223 (2000) 217–237.
- [61] N. Mathialagan, T.R. Hansen, Pepsin-inhibitory activity of the uterine serpins, The Proceedings of the National Academy of Sciences USA 93 (1996) 13653–13658.
- [62] S. Tekin, M.B. Padua, G.R. Newton, P.J. Hansen, Identification and cloning of caprine uterine serpin, Journal of Reproduction and Development 70 (2005) 262–270.
- [63] P.V. Malathy, K. Imakawa, R.C. Simmen, R.M. Roberts, Molecular cloning of the uteroferrin associated protein, a major progesterone-induced serpin secreted by the porcine uterus, and the expression of its mRNA during pregnancy, Journal of Molecular Endocrinology 4 (1990) 428–440.
- [64] J.C. Whisstock, G.A. Silverman, P.I. Bird, S.P. Bottomley, D. Kaiserman, C.J. Luke, S.C. Pak, J.M. Reichhart, J.A. Huntington, Serpins flex their muscle: II. Structural insights into target peptidase recognition, polymerization, and transport functions, Journal of Biological Chemistry 285 (2010) 24307–24312.
- [65] E.U. Susanne, F. Thomas, S. Katy, E. Eva, W. Nadine, J.A. Georg, R. Horst-Diete, R. Myriam, W. Eckhard, H.D.M. Heinrich, B. Stefan, Evidence for estrogen dependent uterine serpin (SERPINA14) expression during estrus in the bovine endometrial glandular epithelium and lumen, Biology of Reproduction 81 (2009) 795–805.

- [66] M.D. Stewart, G.A. Johnson, C.A. Gray, R.C. Burghardt, L.A. Schuler, M.M. Joyce, F.W. Bazer, T.E. Spencer, Prolactin receptor and uterine milk protein expression in the ovine endometrium during the estrous cycle and pregnancy, Biology of Reproduction 62 (2000) 1779–1789.
- [67] H. Khatib, V. Schutzkus, Y.M. Chang, G.J. Rosa, Pattern of expression of the uterine milk protein gene and its association with productive life in dairy cattle, Journal of Dairy Science 90 (2007) 2427–2433.
- [68] T.E. Spencer, F.W. Bazer, Uterine and placental factors regulating conceptus growth in domestic animals, Journal of Animal Science 82 (2004) 4–13.
- [69] S. Ortega, M. Ittmann, S.H. Tsang, M. Ehrlich, C. Basilico, Neuronal defects and delayed wound healing in mice lacking fibroblast growth factor 2, The Proceedings of the National Academy of Sciences 95 (1998) 5672–5677.
- [70] L.A. Okumu, T. Fair, J. Szekeres-Bartho, A.M. O'Doherty, M.A. Crowe, J.F. Roche, P. Lonergan, N. Forde, Endometrial expression of progesterone induced blocking factor and galectins-1, -3, -9 and -3 binding protein in the luteal phase and early pregnancy in cattle, Physiological Genomics 43 (2011) 903–910.
- [71] O.M. Ocon-Grove, F.N. Cooke, I.M. Alvarez, S.E. Johnson, T.L. Ott, A.D. Ealy, Ovine endometrial expression of fibroblast growth factor (FGF) 2 and conceptus expression of FGF receptors during early pregnancy, Domestic Animal Endocrinology 34 (2008) 135–145.
- [72] K.J. McDowell, D.C. Sharp, Maternal recognition of pregnancy, in: A.O. McKinnon, E.L. Squires, W.E. Vaala, D.D. Varner (Eds.), Equine Reproduction, 2, Wiley-Blackwell, UK, 2011, pp. 2200–2210.
- [73] K.W. Walters, J.F. Roser, G.B. Anderson, Maternal-conceptus signaling during early pregnancy in mares: oestrogen and insulin-like growth factor I, Journal of Reproduction and Fertility 121 (2001) 331–338.
- [74] M.A. Horisberger, M.C. Gunst, Interferon-induced proteins: identification of Mx proteins in various mammalian species, Virology 180 (1991) 185–190.
- [75] B.A. Hicks, S.J. Etter, K.G. Carnahan, M.M. Joyce, A.A. Assiri, S.J. Carling, K. Kodali, G.A. Johnson, T.R. Hansen, M.A. Mirando, G.L. Woods, D.K. Vanderwall, T.L. Ott, Expression of the uterine Mx protein in cyclic and pregnant cows, gilts, and mares, Journal of Animal Science 81 (2003) 1552–1561.
- [76] B. Charleston, H.J. Stewart, An interferon-induced Mx protein: cDNA sequence and high-level expression in the endometrium of pregnant sheep, Gene 137 (1993) 327–331.
- [77] G.A. Johnson, M.M. Joyce, S.J. Yankey, T.R. Hansen, T.L. Ott, The interferon stimulated genes (ISG) 17 and Mx have different temporal and spatial expression in the ovine uterus suggesting more complex regulation of the Mx gene, Journal of Endocrinology 174 (2002) R7–R11.
- [78] R.M. Moor, L.E. Rowson, Egg transfer in sheep. Factors affecting the survival and development of transferred eggs, Journal of Reproduction and Fertility 1 (1960) 332–349.
- [79] P.M. VanRaden, R.H. Miller, Effects of nonadditive genetic interactions, inbreeding and recessive defects on embryo and fetal loss by seventy days, Journal of Dairy Science 89 (2006) 2716–2721.
- [80] W.A. King, Chromosomal abnormailities and pregnancy failure in domestic animals, Advances in Veterinary Science & Comparative Medicine 34 (1990) 229–234.
- [81] F. Gayerie de Abreu, G.E. Lamming, R.C. Shaw, A cytogenetic investigation of early bovine embryos: relation with embryo mortality, in: Proc. of 10th International Congress of Animal Reproduction and Artificial Insemination, Urbana IL, vol. II, 1984, p. 82.
- [82] J.L. Robinson, D.B. Dombrowski, G.W. Harpestad, R.D. Shanks, Detection and prevalence of UMP synthase deficiency among dairy cattle, Journal of Heredity 75 (1984) 277–280.
- [83] R.D. Shanks, J.L. Robinson, Embryonic mortality attributed to inherited deficiency of uridine monophosphate synthase, Journal of Dairy Science 72 (1989) 3035–3039.

- [84] B.R. Yadav, W.A. King, K.P. Xu, J.W. Pollard, L. Plante, Chromosome analysis of bovine oocytes cultured in vitro, Genetics Selection Evolution 23 (1991) 191–196.
- [85] J.N. Wiltbank, W.W. Rowden, J.E. Ingalls, K.E. Gregory, R.M. Koch, Effect of energy level on reproductive phenomena of mature Hereford cows, Journal of Animal Science 21 (1962) 219–224.
- [86] L.D. Dunne, M.G. Diskin, M.P. Boland, K.J. O'Farrell, J.M. Sreenan, The effects of pre-and postinsemination plane of nutrition on embryo survival in beef heifers, Animal Science 69 (1999) 411–417.
- [87] W.R. Butler, Review: effect of protein nutrition on ovarian and uterine physiology in dairy cattle, Journal of Dairy Science 81 (1998) 2533–2539.
- [88] J. Kennedy, P. Dillon, K. O'Sullivan, F. Buckley, M. Rath, The effect of genetic merit for milk production and concentrate feeding level on reproductive performance of Holstein-Friesian cows in a grass-based system, Animal Science 76 (2003) 297–308.
- [89] B. Horan, J.F. Mee, M. Rath, P. O'Connor, P. Dillon, The effect of strain of Holstein-Friesian cow and feeding system on reproductive performance in seasonal-calving milk production systems, Animal Science 79 (2004) 453–467.
- [90] D.G. Morris, S. Waters, S. McCarthy, J. Patton, B. Earley, R. Fitzpatrick, J.J. Murphy, M.G. Diskin, D. Kenny, A. Brass, D.A. Wathes, Pleiotropic effects of negative energy balance in the postpartum dairy cow on splenic gene expression: repercussions for innate and adaptive immunity, Physiological Genomics 39 (2009) 28–37.
- [91] D.C. Wathes, Z. Cheng, W. Chowdhury, M.A. Fenwick, R. Fitzpatrick, D.G. Morris, J. Patton, J.J. Murphy, Negative energy balance alters global gene expression and immune responses in the uterus of postpartum dairy cows, Physiological Genomics 39 (2009) 1–13.
- [92] R.D. Geisert, G.L. Morgan, E.C. Short, M.T. Zavy, Endocrine events associated with endometrial function and conceptus development in cattle, Reproduction Fertility and Development 4 (1992) 301–305.
- [93] L.M. McDonald, R.E. Nichols, S.H. McNutt, Studies on corpus luteum ablation and progesterone replacement therapy during pregnancy in the cow, American Journal of Veterinary Research 13 (1952) 446-451.
- [94] A.J.H. Stronge, J.M. Sreenan, M.G. Diskin, J.F. Mee, D.A. Kenny, D.G. Morris, Post-insemination milk progesterone concentration and embryo survival in dairy cows, Theriogenology 64 (2005) 1212–1224.
- [95] R.E. McNeill, M.G. Diskin, J.M. Sreenan, D.G. Morris, Association between milk progesterone concentration on different days and with embryo survival during the early luteal phase in dairy cows, Theriogenology 65 (2006) 1435–1441.
- [96] O.J. Franco, M. Drost, M.J. Thatcher, V.M. Shille, W.W. Thatcher, Fetal survival in the cow after pregnancy diagnosis by palpation per rectum, Theriogenology 27 (1987) 631–644.
- [97] M.J. Starbuck, R.A. Dailey, E.K. Inskeep, Factors affecting retention of early pregnancy in dairy cattle, Animal Reproduction Science 84 (2004) 27–39.
- [98] J. Chagas e Silva, L. Lopes da Costa, Luteotrophic influence of early bovine embryos and the relationship between plasma progesterone concentrations and embryo survival, Theriogenology 64 (2005) 49–60.
- [99] A.K. Goff, Embryonic signals and survival, Reproduction in Domestic Animals 37 (2002) 133-139.
- [100] P.J. Hansen, Regulation of uterine immune function by progesterone lesson from the sheep, Journal of Reproductive Immunology 40 (1998) 63–79.
- [101] E.K. Inskeep, Preovulatory, postovulatory, and postmaternal recognition effects of concentration of progesterone on embryonic survival in the cow, Journal of Animal Science 82 (2004) 24–39.

- [102] G.E. Mann, G.E. Lamming, Relationship between maternal endocrine environment, early embryo development and inhibition of the luteolytic mechanism in cows, Reproduction 121 (2001) 175–180.
- [103] K.P. Nephew, K.E. McClure, T. Ott, D.H. Budois, F.W. Bazer, W.F. Pope, Relationship between variation in conceptus development and differences in estrous cycle duration in ewes, Biology of Reproduction 44 (1991) 536–539.
- [104] S.F. Larson, W.R. Butler, W.B. Currie, Pregnancy rates in lactating dairy cattle following supplementation of progesterone after artificial insemination, Animal Reproduction Science 102 (2007) 172–179.
- [105] P.J. Hansen, Embryonic mortality in cattle from the embryo's perspective, Journal of Animal Science 80 (2002) 33–44.
- [106] D. Cavestany, A.B. El-Whishy, R.H. Foote, Effect of season and high environmental temperature on fertility of Holstein cattle, Journal of Dairy Science 68 (1985) 1471–1478.
- [107] S.E. Dunlap, C.K. Vincent, Influence of post breeding thermal stress on conception rate in beef cattle, Journal of Animal Science 32 (1971) 1216–1218.
- [108] S. Mondal, P. Palta, B.S. Prakash, Influence of season on peripheral plasma progesterone concentrations in Murrah buffaloes (*Bubalus bubalis*), Buffalo Journal 1 (2004) 95–100.
- [109] D. Wolfenson, W.W. Thatcher, L. Badinga, J.D. Savio, R. Meidan, B.J. Lew, R. Braw-Tal, A. Berman, Effect of heat stress on follicular development during estrus cycle in lactating dairy cattle, Biology of Reproduction 52 (1995) 1106–1113.
- [110] D. Wolfenson, B.J. Lew, W.W. Thatcher, Y. Graber, R. Meidan, Seasonal and acute heat stress effects on steroid production by dominant follicles in cows, Animal Reproduction Science 47 (1997) 9–19.
- [111] Y.M. Al-Katanani, F.F. Paula-Lopes, P.J. Hansen, Effect of season and exposure to heat stress on oocyte competence in Holstein cows, Journal of Dairy Science 85 (2002) 390–396.
- [112] S. Sugiyama, et al., Effects of increased ambient temperature on the development of in vitro derived bovine zygotes, Theriogenology 60 (2003) 1039–1047.
- [113] J.L. Edwards, P.J. Hansen, Differential responses of bovine oocytes and preimplantation embryos to heat shock, Molecular Reproduction and Development 46 (1997) 138–145.
- [114] R.M. Rivera, D.J. Hansen, Development of cultured bovine embryo after exposure to increased temperatures in the physiological range, Reproduction 121 (2001) 107–115.
- [115] J.L. Edwards, P.J. Hansen, Elevated temperature increases heat shock protein 70 synthesis in bovine two-cell embryo and compromises function of maturing oocytes, Biology of Reproduction 55 (1996) 340–346.
- [116] C.F. Arechiga, A.D. Ealy, D.J. Hansen, Evidence that glutathione is involved in thermotolerance of preimplantation mouse embryos, Biology of Reproduction 52 (1995) 1296–1301.
- [117] F.F. Paula-Lopes, P.J. Hansen, Heat induced apoptosis in preimplantation bovine embryos is a developmentally-regulated process, Biology of Reproduction 66 (2002) 1169–1177.
- [118] Z. Roth, R. Meian, R. Braw-Tal, D. Wolfenson, Immediate delayed effect of heat stress on follicular development and its association with plasma FSH and inhibin concentrations in cows, Journal of Reproduction and Fertility 120 (2000) 83–90.
- [119] D.J. Putney, J.R. Malayer, T.S. Gross, W.W. Thatcher, D.J. Hansen, M. Drost, Heat stress induced alterations in the synthesis and secretion of protein and prostaglandins by cultured bovine conceptus and uterine endometrium, Biology of Reproduction 37 (1988) 771–782.
- [120] H. Houe, Supervisorship of animals persistently infected with bovine viral diarrhoea virus (BVDV) infections, Veterinary Microbiology 64 (1999) 89–107.

- [121] B. Thur, M. Hilbe, M. Strasser, F. Ehrensperger, Immunohistochemical diagnosis of pestivirus infection associated with bovine abortion and perinatal death, American Journal of Veterinary Research 58 (1997) 1371–1375.
- [122] A. Bielanski, C. Dubuc, In vitro fertilization of ova from cows experimentally infected with a noncytopathic strain of bovine viral diarrhea virus, Animal Reproduction Science 38 (1995) 215–221.
- [123] J.M. Miller, M.J. Van der Maaten, C.A. Whetstone, Infertility in heifers inoculated with modifiedlive bovine herpesvirus- 1 vaccinal strains against infectious bovine rhinotracheitis on postbreeding day 14, American Journal of Veterinary Research 50 (1989) 551–554.
- [124] J.M. Miller, M.J. Van der Maaten, Early embryonic death in heifers after inoculation with bovine herpesvirus-1 and reactivation of latent virus in reproductive tissues, American Journal of Veterinary Research 48 (1987) 1555–1558.
- [125] Anon, VIDA report ADAS, in: Ministry of Agriculture Fisheries and Food Central Veterinary Laboratory. Weybridge, 1988.
- [126] J.F.T. Griffin, P.J. Hartigan, W.R. Nunn, Non-specific uterine infection and bovine infertility. Infection patterns and endometritis during the first seven weeks post-partum, Theriogenology 1 (1974) 91–98.
- [127] H.C. Adler, Genital vibriosis in the bovine. An experimental study of early embryonic mortality, Acta Veterinaria Scandinavica 1 (1959) 1–6.
- [128] M. Engels, F. Steck, R. Wyler, Comparison of the genomes of infectious bovine rhinotracheitis and infectious pustular vulovaginit virus strains by restriction endonuclease analysis, Archives of Virology 67 (1981) 169–173.
- [129] R.M. Roberts, S. Xie, N. Mathialagan, Maternal recognition of pregnancy, Biology of Reproduction 54 (1996) 294–302.
- [130] M.P. Green, M.G. Hunter, G.E. Mann, Relationship between maternal hormone secretion and embryo development on day 5 of pregnancy in dairy cows, Animal Reproduction Science 88 (2005) 179–189.
- [131] W.W. Thatcher, F.W. Bazer, D.C. Sharp, R.M. Roberts, Interrelationship between uterus and conceptus to maintain corpus luteum function in early pregnancy: sheep, cattle, pig and horses, Journal of Animal Science 62 (1986) 25–46.
- [132] C.A. Gray, F.F. Bartol, B.J. Tarleton, A.A. Wiley, G.A. Johnson, F.W. Bazer, T.E. Spencer, Developmental biology of uterine glands, Biology of Reproduction 65 (2001) 1311–1323.
- [133] C.A. Gray, K.M. Taylor, W.S. Ramsey, J.R. Hill, F.W. Bazer, F.F. Bartol, T.E. Spencer, Endometrial glands are required for pre-implantation conceptus elongation and survival, Biology of Reproduction 64 (2001) 1608–1613.
- [134] C.A. Meza-Herrera, T. Ross, D. Hawkins, D. Hallford, Interactions between metabolic status, prebreeding protein supplementation, uterine pH, and embryonic mortality in ewes: preliminary observations, Tropical Animal Health and Production 38 (2006) 407–413.
- [135] M. Mihm, N. Curran, P. Hyttel, P.G. Knight, M.P. Boland, J.F. Roche, Effect of dominant follicle persistence on follicular fluid oestradiol and inhibin and on oocyte maturation in heifers, Journal of Reproduction and Fertility 116 (1999) 293–304.
- [136] N. Ahmad, F.N. Schrick, R.L. Butcher, E.K. Inskeep, Effect of persistent follicles on early embryonic losses in beef cows, Biology of Reproduction 52 (1995) 1129–1135.
- [137] M. Binelli, J. Hampton, W.C. Buhi, W.W. Thatcher, Persistent dominant follicle alters pattern of oviductal secretory proteins from cows at estrus, Biology of Reproduction 61 (1999) 127–134.
- [138] I. Revah, W.R. Butler, Prolonged dominance of follicles reduces the viability of bovine oocytes, Journal of Reproduction and Fertility 106 (1996) 39–47.

- [139] G.A. Perry, T.W. Geary, M.C. Lucy, M.F. Smith, Effect of follicle size at the time of induced ovulation on luteal function and fertility, Proceeding, Western Section. American Society of Animal Science 55 (2002) 45–48.
- [140] D.N. Logue, M.J.A. Harvey, Meiosis and spermatogenesis in bulls heterozygous for a presumptive 1/29 Robertsonian translocation, Journal of Reproduction and Fertility 54 (1978) 159–165.
- [141] I. Dyrendahl, I. Gustavsson, Sexual functions, semen characteristics and fertility of bulls carrying the 1/29 chromosome translocation, Heridatas 90 (1979) 281–289.
- [142] B.E. Ballachey, D.P. Evenson, R.G. Saacke, The sperm chromatin structure assay: relationship with alternate test of semen quality and heterospermic performance of bull, Journal of Andrology 9 (1988) 109–115.
- [143] J.R. Pursley, M.O. Mee, M.C. Wiltbank, Synchronization of ovulation in dairy cows using PGF2alpha and GnRH, Theriogenology 44 (1995) 915–923.
- [144] J.R. Pursley, M.C. Wiltbank, J.S. Stevenson, J.S. Ottobre, H.A. Garverick, L.L. Anderson, Pregnancy rates per artificial insemination for cows and heifers inseminated at a synchronized ovulation or synchronized estrus, Journal of Dairy Science 80 (1997) 295–300.
- [145] J.L. Vasconcelos, R.W. Silcox, G.J. Rosa, J.R. Pursley, M.C. Wiltbank, Synchronization rate, size of the ovulatory follicle, and pregnancy rate after synchronization of ovulation beginning on different days of the estrous cycle in lactating dairy cows, Theriogenology 52 (1999) 1067–1078.
- [146] F. Moreira, C. Orlandi, C.A. Risco, R. Mattos, F. Lopes, W.W. Thatcher, Effects of resynchronization and bovine somatotropin on pregnancy rates to a timed artificial insemination protocol in lactating dairy cows, Journal of Dairy Science 84 (2001) 1646–1659.
- [147] P.S. Baruselli, E.H. Madureira, J.A. Visintin, R.C. Barnabe, R. Amaral, Timed insemination using synchronization of ovulation in buffalo, Revista Brasileira de Saúde e Produção Animal 23 (1999) 360–362.
- [148] V. Paul, B.S. Prakash, Efficacy of the ovsynch protocol for synchronization of ovulation and fixedtime artificial insemination in Murrah buffaloes (*Bubalus bubalis*), Theriogenology 64 (2005) 1049–1060.
- [149] S.Z. El-Zarkouny, J.A. Cartmill, B.A. Hensley, J.S. Stevenson, Presynchronization of estrous cycles before Ovsynch and progesterone in dairy cows: ovulation, pregnancy rates, and embryo survival, Journal of Dairy Science 87 (2004) 1024–1037.
- [150] C. Navanukraw, D.A. Redmer, L.P. Reynolds, J.D. Kirsch, A.T. Grazul-Bilska, P.M. Fricke, A modified presynchronization protocol improves fertility to timed artificial insemination in lactating dairy cows, Journal of Dairy Science 87 (2004) 1551–1557.
- [151] W.R. Butler, J.J. Calaman, S.W. Beam, Plasma and milk urea nitrogen in relation to pregnancy rate in lactating dairy cattle, Journal of Animal Science 74 (1996) 858–865.
- [152] E.R. Jordon, L.V. Swanson, Effect of crude protein on reproductive efficiency, serum total protein and albumin in the high producing dairy cows, Journal of Dairy Science 62 (1979) 58–63.
- [153] J.D. Armstrong, E.A. Goodall, F.J. Gordon, D.A. Rice, W.J. McGaughey, The effects of levels of concentrate offered and inclusion of maize gluten or fish meal in the concentrate on reproductive performance and blood parameters of dairy cows, Animal Production 50 (1990) 1–10.
- [154] C.R. Staples, J.M. Burke, W.W. Thatcher, Influence of supplemental fats on reproductive tissues and performance of lactating cows, Journal of Dairy Science 81 (1998) 856–871.
- [155] K.L. Macmillan, V.K. Taufa, A.M. Day, A.J. Peterson, Effects of supplemental progesterone on pregnancy rates in cattle, Journal of Reproduction and Fertility 43 (1991) 304.
- [156] B.G. Biggers, R.D. Geisert, R.P. Wettemann, D.S. Buchanan, Effect of heat stress on early embryonic development in the beef cow, Journal of Animal Science 64 (1987) 1512–1518.
- [157] S. Santo-Valadez, G.E. Seidel, R.P. Elsden, Effect of hCG on pregnancy rates in bovine embryo transfer recipients, Theriogenology 17 (1982) 85–93.
- [158] J.S. Stevenson, E.P. Call, R.K. Scoby, A.P. Phatak, Double insemination and gonadotropin-releasing hormone treatment of repeat-breeding dairy cattle, Journal of Dairy Science 73 (1990) 1766–1772.
- [159] T. Diaz, E.J.P. Schmitt, R.L. de la Sota, M.J. Thatcher, W.W. Thatcher, Human chorionic gonadotropin-induced alterations in ovarian follicular dynamics during the estrous cycle of heifers, Journal of Animal Science 76 (1998) 1929–1936.
- [160] W.W. Thatcher, F. Moreira, J.E.P. Santos, Strategies to improve reproductive management of dairy cows, in: J. Kennelly (Ed.), Advances in Dairy Technology, vol. 12, The University of Alberta, Edmonton, AB, 2000, pp. 177–193.
- [161] S. Martinod, R.R. Maurer, B. Siegenthaler, C. Gerber, P.J. Hansen, The effect of recombinant bovine interferon-α on fertility in ewes, Theriogenology 36 (1991) 231–239.
- [162] T.K. Schalue Francis, P.W. Farin, J.C. Cross, D. Keisler, R.M. Roberts, Effect of injected bovine interferon-α1 on estrous cycle length and pregnancy success in sheep, Journal of Reproduction and Fertility 91 (1991) 347–356.
- [163] C.M. Barros, G.R. Newton, W.W. Thatcher, M. Drost, C. Plante, P.J. Hansen, The effect of bovine interferon α1 on pregnancy rate in heifers, Journal of Animal Science 70 (1992) 1471–1477.
- [164] F. Moreira, F.F. Paula-Lopes, P.J. Hansen, L. Badinga, W.W. Thatcher, Effect of growth hormone and insulin-like growth factor-I on development of in vitro derived bovine embryos, Theriogenology 57 (2002) 895–907.



# Index

'Note: Page numbers followed by "f" and "t" refer to figures and tables, respectively.'

# A

Abdominal organs bioengineering, 326 bladder, 330-331 challenge of immunosuppression, 337-338 gastrointestinal tract, 328-330 kidney, 331-333 liver, 335-337 pancreas, 333-335 perspectives, 339 tissue engineering and regenerative medicine, solutions for, 325-339 Abp118, 414t-416t A-Brisbane-59-2007 H1N1 influenza A virus, 195-196 A-California-04-2009 virus, 195-196 Acetylcholine, 328 Actagardine, 414t-416t Actinomyces pyogenes, 614–615 Adalimumab, 84-85, 85f, 108-109 Adcetris (brentuximab vedotin), 7-8 Adeno-associated viral vector (AAV1), 9 Adenosine deaminase, 235 Adenosine triphosphate (ATP), 127 Adenosine triphosphate-binding cassette (ABC) transporters, 406-410, 564 - 566Adipogenic cell types, 320 Adipose-derived MSCs (AMSCs), 351, 358 versus bone marrow-derived MSCs, 351-352 Adipose-derived stem cells, 319–320 isolation of, 318-319 Adjuvant, defined, 447-448 Adult stem cells (ASCs), 316, 318, 327 Affinity chromatography, 472

Affinity purification of human antibodies using antigens immobilized on blot membranes, 164-167 AGO proteins, 220-221 AGO1 protein, 221-222 Agrobacterium, 204 AKR1B5 expression, 605 Alanine scanning, 175, 175f Aldo-keto reductase (AKR), 604-605 Alemtuzumab, 80–83 Alginate, 330 Alkaline phosphatase, 320–321 Alpheon (BioPartners GmbH), 37 Amastigote infection, macrophages for, 579 - 580Amgen, 13–14, 61 "Amino acid mixture" method, 144-146 D-Amino acids, 147 L-Amino acids, 147 2'-Amino pyrimidines, 130-131 Ammonium sulfate precipitation, obtaining crude human immunoglobulin by, 163 - 164Amphotericin B, 566-567 Amplification, 169–171 Amylolysin, 414t-416t Animal vaccines, 439, 448-449 Annealing, 531-532 AntagoNAT oligonucleotides, 218-219 Antiamastigote assay, 579-580 automated screening, 580-581 axenic amastigotes, 580-581 high-content screening (HCS), 581 high-throughput screening (HTS), 581 macrophages for amastigote infection, 579-580 propidium iodide dye, 580-581

Antibodies, classes of, 73 Antibody-dependent cytotoxicity (ADCC), 83 Antibody purification, affinity to membrane-blotted antigens, 163 - 167affinity purification of human antibodies, 164 - 167obtaining crude human immunoglobulin, 163 - 164Anti-CD3 IgG, 97-99 Anti-CD20, 83-84, 97-99 Antigen-antibody interactions, 175 Antileishmanial biocompound screening, 563, 574f biotechnological development of new antileishmanial drugs, 569-570 blood cell disruption, 576-577 cytotoxicity assays, 585 experimental models, 577-582 antiamastigote assay, 579-580 antipromastigote assay, 577-578 in silico assays, 582 in vitro assay, 581 in vivo assays, 581-582 *Leishmania* counting in Thoma chamber, 577 Leishmania culturing, 575-576 methods for, 573-585 perspectives, 585-587 primary chemotherapy with antimonials, 564 - 566second generation of antileishmanial drugs, 566-569 amphotericin B, 566-567 miltefosine, 567-568 paromomycin, 569 pentamidine, 567 Antimicrobial peptides, 372 Antipromastigote assay, 577–578 3-(4,5-dimethylthiazol-2-yl)-2,5diphenyltetrazolium, 578 radiolabeled thymidine, 578-579 Antisense RNA (asRNA), 217 Anti-tumor necrosis factor, 84-85

Antitumor treatment, 55 Antiviral treatment, interferon-alfa in, 54 - 55Aptamer, 26, 125–136 application of aptamers, 131-132 in clinical trials, 129t, 133-136 exponential enrichment technique, systematic evolution of ligands by, 125 - 128therapeutics features of, 128-131 Arabidopsis thaliana, 220–221 ARC1779, 135-136 ARC1905 (Zimura), 134 ARC19499, 136 Artemisia annua, 204 Artemisinin biosynthetic pathway, 204 Artificial insemination (AI), 611 Aryloxy benzocycloalkyl azoles, 571 AS1411, 135 Asparaginase, 24 Aspergillus spp., 31 Atryn, 4 Autoimmune thyroid disease (AITD), 218 Autologous cells, 327-328 Axenic amastigotes, 580-581 2,2'-Azino-bis(3-ethylbenzothiazoline-6sulphonic acid), 77

## В

Baby hamster kidney (BHK), 528-529 Baby hamster kidney cell line (BHK-21), 500, 502, 543 adaptation of, to serum-free media (case study), 502-512 adaptation to Cellvento BHK-200, 510-512 Ex Cell 302 from standard culture media, 507-509 VP-SFM from standard culture media, 506-507 BHK-21C13 cells, 543 cell count, 504-505 cell cultivation, 503-504 cell line, 502 cryopreservation, 505

culture media, 503 growth kinetics, 505-506 metabolism analysis, 505 sedimentation tests, 505 strategies for cell adaptation to SFM, 503, 504f Bacillus anthracis, 203, 412 Bacillus licheniformis, 378 Bacillus strains, 418 Bacillus subtilis, 58, 114, 185, 197-198, 217, 378 Bacillus subtilis strain 168, 411 Bacillus subtilis subsp. spizizenii DSM15029, 418 Bacterial vaccines, 448-449 Bacterial vaginosis (BV) therapies, 419-420 Bacteriocins, 403-424 antimicrobial and antibiofilm activities of, 413-421, 414t-416t investigations with liposome-entrapped bacteriocins, 421-423 classification, 404-410, 409t mechanisms of action, 410-412 microbial biofilm control, 404 perspectives, 423-424 resistance mechanisms, 412 Bacteriophages, 203 BamHI, 466-467 Basal media, 500 Batch drying, 535 BblBricks, 197 B-cell activating factor, 85-86 Belimumab, 85-86 Belousov–Zhabotinsky (BZ) oscillating chemical reaction, 191-192 Benzoquinones, 572 Beta-cell replacement, 334  $\beta$ -site APP-cleaving enzyme 1-antisense (BACE1-AS), 250 Bexsero (Novartis), 11 BglBricks, 197, 198f Bifunctional antibody fragments/fusion proteins, 105 Bioactive fusion proteins, 8 BioBricks, 197, 198f

**Bioemulsifiers**, 386 Biofilms, 372-373, 403 current approaches to preventing or removing, 373-375 Biological networks, types of, 186-187 Biological systems, network measures for, 186 Biologics Control Act of 1902, 35 **Biologics Price Competition and Innovation** Act. 36-37 Biomimesis, 288 Biomimetic constructs for tissue engineering, 287-299 collagen-based, 295-296 collagen used as biomaterial, 297-299 **Biomimetic elements** of macroscale scaffold structure, 304-305 nanoscale and microscale scaffold structure of, 299-303 Biopharmaceutical products, 3-18 manufacturing technology downstream processing, 16-17 upstream processing, 14-16 product complexity versus expression system, 12-14 recombinant therapeutic proteins, production processes for, 12-17 technological challenges and future perspectives, 17-18 types of, for human health, 6-12cell therapies, 10 in vitro and in vivo diagnostic products, 11 - 12nucleic acid-based therapies, 8-10 recombinant therapeutic proteins, 6-8 vaccines, 10-11 Biopharmaceutical proteins, 7 Biopharmaceuticals, 3, 23-24 classes, variations, and applications, 24-28, 25f defined, 3 market aspects, 4-5 Biopharmaceuticals and biosimilars, 23 - 42Brazil, regulations in, 40-42 financial market, 31-35

Biopharmaceuticals and biosimilars (Continued) legislation history, 35-37 pharmacogenomics and next-generation biopharmaceuticals, 42 protein expression, heterologous systems for, 29-31 world regulations, 37-39 Biopolymers, 356-358 Biosafety and ethical aspects surrounding synthetic biology, 206 Biosimilars, 28 approved for commercialization, 30t stages of development, 29t Biosurfactants, 371 advantages, 375 as antimicrobial and antibiofilm agents, 371-392 biofilms, 372-373 current approaches to preventing or removing biofilms, 373-375 glycolipids, 381-385 high-molecular-weight biosurfactants, 386 - 387from Lactobacillus, 381-386 lipopeptides (LPs), 377-381 in nanotechnology, 387–392 functionalized nanoparticles, 387-388 gene transfection with liposomes, 391-392 liposomes, 389-391 natural antimicrobial compounds as potential antibiofilm agents, 375-387 perspectives, 392 potential use of biosurfactants in nanotechnology, 387-392 functionalized nanoparticles, 387-388 gene transfection with liposomes, 391 - 392liposomes, 389-391 Biotechnological routes, 29 Bispecific antibody fragments, 104-105 Bladder, 330-331 cell-scaffold approach, 330 Blastocyst, 317, 603

Blood factors group, 24 Blot membranes, affinity purification of human antibodies, 164-167 B-lymphocyte stimulator (BLyS), 85–86 BM86 antigen, 452-453 Boceprevir, 55 Bone marrow-derived MSCs (BMMSCs), 329, 332, 337, 351 versus adipose-derived MSCs, 351-352 Bone marrow-derived progenitor cells, 337 Bone regeneration, 318 Bone remodeling, 327 Bordetella pertussis, 459-460 cultivation and inactivation of, 465-466 extraction and quantification of DNA of, 466 Borrelia burgdorferi, 202–203 Botrylloides leachi, 226 Bottom-up approach, 297 Bottromycin A2, 414t-416t, 419 Bottromycins, 412, 419 Boulogne sore throat. See Diphtheria Bovine somatotropin (bST), 620 Bovine viral diarrhea (BVD), 614–615 Brazil, regulations in, 40-42 Bromide MTT, 578 Bromodeoxyuridine (BrdU), 579 Brucella abortus, 614-615 Brutal immunosuppressive protocols, 337-338 Bulk drying, 535

## С

Caenorhabditis elegans, 220–221 Calcium (CPC) Liquicolor assay, 320 Candida albicans, 381, 383, 386–387 Candida difficile-associated diarrhea (CDAD), 419 Candida lipolytica UC, 383, 386 Candida sphaerica UCP 0995, 383 Candida spp., recombinant vaccines against, 449 Candida tropicalis, 382 Canine rabies, 499 Canonical structures, 100–101 Carbon nanotube-collagen (CNT-Col) gel fabrication process, 298-299 Carnobacteriocin A (CbnA), 414t-416t CCS medium composition, 576t CD11c<sup>+</sup> dendritic cells, 252-253 CD20, 83 CD52, 80-83 CD105, 358 CD271, 358 CDKN1A gene, 247-248 Cell culture, 353 rabies vaccine development in. See Rabies vaccine development, in cell culture Cell culture or embryonated egg-based vaccines (CCEEVs), 526-527 Cell deposition, 350 Cell function, effects of scaffold topography on, 305 Cell microenvironment, 349-350 Cell reprogramming, 10, 349, 354, 358-360 Cell reprogramming techniques, 10 Cell-scaffold method, 326, 339 Cell therapies, 10 Cell-to-cell communication systems, 373 - 374Cell types in organ bioengineering, 326-328 Cellobiose lipids, 383-384 Cells maintenance and characterization. 319-321 adipose-derived stem cells, 319-320 generated bone cells, 320-321 alkaline phosphatase activity, 320-321 calcium content, 320-321 gene expression analyses, 321 histologic analyses, 321 osteopontin concentration, 320-321 Cellular automata (CA), 193 Cellular redox state, microRNA-mediated regulation of, 233-235 Cellular therapy, mesenchymal stem cells in, 352-353 Cellulose, 356-358 Cellvento BHK-200 medium, 501 Certolizumab pegol (CZP), 84-85

Cestodes, recombinant vaccines against, 451 - 452Cetyltrimethyl ammonium bromide (CTAB), 382-383 cGRGDdvc, 152 Chimeric proteins, 198-200, 202-203 "Chimeric" antibodies, 74-75 Chinese hamster ovary (CHO) cells, 4, 56 CHO K1 cells, 115-116 Chlamydomonas reinhardtii, 31 Chondrogenic cell types, 320 Chorionic gonadotropin (CG), 598-599 Chromatin remodeling and transcription regulation, linking, 249 Chromatin remodeling complexes and epigenetics, 246-247 Chromobox 7 (CBX7), 246-247 Chronic lymphocytic leukemia, 80-83 Circular bacteriocins, 406-410 Clarithromycin, 417 Class I Mccs, 404-406 Class II Mccs, 404-406 Class IIb bacteriocins, 420 Cloning, 57-58, 173, 443 Clustered regularly interspaced short palindromic repeat (CRISPR), 198 - 200Co-expression networks, 187 ColA-43864 bacteriocin, purified, 421 Cold trap, 532 COLD-ASSISTED INTRONIC NON-CODING RNA (COLDAIR), 254-255 COLD-INDUCED LONG ANTISENSE INTRAGENIC RNA (COOLAIR), 254 - 255Colicins (Cols), 404-406 Colistin, 379 Collagen, 330 -based biomimetic constructs, 295-296 as biomaterial, 297-299 Combinatorial chemistry, 149, 151 Combinatorial peptide library, 143 Complementary DNA (cDNA), 57-58, 99, 105-108, 616-617 Conformational flexibility, 150

Connectivity, 186 Consensus interferon, 51-52 Consensus IFN-a, 61 Contract manufacturing organizations (CMOs), 17 Corpus luteum (CL) lifespan, 598-599 Corynebacterium diphtheria. See also Diphtheria cloning and expression of, 479 Critical micellar concentration (CMC), 371.391 Crohn disease, 89-90 Crude human immunoglobulin, by ammonium sulfate precipitation, 163 - 164Cryoprotectants, 538 Cryptococcus humicola, 384 Cryptosporidium parvum, 244 **Current Good Manufacturing Practice** (CGMP) regulations, 36-37 Cutaneous diphtheria, 481-482 Cutaneous leishmaniasis (CL), 564 Cutaneous T-cell lymphoma (CTL), 80 - 83Cyclooxygenase-2 (COX-2), 605-606, 612 - 613Cyclophosphamide-hydroxydaunorubicinoncovin-prednisone (CHOP) therapy, 83 Cysticercosis, 167 Cytopathogenic (CP) bovine viral diarrhea, 614 - 615Cytosolic PGES (cPGES), 603 Cytotoxicity assays, 585 Cytotoxicity complement-dependent cellular cytotoxicity (CDCC), 83

## D

Daclizumab, 108–109 Deoxyadenosine triphosphate, 127 Diabody, 104 Differential scanning calorimetry (DSC), 531 DiGeorge critical region-8 (DGCR8), 237 Dihydrofolate reductase (DHFR), 74, 249 3-(4,5-Dimethylthiazol-2-yl)-2,5diphenyltetrazolium, 578 1,2-Dioleoyl-sn-glycero-3-phosphocholine, 391 Diospirine, 572 1,2-Dipalmitoyl-sn-glycero-3phosphocholine (DPPC), 390-391 Diphtheria, 479 cutaneous, 481-482 epidemiology, 482-484 historical aspects, 480 nasopharyngeal, 481-482 pathogen and toxin, 481 types, 481-482 vaccines. See Diphtheria vaccines Diphtheria toxin (DT), 481, 486t Diphtheria vaccines perspectives, 491-492 subunit vaccines, 485-486 technological development of, 487-491 cell growing and protein production kinetics, 490 diphtheria culture and DNA extraction, 487 - 488insert sequencing, 488 plasmid and diphtheria toxin subunit B DNA ligation, 488 plasmid linking to diphtheria toxin subunit B insert cloning, 488 protein expression, 489 purification, 490-491 western blot and protein sequencing, 489 - 490Diphtheria-tetanus-pertussis (DTP) vaccine, 462, 482, 484 Direct immunofluorescence, 547-548 Disease diagnosis, synthetic biology and, 201 - 203Disease networks, 187 Disease prevention, synthetic biology and, 205 - 206Disease therapy, synthetic biology and, 204 - 205Disulfide bonds, 100 "Divide, couple, recombine" method, 144

DNA sequencing, 207 protein identification by, 473 DNA vaccines, 26, 441, 442f DNA-based vaccines, 444t Docosahexaenoic acid (DHA), 618 - 619Double-stranded RNA (dsRNA), 216, 219 Drosha nuclear localization, 237 Drosophila melanogaster, 216, 220-221 Drug-delivery nanocarrier systems, 374-375 Dry matter intake (DMI), 611 DTaP vaccine, 462–463 production of, 463 Durable tolerance, achieving, 338 Duramycin, 414t-416t, 418

#### Е

E10030 (Fovista), 134 Early embryo loss causes of, 611-616 genetic abnormalities, 611 hormonal imbalance, 612–613 infectious agents, 614–615 maternal-fetal interactions, impairment of. 615 nutrition, 611-612 quality of oocytes and sperm, 616 thermal stress, 613-614 reduction strategies, 616-620 development of timed insemination programs, 616-618 nutritional strategies, 618-619 supplementation of luteotrophic agents, 619 supplementation with interferon, 619 use of bovine somatotropin, 620 Early embryonic mortality, 597–599 Ectoparasites, recombinant vaccines against, 452-453 Edman degradation, 147 Eicosapentaenoic acid (EPA), 618-619 Elelyso (glucocerebrosidase/taliglucerase alfa), 4

Embryonic stem cells (ESCs), 316–317, 327, 333, 353-354 obtainment, 317f Emerging economies, pharmaceutical sector in, 4-5 Enbrel (etanercept), 8 Enterocin AS-48, 414t-416t, 420 Enterocin CRL35, 414t-416t Enterococcus faecalis, eradicating, 417 Enzyme-linked immunosorbent assay, 77-78 Ephemerovirus, 524 Epidermicin NI01, 414t-416t Epidermin, 413, 414t-416t, 417-418 Epitope, 72-73  $\varepsilon$ -polylysine ( $\varepsilon$ -PL), 417 Erythropoietin (EPO), 13 Escherichia coli, 4, 13, 23, 31, 56-58, 61, 162, 197-198, 200-202, 204, 217, 443 biofilm, 376 fermentation in, 62 Escherichia coli BL21 pLysS, transformation of, 469 Escherichia coli DH5a, transformation of, 467 - 469Escherichia coli K12-derived strains, 114 Estradiol cypionate (ECP), 618 Etanercept, 27 European Medicines Agency (EMA), 37 Eutectic temperature, 531–532 Ex Cell 302, 501 infection kinetics of cells adapted to, 515 - 516Exopolysaccharide, 376 Exponential enrichment technique, systematic evolution of ligands by, 125 - 128Expressed sequence tags (EST) analyses, 260 Expression system, product complexity versus, 12-14 Extracellular matrix (ECM) scaffold, 287, 326, 329, 335 Extracellular polymeric substances (EPSs), 372 - 373Extravasation, 84

#### F

FarmaBrasil Group, 41 Fasciola gigantica, vaccines against, 452 Fasciola hepatica, vaccines against, 452 Federal Food, Drug and Cosmetics Act, 36 Fengycin, 379 Fetal bovine serum (FBS), 500 Fetal hepatoblasts, 337 Fibroblast growth factor-2 (FGF-2), 609 Filamentous fungi, 31 Filobasidiella neoformans, 384 Financial market, 31-35 First-generation biotechnological products, 41 Fleming, Alexander, 23 Flow cytometry, 78-79 Flublok (Protein Sciences), 11 N-(9-Fluorenylmethyloxycarbonyloxy)succinimide (Fmoc-OSu), 147 Fluorescence resonance energy transfer (FRET), 78-79, 78f 2'-Fluoropyrimidines, 130-131 Follicle-stimulating hormone (FSH), 24 Food, Drug, and Cosmetic Act, 36 Food and Drugs Act of 1902, 36 Foot and mouth disease (FMD) vaccines, 445-446, 454 Foot and mouth disease virus (FMDV), 501 FOVISTA, 127, 134 Foxo3, 238-241 Fragment crystallizable region receptors (FcR), 73 Framework regions (FRs), 100-101 Freeze-drying, 529-541 advances, 538-540 alternatives, 540-541 applications, 535-536 components of freeze-dryer, 533-535 lyophilization, 529-533 annealing stage, 531-532 freezing stage, 530-531 primary drying stage, 532 secondary drying stage, 533 methods, 533-535 use for vaccines, 536-538

Functionalized nanoparticles, 387–388 Fungal infections, recombinant vaccines against, 449 *Fusarium* spp., 31 Fusion proteins, 8

## G

α-Galactosidase, 24 Galectin, 607-608 Gallidermin, 414t-416t, 417-418 Gamma interferon activation factor (GAF), 602 Garvicin A, 412 Gastrointestinal tract, 328-330 GE11, 151-152 GE2270A, 412 Gendicine, 9 Gene therapy, 8-9 Gene transfection with liposomes, 391-392 Generated bone cells, 320-321 Genetic regulatory networks (GRNs), 187 Genome editing, 198-201, 199f, 205 tools, 199f Genome sequencing, 620 Gibson assembly, 195-196, 198f, 207 Glass transition temperature (Tg), 531 Global harmonization, 38 Glucantime, 564, 565f, 577 Glucocerebrosidase, 13-14, 24 Gluconacetobacter xylinus, 357 Glutathione (GSH), 603 Glybera, 4, 9 Glycogen synthase kinase- $3\beta$  (GSK- $3\beta$ ), 237 Glycolipids, 381-385 Glycoproteins, 12-13, 27 Glycosylation, 12-13, 27 Golimumab, 84-85 Gonadotropin-releasing hormone (GnRH), 616-618 GoTaq Flexi DNA Polymerase, 488 Gram-negative colicins, classification scheme for, 407t-408t Gram-negative microcins, classification scheme for, 405t

Gram-positive bacteria, classification of bacteriocins from, 406–410 Granulocyte colony stimulating factor (G-CSF), 13, 337 Graph, defined, 185 Gross domestic product (GDP), 5 Growth factor receptor-bound protein 2 (GRB2), 189 Growth hormone (GH), 610

#### Н

Haemophilus influenzae, 440 Hansen, Gerhard Henrik Armauer, 162 - 163Hansen disease, 162-163 Hansenula polymorpha, 31 Hatch-Waxman Act. 36 Heat-labile molecules, 406-410 Heat stress, 613-614 Heatsynch program, 618 Helicobacter pylori, 189, 217 Helminths, recombinant vaccines against, 451 - 452Hemagglutinin of influenza virus, 441 Hematopoietic stem cells (HSCs), 229-230 Hematoxylin-eosin, 321 Heparin-binding epidermal growth factor (HB-EGF), 481 Hepatitis B surface antigen, 26 Hepatitis C virus (HCV), 54, 218-219 Heterochromatin protein 1 (HP1), 223-224 Heterologous expression systems, 28, 31, 61 Heterologous genes, 203 Heterologous protein, cloning and expression of, 479-492 Heterologous systems for protein expression, 29-31 High-content screening (HCS), 581 High-molecular-weight biosurfactants, 386-387 High-temperature requirement A2 (HtrA2), 151 High-throughput screening (HTS), 581 HindIII, 466-467 Histone H3 at lysine 9 (H3K9), 221-222

Homozygous familial hypercholesterolemia, 9 Hormonal imbalance, and early embryo loss, 612–613 Horseradish peroxidase (HRP), 77 HOTAIRM1, 254 Hoxa6 transcription, 246-247 Hoxa7 transcription, 246-247 Human anti-mouse antibody (HAMA), 108 - 109Human chorionic gonadotropin (hCG), 619 Human embryo kidney (HEK-293), 115 Human embryonic cells (hESCs), 249, 251 Human growth hormone, 24 Human keratinocytes, 327 Human-derived biosurfactants, 376 Hyaluronidase, 24 Hybrid DNA-RNA, 321 Hybridoma cells, 74, 75f Hybridoma technology, 11, 108–109, 112-113 Hydrodynamic focusing, 79, 79f Hydrogen peroxide modulation of microRNA biogenesis by, 236 - 237regulation of specific microRNA by, 238-241, 239t-240t Hydroxyapatite (HA)/b-tricalcium phosphate (b-TCP) scaffold, 319 Hypertriglyceridemia, 9 Hypervariable loops, 100-101 Hypoxanthine-guanine phosphoribosyl transferase (HGPRT) gene, 74

#### I

Ibritumomab tiuxetan, 83–84 IFN-stimulated genes (ISGs), 600–601, 606 IFN-stimulated response element (ISRE), 602 IGF-1 receptors (IGF-1R), 610 IL6-activated cells, 88–89 Immune cell development and functions, impact of microRNAs in, 229–231 Immune IFN, 51–52 Immune responses induced by vaccination, 443–445

Immune system, 71–73 innate and adaptive immunity, 72-73 microscopic simulation of, 194-196 miRNAs in, 231-233 origins of, 71-72 Immunodiagnosis, 162 Immunodiagnostic sector, 97, 100 Immunofluorescence assays, 514 Immunoglobulin, 73, 73f, 101f, 163-164 IgE, 501 IgG, 100 Immunohistochemical staining, 336-337 Immunohistochemistry, 80 Immunosuppression, challenge of, 337-338 Import Drugs Act of 1848, 35 In silico antileishmanial assays, 582 In vitro diagnostic products, 11–12 In vitro selection, 125-126, 126f, 128 In vivo antileishmanial assays, 581-582, 583t-584t In vivo diagnostic products, 11-12 Inactivated viral suspension (IVS), 550-551 Indolicidin, 391 Induced pluripotent stem cells (iPSCs), 10, 248, 316, 318, 327, 329, 353-354 Infectious agents, and early embryo loss, 614 - 615Infergen, 61 Infliximab, 84-85 Injectable inactivated (killed) vaccines, 528 Innate and adaptive immunity, 72-73 Inner cell mass (ICM), 317, 317f Innovation Act and Price Competition of Biosimilars (BPCIA), 35 Insert sequencing, 488 Insulin, 4 Insulin-like growth factor-1 (IGF-1), 610 signaling activation, 238-241 Insulin therapy, 334 Integrin, 153, 299-300, 608 Interferons, 24-26, 51-63 antitumor treatment, 55 clinical applications, 54 IFN-a2b, 52 IFN-alfa, 51-55, 57-58, 61

IFN-alfa2 subtype, 55 IFN-beta, 51-56, 63 IFN-epsilon, 56 IFN-gamma, 51-52, 54, 56, 253 IFN-gamma receptor (IFNGR), 52 IFN-kappa, 57 IFN-lambda, 57 IFN-omega, 57 IFN-tau, 57, 597-598, 600f, 606 IFN-zeta, 57 interferon-alfa in antiviral treatment. 54 - 55modified or new (synthetic) interferons and formulation, 61-63 future perspectives, 63 production and formulation, 57-60 protein structures, 53-54 type I, 52 type III, 53 Interleukin-1 receptor associated kinase-4 (IRAK4), 189 Interleukins (ILs), 24–26 Intestinal organoid units, 329 Intracellular signaling networks, 186 Ipilimumab, 86 Iron oxide nanoparticles, 388 Isopropyl  $\beta$ -D-1-thiogalactopyranoside (IPTG), 466, 469, 471-472 Isothermal desorption, 533 Iterative deconvolution, 146 Iturin, 377-379

J

Jenner, Edward, 11

# Κ

Kadcyla (trastuzumab emtansine), 7–8 Kefauver–Harris amendments, 36 Kidney, 331–333 cell-scaffold strategies, 332–333 cell therapy, 332–333 transplants, 325 Killed/inactivated vaccines, 439 Kocuran-capped AgNPs, 388 Kohler, G., 97 Kynamro, 9–10

#### L

Lactic acid bacteria bacteriocins, classification system for, 410 Lacticin 3147, 414t-416t, 417 Lacticin O, 417 Lactobacillus, biosurfactants from, 385-386 Lactobacillus acidophilus, 385-386 Lactococcin 972, 412 Lactococcin A, 414t–416t Lactococcus lactis, 412 Lantibiotics, 406-410, 412 Large intervening ncRNA (lincRNA), 226 - 227Law for Technological Innovation, 41 Law of Generics, 40 Legislation history, 35-37 Leishmania amazonesis, 569, 580-581 Leishmania counting, in Thoma chamber, 577 Leishmania culturing, 575-576 Leishmania infantum, 568, 576 Leishmania monocytogenes, 413-417, 422 - 423Leishmania Ross, 563 Leishmania tarentolae, 31 Leishmaniasis, 450-451, 563-566, 569 Leish-Tec, 450–451 Leptospirosis, recombinant vaccine against, 449 LeukoScan, 12 LFF571, 414t-416t Libraries from libraries, 149 Lichen galactomannans, 572 Licheniocin 50.2, 420 Life-Saving Medicine Act, 36-37 Linoleic acid, 618-619 Lipopeptides (LPs), 377-381 from Bacillus, 377 Lipopolysaccharide (LPS), 230-231, 244, 376, 379-380, 386 Liposome-entrapped bacteriocins, 421–423 Liposome-entrapped P34, 422–423

Liposomes, 389-391, 404, 422 Listeria-active with YGNGVXC motif, 406 - 410Listeria monocytogenes, 378, 382-383, 419 - 420Live attenuated vaccines, 439 Live vaccines, 527 Liver progenitor cells, 337 Liver regeneration, 335-337 Livestock vaccines, 445-446 Long coding RNA structural labels and partners of transcription factors, 247 - 249Long noncoding RNAs (lncRNAs), 215-218, 226-228, 246-257 and alternative splicing, 249-250 chromatin remodeling complexes and epigenetics, 246-247 in immune system, 252-254 linking chromatin remodeling and transcription regulation, 249 long coding RNA structural labels and partners of transcription factors, 247 - 249long noncoding RNAs and alternative splicing, 249-250 role in the immune system, 252-254 as sensors of environmental conditions, 254 - 255mediators of messenger RNA stability and translation, 250-252 as sensors of environmental conditions. 254 - 255Lotka–Volterra system, 191 Luciferase, 105 Luminal epithelium (LE), 599-600, 606-607 Lunasan and rufisan, 383 Luteinizing hormone (LH), 599-600 Luteotrophic agents, 598-599, 619 Lyophilization, 529-533, 537 of amino acids, 552 annealing stage, 531–532 freezing stage, 530-531 primary drying stage, 532 secondary drying stage, 533

Lyophilized cakes, 552f, 553t Lyoprotectants, 538 *Lysinibacillus fusiformis* S9, 385 Lyssavirus, 499, 524

## М

Macroscale scaffold structure, biomimetic elements of, 304-305 MACUGEN, 127 Mammalian cells, 14, 115–116 Manifold drying, 534-535 4-O-β-D-Mannopyranosylmesoerythritol, 383 Mannose phosphotransferase system (Man-PTS) membrane components, 411 Mannosylerythritol lipid (MEL), 383, 390 MEL-A, 392 MEL-D, 383 Manufacturing process, 16-17 Manufacturing technology downstream processing, 16-17 upstream processing, 14-16 Mass cytometry, 80 Mass spectrometry, 473-474 Masson trichrome procedures, 321 Maternal recognition of pregnancy (MRP), 597 - 602events of, 599f gene regulating, 602–610 cyclooxygenase-2 (COX-2), 605-606 fibroblast growth factor-2 (FGF-2), 609 galectin, 607-608 insulin-like growth factor-1 (IGF-1), 610 integrin, 608 interferon-t, 606 Mx, 610 osteopontin (OPN), 606-607 prostaglandin E synthase (PGES), 603 - 604prostaglandin F synthase (PGFS), 604 - 605serpin, 608-609 Maternal-fetal interactions, impairment of. 615 Mathematics, 190-191

MccB17, 411, 414t-416t MccC7, 414t-416t MccC7-C51, 421 MccE492, 411, 414t-416t MccJ25, 411, 414t-416t MccM, 414t-416t McKinsey company, 5 Mechanosensitive channels (MscL), 411 Melt back, 552 Mepolizumab, 86 Merck, 23 Mersacidin, 414t-416t, 418 Mesenchymal stem cells (MSCs), 300, 332, 350 - 351biomaterials, 356-358 in cellular therapy, 352–353 characterization of, 351 expansion, 353 induced pluripotent stem cells (iPSCs), 353 - 354intrinsic properties, 352 pluripotency of stem cells, 350-354 differences between AMSCs and BMMSCs, 351-352 scaffold interaction with, 358-360 scaffold to cellular therapy, 358 scaffold to reprogram MSCs to neurons, 358-360 seeded on biofunctionalized scaffold. 349-360 sources, 351 substrate, 354-356 Metabolic networks, 187 Metagenomic immune system (MIS), 72 Methicillin-resistant S. aureus (MRSA), 374, 454 2'-O-Methyl ribose pyrimidines and purines, 130-131 Mexican-American war, 35 Micro RNAs, 9, 216, 218-222, 228-246, 234f, 238f, 243f, 256-258 in crossroad between immune functions and oxidative stress, 241-246 reactive oxygen species, aging, and immune senescence, 244-245

reactive oxygen species, inflammation, and cancer, 245-246 TLR pathways and reactive oxygen species activation in inflammatory responses, 242-244 in immune cells and during cellular stress responses, 235 - 241development and functions, 229-231 immune signals influencing microRNA expression, 236 modulation of microRNA biogenesis by hydrogen peroxide, 236-237 regulation of specific microRNA by hydrogen peroxide, 238-241, 239t-240t in immune system, 231–233 -mediated regulation of cellular redox state, 233-235 gene expression, 228-229 Micrococcin P1, 412 Microposts, 303 Microscopic simulation of immune system, 194 - 196Microsomal PGES (mPGES), 603 Milstein, C., 97 Miltefosine, 567-568 α-Minimal essential medium (MEM), 319 Minimum active sequence (MAS), 149 MiR-21, 241 MiR-155, 230-235, 242, 245-246 MiR-200c, 241 Miravirsen, 218-219 Mirtrons, 220 Mitogens, 53 Mix-and-split method, 144 Modern biotechnology, 3 Modified-live vaccines, 527-528 Molecular recognition, alternative nonantibody scaffolds for, 105 Monoclonal antibodies, 4, 24, 32, 71 - 90clinical uses of, 80-90 adalimumab. 84 alemtuzumab, 80-83

belimumab, 85-86 certolizumab pegol (Cimzia), 85 golimumab, 85 ibritumomab tiuxetan, 84 infliximab, 84 ipilimumab, 86 mepolizumab, 86 natalizumab, 86-87 nivolumab, 88 omalizumab, 87-88 pembrolizumab, 88 rituximab, 84 secukinumab, 88 tocilizumab, 88-89 ustekinumab, 89 vedolizumab, 89-90 development, 74-75 Food and Drug Administration-approved, 81t-82t immune system, 71-73 innate and adaptive immunity, 72-73 origins of, 71-72 nomenclature, 75-76 perspectives, 90 in research, 77-80 enzyme-linked immunosorbent assay, 77 - 78flow cytometry, 78-79 immunohistochemistry, 80 mass cytometry, 80 western blot (WB), 78 Monospecific multivalent antibody, fragments of, 104 MRX34, 218-219 Mucosal addressin-cell adhesion molecule-1 (MAdCAM-1), 89-90 Mucosal immunology, 444-445 Multiplicity of infection (MOI), 528 Murine antibodies, 76f Mutacin 1140, 414t-416t, 418 Mutacin B-Ny266, 414t-416t Mx expression, 610 Mycobacterium leprae, 162-163 Mycobacterium smegmatis, 380 Mycobacterium tuberculosis, 203, 422

Mycoplasma pneumoniae, 217 Mycoplasma pulmonis, 378 Myeloma cells, 74 Myeloma creation, breakdown of, 75f

#### Ν

N3 LP, 378 NAD(P)H-dependent oxido-reductase, 604 - 605Naked DNA vaccines, 447-448 Namalwa, 57-58 Nanog, 354 Nanoliposomes, 389 Nanotechnology, biosurfactants in, 387 - 392functionalized nanoparticles, 387-388 gene transfection with liposomes, 391-392 liposomes, 389-391 Naphthoquinones, 572 Nasopharyngeal diphtheria, 481-482 Natalizumab, 86-87 National Agency for Sanitary Vigilance (ANVISA), 40-41 Natural antibodies, structure and applications of, 100–102 Natural antimicrobial compounds as potential antibiofilm agents, 375-387 Natural antisense transcript (NAT), 218-219, 226-227, 251-252 Negative energy balance (NEB), 611–612 Neglected tropical diseases, 177 Nematodes, recombinant vaccines against, 451 Neoblasts, 225 NEttoie Theiler's Pas Salmonella (NeST), 253 Network measures for biological systems, 186 Networks, 185 Neural cell types, 320 Neurospheres, 358-360, 360f Next-generation biopharmaceuticals, pharmacogenomics and, 42

NF-κB target genes, 189–190 Nicotinamide (NAm), 572 Nisin, 410-417 Nisin A, 413, 414t-416t, 417 Nitric oxide, production of, 572 Nivolumab, 88 Nocathiacin, 414t-416t Nod-like receptors, 445 Noncoding RNAs (ncRNAs), 215-262 biogenesis pathways of, 219-228 long noncoding RNAs, 226–228 PIWI-interacting RNAs, 222-226 small noncoding RNAs, 219-222 and evolution of regulatory molecular mechanism accuracy, 259-262 from evolution to human disease therapy, 215 - 219in human disease therapeutics, 255-258 long noncoding RNAs, 246-255 and alternative splicing, 249-250 chromatin remodeling complexes and epigenetics, 246-247 in immune system, 252-254 linking chromatin remodeling and transcription regulation, 249 long coding RNA structural labels and partners of transcription factors, 247 - 249of messenger RNA stability and translation, 250–252 as sensors of environmental conditions. 254 - 255microRNAs, 228-246 in the crossroad between immune functions and oxidative stress, 241 - 246in immune cell development and functions, 229–231 immune signals influencing microRNA expression, 236 in immune system, 231-233 -mediated regulation of cellular redox state, 233-235 -mediated regulation of gene expression, 228 - 229

modulation of microRNA biogenesis by hydrogen peroxide, 236-237 regulation of specific microRNA by hydrogen peroxide, 238–241, 239t-240t Noncytopathogenic (NCP) bovine viral diarrhea, 614-615 Nonsuppressible insulin-like activity, 610 Novy-MacNeal-Nicolle medium composition, 576t NOX-A12, 135 NOX-E36, 135 NOX-H94, 136 Nuclear factor-kB pathway interactome, 189 - 190Nuclear-enriched autosomal transcript 1 (NEAT1), 250, 254 Nucleic acid-based therapies, 8-10 Nukacin ISK-1, 414t-416t Nutrition, and early embryo loss, 611-612

#### 0

Oligonucleotide-based formulations, 26 Oligoribonucleotides, 130–131 Oligosaccharide lipids (GL), 385 Omalizumab, 87-88 Omega-3 fatty acids, 618-619 One bead one compound (OBOC), 144, 148, 152 Oral rabies vaccines, 453-454 Organ bioengineering, cell types in, 326-328 Organ decellularization, 329 Organ transplantation, 325 Organic antimonials, 564 Osteogenic cell types, 320 Osteogenic differentiation, 319 Osteopontin (OPN), 320, 606-607 Ovsynch program, 616-617 Oxytocin (OT), 597-598 receptor, 597-598

#### Ρ

PAE vector, 488 Paenibacillus polymyxa, 380 Pancreas, 333-335 Pancreatitis, 9 Panitimumab, 108-109 Panning phage display library, 168-169 Paraffin, 321 Parasite survival index (PSI), 580 Parasitic diseases, recombinant vaccines against, 450-453 ectoparasites, 452-453 helminths, 451-452 protozoan parasites, 450-451 Parasitic infections, 167–173 amplification, 169-171 cloning, 173 panning phage display library, 168-169 phage titration, 171-172 screening phage displayed peptide library, 168 Paraspeckles, 250 Paromomycin, 569 Pasteur, Louis, 23 Pasteur's vaccine, 499 Patenting biopharmaceuticals, 32 Path length, 186 Pathogen-associated molecular pattern (PAMP), 72 Pathogenic protein inhibition, 26-27 Pediocin-like bacteriocin, 406-410 Pediocin PA1, 414t-416t, 420 Pediococcus acidilactici, 420 PEG 3350 cakes, 552, 554-555 Pegaptanib sodium (Macugen), 26-27, 134 P-element-induced wimpy testis (PIWI), 220 - 221Pembrolizumab, 88 Penicillin, 23 Pentamidine, 567 Pentavalent antimony compounds, 564 Pentostam, 564 Pep5, 414t-416t Peptide libraries, synthetic. See Synthetic peptide libraries Peptide library, synthesis of, 145f Peptide sequences, characterization of, 174 - 177

Peptides as drugs, 149–150 Perfusion culture, 528 Perfusion process, 15-16 Pertussis acellular vaccine, 485 Pertussis disease, 459-460 Pertussis toxin (PT), 460-461, 461f Pertussis vaccine, development of process to produce recombinant component for, 459-474 epidemiology, 461-462 historical aspects, 460 pathogen and toxin, 460-461 pertussis vaccines, 462, 464t recombinant vaccines, 463 technological development (case description), 463-474 Bordetella pertussis, cultivation and inactivation of, 465-466 *Bordetella pertussis*, extraction and quantification of DNA of, 466 cell growth and protein production kinetics, 471-472 gene amplification by PCR, 466 identity confirmation by mass spectrometry, 473-474 optimization of expression, 471-472 preparation of vector and gene for ligation, 467 primers design and vector choice, 466 protein identification by DNA sequencing, 473 protein purification, 472-473 solubility test, 469-470 transformation of Escherichia coli BL21 pLysS and induction of recombinant protein expression, 469 transformation of Escherichia coli DH5a for plasmid replication, 467-469 Pfizer, 23 PG endoperoxidase H synthase-1 (PGHS-1), 603 PG endoperoxidase H synthase-2 (PGHS-2), 603 Phage-displayed peptides, biotechnological role of, 161-177

antibody purification based on affinity to membrane-blotted antigens, 163-167 affinity purification of human antibodies, 164-167 obtaining crude human immunoglobulin, 163-164 characterization of peptide sequences, 174 - 177Hansen disease, 162–163 parasitic infections, 167-173 amplification, 169-171 cloning, 173 panning phage display library, 168-169 phage titration, 171-172 screening phage displayed peptide library, 168 perspectives, 177 phage enzyme-linked immunosorbent assay, 173-174 Pharmacogenomics and next-generation biopharmaceuticals, 42 Pharmacologic treatments, 108-109 Phenolic glycolipid 1 (PGL-1), 163 Philipimycin, 418-419 Physcomitrella patens, 31 Pichia pastoris, 31, 58, 115, 443 PiRNA-induced silencing complexes (piRISCs), 223 PIWI-interacting RNAs (piRNAs), 216, 222 - 226Plasmid vectors, transcriptional unit in, 441f Plasmodium spp., 167 Platelet-derived growth factor (PDGF), 127 Platyhelminthes phylum, 167 Pluripotency, 354 of stem cells, 350-354 differences between AMSCs and BMMSCs, 351-352 Politics for National Biotechnology, 41 Polycomb repressive complex 2 (PRC2), 246 - 247Polyethylene glycol (PEG), 58-60 Polyglycolic acid (PGA), 328-329 Poly-L-lactic acid (PLA), 328-329

Polymerase chain reaction (PCR), 466, 488 Polymyxin B, 379 Polymyxin E, 417 Polymixins, 379-380 Polyvinylidene difluoride (PVDF), 163-164 Pool/split, portion/mix, and split-and-mix, 144 Positional scanning, 146 Positional scanning synthetic peptide libraries (PS-SPCLs), 146-148, 152 Post–Second World War period, 23 Posttranslational modifications, 12-13 Precursor miRNAs (pre-miRNAs), 220, 237 Pregnancy recognition, 597-620 early embryo loss, causes of, 611-616 genetic abnormalities, 611 hormonal imbalance, 612-613 infectious agents, 614-615 maternal-fetal interactions, impairment of, 615 nutrition, 611-612 quality of oocytes and sperm, 616 thermal stress, 613-614 early embryonic wastage, reducing, 616-620 development of timed insemination programs, 616-618 nutritional strategies, 618-619 supplementation of luteotrophic agents, 619 supplementation with interferon, 619 use of bovine somatotropin, 620 maternal recognition of pregnancy (MRP), 598-602, 599f cyclooxygenase-2, 605-606 fibroblast growth factor-2 (FGF-2), 609 galectin, 607-608 gene regulating, 602–610 insulin-like growth factor-1 (IGF-1), 610 integrin, 608 interferon-τ, 606 Mx. 610 osteopontin (OPN), 606-607

prostaglandin E synthase, 603-604 prostaglandin F synthase, 604-605 serpin, 608-609 signaling, 597-598 Presynch–Ovsynch, 617–618 Prevotella copri, 189 Prevotella intermedia, 189 Primary drying, 532 Primary miRNA (pri-miRNA), 219-220 Product complexity versus expression system, 12-14 Progesterone (P4), 597-600, 612-613 Program for Specialized Medicines, 40 Programmed death 1 blockers, 88 Proinflammatory cytokines, 234-235 Prokaryotic systems, 114–115 Promastigotes, 569, 579 ProPharma Programs, 41 Prophylactic vaccines, 26 Propidium iodide (PI) dye, 580-581 Prostaglandin, 597–598 Prostaglandin E synthase (PGES), 603-604 mRNA expression, 604 Prostaglandin F synthase (PGFS), 604-605 ProstaScint, 12 Prostate-specific membrane antigen (PSMA), 132 Protalix, 13-14 Protalix Biotherapeutics/Pfizer, 4 Protein engineering techniques, 7, 13–14 Protein expression, heterologous systems for, 29-31 Protein kinase C, 127 Protein phosphorylation networks, 187 Protein-protein interactions (PPIs), 149, 186 Protein sequencing, 489-490 Protozoan parasites, recombinant vaccines against, 450-451 Protozoans, 31 Provenge (sipuleucel-T, Dendreon), 11 Pseudofactin, 381 Pseudomembrane, 479, 481-482 Pseudomonas, 380, 391

Pseudomonas aeruginosa, 76, 376, 380–382, 388, 417 Pseudomonas fluorescens, 381 Pseudopregnancy, 601–602 Psoriasis, 88–89 Psoroptes ovis, 453 Purified islet cells, implantation of, 334 Putisolvin, 380–381

## Q

Quality of oocytes and sperm, 616 Quasiduplicates, 153 Quorum sensing (QS), 373

### R

Rabies, 499 Rabies vaccine development, in cell culture, 523 - 556case study, 541-556 freeze-dry study, 550-555 virus infection study, 541-549 definition, generations, and use for humans and animals, 525-528 freeze-drying, 529-541 advances. 538-540 alternatives, 540-541 applications, 535-536 lyophilization, 529-533 methods, 533-535 use for vaccines, 536-538 perfusion and multiplicity of infection, 528-529 perspectives, 556 Rabies virus (RABV), 523-529 classification, proteins, and cell infection, 524-525 disease, epidemiology, and burden, 523 Radiolabeled thymidine, 578-579 Rattus norvegicus, 350 Reactive oxygen species (ROS), 233, 241 aging, and immune senescence, 244 - 245inflammation, and cancer, 245-246

Recombinant antibodies, 102-105, 97 - 117alternative nonantibody scaffolds for molecular recognition, 105 bifunctional antibody fragments/fusion proteins, 105 bispecific antibody fragments, 104-105 as emerging molecules for pharmacologic treatments, 108–109 monospecific multivalent antibody, fragments of, 104 natural antibodies, structure and applications, 100-102 new tools for research and clinical analysis, 109-113 perspectives, 117 producing, 113-117 alternative expression systems, 116-117 mammalian cell systems, 115-116 prokaryotic systems, 114-115 yeast systems, 115 simple monovalent antibody fragments, 102 - 104sources of, 105-108 Recombinant DNA (rDNA) technology, 6 Recombinant IFN-beta, 56 Recombinant therapeutic proteins, 6-8 first-generation biopharmaceuticals, 6-7 production processes for, 12-17 downstream processing, 16-17 product complexity versus expression system, 12–14 upstream processing, 14-16 second-generation biopharmaceuticals, 6-7 Recombinant vaccines, 11, 463 Recombinant veterinary vaccines, 439 - 454against fungal infections, 449 against parasitic diseases, 450–453 ectoparasites, 452-453 helminths, 451-452 protozoan parasites, 450-451 against viral diseases, 453-454 bacterial vaccines, 448-449 characteristics of, 444t

perspectives, 454 production, 445-448, 447t types of, 440-443, 444t DNA vaccines, 441 immune responses induced by vaccination, 443-445 subunit vaccines, 443 vectored vaccines, 441-442 Redox signaling, 219, 233, 238 REG1, 135 Regenerative medicine, 349 Regenerative medicine, and tissue engineering, 315-316 Repeat-associated siRNAs, 222 Reprogrammed cells, 10 Resident microbes, 71–72 Resin beads, 144 Reverse transcription-polymerase chain reaction (RT-PCR), 125 Reverse vaccinology, 440, 448-449 RGD (arginine-glycine-aspartic acid), 151 - 153Rhamnolipids (RLs), 376, 381-382 Rheumatoid arthritis (RA), 27-28, 187-189, 188f Ribonuclease III (RNase III) family, 219 Ribosomally synthesized and posttranslationally modified peptides (RiPPs), 410 Riboswitches, 132, 217-218 Ribozymes, 9 Rig-like receptors, 445 Ripcephalus (Boophilus) microplus, 452 RiPP proteusin polytheonamide B, 411 Rituximab, 83-84 RNA interference (RNAi), 219 RNA vaccine, 205, 441 RNA-induced silencing complex (RISC), 220 Ruconest, 4 Rufisan, 383

#### S

Saccharomyces cerevisiae, 4, 31, 115 SA-FF22, 417 Salmonella typhimurium, 217 Scaffolds, 317, 322 and cellular therapy, 358 effects of scaffold topography on cell function, 305 physical structure on cell behavior, 305 to reprogram MSCs to neurons, 358-360 structural organization of, 299-305 biomimetic elements of macroscale scaffold structure, 304-305 nanoscale and microscale scaffold structure of biomimetic elements. 299 - 303Schizosaccharomyces pombe, 221–222 SCM for infection (SCMI), 541-542, 544 Screening phage displayed peptide library, 168 Scrophularia cryptophila, 571 SDS-PAGE, 470-472, 490 Secondary drying, 533-535 Second-generation biotechnological products, 41 Secukinumab, 88 Seeded scaffolds, 320-321, 328-329 Sendai virus, 57–58 Serpin, 608-609 Serratia marcescens, 385 Serum-free media (SFM), 501 cell adaptation to, 503, 504t Shenzhen Sibiono GeneTech, 9 Short interfering RNAs (siRNAs), 216, 251 and microRNAs, 219-222 Shortest path length, 186 Signal transducer and activator of transcription complexes (STATs), 53 Silver nanoparticles (AgNPs), 388 Simple monovalent antibody fragments, 102 - 104Simplified libraries, 153-154 Single-chain antibody fragment (scFv), 99, 102-104, 113f Single-domain antibodies (sdAb), 104 Single-strand oligonucleotides, 9 Skin plaques, 89 Smad protein family, 236 Small interfering RNAs (siRNA), 9, 26

Small intestine, 329 Small molecule compounds (SMIR), 256 - 257Small noncoding RNAs (sncRNAs), 215-222 short interfering RNAs and microRNAs, 219-222 SnoRNA, 249-250 Sodium dodecyl sulfate (SDS), 382-383 Sofosbuvir, 55 Somatotropin, 24 Sonorensin, 418 Sophorolipids (SLs), 384 Spherical nickel oxide nanoparticles, 388 Spiegelmers, 131 Spinner, 543-545, 544f, 545t-546t, 549 Splenocytes, 74, 75f Spodoptera frugiperda, 31 SPOT technique, 148, 175 Spray-freeze dried products, 540 Squibb, 23 Standard culture medium (SCM), 541-542 Staphylococcus aureus, 374-376, 383, 385 - 386Staphylococcus carnosus, 114 Staphylococcus epidermidis strain 224, 420 Staphylococcus spp. strains, 417-418 Staufen 1 (STAU1), 250 Stellate (Ste) gene, 222 Stem cells (SCs), 315-318, 316f, 327 adult stem cells (ASCs), 318 embryonic stem cells (ESCs), 317 induced pluripotent stem cells (iPSCs), 318 isolation of adipose-derived stem cells, 318 - 319osteogenic differentiation, 319 pluripotency of, 350-354 Sterol regulatory element-binding protein 2n (SREBP2) transactivation, 241 Stomach, 328 Streptococcus gordonii cells, eradicating, 417 Streptococcus salivarius, 382 Streptomyces lividans, 114 Structure-activity relationship (SAR), 150 Sublancin, 412

Sublancin 168, 411, 414t-416t Substrate topographic surface on tissue engineering, 287-306 biomimetic constructs, 287-299 collagen-based biomimetic constructs, 295 - 296collagen used as a biomaterial, 297-299 perspectives, 305-306 scaffold, structural organization of, 299 - 305biomimetic elements of macroscale scaffold structure, 304-305 nanoscale and microscale scaffold structure of biomimetic elements. 299 - 303scaffold physical structure on cell behavior, 305 effects of scaffold topography on cell function, 305 Subtilin, 412, 414t-416t, 418 Subtilosin A, 414t-416t, 419-420 Subunit vaccines, 443, 444t, 485-486, 487f Succinyl trehalose lipids (STLs), 384 Surface GE (sGE), 599-600 Surfactins, 377-378 Synthetic biology basic aspects of, 196-201 biosafety and ethical aspects surrounding, 206 and disease diagnosis, 201-203 and disease prevention, 205-206 and disease therapy, 204-205 future perspectives, 206-207 Synthetic peptide libraries, 176–177, 143 - 155peptides as drugs, 149-150 perspectives, 154-155 preparation of, 144-149 recent applications, 151-154 Synthetic polymers, 290t-294t, 330 Systematic evolution of ligands by exponential enrichment (SELEX) technique, 125-127 Systemic lupus erythematosus (SLE), 85 - 86

Systems biology, 183-184 approaches to, 184–185 biological networks, types of, 186-187 from classical modeling to microscopic simulations, 193 future perspectives, 206-207 mathematical modeling, 190-193 microscopic simulation of immune system, 194 - 196networks, 185 measures for biological systems, 186 studies applied to understand key molecules and disease processes, 187 nuclear factor-kB pathway interactome, 189 - 190rheumatoid arthritis (RA), 187-189, 188f

#### Т

Taenia saginata, 162, 167, 175-177, 176f Taenia solium, 177 Taxus brevifolia, 204 Tbca pseudogene, 251 T-cell lymphomas, 80-83 T-cell receptor (TCR), 229-230, 232-233 Tdap, 484 Telaprevir, 55 TentaGel beads, 147 Terminal uridyl transferase 4 (TUT4), 235 3,3',5,5'-Tetramethylbenzidine, 77 Th17 cells, 88 α-Thalassemia, 218 Theiler's Murine Encephalitis Virus Possible Gene1 (Tmevpg1), 253 T-helper subtype 1 (Th1), 89 Thermal stress, and early embryo loss, 613-614 Thermogravimetric analyses, 388 Thermophilin 13, 421 Thiazomycin, 412, 419 Thiomuracins, 412, 419 Thiopeptides, 418-419 Thiostrepton, 412 THP-1 cells, 579-580

Thuricin CD, 414t–416t, 419 Thyroid hormone receptor  $\alpha$  gene (TR $\alpha$ ), 249 - 250TickGUARD, 452-453 Ticks, vaccines against, 452-453 Tissue decellularization, 326 Tissue engineering, 315–322 cells maintenance and characterization, 319 - 321adipose-derived stem cells, 319-320 generated bone cells, 320-321 perspectives, 321-322 regenerative medicine and, 315-316 stem cells (SCs), 316-318 adult stem cells (ASCs), 318 embryonic stem cells (ESCs), 317 induced pluripotent stem cells (iPSCs), 318 isolation of adipose-derived stem cells, 318-319 osteogenic differentiation, 319 substrate topographic surface on. See Substrate topographic surface on tissue engineering Tissue engineering and regenerative medicine (TERM), 325-328 abdominal organs bioengineering. See Abdominal organs bioengineering Tissue plasminogen activator (t-PA), 4 Tocilizumab, 88-89 Toll-like receptors, 445 and ROS activation in inflammatory responses, 242-244 TonB. 411 Top-down approach, 297-298 Transcription factor regulatory networks, 186 - 187Transcriptional activator-like effector (TALE) nuclease, 198-200 Trehalose, 555 Trehalose lipids (TLs), 384 Trematodes, recombinant vaccines against, 452 Trichoderma spp., 31 Trichoplusia ni, 31

Trichosporon montevideense CLOA72, 386–387 Triglyceride-rich lipoproteins, 9 Tritrpticin, 391 Trp53cor1, 250–251 Tumor necrosis factor-α (TNF-α), 233–234 TNF-α receptor, 8 Tumor-homing peptides, 151–152 Turner's syndrome, 33 Type 1 diabetes mellitus, 334 Tyrosine aminotransferase, 582

#### U

Ubiquitin carboxy terminal hydrolase L1 (UCHL1) mRNA, 254–255 Ulcerative colitis (UC), 89–90 Umbilical cord blood mononuclear cells, 327 UniQure, 4 Unnatural amino acids (UAAs), 147, 153 5'-Untranslated region (UTR), 132, 217–218 Upstream and downstream processing, 18 Urothelial carcinoma-associated 1 (UCA1), 250–251 Ustekinumab, 89 Uterine secretions, 615 Uterine serpins (SERPINA14), 608–609

#### V

Vaccination, 439 immune responses induced by, 443–445 Vaccines, 10–11. *See also* specific vaccines Variable fragment (Fv) of antibody, 102–104 Vascular endothelial growth factor (VEGF), 127, 257 Vascular-cell adhesion molecule 1 (VCAM-1), 86–87 Vector NTI (Invitrogen), 466, 488 Vectored vaccines, 441–442, 444, 444t Vedolizumab, 89–90 Vero cell line, 500 *Vesiculovirus*, 524 Veterinary rabies vaccine, 499–519

adaptation of baby hamster kidney cell line-21 to serum-free media (case study), 502-512 adaptation to Cellvento BHK-200, 510 - 512adaptation to Ex Cell 302 from standard culture media, 507-509 adaptation to VP-serum free media from standard culture media. 506 - 507cell count, 504-505 cell cultivation, 503-504 cell line, 502 cryopreservation, 505 culture media, 503 growth kinetics, 505-506 metabolism analysis, 505 sedimentation tests, 505 strategies for cell adaptation to serumfree media, 503 perspectives, 518 rabies virus production, 513-517 cell infection and virus production, 513 - 514cells adapted to Cellvento BHK-200, growth kinetics of, 516 cells adapted to Cellvento BHK-200, infection kinetics of, 516 cells adapted to Ex Cell 302, infection kinetics of, 515-516 cells adapted to VP-serum-free medium, infection kinetics of, 515 comparison of virus titer in serum-free adapted cell, 516-517 control culture, infection kinetics of, 514 - 515culture media, 513 direct immunofluorescence assays, 514 rabies virus titration, 514 virus strain, 513 Veterinary vaccine production, 445-448, 447t Vibrio, 614–615 Vibrio species, 378 Viral diseases, recombinant vaccines against, 453-454

Virus-like particles (VLPs), 11 VP-serum-free media (VP-SFM), 501–502

#### w

Wellferon, 57–58 Western blot (WB), 78, 489–490 Whole-organ decellularization, 326 Whooping cough. *See* Pertussis disease World regulations, 37–39 Wound healing, 327

# Х

X-chromosome reprogramming, 354 Xerovac, 539–540 *Xist* transcription, 246–247 Xylolipids, 385

# Υ

Yarrowia lipolytica, 382–383 Yeasts, 31, 386 expression systems, 115 *Yersinia pestis*, 203 Young's modulus, 295–296 YSI 2700 (Yellow Springs Instruments, EE.UU.), 505

# Ζ

Zinc finger nuclease (ZFN), 198–200 ZnS nanoparticles, 388